### GLOBAL EDUCATION MONITORING REPORT

# Non-state actors in education who chooses? Who Loses?







2021/2

### GLOBAL EDUCATION MONITORING REPORT



# Non-state actors in education

WHO CHOOSES? WHO LOSES?

The present licence applies exclusively to the texts. For the use of images, prior permission shall be requested.

UNESCO is an Open Access publisher and all publications are made available on-line, free of charge through UNESCO's documentary repository. Any commercialization of its publications by UNESCO is for cost-recovery of nominal actual costs for printing or copying content on paper or CDs, and distribution. There is no profit motive.

This publication can be referenced as: UNESCO. 2021. Global Education Monitoring Report 2021/2: Non-state actors in education: Who chooses? Who loses? Paris, UNESCO.

© UNESCO, 2021 Second edition Published in 2021 by the United Nations Educational, Scientific and Cultural Organization 7, Place de Fontenoy, 75352 Paris 07 SP, France

Typeset by UNESCO Graphic design by Optima Graphic Design Consultants Ltd Layout by Optima Graphic Design Consultants Ltd

Cover photo: Jaap Joris Vens / Super Formosa Photography Caption: Different routes to school, the Netherlands.

Illustrations by Housatonic SRL Unipersonale Cartoons by Julio Carrión Cueva (Karry) and Miguel Morales Madrigal

Infographics by Housatonic SRL Unipersonale

ISBN: 978-92-3-100506-0 ISSN: 2618-0685 This publication is available in Open Access under the Attribution-ShareAlike 3.0 IGO (CC-BY-SA 3.0 IGO) licence (http://creativecommons.org/licenses/by-sa/3.0/igo/). By using the content of this publication, the users accept to be bound by the terms of use of the UNESCO Open Access Repository (http://www.unesco.org/open-access/ terms-use-ccbysa-en).

The present licence applies exclusively to the text content of the publication. For the use of any material not clearly identified as belonging to UNESCO, prior permission shall be requested from: publication. copyright@unesco.org or UNESCO Publishing, 7, place de Fontenoy, 75352 Paris 07 SP France.



This report and all related materials are available for download here: Bit.ly/2021gemreport

## Who chooses? Who loses?

### SHORT SUMMARY

Non-state actors' role extends beyond provision of schooling to interventions at various education levels and influence spheres. Alongside its review of progress towards SDG 4, including emerging evidence on the COVID-19 pandemic's impact, the 2021/2 *Global Education Monitoring Report* urges governments to see all institutions, students and teachers as part of a single system. Standards, information, incentives and accountability should help governments protect, respect and fulfil the right to education of all, without turning their eyes away from privilege or exploitation. Publicly funded education does not have to be publicly provided but disparity in education processes, student outcomes and teacher working conditions must be addressed. Efficiency and innovation, rather than being commercial secrets, should be diffused and practised by all. To that end, transparency and integrity in the public education policy process need to be maintained to block vested interests.

The report's rallying call – Who chooses? Who loses? – invites policymakers to question relationships with non-state actors in terms of fundamental choices: between equity and freedom of choice; between encouraging initiative and setting standards; between groups of varying means and needs; between immediate commitments under SDG 4 and those to be progressively realized (e.g. post-secondary education); and between education and other social sectors.

Supporting the fifth Global Education Monitoring Report are two online tools: PEER, a policy dialogue resource describing non-state activity and regulations in the world's education systems; and VIEW, a new website consolidating sources and providing new completion rate estimates over time.

UNESCO is the United Nations Educational, Scientific and Cultural Organization. It seeks to build peace through international cooperation in education, the sciences and culture. UNESCO believes that political and economic arrangements are not enough to secure the lasting and sincere support of the peoples. Peace must be founded upon dialogue and mutual understanding, upon the intellectual and moral solidarity of humanity. In this spirit, UNESCO develops educational tools and cultural and scientific programmes to strengthen bonds among nations, help countries adopt international standards and foster the free flow of ideas and knowledge sharing.

**350 million children** are educated by non-state actors



Since wars begin in the minds of men and women, it is in the minds of men and women that the defenses of peace must be constructed

#### VI 2021/2 • GLOBAL EDUCATION MONITORING REPORT

# Foreword

The theme of this year's *Global Education Monitoring Report* speaks to my entire life trajectory. I am the outcome of a mix of private and public education and life choices. I started my education in a non-state pre-primary and primary school before enrolling in a state secondary school in Sierra Leone, continued through to world-class non-state universities in the United States and then to a multinational company as a research scientist in Kenya. I am now channelling all I learned during these varied experiences into the combined roles I have in the Sierra Leonean government on education and innovation.

With clear oversight of the full range of state and non-state activity in education now as a Cabinet minister, I welcome this report's emphasis on approaches to system governance all over the world. It acknowledges that distinctions between public and private are increasingly blurred, making the arguments that stake positions on clear dichotomies increasingly irrelevant.

I see my objective as Chief Innovation Officer to facilitate and support a vibrant national innovation and entrepreneurial ecosystem that works for all education actors in Sierra Leone. It echoes the call in this report to ensure that all actors – state and non-state – can work together to build an effective, dynamic and equitable education system for the benefit of all. In that sense, this report reads my mind when it calls on governments to create spaces and set the conditions for a variety of actors to interact, coordinate and cooperate. There is much to be gained from maximizing all parties' expertise.

The truth of the matter is that public systems need to innovate to remain relevant, and that requires fresh thinking. One of the first things we did when I arrived in the public education system was to put together a new curriculum framework relevant to the 21st century. We built teacher training modules based on real needs. This report calls for all countries to pick up on such practices. Skills development systems that feed on contributions from governments, employers and the workers themselves are far more likely to keep pace with labour market dynamics.

Anyone in the public sector knows that public education is a conventional institution, which, by definition, is not easily reformed. But with the fast and furious shift to digitization and globalization, education systems need to keep up. More than that, they have a lot to gain if they do so. Partnerships with the private sector can help mobilize resources and expertise across the board, and encourage initiatives that can fast-forward progress at a pace governments may be unlikely to achieve on their own. We have recently issued tablets to administrators to help them track grades, attendance and budgets. We have fully digitized school census data going right back to 2015 that we now interrogate to track those left behind. All these enhance our ability to make an impact at scale within the public sector.

Many of our initiatives would not have left the starting block without funds. But this report calls for caution in our haste to follow the money. What strategic interests are our partners putting first? Does their support align with government priorities, does it avoid duplication or distraction, and will it be flexible to our evolving needs? The questions asked in the recommendations are important for all of us to consider. 'Who is choosing?', we are asked. Sometimes it might not be clear. 'Who is losing?', it asks again. 'Why is it always the same groups?'

Many people have been pretending that non-state actors do not or should not play a role in education today. This report shows that they do already and will continue to do so tomorrow. Education is about the future. Whether you are in the private or public sector, I encourage you to read the 2021/2 GEM Report's recommendations to be sure you are not left behind on the vision it has for change.

Brech

Dr David Moinina Sengeh Minister of Basic and Senior Secondary Education and Chief Innovation Officer, Sierra Leone Chair of the GEM Report Advisory Board

# Acknowledgements

This report would not have been possible without the valuable contributions of numerous people and institutions. The *Global Education Monitoring Report* (GEM Report) team would like to acknowledge their support and thank them for their time and effort.

Invaluable input was provided by the GEM Report Advisory Board's members and its chairperson, David Sengeh. Special thanks also go to our engaged and committed funders, whose financial support makes the GEM Report possible.

We would like to acknowledge the role of UNESCO and its leadership. We are very grateful to many individuals, divisions and units at UNESCO headquarters, notably in the Education Sector and the Bureau for the Management of Support Services, for facilitating our daily work. The UNESCO Institute for Statistics played a key role by supporting access to its data. We would like to thank its director, Silvia Montoya, and her dedicated staff for their close collaboration, notably as part of our partnership on the World Inequality Database on Education and in the framework of the Technical Cooperation Group on the Indicators for SDG 4 – Education 2030, especially in the process of developing national SDG 4 benchmarks. Additional thanks go to colleagues at the UNESCO regional and field office network.

The GEM Report team would like to thank the researchers who produced background papers informing the report's analyses: Desiree Acholla, Philip G. Altbach, Fabian Barch, Jörg Baten, Karina Batthyány, Hélène Binesse, Xavier Bonal, Francesca Bonomelli Carrasco, Federico Bruno, Dana Brakman Reiser, Mark Bray, Anne Campbell, Alejandro Caravaca, Sebastian Cherng, Lee Crawfurd, Martial Dembélé, Hans de Wit, Jonas Edlund, Brent Edwards, Clara Fontdevila, Ulrike Hanemann, Susannah Hares, Marie-France Lange, Thibaut Lauwerier, Daniel C. Levy, Arvid Lindh, Rita Locatelli, Mohammad Naeim Maleki, Zachariah Mampilly, Tatiana Mikhaylova, Christopher Millora, Adriana Morales-Perlaza, Lara Patil, Valentina Perrotta, Annie Rappeport, Serena Rossignoli, Robin Shields, Renu Singh, Geneviève Sirois, Malini Sivasubramanian, Vanessa Sperduti, James Stanfield, Wondwosen Tamrat, Antoni Verger, Qingru Wang, Anthony Welch, Quentin Wodon and Adrián Zancajo.

We are grateful to several institutions and their research staff who also produced background papers: Bamberg University (Annette Scheunpflug and Mark Wenz in cooperation with Jean Kasereka Lutswamba, Samuel Mutabazi, Frederick Njobati, Claude Ernest Njoya, Christine Nyiramana, Onja Raharijaona and Mimii Brown Rubindamayugi), Cambridge Education (Sophie Archambault and Sabine Kube-Barth), Global Initiative for Economic, Social and Cultural Rights (Sylvain Aubry and Meredith Galloway), International Association for the Evaluation of Educational Achievement (Diego Cortés, Yuan-Ling Liaw, Mojca Rožman and Rolf Strietholt), Network of Education Policy Centers (Lana Jurko and Raffaella d'Apolito), National Foundation for Educational Research (Katarzyna Kubacka, Ahmad Jawad Asghar, Jenny Price, Gustavo Lopes and Shannon Lindsey), Al Qasimi Foundation for Policy Research (Marvin Erfurth, Janaan Farhat and Natasha Ridge), Right to Education Initiative (Frank Adamson and Delphine Dorsi), SIL Africa Learning & Development (Manasseh Wekundah and Barbara Trudell), Street Child, University of East Anglia (Anna Robinson-Pant) and University of Glasgow (Yulia Nesterova and Queralt Capsada-Munsech). We are also grateful for the support of the Institute for Health Metrics and Evaluation (Joseph Dieleman, Emmanuela Gakidou, Hussain Jafari, Emilie Maddison, Angela Micah, Erin Mullany, Trang Nguyen, Aislyn Orji, Ian Pollock, Goli Tsakalos).

We are grateful for the engagement and valuable comments of the GEM Report's Group of Friends, whom we asked to provide feedback on the draft recommendations. We are not listing their names, knowing that nobody will have agreed with them in full on such a difficult topic.

A group of independent experts reviewed the draft of the GEM Report's thematic part and provided valuable feedback. For their input we thank Stephen Ball, Gregory Elacqua, Maria Marta Ferreyra, Pooja Gianchandani, Harry Patrinos, Nirmala Rao and Antoni Verger. Prachi Srivastava contributed the think piece that informed the concept note of the report.

Special thanks go to the GEM Report's fellows, supported by the Foundation to Promote Open Society: Nicolás Buchbinder, Elizabeth Buckner and Wei Zhang (2020 cohort); Leena Bhattacharya, Fernanda Gándara, Emmanuel Manyasa, Muhammad Afzan Munir, Saba Saeed and Danqing Yin (2021 cohort).

The report was edited by Andy Quan, whom we thank for his tireless work.

We also wish to acknowledge those who worked hard to support the design, production, printing and translation of the report within and outside UNESCO, including Rebecca Brite, Erin Crum, Julio Carrión Cueva (Karry), Miguel Morales Madrigal, Housatonic SRL Unipersonale (Cecilia Negri, Elena Vasumini), Optima and Strategic Agenda.

Specific thanks go to Arete Stories, Dean Swift, di:ga and Rooftop for their support to the outreach of the GEM Report, and to UNICEF and Save the Children for extensive use of their photos. Thank you to Blossom (Alberto Zanardo), HiTeki (Hossein Aghvami) and Interactive Things (Gerhard Bliedung, Christian Siegrist and Benjamin Wiederkehr) for their web design and development support to the GEM Report.

Finally, we would like to thank the short-term consultants and interns who provided much input to the GEM Report team: Ritika Agarwal, Marie Renée Andreescu, Moussa Garba, Sun Min Lee, Aziah-Katiana Tan and Jieyu Wang. Thanks also to the students of the Université Paris 1 Panthéon-Sorbonne, Sorbonne School of Economics, who contributed to the development of country profiles: Camila Araceli Maciel, Manuel Esteban Arias and Jakob Hannerz.

#### The Global Education Monitoring Report team

Director: Manos Antoninis

Daniel April, Bilal Barakat, Marcela Maria Barrios Rivera, Madeleine Barry, Katherine Black, Nicole Bella, Celia Eugenia Calvo Gutierez, Daniel Caro Vasquez, Anna Cristina D'Addio, Dimitra Dafalia, Dmitri Davydov, Francesca Endrizzi, Constanza Ginestra, Chandni Jain, Priyadarshani Joshi, Maria-Rafaela Kaldi, Josephine Kiyenje, Craig Laird, Katie Lazaro, Kate Linkins, Camila Lima De Moraes, Kassiani Lythrangomitis, Anissa Mechtar, Claudine Mukizwa, Yuki Murakami, Manuela Pombo Polanco, Judith Randrianatoavina, Kate Redman, Maria Rojnov, Laura Stipanovic, Ulrich Janse van Vuuren, Juliana Zapata, Lema Zekrya and Jiaheng Zhou. The Global Education Monitoring Report is an independent annual publication. The GEM Report is funded by a group of governments, multilateral agencies and private foundations and facilitated and supported by UNESCO.



#### For more information, please contact:

Global Education Monitoring Report team UNESCO, 7, place de Fontenoy 75352 Paris 07 SP, France Email: gemreport@unesco.org Tel.: +33 1 45 68 07 41 www.unesco.org/gemreport

Any errors or omissions found subsequent to printing will be corrected in the online version at www.unesco.org/gemreport

#### **Global Education Monitoring Report series**

- 2021/2 Non-state actors in education Who chooses? Who loses?
- 2020 Inclusion and education: All means all
- 2019 Migration, displacement and education: Building bridges, not walls
- 2017/8 Accountability in education: Meeting our commitments
- 2016 Education for people and planet: Creating sustainable futures for all

#### EFA Global Monitoring Report series

- 2015 Education for All 2000–2015: Achievements and challenges
- 2013/4 Teaching and learning: Achieving quality for all
- 2012 Youth and skills: Putting education to work
- 2011 The hidden crisis: Armed conflict and education
- 2010 Reaching the marginalized
- 2009 Overcoming inequality: Why governance matters
- 2008 Education for All by 2015: Will we make it?
- 2007 Strong foundations: Early childhood care and education
- 2006 Literacy for life
- 2005 Education for All: The quality imperative
- 2003/4 Gender and Education for All: The leap to equality
- 2002 Education for All: Is the world on track?

х

# Contents

Short summary	V
Foreword	.vii
Acknowledgements	viii
Contents	. xi

Non-state actors in education	1
KEY MESSAGES	3
CHAPTER 1. Introduction	5
Education is both a public and a private good	9
The right to education involves entitlements and freedoms	9
Support for public education is strong	
Diverse arguments drive debate for or against non-state provision	
Myths about state and non-state actors in education prevail	
Guide to the report	
Recommendations	
CHAPTER 2. Provision	29
Enrolment in non-state schools has been growing	
State and non-state schools differ in student intake and available resources	
The impact of non-state schools on individual and system outcomes is contested	47
Non-state actors offer education goods and services other than schooling	
Conclusion	
CHAPTER 3. Governance and regulation	63
Education system governance is often fragmented	
Regulations do not focus enough on equity and quality	
Weak implementation often undermines regulation	
Private tutoring regulations tend to be underdeveloped	
Conclusion	
CHAPTER 4. Finance	93
Governments finance non-state providers directly and indirectly	
Households face significant burdens and tough choices	
Donors are cautious about supporting for-profit education providers	
The role of philanthropic and corporate activities in education is evolving	
Conclusion	
CHAPTER 5. Influence	115
Networks advocate for competing visions of non-state actors in education	
The business sector exerts strong influence on education	
International organizations and foundations have particular influence	
Conclusion	

CHAPTER 6. Early childhood care and education	133
Non-state actors lead care and education services for children under 3	
Non-state actors are more prominent in pre-primary than in basic education	
Non-state provision challenges governance and regulation	
Public funding of non-state provision takes different forms	
Non-state actors innovate and advocate for ECCE	
Conclusion	
CHAPTER 7. Tertiary education	
More than one in three tertiary students attend non-state institutions	
Non-state tertiary education regulation varies	
Financing modalities have quality and equity implications	
Non-state actors influence tertiary education through multiple mechanisms	
Conclusion	
CHAPTER 8. Technical, vocational and adult education	
Apart from entry-level skills, non-state actors dominate technical and vocational education	
Governing skills development systems with non-state actor participation is challenging	
The right mix of state and non-state financing can support optimal skills development	
Non-state actors are a driving force in adult learning and education	
Conclusion	

Monitoring education in the Sustainable Development Goals	201
CHAPTER 9. Introduction	
Countries have submitted national SDG 4 benchmarks	
COVID-19 has affected the prospects of achieving SDG 4 and the means of monitoring progress	
The SDG 4 monitoring framework continues to develop	
Guide to the monitoring part	
CHAPTER 10. Primary and secondary education	213
Access	
Focus 10.1: How do demand factors prevent universal schooling?	
Learning	
Focus 10.2: Learning does not progress in a linear fashion	
COVID-19	233
CHAPTER 11. Early childhood	241
Focus 11.1: Early childhood care and education starts at birth	
COVID-19	
CHAPTER 12. Technical, vocational, tertiary and adult education	253
Technical and vocational education and training	
Tertiary education	
Focus 12.1: Many families have trouble repaying student debt	
Adult education	
Focus 12.2: Community learning centres have proliferated in many countries	
Focus 12.3: Education leave is a tool to promote adult education in high-income countries	

CHAPTER 13. Skills for work	
Focus 13.1: Computational thinking is an important component of digital literacy	278
COVID-19	
CHAPTER 14. Equity	281
Focus 14.1: Household wealth does not capture all dimensions of poverty	289
Focus 14.2: Many children attend rebel-controlled schools - sometimes their entire school life	290
Focus 14.3: More information is becoming available on the number of children taught in their home language	
COVID-19	
CHAPTER 15. Literacy and numeracy	301
Focus 15.1: Basic numeracy skills have stagnated among Africa's poorest for decades	308
COVID-19	310
CHAPTER 16. Sustainable development and global citizenship	
Focus 16.1: Climate change education aims to equip populations to cope with and mitigate the effects of climate change	320
COVID-19	
CHAPTER 17 Education facilities and learning environments	325
Focus 171: School schedules are part of the learning environment – and influence outcomes	333
Focus 17.2: School infrastructure may be above the minimum standard yet unacceptable and deteriorating	
COVID-19	
CHAPTER 18. Scholarships	
Focus 18.1: Scholarship aid flows are becoming less concentrated	
COVID-19	348
CHAPTER 19. Teachers	
Focus 19.1: Many teachers are trained and qualified but not for the subject they teach	359
COVID-19	
CHAPTER 20. Education in the other SDGs	
Improved energy and infrastructure support education	
Education can support achievement of energy and sustainability objectives	369
Conclusion	372
CHAPTER 21. Monitoring finance	
Public expenditure	
Focus 21.1: Poor countries need to increase tax revenue to fund education	379
Focus 21.2: Education spending should focus on equity	383
Aid expenditure	
Household expenditure	398
Annex	
Statistical tables	401
A*1+ 11	4.50

Aid tables	469
Glossary	479
Abbreviations	482
References	485



# Non-state actors in education

# **KEY MESSAGES**

#### There is no part of education in which non-state actors are not involved.

Put simply, without non-state actors, the education of 350 million more children would fall to the responsibility of the state. But non-state engagement also affects the textbooks they use, the food in their canteens, the additional support they get, the skills they learn and much more.

#### Most people support public education.

Three in four people in 34 middle- and high-income countries would prefer more public spending on education, with support increasing the more unequal the country. Almost 9 in 10 think education should primarily be public.

#### But such support has gradually eroded in several low- and middle-income countries.

Where public schools had been in short supply and their quality had deteriorated, many families voted with their feet. The share of private institutions worldwide increased by seven percentage points in about 10 years: to 17% by 2013 in primary and to 26% by 2014 in secondary education. It has remained roughly constant since. In Central and Southern Asia the share of private enrolment is 36% in primary and 48% in secondary education.

#### Public education is not free.

Households account for 30% of total education spending globally and 39% in low- and lower-middle-income countries. Part is due to wealthier families trying to give their children a competitive advantage. But a large part is spent on pre-primary, primary and secondary education that governments committed to provide free of charge. About 8% of families borrow to pay for education, rising to 12% in low-income countries and 30% or more in Haiti, Kenya, the Philippines and Uganda.

#### Public education is often not inclusive.

Many public education systems fail to prevent stratification and segregation. An index of social diversity in schools, based on Programme for International Student Assessment data, found that Argentina, Brazil, Chile and Mexico had similar high levels of stratification in 2018, although only Chile tends to be criticized for the high share of private institutions in its system.

#### No one type of provider delivers education of better quality than any other.

Data from 30 low- and middle-income countries show that, once household characteristics are accounted for, the apparent premium from attending private school drops by half to two-thirds. In a sample of 49 countries, the richest are almost 10 times likelier than the poor to go to private school. And parents who can choose schools do so because of religious beliefs, convenience and student demographic characteristics rather than quality, about which they rarely have sufficient information.

#### Regulatory, monitoring and enforcement capacity tends to be low where the need is high.

Analysis of 211 education systems for the PEER website shows that regulations tend to focus on registration, approval or licensing (98%), teacher certification (93%), infrastructure (80%) and pupil/teacher ratios (74%). Regulations are least likely to focus on quality or equity: 67% regulate fee setting, 55% prevent selective student admission procedures in non-state schools, 27% ban profit making and only 7% have quotas supporting access of disadvantaged groups. Private tutoring is unregulated in 48% of countries and regulated only in commercial legislation in 11% of countries.

#### Non-state actors are even more present in early childhood, technical, tertiary and adult education.

This is sometimes at the expense of equity and quality. The generally higher cost of non-state early childhood and tertiary education means urban elites are over-represented in these institutions. In the United States, profit-maximizing universities have been linked with a deterioration of student outcomes. Institutions providing private training through market competition or skills development systems, such as Australia's TVET FEE-HELP loan programme and India's National Skill Development Corporation, were forced to rethink accountability and monitoring processes to increase the quality of private provision and improve employability outcomes.

#### Governments need to see all education institutions, students and teachers as part of a single system.

Standards, information, incentives and accountability should help governments protect, respect and fulfil the right to education of all and should prevent them from turning their eyes away from pockets of privilege or exploitation. Publicly funded education does not have to be publicly provided, but disparity in education processes, student outcomes and teacher working conditions should be addressed head-on. Efficiency and innovation should not be a commercial secret; rather, they should be diffused and practised by all. To achieve that, transparency and integrity in the public education policy process need to be maintained.

A schoolboy in school uniform and a young Buddhist monk meet on the street, Bhutan.

CREDIT: Mikkel Ostergaard/Panos Pictures



CHAPTER

# Introduction



## **KEY MESSAGES**

#### Non-state involvement in education generates passionate debate with respect to two questions.

- To what the extent is education a public investment or private consumer good?
- What are the right to education implications regarding state and non-state actors' responsibilities?

#### There is strong support for public education.

- In 10 middle- and 25 high-income countries, 75% of respondents favoured more public education spending, from 52% in the Czech Republic to 95% in the Philippines. Support for more public education spending increases with income inequality.
- Overall, 89% said the primary responsibility for providing school education rested with governments. Reflecting strong exposure to non-state provision, respondents in India (46%), the Philippines (63%) and Chile (76%) expressed the lowest support for public provision.
- Education builds attitudes in support of social cohesion. However, the more educated are often the first to reject public education.

#### Country experiences with non-state actors' involvement in education vary widely.

- In some countries, non-state actors have long been a foundation of the education system for cultural, religious and historical reasons.
- In other countries, the role of non-state actors has been limited.
- A few introduced school choice as a conscious strategy to revamp the education system.
- But for many poorer countries, trust in public education was eroded gradually.

#### Three core issues drive debate for or against non-state provision.

- Proponents argue non-state actors are cost-efficient. Opponents argue that even where there is a clear cost advantage, it is because the underlying issues are not addressed directly.
- Proponents say non-state actors fill genuine gaps, while some prefer education to be adapted to their beliefs and principles. Opponents cite equity and inclusion risks because disadvantaged populations have less access to the options non-state actors offer.
- Proponents assert that public education systems have grown into large centralized bureaucracies and blunt initiative. Opponents argue that non-state innovation is often exaggerated and not replicable.

#### Recurring myths about state and non-state actors in education are questioned in this report.

- State and non-state actors can be clearly distinguished.
- The extent of privatization is known.
- The private sector is to blame for privatization in education.
- Public education is equitable.
- Parents base school choice on robust information about quality.
- Competition leads to school improvement.
- Private schools and universities are better.
- The private sector is a solution to the out-of-school challenge.
- The private sector is a solution to education financing gaps.
- Regulations can address all concerns about non-state provision.

Education is both a public and a private good	9
The right to education involves entitlements and freedoms	9
Support for public education is strong	10
Diverse arguments drive debate for or against non-state provision	11
Myths about state and non-state actors in education prevail	16
Guide to the report	21
Recommendations	22

The Education 2030 Framework for Action, which is the roadmap for achievement of the fourth Sustainable Development Goal (SDG 4), highlights the crucial role of non-state actors in education (**Box 1.1**): 'Country-led action will drive change; however, the ambitious education goal cannot be achieved by governments alone. They will need the support of all stakeholders, including non-state actors' (UNESCO, 2015, §86).

The framework recognizes (§10) that education is:

- 'a public good, of which the state is the duty bearer' and 'a shared societal endeavour, which implies an inclusive process of public policy formulation and implementation', where civil society and the private sector, among others, 'have important roles in realizing the right to quality education' and the state has an essential role in 'setting and regulating standards and norms; and
- 'a fundamental human right and an enabling right' for whose fulfilment countries must 'ensure universal equal access to inclusive and equitable quality education'.

Despite this understanding that fulfilling the right to education requires multiple stakeholders and that the role of non-state actors has increased over the past 30 years, non-state involvement in education generates passionate debate, particularly over two key concepts: the extent to which education is a public or private good, a form of investment or consumption; and how to interpret the right to education with respect to its implications for state and non-state actors' responsibilities.

#### BOX 1.1:

#### This report's definition of non-state actors in education is broad

In this report, 'non-state' is a broad, catch-all term describing individuals and organizations involved not only in education provision, but also in education financing and in influencing the state's direction in its obligation to fulfil the right to education.

As such, the term is used in reference to:

- Individuals who benefit from and/or pay for education (e.g. users or purchasers of goods and services, taxpayers), provide education (e.g. single school proprietors, homeschooling providers) and express views on its content, modality and delivery (e.g. through participation in school management, through the political process).
- Private corporations which also provide education-related goods and services (as owners or managers) and which finance (directly and indirectly) and influence education.
- Philanthropic foundations, independent of private entities in their direction, which mainly influence education policy, but also play limited roles in provision and financing.
- Non-governmental, civil society, trade union and faith-based organizations, which may provide, finance and influence education.
- Academics, researchers and think tanks, even if financed by government, who generate evidence and knowledge on education.
- The media, which exert influence in the debate on the role of non-state actors in education.

The above description makes it clear that the terms 'non-state' and 'private' are not interchangeable. Rather, private actors are a subset of non-state actors.

8

#### EDUCATION IS BOTH A PUBLIC AND A PRIVATE GOOD

Governments have not always led education. As a core part of human existence, education historically was organized spontaneously and informally. From the late 18th century in Europe, states saw the opportunity to develop their economies through an educated workforce and to develop and strengthen a sense of national identity through public schools. This was 'a decisive break with the voluntary and pluralistic form of learning which had preceded them, where church, family and guild had provided for their own needs' (Green, 2013, p. 12).

Governments were prepared to take on the high cost of delivering a public good because of the wider benefits to societies and economies. Without state provision, individuals might not have invested as much in education, depriving societies of their potential. For newly independent countries in the 20th century, building a public education system was the hallmark of emancipation from colonialism. Public education invariably aimed to promote noble ideals or ruling ideologies. The new structures superseded and absorbed traditional education structures managed by local communities and religious organizations.

Education is also considered a private good. Consuming more education improves an individual's opportunities and may exclude others from them. Education becomes a vehicle for differentiation and advancement: Those who manage to climb the education ladder are better placed to achieve a higher standard of living and higher returns. As education systems cannot accommodate everybody on the higher rungs, families do everything they can to ensure that their offspring are the ones who make it to the top.

Such competition generates demand, which in turn leads to the supply of education goods and services. Depending on national context and disposition, markets may emerge in direct provision of education or in other services that confer advantage, such as supplementary tuition. Moreover, education is a costly enterprise, and governments have differed in the extent to which they provide sufficient financing. Some have been forced or even actively chosen to downsize public education systems, shifting the burden to households. As a result, around the world, public authority over education differs, as do citizen expectations of government responsibility for education.

#### THE RIGHT TO EDUCATION INVOLVES ENTITLEMENTS AND FREEDOMS

The right to education has been enshrined in human rights instruments for some 75 years, notably in the 1948 Universal Declaration of Human Rights, the 1966 International Covenant on Economic, Social and Cultural Rights and the 1989 Convention on the Rights of the Child. These texts encompass both entitlements and freedoms. Individuals have the right to free and compulsory primary education, an entitlement that should gradually be extended to higher levels of education as countries' capacity increases. But also recognized are individuals' rights to establish schools and to choose the type of school they prefer for their children, in line with their religious and moral convictions, as long as these schools meet minimum government standards. Such standards should ensure that education is directed to full development of the human personality and sense of dignity, strengthens respect for human rights and fundamental freedoms and enables effective participation in a free society. Education should also promote understanding, tolerance and friendship among all nations and all ethnic, racial and religious groups (United Nations, 1948; 1966).

In 1999, the Committee on Economic, Social and Cultural Rights issued its General Comment No. 13 to elaborate on the right to education and resulting state obligations. It set out four principles: There have to be enough schools with appropriate infrastructure, trained teachers, and teaching and learning materials; the schools have to be accessible to all, without discrimination or physical, technological or financial obstacles; curricula and teaching methods have to be acceptable, relevant, culturally appropriate and of good quality; and education has to be flexible and adaptable to changing societal and community needs (OHCHR, 1999).

Making countries responsible for meeting these principles leaves open several questions on how they should intervene. Should they provide, finance or regulate education? All of these? In what mix? While states have a duty to respect, protect and fulfil their citizens' right to education, a wide range of non-state actors with a variety of forms, arrangements and motivation, from charity to profit, play a significant role in many education systems. Their activities may or may not involve collaboration with the government. Should non-state actors' participation in education be encouraged, contained or prevented? The answers may be specific to country context, education level and type of activity. Governments may respond to popular demand or steer it. Education choices determine children's lives. Parents must not only make simple calculations of financial costs and benefits but also take multiple, interrelated factors into account. Choices regarding what is taught, how, by whom and where reflect the competing world views and aspirations of parents and of other education stakeholders. They concern two main dimensions: control and distribution of resources (socioeconomic ideology) and values and beliefs for changing society (sociocultural ideology). Those who believe that government should have a greater role than market forces in economic governance and distribution, as well as those with liberal rather than conservative values on gender, religion, equality and the environment, tend to support more public spending on education and a primary role for government in education provision.<sup>1</sup>

Education choices are highly political and are reflected explicitly or implicitly in political agendas. In addition to individual ideological and circumstantial factors, understandings of social challenges and how government, people and institutions should relate to one another vary among countries. These understandings influence attitudes on what policies government should pursue and who should benefit from them.

Research on support for public education is overwhelmingly from high-income countries. Among 17 Organisation for Economic Co-operation and Development (OECD) countries, the majority of respondents in all but Finland preferred increases in public education spending. Higher national levels of socioeconomic inequality were associated with stronger preferences for more public spending, especially among the poorest households. Higher national levels of education inequality corresponded to greater preferences for more public spending among the richest households, as they would stand to benefit first (Busemeyer, 2012).

A recent survey of attitudes in Denmark, France, Germany, Ireland, Italy, Spain, Sweden and the United Kingdom confirmed high support for public education spending. It was higher than in other areas: When respondents were asked to prioritize one of eight potential areas for additional spending, education was the top option for 28%, with health care second at 22%. Support varied by education level, being higher for primary and secondary education (62%) than for pre-primary (50%) and tertiary (47%). While 77% of respondents supported school choice, over 60% opposed a significant role for private schools in the national education system. Support for private schools averaged 34%, ranging from 14% in Sweden to 49% in Ireland (Busemeyer et al., 2020).

Analysis of the 2016 International Social Survey Programme (ISSP) special module data on the role of government, commissioned for this report, addressed public education support using a sample of 35 countries, including 10 middle-income countries (Edlund and Lindh, 2021). Overall, 75% of respondents favoured more spending on education, with country shares ranging from 52% in the Czech Republic to 95% in the Philippines. Support for more public education spending increases with income inequality (McCall, 2016); the Czech Republic was among the countries with the lowest inequality and the Philippines among those with the highest. On average, support for more public education spending was stronger in middle-income countries (89%) than in high-income countries (68%) (**Figure 1.1**).

Overall, 89% of adult respondents said the primary responsibility for providing school education rested with governments, while 6% said families and 5% other institutions (private companies and for-profit organizations; non-profit organizations, charities and cooperatives; and religious organizations). Respondents in a few countries were outliers. Reflecting strong exposure to non-state provision, respondents in India (46%),<sup>2</sup> the Philippines (63%) and Chile (76%) expressed the lowest support for public provision (**Figure 1.2**).

Ideology partly explains attitudes in many countries. Sociocultural ideology is twice as likely as socioeconomic ideology (in 80% of countries vs 40%) to be a significant factor in attitudes on public spending and public provision. In poorer countries, the relationship between socioeconomic ideology and attitudes is weaker for public spending and stronger for public provision (**Figure 1.3**). The bottom line is that ideology matters. This is why ideas are at the centre of arguments for or against the role of non-state actors in education.

Before exploring three key debates about non-state provision, a finding that merits reflection relates to the role of education itself. In 46% of the countries analysed, a higher level of education was associated with a positive attitude towards more public spending on education,

<sup>1</sup> The section draws on Edlund and Lindh (2021).

<sup>2</sup> The regional edition of this report on non-state actors in education will be dedicated to South Asia.

#### FIGURE 1.1:

The more unequal a country, the more people favour increases in public education spending

Percentage of adults favouring more or much more spending on education and Gini index of income inequality, selected countries, 2016



Source: GEM Report team based on Edlund and Lindh (2021) and the World Development Indicators.

with a much stronger association in the richest two thirds of countries (64%) than in the poorest third (15%). By contrast, a higher level of education is associated with a positive attitude towards a greater government role in provision in just 9% of countries: 15% of the richest countries and none of the poorest. In practice, the higher the education level of a parent, the higher their income and the higher the tendency, as will be seen, to send children to private school. This poses a moral paradox (Swift, 2003): Education is supposed to build attitudes in support of social inclusion and cohesion; however, a common complaint, especially in countries where private provision is growing, is that elites are the first to reject public education, leading to a societal belief that public education quality is poor and a downward spiral into belief that public education cannot improve.

#### DIVERSE ARGUMENTS DRIVE DEBATE FOR OR AGAINST NON-STATE PROVISION

Proponents and opponents of non-state actors in education argue their cases in relation to the capacity and legitimacy of state and non-state actors to promote efficiency, equity and inclusion, and innovation in education. These issues are seen through the lens of whether people believe education is a good or service to be procured through the market and whether people should be able to choose education.



### ARE NON-STATE ACTORS MORE COST-EFFICIENT IN EDUCATION?

Proponents of non-state activity in education argue that it is inevitable since the state cannot cater for the full range of parental demands for education. Non-state actors have a variety of incentives to offer education goods and services the state cannot provide. Some may be motivated by charity, others by beliefs and ideas, yet others by profit. Regardless, if the supply of these goods and services is responsive to demand, then a market mechanism is possible. Others argue that education does not suit a conventional market. Instead, it has been proposed, and in a few cases attempted, to create a planned education market: Producers, including non-profit and for-profit organizations, compete for public contracts, while consumers buy services with vouchers rather than cash (Le Grand, 2003). Through such a quasi-market, the government may not only try to increase the supply of education but also achieve other objectives, such as cost-efficiency. For this to happen, non-state provision would need to be more cost-efficient than public provision.

#### FIGURE 1.3:

#### Ideology is a key factor in attitudes on government's role in education spending and provision

Percentage of countries where ideology supporting a greater state role in redistribution (socioeconomic) and a liberal approach on gender, religion, equality and environment issues (sociocultural) explains support for more public spending and provision in education, by country income, 2016



In principle, if cost-efficiency in running schools is proven to be feasible, then this raises various issues. Cost-efficiency should not be the preserve and commercial secret of non-state schools; rather, knowledge of such practices should be diffused throughout the education system and be practised by all schools, state and non-state alike.

Where would such gains come from? The largest education cost is teacher salaries. If a case could be made that teachers in a country were paid too much, then this should be a matter for public policy to resolve, not a reason to change the model of provision. Non-state actors may increase cost-efficiency by hiring young or unqualified teachers in a few schools, but this is not a sustainable solution. Alternatively, non-state providers may be tempted to reduce inputs by focusing on subjects whose results are measured, which may matter for their funding, while neglecting other subjects; this is also not sustainable.

Finally, making reliable cost comparisons between state and non-state schools is challenging. For instance, public schools tend to serve more disadvantaged populations, which are costlier to educate, and are more likely to be in rural areas, where costs again are higher.

#### DO NON-STATE ACTORS DELIVER EQUITY AND INCLUSION IN EDUCATION?

The second set of arguments is about equity and inclusion. Those who oppose non-state schools point to problems caused by school choice. If parents can choose the school they want, without any guiding regulations, then the richest are most likely to be able to afford the best, often non-state schools, exacerbating inequality, stratification and segregation.

Recent reports by UN Special Rapporteurs on the right to education argue that privatization promotes segregation and group differentiation due to the cost of access to various types of schools, and that private schools often mislead ill-informed parents about the quality of education they offer (Singh, 2015).

There are equity implications in making education goods or services marketable. Parental decision making requires good information. However, information on school characteristics is lacking or, if it exists, is unequally provided, as more disadvantaged populations have less access to it. Moreover, there are several hard-to-reach populations to whom providers may be reluctant to provide services, such as those living in remote areas (Srivastava, 2020). In 2019, human rights, academic and advocacy experts formulated 10 'guiding principles on the human rights obligations of States to provide public education and to regulate private involvement in education', known as the Abidjan Principles (Skelton et al., 2019). The principles aim to provide a normative framework to inform debate and assessment of privatization's role in education so as to guide countries in implementing related international law (Adamson et al., 2021). They accept that governments can fund non-state schools that comply with government standards. Some of the principles call for mechanisms of public accountability, regulation and compliance monitoring.

Proponents of non-state provision support these mechanisms and argue that, far from violating the right to education, non-state providers contribute to its fulfilment. In this view, international law does not prescribe how states must fulfil their education obligation. Thus governments can choose a mix of public and private education as long as they assume responsibility for regulation that imposes reasonable obligations (Emmerson, 2020).

In many contexts, non-state actors have filled genuine gaps in education provision, often for disadvantaged groups neglected by public systems. Governments are often reluctant to set up schools in informal settlements. A review of slums/underserved areas in Pakistan's eight largest cities found that 25% lacked schools; in the rest, 74% of the schools were private (CHIP Training and Consulting, 2020). Non-state actors also make valuable contributions in crisis and emergency contexts. Following the catastrophic earthquake in Nepal in 2015, non-governmental organizations (NGOs) provided supplementary education structures in locations the government was slow to reach (Street Child, 2021). In El Salvador, in urban areas afflicted by violence and gangs, the share of enrolment in non-state schools is double the national average (USAID et al., 2018).

A separate set of considerations relates to inclusion. Some of those who think government should not have a primary role in education provision challenge its authority to decide on education content or its ability to deliver education of a desired standard. Reasons include religion, political ideology, mother tongue and accessibility for learners with disabilities. Members of the Universal Declaration of Human Rights drafting committee were influenced by the then-recent memory of how public education was used to indoctrinate youth during the Nazi regime in Germany. They wanted to provide a counterpoint to the contested reference to education being compulsory, which had been approved by only a single vote margin in May 1948. In addition, they were motivated by their national and other contexts. Led by Lebanon, where education was traditionally segregated along sectarian lines, the UN General Assembly adopted the paragraph on the parental right 'to choose the kind of education that shall be given to their children', albeit by a small margin, 17 votes to 13 (with 7 abstentions), in November 1948 (Stanfield, 2021).

In many countries, some groups would prefer education to be adapted to their beliefs and principles. Parents may make a case for separate and non-state provision due to concern that the local public school threatens the values of the cultural, ethnic, linguistic or religious community in which they want to raise their child. Governments may argue that this conflicts with their commitment to ensure equitable and inclusive education and interferes with their ability to apply uniform standards in an effort to provide the same quality of education to all children, without exception.

Nevertheless, governments explicitly or implicitly delegate responsibility to non-state actors for various reasons. In a post-conflict environment, a government may allow non-state actors to provide education as a peacebuilding measure, for instance to an ethnic minority. Governments may allow religious communities to run schools, although some of these schools may controversially seek exceptions from the national curriculum, for instance in science or civic education.

In most countries, education for children with disabilities was historically provided by non-state charitable actors. Over time, governments have adopted legislation on inclusive education that expects all children to attend local mainstream schools. But when their transition is not well prepared, these students may suffer from prejudice and stigmatization. Parents who feel public schools are not catering for the needs of their children may place them back into non-state special schools. Who is to blame for parallel education systems? States that inadequately fund mainstream schools, or non-state actors that respond to family demands? Conversely, non-state schools reacting to pressure to compete may exclude some students, including children with disabilities, violating core principles of equity and non-discrimination in education.

Last but not least, non-state actors influence equity and inclusion in education in contrasting ways. Several civil society organizations and, recently, philanthropic foundations have advocated for education to be inclusive, supporting legislation and policy changes. But when such actors' advocacy efforts are so strong that they effectively lead public debate and determine policy change, questions arise as to whether such influence is legitimate or undermines democratic processes.

### DO NON-STATE ACTORS BRING MORE INNOVATION TO EDUCATION?

Proponents of non-state involvement in education assert that it helps increase innovation. Many ideas that have transformed understandings of pedagogy emerged at the margins of public education systems or even outside them. A review of the lives of leading thinkers on education shows that many questioned public education systems' ability to educate people (IBE, 2006). They called for these systems to be fundamentally overhauled or bypassed. Many were ostracized by the very systems they sought to improve, or found more conducive settings in which to apply their ideas, some of which were eventually brought into public education, often heavily diluted.

Public education systems have grown into large centralized bureaucracies that can lose sight of the populations they need to serve. A common criticism is that they blunt initiative, force standardization and demotivate students and teachers. Non-state actors could address unmet needs, offering opportunities for dedicated educators to innovate in education delivery and management without the burden of administrative rules. An analysis of 3,000 innovations in education ('an idea or technology that is a break from previous practice, and is often new in a particular context, even if not new to the world') suggested that 60% were implemented by NGOs, 26% by for-profit organizations and 12% by public actors (Winthrop, 2018, p. 6). NGOs innovate in areas from content (e.g. entrepreneurship education) to inputs (e.g. supplementary reading materials) and from systems (e.g. teacher motivation) to monitoring (e.g. accountability).

However, innovation is often a buzzword, self-reported and possibly exaggerated for fundraising or publicity. Trying to claim a competitive advantage, non-state providers may point to what is not working in public education, sowing distrust. But what is presented as successful innovation in small-scale controlled environments may not be innovative, let alone replicable.

Some non-state actors are testing whether their innovations work in public education systems. Promoting Equality in African Schools, a UK-based NGO operating 32 secondary schools in Uganda and Zambia, works in collaboration with Uganda's education ministry to adapt and implement components of its support and supervision model (Chu and Channa, 2019). Pratham, an Indian NGO, has implemented its Read India, a programme which supports the acquisition of foundational literacy and numeracy skills both directly with schools and communities and indirectly in collaboration with state and local governments (Banerji and Chavan, 2016). Escola Nova 21, an alliance of non-state actors and the Barcelona Provincial Council, aimed to strengthen the public school system in Catalonia, Spain. It consolidated school transformation projects and experimented with protocols for system-wide change (Escola Nova 21, 2020). But these are exceptions; few non-state actors have incentives to support state education.

Delivering innovation is a complex task for public education systems. Changes need to be piloted and tested for scalability. Challenges can include bureaucratic obstacles, organizational capacity gaps, lack of teacher and parental motivation, limited financial means and political meddling and opposition. However, public education systems are not negatively predisposed to innovation by design. There are a variety of mechanisms to introduce change.

Debate on innovation is often obscured by key concepts being referred to in contradictory ways. Standardization is maligned by those opposing what they see as public education systems' rigidity, conformity and lack of differentiation. But those defending equity in education promote standardization in their efforts to abolish selective schools and champion common core curricula to ensure standards are met in all schools. While dismissing the idea that standardization necessarily means uniformity of approach, they suggest that non-state education provision is more likely to lead to standardization in that sense, arguing that competitive pressure, often influenced by private providers, may accelerate a tendency to conform. Ultimately, whether standardization discourages innovation depends on what standards are defined, how they are measured and assessed, what incentives exist to achieve them and what feedback mechanisms are in place so schools can learn from good practices. The risks increase when standards are defined using a narrow set of learning outcomes.

Terms such as 'accountability', 'autonomy' and 'choice' have been both hailed and demonized as organizational principles of education. They are commonly linked to non-state actors' increasing role in education. For instance, the World Bank based its promotion of private education provision on the concepts of local decision making and fiscal decentralization; school autonomy regarding resources, personnel and content; standards and accountability mechanisms; and parental voice (Baum et al., 2014). But granting more authority to local government, schools and parents may exacerbate inequality in allocations between rich and poor districts, urban and rural schools, and more and less educated parents. Choice, the driver of competition in the marketplace, may not work the same way in education. While any of these ideas can be examined on their merits, they do not necessarily justify a bigger role for non-state, and particularly private, education provision.

#### MYTHS ABOUT STATE AND NON-STATE ACTORS IN EDUCATION PREVAIL

This report looks at a wide set of evidence from around the world to review the role non-state actors play in education. As noted above, any actor that does not represent the state is considered a non-state actor with a voice and stake in education, from impoverished parents spending money on education to corporations whose market dominance can shape education systems. The report does not just look at non-state actors in education through the conventional perspective of private schools but also examines other direct and indirect activities, from private tuition and assessment systems to influence on university research agendas and sales of educational toys. It maps activities that affect people of various ages, from young children in care to adults in professional training, which is mainly provided by non-state actors. Ten recurring myths about state and non-state actors in education are questioned throughout this report.

#### **MYTH 1.**

State and non-state actors can be clearly distinguished.



Discussion of non-state actors in education typically involves a binary classification: public and private schools. In practice, the landscape is more complex and distinctions are far less clear-cut. Non-state actors are highly heterogeneous. They enter the education sector for diverse reasons related to ideas, values, beliefs and interests. Many enter into formal or informal organizational arrangements with government, including contracting and public-private partnerships, which blur distinguishing lines. While actors usually have a defined purpose (e.g. provision, finance, regulation, management), with agreed terms of reference (e.g. objectives, time period, resource-sharing), the processes governing them are not neat, orderly, linear, rational or collaborative. Power between and among state and non-state actors is not equally shared or balanced. Interests are neither symbiotic nor focused solely or even primarily on increasing quality or efficiency. They likely result from bargaining between actors having different, incomplete or privileged information. Formal rules are mediated through informal norms and sometimes hidden practices for advantage, including economic gain, legitimacy and extension of influence or power (Srivastava, 2020).

#### **MYTH 2.**

The extent of privatization is known.



Descriptions of trends in the role of non-state actors often rely on the share of private institutions in total enrolment. After the share grew by 7 percentage points to 17% in primary and 26% in secondary education within a 10-year period, it has remained relatively stable since 2014. But these statistics underestimate the extent of privatization if they exclude unregistered institutions and overestimate it if they classify as private institutions that, for all intents and purposes, are public. Nor do they reflect the fact that technical, vocational and adult education largely takes place at work, outside the government's purview. Even in countries with no official non-state provision, including Cuba (Gonzales, 2014) and the Democratic People's Republic of Korea (Hui, 2019), the private tutoring industry has grown in recent years. How do countries account for public school teachers who supplement their income by teaching students after hours? How public is an education system that outsources textbooks, assessment or data management, or even catering and transport? Is a government policy written by a lobbyist still considered public?

#### MYTH 3.

The private sector is to blame for privatization in education.



Opponents of non-state education provision often blame providers for private schools' growth, but the growth seems to be a symptom, not the cause. The vast majority of private providers are single-proprietor schools. They emerged in response to genuine parental concerns about public school quality lowered by neglect. Many low- and middle-income countries' problems in public education may not be of their own making. Severe structural adjustment programmes often reduced funding available to public education and other services. Institutions that took decades to build quickly unravelled and were impossible to restore. When the decline in quality became clear, rich and, to a lesser extent, also poorer households left the public system, which undermined its support and left it underfunded. Related factors that may have exacerbated such situations include elitism among political leaders, which increased their tolerance for inequality and reduced their commitment to protect public education and the disadvantaged populations that benefited from it.

MYTH 4. Public education is equitable.



Another myth is that public education is free. Households often incur high education costs through hidden fees, avoidable out-of-pocket payments and additional expenditure to compensate for what public schools do not offer. Few policies and programmes (none in poor countries) target marginalized populations for benefiting from public education expenditure. While it is common to criticize education systems that have opened the doors to non-state providers, which exacerbate inequality, many public education systems fail to prevent stratification and segregation. An index of social diversity in schools in Latin America, based on the results of the Programme for International Student Assessment, found that Argentina, Brazil, Chile and Mexico had similar high levels of segregation in 2018 (OECD, 2019), although only Chile tends to be criticized for the high share of private institutions in primary and secondary enrolment.

#### **MYTH 5**.

Parents base school choice on robust information about quality.



A foundational assumption among those who support non-state schools and school choice is that parents, as consumers, have access to information about the best schools and use that information efficiently. In practice, as the previous muth suggests, this is a fallacy. Differences between schools are small. Although a visit to a school may reveal a lot about its quality, few information sources, if any, can identify better schools from a distance. What defines a better school is contested and cannot be summarized in a single score. The data required to identify the value schools add to children's academic outcomes are too complex for most countries to manage, let alone communicate. In any case, parents often ignore such information. They choose schools that appeal to them for other reasons: religious beliefs, convenience and students' demographic characteristics. In fact, parents often look for schools whose students have the social status to which they aspire and take advice from their social networks. Finally, in practice, school choice is not available for many households, notably those in rural areas.

#### **MYTH 6**.

Competition leads to school improvement.



S DO NOT NECESSARILY IMPROVE WHEN THEY HAVE TO COMPETE

Another tenet of supporters of non-state activity in education is that change is better than the status quo. Public institutions transform slowly; public education institutions are particularly slow to respond to individual and societal demands for relevant skills. Civil servants who do not perform are often protected, and vested interests block innovations that could increase efficiency and guality. Alternative providers and the opportunity to choose them should ensure that public schools accelerate needed reforms. In everyday life, accountability and healthy competition motivate some people to improve. In the economic sphere, firms compete to survive, as profit making is why they exist. But it is not clear how such dynamics play out in education. Studies that demonstrate system-wide effects of competition are rare, due to the complexity of the subject matter, and findings have been inconclusive. Worse, competition can lead non-state schools to pander to parents' aspirations, against good pedagogical practice. For instance, they may choose English as the language of instruction no matter what language learners speak at home.

#### **MYTH 7.**

Private schools and universities are better.



Whether private schools and universities are better than public ones is much debated. Comparison of public and private school examination pass rates is the usual evidence relied upon for school league tables as reported by the media and read by parents. The same is true of university rankings, which have grown in number and popularity as basic reference points of performance. In practice, comparisons should take into account differences between school and university types. Student intake varies, with better-off, well-educated and highly aspirational parents far more likely to choose a private school. Private schools, in turn, may be able to screen students to maximize the possibility of top results. Often (though not always) better resourced, they can more easily fulfil the potential of easy-to-teach students. When such factors are controlled for, the gap between public and private schools is usually slashed or eliminated. And this is just one comparison criterion. Should only final results be compared, not the contribution of each school and university type to improvement of students' results? Should academic outcomes be compared, but not life skills? At what point in life should results be compared?

**MYTH 8.** 

The private sector is a solution to the out-of-school challenge.



With more than 350 million primary and secondary school students enrolled in private institutions, it is sometimes argued that non-state actors can help increase enrolment levels and contribute to achievement of SDG 4. It is true that crisis is inevitable if these students switch over to the public education system, as happened in countries where students left private schools during the COVID-19 pandemic due to financial issues or dissatisfaction with the schools. However, non-state actors are not operating at the scale necessary to contribute to universal primary and secondary education completion. Private schools are booming in urban areas, where enrolment levels are already close to universal. But they are largely absent in rural areas. And in low- and middle-income countries, children from the richest 20% of households are 10 times more likely to attend a private school than their peers from the poorest 20%. The potential of low-fee private schools to improve equity has been exaggerated.

#### **MYTH 9.**

The private sector is a solution to education financing gaps.



The Education 2030 Framework for Action holds high hopes that the 'private sector, philanthropic organizations and foundations can play an important role, using their ... business expertise and financial resources to strengthen public education', because 'the private sector has emerged as a contributor with significant potential to complement resources for education'. The Global Partnership for Education developed its Private Sector Strategy to 'engage, co-create and leverage private sector ... finance' to advance its goals. Such documents suggest that private-sector organizations can fill the financing gap to achieve SDG 4. There is no evidence so far that they are willing or able to do so. But the private sector could make many other contributions. The obvious one is paying tax, especially in low- and lower-middle-income countries where domestic revenue mobilization rates are low and opportunities for tax evasion and avoidance are rife. Corporations could refrain from corruption and from lobbying that distorts public spending in education. The private sector leads in terms of skills development but could do more. Businesses could take a stronger lead in offering childcare services as required by national regulations.

**MYTH 10**.

Regulations can address all concerns about non-state provision.



There is consensus that non-state activity in education should be regulated. Not everyone should be allowed to open a school. Not every building is suitable to host students. Not every textbook should be used in classrooms. Getting these basic conditions met is challenging, as governments tend to do little monitoring of any regulations imposed. But the main concern is that regulations do not meaningfully address how to promote equity and quality. Non-state education activity can have system-wide implications on equity and quality. Yet few governments monitor whether the flight of wealthier households to private schools segregates the education system. Few governments monitor how much money households pay out of pocket for education and what this means for equity. Many governments allow selective school admissions. Few regulate private supplementary tuition or lobbying, which remains largely undefined under the guise of partnerships. Even fewer have the resources to implement and enforce regulations effectively. Finally, regulations cannot fix problems. They provide guidance to actors who have to abide by them to ensure equitable education.

#### **GUIDE TO THE REPORT**

About a quarter of a century ago in the United States, when evidence started emerging about the unequal effects of new organizational forms in public education based on school choice, the authors of an early study aptly summarized the findings with two questions: Who chooses? Who loses? (Fuller and Elmore, 1996). As more evidence accumulates on the mechanics, effectiveness and consequences of school choice around the world, the *Global Education Monitoring Report* takes these questions to a global audience. Recommendations at the end of this chapter refer to the most important debates in relation to achieving SDG 4.

The thematic part of the report is organized into seven chapters. The first four address key aspects of non-state activity in primary and secondary education – provision, regulation, financing and influence – and the last three review these aspects at other education levels, which tend to receive less attention: early childhood education, tertiary education and technical, vocational and adult education.

Chapter 2 analyses three kinds of **provision** of education goods and services. First, core education services are examined. A typology of non-state schools is developed, based on their relationship with the state, motivations of their establishment and costs to households. The prevalence of non-state providers and their key characteristics, such as resources and outcomes, are monitored. Second, other learning-related support goods and services are reviewed, including supplementary private tutoring and textbook publishing. Third, the chapter addresses other support goods and services, increasingly procured privately, such as infrastructure and catering. The effects of different kinds of non-state provision on access, equity, inclusion, quality, learning and efficiency are discussed.

Chapter 3 focuses on **governance and regulation** of non-state actors in education and whether existing mechanisms are fit for purpose. It approaches the issue from a system-wide perspective, recognizing that both public and private actors need to be held accountable for compliance with minimum standards. The analysis covers entry and exit, affordability, teachers and curricula, and quality assurance. It is based on profiles of regulatory practices in 211 education systems, prepared by the GEM Report team and available at the Profiles for Enhancing Education Reviews (PEER) website. Special attention is paid to regulation implementation and enforcement.

Chapter 4 examines **financing** aspects of non-state activity in education. It looks at the ways governments support non-state providers, the extent to which households shoulder a share of total education costs, engagement of non-state actors by donors and the role of corporations and philanthropic foundations as education funders.

Chapter 5 reviews an often overlooked factor, the role of **influence** in swaying public opinion and public policy for or against non-state actors in education. It describes diverse channels, such as advocacy, research, funding, lobbying and the selling of goods and services, and various non-state actors, including corporations, teacher unions, civil society organizations, think tanks, private-sector associations and the media.

Chapter 6 looks at **early childhood care and education**, where non-state provision is more prevalent than in primary and secondary education in many countries, and where compliance with standards needs to be closely monitored.

Chapter 7 unpacks the special case of **tertiary education**, where expansion of private provision has been rapid in several countries, posing particular challenges for governments that wish to promote equity and assure quality. Post-secondary teacher training institutions are another area in which non-state provision has emerged.

Chapter 8 looks at **technical**, **vocational and adult education and training**, where non-state actors are dominant. Public provision is limited to general skills, while public funding mechanisms try to encourage businesses to provide more on-the-job training and individuals to continue developing their skills.

Finally, the monitoring part of the report, Chapters 9 to 21, serves two purposes. It updates on progress towards the **SDG 4 targets**, including the interrelationship of education with three **other SDGs** and the evolution of education **financing**, based on data mostly up to 2019. In addition, it summarizes the latest evidence on the **impact of COVID-19** on each SDG 4 target. This is the most serious crisis ever to have hit all the world's education systems at once, and has also negatively affected data systems. As the crisis continues to unfold, insights are emerging from scattered evidence.



#### RECOMMENDATIONS

The diversity of non-state activity in education is often not understood. Far from a simple public vs private dichotomy, there is a variety of non-state school types in terms of management, ownership, financing, motivations, profit orientation and fee charging, as well as various types of relationship with the state. Moreover, the role of non-state actors extends well beyond provision of schooling to many other interventions (e.g. from learning assessments to supplementary tuition), at various education levels (e.g. from young children receiving care to adults learning foreign languages) and through multiple channels of influence (e.g. from lobbying to research). The question for policymakers is not just about whether non-state involvement in education meets agreed standards of quality, but also how non-state actors help or hinder efforts to ensure equity and inclusion in education.

Two strategic directions, relating to funding and provision, stand out in relation to governments' task of protecting and fulfilling the right to education. First, governments pledged in 2015 that all children and young people would have free, publicly funded access to a year of pre-primary and 12 years of primary and secondary education. However, with one in three countries devoting less than 4% of GDP and 15% of total public spending to education – the internationally agreed minimum benchmarks – it is clear that many are not matching this commitment with the required funding.

Second, governments need to decide how strong a role they will play in delivering and managing education. Their perspectives vis-à-vis school choice and non-state actors vary widely. In some countries, for cultural, religious and historical reasons, non-state actors have long been a foundation of the education system; in other countries, the role of non-state actors has been limited. A few introduced school choice as a conscious strategy to revamp the education system.

But for many poorer countries, a switch to school choice happened almost imperceptibly. Just as campaigns to accelerate mass education were taking off, drastic cuts in public education spending in the 1980s and early 1990s led to the erosion of public education quality. Communities, especially but not exclusively those better off, began to meet their own education needs, rejecting the public education system. With demand for schooling expanding at unprecedented rates, governments decided that oversight was beyond their means. Trust in governments to provide education was broken and hence support for public financing of education was undermined.

Various non-state actors have become more visible in many aspects of education. Businesses make choices about whether education is a lucrative activity and how to market their goods and services, but also to whom they are answerable: just shareholders or others as well? NGOs and civil society organizations choose priorities and decide how to address them: Should they fill gaps or advocate for the state to do so? Foundations also set priorities and choose how to influence society and how closely to work with education systems. Teachers and their organizations make choices that can strengthen or erode trust in public education systems.

The report's rallying call – Who chooses? Who loses? – is an invitation for policymakers to question relationships with non-state actors in terms of fundamental choices: between freedom of choice and equity, the two poles of education as a human right; between encouraging initiative (i.e. improving quality anywhere in the system) and setting standards (i.e. improving quality for all learners); between population groups of different means and different needs; between their immediate commitments (i.e. 12 years of free education under SDG 4) and those that are to be progressively realized (e.g. post-secondary education); and between education and other social sectors.

At the same time, as evidence in this introduction suggests, most parents would like their children to have access to a local, inclusive public school of good quality; they do not want uneven distribution of school quality, which forces parents to search for a school that would be better than the local school. They would prefer a guaranteed place for their children at the local school over having to go through a selective and competitive admission process. Although parents may vary in the priority they give to their children's education, no parent wants to spend out of pocket for what the education system should be providing.

With these thoughts in mind, the following recommendations were framed to help #RighttheRules to ensure that equity in education is protected in financing, quality, governance, innovation and policy making. The aim is to harness the contributions non-state actors can make to deliver education of quality without sacrificing equality. Mobilizing this potential could also challenge governments to purposefully address low quality and inequality that afflict public provision. The recommendations are primarily aimed at governments, whose responsibility it is to protect and fulfil the right to education. However, they are also meant to be used as an advocacy tool by all education actors committed to supporting progress towards SDG 4. As such, the recommendations call on all actors, state and non-state alike, to play #RightbytheRules.

Establishing a strong legal framework and an enabling policy and regulatory environment built on standards, information, incentives and accountability must be done in a participatory, transparent and equitable manner through coordination, collaboration and cooperation. Mechanisms need to be coherent across government bodies and programmes; reflect input from non-state actors, parents and communities; and incorporate good practice from other countries. They should not dilute government responsibilities for guaranteeing the right to education.

Compared to the attention given to choice and market mechanisms as levers to promote efficiency and innovation in education, this report looks at state and non-state actors through the lens of equity and inclusion, two commitments to which all countries subscribed in 2015. The recommendation is therefore that, in their laws, policies and programmes related to non-state actors, governments need to provide clear answers to these five core questions from an equity and inclusion perspective.

#### 1. DOES THE FINANCING OF EDUCATION FAVOUR SOME LEARNERS AND EXCLUDE OTHERS?

Fulfil the commitment to make 1 year of pre-primary and 12 years of primary and secondary education free – but publicly financed need not mean publicly provided if equity can be ensured



In many countries, far from education being free, families are forced to spend a significant amount, even when they send their children to public schools: more than 60% of total education spending in Kenya, Zambia and Zimbabwe comes from households whose children are attending public schools. About 16% of households save and 8% borrow to pay fees in low- and middle-income countries, often a consequence of governments not spending enough.

Governments should make education of good quality free at the point of access. They need to ensure that households do not pay for education goods and services that their countries have committed to make available free of charge. Eliminating formal fees is an obvious starting point but is rarely enough, as such fees are usually a small part of total costs. Providing learning materials to all for free and not imposing costly requirements that do not contribute to learning, such as uniforms, are other necessary steps.

Governments need to monitor out-of-pocket education spending with household income and expenditure surveys. Formal payments are often the only ones to which governments pay attention. They often turn their eyes away from other less well documented costs that increase inequality, such as private supplementary tuition. The effectiveness of policies that aim to target resources at disadvantaged learners needs to be evaluated and not assumed. All providers, state and non-state, must offer the same conditions to students. Support to strengthening public education systems has to be a priority. But a commitment for education to be publicly funded does not mean that all education must be publicly provided. In Finland and the Netherlands, funding depends not on school type but on number of students, with additional funds allocated for students from disadvantaged backgrounds and with special education needs. A foundation for equity is that all education institutions be treated as part of a single system with common rules and common financial support and oversight mechanisms.

Any attempts to diversify provision should be designed in a way that ensures equity. Countries that encourage the diversification of provision, for instance by contracting out the management of public schools, subsidizing the operational costs of private schools or funding households to attend the school of their choice, can easily end up benefiting learners who are well off. Three common design flaws need to be avoided in any mechanism that provides financial support to non-state actors: explicit or implicit student selection; explicit or implicit fee charges; and the operation of profit-seeking schools. Each of these poses a threat to equity.

Schools should not select students. Countries are committed to non-discrimination in education, a principle that must be reflected in school admission policies. This challenge affects state and non-state schools alike. At the same time, the right of families and students to choose is enshrined in conventions and legislations. Yet school choice can also exacerbate inequality. Poorer families tend to have fewer options to choose from. They have access to less accurate information on which to base their choice or may lack the skills to use information effectively. Even when information on schools is publicly available, it may be irrelevant or misleading: Measures of school quality tend to be narrow, naive and subject to schools manipulating data to make themselves appealing. To limit unintended consequences, governments should finance schools adequately and ensure that all schools meet quality standards. Choice should not be used to compensate for weak government accountability mechanisms in education.

Non-state providers funded by the state should not charge any fees. While all countries should aim to ensure that pre-primary, primary and secondary education are free, many are far from this ideal. Several countries, often with higher than average income inequality, have many independent fee-charging private education institutions. Even government-dependent private institutions
charge fees. Along with ensuring timely and adequate government funding for aided schools, as an intermediate step all governments should cap school fees and ban add-on fees. As of 2021, 55% of countries did not impose a cap on the cost of childcare and 30% of countries did not have regulations controlling fee levels in primary and secondary education. Several states in India have fee regulatory committees. Alternatively, governments could regulate fees indirectly by establishing a reference price to calculate support levels for poor students in non-state schools, as in Chile and Côte d'Ivoire, or introduce progressive fee and other financial support policies that target the most disadvantaged.

Profit making is inconsistent with the commitment to guarantee free pre-primary, primary and secondary education. Ultimately, governments should value equity more than businesses' equity value. Interim measures, such as regulating or banning profit making, can be used to address school choice policies that exacerbate inequality. Currently, only 17% of countries ban for-profit provision in pre-primary education and 28% in primary and secondary education.

#### 2. DO ALL LEARNERS RECEIVE THE QUALITY OF EDUCATION THEY ARE ENTITLED TO, OR ARE SOME SHORT-CHANGED?

Establish quality standards that apply to all state and non-state education institutions



Much attention is given to whether non-state schools are better than state schools. Non-state school admission selectivity makes this question difficult to answer. Most studies suggest that, if there are any differences between state and non-state schools, they are small once student background is taken into account. The main question for governments is whether each school meets the same standards. Equitable access to quality is a government responsibility. Governments need to establish quality standards that apply to all education institutions. Quality standards, covering not just inputs but also processes and results, protect those who have the most to lose. They should also cover safety (from teacher conduct to infrastructure) and inclusion (from non-discrimination to curriculum). These standards should not be so onerous that they divert resources from teaching and learning, especially in disadvantaged schools. They should relate to where schools are and help them improve. Their achievement should be assessed for each school, state or non-state, and publicly reported to increase accountability.

Teachers should be valued as professionals in all schools. Teacher qualifications and professional development opportunities should not vary by provider. In some countries, non-state providers have based their cost-efficiency on recruiting unqualified teachers at low salaries. Segmented teacher labour markets and wide inequality in teacher pay and conditions are strong signs of a malfunctioning education system. Governments need to gradually address all root causes of such imbalances. Whether in the private or the public system, teachers should have their labour rights protected. This was highlighted during the COVID-19 pandemic, when many private school teachers had their salaries cut or contracts terminated.

Quality assurance mechanisms need to be in place to monitor and enforce standards. Some non-state providers may enjoy considerable autonomy for historical or cultural reasons and be reluctant to accept scrutiny, whether they are faith-based or profit-oriented. Government oversight through school inspections, evaluations and learning assessments should be common to all providers. State capacity to implement these mechanisms should be factored into their design.

Countries need stronger quality assurance processes in technical, vocational and tertiary education. As governments increasingly subsidize individuals or contract with companies to promote training, they need to protect the most disadvantaged, who are vulnerable to fraud. Likewise, for-profit universities have come under scrutiny for offering education at the lowest end of the quality spectrum and engaging in malpractice. In Brazil, competition authorities have been struggling against market concentration. In the United States, 7 of the 10 biggest for-profit education companies have been found guilty of deceptive business practices, including predatory recruitment and marketing strategies. Governments need to prevent private supplementary tuition from having a negative impact on system guality and equity. Countries need to monitor private tutoring through surveys. Policy responses could include measures to support those at risk of not benefiting from such services, as in Japan; tutor teaching permit requirements, as in Malaysia; tutor standards, as in the Russian Federation and Ukraine; and online registers for better oversight, as in the Republic of Korea. China recently banned online private tutoring out of concern that it would negatively affect public education. Bans on private tutoring by serving teachers have been introduced, including in Kenya and Pakistan. However, enforcing such bans could lead to an informal market. The priority should be on addressing root causes, such as low teacher pay and high-stakes final examinations.

#### 3. ARE REGULATIONS EFFECTIVE AND FEASIBLE OR DO THEY HAVE UNINTENDED CONSEQUENCES THAT HARM DISADVANTAGED LEARNERS?

Establish common monitoring and support processes that apply to all state and non-state education institutions



Regulation of non-state activity in education should ensure that equity and inclusion standards are met in infrastructure, curricula, learning materials or teacher certification. But in practice, rules are often poorly designed or weakly implemented, leaving the door open for misconduct. Regulations need to treat non-state providers with trust as partners, rather than antagonists. And they need to protect the most vulnerable from exploitation. Governments need a clear vision and framework of how they want to engage non-state actors and communicate this vision through regulations. It is inconsistent to promote access to public education when a parallel private supplementary tuition system is prospering; to promote equity when a private childcare market offers unequal chances to young children from the start; to promote quality when private provision creates segregation and offers high standards only for some; and to promote inclusion when some schools can set their own admission criteria and fees. Governments should develop clear policy on such issues and use regulations to address them. The regulations should focus not on administrative details and unrealistic input standards, such as land and infrastructure requirements, but on education processes and results related to health, safety, guality and equity. Regulations should be periodically reviewed and incrementally adjusted in a transparent and participatory way, with input invited from state and non-state schools.

Education providers should always be regulated as education entities by education authorities and never just as commercial entities by market regulators. Some providers are regulated as businesses in early childhood care and education, private supplementary tuition and vocational training. Similarly, other providers are supervised by social protection ministries or religious authorities. Fragmentation of responsibilities implies that the education system is not seen as a whole, leading to both duplication and gaps.

Regulations need to be simple, transparent and efficient. The paradox is that regulatory capacity is lowest where the need for it, and the potential for corruption, is highest. Where capacity to monitor and enforce impractical rules is lacking, regulations become irrelevant and counterproductive. Governments need to develop more feasiblu monitored standards. listening to all stakeholders in an inclusive process. In sub-Saharan African cities, including Accra, Kampala and Lagos, bribes to government officials are common and only a fraction of non-state education providers are registered. Lack of monitoring capacity has also led to corruption in the Global North in cases involving non-state actors in vocational, technical and tertiary education, with issues such as illegal admissions, aggressive marketing, unfair treatment of staff and embezzlement of subsidies.

Governments need to be honest about the causes of the phenomenon they want to regulate. Just as they set common standards for state and non-state schools, they should also set common processes of monitoring and support. They must have oversight of all learners and cannot ignore those learners, often more disadvantaged, who are served by non-state providers. Governments also need to build a relationship of trust with non-state providers, encouraging them to register, eliminating arbitrariness in rules and communicating the right incentives for them to run their schools effectively for learners' benefit. And they need to protect children whose families have placed them in private schools and convey the message that they care for all children's education, irrespective of what type of school they attend.

## 4. ARE GOOD IDEAS FOR EDUCATION NURTURED OR STIFLED?

Facilitate the spread of innovation through the education system for the common good



Nobody has a monopoly on good ideas. Education is a social endeavour and a complex system. Policymakers should be able to identify good practice and innovation and to give good ideas time and space to develop. The challenge for policymakers is to encourage innovation, especially when the general public is likely to prefer conformity over experimentation. But what conditions promote or prevent innovations that tackle the big challenges of education systems? How can such innovations reach all and not just a few?

The government should work in partnership with all actors to build an education system that works for all, prioritizing a consultative approach. A culture of trust needs to be built to promote innovation. Creating conditions and offering platforms for multiple actors to interact and cooperate can help the public education and training system benefit from different views and different sources of expertise and innovate to remain relevant. This is especially important in the parts of the system most affected by rapid change, such as skills development.

To start with, governments need to nurture innovation in the public education system. Governments need to convey the message that they are committed to excellence. They should monitor learning and its determinants, evaluate where good practices are taking place, compile information about such practices, provide resources enabling practitioners to exchange experiences, and pilot good ideas and scale them up. Governments need fresh ideas in education and should bring together those who can develop them. To achieve that, they need to train education officers to be able to identify and develop such ideas.

Governments should also look for lessons from non-state actors. Autonomous, contextualized and flexible approaches to teaching and learning, especially as regards marginalized learners, can generate new insights. While governments may not run the non-state part of the education system, they are responsible for oversight. Admittedly, many governments have low capacity to monitor and evaluate the public school system, let alone a non-state one.

The government's role is to create the right environment to produce innovation. Education should not be seen as a market where education 'producers' have incentives to outcompete other providers. Instead, new ideas need to be shared, tested and, if proven, adopted, with the state facilitating the spread of ideas through the education system and non-state actors volunteering ideas for the common good rather than solely for economic motives.

#### 5. ARE ALL VOICES GIVEN EQUAL OPPORTUNITIES TO SHAPE THE PUBLIC DEBATE IN EDUCATION?

Maintain the transparency and integrity of the public education policy process so as to block vested interests



The mirror image of the case for innovation is the need to protect education from narrow economic and political interests. Just as policymakers should be open to multiple voices, they need to be wary of both loudspeakers and whisperers. Just as all actors should have a genuine seat at the table, it is also essential for communications with public officials about education legislation, policy and regulation to be transparent. Those with vested interests may be working to increase their market share or political power rather than for the public good. So how can public policy processes retain their integrity? The relevance of this question increased during the COVID-19 pandemic, which confronted governments with difficult decisions, such as choices over education technology. Policymakers need to take into account insights and perspectives from all stakeholders, not just the powerful. Some strong actors have the potential to sway policymakers enough to force their views into policy. This can be done by lobbying behind closed doors or through revolving doors enabling lobbyists and policymakers to swap seats. Some actors popularize the idea that education quality has fallen so much that only drastic solutions outside government can help. As the negative experience of contracting out state schools to non-state managers in Liberia shows, there is no quick route to resolving complex education challenges. Still, it is counterproductive to discredit non-state actors' involvement even when it offers quality-enhancing and equity-oriented solutions for the system.

Governments need to monitor and safeguard against lobbying by vested interests to prevent it from unduly influencing public policy. To maintain trust in public policy processes, a range of measures to promote transparency can be applied, depending on capacity. These include open and inclusive consultations; legislative committee hearings; government or court petitions; freedom of information acts promoting disclosure of donations to political parties and meetings with senior government officials; and rules against government officials who leave office taking positions from which they could derive private benefit, and against lobbyists and their sponsors taking public office. These recommendations also apply to international organizations, including UNESCO, all of which need a clear policy on engaging with non-state actors that prioritizes equity and inclusion.

Schoolchildren smile in an Islamic school in Sri Thammarat, Thailand.

CREDIT: Tarik Abdel-Monem



### CHAPTER



# Provision

## **KEY MESSAGES**

#### Non-state actors exist in multiple forms and provide education to 350 million children.

- Non-state schools differ by management, financing, values, profit intent and fees charged.
- The shares of private institutions worldwide increased from 10% in 2002 to 17% in 2013 in primary education and from 19% in 2004 to 26% in 2014 in secondary education. They have since remained roughly constant.
- For-profit schools are rare in middle- and high-income countries. Exceptions include Chile, where 26% of 15-year-olds attend such schools, and the United Arab Emirates, at 39%.
- In the last decade, international schools' numbers have almost doubled. Low-fee private schools are also proliferating. About 18% in Ghana and 70% in Lagos, Nigeria, are unregistered.

#### Private schools tend to have better resources and attract richer students.

- The richest households are 10 times more likely than the poorest to attend private primary school. Accounting for peer effects and selection almost eliminated the private school advantage in 40 richer countries analysed and slashed the premium of attending private school by half to two thirds in 31 low- and middle-income countries.
- In Latin American countries, public school students expect to participate in politics more than their private school peers.
- A review of 14 countries found that the top reasons for choosing a school were academic quality, teacher quality, location and safety. Religion, ethnicity and culture also mattered. In seven sub-Saharan African countries, parents' satisfaction rates were higher for faith-based than for public schools.
- Countries with choice policies have seen segregation increase. From 1998 to 2015, segregation of the lowestincome students increased by about 15% in large districts in the United States, partly due to charter school growth.

#### In addition to provision, non-state actors are involved in ancillary services.

- Private tutoring, which is increasing even in regions where it had been uncommon, such as sub-Saharan Africa, affects education system quality, teacher behaviour and equality. In India and Myanmar, teachers offering tutoring put less effort into regular lessons.
- The textbook industry is increasingly commercialized. Oversight varies dramatically by country: 12 of 50 countries have no process for approving instructional materials.
- In recent years, many companies have been linking textbooks, assessment systems and online learning.
   Market concentration can increase inequality and reduce accountability.
- Governments are outsourcing more support services in education, including transport, technology and food, but quality is not always protected. Several studies have found that districts in the United States with privately managed cafeterias offer less healthy meals.

Enrolment in non-state schools has been growing	2
State and non-state schools differ in student intake and available resources40	)
The impact of non-state schools on individual and system outcomes is contested 47	7
Non-state actors offer education goods and services other than schooling	1
Conclusion	2

Key to discussion about non-state actors in education is the perception that the number of students in private schools has been growing. However, at least three important issues on provision of education services by non-state actors deserve closer scrutiny. First, schools around the world defy simple categorization. Multiple criteria can be used to classify schools, revealing high diversity. Thus, determining the impact of non-state expansion in education provision is not straightforward.



Second, non-state schools have multiple effects, not all of which have been explored with the same depth of analysis. The effects on individual student outcomes, both in the short and, to a lesser extent, long run, including in domains other than education, have received the most attention. However, there are important and sometimes intractable questions on the effects of non-state providers on education systems, notably on equity, quality and efficiency. Is a parallel non-state education system an isolated arrangement for the children of wealthy or ambitious parents? Does it cause the state system to improve or deteriorate?

Third, non-state actors' engagement in education provision is not limited to instruction. They are also involved in other learning and support goods and services. This chapter therefore also maps ancillary services that routinely involve non-state actors, such as private supplementary tuition, learning materials and outsourced goods and services, from cleaning to catering.

#### ENROLMENT IN NON-STATE SCHOOLS HAS BEEN GROWING

Non-state education providers defy easy categorization. Criteria such as providers' relationship with the state, their motivations and their price can be used to group them (**Figure 2.1**). Yet all these criteria are on a continuum and the boundaries between types are not always clear. The degree to which any typology can be generalized across countries depends on data availability and national education systems' unique characteristics; other criteria may be more important in some national contexts than those captured here. However, regardless of typology, evidence suggests that the role of non-state actors in education provision has been growing. The UNESCO Institute for Statistics (UIS) is the main source of data on enrolment in public and private institutions. It defines private education institutions as those that are 'not operated by a public authority but controlled and managed, whether for profit or not, by a private body (e.g. non-governmental organisation, religious body, special interest group, foundation or business enterprise)'. The share of private institutions worldwide increased by 7 percentage points in about 10 years, from 10% in 2002 to 17% in 2013 in primary education and from 19% in 2004 to 26% in 2014 in secondary education, but has since remained roughly constant.

There is significant regional variation. Central and Southern Asia is the region with by far the highest share of private enrolment: 36% in primary and 48% in secondary education in 2019. It has also

#### 66

Central and Southern Asia has the highest share of private enrolment: 36% in primary and 48% in secondary education

99

experienced the largest absolute increases since 2000. The share generally does not exceed 20% in other regions. The fastest increases in primary education enrolment, with shares doubling or more, were observed in Northern Africa and Western Asia and in Eastern and South-eastern Asia, but the shares remain among the world's lowest at about 10%. Europe and Northern America had the lowest rate of increase in primary enrolment (remaining constant at about 10%) but the highest rate of increase in secondary enrolment (from 9% in 2006 to 15% in 2014). The share of private institutions grew more slowly than the global average in sub-Saharan Africa and in Latin America and the Caribbean (**Figure 2.2**).

#### FIGURE 2.2.

**Private enrolment shares are highest in Southern Asia** Percentage of enrolment in private institutions, by education level, 1990–2019



No relationship exists between the share of private institutions and country income. While the share is highest in lower-middle-income countries, influenced by India, low-, upper-middle and high-income countries report similar shares. However, some poorer countries have low overall enrolment rates and above average shares of private enrolment, which may suggest that private institutions fill a gap in demand. In 13 countries, at least 20% of lower secondary enrolment is in private institutions, while the net enrolment rate is below 50%; of those, 10 are in sub-Saharan Africa, mostly western Africa, 2 are in Central America (Guatemala and Honduras) and 1 is in Southern Asia (Pakistan) (**Figure 2.3**).

One dimension of non-state provision usually not captured in the data relates to non-state actors operating schools that are owned by the state. Analysis for this report found that this was the case for private actors in 29 out of 96 countries, for faith-based actors in 17 out of 83 countries and for other non-state actors in 19 of 81 countries.

The UIS definition of private schools focuses on management and operation but ignores the distinction between private schools that receive public funding and those that do not. Analysis for this report found that there are government-aided non-state schools in 171 out of 204 countries: These include private schools in 115 countries, faith-based schools in 120 countries; and non-governmental organization (NGO) and community schools in 81 countries. The Programme for International Student Assessment (PISA) defines private schools that receive less than 50% of their funds from public sources as 'private independent' and the rest as 'private dependent' (OECD, 2019a). In a sample of 50 education systems with an average private share of 19%, analysis of 2018 PISA data suggests that about 46% of the private secondary schools were dependent and 54% independent.

As with management, the reverse relationship where non-state actors fund schools owned by the state is less well known. Analysis for this report found that this was the case for private actors in 31 out of 77 countries, for faith-based actors in 8 out of 57 countries and for foundations or philanthropic organizations in 21 out of 64 countries. However, available information does not allow an assessment of the volume of this support.

#### FIGURE 2.3:

### Some poorer countries have a high share of private enrolment in lower secondary education even though overall enrolment is low

Share of enrolment in private institutions and adjusted net enrolment rate, lower secondary education, 2018 or latest year



## Faith-based schools can be found in 124 out of 196 countries

#### NON-STATE SCHOOLS DIFFER IN THEIR FOUNDERS' HIGHLY DIVERSE MOTIVATIONS

Many schooling systems have their roots in religious or community education objectives. Faith-based and community or NGO schools are typically motivated by values and charitable duties.

Faith-based schools continue to be significant in much of the world. Analysis for this report found that they could be found in 124 out of 196 countries. Christian faith-based schooling is state-owned and subsidized in countries such as Belgium (Wodon, 2021). Global primary and secondary enrolment in Catholic schools is estimated to have increased by 43% between 1995 and 2016 (Wodon, 2019). As of 2018, 35 million children were in Catholic primary schools, the majority in English-speaking eastern Africa. About 19 million were enrolled in Catholic secondary schools, some public and some private (Wodon, 2021) (Figure 2.4).

An analysis of faith-based schools for this report found that one in six primary and secondary schools is operated by Christian denominations in Cameroon, Madagascar and the United Republic of Tanzania, and more than one in two in the Democratic Republic of the Congo and Rwanda. Government and churches cooperate on all dimensions of education systems, including admission processes, teacher gualifications and salaries, and control of curriculum standards, although such cooperation is more effective in some countries than in others. In Cameroon, government has not reimbursed churches for teacher salaries, training and school construction, prompting requests for compensation as part of debt relief negotiations. By contrast, state-aided church schools in Rwanda receive timely salaries and national curricula are planned collaboratively (Scheunpflug and Wenz, 2021). Finally, religion is a common reason for some families to opt out of the education system altogether, choosing homeschooling instead (Box 2.1).

#### FIGURE 2.4.

#### Catholic faith-based schools are not always private

Percentage of students enrolled in Catholic institutions and in private institutions, selected countries, 2019 or latest year

99



#### BOX 2.1:

#### Some families opt out of state and non-state schools to educate children at home

Parents homeschool when they feel the education system does not fulfil their aspirations for safety, quality or content. It can be motivated by practical concerns, such as remote location or severe disability, but also by philosophical, religious or political reasons. About 3% of the student population in the United States is homeschooled, the world's largest share (Ray, 2017). A 2016 survey found that the main reasons for homeschooling were concern about school environment (34%), dissatisfaction with academic instruction (17%) and desire to provide religious (16%) and moral (5%) instruction (US Department of Education, 2019).

The evidence base for homeschooling in other contexts is limited, but interest is growing (Kunzman and Gaither, 2020), not least because of advocacy by a global conservative movement (Permoser and Stoeckl, 2020). In Brazil, although homeschooling had been illegal, around 22,000 students were in home education in 2019. A 2018 Supreme Court ruling had declared that it was not unconstitutional but needed further regulation. A draft law allowing homeschooling, a government priority, was approved by the Constitution and Justice Commission of the lower house of the National Congress in July 2021 (Costa, 2021).

Homeschooling can also reflect strong dissatisfaction with mainstream education. In the United Kingdom, a survey found that 24% of home educators had chosen homeschooling because of circumstances such as health reasons or special education needs (Smith and Nelson, 2015). In Australia, the primary motivations appeared to be concerns about school environment and curriculum quality (Jackson, 2017).

Home education was a necessity during the COVID-19 lockdown periods, which exacerbated gender imbalances as mothers took a higher share of responsibility for homeschooling. The US Census Bureau Household Pulse Survey found that homeschooling increased to 5.4% in April/May 2020 and 11% in September/October 2020 (Eggleston and Field, 2021). Homeschooling had adverse employment effects for mothers but not fathers (Petts et al., 2020).

99

Islamic education is provided in formal and non-formal settings at the primary and secondary levels in more than 50 countries with majority and minority Muslim populations, including in western Africa (e.g. Nigeria and Senegal) (d'Aiglepierre and Bauer, 2018), Northern Africa and Western Asia (e.g. Morocco, Saudi Arabia and Turkey), Southern Asia (e.g. Bangladesh and Pakistan) and South-eastern Asia (e.g. Indonesia and Malaysia). As with Christian faith-based schools, madrasas' organizational arrangements vary from state-owned to private subsidized and private independent, so they cannot all be labelled public or private (Daun, 2018). In Pakistan, 4.1 million students in 2017/18 were studying in more than 31,000 deeni madaris (religious schools), free religious education institutions for the poor from the pre-primary to post-secondary level. They represent 8% of overall enrolment and 18% of enrolment in private education institutions (Pakistan Ministry of Federal Education and Professional Training, 2021).

#### 66

In Pakistan, 18% of enrolment in private education institutions is in religious schools

In addition to management and financing, Islamic faith-based provision varies considerably in pedagogy and in the balance between teaching of religious and secular content. Until the mid-2000s, Malaysia used to have five types of religious schools: federal, state, State Islamic Religious Council, people's and private. The last two taught the national curriculum but were otherwise independent of government. The government has been focusing on the integration of religious and secular education (Hamid, 2018). By 2019, religious schools had been grouped into two broad categories: 314 national primary and secondary religious schools under the Ministry of Education and the state Religious Departments, and 221 state-aided religious schools jointly controlled by the ministry and the state religious authority or school board of trustees (PEER country profiles).

Nigeria has a variety of state, state-aided and independent Islamic faith-based schools. In 2018, nearly 2 million students were enrolled in Islamiyya primary schools (one third of the students were in private schools), 109,000 in Islamiyya lower secondary schools (half in private schools) and 364,000 in Tsangaya, or traditional Koranic schools (more than two thirds in private schools) (Nigeria UBEC, 2019). The state schools integrate religious and secular 66

## In Zambia, more than 2,400 community schools in 2015 accounted for about 20% of total primary enrolment

education and are jointly managed by the government and religious individuals or organizations. State-aided Islamiyya schools are also integrated and follow a government-approved curriculum. Independent faith-based schools include Tsangaya schools, Islamiyya schools with broader coverage of religious issues, and integrated Islamiyya schools (PEER country profiles).

Similar variety is observed in NGO schools' relationships with the state, further defying easy categorization. Analysis for this report found that there were NGO and community schools in 74 out of 196 countries. They tend to operate on the edges of the formal education system and focus on marginalized, out-of-school children, offering flexible approaches to learning (Rose, 2007). They may be spontaneous grass-roots interventions or calculated responses to donor funding. In the best cases, successful interventions have eventually been absorbed into the mainstream education system, meeting national standards and complementing public services.

In Southern Asia, there are two well-documented cases of large, non-formal, spontaneously developed education systems being gradually mainstreamed. In Bangladesh, after a cyclone, war and famine struck the country in the early to mid-1970s, several NGOs began charitable work in rural areas, including non-formal education programmes. By the early 2000s, these reached 1.5 million, or 9%, of the 16 million primary education students. About 70% of those attending NGO schools were girls. The system was gradually formalized, with NGO school students participating in primary education graduation examinations from the early 2010s, enabling them to transition to formal secondary education. By 2018, the system had begun to contract; the 674,000 students in NGO schools represented less than 4% of total enrolment (Bangladesh Directorate of Primary Education, 2019; Campaign for Popular Education, 2005).

In Afghanistan, a community-based education system responded to the major barriers to access for girls and children in remote areas after the end of the 1996–2001 Taliban regime. With collaboration by local government, communities and NGOs, the system reached 334,000, or 5%, of the 6.3 million primary education students in 2015. In 2018, a policy was developed to formalize this system. The National Education Strategic Plan 2017–21 envisaged expansion of the system in the short term and its eventual contraction by 2030 (Afghanistan Ministry of Education, 2016; Bakshi, 2020). But the latest political developments in the country put these plans at great risk. Conflicts and crises, often the backdrop of NGO involvement in education, overstretch governments' limited capacity and their ability to absorb services (**Box 2.2**).

99

Community schools were key to education expansion in sub-Saharan Africa. Some countries' education systems have a long history of community-operated and -supported schools, such as harambee secondary schools in Kenya and the United Republic of Tanzania, and écoles spontanées in Cameroon and Chad (Rose 2007; Martin, 2003). In Zambia, after a serious financial crisis followed by structural adjustment policies that constrained the state's capacity for delivery of essential services, a community school movement emerged, growing from 38 schools in 1996 to more than 2,400 in 2015 and accounting for about 20% of total primary enrolment. Only a few became state schools, although all had to follow the national curriculum after 2011 and received limited additional government funding. NGOs and local communities are relied upon for financing and management support, infrastructure and capacity development (Bamattre, 2018). The Zambia Community Schools Secretariat is responsible for coordinating ideas, funding support and curriculum reform with the government (Chilangwa, 2019).

Large-scale interventions are an exception. More commonly, NGOs address education of particular groups, such as children with disabilities, for whom public services may be lacking. International NGOs, funded from their own resources or from donor programmes, have often initiated education provision through community-based rehabilitation in some poorer countries.

In 2020, Humanity and Inclusion, an international NGO advocating for education of children with disabilities, implemented 52 projects in 27 countries (Humanity and Inclusion, 2021). In Central and Eastern Europe and the Caucasus, NGOs have led service provision, in some cases transforming and adjusting to the transition from a medical to an inclusive approach to education for children with disabilities. In Armenia, where inclusive education reforms have focused on placing children with disabilities in mainstream public schools, cooperation between local schools and NGOs operating at the grass-roots level has

#### Non-state education provision in emergency contexts offers relief, but concerns for sustainability remain

Non-state actors frequently provide education in fragile and conflict-affected countries, e.g. in refugee or internally displaced people's camps. Most non-state initiatives are relatively small; their scope, scale and long-term sustainability are critical issues, especially as they depend on humanitarian funding, which tends to be short-term and unpredictable (UNESCO, 2019). Strong involvement of non-state actors can even create disincentives for the state to fulfil its responsibilities.

In Afghanistan, Street Child, an NGO, worked with the government to support 100 community education centres in which an accelerated course compressed six years of schooling into three. Students were then referred to local education authorities to prepare for transition to nearby state schools. The centres demonstrated strong continuity, with 90% of students in Kabul's informal settlements transferring to state schools and 85% retained after one year. Yet the state lacked capacity and did not commit to ensuring that the centres would be absorbed into the state sector (Street Child, 2021).

A significant challenge in emergencies is the extent to which programmes use local NGO capacity, which contributes to long-term national sustainability. US donor agencies, to reduce risk, tend to work with international service providers, selecting local contractors only in countries with higher capacity levels (Harris and Brunjes, 2021). The US Agency for International Development introduced procurement reforms to direct more funding to local contractors. However, because of concern over increased risk, lower accountability and limited control over corruption (Dunning, 2013), the agency still typically works with selected NGO partners, mostly US based (USAID, 2019).

In Haiti, after the 2010 earthquake, the government received only 1% of humanitarian aid; instead, many NGOs and private contractors provided services. The number of NGOs operating in Haiti is unknown. Estimates range from the 343 NGOs registered with the Ministry of Planning to over 10,000, according to the Catholic Institute for International Relations. Criticism of the proliferation of NGOs and private actors has focused on where funding went, how lack of coordination hampered education delivery and the creation of parallel NGO structures, which further weakened government channels (Hsu and Schuller, 2020; Ramachandran and Walz, 2015).

Responses to the Syrian refugee crisis also included substantial non-state activity. A 2016 mapping of NGOs engaged in education found 144 organizations, of which 45% were civil society organizations, 32% businesses and 10% foundations. In total, 61% of the private actors did not have education as their mandate. Greater private participation in Syrian refugee education has resulted in less coordination and overdependence on technology solutions (Menashy and Zakharia, 2017, 2021).

been fragmented (Nazaryan, 2021). Not all NGOs manage to make such a transition smoothly and to engage constructively with government services.

NGOs have led education provision to excluded populations that the government education system does not serve, such as street children. In Kenya, education of street-connected children is left to NGOs, charities and religious groups working locally (Burkholder, 2020). The Consortium of Street Children, a global alliance working towards promoting street children's rights worldwide, has highlighted how formal and non-formal services provided by NGOs for street children and families were suspended during COVID-19, even as informal support networks struggled to cope (Edmonds and Macloed, 2021).

NGOs, community-based organizations and donors have been instrumental in advocating for education provision to ethnic and linguistic minorities. In the Plurinational State of Bolivia, Guatemala, Mexico and Peru, over 1 million students from indigenous communities attend intercultural bilingual schools, with growing government engagement (Cortina, 2017). Guatemala's Abriendo Oportunidades programme, launched by the Population Council and local and international partners, focuses on indigenous girls aged 8 to 18, using a culturally relevant, rights-based curriculum to build life skills and assets and provide hands-on professional training (Population Council, 2020).

## For-profit primary and secondary schools are an exception

A fundamental argument against for-profit independent private schools is that they undermine equity and interfere with governments' role as duty bearers of education. But providers' profit orientation is often not easy to discern. First, data may not distinguish between for-profit and non-profit (**Chapter 3**). Second, registration requirements may mask differences in actual profitability or wealth. In the United Kingdom, 'for-profit' is supposedly a small portion of the education system: 72% of independent schools are registered as charities (ISC, 2021).

#### 66

In Sweden, 76% of independent school students in 2019/20 attended schools run as limited companies

However, schools that do not operate on a for-profit basis may have operational profit while others may opt to be classified as 'for-profit' primarily to increase autonomy and reduce the regulation associated with operating as a charitable trust (Martin and Dunlop, 2019).

A question addressed to head teachers in the 2015 PISA offered insight into the extent of for-profit education in middle- and high-income countries: It suggested that such schools were a minority. Exceptions included Chile and the United Arab Emirates, where 26% and 39% of 15-year-olds, respectively, attended for-profit schools (**Figure 2.5**). The presence of sizeable for-profit provision may indicate government support to market-based education policies. In Sweden, 76% of independent school students in 2019/20, amounting to 14% of primary and secondary school enrolment, attended schools run as limited companies (Holmström, 2020).

The Gulf Cooperation Council countries, which have the world's highest immigration rates, usually have specific rules regarding education of non-nationals. In the United Arab Emirates, where almost 90% of the population has an immigrant background, education provision relies on for-profit international schools (Kamal, 2018). Except for Bahrain, which has not charged school fees or imposed any restrictions on immigrant enrolment in public schools, curricula and fees tend to be stratified by immigrant community in the other countries. For instance, Indian curriculum school fees are typically 75% lower than those at British and American curriculum schools (Kippels and Ridge, 2019).

#### NON-STATE SCHOOLS VARY BY LEVEL OF FEES CHARGED

Profit making in education is often confused with fee charging. Yet the two are not as closely related as might be imagined. Many non-profit schools charge fees. Conversely, the extent to which fees are charged may be the result of a policy decision to cover the cost of enrolment in a non-state school (**Chapter 4**).

The fee spectrum varies on the basis of consumer ability to pay. Two growing phenomena on the opposite ends of the spectrum are elite international private schools and low-fee private schools. The number of international schools increased from 7,655 in 2011 to 12,372 in 2021. More than 57% are located in Asia (ISC Research, 2021). Elite international schools generally provide international curricula, such as the International Baccalaureate, and use English as a medium of instruction (Bunnell et al., 2020). Large chains of global private schools include the GEMS Education group, which runs over 70 schools; Cognita in Asia, Europe and Latin America; and SABIS, which operates in 20 countries (L.E.K. Consulting, 2020).

#### FIGURE 2.5:

99

**Few countries have a sizeable for-profit sector** Percentage of students, by type of secondary school attended, selected countries, 2015



Notes: The analysis focuses on education systems where at least 10% of 15-year-olds were enrolled in non-state schools. Private schools are defined as those managed directly or indirectly by an NGO, church, trade union, business or other private institution. Head teachers answered whether schools were run by religious institutions, other not-for-profit or for-profit organizations.

GEM StatLink: https://bit.ly/GEM2021\_fig2\_5 Source: GEM Report calculations based on the 2015 PISA database.

## In Lagos, Nigeria, only some 30% of private schools were approved

Such schools may not integrate fully into national education systems. In Saudi Arabia, international schools' co-educational environments contrast with mainstream schools' single-sex organization (Hammad and Shah, 2018). In Sri Lanka, enrolment in international schools increased by 73% between 2017 and 2019, fuelled by the increasing popularity of English instruction and foreign curricula. Only 7 of the 389 international schools were registered with the Ministry of Education; the rest were registered as private businesses (Central Bank of Sri Lanka, 2020).

99

'Low fee', 'low cost' or 'affordable' private schools are terms describing a broad range of modestly priced schools in low- and middle-income countries. The terms have also been used to describe the lower-cost end of private schooling in Australia, Canada, the United Kingdom and the United States (Walford, 2011). There is no commonly agreed threshold for a private school to be considered low fee. Various qualitative descriptive criteria on ownership structure and quantitative objective criteria on fee levels have been used, although relevant information is usually patchy (Acholla, 2021).

The vast majority of low-fee private schools are single-proprietor schools. Their operational professionalism, infrastructure quality and school manager and proprietor background can differ considerably. Most such schools in urban areas of sub-Saharan Africa and South Asia are not registered (Härmä, 2021). In Lagos, Nigeria, a 2011 school census found that only 26% of private primary and secondary schools were registered. Even by 2018, only some 30% of private schools were approved (Ugbodaga, 2018). In Ghana, 18% of 36,000 private basic education providers in 2018/19 were not registered (Acholla, 2021).

Larger-scale operations include a variety of arrangements, from partnerships to chain schools. Church organizations, instead of individual entrepreneurs, may establish low-fee private schools, for instance in post-conflict, low-income contexts (Härmä, 2021), accompanying them with significant financing and other support from faith-based financial intermediaries (Acholla, 2021; Sivasubramaniam, 2021). Corporate chain schools include small and large national networks (e.g. APEC Schools in the Philippines; Silverleaf Academy in the United Republic of Tanzania; MA Ideal Schools in Hyderabad, India; and Samata Schools in Nepal); not-for-profit regional and international operator models (e.g. Promoting Equality in African Schools in Uganda and Zambia, the Forum for African Women Educationalists in Rwanda and The Citizens Foundation in Pakistan); and international for-profit chains with plans for considerable expansion (e.g. Omega Schools in Ghana and Bridge International Academies in India's Andhra Pradesh state as well as Kenya, Nigeria and Uganda) (Acholla, 2021). Discussion of low-fee private schools has sometimes exclusively focused on the last model (Espindola, 2019). Even though international chains make up only a small market segment and are unlikely to become major providers in low-income settings (Hares and Crawfurd, 2021), they have been disproportionately emphasized in research and policy discourse due to their profit orientation and potential influence on donor resource allocation decisions (UNESCO, 2017; Verger et al., 2018).

As these qualitative definitions of low-fee private schools have unclear boundaries, quantitative definitions have been proposed to define affordability using fee levels (**Table 2.1**). The Center for Indonesian Policy Studies calculated a threshold for Jakarta from the provincial monthly minimum wage (Rahman, 2016). India's National Institute of Public Finance and Policy used fee distribution at schools attended by children living in households in the bottom 40% of the consumption expenditure distribution, and identified two thresholds: 'basic' schools costing less than US\$115 a year, and 'standard' schools costing less than US\$215 a year (Bose et al., 2020).

#### STATE AND NON-STATE SCHOOLS DIFFER IN STUDENT INTAKE AND AVAILABLE RESOURCES

Once policymakers have identified an operational school typology that suits their needs, the key question is what effects non-state schools have on students and the education system. If an analysis is to be meaningful and relevant for policy, it should be noted that state and non-state schools differ in more than just operations, management, financing, motivation and fee levels. State and non-state actors also differ in student intake and access to resources. Only if such differences are accounted for can a comparison of state and non-state school performance be accurate.

## PUBLIC AND PRIVATE SCHOOLS DIFFER IN STUDENT INTAKE

Most households do not get to choose their schools, and few poor children have access to private schools. School and household surveys covering primary schools globally show that for a sample of 42 countries where

#### **TABLE 2.1:**

Selected low-fee private school definitions with qualitative and quantitative criteria

Qualitative criteria on ownership	Single proprietor	Differentiated by business and education background and skills, financial sustainability, fee range and government oversight	Emergent Established
	Larger scale	Differentiated by expansion strategy, financing operations and availability, involvement of government and international actors	Cooperative partnership Corporate chain
Quantitative criteria on fee levels	Relative	<ul> <li>&lt;11% of income of families at poverty line, US\$60/year (Tooley, 2013)</li> <li>&lt;10% of household income (Lewin, 2007)</li> <li>&lt;4% of the household budget (Barakat et al., 2014)</li> </ul>	
	Absolute	<ul> <li>&lt;1 day's earnings for monthly primary school fee (Srivastava, 2007)</li> <li>&lt;2 days' earnings for monthly secondary school fee (Srivastava, 2007)</li> <li>&lt;50% of minimum wage (Stern and Heyneman, 2013)</li> <li><us\$1.25 (low="" (tooley="" (very="" 2016)<="" <us\$2="" and="" cost)="" cost);="" day="" li="" longfield,="" low=""> </us\$1.25></li></ul>	

Source: GEM Report team based on Acholla (2021).

#### FIGURE 2.6:

#### Private schools serve the rich

Share of primary enrolment in private institutions, by wealth quintile, 2013–15



Notes: LLECE, 2013 = Laboratorio Latinoamericano de Evaluación de la Calidad de la Educación; PASEC, 2014 = Programme d'Analyse des Systèmes Educatifs de la CONFEMEN; TIMSS, 2015 = Trends in International Mathematics and Science Study.

GEM StatLink: https://bit.ly/GEM2021\_fig2\_6

Source: Patel and Sandefur (2020) based on analysis of citizen-led assessments, LLECE, PASEC and TIMMS data.

median private school attendance was 17%, the richest households (39%) were 10 times more likely than the poorest (4%) to attend private school (**Figure 2.6**). In 31 of the 42 countries with data, less than 7.5% of the poorest households' children attended private school.

In some countries, the extent of rich households' exit from public education is astonishing. In Congo, 87% of the richest but 6% of the poorest attend private school. In Honduras, the figures are 92% and 3%: Nearly all rich children go to private school and all poor children to public school. Even in countries with a high share of private institutions, the gap between richest and poorest persists: In Chile, where one in two children is in private primary school, 87% are from the richest households and 20% from the poorest. Only in Indonesia is the opposite observed: As discussed in the previous section, poorer children are served by faith-based institutions.

The belief that private schools cater for poorer segments of the population is mistaken. A review of Ugandan secondary education since 1950 found that new private or Catholic schools were not constructed in underserved areas (Wodon, 2020a). Part of the explanation may be that access to private schools is limited in rural areas, where most of the poor live (Härmä, 2021). But even in rural areas, as Annual Status of Education Reports in Pakistan show, the richest are three to four times more likely to attend private school. Even in urban Lagos, Nigeria, where 73% of children attend private school, only 33% of the poorest students attend a medium-cost and 17.5% a high-cost private school (Cambridge Education, 2021).



## Parents use different criteria when they can choose school

School choice is not open to all parents. Socioeconomic status is directly related to whether parents can choose between schools in the first place, what criteria they use and what information they base their decisions on. It is generally assumed that access to more schooling options allows parents to find a school that matches their preferences, to stay engaged and to see their children achieve better outcomes. However, poor parents usually lack choice and have limited access to information, and their dissatisfaction with public services is not heard. In Nepal, richer parents who chose medium-priced private schools were the most engaged and satisfied, while poorer parents, who had no choice and whose children went to poor-performing public schools, were dissatisfied and disengaged (Joshi, 2014a). Even when they can choose, choice by itself is not empowering if it is limited. For example, while parents in some poor urban areas in the United States have school choice, the local schools are not as good as those available to rich families (Ellison and Aloe, 2019; Jabbar and Lenhoff, 2020).

Where parents can make a choice, they base it on perceived or actual school quality but also on other dimensions that may be unrelated to quality, relying on objective information as well as shortcuts such as social networks or visual cues from the state of school infrastructure (Joshi, 2014a, 2014b). A review of 26 parental choice studies from 14 countries found the top reasons for choosing a school were academic quality, teacher quality, location and safety (Rohde et al., 2019). When choosing schools, parents in 17 Organisation for Economic Co-operation and Development (OECD) countries prioritized a safe environment (92% classified it as important or very important), an active and pleasant school climate (89%) and academic achievement (81%) (OECD, 2019b).

To evaluate the quality of education experience, parents refer to class size, teacher quality and effort, school responsiveness, discipline and safety, and language of instruction. In Kinshasa, Democratic Republic of the Congo, 45% of parents with children in private schools said they would prefer to send their children to public schools, but only if they

66

In a sample of 42 countries, the richest households (39%) were 10 times more likely than the poorest (4%) to attend private school

99



provided adequate security and quality, in terms of teacher quality, effort and absenteeism (Cambridge Education, 2021). In India, a child is more likely to attend private school if the teacher is local and almost always present (Kumar and Choudhury, 2020).

Parents' self-reported preferences may differ from their behaviour. Research in the United States has shown parents reporting they value academic guality but their actual choices suggesting they care more about schools' demographic composition than they are willing to admit (Jabbar and Lenhoff, 2020). Such findings also emerge from analysis of education systems where parents can choose between public schools, such as in New York, where parents choose schools enrolling high-achieving students (Abdulkadiroğlu et al., 2020). Analysis of school choices in Santiago, Chile, also found that parent decisions were more influenced by school demographics than academic reputation (Elacqua et al., 2006). Parents in Nepal focused on academic reasons for school selection, but some clearly stated that choosing a public school could have negative social status ramifications (Joshi, 2014b).

#### 66

## In New York, parents choose schools enrolling high-achieving students

"

Religion, ethnicity and culture are important determinants of choice. In Chile, school quality and religious education are more important for richer parents, while proximity to school is more important for poorer, indigenous and less educated parents (Hofflinger et al., 2020). A study of Islamic, Christian and public schools in Burkina Faso and Ghana showed that religious knowledge was an especially important reason for choosing an Islamic school (Gemignani et al., 2014). In seven sub-Saharan African countries, parents expressed higher satisfaction rates for faith-based than for public schools (Wodon, 2020c). In Malaysia, location and ethnicity, as reflected in medium of instruction, and school 'reputation' were the top reasons for choosing Chinese- and Malay-medium primary schools (Ting and Lee, 2019). An analysis of 15,000 students starting primary school in the Netherlands found that more



educated parents had stronger preferences for school quality than less educated parents and that religious denomination and education philosophy were more important predictors of choice (Borghans et al., 2015).

Gender-related cultural considerations can play a role when parents make decisions with significant household expenditure implications. An analysis of Learning and Achievement in Punjab Schools project data found that being a girl decreased the probability of attending a private school by six percentage points (De Talancé, 2020). In Haryana, India, it was found that parents sent sons to English-medium private and daughters to Hindi-medium public schools (Narwana, 2019). In Indonesia, girls are more likely to be enrolled in non-state madrasas, which may suggest that religion was a reason behind their parents' choice (Asadullah, 2018). In Sierra Leone, parents who highlighted a safety preference for girls were more likely to choose government over private schools (Dixon and Humble, 2017). In Egypt, young women said their parents, concerned about safety, had chosen closer public schools instead of far-away private schools for which they would have needed public transport (Krafft et al., 2019).



#### TEACHER CHARACTERISTICS DIFFER BY SCHOOL TYPE AND CONTEXT

Public and private schools often differ in teacher characteristics, such as age, sex, training and working conditions. In 2018, public school teachers in education systems that had participated in the Teaching and Learning International Survey (TALIS) had two more years of experience than their counterparts in private dependent schools and three more years than private independent school teachers. In Georgia, public school teachers had seven more years of experience than private school teachers. In Brazil, Japan and the Republic of Korea, there were more male teachers, but in France, Kazakhstan and Turkey more female teachers, in private schools (Cherng and Barch, 2021). In Burkina Faso, 48% of public and 40% of private school teachers were female, but the gap was 30 percentage points or more in some regions, such as Plateau Central and Nord (Lange et al., 2021).

Teacher education and professional development, which have been affected by outsourcing of such programmes to non-state providers (**Chapter 7**), may also affect education quality in state and non-state schools. In the 2018 TALIS, lower secondary school teachers reported that the quality of teacher education and professional development received was above average for those working in private independent schools, average for those in public schools and below average for those in private dependent schools (Cherng and Barch, 2021). Differences in contracts and salaries also create differences in teacher working conditions between state and non-state schools (**Box 2.3**).

Some research indicates that high levels of teacher absenteeism in public schools lead to an advantage for private schools. For instance, in the United Republic of Tanzania, 20% of public school teachers but only 7% of private school teachers were absent during unannounced school visits (Sabarwal et al., 2020). However, absenteeism in public schools may reflect systemic challenges rather than lower teacher effort (UNESCO, 2017). In addition, private schools are located primarily in urban areas, where attendance is easier both to ensure and to monitor (Lange et al., 2021).

Several studies reflect on the benefits of private schools' greater autonomy and freedom from major bureaucratic challenges. The administrative workload is higher in public schools, which typically experience more bureaucratic control (Singh, 2021). Evidence from the 2018 TALIS shows that in 32% of public but only 20% of private schools, teachers said they lacked time to provide instructional leadership (**Figure 2.8**).

Further evidence from the 2018 TALIS showed that, at the lower secondary level, teachers and head teachers in independent private schools reported the best school climate, followed by public schools, then government-dependent private schools – further proof that even within the private sector, sizeable structural differences exist. In the Russian Federation, 34% of teachers in independent private schools strongly agreed that their school had a collaborative culture with mutual support, compared with 18.5% in public and 12.5% in private dependent schools (Cherng and Barch, 2021).

#### PUBLIC AND PRIVATE SCHOOLS TEND TO DIFFER IN ACCESS TO RESOURCES

With respect to school resources other than the teaching force, public schools are more likely to report shortages in support personnel, instructional materials, technology and overall infrastructure. Among head teachers in lower secondary schools that took part in the 2018 TALIS, those in public schools were twice as likely to report lack of technology for instruction (30% vs 14% in private schools) and support personnel (35% vs 17%). One in five head teachers in Italian public schools reported a shortage or inadequacy of instructional materials, while none did so in private schools (Cherng and Barch, 2021).

Access to information and communication technology is another difference. In India, private companies offer education technology products to private schools, which already have much higher access to computer equipment (FICCI and Nielsen India, 2016). In Latin America, the average number of computers per student in private schools is double that in public schools. The ratio remained constant between 2009 and 2018. Only Chile achieved equal ratios in public and private schools (Buchbinder, 2020) (**Figure 2.9**).

Class size can make all the difference in how attentive and effective teachers can be. Across 32 education systems in 2018, the average class size was slightly lower in private secondary schools than in public schools, but some countries had larger gaps. In the Russian Federation, the average class size was 12 students in private and 19 in public schools (OECD, 2020a).

Differences in water and sanitation facilities vary. Burkina Faso and the Gambia had better water supply in public schools, while Kenya, Mali and Senegal had much better coverage in private schools. In Mali, 87% of private schools, 68% of public schools, 49% of madrasas and 37% of community schools had functional toilets. In Bhutan, in 2016, monastic institutions had higher coverage than non-monastic schools for basic water, but lower for basic sanitation (UNICEF and WHO, 2018).

Lack of disaggregated data by school type limits the ability to provide a fuller picture of differences in a range of other inputs, especially in poorer countries with numerous low-fee private schools that have reportedly poor facilities and resources (Härmä, 2021). In analysis carried out for this report, parents in Kinshasa, Democratic Republic of the Congo, and Lagos, Nigeria, criticized low-fee private schools for substandard and limited facilities (Cambridge Education, 2021).

#### Contractual differences affect teacher working conditions in private and public schools

Differing school contract arrangements can have varied effects. Some see public-sector tenure and contract security as key reasons for the limited accountability regarding teacher performance in public schools. Others believe precarious contract teaching negatively affects both public and private schools.

There are contractual differences between public and private school teachers in many countries. Overall, 81% of public lower secondary school teachers surveyed in richer education systems for the 2018 TALIS were employed full time, compared with 54% of their independent private school peers. In Turkey, 96% of public school teachers, but only 27% of those in private schools, are permanently employed (Cherng and Barch, 2021) (**Figure 2.7**). Teachers in low-fee private schools in India, Kenya, Nigeria and Pakistan are paid between one eighth and one half of what public school teachers receive (GCE, 2016). In the Philippines, the APEC Schools chain pays teachers half as much as public schools (Riep, 2015; Singh, 2021). In Uganda, the monthly salary at Bridge International Academies was less than half that at public schools (Riep and Machacek, 2016).

#### FIGURE 2.7:



**Private school teachers are less likely to be fully employed** Percentage of fully employed lower secondary school teachers, by type of school, 2018

GEM StatLink: https://bit.ly/GEM2021\_fig2\_7

Source: Cherng and Barch (2021) based on the 2018 TALIS.

Inequality in contract arrangements also exists among private school teachers. A review of 15 sub-Saharan African countries showed some 20% of teachers worked in private schools, from 5% in Namibia to 38% in the Gambia. For the fixed-term contract teachers among them, median monthly earnings amounted to about 71% of those of permanent teachers (Evans et al., 2020). Even among public school teachers, many are hired under conditions similar to those of private school teachers, with lower salaries and fewer benefits (Lauwerier and Akkari, 2019). This practice creates a parallel cadre of undertrained, underpaid, younger, inexperienced teachers (Chudgar et al., 2014).

Non-state school teachers on temporary contracts are at risk of lower motivation. They work under pressure to ensure that their students perform better than those in public schools and feel at greater risk of losing their jobs (Lange et al., 2021). In Bogotá, Colombia, teachers in 25 Concession Schools, established by private operators with public funds, reportedly have poorer working conditions (Edwards Jr and Termes, 2019). In Guinea, while public school teachers complain about poor infrastructure and overcrowded classrooms, private school teachers face a lack of job security and poor salaries (Botta Somparé and Wotem Somparé, 2018). Reliance on greater effort from teachers with lower qualifications through unequal contractual arrangements is unsustainable, especially if non-state school teachers begin to demand rights comparable to those of their state school counterparts (Singh, 2021).

#### **BOX 2.3 CONTINUED:**

Richer countries are not immune to such differences between teacher working conditions in the state and non-state sectors. In the United Kingdom, analysis of 18,000 English schools that compared public schools managed by local education authorities with academies, which are also publicly funded but self-governed, found a greater percentage of unqualified teachers in the latter, resulting in wider inequality in student access to qualified teachers (Martindale, 2019). In the Republic of Korea, outsourcing of education services has been found to undermine teacher professionalism (Bates et al., 2019).

#### THE IMPACT OF NON-STATE SCHOOLS ON INDIVIDUAL AND SYSTEM OUTCOMES IS CONTESTED

Information on differences in student and school characteristics between state and non-state schools offers much-needed context for tackling questions on non-state actors' contribution in education. In particular, do non-state schools improve individual student results? Is the non-state sector helping pull education systems up or does it lead to the creation of parallel systems with potentially negative effects on state schools?

#### THE GAP IN LEARNING OUTCOMES IS REDUCED IF STUDENT BACKGROUND IS TAKEN INTO ACCOUNT

The considerable differences between public and private schools in inputs such as student intake and material or human resources need to be considered when comparing differences in outputs such as completion rates and outcomes such as learning achievement. The variety of non-state schools and of the conditions that led to their emergence implies such comparisons are not straightforward. However, on the whole, the research provides limited evidence that private schools confer an advantage.



66 -

In richer countries, differences in learning outcomes between public and private schools are typically minimal after controlling for school and student background

In upper-middle and high-income countries, differences in learning outcomes between public and private schools are typically minimal after controlling for school and student socioeconomic background. An analysis of 2012 PISA mathematics scores in 40 countries found that accounting for peer effects and selection effectively eliminated the private school advantage in all but a handful of countries (Sakellariou, 2017). An analysis of 2018 PISA reading scores that accounted for school and student socioeconomic profiles found that public school students scored higher (OECD, 2020b). In the United Kingdom, chains of publicly funded but self-governed secondary schools, known as academies, were not found to do better than publicly funded schools managed by local education authorities (Andrews, 2016).

99

Even where private schools maintain an advantage over public schools, it is important to look at which students benefit. Analysis of how competitive pressure, administrative autonomy and staffing practice differences affect the public-private gap in Australia, Portugal and Spain using TALIS data found that such school factors explained the gap only among better performing students rather than those whose results most needed improvement (Delprato and Chudgar, 2018).

In the United States, there has been interest in two flagship policy instruments: charter schools, which are publicly funded, self-governed and accountable for results; and vouchers, which are funds offered to parents for their child to attend their preferred school. Research shows variable results. In some states, such as New York,

#### **FIGURE 2.8:**

#### Public school head teachers are less likely to provide instructional leadership

Percentage of head teachers who agreed or strongly agreed that they lacked time for instructional leadership, 2018



Source: Cherng and Barch (2021) based on the 2018 TALIS.

#### FIGURE 2.9.

**Students in private schools in Latin America have higher access to computers** Computer per student ratio in secondary schools, by sector, Latin America, 2009–18



Notes: The Latin American average is a simple average of eight participating countries. The OECD average is a simple average of the OECD countries that participated in all rounds of PISA, excluding Mexico.

GEM StatLink: https://bit.ly/GEM2021\_fig2\_9 Source: Buchbinder (2020). charter schools have improved student achievement, but in others, such as Arizona, there has been no positive impact. Virtual charter schools, which use technology to deliver education at students' homes, are found to have a substantial negative effect on achievement. By contrast, so-called 'no excuses' charter schools, which emphasize standardized tests, behavioural standards and longer instruction time, have produced outcome improvements, especially for disadvantaged groups. Non-profit charter school networks are more effective than for-profit charter schools at delivering learning gains (Ferrare, 2020). For instance, an analysis of 43 lower secondary schools in the Knowledge is Power Program network found large gains in mathematics and reading (Clark Tuttle et al., 2013).

A survey of studies on small- and large-scale voucher programmes showed mostly no effects on test scores, with some programmes even showing large negative effects, though there was some positive impact on graduation rates for African American students (Epple et al., 2017). A longitudinal analysis found that the better academic, social, psychological and attainment outcomes at age 15 enjoyed by children in private schools disappeared after controlling for student sociodemographic characteristics (Pianta and Ansari, 2018).

In low- and middle-income countries, a 2014 review found moderate evidence that children in non-elite private schools learned more than their peers in government schools (Day Ashley et al., 2014) but a more recent review of 13 studies, mostly from South Asia and sub-Saharan Africa, supported the conclusion that effects on learning were mixed (**Table 2.2**). Furthermore, using data for 31 countries, the analysis showed that the estimated premium from attending private school dropped by one half to two thirds after adjusting for household wealth, which determines who gets selected into the schools (Patel and Sandefur, 2020). The overwhelming conclusion was that learning levels were poor overall, raising concerns about the education system that were unrelated to school type.

Some results were slightly more positive for private schools. A meta-analysis of 17 studies of different types of non-state provision (government aided and unaided, registered and unregistered, faith-based and NGO) on foundational literacy and numeracy outcomes in South Asia showed a moderately positive effect on learning, larger for language than mathematics, after controlling for household and school factors (Borkowski et al., 2021). In sub-Saharan Africa, results depended on context and focus. In Kenya, citizen-led learning assessment data showed a significant advantage for private school students in grades 2 to 4, even after controlling for socioeconomic background. A more sophisticated technique found an advantage in mathematics but not in English (Wamalwa and Burns, 2018). But further analysis showed that the poorest were least likely to be learning. In mathematics, the private school advantage for the wealthiest 20% of students was four times that of the poorest 20% (Baum and Riley, 2019).

In Nigeria, schools in the Bridge Academy chain performed better than similarly priced local private schools in reading but not in mathematics (Lipcan et al., 2018). Students at Rising Academies, a chain in Sierra Leone, performed significantly better than those in other private and public schools in both reading and mathematics (Johnson and Hsieh, 2019). In South Africa, low-fee independent schools appeared to perform better than public schools, but comparing schools in similar fee brackets showed a more mixed impact (van der Berg et al., 2017). An analysis of various providers in Uganda found that students from the Promoting Equality in African Schools network performed slightly better than other local public-private partnership schools in terms of value added to test scores (Crawfurd, 2017).

An evaluation of Fe y Alegría faith-based schools in Colombia, where the network originated, found they had a significant positive effect on mathematics test scores, at a cost similar to public schools. In Peru, analysis of a lottery selection process into grade 1 for Fe y Alegría schools found a large positive effect in reading and mathematics (Lavado et al., 2019). In sub-Saharan Africa, despite higher parental satisfaction with Catholic schools, test scores in faith-based schools, while higher than in public schools, tended to be lower than in private independent secular schools – but scores overall were low, with school type not making a big difference (Wodon, 2020c).

## Learning outcome comparisons in civics and technology are less common

Beyond core subjects such as reading and mathematics, there is little evidence on the impact of private school attendance on other content domains. Analysis for this report of International Civic and Citizenship Education Study (ICCS) data, mostly from European countries, shows that civic knowledge and perceptions vary quite significantly between public and private schools, albeit in different ways by country. The ICCS

#### TABLE 2.2

Summary of evidence on private schooling impact on learning outcomes in selected low- and middle-income countries

Country	Outcome assessed	Effect on learning	
Ethiopia	Mathematics and Peabody Picture Vocabulary Test; data collected between 2002 and 2013 for children who were 12 years old in 2013	<i>Mixed.</i> Positive in mathematics but not on the vocabulary test (Eigbiremolen, 2020).	
Andhra Pradesh and Telangana, India	Mathematics, functional English and transferable skills; grade 9	Positive (Rolleston and Moore, 2018). Mixed. Positive effects on solving arithmetic problems, but not on more challenging reasoning-based questions (Kumar, 2018).	
Delhi, India	Mathematics, Hindi and English; children aged 5 and 6 when voucher programme started	Mixed. After four years, positive for English and negative for Hindi (Dixon et al., 2019). After six years, no positive effects for English, negative effects on Hindi strengthened (Crawfurd et al., 2021).	
Rajasthan and Orissa, India	Mathematics; grade 9	Mixed. Positive effect in rural and urban Rajasthan; no effects in Orissa (Azam et al., 2016).	
Nairobi, Kenya	English and Kiswahili literacy and mathematics; primary school	Null. Low-cost private schools produced no significant improvement over public schools except schools in a specific instruction improvement programme (Zuilkowski et al., 2020).	
Liberia (Partnership Schools)	English and mathematics; kindergarten to grade 5	Mixed. Some learning gains after one and three years but large variation among providers (Romero et al., 2020).	
Punjab, Pakistan (Education Fund)	Five subjects; grade 5	Negative. Slight decline in overall test scores (Crawfurd, 2018).	
Peru	Mathematics and Peabody Picture Vocabulary Test; data collected between 2002 and 2013 for children who were 12 years old in 2013	Null. No private school effects on learning after controlling for prior achievement (Eigbiremolen et al., 2020).	
U. R. Tanzania	Kiswahili, English and mathematics; grade 7 examinations and grade 9 assessments	Positive. Secondary school private students performed better in all three subjects (Brandt, 2018).	

Source: GEM Report team based on Hares and Crawfurd (2021).

focuses on four content domains (society and systems, principles, participation and identities), which are combined to develop an aggregate civic knowledge score. Private schools had significantly higher scores in countries where they accounted for at least 10% of the student population. The exceptions were Indonesia, Luxembourg and the Russian Federation, where public school students had an unconditional advantage in civic knowledge. However, after accounting for student background and pupil/teacher ratio, the overall advantage was cut by 45%, and in education systems including those of the Flemish Community of Belgium, the Czech Republic and Italy, the advantage disappeared (IEA, 2021) (**Figure 2.10**).

With respect to student values, beliefs, attitudes, behaviours and behavioural intentions related to aspects of civics and citizenship, there is more variation in observed gaps between public and private schools. Private school students in 10 of 17 countries have higher expected participation in elections; only in Indonesia do public school students have higher expected participation. By contrast, public school students in 8 of 17 countries have a higher expectation of active participation in political activities (joining a political party, a trade union or an organization for a political or social cause, standing as a candidate in elections or helping during an election campaign); only in Ireland do private school students have a higher expectation. Two interesting cases are Mexico and Peru, where public school students have a lower expectation of voting but a higher expectation of actively participating in political activities (IEA, 2021).

## Few long-term outcomes of private school attendance have been found

The few studies that have tried to assess the long-term impact of school choice on educational attainment and other future outcomes are mostly from high-income

#### **FIGURE 2.10:**

#### Some of the advantage private schools enjoy in civic knowledge is linked to socioeconomic differences

Private-public civic knowledge score gap after accounting for student background and pupil/teacher ratio, 2009 or 2016



GEM StatLink: https://bit.ly/GEM2021\_fig2\_10 Source: IEA (2021).

countries (Egalite and Wolf, 2016). Students who participated in voucher programmes in the US states of Florida and Milwaukee were more likely to enrol in and graduate from university than those from public schools. However, the impact on tertiary education completion was negative for low-income students (Chingos et al., 2019). In Chicago, lottery winners from disadvantaged backgrounds who attended 'no excuses' charter schools with strict disciplinary practices were 10 percentage points more likely than lottery losers to enrol in tertiary education and to remain there at least two years (Davis and Heller, 2019). Analysis of data from a large US public university found that students from Catholic secondary schools, including those from disadvantaged backgrounds, were more likely to graduate and to earn a degree in science, technology, engineering or mathematics than those who had attended other private religious, other private non-sectarian and public secondary schools (Fleming et al., 2018).

Other studies have not been as positive. An analysis using longitudinal surveys from Australia showed

no statistically significant association between type of school attended and employment status, occupation or earnings at age 24, after controlling for educational attainment (Chesters, 2018). Longitudinal analysis from England and Wales (United Kingdom) found that most of the apparent long-term effects of having attended private, selective and faith-based schools in the 1980s disappeared after controlling for student characteristics (Sullivan et al., 2018).

Two recent studies based on longitudinal data from England also show no private school advantage in social outcomes. At age 21, private school attendance had made people less risk-averse and more prone to begin drinking alcohol at younger ages. It had not contributed to differences in social-emotional development (von Stumm and Plomin, 2021). At age 25, having attended a private school yielded a 17% wage premium and a 12% point lower chance of downward social mobility, but was not associated with charitable giving or participation in voluntary groups (Green et al., 2020).

#### FIGURE 2.11:

Differences in the socioeconomic composition of public and private schools remain constant

Gap in PISA economic, social and cultural status index between private and public schools, 2009 and 2018



Note: Student socioeconomic status in PISA is estimated with an index of economic, social and cultural status, a composite score derived from information on parental education, parental occupation and home possessions, including number of books and other education resources available at home. The values of the index are standardized to have a mean of zero and a standard deviation of one for the population of students in OECD countries.

GEM StatLink: https://bit.ly/GEM2021\_fig2\_11

Source: GEM Report team estimates based on OECD (2020b).

#### NON-STATE PROVIDERS CAN HAVE SYSTEM-WIDE EFFECTS

Non-state actors can potentially have effects not only on individual education outcomes but across the education system. In addition to the example discussed previously, concerning the impact of contract teachers on the teaching profession, two effects that have been studied closely are segregation and competition.

## Sorting and segregation are an unintended consequence of school choice

In 53 of 66 education systems that took part in the 2018 PISA, students who attended private schools had a significantly higher value on a socioeconomic status index than those who attended public schools. The gap was particularly high in Latin America, notably in Brazil, Colombia, Costa Rica and Panama. It is generally persistent, although the number of education systems where it decreased between 2009 and 2018 was higher than those where it increased. Countries where it noticeably shrank in the 2010s included Kazakhstan, Poland, the United Kingdom and the United States, while countries where it noticeably increased included Germany, Jordan, Luxembourg and Switzerland (**Figure 2.11**).

Social segregation can also exist in education systems that are primarily public (OECD, 2019a) but a number of studies have argued that school choice policies and the growth of the non-state sector are stratifying education systems (Hernández, 2019; Srivastava, 2020).

Segregation is viewed as a key consequence of the universal voucher policy in Chile (Zancajo, 2019). School choice exacerbated student segregation beyond levels linked to residential segregation (Santos and Elacqua, 2016). In Sweden, 29 of 30 municipalities contained

66

School choice policies and the growth of the non-state sector are stratifying education systems strongly segregated lower secondary schools; in 16, segregation appeared to have been largely driven by school choice. Children of foreign-born and/or less educated parents were almost exclusively in municipal schools, while Swedish-born and well-educated parents had opted out of municipal schools, choosing independent schools (Kornhall and Bender, 2019). Charter schools in the United States are believed to have exacerbated existing segregation (Wells et al., 2019). District-level data show that segregation of poorest students increased by 15% in large school districts between 1998 and 2015, with the presence of charter schools being a key factor (Marcotte and Dalane, 2019).

#### Competing with non-state schools may not help improve state schools' outcomes

One appeal of school choice is the notion that competition will lead public schools to improve. Schools may change operations to improve quality, change extracurricular activities, emphasize recruitment and marketing, and even try to attract the best students. However, education has certain characteristics that make it less conducive to replicating the kind of market competition that may yield benefits in economic activities. Chile's voucher programme had a negative impact on public schools. In municipalities with a higher share of private school enrolment, public schools had lower test scores, the gap in test scores between elite private and public schools was wider and the socioeconomic gap between public and private school parents was greater (Hsieh and Urquiola, 2006). A comparison of municipality test score performance between 2002 and 2013 found that scores increased with resources, such as parental education, but not with school competition (Hofflinger and von Hippel, 2020). The introduction of targeted vouchers in 2008 prompted schools to compete for students, since targeting made more expensive schools more attractive to poorer students (Urquiola, 2016).

In Nepal, public school outcomes did not seem to be associated with the extent of private competition. But the gap between public and private school outcomes was higher in localities with higher growth in private education, suggesting stratification (Joshi, 2020).

By contrast, in Sweden, a longitudinal analysis from 1988 to 2009 found that grades by subject and average attainment at age 24 increased with the share of enrolment in independent schools, with competition having increased productivity instead of resulting in sorting (Böhlmark and Lindahl, 2015). In the US state of Ohio, there was modest improvement in mathematics and reading scores of students who had the option but had not taken up a voucher, a result which could



#### SCHOOLS DO NOT NECESSARILY IMPROVE WHEN THEY HAVE TO COMPETE

be attributed to the effect of competition (Figlio and Karbownik, 2016). Analysis of the impact of charter schools in New York City, United States, found that positive effects on mathematics and English among traditional public school students were largest when charter schools were in the same location (Cordes, 2018).

The mere presence of private or other schools in near proximity may not be a sufficient incentive for public school authorities to take any action if they do not have financial resources or autonomy to respond. An analysis of the Foundation Assisted Schools programme in Pakistan, a public-private partnership, showed that its schools had a negative impact on enrolment at nearby public schools (Ansari, 2021). In New Orleans, United States, most school principals felt the large number of charter schools presented strong competition for students. Over half reported competing with private schools (Jabbar and Li, 2016).

In some instances, competition may focus on teachers. In the United States, teacher quality declined in difficult-to-staff schools after charter schools were introduced (Austin, 2020). In the state of North Carolina, after a charter school opened nearby, hard-to-staff public schools hired fewer new teachers and experienced small declines in teacher quality. At the same time, schools increased teacher compensation to retain teachers of good quality (Jackson, 2012). In New Orleans, major reforms removed teacher bargaining power and protections to enable more flexibility and variation in hiring strategies. The reforms seem to have led to changes in hiring patterns, with charter schools seeking candidates outside the city and public schools looking locally (Jabbar, 2018).

Competition between public and private schools can take several forms, including choice of instruction language. Some public schools switch to international languages, a marketing tool commonly used by private schools, but one that represents a negative consequence of competition. Such responses have been observed in Lebanon (Bahous et al., 2011), Morocco (Chakrani, 2017), Nepal (Joshi, 2016) and the Philippines (Termes et al., 2020).

Some limited evidence exists on how expanding public school choice affects private schools. After open enrolment was introduced in the Canadian province of British Columbia, students had greater public school choice and could enrol outside their catchment area. This increase in public school competition led to reduced enrolment in private secular and Catholic schools, but did not change demand for other faith and other Christian private schools (Cohn, 2020).

#### NON-STATE ACTORS OFFER EDUCATION GOODS AND SERVICES OTHER THAN SCHOOLING

Non-state actors do not only provide education services. They also provide services that support learning, such as supplementary tuition, and goods used in the classroom, such as teaching and learning materials and technology inputs. This engagement can be significant in some countries, with potential impact on education system performance and development.

## SUPPLEMENTARY TUITION IS EDUCATION PRIVATIZATION BY ANOTHER NAME

Demand for supplementary tuition has been associated with students' need to prepare for high-stakes examinations. Households increasingly resort to such support over longer periods to give children a comparative advantage. This demand is linked to dissatisfaction with the perceived quality of education in public and even private schools.<sup>1</sup>

Two broad models of provision are teachers and companies. In low- and middle-income countries, including Cambodia, Egypt, Ghana, Myanmar, Turkey and Uzbekistan, teachers are directly involved in tutoring. Such services often emerged in rapidly expanding but underfunded education systems, where low salaries led teachers to seek additional income, among other reasons, as in Yangon, Myanmar, where 48% of teachers from eight schools reported that they offered tutoring services (Bray et al., 2020). As the practice of tutoring becomes embedded into the schooling culture, it is often hard to dislodge.

In high-income countries, including Australia, France, Japan, the Republic of Korea, the United Kingdom and the United States, enforcement of regulations against teachers offering such services has led to professionalization, with tutoring companies dominating (Zhang, 2020). In some countries, including Australia, parents who might previously have thought it irresponsible to undermine public education are influenced by marketing that has in effect legitimized tutoring (Doherty and Dooley, 2018). From the teachers' perspective, income is not the only reason to engage in tutoring. About one third of Czech teachers had an additional job to supplement their income, for example, but younger teachers who moonlighted as private tutors also did so for experience in teaching core subject areas (Šťastný et al., 2021).

In the Russian Federation, 42% of recent school graduates used private tutoring to prepare for university entrance examinations

While these examples suggest that private tutoring is nearly universal, there is no globally comparable estimate of attendance. This is at least partly because such services are not standardized and vary by provider type (e.g. individual or institutional tytors), form (e.g. offline or online), focus (by subject) and duration (e.g. timing or number of hours). Individual country estimates thus remain the main source of information on the scale of the phenomenon. In the Russian Federation, 42% of recent school graduates reported using private tutoring services to prepare for university entrance examinations (VCIOM, 2019). Conversely, it is estimated that 28% of teachers provide private lessons (ISSEK, 2018). Studies typically show that tutoring increases by level of education. In Egypt, 36% of primary, 53% of lower secondary and 84% of general upper secondary students received supplementary tutoring in 2013/14 (Sieverding et al., 2019).

The phenomenon is also spreading in regions where it had been uncommon, such as sub-Saharan Africa (**Box 2.4**) and northern Europe. Results from a German survey of 4,000 parents in 2014/15 showed some 5% of primary and 18% of secondary school students received tutoring (Klemm and Hollenbach-Biele, 2016). In England and Wales (United Kingdom), 27% of 11- to 16-year-old students, and as many as 41% in London, had received private tutoring at least once in their academic careers (Sutton Trust, 2019).

Tutoring prevalence has also increased over time in many countries. In Bangladesh, the share of households that paid for private tutoring increased between 2000 and 2010 from 28% to 54% in rural areas and from 48% to 67% in urban areas (Pallegedara and Mottaleb, 2018). In Cambodia, 18% of primary, 54% of lower secondary and 72% of upper secondary school students received supplementary tutoring in 2015, an average increase of 7 percentage points since 2004 (Marshall and Fukao, 2019). In Sri Lanka, the percentage of households spending on tutoring increased between 1995/96 and 2016 from 41% to 65% of urban households and from 19% to 62% of rural households (Abayasekara, 2018). In Ukraine, the share of students who reported private supplementary tutoring from secondary school teachers increased from 34% in 2006 to 53% in 2011 and almost 80% in 2016 (OECD, 2017).

Data from the world's two most populous countries show variation in tutoring prevalence by education level, location, school type and over time. In China, one third of primary and one half of upper secondary school students receive supplementary tuition; among the latter, students in urban areas are twice as likely to receive tuition as their peers in rural areas. In India, 30% of upper secondary school students receive tuition, with a 10 percentage point gap favouring urban over rural students (**Figure 2.12a**). In rural India, the prevalence rate increased only slowly between 2010 and 2018, but students from both public and private schools receive supplementary tuition (ASER, 2019) (**Figure 2.12b**).

Private tutoring growth is also linked to a rise in individual-driven learning linked to specific skills and an increased desire to learn at one's own pace. In the Republic of Korea, student dissatisfaction with a standardized mathematics curriculum and its pace drove many students to seek customized teaching (Kim and Jung, 2019).

#### BOX 2.4:

#### The private tutoring landscape is changing in sub-Saharan Africa

Supplementary tuition has been common mainly in affluent parts of sub-Saharan Africa, such as Mauritius, where 81% of grade 6 students received private tutoring in 2013. While weak household purchasing power was a factor limiting its expansion, data from the Southern and Eastern Africa Consortium for Monitoring Educational Quality (SACMEQ) survey show that such services increased rapidly, from low levels, between 2007 and 2013 even in some of the poorest countries, such as Malawi (from 5% to 14%) and Mozambique (from 10% to 21%). These relatively low averages mask high prevalence in urban areas. In urban Ethiopia, 67% of upper primary students had received private tutoring.

Tutoring may be fuelled by inadequate teacher salaries and high urban unemployment, a context in which private tutoring is temporary employment. In Benin, 60% of tutors were university students, secondary school students, informal sector workers or unemployed. In Ouagadougou, Burkina Faso, the corresponding rate was 49%.

The region's lower internet penetration and limited use of mobile technology for learning means face-to-face delivery far exceeds technology-based provision. However, a few companies maintained provision during COVID-19 on the internet while physical tutoring centres were shuttered.

Source: Bray (2021).

#### **FIGURE 2.12.**

**Supplementary tuition is common in China and India** a. Supplementary tuition participation rates by education level and location, China and India, 2017



GEM StatLink: https://bit.ly/GEM2021\_fig2\_12

Sources: 2017 China Education Finance Household Survey subject based off-campus training participation rates and 2017/18 India National Sample Survey coaching rates.

### The impact of tutoring on individual student performance is unclear

Researchers have sought to understand the impact of private tutoring on individual student outcomes, such as test scores and examination pass rates. Some have detected positive effects for those furthest behind. In India and Pakistan, private tuition improved learning outcomes more among poorer than richer children, even though the worst-performing richest children still outperformed the poorest children who receive tuition (Alcott and Rose, 2015). Using China Education Panel Survey data, a rigorous quasi-experimental study showed private tutoring positively associated with higher English scores for rural lower secondary students. Across most groups, private tutoring was also associated with lower frequency of self-reported negative emotions (Sun et al., 2020).

Other studies have been more equivocal. Analysis of two longitudinal data sets in Germany found that longer tutoring duration or more qualified tutors had no systematic positive effect on secondary school student performance. Rather, tutoring was more effective when students had greater prior knowledge and were coached by trained teachers with more content knowledge and stronger experience in teaching (Ömeroğulları et al., 2020). A similar conclusion was reached in the United States, where tutoring's effects were stronger for students of higher socioeconomic status, who were more likely to use it in the first place (Choi, 2018). In the Republic of Korea, private supplementary tutoring was positively associated with secondary school students' cognitive achievement in Korean, English and mathematics, but negatively associated with their affective achievement, i.e. their interest in and attitude towards the subject (Kim and Hong, 2018).

In India, analysis of tutoring in Delhi slums observed that prices determined attendance or dropout but found no evidence that tutoring improved test scores (Berry and Mukherjee, 2019). An assessment in Sri Lanka showed that, on average, five months of tutoring had no impact on year 5 student examination scores (Cole, 2017).

Regardless of evidence, an important parameter is household and student perception of tutoring's value for test preparation. In China, students reported a higher level of self-confidence and better relationships with their parents (Zheng et al., 2020). In Germany, support was positively associated with grade 10 students' satisfaction with their schools, even though it did not improve their test scores (Guill et al., 2020). In England (United Kingdom),

b. Percentage of students who take private tuition by education level and school type, rural India, 2010–18

60

50

40

% 30

20

10

n

2010 2012 2014 2016 2018

Primaru

Source: ASER (2019)

Public

Drivato

2010 2012 2014 2016 2018

Lower secondaru

grade 6 students viewed tutoring initially as a means of passing competitive grammar school examinations and meeting parental expectations but, over time, they noted a positive impact on self-esteem and interest in learning (Hajar, 2018).

#### Tutoring can undermine education system performance

There are concerns that instruction-related support services may negatively affect student and teacher behaviour. Shadow education, a term used to describe private supplementary tuition, may overshadow core education provision (Bray, 2017). Students may prioritize tutoring time and reduce their school attendance before high-stakes examinations. In Egypt, tutoring has become a social norm (Sieverding et al., 2019). In India, schools feel intense pressure from private tutoring and coaching institutions. In urban Maharashtra state, children were reportedly more respectful at coaching centres and to tutors than to teachers, and skipped lessons to attend coaching centre classes (Bhorkar and Bray, 2018).

In some countries, teachers favour tutoring over classroom lessons. In Myanmar, teachers often assumed that students who needed extra support would receive it. They put less effort into teaching regular lessons and more into fee-paying work, which led to accusations of favouritism and corruption. Teachers who were tutors were overworked and stressed, which affected their formal teaching time (Bray et al., 2020). In Nepal, teachers who offered tutoring covered less material in school to increase demand for tutoring. Poorer students who did not receive tutoring did worse on examinations (Jayachandran, 2014).

Corruption surfaces in other forms. In Shanghai, China, schools and tutoring providers collaborated to circumvent rules that prohibited tracking of students into elite and non-elite categories. Some elite primary schools contracted with tutoring companies to provide additional lessons for the best-performing students, then secretly sent higher performers to unofficial high-track classes. They colluded with tutoring companies to select a small group of students with the best potential to enter top-ranking secondary schools and gave them much more intensive training (Zhang and Bray, 2018). In an extreme example, many upper secondary schools in Kota, India, only enrol students

#### 66

## In Egypt, tutoring has become a social norm

99

so they have the right to sit public examinations; the students attend coaching centres (Rao, 2017).

Private tutoring increases the gap between haves and have-nots. In Cambodia, analysis of 138 lower secondary schools found students primarily received tutoring from teachers within the school. Teachers with university education and higher subject matter knowledge were more likely to do fee-based tutoring, charging more per session, thus increasing inequality (Marshall and Fukao, 2019). In Shanghai, China, one of the two most popular tutoring companies charged US\$1,200 a month for grade 9 students, effectively excluding most families. An entrance examination excluded low achievers (Zhang and Bray, 2018). In Nepal, there are clear signs of major differences in education spending by income guintile, in private vs public schooling access and in access to private tuition (Mottaleb and Pallegedara, 2019). Recognizing the potential equity challenges and the need to support low-income communities, Japan has publicly funded after-school programmes. Under the Chiiki Mirai Juku (Community Tutoring School for the Future) programme, municipalities contract non-profit organizations to deliver free tutoring for children in need (Feldhoff, 2017).

#### TEXTBOOK PROVISION HAS BECOME MORE COMMERCIALIZED

Teaching and learning materials are key for education quality and national identity development. Government's role in textbook policy, procurement and distribution ranges from acting as a primarily state-owned and -controlled enterprise to being part of a mixed system of public and private publishing. In a few cases, the private sector manages most aspects of the book supply chain.

Analysis of approval processes for reading instruction materials in 50 education systems that took part in the Progress in International Reading Literacy Study identified 12 with no process for approving instructional materials. They tended to be high-income countries emphasizing school autonomy and leaving textbook production largely to commercial providers, with government involved in providing guidelines and approving proposals. Finland has no approval process and commercially published textbooks are the primary teaching and reading materials. In Spain, commercial providers produce textbooks under public supervision. The Centre for Educational Research and Documentation assists in development and dissemination of curriculum materials and guides (UNESCO, 2013). Public authorities provide a recommended list but do not prescribe textbooks for use (Mullis et al., 2017). In Latin America and the Caribbean. education ministries develop the curriculum and mostly hire private publishers to develop textbooks (UNESCO, 2013).

#### Out of 50 education systems, only 12 had no process for approving reading instructional materials

Other countries, including in the Arab States, the Caucasus and Central Asia, have a centralized process. The Bahrain Ministry of Education is involved in production, compilation and printing, and provides all public school students with free Arabic-language textbooks. Azerbaijan's government is legally obliged to prepare, publish and disseminate free textbooks and other materials. The Kazakhstan Ministry of Education approves a list of textbooks from which local education organizations choose, then local executive bodies make the purchases (Mullis et al., 2017).

99

The perceived sensitivity and importance governments attribute to certain topics affect the extent of the role they take in textbook provision. Governments deem mother tongue, mathematics, social studies and history textbooks to be high priority and may be especially involved in their development. In the Republic of Korea, the government provides and more closely controls primary school than secondary school textbooks. Singapore's Ministry of Education publishes textbooks for languages and ethics. In Viet Nam, the Ministry of Education and Training has published 32 grade 1 textbooks in the general education curriculum, of which 24 were approved by the Viet Nam Education Publishing House (Trinh, 2019). As part of an effort to decouple textbooks from examinations, a multiple textbook policy has been tried, to promote the role of commercial publishing (Smart and Jagannathan, 2018).

In South Asia, textbook development, publishing and printing are transitioning from a relatively tightly state-controlled system to one where corporate publishers are starting to play a major role. In Bangladesh and India, textbooks are free and cheaply manufactured for single use in public schools. But the presence of a large private school sector supports the existence of a commercial publishing sector (Smart and Jagannathan, 2018).

## Textbook procurement challenges national publishing industries

While the aims of moving from state control to diverse suppliers and open competition are to reduce cost and improve quality, the transition is not always smooth or linear. In Indonesia, after a multiple textbook policy was poorly enforced, it was decided to recentralize textbook publishing in 2013. However, collusive practices with business interests followed, leading to higher prices. The government has therefore considered producing its own textbooks to compete with those developed by private providers (Smart and Jagannathan, 2018).

Competition was introduced in the Russian Federation in the 1990s. Schools could choose from a diverse list of textbooks vetted by the central government (Kaplan, 2004). When the vetting process became stricter in 2013, Prosveshcheniye, the former state-owned publisher, which had been privatized in 2011, re-emerged as a dominant provider (Moskwa, 2019). The company has 40% of the education market and claims its products are in use in every school in the country (Prosveshcheniye, 2017).

In Rwanda, publishing houses began producing and distributing textbooks through a tendering process in 2009. In 2017, the Ministry of Education decided textbooks would be produced through the Rwanda Education Board rather than independent textbook publishers. The ministry cited delays in textbook delivery, lack of government copyright, quality issues and cost efficiency. Its decision jeopardized many independent publishers whose primary business was linked to textbook development (Nsabimana, 2018).

India had over 5,000 primary and secondary school textbook publishers in 2015 (FICCI and Nielsen India, 2016). The National Council of Educational Research and Training (NCERT) publishes textbooks used in public schools. The government recommends NCERT textbooks for public and private schools participating in the Central Board for Secondary School examinations and cautions against expensive privately published textbooks. Private schools and parents are sceptical, pointing at production delays, errors and outdated quality on the part of NCERT textbooks as reasons to use other publishers' books (Frontlist, 2019; Mufti, 2017; Vinayashree, 2017). At the same time, parents also express concern that the textbook industry works with private schools to require costly books that increase financial burden (Mishra, 2019).

In sub-Saharan Africa, as part of structural adjustment programmes, the World Bank promoted international tenders, which ended textbook publishing monopolies but initially favoured foreign publishers since international languages were used for instruction (Thierry, 2020). A World Bank survey of 19 anglophone and francophone countries found that most had engaged private publishers in textbook production (World Bank, 2008). The interplay between international publishers, donors and local interests has often complicated this transition. Ethiopia's state-controlled Educational Materials Production and Distribution Agency had provided textbooks since the 1970s, but the Ministry of Education established a procurement process in 2009 with support from development partners. A World Bank analysis found that textbook development, printing and distribution significantly improved as a result of competitive bidding. In 2013/14, the textbook-to-student ratio reached 1:1 for the first time in 7 primary and 13 secondary school subjects. The bidding process and criteria led to the selection of international publishers, since local competitors lacked capacity (Woldetsadik and Raysarkar, 2017).

The United Republic of Tanzania marketized textbook provision with donor influence in the early 1990s, aiming to build a local publishing industry, but donor practices helped protect multinational companies' dominant position. Eventually, after corruption scandals involving international publishers, such as Oxford University Press and Macmillan in the early 2010s, nine local publishers were selected to supply standard textbooks to schools. But blurred boundaries and tensions between national and international publishers, elite groups and government offices led to collusion in the textbook market. In 2014, the government discontinued the market-based system and moved back to full state control of textbook provision (Languille, 2016). Most textbook publishing in Kenya, Rwanda, Uganda and the United Republic of Tanzania is now in the hands of regional, African-owned publishers. The international publisher Longman, for instance, sold its Kenyan company and local copyrights to Longhorn, a Kenyan-owned company (Read, 2015).

In francophone Africa, the French publishing industry is dominant. Ties between national governments and international publishers make it difficult for local publishers to compete. Gabon's textbook industry is dominated by Edicef, a textbook publishing arm of French-owned Hachette Livre, one of the world's largest publishers. Hachette's local partner is Éditions Gabonaises, which has a monopoly and receives state subsidies (Gary, 2018). In Cameroon, the official list of recommended textbooks is dominated by French publishing giants such as Edicef, Belin and Nathan (Nkeck Bidias, 2019). Incentives have provided some room, albeit limited, for local publishers. In Senegal, policies allow calls for tenders to favour national over international publishers (de Lesseux, 2018). Mali's government engages in partnerships with private companies to build publishing capacity (Canada Government, 2018).

In Côte d'Ivoire, Hachette owns a 70% stake of the leading publishing house NEI-CEDA, formed by a merger of Nouvelles Editions Ivoiriennes and Centre d'Edition et de Diffusion Africaines in 2011 (Hachette Livre, 2019). Local publishers say the government's close ties with NEI-CEDA has allowed its continued textbook market domination since the 1980s (Niakaté, 2018). The government has allowed national bids to be slightly higher than international ones in recent tenders for projects financed by the World Bank (Rogez, 2020). However, an analysis of the list of required textbooks in 2020/21 for this report shows that 56% were still published by NEI-CEDA, despite the presence of other national actors, including Eburnie and Frat-Mat (Cote d'Ivoire Ministry of National Education, 2020).

## A few big providers push for digitized and standardized education content

In the larger publishing industry, trends in the textbook publishing market are difficult to detect due to rapid technological developments. Houghton Mifflin Harcourt, Cengage, McGraw Hill, Pearson and Scholastic dominate English-language education publishing. Individual publishers prominent in other large markets include Somos Educação in Brazil, Central China Publishing & Media Investment Holding Group Co Ltd, Hachette in France, the S Chand Group in India and Prosveshcheniye in the Russian Federation (Wischenbart, 2019). In Brazil, the national competition regulator authorized the takeover of Somos Educação by Kroton Educacional, the country's private education sector market leader, arguing that the transaction would not decrease competition (BSIC, 2018; Competition Policy International, 2018). S Chand provides textbooks and boasts of its national distribution and sales network, including over 5,000 distributors and dealers (S Chand and Company Ltd, 2021).

66

In Ethiopia, the textbook bidding process and criteria led to the selection of international publishers, since local competitors lacked capacity

"

In recent years, many companies have been linking textbooks, assessment systems and online learning for students and teachers

In recent years, many of these companies have expanded and modified their roles in education systems by linking textbooks, assessment systems and online learning for students and teachers. In addition, parents have found their focus on customized learning needs appealing (Wischenbart, 2019). In the United States, the five established players are viewed as operating like a cartel. The average school district cost per student per year is around US\$250, a significant amount for poor districts (Zook, 2017). Market concentration can increase inequality. Textbook pricing causes a vicious cycle: Schools that cannot afford the latest textbooks, which are the basis for many standardized tests, produce poorer test scores, whereupon school funding is reduced. The major providers are also part of consortiums including companies that develop examinations and grade them. This gives schools and teachers extra incentive to use books from the big publishers (Broussard, 2014). Personalized learning is also being developed although evidence is weak that it is effective (Pane, 2018).

Pearson, the global market leader in education publishing, changed its slogan from 'world's largest publisher of textbooks and online teaching materials' to 'world's digital learning company', with more focus on online schooling and assessment. Its immense product range includes English learning apps, learning platforms, tutoring apps, classroom assessment, certification, virtual schools and online learning services (Pearson, 2019). Its 2025 digital strategy focuses on routinization of teaching and on expanding complementary teaching expertise to use of technology tools. These developments could diminish the broader purpose of public schooling in favour of personalized learning with education platforms, potentially locking customers into proprietary online services (Sellar and Hogan, 2019).

As part of digital expansion strategies, large firms market platforms and applications broadly as 'solutions' directed at students, including tutoring, language learning and personalized assessment, and at teachers and school leaders for professional development and teacher training in digital learning. Some companies, including Somos Educação, provide solutions and platforms relating to core content, digital learning, teacher improvement, bilingual education, social and emotional learning, and e-commerce, reaching over 3,400 private partner schools (Wischenbart, 2019). S Chand is blending online and offline interventions with apps and digital initiatives; its vision includes stronger integration of the provision of analytics and insight, digital content, live classes, assessment and teacher content (S Chand and Company Ltd, 2021).

Beyond textbook publishers, technology giants such as Amazon, Google and Microsoft have entered the online education sector, a trend strengthened by the expansion of online learning during the COVID-19 pandemic (Williamson and Hogan, 2020).

Scripted curricula, with their highly structured lessons, and related technology are the focus of much criticism and concern. Teachers and school leaders in Australia suggest such services are leading to deprofessionalization of public school teachers, reducing their autonomy and increasing their concern over being replaced (Hogan et al., 2018). Scripted curricula are viewed as constraining teacher and student intellectual participation in the classroom (Fitz and Nikolaidis, 2020). In the US state of New York, schools offered scripted EngageNY modules to help meet the Common Core State Standards. Interviewed educators appreciated some aspects of the modules, but were alarmed by the presumption that inequity caused by disability and poverty could be redressed by holding students to high standards (Timberlake et al., 2017).

Similar trends can be seen in low- and middle-income countries, where a possible goal of low-fee private schools is to 'teacher-proof' lesson plans so that even unqualified people can teach them, as seen in the detailed developed lesson plans of Omega Schools, Rising Academies and Bridge International Academies (Härmä, 2021). Analysis of the Bridge International model has found that it uses scripted lessons, GPS technology and automation to create a market for its schools and reduce teacher autonomy (Riep, 2017).

99
66

State and local education authorities in the United States issued nearly 54,000 bids and calls for proposals in 2019

### GOVERNMENTS ARE OUTSOURCING MORE SUPPORT SERVICES IN EDUCATION

The choice whether to deliver support services in education internally or to outsource them relates to sound public financial management. Support services represent the largest expenditure in education budgets after teacher salaries, yet surprisingly little is known about good practice (LaRocque, 2008). Analysis of education technology procurement process experiences in the United States found that school districts and schools were typically overwhelmed by thousands of education technology vendors marketing a wide array of products (Morrison et al., 2019).

State and local education authorities in the United States issued nearly 54,000 bids and calls for proposals in 2019 for companies in 12 industry groups, 10 of which were not related directly to classroom teaching and learning (Parkin and Irby, 2019). In Michigan, which tracked key support service contracting over time, the percentage of districts contracting any of three key services (food, custodial and transport) to private-sector providers doubled in 10 years, from 36% in 2005 to 70% in 2015 (Hohman and Slasinski, 2019) (**Figure 2.13**).

Critics of outsourcing fear that privatization could undermine public services and professionalism. A systematic review of 14 studies focused on contracting within social services, such as education, employment and health care, found the documented consequences to be primarily negative. They included changed workforce composition (smaller workforce, replacement of experienced employees with younger, lower-paid people), poorer working conditions (higher pace and performance demands, poorer safety measures), worse salaries and benefits and reduced employee satisfaction (less job security, more stress and burnout) (Vrangbæk et al., 2015). It should be noted that existing literature on outsourcing primarily focuses on the United States, and seldom includes analysis of non-instructional education services.

By focusing on cost efficiency, outsourcing could free up time and money for instruction. An analysis of service contracting in 1,000 Texas state schools found an association between an increase in district contracting by 1 percentage point and a decrease in resources allocated to core instructional functions by 0.05 percentage point. However, more contracting was also associated with improved educational performance (Rho, 2013).

**99** 

### **FIGURE 2.13:**

School district outsourcing of support services has risen over time in the US state of Michigan Percentage of districts that have outsourced support services, 2005–19



Contracting of custodial services usually leads to worse labour standards and poorer outcomes. An Australian analysis found that key results of increased contracting of cleaning staff were that contractors proliferated, incidence of underpayment increased, cleaning hours were reduced and occupational health and safety standards were lowered (Gerrard and Barron, 2020). A decision to privatize cleaning services to two major firms in Chicago, United States, in 2012 led to dirtier schools and discussions of ending the contracts in 2020 (Ravitch, 2020). Food catering contracting has major consequences for student health and labour standards. In the United States, seven states account for nearly 50% of all food service outsourcing in education. School officials cite high labour costs as the main reason for outsourcing. Over the 50 years since school systems started outsourcing, key consequences have included declining labour standards, increased use of branded, pre-packaged food and increased marketing in an effort to cultivate lifelong consumers. Several studies have found that districts with privately managed cafeterias offer less healthy meals (Gaddis, 2020).

Analysis conducted with 446 social actors responsible for organic food acquisition in Santa Catarina, Brazil, perceived outsourcing as hindering programme quality, as it decreased family farm sales and exposed children to food procured by companies (de Andrade Silverio and Araújo de Sousa, 2014). The choice of contractors has a wide-ranging impact. Analysis from the US state of California found that contracting with healthier school meal providers increased student achievement (Anderson et al., 2018). A study of primary school students in Naples, Italy, showed that the larger the catering company, the lower student satisfaction with food (Maietta and Gorgitano, 2016).

Children's health and physical education form another area with substantial outsourcing. Analysis of health and physical education service outsourcing in Australia, Ireland, the United Kingdom and the United States, as well as Hong Kong, China, found that schools' health-related work was highly marketized and networked, which had consequences for teachers' work (Macdonald et al., 2020). In Australia, for example, involvement of sport organizations gradually changed from providing physical equipment in the 1950s to suggesting curriculum material in the 1990s and, more recently, direct engagement of external institutions in the school system, despite a lack of teaching qualifications. Four in five schools had outsourced some aspects of sports education in Queensland (Sperka, 2020).

### CONCLUSION

Non-state education provision, which has been increasing in much of the world, is an important phenomenon with historical roots. The distinction between state and non-state providers is not always clear. Understanding differences between providers can help improve understanding of trends, impact on students and systems, and appropriate policy responses to improve access, equity and quality.

Comparisons of state and non-state schools' effectiveness have been inadequate. They have not accounted for diversity in providers' objectives, schools' student intake and available resources. After factoring in these differences, non-state schools do not hold a sizeable advantage. Meanwhile, few studies have accounted for the relative cost of education provision or looked beyond the most rudimentary outcomes.

Moreover, the impact of non-state involvement should be examined from the perspective not just of the individual student but also the overall system. While non-state provision of core education can fill gaps in the short or medium term, it leads to segregation and inequality and has not yet convincingly managed to lift the quality of all schools, including public ones.

Beyond core education, ancillary goods and services play diverse, often contrasting roles in challenging, supporting or undermining education system performance and the equity record for teachers and learners. The push for content digitization, led by large publishing and technology companies, is accelerating the need to address equitable access to technology and digital learning.

One of the most valuable lessons from the COVID-19 crisis is how fragile systems are when governments allow a laissez-faire system to develop, without proper regulation and financing. This jeopardizes the education of many students and teachers' jobs. Building more crisis-resilient education systems should be a priority.

Students learn in a crowded classroom in Kajiado County, Kenya. mmater

12mms

Chales,

CREDIT: Jake Lyell/Alamy Stock Photo



CHAPTER



# Governance and regulation



### **KEY MESSAGES**

#### Good governance and effective regulation are key for delivering equitable education of good quality.

- In 94 countries, sector plans or strategies envisage intervention by non-state actors in provision or other service delivery.
- Distribution of responsibilities for education provision is often fragmented among ministries and by type of actor. In 13% of countries, multiple authorities share responsibility for quality assurance of non-state activities in education.
- Not all approaches to education system governance treat all students, teachers and schools uniformly.

#### Governance can be affected by funding arrangements.

- In most countries, non-state actors receive government funding, which takes various forms: per-student subsidies in 79% of countries, subsidies to parents in 23%, loans and gifts to schools in 27% and support for teacher salaries or operational expenses in 61%.
- Although public-private partnerships' impact ultimately depends on their design, a scoping review of 98 studies on partnership modalities found that, in at least two thirds, the impact of subsidy, voucher and charter school programmes on equity was negative.

### Equity and quality are often low on any list of regulations' aims.

- Nearly all systems stipulate requirements for entry and operation, including registration and licensing, but only 53% regulate teacher accreditation. About 55% of countries prevent selective admission procedures. Some 27% prohibit profit making in primary and secondary education and 67% cap fees.
- At least 27 countries' statistics recognize unregistered schools. In Uganda, 14% of primary and 13% of secondary schools are unregistered.

#### Weak implementation and accountability undermine regulations' effectiveness.

In some low- and lower-middle-income countries, complex, expensive or long registration procedures deter providers from obtaining official recognition. Nigeria's Lagos state government had approved just one in four of some 20,000 private schools as of 2021.

### Governments hold education providers accountable for compliance with standards.

- In 81% of countries, regulations mandate participation by non-state schools in large-scale assessments.
- Almost all countries apply sanctions, school closure or licence withdrawal if non-state schools do not comply with regulations. Some 54% of countries also regulate such closures' duration.
- About 90 countries have codes of ethics or conduct for teachers and school personnel, which often cover non-state providers.

#### Private supplementary tutoring is rarely regulated.

- Private tutoring is unregulated in 48% of countries. Only 53 countries regulate it in education legislation, while 19 regulate it only under commercial law.
- In 31% of countries, regulations specify tutors' required qualifications; 10 countries explicitly ban teachers from tutoring.

Education system governance is often fragmented	67
Regulations do not focus enough on equity and quality	75
Weak implementation often undermines regulation	86
Private tutoring regulations tend to be underdeveloped	90
Conclusion	91

The more non-state activities in education expand, the more the spotlight turns to the role of government, as the primary duty bearer of the right to education, to ensure education system steering, oversight, monitoring and accountability. Through legislative and policy-setting functions, governments are expected to establish sound regulatory frameworks ensuring that all actors operate under the same conditions, that the right to education can be enjoyed by all and that public funds are used efficiently, effectively and equitably.

The scope of governance in the realm of public goods and services is provision, distribution and regulation. Although it should be noted that regulation also affects provision and distribution, regulation can be defined as the part of governance that deals with the 'flow of events and behaviour' (Braithwaite et al., 2007). Just as the 2009 Education for All Global Monitoring Report defined education governance as the 'processes, policies and institutional arrangements that connect the many actors in education' (UNESCO, 2008), it is clear that governance and regulation capture both formal and informal rules and norms.

Good governance is expected to achieve multiple objectives. Different organizations emphasize different aspects. A recent comprehensive definition includes participation; responsiveness to the public in terms of delivering services and addressing complaints; efficiency and effectiveness; openness and transparency; respect of the rule of law; ethical conduct; development of competence and capacity; innovation and openness to change; sustainability and long-term orientation; sound financial management; promotion of human rights, cultural diversity and social cohesion; and accountability (Council of Europe, 2018). This chapter addresses good governance with respect to non-state activity in education, paying particular attention to promotion of human rights through regulation. Human rights frameworks enshrine both the right of parents to choose their children's education and the right of all children to receive education of good quality; at least one year of pre-primary and 12 years of primary and secondary education should be free. Yet these rights may not align with each other. Where parents have unlimited school choice, the most marginalized populations are at risk of segregation and discrimination, amplifying inequality. Where schools are free to apply fees and select their students by setting their own admission criteria, they grant privileges to the wealthiest and the top achievers. The exercise of the liberty to choose, establish and manage schools may fulfil the obligation expressed in Article 13 of the 1966 International Covenant on Economic, Social and Cultural Rights to 'conform to such minimum standards as may be laid down by the State', but may contravene education systems' obligation to pursue equity and inclusion and adhere to the tenets of availability, accessibility, acceptability and adaptability.

Not all approaches to education system governance treat all students, teachers and schools uniformly. In some cases, distribution of responsibilities for education provision is fragmented among ministries and by type of actor. Some countries turn a blind eye to regulatory gaps that lead to neglect of equity and quality. Others over-regulate non-state actors, at least on paper, but without achieving the results ostensibly desired – and without taking the same approach in oversight of public education institutions. Still others pursue governance approaches where the borders between actors and levels are blurred, with intersecting and overlapping powers, interests and responsibilities (Zancajo et al., 2021). It is a challenge for governments to tread the fine line between too little oversight, which can cause unplanned expansion that exacerbates inequality, and excessive regulation, which can deter non-state providers from contributing to the education system.

This chapter discusses regulations related to the design of rules for individual establishments and operations (e.g. infrastructure, licensing, funding, financial operations, profit making and tax incentives); the design of the education system as a whole, of which non-state providers are a part, which includes rules for equitable access (e.g. school admission, school choice and fees) and rules for equitable quality (e.g. workforce, curriculum and learning materials); and the implementation of these rules, including follow-up actions (e.g. quality assurance, sanctions and accountability) (**Figure 3.1**). The analysis draws on a review of 211 education systems for the *Global Education Monitoring Report's* Profiles for Enhancing Education Reviews (PEER) website.

### EDUCATION SYSTEM GOVERNANCE IS OFTEN FRAGMENTED

Countries govern non-state actors in education in various ways, often linked to how these actors emerged and began operations. Modern education systems are characterized by governance at multiple levels, with reforms in recent decades trending towards decentralization, in which central authorities maintain responsibility for monitoring and regulation but shift responsibility for implementation to local levels. Decentralization can take various forms, focusing on the relationship with local authorities (e.g. deconcentration, devolution and delegation), with schools (e.g. school-based management, self-governance and outsourced management) and with parents (e.g. school choice). Non-state education provision is regulated to varying degrees, usually reflecting whether and how such institutions are financed. The exception is where an unregulated market operates, notably in the case of supplementary tuition.

### HISTORY, CULTURE AND IDEOLOGY AFFECT HOW STATES GOVERN NON-STATE ACTORS

Countries vary in how central governments manage non-state education providers. Globally, 39% of countries have a single department, division or agency on non-state providers at the national level under the ministry in charge of primary and secondary education, with regional shares as high as 56% in sub-Saharan Africa and 73% in Northern Africa and Western Asia. Education ministries are exclusively responsible for quality assurance of non-state actors in 83% of countries, while in 13% multiple authorities share responsibility.

In some countries, responsibilities for governing non-state providers are shared between ministries, departments or other actors. Fragmentation can have a negative effect on system oversight if responsibilities overlap, are unclearly articulated or replaced in practice by informal rules. This is often the case in countries where lengthy conflict has resulted in destabilized public administration. In the Democratic Republic of the Congo, the weakening of the central government after years of conflict and the emergence of non-state actors as the main providers resulted in 'negotiated' governance between state- and non-state actors (Titeca and de Herdt, 2011). Provincial education authorities, representatives of faith-based organizations and private actors play a significant role in the management of local service delivery, leading to considerable provincial differences in school system oversight (Cambridge Education, 2021).

In 96 countries, sector plans or strategies describe roles for non-state actors in direct education provision or other service delivery. But countries differ in how they engage with non-state actors in education. History, culture and ideology, often combined, affect how the state governs non-state actors. In 1974, shortly after independence, Bangladesh passed the Primary Schools (Taking Over) Act nationalizing 36,000 community primary schools, partly to signal a commitment to secularism (Hossain et al., 2002). Its 2010 education policy further pushed for more primary schools to become government schools, stating that the 'responsibility of primary education cannot be delegated to private or NGO sectors' (Bangladesh Ministry of Education, 2010, p. 6). Non-governmental organizations (NGOs) have been catering for millions of disadvantaged students at the primary level, and secondary schools, whether secular or religious, have remained almost exclusively in private hands, albeit subsidized by the state.

66

In 96 countries, sector plans or strategies describe roles for nonstate actors in direct education provision or other service delivery

99



Morocco's 2008 Education and Teaching Charter set a target for the share of private institutions in total enrolment to reach 25% by 2024 throughout the education system. The 2015–30 strategic vision calls private education 'a partner of public education in the generalization and realization of equity'. Between 2009/10 and 2018/19, the number of private schools increased by 91%, compared with 15% for public schools. Willingness to engage with the private sector is most explicit in early childhood education, with the education ministry's Directorate of Cooperation and Promotion of Early Childhood Education being renamed Directorate of Cooperation and Promotion of Private School Instruction (Abdous, 2020).

Since the early 1980s, many countries have granted schools more autonomy to make decisions about curricula, resource allocation and, in some cases, teachers. Globally, 74% of countries have regulations mandating some form of school-based management in non-state schools (**Figure 3.2**). Regulations explicitly require boards for all types of non-state schools in 45% of countries but for government-aided schools in only 14% countries. About 64% of countries also mandate the composition of school boards.

The boards' roles vary but can be substantive in some countries. In the Flemish Community (Belgium), school

boards enjoy considerable autonomy. For instance, equity-oriented grants are disbursed to schools according to norms, yet major differences in financial conditions exist between schools, as they can choose how to allocate such funding (Shewbridge et al., 2019). In Oman, a 2017 ministerial decision requires schools to have a board of at least five trustees, responsible for reporting to the education ministry. One board member must represent parents and at least two members are required to have experience in education. School owners, partners or stakeholders can be board members and can vote on governance procedures. Every private school must also create a parental council for which the education minister stipulates certain conditions regarding composition and organization. In Zimbabwe, where the non-governmental sector accounts for 22% of primary and 27% of secondary schools, run mostly by individuals, trusts, farms, missions and companies (Zimbabwe Ministry of Primary and Secondary Education, 2021), School Development Committees foster parental and community involvement in school management (PEER country profiles).

Faith-based schools account for a significant share of enrolment in many countries, and this has an impact on governance models (Wodon, 2021a). Globally, in 22% of countries, a ministry or other authority of religious affairs, distinct from the education ministry, governs

### FIGURE 3.2:

#### Most countries require non-state schools to have boards

Percentage of countries where non-state schools must have some form of school-based management, by region, 2021



99

faith-based schools. In Northern African and Western Asia (e.g. Egypt and Tunisia), where these tend to be state schools, the share is 70%, while in the rest of Asia it is over 50%. In Cambodia, religious schools must be registered with the Ministry of Cults and Religion and are under the authority of the ministry's General Inspectorate of National Buddhist Education (UNICEF, 2020). In Indonesia, the Ministry of Religious Affairs is responsible for public and private Islamic schools (Jackson, 2016). School policy towards faith-based schools varies in sub-Saharan Africa (**Box 3.1**).

In discussions about the roles of non-state and state education actors in fragile settings or emergencies, the issues of coordination and support often arise. How can duplication be avoided and gaps identified? After Hurricane Mitch in Honduras in 1998, clear roles and responsibilities contributed to the success of the Honduran Programme of Community Education in delivering education services to rural and deprived areas with direct participation by parents in their administration. The Ministry of Education manages the programme's coordination, financing and monitoring (Honduras Ministry of Education, 2020).

### 66

International NGOs catering for disadvantaged populations make various arrangements with governments, from replacing state services to strengthening the state system International NGOs catering for disadvantaged populations make various arrangements with governments, from replacing state services to strengthening the state system. In Sierra Leone, Street Child, an international NGO, constructed more than 300 schools between 2010 and 2014 in rural and remote areas lacking state provision or oversight. As Street Child increased the scale and scope of its work, concerns about sustainability exposed the limitations of operating outside the state. Street Child has since begun coordinating and collaborating with the state to ensure the schools' legitimacy, complementarity and ability to respond to crises. More than 30 schools have been approved and adopted by the state and had their infrastructure and staff verified to assure compliance with state standards (Street Child, 2021).

In Somalia, as a result of the civil war and collapse of state authority, former teachers established umbrella associations. In 2018, there were 14 such associations with more than 1,000 affiliated schools serving over 250,000 students, of which the largest were the Formal Private Education Network and the Schools Association for Formal Education, established in the 1990s. By contrast, the Ministry of Education, Culture and Higher Education manages 93 schools with 32.000 students. The associations issue certificates and degrees recognized abroad and have helped rebuild the syllabus. Schools need to be registered with the state and comply with association standards to be umbrella members. The ministry established the Department of Umbrella and Private Education to regulate the non-state education system and coordinate with private education institutions, while the Curriculum and Quality

### Relations between government and religion on faith-based schools have evolved in sub-Saharan Africa

Public policy towards faith-based schools varies in sub-Saharan Africa (Scheunpflug and Wenz, 2021). In Cameroon, 26% of schools are church-run. The government has not fulfilled a commitment to fund their teachers. As the church education system is dependent on tuition fees, teachers are often not adequately paid, especially in areas where the population is poorer (Lange et al., 2021; Wodon, 2020). In Comoros, the National Advisory Council on Education reviews all draft laws relating to public or private education, but the ministry in charge of Islamic affairs supports education in Koranic schools. In Djibouti, a 2020 law stipulates that the Ministry of Muslim Affairs, Culture and Waqf Assets 'elaborates and proposes the basic foundations of Islamic education and the dissemination of Arab-Islamic culture'. In Madagascar, where faith-based schools play an important role in the education system, the education ministry supervises directorates for each of three faiths: Catholic, Protestant and Muslim; the state pays part of teacher salaries.

The Democratic Republic of the Congo nationalized religious schools in 1974, but in 1977 the state and four religious communities – Catholic, Protestant, Kimbanguist and Muslim – signed a convention introducing a state-owned and regulated but jointly managed system (Scheunpflug and Wenz, 2021). The convention divided the public education sector into one state- and four faith-based networks, each with its own administration. Public schools that belong to a faith-based network are run by both a religious and a state education office, complicating management. The networks are indirectly in charge of distributing salaries for more than half the teachers, and can establish schools and recruit staff locally with no state oversight. Religious networks also own private schools, which compete with public schools to attract students. In 2019, the government introduced free education, abolishing all school fees for primary education. Religious networks need to register schools and teachers with the provincial education directorates to receive state funding (Cambridge Education, 2021).

In Kenya, churches transferred their schools to local authorities under the 1968 Education Act, instead becoming sponsors. The 2013 Basic Education Act defines a sponsor as a 'person or institution who makes a significant contribution and impact on the academic, financial, infrastructural and spiritual development of an institution of basic education'. It increased state authority over schools and gave each new County Education Board the right to name the chairs of schools' governing boards and the Teachers Service Commission the power to appoint head teachers, as well as a bigger role in their management. While sponsors, such as churches, are consulted over decisions concerning teachers, they no longer have veto power over head teachers and their responsibilities are often limited to issues related to infrastructure and the spiritual care of the students (D'Agostino et al., 2019).

In Rwanda, Catholic schools not run by religious orders were nationalized in 1966, but a smaller number of religious order and Protestant schools were not. Faith-based schools grew in number after the genocide. There is strong cooperation between church and state. The government pays teacher salaries at church-run schools. In primary education and government-aided secondary schools, tuition fees are prohibited. Non-state faith-based schools, categorized as government-aided schools, receive support for construction and renovation. In the United Republic of Tanzania, all schools were nationalized during one-party rule, but Christian schools became independent after 1985.

Assurance Department has sections on curriculum, teacher management and supervision of both public and private institutions (Altai Consulting, 2018).

In Turkey, non-state organizations and private enterprises set up accredited temporary education centres for Syrian refugees, initially under loose state supervision. In September 2013, a regulatory circular announced that education services, in or out of camps, had to be planned, coordinated and monitored by the Ministry of National Education and staff it appointed locally (Tezel Mccarthy, 2017). In the process, non-state actors operating such centres gained recognition. However, as the centres' education quality was questioned, the Ministry of National Education adopted a policy in September 2016 to gradually phase them out and enrol all Syrian students in public schools (UNESCO, 2019).

### FUNDING OF NON-STATE PROVIDERS CAN AFFECT GOVERNANCE ARRANGEMENTS

In most countries, non-state actors receive government funding, which takes various forms: per student subsidies in 79% of countries, subsidies to parents in 23%, loans and gifts to schools in 27% and support to teacher salaries or operational expenses

66

In most countries, non-state actors receive government funding, which takes various forms

,,,

(e.g. textbooks or technological equipment) in 61%. In 83% of the countries where governments support non-state schools, faith-based schools are among the beneficiaries. Since 1976, El Salvador's Ministry of Education has paid some Catholic school teacher salaries. The government sets a quota of teaching positions; archdioceses decide how to distribute them among schools. The church typically owns the infrastructure, and it manages the schools through school leadership councils and can hire and fire teachers. However, the government regulates the schools through directives and inspects their operations (USAID et al., 2018).

Non-state and, especially, faith-based schools can be effective in lobbying. In Australia, the school funding mechanism has historically favoured private schools over public schools. Following a major review in 2011, a 2013 education law set the country's first nationwide minimum capitation grant per student, with a formula to determine additional support to disadvantaged students (affected by poverty, disability, indigeneity and low English proficiency) and schools (rural and small). Private schools' minimum capitation was to be adjusted downward to take their ability to charge fees into account. However, the reform was diluted. A government commitment that no school would lose resources entrenched historically unequal allocations. The Catholic school sector was given discretion in applying the formula. States were not forced to comply with federal targets. And federal funding that would have helped underfunded schools catch up was postponed by five years, meaning overfunded schools would lose their advantage at a very slow rate (Goss et al., 2016). Ultimately, private schools received a larger increase in funding than public schools between 2007/08 and 2016/17 (Baker, 2019).

In most cases, the only education planning criteria for authorizing new, publicly funded private providers is demand. In Pakistan's Punjab province, the Punjab Education Foundation is a semi-autonomous organization promoting public-private partnerships to expand education access and improve quality. Its programmes include a voucher plan, a private school subsidy programme, a public school management outsourcing mechanism and the New School Program (Ansari, 2020), which offers incentives to individuals and organizations to set up schools, subject to basic conditions related to the applicant's education level and residence. The schools must be in areas with no formal school within a one-kilometre radius and serve a population of at least 350 people, as the aim is to reach populations that would not otherwise have access to school. Since 2007, more than 2,500 schools have been established (Punjab Education Foundation, 2021). However, there is a risk that such demand-based approaches in authorizing providers could increase school segregation and social stratification between schools.

In Peru, the government negotiates with selected not-for-profit actors to operate public schools, i.e. schools that either fully operate under education ministry regulations and have their teacher salaries covered by the government (en convenio) or receive some government financial support for general funds, teacher salaries, equipment and buildings (financiación mixta). The schools must have previously obtained formal certification and should respect minimum requirements set in the private education institution regulations. The operating agreement includes regular inspections of facilities, curriculum and learning materials, and a calendar aligned with that of public schools. Funds are transferred not on the basis of the number of students enrolled but on a fixed-fee, case-by-case basis (Rossignoli, 2021).

In Uganda, a public-private partnership took the form of a per-student capitation grant to private secondary schools to enrol qualifying students at no added charge. Between 2007 and 2016, 850 schools took part in the programme, which accounted for 46% of enrolment. A 2010 framework policy called for bidding to select schools based on needs identification, value assessment and options analysis, to be followed by negotiation of terms with successful applicants. However, capacity was lacking for this, as well as a clear accountability structure, especially for local authorities that disbursed funds (ISER, 2016, 2017). Moreover, neither the implementation guidelines nor the rationale behind partner school selection was clear to stakeholders interviewed for an evaluation (O'Donoghue et al., 2018). The programme was phased out starting in 2018 (Ahimbisibwe, 2018).

To improve funding of non-state providers, stricter criteria are needed that consider not only the level of education demand but also how schools requesting public funding can improve equity and diversity (Zancajo et al., 2021). In Ireland, public funding for private providers was conditional on basic requirements, such as facility standards and official curriculum implementation (Buchanan and Fox, 2008). Since 2011, the Department

### 66

## Public-private partnerships for education are found in relatively few countries

of Education and Skills has introduced a bidding process for authorizing access to public funding by new private providers. It identifies areas where school supply needs to increase because of demographic change and releases competitive calls. This process allows public authorities to increase control over the publicly funded school supply and to guarantee diversity among new providers (Ireland Department of Education and Skills, 2020).

Procurement or bidding processes have also been used to select new publicly funded providers in the niche area of charter school programmes in some low- and middle-income countries. Competitive calls have focused on selecting organizations to manage publicly owned schools. But implementation problems have been reported. In the Partnership Schools for Liberia (PSL) programme, competitive tenders were adopted because of controversy about an initial agreement between the government and a school chain, Bridge International Academies, that suggested a monopoly on contracting (Romero et al., 2020b). Even so, the bidding process was not considered sufficiently competitive and transparent (Cameron, 2019).

The Education Management Organization's charter school programme in Pakistan's Sindh province uses a two-phase competitive bidding process. First, bidders' technical capacity is assessed on the basis of their management experience, proposed work plan, professional profiles and strategies to achieve performance and efficiency targets. Then qualifying organizations are assessed based on financial proposals (LaRocque and Sipahimalani-Rao, 2019).

Public-private partnerships for education are found in relatively few countries: Voucher programmes exist in 8 countries, starting with Chile in the 1980s (Elacqua, 2012), while charter schools are found in 14. But there is concern that governance arrangements involving public-private partnerships have a negative impact on equity. Although the impact ultimately depends on their design, a scoping review of 98 studies on public-private partnership modalities found that, in at least two thirds, the impact of subsidy, voucher and charter school programmes on equity was negative (Verger et al., 2021). Examples from England (United Kingdom) (**Box 3.2**) and the US state of Louisiana (**Box 3.3**), where governance reforms transformed most schools into academies and charter schools, respectively, highlight the challenges.

A core argument for public-private partnerships is their perceived cost-effectiveness. But not everyone believes that this is true or that the way they cut costs is worthwhile. As teacher salaries are by far the largest cost item in education, cost-efficiency would need to be driven by lower teacher salaries. But teachers are paid less than other professionals (**Chapter 19**). A global survey suggested that in 28 of 35 countries, the actual starting wage for teachers was lower than that perceived as fair (Dolton et al., 2018). Still, there are exceptions to this rule: In some countries, large salary gaps between the state and non-state sectors or, within the state sector, between permanent and contract teachers indicate segmented teacher labour markets with some teachers overpaid (Crawfurd and Pugatch, 2020).

Much of the research to this effect comes from South Asia. Teachers in the region make less than other professionals, on average (Beteille et al., 2020), yet there are large pay inequalities among teachers. Employment of young, ungualified female teachers with limited employment alternatives has been recognized as a key reason for the low-cost private school business model in Pakistan (Andrabi et al., 2007). In Punjab province, four public-private partnership programmes paid between US\$3 and US\$9 per student per month, compared with the government cost of US\$9 per student per month in public schools. The lower cost per student was driven by lower salary costs (Crawfurd and Hares, 2020). In India's Andhra Pradesh state, voucher recipients in private schools were found to perform more or less at par with non-recipients in public schools at less than one third of the unit cost of public schools and one sixth of the teacher salary cost (Muralidharan and Sundararaman, 2015). Still, it remains unclear whether public-private partnerships are an effective way to address a segmented teacher labour market. Effective governance calls for treating the education system as one, finding direct and sustainable ways to address imbalances and inefficiencies.

99

### BOX 3.2:

### Academies in England - a radical governance reform which may not be equitable

Academy schools in England (United Kingdom) are an example of education reforms that have resulted in substantial transformation of governance in the past 20 years. The schools enjoy considerable autonomy and, while they are part of the state system, operate outside local authority control, establishing a distinct governance setting.

The first 'sponsored' academies were established in 2002/03 as a remedial programme to improve secondary schools in disadvantaged areas (Eyles et al., 2018). Authorized and funded directly by the government, they were set up as charitable trusts with focuses in particular curriculum areas. Then the 2010 Academies Act allowed any public primary or secondary school to become an academy. The act introduced two 'conversion' patterns. Schools that Ofsted, the education standards authority, rated as 'outstanding' (or, increasingly, 'good, with outstanding features'), known as 'converter' academies, had their applications fast-tracked and did not need a sponsor. Schools rated as requiring improvement or 'inadequate' could volunteer to be converted by seeking a sponsor ready to raise their low standards. Potential sponsors only had to discuss the possibility of conversion of one or more schools with parents or staff and register their intention to apply at the Department for Education (Ladd and Fiske, 2016).

Sponsored and converter academies enjoy similar status and have similar obligations. In principle, they have more autonomy than other public schools, as they can choose their own curriculum and do not have to align with the statutory guidance on teacher pay and conditions. A contract between a non-profit trust (exempt charity) and the secretary of state for education is necessary to get funds and authorize the academy. The trust can run one or multiple academies with a single contract, or single-contract academies can group into umbrella trusts or other forms of collaborative partnerships (West and Wolfe, 2018). As of 2019/20, 43% of state-funded schools (enrolling 53% of students) were academies: 78% of secondary and 36% of primary schools. Of those, 84% were part of a multi-academy trust (National Governance Association, 2020). There are 29 multi-academy trusts with more than 25 schools each (British Education Suppliers Association, 2021).

In 2014 most schools said conversion was a way to 'gain greater freedom to use funding as they see fit' and 'raise educational standards' (Eyles et al., 2017, p. 129). Yet the vast majority of school leaders thought that their team, followed by the head teacher and governors, should influence individual school policy; private actors were ranked by just 1% of respondents (IPSOS Mori and the Key, 2015). By law, a multi-academy trust is one legal identity and single schools within them cease to exist as separate legal identities. Decision-making power rests with the governing body at the central level. Diminished school autonomy has been accompanied by increased school control by fewer people without real oversight. The 2016 Education and Adoption Act increased the powers of the secretary of state for education and the eight regional school commissioners who act on his behalf but were not appointed through a democratic parliamentary process (West and Wolfe, 2018).

By 2017, about 70% of such schools were converter academies. As the programme focus shifted from addressing underperformance to increasing autonomy, academies seem to have enrolled more affluent pupils (Bertoni et al., 2020; Eyles et al., 2018). The pre-2010 academy programme was associated with improved learning achievement in secondary schools (Eyles and Machin, 2019), but there was no achievement effect in primary converter academies (Eyles et al., 2018).

A 2016 white paper, *Educational Excellence for Everyone*, expressed an intention to convert all primary and secondary schools into academies by 2022. Despite changes in state funding allocation, inequality persists and tends to deepen according to school status. For instance, local authorities have to transfer minimum per-pupil funding to their schools while academy trusts do not, and there is no clarity on redistribution of block funding. Average per-pupil funding in real terms fell by 1.2% for the most deprived fifth of schools but increased by 2.9% for the least deprived (NAO, 2021).

A second counterargument is that most analyses rely on inadequate costing data, preventing a clear evaluation of cost-effectiveness (Levin and Belfield, 2015). Private school chains, in particular, are often backed by a well-funded central office that provides curriculum and other support that is not accounted for in overall costings (Crawfurd and Hares, 2020). An evaluation of the previously mentioned PSL programme in Liberia, through which 93 public schools were contracted out to 8 private providers, illustrates the challenge. It was difficult to estimate how providers cross-subsidized their activities. Improper behaviour skewed data: One contractor dismissed three quarters of teachers in its schools, possibly the lowest-performing ones, and pushed the resulting excess of pupils to neighbouring non-PSL schools. Learning outcomes varied widely by contractor, with costs varying. After three years of operation, PSL's average cost was around three times the government cost per child for only a modest increase in learning outcomes (Romero et al., 2020a).

A final counterargument is that private schools cost governments less because households bear some of the costs (**Chapter 4**). Analysis of 38 Organisation for Economic Co-operation and Development countries and

### BOX 3.3:

### Changes in school management in Louisiana had a controversial impact on students

In the US state of Louisiana, a 2003 reform established a Recovery School District with the objective of transforming schools classified as failing under the accountability measures of the No Child Left Behind Act. In the aftermath of Hurricane Katrina, when most schools in New Orleans were damaged, a 2005 act raised the minimum performance score required for takeover by the Recovery School District. State-takeover authority was also extended to districts with more than 30 'failing' schools and with at least 50% of their student population in schools below the minimum performance score (Sanders, 2018). These schools would either be directly managed by the Recovery School District or converted to charter schools overseen by it (The Learning Landscape, 2015).

Between 2005 and 2015, the number of charter schools increased from 20 to 130. In New Orleans, 91% of students were enrolled in charter schools, often Knowledge Is Power Program (KIPP) and KIPP-like 'no excuses' schools taking the approach that poverty should not be an excuse for low expectations or performance. One study, which matched schools that had been taken over with schools that had not, estimated large gains in learning in the former (Abdulkadiroğlu et al., 2016). Nevertheless, by 2013/14 the Recovery School District schools were still at the 17th percentile of student testing ranking and had low graduation rates (Deshotels, 2015). Many other studies have also found a positive impact of urban charter schools on test scores in the United States, while pointing out that this is not the only finding that matters (Cohodes and Parham, 2021).

The reform increased segregation and resulted in a tiered system of public schools, undermining education opportunities for Black students (Gumus-Dawes et al., 2013). Test-based accountability increased along with charter school expansion. Decisions on whether a school should close started being based on school performance on standardized achievement tests, putting high pressure on schools. Sanctions for poor school performance and strict discipline raised school suspension rates and other corrective actions (Hernandez, 2019), pushing vulnerable students out of the system and negatively influencing relations with educators (Coco et al., 2020). When the teacher union contract expired and charter schools were not required to hire certified teachers, new teachers were less qualified (Barrett and Harris, 2015) and predominantly white: The share of black teachers dropped from 71% to 49%, while the share of students from non-white minority groups remained at around 95% (Hernandez, 2019).

Although a Charter School Performance Compact requires the Department of Education and charter school authorizers to monitor schools' performance, lack of fiscal oversight meant that mismanagement and fraud, including use of funds for personal gain, support to charter businesses and services not provided, amounted to tens of millions of dollars (Center for Popular Democracy and Coalition for Community Schools, 2015). In May 2016, a law was passed to return the schools from the Recovery School District to the original boards as from July 2018, after 13 years of unsolved problems (Sanders, 2018).

partner countries found that, in 2016, Catholic schools represented savings to governments of US\$63 billion per year, in purchasing power parity terms, but that the savings were the result of shifting the burden to households (Wodon, 2020).

Questions also arise regarding governance of contracting and outsourcing procurement of other goods and services in education and the extent to which these practices reduce costs. In 1992, the United Kingdom adopted the private finance initiative (PFI), a procurement model based on public-private partnerships. These included nearly 100 education deals, valued at GBP 3.5 billion (Pratap and Chakrabarti, 2017). The model was discontinued in 2018 when Carillion, a key partner and one of the country's oldest construction firms, collapsed. Financial practices were strongly criticized, as the largest offshore funds of PFI projects made profits totalling GBP 1.8 billion in 2011–15 but paid no tax (Sherratt et al., 2020). A review of public-private partnerships for education infrastructure projects in Ireland found that they were no faster and produced no better value for money than traditional procurement processes (O'Shea et al., 2019).

Besides infrastructure, there is strong interest in contracting out non-instructional services to reduce costs. Although interesting cases are emerging from poorer countries (**Box 3.4**), most analyses come from high-income countries, especially the United States. In New Orleans, contracting increased total operational spending in public education, which in turn increased administrative spending, with less devoted to instruction (Buerger and Harris, 2020). Privatizing education transport in the state of Minnesota did not reduce costs (Thompson, 2011).

66

Questions arise regarding governance of contracting and outsourcing procurement of other goods and services in education and the extent to which these practices reduce costs

99

### BOX 3.4:

### Outsourcing teacher salary payments in the Democratic Republic of the Congo did not pay off

Especially in fragile contexts, weak systems for paying teacher salaries are a key challenge. Teachers can go months without pay due to bureaucratic rigidity and lack of access to the financial system. Some countries have tried to tackle the challenge via private engagement, such as using mobile money payment programmes in poorer countries to improve financial inclusion and pay teachers on time (Trucano, 2014).

In 2011, a public-private partnership between the state and the banking sector in the Democratic Republic of the Congo replaced arrangements between faith-based organizations and teachers for teacher salary payments. While the programme improved the situation in cities, it did not address the main problem: how to pay teachers in remote rural areas lacking access to banking. The programme gave banks an incentive to open new branches in Kinshasa but not in the countryside. It overlooked the fact that only 4% of adults held bank accounts in 2014, with two thirds of them in Kinshasa and in Katanga province. To cover the remaining areas, banks set up temporary payment points for teachers, tried to subcontract with mobile phone companies for payment and subcontracted other services as well. After a few years, transaction costs were reported to have increased significantly. Instead of getting direct payments from faith-based school networks, teachers had to make most arrangements individually. This eventually led to a change in which the Catholic Church used its networks and infrastructure to deliver payments (Brandt and De Herdt, 2020).

In fact, using a sample of 343 school districts over 12 years with data before and after outsourcing, further analysis showed that total transport costs increased by 21% due to outsourcing, raising per-pupil costs by about 16% (Puozaa, 2016). Food service companies also failed to deliver cost savings, often spending less on food and labour but more on fees and supplies. Any savings were due to hiring hourly workers, paying them less and giving them fewer benefits, while the quality of food deteriorated (Gaddis, 2020).

### REGULATIONS DO NOT FOCUS ENOUGH ON EQUITY AND QUALITY

The aim of regulation is, ideally, to mobilize all available resources to promote educational development that serves the academic, socioemotional and well-being needs of every learner and educator while also achieving desirable system outcomes related to equity, inclusion, quality, innovation and efficiency.

This report reviewed regulations covering non-state school establishment, operation, quality, equity and accountability in 211 education systems. It found that some areas are comprehensively covered by regulation. Nearly all systems stipulate requirements for entry and operation, including registration and licensing. Regulations on classroom size and pupil/teacher ratios are also common. By contrast, few countries regulate teacher accreditation, curriculum or admission procedures, which could promote more equitable access (**Figure 3.3**).

### ESTABLISHMENT REGULATIONS CAN BE VERY BUREAUCRATIC

Education providers need to register before they can receive a licence to operate a school and to fulfil legal (for

individual or corporate owners) and material conditions. In Chile, education establishments are recognized when they have proprietors (legal persons under public or private law or municipalities), mandatory minimum capital and adequate infrastructure. The proprietor must have a legal representative and an administrator who hold professional or bachelor's degrees and have not been convicted of anti-union practices, non-payment of social charges, violation of fundamental rights, or crimes against family, public morality and sexual integrity or related to drugs. The Ministry of Education keeps a Public Register of Proprietors and a Public Register of Officially Recognized Educational Establishments with relevant information about each school. There is no limit to the number of schools a proprietor can manage. In Honduras, non-state schools can be established and administered by any physical or legal persons, including communities, businesses and non-profit organizations. The Non-Governmental Education Institutions Regulations require them to have financial and pedagogical capacity and no pending legal liabilities.

In 80% of countries, regulations focus on space requirements, such as land, building and classroom size; in 74%, there are requirements on pupil/ teacher ratios or the number of pupils per classroom (**Figure 3.4**). In India, states have various regulations related to space. Andhra Pradesh stipulates minimum land area per student. Rajasthan and Karnataka set the minimum size of owned land. Uttar Pradesh uses two criteria to recognize a school: the minimum per-student area (9m<sup>2</sup>) and classroom size (180m<sup>2</sup>) (Ambast et al., 2017). In Madagascar, under a 2017 decree, private schools must be located in a quiet, healthy and secure place and the applicant must submit a plan of the premises and a title deed or an administrative certificate of the site's legal status. Regulations also cover water and sanitation, which can have important effects on girls' continuing education. Single-sex toilets are required in 47% of countries with available information. In Fiji, a 2011 policy sets minimum infrastructure standards, including proper sanitary facilities, one toilet for every 15 children, separated by gender, and safe drinking water. Schools must also comply with school health and occupational health and safety policies.

Various levels of government share responsibility for registering schools but often the highest level is involved in final decisions. In Albania, establishing private institutions, including those of religious communities and those with foreign language programmes, requires a Council of Ministers decision on a proposal by the education minister. In Algeria, each school must seek approval at *wilaya* (province) level on the application's admissibility, and a licence is issued by the minister if approved. Bhutan's 2018 guidelines call for an expression of interest and a master plan to be submitted to the *dzongkhag* (municipality), which assesses land ownership and registration, before joint verification with the Ministry of Education. The Ministry of Economic Affairs must approve a detailed project report for the proposal to obtain a licence. More requirements must be met for secondary schools. Israel grants a non-transferable licence for a minimum of a year. Long-established institutions that receive positive reviews ('green') can receive longer-term licences, while one-year licences are given



GEM StatLink: https://bit.ly/GEM2021\_fig3\_3

Source: GEM Report team analysis based on the PEER country profiles on non-state actors in education.

### -2



Land, building, or classroom size

GEM StatLink: https://bit.ly/GEM2021\_fig3\_4

20

0

Oceania

to institutions that are new or growing or have received conditional or negative reviews ('red'). Licences are subject to Ministry of Education renewal and inspection.

Central and

Southern Asia

Eastern and

South-eastern Asia

Source: GEM Report team analysis based on the PEER country profiles on non-state actors in education.

Latin America

and the

Caribbean

Pupil/teacher or pupil/classroom ratio

In some countries, requirements to operate non-state schools are complex. In Accra, Ghana, the Guidelines for the Establishment of Private Schools require approval from the government two years before the school is set up; other requirements concern teachers, land and infrastructure. More recent guidelines added conditions on pupil/class ratios, boards of governors and limits on school fee setting without approval from the Ghana Education Service Council (Härmä, 2019). Many schools lack information about registration procedures (Marchetta and Dilly, 2019).

In Delhi, India, 28 certificates or approvals from 6 departments are needed to open a private school.

These require 125 documents, of which 29 are for the Essentiality Certificate, 14 are to secure approval for the Scheme of Management and 82 are to obtain a Certificate of Recognition. Documents for the Essentiality Certificate alone include a copy of the Memorandum of Association and Rules and Regulations of Society, a copy of the Society Registration Certificate, the List of Governing Body Members, proof of no blood relationship among the society members, a complete list of society members with biometric data and signatures, and an undertaking that the school has provision for adequate drinking water and separate toilets for boys and girls. Obtaining the documents involves 16 officials from 3 separate Directorate of Education offices. The official estimate is that the process should take four months at most, but some cases have taken over five years (Centre for Civil Society, 2019).

World

Single-sex toilet

Northern

Africa and

Western Asia

Furone and

Northern

America

Sub-Saharan

Africa

Other infrastructure

66

In Delhi, India, the official estimate is that the private school approval process should take four months at most, but some cases have taken over five years

"

#### FIGURE 3.5:

### Just over a quarter of countries prohibit schools from making profits

Percentage of countries with regulations on school profit making, by region, 2021



### FINANCIAL REGULATIONS ACCOMPANY STATE SUPPORT

Specific rules exist for financial operations related to profit making, fee setting and incentives, especially where governments support non-state providers.

#### A few countries ban profit making

Some 27%, or 52, of 194 countries prohibit profit making by law in primary and secondary education (**Figure 3.5**), although usually only for government-supported schools. Over the past 10 years, 21 countries have introduced or amended such regulations. The 2013 Australia Education Act stipulates that independent schools must be not-for-profit to receive public funding. Chile's 2015 Inclusion Act states that proprietors receiving public money are prohibited from profit making and must allocate public contributions entirely to education. In China, 2018 regulations banned profit making in grades 1 to 9; the 2021 private education law further restricts private schools' profit-making opportunities and requires them to reinvest part of the profit in school development (Che, 2021; Jones et al., 2021). The 2014 framework law in the Democratic Republic of the Congo stipulates that fee revenue must be used to improve education quality. India's 2009 Right to Education Act requires private schools to register as non-profit societies or trusts and to use any surplus for charitable objectives. Yet charity governance is weakened by the lack of a central legal framework, inadequate staff and poor political commitment (Central Square Foundation, 2020).

In the Russian Federation, education organizations, which can generate revenue through service provision, have non-profit status under the education law and cannot distribute profit among their members. However, the 1994 Civil Code allows non-profit organizations to generate profit if it is used to achieve organizational goals, and the Non-profit Organizations Law specifies that profit can be generated from goods and service provision (PEER country profiles).

### 66

Caps and restrictions on non-state school tuition fees exist in 67% of countries

,,

### Fee levels are often regulated yet schools are not more accessible

Caps and restrictions on non-state school tuition fees exist in 116, or 67%, of 173 countries. Education ministries often must approve fee increases or can set limits on them. Bhutan's education ministry must approve private actors' fees and can specify that the revenue must be spent on, for instance, learning materials or teacher development programmes. In Dominica, private and private-aided schools cannot increase or add fees without the minister's approval and must give at least one term's notice. The Dominican Republic Ministry of Education can regulate and set private school fees. In Israel, a 2020 circular on parental payments specified that the ministry approves tuition fees on the basis of enrolment, profile and percentage of budget covered by the government; schools fully funded by the state cannot charge fees. A 2021 circular further mentioned a school may charge payments to purchase voluntary services as to the level approved by the Ministry of Education.

Nevertheless, in practice tuition restrictions do not appear to increase the probability of poorer learners having access to non-state schools. In Comoros, where the government had to approve independent private school tuition fees, 2% of students from the poorest households but 24% from the richest attended such schools (CONFEMEN, 2010). Schools in the Partnership Schools for Liberia programme were prohibited from charging fees, including for textbooks and uniforms, yet 48% of parents reported they had paid some fees (Romero et al., 2020b). Ugandan private schools taking part in the Universal Secondary Education programme were meant to stop charging fees in exchange for the support they received, but continued asking for informal fees and parental contributions for basic services (Omoeva and Gale, 2016). Richer households were thus more likely to benefit from the programme: The richest quintile received 20% of the subsidy and the poorest guintile 10% (Wokadala and Barungi, 2015).

In Pakistan, the 1984 Punjab Private Educational Institutions Ordinance stipulated that fees could not increase by more than 5% per year without formal provincial government approval. A 2016 amendment further regulated the fee structure, prohibiting schools from making parents pay extra for uniforms and textbooks from particular providers. But fee paying is an unofficial requirement for receiving vouchers; participating schools charge extra fees for a year before allowing students to apply and possibly be admitted to the voucher programme. Most of these schools' students can afford to pay some fees (Afridi, 2018; Ansari, 2020).

### Most countries grant fiscal and tax advantages to non-state providers

Non-state providers enjoy tax exemptions and other fiscal incentives in 173, or 87%, of 200 countries. Pakistan has established five foundations to support the establishment and maintenance of non-state primary and secondary schools through grants, loans, state infrastructure, community facilities and other services. The Philippine government provides grants, subsidies and tax incentives to non-state schools, in the form of vouchers, management contracts, grants, loans and technical assistance. A 1998 act set criteria for receiving support, including tuition fees, school performance and location, and students' socioeconomic needs. The Department of Education annual madrasa fund also subsidizes private madrasas' operational costs.

Such incentives can be costly for governments. One estimate suggests that tax incentives result in forgone government revenue of some US\$1.2 billion a year in Ghana, US\$1.1 billion in Kenya, US\$4 billion in Pakistan and US\$272 million in Uganda (Balsera, 2017). By contrast, the United Republic of Tanzania recently introduced a 30% income tax for private schools whose revenue exceeds a certain tax-free margin (Rossignoli, 2021).

### FEW REGULATIONS AIM TO PROMOTE EQUITABLE ACCESS

Some countries try to prevent selective admission procedures or directly enhance access to non-state schools.

### Some regulations deter, but may not prevent, selective admission

Globally, in 78 of 172 countries, or 55%, schools cannot determine their own admission processes (**Figure 3.6**). This applies to non-state schools in 46 countries, while in 9 countries only for government-aided schools. All countries espouse the principle of non-discrimination to ensure fair access to education, but faith-based schools have more freedom to choose students.



99

### 66

### In Chile, the 2015 inclusion law prohibited selective admission

In Chile, private subsidized schools selected students by socioeconomic characteristics or religious beliefs through interviews, tests or play sessions (Carrasco et al., 2017; Santos and Elacqua, 2016) until the 2015 inclusion law prohibited selective admission. In Peru, private schools are free to establish their own admission process as long as they comply with the guidelines in the 2021 private basic school regulations regarding non-discrimination and prohibition of assessing grade 1 students. In the Bolivarian Republic of Venezuela, private school establishment and operation rules prohibit admission criteria that discriminate against students on the basis of gender, race, religion or economic status.

Ireland's education system was long dominated by the Catholic Church. But increasing ethnic and religious diversity led to the 2018 Education (Admission to Schools) Act requiring all schools to have an admission policy in line with the 2000 Equal Status Act, which means they must explicitly state that they will not discriminate on the basis of gender, civil or family status, sexual orientation, religion, disability, race or special education needs. Admission criteria must also not include whether parents were pupils at the school or when a child's name was placed on the enrolment list (Doyle et al., 2020; Kenny, 2020). At the same time, minority students still have access to schools of their faith. Special statements are permitted for religious schools, which can refuse students if they can prove doing so is 'essential to maintain the ethos of the school' (Ireland Government, 2018). In the Netherlands, private faith-based schools, unlike public schools, can prescribe principles to which teachers and students should adhere, but cannot discriminate on the basis of sexual orientation or gender identity (Netherlands Government, 2021).

In England (United Kingdom), privately managed academies and free schools can control their admission criteria when they are oversubscribed, as long as they comply with the School Admissions Code, the Human Rights Act and the Equality Act. Schools are not allowed to interview parents (Department for Education, 2021). Oversubscribed schools must make their student selection criteria public and parents can appeal (Roberts and Danechi, 2021). Initially, schools could automatically select students who had ranked that school first in their preferences. This policy was banned in 2008 out of concern that it favoured privileged families. However, the policy ban had the opposite effect, giving privileged families a further edge by increasing competition for the most popular schools (Terrier et al., 2021). Typically, the main criterion is distance from school (Burgess et al., 2020). The same is true for schools managed by local governments, although in practice free schools can also select their preferred feeder schools and catchment areas (Morris, 2014). Selection mechanisms based on distance from school have a strong effect on the housing market. An increase in test scores by one standard deviation raises house prices by 3% (Black and Machin, 2011).

Namibia's 2020 Basic Education Act prohibits private school admission policies from using race, ethnicity, religion, sex or socioeconomic status as criteria. Schools that violate the rule can be fined up to US\$1,350 and the owner can be imprisoned for up to two years. In Seychelles, which has four primary and three secondary private schools, the 2004 Education Act allows the minister to determine the maximum number of students admitted, and the principal secretary can launch an inquiry if a private school is accused of discriminating on the basis of race, religion or political affiliation by expelling or refusing to admit a student.

### Some countries support disadvantaged learners' access to non-state schools

Globally, one third of countries subsidize tuition fees and 40% subsidize other costs, such as transport and meals, to support students from particular backgrounds (**Figure 3.7**). Just 7% of countries (but 27% in Central and Southern Asia) have quotas for disadvantaged students, and 4% of countries have regulations against gender-based exclusion.

Argentina's Progresar programme supports young people who are unemployed or have a very low income, as well as other disadvantaged groups, through benefits such as scholarships to complete compulsory education. In 2020, the programme was extended to include government-dependent private schools. Mexico's 2019 General Education Law states that all private schools must grant scholarships to at least 5% of students enrolled. In Peru, private schools must give scholarships to students who lose their parents or guardians and can prove they cannot pay school fees.

In India, the 2009 Right to Education Act mandated private schools to reserve 25% of places for children from poor and disadvantaged groups and to reimburse their fees (**Box 3.5**). Nepal's 2018 Free and Compulsory

#### FIGURE 3.7:

**One in three countries helps disadvantaged students pay private school tuition fees** Countries with regulations supporting disadvantaged students' access to private schools, by region, 2021



### In India, the ambitious equity objectives of the Right to Education Act provisions on private schools are not yet met

Section 12(1)(c) of the 2009 Right to Education (RTE) Act stated that non-minority private unaided schools in India should reserve at least 25% of seats in entry-level grades for children from poor and disadvantaged backgrounds. State governments were to reimburse schools either the non-state school fees or the state school per child spending, whichever is lower.

The policy ran into several implementation problems (Day Ashley et al., 2020). Between 2010 and 2016, about one quarter of litigation under the act in High Courts and the Supreme Court concerned this provision (Ambast and Gaur, 2017). State education departments have not had enough staff to inform applicants, process applications, monitor private school compliance with infrastructure and teacher norms, and deal with appeals (Mehendale et al., 2015).

Schools have complained that reimbursements are slow, difficult to obtain and below expected levels, leading them to raise fees for the other students. It has also been argued that state governments have underestimated the per-student cost in public schools and not made their calculations public (Kingdon and Muzammil, 2018).

Not only is there lack of awareness with respect to procedures, but also application procedures and eligibility proof have been too complicated, sometimes giving rise to corruption claims (Mehendale et al., 2015). In many states, the admission process takes place through the computerized RTE lottery, which requires successful applicants to approach the schools designated for admission within one or two weeks. Despite trying to reduce schools' discretion in selecting applicants, the system presents challenges for families as it requires computer proficiency and access to internet (Wad et al., 2017).

By 2021, only 16 of the 36 states and territories were implementing the measure, according to the National Commission for Protection of Child Rights (Sharma, 2021). Maharashtra had 69% of reserved seats filled in 2020, the highest percentage, but only 41% of private unaided schools had registered for RTE admissions. The COVID-19 crisis exacerbated some challenges, with schools not allowing so-called RTE students to join online classes and opposing an order against fee collection during the lockdown (Sarasvati, 2020). Across India, it was estimated that 4.3 million children benefited in 2018/19, although Karnataka, Madhya Pradesh, Rajasthan and Tamil Nadu accounted for 68% of the total. In Uttar Pradesh, no more than 2% of seats were filled (Indus Action, 2018, 2019).

Schools appear to have invested little to ensure integration of children and there are concerns that RTE students may be subject to patronizing attitudes and discrimination (Mehendale et al., 2015). However, there are indications of success in social cohesion (Joshi, 2020) and the measure is generally perceived to have been positive, at least in low-fee if not elite schools. Beneficiary children, while not among the poorest, appear to have ended up in schools they might otherwise not have had access to (Dongre et al., 2019). But concerns remain whether they will have the support they are likely to need in higher grades.

Education Act required private schools up to grade 12 to offer scholarships to 10% (for schools with up to 500 students), 12% (between 500 and 800 students) or 15% of their students (above 800 students) (Jha, 2019). However, private schools have contested the law and refused to comply with it (Singh, 2019). In Pakistan, the 2017 National Education Policy stated that all for-profit schools must provide free education to at least 10% of their enrolment, favouring 'deserving children'; similar commitments were made in provincial laws in the Islamabad Capital Territory in 2011, Sindh in 2013 and Punjab in 2014. The Philippines has subsidized private school access since the enactment of the 1989 Government Assistance to Students and Teachers in Private Education legislation to support implementation of the 1988 Free Secondary Education Act. Two of its components target students: Education Contracting Services (ESC) and, since 2013, the Senior High School Voucher Program (SHS VP). ESC is a tuition fee subsidy, averaging US\$160 per year, which allows students to move from 'congested' public to private secondary schools. In 2017/18, it supported 970,000 students in 3,300 private schools. The Private Education Assistance Committee, which implements it,

66 -

In the Philippines, a 2019 audit found there were no guidelines in the Education Contracting Services programme for targeting disadvantaged students

99

#### FIGURE 3.8:

**Regulations on quality vary between state and non-state schools in OECD countries** Number of OECD countries with a quality regulation, lower secondary education level, 2018



is responsible for certifying private schools and deciding places per school (Saguin, 2019). A 2019 audit found there were no guidelines for targeting disadvantaged students and no procedures for identifying congested public schools, while performance indicators were inadequate (Philippines Commission on Audit, 2019).

SHS VP provides financial support for grade 10 students wishing to enrol in private schools in grades 11 and 12. All grade 10 public school students are eligible to receive the full voucher amount, while private school students receive 80%. In 2017, more than 1.2 million voucher recipients were enrolled in grades 11 and 12, equivalent to 47% of total and 94% of private school enrolment. As neither the voucher value nor eligibility varies by background, this component of the programme probably also benefits richer students, exacerbating inequality (World Bank, 2020).

### ENSURING QUALITY IN NON-STATE SCHOOLS IS A CORE AIM OF REGULATION

Quality assurance mechanisms and processes for non-state schools focus on education inputs, processes and, to a lesser extent, outputs. In OECD countries, public lower secondary schools are subject to regulations on curriculum, teacher certification, class size, national assessments and examinations somewhat more frequently than non-state schools (**Figure 3.8**).

### Alignment with the national curriculum is a focus of regulation

The curriculum is a key area of regulation due to the key role it plays in shaping national identity. Just 9% of countries had no regulations mandating the use of national curricula in non-state schools, while 15% mandated it only for government-aided schools.

### 66

Two issues that drive a wedge between public and private schools in many countries are religion and the medium of instruction

Chile has made the curriculum for pre-primary, basic and lower secondary education mandatory for all officially recognized schools. Schools must safeguard the principle of non-discrimination and meet national learning standards. The 2007 general education law provides for schools to develop their own plans and programmes as long as they are in line with the baseline curriculum's general goals. Private establishments can set complementary aims, such as a third language or emphasis on arts, science or religion, as long as the Ministry of Education approves. China's 2021 private education law mandates the use of the national curriculum in private schools.

In Australia's New South Wales state, registered private schools must align their education programme with the Education Standards Authority syllabuses and guidelines approved by the Department of Education. Registered schools not accredited to provide a grade 11 or 12 certificate can seek approval to modify one or more syllabus outcomes. In Queensland state, private schools' education programmes must be consistent with the Melbourne Declaration on Educational Goals for Young Australians and implement the national curriculum or one recognized by the Australian Curriculum, Assessment and Reporting Authority.

Two issues that drive a wedge between public and private schools in many countries, notably in Northern Africa and Western Asia, are religion and the medium of instruction. Although Arabic is the official language of instruction in much of the region, English and French are widely used in private schools. Most countries mandate the teaching of Arabic and Islamic studies, although schools can follow a curriculum of their choice that is accredited elsewhere, subject to approval. In Kuwait, non-state schools must follow the general education objectives, respect Islamic values and teach the local culture and Arabic. In Saudi Arabia, private schools are required to teach Arabic and Islamic studies even when they follow an international curriculum (Arab News. 2013). The United Arab Emirates makes Islamic studies (for Muslim students), Arabic and social studies (for all students) compulsory in private schools. Oman and Sudan forbid non-state schools from using reading materials that contradict Islamic values (Al Qasimi Foundation for Policy Research, 2021).

In Lebanon, the national curriculum is delivered in Arabic, with the first foreign language (French or English) introduced from grade 1 and the second from grade 7. Yet in many private schools, the second foreign language starts in the primary grades. After the primary grades, the curriculum is taught in both Arabic (humanities and social sciences) and the first foreign language (mathematics and sciences) (Zakharia, 2016). In Morocco, Arabic is the official language of instruction, but expensive and selective bilingual private schools (teaching in French and English or Spanish) have emerged to provide a comparative advantage for the well-off in access to tertiary education and more lucrative labour market opportunities (CMEPT, 2014).

Fiji's non-state schools may follow their own curriculum, subject to ministerial approval, but they must teach specific subjects relating to health, civic education and issues of national interest. In addition, primary schools must teach iTaukei and Fiji Hindi as compulsory subjects. In Pakistan's Sindh province, non-state schools must teach Sindhi and the curriculum 'shall be at least, at par with the curriculum approved by Government' (Sindh Government, 2002, p. 4).

Religion is treated in various ways in education systems, and regulating faith-based schools is often challenging when it comes to issues including curriculum, safety and admissions (Box 3.6). Some countries emphasize secular education in both state and non-state schools. In France, which adheres to the principle of laïcité, religion is not taught as a school subject. Fiji's 1978 Education Act allows schools to provide non-compulsory religious instruction, but most subjects must remain secular or the institution runs a risk of not being recognized as a school. By contrast, in Tuvalu, the minister approves the curriculum of the two non-government schools after consulting with the Educational Advisory Committee; the director of education may cancel their registration if the curriculum is 'persistently and materially departed from or religious instruction is not provided as part of the curriculum'.

### Regulations also address teacher certification and working conditions

Practically all countries (93%) have regulations on teacher certification or training for primary and secondary school teachers in non-state schools. In the past decade, regulations have been adopted or amended in 80 countries. About 53% of countries have regulations on non-state school teacher accreditation, with regional shares ranging from 30% in Northern Africa and Western Asia to 82% in Eastern and South-eastern Asia.

### 66 -

### About 53% of countries have regulations on non-state school teacher accreditation

**?**?

#### Maintaining faith-based school autonomy and accountability is often a challenge

Faith-based schools, a core part of education in many countries, can be controversial. Opposition to state funding is often strong, as faith-based schools may impose particular religious teachings and be seen as raising barriers between groups, especially through selective admission. Another concern is that religious schools may be less open to scrutiny than others, especially when they do not receive public funding (Maussen and Bader, 2015). Examples from around the world suggest that checks and balances are sometimes insufficient.

Selective admission policies based on religious beliefs are common in faith schools. In England (United Kingdom), such schools can rank applicants by their observed religiosity using complex criteria (Burgess et al., 2020). The Department for Education has published warning notices, for instance, for an Islamic school that had separate entrances for boys and girls and segregated children by sex for all classes and activities (Lowrie, 2017) and a Jewish nursery school that also separated children by sex, reinforcing gender stereotypes (National Secular Society, 2019). In the US state of North Carolina, a Catholic school's admission policies explicitly excluded families adhering to non-Christian faiths or behaving in ways seen as 'deviate and perverted' (Kahlenberg, 2020).

Concerns are often raised about faith-based schools with respect to child protection. Lawsuits for sexual abuse and corporal punishment have been lodged in Argentina and Chile (Bishop Accountability, 2021); in France, where a traditionalist Catholic school was closed after allegations of child abuse (RFI, 2017); the United Kingdom (IICSA, 2019); and the United States (ProPublica, 2021). Accounts of corporal punishment and sex abuse have been reported in Myanmar's monastic Buddhist schools, which enrol at least 300,000 children and where clerical immunity impedes investigation (Phillips, 2018). Senegal's Koranic schools, or *daaras*, remain outside the national education system despite attempts at integration (RFI, 2019). It is estimated that 17 students died from 'violence, acts of neglect or endangerment' in 2017–20, leading to calls for teachers to break silence on the issue (Senghor, 2020). Cases of teachers forcing students to beg and subjecting them to severe abuse and neglect were highlighted (Human Rights Watch, 2019).

Corporal punishment is a major concern in South Asia. In Bhutan, a 2010 Ministry of Education assessment found that 11 of 126 monastic schools used punishment such as beating, spanking and whipping. In India, a 2007 Ministry of Women and Child Development survey of children in 11 states found that at least half the girls had experienced some form of violence, including in private and religious schools (UNICEF, 2016). In Pakistan, sexual abuse in schools and madrasas has been a long-standing and pervasive issue (Zafar, 2020), with reports of police being bribed to not pursue justice against clerics (Gannon, 2017).

Tensions often emerge between national curriculum requirements and faith-based schools, particularly in subjects such as religion, science and sexuality education. In England, a report on over 600 faith-based state secondary schools (Catholic, Church of England, Muslim and Jewish) found that in two thirds, sex and relationship education was taught according to religious principles, including that sex outside a religious marriage was wrong, that divorce was not recognized and that homosexuality was wrong (National Secular Society, 2018). In Birmingham, England, a textbook from an unregistered Islamic school advocated killing homosexuals (Titheradge, 2018). In the United States, ultra-Orthodox Jewish yeshivas (private religious schools) provided rudimentary instruction in mathematics, science and English for 90 minutes at the end of the school day, with classes sometimes cancelled on religious occasions (Hassan, 2021; New York Times, 2021).

Religious schools are also scrutinized due to concern that some spread religious hatred and radicalize youth. In Kenya, institutions linked to extremist organizations were found to have used madrasas to disseminate radical jihadist ideology (Mkutu and Opondo, 2019). In Pakistan, a survey of students found that those in madrasas held more extremist views than those in secular schools (Hanif et al., 2021). A survey of madrasas students in Punjab province found that 45% agreed that madrasas took part in political activities and 42% that they were involved in extremist activities (Qadri, 2018).

In Burkina Faso, non-state school teachers must have the required diploma or capacity qualification. In addition, they must receive continuous, certified training. Permission to lead or teach is granted by the regional basic education and literacy director. But in several other sub-Saharan African countries, many private school teachers tend not to receive professional training (Lange et al., 2021). In Guinea, private schools are less demanding about staff qualifications than public schools (Somparé and Somparé, 2018). Between 80% and 90% of Bridge teachers in Uganda were unlicensed (Education International, 2017).

In Gulf Cooperation Council countries, to be eligible for teaching positions teachers must meet ministries' licence and certification criteria. In Bahrain and the United Arab Emirates, a bachelor's degree in education is required for primary school teachers

99

and a degree in the subject they are teaching, with a minor in education, for secondary school teachers. Bahrain's education ministry requires teachers to pass subject proficiency exams. The United Arab Emirates education ministry recently introduced licensing requirements, indicating a move towards greater standardization of teacher training and qualifications (Al Qasimi Foundation for Policy Research, 2021).

About two thirds of countries have regulations on non-state school teacher salaries. But only in 28% of countries are non-state teachers covered under similar laws with state school teachers in terms of working conditions. Only in 15 countries do the regulations concern all types of non-state schools. In 23 countries, only teachers working in government-aided schools get equivalent treatment.

In Chile, the 2016 law that introduced the national teacher policy provided for more rigorous entry requirements, regular evaluations and government-supported professional development (Mizala and Schneider, 2019). Despite strong opposition from the private school sector, the final version of the law covered all teachers in both public and private subsidized schools (Treviño et al., 2018) abolishing differences by contract type (public vs private law) under which they had been recruited. But in El Salvador and Guatemala, private school teachers are under the private work law and are not considered civil servants. In the Philippines, non-state schools recruit their own teachers, who must meet a 2011 manual's gualifications and standards. Teacher employment conditions in non-state schools are regulated through guidelines on salary scales, teaching hours and professional development. Unlike teachers in state schools, however, non-state teachers do not have to be trained in the national curriculum and have more freedom regarding their teaching methods.

Francophone sub-Saharan African countries have a variety of teacher working arrangements, but four contract types are generally most common, as in Togo: civil servant, private, temporary, and voluntary or community. Their conditions vary in terms of job security, salary scale, social security and pensions, family benefits and access to continuing professional development. Voluntary or community teachers tend to have salaries below the minimum wage and no social security or medical insurance. The number of private school contract teachers has been increasing; such teachers tend to be less qualified, have less teaching experience and be less trained than their public school peers (Lange et al., 2021).

### 66 -

A survey of 1,000 schools in five sub-Saharan African cities found that 56% were unregistered, and one third of those had been operating for at least five years

The Ethiopian government favours granting the right to pensions to all contract teachers in both private and public schools, although private schools have autonomy regarding salaries, which vary widely. In Sudan, 50% of the teachers in private schools are on contract. In the United Republic of Tanzania, independent private schools can set teacher salaries and deploy and dismiss teachers, as outlined in the Education and Training Policy. In Uganda, private schools prefer to hire men rather than women. All teachers working in private schools are on contract, most often verbal (International Task Force on Teachers for Education 2030, 2020).

The COVID-19 crisis heavily influenced private-sector teachers' working conditions. Following school closures and non-payment of school fees, school owners struggled to pay teachers. In India's Karnataka state, the Associated Management of Primary and Secondary Schools revealed that as of September 2020, 50% of teachers had been fired and the remaining 50% were on half salary (Shrinivasa, 2020). Similar issues were found for Catholic schools in Burkina Faso, Cameroon, Côte d'Ivoire, Senegal and Togo, which faced severe financial constraints leading to serious difficulty in ensuring that teachers were paid (Lange et al., 2021; Wodon, 2020).

### WEAK IMPLEMENTATION OFTEN UNDERMINES REGULATION

Despite the extensive presence of regulations, they often remain largely theoretical, with implementation lagging. Effective complaint and redress mechanisms may not exist or be functional.

### MANY NON-STATE SCHOOLS REMAIN UNREGISTERED

As school registration in poor countries is often cumbersome, slow and corruption-prone, many schools operate without a licence and without meeting minimum conditions. An audit of four districts in India's Jharkhand state found that 352 of 547 private schools in operation had applied for recognition in 2013–15, but only 101 met requirements. In mid-2016 their proposals were caught in a bottleneck at the Directorate of Primary Education. In Gujarat state, four years after the 2012 State Right to Education Rules took effect, more than 2,000 independent private schools were operating without a certificate of recognition (India Comptroller and Auditor General, 2017).

The situation is similar in sub-Saharan Africa. As more stringent establishment regulations are introduced, many informal schools remain unregistered or unrecognized, with governments lacking capacity to enforce the rules and resources to staff inspectorates (Baum et al., 2018). A survey of 1,000 schools in five cities found that 56% were unregistered, and one third of those had been operating for at least five years (CapPlus, 2017). In Burkina Faso, government inspection visits in 13 regions revealed that 649 schools did not meet minimum registration criteria. Of those, 65% were deemed 'unrecognized, but recoverable' with ministry support; the other 35% were 'unrecognized and irrecoverable' and ordered to close (Ouédraogo, 2018).

Nigeria faces considerable challenges with registration. In Lagos state, the number of new private schools registered more than doubled between 2019 and 2020, from 729 to 1,660. But as of 2021, the government had approved just 1 in 4 of about 20,000 private schools in the state (Premium Times, 2021). In a survey of private school head teachers, 62% disagreed or strongly disagreed that the application process was easy, and the share was 57% on a question about most requirements being achievable for all types of schools (Binci et al., 2016). In the Ajeromi-Ifelodun local government area, less than 1% of schools paid the required fees, and 74% of unregistered schools cited lack of purpose-built buildings, inadequate infrastructure and failure to own their land as reasons for their unregistered status (Baum et al., 2018). About 40% of surveyed schools in the Makoko slum said they would remain unregistered, mainly because it was impossible for them to meet the regulations (Härmä and Adefisayo, 2013). In Abuja, Federal Capital Territory, 66% of non-state schools were unregistered in 2016 (Härmä, 2019).

While unregistered schools tend to operate outside the oversight of government authorities and are left out of official administrative data, this report found that at least 27 countries recorded such schools in their statistics. Burkina Faso's education ministry publishes an annual list of unauthorized schools and publicizes it through its website, the news media and Facebook (Lange et al., 2021). Uganda's Ministry of Education and Sports categorizes non-state schools by registration status as licensed, registered or unregistered. In 2017, 14% of primary and 13% of secondary schools were documented as unregistered.

Responses have varied, with some countries relaxing registration requirements and assisting schools to meet them, and others ordering unregistered schools to close immediately. In Ghana, policy implementation was flexible, with schools applying for registration after starting operations and attracting students. But at the same time, schools reported that government officials took bribes to overlook serious deficiencies during the application process and routine inspections. About 67% of schools reported having been inspected in 2017 (Härmä, 2019). In Uganda, non-state schools are rarely shut for not meeting standards. Government officials help them reach minimum standards and develop a school improvement plan. In addition, minimum land requirements are assessed flexibly, and providers are given extra time to comply with any remaining standards.

By contrast, in Kenya, following a fatal accident in an unregistered school, the government ordered all unregistered schools closed without distinguishing between those that were already halfway through the government registration process and those that had not engaged at all, even with the Teachers Service Commission (Cheruiyot, 2019). In Johannesburg, South Africa, two illegal schools were shut down in 2020 for not having registered with the Gauteng Education Department. In addition, some teachers lacked legal identification documents and the South African Council for Educators documents required to teach (Maphanga, 2020).

Yet countries often lack the capacity to enforce closures, as in Nigeria. In Abuja, the Department of Quality Assurance had the authority to close 550 schools for operating illegally but not the resources to carry out the closures and offer other provision (Härmä, 2019). Still, in 2021, the Osun state government shut more than 600 nursery and primary schools for failing to meet minimum standards, declaring 'war on mushroom schools' and 'zero tolerance for quacks in the system and schools that do not meet the required standard'. Authorities said anyone wishing to establish a new school would need a Teacher's Registration Council of Nigeria certificate (Premium Times, 2021).

### QUALITY ASSURANCE AND SANCTION MECHANISMS AIM TO RAISE STANDARDS

Globally, all countries mandate inspection of primary and secondary non-state schools, although only in 81% does this obligation concern all types of non-state schools. Countries often determine the inspection content and standards schools should meet, as well as potential action, from warnings to sanctions. The most extreme form of sanction, school closure or licence withdrawal, is also the most common, available in practically all countries. In 54% of countries, regulations also determine closures' duration.

In Burkina Faso, a 2004 order allows the education ministry, on advice from the Permanent Commission on Private Education, to close private basic schools that do not meet specifications. Private schools are regularly shut down, with the ministry usually citing conditions such as overcrowded, unsanitary or dangerous premises. Even though teachers do not always have written contracts, are not paid according to official salary scales and are not declared in the social security system, these are not cited among closure reasons (Lange e al., 2021). By contrast, Mauritania has no regulation requiring government-dependent private schools to undergo inspection. Should such schools be inspected, no sanctions based on inspection results can be administered, and they are also exempt from sanctions based on examination results (World Bank, 2016).

In Northern Africa and Western Asia, countries including Algeria, Libya and Iraq require non-state schools' students to take state-level secondary school examinations. That is also the case in Sudan, although it generally exempts international schools from this requirement as the curriculum may require other examinations. In the United Arab Emirates, which has a sizeable non-state education sector, the secondary school completion certificate in non-Arab private schools is equivalent to the national certificate 'in accordance with the conditions specified by the Ministry' (United Arab Emirates Cabinet, 2008).

In Bhutan, private schools must assess and review their own performance. Most schools are subject to annual ministry assessments as well, and students sit national examinations. If the ministry receives notice that a private school has failed to comply with any guidelines, it investigates the concerns, gives the school a written warning if it is found to be non-compliant, and may recommend suspension or termination of the licence and eventual school closure.

India's 2009 Right to Education Act states that schools cannot be established or recognized unless they fulfil norms and standards in an annex to the act. If non-compliance exceeds three years, the authorities must withdraw recognition, whereupon the owners are liable for substantial fines if they continue to run the school. In Karnataka state, the education minister threatened to close over 1,400 private schools not in compliance with the act (Deccan Chronicle, 2014). In 2016, 1,170 private schools were shut down in Punjab state. The National Independent School Alliance, which represents 55,400 private schools across India, reported that more than 15,000 schools were given closure notifications for non-compliance (Iyer and Counihan, 2018).

In the Philippines, private accreditation has been in effect since the 1970s to address concerns about government regulation. In the past, government could limit the entry of private schools to protect public schools from competition. Recognized non-state schools can voluntarily apply for accreditation, a form of recognized quality certification, under the Federation of Accrediting Agencies of the Philippines. If granted, it can give the school increased autonomy, deregulated status (with schools not subject to periodic evaluations) and state financial aid. The degree of autonomy of private schools, compared with government standards, depends on the level of accreditation they receive. However, accountability issues remained between the federation and the Department of Education (Rossignoli, 2021).

All non-state schools in Fiji are subject to external evaluation by officers authorized by the permanent secretary; they may enter and inspect at any time with or without notice. The 2014 Schools Standard Monitoring and Inspection Policy requires schools also to submit self-assessment reviews to the district education officer, who must verify them at least once every three years. Schools not complying following an official warning may have their registration certificate cancelled or be ordered closed by the permanent secretary. In Georgia, accreditation by the National Centre for Educational Quality Enhancement is voluntary, but required for schools to be considered for the government's voucher programme. Authorized and accredited schools are entitled to issue certificates.

In Bogotá's Concessionary Schools, the charter school experiment in Colombia, contracts are not renewed if schools do not meet learning outcome standards. Of the 25 charter schools that started operating in 2000 under 15-year contracts, 3 had their contracts not renewed after the government's programme evaluation in 2014; they then came under direct public management (Edwards Jr and Termes, 2018). Ecuador's National Educational Authority defines education guality standards and indicators to be used for evaluations conducted by the National Institute of Educational Assessment, related student academic performance, alignment with the mandatory national curriculum, and teacher and manager performance. In Peru, although supervision of private schools is a state responsibility, since 2014 education decentralization has challenged the effectiveness of private provider supervision, as local ministry units lack capacity and resources (Balarin, 2015).

Despite such regulations' validity, sudden closures can put many children's education at risk, as was seen in countries where financial difficulties caused by the COVID-19 pandemic led to closures. With some students dropping out and others transferring to public institutions, private institutions faced a reduction in tuition revenue (Wodon, 2021b). In India's Punjab state, the education department warned 38 private schools that charged fees during the lockdown, despite having been ordered not to do so, that their licence would be withdrawn if they did not cancel the fees (Sood and Prakash, 2020).

### TEST-BASED ACCOUNTABILITY MAY FAIL IN ITS AIMS

Examination results are one accountability mechanism used to ensure that non-state schools meet standards. But few systems have the capacity to use this mechanism effectively, and test-based accountability is generally plagued by conceptual and practical problems. In Burundi, all schools are subject to inspection. In addition, private secondary school students have the right to sit national examinations after grades 6 and 10. Private schools with a pass rate of less than 30% at the end of each school year may be closed; in 2016/17, private schools with results below 20% were closed (Agence Bujumbura News, 2017).

In Chile, the 2008 Preferential School Subsidy Law granted extra funding to schools that enrolled disadvantaged students, increasing the voucher's value if schools adopted tighter accountability measures based on ranking schools and sharing the information publicly. Underperforming schools lose autonomy and must contract with external school improvement services. If they do not move to a higher performance category within three years, the ministry can ask families to consider other schooling options. If the situation remains the same for a fourth year, the ministry revokes the licence and withdraws public funding (Elacqua et al., 2018). There is evidence that these provisions led to strategic behaviour, especially among underperforming schools, such as teaching to the test and assigning the best teachers to students who were to be externally evaluated (Elacqua et al., 2019; Falabella, 2020).

In Pakistan, financial assistance under the Punjab Education Foundation's programme depends on the school's performance on the annual quality assurance test to hold participating schools accountable. Funding is automatically discontinued if the school fails the test twice in a row (Afridi, 2018; Ansari, 2020; Barrera-Osorio and Raju, 2015). In practice, however, the competitive pressures of this test-based funding model have been shown to work against equity. In interviews, providers said the pressure to perform well on the test and minimize the risk of losing government funding led to selecting students exclusively based on academic ability as determined by screening in admissions. About 90% of interviewed providers identified the test as one of the biggest issues limiting access of out-of-school children (Afridi, 2018).

The US states of Indiana and Louisiana are among 23 states offering private school choice (Cunningham, 2014; Olneck-Brown, 2020). In their voucher programmes,

66 -

In Burundi, private schools with national examinations results below 20% were closed in 2016/17

#### FIGURE 3.9:

#### One in two countries does not regulate private supplementary tuition

Percentage of countries by distribution of private supplementary tuition regulation, 2021



GEM StatLink: https://bit.ly/GEM2021\_fig3\_9 Source: GEM Report team analysis based on the PEER country profiles on non-state actors in education.

schools must reach a minimum score on a student learning assessment to be able to admit new 'choice' students the following year (Cunningham, 2014). School closures are frequently used to raise district-level performance: 1,522 low-performing schools were closed in 26 states from 2006 to 2013, of which 318 were charter schools and 1,204 were traditional public schools (Han et al., 2020).

### PRIVATE TUTORING REGULATIONS TEND TO BE UNDERDEVELOPED

Private supplementary tuition is an ancillary service that has grown exponentially in recent years. Tutoring services can circumvent regulation. In many countries, roles and responsibilities are fragmented and opaque, and there is limited awareness that some practices are illegal. Analysis of country profiles for this report found private tutoring to be unregulated in 81, or 48%, of 169 countries. In 53 countries, regulations are within education legislation; 16 of those also include it in commercial regulation, while 19 other countries regulate private tutoring only under commercial laws (**Figure 3.9**). Regulation of the sector, on the whole, is limited (Zhang, 2020).

In China, more than 75% of primary and secondary school students attended after-school tutoring sessions in 2016. In 2021, the government introduced comprehensive legislation to limit a sector that was expected to grow from US\$120 billion in 2019 to US\$155 billion by 2025 (Zhu and Yang, 2021). The new law bans profit making by firms teaching compulsory curricula, such as mathematics, science and history, and prevents new licences from being issued (Bloomberg News, 2021). Profit-making companies providing tutoring have to change their structure and become non-profit to continue operations. The government also created a department exclusively to regulate and monitor Chinese private tutoring companies. Just before the law came into effect, 15 firms, including Zuoyebang and Tencent, were fined a total of US\$5.7 million after allegations of fraud concerning teacher qualifications and inflation of costs (Clarke, 2021; France 24, 2021).

In Denmark, tutoring companies are registered as corporations with the Danish Business Authority and additional registration is required as a child-related business. By contrast, the Russian Federation requires tutoring institutions to have an education licence. The education law distinguishes between individual entrepreneurs providing direct educational services and companies with staff carrying out educational activities. Those in the first category (i.e. private tutors) are not required to license their services but must register with tax authorities. The law forbids individuals who are not allowed to teach school, or suspended from teaching, to provide educational activities as individual entrepreneurs. In Ukraine, the law sets conditions for the exercise, management and attendance of out-of-school and extracurricular education, including citizens' rights to receive such education, the basic principles of relevant state policy and the creation of regulatory frameworks concerning supplementary education organizations (PEER country profiles).

66

In China, a new law bans profit making by firms teaching compulsory curricula, such as mathematics, science and history

99

Regulations cover other aspects of tutoring as well. The Republic of Korea takes into consideration the number of people receiving tutoring and requires registration only for companies, not for the self-employed, though the latter need to meet certain requirements. Other regulations concern the premises where tutoring is provided (e.g. in Ethiopia, the Russian Federation and Uzbekistan) or the times when it is carried out (e.g. in the Republic of Korea and the Russian Federation). In Kazakhstan and Uzbekistan, tutoring institutions must create collegial bodies bringing tutors together for professional development and discussion on curriculum and pedagogy. Some regulations also consider the content of the curriculum delivered (e.g. Kazakhstan and Pakistan).

Regulations on tutors' gualifications are less common. Analysis of country profiles for this report suggests they exist in 28 of 91 countries. Malaysia requires tutors to have a teaching permit and their managers to have at least three years of teaching experience or at least six months in a related field. Similar requirements exist in the Republic of Korea. The Russian Federation and Ukraine require tutors to meet specific standards. In some countries where the market for private tutoring has been expanding, governments have tried to introduce systems to monitor and ensure instruction quality using, for instance, online registers and platforms, as in China and the Republic of Korea (Zhang, 2020). The Australian Tutoring Association's code of conduct provides guidelines on advertising standards, qualifications, consumer information, refund policies and business ethics (Australian Tutoring Association, 2021). The Japan Juku Association publishes voluntary standards and child and data protection guidelines for use by tutoring companies (Bray and Kwo, 2014). The United Kingdom, The Tutors' Association was established in 2013 to professionalize the burgeoning industry with a code of ethics for tutors and a code of practice for companies (Tutors' Association, 2021).

On the controversial issue of whether serving public school teachers should be allowed to provide supplementary tutoring, four approaches have been identified to prevent corruption and poor teaching in classrooms: prohibition, discouragement, permission if approved, and laissez faire (Bray and Kwo, 2014; Zhang, 2020). In Ukraine, where there is no explicit prohibition of public school teachers tutoring their pupils (OECD, 2017), 36% of students reported receiving private supplementary tutoring by their schoolteachers in 2016 (Civil Network Opora, 2016). In Mauritius, despite repeated attempts to address the issue, a report called private tutoring a parallel education system with 'a turnover of billions of rupees' (Think Mauritius, 2019, p. 42). Analysis of PEER country profiles shows that 10 countries explicitly ban all teachers from being tutors; 11 other countries ban public school teachers from private tutoring and 10 countries also ban teachers, but the type of teacher remains unclear. In India, steps to ban government teachers from doing private tutoring started in 1996. The High Court in Tripura state banned all teachers in government-dependent and private schools from tutoring in 2015 (Barman, 2020). The Russian Federation's 2012 education law, amended in 2020, forbids teachers from providing paid education services to pupils of the same school 'if this leads to a conflict of interest', which may occur when teachers' interest in material benefits negatively affects their primary work. Thus teachers in some cases can tutor their own students for a fee.

### CONCLUSION

Ensuring inclusive, equitable and good-quality education requires all actors to make a concerted effort to meet their responsibilities. While accountability starts with governments, it is important to recognize the multiplicity and interdependency of actors in education.

How responsibilities in education are shared varies greatly among countries. Improving responsibility sharing can lead to more efficient and equitable education systems. Engaging non-state actors in education has advantages but requires sound governance and regulatory frameworks. Yet in many contexts, regulation is not developed. In others, even when regulations are in place, government has insufficient financial and human resources to ensure compliance. Corruption and bribery take hold, leaving the most vulnerable exposed to the risk of unaffordable and low-quality education. Registration processes that are complex, unaffordable and long can hamper licensing and approval and fuel the operation of unrecognized schools.

Ultimately, actors depend on each other to reach shared education goals. Meeting those goals requires collaboration and communication. Governments need to view the education system as a single entity, ensuring that standards are set and applied in both state and non-state schools and that all education actors are held accountable for their activities and results. Effective regulatory frameworks that support compliance and ensure effective complaint and redress mechanisms exist and can be replicated. This is necessary to ensure that every student learns.

A boy waves goodbye to his parents as he leaves for school, Indonesia.

CREDIT: Robbi Akbari Kamaruddin/Alamy Stock Photo



### CHAPTER



# Finance



### **KEY MESSAGES**

#### Governments fund non-state actors in various ways.

- Governments may support none, some or all non-state providers. In Canada, government covers 30% of
  private but 94% of public school expenditure. Sweden covers all expenditure in both private and public schools.
- Countries vary in the costs they subsidize. In Bangladesh, the government pays teacher salaries for non-state secondary schools and madrasas, which account for 96% of enrolment. In India, 6% of primary and secondary schools are 'government aided', receiving grants for teacher salaries, but only six states cover expenses beyond tuition reimbursement.
- Some governments tax private schools, while others facilitate school choice with financing mechanisms. Chile has had a voucher policy since 1981. In 2008, schools received additional funding based on their share of marginalized students. In 2015, Chile regulated admission processes and eliminated fee charging for schools receiving subsidies.

#### Households cover a large share of education financing, particularly in the poorest countries.

- An estimated one third of household education expenditure in low- and middle-income countries comes from households with children in public schools. In 15 countries, 39% of education expenditure by households with children in state schools was for uniforms and supplies.
- Household education spending is unequal. While households in the poorest 20% spent practically nothing on education in Argentina, Costa Rica, the Philippines and Zambia, the richest 20% spent between 0.5% and 1.7% of GDP.
- Additional fees are charged in public schools in high-income countries too. According to the 2018 PISA, public schools took obligatory financial contributions from parents in 7 of 59 countries and voluntary financial contributions in 38 of 59 countries.
- Households are increasingly spending on private tutoring. In the Netherlands, household spending on tutoring increased by 160% between 2005 and 2016. In Myanmar, tutoring represented 42% of total household education spending.

#### Private providers usually rely on household out-of-pocket expenditure.

- Most private secondary schools receive at least 80% of their revenue from fees in 28 of 51 upper-middle- and high-income education systems.
- COVID-19 challenged private schools that rely on school fees. In Panama, up to 40% of parents could not pay monthly fee instalments. In Ecuador, public school enrolment was up by 6.5%, or 120,000 students, at the beginning of the 2020 school year.
- About one in six families saves to pay school fees; around 8% of households also borrow, rising above 30% in Haiti, Kenya, the Philippines and Uganda.

#### Donor strategies envision a strong private-sector role but are cautious on for-profit providers.

- The Global Partnership for Education, International Finance Corporation and European Parliament have recently declared they would not support for-profit actors.
- The World Bank and Asian Development Bank support blended financing and public-private partnerships, although evidence of their benefits in education is unclear.

Governments finance non-state providers directly and indirectly	96
Households face significant burdens and tough choices	100
Donors are cautious about supporting for-profit education providers	108
The role of philanthropic and corporate activities in education is evolving	111
Conclusion	.114

Any countries are grappling with whether to offer public funds to non-state education providers and whether such funding compromises equity in education. Achievement of Sustainable Development Goal 4 relates to how much money households pay for education and whether it is directly out of pocket or indirectly through taxation. As philanthropic actors gain influence, meanwhile, their methods and motives are being examined.

This chapter analyses government's role in funding private education institutions through various mechanisms, including public-private partnerships (PPPs), along with the effect of COVID-19 on non-state education financing. It analyses household contributions and their consequences for inequality. Finally, donor and philanthropic financing of non-state provision and efforts to leverage private finance are discussed.

### GOVERNMENTS FINANCE NON-STATE PROVIDERS DIRECTLY AND INDIRECTLY

Governments vary in their decisions whether to fund non-state education providers. They may or may not distinguish between state and non-state schools in their finance allocations. They may or may not subsidize private schools. They may contract out one or more aspects of education management. They may or may not finance students to attend the public or private schools of their choice (Patrinos et al., 2009).

Governments typically fund privately managed schools less, or less often, than publicly managed schools. In 2017, Organisation for Economic Co-operation and Development (OECD) countries spent US\$9,515 per student in public primary and secondary schools, on average, but US\$6,064 per student in private schools. However, there is substantial variation: In Canada, government covers 30% of private but 94% of public school expenditure, while Sweden covers all expenditure in both private and public schools (OECD, 2020a).

No students attended private schools that received at least 80% of their funds from government in 29 of the 68 education systems that took part in the 2018 Programme for International Student Assessment (PISA). In Germany, no students in private schools but 96% of students in public schools attended institutions where government provided at least 80% of funding. All students in Finland attended public or private schools where the government provided at least 80% of the funds (**Figure 4.1**).

Among education systems that have participated in multiple rounds of PISA, only a few, including those of Peru, Qatar and the United Arab Emirates, saw the enrolment share increase in independent private schools, which receive less than 50% of their funding from government. More countries experienced increases in the enrolment share of dependent private schools, which receive at least 50% of their funding from government. In Chile, Hungary, Sweden and the United Kingdom, enrolment in dependent private schools grew faster than enrolment in public schools. In Indonesia and the Republic of Korea, it grew faster than in independent private schools (OECD, 2020b) (**Figure 4.2**).

66 -

Governments typically fund privately managed schools less, or less often, than publicly managed schools

"
#### FIGURE 4.1:

#### Few countries provide substantial financing for privately managed schools

Percentage of students attending public or private schools that receive more than 80% of their funding from government, 2018



Source: GEM Report team analysis of the 2018 PISA database.

#### FIGURE 4.2:

**Enrolment in government-aided private schools has increased substantially in some countries** Enrolment shares in public and independent and dependent private schools, selected countries 2000–18



Note: Data come from school head teacher responses. Dependent private schools receive at least 50% of their funding from government. Independent private schools receive less than 50% of their funding from government.

GEM StatLink: https://bit.ly/GEM2021\_fig4\_2

Source: GEM Report team analysis of the PISA database.

#### BOX 4.1:

# Chile has been gradually changing its school financing policies to improve equity

Chile introduced a universal voucher policy in 1981. The school financing system followed the student and offered the same amount of subsidy per pupil to dependent private schools (53% of students in basic education in 2015) as to public schools administered by municipalities (39%) (Santiago et al., 2017). However, the policy allowed for-profit dependent private schools to charge top-up fees, which led to significant social segregation and excessive profitmaking. In 2013, an estimated US\$500 million of public funding to for-profit schools was retained by school owners instead of being reinvested in education (Stuardo and Cayuela, 2019).

In response to pressure from a major social movement since the mid-2000s, the government has tried to address these consequences. In 2008, the Preferential School Subsidy programme recognized the higher cost of educating students from poor families and increased the voucher amount, and schools received additional funding based on their share of marginalized students. In 2015, the Inclusion Law sought to reduce segregation by regulating admission and selection processes and eliminating fee charging for schools receiving state subsidies, requiring for-profit schools to convert to non-profit status: 97% of them had begun transferring to a non-profit legal entity by 2018 (Stuardo and Cayuela, 2019).

Some evaluations suggest improved equity and effectiveness. Analysis of the 2008 reform found an increase in public schools' and non-fee-charging private schools' test scores (Murnane et al., 2017). Another found that eliminating top-up fees and increasing the voucher value for poorer students raised competition in poor neighbourhoods by giving poorer families a choice of schools that were previously unaffordable to them (Neilson, 2013). But an audit found schools were not using revenue for permitted additional improvement or better instruction quality for disadvantaged students (Feigenberg et al., 2019).

Finally, the 2016 National Teacher Policy incorporated all dependent private school teachers in the civil servant career path at an estimated cost of about 1% of gross domestic product (GDP). Increases in teacher minimum salary levels affected private schools (Mizala and Schneider, 2020).

Governments in many middle- and high-income countries promote school choice through financing mechanisms, with some emphasis on disadvantaged students. Chile has a well-known financing mechanism, which has gradually been reformed in the past 15 years to address the high degree of inequality and segregation brought about by its initial design (Santiago et al., 2017) (**Box 4.1**). Argentina's system of state funding to private education started to offer financial help to private schools serving poor households in 1947 (Narodowski and Moschetti, 2015). In Buenos Aires, 50% of students are enrolled in private schools and 75% of them attend subsidized private schools. These schools can receive as much as 80% to 100% of the cost of teacher and head teacher salaries, and some are authorized to charge low monthly fees (Moschetti and Verger, 2020).

In the Netherlands, the constitution allows public and private schools to receive equivalent public funding. While about one third of students attend public schools and two thirds attend Christian and other dependent private schools, funding levels are related not to school type but to student numbers. Schools receive block grants for staff and operating costs and additional funds for students from disadvantaged socio-economic backgrounds and with special education needs (OECD, 2016).

In India, a minority of private schools had been receiving grant-in-aid and explicit or implicit subsidies since the 1970s. The 2009 Right to Education Act required private schools to offer 25% of grade 1 places to children from low-income families; in exchange, the government reimbursed tuition costs for these families (Sarin et al., 2015).

Certain PPP programmes, aimed at bringing private education under public financing – such as concessional schools in Bogotá, Colombia, the Philippines' Education Service Contracting programme and Uganda's universal secondary education programme – also had the objective of expanding access to private schools for poorer students (Aslam et al., 2017).

# GOVERNMENTS FINANCE ONLY SOME NON-STATE SCHOOL EXPENDITURE

A review by the GEM Report team shows that only 84% of countries have government 'aided' or 'subsidized' private schools, with clearly defined payments for teacher salaries and/or non-salary items, such as textbooks and other learning materials.

In countries where the private sector is the dominant education provider, governments tend to pay teacher salaries, the largest recurrent cost. In Bangladesh, where the handful of public secondary schools account for just 4% of enrolment, over 16,000 non-state secondary schools and 7,600 madrasas receive monthly payments for teacher salaries; an ongoing World Bank-supported programme is aiming to rationalize and make the monthly pay order mechanism more strategic to fill teacher gaps but progress has stalled due to the COVID-19 pandemic (Bangladesh Ministry of Education, 2018; World Bank, 2020). In other countries, salary costs are not covered. In Haiti, non-governmental organizations (NGOs), churches and for-profit operators manage over 85% of primary schools (World Bank, 2018). The 2017–27 education plan specifies that public funding can be provided to non-state institutions to subsidize teacher salaries, especially in low-tuition private schools.

Access to funding is contingent, among other factors, on performance, school registration status and fee levels (Haiti Ministry of National Education and Professional Training, 2018). In practice, however, the system has lacked adequate domestic financing, and donor funds cannot be used for salaries (World Bank, 2018).

The government-aided private sector is much smaller in other countries. In India, 6% of primary and secondary schools – but as much as 22% in Maharashtra, 28% in Meghalaya and 43% in Kerala states – were classified as 'government aided' in 2019/20, receiving grants for teacher salaries (India Ministry of Education, 2021). However, aided schools have gone for years without non-salary grants, putting their financial sustainability in question (Dore, 2013), as they are not allowed to increase fees to recoup costs. Only six states cover expenses beyond tuition reimbursement, such as textbooks, stationery, uniforms and transport. Reimbursement delays can be lengthy: In Jaipur, Rajasthan, reimbursement processing was six months behind (Sarin et al., 2015).

Côte d'Ivoire, which had been subsidizing private schools, began in the late 1990s to link subsidies to teacher qualifications, class size, a fee range and exam performance (Sakellariou and Patrinos, 2009). The number of students in subsidized secondary schools quadrupled between 2010/11 and 2017/18 (Education Partnerships Group, 2019). But the expansion challenged budgets, while the programme did not target equitable access. Moreover, many private schools have poor facilities and face financial problems due to late payment of subsidies and difficulties in raising tuition fees on time, which affects the timely payment of teacher salaries (Koutou and Goi Bi, 2019).

Governments seldom cover non-state schools' capital expenditures related to construction, renovation or equipment. Privately managed schools are not eligible for public capital funding in 7 out of 17 OECD education systems (OECD, 2018c). For others, eligibility and financing varies. In the Czech Republic, capital funding of private schools is left to local authorities' discretion. In Denmark, private schools receive an activity-based capital grant from central authorities. In Sweden, dependent private schools receive capital funding similar to that of public schools. The Flemish Community of Belgium requires private providers to use publicly subsidized facilities for education activities for at least 30 years. They then can opt to sell the building without returning the equity to the government. School facility ownership can thus become a difficult question (Nusche et al., 2015; OECD, 2017, 2018a).

In Indonesia, central, provincial and district governments finance basic education through four mechanisms: direct one-off capital investments and ongoing salary payments; subsidies for public utilities such as electricity; grants for selected operational costs; and scholarships or other assistance to poor families. Public schools receive funds from all these mechanisms. But madrasas and Islamic boarding schools, known as pesantren, accounting for 35% of all private schools and under the jurisdiction of the Ministry of Religious Affairs, have only partial access to funds from the last two mechanisms. They are excluded from provinceand district-level funds, which are decentralized by the Ministry of Education but not the Ministry of Regional Affairs. As a result, madrasas and pesantren raise funds privately and pass on costs to families (Joshi, 2018).

While some countries promote school choice, others treat private schools like companies and tax them. Uganda's government introduced a corporate tax on private primary and secondary schools in the 2014/15 budget. Interviews and focus groups conducted in three districts with proprietors and other stakeholders suggest that the tax led to an increase in school fees (Kasozi-Mulindwa and Okware, 2019).

#### PRIVATE SCHOOL FUNDING HAS BEEN SUBSTANTIALLY AFFECTED BY COVID-19

The COVID-19 crisis affected private schools, especially those relying on school fees. An early synthesis of media reports suggested that private school and contract teachers had lost their jobs or experienced salary cuts in at least 25 countries, including Cameroon, Mozambique, Niger, Viet Nam and Zambia (Carvalho and Hares, 2020).

66

In India, 6% of primary and secondary schools, but 43% in Kerala state, were classified as 'government aided' in 2019/20

"

66

Afghanistan, Canada, Ireland, Pakistan and Panama excluded private schools from additional education funds provided as COVID-19 relief

In Nigeria, most of the 715 low-fee private schools that received funding from Sustainable Education & Enterprise Development (SEED) had not collected fees since school closures began and could not pay rent, thus facing possible eviction (Niazi and Doorly, 2020). In the Dominican Republic, Ethiopia, Ghana, Mozambique, Nigeria, Senegal, Uganda and Zambia, 99% of 657 low-fee private schools reported a significant fall in revenue, 80% on average (Opportunity EduFinance, 2020).

Several countries tried to ensure that private school and contract teachers continued to be paid. Nigeria launched a stimulus package with low-interest loans to pay private school teachers. Côte d'Ivoire agreed to pay more than 10,000 contract teachers who had missed three months of salary (Carvalho and Hares, 2020). In Ghana, private schools received support as part of a generalized programme for small and medium-sized enterprises (Niazi and Doorly, 2020). Viet Nam expanded cash transfer programmes to cover private school teachers. In Rwanda, private school teachers earning below a given threshold were exempted from personal income taxes (IMF, 2021). The office of the United Nations High Commissioner for Refugees ensured that teachers still received pay while schools were closed in Burkina Faso, Chad, Malawi and Mozambique (Carvalho and Hares, 2020). The Ghana National Association of Private Schools successfully advocated for inclusion in support to small and medium-sized enterprises. The Kenya Private Schools Association launched an online platform supporting distance learning for all private schools (Niazi and Doorly, 2020). Other governments, however, including those of Afghanistan, Canada, Ireland, Pakistan and Panama, excluded private schools from additional education funds provided as COVID-19 relief (Carvalho and Hares, 2020).

In some countries, governments focused on how to deal with the large transfer of students from private to public schools during the COVID-19 crisis. In Peru, a survey of 12,000 parents who requested a transfer of their children to public institutions found that one key reason was objection to the quality of private institutions' online education services. The Ministry of Education added more than 100,000 spaces in public institutions and estimated that over US\$19 million would be needed to hire at least 1,700 executives, 1,500 teachers, 2,000 education assistants and 900 administrative staff. It developed an algorithm, based on availability of spaces, to assign students to the closest public school (Alvarez et al., 2021).

# HOUSEHOLDS FACE SIGNIFICANT BURDENS AND TOUGH CHOICES

Households invest resources to ensure that children have access to education of the highest possible quality and gain a competitive advantage that will improve their life chances. The extent of direct household financial contributions is likely related to the extent of government provision, its focus on equity and taxation's redistributive effect. The distribution of household burden, along with households' opportunity to choose, affects the entire education system. Out-of-pocket spending on education may be motivated by low government spending, which forces parents to pay, or by wealth, ambition or peer pressure to pursue advantage for children.

Analysis carried out for this report shows that households account for 16% of education expenditure in high-income countries and 36% in low- and middle-income countries. As a share of GDP, this spending amounts to 1% in high-income and 2.3% in low- and middle-income countries (see **Chapter 21**). In primary and secondary education, it accounts for 1.2% of GDP in El Salvador, 1.5% in Morocco, 1.8% in India and 2.5% in Ghana.

66

Households account for 10% of primary and secondary education expenditure in high-income countries and 23% in low- and middle-income countries 4

99

There are four notable issues. First, not all spending comes from households with children in non-state primary and secondary schools. An estimated one third of household expenditure in low- and middle-income countries comes from households with children in public schools.

While households with children in private schools accounted for about 80% of spending in Guatemala and Pakistan, households with children in public schools accounted for about 60% of spending in China and Kenya (**Figure 4.3**).

Second, household spending covers a wide range of costs, from formal and informal fees to uniforms, textbooks, stationery, transport, private supplementary tuition and boarding. Comparative analysis is difficult because surveys categorize items differently and some are more important in some countries, and some school types, than others, e.g. in 15 low- and middle-income countries, 39% of education expenditure by households with children in state schools was for uniforms and school supplies, compared with 17% for households with children in non-state schools (**Figure 4.4**).

Third, most spending goes to school fees for households with children in non-state schools; items other than fees dominate spending for households with children in state schools. In a sample of 24 low- and middle-income countries analysed for this report, fee payment to schools accounted for 64% of household education expenditure: 45% for households with children in state schools and 71% for households with children in non-state schools. Fees to public schools were negligible in Argentina, Ethiopia and Nigeria but sizeable in Kenya, Uganda and Zambia.

#### FIGURE 4.3:



In 24 low- and middle-income countries, fee payment to schools accounted for 64% of household education expenditure

Fourth, household education spending is unequal, with urban and rich households accounting for large shares. Cities have a longer history of non-state education provision due to higher population density and higher demand for education. Household expenditure as a share of GDP is higher in urban than in rural areas, and a high share of expenditure in urban areas for primary and secondary education goes to non-state providers; in Guatemala and Pakistan, households in urban areas spend five time as much if their children attend non-state than if they attend state schools (**Figure 4.5a**). While households in the poorest 20% spent practically nothing on education in Argentina, Costa Rica, Philippines and Zambia, the richest 20% spent between 0.5% and 1.7% of GDP (**Figure 4.5b**). Gender is not a major determinant of household spending patterns, although in some countries gender biases continue to affect allocations within households (**Box 4.2**). Many households' ability to pay was affected during the COVID-19 pandemic.

#### **PUBLIC EDUCATION IS OFTEN NOT FREE**

Governments are expected to cover the full operational cost of public schools, although there are cases where schools rely at least partly on fees. In Lebanon, 38% of public upper secondary school students attended schools that received at least 80% of their funds from fees in 2018; in Mexico, the corresponding share was 29%. About 1 in 10 students in Jordan and Morocco attends such schools (**Figure 4.6**). Even schools that are fully funded by government are typically not free for parents. Requirements to pay for textbooks and school uniforms are well documented.

#### FIGURE 4.4:

The main spending item is fees for households with children in non-state schools but other costs for households with children in state schools

99

Household primary and secondary education expenditure as share of GDP, by school type and expenditure item, selected countries, 2010s



GEM StatLink: https://bit.ly/GEM2021\_fig4\_4

Source: GEM Report team analysis based on Institute for Health Metrics and Evaluation (2021).

#### FIGURE 4.5:

#### Urban and rich households account for much of household education expenditure

Household primary and secondary education expenditure as share of GDP, by household characteristics and school type, selected countries, 2010s





#### b. By household wealth quintile

#### Intra-household differences in education spending reflect changing gender norms and opportunities

Household out-of-pocket education spending is not always equally split between family members (UNESCO, 2020). The bias is not consistently against girls or boys. It is driven by expectations of employment and earning prospects, along with factors such as elderly care (Rodríguez Takeuchi, 2020). In Latin America and the Caribbean, where boys lag behind girls in education attainment, households spend less on secondary education for boys (Acerenza and Gandelman, 2019). But in countries with discriminatory gender norms at girls' expense, families prefer to send sons, not daughters, to private schools. A study of the effects of privatization between 2010 and 2015 on girls' access to schooling in Liberia, Malawi, Mozambique, Nepal and the United Republic of Tanzania found boys had been prioritized when parents had to pay for education (Iversen and Begue, 2017).

In India, longitudinal research suggests that gender differences in household education expenditure have shifted over time. Decisions favouring boys shifted between 1995 and 2014 from which child to enrol in school to which child to enrol in private school (and how much to spend). Decisions also shifted from primary to secondary school children (Datta and Kingdon, 2019). In Indonesia, girls are more likely to be enrolled in madrasas and boys in private secondary schools. The decision is linked to madrasas' lower cost and to cultural preferences (Asadullah, 2018).

Gender gaps also exist in spending on private supplementary tutoring. In India, the gender gap at girls' expense in the probability of private tutoring was two percentage points in primary and lower secondary and four percentage points in upper secondary education (Azam, 2016). In the Republic of Korea, gender gaps have been narrowing. The gap in non-school education expenditure on academic subjects changed from US\$10 per month favouring firstborn boys in 2007 to US\$1 in 2012 and US\$6 favouring girls in 2016 (Choi and Hwang, 2020).

#### FIGURE 4.6:

#### Private schools rely mainly on school fees in the majority of countries

Percentage of students who go to secondary schools that receive at least 80% of funding from fees, by type of school, 2018



Low-income countries have lacked the funding to make public education fee-free. In Uganda, households regularly pay informal cash user fees even though this is prohibited by law. The share of rural public schools collecting fees increased from 40% in 2005/06 to 80% in 2011/12. Higher fees were more likely to be charged where there were competing private schools in the community. High informal fees in public schools, corresponding to 4% of total household consumption expenditure, have been associated with a 13% decrease in school attendance among children from poor households but no decrease among children from non-poor households (Sakaue, 2018). In rural areas of the United Republic of Tanzania, despite a government policy of fee-free education, more than three guarters of families viewed primary school contributions as mandatory, noting that children could be dismissed or punished if contributions were delayed (Lindsjö, 2018).

Eliminating informal user fees can have substantial benefits. A scholarship programme in the Gambia that covered informal fees for uniforms, books and other supplies, complementing a policy to eliminate formal school fees for girls, increased female enrolment by 13% and the share of those who took the grade 9 examination by 11 percentage points (Giordono and Pugatch, 2015).



In high-income countries, public schools often charge additional fees. The 2018 PISA found public schools took obligatory financial contributions from parents in 7 of 59 countries and voluntary contributions in 38 of 59 countries (OECD, 2019c). In Australia, parents contribute significant funds, exacerbating inequality between schools (Thompson et al., 2019). In Melbourne, school councils can request three categories of additional payments from parents: essential student learning items, optional items and voluntary contributions (Hedges et al., 2020). While the average parental contribution in government schools in Melbourne from 2013 to 2016 was over US\$700, parental contributions in schools in well-off areas were more than triple those in schools in the poorest areas (Rowe and Perry, 2020). A survey of school business leaders in the United Kingdom found that 18% of schools had asked parents for voluntary contributions to mainstream activities. Such parental contributions had become necessary due to substantial budget cuts (Coughlan, 2017). In the United States, parent-teacher association revenue tripled between the mid-1990s and 2010 to over US\$425 million, exacerbating inequality between schools (Brown et al., 2017).

# PRIVATE SUPPLEMENTARY TUITION IS A LARGE COST FOR MANY HOUSEHOLDS

The analysis so far has shown that fees and other payments to schools are only one share of total household spending, but the largest share for households that send their children to private schools.

Private supplementary tuition is the largest household education expenditure item in some countries. In Bangladesh, the share of urban households paying for private tuition increased from 48% in 2000 to 67% in 2010 while the corresponding share of rural households doubled from 27% to 54%; for the poorest quartile it quadrupled from 10% to 40%. Overall, average expenditure increased by 80% in real terms (Pallegedara and Mottaleb, 2018).

In China, households allocated about one third of their total education expenditure to costs outside school in 2017, ranging from 17% in rural areas to 42% in urban areas. The private tutoring industry cost families US\$70 billion (Wei Yi, 2018). Among students in the last year of lower secondary school in four wealthy areas that took part in the 2015 PISA, 64% took private supplementary tuition (Liao and Huang, 2018). In Egypt, as a share of average expenditure per capita, among students in general secondary education, those form the richest quintile of households spent 51% on private lessons and the poorest 29% (Sieverding et al., 2019). In the Netherlands, household expenditure on private supplementary tutoring increased by 160% between 2005 and 2016 (Elffers and Jansen, 2019).

66

In the Netherlands, household expenditure on private supplementary tutoring increased by 160% between 2005 and 2016

"

In Myanmar, tutoring represented 42% of total household education spending in 2009/10: 58% of grade 9 and 79% of grade 11 students felt that private tutoring was a moderate or heavy financial burden (Bray et al., 2020). In Sri Lanka, private tutoring accounted for half of households' education spending in 2012/13; rural, Tamil and Muslim households, and those with less educated heads, were less likely to spend on such education (Pallegedara and Kumara, 2020).

#### PRIVATE PROVIDERS RELY ON HOUSEHOLD OUT-OF-POCKET EXPENDITURE

The extent to which private schools rely on fees for their operation varies greatly. Data from the 2018 PISA show the majority of private secondary schools receive at least 80% of their revenue from fees in 28 out of 51 education systems. There are marked contrasts between countries in a given region that choose to support private schools versus those that do not (e.g. Argentina vs Uruguay, Australia vs New Zealand, Latvia vs Belarus) (Figure 4.6).

In Dubai, United Arab Emirates, the government provides free public education only for the some 10% of the population who are citizens (Winchip, 2020). A highly stratified for-profit private sector provides education to the immigrant population (Kippels and Ridge, 2019): The average fee for schools rated 'outstanding' is about US\$13,000, more than triple that for schools rated 'acceptable' and more than quintuple the average for schools rated 'unsatisfactory' (Azzam, 2017). In a survey of 3,000 households, 64% said they received no financial support from their employers for education costs (Rizvi, 2019).

In poorer countries, private schools for low-income households grew out of lack of access to public schools, often in slums. Analyses of parental behaviour highlight how households sacrifice other necessities to educate their children in private schools. In Lagos, Nigeria, poor parents, unlike their richer peers, could not afford medical expenses after spending on schools (Härmä and Siddhu, 2017). Analysis of 47 schools in Nairobi, Kenya, found the financial burden on households for children to attend low-cost private schools was too high to warrant the resulting marginal improvement in English, Kiswahili or mathematics (Zuilkowski et al., 2020). The poorest parents in Kenya and other low- and lower-middle-income countries often have to resort to schools that are unregistered and cheaper but likely to have poor facilities and offer lower-quality instruction (Figure 4.7).

#### FIGURE 4.7:

# Unregistered schools are likely to be cheaper than registered low-fee private schools

Average low-fee private school annual fees, selected countries, 2016–20



Notes: The sample includes 1,130 schools, of which 55% were registered Fees were adjusted to 2020 constant US dollars.

GEM StatLink: https://bit.ly/GEM2021\_fig4\_7 Source: Acholla (2021), based on data collected by independent researchers, Opportunity International and CapitalPlus Exchange.

Poor parents employ a variety of strategies to cope with expenses. Some negotiate discounts based on the number of children they enrol. Others pay to secure a place in the school, leaving nothing for uniforms, textbooks and other materials. Some accrue a debt to the school and pay it off in small instalments. Proprietors in India and Nigeria make plans for dealing with unpredictable payments (Härmä, 2020).

About one in six families saves to pay school fees. Analysis of the 2014 Global Findex Database for this report has found that about 8% of households also borrow, with shares ranging from 6% in high-income to 12% in low-income countries. Similar proportions of the poorest 40% and richest 60% of households borrow to pay fees in low- and lower-middle-income countries, although slightly more of the richest manage to save. In Haiti, Kenya, the Philippines and Uganda, 30% of households or more borrow for school fees (**Figure 4.8**).

"

About one in six families saves to pay school fees

#### **FIGURE 4.8:**

#### About 16% of households save and 8% borrow to pay for school fees in poor countries

Percentage of households that saved or borrowed to pay for school fees, low- and lower-middle-income countries, 2014



Source: GEM Report team analysis based on the database compiled by Demirgüç-Kunt et al. (2014).

Savings accounts, digitization of payments and innovative platforms aim to support families in paying school fees (Braniff, 2016; Mbole and Kimathi, 2019). In Uganda, a loan product, ReadyPay, allows those who have repaid solar home system loans to use the equity as collateral for loans to pay school fees, enabling education spending by families traditionally lacking access to credit (Mattern and Garcia, 2021). An emerging array of financial service providers is geared particularly towards supporting low-fee private school proprietors (**Box 4.3**). Such solutions focus on smoothing household expenditure so parents can pay fees more often in smaller amounts. They neither reduce nor eliminate households' financial burden.

# COVID-19 HAS PRESENTED A MAJOR FINANCIAL CHALLENGE FOR HOUSEHOLDS

It has been estimated that during the COVID-19 crisis, household spending on education may have fallen by between 6% and 10% in Latin America, South Asia and sub-Saharan Africa (Le Nestour et al., 2020). Many

### BOX 4.3:

# Financial service providers emerge to support low-fee private schools

Private schools can raise funds in countries with developed financial systems, but in others the lack of access to capital can be a severe challenge, especially for low-fee private schools. Financial service providers have sprung up as a result, with a focus often on religion, charity or equity. Edify and Opportunity International are the largest education support providers lending to single-proprietor low-fee private schools in 15 countries, providing funds for technology, working capital, and infrastructure improvement and expansion.

Besides financing, these and other institutions, such as the Center for Indonesian Policy Studies, Kashf Microfinance Bank in Pakistan and SEED in Nigeria help develop proprietors' capacity in school improvement planning and business administration. SEED recognizes providers' diversity and classifies them as aspiring, emerging or thriving. It uses this classification to provide school transformation programmes, with training, coaching, finance and partnerships appropriate for their level (Acholla, 2021; Sivasubramaniam, 2021). 66

During the COVID-19 crisis, household spending on education may have fallen by up to 10% in Latin America, South Asia and sub-Saharan Africa

99

households, experiencing job and income losses, struggled to pay fees and left private schools. In Panama, private school associations reported that 35% to 40% of parents could not pay monthly fee instalments (Alvarez et al., 2021). In Ecuador, public school enrolment was up by 6.5%, or 120,000 students, at the beginning of the 2020 school year (Olsen and Prado, 2020).

Some private schools may have been well placed to move to remote and hybrid education. Even so, families sometimes felt the service received was not worth the price. In Colombia, parents expressed dissatisfaction with teaching quality and sued private schools to demand fee reductions. In Peru, 90% of parents with children in private schools were dissatisfied with virtual classes (Alvarez et al., 2021).

India's Supreme Court rejected parents' plea for more time to pay schools due to COVID-19 (Thomas, 2020). A survey of 1,052 school proprietors in five states, conducted in February and March 2021, showed that nearly all schools struggled with fee collection; in Haryana state, more than 80% of parents failed to pay fees. School owners surmised that parents had transferred children to public schools or less expensive private schools, or perhaps paid instead for private tutoring and coaching centres (Centre for Civil Society, 2021).

Anticipating such challenges, the Colombian government extended a credit line in July 2020 for up to six months to address difficulties in paying fees, with high forgiveness percentages for poorer families; however, just 2% of families with children in private schools benefited (Alvarez et al., 2021). The Organic Law of Humanitarian Support in Ecuador proposed that parents of private school pupils who lost their jobs would receive support of up to 25% of the monthly cost of tuition. Parents who decided to withdraw children from private schools would be guaranteed access to state schools. The Mexico City Federal District government administered a monthly direct payment for any student who moved from a private to a public school (Berlanga et al., 2020). In October 2020, Panama helped private schools by providing direct support to families paying monthly fees, but the amount was only about 3% of the funding requested (Alvarez et al., 2021).

## DONORS ARE CAUTIOUS ABOUT SUPPORTING FOR-PROFIT EDUCATION PROVIDERS

Aid has remained constant as a share of donor countries' gross national income. As poorer countries grow faster than richer countries, the relative significance of aid as a source of funding, including in education, has been decreasing. Donors are therefore open to ideas that could help improve aid efficiency. While some donors view for-profit providers with caution, a few have been attracted to the idea of non-state provision and its potential to maximize private investment's effectiveness.

The Japan International Cooperation Agency's 2015 position paper noted that more private schools and enhanced private sector and NGO involvement posed the challenge of simultaneously ensuring equitable access and quality. However, it acknowledged that education cooperation was changing to include private sector actors in multi-stakeholder partnerships (JICA, 2015, 2016).

The US Agency for International Development has a proactive strategy for private sector engagement in education, which starts from the premise that non-state schools fill a gap in provision. Aligned with its overall private sector engagement policy, the strategy describes significant private capital invested in non-state schools and seeks to leverage and shape this investment through strategic partnerships to help reach the most marginalized. Non-state schools are viewed as especially important in crisis and conflict contexts (USAID, 2018b) (**Box 4.4**).

The United Kingdom's 2018 policy on aid to education notes that non-state actors are a major provider of schooling in sub-Saharan Africa and South Asia and refers specifically to the Punjab Education Fund in Pakistan. The policy aims to support decision makers in developing good regulatory arrangements and to engage PPPs and non-state schools in improving access to poor and marginalized children (DFID, 2018). The CDC Group, the UK development finance institution, has invested about US\$60 million or 2% of its portfolio in education; it champions private schools, whether for-profit or not, as part of its impact investment strategy (CDC Group, 2019).

Indeed, much support for private actors is provided through the impact investing activities of development finance institutions (Smith and Baker, 2017); the International Finance Corporation (IFC), part of the World Bank Group, is the largest. Out of an education portfolio of almost US\$1.2 billion, most of its large projects in education technology and tertiary education were directed to

# BOX 4.4:

#### Donor support to non-state providers in crisisand conflict-affected contexts struggles to build sustainable service delivery structures

In the absence of a strong national system, donor funding often maintains education continuity in crisis- and conflict-affected contexts. Analysis of policy documents found that 11 of 16 donors listed education as a priority area in their humanitarian and development policies (Dupuy et al., 2020). These arrangements usually involve funding a combination of non-state education providers.

One concern is that donors uncritically engage with the private sector rather than try to improve state education provision (Unterhalter and Robinson, 2020) or build governments' capacity for supervision of nonstate providers (Menashy and Zakharia, 2021). In Haiti, non-governmental and faith-based organizations manage over three quarters of the school system. Donors are a key source of funding for the system. About 20 technical and financing partners invested US\$587 million in the education sector between 2010 and 2015, in the aftermath of the 2010 earthquake (USAID, 2018a; World Bank, 2018). A relatively small part went to the fledgling public education system. A US\$50 million Inter-American Development Bank project devoted funds to build just 20 public schools (IDB, 2011).

Coordinating and monitoring the work of non-state providers in emergencies can be challenging and negatively affect equity. For instance, teachers of Syrian refugees in Turkey were receiving different salaries depending on which NGO was employing them (Nicolai et al., 2020). Often government leadership in such coordination mechanisms is weak, while different NGOs take the leadership in coordinating education activities in different parts of the country, as a review of such arrangements in the Democratic Republic of the Congo has shown (Khan et al., 2020). In some contexts, international NGOs may undermine the capacity of the state to deliver public services, raising sustainability concerns (Campbell et al., 2019).

Sustainability is also undermined by fragmentation and the small scale of projects involving NGOs. A review of nine NGO-implemented peace education projects to counter violent extremism in Pakistan found that a few had produced valuable education materials, but they were overdependent on aid funding and came to an end when funding dried up, as strong links with the government had not been established (Ahmed and Shahzad, 2021). Fragmentation also results in difficulties to assess the effectiveness of non-state interventions. As non-state actors compete for funds, evidence may be distorted. The results of partnerships with non-state actors in emergency contexts are presented as success stories even when outcome improvements are marginal or, worse, the emergency context is used as an excuse to promote inappropriate solutions, as in the case of technology in the Syrian refugee crisis (Menashy and Zakharia, 2020).

non-state actors, while it allocated 15% to private school chains (Hares and Crawfurd, 2021). However, the IFC froze investment in private, fee-charging pre-primary, primary and secondary schools in 2019. The decision, which covered direct investment, advisory work and indirect investment linked to private equity, resulted from years of pressure from civil society organizations concerned about promotion of private education providers, especially commercial operations such as Bridge International Academies (Ron Balsera, 2020). In addition, the Compliance Advisor Ombudsman, an independent accountability mechanism that evaluates the impact of IFC projects, received complaints from current and former teachers and parents in Kenya (EACHRights, 2018). The World Bank Independent Evaluation Group has undertaken a recent evaluation of IFC investment in pre-primary, primary and secondary education, focusing on alignment with the organization's strategic objectives; results are expected soon. The IFC will continue funding private actors in other parts of its education portfolio (IFC, 2020) (Figure 4.9).

The IFC decision is characteristic of the tension surrounding use of public funds for funding private education, especially in organizations with multiple members. The Global Partnership for Education (GPE) has developed a private sector strategy, taking a broad view of private actors and aiming to use its convening power to advance thought leadership, best practices and collaboration with the private sector. However, opposition during negotiation of this strategy led to a clause prohibiting use of GPE funds to support for-profit provision of core education services (GPE, 2019). In 2018, a European Parliament Resolution instructed the European Commission to not fund for-profit education actors, making special reference to international commercial operations. The resolution noted the importance of core education - including elements such as school supplies, transport and food - being free of charge (European Parliament, 2018).

66 -

In 2018, a European Parliament Resolution instructed the European Commission to not fund for-profit education actors

**99** 

A few donors have been attracted to the idea of non-state provision

"

# DONORS ARE EXPERIMENTING WITH PUBLIC-PRIVATE PARTNERSHIPS

Some donors have looked into using their funds as a catalyst to raise financing through PPPs. Such arrangements also attract cash-strapped governments interested in raising private capital to improve and expand public infrastructure. Accounting measures allow them to keep costs and liabilities off their balance sheets and postpone recording fiscal costs of infrastructure services (Romero and Vervynckt, 2017). PPP initiatives to reduce infrastructure backlogs and improve schools are under way, for instance, in the Philippines (Philippines Department of Education, 2021) and South Africa (van Tonder, 2019). Egypt's PPP Central Unit approved a project in 2016 for up to 1,000 schools by 2030, but its first phase – 24 schools in 6 governorates – has been delayed (Enterprise, 2019).

While PPP advocates argue that contracts lead to improved efficiency, the evidence is not robust. Private finance through infrastructure PPPs in sub-Saharan Africa has not played as strong a role as hoped, forcing governments to fill gaps. Former proponents now concede that private partners are unlikely to support activities with a potential poverty-reducing impact unless offered substantial incentives by governments and donors, leading to the conclusion that 'a government capable of fully playing its role in designing, developing, implementing and regulating PPPs is probably better off using traditional public procurement to achieve the same objectives' (Leigland, 2018, p. 128). The International Monetary Fund (IMF) acknowledged in 2018 that PPPs might not always be efficient, fiscal risks could be sizeable, and PPPs suffered from the same management challenges as traditional public investment (Irwin et al., 2018). Yet the IMF, World Bank, Asian Development Bank and OECD still promote PPPs (Rosell and Saz-Carranza, 2020).

PPPs have also been used to finance social impact bonds, in which a private investor provides capital up front; repayment is contingent on results, such as completion rates or achievement of specific learning outcomes. An example funded by multilateral financial institutions is the Brazil Secondary Education Social Impact Bond in São Paolo. The India Educate Girls Development Impact Bond in Rajasthan ran from 2015 to 2018 with a capital commitment of US\$280,000 (Gustafsson-Wright et al., 2017; Gustafsson-Wright et al., 2015). India has offered more opportunities to experiment with impact bonds due to its regulatory environment and incentives for corporate engagement (**Box 4.5**).

Social impact bonds' effectiveness has been questioned, however (Dey and Gibbon, 2018), as has the effectiveness of results-based financing in education (UNESCO, 2018). On average, despite optimistic predictions, US\$1 of public investment from bilateral and multilateral donors mobilizes only 75 cents of private investment in low- and middle-income countries and just 37 cents in low-income countries. As little of such funding goes to education, better targeting of aid should remain the priority (Attridge and Engen, 2019).

### BOX 4.5:

#### India has become a laboratory for results-based financing in education

India has served as a testing ground for social impact bonds in education. The Quality Education India (QEI) Development Impact Bond, running from 2018 to 2022, is the largest of its kind in education at a cost of US\$11 million. It builds on lessons from the Educate Girls bond, which reached 7,000 children (Edwards, 2019). QEI aims to reach more than 300,000 poor children of primary school age and improve their learning outcomes (Ecorys, 2019b). It includes a broad coalition: The risk investor is the UBS Optimus Foundation, while the British Asian Trust, the Michael and Susan Dell Foundation, Comic Relief, the Mittal Foundation, the Larry Ellison Foundation and British Telecom are the outcome funders. The UK Foreign, Commonwealth and Development Office is providing technical assistance.

The novelty of the funding mechanism and the reputational risks involved led to the choice of providers with a track record: the Kaivalya Education Foundation, Gyan Shala, Society for All Round Development, Educational Initiatives and Pratham Infotech Foundation. An evaluation found significant costs associated with designing and launching the mechanism and fundraising (Ecorys, 2019a, 2019b). First-year results, which found mixed effects on learning, led to some underperforming programmes being dropped, but the results were considered to be on track (Edwards, 2019). Second-year results, released in August 2020, were positive: Learning gains were much higher, with children learning twice as fast as their peers in control schools and substantially outperforming targets (Quality Education India Development Impact Bond, 2020).

More recently, Social Finance India, an impact financing platform launched by the Tata Trusts and the Global Steering Group for Impact Investment, aims to provide US\$2 billion by 2030, of which US\$1 billion will be directed at non-state education service providers through the India Education Outcomes Fund, and US\$1 billion will support the India Impact Fund of Funds to catalyse debt-for-impact in education, health and housing (Mehendale and Singh, 2020).

#### FIGURE 4.9:



Development finance institution investment in education, 2012–16



Notes: FMO = the Netherlands Development Finance Company; CDC = UK's development finance institution; Proparco = a subsidiary of the French Development Agency focused on private sector development; OPIC = Overseas Private Investment Corporation, now the US International Development Finance Corporation.

GEM StatLink: https://bit.ly/GEM2021\_fig4\_9

Sources: Kenny et al. (2018) on the total amount of expenditure; Hares and Crawfurd (2021) for the breakdown of IFC expenditure based on an analysis of its submissions to the International Aid Transparency Initiative.

### THE ROLE OF PHILANTHROPIC AND CORPORATE ACTIVITIES IN EDUCATION IS EVOLVING

Corporate and philanthropic actors include corporations (transnational, multinational and local), individual entrepreneurs, angel investors, private foundations (international and regional), venture philanthropies, corporate philanthropies, corporate social responsibility units, social innovation funders, impact investment funds, social enterprise organizations, and consulting companies (Srivastava and Read, 2019).

The boundaries between these categories are sometimes hard to discern, while the motivation for actors' engagement ranges from charity to corporate interest. Philanthropic foundation activities tend to be unrelated to business interests, even when they draw their wealth from a business operation. Corporate activities tend to be closely tied to business interests. In some cases, they channel funds through NGOs or international organizations; in others they engage in corporate social responsibility activities, and may directly supply goods and services (UNESCO, 2012).

Some rich individuals and selected companies have engaged in initiatives that draw attention to their contribution but also aim to mobilize peers to act accordingly. The Giving Pledge, initiated by Bill Gates and Warren Buffet in 2010, encourages extremely wealthy individuals to provide most of their fortune to philanthropic causes. The similarly

66

While public-private partnership advocates argue that contracts lead to improved efficiency, the evidence is not robust

**?**?

# Corporate activities tend to be closely tied to business interests

intended Pledge 1% was initiated by technology corporations, including Salesforce and Atlassian, in 2014 (Patil and Brakman Reiser, 2021).

"

As philanthropic and corporate activity has grown, there is a need to understand how much such non-state actors are providing to support education, along with how they are competing with, complementing or collaborating with governments and donors (see **Chapter 5**).

#### PHILANTHROPIC FOUNDATIONS' SPENDING IS GROWING BUT IS STILL SMALL

Despite perceptions that the amount philanthropic foundations spend on education is growing, it remains relatively insignificant. In 2015, around US\$2.6 billion, or 13% of the total Fortune Global 500 companies' corporate social responsibility budget, was spent on education. Of that, 30% went to primary and secondary education. Spending was mostly linked to companies' countries of origin and tended to align with their business interests and support their supply chains (Varkey Foundation et al., 2015). A survey of 110 business leaders in 22 middle-income countries noted that half were motivated to provide long-term education solutions to their countries; one fifth cited religious and philosophical motives; another fifth noted the importance of education for their firms' development objectives. About 6 in 10 had established tax-exempt foundations to promote non-profit education. Several recognized the benefits of such activity for their reputation (Giacomin et al., 2019).

The OECD Global Survey of Private Philanthropy for Development collected information on support to education from 143 foundations. It estimated that education received US\$2.1 billion over the three years from 2013 to 2015. This was equivalent to 9% of all philanthropic giving, making education the second-most-supported sector after health. From the 26% that funded basic and secondary education, the top recipients were Brazil, India and Kenya (Figure 4.10) (OECD, 2019b). A second survey collected information on support to education from 88 foundations. They are estimated to have disbursed a total of US\$2.3 billion in 2016–19 or US\$580 million per year on average. International philanthropy was allocated to primary and secondary education (37%); post-secondary education (31%); and other education levels and issues (32%). The top 15 foundations education provided 70% of

the total, while the most common funding vehicle was grants. National foundations focus entirely on post-secondary education (OECD, 2021).

# CORPORATE FINANCING OF EDUCATION IS CLOSELY SCRUTINIZED

Large corporations and wealthy individuals are increasingly positioning themselves to influence education systems, including through financing. However, critics say they contribute to erosion of democratic accountability and offer short-term, partial solutions, similar to the unequal distribution of opportunities in social and economic life to which they have contributed (Giridharadas, 2019). Conversations about corporate social norms, transparency and tax responsibility have barely begun (McCluskey, 2015; Meiring and Krugel, 2018).

Regarding corporate activity, public education systems might be better funded if governments could rein in large corporations' tax avoidance and evasion. It is estimated that the global tax system loses US\$427 billion a year, of which US\$245 billion is linked to corporations transferring profits to tax havens and US\$182 billion to private individuals. Low- and lower-middle-income countries lose the equivalent of 5.8% of their collected tax revenue (Tax Justice Network, 2020).

Official IMF estimates put the cost of tax havens even higher, between US\$500 and US\$600 billion a year, taking legal tax evasion into account. The IMF estimated that low- and middle-income countries lost US\$200 billion a year in tax revenue, which exceeds total official development assistance (Shaxson, 2019). The agreement reached in July 2021 for a global minimum corporate tax is an important step towards rectifying that situation. But indirect losses are likely even higher. Analysis of IMF data revealed that, while total official foreign direct investment in 2017 amounted to US\$40 trillion globally, around US\$15 trillion was phantom investment in shell corporations with no real activity (Damgaard et al., 2019).

Corporations, rather than individuals, are the major donors. Of the top 20 wealthiest individuals in the United States, 10 devoted no more than 0.1% of their net worth to charitable giving in 2018 (Khan, 2020). Tax incentives encouraging corporations to make charitable contributions have led to growth in such giving. In China, the number of Chinese foundations

99

The IMF estimated that low- and middle-income countries lost US\$200 billion a year in tax revenue due to tax havens

increased by 430% between 2006 and 2016 (Global Chinese Philantrhropy Initiative, 2017). Corporations qualify for significant tax advantages and have the resources and sophistication to meet their requirements. As Chinese corporations' founders are strongly identified with them and have continuing connections, wealthy founders pursue philanthropy through their corporations instead of, or in addition to, individual efforts (Johnson and Saich, 2018). Corporate social responsibility activities, moreover, are seen as a way to cleanse reputations besmirched by tax avoidance and wealth accumulation. Firms even anticipate consequences of aggressive tax avoidance by increasing corporate social responsibility activities (Col and Patel, 2019).

In India, Section 80G of the Income Tax Act has provided incentives to charitable organizations and donors, with organizations receiving substantial tax exemptions on income (Patil and Brakman Reiser, 2021). India was also the first country to legally mandate

corporate social responsibility through Section 135 of the Companies Act 2013. Indian companies with a net worth above US\$79 million, annual turnover of over \$140 million or net profit of US\$0.7 million have to give 2% of average net profit over three years to charity (Chadha and Nandwani, 2020). In the first four years of implementation, 64% of eligible companies reported on corporate social responsibility activities. The reports showed that 24% had their own foundations and 52% had established corporate social responsibility departments (Samhita Social Ventures, 2018). Education has been the top recipient sector, absorbing 37% of total corporate social responsibility expenditure, or US\$626 million a year, on average - but this was equivalent to only 1.5% of the states' total government education expenditure (Chadha and Nandwani, 2020). Moreover, since three relatively prosperous states received 60% of all domestic philanthropy, current corporate social responsibility funding may be adding to education financing inequality (OECD, 2019a).

99

#### **FIGURE 4.10:**

**Brazil, India and Kenya receive the largest shares of philanthropic financing for basic and secondary education** Philanthropic funding to primary and secondary education, by recipient country, 2013–15



### **CONCLUSION**

Non-state actors in education receive and disburse funds. Government financial support for non-state providers varies considerably and largely depends on whether governments decide to fund non-state actors at all, a decision shaped by history and politics. Governments may support none, some or all non-state providers and the type of costs they cover may vary. Some high-income countries' governments have financially supported non-state providers through aid programmes, even when this runs against other aid policy objectives, such as promoting equity.

Some governments have opted to help families pay for private schooling as a way to support school choice policies. Private schools in most countries rely on fees, a fact that became particularly evident during the education crisis brought about by COVID-19. But households do not pay just for private school tuition. They also incur non-school expenditure, such as supplementary private tuition, and many pay public school fees that are often informal but perceived as obligatory. Richer households are more likely to spend out of pocket, further exacerbating inequality, while the burden is much heavier for poor households that incur such costs.

Equality and efficiency are themes that emerge in discussions of the role of philanthropic and corporate financing of education. Public-private partnerships have proved to have less impact than expected in infrastructure projects. There is interest in blended approaches that try to catalyse private funding for achievement of education outcomes, although these are at the experimental stage.

Books on business published by Pearson, a multinational publishing and education company with headquarters in London. WE F

5/10

1e

CREDIT: Chris Batson/Alamy Stock Photo



# CHAPTER



# Influence



# **KEY MESSAGES**

#### Actors try to influence education through advocacy, lobbying, research and funding.

A network led by the International Finance Corporation framed its approach on public-private relationships in education by stating that education was a consumer good. But other advocacy networks frame privatization as a threat to the right to education.

#### No actor group is homogeneous in its attitudes, approaches and influence.

- In a survey of Global Campaign for Education members, 43% expressed a negative view of for-profit provision but 16% were supportive; likewise, 35% expressed a negative view of public-private partnerships, while 31% were supportive. A third expressed a mixed view.
- Among non-state actors that influenced the Indian government's emphasis on foundational literacy and numeracy, some focused on scaling up interventions in the public education system, while others advocated for an enabling environment for low-fee private schools.
- Teacher unions are leading advocates for public education, questioning the use of public-private partnerships in Latin America at the expense of equivalent public institutions. In some countries, unions are criticized as undermining efforts to strengthen public education.

International organizations, foundations and think tanks tread a fine line between pushing their agendas and facilitating dialogue for policymaking.

- The World Bank uses loan conditions, technical assistance, research studies and events to strengthen its position as a knowledge broker. Analysis of its recommendations in 10 countries showed it promoted more private provision in 9 and less regulation in 6.
- In the United States, the Bill and Melinda Gates Foundation backed the charter school movement. Brazil's Lemann Foundation mobilized broad consultation in pursuit of the goal of national core curricular standards.
- Ark, a manager of 39 schools in England (United Kingdom), has an international arm that advises governments how to outsource management of public schools, for instance in Liberia and South Africa.

#### The focus of non-state actors' influence on national education systems varies.

- Faith-based organizations opposed government policies on gender in Peru's curriculum, affected school meal menu choices in India and lobbied for more public funding in Canada.
- Non-profit organizations may facilitate adoption of technology-related innovations, promote standards in financial education or support low-fee private schools.

#### The business sector exerts strong influence on education.

- Education technology firms are frequently presented as enablers and disruptors. But their claims on products can be misleading. Just 2% of 10,600-plus products reviewed in a US government clearinghouse were classified as strongly or moderately effective.
- Business organizations often express support for school choice, competition and for-profit education, standardized assessment and publication of results, as in Japan.
- School sponsorships are debated. The benefits of financial or in-kind support can be reversed if accompanied by marketing, as in Australia, China and the Philippines.

Networks advocate for competing visions of non-state actors in education	.118
The business sector exerts strong influence on education	.125
nternational organizations and foundations have particular influence	.127
Conclusion	.132

99

A rguments regarding efficiency, innovation and equity are at the core of debates on the role of non-state actors in education, which are usually characterized by acrimony and mistrust. Two very different views of the world clash and it is rare to see agreement, compromise or admission of error. Civil society organizations, teacher unions, public agencies, international organizations, businesses, non-governmental organizations (NGOs), media, philanthropies, political parties, academics, private-sector associations and think tanks are among those trying to shape discourse and influence opinion in favour of or against a stronger role. At the same time, none of these groups are homogeneous.

Actors try to influence education through advocacy and lobbying networks, research, and funding, which is often associated with sales of goods and services. They may use some of these approaches, depending on their comparative advantage, or all of them in some combination. This chapter describes various actors' attempts to exercise influence, sway views and mobilize support behind competing aspirations and interests. Examples highlight the blurring of identities among actors. Individuals may change hats, switching from representing state organizations to non-state ones. Non-state actors may be dedicated to supporting state education. Organizations entrusted with protecting and promoting public education may undermine it with their actions. The same principles are invoked to defend radically opposed perspectives on education, often by manipulating language.

66

Actors try to influence education through advocacy and lobbying networks, research, and funding In this competition of political ideas and economic interests, actors seek to form coalitions and make their views authoritative, legitimate and attractive to others to achieve their aims (Hill and Jochim, 2009). But the battle for opinion is often behind the scenes, which is why this report recommends that governments, legislators and the education community at the national and global levels should strive to maintain the transparency and integrity of the public education policy process so as to keep vested interests at bay. Over and above any other consideration, decision makers need to be conscious of every learner's right to education and take responsibility for fulfilling it. They should not be held hostage by anyone using force, deception, money or other illegitimate means to promote narrow, group-specific interests. Ultimately, governments are accountable for ensuring that all views are given a public hearing and are taken into account in decision making.

# NETWORKS ADVOCATE FOR COMPETING VISIONS OF NON-STATE ACTORS IN EDUCATION

Advocacy approaches vary in their ambition, sophistication and target audiences. They range from community sensitization by grass-roots organizations to organized lobbying by technology corporations. Networks frame their approach to the preferred role of non-state actors. For instance, a network led by the World Bank's International Finance Corporation (IFC) framed its approach on public-private relationships in education by stating that 'education is a consumer good, and that the student is the principal consumer through parents' (Robertson and Verger, 2012, p. 30). By contrast, other advocacy networks have opposed the privatization and commercialization of education, framing privatization as a threat to the right to education (Brehm and Silova, 2019). Network members mutually reinforce their messages by cross-referencing each other, exacerbating polarization. At the same time, these networks are accountable to their members, which face differing circumstances in their respective contexts. The case of the Global Campaign for Education (GCE) is interesting in this regard. Founded on a rights-based approach to education, it considers 'strong, public systems' to be one of its six strategic areas. Its members, however, hold a wider variety of views vis-à-vis non-state and, particularly, private actors than might be thought (**Box 5.1**).

The People's Action for Learning (PAL) Network is an organization whose members, individually and collectively, have been effective in raising awareness of the major education challenges facing African, Asian and Latin American countries (PAL Network, 2020). Its members implement a variety of national education activities but share a common activity: citizen-led assessments, i.e. household surveys, that measure children's foundational learning skills and challenge government shortcomings and poor records in collecting and sharing data on education outcomes. While the PAL Network is a non-state actor, it includes diverse perspectives on government's role. Its members vary widely not only by background and their attitude towards assessments, but also by approaches to action and partnerships. In Senegal, members collaborate with the central government and have been drawn into national education reform. In India and Pakistan, members cooperate with regional education authorities. In Mexico, they operate mainly at the local level and independently from education policies (Alcott et al., 2018).

Pratham, an Indian member of the PAL Network, has been conducting ASER, the world's largest citizen-led assessment, since 2005. Its results have made uncomfortable reading for successive governments but its approach to scaling up cost- and education-effective models routinely includes government and is supportive of the public education system. ASER research and data have been used as major advocacy tools in parliamentary discussions and quoted by government sources. They had a major influence over the inclusion in the 2020 National Education Policu of a reference to foundational literacu and numeracy as an 'urgent and necessary prerequisite' to learning (India Ministry of Human Resource Development, 2020). The Central Square Foundation is another non-state actor that has contributed to the government's strategic shift of focus. But it has explicitly advocated for an 'enabling environment for

### 66

In 2015, Brazil's National Confederation of Educational Institutions, which represents private schools, contested obligations under the disability law on inclusion of people with disabilities in mainstream education.

99

the affordable private school system' as a key lever of progress towards foundational learning (Central Square Foundation, 2020). Both organizations' board members have almost exclusively private-sector backgrounds.

By contrast, private-sector backgrounds are markedly absent among the drafting committee members and signatories of the Abidjan Principles on the human rights obligations of States to provide public education and to regulate private involvement in education (Skelton et al., 2019). The text, drafted by human rights law experts, was conceived as a response to 'private involvement in education ... which, if left unchecked, could gravely impair the progress made in the realisation of the right to education'. NGOs linked to the principles have been among those that criticized Bridge International Academies, a chain of over 500 for-profit schools then operating in India, Kenya, Liberia, Nigeria and Uganda. They lobbied the Kenyan and Ugandan authorities t10 not allow Bridge to run schools and had successful challenges in both countries' courts (ActionAid Uganda et al., 2018; Ball, 2017). They also lodged a complaint with the Office of the Compliance Advisor Ombudsman, a recourse mechanism for IFC projects, on the grounds that Bridge was violating curriculum, health and safety and labour standards (Chapter 4).

In Buenos Aires, Argentina, private schools, even though they are state-subsidized, screen applicants through interviews and tests. In 2018, the NGO Civil Association for Equality and Justice filed a class action suit against the Ministry of Education of the Autonomous City of Buenos Aires, calling the practices illegal and local governments negligent. In 2015, Brazil's National Confederation of Educational Institutions, which represents private schools, contested obligations under the disability law on inclusion of people with disabilities in mainstream education. It argued before the Supreme Court that inclusion was a state responsibility and infringed upon its members' freedom to manage schools. They claimed that non-discrimination was not

#### NGOs hold diverse perspectives on the issue of private education provision

No actor group holds a uniform position on non-state providers in education. Civil society organizations are often critical of, or at least cautious about, the role of private and for-profit actors, voicing concern over growing privatization and commodification, and arguing that education is a public good that needs to remain under democratic control (Croso and Magalhães, 2016).

As a movement grounded in the rights-based approach to education, the Global Campaign for Education, a platform that helps coordinate civil society organizations, stresses that the state has the primary responsibility of ensuring that education of good quality is free and accessible for all. However, many of its members work in countries where non-state and private actors are prominent, and they need to take that into account in their advocacy (Pearce, 2017).

A survey for this report of GCE members demonstrates a relatively diverse range of positions (National Foundation for Educational Research, 2021). Among 49 respondents, 43% expressed a negative view of for-profit provision but 16% were supportive; likewise, 35% expressed a negative view of public-private partnerships, while 31% were supportive. About one third of respondents in both cases expressed a mixed view. By contrast, two thirds of respondents were supportive of non-profit actors, such as community and NGO schools, appreciating their contribution to supporting children in hard-to-reach areas (**Figure 5.1**).

Most members prioritized pressuring governments to ensure that non-state actors did not violate the right to education. Accordingly, among their priority areas of work, some members gave high ranking to offering support to government to regulate or even fund non-state actors (**Figure 5.2**). Nigeria's Civil Society Action Coalition on Education for All said that, while it opposes fees in education, non-state actors are already involved in education and the state cannot guarantee all citizens their right to education, so the coalition's efforts focus on making sure non-state schools provide a safe learning environment of good quality.

#### FIGURE 5.1:

#### Civil society organization views on non-state providers are not uniform

Positions of Global Campaign for Education members on three issues related to the engagement of non-state actors in education, 2020



#### **BOX 5.1 CONTINUED:**

#### FIGURE 5.2:

# Civil society advocacy activities put pressure on governments but also encourage them to regulate and even fund non-state providers

Positions of Global Campaign for Education members on three issues related to the engagement of non-state actors in education, 2020



Note: Based on replies of 29 respondents who were asked to rank areas of work as first, second and third priority.

GEM StatLink: https://bit.ly/GEM2021\_fig5\_2

Source: National Foundation for Educational Research (2021).

GCE members also mentioned the role of media and grass-roots organizations in raising awareness about rules and regulations related to nonstate actors. In India, the campaign strategy is to raise awareness and develop capacity through school management committees as a way to support non-state schools in strengthening their learning environments.

Informing and influencing political parties is a central part of GCE members' mandates. During the COVID-19 pandemic, the National Congress of Brazil approved a constitutional amendment to make permanent the Fund for Maintenance and Development of Basic Education and Promotion of Education Professionals, a tool aimed at equalizing education resource allocations between states and municipalities. Face-to-face campaigns on the amendment were interrupted, so the National Campaign for the Right to Education, Brazil's GCE member, mobilized its own members to engage directly with legislators via WhatsApp and social media. The Bangladeshi GCE member, the Campaign for Popular Education, has acted as a bridge between non-state actors, which it counts among its members, and the government; interactions between them were previously rare.

Some GCE members engage with the international community. The Pakistan Coalition for Education reported engaging with donor countries regarding their support to non-state actors in Pakistan to ensure that rights-based considerations were central to their aid allocation decisions. However, other evidence suggests that civil society organizations have had limited impact on shaping donor decisions (National Foundation for Educational Research, 2020).

binding, but an option. The court dismissed the action on the grounds that non-state providers are obliged to share responsibility for building an inclusive society (ACIJ, 2019). National advocacy activities in support of public education also take place in the Global North (**Box 5.2**). Teacher unions have been leading advocates for public education. For instance, the National Education Association, one of the two largest national unions in the United States, has been active in collective bargaining and advocacy campaigns to improve

#### BOX 5.2:

#### In Quebec, advocates have challenged public funding of private schools and private costs in public schools

In Quebec, Canada, a 2019 bill was aimed at regulating parental financial contributions in public schools (National Assembly of Québec, 2020), which affected at least 20% of children who attended selective public schools (Québec Government, 2020) offering 'special projects' (e.g. arts, sport, international). Private schools, attended by 12% of students in the province, remain largely state funded. About 65% of private schools receive direct state subsidies that amount per pupil to about 60% of the subsidy per public school pupil (Québec Ministry of Education, 2021a), but the share rises to 79% when additional allowances for special programmes or educational services and tax credits (Québec Ministry of Education, 2021b) are taken into account (Mouvement l'école ensemble, 2021).

The Commission des droits de la personne et des droits de la jeunesse (Human Rights and Youth Rights Commission) argued against the bill, maintaining that legalization of fees charged to parents since 2007 for participation in 'special projects' violated the right to free public education guaranteed by the Charter of Human Rights and Freedoms (Human Rights and Youth Rights Commission, 2020). Mouvement l'école ensemble (School Together Movement), a parents' initiative, also opposed the bill and called for ending all direct or indirect public funding of private schools, as well as public school selective admissions (Vigneault, 2020). It published a report on equity in Quebec's education system compared with other Canadian provinces (Mouvement l'école ensemble, 2019). In March 2020, it asked the UN Committee on Economic, Social and Cultural Rights to hold the Quebec government accountable for what amounted to a three-tier school system – private, public and selective public (Vigneault, 2020).

teacher working conditions. It has sponsored projects supporting public education through its Great Public Schools Fund grant programme, which as of 2018 had awarded 120 grants worth US\$32 million (Coons, 2018). It is also engaged in influencing federal and state politics, contributing to election campaign finance and supporting politicians who oppose policies friendly to school choice and to private education interests, regardless of party affiliation (Marianno, 2018).

Teacher unions have also actively opposed commercialization in public education, which they believe undermines it. Education International, the global teacher union federation, questions the extensive use of public-private partnerships in Latin America instead of equivalent public institutions. It has criticized the relationships of governments in Costa Rica with Fundación Omar Dengo, a private foundation selling digital education materials and teacher training services to the education ministry; in the Dominican Republic with Acción Empresarial por la Educación, a business organization providing teacher education courses through the Aprendo project; and in Uruguay with Fundación Ceibal, a publicly funded centre that has contracted with Global Learning Network, a company that sells a standardized pedagogical model, teacher training, digital resources and evaluation processes (Education International, 2021a, 2021b, 2021c).

In some countries, unions are criticized as undermining efforts to strengthen public education. Greece is one of three countries in Europe without a teacher appraisal system (European Commission/EACEA/Eurydice, 2018). The use of school inspections as a means of political repression and control tarnished inspections' reputation and led to their abolition in the 1980s. Successive governments, sometimes on external advice, have tried to establish an alternative support and accountability mechanism but have faced opposition by teacher unions (OECD, 2017; Papakonstantinou and Kolympari, 2019). Opinion polls suggested that 75% of the population agreed with the proposal that teachers should be evaluated by their head teacher and two councillors (Alfavita, 2021). Yet when the government passed a law in 2021 on a new school and teacher evaluation process, the primary school teachers union tried to undermine it by instructing schools to submit an identical school self-assessment and improvement plan; 250 of the country's 8,300 schools did so (Lakasas, 2021).

Another contested issue is the effect of certain union activities on the quality of public education and people's trust in it. A US literature review found no significant effect of strike action on student learning (Cowen and Strunk, 2015). Some studies showed that more low-quality teachers are dismissed and more high-quality teachers retained in more unionized districts, raising education outcomes (Han, 2020). In states with laws restricting collective bargaining by public employees, strong unions had a positive impact on student learning (Han and Keefe, 2020). But studies of teacher strikes in Canada (Baker, 2013), Colombia (Alvarado et al., 2021) and South Africa (Wills, 2014) found negative effects on student learning. An analysis of some 1,500 strikes over 32 years in Argentina estimated that exposure to the average strike duration of 88 days during primary school resulted in students' later earnings

being reduced by 3% (Jaume and Willén, 2019). Disruption to child care (Jaume and Willén, 2021) had a negative impact on parental perceptions of public education, especially in disadvantaged groups, where parents cited private education's stability in their decision to move out of public schools (Moschetti and Verger, 2020; Narodowski et al., 2016).

Teach for All, a global organization with a network of member organizations in 60 countries, provides high-achieving recent graduates as teachers to under-resourced schools, positioning them as future education leaders and change-makers. Of its 83,000 alumni, 73% still work in education (Teach for All, 2021). The US organization, Teach for America, has served as a training ground for young people who may then enter other careers, for instance in the corporate and philanthropic sector, from where they can also shape public education reforms. In the bitterly divided world of education politics, the organization does not endorse any policy direction and claims independence from its funders, which include the Walton Family Foundation, a strong supporter of school choice approaches (Sawchuk, 2016). But some point to several charter school networks that have been founded by or in partnership with Teach for America alumni. In New Orleans, where public schools were converted into charter schools (Chapter 3), experienced teachers were replaced with alternatively certified teachers and Teach for America fellows. Critics also question the promotion of ideas developed in the US context that may not be applicable elsewhere (Crawford-Garrett and Thomas, 2018).

While non-state actors aim to influence policies through advocacy, in some contexts they become de facto policymakers (Moschetti et al., 2019). As a result of education governance reforms in England (United Kingdom) (Chapter 3), Ark has emerged as a manager of 39 schools (Ark, 2021). It engages in policy conversations with governments, contributes technical advice, writes reports, conducts evaluations and proposes solutions (Junemann and Ball, 2013). In 2015, it launched the international Education Partnerships Group, which commissioned a major review of public-private partnerships (Aslam et al., 2017). The group has been involved in high-profile initiatives in Africa and South Asia (EPG, 2021) (Box 5.3), including in designing Partnership Schools for Liberia, even carrying out its monitoring due to 'limited government capacity' (Ark, 2017).

# BOX 5.3:

#### An exported public-private partnership model – Collaboration Schools in Western Cape province, South Africa

Western Cape outperforms other South African provinces in education, yet significant inequality persists. In one effort to improve the education system, in 2016 the provincial government launched the Collaboration Schools programme, a public–private partnership involving non-profit school operators funding and supporting underperforming no-fee public schools, with support from the Education Partnerships Group (Ark, 2021). The programme started with 5 and expanded to 14 primary and secondary schools in 2019, with a target of covering up to 15% of all public schools in the province (Western Cape Education Department, 2020).

The reform takes two forms. First, poorly performing 'transition' schools are handed over to private operators to manage. The intent is to improve learning outcomes through focused additional support, including for more teachers in addition to those already working at the school. Second, new schools are immediately handed over to a school operating partner to manage a block grant for teacher salaries and school management. New teachers in transition and new schools are school employees hired on a contractual basis and no longer considered civil servants.

The funding model borrowed policy elements from India and the United Kingdom (Junemann and Olmedo, 2020). The governance structure required significant legislative changes. Amendments to the Provincial Schools Education Bill included establishment of an independent Schools Evaluation Authority and two new types of schools, Collaboration Schools and donor-funded schools. The bill also provided for representation by Collaboration School Operating Partners on school governing bodies, giving them voting rights of at least 50% and effective control over the schools.

Public hearings were contentious (Parliamentary Monitoring Group, 2018). Opposition was expressed by the teacher union and civil society organizations, which claimed the amendments were inconsistent with the Constitution and the 1996 Schools Act. One argument was that giving the education minister power to identify a school as a Collaboration School entailed a conflict of interest, as the minister would also appoint the Evaluation Authority. Another argument was that a competitive public school policy posed a risk of leaving the most vulnerable behind. In July 2019, the Equal Education Law Centre challenged the amended bill before the Western Cape High Court (Equal Education, 2018). The decision, due in October 2020, has not yet been published.

An important implication of the model is the blurring of lines between state and non-state actors. The responsibility for turning public schools around was handed over by the state to private actors. While the state had the regulatory responsibility to manage the contract, the nature of accountability for public service delivery was changed (Sayed and Soudien, 2021). While non-state actors aim to influence policies through advocacy, in some contexts they become de facto policymakers

99

The report of a forum it helped organize with the former UK Department for International Development with support from the UBS Optimus Foundation to discuss how to expand non-state provision stated: 'Too often, the debate around the role of non-state provision in basic education has been driven by ideology rather than robust evidence'. It noted that evidence that non-state provision was better or could reach the poorest children was limited (Wilton Park, 2017, p. 1).

Ark, alongside the IDP Foundation, Imaginable Futures and the Vitol Foundation, is a founding member of the Global Schools Forum, whose mission is to strengthen and support non-state actors. A global network of 58 members supporting 17,000 schools in 46 countries, it aims to share expertise, knowledge, data and evidence, shaping and influencing global dialogue. It organizes conferences promoting low-fee private schools in the Global South and advocates for them to be elevated on political agendas (Global Schools Forum, 2021).

Edify, a faith-based social enterprise that works to improve and expand Christian education globally, also promotes low-fee private schools. It is active in Burkina Faso, Ethiopia, Ghana, Liberia, Rwanda and Uganda, in sub-Saharan Africa; the Dominican Republic, Guatemala and Peru, in Latin America; and south-western India. Its pilot programme in Uganda, begun in 2019, reaches almost 700 schools (Sivasubramaniam, 2021). Edify Ghana also works with about 700 schools (Brion, 2020) and provides capital for them to expand, partnering with local micro lenders, including Sinapi Aba, a local Christian micro-lending institution. It has also contributed to classroom construction in the Omega private school chain. Edify provides teacher training, and accounting and school leadership training to school proprietors (Sivasubramaniam, 2021).

Members of the faith-based International Society for Krishna Consciousness have held management positions in Akshaya Patra, a non-profit organization that runs the world's largest non-state school lunch programme, serving 1.8 million children in 19,000 government and government-aided schools in at least 14 Indian states (Srinivasaraju, 2020). The organization and menus adhere strictly to religious values. Akshaya Patra has been criticized, and entered into disputes with state governments, over allegations that its food is not culturally relevant for the communities served and does not meet contractual nutrition guidelines. Nevertheless, state governments continue to contract with the organization, whose advantages include its large scale, competitive prices, donor funding and 20 years of experience (Nathan, 2019).

Religion in education is a divisive issue in much of the world. Grass-roots religious organizations have emerged to advocate on curriculum and learning materials. In Peru, groups including National Pro-Family Coordination and Parents in Action, using the slogan 'Don't mess with my children', demanded the withdrawal of a gender-based national curriculum in 2016 (Muñoz, 2020). In 2012, prior to the curriculum's introduction, the government had introduced a national textbook repository to which 8,500 private schools were to upload textbooks. But, backed by these grass-roots organizations, many schools refused to comply (Ciriaco, 2019).

The Escola sem Partido (Non-partisan School) movement in Brazil was begun in 2004 by activists who believe that teachers are too often politically militant and that parents have a right to make sure their children receive a moral and religious education (Fernandes and Ferreira, 2021). The movement has developed draft legislation for the local, state and federal levels (Escola Sem Partido, 2021), inspiring several municipalities to approve laws prohibiting schools from discussing gender in classrooms (Cancion, 2018). A 2016 Alagoas state law banned political indoctrination in classrooms and introduced ethics courses for teachers (Brazil Ministry of Education and Culture, 2016; Jungueira, 2016). These laws have been challenged in state and federal courts as a form of teacher censorship and judged unconstitutional in some cases (Cancion, 2018; Cappelli, 2020). Still, some politicians have encouraged students to film and report teachers, leading teachers to censor themselves and avoid controversial topics for fear of reprisal (Fagundez, 2018).

In the Canadian province of Ontario, various movements and networks have campaigned, so far in vain, to broaden public funding of non-state denominational schools beyond Catholic private schools. In 1985, for historical and constitutional reasons, the government decided to extend public funding to the end of secondary school for Ontario's Catholic schools but not those of other religious groups. Despite an appeal to the Supreme Court, the decision was found to be constitutional (Dickinson and Dolmage, 1996; MacLellan, 2012). In the 1980s, the Multi-Faith Coalition for Equity in Education brought together Hindu, Sikh, Muslim, Mennonite and Reform Protestant parents, the Canadian Jewish Congress and the Ontario Alliance of Christian Schools to lobby for state funding to be extended to non-Catholic faith-based private schools meeting provincial requirements. Other associations, including the Canadian Civil Liberties Association, advocate for constitutional amendments to eliminate public funding to religious schools.

# THE BUSINESS SECTOR EXERTS STRONG INFLUENCE ON EDUCATION

In 2013, The Washington Post reported that the investment bank IBIS Capital estimated the size of the 'global education market' then at US\$4.4 trillion and projected a value of US\$6.3 trillion by 2017, which was described as '[q]ood news for those who see school reform as a way to make big bucks' (Strauss, 2013). As this report shows (Chapter 4 and Chapter 21), references to global education spending as a market are misleading, as most of it is unrelated to market opportunities: 70% of spending is allocated by governments, mostly to salaries, and much of the remaining 30%, which households spend, goes to routine costs. Textual analysis of reports by leading service providers, such as HolonIQ, EdTechX and Emerge Education, shows that they use language depicting education technology as essential and the firms as enablers and disruptors. When optimistic projections are not fulfilled (e.g. by 2019 the IBIS Capital estimate had reached US\$5.3 trillion), responsibility is implicitly placed on governments (Mármol Queraltó, 2021) as a way of maintaining indirect pressure on them to increase procurement.

Corporate investment in education has been increasing, albeit from a low base. Venture capital investment grew from US\$2 billion in 2014 to US\$4 billion in 2018, concentrated in China (50%), the United States (20%), India (10%) and Europe (8%) (HolonIQ, 2019). Before COVID-19 struck, global primary and secondary education testing and assessment procurement was expected to grow by US\$6 billion between 2020 and 2024 (with some 45% of the growth in Northern America) as a result of personalized learning analytics linked with

### 66

Corporate providers' claims on the effectiveness of digital learning products can be misleading

access to big data and collaboration between schools and assessment companies (Technavio Research, 2020). COVID-19 has since been called the 'mover and shaker' of the market (Educapital, 2020) and education technology one of the few areas, alongside logistics, entertainment and data protection, that could see increased venture capital interest (KPMG, 2020). Digital learning platform providers responded promptly to support the switch to online learning, especially in richer countries, and helped maintain learning continuity. Governments need to ensure user privacy in this context (Coleman, 2021).

99

Questions remain on the best products for schools and teachers to provide remote learning options during school closures. There is a wider issue as well: Corporate providers' claims on the effectiveness of digital learning products can be misleading. Government needs to help ensure their quality, especially when public funds are used to help schools procure them. In England (United Kingdom), the British Educational Suppliers Association, with support from the Department for Education, has a website giving schools and educators access to products. Recognizing that 'finding the right EdTech for your school can be difficult', it assures users that each featured product 'comes complete with a case study showcasing how other teachers have used the product'. The website also features products and services for India's Central Board of Secondary Education (LendED, 2021).

In the United States, the 2002 No Child Left Behind Act required districts and schools to use programmes backed by 'scientifically-based research' to qualify for federal funding. Its successor, the Every Student Succeeds Act of 2015, added a requirement to use 'evidence-based interventions'. However, companies exercise strong influence over studies that evaluate their products, then promote their effectiveness through websites, brochures and trade fairs. There is considerable need for independent quality assurance.

The US government established a What Works Clearinghouse (Mayo-Wilson et al., 2021; Polanin et al., 2021). Such efforts are not without critics. Some raised process issues, questioning the authority of federal 66 -

In Germany, the civil society initiative Hochschulwatch collates information in a public database on the EUR 1.5 billion that universities receive annually from industry sponsorship and research fundings

agencies, which rely on private contractors, to select which research findings to showcase (Schoenfeld, 2006). Others questioned the ability to draw conclusions from small numbers of studies, even if good standards were respected (Reeves and Lin, 2020; Slavin, 2008). A major criticism is that, of the Clearinghouse's 10,600-plus product review studies by early 2020, only 188 products, less than 2%, had been classified as strongly or moderately effective (García Mathewson and Butrymowicz, 2020).

Academic research is expected to generate knowledge that informs debates. However, network analysis of citation patterns suggests research findings' uptake can be selective and biased (Verger et al., 2019). The framing of research, for instance whether education challenges are presented in terms of a 'learning crisis' or a 'narrowing of education', and topic selection, such as whether the focus is on teacher accountability or teacher professional development, influence findings.

Policy-oriented research can be heavily influenced by funders (Reckhow and Tompkins-Stange, 2018). While industry funding may be necessary for research, transparency is needed to avoid undue influence. In Germany, the civil society initiative Hochschulwatch (Higher Education Watch) collates information in a public database on the EUR 1.5 billion that universities receive annually from industry sponsorship and research funding, and on industry representation on institutions' boards (Hochschulwatch, 2021). University sponsorship is unregulated at the federal level. German states take different approaches. North Rhine-Westphalia and Bremen are developing rules for higher education institutions to declare industry partnerships. In Bavaria, universities are expressly excluded from the state's official report on sponsorship contributions to public budgets. Together with the Stifterverband, an association of corporate, philanthropic and private

research funders, Hochschulwatch develops best practice guidance, awareness building for conflicts of interest, self-audit tools and codes of conduct for sponsored chairs. It aims ultimately to ensure that higher education institutions are included in the remit of public freedom of information and public accountability legislation and that third-party funding agreements are publicly available.

The private sector is a significant player in education in other ways. The Japan Business Federation (Keidanren) represents over 1,400 mainly large companies listed on the First Section of the Tokyo Stock Exchange, (Keidanren, 2021). Along with the Japan Chamber of Commerce and Industry and the Japan Association of Corporate Executives, it has a strong influence on policymaking. Keidanren's vice chairman heads the Central Council for Education under the Ministry of Education, Culture, Sports, Science and Technology (MEXT, 2021). In the mid-2000s, the federation expressed strong support for school choice, competition and for-profit education, along with mandatory standardized assessment and, a proposal opposed by the government, the publication of results to inform consumers (Takayama, 2013). Keidanren's latest proposals for primary and secondary education reform cover a broad range of issues, from coding to global citizenship (Keidanren, 2020). Its recommendations include greater investment in the GIGA School programme, through which the ministry aims to develop digital talent in collaboration with the private sector to meet the needs of an increasingly diverse society and address widening disparity (MEXT, 2020).

For-profit actors use lobbyists to promote commercial interests in public policy, including through the narrative they use to describe education. Over time, through policy feedback and lock-in effects, the state becomes dependent on these actors. In Sweden, once enacted, deregulation of education led to a self-reinforcing process, changing public opinion on the delivery of public service and giving business instrumental power over public policy (Busemeyer and Thelen, 2020).

School sponsorships are much debated. Financial and in-kind contributions can support beneficiary schools, but the benefits can be reversed if marketing accompanies the support (Bakir et al., 2017). In an Australian survey of teachers and head teachers, some

66

School sponsorships are much debated. Financial and in-kind contributions can support beneficiary schools, but the benefits can be reversed if marketing accompanies the support

99

99

supported in-kind provision of education technology and sports equipment but most opposed financial support from supermarket and fast food chains in exchange for use of companies' logos (Hogan et al., 2018). In China, the tobacco industry's corporate social responsibility activities undermine tobacco control efforts. As part of the Project Hope rural development programme, the industry sponsored more than 100 primary schools, with new school buildings featuring company logos and pro-tobacco messages. A study of schools in Yunnan province found that parents and teachers perceived no negative impact on children from such sponsorship (Fang et al., 2020).

In the Philippines, the Adopt-a-School programme provides tax incentives to private actors in exchange for financial support to schools. While partner companies, including Coca-Cola, San Miguel and Stanfilco, cannot promote their products in school as signatories of the Philippines Pledge on in-school food marketing, participation criteria do not explicitly prohibit alternative promotion strategies or brand visibility (Reeve et al., 2018). As part of Brigada Eskwela, a government-supported volunteer programme, the snack company Mondelez supports 19 schools but has supplied its products to 40 other schools as part of school feeding initiatives (Vivas, 2020).

The Global Business Coalition for Education (GBC-Education) calls on its more than 200 members' expertise, leadership and resources to make education politically prominent at the national and global levels. Its areas of interest include early childhood, youth skills, emergencies and disability inclusion. It describes four pathways to business engagement in education: investing financial resources through 'global funding tools' and domestic financing for education; delivering in-kind and technical support, including through employee engagement; shaping global strategies and domestic policies to ensure that business has a voice on education; and championing education and becoming thought leaders (Global Business Coalition for Education, 2021). In the area of global financing, GBC-Education has supported efforts by the UN Secretary-General's Special Envoy for Global Education, Gordon Brown, to establish the International Finance Facility for Education, an innovative approach that would expand development banks' ability to lend middle-income countries money for education projects and make loan terms more attractive for countries (Global Business Coalition for Education, 2019). Some observers say coalition members' impact might be greater if they focused on promoting a culture of transparency in tax affairs (ActionAid, 2018) (Chapter 21).

Mobilizing innovative financing is the starting point of the Education Outcomes Fund (EOF), an initiative with roots in venture capital and private equity. Supported by the Global Steering Group for Impact Investment (GSG, 2019), it focuses on explaining the advantages of paying for results in education (GSG, 2021). While a results orientation is an aid effectiveness principle, paying for results is not, and it goes against the principle of country ownership (Chapter 21) (Savedoff, 2019). The approach assumes that countries need incentives when in fact theu are financially constrained. Unlike in other development interventions, where the link between inputs and results is observable, it is not straightforward in education. Finally, the potential for unintended consequences is considerable if efforts are focused on one education indicator (UNESCO. 2018). UNICEF, which has hosted EOF since 2020, has long embraced public-private partnerships. The private sector accounts for 22% of its revenue (UNICEF, 2020), including corporate sponsorship.

Cooperation between the United Nations and the business sector is based on principles included in two documents: the Global Compact, which provides an overall value framework (human rights, labour standards, environment and anti-corruption), and the Guiding Principles on Business and Human Rights, endorsed by the Human Rights Council (United Nations, 2015). Individual UN organizations, such as the Food and Agriculture Organization and the World Health Organization, have publicly available strategies and guidelines for staff and partners on engagement with non-state actors, including private entities. Such guidance addresses risks, including conflict of interest, influence on norms and standard setting, credibility and reputation, and whitewashing of partners' images (FAO, 2013; WHO, 2016, 2018a, 2018b). Outside the UN framework. the Global Partnership for Education also has a detailed private-sector engagement strategy (GPE, 2019).

### INTERNATIONAL ORGANIZATIONS AND FOUNDATIONS HAVE PARTICULAR INFLUENCE

The World Bank influences policy design and implementation. World Bank loan conditions have shaped government programme operations, and its pilot projects have influenced other development institutions' strategies. Regarding private education, World Bank positions have for decades influenced policy directions on school fees and schooling provision by private actors. In the 1980s, fees in public education were seen as desirable. The official World Bank approach

### 66

The World Bank's use of technical assistance, project reports, research studies and international events strengthens its position as a knowledge broker

to private participation shifted in the early 2000s, with the group explicitly withdrawing support of school fees after strong opposition from other international actors, notably civil society. With respect to private provision, World Bank education strategies have evolved from strongly endorsing private participation to acknowledging risks but still recommending public-private partnerships as a useful strategy (Edwards Jr et al., 2021).

The World Bank's use of technical assistance, project reports, research studies and international events strengthens its position as a knowledge broker. A module on private-sector engagement, which is among the group's top 13 policy areas for promoting learning under the Systems Approach for Better Education Results, has as its premise that 'private educators provide a significant contribution to education and that improved interaction between government and private schools is essential for increasing equity and quality' (Baum et al., 2014, p. 8). A World Bank analysis of 10 countries recommended expanding private provision in all but Zambia; increasing public funding of private actors in all but Eswatini and Zambia; and reducing regulations in all but Bangladesh, Mauritania, Eswatini and Zambia (DeCoster, 2019).

Philanthropic organizations also influence policy through various means. In the United States they have promoted charter schools through knowledge production, support for advocacy campaigns and dissemination of good practices (Verger et al., 2016). The Bill and Melinda Gates Foundation is among those backing the charter school movement; its financial support to the campaign by the Washington Coalition for Public Charter Schools was key to passage of the 2012 charter school ballot initiative in Washington state (Au and Lubienski, 2016). Challenges for foundations engaging in education policy include aligning their efforts with national education policy processes (i.e. avoiding being seen as pushing their own agendas), engaging with multiple stakeholders and finding the right political moment. Brazil's Lemann Foundation mobilized broad consultation in pursuit of

# BOX 5.4:

99

#### Non-state actors influenced the introduction of minimum standards in Brazil

Brazil has a high degree of socioeconomic and education inequality. A decentralized structure, with responsibility for basic education delivery mostly given to municipalities, means disadvantaged local governments struggle to catch up with richer ones. The seventh goal of the National Education Plan 2014–24 provided an impetus to address this challenge (OECD, 2021).

In 2013, the Lemann Foundation, a Brazilian philanthropic organization, helped establish and fund the Movimento pela Base Nacional Comum (Movement for a National Common Base), which was instrumental in the introduction of national learning standards (Costin and Pontual, 2020). Members succeeded in getting the National Education Plan amended to include a schedule for establishing such standards. The foundation organized events advocating for the standards' introduction, developed knowledge products and pitched them to officials involved in drafting, and built a network of politicians and experts to support the standards (Tarlau and Moeller, 2019), creating an influential network of policy ideas (Avelar and Ball, 2019). Throughout, the foundation's favouring of technical over political debate allowed it to influence the process through its expertise (Tarlau and Moeller, 2019).

Proponents of the standards argued that they would lead to high academic expectations for all, fostering equity in education. Opponents maintained that standards alone would not suffice, given the wide inequality in distribution of human and financial resources in the education system (Ready, 2018). The first version of the Base Nacional Comum Curricular (BNCC, National Common Curricular Base) was followed by an online consultation with 12 million contributions (Costin and Pontual, 2020); a technical team of university specialists then integrated comments. The BNCC's second version was published at the height of a political crisis in 2016. Before the change of government, control over drafting was transferred from the national to regional and local levels, which helped the process survive the change. Local seminars were held to collect teacher feedback. The BNCC was adopted in 2017, offering a strong example of how non-state actors' influence over education reforms has increased (Avelar and Ball, 2019).

the goal of national core curricular standards (**Box 5.4**). An approach in Uruguay favoured a greater role for private actors in legislation (**Box 5.5**).

Various non-profit organizations focus on influencing public education policy concerning innovation. Since 2017, HundrED, a global non-profit organization based

# BOX 5.5:

#### Building support for private education in Uruguay

Uruguay's share of private institutions in enrolment is 12% in secondary education, well below the average globally and for Latin America, perhaps due in part to early separation of church and state. Public schooling has moreover had a strong symbolic role in nation building in a context of high immigration, becoming associated with democratization and social cohesion. Private actors have played little part in education governance. Unlike in Argentina, Chile and Colombia, the regulatory framework has not encouraged participation of private actors in education (Bordoli et al., 2017).

The Administración Nacional de Educación Pública (ANEP, National Public Education Administration), the country's highest education authority, is an autonomous body established as a guarantor of education planning independence from political meddling. A feature of the governance structure is participation of teacher unions. Unions have blocked private actors' involvement in education in the past (Marrero, 2014; Moschetti et al., 2019). Critics believe ANEP is not fully autonomous and its purpose is not being served, since political stand-offs affecting education still occur (Bogliaccini and Rodríguez, 2015).

Since the early 2000s, a tax exemption regime has favoured the private sector and new regulations have allowed public–private partnerships relating to education provision, along with a type of indirect public funding for private education actors through tax exemption of donations (Bordoli et al., 2017). These developments have led to opportunities to shift the discourse to support private actors in education.

In recent years, new stakeholders, such as the citizen initiative Eduy21 and the Centre for the Study of Economic and Social Reality (CERES), a think tank with links to the Brookings Institution, a US think tank, have been pressing for market-friendly education policy, maintaining that the public education system has failed to reform to meet contemporary needs (Moschetti et al., 2019). CERES and Eduy21 have organized events at various levels to build a policy network and gather media attention to their narrative that education is in crisis. In 2017, CERES began holding citizen meetings to propose policy solutions on education resulting from policy research seeking evidence for their stance on education reform (Moschetti et al., 2019).

This change in approach is evident in a sweeping July 2020 emergency law, whose 476 articles include 78 on education. The law changed core aspects of education, primarily concerning the approach to public education. For instance, the word 'public' no longer appears in references to what is now known as the 'national education system'. Private actors are now included in coordination mechanisms and departmental education commissions.

Article 14 of the 2008 Education Law stipulated: 'No agreement or treaty, bilateral or multilateral, will be signed with any State or international organization that directly or indirectly would entail consideration of education as a for-profit service or encouraging its commercialization' (Uruguay Ministry of Education and Culture, 2008). Article 129 of the new law states: 'No bilateral or multilateral agreement or treaty shall be signed with States or international organizations that reduce education to the status of for-profit service' (Uruguay Government, 2020). Other articles restrict participation by social actors in education policy formulation and propose regulating teachers' work (Martinis, 2020).

in Finland, has been reviewing, selecting, compiling and disseminating innovations in areas including collaborative learning, creative and critical thinking, project-based learning and play. Each year, it identifies and features 100 innovations for their impact and potential to be scaled up; 69% of those selected in 2021 were from non-profit organizations. HundrED examines what enables efficient adoption of innovations in public education systems and how systems can support innovations' implementation (Leponiemi et al., 2020).

Other organizations focus on enabling system change in various areas. In Côte d'Ivoire, the Jacobs Foundation, as part of its Transforming Education in Cocoa Communities programme, worked with the government to scale up a remedial education programme to improve foundational literacy and numeracy skills (Curtiss Wyss and Perlman Robinson, 2021). In India, Peepul, in a public-private partnership with the Delhi municipal corporations, runs three model schools serving marginalized children to incubate innovative ideas with the aim of eventually testing them in public schools. Three other programmes, also run in partnership with state governments, focus on teacher skills, accountability and incentives, with mentoring a part of the approach (Peepul India, 2020).

Another area in which NGOs have tried to influence public education policy agendas is financial education. In the United States, Jump\$tart, a non-profit coalition, has developed financial education standards for primary and secondary education curricula (Alsemgeest, 2015).

# In 2019, US charitable contributions to education reached US\$64 billion

99

The Council for Economic Education has developed national standards for grades 4, 8 and 12 and has worked with schools and districts to design and implement financial literacy curricula, while providing its own programmes to students and educators (Kasman et al., 2018). As of early 2020, 21 states had included the subject in higher education curricula; about the same number were discussing bills on introducing financial education in schools (Carrns, 2021).

The growth of wealth in some of the world's largest economies can be linked to technology companies, which place a special emphasis on education. Technology philanthropists, given their firms' large scale and universal access to their products and platforms, can quickly have a far-reaching impact (Patil and Brakman Reiser, 2021). In 2019, US charitable contributions to education reached US\$64 billion. In China, education consistently receives the greatest share of philanthropic support. The rapidly growing philanthropic sector is dominated by corporate giving from technology firms, which both contribute funds and use their platforms to engage and encourage donors. Corporate contributions to education reached US\$23 billion in 2017. Charles Chen Yidan, the driving force behind the Tencent Foundation and corporate philanthropic activities, focuses on higher education.

Data on philanthropic financial flows (Chapter 4) underestimate philanthropies' influence and impact in leveraging networks, technology platforms and products for policy advocacy (Patil and Brakman Reiser, 2021). Philanthropic foundations do not just fund specific activities but also have strategic interests. For instance, a survey of 44 foundations found that 39 were also engaged in advocacy. Increasingly, they are interested in long-term impact and sustainability: 14 mentioned that they often collaborated with government and 18 with donors; others transitioned to for-profit models or developed grantee financing models. Philanthropic work has also become much more data-driven and evidence-based: two thirds of the foundations had an evaluation person, unit or department and had conducted monitoring and evaluation on all their programmes and grants. But foundations rarely disclose information readily (OECD, 2018).

In parallel with philanthropic foundations focusing more on non-profit work, corporate activity with philanthropic intent has grown (Srivastava, 2020). In the United States, the philanthropic sector has close ties with the business sector and thus is strongly framed around corporate profitability. 'Venture philanthropy' and 'charitable companies' are terms used for hybrid forms of funding, combining investment capital logic with philanthropic objectives and activities, such as education. They are characterized by high involvement of investors, who often provide intellectual and social capital on top of financial resources.

The highest-giving venture philanthropies include a broad mix of foundations from family fortunes. These mega-philanthropies fund other education-related venture philanthropies, such as the Charter School Growth Fund and the NewSchools Venture Fund. Venture philanthropies include not only national foundations built on individual family fortunes but also local philanthropic efforts of business leader collectives, such as Chicago's Renaissance Schools Fund (now Kids First Chicago) and the Chicago Public Education Fund. The Chan Zuckerberg Initiative focuses heavily on education, particularly personalized, technology-assisted learning. A grantee, Summit Public Schools, has developed project-based teaching and student-directed learning, which have been used by nearly 400 schools in 38 states and the District of Columbia (Pane, 2018; Schultz, 2018).

Some foundations promoting a market approach to education support third-party mobilization, presented falsely as grass-roots initiatives. Such front groups claim to represent specific agendas, such as those of parents and students, but in fact pursue the interests of another organization (Camera, 2020). In the United States, Parent Revolution primarily receives philanthropic funding, which is used to hire private firms to collect petitions or parent signatures to lobby for changes in education (Lubienski et al., 2012). A project investigating the role of policy advocacy networks in five US cities showed that funders used intermediary organizations to connect knowledge producers with knowledge users and that various strategies of the networks aimed to incentivize non-state actors' actions (Lubienski, 2019).

Non-state actors often target parliamentary committees. In the two US federal legislative bodies, hearings are a key direct opportunity to influence policy (Wang, 2020). Committees, which are groups of specialist members, and caucuses, whose members are committed to common aims, also review proposed legislation. For instance, the Congressional Caucus on Education The media play a major role in shaping opinion about education policy, representing actors' interests or serving as channels for them through coverage of events and views

Innovation and Opportunity 'seeks to highlight how applying market-based principles to education can enhance student achievement' (Legistorm, 2020). In 2017, the Black Caucus expressed concern that the incoming secretary of education could 'fundamentally weaken the U.S. public school system' (Congressional Black Caucus, 2017) because of previous promotion and sponsorship of a policy favouring school choice in Michigan state (Michigan Campaign Finance Network, 2006). Most lobbying takes place at state level. Coalitions of school choice supporters, mainly foundations and rich donors, and of opponents, largely from the labour movement, have played major roles in state legislation (Kirst, 2007; Vergari, 2007). But various actors' positions have evolved considerably, especially as more data became available, forcing advocacy to be more evidence-based (DeBray-Pelot and McGuinn, 2009).

Global Home Education Exchange is an international NGO supporting homeschooling (GHEX, 2020). One member, the US-based Home School Legal Defense Association, is the largest such advocacy organization, representing the interests of families who opt for homeschooling. Founded in the early 1980s, it assists more than 100,000 member families through personalized support, resources and advocacy. It has a global outreach team mapping relevant legislation (HSLDA, 2021), finances advocacy in various countries (McShane, 2020) and provides technical assistance on drafting legislation. Panama's January 2021 adoption of a law recognizing home education reportedly followed testimony by association representatives at the National Assembly (Donnelly, 2021; Lara, 2021).

In Italy, Agorà della Parità is a forum of more than 50 non-profit organizations of managers and parents in 'independent schools with parity', a type of hybrid public-private school whose diplomas are recognized as equivalent to those of public schools and which serves 10% of the total compulsory school-age population (Istat, 2019). In 2021, they campaigned to amend the COVID-19 emergency funds legislation to include non-state schools, as it was initially reserved for the public education system (Linkiesta, 2021). As a result, the law allocated EUR 60 million to independent schools with parity, in addition to EUR 350 million for public schools, conditional on transparency measures requiring the schools to publish data on their assets, spending and personnel (Italy Parliament, 2021). Allocation of public funds to independent schools with parity, more than half of which are Catholic, is a divisive subject in Italy. The state does not directly support non-state schools' expenses, but independent schools with parity have been recognized by law as providers of a public educational service (Huffington Post, 2020).

,,

The media play a major role in shaping opinion about education policy, representing actors' interests or serving as channels for them through coverage of events and views, but also through the language they employ, for instance on market opportunities in education (Connell, 2013; Olmedo, 2013). In Chile, two mainstream media outlets, La Tercera and El Mercurio, supported the voucher system (Grau and Olmedo, 2012) and depicted a stronger role for the state as hampering education freedom (Cabalin, 2015). Some media networks have had direct links with the education industry, e.g. the Financial Times and its former owner, the textbook publisher Pearson (Verger et al., 2016). In India, English-language newspapers and the entertainment industry have traditionally reflected affluent audiences' interests, covering topics such as private school fees and tutoring. English news broadcasts on Indian television often feature experts who support private education and invite private school heads and sector entrepreneurs to comment on national policy (Thapliyal, 2018).

### **CONCLUSION**

Various actors take part in the often tense debate on public education and non-state provision. The language employed by proponents and opponents of a greater role for non-state actors is carefully selected to appeal to public opinion. Using such narratives, actors exercise influence through advocacy, lobbying, research, financing and marketing. The challenging question for government officials is whether sufficient mechanisms are in place to ensure that such activities are legitimate, that the transparency of the public education policy process is not compromised and that narrow vested interests do not gain ground at the expense of equity and inclusion for learners.
Children play and learn at a reading club facilitated by a preschool teacher from Save the Children in a rural village in Malungon, Mindanao, Philippines.

CREDIT: Save the Children



#### **CHAPTER**



# Early childhood care and education

# **KEY MESSAGES**

#### Non-state actors dominate education for children under age 3.

- In 2018, as a proportion of total enrolment among children under 3, private institutions accounted for 57% in high-income countries and 46% in middle-income countries.
- Laws require at least some employers to support or provide childcare for employees in 26 of 189 countries. But work-based ECCE is linked to formal employment, which accounts for only 30% of the employed in low- and middle-income countries.

#### Non-state provision is increasing in pre-primary education.

- The share of private institutions in total pre-primary education enrolment increased from 28.5% in 2000 to 37% in 2019.
- In Oceania, some countries have close to 100% of preschool students enrolled in non-state institutions. In Samoa, 62% of centres belonged to missions and 38% were private.
- Trends vary. In some countries, such as Algeria and Colombia, non-state provision is decreasing, while in others, such as China and Peru, it is increasing.

#### Non-state provision of early childhood care and education remains unaffordable for families.

- In Ghana, private provision costs 6% of annual consumption for the richest and 17% for the poorest; the equivalents in Ethiopia are 4% and 21%.
- In OECD countries, the net childcare cost for a two-earner family with two children aged 2 and 3 was 17% of women's average earnings in 2019, ranging from zero in Germany and Italy to about one third in Ireland and Slovakia, and half in Japan and the United Kingdom. In the United States, it is 56% for a Black family.

#### Most countries struggle to govern fragmented provision systems.

- Many countries lack comprehensive regulatory frameworks or fail to implement them.
- Regulations tend to focus on administrative requirements related to registration, approval and licensing. Non-state providers register as business or commercial entities in some countries. In Lusaka, Zambia, centres must register through the City Council's health and commercial sectors and their education orientation is not considered.
- In 42 countries, multiple authorities are responsible for quality assurance. Non-state providers may even be subject to separate quality assurance processes, as in the Philippines.

#### Non-state actors receive and spend funds, but the costs still often fall to families.

- Governments provide subsidies, grants, taxes and incentives, such as start-up funding or public land, to encourage non-state provision. A voucher programme in the Bahamas is provided only to quality-approved private preschools.
- In total, 79 countries regulate ECCE fees. Norway allows private kindergartens to have a 'reasonable' profit but any government grants and fees received must go towards meeting government objectives and conditions to benefit children.

Non-state actors lead care and education services for children under 3	137
Non-state actors are more prominent in pre-primary than in basic education	39
Non-state provision challenges governance and regulation1	44
Public funding of non-state provision takes different forms1	50
Non-state actors innovate and advocate for ECCE1	153
Conclusion 1	155

n the first few years after birth, children experience their most rapid growth in cognitive development, yet are at their most vulnerable and fragile. Care and education of young children have traditionally been the responsibility of the family, particularly women. In recent years, however, access to early childhood care and education (ECCE) services of good quality has been increasingly recognized as providing children, especially disadvantaged children, with additional developmental benefits, such as socioemotional and cognitive stimulation. ECCE services can also free mothers' time for formal work, making such services' development even more urgent, given the potential economic gains for families and economies.

66

By 2030, the number of children up to primary school entry age who need childcare but lack access is expected to increase by 25 million in low- and lowermiddle-income countries

By 2030, the number of children up to primary school entry age who need childcare but lack access is expected to increase by 25 million in low- and lower-middle-income countries (Devercelli and Beaton-Day, 2020). Low enrolment in pre-primary education in these countries is linked to lack of supply rather than of demand. Many families are willing to pay for access to ECCE services. Gaps in provision have led to the entry of private providers, non-governmental organizations (NGOs), faith-based organizations and community groups. But a strong non-state role in ECCE provision risks entrenching unequal access at disadvantaged groups' expense. Governments worldwide are gradually coming to terms with the need to expand free and compulsory ECCE to improve access by reducing costs, along with monitoring whether non-state providers meet standards. As ECCE services continue to grow, understanding the opportunities and challenges that non-state actors present is critical.

This chapter reflects the themes covered in this report in the context of early childhood. It explores how non-state actors operate as providers, businesses, funders, influencers and innovators. It also discusses the shifting role of government as ECCE systems expand. Attention is paid to children under age 3 who attend day care, nursery or playgroups and to children from age 3 up to school starting age who attend kindergarten or preschool. Dimensions of quality and equity are discussed throughout.

#### NON-STATE ACTORS LEAD CARE AND EDUCATION SERVICES FOR CHILDREN UNDER 3

Standardized, comparable data on care and education services for the youngest children do not exist for many countries, especially in comparison with other education levels. This is due to the variety and, often, the informality of some types of provision. In principle, programmes have an adequate education component, i.e. services provided by trained or accredited staff with pedagogical qualifications, for at least 2 hours per day and 100 days a year. Whether available in school-, centre- or home-based settings, they should be governed by a regulatory framework recognized by national authorities. However, some countries recognize programmes that are integral to their ECCE systems but whose education component is inadequate. Other countries have informal settings or unregistered ECCE services, which are not included in the data but are vital to some families from an education perspective.

In high-income countries, non-state actors have dominated care and education services for the youngest children. Private institutions in 33 high-income countries accounted for 57% of total enrolment for children under 3 in 2018. The phenomenon has historical roots. In France and Italy, private charities and churches introduced ECCE services before the state gradually enacted legislation, introducing and expanding its services (Kamerman, 2006).

In Australia, Ireland, the Netherlands, New Zealand and the United Kingdom, the for-profit private sector is mainly responsible for non-state ECCE provision. In New Zealand, private institutions account for 99% of enrolment for children under 3. Among them, the share of for-profit private-owned services has increased from 23% in 2002 to 41% in 2019 at the expense of community-owned services (Gallagher, 2017; Neuwelt-Kearns and Ritchie, 2020). In the United Kingdom, 82% of young children are enrolled in private institutions. In England, private companies accounted for 755,000 childcare places or 46% of the total in 2019. In that year, the value of the childcare market was estimated at GBP 6.7 billion with private for-profit providers accounting for 82% of the total (Department of Education, 2019; LaingBuisson, 2020).

66 -

Private institutions in 33 high-income countries accounted for 57% of total enrolment for children under 3 in 2018

99

Faith-based organizations and NGOs are the main providers in many countries. In Germany, 73% of enrolment was in private institutions in 2017. About one third of providers for children under 3 were Catholic or Protestant church programmes (Blome, 2018; Strehmel, 2019). Another third were NGOs. Only 3% of providers were in the for-profit sector (European Commission/EACEA/Eurydice, 2019).

Home-based services are a major part of ECCE provision in several countries (Kaneko et al., 2020). In 2012, 30% of children under 3 in the United States were in home-based services: 15% with an unpaid provider with whom the child had a prior relationship, 7% with a paid provider with a prior relationship and 7% with a paid provider with no prior relationship (Paschall, 2019). In 2019, 26% of children under age 3 in the European Union received childcare provided by a professional childminder at the child's home or at the childminder's home as well as care provided by other relatives, friends or neighbours: 19% for less than 30 hours and 7% for 30 hours per week or more (Eurostat, 2021). In France, childminders are the main childcare providers for children under 3. Families pay directly but can receive a subsidy through the childcare supplement of the childcare allowance, whose size is based on the family income (European Commission/EACEA/Eurydice, 2021b, 2021c).

Although state provision dominates in Finland, enrolment with private providers increased from 13% in 2013 to 24% in 2019. While private provision includes local businesses and non-profit providers, the introduction of chains has fuelled its growth. The combined revenue of the three biggest for-profit chains increased from EUR 46 million in 2015 to EUR 146 million in 2019 (Ruutiainen et al., 2021). This rise is linked to a parental allowance for private services and increased outsourcing of ECCE services by municipalities seeking to reduce costs and promote parental choice based on interest areas (e.g. language and music), alternative pedagogies (e.g. Montessori) and location convenience (Kumpulainen, 2018). Israel's early childhood educational development gross enrolment ratio almost doubled from 33% in 2013 to 62% in 2018. But only 25% of infants and toddlers were in certified day-care centres, which are the responsibility of the Ministry of Labour, Social Affairs and Social Services and must meet specific criteria (Vaknin, 2020).

While the share of non-state provision has remained stable or slowly increased in many high-income countries, it has decreased in Chile and Denmark. In Chile, the share of private enrolment decreased from 30% in 2013 to 10% in 2018, while the gross enrolment ratio In 33 middle-income countries, non-state actors account for 46% of enrolment in early childhood educational development programmes

increased from 19% to 25%. The Chile Crece Contigo (Chile Grows with You) programme was introduced in 2009 to extend access to ECCE provision to vulnerable children under age 4 (Chile Government, 2021). About 23% of all centres are managed by the private, non-profit Integra Foundation, which receives state funding directly and through an administrative agreement (Chile Undersecretary of Early Childhood Education, 2019).

In 33 middle-income countries, 19% of children under 3 are enrolled in early childhood educational

development programmes; non-state actors account for 46% of enrolment, with national shares ranging from less than 2% in Azerbaijan, the Russian Federation and Ukraine to 100% in Dominica and Turkey, albeit at low enrolment levels. Overall, low- and middle-income countries' governments seldom sponsor childcare, which is considered unaffordable in light of other budget pressures (Penn, 2021). Employer-based provision, relatively common in richer countries, is only gradually emerging in poorer countries (Box 6.1). Trends vary in ECCE provision for children under 3. Some countries, such as El Salvador, have been shifting towards greater state provision, while in others non-state actors dominate. In Jamaica, all ECCE services are run by for-profit, religious and not-for-profit private providers (World Bank, 2019a). In South Africa, where 38% of children under 4 are in formal childcare according to the 2018 General Household Survey, anecdotal evidence suggests most provision is private and often not registered, especially for disadvantaged populations (Alfers, 2016; Statistics South Africa, 2019).

#### BOX 6.1:

#### Employer-based day-care services are sometimes mandated but not available to most children

Employer-based ECCE services and incentives include childcare vouchers, subsidies, breastfeeding support and on- or off-site childcare centres run or sponsored by the employer. Employers providing childcare programmes can benefit from tax incentives through deductions or credits and from subsidies, in-kind support and a positive public image (IFC, 2019; UNESCO, 2020). Laws require at least some employers to support or provide childcare for their employees in 26 of 189 countries. In some countries, employers must provide ECCE services for children until a certain age, e.g. age 2 in Chile and Paraguay (IFC, 2019).

99

In Bhutan, work-based childcare centres involve a partnership among companies, UNICEF and the Ministry of Education (Tshomo, 2017). Operational guidelines state that the centres may not charge fees other than those approved by the Ministry of Education (Rao et al., 2020). India's Maternity Benefit Act (1961, 2017 Amendment) requires all employers with 50 or more employees to provide childcare services for children under age 5 on company premises or in the employees' community. The government provides tax incentives, implementation guidelines and sanctions for non-compliance, but employers say they lack guidance on quality issues, such as curriculum, standards and selection of third-party providers (IFC and Bright Horizons, 2019). In Sri Lanka, the 1939 Maternity Benefits Ordinance obliges employers 'with a prescribed number of women workers' to establish and maintain a crèche (IFC et al., 2018). In practice, employers tend to outsource ECCE services to private preschool providers. Some employers have developed a 'workplace consortium model' to share operational costs (Warnasuriya et al., 2020).

Mandates for such services are often not enforced. In Cambodia, employers with at least 100 female employees are supposed to establish a day-care centre or cover employees' day-care costs. An assessment of factory compliance found that 43% had no functioning nursing room or day-care facility and paid no childcare allowance. Another 38% had no functioning nursing room but paid a childcare allowance, though in more than 40% of those cases, payments were incorrect (ILO and IFC, 2018). Another survey found that just 22% of firms with more than 100 female employees had on-site or nearby childcare, and that among the 47% of companies that reported paying a childcare allowance, the amount ranged from US\$3 to US\$25 per child per month (IFC, 2020).

The main challenge is still the fact that work-based ECCE services are linked to formal sector employment, which accounts for 39% of the world's employed population and 30% in low- and middle-income countries (Devercelli and Beaton-Day, 2020; ILO, 2018). In Ghana, private institutions account for 98% of the gross enrolment ratio in ECCE services for young children. In 42 markets in Accra, women who work as porters, street vendors and traders have limited childcare options, with just 7 childcare centres, of which 3 are for-profit. The cost can be high, especially without support from the municipality or the Market Traders Association (ILO and WIEGO, 2019). The need for childcare provision in African cities' informal settlements, such as those in Nairobi, Kenya, is a major policy concern (Hughes et al., 2021). In Gisenyi, Rwanda, UNICEF and Action pour le Développement du Peuple, a national NGO, established six ECCE centres near markets where mothers working as traders across the border in the Democratic Republic of the Congo can leave their children under the supervision of trained caregivers (UNICEF et al., 2021).

Small-scale, community-based childcare programmes also provide ECCE. In Uganda, 7% of ECCE centres are community-based (Uganda Ministry of Education and Sports, 2017).In some countries, governments have embraced and developed such programmes to leverage the potential of an approach that integrates care, education, health and nutrition (Hayden and Wai, 2013). In Latin America, such programmes became popular in the 1980s in Colombia (Hogares Comunitarios de Bienestar), Guatemala (Programas Hogares Comunitarios) and Nicaragua (PAININ) (Diker, 2001). In Peru, where 17% of enrolment was in private institutions in 2019, government and communities jointly run the Cuna Más home visiting programme, offering childcare, education playgroups and comprehensive care such as nutrition, safety, protection

and health (Josephson et al., 2017).

Informal arrangements with relatives such as grandparents and siblings or with friends, neighbours, babysitters or nannies may be the only option available. In Organisation for Economic Co-operation and Development (OECD) countries, 26% of children under 3 are in informal childcare arrangements (OECD, 2019b), especially in countries with little ECCE provision (Figure 6.1). In the United Kingdom, 35% of children were in informal childcare. Grandparents were the most common carers, sometimes in combination with formal provision, which could suggest after-school care (Simon et al., 2015). However, these informal arrangements are not included in the definition of ECCE. Untrained nannies and domestic workers may not be able to provide developmentally appropriate care. About 10% of domestic workers in the United States are nannies. With a median age of 26, they are likely to be women and receive the lowest pay among all domestic workers (Wolfe et al., 2020).

In the poorest countries, many children receive little to no care. In Chad and the Democratic Republic of the Congo, half of children under 5 had been left alone or with a sibling in the past week. The use of siblings as caregivers can hamper the older sibling's schooling and right to play, and siblings' inexperience can have negative implications for the young child's learning and well-being (Gromada et al., 2020).

In OECD countries, 26% of children under 3 are in informal childcare arrangements

99

#### FIGURE 6.1:

## Scarce provision of ECCE services is associated with a higher share of informal childcare

Proportion of children under 3 using informal childcare arrangements and gross enrolment ratio of early childhood educational development programmes, selected countries, 2017



Note: Data on childcare are for 2009 in the Republic of Korea, 2011 in the United States, 2014 in Malta and Switzerland and 2015 in Iceland. Informal childcare refers to unpaid care, except in Australia and the Republic of Korea, where it means care by relatives, friends or neighbours, paid or unpaid. Early childhood educational development programmes are typically designed for children under 3.

GEM StatLink: https://bit.ly/GEM2021-fig6\_1 Source: OECD (2019b) and UIS database.

#### NON-STATE ACTORS ARE MORE PROMINENT IN PRE-PRIMARY THAN IN BASIC EDUCATION

Between 2000 and 2019, the share of private institutions in total pre-primary education enrolment increased from 28.5% to 37%, a much higher proportion than in primary (19%) or secondary (27%) education. Shares range from less than 1% in Eastern European countries, including Belarus, the Republic of Moldova and Ukraine, to more than 95% in many countries, notably including Small Island Developing States in the Caribbean (e.g. Anguilla, Antigua and Barbuda, and the British Virgin Islands) and the Pacific (e.g. Samoa, Tonga and Vanuatu) (**Figure 6.2**).

66

#### **FIGURE 6.2:**

The share of private institutions in early childhood enrolment is high, especially for the youngest children

Proportion of children enrolled in private institutions, pre-primary education, 2019 or latest year



As this increase in the share of private institutions occurred, the pre-primary education gross enrolment ratio increased from 34% to 62%, with enrolment in absolute terms doubling from 106 million to 214 million. Eastern and South-eastern Asia, at 55%, has the highest share of private institutions in total enrolment. By contrast, about one quarter of children in Central and Southern Asia, Europe and Northern America, and Latin America and the Caribbean are enrolled in private pre-primary institutions (**Figure 6.3**).

In Oceania, some countries have close to 100% of preschool students enrolled in private institutions, at various levels of coverage: gross enrolment ratios exceed 90% in Australia, New Zealand and Vanuatu but are below 50% in Samoa and Tonga (**Figure 6.4**). Much of the provision in Pacific island states, as in Vanuatu, is community-based; community members contribute to infrastructure, classroom resources and teacher salaries (UNICEF, 2017; World Bank, 2012). In Samoa, 62% of early childhood education centres belonged to missions and 38% were private. The government provides curriculum and teacher support and finances these non-state providers through the One Government Grant (Samoa Ministry of Education Sports and Culture, 2019, 2020).

Eastern and South-eastern Asian countries have made big efforts to increase pre-primary education enrolment with different strategies. Since the 1990s, private and NGO kindergartens in China have supplanted state-run kindergartens under the 'walking on two legs' policy (Li et al., 2016). The pre-primary gross enrolment ratio increased from 43% in 2006 to 89% in 2019, while the share of private institutions increased from 31% to 57%. By contrast, in Viet Nam, where the gross enrolment ratio rose from 33% in 1995 to 96% in 2019, the share of private enrolment fell from a high of 60% in 2003 to a low of 12% in 2014 before recovering slightly to 16% by 2019 (Figure 6.5). Since 2002, the government has increased funding to public kindergartens, focusing on the most vulnerable children (Boud and Phuong Thao, 2017). A similar change in the Philippines came with the universalization of pre-primary enrolment, which began in 2012/13 (Philippines Congress, 2012), leading to a decrease in the share of private institutions from 49% in 2000 to 12% in 2019.

Among world regions, Northern Africa and Western Asia, led by Algeria and Egypt, recorded the largest drop in the share of private institutions in pre-primary

#### FIGURE 6.3:

The role of private institutions in pre-primary education expansion varies by country and region

Share of private institutions in pre-primary education enrolment, by region, 2000–19



#### FIGURE 6.4:

## In Oceania, most countries rely on non-state pre-primary education providers

Share of private institutions in pre-primary education enrolment and pre-primary education gross enrolment ratio, selected countries in Oceania, 2019 or latest year



Between 2000 and 2019, the share of private institutions in total pre-primary education enrolment increased from 28.5% to 37%

education enrolment, from 53% in 2000 to 36% in 2019, even as the gross enrolment rate almost doubled from 17% to 32%. The share of private institutions also fell in Jordan and Morocco, from 100% in 2000 to 72% and 84%, respectively, in 2019. In Jordan, the private sector generally provides kindergarten for children under 5 while the state also provides kindergarten the year before primary school (Jordan Ministry of Education and UNICEF, 2020). By contrast, between 2000 and 2018 the share of private institutions in pre-primary education increased in Israel from 5% to 36% and in Kuwait from 26% to 45%.

Other regions have maintained a stable share of private enrolment in pre-primary education. Private enrolment has remained at about 25% in Latin America and the Caribbean, although the average masks divergent paths. Peru's share of private provision doubled from 16% in 2000 to 31% in 2013. But in Colombia the share of private provision fell by half, from 41% in 2000 to 21% in 2019, due to a strategy, law and corresponding funding supported mainly by NGOs (Cárdenas and Cadena, 2020).

Unlike in primary and secondary education, the share of private pre-primary provision has remained stable in sub-Saharan Africa at about one in three students, while the enrolment ratio has been growing. In some countries where enrolment growth was fast, public provision was prioritized. In Ethiopia, the gross enrolment ratio increased from 4% in 2010 to 29% in 2015, while the share of private enrolment fell from 95% to 18% following the introduction of the National Policy Framework for Early Childhood Care and Education in 2010. The policy promoted the O-Class or pre-primary classes attached to government primary schools (Kim et al., forthcoming). In the United Republic of Tanzania, which introduced free preschool in 2014, the pre-primary gross enrolment ratio rose from 33% in 2014 to 42% in 2019, while the share of private enrolment increased only slightly, from 5% to 7%.

#### FIGURE 6.5:

"

The role of private institutions in pre-primary education expansion varies by country and region Share of private institutions in pre-primary education enrolment and pre-primary education gross enrolment ratio, selected countries in Eastern and South-eastern Asia and in Latin America, 2000–19



99

Official administrative data seldom capture unofficial non-state providers. Household surveys show that administrative data underestimated the share of non-state enrolment in six of seven sub-Saharan African countries (Baum, 2021; Baum et al., 2018; King et al., 2020). On average, the difference between the two sources was 20 percentage points. In Malawi, the share of private institutions in the pre-primary education enrolment ratio was 7% according to administrative data but 63% according to household survey data in 2016–17 (**Figure 6.6**). Unregistered providers pose concern about quality, governance, regulation and equity.

#### FIGURE 6.6:

# A substantial share of pre-primary enrolment is in unregistered non-state institutions

Share of non-state institutions in pre-primary enrolment, household and administrative data, selected countries, 2010s



Note: In Nigeria, the most recent available administrative data on non-state enrollment come from 2010/11, eight years prior to the household data collection; hence, the gap should be interpreted with caution.

GEM StatLink: https://bit.ly/GEM2021-fig6\_6 Source: Baum (2021).

#### RURAL AND POORER CHILDREN HAVE MUCH LOWER ACCESS TO NON-STATE PROVISION

Non-state provision has mainly addressed demand in urban areas, where such services tend to be more commonly available, and from richer households, which can afford them. High-income countries also suffer from what is known in Northern America as 'childcare deserts': Provision for young children in rural communities is likely to be home-based or non-existent. In the United States, three in five rural communities lack adequate childcare supply and mainly rely on home-based care (Home Grown, 2020; Malik et al., 2018).

#### 66

A review of nine low- and middle-income countries found that the probability of attending private preschool was significantly higher for urban than for rural children in five countries

99

A review of nine low- and middle-income countries for this report found that the probability of attending private preschool was significantly higher for urban than for rural children in five countries (Ecuador, Ghana, Nigeria, South Africa and Uganda) (Baum, 2021). Yet, within urban areas, households may have access only to private arrangements. For instance, the gross enrolment ratio in private kindergartens was well above the national average (10%) in the three urban regions of Ethiopia in 2019/20: Addis Ababa (104%), Dire Dawa (31%) and Harari (45%) (Ethiopia Federal Ministry of Education, 2020). Poorer households living in informal settlements usually have no access to public childcare at all (Caddu, 2017). For instance, 94% of providers in Mukuru, a large slum in Nairobi, Kenya, were private; households had access to five private facilities within walking distance, on average (Bidwell and Watine, 2014).

The cost of private pre-primary education can be high. Families spend 9.4% of their annual budget on it in Ghana (compared with 5.1% for those whose children attend public preschool) and 4.9% in Ethiopia, on average. For poor households, the cost is usually unaffordable. As a share of annual consumption, it is 6% for the richest and 17% for the poorest in Ghana; the equivalents in Ethiopia are 4% and 21% (Figure 6.7). Non-state pre-primary fees may vary by type of provider. In Morogoro, United Republic of Tanzania, the annual fee, in purchasing power parity terms, is US\$444 in for-profit, US\$302 in faith-based, US\$253 in community and US\$246 in NGO schools. Tuition varied by school registration and approval status, class size, pupil/teacher ratio and percentage of certified teachers. Children from the richest 20% of households were 2.5 times more likely to attend private provision than their peers in the poorest 20% (Baum, 2021).

Poor families face the same challenge in high-income countries. On average, in OECD countries, the net childcare cost for a two-earner family with two children aged 2 and 3 was 17% of women's average earnings in 2019. This cost varies greatly across countries, from zero in the Czech Republic, Germany and Italy to about one third in Ireland, Slovakia and Switzerland, and one half in Japan and the United Kingdom. Some of these countries target support to effectively eliminate the cost for to poorer families (e.g. Switzerland), while in others the cost is as high for low as for median earners (e.g. Slovakia). Despite some measures, in countries with high private provision, such as Ireland and the United Kingdom, the net childcare cost for poor single-parent families is 25% of women's earnings (OECD, 2020).

In countries that provide little support, families make the difficult choice between paying for ECCE services or forgoing salary to care for children. In the United States, parents spend about US\$42 billion on ECCE, while the federal, state and local governments spend US\$34 billion (Gould and Blair, 2020). The total annual cost to households is about US\$9,600 per child under 5, representing 18% of median household income or 64% for individuals earning the minimum wage (Schulte and Durana, 2016). A median-income Black family with two young children would have to spend 56% of its income on childcare, much higher than any other group (Novoa, 2020).

#### FIGURE 6.7:

# The cost of non-state pre-primary provision can be too high for the poorest

Median private pre-primary education cost for one child as a share of average household consumption expenditure, by household consumption quintile, selected sub-Saharan African countries, 2010s



#### 66

In the United States, a median-income Black family with two young children would have to spend 56% of its income on childcare

Few ECCE centre leaders in 9 OECD countries reported that at least 1 in 10 of their students was from a disadvantaged socioeconomic background. But in four of those countries (Chile, Denmark, Israel and Turkey), the percentage of leaders reporting they had children from disadvantaged backgrounds was significantly higher in public than in private ECCE centres (OECD, 2019c). The Netherlands has two-tiered ECCE provision for the youngest children: Wealthier areas have access to fee-paying, full-time private childcare while part-time playgroups are more common in poorer areas (Penn, 2019).

#### NON-STATE PROVISION CHALLENGES GOVERNANCE AND REGULATION

The range and number of non-state actors in ECCE and the volume of non-state provision complicate both governance and regulation. Yet good governance and regulation arrangements can improve access, equity and quality and prevent the emergence of parallel systems linked to type of provider and parental ability to pay.

# STRONG GOVERNANCE IS NEEDED WHEN NON-STATE ACTORS DOMINATE PROVISION

ECCE involves a complex set of services that most countries struggle to govern well, especially for children under 3. Good governance demands strong coordination, collaboration and cooperation that is horizontal (between sectors) and vertical (i.e. responsibilities decentralized to regional and local actors) (Britto et al., 2017; Yoshikawa et al., 2018). However, how non-state actors are governed varies by country. The multiplicity of non-state actors and volume of non-state provision make governance ever more complex, ranging from single sector and centralized to mixed, multilayered and multisector models.

Governance may be led by the education ministry, another ministry (e.g. social policy or women's affairs) or an agency, sometimes linked to the president's office, notably in Latin America (e.g. Chile, Colombia and Uruguay) but also elsewhere (e.g. the Philippines and Seychelles) (Loizillon and Leclercq, 2016; Marzonetto and Rodríguez, 2017; OECD, 2016; Seychelles Government, 2011; Vargas-Barón, 2015). In Japan, the Cabinet Office led implementation of the Comprehensive Support System for Children and Childcare to ensure

99

consistency between education and health ministry regulations (Japan Government, 2014). In Singapore, the independent Early Childhood Development Agency, overseen by the Ministries of Education and of Social and Family Development, governs and regulates non-state provision for children under age 7 in both kindergartens and childcare centres (ECDA, 2020).

Faith-based provision often leads to fragmented governance. In Mali, the Ministry of National Education is responsible for preschools while the Ministry of Territorial Administration and Local Government supervises Koranic schools. ECCE in Somalia is delivered through traditional Koranic schools, integrated Koranic schools, and privately owned kindergartens and nursery schools run by local NGOs and private foundations in large urban centres. The federal Ministry of Education, Culture and Higher Education is responsible for ECCE policy, service standard guidelines, monitoring and evaluation, while the Ministry of Religious Affairs develops integrated Islamic religion-based ECCE curriculum and appoints and supports training of ECCE trainers (PEER country profiles).

Decentralization adds further complexity. Clear roles and responsibilities are needed to avoid overlaps and misalignment. In Norway, governance of state and non-state providers is delegated to municipalities, which can decide whether to increase the share of private provision to address parental demand (Norway Ministry of Education and Research, 2014). In the United States, the fragmented governance structure, coupled with a reliance on private providers, has produced different regulatory frameworks and separate support systems at federal, state and district levels (Karch, 2013; Miller et al., 2017).

Governments engage non-state actors through partnerships. The Bangladesh Early Childhood Development Network, a partnership of 172 members from government, NGOs and international organizations, was established in 2005 for advocacy and stronger cooperation through information sharing, technical guidance and capacity building. It helped develop the 2008 Operational Framework for Universal Pre-primary Education, which included national standards and focused on introducing a year of pre-primary education with clear roles and responsibilities for non-state actors to complement government capacity (Zahar and Khondker, 2017). In Brazil, Rede Nacional Primeira Infância (National Early Childhood Network), established in 2007, increased its membership from 10 to more than 200 organizations in about 10 years. It aims to have the National Early Childhood Plan referenced at all levels of government by 2022 (Rede Nacional Primeira Infância, 2018). Costa Rica's Red Nacional de Cuido y Desarrollo Infantil (National Care and Child Development Network) was established by law in 2014 to bring together various providers and ministries; however, efforts to ensure cohesion and coordination are still needed (FLACSO, 2020).

Engaging community-level stakeholders, including parents, can support governance at the local level (Vargas-Barón, 2015). Analysis for this report found that several countries, including Bhutan, Côte d'Ivoire, Lesotho, Morocco, New Zealand, Niger, the Republic of Korea, Togo and Tunisia, expected ECCE centres to include parents in their decision making through management committees or elected representatives.

Non-state providers also govern themselves. In France (Fédération Française des Entreprises de Crèches, 2021) and the Netherlands (Expertisecentrum Kinderopvang, 2021), associations of non-state providers have created platforms to monitor centres' quality and track progress. In the United Kingdom, the National Day Nurseries Association, a charity representing private nurseries' interests, has quality assurance regimes emphasizing regulation compliance, administrative procedures and personal development of staff (National Day Nurseries Association, 2019). In the United States, the National Association for the Education of Young Children accredits programmes through its assessment system to promote quality and improvement of ECCE services and inform parents' choices (NAEYC, 2021).

# THE QUALITY OF NON-STATE PROVIDERS IS HIGHLY VARIABLE

Quality refers to structural and process factors linked to positive child outcomes. Structural factors are inputs, such as educator/child ratios, class or group sizes, teacher qualifications, materials, physical environments and infrastructure, safety, and attention to water, sanitation and hygiene. Process factors are related to learning and include activities children engage in,

66

The quality of provision is related not to provider type but to country context and policy environment

"

stimulation and staff-child interactions, all of which are difficult to measure (UNESCO, 2016b). The challenge for policymakers is to set standards and ensure they are met so as to minimize variation in quality between types of providers and within each provider type.

Evidence is split on whether non-state providers deliver services of good quality. In practice, the answer is related not to provider type but to country context and policy environment. In Brazil, the value of the infrastructure quality index was significantly higher in private contracted crèches and preschools than in public ones (Evans and Kosec, 2012). By contrast, a study in rural India found more toys and games in public centres (Gupta, 2020). In Mongolia, private kindergartens scored lower on a measure of classroom quality (e.g. programme structure, activities, literacy and math) but higher on a measure of interaction quality (World Bank, 2017). In Portugal and the United States, public provision was characterized by higher levels of process quality (Slot, 2018). Public ECCE centres for 3- to 6-year-olds in OECD countries overall provided better staff-child interactions than private ones (OECD, 2018).

Studies on structural factors in sub-Saharan Africa also had contradictory results. A survey of 37 preschools in Addis Ababa, Ethiopia, found that private preschools, which were in the majority, performed better overall but neither private nor public preschools met standards (Admas, 2019). In Ghana, private preschools had better facilities, which could be considered a result of the 2011 National Policy on Public-Private Partnerships (Pesando et al., 2020). In Kenya, no significant differences between public and private centres were found: Classroom and furniture quality was substandard in both. Public centres had better water, sanitation and play facilities, however, while private providers did not meet government service standard guidelines (Sitati et al., 2016). In Nigeria, for-profit private schools had higher pupil/teacher ratios (Baum, 2021).

High-income countries tend to have regulations on educator qualifications, certification and working conditions. In Finland, the Early Childhood Education and Care Act, which regulates all providers, ensures there are usually no significant differences in workforce education levels between publicly and privately managed centres (Finland Government, 2018). Spain, on the other hand, has introduced minimum qualifications for public but not private centre heads, except in Asturias region (European Commission/EACEA/Eurydice, 2019). Studies have not reached consistent conclusions on teaching practices. In Denmark, staff used more practices associated with process quality in publicly managed preschools than in privately managed facilities. In Iceland, less facilitation of numeracy development was reported in public than in private centres (OECD, 2019c).

In many low- and middle-income countries, private ECCE educators tend to be less prepared and have fewer professional development opportunities than public-sector peers. Only 8% of private but 75% of public kindergarten teachers do the Ghana Education Service's training programme (Ghana Ministry of Finance, 2019), as there are no minimum requirements for private ECCE teachers (Wolf et al, 2018). A study of 139 schools in Punjab, Pakistan, found that more pre-primary teachers in government schools had higher levels of education, even though teachers generally had low levels of education and received little specialized ECCE training (Jamil and Saeed, 2018). In Zambia, private preschool teachers increasingly are certified by unaccredited private colleges, while teachers in public preschools must be certified through the Zambia Preschools Association (Edwards et al., 2019b).

Research on teaching practices is scarce relative to high-income countries. In Ghana, public preschools scored higher for 'praising children for positive behaviours, stimulating classroom interactions, promoting cooperative learning, and using specific tools to facilitate learning' (Pesando et al., 2020). There may be differences within the private sector. In Kenya, approved private preschools scored significantly higher than unapproved schools on teacher-student interaction indicators, such as wait time between activities, teachers' responses to children's questions, teachers' use of open-ended questions and teachers helping students work through problems (Baum, 2021).

Many countries aim to ensure that non-state provision complies fully or partially with the national curriculum or learning standards. Denmark's curriculum emphasizes the importance of play, curiosity and social relations and applies to both state and non-state providers (Denmark Ministry of Children and Education, 2020). The Philippines encourages use of a holistic curriculum for children up to age 4 (Philippines Congress, 2013; Philippines ECCD Council, 2015) and monitors national standards and competencies through the government-validated Philippine Early Childhood Development Checklist (Philippines ECCD Council, 2019). Introducing a curriculum appropriate for a given development stage and context is important (Joo et al., 2020). Tunisia prohibits formal instruction in private preschools of the kind delivered in primary schools (PEER country profiles). In India, public preschools used more play-based activities, while private preschools had a more formal instructional style (Gupta, 2020). However, the National Education Policy approved in 2020 introduced a foundational stage for 3- to 8-year-olds, which will include 'aspects of more formal but interactive classroom learning' (India Ministry of Human Resource Development, 2020, p. 11).

The use of English as medium of instruction in non-state preschools, often featured in marketing, is another example of tension between developmentally appropriate curricula and popular perceptions of quality. In Brazil, where the Common Base National Curriculum envisages teaching of a foreign language only from grade 6 onwards, education in English is growing in private early childhood education centres (Padinha and Goia, 2021). In Ghana, most public and private preschools reported English and a local language in their instruction, while 22% of private schools but no public schools reported teaching exclusively in English (Wolf et al., 2018).

#### **REGULATIONS HAVE A WEAK FOCUS ON QUALITY**

As ECCE providers are diverse, regulatory frameworks should aim to define minimum standards and set quality assurance mechanisms to monitor quality and equity of provision (Ponguta et al., 2019; UNICEF, 2019). Countries increasingly mention non-state actors in their policy documents. A review of regulations for this report looked at their focus on establishment, financing, quality, accountability and sanctions.

Some countries have distinct regulations for provision for children under 3 and age 3 and up. In Hong Kong, China, the Department of Social Welfare supervises care centres for children under 3 based on regulations of the 1997 Child Care Services Ordinance, amended in 2020, and the 1976 Child Care Services Regulations. The Education Bureau supervises kindergartens for older children, mostly under the 1971 Education Ordinance and Education Regulations, which are similar to those that apply to schools. In the United Republic of Tanzania, non-state provision for children under 4 is regulated by the Ministry of Health, Community Development, Gender, Elderly and Children and the 2019 Law of the Child Act, while pre-primary education is regulated by the Ministry of Education, Science and Technology and the 1978 National Education Act (PEER country profiles). Multiple sets of regulations from different ministries can be a source of contradiction and discontinuity, for instance in Latin America (CLADE and OMEP, 2020).

In some countries, non-state ECCE provision is regulated within the same framework as higher levels. Bahrain's Private Educational and Training Institutions Law regulates non-state nurseries, kindergartens, and primary and secondary schools (Bahrain Government, 1998). The regulatory framework for early childhood to secondary education in the Dominican Republic is defined in the 2000 Private Education Institutions regulation and 1997 General Education Law (PEER country profiles). In Ukraine, where preschool education covers children aged 1 to 6 and enrolment in private ECCE centres is low, non-state providers are governed by the same education law as other levels and school types (PEER country profiles).

Many countries lack comprehensive regulatory frameworks or fail to implement them. In Eswatini, the Multi-Sectoral Early Childhood Development and Education Framework 2018-22 contains registration quidelines and minimum standards applicable to all ECCE services, irrespective of ownership, but is yet to be formally adopted (Eswatini Government, 2018; World Bank, 2021). In Sri Lanka, absence of a multisector regulatory framework means the scope of work of several ministries (education, health, and women and child affairs) and provincial councils overlaps (UNESCO, 2016a). High-income countries are not immune to such challenges. A study of 55 childcare providers in New York state, United States, revealed that providers faced conflicting or arbitrary interpretations of regulations by local licensors and inspectors as well as lack of discussion or reconsideration when disagreements emerged (Shdaimah et al., 2018).

66 -

Many countries aim to ensure that non-state provision complies fully or partially with the national curriculum or learning standards

"

# Regulations tend to focus on administrative requirements

Analysis for this report shows that regulations on the registration, approval and licensing of non-state providers exist in at least 181 of 196 countries. Standards and requirements variably cover health and safety standards, teacher qualifications and training requirements, group size or pupil/teacher ratios, financing, and zoning and infrastructure standards.

In Australia, a person may apply for provider approval to the regulatory authority of the state or territory and pay the prescribed fees. The person's criminal and financial history are examined to assess character and fitness. The authority must refuse approval if it believes the service would pose an unacceptable risk to children's safety, health or well-being. Once granted, approval remains in force until cancelled in the event of mismanagement, risk to safety or non-compliance with the law. In Bosnia and Herzegovina's Sarajevo canton, a preschool can be established with prior government consent and approval from the Ministry of Education as long as it has enough children to form at least two classes; standard facilities, equipment and teaching aids; employees with standard qualifications; and justification for its establishment, such as a need for a preschool in a particular area.

In Mauritania, the Pre-school Education Service of the Ministry of Social Affairs, Children and the Family is responsible for registering and approving public and private kindergartens and preschools. Applications for private kindergartens must include a request addressed to the ministry, a morality investigation by police, a criminal record review, a medical certificate and a study of the project describing the building, security, equipment and location, ensuring that it is situated far from traffic crossings and petrol stations, and at least 500 metres from the closest similar facility (PEER country profiles).

In some cases, entry and establishment requirements may differ for children under 3. A review of 100 countries' regulations on childcare quality and safety standards found them likely to be more comprehensive and focused on quality for children of pre-primary school age than for those under 3 (World Bank, 2019b). Israel's 2018 Daycare Supervision Law has requirements for obtaining a licence and key conditions for every private day-care centre for children under 3 (The Knesset, 2019).

Entry and operation requirements may differ by provider type. Cambodia has separate regulations and decrees for community preschools. In Chile, a 2010 decree states

#### 66

In 181 out of 196 countries, there are regulations on the registration, approval and licensing of non-state providers

that by 2022, state-subsidized centres must obtain an official recognition certificate and comply with certain technical, pedagogical, legal and infrastructure requirements, while non-aided, independent centres must obtain authorization to operate (PEER country profiles). In Kyrgyzstan, the government lowered licensing requirements and introduced public-private partnerships and private-sector contracting (UN Women, 2019). Non-state providers, including part-time preschool programmes in Saskatchewan, Canada, and ECCE centres in Kazakhstan, may even be exempt from entry and establishment requirements (PEER country profiles). Certain requirements can be counterproductive. In South Africa, entry requirements act as a disincentive for potential non-state providers to enter and operate officially, but unregistered operation can lead to guality, safety and equity issues (BRIDGE et al., 2020).

Non-state providers register as business or commercial entities in some countries. In Lusaka, Zambia, low-fee private centres must register through the City Council and its health and commercial sectors. They can operate as 'commercial enterprises' whose education orientation is not considered (Edwards et al., 2019).

#### Quality assurance mechanisms tend to be superficial

Quality assurance mechanisms, such as internal and external inspections, can support state and non-state actors in monitoring and strengthening the quality of non-state provision (Neuman and Devercelli, 2013). Analysis for this report shows that the education ministry is responsible for assuring the quality of non-state ECCE provision in 162 of 192 countries; in 42 countries, multiple authorities are in fact responsible for quality assurance at this level.

Croatia's Early Childhood Education and Care Act has a quality assurance process that covers state and non-state actors and is used in place of a licensing system. The Ministry of Science and Education carries out centre inspections, including expert pedagogical supervision and health inspections. In the Czech Republic, registered kindergartens are regularly inspected and must report on enrolment, staff and children (PEER country profiles).

In 162 of 192 countries, the education ministry is responsible for assuring the quality of non-state ECCE provision

99

Singapore's Nurturing Early Learners Kindergarten Curriculum Framework (revised in 2012), Early Years Development Framework (launched in 2011) and Singapore Preschool Quality Assurance Framework and Quality Rating Scale are used to monitor and assess programme quality (Tan, 2017). In the United Arab Emirates, the Educational Inspection Directorate (Early Childhood Educational Institutions) oversees quality assurance and carries out general compliance inspections, focusing on compliance with key regulations on inputs, process and provision quality, and targeted compliance inspections, which respond to complaints or observations (United Arab Emirates Ministry of Education, 2018).

While the level of responsibility for assuring quality is usually national, many European countries assign this task locally. Local authorities must assess service provider plans before granting operating licences in Germany; they develop regulations and procedures in Lithuania. Most European countries also require regular internal evaluations (European Commission/EACEA/Eurydice, 2019).

Child assessment is rarely used. In Belgium, the French Community's Reference System of Initial Skills proposes learning benchmarks for children aged 2.5 to 5, encouraging evaluation based on observation (European Commission/EACEA/Eurydice, 2021a; Wallonia-Brussels Federation, 2020). An analysis of 10 South-eastern Asian countries found that countries mostly used checklists and observations to assess children's learning. In Brunei Darussalam, children in both state and non-state schools took written and verbal tests (SEAMEO INNOTECH, 2020). In South Africa, assessment is based on the National Integrated Early Learning and Development Standards and the South African National Curriculum Framework for Children from Birth to Four, which provide key developmental results in six areas: communication, exploration, well-being, identity and belonging, creativity, and knowledge and understanding (PEER country profiles).

Regulations in low- and middle-income countries generally specify minimum standards, including for learning materials. For instance, Nepal's Curriculum Development Centre must approve books used by non-state centres (PEER country profiles). But regulations usually do not establish quality assurance mechanisms, and those that do tend not to focus on outcomes. In Brazil, requirements for teacher qualifications and class size are not actively enforced and many private preschools remain independent and unregulated (Bastos and Straume, 2013). Mongolia inspects private kindergartens to ensure compliance with infrastructure and safety standards, and audits their finances, but pays little attention to the learning environment, teaching practices or quality outcomes (World Bank, 2017).

Some countries have followed a structured approach to regulating, monitoring and facilitating improvement of private ECCE provision. In Jamaica, where provision is largely in non-state actors' hands, the Ministry of Education, Youth and Information requires inspectors and their senior supervisors to have at least one specialized degree. Additional staff make monthly monitoring site visits and conduct teacher training. School oversight focuses on 12 national standards and refers to both structural and process dimensions, including interactions and relationships among children, teachers, parents, caregivers and community members. The Early Childhood Commission's information system tracks compliance with the standards (Anderson et al., 2017; Araujo et al., 2013).

In Sri Lanka, 79% of pre-primary education students were in private institutions in 2018. Early Childhood Development Standards for 3- to 5-year-olds were approved in 2017. The Children's Secretariat trained officers and preschool educators, provided them with detailed guidelines and supported them with a system to assess achievement of learning and developmental milestones, which had been implemented in 3,800 Early Childhood Development Centres by mid-2020 (Warnasuriya et al., 2020).

Non-state providers may be subject to separate quality assurance processes. In the Philippines, the recognition and accreditation process of private centres for children under 4 involves a three-month internal assessment and a two-day external assessment that cover health and nutrition, safety, physical environment, interactions and relationships between staff and children, staff qualifications and development, curriculum, parental involvement and centre management (Philippines ECCD Council, 2015). Each centre is rated as satisfactory, very satisfactory or outstanding (PEER country profiles).

Inspection coverage of registered non-state providers varies. In Lagos, Nigeria, where the quality assurance system covers such providers, the Lagos State Ministry of Education's Quality Assurance Department has inspected 56% of private preschools, with those charging higher

fees more likely to have been inspected (68%) than those charging lower fees (48%). In Nairobi, Kenya, community schools are inspected more often than religious, charity, or for-profit schools. However, many providers are unregistered and thus not inspected (Baum, 2021).

Overall, ECCE regulations tend to neglect equity issues. More stringent regulations on quality may even exacerbate inequality. A survey of 141 ECCE providers in Istanbul, Turkey, found that stricter structural regulations, measured by the requirement to have a garden on the premises, was significantly associated with higher fees and lower percentages of children from poorer households enrolled (Aran et al., 2016).

#### Sanctions are applied for violation of standards

Analysis for this report indicates that 148 of 194 countries have regulations to sanction providers that do not comply with government standards. Some include a process allowing providers to redress issues within a certain period; if there is no improvement, their licence can be revoked or centre closed.

In Bahrain, kindergartens violating regulations receive a warning and 10 days for redress. If they fail in this, or repeat the violation, they are placed under Ministry of Education supervision and the provider loses ownership until the violation is corrected. Ecuador requires an annual inspection to assess whether the national curriculum is used and learning objectives are met. If a centre is found to be in violation, sanctions are applied, including written reprimand, temporary suspension, or definitive suspension if children's rights are violated. In Samoa, the Education Act 2009 states that a licence can be withdrawn if a centre no longer meets a registration criterion. The Tunisian government may close centres that fail to comply with an obligation in the 2002 Framework Law, with moral precepts or with hygiene and safety rules. In Uruguay, ECCE centres violating legal and regulatory norms are subject to observation, warning, fines, temporary closure, revocation of authorization and closure. Once an authorization has been revoked, the centre must close within 10 days (PEER country profiles).

In New Zealand, a centre operating without a licence can be fined up to US\$35,000. A licence or certification may be suspended or cancelled if the centre is found not to comply with minimum standards set out in the 2020 Education and Training Act or the 2008 Education (Early Childhood Services) or 2008 Education (Playgroups) regulations. A review officer can inspect the office or other premises of any centre-based or home-based ECCE

#### 66

In 148 of 194 countries, there are regulations to sanction providers that do not comply with government standards

service to assess curriculum delivery, staff qualifications, and health and safety. All adults in home-based centres, whether caregivers or not, are subject to police vetting and document review (PEER country profiles).

#### PUBLIC FUNDING OF NON-STATE PROVISION TAKES DIFFERENT FORMS

Globally, ECCE expenditure data are scarce; for private expenditure, only data for high-income countries are available. Among OECD countries, only in Israel and the United Kingdom do households spend more than governments on early childhood educational development programmes (Figure 6.8a), although fees are a source of ECCE centres' income in most countries. A survey showed that while more than 9 in 10 centres for children under age 3 in Israel received fee income, the same was true for 8 in 10 centres in Germany and almost all centres in Norway (OECD, 2019c). Government accounted for 83% of total pre-primary education spending in 2016 and for more than 50% in all OECD countries except Japan (48%), where private expenditure on pre-primary education came from both households (65%) and businesses and foundations (35%) (Figure 6.8b). In response, the government made preschool education free for all children and childcare free for low-income families in 2019 (Suzuki, 2019).

Governments use subsidies, grants, taxes and various incentives, from start-up funding to public land, to encourage non-state providers to operate in the ECCE sector. A review of regulations for this report found that 131 of 194 countries offered such incentives. It is estimated that 35% of governments support private childcare centres (ILO and WIEGO, 2019). In Europe, the private sector in southern countries, such as Portugal and Spain, tends to be self-financing, although Malta's Free Childcare Scheme is applicable to both state and non-state providers, the aim being to provide equitable services to families irrespective of their socioeconomic background. Entitlement to free childcare hours is based on the employment hours of mothers or single fathers (PEER country profiles). Nordic countries transfer funds to municipal governments, giving them autonomy to decide how to subsidize state and non-state childcare. Denmark has four types of day-care institutions: municipal, self-governing, outsourced and private. A municipality that cannot offer day-care services directly must either fund a day-care setting in another municipality, cover parents' expenses for private day care or give them a subsidu for taking care of their own children. They are required to report the number of children supported to the municipality each year. In 2020, the government agreed on a new parental payment programme for day care to help vulnerable groups affected by the COVID-19 crisis. The programme enables both municipalities and private providers to waive or reduce their fees to households experiencing financial difficulties (PEER country profiles).

In Indonesia, where ECCE enrolment is almost exclusively in private institutions, the government began providing operational subsidies to early childhood education centres in 2011, investing in facility improvement and programme development. However, at US\$42 per child (UNESCO, 2019), this subsidy covers a fraction of the annual cost, which is estimated at US\$151 in informal play groups and US\$256 in formal kindergartens (World Bank, 2020). In Japan, prefecture governments provide subsidies, known as facility-type benefits, to public and private day-care centres, kindergartens and integrated centres. In addition, private day-care centres receive subsidies through municipalities for management costs. Previously, subsidies were provided to parents (Abumiya, 2015).

New Zealand introduced two types of public subsidies in the mid-2000s. The first encourages providers to increase the share of registered teachers and improve quality (New Zealand Ministry of Education, 2020). The second subsidizes centres so they can provide 3- to 5-year-olds up to 6 hours of ECCE per day and up to 20 hours per week without their families being charged additional fees (Education Counts, 2021). An evaluation found that the policy had increased the labour force participation rate of mothers with two children but had unexpectedly decreased the participation and earnings of mothers with one child (Bouchard et al., 2021).

In South Africa, the Department of Social Development provides a subsidy of US\$305 per child per year for those attending registered early childhood development centres and whose families are below an income threshold. It is estimated that the policy reached almost 80% of children in registered centres but just over 30% of all children receiving early childhood care and education, as the majority attend unregistered

#### FIGURE 6.8:

# Public funding of early childhood care and education exceeds private funding in richer countries

Share of public sources in total expenditure (2015) and of private institutions in total enrolment (2018), selected middle- and high-income countries

a. Early childhood educational development programmes



GEM StatLink: https://bit.ly/GEM2021-fig6\_8a



centres, which find registrations conditions onerous (South Africa Department of Social Development, 2020; Wills and Kika-Mistry, 2021). Trinidad and Tobago's Ministry of Education operates and/or assists 138 ECCE centres and also subsidizes teacher and caregiver salaries in 63 centres managed by Service Volunteered for All, an NGO, which is responsible for the remaining operational and infrastructure costs. No support is provided to 691 private centres (Araujo et al., 2013; Trinidad and Tobago Ministry of Education, 2021).

Some governments may use tax exemptions for certain provider types as an incentive. Ukraine's Ministry of Education and Science plans to expand the network of private preschools by 2024 by making preferential arrangements to pay premises' rent and giving providers a tax exemption under Article 197 of the tax code (PEER country profiles).

Government contracting is a common way to provide and fund ECCE services. In Brazil, 8.2% of children in day-care and pre-primary services attended government-contracted private providers, which accounted for 29% of total private provision (INEP, 2020). In the Philippines, the ECCD Program Contracting Scheme enables accredited, community-level private providers to offer services and benefit from subsidies to establish and manage centres and receive technical and operational assistance from the education, social welfare and health departments (PEER country profiles).

Voucher programmes seek to increase access to pre-primary education through non-state providers. Under the Bahamas' Universal Pre-primary Education Initiative, when there are no more places in government-operated schools, vouchers worth US\$2,000 are provided for up to 1,000 children per year and paid directly to approved private preschools that meet national standards (Bahamas Ministry of Finance, 2019). In Hong Kong, China, the Pre-primary Education Voucher Scheme was found to contribute to inequality (Wong and Rao, 2015) and was replaced with the Free Kindergarten Education Policy in 2017, changing it from a demand- to a supply-funded system. Previously, the 84% of kindergartens run by non-profit organizations received per-student vouchers to cover operational costs on condition that fees were regulated. The new policy provides subsidies to kindergartens for half-day services for all 3- to 6-year-olds (Rao and Lau, 2018).

66

### In total, 79 countries regulate ECCE fees

99

Government regulations deter non-state providers from charging high fees and taking advantage of families. Analysis for this report found that 79 countries regulate ECCE fees. Ghana, the Philippines and Viet Nam limit the level of tuition and other fees charged by independent private schools that do not receive government subsidies; they also require them to consult the government about any fee increase (PEER country profiles). In the Philippines, the Early Years Act requires government to monitor tuition through the ECCD Council to ensure fees remain 'affordable' and 'within reasonable limits', although they do not specify a standard fee (Philippines Congress, 2013). Subnational governments also set such restrictions. In Lagos state, Nigeria, pre-primary education tuition and other fees may not exceed US\$120 a year (Baum, 2021).

The extent to which profit orientation is tolerated varies. In Canada, regulations tend not to cover profit orientation but operators must be not-for-profit corporations to get a licence in Newfoundland and Labrador province and Nunavut territory (PEER country profiles). In Iceland, for-profit provision is allowed only if the profit is reinvested in development of the ECCE centre, although loopholes can enable providers to expand activities, for instance through new buildings, that offer other opportunities for profit (Dúrfjörð and Magnúsdóttir, 2016). Ireland's 2013 Child and Family Agency Act bans for-profit provision. In Greece, private centres can be established as for-profit or non-profit entities with provisions corresponding to commercial companies, but the government may provide financial assistance only to centres established by non-profit entities (PEER country profiles). Norway's Kindergarten Act allows private kindergartens to have a 'reasonable' profit but any government grants and fees received must go towards meeting government objectives and conditions to benefit children. South Africa's 2015 National Integrated Early Childhood Development Policy allows for-profit ECCE provision, but such providers qualify for government aid only if they are contracted by the state to provide services to vulnerable groups.

To address equity, some countries implement policies to give children from a specific population or background access to ECCE through direct or conditional cash benefits, in-kind payments, vouchers, parental leave and tax credits. The Netherlands gives tax credits to parents for non-state ECCE (Knijn and Lewis, 2017). Saudi Arabia introduced income-related vouchers to increase enrolment of children from low-income families in registered private ECCE centres; the Qurrah programme allocates childcare vouchers to working women who earn less than US\$2,100 per month (Saudi Arabia Government, 2021). However, evidence from Finland, where regulation aims to ensure equity between different types of ECCE services, shows that admission policies and tuition fees lead to stratification of consumers between the public and private sectors (Ruutiainen et al., 2021).

Expanding public provision may be the most direct route to supporting families. In total, 63 countries have legal provisions for free pre-primary education and 51 for compulsory pre-primary education. Most are located in Europe and Northern America and in Latin America and the Caribbean (UNESCO, 2021).

#### NON-STATE ACTORS INNOVATE AND ADVOCATE FOR ECCE

Non-state actors, such as civil society, grass-roots organizations, academia, foundations and corporations, contribute to ECCE through innovations for access and quality and through profile-raising advocacy.

Historically, committed educationists, such as Friedrich Fröbel, Maria Montessori, Johann Heinrich Pestalozzi and Rudolf Steiner, who built the foundations of early childhood education pedagogy and organization, worked outside or at the margins of the formal public education system to pursue their vision of child-centred and self-directed learning. The groundbreaking Reggio Emilia approach, which emerged in northern Italy in the late 1940s, stemmed from activities of grass-roots organizations embraced by the local municipality (Aljabreen, 2020).

Academic researchers in the United States drew attention to early childhood education programmes' long-term effectiveness, encouraging public authorities to scale up such programmes (Schweinhart, 2016). Scientific research, notably in neuroscience, has confirmed many of the insights of these empirical approaches on developmentally appropriate experiences for young children, notably how life experiences affect the brain and their implications for learning, behaviour and development (Center on the Developing Child, 2021; Lancet, 2016; Stack, 2013).

66

In Slovenia, non-state actors push for inclusion of Roma children in preschool education and their smooth transition into primary school

99

Around the world, governments, including those of Chile and Jordan, involve external, non-state experts to build consensus and ownership of national ECCE strategies and policies under development (Herrera et al., 2016; Sultana, 2009). In North Macedonia, a country with major inequality in access to ECCE, non-state actors, with European Union support, mobilize networks of service providers, civil society, policymakers and teacher training faculties to promote equity and guality in ECCE (Step by Step, 2019). In Slovenia, non-state actors push for inclusion of Roma children in preschool education and their smooth transition into primary school, with engagement of Roma NGOs (Ionescu et al., 2019). In the United States, interest groups contribute to ECCE policy formation (Karch, 2013). NGOs play an important role in promoting ECCE for those left behind (Box 6.2).

But not all non-state actors' influence is positive. Businesses influence families with advertising. Governments do not regulate whether supposed learning products are appropriate for young children's developmental stage (**Box 6.3**).

Workforce unions, including those of childcare workers, are another influence on ECCE. In Canada, the Centrale des sundicats du Québec, whose members includes 10,000 home day-care workers, went on strike in September 2020 to demand higher pay; their wages were below those of untrained workers in day-care centres and even below the minimum statutory level (Education International, 2020). In the European Union, childcare professional unions are consulted on issues affecting their members as part of European Commission efforts to improve employment quality (European Public Service Union, 2017). The Norwegian Union of Teachers and Union for Kindergarten Assistants are among organizations invited to policy discussions on legislative amendments and strategy implementation (Norway Ministru of Education and Research, 2014). The Tanzania Teachers Union. in collaboration with the international teacher union federation. Education International. and with funding from the international NGO Comic Relief, implemented a four-year project in two rural districts to develop a competency profile for ECCE educators, raise their professional teaching standards and support a cadre of about 350 teachers in earning diplomas (Education International, 2021).

International non-state networks, such as the Early Childhood Development Action Network, carry out global advocacy efforts with an engaged membership (Zonji, 2018). These efforts are replicated at regional level by organizations including the Africa Early Childhood Network, the Arab Network for Early Childhood

#### Non-state actors provide ECCE for those most in need

Many non-state actors advocate for children and families who may be excluded from the formal ECCE system. Some children grow up in difficult environments; exposure to extreme stress affects their socioemotional and educational development and behaviour (Haskins, 2014; Murray et al., 2012; Zeanah et al., 2017). Non-state actors can tailor support through ECCE.

Nurseries and co-residence programmes for children with incarcerated mothers allow mothers to live and interact with their children to support their development during crucial stages (Goshin et al., 2014; Poehlmann-Tynan and Turney, 2021). Studies in Brazil and Kenya indicate not all children of imprisoned mothers have such opportunities (Cheruiyot, 2019; Stella et al., 2016). However, contact with an incarcerated parent may not always be in the child's best interest and may even be detrimental (Dwyer, 2014; Martin, 2017; Smith and Gampell, 2011).

Eligibility for access to prison nurseries varies among countries. Some allow women to access these services only if they give birth while in prison, have children of a certain age or have a certain type of sentence (Halter, 2018; Warner, 2015). Bulgaria's Child and Space Association works with Roma mothers and fathers in prison to develop their parenting skills and ensure the children's well-being and rights (Brett, 2018). Chile's non-state Integra Foundation works in penitentiaries to provide mothers and children under 2 with a tailored nutritional plan and developmentally appropriate education (Chile Undersecretary of Early Childhood Education, 2019).

NGOs and faith-based organizations help children born into difficult situations. Romania's Bucharest Early Intervention Project has highlighted the importance of supporting young children in institutions (Zeanah et al., 2017). In the Philippines, the Kaisahang Buhay Foundation provides poor working parents who may lack family support with a range of services, including day care with an education component, to encourage them not to place their children in institutions (Bold, 2006; Kaisahang Buhay Foundation, 2021).

#### BOX 6.3:

#### Business influence on early childhood development can be harmful

The influence of non-state actors in ECCE is broader than direct delivery of services. Companies influence and often mislead families through advertising, exposing them to products allegedly geared towards children's learning and development. Advertisements may claim, for instance, that certain games teach young children early science or coding and promote them as key to their learning without research to support such claims; they may instead hinder children's socioemotional and educational development (Clark et al., 2020; Lieber, 2020). The use of branding, logos and social media to target children from an early age influences children's and parents' perceptions of toys and their educational value (Burroughs, 2017; Vaala and LaPierre, 2014).

Young children are increasingly introduced to digital and analogue tools for learning, such as interactive media, applications and other software, television programmes and touchscreen devices, linked to the growing education entertainment industry. A study in the United States, tracking children's use of media from birth to age 8, found that 2- to 4-year-olds used screens for about 2.5 hours a day, on average, increasing with age, and that 73% used screens to view TV and video, compared with 1% for homework (Rideout and Robb, 2020). An analysis of more than 10,000 students found that pre-primary school-age children from low-income families and Black children at all income levels, as well as hyperactive children and those exhibiting aggressive behaviour, were more likely to become frequent users of technology by the end of primary school (Morgan et al., 2021).

Start-up companies have begun to tap into this market, which is expected to grow. One firm in the United States raised US\$50 million in 2020 to develop and scale up a personalized early learning app for 2- to 8-year-olds to improve literacy skills (Rauf, 2020). In China, platforms such as Zhang Tong Jia Yuan, iBeiliao and Zhihuishu aim to improve communication between kindergartens and parents, instructing parents through customized parenting plans. Some offer free trials to kindergartens (Deloitte, 2018).

The use of education entertainment and digital learning tools for the youngest children has raised concern regarding limited parent-child interaction, poor app design, inadequate inclusion of learning content, cyber safety and health issues linked to overexposure to digital devices (Britto et al., 2013; Liu and Hwang, 2021; Reid Chassiakos et al., 2016).

# Foundations play an important role in ECCE advocacy

Development and the Asia-Pacific Regional Network for Early Childhood, which count non-state actors among their members. They help share information and provide additional capacity, expertise and resources to translate and adapt ECCE developments into national contexts (Vargas-Barón, 2015).

Foundations are active in these networks, playing an important role in ECCE advocacy. The Bernard van Leer Foundation has seven decades of experience in stressing the importance of early years, providing financial support and pooling expertise. Its current strategy focuses on parent education, bringing an early childhood focus to urban planning, and advising on scaling up early childhood development initiatives. For instance, it has researched how flagship public programmes such as Criança Feliz in Brazil (Datla, 2021) and Cuna Más in Peru (Datla, 2018) can become sustainable, and how to ensure that slums and refugee settlements have close access to ECCE (Arup and Bernard van Leer Foundation, 2020). The Aga Khan Foundation, active notably in Central Asia and Eastern Africa, and the Open Society Foundations, working especially in Central and Eastern Europe and the Caucasus through the International Step by Step Association, are other examples of non-state actors mobilizing ECCE support and advocacy (Vargas-Barón, 2015).

#### CONCLUSION

99

Where the public sector has lacked capacity to provide ECCE services for families and children from birth to school age, non-state actors have responded to demand. From non-profit, non-governmental, faithbased and community-based organizations to forprofit private actors, non-state actors' services range from unregistered and informal childminding to formal preschools. Yet, despite the presence of this range of providers, access for many disadvantaged families is challenging due to high costs and lack of financial support. Furthermore, quality across the diverse spectrum of state and non-state providers is highly variable, necessitating uniform quality standards and stronger regulation and quality assurance mechanisms.

Non-state provision continues to grow in many countries and the role of non-state actors in influencing policy development is evident. While government efforts to expand public provision in pre-primary education have helped enrolment grow in some contexts, governance and monitoring efforts need to be accelerated for the full range of non-state providers and target populations. As COVID-19 continues to cast its shadow on the sector, it will be important to ensure effective arrangements. Aya, Yamama, Balsam and Aya (L to R) are Syrian refugee students studying nursing at Luminus Technical University College, Jordan. They say the lab will help them replicate the environment of a hospital where they will work in the future.

CREDIT: UNHCR/Lilly Carlisle



# CHAPTER



# Tertiary education

# **KEY MESSAGES**

Non-state actors account for over one third of students in tertiary education, a share that remained stable in the 2010s.

The private share is higher in tertiary education than in secondary in 96 of 131 countries.

#### Non-state tertiary education can challenge system quality.

- Opinions about the benefits of profit orientation in the sector are split. In the United States, student outcomes deteriorate as the incentive for profit maximization increases.
- Non-state institutions' focus on responding to labour market needs may blunt the drive for innovation. In India, about 40% of private colleges offer only one field of study.
- Substantial numbers of teachers in the Global South are trained in non-state institutions. Chile and Mexico have banned this practice due to quality concerns.
- Staff in non-state universities are likely to work only part time, often moonlighting from public institutions, which threatens teaching quality. Mozambique has banned the practice.

#### Non-state providers may improve or hinder access to tertiary education.

- Non-state providers are more likely to select those who can afford to pay. In Colombia, the share of students enrolled in non-state institutions was 17 percentage points higher in urban than in rural areas in 2018.
- Non-state institutes can provide more flexibility. In Brazil, 68% of students in non-state universities are enrolled in evening classes, which allows them to work during the day.

#### Regulations can protect quality and affordability, depending on context, capacity and resources.

- Quality assurance mechanisms are recent and tend to focus more on establishment and operation than on quality and results. Bangladesh had no accreditation framework until 2017. Most of Peru's 32 private institutions were not accredited as of mid-2019.
- Equity-promoting regulations are not common, and usually exist only for providers that receive public funding. Romania's government reserves tuition-free places for students from rural secondary schools and Roma students, but only in public universities.

#### The way private tertiary education is financed has implications for equity and quality.

- Governments fund non-state institutions directly, via tax breaks and research grants, or indirectly, through scholarships and student loans. In Australia, the government covers 55% of spending on higher education institutions, nearly one-third of which is aid to households.
- Non-state institutions also raise income through loans and bonds. University bond issuance had reached US\$11.4 billion by mid-2019, more than doubled from the year before.

#### Households are taking on larger shares of tertiary education costs.

- Over 70 countries operate student loan programmes, most often government-subsidized, which may lead to hikes in tuition fees, as in Brazil.
- Some non-state actors help cover student costs with scholarships and loans. In Bangladesh, non-state universities must earmark 2% to 5% of revenue for scholarships or grants.

More than one in three tertiary students attend non-state institutions	.160
Non-state tertiary education regulation varies	. 170
Financing modalities have quality and equity implications	174
Non-state actors influence tertiary education through multiple mechanisms	. 178
Conclusion	.180

N early all countries ensure tertiary education through a combination of state and non-state actors. But the participation of non-state actors raises questions and poses challenges to the achievement of quality and affordability, the key dimensions of Sustainable Development Goal target 4.3. Moreover, the line between state and non-state tertiary education is blurry, and non-state actors are involved not only in service provision but also in financing and influencing public institutions and the sector as a whole.

Non-state institutions have implications for system quality. For instance, if they focus on particular skills to respond to labour market needs, they may blunt innovation. Academic staff of non-state institutions are less likely to be full-time professors. Non-state institutions also have implications for affordability. For instance, they may target those who can afford to pay for tertiary education, even if, in some cases, they target groups at risk of exclusion.

Regulation of non-state tertiary education varies, as does the use of quality assurance mechanisms to ensure minimum standards. Equity-promoting regulations are not common. Financing modalities of non-state institutions, such as their access to public funding and the degree to which they depend on fees, also have quality and equity implications. Some institutions are supported by non-state actors beyond fees. In addition, households are supported in taking on a larger share of tertiary education funding, including of non-state institutions, through loans, scholarships and income share agreements.

#### MORE THAN ONE IN THREE TERTIARY STUDENTS ATTEND NON-STATE INSTITUTIONS

Measuring the extent to which non-state actors are involved in provision of tertiary education is challenging. The diversity of institutions means the distinction between state and non-state provision is blurry. The UNESCO Institute of Statistics (UIS) defines private institutions as those controlled and managed by a non-governmental organization (NGO) or whose governing board consists mostly of members not selected by a public agency. By this definition, about 33% of students are enrolled in private institutions globally, a number which has remained constant in recent years. The highest shares are in Central and Southern Asia (49%) and Latin America and the Caribbean (54%) (**Figure 7.1a**). But regional averages mask large differences between countries (**Figure 7.1b**).

Although there are insufficient comparable data to establish a long-term trend, the private share appears to have remained relatively stable in the 2010s. It is increasing in some countries and decreasing in others (**Box 7.1**). Policies in some countries have focused on expanding public provision of tertiary education, which has reduced the size of the non-state sector. In Colombia and the Philippines, expansion of public provision through establishment of new public institutions led to a decrease in the share of students enrolled in non-state education, even if the absolute number continued to rise (Levy, 2015b).

The private share in tertiary education is higher than that in primary in 99 of 131 countries and higher than in secondary in 96

The private share in tertiary education is higher than that in primary in 99 of 131 countries and higher than in secondary in 96. Unlike in compulsory education, where governments prioritized the expansion of state schools or, in some cases, the takeover of non-state schools, changing non-state universities' status was a lower priority. In Bangladesh, the 2010 National Education Policy reinforced non-state tertiary providers' role but called for nationalization of primary education. In 2014, the government nationalized nearly 30,000 non-state primary schools (BRAC, 2021).

Where the number of non-state providers is growing, the common explanation is that it is in response to demand for 'different', 'better' or 'more' tertiary education. It is accordingly associated, respectively, with three main types of non-state institutions: religious-cultural, elite and demand-absorbing, a term commonly used to describe the mass of smaller, non-denominational institutions (Levy, 2006; Pachuashvili, 2009).

#### FIGURE 7.1:

**?**?

**Regional trends in the share of private enrolment in tertiary education mask divergent national trends** *Percentage of enrolment in tertiary education in private institutions* 



GEM StatLink: https://bit.ly/GEM2021\_fig7\_1a Source: Buckner (2021), based on data from UIS and PROPHE.



#### BOX 7.1:

#### In Bangladesh and Poland, the share of non-state institutions in tertiary enrolment has decreased

Bangladesh began allowing non-state tertiary institutions to operate in 1992 (Ahmed, 2015). After their initial growth, their share of tertiary enrolment almost halved, from 64% in 2001 to 35% in 2016, while the tertiary education gross enrolment ratio rose from 7% to 18%. The government has absorbed much of the demand growth by increasing the number of public institutions – particularly professional and technical institutions, which in the non-state sector have declined – and by charging considerably lower fees than non-state institutions (Bangladesh University Grants Commission, 2018; BRAC, 2021) (Figure 7.2).

In Poland, the share of non-state institutions in tertiary enrolment fell from 33% in 2008 to 27% in 2018, largely because of a demographic shift (Kwiek, 2016, 2018). Overall tertiary enrolment declined by 31% in that period: i.e. the 2008 public sector would have accommodated nearly all of current tertiary demand. In countries where total enrolment is falling, small non-state institutions are more likely than larger ones to be affected. This is particularly true when public institutions offer lower fees and reasonable or superior quality. Falling enrolment also allows governments to increase public spending per student in public institutions, at least temporarily.

#### FIGURE 7.2:

#### The share of non-state enrolment may decrease regardless of the overall enrolment trend

Share of enrolment in tertiary private institutions and total enrolment in tertiary education, Bangladesh and Poland, 1998–2018



Source: UIS database.

#### RELIGION- OR CULTURE-ORIENTED INSTITUTIONS ARE LINKED TO HISTORY AND TRADITION

Religiously affiliated institutions have a long history, in some cases going back to the very origin of universities in Europe and Asia (Altbach et al., 2021). Today, they are still prominent in many countries. In South-eastern Asia, where religiously affiliated institutions are common, there are Islamic tertiary education institutions in Indonesia, Catholic institutions in the Philippines and Buddhist institutions in Thailand (Asian Development Bank, 2012; Welch, 2021).

Religious organizations have historically been important non-state providers of tertiary education in Latin America and sub-Saharan Africa, with the latter considered a hotspot for the expansion of Christian tertiary education today (Carpenter, 2017; Durham and Sampaio, 2000; Levy and Tamrat, 2021). In Kenya, missionaries founded the first non-state tertiary education institutions during the colonial period and the country still carries this legacy: Today, most non-state institutions in the country are run by religious organizations (Carpenter, 2017; Onsongo, 2007).

Culturally oriented non-state institutions have been established by various ethnic groups, particularly in countries with heterogeneous populations. Ethnic communities in Indonesia and Malaysia own non-state institutions and provide scholarships to members of their communities (Asian Development Bank, 2012; Welch, 2021). In Malaysia, the establishment of ethnic quotas for public institutions after independence restricted access to non-Bumiputra (ethnic Malay), thus pushing ethnic Chinese and Indian Malaysians into the non-state sector (Jie, 2018; Welch, 2021). Some estimates indicate 85% of students in public institutions are Bumiputra and over 90% of those in non-state institutions are non-Bumiputra (Welch, 2021).

A population's ethnic and linguistic heterogeneity can influence governments' disposition towards non-state education. After regime change in the Baltic countries in 1989, language laws prohibited public sector instruction in languages other than the national language. Hence, most demand for non-state provision in those countries came from Russian language providers. Latvia and Estonia, which had sizeable Russian-speaking minorities, allowed the establishment of Russian non-state tertiary institutions soon after the regime change (Pachuashvili, 2011). Language still plays an important role in Latvia's non-state sector: In 2017, of students in non-tertiary institutions, 29% studied in Russian, 19% in English and 10% in multiple languages. In 2018, a new law extended the national language obligation to non-state institutions, with minor exceptions for EU official languages, effectively excluding Russian provision (Kuzmin, 2018).

#### LIMITED BUDGETS AND MARKET-FRIENDLY POLICIES ENCOURAGE MASSIFICATION

The proliferation of non-state institutions that are neither elite nor religiously or culturally affiliated, often referred to as demand-absorbing institutions, is a relatively recent, fast-growing phenomenon that emerged in response to rising demand and in the context of tight public budgets (Levy, 2013; Welch, 2021). In many low- and middle-income countries, policies promoting non-state involvement in tertiary education became popular in the 1990s, often with the encouragement of international financial institutions (Task Force on Higher Education and Society, 2000).

Historically, non-state tertiary education institutions in Latin America were either Christian or secular elite institutions. But expansion since the 1960s has been marked by smaller non-state institutions aimed at absorbing excess demand (Durham and Sampaio, 2000; Levy, 1986). Two factors contributed to this development. First, governments increased the non-state sector's flexibility to create institutions or programmes quickly. Second, in an adverse fiscal context, the International Monetary Fund and World Bank recommended reducing public funding for tertiary education (Corbucci et al., 2016; Ferreyra et al., 2017).

A similar pattern has been observed in sub-Saharan Africa, especially in anglophone countries (Levy, 2007; Varghese, 2016). After the post-independence nationalization of tertiary education, smaller, non-elite and non-religious institutions have accounted for most of the surge in tertiary enrolment (Irene and Hussain, 2020). Such institutions have also been at the forefront of expansion of the non-state sector in Central and Eastern Europe since the transition in the 1990s, in part because of a market-friendly shift in policies (Kwiek, 2016). Similar factors explain expansion in India since the 1990s, after the post-independence strategy of expanding tertiary education through public institutions was reversed (Ravi et al., 2019; Varghese and Panigrahi, 2019).

66 -

Religious organizations have been important non-state providers of tertiary education in Latin America and sub-Saharan Africa

Such institutions have also grown in contexts of crisis or major disruptions in public services. In Côte d'Ivoire, the political crisis of 2010/11 effectively shut down many public institutions for infrastructure repairs, which fuelled non-state enrolment (World Bank, 2017). Lebanon's civil war led to major disruptions in the public tertiary education system, leading to dismemberment of the national university in 1974 and the rise of both commercial and religious non-state institutions (Buckner, 2011).

#### DEMAND FROM PRIVILEGED GROUPS SOMETIMES LEADS TO ELITE NON-STATE INSTITUTIONS

Non-state elite institutions may arise in response to perceived decline in quality of public universities, a concern sometimes expressed by the most affluent segments of society, for example, in Latin America in the mid-20th century (Durham and Sampaio, 2000; Levy, 2006). In South Africa, the expansion of the non-state sector in the mid-1990s has also been partly attributed to demand from the white population for 'better education' (Tamrat, 2017).

Non-state elite institutions are the rarest type of non-state institution. In most countries, public universities enjoy the highest prestige. Some countries, however, have top-tier non-state universities, such as the Ivy League institutions in the United States and Catholic universities in Latin America and other parts of the world (Altbach et al., 2021). In countries with a long tradition of religious tertiary education, it is common for institutions to be both religiously affiliated and elite. In the Democratic Republic of the Congo, for example, the most prestigious institutions are, and have historically been, religious (Gérard, 2020).

Countries' attempts to increase the prestige of their tertiary education system through non-state institutions is a more recent phenomenon. India's government introduced a plan to select 20 institutions to become world-class, half of them non-state (Chattopadhyay, 2019). In 2014, the Japanese government began the Top Global University Project to boost the global status of selected universities. Among the 37 institutions selected, 14 are non-state (MEXT, 2014). In Morocco, the non-state tertiary education sector, originally composed of small vocation-oriented institutions, is being transformed to cater for national elites, the Moroccan diaspora and international students. The government offers incentives to non-state institutions to follow the American elite university model, requiring them to have a full campus, student dormitories and a research centre (Buckner, 2018).

#### 66 -

Some countries have recently attempted to increase the prestige of their tertiary education system through non-state institutions

One recent type of elite institution is the international campus, facilitated by the 1995 General Agreement on Trade in Services, which recognized tertiary education as a tradable service and encouraged a wide range of cross-border non-state provision (Altbach et al., 2021). This has since been strengthened by the inclusion of tertiary education in other international trade negotiations, including the Trans-Pacific Partnership (Australia DFAT, 2016) and Comprehensive Economic Trade Agreement (Davies, 2018). After joining the World Trade Organization in 2000, China opened its tertiary education market and joined the quest for 'world-class universities' by encouraging collaboration between local and leading foreign institutions. The number of institutions based on such efforts grew from 1 in 2004 to 12 in 2019 (Mok, 2021).

#### LINES BETWEEN STATE AND NON-STATE TERTIARY EDUCATION ARE OFTEN BLURRED

Diverse ownership, control and governance arrangements in tertiary education often blur the lines between state and non-state providers. A well-known example is the United Kingdom, where most universities are regarded as public and receive significant government funding but are controlled by non-state actors and therefore considered private in international statistics (Knight, 2006). The University of Nairobi, the oldest and biggest public university in Kenya, receives the majority of its funds from private sources and applies a business approach to governance, focused on income generation and increasing the institution's entrepreneurial practices (Provini, 2019).

The International Islamic University Malaysia is a public institution by law but governed by the Companies Act, with a board of governors that includes five members from Muslim countries and a representative of the Organisation of Islamic Cooperation (Welch, 2021). In Viet Nam, a university was founded as a training centre for a subsidiary of a state-owned enterprise. The subsidiary was later privatized, but the university still claims affiliation with the parent state-owned enterprise (Chau et al., 2020).

Within the non-state sector, it is hard to distinguish for-profit from non-profit institutions. In Chile, large for-profit multinational companies such as Laureate

#### Within the non-state sector, it is hard to distinguish for-profit from non-profit institutions

Education owned universities, even though the country does not allow for-profit universities (Bernasconi, 2013). A recent decision clarified that universities could be owned and controlled by a for-profit company but could not be operated as for-profit institutions (Educación 2020, 2018; Hurtado, 2020). In Malaysia, a for-profit company held by a non-profit entity owns two institutions (Welch, 2021). Many non-profit institutions use loopholes to distribute profit, leading some countries, such as Brazil (since 1997), to permit for-profit institutions to operate in hopes of collecting taxes (Almeida de Carvalho, 2013). Even within the for-profit sector, ambiguity over what is legally considered a tertiary education institution, rather than a business providing training, hampers regulation, as the former may come under education law and the latter under business law (Levy, 2015a).

The rise in cross-border provision of tertiary education further complicates the lines of ownership and regulation. Branch campuses of public tertiary education institutions in Australia, the United Kingdom and the United States are deemed private institutions in Malaysia. In Viet Nam, RMIT University and British University Vietnam, operated by public institutions, are considered private universities, while the Vietnamese-German University and the University of Science and Technology of Hanoi, also known as the Vietnam-France University, are listed as public (Welch, 2021).

99

# THE EMERGENCE OF NON-STATE INSTITUTIONS HAS IMPLICATIONS FOR SYSTEM QUALITY

As with other levels of education, a common argument for non-state actors is that they can increase competition and improve system quality. However, competition in tertiary education is imperfect, as entry costs are high and services are highly differentiated in terms of location, programme type, student ability, fields of study and rigour (Ferreyra et al., 2017). For these and other system-level factors, the potential for improvement may not be fulfilled.

# The impact of profit orientation on quality is a subject of intense debate

Supporters of for-profit institutions argue that the need to attract students creates competition and incentives to improve quality. Opponents say this mechanism does not work in a market as imperfect as that of tertiary



"

education (Altbach et al., 2021). Decisions to pursue tertiary studies are determined by short-term financial constraints as well as by information constraints – which also usually work at the expense of disadvantaged students – and not by long-term academic objectives and potential returns (Knobel and Verhine, 2017).

But not all for-profits are the same. Different types of providers have different levels of profit-maximizing incentives. At one end of the spectrum, individuals, families and privately held companies may set up for-profit tertiary education institutions with socially responsible or corporate visions that are compatible with prioritizing quality and student outcomes over maximizing profit (Tamrat, 2018). At the other end, publicly traded and private equity-owned firms have stronger incentives to maximize profit because of their payout responsibilities to shareholders and investors (Eaton et al., 2018).

In the United States, where the sector is highly subsidized (around 90% of for-profit institutions' revenue comes from federal grants and federally guaranteed loans), student outcomes deteriorate as the incentive for profit maximization increases. For instance, acquisition of a for-profit institution by a private equity-owned firm led to declines in graduation rates, loan repayment and labour market earnings as a result of several mechanisms, including a decrease in education inputs such as number of faculty per student and the share of expenditure devoted to instruction (Eaton et al., 2018).

# A focus on skills to respond to labour market needs may blunt innovation

Students in fee-charging institutions may need to learn skills that are rewarded in the labour market to be able to recuperate the cost of their studies after graduation. However, the focus of non-state institutions on providing such skills blunts their drive for innovation (Altbach et al., 2021). This can be observed in the fields of study offered and the focus on employability as a marketing strategy and source of legitimacy. Graduate employability is an indicator used in South-eastern Asia to rank institutions (Welch, 2021).

Demand for specialized and professionally oriented programmes shapes tertiary education growth in many countries (Irene and Hussain, 2020; Muzammil, 2019). Smaller institutions with little capital choose to offer only fields of study that require little upfront investment in infrastructure and are less regulated by governments and professional associations (Asian Development Bank, 2012; Teixeira et al., 2016). In India, public and government-aided colleges generally provide general education, whereas private colleges provide professional courses. About 40% of private colleges offer only one field of study, mostly education (Muzammil, 2019); 67% of teacher training colleges are private (Henry et al., 2020) (**Box 7.2**). In Indonesia, two thirds of total enrolment in non-state institutions is in education, social sciences or business (Welch, 2021). In Viet Nam, state universities offer medicine, engineering or biotechnology, leaving other fields to private universities, which are mostly smaller and more vocation-oriented (Altbach et al., 2021; Levy et al., 2020). Ethiopia's government prohibits non-state institutions from offering degrees in fields such as law and teacher education (Tamrat and Levy, 2017).

Non-state institutions create networks with employers, e.g. inviting them to teach and working with them to adapt programmes and curricula to improve outcomes for graduates. In Germany, most non-state institutions offer degrees in business administration and communication in close cooperation with the private sector (Altbach et al., 2021). In Morocco, CIRA, one of the largest private education service providers, brings in professionals to lecture so that what is being taught is in line with up-to-date practical knowledge (Kalla, 2021). Similar practices in Senegal lead to programmes changing much more often than at public universities (Dia and Goudiabu, 2020). Career counselling, job fairs and alumni networks are other ways to connect to the labour market and have been used, for instance, in Kenya, Morocco and Peru (Zuniga et al., 2021). In Saudi Arabia, non-state institutions have increased graduates' employment rate by changing programme types and teaching methods, using English for instruction, and providing structured work experience and career counselling (Jamjoom, 2016).

# Academic staff of non-state institutions are less likely to be full-time professors

Non-state tertiary education providers tend to rely more than public universities on part-time academic staff (Levy and Tamrat, 2021; Salto, 2018; Welch, 2021). In Senegal, less than 20% of teachers at non-state institutions have full-time permanent contracts (Dia and Goudiaby, 2020). In the United States, non-state institutions have higher shares of part-time faculty, on average, and the share grows the more institutions rely on fees (Liu and Zhang, 2013). Reliance on part-time staff may reflect these institutions' stronger link with the labour market, as the emphasis on practical learning can lead them to hire working professionals as lecturers. It can also be a cost-saving strategy (Teixeira et al., 2016).

Working multiple jobs, or moonlighting, is a factor in high shares of part-time contracts. It allows newer and smaller non-state institutions to gain legitimacy by employing respected public university professors.

#### Teachers in the Global South are often trained in non-state tertiary education institutions

Non-state actors play an increasingly important role in teacher education in the Global South. For instance, non-state teacher training institutions operate in at least 22 sub-Saharan African, 17 Latin American and 7 South Asian countries. Non-state actors have made an important contribution to teacher education programmes in conflict-affected countries. In Afghanistan, non-state teacher training colleges were established in each province, along with rural college satellites to facilitate access for those in remote areas. In Angola and Mozambique, DAPP, a non-governmental organization (NGO), has played a key role in teacher training, in collaboration with the governments. In Somalia, where the main public institutions for teacher education were destroyed during the civil war, non-state actors have trained most teachers since 2002.

Teacher training programmes tend to be government regulated: State and non-state providers largely follow a centralized curriculum or qualification framework. In India, the government regulates minimum qualifications for trainers in both sectors, as well as the level of fees. In Mozambique, state and non-state institutions follow the same criteria and conditions for admission. In recent years, non-state teacher training institutions have been closed in Chile, Colombia and Ecuador for failing to meet minimum quality standards. In Costa Rica, poorly regulated non-state institutions offer programmes from which students graduate in considerably less time than required by public programmes.

Non-state teacher training programmes are increasingly available by distance, which raises concern about the lack of a practical component. In response, some countries, including Chile and Mexico, have banned such programmes. In Brazil, where the law gives preference to teacher education conducted in person, 67% of entrants in initial teacher education enrolled in distance courses; of those, over 95% were at non-state institutions (**Figure 7.3**). In Botswana, difficulties in regulating the large number of online programmes offered by non-state institutions leave many unaccredited and likely substandard. Pakistan developed national standards in 2016 to accredit distance teacher education programmes and thus increase regulatory oversight over them.

Source: Sirois et al. (2021).

#### FIGURE 7.3:



Number of entrants to initial teacher training courses in Brazil by sector and type of programme, 2010–19



In Malaysia, the share of moonlighting professors at non-state universities ranges from 20% to over 80% in smaller and newer institutions (Welch, 2021); the same practice is observed in Nigeria (Amini-Philip, 2019). In Poland, employing public institution faculty helped the non-state sector expand quickly in the 1990s; it has been estimated that if professors could hold only one full-time position, non-state institutions would lose at least half their faculty (Jablecka, 2007).

Moonlighting can have negative effects on institutional capacity, education quality and student support (Amde et al., 2018). In 2015, Mozambigue's government outlawed it, requiring professors to be registered and teach at only one university (Makoni, 2015). Professors contested the measure as it threatened their income (Júnior, 2015). As implementation of the regulation has been incomplete, the practice reportedly continues (Amde et al., 2018; Club of Mozambique, 2018). Moonlighting can lead to excess workloads for professors and be detrimental to their well-being (Elder and Kring, 2016; Mulokozi, 2015). Professors in Kenya, where moonlighting is common, admitted that the resulting fatigue reduced their work quality (Wesangula, 2015). However, there are also positive effects on job satisfaction, as moonlighting increases professors' income and enables more collaboration and networking opportunities (Amini-Philip, 2019; Elder and Kring, 2016; Sakyi and Agomor, 2020).

The stronger labour market focus and higher share of part-time professors at non-state institutions are also reflected in a greater focus on professional and vocational education programmes that do not award research degrees (Seeber, 2016). In Brazil, academic research is almost entirely conducted at public institutions, which account for about one quarter of students (Moura, 2019). Between 2013 and 2018, the 15 tertiary education institutions with the largest research output were public; they accounted for over 60% of articles published in international journals (Web of Science Group, 2019). In Malaysia, where one quarter of non-state institutions have fewer than 1,000 students (Welch, 2021), a proposed law requested non-state universities to conduct research as well as teaching (Kean and Soe, 2018).

66

Moonlighting can have negative effects on institutional capacity, education quality and student support

. 99

#### NON-STATE PROVIDERS TARGET THOSE WHO CAN AFFORD TO PAY FOR TERTIARY EDUCATION

Non-state providers tend to reserve access for those who can afford to pay. In some contexts, some may try to increase access for those who had been excluded. But in both cases, there is a risk of segregation in the tertiary education system.

The wealth gap in access to tertiary education is high in much of the world. Analysis of data from the World Inequality Database on Education shows the median gap in tertiary attendance between the richest and poorest quintiles to be 21 percentage points. Non-state providers exacerbate such gaps. In Latin America, students from the richest households make up a considerably higher share of enrolment in non-state institutions than in public ones (**Figure 7.4**). Over the past 15 years, the largest participation increases among the richest quintiles have been in the countries where the share of students in non-state tertiary education has increased the most, including Honduras, Paraguay, Peru and Uruguay (SEDLAC, 2018).

#### FIGURE 7.4:

#### Rich students make up a large share of enrolment in non-state tertiary education institutions in Latin America

Percentage of students from the richest 20% of the population, selected Latin American countries, 2018



Source: GEM Report team calculations based on SEDLAC (2018).
This pattern is also observed in other regions. In China and Viet Nam, non-state tertiary education institutions have widened access mostly to wealthier students of lower ability, contributing little to social mobility (Henaff et al., 2020; Welch, 2021; Zha, 2006, 2011). In India, among students enrolled in tertiary institutions in 2014, about 25% from the two poorest quintiles, but 42% from the richest quintile, attended independent private institutions. The vast majority of students from Scheduled Tribes and Scheduled Castes attended government institutions or government-dependent private institutions (Henry et al., 2020). India's non-state sector appears to have increased access for the most privileged groups first, then extended it to the most privileged within each group that benefit from affirmative action in the government-aided institutions, such as Scheduled Tribes and Scheduled Castes (Gérard, 2020).

An analysis of the relationship between non-state provision and overall participation in tertiary education around the world, conducted for this report, shows that the relationship varies by country income. A greater share of non-state actors in total enrolment is associated with lower inequality in attendance rates in high-income countries, where enrolment levels are already high, but greater inequality in upper-middle-income countries, where participation among the richest is still not universal. No relationship emerged in low- and lower-middle-income countries, where enrolment levels are very low and where over three quarters of inequality can be explained by differences in secondary completion rates between poorest and richest (Buckner, 2021).

Inequality in tertiary education can also manifest in access to education services, conditions for learning and graduating, and labour market outcomes. In many countries, non-state actors establish and reinforce tiers of institutions, e.g. through fee differentiation, as in Mexico (Gérard et al., 2020). A vicious cycle of reinforcing quality differences is created as most institutions targeting poorer students rely almost entirely on student fees, while elite institutions can count on additional sources of revenue, including donations and public funds. In the Democratic Republic of the Congo, small non-state institutions charge US\$150 in annual student fees, while bigger, mostly religion-affiliated universities charge US\$1,000. In Senegal, fees range from US\$330 outside Dakar to US\$4,500 in Dakar (Gérard, 2020). Non-state institutions outside Dakar tend to have academic staff who are less qualified and receive lower, less regular pay (Dia and Goudiaby, 2020).

Non-state provision tends to be concentrated in urban centres, as can be observed in many countries of sub-Saharan Africa, Latin America, South and South-eastern Asia, and Western Europe (Gérard, 2020; Muzammil, 2019; Onsongo, 2007; SEDLAC, 2018; Teixeira et al., 2016; Welch, 2021). In Senegal, two thirds of non-state institutions are in Dakar (Dia and Goudiaby, 2020). In India's North Eastern Region, private unsubsidized institutions took in 10% of students in rural areas and 18% of students in urban areas in 2014 (Muzammil, 2019). In Colombia, the share of students enrolled in non-state institutions was 17 percentage points higher in urban areas than in rural areas in 2018; in Peru, the difference was 23 percentage points (SEDLAC, 2018). Richer households that can afford the higher cost of tertiary education live in urban areas and students can more easily find work there while studying or after graduation (Ahmad and Shah, 2016).

# Some non-state providers target groups at risk of exclusion

Despite an overall tendency to favour more privileged students, some non-state institutions provide access to groups that are otherwise discriminated against or excluded from tertiary education as a result of gender, race, poverty, displacement or remoteness. In Kenya, rapid expansion of non-state tertiary education since the 1990s has been associated with increased opportunities for female participation, due partly to perceptions that such institutions are safer and provide more discipline and partly to a focus on the humanities and social sciences, which are popular among female students (Onsongo, 2007; Tamrat, 2017). In Saudi Arabia, non-state institutions have helped expand access for women by offering female-only courses not available in the public system (Jamjoom, 2016). In South Africa, a social entrepreneur founded CIDA City Campus, a low-cost university targeting poor black students, while Tsiba Education, a non-profit institution financed by corporate sponsors, offers business programmes to financially and educationally disadvantaged students (Salmi and Sursock, 2018).

66

A greater share of non-state actors in total enrolment is associated with greater inequality in attendance in upper-middle-income countries

#### 66 -

In Kenya, rapid expansion of non-state tertiary education since the 1990s has been associated with increased opportunities for female participation

99

In Colombia, UNIMINUTO, a faith-based university enrolling about 100,000 students, provides access to students in informal settlements, small cities and rural areas (Casanova et al., 2015; UNIMINUTO, 2020). Building on this initiative, the government encouraged the creation of over 200 regional centres called CERES, which are partnerships among tertiary education institutions, local government and private companies (Salmi, 2020). A collaborative global partnership, Jesuit Worldwide Learning: Higher Education at the Margins, offers an online programme that allows refugees in camps in Kenya to attend tertiary education (Salmi, 2020). Kiron Open Higher Education, a German non-profit organization, offers refugees a two-year online programme that helps them complete their studies at a host-country institution (Unangst, 2017). In Senegal, Al-Azhar University, open only to students who complete an Arabic upper secondary education, has developed an engineering programme that includes a preparatory year to improve students' French language skills before they follow the official three-year curriculum (Dia and Goudiaby, 2020).

Non-state providers demonstrate innovation in providing flexibility. In Brazil, 68% of students at non-state universities take evening classes – which allow them to work during the day – compared with 36% at public institutions. Similarly, 56% of non-state institution enrolment is in distance education, compared with 14% for public institutions (INEP, 2020). In Argentina, institutions are distinguished by offering flexible hours and distance education (Altbach et al., 2021).

Despite short-term positive effects of increasing access to tertiary education, creating separate institutions for different groups can jeopardize development of social cohesion. In Indonesia, Malaysia and Thailand, concerns have emerged about the rise of extremism in some religion-affiliated non-state tertiary institutions despite government countermeasures (Welch, 2021).

#### NON-STATE TERTIARY EDUCATION REGULATION VARIES

Tertiary education regulatory frameworks reflect government views of the sector and of non-state actors. Countries form a continuum from those where the government supervises and centrally coordinates non-state providers, assigning them variable roles within an overarching national strategy, to those allowing competition, choice and autonomy for non-state providers, as in Chile, Colombia, Japan, Malaysia, the Philippines and the Republic of Korea (Ferreyra et al., 2017; Welch, 2021).

In countries with historically centralized structures, non-state actors are viewed with mistrust and placed under strict regulations, which often favour public institutions. In Argentina, non-state universities need approval from the accreditation agency before being established and the authorization is probationary for up to six years (Altbach et al., 2021). Azerbaijan's government controls the types of programmes non-state institutions can offer and even the number of students they can enrol (Salmi, 2017). In Ethiopia, non-state providers had a leading role in introducing distance education but in 2012 were banned for two years from offering that mode of study (Tamrat and Teferra, 2019).

By contrast, in Senegal, the government views non-state institutions as complementary and as accessories to the public system. Accreditation regulations encourage the establishment of institutions following market principles. Nevertheless, the government retains strong control over their operations. For example, it can allocate any number of students who do not qualify for state institutions to non-state institutions, with little say from the students themselves or the receiving institutions (Dia and Goudiaby, 2020).

In some cases, there is a gap between the regulations and a country's discourse on non-state actors. Tunisia's government has tried to present itself as encouraging of non-state provision, but regulations are quite restrictive. Institutions are required to be for profit, cannot have regional branches and can offer only one field of study. The government put new programme approval processes on hold and increased the minimum investment required for new institutions, with little consultation (Buckner, 2018).

66

In Argentina, non-state universities need approval from the accreditation agency before their creation and the authorization is probationary for up to six years Moreover, country views of non-state actors change over time, a fact reflected in their regulatory frameworks, as in Viet Nam (**Box 7.3**). In Egypt, the first non-state institutions were viewed as a low-quality last resort for those who could not get into public institutions. In 1970, the legal framework required state approval of institutions' fee structure, subjects, course content, student cohort size and faculty hires. With time, openness to more elite non-state universities has been matched by an adapted legal framework allowing more flexibility and autonomy (Altbach et al., 2021). In post-communist Central and Eastern Europe, non-state institutions emerged in a regulatory vacuum, leading to significant challenges to which governments responded with delay (Levy, 2013) (**Box 7.4**).

#### BOX 7.3:

#### Non-state tertiary education regulation in Viet Nam has reflected changing attitudes towards private economic activity

The regulation of non-state tertiary education reflects wider political considerations. In Viet Nam, rapidly changing but ambiguous regulations have reflected a gradual embrace of economic privatization (Altbach et al., 2021; Chau et al., 2020). The first non-state university was founded by a group of intellectuals in 1988, two years after the launch of the major market-oriented economic reform called Doi Moi. It took a further five years for the country to regulate non-state tertiary education provision (Quang, 2017).

The first legal frameworks in the early 1990s regulated three types of non-state institutions: 'people-founded' institutions established by social and professional associations; state-established but privately financed 'semi-public institutions'; and 'private institutions' set up by individuals or companies. Private institutions effectively began operating only in 2005, and the following year the government outlawed people-founded and semi-public institutions, urging them to adopt the for-profit model. The transition caused regulatory ambiguity and led to many closures (Chau et al., 2020; Quang, 2017). Then, in 2012, non-profit universities were allowed again, sowing confusion among private institutions.

A shift towards a market-friendlier environment gained momentum in the 2010s. A target of the Higher Education Reform Agenda was that 40% of enrolment should be in private institutions by 2020 (Welch, 2021). In 2014, public institutions were encouraged to seek non-state funding under a new financial autonomy programme, which has led fees to increase considerably (Altbach et al., 2021). Today, the country has 65 non-state universities that enrol about 15% of students. Big private corporations and education conglomerates have entered the market, but there are no religious or civil society universities (Chau et al., 2020; Henaff et al., 2020).

#### 66

Unaccredited institutions tend to be the ones attended by disadvantaged populations, raising equity concerns

QUALITY ASSURANCE MECHANISMS AIM TO ENSURE MINIMUM STANDARDS 99

Most countries have regulatory frameworks for the establishment, operation and closure of non-state tertiary education institutions, aimed at ensuring minimum quality standards. In some countries, this is a recent development. Bangladesh had no accreditation framework to assess whether universities or programmes met a minimum quality standard until 2017 (Bangladesh University Grants Commission, 2018). Quality standards for non-state institutions are often different than for public institutions (Asian Development Bank, 2012; Ferreyra et al., 2017).

In some countries, explosive growth in non-state provision has posed a regulatory challenge when resources to accredit and monitor such institutions are lacking (Levy, 2013). About 90% of Indonesia's 4,500 tertiary education institutions are non-state, enrolling 59% of students as of 2018. A national accreditation board formed in the mid-1990s to accredit only non-state institutions has since assumed responsibility for all programmes. In 2009, there were 3,000 academic programmes to be evaluated; by 2013, some 20% of accreditation decisions were late. Emergency measures allowed some institutions to operate without accreditation. In 2017, less than 2% of tertiary education institutions had the highest level of accreditation. Some had no accredited programmes. In late 2018, in an effort to control the growing quality problems in the non-state sector, the government announced that it would revoke the permits of around 1,000 non-state tertiary institutions (Dilas et al., 2019). A rapid increase in unaccredited non-state institutions posed similar challenges in Libya, leading the government to close 20 non-state universities and colleges failing to meet quality standards in 2021, a decision with important implications for their students and academic staff (El-Galil, 2021).

Lack of resources or inadequate capacity of regulatory agencies can be a challenge. In the Democratic Republic of the Congo and Mexico, some institutions operate while accreditation is pending or with varying

#### BOX 7.4:

# In Central and Eastern Europe, the growth of non-state tertiary education was unregulated in the early transition years

The end of state socialism led to unprecedented growth in non-state tertiary education in most countries of Central and Eastern Europe. The rapid shift away from the state-centred model often took place in the absence of a legal framework (Dobbins and Knill, 2009; Slantcheva and Levy, 2007). Within five years after 1989, the share of non-state tertiary education institutions had reached 10% in Bulgaria, Hungary and the Russian Federation and over 20% in Poland (Slantcheva and Levy, 2007). In the Russian Federation, none of the over 150 tertiary education institutions were accredited by 1994 (Tomusk, 2003).

In Albania, the government tried to meet rising demand for tertiary education by expanding the public sector in the 1990s. Insufficient funding, however, and strict control over the number of new public institutions limited the supply. In 2005, a change in government led to market-favourable policies. By 2013, nearly 50 non-state tertiary education institutions had obtained licences (Kajsiu, 2015) and the share of students in non-state institutions rose from 1% in 2004 to 21.5% in 2012. However, this expansion raised concerns over quality and fraud (Clark, 2016; Erebara, 2014; Kajsiu, 2015). After an investigation, 18 non-state institutions suspected of delivering fake or dubious diplomas were closed in 2014 (Erebara, 2014). A 2015 law aimed to merge the state and non-state sectors, make all tertiary education institutions non-profit, increase funding by allowing public institutions to raise fees and enable non-state institutions to compete for public funding. However, several of these aims were not implemented (Raxhimi, 2019).

In Romania, expansion was even more rapid. By 1995, one quarter of tertiary education enrolment was in non-state institutions (Slantcheva and Levy, 2007). Non-state actors initially simply declared themselves to be tertiary education providers. The government did not establish a legal framework for accrediting institutions and recognizing diplomas until 1993, and the law came into effect only in 1996 (Nicolescu, 2007; Viiu and Miroiu, 2015). In 2000, a regulation imposed a 10% tax on non-state institutions' revenue (Fried et al., 2007). Later, regulations emphasized the importance of quality assurance mechanisms and allowed non-state institutions to compete for public funds, which led to the closure of several substandard non-state institutions and helped improve the legitimacy of the remaining ones (Korka and Nicolescu, 2007; Viiu and Miroiu, 2015).

Yet cases of fraud have continued to taint the sector. In 2009, over 100,000 graduates of Spiru Haret University, the largest non-state institution, had their diplomas annulled and labelled as illegally granted after irregularities were exposed (UWN, 2009). In 2019, a Spiru Haret pro-rector was arrested as part of an investigation on fraudulent examinations. Other institutions have been accused of fraudulently granting diplomas to students who did not attend classes and did not speak Romanian (Romania Insider, 2017, 2019). The share of students enrolled in non-state institutions declined from its peak of 42% in 2009 to 13% in 2018. This trend reflected an overall enrolment drop resulting from demographic decline and a decrease in the share of students passing the upper secondary education exit examination, from 81% in 2009 to 44% in 2011, attributed to a series of anti-fraud measures (Salmi et al., 2015).

degrees of temporary accreditation (Gérard, 2020). Often, unaccredited institutions tend to be the ones attended by disadvantaged populations, raising equity concerns. In Peru, of the 32 non-state institutions attended by relatively less well-off students, most were for-profit and had not obtained accreditation by mid-2019 (Benavides and Hagg Watanabe, 2020). In India and Mexico, government accreditation of non-state institutions is voluntary (Gérard et al., 2020; Ravi et al., 2019). In Guatemala, the main public university is in charge of granting licences and overseeing non-state institutions (Ferreyra et al., 2017), raising questions of conflict of interest.

Some countries outsource accreditation to non-state actors. In Mexico, where accreditation is not mandatory, non-state institutions seek accreditation from an association of private tertiary education institutions or accrediting agencies in the United States (Gérard et al., 2020). The Philippines has several non-state accrediting agencies, including two church-based agencies (Tanhueco-Tumapon, 2020). In the United States, quality assurance has been undertaken by non-state associations since the 19th century (Xiaoying and Abbott, 2016). The American Bar Association (ABA) does its own accreditation for law schools. In most states, only graduates of an ABA-accredited school are allowed to sit the bar exam (The Princeton Review, 2021). Some argue that competition between accrediting agencies may encourage innovation and efficiency, while others point to the advantages of a government-owned quality assurance structure, including harmonization of standards and reduced transaction costs (Xiaoying and Abbott, 2016).

Once in operation, quality control tends to be input- and output-based, focusing on infrastructure, student numbers and academic staff qualifications. Outcome-based quality assurance, focusing on student 66

Market concentration is another regulatory challenge presented by for-profit institutions that can have major implications for system quality

assessment, student surveys or research results, is not common and can be controversial. Germany, for instance, has not favoured direct large-scale evaluations of student skills (Pereira et al., 2018). A proposal by the Organisation for Economic Co-operation and Development (OECD) for an international higher education student assessment faced strong opposition (Altbach, 2015; Van Damme, 2015). But in Brazil, a mandatory student exit exam is used to monitor institutions' quality and allows the government to assign a grade to each programme (Pereira et al., 2018).

Outcome-based quality assurance is also more common when it is linked to public funding. In the United States, the 2016 Gainful Employment Rule introduced graduate employment and student loan repayment rates as part of the criteria for non-state institutions' eligibility for federal student loans, thus adding a layer of oversight and leading to closures of several for-profit institutions of poor quality (Jakiel, 2016; Kreighbaum, 2019). In Poland, the government began focusing evaluations and funding on variables such as research productivity, partly because non-state institutions were becoming too teaching-oriented and focused only on attracting students (Tarlea, 2017).

Regulatory frameworks may increase institutional autonomy in an effort to improve quality. In India, the government approved a plan in 2018 that rewards regulation-compliant state and non-state institutions with more autonomy to create new courses, collaborate with international institutions and enter research collaborations with industry (Chattopadhyay, 2019; Henry et al., 2020). In the Philippines, where the non-state tertiary education sector enrols over half of students, institutions that demonstrate continued adherence to high standards of scholarship, research and instruction are granted special regulatory treatment and autonomy, including financial autonomy in fee setting, through progressive deregulation and reduced government monitoring and evaluation, except where violations or complaints are reported (PEER country profiles).

Regulatory frameworks can also be used to ensure the quality of for-profit institutions and guard against a misalignment between profit maximization and students' and taxpayers' interests (Eaton et al., 2018). Some countries, including Argentina, Chile and India, outlaw for-profit tertiary education provision altogether (Educación 2020, 2018; Gérard, 2020; Mohanty, 2020). However, once for-profit provision is established, it becomes difficult to eliminate. In Peru, an initial version of the 2014 tertiary education law proposed outlawing for-profit institutions but was overturned (Gérard, 2020). Profit-making can lower institutional quality through insufficient investment in core services in exchange for higher investor return (Eaton et al., 2018). In the Philippines, the Commission on Higher Education stipulates that 70% of revenue from fees must be allocated to payment of staff salaries and 20% to infrastructure improvement, meaning no more than 10% is to be allocated to return on investment (PEER country profiles).

99

For-profit institutions may resort to deceptive business practices, including predatory recruitment and fraudulent marketing strategies. In the United States, 7 of the 10 biggest for-profit companies in the sector have been investigated on allegations (Halperin, 2016) and found to have indulged in such practices: Apollo/ University of Phoenix (US Federal Trade Commission, 2019)), EDMC (US Department of Education, 2018), ITT Tech (Douglas-Gabriel, 2018), Kaplan (US Department of Justice, 2015), Career Education Corporation (Pennsylvania Attorney General, 2019), DeVry (US Federal Trade Commission, 2016) and Bridgepoint Education (United States Department of Education, 2018). In Nigeria, the National University Commission helped close many of the country's illegitimate institutions (Varghese, 2016). In Viet Nam, several university officials were recently prosecuted on charges of selling diplomas or other types of corruption and wrongdoing (Tho, 2021).

Quality assurance regulations also need to tackle the rise in fake, unregistered for-profit institutions that often enrol marginalized people. In 2017, Pakistan's Higher Education Commission issued a public notice with a list of 153 illegal universities and degree-awarding institutions (Khan, 2017). In South Africa, the number of fake and unregistered institutions led the government to launch an awareness campaign and, in 2019, promulgate the National Qualifications Framework Amendment Act, making it a crime to claim or hold a fraudulent qualification (de Wet, 2019; TimesLIVE, 2018).

Market concentration is another regulatory challenge presented by for-profit institutions that can have major implications for system quality. In Brazil, where for-profit institutions account for about half of tertiary education enrolment, the 10 largest companies, many of which are international, made up 30% of total tertiary education enrolment and over 60% of enrolment in for-profit institutions (Cunha, 2018; INEP, 2020). That year, the competition regulation authority blocked the acquisition of the second-largest company by the largest, which would have increased market concentration above 30% (Martello, 2017).

# EQUITY-PROMOTING REGULATIONS ARE NOT COMMON

In addition to helping ensure minimum quality standards, regulations can help promote equitable access. Many countries have quotas or special admission criteria to improve disadvantaged groups' access to tertiary education. However, the criteria are not always extended to non-state providers. Brazil's affirmative action policy, which guarantees 50% of posts to disadvantaged students, applies only to public institutions, which enrol just about one quarter of students (Brazil Ministry of Education and Culture, 2021). In Indonesia and Viet Nam, public universities must provide financial aid grants to at least 20% of their student population and scholarships to at least 10% (Salmi, 2020). Romania's government reserves fee-free places only in public universities for students from rural secondary schools and Roma students (Altbach et al., 2021).

Where regulations do apply to non-state institutions, they are usually enforced only for non-state providers that receive public funding. In India, quotas reserved for students from Scheduled Castes and Scheduled Tribes apply only to public institutions and non-state institutions that receive government funding (Henry et al., 2020; UNESCO, 2017). Supported by the Constitution, however, some regions have also applied the quota system to independent non-state institutions. The Bihar state government enforces quotas for women in non-state institutions in addition to categories determined by law (Henry et al., 2020).

Some countries apply admission policies to all non-state tertiary education providers. In the Plurinational State of Bolivia and Ecuador, non-state tertiary institutions are required by law to provide scholarships to 10% of students (Ferreyra et al., 2017). Non-state universities in Mexico must provide grants or scholarships to at least 5% of students. In England (United Kingdom), institutions must commit to spend a fixed proportion of fee income on scholarships and bursaries (Salmi, 2020). Another equity-related regulation is the capping of fees to keep institutions affordable and accessible to a larger share of the population. This is common for public universities but can also apply in non-state institutions. In India, several states have fee regulatory committees that oversee non-state institutions' fees to stop them from charging exorbitant amounts (Muzammil, 2019); the same is true in Azerbaijan (Salmi, 2020). In Kenya, the regulator rejected a recent request by public and non-state universities for a fee increase because it could restrict access to students from poor families (Nganga, 2019). Governments may indirectly regulate fee levels, as in Chile and Côte d'Ivoire, by establishing a reference price that is used to calculate scholarship amounts for poor students in non-state institutions (Salmi, 2020).

#### FINANCING MODALITIES HAVE QUALITY AND EQUITY IMPLICATIONS

There is heated debate over the degree to which governments and households should share the financing of tertiary education. In most countries, wealthier groups enjoy disproportional access to tertiary education; public financing of tertiary education therefore risks exacerbating inequality. While there are both private and social returns to investment in education, the relative share of returns appropriated by individuals through higher wages is larger in higher education than in basic education. In terms of allocating the education budget, a case can be made for prioritizing pre-tertiary education on equity grounds, letting those who benefit more from tertiary education be responsible, at least in part, for financing it.

Opponents of such cost sharing believe government should be responsible and public provision is the best way to guarantee that access to tertiary education does not depend on ability to pay. Inequality, in this view, should be addressed by increasing the number of participants in tertiary education; systems require more, not less, government presence (Johnstone, 2004). How these arguments play out depends on countries' historical, political and economic context. The debate is usually one of degree – it is rare to find advocates of a system financed entirely by the government or by households (Altbach et al., 2021).

Most non-state institutions, especially those that are smaller and non-elite, rely on fees for their funding. Non-state institutions in some countries, including Colombia and Mexico, rely exclusively on households (Figure 7.5). In a survey of 109 countries, non-state institutions described over-reliance on household funding as a financial risk during the COVID-19 crisis (Marinoni et al., 2020). In Ethiopia, where nearly all non-state institutions rely almost exclusively on fees, COVID-19 has caused a serious financial blow. Pressured by students, non-state institutions agreed to a 25% reduction in fees and many cut employee salaries by over 50% (Tamrat, 2021). In Ghana, about 50% of non-state institutions' students left campus with unpaid fees. In Uganda, non-state universities reported difficulties paying bills and teacher salaries (Levy et al., 2020; Tamrat, 2021). In addition to a drop in fee revenue, household spending on room and board declined. In the United States, most higher education institutions with on-campus housing receive between 10% and 25% of their revenue from such auxiliary sources (Kelchen, 2020). The degree to which institutions rely on fees is at least partly related to their access to government funding. In some countries, non-state institutions receive at most minimal public funding. In Argentina, non-state universities cannot receive any public funding, direct or indirect (Altbach et al., 2021). In most countries, however, non-state institutions have at least some access to public funds. In Western and Northern Europe, it is common for tertiary education institutions that are privately controlled to receive at least 50% of their funding from public sources. In Finland, Iceland and the United Kingdom, all privately controlled institutions receive at least 50% of their funding from the public sector (OECD, 2020).

Governments finance non-state institutions directly, through tax breaks and research grants, or indirectly, through students, in the form of scholarships and loans. In Australia, the government is the initial source of

#### **FIGURE 7.5:**

Non-state institutions in middle- and high-income countries rely far more on household funding than state institutions Share of tertiary education institutions' revenue that comes from households, by sector, selected countries, 2017



Note: Percentages in parentneses represent the share of students enrolled in private institutions. Private institutions include both completely ind and government-dependent private institutions.

GEM StatLink: https://bit.ly/GEM2021\_fig7\_5

Source: GEM Report team calculations based on OECD (2020).

99

55% of total spending on tertiary education institutions, with nearly one third of this transferred to households in the form of financial aid (OECD, 2020). In Indonesia, where most students are enrolled in non-state institutions, the government subsidizes a small share of the academic staff as civil servants and allows institutions to compete for public funds. In Malaysia, students enrolled in non-state institutions are eligible for the public student loan programme, which covers about half of these institutions' income. Thailand's government has a fund specifically dedicated to non-state tertiary education institutions to help them improve facilities and human resource development (Welch, 2021).

The degree of public funding may also depend on the type of institution. In Belgium, the government subsidizes church-based institutions, which enrol over half of all students. In some Latin American countries, including Chile, the Dominican Republic and Nicaragua, the state provides the same level of funding to the oldest non-state universities as to public universities (Salmi, 2020). In Japan and the Republic of Korea, prestigious non-state universities receive substantial government funding (Welch, 2021).

Access to public funds can help improve the quality of non-state provision. An important example is eligibility for public research grants, which increase the likelihood of non-state institutions engaging in research and, in turn, their ability to attract better-qualified professors (Teixeira et al., 2016). Moves towards competitive neutrality have been made by countries including Indonesia and Thailand (Welch, 2021). Governments may also target or condition funds to foster a specific agenda. The University Grants Commission in India (India University Grants Commission, 2021) and the Fund for Students with Disabilities in Ireland (Ireland Higher Education Authority, 2021; Salmi, 2020) fund both public and non-state institutions to provide services aimed at students with disabilities.

#### HOUSEHOLDS ARE SUPPORTED IN TAKING ON A LARGER SHARE OF TERTIARY EDUCATION FUNDING

Household funding of tertiary education is increasing for both public and private institutions. Yet shifting tertiary education costs to households does not necessarily mean all households ultimately pay, or pay equally. Student financial support systems, whether financed and supported by the government or non-state actors, have major implications for equity (Chapman, 2016).

Institutions can be funded directly through fee subsidies, although such aid to students who attend non-state institutions tends to be limited. It usually happens in

#### 66

Over 70 countries operate student loan programmes, most of them government subsidized

countries with larger non-state sectors, where the government provides targeted fee subsidies to students of both state and non-state institutions. In Brazil and Chile, where over 70% of students are in non-state institutions, governments offer targeted fee subsidies to low-income students attending selected non-state institutions (Ferreyra et al., 2017; Salmi, 2020).

Some countries, again mainly those with a large non-state sector, channel public funds directly to students of state and non-state institutions through scholarships, grants and loans. In Côte d'Ivoire, the government offers scholarships for students taking the short professional course towards the Brevet de Technicien Supérieur diploma in non-state institutions (Salmi, 2020). In Colombia, the government student loan agency, ICETEX, administers scholarships for underserved groups to attend state and non-state tertiary institutions. The Álvaro Ulcué Chocué Fund, for instance, supports indigenous groups (ICETEX, 2020).

Over 70 countries operate student loan programmes, most of them government subsidized (Salmi, 2020; Ziderman, 2017). Loans are often available to students of both types of institutions, although borrowing limits can deter enrolment in more expensive institutions. In Viet Nam, low-income students from both state and non-state institutions have access to public student loans, although the maximum amount is relatively low, covering about 84% of the education cost in public institutions and 52% in non-state institutions (Doan et al., 2020). Loans' ability to increase equitable access to education is mixed (**Chapter 12**). Moreover, increased availability of student loans may lead to fee hikes, as in Brazil (de Mello and Duarte, 2020) and the United States (Eaton et al., 2018; Goldin and Cellini, 2014).

Non-state actors can help cover tertiary education costs through scholarships paid for by companies, foundations, NGOs, philanthropists and non-state tertiary education institutions themselves. The Strategic Plan for Higher Education in Bangladesh: 2018–2030 establishes that non-state institutions must earmark 2% to 5% of their revenue for scholarships or grants (Bangladesh University Grants Commission, 2018). Non-state tertiary institutions provide scholarships to up to 30% of their students in Uruguay (Ferreyra et al., 2017). In the United States, non-state non-profit institutions have been increasing the share of fee revenue used for financial aid, which reached 46% in 2018 (NACUBO, 2019).

Non-state actors also grant student loans. Commercial banks, for instance, often disburse and collect loans (Chapman, 2016; Ziderman, 2017). An example of an entirely non-state alternative to financing tertiary education is the nascent development of platforms for peer-to-peer loans, where students borrow directly from an individual without going through a financial institution (Assomull et al., 2015). Another example is financing education through equity instead of debt, or income share agreements (**Box 7.5**).

# SOME NON-STATE ACTORS FUND INSTITUTIONS BEYOND FEES

Non-state tertiary education institutions look for ways to diversify their sources and types of revenue beyond fees. One way is to raise capital through loans. Multinational corporations are important creditors in the tertiary education market, particularly in low- and middle-income countries. The World Bank's African Higher Education Centers of Excellence project focuses on strengthening selected institutions' ability to deliver quality and build research capacity in priority areas (Salmi, 2017). The programme was expanded in 2019 to science, technology, engineering and mathematics institutions in Burkina Faso, Djibouti, Ghana, Guinea and Senegal; more than US\$450 million will have been disbursed by its end (World Bank, 2019b). The International Finance Corporation provides loans specifically to non-state tertiary education institutions. By 2017, it had financed nearly US\$1.4 billion in technical, vocational and tertiary education projects. By far the top recipient region is Latin America and the Caribbean (48%), followed by East Asia and the Pacific (13%) and Europe and Central Asia (12%) (IFC, 2018).

Besides loans, bonds are increasingly an alternative source of funding for tertiary education institutions (Katsomitros, 2018). The bond market reached record numbers during the COVID-19 crisis. By mid-2020, bond issuance by universities worldwide had reached US\$11.4 billion, more than double the amount in 2019. The main players in the bond market were institutions in Australia, Brazil, Canada, Singapore and the United States (Bahceli, 2020). Access to bond markets, however, may increase the gap between institutions, as those with better reputations and credit ratings are more likely to attract investors. In the United States, for example, public institutions tend to have higher credit ratings than non-state institutions, even if all non-profit institutions benefit from tax exemptions (Howard, 2020).

#### BOX 7.5:

# Income share agreements are a market-based initiative to fund tertiary education

Scholarships and loans may be insufficient to remove all financial barriers to tertiary education. Scholarships require donor funding, are non-refundable, tend to have a limited reach and are often focused on specific groups. Loans require collateral and co-signers may be inaccessible to some students.

Income share agreements were designed to fill the gap. They finance tertiary education through equity instead of debt. Investors fund students' tertiary education in exchange for a fixed share of their future income for a defined number of years. The period of the payment and share of income depend on the programme's estimated value. As with income-contingent loans, repayment depends on the graduate's income, but unlike in loans, the payment is not based on principal that is owed. To avoid exorbitant repayments, many institutions cap total repayment at 1.5 to 2.5 times the initial amount borrowed (Salmon, 2020).

Income share agreements can be made with education providers, potential employers or direct investors, who become asset owners of students' future earnings. Such agreements are increasingly adopted in Latin America and the United States (Salmon, 2020) and expanding to African countries. Brighter Investment, for example, pools potential donors and connects them to students from developing countries wishing to finance their degree in approved programmes through income share agreements (Salmi, 2017). In 2019, Brighter Investment signed a partnership with Ghana's government to provide information on top-performing students, facilitate their placement with relevant employers, use tax records and pension savings to locate them and enforce repayments through the Ministry of Finance (Mathot, 2019).

Many income share projects have an equity focus. CHANCEN International, a non-profit based in Germany, offers income share agreements to women who attend the Akilah Institute for Women in Kigali, Rwanda. Graduates who earn above US\$80 are expected to repay 9% of their net monthly income for 8 years (CHANCEN, 2021). CHANCEN is raising funds to expand the programme to 10,000 students in sub-Saharan Africa (CHANCEN, 2020).

Proponents of income share agreements note that not only do they provide insurance against uncertain earning prospects, but the risks are taken by investors who then have a strong incentive to help graduates find good jobs. Critics say such agreements restrict opportunities to students and programmes deemed economically valuable. They also call for regulations on tertiary education financing to be updated to cover such agreements (Salmon, 2020). By mid-2020, bond issuance by universities worldwide had reached US\$11.4 billion, more than double the amount in 2019

99

Public-private partnerships also diversify tertiary education institutions' revenue. In China, public tertiary education institutions generate income by partnering with private entrepreneurs to develop and sell housing around the institution (Welch, 2021). In Mexico, public institutions have incorporated non-state institutions into their programmes in exchange for an incorporation fee calculated as a percentage of the non-state institutions' workforce (Gérard, 2020). In the United States, public and non-state institutions' endowment funds are so important that they have been called hedge funds with a university on the side (Gilbert and Hrdlicka, 2017). Critics say some universities have shifted their priorities towards investment and away from education and research. Another criticism is that endowment funds benefit from tax exemptions, and much of their income is used for high compensation packages and bonuses for money managers instead of students, staff or the institution (Samuels, 2013; Taylor, 2016).

Some public universities in sub-Saharan Africa have generated income through market activities. The University of Ghana privatized services such as guest houses and land leases for private hostels (Varghese, 2016). In Kenya, the University of Nairobi commercialized products (bookstores, restaurants) and services (clinical chemistry, diagnostic radiology, veterinary farm and a funeral parlour) (Provini, 2019). In Nigeria, the University of Ibadan has offered consulting and set up endowments (Varghese, 2016). Throughout the region, to promote and coordinate their income-generating activities, institutions have established companies, which exert considerable influence on institutional governance (Wangenge-Ouma, 2018).

Philanthropic foundations are another direct source of funding to tertiary education institutions. Based on an annual survey by the Council for Advancement and Support of Education, in the United States, foundations gave US\$16.4 billion to tertiary institutions in 2020 and alumni contributed US\$11 billion, jointly making up 56% of the total raised by tertiary institutions overall (Kaplan, 2021). An OECD survey of 143 foundations working in developing countries found that philanthropists donated about US\$549 million between 2013 and 2015 to tertiary education in the form of scholarships or funding for universities, colleges and polytechnics. A significant proportion of philanthropic funding is channelled through intermediary organizations, such as the World Bank or bilateral donors, in order to pool resources and increase the scale of efforts (OECD, 2019).

Foundations have also been working with governments to better target financing. Education Sub Saharan Africa was founded in 2016, using Robert Bosch Foundation seed funding, to develop tertiary education in the region with the aim of increasing the visibility of research on education by scholars in sub-Saharan Africa, gather evidence on academic staff in tertiary education and assess scholarships' effectiveness and accessibility. Other donors have since joined it, including the Jacobs Foundation, the Mastercard Foundation and the Schaufler Foundation, partnering at the regional and local levels (ESSA, 2019; OECD, 2019).

#### NON-STATE ACTORS INFLUENCE TERTIARY EDUCATION THROUGH MULTIPLE MECHANISMS

Non-state actors influence tertiary education in many ways. There is concern that commercial private actors can influence non-state tertiary institutions towards their interests, such as research priorities or enrolment expansion, rather than education quality. Yet other non-state actors can strengthen equity or the sector overall.

In recent years, many governments have boosted incentives for closer relationships between non-state actors and research. In Bangladesh, the government encourages state and non-state universities to collaborate with industry through collaborative research, contract research and consultancy. While the Strategic Plan for Higher Education envisages transparency and ethics guidelines, they are not yet in place (Bangladesh University Grants Commission, 2018; World Bank, 2019a). In France, the 2020 PACTE law, aiming to facilitate private sector access to public research, allows public university researchers to devote 50% of their time to work for a private company and to own up to 32% of its capital (Caulier, 2020). However, non-state funding of tertiary education research is not without controversy. In clinical research sponsored by pharmaceutical or medical device companies, there is an increased likelihood of finding positive results (Lundh et al., 2018). Critics have also argued that allowing private companies to contract research on specific topics undermines institutional autonomy (Oliveira, 2015).

The for-profit sector has seen the emergence of powerful publicly traded tertiary education conglomerates with strong lobbying powers

99

Non-state actors lobby for their own operation and expansion. The Brazilian association of non-state tertiary education providers is pressuring government to change the accreditation process so that they can be accredited by a non-state agency (O Sul, 2019). A strategic aim of the International Federation of Catholic Universities, which has over 200 members, is to engage with international bodies such as the OECD, UNESCO and the World Economic Forum to increase public support for Catholic universities (FIUC, 2021). The influence of non-state actors through funding can also be exerted via participation in discussions and consensus building. Staff of several scholarship programmes, for example, develop long-term relationships and engagement with partner universities and may influence practice, priorities or curricula (Campbell, 2021).

The for-profit sector has seen the emergence of powerful publicly traded tertiary education conglomerates with strong lobbying powers. In the United States, where for-profit enrolment accounts for about 5% of the tertiary education student body (NCES, 2019), most top donors lobbying on behalf of for-profit education are owners of for-profit tertiary institutions. Politicians have received generous donations from for-profit tertiary education companies (Arke, 2020; Halperin, 2016). Since 2010, the shrinking for-profit market has pushed many companies to establish tertiary education institutions in developing countries (Green, 2018; Knobel and Verhine, 2017). At its peak, when it went for a public offering, the biggest tertiary education company in the world, Laureate, based in the United States, had 95% of its 1 million students enrolled abroad (Debter, 2017). It has since downsized, having closed or sold foreign units in several countries, such as India and Turkey, and is in the process of selling remaining units in Brazil and the United States: As of mid-2021, 184,000 students were enrolled in Laureate institutions and it planned to maintain just four universities in Mexico and Peru (Laureate, 2021).

Indeed, an important destination for such companies has been Brazil (Knobel and Verhine, 2017), where the for-profit sector accounts for over half of tertiary education enrolment (INEP, 2020). The 10 largest tertiary education companies in Brazil had estimated revenue of over US\$3.3 billion. The largest, Kroton, enrolled over 800,000 students (Cunha, 2018). Tertiary education companies are thought to exert substantial influence over politicians. Before taking office, the finance minister had considerable investment in private education companies, many of which had been investigated under corruption and fraud allegations, while his sister has been vice president of the National Association of Private Universities (Chiaverini, 2018; Guasco Peixoto, 2018). Other politicians across the political spectrum have received sizeable campaign donations from large for-profit companies, raising concerns about potentially undue influence on regulation (Leray de Lima, 2018). The media has questioned donations to candidates who helped increase public student loan availability, which in turn helped institutions increase enrolment (Pompeu et al., 2016).

Governance reforms in several countries reflect a shift towards more businesslike processes and a management style that borrows from the private sector, including in the composition of institutions' governing boards. As early as 1992, the United Kingdom had reformed the composition of tertiary education governing boards by requiring the majority of members to be independent, with 'demonstrated capacity in industrial, commercial or employment matters of a given profession' (Bennett, 2002). The Canadian Association of University Teachers has expressed concern about the shift towards corporate-like management of universities through the growing presence of board members from the for-profit corporate sector (CAUT, 2018).

In Ethiopia, Addis Ababa University introduced a 'business process re-engineering' approach to management in an attempt to optimize efficiency and improve work processes. In Kenya, a results-based management approach introduced performance contracts between the state and public institutions that focus on measured performance and outputs. The South African government introduced a performance-based funding framework that set specific enrolment and output targets. The success of such reforms has been mixed. In Ethiopia, some scholars disapprove of the shift in attention from academic to operational competence. But in Kenya, the reforms are credited with having propelled the University of Nairobi to the top tier of national rankings (Varghese, 2016).

Civil society actors have been important advocates for tertiary education reform. The Romanian Coalition for Clean Universities, for example, has played an important role in monitoring the integrity and transparency of tertiary education institutions and has issued an integrity ranking to increase awareness and accountability (Mungiu-Pippidi and Dusu, 2011). Non-state actors also head several equity-related advocacy initiatives in tertiary education. In 2016, UNIMED, a network of 141 universities in 23 countries, together with the University of Rome La Sapienza, the University of Barcelona, Campus France, the European University Association and the United Nations High Commissioner for Refugees, established the inHERE consortium and project to strengthen knowledge and communication on refugees and displaced students (Salmi, 2020).

Research networks and academic societies help strengthen the tertiary education sector by organizing conferences and seminars and publishing peer-reviewed journals. Low- and lower-middle-income countries, however, lack such organizations. While research organizations make up about 17% of education non-profit organizations in rich countries, they are virtually non-existent in poor countries (Owens, 2017).

#### CONCLUSION

Non-state actors play important roles in the provision, financing and management of tertiary education. In most cases, their roles are so intertwined and interdependent with those of the government that the distinction between the two types of actors is often blurred.

Non-state provision of tertiary education accounts for more than one third of students worldwide, a considerably higher share than in primary or secondary education. Non-state actors are important players in the financing of tertiary education, not only through increasing participation by households, but also through market mechanisms and public-private partnerships. As a result, these actors play a significant role in influencing regulations and policymaking, and in shaping the tertiary system as a whole.

Participation of non-state actors in tertiary education raises both challenges and opportunities. The same is true regarding their influence on regulation and policymaking. Governments have several tools to ensure that, regardless of how state and non-state actors share responsibility, the tertiary education system continues to strive for more quality and equity.

Metalwork apprentices work at the vocational training centre of the Remscheid metal and electrical industry in Germany.

8-

E.

CREDIT: Rupert Oberhäuser/Alamy Stock Photo





# Technical, vocational and adult education



# **KEY MESSAGES**

#### Initial skills development is dominated by public institutions.

- In OECD countries, about 20% of students in upper secondary vocational education and 40% of students in post-secondary non-tertiary vocational education were enrolled in private institutions.
- In some countries, technical and vocational education has expanded through private providers. Using tripartite dialogue, Morocco restructured training institutions to make them more responsive to companies' needs.

#### Continuing skills development is dominated by private institutions.

- In Europe, employer-sponsored job-related training amounted to 85% of non-formal job-related training in 2016. The share of firms providing internal and external continuing vocational training increased from 48% in 2005 to 59% in 2015.
- In lower-middle-income countries, almost one in three firms provide training to full-time permanent employees; in low-income countries, only one in four enterprises does so.

Cooperation with non-state actors is crucial for anticipating skills demand and integrating non-formal learning into recognized frameworks.

- Linking skills with qualifications and ensuring accreditation of learning acquired outside the formal system is key. However, employers are not very willing to accredit informal and non-formal learning, which remains a task for public authorities.
- Quality assurance mechanisms need to be based on enforceable accreditation standards and on credible skills testing and certification. In Uganda, a complex process meant only one in four private training providers was registered.
- Collaboration with non-state actors tends to focus on skill needs identification and standards definition. But
  in the Republic of Moldova, a social dialogue platform with employer organizations and trade unions through
  skills committees has also helped define curricula.

#### The right mix of state and non-state financing supports optimal skills development.

- Governments finance training directly through grants to public centres or competitive processes. Some of the costs may be recovered through tuition or levies on firms. Around 70 countries use levy-financed training funds.
- Governments may subsidize students' or workers' training costs or offer incentives to firms to invest in training. Training represents less than 2% of firms' total labour cost in Europe.
- Individual learning accounts, vouchers and grants cover direct or indirect training costs. In Singapore, over half a million students have benefited from a US\$365 per student credit to be spent on skills and training participation in the resident workforce.

#### Non-governmental and community organizations are the main providers of adult education.

- Almost 8 in 10 countries have reported engaging with non-government actors for adult education since 2015, benefiting from their networks and flexible learning approaches. But there is limited consultation with non-state adult education providers.
- Private providers are most active in for-profit markets. About 40% of English language learners in Argentina and Peru study in private institutions.

Apart from entry-level skills, non-state actors dominate technical and vocational education	184
Governing skills development systems with non-state actor participation is challenging	189
The right mix of state and non-state financing can support optimal skills development	192
Non-state actors are a driving force in adult learning and education	195
Conclusion	200

Adults acquire skills for work and life through a variety of means. Formal technical and vocational education and training (TVET) tends to produce graduates with traditional skills. But in practice, economies constantly create jobs with new skills requirements that need to be filled as soon as possible, at a pace formal systems cannot match (ILO, 2020c). As firms are the main provider of workers' skills, a key policy question is how to ensure that the level of such investment is optimal for society. Public authorities cannot realistically provide and finance the acquisition of all skills. Finally, some, maybe even most, skills are routinely acquired outside formal or non-formal contexts, as individuals follow informal paths to lifelong learning opportunities, which Sustainable Development Goal 4 aims to promote.

This chapter consists of two parts. The first discusses the role of non-state actors in formal and non-formal development of skills for work. It looks at their role both in provision and in regulatory, governance and financing mechanisms. The second part addresses the role of non-state actors in adult learning and education. Such actors play a far more dominant role in the education of adults than in that of children and youth, and the government thus has a lesser role in provision, financing and regulation but a greater role in promoting education, training and learning activities.

66

Most students in formal vocational education attend public institutions, although the share of enrolment in private institutions increases at higher education levels

#### APART FROM ENTRY-LEVEL SKILLS, NON-STATE ACTORS DOMINATE TECHNICAL AND VOCATIONAL EDUCATION

TVET takes a variety of forms depending on the national context. It 'can take place at secondary, post-secondary and tertiary levels and includes work-based learning and continuing training and professional development which may lead to qualifications' (UNESCO, 2016a, p. 9) (**Figure 8.1**). Governments provide and regulate services, aiming to promote equity and quality. Yet their impact is often limited beyond entry-level skills, and they respond slowly to changing demand. As workers need to upgrade their skills, direct involvement of non-state actors in skills development increases and collaboration with government becomes necessary (Dunbar, 2013; Glick et al., 2015).

In 2019, 5% of 15- to 24-year-olds were enrolled in TVET programmes, with regional shares ranging from 1% in sub-Saharan Africa to 18% in Europe. Globally, about 1 in 100 lower secondary school students (ISCED level 2) and 1 in 5 upper secondary school students (ISCED level 3) were enrolled in vocation-oriented programmes; in Organisation for Economic Co-operation and Development (OECD) countries, the ratio was more than 2 in 5 students in upper secondary education in 2018 (43%). By contrast, 92% of students in post-secondary non-tertiary programmes (ISCED level 4) and 96% in short-cycle tertiary education (ISCED level 5) were enrolled in vocational programmes in OECD countries (OECD, 2018). Worldwide, 89% of the students enrolled in post-secondary education (ISCED 4) were in a vocational track in 2018.



Most students in formal vocational education attend public institutions, although the share of enrolment in private institutions increases at higher education levels. According to UNESCO Institute for Statistics (UIS) estimates, the share of private institutions in post-secondary, non-tertiary enrolment was 38.5% in 2019, with shares ranging from 12% in East Asia to 76% in Oceania. In OECD countries, on average, 20% of students in upper secondary vocational education but 40% of students in post-secondary non-tertiary vocational programmes and 44% of students in short-cycle tertiary vocational programmes were enrolled in private institutions (**Figure 8.2**).

Some countries have made considerable efforts to expand public TVET provision and attract more students by making the system more responsive to labour market demands. The Czech Republic enhanced students' ability to transfer between schools and improved cooperation between schools and employers. To make TVET more suitable for adults, shorter one- to two-year programmes were established for those who wanted to get a vocational qualification and had already completed secondary education; enrolment in these programmes quadrupled between 2011 and 2018 (Kaňáková et al., 2019). In the European Skills Index, which measures TVET system capacity to develop, activate and match skills, the Czech Republic ranked first between 2018 and 2020, due especially to its performance in skills matching (Cedefop, 2020).

Estonia modernized its vocational education and expanded post-secondary skills development

#### FIGURE 8.2:

#### In OECD countries, post-secondary students are more likely in vocational than in general programmes to enrol in private institutions

Share of enrolment by type of institution, programme orientation and education level, OECD countries, 2018



GEM StatLink: https://bit.ly/GEM2021\_fig8\_2 Source: GEM Report team calculations based on OECD (2018). opportunities to adult learners. The 2013 Vocational Educational Institutions Act set out general principles for the reform, including a framework for establishing and closing vocational schools, as well as their financing, and quality assurance criteria based on the teaching workforce. In 2016/17, of the country's 34 TVET institutions, 29 were public (26 run by the Ministry of Education and Research and 3 by municipalities) and 5 private. Programmes for post-secondary non-tertiary education were established, with provision by five professional institutions. Following this restructuring, TVET programme enrolment of students older than 25 increased (Musset et al., 2019). The Lifelong Learning Strategy 2020, launched in 2014, emphasized the need to make TVET relevant, aligning skills development with lifelong learning objectives. In 2016, new skills demands, based on the System of Labour Market Monitoring and Future Skills Forecasting, were identified to inform TVET provision (UNESCO, 2019).

Updated curricula and engagement with employer organizations can address new needs (ILO, 2020b). Well-established dual vocational systems, based on tripartite regulatory arrangements and alternance between an education institution and a workplace, can deal with changes. In Austria, Germany, Iceland, Norway, Sweden and Switzerland, the unemployment rate of young adults aged 25–34 years with a vocational secondary or post-secondary non-tertiary qualification is less than half that of people with a corresponding general qualification (OECD, 2020a).

In other countries, TVET has expanded through private providers in various ways. In South Africa in the mid-1990s, the White Paper on Education and Training separated initial vocational education, supported by public further education and training (FET, later TVET) institutions, from continuing vocational skills development at the workplace, dominated by private actors. The latter were favoured by legislation allowing private providers to apply for accreditation of their qualifications and transferred apprenticeship from initial to continuing education (Needham, 2019).

In South Asia, demographic changes and rapid economic transformation have led national skills development strategies to acknowledge the growing role of private actors, both for-profit and not-for-profit. While the strategies aim to improve training quality and flexibility for a more competitive workforce, they also recognize the extensive dependence on private actors for training provision, encouraging their further expansion (Béteille et al., 2020). The role of non-state actors in promoting equitable access to TVET varies accordingly by context (**Box 8.1**).

The private sector delivers learning services through traditional procurement but also promotes training initiatives, contributing to the development of content and teaching (Glick et al., 2015). Over time, private institutions have complemented or even substituted for the vocational education system, filling skills gaps where public institutions have limited resources or capacity for timely responses to the demands of rapidly changing global industries and international standards (Fernandez-Stark and Bamber, 2018). As information technology companies emerged in India in the late 1970s and early 1980s, public vocational education institutions had to involve them to be able to respond to increasing demand for trained and skilled professionals. Conversely, private providers have been less involved in more established sectors, such as construction, manufacturing and retail (Flake et al., 2017). In the emerging gas sector of the United Republic of Tanzania, only 19% of TVET institutions are public. Private companies or individuals provide more than one third of the training for relevant occupations, faith-based institutions 31% and civil society organizations (CSOs) 12% (VSO Tanzania, 2014).

Expansion of TVET provision has been supported by public-private partnerships (PPPs), mostly coordinated by the state. Each actor in a PPP contributes according to its resources, expertise and interests (Fernandez-Stark and Bamber, 2018). Unlike in traditional forms of cooperation on workplace learning and/or apprenticeships, provision-oriented PPP coordination mechanisms address career services, orientation and guidance, and training renewal (ETF, 2020a). Morocco signed agreements with the private sector as part of the 2009-15 National Pact for Industrial Emergence implementation. Using tripartite dialogue, the government restructured training institutions to make them more responsive to companies' needs. In line with the 2016-21 National Strategy for Vocational Training, delegated management institutes were set up in relevant sectors, in partnership with professional associations, whose representatives became part of the institutes' management. In the automotive industry, for instance, the reform led to increased participation (enrolment more than doubled from 2017 to 2018) and a higher student employability rate (ETF, 2020b).

66 -

Expansion of TVET provision has been supported by public-private partnerships, mostly coordinated by the state

**?**?

#### BOX 8.1:

#### Depending on the country, non-state actors may encourage or obstruct equity in TVET

The extent to which formal TVET is accessible to groups at risk of exclusion, and hence its impact on their equity, varies by country. In poorer countries, tight entry requirements and inadequate provision mean the few places in formal public TVET tend to be taken by urban, relatively well-off youth. In much of sub-Saharan Africa, rigid academic achievement requirements for access to formal school-based training, combined with a concentration of training institutions in cities, make TVET inaccessible to poor and rural youth (Afeti, 2018). In South Asia, the minimum requirement for formal TVET tends to be eight years of schooling, which excludes the poorest and those living in rural areas (Mehrotra, 2017). In some cases, non-state actors respond to this challenge. In Indonesia, religious institutions, which cater for more disadvantaged populations, have been a partner in education expansion. Formal TVET is provided in both state and religious vocational secondary schools, which all offer national secondary education certification (Triyono and Moses, 2019).

CSOs cater for population groups whose needs are not covered by government policies. A survey of 85 relevant CSOs in Northern Africa and Western Asia showed that most of their activities are dedicated to initial TVET provision and to training for groups in disadvantaged situations, such as vulnerable youth, the unemployed, migrants and refugees, and people with disabilities (Foubert and Folisi, 2019). Non-governmental organizations (NGOs) and private agencies also tend to be important implementing actors in low- and middle-income countries for vocational training programmes that are targeted at women and outsourced by governments. Quality control and monitoring procedures for these programmes are often challenging (Chinen et al., 2017). Some businesses with philanthropic arms try to provide TVET with an equity lens. Google's philanthropic arm funds provision of digital skills to disadvantaged rural communities in the 10 countries of the Association of Southeast Asian Nations (ASEAN) through the Go Digital ASEAN initiative, launched in 2019 (The Asia Foundation, 2020).

In richer countries, formal TVET tends to target more disadvantaged populations. Yet some students, particularly those with immigrant backgrounds, do not benefit from equal access. Access to apprenticeships, in particular, has discriminated against this group, even in highly regulated dual-system models, where training takes place at both education institutions and workplaces. Enterprises, which make the final selection of apprentices, may contribute to perpetuating discrimination. In Germany and Switzerland, applications of immigrant students, labelled as foreign youth, were more likely to be rejected, especially by small and medium-sized enterprises (Imdorf, 2017). Social partners and NGOs advocate for the relevance of work-based learning and the need for preparatory programmes (Jeon, 2019). In Germany, federal Youth Integration Courses target young migrants and refugees to promote access to TVET. Providers are requested to partner with counselling services to help migrants and refugees receive practical information about further educational, work-related and vocational opportunities and to cooperate with local Youth Migration Services, a social support network whose professionals visit the class at least once a week to build relationships with the participants (Germany Federal Office for Migration and Refugees, 2015).

### EMPLOYERS ENGAGE IN FORMAL AND INFORMAL APPRENTICESHIPS

Apprenticeship refers generally to a form of vocational education and training combining on-the-job and off-the-job learning, enabling students to develop knowledge and skills for a specific occupation. Engaging private businesses in the design of training through communication with TVET institutions ensures a smoother school-to-work transition, reducing potential skills mismatch (ILO, 2019b).

Formal apprenticeships are not very widespread. Analysis of the International Labour Organization (ILO) School-to-Work Transition Survey data for this report found that fewer than 1 in 5 of 15- to 35-year-old participants in 33 countries did at least one apprenticeship as part of their education.

Youth from low-income countries are less likely to participate in formal apprenticeships, possibly because

labour markets and training systems tend to be informal (see **Chapter 12**). Informal apprenticeships, for their part, are bound not to a curriculum but to social norms and traditions, and their completion does not lead to officially recognized qualifications (ILO, 2019b), such as in Africa (AFDB, 2020). Often the only form of skills acquisition available, informal opportunities are based on an agreement between a master craftsperson and a trainee, who commits to work as an apprentice in exchange for being instructed by the master. In Senegal, informal apprenticeship targets people without education, serving 418,000 youth, compared with 54,000 in formal TVET (World Bank, 2018a).

66 -

Fewer than 1 in 5 of 15- to 35-year-olds in 33 countries did at least one apprenticeship as part of their education.

99

The adjustment of skills development to current needs is mainly occurring outside traditional education

If not adequately regulated and recognized, the incentives linked to apprenticeship can be eroded, reducing potential participation. In Egypt, traditional, mostly informal apprenticeship is associated with cheap labour and is not considered an attractive opportunity. The learning process is reported to be mostly passive and the knowledge transfer from master to apprentice to be weak. Remuneration is extremely low, with no guarantee of employment. The workforce apprenticeship participation rate is estimated at 0.1%, compared, say, with 4.8% in Denmark (Chankseliani and Anuar, 2019). Countries including Botswana, Ghana, Kenya, South Africa and Zimbabwe have regulated distinct apprenticeship systems, parallel to other formal TVET programmes. Others, such as Namibia and Rwanda, have planned or are setting up national apprenticeship and internship frameworks based on employer engagement to make them more attractive and the training more credible (Arias et al., 2019).

Intermediary organizations can facilitate dialogue with employers to improve learning, providing information and ensuring the right match, especially assisting small and medium-sized enterprises. They may include for-profit and independent agencies or employers' associations (ILO, 2019a). Denmark's Local Trade Committees, which include representatives of employers and employees, ensure close contacts between the parties. The Association of German Chambers of Commerce and Industry is in charge of supporting trainees and mediating with firms (Chankseliani et al., 2017). In England (United Kingdom), Apprenticeship Training Agencies, introduced in 2012, are interim placement agencies that operate for or not for profit. They are responsible for recruiting, employing and arranging training for apprentices on behalf of employers. They are listed on a national registry and are mainly funded by employers, whose payments includes a management fee and apprentice remuneration of at least the minimum apprenticeship wage (ILO, 2019a).

Beyond facilitating dialogue, intermediary organizations play a significant role in mobilizing parties, especially in countries without a solid apprenticeship tradition. Inspired by the Swiss model, the not-for-profit apprenticeship intermediary CareerWise aims to engage and connect local employers, apprentices, school districts and government entities to make apprenticeships possible in the US state of Colorado. The intermediary organization helps attract and mobilize employers to recruit apprentices from schools, assisting them throughout the three-year work-based learning programme and ensuring high retention rates (Katz and Elliott, 2020).

## PRIVATE BUSINESSES DOMINATE CONTINUING SKILLS DEVELOPMENT

While they have an increasing role in formal TVET, private businesses have always led on continuing training that provides knowledge and skills, or develops, updates and improves existing ones, for youth and adults in the labour market. The role of business has become even more vital in the context of digitalization and globalization, which have brought about an increase in the relative share of both high- and low-skill jobs in industrialized countries, at the expense of middle-skill occupations (Nedelkoska and Quintini, 2018). Formal TVET is targeted at low-skill occupations, which are vulnerable to automation and obsolescence (ILO, 2020b; UNESCO-UNEVOC et al., 2019).

The adjustment of skills development to current needs is mainly occurring outside traditional education (World Bank, 2019b), with non-formal and employer-sponsored training prevailing over formal education and training. In Europe, employer-sponsored job-related training increased between 2007 and 2016, reaching around 85% of non-formal job-related training. Adults were about six times likelier to have participated in non-formal training than in formal education in 2016 (Eurostat, 2021). The share of enterprises providing internal and external continuing vocational training increased steadily from 48% in 2005 to 59% in 2015, according to the Continuing Vocational Training Survey (CVTS, 2005, 2015).

In Australia, although employers reported being satisfied with the TVET system, one in two preferred unaccredited training to meet their staff skills development needs, mostly without relying on external providers (NCVER, 2019). More employers reported being satisfied with this arrangement than those who had used nationally recognized training. While recognized training is used for professional development and to meet mandatory requirements, non-formal education serves highly specific and relevant training needs, and its delivery is more flexible (White et al., 2018). In Belgium, a 10-year

99

study showed that firms that increased the share of employees who participated in on-the-job-training by 10 percentage points saw their productivity grow by between 1.7% and 3.2% (Konings and Vanormelingen, 2015).

In lower-middle-income countries, almost one in three firms provide training to full-time permanent employees; in low-income countries, only one in four enterprises provides training (World Bank, 2019a). An employer survey conducted as part of the World Bank's STEP Skills Measurement Program in Armenia, Azerbaijan, Georgia, Sri Lanka, Viet Nam and China's Yunnan province showed that employers tended to prefer on-the-job training over external programmes provided by formal public or private providers, especially as both the general and vocational education systems were not perceived as responsive to employers' skills needs (Sanchez Puerta et al., 2016).

Firm size is correlated with the offer of training. On average across OECD countries, 60% of adults in firms with at least 250 employees participated in at least one non-formal, job-related, employer-sponsored education or training activity, twice the rate for employees in enterprises with fewer than 10 employees (OECD, 2020a). A study in Viet Nam found that household firms were less likely to provide vocational training due to their small scale and limited resources (Nguyen et al., 2020). Yet traditional measures of training may underestimate families' informal transfer of knowledge and skills in traditional sectors (Pilz et al., 2015). Informal learning can involve family social capital and competitive advantage through passing unique knowledge from one generation to the next. Investing in formal training is unattractive where the cost of training is high and employability is ensured despite the informality of skill acquisition (Chankseliani and Anuar, 2019).

#### GOVERNING SKILLS DEVELOPMENT SYSTEMS WITH NON-STATE ACTOR PARTICIPATION IS CHALLENGING

The dominance of non-state actors in provision makes the task of governing and regulating skills development systems challenging (ETF, 2013). Governance refers to how decisions are taken to steer systems towards achieving national economic development priorities, including their ability to match demand with supply, and ensuring quality while achieving social objectives such as equity and inclusion. Governance needs to be integrated: horizontally, bringing state and non-state actors together to communicate and negotiate their respective needs; vertically, articulating the central,

#### 66

The dominance of non-state actors in provision makes the task of governing and regulating skills development systems challenging

regional and local levels; and over time, meeting the need for initial and continuing skills development over a life cycle (Hawley-Woodall et al., 2015).

99

Formulating and implementing national qualification frameworks has made TVET governance more participatory and fit for purpose, bringing together diverse actors by constituency, sector and government level (ETF, 2013). Linking skills with standardized national qualifications helps balance supply and demand, promote quality and ensure a common and consistent mechanism accepted by all parties (UNESCO and ILO, 2018). Such frameworks have also driven recognition and validation initiatives (Cedefop et al., 2019).

More than 150 countries have developed national qualification frameworks to improve the transparency and relevance of skills and knowledge. Countries have also established programmes that acknowledge and promote flexible learning pathways to enable learners to accumulate and apply skills. Making the outcomes of non-formal and informal learning acquired outside formal public settings visible and socially recognized can help reduce the cost of providing continuing training and qualifications to low-skilled workforces (UIL et al., 2019).

Private sector initiatives to validate competencies allow individuals to gain access to new jobs, be more flexible and reposition themselves in the job market (Hawley-Woodall, 2019). An inventory of 39 mainly European education systems showed that more than 90% had initiatives to validate competencies generated outside official education and training frameworks. However, non-formal and informal learning acquired in the labour market was still less recognized than learning in formal education, and mostly related to occupational standards. Employers are less willing to deal with accreditation of informal and non-formal learning, which remains a task for public authorities (Cedefop et al., 2019).

When individuals change jobs and need to reposition themselves on the labour market by acquiring new skills, it is important to provide processes of assessment and recognition of prior learning, including accreditation (Aarkrog and Wahlgren, 2015; UIL et al., 2019). Particularly for workers moving through the labour market without academic or formal qualifications, accreditation of skills and knowledge gives them an opportunity to gain a qualification and get access to further education and training opportunities (UIL et al., 2015).

To set such processes up and ensure that their results are used, all actors need to work together. In Mauritius, a mechanism recognizes competencies acquired outside formal learning. After a pilot project funded by the National Empowerment Foundation, the mechanism was widely accepted, including by trade unions. Skills and knowledge are assessed against defined standards and if the learning requirements are met, the learning outcomes are officially accredited. Through this recognition, validation and accreditation of prior learning, workers can obtain qualifications at levels 2 to 4 of the national qualification framework and re-enter the education and training system at the appropriate level (UIL et al., 2015).

National qualification frameworks may not be effective without quality assurance mechanisms to ensure coherence and build trust in education and training systems' output. In quality assurance, education and training providers undergo audit, verification and monitoring processes to determine whether specific rules and occupational or educational standards have been respected. Procedures for assessment of achieved outcomes and for accreditation are also controlled (Bateman and Coles, 2013).

A review for this report of quality assurance systems in Bangladesh, China, India, the Philippines, Rwanda, South Africa and the United Republic of Tanzania shows that all seven have adopted a centralized approach. While they all supervise accredited providers, the emphasis differs. The Bangladeshi and Tanzanian approaches focus on facilities and equipment, while the Chinese, Rwandan and South African approaches focus on teacher qualifications' suitability for the curriculum and assess teacher quality improvement plans. In Bangladesh, South Africa and the United Republic of Tanzania, all providers must use an internal quality assurance mechanism to evaluate their programmes (Nesterova and Capsada-Munsech, 2021). But private providers do not always align with the requirements.

66

In Uganda, only one in four private training providers was registered

99

Bangladesh's 2011 skills development policy introduced a quality assurance system to ensure consistent quality standards in training provision and assessment. It consists of registering and auditing public and private training providers and accrediting learning and assessment programmes and centres. To obtain registration by the Bangladesh Technical Education Board, providers have to prove their educational, financial and management capacities. In addition, their programme outcomes must be in line with the National Technical and Vocational Qualification Framework (Bangladesh Technical Education Board, 2012). However, this process applies only to formal TVET programmes. Public, private and NGO-run training programmes can issue certificates independently (Haolader et al., 2017). Moreover, only one in four non-state institutions has recognized the gualification framework as a skills development model, thus challenging effective implementation of the quality assurance mechanism (Nesterova and Capsada-Munsech, 2021).

To be effective, such processes need to be based on enforceable accreditation standards and on credible skills testing and certification. In Uganda, many private providers saw no benefit in registering, as the process was too complex. As a result, only one in four private training providers was registered (Arias et al., 2019). The 2011-20 Strategic Plan recognized the limitations of the highly centralized and costly system and the challenges it posed for private institutions (Uganda Ministry of Education and Sports, 2011). To encourage private providers to apply for formal recognition and to improve training quality, the Uganda Association of Private Vocational Institutions was involved in the process. The 2019 TVET policy further simplified governance, merging national assessment bodies into the Technical and Vocational Assessment Board, which redefined its regulations to make the system clearer and more efficient (Uganda Business and Technical Examinations Board, 2019; Uganda Ministry of Education and Sports, 2019).

#### PARTICIPATORY GOVERNANCE IS OFTEN LIMITED TO DEFINING STANDARDS

The relative degree of engagement of state and non-state actors is one criterion for categorizing approaches to governance of skills development systems in high-income countries. Engagement may be low or high on both sides, or one may play a predominant role (Busemeyer and Trampusch, 2012). Private sector and employer involvement in content development has often been considered an opportunity to align skills development systems with labour market needs (Dunbar, 2013). Low- and middle-income countries, however, tend to view state actors as the most important in TVET planning and implementation. The quality assurance systems in Bangladesh, China, India, the Philippines, Rwanda, South Africa and the United Republic of Tanzania reviewed for this report have all adopted policies allowing private education institutions to operate, while development plans support the expansion of TVET through PPPs. Even so, their TVET systems remain highly centralized, with limited opportunities for non-state actors to shape them (Nesterova and Capsada-Munsech, 2021). Interactions between businesses and education institutions tend to focus on skills identification rather than on curriculum development (Sanchez Puerta et al., 2016).

There are signs of collaboration in some regions. An assessment of highly centralized models of planning and managing TVET policies and systems in Northern Africa, Central and Eastern Europe, the Caucasus and Central Asia identifies a weak trend towards more participatory governance approaches. In eight countries reviewed (Albania, Jordan, Kazakhstan, Morocco, the Republic of Moldova, Serbia, Tunisia and Ukraine), non-state actors, especially employers, were involved in consultative and/or executive advisory functions in tripartite councils (Arribas and Papadakis, 2019).

The TVET system in the Republic of Moldova has undergone significant restructuring and modernization to enhance the institutional role of social, regional and local partners in policy formulation. Within a reformed legal framework – consisting of the 2014 Education Code, 2011–2015 Strategy for Consolidated Educational Development, Vocational Education and Training Development Strategy for 2013–2020 and Small and Medium-sized Enterprise Development Strategy 2012–2020 – a social dialogue platform for vocational education and training has institutionalized partnerships with employer organizations and trade unions through skills committees, which help define occupational standards and curricula (Arribas and Papadakis, 2019).

TVET reform in Mongolia was based on extensive consultation with key stakeholders to inform new legislation and financing mechanisms for a demand-driven system. A memorandum of understanding was signed in 2007 by all social partners, including ministries, employers and trade unions, paving the way for participatory governance at the system level. The 2016 national TVET programme has since tried to strengthen this social partnership to help expand provision and enhance quality (UNESCO and Mongolia Ministry of Labor and Social Protection, 2019).

#### 66

Interactions between businesses and education institutions tend to focus on skills identification rather than on curriculum development

An assessment of 12 Eastern and South-eastern Asian countries showed that occupational or competency standards had been developed in the vocational and training sector with the involvement of multiple actors. Most countries had engaged with industry representatives (e.g. Cambodia, Indonesia and the Philippines), tripartite groups of employer and employee organizations (e.g. Republic of Korea) or public and private training institutions and trainers (e.g. Malaysia, Thailand and Viet Nam) (Bateman and Liang, 2016).

99

In the Philippines, private sector participation is institutionalized in policymaking and the design of industry-based training programmes. Private actors account for 14 of the 22 board members of the Technical Education and Skills Development Authority, the national policymaking body. The authority sets development plans, competency standards, curricula and assessment procedures. The training curriculum for the semiconductor and electronics industry, designed with industry representatives, proved successful in terms of trainee employability (OECD, 2017).

Skills systems that rely on tripartite approaches, engaging governments, employers and workers, cope well with changing economic and labour market dynamics. Employer bodies are more likely to contribute when they are engaged in policy formulation and system development, rather than in an advisory role only. In Singapore, employer organizations are involved in national skills planning as part of the inclusive Future Economy Council, led by the deputy prime minister and finance minister. All members are involved in defining and adjusting the skills sets that will drive the country's economic and industrial development over the coming 5 to 10 years (ILO, 2020a).

With the aim to share capacity and resources, PPPs have been set up to encourage cooperation for a better understanding of the need for labour market competencies and qualifications. In Viet Nam, the car manufacturer VinFast established a centre in 2018 to train mechanical and mechatronic engineers. Through an agreement with the Association of German Chambers of Commerce and Industry, trainers obtain certificates based on German standards (Viet Nam News, 2018). The company and government have since initiated a dialogue on developing a joint training programme, funded by the company and implemented at five public vocational colleges (Viet Nam News, 2020).

In the Netherlands, TVET reform in 2010 initially aimed to create two PPPs: centres for innovative craftsmanship in upper secondary vocational education and centres of excellence in higher professional education. While they received grants, companies were invited to co-finance them. This made education institutions responsive to firms' specific competency needs. The focus is on innovative projects, such as smart organization in home care, three-dimensional printing, digitization in the security sector and skills for the green sector. A regional investment fund helped increase the number of such PPPs, encouraging all TVET centres to compete. Between 2011 and 2019, the number of centres increased from 7 to more than 160, which train about 84.000 students and involve more than 9,800 companies and 5,000 teachers (ETF, 2020b).

Sector skills councils are another kind of cooperation, aimed at improving training quality and relevance, anticipating needs between and within sectors (ETF, 2020a) and fostering social dialogue (ILO, 2020a). With varying composition and functions, the councils gather actors in a given industry to contribute to the development of skills training and education for the workforce in that sector (ILO, 2019c). Sector skills councils in Poland, established in 2016, promote cooperation among social partners, entrepreneurs, and public and education institutions. They provide recommendations on skills needed and related learning opportunities, and play an advocacy role for skills use and practices. The sector skills council on finance was involved in implementing the Sectoral Qualification Framework and its inclusion in the Integrated Qualification System, in line with European standards (OECD, 2019c). With support from the European Social Fund, the Polish Agency for Enterprise Development increased the number of councils from 7 sectors to 17 (Chłoń-Domińczak et al., 2019; Polish Agency for Enterprise Development, 2021).

#### THE RIGHT MIX OF STATE AND NON-STATE FINANCING CAN SUPPORT OPTIMAL SKILLS DEVELOPMENT

Skills provision remains below optimal levels, despite the productivity gains it brings and the multiplicity of actors that can be deployed to deliver it. Ultimately, this is a

result of insufficient funding. There are several obstacles to overcome. The cost per student in formal initial TVET is generally high, as expert educators are in short supply and expensive equipment quickly becomes obsolete. Firms are reluctant to invest in training out of concern that trained workers may take their skills to competitors offering higher wages. Workers' investment in training is limited by lack of money and time and by social norms. The situation is unsatisfactory from a social and economic perspective. Any solution involves not just more funds but also consensus on how to share costs.

Governments finance TVET directly through grants to public centres or competitive processes, sometimes pitting public TVET centres against private ones. Some of the costs may be recovered through tuition or levies on firms. Governments may subsidize students' or workers' training costs or offer incentives to firms to invest in training (Ziderman, 2016) (**Figure 8.3**).

#### DIVERSIFYING TRAINING INSTITUTIONS' FUNDING HAS ADVANTAGES AND RISK

Formal TVET institutions have traditionally been funded through direct government allocations, as in Eastern Europe, Northern Africa, Western Asia, Central Asia (Arribas and Papadakis, 2019), sub-Saharan Africa (Arias et al., 2019) and East Asia and the Pacific (Palmer, 2017). However, there are signs of diversification.

Earmarked training levies on firms are a common complementary source of funding for public and private TVET institutions. They require both a large formal sector to provide enough funds and a strong administration to use them effectively. In Senegal, up to 2015, only 5% of the amount raised by a 3% levy on firms yielded had been transferred to TVET; as little as half of that was allocated to training, as operating costs absorbed the rest (UNESCO, 2018).

Training levies on firms have been used to set up national training or skills development funds that can operate with relative autonomy from the government budget. However, this is often not enough. In the United Republic of Tanzania, more than 80% of the Vocational Education and Training Fund is financed through a levy. But just one third of the revenue raised goes to the Vocational Education and Training Authority; the rest goes to the general budget. Companies with four or more workers contribute 6% of their payroll. Employers protested that the levy had been introduced with limited consultation, that it was used to cover expenditure unrelated to training, that public TVET centres were prioritized over other providers and that the private sector had not been engaged in the fund's management (Arias et al., 2019).

#### FIGURE 8.3:

Individuals, firms, governments and donors fund skills development Skills development financing mechanisms



#### 66

# Some countries involve non-state actors in training delivery through competitive procurement of vocational education services

Such funds may work better when they are for a particular industry (Krishnan and Gelb, 2018). In Zambia, the establishment of a training fund for the construction industry has been well supported by employers. This sector-specific fund has avoided the high administration costs of a general fund (UNESCO, 2018). Levy-financed training funds that are managed at least in part by employers are usually considered more successful. In Brazil, national employer federations play a major role and have a strong sense of ownership in the industry, commerce, agriculture and transport sector training funds (Palmer, 2020).

Some countries involve non-state actors in training delivery through competitive procurement of vocational education services. In Latin America and the Caribbean, the Jóvenes youth training programme has been implemented in eight countries since the 1990s. It targets unemployed and socio-economically disadvantaged youth with low education attainment. Firms participate in training design via memoranda of understanding, while private trainers are recruited through competitive bidding under public oversight (Glick et al., 2015). The programme provides comprehensive training, including counselling and soft skills, in line with labour market needs. Competition has increased participation by training providers (OECD, 2020b). Market competition has not always led to desirable results. In Australia, the TVET FEE-HELP loan programme provided loans for TVET students. It was expanded in 2012 through the National Agreement for Skills and Workforce Development, with promotion of subsidized training linked to approved providers. There was little control of provider eligibility, however, and a market of low-value private courses resulted, with some providers using deceitful practices, including fraudulent marketing and inflated costs (Australian National Audit Office, 2016; UNESCO, 2017a). Manu students accumulated debts unwittingly and acquired no marketable skills. After an inquiry by the Senate Education and Employment References Committee, the programme was replaced by TVET Student Loans in 2017. Stricter provider selection and course eligibility criteria were introduced and loan amount capped. Registered providers must publish their tuition online (Australia Department of Employment Skills Small and Family Business, 2017).

99

To train a growing workforce, private actors are mobilized to set up skills development systems. One of the best known recent initiatives is the National Skill Development Corporation in India. It was set up in 2008 as a non-profit and industry-led PPP to attract and catalyse private investment for the establishment of a skills development ecosystem. In a sector considered

99

financially risky and of limited profitability, private training institutions received soft loans to offer training with a low capital investment (Nambiar, 2021). Yet the generous government incentives, which covered three quarters of costs for the first three years, have not led to financial engagement and have raised concerns about quality (Mehrotra, 2018). The corporation's broad, evolving role has hampered its functioning, leading to limited accountability in government fund use and loan disbursements. To overcome these shortcomings, monitoring processes and indicators have been redefined to consider not only trainee numbers but also learning outcomes and student employability. The corporation has also introduced development impact bonds as alternative funding mechanisms linked to measurable objectives (Nambiar, 2021).

# INCENTIVES TO FIRMS DO NOT HELP BRIDGE THE TRAINING GAP

Enterprises tend to underinvest in training. In Europe, the Continuing Vocational Training Survey shows that training accounts for 2% of the total labour cost in firms with 250 or more employees, 1.5% in those with 50 to 249 employees and 1.2% in smaller firms. Net contributions to training funds are a tiny part of a company's total labour cost (CVTS, 2015). Investment in training is also limited compared with that for other assets. In 2018, on average across EU countries, the United Kingdom and the United States, only 9% of business investment was allocated to employee training, less than a fifth of the amount spent on machinery and equipment (EIBIS, 2018).

The free-rider problem partly explains this situation. Firms fear losing the return on their investment if the workers trained move to another employer. Institutional factors may also play a role. Incentives for training are lower when many workers are on temporary contracts. Enterprises responding to a European Investment Bank survey said they underinvested in training because of lack of need, expectations about public institutions, cost barriers and limited access to credit. Firms in countries with strong investment in research and development and a high share of people with tertiary education tend to invest more in training. Even then, the incentive to provide training is consistently lower than that for drawing required skills directly from the labour market (Brunello and Wruuck, 2020; EIBIS, 2018).

Levy-grant programmes directly support employers in providing continuing on-the-job training or arranging external training for their workforce. They involve cost reimbursement or exemption from levies (Ziderman,

#### 66

The incentive to provide training is consistently lower than that for drawing required skills directly from the labour market

2016). In Singapore, the Skills Development Levy is a compulsory monthly payment by companies, proportional to the number of employees – and additional to social contributions – which finances the Skills Development Fund. The levy finances workforce continuing education programmes through grants for firms to pay for employees' external training in the Continuing Education and Training system. Critically, the training must be relevant to the country's economic development and generate certifiable skills (Kuczera and Field, 2018; SkillsFuture Singapore, 2020a).

In Southern African Development Community countries, including Botswana, Malawi, Mauritius, Namibia and South Africa, reimbursement of direct training costs is the most common form of employer training incentive. In South Africa, a programme gives employers a grant accounting for 20% of the levy they paid (Palmer, 2020).

France encourages employers to offer apprenticeships through an apprenticeship development fund and a training levy collected through an apprenticeship tax. The funds are distributed to intermediary bodies and regional entities that give employers tax credits of EUR 1,600 and allowances of at least EUR 1,000 per apprentice. Employers also benefit from social security contribution exemption, and can allocate some of their contributions to the apprentice tax to local training institutions (Kuczera and Field, 2018).

## PAYMENTS TO STUDENTS AND WORKERS ARE ANOTHER WAY TO INCREASE TRAINING

Affordability is a constraint for students and workers who want to invest in training. In East Asia and the Pacific, student fees vary greatly within and across countries, depending on the type of provider. In Fiji, Papua New Guinea and Solomon Islands, tuition is the largest source of funds for private providers. In the Philippines, student fees make up between 0.5% and 5% of public TVET funding, compared with almost 70% in the for-profit TVET system (Palmer, 2017). Governments can support training through direct payments to individuals, with or without learner contributions. Individual learning programmes include individual learning accounts, saving or lifelong learning accounts and entitlements, such as vouchers and grants, covering direct or indirect training costs (**Chapter 12**).

Individual learning accounts allow the right to training to be accumulated over time, with resources mobilized when the individual takes a course. France introduced the Compte Personnel de Formation in 2015. Financed through a training levy on firms, it enables individuals to transfer their training rights across jobs, regardless of employment status. All labour force participants are eligible to benefit from the account, which entitles them to EUR 500 a year and up to EUR 5,000 in all, for certified training. The programme was reformed in 2018 to address persistent inequality in participation and disadvantages linked to employment status. Such accounts allow employees to accumulate resources that employers and sometimes governments complement, when the account holder uses the funds for continuing training (OECD, 2019a).

In the United States, Lifelong Learning Accounts were established in Chicago, New York City and Washington state to help workers take work-related training. Another state, Maine, launched Lifelong Learning Accounts in 2005, coordinated by the state Department of Labor with support from the Maine Centers for Women, Work, and Community, which help workers with education planning. These accounts are linked to NextGen, a higher education saving programme, which allocates training contributions to low- and middle-income workers (Fitzpayne and Pollack, 2018).

Voucher programmes make direct payments to individuals to cover costs such as programme fees, learning materials, and travel or daily allowances, conditional on participation. Singapore's SkillsFuture Credit provides each citizen over age 25 with an opening credit of US\$365 to be spent on skills development (Fitzpayne and Pollack, 2018). Individuals can choose among eligible courses in several areas, provided by diverse actors. In 2019, half a million citizens attended continuing training and benefited from the credit (SkillsFuture Singapore, 2020b; SkillsFuture Singapore and Workforce Singapore, 2020). Training participation of the 15- to 64-year-old resident workforce grew from 35% in 2015 to 48.5% in 2019 (Singapore Ministry of Manpower, 2019).

The degree to which individual incentives are effective and equitable depends on the design. In Liberia, a project aimed at supporting women's transition to work, which provided training, job placements, internships and allowances for transport and childcare, increased 66

In Singapore, half a million citizens attended continuing training and benefited from the SkillsFuture Credit programme in 2019

99

employability (Elder and Kring, 2016). Proximity, flexible schedules and clear links to employability are needed. In Kenya, half the women participating in a vocational training voucher programme mentioned proximity to a training centre as one of the main reasons for enrolling. Pakistan's Skills for Employability programme failed to attract learners, even though they were entitled to a voucher. Daily stipend levels were raised and training was provided in rural centres, but still only one in four of those targeted enrolled (World Bank, 2019b).

#### NON-STATE ACTORS ARE A DRIVING FORCE IN ADULT LEARNING AND EDUCATION

The 2015 UNESCO Recommendation on Adult Learning and Education (ALE) states that ALE 'involves sustained activities and processes of acquiring, recognizing, exchanging, and adapting capabilities' in three domains: literacy and basic skills, continuing training and professional development, and active citizenship, 'through what is variously known as community, popular or liberal education' (UNESCO, 2016b, pp. 6–7). As with TVET, with which it overlaps to a large extent, ALE refers to work- and non-work-related skills that can be acquired through a continuum of formal, non-formal and informal education.

The vast range of opportunities and scarcity of data make it difficult to assess the size of non-state involvement in ALE provision. Results from the 2016 Adult Education Survey in Europe show that fewer than one in five adults participated in non-job-related education, compared with four in five who continued learning for work purposes. Most non-formal education and training was delivered by non-state actors, such as employers and employers' organizations (38%), commercial institutions (10%), non-profit associations (7%) and individuals (5%) (Eurostat, 2021).

In OECD countries, 11% of adults had participated at least once in non-job-related non-formal education, while 38.5% took part at least once in job-related training. Respondents in Hungary and Switzerland were the most active in non-job-related learning, with one in four engaged. Adults in Greece and Lithuania were the least likely to be engaged (OECD, 2021). However, aggregate estimates are less useful to describe specific contexts where non-state actors play a dominant role. Two examples from this vast area are covered here: the role of NGOs, CSOs and communities in adult literacy, especially in non-dominant languages; and the growing presence of for-profit actors in commercialized fields, such as foreign languages.

#### NON-GOVERNMENTAL AND COMMUNITY ORGANIZATIONS DOMINATE ADULT LITERACY PROGRAMMES

Non-state actors, such as NGOs and CSOs, reach out to vulnerable groups of adults traditionally excluded from formal education. Governments often rely on their services to deliver national adult literacy and second chance programmes. NGOs played a key role in enhancing adult education centres in Belarus, the Republic of Moldova and Ukraine (Lukyanova and Veramejchyk, 2017). In the Democratic Republic of the Congo, the government's 2012 literacy and non-formal education strategy was almost entirely implemented through NGOs (UNESCO, 2017b). The Islamic Republic of Iran reported outsourcing most of its adult education provision to NGOs (UIL, 2019a). Morocco contracted almost 1,200 NGOs to provide literacy instruction (UNESCO, 2017b). These organizations engage with communities, mobilizing influential local leaders to identify community needs and encourage participation (**Box 8.2**). Pakistan's National Literacy Programme, for example, is based on close partnership with communities, which are involved in its development and implementation in community-based adult learning centres (Hanemann, 2015).

CSOs have historically challenged government literacy policies, bringing about major change to adult education provision, notably in Latin America. In Brazil, social movements such as the Movimento de Cultura Popular (Popular Culture Movement) and Movimento de Educação de Base (Basic Education Movement) started in the 1960s. They opposed top-down adult education campaigns, arguing for the transformative and creative power of people and communities through popular education. Popular education had clear political connotations, representing emancipation from external imposed pedagogical ideas and the possibility of social transformation (Streck and Zanini Moretti, 2018).

66

Most non-formal education and training in Europe was delivered by non-state actors, such as employers and employers' organizations, commercial institutions, non-profit associations and individuals

"

#### BOX 8.2:

#### Communities are instrumental for the development of adult literacy centres

Globally, communities play a major role in adult education. In Indonesia, community learning centres rely on community initiative but need to obtain formal permission from provincial or local government; some are run by NGOs. In Thailand, community and religious leaders participate in community learning centres' management, following ministerial guidelines and regulations on resources (NILE and UIL, 2016).

In some countries, the state is instrumental in establishing community learning centres and ensuring their financial sustainability. In the Republic of Korea, lifelong learning centres are included in community-based projects and are co-financed by local and national authorities. Mongolian lifelong learning centres may be affiliated with formal education institutions, relying on their facilities and staff (NILE and UIL, 2016). Japanese community learning centres, called kominkan, are mostly funded by local authorities, although it has been possible to open private centres, financed by learner fees, since the 1990s (Stromquist and Lozano, 2018).

Faith-based institutions have played a major role in contexts where they can benefit from strong support by local communities. In Afghanistan, religious literacy has the longest tradition of adult literacy provision, with many mosques engaged in education. The values-oriented courses foster participation for groups that otherwise would be excluded from education opportunities, such as women and villagers. Religious centres are neither monitored nor financed by the state but rely on community support. Religious authorities also play a role in authorizing courses. An NGO or a CSO intending to organize a literacy course in a village may need to obtain agreement and support from the local religious leader, elders and the head of the village (Robinson-Pant et al., 2021).

In some contexts, NGOs and CSOs had followed original approaches to challenge the role of government as the main education service provider but subsequently these approaches were influenced by donors (Zarestky and Ray, 2019). India experienced a rapid proliferation of NGOs, which became essential for delivering education services. However, their increasing reliance on external donors has affected their teaching and learning processes as well as the content, which have become more formalized and skills-oriented in an attempt to meet donor standards (Bhutani Vij, 2020).

### Non-state actors are under-represented in adult education policy formulation

Regulatory frameworks have implications for the quality, scope and nature of non-state activities in adult literacy. They vary significantly by country. In the Philippines, formal registration of NGOs delivering literacy programmes is voluntary, as they are primarily seen as campaigning and advocacy organizations. In Afghanistan, philanthropists and religious centres operate independently and in parallel with other programmes, without government support or monitoring (Robinson-Pant et al., 2021).

In contrast, NGO representatives in Colombia, Ecuador and Peru voice concerns about a tendency to over-regulate, formalize and standardize adult basic education activities. Increasingly, public authorities have a supervisory role and restrict non-state actors' autonomy to adapt learning content to individual learner or group needs and develop contextualized materials (Hanemann, 2021). Standardized teaching and limited flexibility help ensure education quality, as programmes can be better monitored across providers (Fejes et al., 2016). Yet they contradict the ideal of a contextualized, flexible and participatory learning approach (Rogers, 2019).

Stakeholder participation is an element of good governance. The Global Report on Adult Learning and Education found that almost 8 in 10 countries reported engaging with other actors in programme organization or the establishment of adult education councils (UIL, 2019a) (**Figure 8.4**). Flanders (Belgium) has a complex structure, with responsibilities split among the federal level, regions and communities. Provinces, cities and municipalities oversee implementation and multiple

66

Increasingly, public authorities restrict nonstate actors' autonomy to adapt learning content to individual learner or group needs

"

actors operate. They include Catholic centres (publicly financed and privately managed), Go! Education (publicly financed and independently managed) and private adult education institutions. Consultative bodies, such as the Flemish Education Council, representing all actors, act as focal points for the government. Representatives of some beneficiaries, especially more marginalized groups, are often not involved, hampering the inclusiveness and effectiveness of policy formulation (OECD, 2019b).

In El Salvador, the Policy for Continuing Education of Young People and Adults stresses the need to strengthen intersectoral cooperation across civil society and local and education authorities for effective implementation (El Salvador Ministry of Education, 2015). In Uganda, under the leadership of the Ministry of Education and Sports, partners cooperate to improve coordination and quality assurance in adult literacy. At the national level, a Sector Working Group is responsible for planning, reviewing and monitoring activities that further the national strategy of poverty alleviation. At local levels, the structure is replicated by the Community Mobilization and Empowerment Committee (UIL, 2019a).

#### FIGURE 8.4:

#### Almost 8 in 10 countries reported having increased stakeholder participation in adult education and learning policies and programmes



Proportion of countries reporting governance improvement, 2015–19

#### Non-state actors have helped promote non-dominant languages in adult literacy

Non-state actors have played a critical role concerning non-dominant languages and promotion of home languages for instruction in adult education. Thanks to their proximity to communities and their flexible, diverse approach, NGOs and CSOs have been active in promoting local language use in literacy learning.

In Cameroon, the use and promotion of non-dominant languages for literacy activities rely exclusively on initiatives and resources of communities and CSOs. The National Association of Language Committees includes NGO and community representatives engaged in mother tongue adult literacy, although their scope remains limited without adequate political support and integration into development programmes (Robinson and Vü, 2019).

Papua New Guinea has more languages than any other country. Its literacy programmes have always dealt with the issue of mother tongue learning. NGOs and CSOs have engaged in development of local languages and communities through literacy projects. The government acknowledges multilingualism's relevance, but this has not translated into adequate financial commitment. Despite the absence of a formal outsourcing policy, non-state actors have taken over implementation of mother tongue literacy programmes in the country (Robinson, 2019).

Some organizations have advocated for integration of local languages into national literacy programmes. In Chad, the Federation of Associations for the Promotion of the Guéra Languages, a CSO promoting 26 languages and mother tongue literacy projects in the Guéra region, contributed to defining the National Literacy Plan in 2012 and to maintaining a strong focus on each community's identity (UNESCO, 2017b).

Globally, engagement of actors such as NGOs in development of implementation plans has been limited (Aitchison, 2017; Rossel and CEAAL, 2017; UIL, 2020). Even where participation and dialogue with civil society are institutionalized, as in Europe, consultation is limited and held only in the early stages of the policy cycle (Hanemann, 2021). Selected non-formal providers in 16 Central, Eastern and Southern European countries expressed a need for stronger cooperation with civil society in decision making for effective implementation of the European Agenda for Adult Learning (EAEA, 2014).

In West and Central Africa, many countries have adopted the faire-faire decentralization and outsourcing strategy for adult literacy and education. The approach is based on a clearly defined task and role division among government, providers and local communities. While government supervises and distributes resources, provision rests with the other parties (Aitchison, 2017). The approach ensures more flexibility in programming, allowing NGOs and CSOs to tailor education provision to community needs, including local language use (**Box 8.3**).

NGOs and CSOs can move and work relatively easily across sectors, exploring learning approaches that can promote understanding, skills and actions in areas such as health, women's empowerment and the environment. The non-formal approach to adult education provision calls for learning programmes to be flexibly delivered outside formal schooling and across sectors (Rogers, 2019). In Senegal, partnership with NGOs within the faire-faire framework had an impact on curriculum and monitoring. The curriculum of the Centre d'Apprentissage Populaire, an initiative for out-of-school youth run by the NGO Alphadev, covered gender-based violence and health. It was partly based on REFLECT, a participatory approach using Paulo Freire's principles, linking social literacy and empowerment. In 2018, a positive evaluation of the programme led to inclusion of its literacy outcomes in Senegal's national statistics. The approach also influenced the design of the national literacy and training programme. Collaboration between government and NGOs in addressing governance issues contributed to these outcomes (Robinson-Pant et al., 2021).

Opening adult education programme implementation to competition through procurement has influenced the sector's organization. In Sweden, the Adult Education Initiative promoted contracting of adult education services through municipal procurement to increase competition among providers, reduce costs and improve quality and pedagogical innovation. This approach has led to a rapid increase in the presence of non-state providers, mostly for-profit, in education in the last 20 years (Fejes and Holmqvist, 2019). Principals of competing folk high schools were increasingly recruited from outside education among professionals who had similar experience in business or the public sector. Teacher contract stability became less certain and teaching methods more standardized (Fejes et al., 2016). Contracting through procurement requires strong oversight to ensure the quality of new providers (Andersson and Muhrman, 2019). In the Philippines, the Alternative Learning System (ALS) is implemented by the Department of Education through district coordinators; mobile teachers and service providers, including private and public universities; local government agencies; and community groups, within the ALS Unified Contracting Scheme. The department sets minimum requirements for hiring and training facilitators and for monitoring and reporting, but supervision is not equally ensured (World Bank, 2018b). Learning facilitators in programmes delivered directly by the department are trained and vetted public employees, who are more likely to be monitored than their peers in procured programmes. While there is no evidence that learning outcomes differ, contracted facilitators are paid substantially less than those in the state-delivered programme and do not benefit from an institutional incentive structure. They tend to be relatively inexperienced young women who aspire to enter the regular teaching profession (Tenazas et al., 2016).

#### Businesses support adult literacy

Private companies may engage in adult education through community development, often as part of corporate social responsibility initiatives. For instance, companies in extractive industries do this in response to criticism of their environmental and human rights record. A study of the 10 largest Canadian mining companies engaged in adult education programmes suggests that their initial engagement with communities has evolved to a more substantial role in education service provision. Limited regulation, however, combined with the voluntary nature of the interventions hampers effective planning, monitoring and reporting, and overall sustainability. In addition, the focus is more likely to be on professional development than on other community needs (Walker and Sarkodie, 2019).

Information and communication technology (ICT) companies also engage in adult education (Hanemann, 2021). Since 2000, Tata Consultancy Services, a multinational technology service and consulting firm based in Mumbai, India, has delivered computer-based functional literacy programmes using its multimedia software and e-learning system. Through its centres and local instructors – volunteers trained to use the software and equipment – people take courses to learn to read in their home language. The courses are free and government certified. The programme currently operates in 18 states in India and in Burkina Faso (UIL, 2019b). In the United States, Cell-Ed provides literacy and basic skills programmes through mobile phone messages to adult immigrants and to people with limited literacy skills and low income, with financial support from charities and donations. Since it does not require internet connection, it is able to target the most disadvantaged youth and adults (UIL, 2018). Its target during the COVID-19 crisis was to reach 1 million adults, in partnership with the New York and California state governments, the Barbara Bush Foundation and the Dollar General Literacy Foundation (Cell-Ed, 2020).

## THE PRIVATE SECTOR HAS EXPANDED ITS ROLE IN LANGUAGE LEARNING

Language learning has become increasingly relevant as part of continuing education, both for personal development and for economic and social growth. English learning is a growing phenomenon, as English is a common language of communication between people with other primary languages. Among the estimated 1.5 billion English-language learners worldwide, some 750 million learn English as a foreign language, using it occasionally for work or pleasure, while 375 million speak it daily as a second language (Beare, 2019). The number of people assessed is also increasing. In 2018/19, the British Council tested 3.9 million learners, almost 400,000 more than the year before (British Council, 2019). In 2019, learners of French as a foreign language increased by 3.2% from 2018, with 210,000 candidates interested in getting their skills certified, up 3.5% from the previous year (Fondation des Alliances Françaises, 2020).

The growing relevance of language learning and assessment has created a market and attracted for-profit firms. Pearson, a British company engaged in the education sector worldwide, purchased one of Brazil's largest private networks of English language schools in 2013 and bolstered its presence in China by acquiring an English language test preparation provider (Santori et al., 2016). Latin American countries have developed strategies and launched public programmes to expand access to English language learning, but the

66

Contracting adult education through procurement requires strong oversight to ensure the quality of new providers

"

#### 66

# Around 40% of English learners in Argentina and Peru studied with private language institutions

number of students relying on private instruction is also increasing exponentially. A British Council survey found that around 40% of English learners in Argentina and Peru studied with private language institutions. Cost is the greatest barrier but that has not prevented language academies from proliferating across the region. In Quito, Ecuador, private English language institutions increased sixfold in 10 years. Private language schools are responsive, delivering programmes tailored to learners' level and needs. They usually rely on learning materials and curricula from private providers, including Pearson, McGraw Hill and operations affiliated with Cambridge and Oxford universities (Cronquist and Fiszbein, 2017).

The number of people learning languages through mobile applications has surged, creating a profitable market for ICT companies. Duolingo, a free language learning application launched in 2012, counted nearly 300 million downloads in 2019. It offers language learning in 36 languages and 180 countries. In 2016, it started working on an assessment tool, the Duolingo English Test (DET), aiming to compete with the Test of English as a Foreign Language (TOEFL), a proficiency examination taken by foreign students applying to English-language universities. In 2019, more than 180 universities accepted DET as a substitute for TOEFL. Duolingo is an example of a private company that parlayed mobile-assisted learning into a larger position on the market for language learning and assessment (Adams, 2019).

Studies on the effectiveness of mobile-assisted language learning are limited, however, and existing research has been often commissioned by providers. A recent study on a sample of learners who chose to study Turkish for the first time using Duolingo shed light on the advantages and limitations of mobile-assisted language learning. Flexibility of place and time was a clear benefit, but learners' motivation declined over time. In addition, pedagogical shortcomings hampered acquisition of certain skills. Such applications cannot fully replace instructional expertise (Loewen et al., 2019).

#### CONCLUSION

Non-state actors play a major role in TVET and in adult learning and education. Cooperation with government is evolving towards more complex mechanisms, such as PPPs and expanded outsourcing strategies, that enable enhanced expertise and resources in response to new education and learning needs.

In TVET, non-state actors play a critical role in providing continuing non-formal training and expanding existing formal vocational education systems. Beyond provision, cooperation with non-state actors has become crucial in understanding labour markets, anticipating skills demand and improving education provision effectiveness through definition of new standards and integration of non-formal and informal learning into recognized frameworks. Mobilization of non-state actors has also led to diversification of TVET financing mechanisms.

NGOs and CSOs have traditionally been active in ALE provision, whether promoting spontaneous initiatives or cooperating with public authorities. New profitable markets are attracting businesses to areas such as language learning. Despite their dominant role in provision, regulation and governance, however, ALE remains a public prerogative.

99



# Monitoring education in the Sustainable Development Goals

In the winning photo from the 2021/2 GEM photo competition, the Kamara family participate in the daily interactive Reading on the Waves radio programme, a family literacy initiative implemented jointly by CODE, Farm Radio International, the Association of Language and Literacy Educators (Sierra Leone) and the We Care Foundation (Liberia) that has helped to sustain learning during COVID-19 school closures in Sierra Leone and Liberia.

CREDIT: Stephen Douglas

17



### CHAPTER



# Introduction
# **KEY MESSAGES**

In 2015, as part of the Education 2030 Framework for Action, countries committed to establishing intermediate benchmarks for SDG 4 indicators. Setting SDG 4 benchmarks serves multiple objectives:

- To capture each country's intended contribution to the global education agenda, following the approach of the climate change agenda's nationally determined contributions.
- To contextualize progress monitoring, taking into account countries' various starting points.
- To link monitoring of national, regional and global education agendas to promote coherence.
- To focus attention on data gaps for selected key education indicators.
- To highlight the need to strengthen national plans, which may lack specific targets.
- To support dialogue through the global SDG 4 cooperation mechanism and spur collective action.

The UIS and GEM Report teams have mobilized the international community towards setting benchmarks on seven SDG 4 indicators. By October 2021, responding to an invitation by UNESCO, 39% of countries had submitted national benchmark values, 10% had committed to submit them and 15% had committed as part of the EU and CARICOM regional education monitoring frameworks.

Also by October 2021, schools globally had been at least partially closed for 55% of total days due to the COVID-19 pandemic.

#### Official SDG 4 statistics still reflect the pre-pandemic situation:

- Two thirds of national education statistics units had to delay data collection or postpone it to the following school year.
- Delays in household survey fieldwork and issues of data interpretation will affect understanding of progress towards SDG 4.
- Learning assessments have been affected as well. For instance, the 2021 round of PISA was postponed by a year.

Various face-to-face and phone surveys provide initial evidence on potential dropout upon school reopening. In Ethiopia, Ghana and Senegal, dropout rates did not change but repetition increased.

The multiplicity of information sources, coupled with differences in study methodologies, samples, timing and contexts, means understanding the impact of COVID-19 remains challenging.

The UIS, with support from the GEM Report, is reporting the proportion of children who are 'prepared for the future', meaning those who have completed primary or lower secondary school and achieved minimum proficiency in reading and mathematics – in other words, a measure combining the two target 4.1 global indicators. In sub-Saharan Africa, while 29% of primary school students meet learning objectives, high dropout rates mean that only 18% of all primary school-age children do.

A s the midpoint nears for achieving the 2030 Agenda for Sustainable Development, the monitoring framework development has significantly advanced, and countries have begun setting targets. However, the COVID-19 pandemic has presented major setbacks. Not only are the standard tools used to monitor education progress affected, but the targets set may need reconsideration. This introductory chapter presents key issues and provides background on the assessment of global education progress in the 12 chapters that form the monitoring part of this report.

# COUNTRIES HAVE SUBMITTED NATIONAL SDG 4 BENCHMARKS

The UN Secretary-General's 2014 Synthesis Report, which provided the foundation for the 2030 Agenda for Sustainable Development, called for a 'culture of shared responsibility, one based on ... benchmarking for progress' (§146). It also requested alignment of the four 2030 Agenda monitoring levels: global, regional, thematic and national (United Nations, 2014).

The education sector responded in 2015 in the Education 2030 Framework for Action, which called on countries to establish 'appropriate intermediate benchmarks (e.g. for 2020 and 2025)' for SDG indicators, seeing them as 'indispensable for addressing the accountability deficit associated with longer-term targets' (§28) (UNESCO, 2015). The framework implicitly recognized that there were known progress rates but also different national starting points: Countries' progress should be assessed relative to these starting points as well as in terms of whether they can improve on historic progress rates.

The importance of nationally determined contributions, underlined by the climate change agenda, has helped rally countries in recent years. The contributions 'embody efforts by each country to reduce national emissions and adapt to the impacts of climate change' (UNFCCC, 2021a, 2021b). The SDG 4 benchmarks, envisaged in the Framework for Action, bring this approach to education.

Setting SDG 4 benchmarks serves multiple objectives. First, as already noted, benchmarks capture the contribution each country is prepared to make to the global education agenda, given initial conditions. Ideally, national benchmarks should be more ambitious than 'business as usual' and ensure that progress be made relative to past trends. Second, they help contextualize the monitoring of progress so that it is related to what countries intend to achieve. Third, through dialogue with countries and regional organizations, a benchmark-setting process can link national, regional and global education agendas and monitoring frameworks to promote coherence and mutual understanding of different contexts. Fourth, tracking baselines and benchmarks can focus attention on data gaps for selected key education indicators. Fifth, benchmarking can highlight the need to strengthen national plans, which may lack specific targets. Finally, and most importantly, benchmarking is a key tool to support dialogue through the global SDG 4 cooperation mechanism, which is undergoing reform, and to spur collective action.

The UNESCO Institute for Statistics (UIS) and the Global Education Monitoring (GEM) Report teams have worked to mobilize the international community towards SDG 4 benchmarks. In 2019, at the meeting of the Technical Cooperation Group (TCG) on SDG 4 indicators, seven SDG 4 indicators meeting the criteria of policy relevance and data coverage were endorsed for benchmarking (**Table 9.1**).

In October 2020, the Global Education Meeting Declaration urged countries to 'accelerate the progress and propose relevant and realistic benchmarks of key SDG 4 indicators' (§10) (UNESCO, 2020). The UIS and GEM Report teams led extensive consultations in the first half of 2021, in collaboration with UNESCO regional offices in Bangkok, Beirut and Santiago, as well as regional organizations: the African Union, the Caribbean Community (CARICOM), the Central American Education and Cultural Coordination of the Central American Integration System, the European Union (EU), the Pacific Community and the Southeast Asian Ministers of Education Organization. This approach has advanced the agenda the GEM Report had proposed on an enhanced role for regional organizations in SDG 4 (UNESCO, 2017). The UIS has initiated a series of reports showing the alignment of SDG 4 with regional education monitoring frameworks, such as the Continental Education Strategy for Africa 2016-25 (UIS, 2021a); the next reports in the series will cover Asia and the Pacific, the Arab States, Latin America and the Caribbean, and Europe and Northern America.

66 -

Benchmarks capture the contribution each country is prepared to make to the global education agenda

99

#### **TABLE 9.1:**

SDG 4 benchmark indicators

	Indicator	Benchmark values
Early childhood	(4.2.2) Participation rate in organised learning (one year before the official primary entry age), by sex	1
Primary and secondary education	(4.1.1) Proportion of children and young people in: (a) grades 2/3; (b) end of primary; and (c) end of lower secondary achieving at least a minimum proficiency level in: (i) reading and (ii) mathematics, by sex	6
	(4.1.2) Completion rate (primary, lower secondary, upper secondary education)	3
	(4.1.4) Out-of-school rate (primary, lower secondary, upper secondary education)	3
Equity	Gender gap in upper secondary completion	1
Teachers	(4.c.1) Proportion of teachers qualified in basic education by education level	4
Finance	Total public spending on education as share of (a) GDP (b) total public spending	2

Notes: The last column lists the total number of benchmark values if all education levels and subjects are included. The gender gap indicator was endorsed in 2021 but not in time for countries to submit benchmarks on it. Thus, countries were requested to submit 19 benchmark values each for 2025 and 2030. Source: UIS (2021d).

In mid-2021, UNESCO invited countries to submit national benchmark values by 1 October 2021 for six of the seven indicators for 2025 and 2030. Countries were instructed to submit the targets included in their national education sector plans, regardless of the target year. To help with this exercise, the UIS and GEM Report teams sent a template with baseline and recent values by country and indicator. If countries had no targets in their plans on these indicators, they could use two indicative values for discussion: where countries would be if they continued at average progress rates (minimum or 'business as usual') and if they followed the progress rates of the fastest-improving one third of countries (feasible). Finally, national targets from publicly available national education sector plans were also compiled in case countries did not report back.

National benchmark values were submitted by 39% of countries by the deadline. On average, 11 of the requested 19 benchmark values were submitted. In addition, 10% of countries initiated the process and committed to submit their benchmark values. A further 15% of countries did not submit benchmarks but, as EU and CARICOM members, had already committed to targets as part of regional education monitoring frameworks, which have been aligned to SDG 4 for at least some indicators. National plans yielded at least some targets for benchmark indicators for 18% of countries. About 12% of countries had plans without targets and 8% had no plans (**Figure 9.1**).

Information on baseline values and submitted national benchmark values for 2025 and 2030 features in the Global Education Observatory (GEO), a new

#### FIGURE 9.1:

National plans without targets

#### Two in three countries engaged in the SDG 4 benchmark-setting process

Proportion of countries by status of submission of national SDG 4 benchmarks by October 2021

Submitted benchmarks No national plans 8% 12% 39% 39% 18% 14% 10% 10% Tegional plans with targets Regional benchmarks (EU and CARICOM) Committed to submit benchmarks

GEM StatLink: https://bit.ly/GEM2021\_fig9\_1 Sources: UIS and GEM Report teams.

99

0

gateway to education-related data (UIS, 2021c). The GEM Report websites on education progress (SCOPE), education inequality (WIDE) and education profiles (PEER) can be accessed through GEO, which is building on data from a range of sources aiming to improve SDG 4 progress monitoring.

The UIS and GEM Report will release a baseline report analysing the results of this process in early 2022, highlighting the aims of countries, regions and the world vis-à-vis the key SDG 4 indicators in 2025 and 2030. The findings will be a critical issue, as SDG 4 is expected to be at the centre of discussions when the UN Secretary-General convenes the Transforming Education Summit in September 2022. Such reporting will become part of the global cooperation mechanism to inform policy dialogue.

A range of challenges will need to be tackled from 2022 onwards. A process will be outlined to help countries develop missing education targets and try to address misalignment between national and global indicators through dialogue and capacity development. Benchmarks will also be set for the seventh indicator, the gender gap in the upper secondary completion rate. Engagement with the education agendas of regional organizations, which played a critical role in facilitating this process, will be strengthened. Finally, at least some countries may need to account for the potential effects of COVID-19 in national benchmarks as data emerge.

# COVID-19 HAS AFFECTED THE PROSPECTS OF ACHIEVING SDG 4 AND THE MEANS OF MONITORING PROGRESS

UNESCO has been monitoring school closures since the declaration of the pandemic. Aside from the quarter of total days accounted for by closure for academic breaks, schools were closed for 28% of total days and partially closed for 26% between March 2020 and October 2021. The peak of 95% was reached in April 2020; between September 2020 and August 2021, schools were closed or partially closed for half of total days. As of October 2021, schools were entirely closed for 7% of total days and partially closed for 24% (**Figure 9.2**). A major caveat is that many countries classify their schools as partially open even when the vast majority may have been closed.

66

Between September 2020 and August 2021, schools were closed or partially closed for half of total days



The main tool to monitor national policy responses to the COVID-19 crisis has been a joint survey administered by UNESCO, UNICEF and the World Bank. The surveys reached up to 150 countries over three rounds, covering May to June 2020, July to October 2020 (UNESCO et al., 2020) and February to April 2021 (UNESCO et al., 2021); the Organisation for Economic Co-operation and Development joined in the third round. The survey provides an overall picture but also has clear limitations, notably self-reporting and a lack of real-time continuous

notably self-reporting and a lack of real-time continuous observation. In addition, as countries were the units of observation, the surveys did not assess differences within countries: between regions, which is especially relevant in decentralized countries; between schools, including between public and private schools; and, critically, between students, which would highlight the unequal distribution of the consequences. Moreover, the surveys could not capture cases where national policies were not implemented according to plan or where multiple decision makers were involved. Some regional studies have tried to address these challenges; one example is a set of 1 continental, 3 regional and 14 country studies in Asia (UNESCO and UNICEF, 2021).

The official data presented in the monitoring part of this report reflect the UIS February 2021 release, which reflects the pre-pandemic situation, providing a baseline for assessing the scale of the disruption, but not capturing the disruption itself.

Official SDG 4 statistics, which are released with an unavoidable lag, now bear the additional effect of the pandemic on regular data collection. The UIS assessed education ministry planning units between June and September 2020 to determine the impact of lockdown on data collection. Among 129 countries responding, two thirds of national education statistics units said they had had to delay data collection or postpone it to the following school year due to either a moderate or severe effect on their ability to meet reporting requirements (Figure 9.3). Even after postponing their annual school census, many countries struggled to maintain high and timely school response rates (UIS, 2020c). Important conceptual questions arose even among less affected systems, such as who counts as a student when schools are closed, who listens to educational radio programmes, and who is actively participating. Standard SDG 4 indicators, including those on learning environments, do not expect these environments to be at home rather than at school. Administrative data are often not well suited to address many of these questions in an emergency context such as COVID-19.

Survey administration was also severely affected by the pandemic and standard operations were often stopped. Nearly two thirds of countries for which the main data collection mode was face-to-face interviews halted labour force survey data collection, at least temporarily, while among countries with some remote data collection capacity, about one guarter did so (Discenza and Walsh, 2021). In surveys using alternative online or phone techniques, the range of questions asked was narrowed, response rates declined and samples were biased, as disadvantaged populations are harder to reach. Comparability was also affected by the likelihood that respondents answer differently face to face than in phone interviews. Enumerators were not only trained less but also monitored less (Gourlay et al., 2021). Still, more than 60% of countries increased the use of remote data collection in labour force surveys (Discenza and Walsh, 2021).

#### **FIGURE 9.3:**



**The pandemic affected the ability of almost all low- and lower-middle-income countries to report education statistics** Proportion of countries by effect of COVID-19 on ability to provide national education statistics for global reporting, 2020 Some large household survey programmes switched to phone surveys. The World Bank Living Standards Measurement Survey carried out longitudinal phone surveys in Burkina Faso, Ethiopia, Malawi, Mali, Nigeria and Uganda. These included up to five education questions covering pre-pandemic attendance and activities during school closures, such as teacher assignments, use of mobile learning applications and access to radio and television programmes (World Bank, 2021). The UNICEF Multiple Indicator Cluster Surveys (MICS) programme introduced the MICS Plus longitudinal phone survey in Belize, Georgia and Mongolia (UNICEF, 2021). But the Demographic and Health Survey (DHS) programme did not do phone surveys. After suspending field activities for seven months, it resumed data collection in Gabon and Rwanda in September 2020, but did not restart operations in other countries, including Angola, Cambodia, Côte d'Ivoire, Lesotho, Mozambique, Myanmar and the United Republic of Tanzania, until 2021 (ICF, 2020).

More than 25 MICS and DHS surveys had been planned for 2020 or were already under way when the pandemic struck. Thus pandemic-related fieldwork delays and methodology issues will affect understanding of progress towards SDG 4. For example, depending on school closure timing, there may be misunderstanding of questions related to attendance, compared with previous survey rounds. Data analyses will need to clearly identify when the fieldwork was conducted and how schools involved were operating at the time.

Learning assessments have been affected as well. For instance, the 2021 round of the Programme for International Student Assessment was postponed by a year; its results will represent the first major cross-national assessment of what the pandemic's impact was on learning outcomes, which countries were affected and whether the consequences were borne primarily by more disadvantaged students, exacerbating inequality. Nevertheless, research carried out despite the challenges contributes to an emergent understanding of the magnitude of learning loss (**Chapter 10**).

66

Pandemic-related survey fieldwork delays and methodology issues will affect understanding of progress towards SDG 4

. 99

#### A RANGE OF SOURCES PROVIDE INSIGHTS INTO COVID-19'S EFFECTS

While a comparative global picture on the short-term effects of COVID-19 will take more time, a number of research activities gave indications of the impact. Insights from these diverse sources are provided in dedicated sections for each SDG 4 target under Chapters 10 to 19.

Analyses initially relied on pre-pandemic information and simulations. For instance, evidence from student background questionnaires on learning assessments showed students' level of access to devices, use of computers for studying, and home environments in middle- and high-income countries. Some household surveys provided more nuanced information on access to radio, television and the internet, including internet distribution between the poorest and richest, and its speed and reliability. Globally, only one in three children, and one in six of the poorest, had internet access; i.e. the most effective available distance learning modalities excluded most learners, and efforts to expand such modalities would increase inequality in the short to medium term (**Figure 9.4**).

#### FIGURE 9.4:

**Two in three children and youth lack internet access** Percentage of children and young people up to age 25 with internet access at home, by country income group, 2020



Source: UNICEF and ITU (2020).

Studies of the impact on learning of breaks in the academic calendar, whether for recurring reasons such as holiday recess or emergencies such as earthquakes, were used to simulate the potential impact of school closures. Some studies collected teachers' subjective views. In one such large-scale study of 1,300 primary and 900 secondary schools in England (United Kingdom), 18% of students from the most deprived schools were considered by their teachers to be four to six months behind, compared with 5% from the least deprived schools (Sharp et al., 2020).

As the pandemic unfolded, more sophisticated surveys added insights on time use. For instance, internet surveys highlighted decreases in time spent on education and learning during the first COVID-19 wave - by one half in Germany and one third in Italy - and increases in unproductive time spent on-screen, especially if children were low-achieving before the pandemic or did not benefit from parents' presence; inequality in support packages from schools was also recorded (Grewenig et al., 2021; Mangiavacchi et al., 2021). Information from online learning platforms is generally not available due to privacy restrictions. One analysis of user activity log data in Japan showed secondary school students' study time increased during lockdown but by more for students who had prior access to the platform at home (Ikeda and Yamaguchi, 2021).

A phone survey of children of secondary school age in Ecuador found that 23% of those lacking internet access had spent zero time doing schoolwork, compared with 9% of those with access. The poorest were more likely to be at work than in education and girls were more likely than boys to have spent time doing household chores (Asanov et al., 2021). In Accra, Ghana, private school teachers relied more than public school teachers on online classes (42% vs 6%) and material via WhatsApp (62% vs 16%); by contrast, public school teachers relied on radio and television (78% vs 26%) and hard copy materials (78% vs 32%). Private schools were also more likely to organize remedial or after-school classes upon reopening (44% vs 33%) (Aurino et al., 2021).

In the Lao People's Democratic Republic, 45% of all children enrolled in school before the pandemic, but 30% of children from an ethnic minority, were engaged in some education or learning activity during school closure. Of those who were engaged in education and learning, 34% of urban children but 18% of rural children engaged in online learning (World Bank, 2020). In Mongolia, schools closed until the end of the academic year 2019/20 and again in February 2021. A phone survey at the time of the second closure found that only 71% had attended any learning the week before. Of those, 88% attended television lessons and 23% interactive digital lessons (Mongolia National Statistics Office and UNICEF, 2021).

Various face-to-face and phone surveys provided initial evidence on potential dropout upon school reopening. In Ethiopia, a survey split between a period of school closure and a period of partial reopening found that, by the second half of November 2020, almost all children had either returned to school or, if not yet, intended to do so when school reopened (Agness et al., 2021); these findings were similar to those of a September 2020 survey (Akmal et al., 2020). In Ghana and Senegal, dropout rates did not change, remaining low at 2%, but repetition tripled in Ghana, from 3.5% to 10.5%, and doubled in Senegal, from 6.3% to 11.4% (Abreh et al., 2021; Mbaye et al., 2021).

The multiplicity of sources, coupled with differences in study methodologies, samples, timing and contexts, means understanding the impact of COVID-19 remains challenging. Further insights into how different SDG 4 targets have been affected are provided in the following chapters, including a review on the critical issue of learning loss.

# THE SDG 4 MONITORING FRAMEWORK CONTINUES TO DEVELOP

Following the 2020 Comprehensive Review by the Inter-agency and Expert Group (IAEG) on SDG Indicators, the global indicator set will not change until 2025. Two small but important adjustments were made as a result of the 2020 review. First, the metadata of SDG indicator 4.1.1 were updated so that the minimum proficiency level in reading and mathematics is reported not just as a proportion of students but also as a proportion of the cohort of children who complete each level of education (SDG indicator 4.1.2) and learn (UNSD, 2021). For example, the global share of children who achieve the minimum proficiency level in reading is 51% among students but 43% when accounting for children who do not complete primary school. In regions with universal completion, the adjustment has no impact on the indicator, but in sub-Saharan Africa, the value of indicator 4.1.1b drops from 29% to 18% (Figure 9.5). The completion rate definition refers to timely completion (i.e. three to five years after official graduation age) but can also be expressed in terms of ultimate completion to include children, notably in the poorest countries, who complete school even later (Chapter 10).

#### FIGURE 9.5:

One in two children does not achieve minimum proficiency in reading

Percentage of children and adolescents in school and the population who achieve minimum proficiency in reading, by region and completion status, 2019



The second adjustment was of the target age group of SDG indicator 4.2.1, on the percentage of children developmentally on track. The age group was not aligned with the indicator's updated methodology and available data sources; it has been adjusted from children under 5 to children aged 24 to 59 months (**Chapter 11**).

The TCG, which is the IAEG's counterpart on education and is co-chaired by the UIS and the GEM Report, met twice virtually in 2020 and 2021. The organization was improved with five working groups addressing issues related to different data sources: administrative data, learning assessments, household surveys, teacher data and expenditure data (UIS, 2020a, 2020b). Key issues examined included the SDG 4 benchmarking process, the calculation of regional and global averages and the use of population data. It also examined two UIS data collection process innovations to increase coverage. First, the UIS has compiled and harmonized data directly from national reports to fill data gaps; this approach yielded around 6,000 additional data points across 170 countries and the source documents can be accessed in a new repository (UIS, 2021e). Second, it has piloted a template that, if successful, could replace the current UIS survey to simplify and accelerate the data submission process and reduce data gaps.

# **GUIDE TO THE MONITORING PART**

As with each edition of the GEM Report, the next 12 chapters provide an update on progress in education in the SDGs. Chapters 10 to 19 review progress towards the seven targets (4.1 to 4.7) and three means of implementation (4.a to 4.c). Each chapter also focuses on selected monitoring-related aspects and provides an overview of emerging evidence on the effects of COVID-19 on each target. Chapter 20 discusses issues related to education in three other SDGs (energy; industry, innovation and infrastructure; and sustainable production and consumption) and Chapter 21 reviews education financing.



CREDIT: UNICEF/Alessandro Potter

# **KEY MESSAGES**

Out-of-school numbers have barely changed in a decade. Global out-of-school figures are partly based on imputation, but the stagnation cannot be explained away as an imputation problem.

The primary completion rate in sub-Saharan Africa increases from 65% to 76% if those who reached the last grade with more than five years' delay, due to late entry and repetition, are included.

The UIS is pursuing multiple approaches to align results from various learning assessments to facilitate reporting against the minimum proficiency level in reading and mathematics up to grade 9.

A new regional learning assessment, the Southeast Asia Primary Learning Metrics, shows that at least 80% of students in Cambodia, Myanmar and the Philippines do not achieve minimum proficiency in reading.

By October 2021, schools had been open for less than 5% of total instruction days in many Latin American countries, including Brazil, Ecuador and Panama, but also in Bahrain, India and the Islamic Republic of Iran.

Averaged over seven high-income countries, COVID-19 learning losses were equivalent to 30% of a school year for mathematics and 35% for reading if schools were closed for eight weeks.

In rural Karnataka state, India, the percentage of those able to read a grade 2 text fell among grade 4 students from 33% in 2018 to 18% in 2020.

# **CHAPTER 10**



# Primary and secondary education

By 2030, ensure that all girls and boys complete free, equitable and quality primary and secondary education leading to relevant and effective learning outcomes

#### **GLOBAL INDICATOR**

**4.1.1** – Proportion of children and young people (a) in Grade 2 or 3; (b) at the end of primary education; and (c) at the end of lower secondary education achieving at least a minimum proficiency level in (i) reading and (ii) mathematics, by sex

**4.1.2** – Completion rate (primary education, lower secondary education, upper secondary education)

#### THEMATIC INDICATORS

4.1.3 – Gross intake ratio to the last grade (primary education, lower secondary education)

**4.1.4** – Out-of-school rate (primary education, lower secondary education, upper secondary education)

4.1.5 – Percentage of children over-age for grade (primary education, lower secondary education)

4.1.6 - Administration of a nationally representative learning assessment (a) in Grade 2 or 3;
(b) at the end of primary education; and (c) at the end of lower secondary education

4.1.7 – Number of years of (a) free and (b) compulsory primary and secondary

The global target of universal access and completion has been extended since 2015 beyond primary to secondary education and deepened to include good quality that leads to relevant learning outcomes. This chapter first tackles attendance and completion, then discusses learning outcomes, drawing attention to their interaction with attendance and completion.

# ACCESS

Millions of children, adolescents and youth around the world do not have any access to school. Prolonged school closures in response to COVID-19 have further slowed efforts to achieve the goal of universal schooling. For the last school year before the pandemic, 64 million children of primary school age, 63 million adolescents of lower secondary age and 132 million youth of upper secondary age were out of school. The figures have barely budged

in a decade, continuing the stagnation that began after sustained declines during the 2000s (**Figure 10.1**). The various reasons for not attending school include not only lack of provision, but also provision of unappealing quality or low relevance (**Focus 10.1**).

Not all countries regularly report detailed enrolment data that allow for estimation of out-of-school children and out-of-school rates. Countries for which administrative data are missing include some with large populations of out-of-school children. Global out-of-school figures are therefore estimates and partly based on imputation. One consequence of imputation based on the last value actually observed is a bias towards a flat trend: If real-life improvement in a large country remained unreported, global estimates would continue to reflect the higher out-of-school figure from the past, when the country last reported data.



The stagnation in estimated global out-of-school rates cannot, however, be explained easily as a problem of imputation. Even among countries with recent data, some that have large out-of-school populations, such as Indonesia and South Africa, have not shown improving trends. And for all countries with large out-of-school populations whose contribution has been imputed for recent estimates of the global figure, the trend was stagnant even before reporting stopped (**Figure 10.2**).

To date, global and regional estimates of out-of-school rates (indicator 4.1.4) and numbers reported by the UNESCO Institute for Statistics (UIS) have been based exclusively on administrative enrolment data and UN population estimates. A collaborative project between the GEM Report and the UIS is under way to integrate household survey data into these estimates, triangulate sources, fill gaps in the administrative data and develop a coherent time series. However, this may not change the overall stagnation finding. India and Nigeria are among the countries with the largest absolute numbers of out-of-school children for which estimates based on administrative data were not available. In both countries, recent household surveys give little reason to expect large improvements obscured by administrative data gaps. Data released for India in 2020 confirm that the out-of-school rate remains at 5%.

### **FIGURE 10.2:**

Even if household survey data are substituted for missing administrative data, this may only confirm that out-of-school rates have stagnated

Out-of-school children rate, by recent availability of administrative data, 2000–19



Note: The countries are those with more than 1 million primary school-age children out of school.

GEM StatLink: https://bit.ly/GEM2021\_fig10\_2

Sources: UIS database for administrative data; GEM Report team analysis for household survey data: India (2015–16 DHS), Kenya (2019 census) and Nigeria (2016–17 MICS and 2018 DHS).

The GEM Report has consolidated data sources to estimate a consistent time series of the completion rate, which is global indicator 4.1.2 (**Box 10.1**). Primary completion rates are approaching or exceeding 90% in all regions except sub-Saharan Africa, where only two of three children complete primary school. The share has been increasing by roughly one percentage point per year. However, the region is some 20 years behind Central and Southern Asia, where, starting from a similar level, primary school participation and completion expanded relatively rapidly during the 2000s (**Figure 10.3**)

Yet sub-Saharan Africa may already be in a better position than Central and Southern Asia was in 2000 in relation to whether children eventually complete primary school. Indicator 4.1.2 measures reasonably timely completion, within three to five years of the theoretical age for the final primary grade. Children who enter school several years late or repeat several grades, or both, may find themselves completing primary school even beyond that period. Estimates for this report indicate that the primary completion rate in sub-Saharan Africa increases from 65% to 76% if those who reached the last grade with more than five years' delay are included. Taking late completers into account increases completion by four percentage points in Central and Southern Asia and in Latin America and the Caribbean.

# 66

Primary completion rates are
approaching or exceeding 90% in all
regions except sub-Saharan Africa,
where only two of three children
complete primary school

99

Delayed completion is associated with worse schooling outcomes. Over-age participation is therefore of interest in its own right and captured by indicator 4.1.5, the percentage of children who are at least two years over-age for their grade. Discrepancies by a single year are ignored because even children of the same nominal age can differ in actual age by nearly a whole year. Moreover, if the reference dates for entry eligibility, the beginning of the academic year and data collection are not perfectly aligned, children belonging to the same statutory school entry cohort will report two different ages even if they all entered school on time.

#### **FIGURE 10.3:**



# BOX 10.1:

#### A new resource to understand completion rate estimates

In 2015, the United Nations called for a data revolution that would encourage use of multiple sources and enable inclusion of aspects previously neglected in global monitoring, such as equity. In recent years, the GEM Report has focused on ways to make more efficient and effective use of information from multiple survey sources to support SDG 4 monitoring.

In 2020, the Inter-agency and Expert Group on SDG Indicators approved a UIS proposal to adopt the primary, lower secondary and upper secondary completion rate as a second global indicator for target 4.1. The proposal anticipated the possibility of a statistical model being needed to address challenges related to use of household survey data, such as time lags and inconsistent estimates among sources, but also opportunities such as the ability to build coherent long-term time series and to disaggregate by population groups. The GEM Report team has developed a model (Dharamshi et al., 2021) that adapts to education some principles previously used to address the issue of multiple sources for estimation of health indicators, such as child and maternal mortality (Alkema et al., 2016; You et al., 2015). The estimates are used to support the UIS in reporting regional and global aggregates.

A new website, VIEW (www.education-estimates.org), introduces the model and its results to help make the approach more accessible to countries. Maps plot differences among countries while graphs highlight the sources that enter into the calculation of national estimates. A similar resource will showcase the work under way by the GEM Report and UIS to synthesize out-of-school rates from administrative and survey sources.



# 66

Globally, 10% of children and adolescents are over-age by at least two years at both the primary and lower secondary levels

Globally, 10% of children and adolescents are over-age by at least two years at both the primary and lower secondary levels. Over-age enrolment has been declining slowly in all regions but remains high in sub-Saharan Africa, where 23% of children in primary school and 31% of adolescents in lower secondary general education are significantly over-age (**Figure 10.4**). This explains why the region has the largest gap between timely and ultimate completion rates.

In the Lao People's Democratic Republic, the percentage of over-age primary students dropped from 35% to 7% between 2010 and 2020. The country identified over-age enrolment as a crucial challenge, introduced an 'age 6 entry into grade 1' policy in the Education Sector Development Framework and elaborated on it in the 2011 Education Sector Development Plan. The policy emphasized preschool access for 5-year-olds to ensure school readiness (Somsanith and Noonan, 2020). Participation in pre-primary education doubled from 36% in 2010 to 71% in 2020.

# FOCUS 10.1: HOW DO DEMAND FACTORS PREVENT UNIVERSAL SCHOOLING?

In 2020, 260 million children, adolescents and youth were out of school. Examining the reasons and the barriers that keep them from education can help governments design better policies to address the problem. Policies focusing on provision, based on the idea that if schools are built, students will come, have limited success if they do not respond to community demand for schooling. Communities often already prioritize education, even in emergency situations. But schools of low quality and with inappropriate curricula can lead to disillusionment and disengagement.

It is possible to distinguish three types of barriers to schooling: situational (life circumstances), dispositional (personal attitudes) and institutional (structural conditions) (UNESCO, 2020). An analysis for this report of reasons given by secondary school-age adolescents (or their parents) in Malawi, Nigeria and Sierra Leone sheds light on why they were not in school. Among those who have never attended school, half cite dispositional barriers related to an ostensible lack of value or interest in education as their main reason for being out of school. Only in Nigeria do institutional barriers, such as lack of schools nearby, play a role in keeping youth from ever

#### **FIGURE 10.4:**



"

99

going to school. Among those who have attended school at some point but dropped out, dispositional barriers remain important in Malawi but their importance falls by half in Sierra Leone and by two thirds in Nigeria. Situational barriers, notably lack of resources, keep at least 40% of adolescents from going back to school in Nigeria and Sierra Leone. Marriage and pregnancy keep about 10% of adolescents, mostly girls, away from education (**Figure 10.5**).

Different types of barriers require different policy responses. As countries moved towards making education compulsory, as is the case in at least 159 countries (UNESCO, 2019), they increased the number of schools available. After Malawi made education compulsory in 2013, the government rolled out school construction programmes to increase the school supply (Malawi Government, 2013). Countries have also introduced automatic promotion, overturning another institutional barrier that leads to repetition and eventual early school leaving. In Sierra Leone, 5% of those who left school did so because they failed exams and 2% were dismissed.

# 66

Many countries have tried to address the largest situational barrier by reducing the cost of schooling, although only 46 guarantee free education for 12 years

Many countries have tried to address the largest situational barrier by reducing the cost of schooling, although only 46 guarantee free education for 12 years (UNESCO, 2019) and household costs unrelated to fees remain a major burden (Chapter 4). Since 2019, at least five sub-Saharan African countries – Mozambique, Sao Tome and Principe, Sierra Leone, Uganda and Zimbabwe – have either revoked policies that restricted pregnant girls' access to school or implemented policies allowing them to stay in school. In addition to ensuring all schools prioritize admission of young mothers or girls after pregnancy, Uganda's 2020 revised policy on teenage pregnancy included guidance to schools on how to combat discrimination and stigma affecting students who are pregnant or new parents (Human Rights Watch, 2021).

#### **FIGURE 10.5:**

Lack of interest, motivation or valuing of education is a commonly cited reason for never having attended school Share of respondents (out-of-school 12- to 17-year-olds or their parents) by main reason cited for not attending school, 2016–19



GEM StatLink: https://bit.ly/GEM2021\_fig10\_5

Sources: GEM Report team analysis based on data from the 2016-17 Malawi Fourth Integrated Household Survey, the 2018–19 Nigeria Living Standards Survey and the 2018 Sierra Leone Integrated Household Survey.

Addressing dispositional barriers can be more challenging. In some contexts, there may not even be recognition that they need to be addressed. A 'lack of interest' in education can be used as an excuse to shift the blame for low education achievement from policymakers to individuals, especially members of marginalized groups who are stigmatized as lazy. Dispositional barriers may reflect not lack of demand but rather hidden costs or the poor quality of the schooling on offer. In the Ugandan National Panel Survey, the response option 'Education is not useful' could be a criticism of education in general or of the specific education provided in the local context – or both (Uganda Bureau of Statistics, 2019).

Lack of interest may also reflect negative learning experiences. In South Africa, 23% of students who dropped out before age 18 cited poor academic performance as the main reason (Statistics South Africa, 2018). In Nigeria, 7% of children who left school did so because they reportedly 'could not learn' (Nigeria National Bureau of Statistics, 2019).

Lack of opportunity for continuing education may also explain low motivation. Studies have shown that the availability of secondary schools nearby increases the likelihood that students will complete primary education (Mukhopadhyay and Sahoo, 2016). Even if students can continue their education, though, they may not consider doing so, as research has long shown that students' socioeconomic backgrounds heavily influence their aspirations (Gölz and Wohlkinger, 2019; Kao and Tienda, 1998). In rural China, migrant children who do not feel capable of achieving educational success are likely to dismiss education as irrelevant (Chen, 2020).

At least partly due to such challenges, dispositional barriers are less often the focus of policies designed to increase attendance. But their prevalence among responses demonstrates that supplying affordable education of good quality is not enough. It is also important to address socioeconomic barriers keeping families and students from wanting to attend or believing they can do so.

# LEARNING

SDG global indicator 4.1.1 is the percentage of students meeting a minimum proficiency level in reading and mathematics in grade 2 or 3, at the end of primary and at the end of lower secondary education. Three major cross-national learning assessments were conducted in 2019, just before school closures related to COVID-19 moved learning to the home for most children and led parents to play a bigger role in their children's learning. However, parental input into learning has always been substantial, even before school closures (**Box 10.2**).

Across 15 francophone countries in sub-Saharan Africa, the second round of the Programme d'analyse des système éducatifs de la CONFEMEN (PASEC) assessed skills in reading and mathematics in grades 2 and 6. The seventh round of the Trends in International Mathematics and Science Study (TIMSS) assessed grade 4 and 8 students' skills in 64 middle- and high-income countries. The Southeast Asia Primary Learning Metrics (SEA-PLM), a new regional learning assessment, tested grade 5 learners in Cambodia, the Lao People's Democratic Republic, Malaysia, Myanmar, the Philippines and Viet Nam in mathematics, reading, writing and global citizenship skills. The Southeast Asian Ministers of Education Association and UNICEF initiated SEA-PLM in 2012 but it took eight years of planning, curriculum review, assessment framework development, field trials and data collection before data could be released. In other words, the challenges of creating a cross-country assessment from first principles cannot be underestimated. However, considerable progress has been made in the past three years (Box 10.3).

Mathematics results from the three assessments show large differences between countries. Among francophone sub-Saharan African countries, in Burkina Faso and Senegal 1 in 4 grade 6 students reached the minimum proficiency level, but results were much lower in other countries, notably below 5% in Chad, Côte d'Ivoire and the Democratic Republic of the Congo. In South-eastern Asia, about 1 in 10 grade 5 students in the Lao People's Democratic Republic and Myanmar and about 1 in 6 in Cambodia and the Philippines achieved minimum proficiency, compared with 2 in 3 Malaysian and 9 in 10 Vietnamese students.

# BOX 10.2:

#### How much support for schoolwork were learners getting at home before the pandemic?

In recent decades, the time parents spend with their children has been increasing in many countries, at least partly because of a growing understanding of the positive effects of parental time on children's development (Dotti Sani and Treas, 2016; Ortiz-Ospina et al., 2020). One type of interaction, helping children with their education, took on particular significance when the COVID-19 pandemic began.

Some studies have found that parental involvement in children's education is generally associated with positive outcomes, such as higher rates of homework completion and improved academic performance (Patall et al., 2008). However, this is not always the case. The effect depends on factors such as the child's age and ability, the subject matter, and the type of homework and parental involvement, as well as parents' understanding of the material (Gonida and Cortina, 2014; Robinson and Harris, 2014).

Not everyone spends the same amount of time helping their children with their education. First, there are wide discrepancies at the country level, which may reflect cultural and economic backgrounds. In India, 95% of parents spend at least some time helping their children academically and do so on average for 12 hours per week. In Japan, the same is true for only 55% of parents and less than three hours per week (**Figure 10.6**). Across countries, there is a positive association between the share of parents who help their children with their education and the intensity, in terms of hours, with which they do it.

Within countries, there are also significant differences by family socioeconomic background. Richer parents with higher levels of education are more likely to help their children with their education (Varkey Foundation, 2018). This relationship tends to hold even if parents spend more time working outside the home and have higher opportunity costs, suggesting that at least part of the gap may be attributed to differences in parenting values among this group (Bonke and Esping-Andersen, 2011; Brown, 2006; Buchanan et al., 2018; Cha and Song, 2017; Guryan et al., 2008). Another explanation for the gap is that less educated parents may feel unprepared or unable to help their children. Lack of knowledge in the subject may be their main barrier to helping their children academically (Varkey Foundation, 2018).

#### **FIGURE 10.6:**

**The larger the number of parents helping their children with education, the likelier they are to spend more time doing it** Share of parents who help their children with their education and average number of hours per week spent helping, selected countries, 2017/18



Note: The data were collected from an online survey and therefore probably represent only the urban online population in countries with low internet penetration, such as India, Indonesia, Kenya, Peru and Uganda.

GEM StatLink: https://bit.ly/GEM2021\_fig10\_6

# BOX 10.3:

#### UIS has been making progress in aligning results of various learning assessments

Over the past few years, the UIS, through the Global Alliance to Monitor Learning, a working group of the Technical Cooperation Group on SDG 4 indicators, has systematically tried to overcome the challenges countries face in reporting on global indicator 4.1.1. Its efforts build on the development of the global proficiency framework, a core concept defining the minimum proficiency levels learners are expected to reach up to grade 9 in reading and mathematics. These proficiency levels build on the Global Content Frameworks of Reference for reading and mathematics, developed by the UNESCO International Bureau of Education, along with national and cross-national content and assessment frameworks (UIS and USAID, 2020a, 2020b).

Since a single test for all learners in the world is neither feasible nor desirable, the UIS has been pursuing multiple approaches, in increasing order of complexity and robustness, to align results from different assessments, on the principle that countries should be able to use any assessment, provided it meets quality criteria.

The first and simplest step was for all major cross-national assessment programmes to agree how their respective proficiency levels aligned with the global minimum proficiency level. The consensus they reached, which is based on a comparison of the narrative descriptions of each proficiency level, is the basis for most data now reported on global indicator 4.1.1 (UIS, 2018).

This agreement does not take into account national assessments, which are not directly comparable because they differ in curriculum coverage and in assessment frameworks and items. Hence the second approach has been to introduce a method allowing countries to link their assessment to the global proficiency framework. Known as 'policy linking', the approach includes a workshop at which national curriculum and assessment experts and mainly teachers review how the national assessment items align with the global proficiency levels. Each participant subjectively assesses the level of knowledge and skills required for students to answer each item correctly. Participants then rate whether students who reached the minimum proficiency level would be able to answer each aligned item correctly (UIS and USAID, 2020c; USAID, 2021). This standard-setting exercise allows the estimation of the proportion of students above the minimum proficiency level. More than 10 countries have piloted the approach in primary and lower secondary education, including Bangladesh, Djibouti, India, Lesotho, Nepal and Rwanda.

More sophisticated statistical approaches try to link entire assessments. The third approach involved students from two Latin American countries, Colombia and Guatemala, which took part in the Regional Comparative and Explanatory Study (ERCE), and three francophone African countries, Burundi, Guinea and Senegal, which took part in PASEC. The students also took the Progress in International Reading Literacy Study (PIRLS) and TIMSS assessments, administered by the International Association for the Evaluation of Educational Achievement (IEA). The aim is to equate the ERCE and PASEC scores with the TIMSS and PIRLS scores (UIS, 2020). Results are expected in early 2022.

The fourth approach goes a step further. In the context of Monitoring Impacts on Learning Outcomes (MILO), a project aimed at quantifying the impact of COVID-19 on students at the end of primary school in Africa, an assessment was administered in Burkina Faso, Burundi, Côte d'Ivoire, Kenya, Senegal and Zambia. To measure learning loss, the study linked data from national assessments prior to the pandemic with the MILO assessment, with emphasis on assessing the proportion of students meeting the minimum proficiency level. Results are expected in early 2022. As part of the project, a subset of MILO assessment items was selected from the UIS Global Item Bank, which are linked to the global proficiency framework (ACER and UIS, 2021). Country access to items from the Global Item Bank that are already aligned to the standards will increase their flexibility in continuing to run their national assessments, while ensuring they can report on global indicator 4.1.1.

In the global set of countries covered by TIMSS, results in middle-income countries such as Morocco, Pakistan and South Africa were not above those in low-income countries. The proportion of students achieving minimum proficiency in Gulf States ranged from 21% in Kuwait to 53% in the United Arab Emirates. This was below high-income countries but also middle-income countries in the Caucasus and Central Asia, which ranged from 53% in Georgia to 71% in Kazakhstan. Other high-income countries that lagged relative to their peers included France and New Zealand, where at least 4 in 10 students did not reach minimum proficiency. By contrast, children in Japan, the Republic of Korea and Singapore achieved near universal proficiency.

There was a small gap, on average, in favour of boys. Girls were at a particular disadvantage in Burundi and Gabon. Notable gaps were also observed in Bosnia and Herzegovina, Canada and Chile. But boys performed worse in South-eastern Asian countries, especially Cambodia and Malaysia, and in Oman, Saudi Arabia and South Africa (**Figure 10.7**).

66

While 41% of Senegalese and 30% of Cameroonian grade 6 students achieve minimum proficiency in reading, the two countries are level at about 24% when dropout rates are taken into account

Average results were similar in reading, with the small gender gap reversed. Among the sub-Saharan African and South-eastern Asian countries that took part in PASEC and SEA-PLM, girls had worse scores than boys only in Burundi and the Democratic Republic of the Congo. Boys did much worse in Benin, Cameroon and Senegal, and especially in Cambodia and Malaysia (**Figure 10.8a**). Relatively large gaps were observed in other countries, including Madagascar, Niger and the Philippines, but in a context of low overall learning levels. These results refer to children who have reached the upper grades of primary school. Target 4.1 focuses on completion 'leading to relevant and effective learning outcomes', linking the two results. Combining information on completion and learning, on the assumption that children who do not reach the end of primary school have not achieved the minimum learning proficiency level, gives a more realistic picture of learning achievement for all children of primary school age. For instance, while 41% of Senegalese and 30% of

#### **FIGURE 10.7:**

**Universal achievement of minimum proficiency in mathematics is out of reach** Percentage of students at the end of primary school at or above minimum proficiency level in mathematics, by sex, 2019



Source: GEM Report team analysis of the 2019 rounds of PASEC, SEA-PLM and TIMSS.

99

Cameroonian students achieve minimum proficiency in reading, the two countries are level at about 24% when dropout rates are taken into account, as dropout is high in both countries but higher in Senegal. In Côte d'Ivoire and Guinea, 22% of all students but 13% of all children achieve minimum proficiency (Figure 10.8b).

In 2021, the Inter-agency and Expert Group on SDG Indicators agreed that global indicator 4.1.1 could be disaggregated by completion status to draw attention to the learning achievement of all children, not just those fortunate enough to have progressed through the education system. This disaggregation captures the spirit of SDG target 4.1, which calls for all children to complete each level of education and achieve relevant outcomes. The assumption that children who leave

school early have not achieved minimum proficiency is also supported by evidence provided by the UNICEF Multiple Indicator Cluster Surveys (MICS). Their module on foundational learning skills assesses whether children aged 7 to 14 have foundational skills whether they are in school or not. In most low- and lower-middle-income countries, no out-of-school children have such skills. Even in countries such as Lesotho and Zimbabwe, the apparent sizeable minority of out-of-school children with such skills is an overestimate. Foundational skills as defined by MICS are well below the minimum proficiency level of indicator 4.1.1. In addition, many children in the sample have left school after completing primary education, by which time they should have developed foundational skills (Figure 10.9).

#### **FIGURE 10.8:**

The minimum proficiency level indicator ignores children who do not complete primary school

Percentage of students at the end of primary school at or above minimum proficiency level in reading, selected sub-Saharan African and South-eastern Asian countries



b. By completion status, 2014 and 2019

Source: GEM Report team analysis of the 2019 rounds of PASEC and SEA-PLM.

GEM StatLink: https://bit.ly/GEM2021\_fig10\_8b

GEM StatLink: https://bit.ly/GEM2021\_fig10\_8a

In PASEC, almost all participating countries improved grade 2 performance in both subjects but especially in mathematics (**Figure 10.10**). In Congo, Côte d'Ivoire, Niger and Senegal, the average improvement between 2014 and 2019 was sufficiently rapid that if sustained, the target could be achieved by 2030; Benin came close. However, all countries were off track at grade 6. Existing methodologies do not allow results between grades to be linked, even within the same survey, so caution is needed in interpreting these results. In addition, evidence about how individual learning progresses by grade is incomplete (**Focus 10.2**).

The reported progress among all children (**Figure 10.8b**) raises the question whether it took place because more children completed primary school or because those in school learned more. Analysis for this report suggests that in 7 of 10 countries analysed, progress in completion contributed more. Only in Benin, Congo and Niger did learning gains have a bigger role. But the accuracy of such analysis is hampered by the relatively short period, the small magnitude of change and the imprecision of the estimates.

#### **FIGURE 10.9:**

**Children who have never been to school or leave school early do not acquire even rudimentary literacy and numeracy skills** Percentage of children aged 7 to 14 with foundational reading and numeracy skills, by education status, selected low- and lowermiddle-income countries, 2017–19



Note: A child with foundational literacy skills can (a) read 90% of words in a story correctly, (b) answer three literal comprehension questions and (c) answer two inferential comprehension questions. A child with foundational numeracy skills can successfully perform (a) a number reading task, (b) a number discrimination task, (c) an addition task and (d) a pattern recognition and completion task.

GEM StatLink: https://bit.ly/GEM2021\_fig10\_9 Source: MICS Survey Finding reports.

#### **FIGURE 10.10:**

In francophone sub-Saharan African countries, reported progress in learning outcomes among grade 2 and grade 6 students has not been in the same direction

Percentage of students achieving at least minimum proficiency, by grade and subject, 2014 and 2019



GEM StatLink: https://bit.ly/GEM2021\_fig10\_10 Sources: PASEC (2015, 2020). The TIMSS assessment's low international benchmark in mathematics is below the target 4.1.1 minimum proficiency level. The average annual growth between 2015 and 2019 in the share of students achieving the TIMSS low benchmark was 0.3 percentage points at grade 4 and 0.5 percentage points at grade 8. Countries exceeding these averages included Chile, where the share grew from 41% in 2003 to 57% in 2011 and 70% in 2019, i.e. its growth rate was at least three times faster than the average. Elsewhere, as in Jordan and Romania, there was little or no growth. Reaching the last 10% is proving challenging even in well-resourced settings. In the United States, 86% of students achieved the TIMSS low international benchmark in 1995 and 87% in 2019; in New Zealand, the share declined steadily from 89% in 1995 to 82% in 2019 (Figure 10.11).

To improve country coverage of learning outcome data for monitoring of indicator 4.1.1, UNESCO, the World Bank and UNICEF agreed to establish a Learning Data Compact in mid-2021. The initiative aims to streamline financial and capacity development support in an effort to enable all countries to measure learning in at least two subjects, in at least two grades and over at least two rounds in five years (UIS et al., 2021). For 2015-19, 46% of high-income, 17% upper-middle-income, 29% of lower-middle-income and 37% of low-income countries had at least two subjects in at least two grades assessed. Low-income countries tend to prioritize earlier grades (Figure 10.12). High-income countries tend to prioritize assessments in secondary education, even though a considerable number of primary school students fall below minimum proficiency level.

#### **FIGURE 10.11:**

#### Average progress on learning is slow, and often stalls before the goal is reached

Percentage of grade 8 students who achieved the TIMSS low international benchmark in mathematics, selected countries, 1995–2019



Note: The figure includes countries that have participated in more than one grade 8 TIMSS round since 1995. Percentages are estimates; standard errors are not displayed.

GEM StatLink: https://bit.ly/GEM2021\_fig10\_11 Source: IEA and UNESCO (2020). 66

High-income countries tend to prioritize assessments in secondary education, even though a considerable number of primary school students fall below minimum proficiency level

For instance, the percentage of grade 4 students below this level in mathematics in European countries in 2019 ranged from 15% in Latvia and 16% in the Netherlands to 35% in Spain and 43% in France, according to the 2019 TIMSS (Mullis et al., 2020).

Foundational learning outcomes during the primary school years are seen as a linchpin for achieving SDG 4, but the extent to which this justifies an exclusive focus on these skills is hotly debated (CGD, 2021). In practice, no country that has achieved minimum proficiency in reading at the end of primary for at least two-thirds of students has a pre-primary enrolment ratio below 75%, an out-of-school youth rate above 25% or a pupil/ qualified teacher ratio above 25 (**Figure 10.13**). Other targets, such as literacy in the home, can similarly be related to foundational learning (Friedlander, 2020). High-quality pre-primary schooling has a positive effect on schooling outcomes, but just raising pre-primary attendance may not. Higher upper secondary participation may reflect primary schooling of good quality rather than the other way round. Increased staffing ratios do not on their own improve learning. Nevertheless, these relationships illustrate that education systems tend to operate as interconnected systems. There is no precedent for achieving good foundational learning outcomes without relatively strong pre-primary and secondary education or adequate teacher supply. These and other dimensions related to SDG 4 targets other than 4.1 may not be necessary conditions for foundational learning in theory, but have so far proven to be in practice.

99

#### **FIGURE 10.12:**

Poorer countries prioritize earlier grades for learning assessments

Percentage of countries with learning assessment data on mathematics reported by UIS for any year in 2015–19, by income group



Source: UIS database.

#### **FIGURE 10.13:**

No country ensures close to universal foundational learning skills without good indicators in other aspects of the education system

Percentage of students at the end of primary achieving at least a minimum proficiency level in reading relative to other education indicators, 2015–19



GEM StatLink: https://bit.ly/GEM2021\_tig10\_ Source: UIS database.

# FOCUS 10.2: LEARNING DOES NOT PROGRESS IN A LINEAR FASHION

Target 4.1 calls for monitoring learning outcomes at three points in a learner's school trajectory: in the early grades (4.1.1a), at the end of primary school (4.1.1b) and at the end of lower secondary school (4.1.1c). This reflects an understanding that those who lag in the beginning may catch up and that those who are on track halfway through their schooling can fall behind. Measurements at three points over the course of a schooling trajectory are a reasonable compromise in terms of systematic international monitoring. In principle, however, learning progression would be best understood using annual assessment data from each grade. Such data are rarely available, especially in low- and middle-income countries, while longitudinal data for individual students are rarer still (Bau et al., 2021).

An analysis of learning profiles of Peruvian children based on receptive vocabulary tests administered at ages 5, 8, 12 and 15 shows how such skills develop. The shape of the growth curve is non-linear with a decreasing rate of change, i.e. receptive vocabulary develops more rapidly earlier in life and learning gains decrease with age (**Figure 10.14**). Results are consistent with studies showing decreasing growth rates for reading achievement as students move from early to later grades (Bloom, et. al., 2008).

An average flattening of learning in higher grades is in line with models of adolescent development (Buchmann and Steinhoff, 2017). Theoretical models and empirical evidence reveal that students' expectations and motivation during adolescence influence their engagement and achievement in school (Eccles and Wigfield, 2002). Student engagement declines during adolescence and achievement growth tends to be slower as a result of the decline in student motivation

66

The degree of decline in student engagement during adolescence is greater for mathematics than for reading, and particularly affects students from disadvantaged backgrounds

99

(Akos et al., 2015). The degree of decline seems to depend on the subject. It is greater for mathematics than for reading, and particularly affects students from disadvantaged backgrounds. In terms of policy, studies call attention to subject-specific interest and the influence of instructional practices, teacher support and parental support on student engagement in school (Lam et al., 2016). Similarly, they highlight the need to look at student beliefs, including self-efficacy, self-concept and motivation, in addition to student achievement.

While the average pattern is somewhat predictable, individual trajectories display significant variability. There are differences between children in initial achievement levels and the amount and timing of achievement growth, resulting in qualitatively different learning profiles (Helbling et al., 2019). Examples of fast and slow progression exist among those starting at high as well as low levels. One implication of this heterogeneity is that cross-sectional information on which learners have higher achievement at a given point does not necessarily reveal who has gained the most learning until now or who will have the highest achievement in the end.

With some limitations, cross-sectional data on learning outcomes for children of different age cohorts offer an alternative for characterizing learning profiles (Kaffenberger, 2019; Silverstein, 2021). However, interpretation of learning trajectories that are not based on individual longitudinal observations is not straightforward. Consistent improvements over cohorts result in cross-sectional learning profiles that are flatter than individual trajectories, i.e. if successive cohorts reach the same level at earlier and earlier points in the course of their schooling. Conclusions that flat learning profiles imply that students 'do not learn anything' in school should be drawn only with great care.

When assessment data are available only for those who remain in school, interpretation becomes still more challenging. Assumptions of how learning interacts with dropout drive the conclusions. It may seem plausible in many contexts that the lowest performers are more likely to drop out and that the gradient of learning over grades will be generally overestimated in cross-sectional school-based assessment profiles. In addition, if low performers who leave school disproportionately at transition points in the education system, such as the end of primary school, follow the 'slow-fast' trajectory, then keeping them in school would lead to steeper average learning gains in later grades, apart from other potential positive consequences.



Notes: The purple line shows the average score. The orange/green line shows examples of fast-slow/slow-fast individual learning trajectories. GEM StatLink: https://bit.ly/GEM2021\_fig10\_14 Source: Young Lives.

As COVID-19 struck, schools were forced to close all over the world and governments scrambled to respond to the emergency and to prepare for school reopening. A year and a half later, following successive waves of the pandemic, major differences between countries have emerged regarding the percentage of total instruction days when schools were fully or partially closed or open. By the end of October 2021, Bangladesh, Kuwait, the Philippines and the Bolivarian Republic of Venezuela were the countries with the longest duration of full school closures (between 85% and 93% of total instruction days). Wealthier countries, such as Chile, Oman, the Republic of Korea and the United States, but also Ghana, had the longest periods of partial school closures (at least 75% of total instruction days). The countries where schools remained fully open for less than 5% of total instruction days are mainly in Latin America (including Brazil, Ecuador and Panama) but also included Bahrain, India and the Islamic Republic of Iran (Figure 10.15). Schools in Belize, Jamaica and Uganda were still closed as of November 2021. Some countries never closed their schools, including Belarus, Burundi and Tajikistan.

Oceania was the region where schools remained open the longest, an average of 85% of regular instruction days, followed by sub-Saharan Africa (57%) and Europe and Northern America (53%). Latin America and the Caribbean was the region where schools remained open the least (25%); the next lowest share was that of Northern Africa and Western Asia (31%). Overall, instruction days lost did not vary by level of education.

School closures' duration and the ability of governments, teachers and families to support students during the pandemic varied greatly between and within countries, resulting in differing effects on learning. Nearly all countries turned to some form of remote learning, but availability of and access to such programmes were limited in low- and middle-income countries, particularly in rural areas (UNESCO et al., 2021b; UNESCO et al., 2021c) (**Chapter 14**).

In sub-Saharan Africa, phone survey data from six countries showed variation in uptake of various remote learning solutions. Education television and radio programmes were most popular in Burkina Faso, where 40% of households reported their children following them. The same share of households in Ethiopia and Nigeria followed radio alone. No child reported following radio programmes in Mali; instead, 35% of children continued their learning through teacher assignments, even though in all six countries an average of just one contact with the teacher was reported. Tutoring was especially popular in Nigeria (39%). The use of mobile learning apps received much media attention but was the least common remote learning approach, used by no more than 17% of children in Nigeria and 12% in Ethiopia and by barely any in Burkina Faso, Malawi, Mali and Uganda (Dang et al., 2021).

A case study of Sierra Leone found that despite the availability of distance learning modalities, children preferred self-study with familiar resources from school, such as teacher notes and textbooks. As a result of experience with the Ebola outbreak, the country had interactive radio programming that could accommodate local languages. Teachers were retrained and the curriculum was adjusted. But only about one third of students had access to radio lessons, with children from wealthier families (41%) almost three times as likely to tune in as those from poorer families (15%), for whom financial constraints often made radios and batteries unaffordable. In any case, signal and network coverage gaps left one in three children unreachable. The government is procuring more transmitters for universal coverage (Sengeh, 2021; Sierra Leone Ministry of Basic and Senior Secondary Education, 2021).

In Northern Africa and Western Asia, countries favoured a mix of interventions to ensure learning continuity, which includes face-to-face, hybrid and full remote learning modalities based on different contexts and education levels. For instance, Jordan adopted the blended approach and launched the Learning Bridges learning programme that provided both online and offline educational resources to students in grades 4 to 9. Printed learning materials with a QR code were distributed to students on a weekly basis to access audio content and extra learning resources in addition to textbooks (UNESCO et al., 2021a).

In Asia, online and television lessons were the most common instruction methods. Internet access varied from 41% in lower-middle- to 68% in upper-middle- and 86% in high-income countries (ADB, 2021). Countries in Southern Asia were particularly affected. Bangladesh's government made remote learning available through radio, television, mobile phones and the internet. But less than 50% of children aged 5 to 15 had access to a radio, computer or television. About 91% of children in the richest but only 9% in the poorest families had television access (Rahman and Ahmed, 2021). In India, 75% of teachers reported that their biggest challenge was reaching students and 51% maintaining class discipline online (UNESCO and UNICEF, 2021b). In Nepal, 58% of households had internet



Source: GEM Report team analysis of UNESCO database of school closures.



Source: GEM Report team analysis of UNESCO database of school closures.

Eastern and South-eastern Asia and the Pacific faced COVID-19 in the middle of rapid technological transformation. Throughout East Asia and the Pacific, it was estimated that 80 million children were not reached by any distance learning modality. In the Lao People's Democratic Republic, access to the internet via a device at home had increased from less than 2% in 2017 to 43% in 2020. But as only 49% of rural households owned a television, learning was seriously disrupted (UNESCO and UNICEF, 2021d). The Learning from Home programme was the core of Indonesia's national response. Concerns that the curriculum was not sufficiently adapted, given that learners were below its standards even during in-person teaching, led to the release of an emergency curriculum in August 2020 that focused on foundation skills (UNESCO and UNICEF, 2021c).

In China, the number of internet users increased by 86 million in the year to June 2020. Four state-owned mobile operators added base stations to improve network signal coverage and expand bandwidth, and discounted or free internet packages were offered for children from poor families or remote areas. A national programme, Disrupted Classes, Undisrupted Learning, made a range of tools available to students, depending on internet access. The National Network Cloud Platform and National Network of Cloud-Based Classrooms offered e-textbooks for grades 1 to 12 and teacher-developed resources. A public-private partnership arrangement provided free access to 95 companies' products and services. The curriculum was adapted, balancing time, structure and content of distance learning. China Education Television's Channel 4 aired 14 hours of programmes per day, covering a population of almost 300 million in rural areas. An assessment of this massive effort identified several potential areas for improvement, including better integration of the online and television modalities and stronger curation of the resource bank, taking into account local needs, teacher capacity and student learning level. The assessment found that the multiplicity of platforms meant a recommended list, supported by operation manuals and training, would have been more efficient (UNESCO and UNICEF, 2021a).

In Latin America, where schools were closed the longest, countries designed a range of remote learning strategies depending on information and communication technology (ICT) resource availability. Yet large gaps between the poorest and richest households were reported in access to the internet (45% vs 98%) and computers (29% vs 94%). Chile, Colombia and Uruguay opted for online learning solutions, whereas countries with less connectivity offered television and radio lessons (IADB, 2020; British Council, 2021).

# STUDENT LEARNING IS AT STAKE DUE TO PROLONGED SCHOOL CLOSURES

The costs of COVID-19 are many, and the lost lives and economic hardship for societies and households have received much attention. Yet the potential impact of school closures on young people's learning could be one of the pandemic's most costly long-term consequences, potentially spanning generations. Many countries are bracing for a major drop in learning levels, as remote learning modalities are only an imperfect substitute for lost classroom instruction time and are not always available, accessible and affordable in the first place.

Evidence on how much school closures have derailed efforts to achieve SDG 4 relied initially on three groups of studies. The first drew conclusions from related research on out-of-school contexts, such as effects of summer breaks, school closures due to natural disasters and, more generally, student absenteeism. The second simulated consequences of school closures by postulating scenarios for student learning and the effectiveness of remote programmes during the pandemic. The third was based on survey responses from teachers, families and students, capturing perceptions of school closures' effects.

Simulation exercises have pointed to potentially significant learning losses. One based on Early Grade Reading Assessment data for Ethiopia, Kenya, Liberia, Uganda and the United Republic of Tanzania indicated learning losses caused by a full year of school closure could cumulatively amount to almost three years in the long term (Angrist et al., 2021). Another simulation, using PISA for Development data, indicated three months of school closures could translate into a year of learning loss (Kaffenberger, 2021). With Progress in International Reading Literacy Study data, one set of simulations foresaw learning loss exceeding the actual number of schooling days lost by at least 25% in South Africa (Gustafsson and Nuga, 2020). Simulations in Asia and the Pacific projected learning losses ranging from 8% of a learning-adjusted year of schooling in the Pacific, where schools mostly stayed open, to 55% in South Asia, where school closures were the longest (ADB, 2021).

As post-COVID data start trickling in, evaluations of actual changes in student performance observed before and after school closures are emerging. But these studies are not directly comparable. Variation in study design and context restricts what can be learned. Some studies look at short-term effects right after students returned to school, while others examine long-term effects as students advanced in school after reopening. Studies vary by subject and school level and differ in the contextual information collected from teachers, parents and students, especially on how students continued to learn, which is essential to understand the mechanisms underlying school closures' impact.

Most studies using post-COVID data have been conducted in high-income countries (Hammerstein et al., 2021; Zierer, 2021). Overall, results point to typically – but not exclusively – negative and heterogeneous effects that vary by subject, education level and context. Studies from Australia, Belgium, China, Germany, the Netherlands, Switzerland and the United States, at an average school closure duration of eight weeks, point to learning losses equivalent to 30% of a school year for mathematics and 35% for reading (Hammerstein et al., 2021), assuming that a year of learning is equivalent to 0.4 standard deviations.

A study in Japan found that grade 4 to 6 students made up short-term learning losses in mathematics by the end of the school year (Asakawa and Ohtake, 2021). Evidence of positive effects of remote learning was found in Australia, Germany and the Netherlands (Gore et al., 2021; Meeter, 2021; Spitzer and Musslick, 2021).

However, there is clear evidence that effects differ by education level and socioeconomic status. Primary school students seem to have been more affected than secondary school students, likely due to their less developed self-regulation skills (Tomasik et al., 2020). Likewise, school closures were more detrimental for students from disadvantaged socioeconomic backgrounds, who had less ICT access, lower ICT skills and less support from parents (van de Werfhorst, 2021). In France, where schools closed for only about two months in 2020, the annual large-scale national assessment after schools reopened for the 2020/21 academic year showed a small decline in learning proficiency among grade 2 students in reading but no effect in mathematics. Among grade 6 students, results in both subjects actually have increased (France Ministry of National Education, 2021a, 2021b). A large-scale national assessment in Italy found that the share of secondary school students falling below minimum proficiency grew by five percentage points in lower secondary and nine in upper secondary in both reading and mathematics. Losses were higher among the most disadvantaged students and, within that group, especially among those who initially had a higher level of proficiency (INVALSI, 2021).

There is a dearth of direct learning assessments in low- and middle-income countries. The results of the largest study to date, Monitoring Impacts on Learning Outcomes, managed by the UIS, are not expected before early 2022. The study assessed students at the end of primary school in Burkina Faso, Burundi, Côte d'Ivoire, Kenya, Senegal and Zambia, enabling comparison with national assessments prior to the pandemic (Box 10.3).

Individual country studies have shown that learning slowed down. In Ethiopia, a study assessed about 3,000 grade 6 students who had returned to school and been tested before the pandemic at the beginning and end of grade 4. Their learning level had increased but the sampled students had benefited from 45 days of catch-up classes. The results showed that learning growth was below the level that might have been expected and that the effect was worse in rural areas, widening the urban-rural gap (Kim et al., 2021). A study of about 1,000 students in rural Kenya found that 53% of students showed declines in mathematics achievement, with an average loss of 1.1 years of learning, affecting grade 4 students (69%) more than grade 8 students (31%) (Whizz Education, 2021). In South Africa, grade 2 and 4 students lost between 57% and 81% of a year of reading skills in 2020, relative to their pre-pandemic peers; the effect was larger for those with a higher initial level of skill (Ardington et al., 2021).

The long-standing Annual Status of Education Report (ASER) citizen-led assessments in South Asia show that learning levels have declined in the early grades. In rural Karnataka state, India, the share of grade 3 students unable to read rose from 9% to 17% and those unable to recognize a single-digit number from 5% to 11% between

2018 and 2020. The percentage of those able to read a grade 2 text fell among students of all grades but the decline was worst among grade 4 students (from 33% to 18%); a similar but smaller effect was observed in the percentage of students able to carry out division (ASER, 2021) (**Figure 10.16a**). In Pakistan, the ASER household survey in 16 districts found similar learning losses in foundational skills in grades 1 and 3 but not in grade 5. There was some evidence that the impact was stronger for girls. Among grade 3 students, the share of girls who could read a text in Urdu, Sindhi or Pashto fell from 21% to 14% while the share of boys fell from 17% to 16% (ITA, 2021). (**Figure 10.16b**).

In Latin America, analysis of quarterly standardized tests in São Paulo, Brazil, showed that secondary school students learned only 27.5% of what they would have learned in school had there been no pandemic; students whose schools reopened suffered a lower learning loss (Lichand et al., 2021). In Colombia, students performed

a. Students by grade and selected skills in reading and

mathematics, rural Karnataka, India, 2018 and 2020

five points below the previous year, or 0.1 in standard deviation terms, which is about one quarter of a school year. Students with internet and computer access at home obtained, on average, one point more on examinations (Abadía-Alvarado et al., 2021).

This disparate evidence, when combined, confirms that school closures had a negative impact on student learning. If loss is defined in terms of the SDG 4 minimum proficiency level, the impact may be greater in middle-income countries than in low-income countries, where initial levels were very low, and in high-income countries, where schools stayed closed for shorter periods and students had more access to online learning. Still, many aspects remain unknown, including whether learning levels will bounce back or COVID-19 will have a long-term impact on learning. Ultimately, the international community will need to wait for the results of large-scale international assessments, expected to be released from late 2023, for more answers.

#### **FIGURE 10.16:**

#### In South Asia, learning achievement levels have fallen in early grades

- 2018 80 - 2020 70 60 50 % 40 30 20 10 0 1 2 3 4 5 6 7 8 1 2 3 4 5 6 7 Can read a grade 2 text Can do division

b. Children in grades 1, 3 and 5 by selected skills in reading and mathematics, Pakistan, 2019 and 2021



GEM StatLink: https://bit.ly/GEM2021\_fig10\_16 Sources: ASER (2021) for Karnataka, India; ITA (2021) for Pakistan.

# REMEDIATION MEASURES ARE NEEDED TO SUPPORT DISADVANTAGED STUDENTS

When the pandemic struck, many countries were faced with difficult choices. The Plurinational State of Bolivia cancelled the academic year, as weak connectivity in rural and remote areas made virtual classes impossible (Dube, 2020; Eulich, 2020). Kenya decided in July 2020 that the school year would be repeated but subsequently announced that two terms would be covered by the end of May 2021 so that all learners would complete a new school year between June and December 2021 (New York Times, 2020; Oduor and Gikandi, 2020). In 2020, up to 70% of countries had adjusted or planned to adjust instruction time and 24% reported that they would increase class time when schools reopened. Papua New Guinea halved the two-week breaks, while Rwanda shifted the academic year from January-December to September-June (Nugroho et al., 2020).

By 2021, 41% of countries reported having extended the academic year, while 42% of countries reported using learning loss mitigation measures, such as prioritizing certain areas of the curriculum or certain skills (UNESCO et al., 2021c). As part of its recovery programme, Bangladesh introduced an abridged syllabus, which will lead to a curriculum and learning assessment reform by 2023 (Daily Star, 2021). In India, the Central Board of Secondary Education compressed the curriculum by 30% in 2020 and was followed by state education boards such as in Odisha (Kalita, 2020). In Pakistan, the curriculum was condensed to a little over half its original size focusing on three subjects (World Bank, 2021a). As schools reopened, two thirds of countries reported implementing remedial measures in primary and secondary education (UNESCO et al., 2021c). In Italy, disadvantaged students in lower secondary schools were provided free individual tutoring online during the lockdown. Tutors were university students who volunteered for three to six hours per week. An evaluation found that the programme increased academic performance, especially for more disadvantaged students, and had a positive impact on student socioemotional skills and psychological well-being, especially for immigrant students (Carlana and La Ferrara, 2021). In 2020, the Department of Education in England, United Kingdom, introduced a GBP 1 billion programme. Two thirds of the funds were directed at a universal catch-up premium allocated to schools on a per-pupil basis and the rest to the National Tutoring Programme to support up to 6 million 15-hour tutoring courses for the most disadvantaged children. An additional GBP 700 million was allocated in 2021, including for secondary schools to provide summer schools (House of Commons, 2021).

In Cambodia, remedial learning packages first assess five core competencies in Khmer and mathematics. Based on the results, teachers group students and give them exercises at their respective levels of proficiency for 12 hours per month. In Chile, the national remediation programme consists of three phases: catch-up to grade-level learning, new content learning and formative assessment (World Bank, 2021b). In the Philippines, the Department of Education issued guidelines for six-week remedial classes aimed at students who scored below 75% (Philippines Department of Education, 2021).
Princess Lara and her mother attend twice-weekly reading and parenting sessions as part of Save the Children's First Read early childhood education programme in the Philippines.

CREDIT: Save the Children

## **KEY MESSAGES**

Globally, 75% of children were enrolled in pre-primary education one year before the official primary entry age in the school year ending in 2019, but in sub-Saharan Africa and in Northern Africa and Western Asia the rate was about 50%.

The average attendance rate for children aged 36 to 59 months in 61 low- and middle-income countries was 37%, with a 16 percentage point gap between urban and rural areas and a 34 percentage point gap between the richest and poorest quintiles.

Home learning environments are often poor: In about 70 low- and middle-income countries, only 23% of children under 5 had at least three books at home and only 62% were engaged in four activities or more by an adult in the household.

In Brazil, 1.8 million children under 3 are left out of day care for lack of places; only 32% of children under 3 are enrolled. The poorest children are most affected: 34% lack places, compared with 7% of the richest.

COVID-19 challenged early childhood education in terms of adapting remote learning for young children, monitoring and assessing child development and dealing with disadvantaged home environments with insufficient support.

Only 55% of countries provided pre-primary school teachers with instructions to ensure learning continuity during the pandemic, compared with nearly 70% of countries for other levels of education.

#### **CHAPTER 11**



# TARGET 4.2 Early childhood

By 2030, ensure that all girls and boys have access to quality early childhood development, care and pre-primary education so that they are ready for primary education

#### **GLOBAL INDICATOR**

**4.2.1** – Proportion of children aged 24–59 months who are developmentally on track in health, learning and psychosocial well-being, by sex

**4.2.2** – Participation rate in organized learning (one year before the official primary entry age), by sex

#### THEMATIC INDICATORS

**4.2.3** – Percentage of children under 5 years of age experiencing positive and stimulating home learning environments

**4.2.4** – Gross early childhood education enrolment ratio in (a) pre-primary education and (b) early childhood educational development

**4.2.5** – Number of years of (i) free and (ii) compulsory pre-primary education guaranteed in legal frameworks

Early childhood care and education (ECCE) is the foundation of all other education development outcomes. Global indicator 4.2.1 aims to capture child preparedness for school, a concept reflected in the Early Childhood Development Index (ECDI) for children aged 36 to 59 months and measured through 10 questions in the UNICEF Multiple Indicator Cluster Surveys (MICS). The indicator is defined as the percentage of children developmentally on track in at least three of four domains: literacy-numeracy, physical, socioemotional and learning development. Given the weaknesses of this measure, UNICEF, as custodian agency, developed ECDI 2030, a more robust tool for children aged 24 to 59 months, which was approved by the United Nations Statistical Commission in March 2020. It consists of 20 questions in three domains,

learning, health and psychosocial wellbeing, and will also be assessed through future rounds of MICS.

However, collecting data and establishing trends with the new tool will not be possible until the end of the decade. In the meantime, the old ECDI is the only source allowing analyses of trends in early childhood development in the 2010s for a large number of countries. In most countries, the gap in the percentage of children developmentally on track between the poorest and richest families either stagnated or even increased. in Bangladesh, Chad, Kyrgyzstan and the Lao People's Democratic Republic, the wealth gap declined, although in Kyrgyzstan this was the result of a deteriorating trend for both groups (**Figure 11.1**).



#### **FIGURE 11.2:**

**Countries with many children developmentally off track tend to also have lower pre-primary participation rates** Participation rate in organized learning one year before the official primary entry age and proportion of children aged 36 to 59 months who are developmentally on track, selected countries, 2016–19



Overall, just 40% of children in Burundi and just over 50% of children in Benin, Guinea, Sierra Leone, Timor-Leste and Togo are developmentally on track. Countries with the largest number of children who are not developmentally on track also tend to have low levels of pre-primary education that could help prepare them to be ready for school (**Figure 11.2**).

Globally, 75% of children are enrolled one year before the official primary entry age, which is global indicator 4.2.2, but only about one in two in sub-Saharan Africa and in Northern Africa and Western Asia. By contrast, 95% of children were enrolled in pre-primary education one year before the start of primary school in Latin America and the Caribbean (**Figure 11.3**).

Meanwhile, 61% of pre-primary school-age children are enrolled in pre-primary school, which is thematic indicator 4.2.4. The gap between the two indicators can be explained by the fact that the respective age groups differ. The reference age group of the pre-primary gross enrolment ratio is larger, varies by country and ranges from one to four years before the start of primary school. Enrolment rates tend to be lower among younger children. The gap between the two indicators is zero in Eastern and South-eastern Asia but 24 percentage points in Central and Southern Asia. The gap between the enrolment one year before entering primary school and the attendance of children aged 36 to 59 months in pre-primary school exceeds 60 percentage points in Bangladesh, Cambodia, El Salvador, Paraguay and Rwanda.

Just one in five children of pre-primary school age is enrolled in pre-primary education in low-income countries, but even in high-income countries, one in six children is not enrolled (**Figure 11.3**). The European Union has multiplied efforts to increase enrolment, culminating in the approval of the EU Strategy on the Rights of the Child and the European Child Guarantee (**Box 11.1**). One in four children is not in school the year before they are expected to start primary education Participation rate in organized learning one year before the

official primary entry age and gross pre-primary education enrolment ratio, by region and income group, 2020



GEM StatLink: https://bit.ly/GEM2021\_fig11\_3 Source: UIS database.

#### 66

More than 90% of children had benefited from three or more years of early childhood education in the Republic of Korea and Sweden but only 12% in Saudi Arabia and 8% in Turkey

99

Relatively low ECCE participation partly reflects that only a minority of countries guarantee even one year of free and compulsory pre-primary education, which indicator 4.2.5 aims to capture. A recent review of legal frameworks in 193 countries found that 63 countries adopted legal provisions for free pre-primary education and only 51 countries adopted compulsory pre-primary education (UNESCO, 2021b).

While each country should ensure all children complete at least one year of pre-primary education, many middle- and high-income countries already offer more opportunities for early childhood education (**Focus 11.1**). In France, Hungary, Israel and Mexico, entitlement to preschool begins at age 3 (OECD, 2020a). In Latin America, Argentina, the Plurinational State of Bolivia, Brazil, Colombia, Cuba, the Dominican Republic and El Salvador consider that children have the right to education from birth. Others, such as Costa Rica and Honduras, grant the right once the child is 3 or 4 years old. Analysis of a retrospective question addressed to grade 4 students, as part of the

#### BOX 11.1:

#### European Union countries have committed to universal early childhood education

In Europe, the emphasis on universal and inclusive early childhood education and care has its roots in an economic strategy aiming to promote female labour market participation. In its 2011 Communication, the European Commission urged universal access to high-quality inclusive services and made special references to children of migrant families ('early language assistance ... is an important part of improving their school-readiness'), children with special education needs ('paving the way for their later integration in mainstream schools') and the Roma (European Commission, 2011). The percentage of children enrolled from age 4 to the starting age of primary school increased from 93% in 2010 to 95% in 2019, while in Croatia, it increased from 70% to 82% and in Poland from 76% to 95% (Eurostat, 2021).

In the past three years, several declarations focus on ensuring the last 5% of young children also benefit from education. The 2018 European Council Recommendation on Common Values, Inclusive Education and the European Dimension of Teaching called for inclusive and high-quality education and training 'at all levels and from an early age' to ensure 'social inclusion by providing every child with a fair chance and equal opportunities to succeed' (European Council, 2018). The 2019 European Council Recommendation on High-Quality Early Childhood Education and Care Systems reflects the quality framework proposed by an expert working group in 2014 (European Commission, 2014; European Council, 2019). The European Commission's Action Plan on Integration and Inclusion 2021–2027 includes in its aims to encourage the participation of children with a migrant background in early childhood education services, which are bolstered to support cultural and linguistic diversity (European Commission, 2020). Finally, a new comprehensive EU Strategy on the Rights of the Child and the European Child Guarantee also cover early childhood education. Principle 11 of the European Child Guarantee calls upon countries to guarantee free and effective access to inclusive, non-segregated early childhood education of good quality (European Commission, 2021a).

Trends in International Mathematics and Science Study (TIMSS), found that more than 90% of these children had benefited from three or more years of early childhood education in the Republic of Korea and Sweden but only 12% in Saudi Arabia and 8% in Turkey (**Figure 11.4**).

The poorest children and those who live in rural areas are overrepresented among the millions of children deprived of early childhood education. Among 61 countries that took part in either a MICS or Demographic and Health Survey (DHS) in 2012–19, the average percentage of children aged 36 to 59 months attending early childhood education was 37%. Children living in urban areas were more likely to attend (47%) than those in rural areas (31%). Urban-rural disparity was particularly large in countries where overall early childhood education attendance was still very low, including Burundi, Côte d'Ivoire, Guinea, Guinea-Bissau, Iraq and Sierra Leone, where children living in urban areas were at least six times more likely to attend than their rural counterparts. Disparities in attendance based on wealth were even greater. The average share of the richest children attending early childhood education (58%) was more than twice that of the poorest children (24%). particularly large disparities were observed in Cameroon, Congo, Lesotho, Nigeria,

Serbia and Turkmenistan, with their level of attendance at least 60 percentage points lower than that of the richest children (**Figure 11.5**).

Learning starts at home, as children manipulate objects and materials, develop language and explore the world around them. During crucial formative years, children develop the cognitive and wider skills that will prepare them for school. Children with disadvantaged backgrounds have lower levels of communication, language and literacy skills.

Thematic indicator 4.2.3 examines whether children experience a positive and stimulating home environment, in the form of adult engagement in a range of activities: reading or looking at picture books; telling stories; singing songs; taking children outside the home; playing; and naming, counting and/or drawing. Parenting practices, such as reading to children, using complex language, responsiveness and warmth in interactions, are all associated with better developmental outcomes (Winter, 2010). Having someone read books to them is important for children and helps them develop and improve their reading skills (Fletcher and Reese, 2005). This is particularly so for children living in the poorest households. In the

#### FIGURE 11.4

**Children in many high-income countries have at least three years of preschool** Grade 4 students with at least three years of early childhood education, 2019



#### **FIGURE 11.5**

#### Poor children and children in rural areas are less likely to attend early childhood education

Percentage of children aged 36 to 59 months currently attending early childhood education, selected countries, 2012–19

a. By location



Sources: UNICEF database, DHS and MICS country reports.



#### **FIGURE 11.6**

#### In some countries, many children under 3 participate in ECCE

Percentage of children under 3 enrolled in early childhood education, selected middle- and high-income countries, 2018



#### 66

In Punjab, Pakistan, 29% of the poorest children to whom someone was reading books achieve foundational reading skills, compared with only 15% of children with no such support

province of Punjab, Pakistan, 29% of the poorest children to whom someone was reading books achieve foundational reading skills, compared with only 15% of children with no such support. The differences were less marked for children in the richest households (Brossard et al., 2020).

In 2012–19, on average 62% of children were engaged in four activities or more by an adult in the household among the 71 countries with MICS or DHS data. The percentage was lower than 20% in the Gambia, Sierra Leone and Togo. Children in the richest households (73%) were more likely to be engaged in learning and stimulation activities by adults than those in the poorest households (49%). Wealth disparities in early stimulation were significant in countries including Guinea, Haiti, Mauritania and Paraguay, where the percentage of children with whom an adult had engaged in four or more activities was twice as high in the richest as in the poorest households.

An important constraint on stimulating activities, such as joint reading, is the availability of books. Across 70 low- and middle-income countries, 23% of children had at least three books at home. In half of these countries, the share was fewer than 1 in 10 children and in 8 countries in sub-Saharan Africa, fewer than 1 in 100.

"

#### FOCUS 11.1: EARLY CHILDHOOD CARE AND EDUCATION STARTS AT BIRTH

While Article 28 of the UN Convention on the Rights of the Child (UNCRC) does not explicitly extend the right to education beyond primary, secondary and higher education, Article 6 states that children have a right to develop to 'the maximum extent possible' and Article 29 provides that education should be directed to the development of a child's abilities 'to their fullest potential'. This has been interpreted to support the right to education as 'beginning at birth' in the UNCRC General Comment No. 7 (United Nations, 2006). Decades of research in neuroscience has generated strong evidence on early learning and factors enabling child development. By the time a child reaches age 3, 90% of their brain is developed (Winter, 2020). Participation of children under 3 in ECCE programmes tends to be limited, though it reaches over 20% for ages 0 to 1 and over 60% for age 2 in several middle- and high-income countries (**Figure 11.6**). In Cuba, where ECCE programmes have been a priority for over 50 years, more than 80% of 0- to 1-year-olds and 100% of 2-year-olds are enrolled. Strong intersectoral interventions focusing on education, health and child protection are available from pregnancy, and strong community and family involvement, particularly through the Educate Your Child programme, increase the reach of ECCE in rural or remote areas (Laire, 2016).

Other countries have tried to increase access through regulation. Denmark, Estonia, Finland, Germany, Latvia, Norway and Slovenia guarantee places in ECCE in their regulatory frameworks for all children after the end of parental leave (OECD, 2020a). Ghana and Mozambique have included the right to ECCE access from birth in policy objectives, to be accomplished gradually (Ghana Ministry of Women and Children's Affairs, 2004; Mozambique Ministry of Education and Human Development, 2020).

Affordability of ECCE for the youngest is a challenge; private sources account for a considerable share of spending (**Chapter 6**). In Colombia and Israel, private sources account for over 80% of spending in early childhood educational development programmes, compared with less than 30% for pre-primary (OECD, 2019). Especially for children under 3, demand exceeds public supply in most countries, and public subsidies may not be granted from birth. In Slovenia, preschool is free of charge for all children aged 3 to 6. Luxembourg offers free preschool from age 3, and since October 2017, a voucher system finances 20 hours of free childcare in the non-formal education sector to children aged 1 to 4 (Luxembourg National Youth Services, 2020).

In Brazil, 1.8 million children under 3 are left out of day care for lack of places; only 32% of children under 3 are enrolled. The poorest children are most affected, with 34% lacking ECCE places, compared with 7% of the richest (IBGE, 2018). In Rio de Janeiro, where a lottery system was introduced to increase access in low-income neighbourhoods, access remains a major challenge. Initiatives are being taken to address this by increasing the use of community mothers and informal childcare (Attanasio, 2017; Walker, 2019).

66 -

In Brazil, 34% of the poorest children lack ECCE places, compared with 7% of the richest

"

Even in high-income countries, access to ECCE still depends very much on socioeconomic background. In France and Ireland, the difference in participation between 0- to 2-year-olds in low- and high-income households is over 50 percentage points. In Denmark and Sweden, by contrast, only 2% of low-income households report wanting more access to childcare but being unable to afford it (OECD, 2020b). Local policies can help. Aarhus, Denmark, ran the project A Safe Start in Life in 2014–17 to develop cooperation between health visitors and nursery educators to support parents with newborn children, enhancing their trust and knowledge on activities that can enhance their children's well-being (European Commission, 2021b).

Types of participation vary. In many countries, programmes for the youngest are available only part time, even just a few hours per week. In the Netherlands and the United Kingdom, children under 3 participate in ECCE services for less than 20 hours a week, on average (OECD, 2017). Although evidence on the benefits of full-time vs part-time for the youngest children is not conclusive, the number of hours per week that children under 3 spend in ECCE programmes can play an important role in parents' labour market participation. In high-income countries, this number is closely correlated with women's full-time employment rates (OECD, 2018).

Official participation figures, however, tell only part of the story. An ECCE programme must meet criteria regarding intensity, intentional education objectives and staff to be considered in the International Standard Classification of Education (ISCED). ISCED level 0 has two subcategories: ISCED 01 covers early childhood education development programmes, generally for children up to age 2, while ISCED 02 is pre-primary education, generally from age 3 to the start of primary school. Especially for the youngest children, participation may be high in registered programmes that offer formal services outside the scope of ISCED, such as crèches in France or amas in Portugal. In Japan, 2% of children under 3 are enrolled in ISCED 01 while 32% are enrolled in registered programmes outside ISCED's scope (OECD, 2021).

ECCE for the youngest often lacks formal recognition and may not be fully captured in data. The issue of unregistered non-state providers, particularly of services for children under 3, has been recognized in many countries, from pre-primary education in Nepal (World Bank and UNICEF, 2020) to unlicensed home childcare in Canada (Varmuza et al., 2019). In India, the large number of unregistered non-state ECCE centres poses a challenge regarding accountability and administrative data collection (Rao et al., 2021).

#### COVID-19

At the height of preschool and kindergarten closures in early April 2020, more than 180 million children had their pre-primary schooling disrupted (Nugroho et al., 2021). On average, preschools were closed for 78 days in 2020, ranging from 46 days in high-income countries to 122 days in lower-middle-income countries (UNESCO et al., 2021). One study estimated that closures between March 2020 and February 2021, equivalent to 19 billion person-days of instruction lost, would result in 11 million more children being developmentally off track, with losses concentrated in low- and lower-middle-income countries, exacerbating existing global inequity (McCoy et al., 2021).

By mid-2021, over 60 countries had not fully reopened pre-primary schools. Although they lost more instruction days in 2020 than primary and secondary school students, the youngest learners in low- and middle-income countries were less likely to have access to remote learning opportunities (Nugroho et al., 2021). Globally, 46% of countries reported that at least 75% of pre-primary students followed remote education when schools were closed in 2020, well below the 64% of countries reporting the same for primary education students (UNESCO et al., 2021). A survey of 26 low- and middle-income countries found that 19 had national online learning platforms and 14 had learning content for young children but just 6 had guidance or training for teachers, and only Pakistan monitored participation (Galevski et al., 2021). It was reported that, in well-resourced settings, early childhood educators adapted to distance teaching and gained in confidence even with limited or no guidance (McKenna et al., 2021). But in most countries, there were too many obstacles to ensure learning continuity.

Even where remote learning was available, challenges included a lack of teacher training, difficulties in adapting remote learning for young children, and disadvantaged home environments with insufficient support. Pre-primary teachers received instructions to ensure continuity of learning in 55% of countries, compared with nearly 70% of countries for other levels of education (UNESCO et al., 2021). Lack of preparation for distance teaching was a major challenge, especially how to use technology for developmentally appropriate activities, but also how to work with caregivers who may be unable to provide feedback on how well their children are responding and learning (Atiles et al., 2021). In remote mode, teachers only receive activity results, such as images of drawings, videos and activity sheets, but cannot monitor and assess child development in person (Oktavianingsih and Arifiyanti, 2021). In an online survey of early childhood educators,

58% in Asia and the Pacific could not monitor learning progress, making this their biggest challenge after lack of internet and devices; in sub-Saharan Africa, 46% of educators reported this challenge (UNESCO, 2021a).

Where instruction was moved online, educators noted that young children lacked the self-regulation and skills to look at a screen for long periods and that it was challenging to explain activities. A survey of parents in China showed that, while they used free online learning resources guided by preschool teachers, 85% said their children spent no more than 30 minutes at a time, with about one third spending less than 15 minutes online per day (Dong et al., 2020). Remote learning classroom size played a significant role: A study in the United States found that in classes of more than 10 pupils, children had fewer turns, were less actively engaged and lost interest sooner. Toddlers could remain engaged for sessions of 15 to 20 minutes and preschool-aged children for up to 30 minutes. Children responded well in both age groups to songs, engaging stories and music/movement (Szente, 2020).

Policies and measures to support parents and caregivers were put in place in about three quarters of countries: 62% of countries reported providing materials to guide parents in home-based learning, but only 44% of low-income countries provided such materials to at least some parents, compared with 71% of high-income countries. Among the most frequently used measures were provision of guidance, tips or materials for continued learning at home. Parental guidelines to support learning at home were reinforced through regular phone follow-up conducted by schools in 45% of high-income but just 22% of low-income countries (UNESCO et al, 2020).

Beyond the long-term impact on learning, the closure of early education facilities and limited interactions with extended families deprived children of social and cognitive stimulation beyond their homes (Yoshikawa et al., 2020), a lack that has been likened to a 'shadow pandemic' (Howard-Jones et al., 2021). Children thrive in the company of other children and responsive adults, and remote learning cannot replace those experiences (Pascal and Bertram, 2021). Many children faced social isolation and increased stress levels. A study in the US state of Massachusetts found that 61% of parents noticed a negative impact on children's socioemotional development and 53% of early childhood educators noticed behavioural changes; of those educators, 77% reported negative changes, such as increased temper tantrums, crying and difficulty separating from parents, and 23% reported positive changes related to adaptability and resilience (Hanno et al., 2021).

Increased stress is particularly critical for children with disabilities and mental health issues and those exposed to protection risks. In the United States, while almost half of educators surveyed said they used strategies to engage differently with children at risk or with special needs, only 6% collaborated with service providers such as speech therapists (McKenna et al., 2021). Reaching vulnerable students remotely would require a multi-stakeholder approach and collaborations between teachers and specialized institutions. In Cameroon and North Macedonia, radio and television programmes and printed learning materials were not adapted for children with visual and hearing impairments (Galevski et al., 2021). For some children, losing access to teachers meant a risk to their safety. Analysis of data on domestic violence victims reported to the Mexico City attorney general's office showed a decline in the incidence of child maltreatment reports by 21% to 30%, with larger reductions in poor municipalities, which could be attributed to the role educators play in early detection and reporting (Cabrera-Hernandez and Padilla-Romo 2020).

Final-year nurse trainees take an examination at the College of Medicine and Allied Health Sciences in Freetown, Sierra Leone.

CREDIT: UNICEF/Olivie Asseli

## **KEY MESSAGES**

Between 2012 and 2015, almost 1 in 5 respondents aged 15 to 35 in 33 low- and middle-income countries said they had had at least one internship or apprenticeship with an employer as part of their education.

In a quarter of countries, vocational secondary schooling does not give access to tertiary education, reducing its attractiveness.

The tertiary education gross enrolment ratio reached 39%, having grown around one percentage point per year since 2000, and ranging from 9% in sub-Saharan Africa to 78% in Europe and Northern America.

About 40% of students in 19 mostly high-income countries take out tertiary education loans. In Brazil and Chile, over 40% of student loans had at least three months of payment delays.

In 35 European countries, 90% of adult education participation was employer-sponsored and 40% was employer-provided.

Australian, British, Korean, Russian and Swiss adults with tertiary education were twice as likely as those with secondary education to have 10 or more training occurrences in their lifetime.

Technical and vocational education and training suffered during the COVID-19 pandemic because up to 80% of programmes focus on practical skills, which should be acquired in person.

In the United States, undergraduate enrolment fell by 6.5% between 2019 and 2021 and by at least 13% in four-year private for-profit and two-year public institutions.

#### **CHAPTER 12**



### **TARGET 4.3**

# Technical, vocational, tertiary and adult education

By 2030, ensure equal access for all women and men to affordable and quality technical, vocational and tertiary education, including university

#### **GLOBAL INDICATOR**

**4.3.1** – Participation rate of youth and adults in formal and non-formal education and training in the previous 12 months, by sex

#### THEMATIC INDICATORS

- **4.3.2** Gross enrolment ratio for tertiary education by sex
- 4.3.3 Participation rate in technical-vocational programmes (15- to 24-year-olds) by sex

Target 4.3 and its indicators concern education opportunities outside of and beyond general academic primary and secondary school, namely technical and vocational education and training (TVET), tertiary education and adult education. In many countries, most learners have few such opportunities. Large differences between countries exist, likely reflecting differing constraints more than preferences.

## TECHNICAL AND VOCATIONAL EDUCATION AND TRAINING

TVET fosters economic growth, including by enhancing key dimensions of social inclusion (Ali Asadullah, 2019). Yet it remains underfunded and often neglected. In low-income countries, only 1% of young people benefit from TVET, on average (**Figure 12.1**). To some extent, this reflects low overall levels of secondary participation. However, TVET opportunities linked exclusively to general secondary schooling do not provide a genuine alternative. Moreover, one of the most common criticisms of low-performing TVET provision is that it is overly school-based and disconnected from employers' needs (ILO, 2020). Availability of apprenticeships, which are designed to strengthen such connections, remains low (**Box 12.1**).

In the last 15 years, some countries have substantially increased the share of participation by 15- to 24-year-olds in vocational training. All these countries made considerable efforts, supported by political will and commitment, to expand vocational education provision and to reform the existing system of training (**Figure 12.2**).

In Uruguay, an Inter-American Development Bank loan paved the way to systemic reform and relaunch of the TVET system in the late 1990s (Cardozo, 2008), followed by a new education legislative framework in 2008 (INEEd, 2014). Over the years, reform of the country's TVET system included creation of the National Institute for Employment and Vocational Training; the launch of workplace learning, including apprenticeship; recognition of prior learning; and the development of the National Qualification Framework, which is expected to ensure the quality of training provided by about 1,000 institutions operating in the country, including private ones (ILO, 2019).

Donor assistance also played a key role in fostering the modernization of Armenia's TVET system. Between 2007 and 2013, the European Union (EU) provided around EUR 40 million to refurbish and transform TVET

#### **FIGURE 12.1:**

**Very few benefit from TVET in low-income countries** Participation rate in technical and vocational education, 15- to 24-year-olds, and total net enrolment rate in upper secondary, selected low-income countries, 2015–19



colleges into regional centres of excellence (Arribas and Papadakis, 2019). Within a redefined regulatory framework in alignment with the European vocational education area, Armenia more than doubled the proportion of secondary vocational students between 2011 and 2019 (ETF, 2020).

Modernizing existing TVET centres and establishing new ones were instrumental in attracting students in Burundi and Indonesia. In Burundi, which had 20,000 students in vocational programmes in 2011 (World Bank, 2014), the number reached about 80,000 in 2017 after Trade Education Centres were established in each municipality and partnerships with the business sector helped strengthen local training provision (Burundi Ministry of Education and AFD, 2018). Indonesia expanded and revitalized secondary vocational schools and post-secondary community colleges as part of a major relaunch of the TVET system, in line with an extension in compulsory schooling (OECD and Asian Development Bank, 2014). Partnerships with private firms helped expand participation and improve matching of vocational curricula with labour market needs (Triyono and Moses, 2019).

#### BOX 12.1:

#### Apprenticeships remain rare in low- and middle-income countries

Apprenticeships are one way to deliver TVET outside the school environment.<sup>1</sup> Well-designed apprenticeships can respond to skills shortages, support youth in acquiring complex skill sets, smooth school-to-work transitions and, ultimately, reduce youth unemployment (Aivazova, 2013). Austria, Germany and Switzerland, which have a strong apprenticeship tradition, have managed to maintain high youth employment rates. All three countries use vocational training programmes efficiently, with a special focus on potential school dropouts (Dolado, 2015).

In many low- and middle-income countries, apprenticeships are less common and less formalized, in line with high levels of labour market informality, as well as less connected with school-based TVET. As a consequence, few statistics on their prevalence are available. One possible source is the School to Work Transition Survey, which the International Labour Organization (ILO) carried out in 33 countries between 2012 and 2015. It included the question: 'Did you have one (or more) internship(s)/apprenticeship(s) with an employer as part of your education?' On average, almost 1 in 5 respondents aged 15 to 35 said they had.

Prevalence tended to be higher in upper-middle-income countries in Central and Eastern Europe. In the Republic of Moldova, Serbia and Ukraine, between 40% and 50% of young adults had had an internship or apprenticeship. Prevalence varied by self-assessed poverty. In households that reported themselves as poor, 11% did an internship or apprenticeship, compared with 28% in well-off households. In high-income countries such as the United Kingdom, high-status internships are one strategy allowing privileged young people to mobilize family resources to get ahead of their peers (Wright and Mulvey, 2021).

The percentage of people having done an internship or apprenticeship increases to, then peaks around, age 30. Gender differences are minimal, on average, though men are twice as likely as women to report an internship or apprenticeship in Liberia (15% of men and 7% of women) and Malawi (9.5% of men and 4% of women).

Some evidence indicated that apprenticeships served as an alternative to formal TVET. They were least common among graduates of vocational secondary (8%) and post-secondary education (11%). Possibly due to conflation with internships, they were much more common among those with general secondary (36%) or post-secondary education (23%) than among those with only primary education (19%). The percentage of people who did an apprenticeship or internship was much higher for those working in the formal sector.

A subsample of respondents completed a diary describing their complete history of labour market activities that lasted longer than three months. While this record still conflated apprenticeships and internships, the minimum three-month duration, together with the fact that training undergone while studying was excluded, likely filtered out internships. Using this stricter criterion, the prevalence of apprenticeship was only 2% of the small sample of diary completers, which does not allow for further disaggregation. However these data are considered, apprenticeship training does not yet deliver on its potential in low- and middle-income countries.

1 This box is based on Bonomelli Carrasco (2021).

As in Indonesia, compulsory schooling extension required a substantial increase in the number of vocational schools in Costa Rica (OECD, 2017). The rise of TVET participation was also an outcome of increased funding and new coordination mechanisms engaging social and economic parties. Entry requirements became less rigid (Guzmán, 2011) and provision was expanded through new technical services and specializations (Alvarado Calderón and Mora Hernández, 2020; Beirute Brealey, 2018). As a result, the percentage of students enrolled in the public Colegios Técnicos Profesionales increased from 22% in 2010 to 32% in 2018 (Camacho-Calvo et al., 2019).

Strong cooperation across sectors and an emphasis on students from the most disadvantaged households

characterized the relaunch of vocational training in Brazil. Set up in 2011, the Programa Nacional de Acesso ao Ensino Técnico e Emprego (PRONATEC) provided free vocational courses in federal institutes and financed training and apprenticeship experiences to address the persistence of dropout and low training participation (Rambla et al., 2020). PRONATEC aimed to foster cooperation across ministries, which could request provision of specific training programmes in various sectors. However, training tended to reflect training providers' capacity, rather than identified needs (OECD, 2020); expected participation and quality were not reached (OECD, 2015). In 2014, the government began reducing PRONATEC's funding (UNESCO-UNEVOC and National Council for the Federal Network of Vocational Institutions, 2018).

#### **FIGURE 12.2:**

### Some countries have increased TVET participation in recent years

Participation rate in technical and vocational programmes (15- to 24-year-olds), selected countries, 2005–19



Vocational secondary schooling may not seem an attractive option if seen as a 'dead end' (Field and Guez, 2018). This is particularly so if, unlike a general secondary certificate, a vocational secondary diploma does not offer the option of continuing directly into tertiary education. In a quarter of countries, this is the case for all forms of vocational secondary schooling (**Figure 12.3**). By contrast, in 30% of countries, all vocational secondary school graduates enjoy direct access to tertiary education. Elsewhere, vocational secondary school pathways with and without access to tertiary education coexist. The UNESCO, OECD and Eurostat data collection instrument requests details about TVET enrolment with direct access to tertiary, but the data are not routinely published.

#### **FIGURE 12.3:**

#### In a quarter of countries, vocational secondary schooling does not give access to tertiary education Number of countries, by presence of vocational secondary schooling programmes with and without tertiary access,

2019 Some vocational No vocational secondary school secondary school programmes give programmes give direct access direct access to tertiary to tertiary 43 countries 73 countries 49 countries All vocational secondary school programmes give direct access to tertiaru Note: Analysis based on presence of ISCED levels 351, 352 and 353 without and level 354 with direct access to tertiary GEM StatLink: https://bit.ly/GEM2021\_fig12\_3 Source: UIS database

Where vocational secondary graduates lack access to tertiary education, the route may be opened in various ways. Existing forms of vocational secondary education may be upgraded to satisfy tertiary education entry requirements (e.g. through integration with general upper secondary, as in Brazil); existing forms of post-secondary education open to vocational secondary graduates may be upgraded to tertiary level; new forms of vocational secondary with direct access to tertiary may be introduced; or new forms of tertiary education with less restrictive entry requirements may be introduced, similar to the 'universities of applied sciences' in European countries, including Germany, the Netherlands and Switzerland (Field and Guez, 2018). In practice, reforms on the tertiary education side have been more common, such as upgrading vocational degrees (e.g. in nursing) to tertiary level or establishing alternative but fully recognized tertiary institutions in the form of open universities.

66 -

In 30% of countries, all vocational secondary school graduates enjoy direct access to tertiary education

In Tajikistan, a public university with a technology specialization created a lucée within the university structure in the 1990s to provide basic technical education for students in grades 9 and 10. The final exam doubled as an exam for admission to the university (Sabzalieva, 2020). More recently, Denmark's EUX programme, piloted in 2010 and expanded to 24 occupations in 2015, created a vocational pathway into university (Jørgensen, 2015). This hybrid programme improved on a previous initiative that conferred both vocational and tertiary education access gualifications but was seen as saving too little time compared with obtaining the qualifications one after the other. The existence of pathways is insufficient, however. EUX is demanding, and reaches only 2% of TVET students (Jørgensen, 2017). In Estonia, pathways are open in principle, but in 2016, only 21 TVET students took advantage of an extra year leading to a higher education entrance qualification (Musset et al., 2019).

#### **TERTIARY EDUCATION**

Global participation in tertiary education reached 118 million women and 110 million men in 2019. The gross enrolment ratio for tertiary education was 39%, continuing a steady average growth of around one percentage point per year since 2000. Regional values ranged from 9% in sub-Saharan Africa to 78% in Europe and Northern America.

These figures derive from administrative data on tertiary enrolment. Survey data on attendance paint a different picture in many countries (Figure 12.4). Aside from errors in either enrolment or attendance data, enrolment may underestimate attendance if many students attend institutions that are not counted in official statistics because, for example, they lack recognition or accreditation. Conversely, enrolment may overestimate attendance if a significant number of people are enrolled only nominally, especially where tuition is free and student status comes with perks such as subsidized transport. In some German states, the introduction of fees and subsequent reintroduction of tuition-free tertiary education over the past 20 years was accompanied by debates about nominal but inactive enrolment, including the legal and practical advantages and disadvantages of enrolling purely to benefit from perks related to student status. In Germany, in the open-inscription degree subject of physics, less than 30% of new entrants

in the summer terms 2013, 2014 and 2015 attended courses or sat exams (Düchs and Ingold, 2016).

Gross participation ratios compare the number of people of any age enrolled at, or attending, a given level of education with the size of the corresponding age group. Where available, net ratios that consider enrolment and attendance only by individuals of the appropriate age are preferred, and have all but replaced gross participation ratios in primary and secondary education. But in tertiary education, enrolment data by age are rarely available for international statistics, so net enrolment ratios cannot be calculated. Moreover, unlike at lower levels of schooling, especially compulsory, the normative expectation of enrolment at a specific age is much weaker. For purposes of gross participation ratios in tertiary education, the five years immediately following the upper secondary age bracket are customarily used.

However, tertiary study at higher ages is common. In many countries in sub-Saharan Africa, young people are more likely to be attending tertiary education after the conventional five-year age bracket than even two years into it (**Figure 12.5**). In half the countries with recent data, attendance in the older age group is twice as high or even more. Attendance at older ages can reflect delayed graduation, but also result from later entry, which is not inherently problematic. However, large differences in the age profile of new and continuing students in tertiary education can distort comparisons of gross participation ratios.

Target 4.3 calls for tertiary education to be affordable. Unlike lower levels of schooling, where private providers are often publicly funded or directly contracted by the public sector, private tertiary education almost always requires students or their families to cover the full cost of provision. Even high tuition is often seen as a worthwhile investment, given the promise of higher lifetime incomes – although this promise is not met for all. However, being able to afford tertiary education from a lifetime perspective does not make it affordable upfront. The economics case for cost sharing in tertiary education depends crucially on prospective entrants not facing credit constraints. Student loans of various kinds have grown into a trillion-dollar market. However, they do not always fulfil their purpose in practice. Poorly designed student loan or equivalent arrangements may be not just unsustainable for many graduates but also a poor revenue stream for the government (Focus 12.1).

#### FIGURE 12.4:

Tertiary education enrolment data may overestimate or underestimate actual attendance

Gross attendance and enrolment ratios in tertiary education, 2015–19



#### **FIGURE 12.5:**

#### Delayed tertiary education is the norm in many countries in sub-Saharan Africa

Ratio of tertiary education attendance rates one year after and two years into the theoretical age range for tertiary education, selected sub-Saharan African countries, 2015–19



## FOCUS 12.1: MANY FAMILIES HAVE

Student loans have become an increasingly popular instrument to promote affordable access to tertiary education. They are currently available in over 70 countries, to varying degrees. The average share of tertiary students receiving a public or government-sponsored student loan in 19 mostly high-income countries was 40%, ranging from 1% in Switzerland to 42% in South Africa and 84% in New Zealand (**Figure 12.6**). Student loans can increase access at a lower cost to governments than tuition subsidies or scholarships and grants and can even help improve equity, completion rates and graduates' labour market outcomes. Student loans that also cover living costs allow students to be financially independent from their parents (Ziderman, 2017).

**TROUBLE REPAYING STUDENT DEBT** 

Often, however, student loans fall short of their promises of increased access and affordability. In Brazil, of the 2.2 million new student loans issued between 2009 and 2015, fewer than half translated into new enrolments; the majority of new loans were taken by students who were already enrolled or would have enrolled anyway (Brazil Ministry of Economy, 2017). In the United States, much of the increase in national student debt fed tuition spikes rather than new enrolments (Lucca et al., 2017).

Student loans may also have different uptake rates by population group or programme type. Studies show that the risk-averse, who may be over-represented in some ethnic or immigrant groups or among poorer households, are less likely to take on student loans (Boatman et al., 2017). In Japan, debt aversion in poorer families diverts students from general to vocational high schools, hindering their prospect of entering tertiary education (Furuta, 2021).

The high levels of debt many students carry into adulthood call into question whether student loans are actually affordable. Average debt at graduation varies considerably by country and depends on the level of tuition and living costs, the amount of aid or tuition subsidies, and loan repayment conditions. In England (United Kingdom), where nearly 95% of tertiary education students have a student loan, the average debt at graduation is US\$50,000 (OECD, 2019). The level can quickly increase if students have trouble repaying, as institutions may add late fees or, as in the United

"

#### **FIGURE 12.6:**

#### Almost all students receive a loan in some countries, almost none in others

Share of tertiary students who receive public or government-sponsored student loans, selected countries, 2018 or latest available year



States, use for-profit debt collection agencies that can add 30% to 40% to the original amount (Kolodner, 2021).

More than the absolute value of the debt, what matters is students' ability to repay it. In many countries, the proportion of borrowers' income required to repay loans is excessive, especially for the least well-off graduates. In Brazil, a simulation exercise estimated that the repayment burden for female graduates in the bottom guintile ranged from 100% to 55% between ages 25 and 37 (Dearden and Nascimento, 2019). In Indonesia, it can vary from around 30% in a relatively high-income area (Java) to around 85% in a relatively low-income area (Sumatra) (Chapman, 2016). In Viet Nam, simulations show repayment burden varying between 20% and 85%. Even graduates in developed countries face high repayment burdens, ranging from 50% for public lawyers in the United States to 70% for women in former East Germany (Chapman, 2016; Chapman and Lounkaew, 2015).

In many countries, the proportion of borrowers' income required to repay student loans is excessive, especially for the least well-off graduates

"

High levels of student loan defaults and payment delays, therefore, are hardly surprising. In Brazil and Chile, over 40% of the student loan contracts in the amortization phase had at least three months of payment delays, enough to harm borrowers' credit reputation (**Figure 12.7**). The situation is even more concerning for those who drop out, as they become more likely to have lower earnings, delay payments and default on their loan. In Chile, over 70% of borrowers who dropped out have notable delays in student loan payments, compared with 34% of those who graduated (Ingresa, 2019). In the United States, late payments and defaults are highest at low levels of debt – around US\$2,000 – likely from students who dropped out early (D'Amato, 2021).

Difficulties in repaying student loans can snowball into further financial troubles, as households have less disposable income and a damaged credit reputation, both of which can hinder access to further credit. Student loan borrowers are less likely to buy homes (Mezza et al., 2019) and save for retirement (Rutledge et al., 2018). And as easy as it may be to get a student loan, in many countries it is nearly impossible to get rid of them, even through debt settlement procedures or bankruptcy (Eurofound, 2020; Latham, 2020).

Accumulated across many families, a high level of outstanding student debt could pose risks to a country's economy. In the United States, where student debt nearly tripled between 2007 and 2019 (Latham, 2020) to US\$1.6 trillion (Federal Reserve, 2020), there are concerns that this is hampering small businesses' growth (Ambrose et al., 2015) and jeopardizing households' ability to weather financial crises (Elliott and Nam, 2013). Once student debt

#### **FIGURE 12.7:**

**Repayment difficulties are becoming more common in Brazil and Chile** Number and share of student loan contracts with at least 90 days of delay, Brazil, Chile and Colombia, 2012–20



GEM StatLink: https://bit.ly/GEM2021\_fig12\_7

Sources: Information requests to the governments of Brazil (FNDE, 2020), Chile (Ingresa, 2020) and Colombia (ICETEX, 2020).

has ballooned, there is no easy way out for governments, as forgiveness programmes may be considered regressive and unfair, and may lead to increases in tertiary education costs (D'Amato, 2021; Lowrey, 2020).

Some countries have tried to address these concerns by imposing a debt ceiling, but maximum loan amounts are often not enough to cover both fees and living costs. Under-borrowing can cause its own problems, such as lowering access to loans. It may also push students to work longer hours (Black et al., 2020), which can have perverse effects on academic performance (Callender, 2008; Kalenkoski and Pabilonia, 2010), physical and mental health (Oviatt et al., 2017) and completion time (Darolia, 2014). In the United States, student loan caps have led parents to increasingly take on debt to help finance their children's education - usually at a time in their lives when income is stagnant and soon to decline as they enter retirement. Between 2014 and 2019, the number of recipients of the federal loan programme Parent PLUS increased by 13% and the amount borrowed by 36%. The increase in borrowing has been accompanied by a rise in defaults: In 2019, US\$96 billion was outstanding from 3.6 million parents (Fletcher et al., 2020).

A more promising policy reform has been a shift from the widely used time-based repayment loans (mortgage style) to income-contingent loans. A growing number of countries have adopted full or partial income-contingent loan systems, including Australia, Ethiopia, Hungary, the Netherlands, New Zealand, the Republic of Korea, South Africa, the United States and England (United Kingdom) (Britton et al., 2019; Yizengaw, 2007; Ziderman, 2017).

In addition to lowering administrative costs for governments, income-contingent loans fix repayment burdens to reduce the possibility of facing repayment hardships in the future. Australia set the maximum repayment proportion of annual income for such loans at 8%, New Zealand at 9% and England and Wales (United Kingdom) at 10%. By providing insurance against low earnings or unemployment, the introduction of such loans has increased overall access to tertiary education in some countries. In Australia, the introduction of tuition fees combined with income-contingent loans increased overall tertiary enrolment, including for students from low-income households (Chapman, 2016).

Success stories in high-income countries have helped advance this type of loan reform in low- and middle-income countries. Experts from Australia and the United Kingdom, for example, helped develop income-contingent loan proposals adopted in Brazil and

Ethiopia, and are involved in public debates on the topic in Colombia and Malaysia (Chapman and Dearden, 2018; Filizola, 2019; IPEA, 2019; Woodhall, 2007). However, the suitability of income-contingent loans in less developed countries is debated. Income-contingent loans require a universal regime of income taxation or social security collection that identifies, tracks and collects repayments (Ziderman, 2017). Poorer countries face challenges such as large informal sectors or insufficient short-term government funding of the type needed for such programmes. In Ethiopia, the graduate tax (a modified type of income-contingent loan) introduced in 2003 has had low cost recovery (Portela and Gebremedhin, 2020), likely due to the federal system's weak collection capacity, lower-than-expected graduate earnings and high levels of brain drain (Woldegiorgis, 2008).

Student loans can be a powerful tool to alleviate financial constraints and increase access to tertiary education, but policies and design features have important implications for programmes' ability to make tertiary education affordable and increase access equitably.

#### **ADULT EDUCATION**

Indicator 4.3.1 refers to participation in formal and non-formal adult education and training in the past 12 months. A long-standing challenge is poor data coverage outside Europe, combined with limited comparability of alternative data sources, especially where they refer to participation in the past month instead.

Although participation in adult education is significantly higher in high-income countries, most of it is sponsored by an employer. In 35 European countries, employer-sponsored training accounted for nearly 90% of the total (**Figure 12.8**). But employer sponsorship alone does not overcome other constraints, such as childcare needs, or cover all direct costs. Accordingly, not all sponsorship that is offered is used (Pulkkinen, 2021).

66

A more promising policy reform has been a shift from the widely used time-based repayment loans to income-contingent loans

**?**?



#### Most participation in job-related training in Europe is paid for by employers

Adult participation in job-related training, by employer sponsorship, European countries, 2016



It is clear that private employers play a significant role as sponsors, although data include the public sector. Employers and their associations also accounted for 40% of job-related training provision in the 35 countries studied (**Figure 12.9**). Education and training policies thus need to target individuals who are outside the labour market and lack access to employer-sponsored training (Henehan, 2020). Community learning centres can address this need (**Focus 12.2**). Even for the employed, time to pursue training may be as important as sponsorship, showing the need for public intervention in the form of education leave (**Focus 12.3**). In theory, employers may be expected to underinvest in training for their staff without subsidies. In practice, the evidence is inconclusive.

66 -

Employers and their associations accounted for 40% of job-related training provision in 35 European countries

**?**?

The standard data sources for indicator 4.3.1 provide a snapshot of training participation, but not always its distribution. The Comparative Panel File (CPF) harmonizes the longest-running national longitudinal surveys from Australia, Germany, the Republic of Korea, the Russian Federation, Switzerland, the United Kingdom and the United States. It shows information on participation in training, disaggregated by characteristics such as education attainment, employment status and, except for in United States, job security (Turek et al., 2021). On average over 2010–19, 15% of respondents reported having participated in training in the previous year, while 31% of those observed over at least 10 years reported having engaged in non-formal work-related education at least once. A single training occurrence (40%) or at most two (20%) in a lifetime is the majority experience, but adult education is a recurrent pursuit for a significant minority. Four or more lifetime training episodes are reported by 28% of participants, and 29% of training periods last three or more years in a row.

#### **FIGURE 12.9:**

Source: Eurostat database.



Non-state actors provide most job-related training Adult participation in job-related training, by provider type, European cou Education qualification is the most influential determinant of training. In a finding consistent with other internationally comparable data, adults with higher educational attainment are more likely to continue learning. In addition, they disproportionately fill the ranks of frequent learners: People with tertiary education (66%) are twice as likely as those with secondary education (33%) to have 10 or more training occurrences in their lifetime (**Figure 12.10**). As a result, frequency of training, like participation overall, tends to further increase education inequality. Those who are already more educated are more likely not only to get further training but also to get more of it.

#### FOCUS 12.2: COMMUNITY LEARNING CENTRES HAVE PROLIFERATED IN MANY COUNTRIES

Community learning centres (CLCs) are increasingly recognized as playing an important role in providing education opportunities meeting local communities' needs. Found worldwide, they have a long tradition in many contexts, such as folk high schools in Europe, popular education centres in Latin America and community development centres in Northern Africa and Western Asia (Gartenschlaeger, 2017). Despite the differing terminologies and origins, community engagement remains a standard feature (NILE and UIL, 2016).

CLCs' functions and their contributions to the SDGs were explicitly recognized by the Belem Framework for Action at the 2009 Sixth International Conference on Adult Education, which called on countries to create multipurpose community learning spaces or centres to provide inclusive access to and participation in the full range of adult learning and education programmes (UIL, 2010). The Education 2030 Framework for Action reaffirmed the need to 'make learning spaces and environments for non-formal and adult learning and education widely available, including networks of community learning centres' (UNESCO, 2015). Embracing an intersectoral approach to education beyond formal schooling, CLCs can act as learning, information dissemination and networking hubs. The learning experience in the community is well-suited for multiple activities aimed at increasing members' overall empowerment, and their multidisciplinary approach to education includes learning for work, livelihoods and health (UNESCO, 2016).

#### **FIGURE 12.10:**



Adults with a tertiary education qualification are likely to be frequent training participants Training occurrences in lifetime, by education level, selected high-income countries, 2010–19

Note: Data are averaged over Australia, the Republic of Korea, the Russian Federation, Switzerland and the United Kingdom.

GEM StatLink: https://bit.ly/GEM2021\_fig12\_10

Source: GEM Report team calculations based on the CPF (2021) database.

## In Indonesia, most community learning centres are led by non-state initiatives, often NGOs

Since 2000, the number of CLCs worldwide has grown rapidly, particularly in Asia. In Viet Nam, their number rose more than 14-fold between 2002 and 2010; by 2014/15, there were 11,000 centres (NILE and UIL, 2016). Nepal experienced similar exponential growth, from fewer than 20 CLCs in 2000 to 1,900 in 2012 (Govinda, 2017).

The establishment and management of CLCs has been bolstered by local and national government authorities and non-state actors, such as non-governmental organizations (NGOs), which have supported community engagement with financial and human resources. In some countries, the role of non-state actors is more prominent. In Indonesia, most CLCs are led by non-state initiatives, often NGOs. Similarly, NGOs run all 5,000 or so centres in Bangladesh (NILE and UIL, 2016).

Despite their growth, CLCs vary in distribution. In Nepal, an evaluation of the national literacy campaign pointed out that progress in literacy strategy implementation had been hampered by lack of funding to local CLCs and consequent limited availability of classes in rural areas (Nepal Ministry of Education and UNESCO, 2017). By contrast, every subdistrict in Thailand has a CLC that acts as an information hub and education institution, providing non-formal education, including cultural activities, vocational training, disaster prevention courses and environmental preservation training. In Cambodia, local communities managed to keep CLCs active, despite limited financial support, to deliver training in tailoring, traditional music, hairdressing, stone carving, weaving, carpentry, information and communication technology (ICT) and English. Aiming to provide training even to remote areas, CLCs launched the initiative Mobile Life Skills, delivering training by van (Govinda, 2017).

In contexts where communities were unserved, centres have been set up to pursue knowledge at the local level. Morocco has more than 16,500 literacy centres, but only 8,700 in rural areas. To meet the need for a variety of courses in rural areas, 200 CLCs have been set up across the country, delivering literacy and post-literacy classes, mostly for women, along with professional training and preschool classes (Chaker, 2017). In Ukraine, adult education centres were concentrated in big cities on state education institutions' premises. Parallel centres run by NGOs have been established to provide local communities with vocation-oriented programmes and with cultural and leisure courses (Lukyanova and Veramejchyk, 2017).

In Indonesia, besides CLCs, community-based Vocational Training Centres (Balai Latihan Kerja) were relaunched as part of a Ministry of Manpower programme in 2017 to encourage vocational education in remote areas and to train and reskill qualified workers. As of 2020, 2,127 centres had been established nationally and 787 more were planned in 2021 (Chau, 2021). Developed in the 1970s and 1980s to address manufacturing sector needs, they provide non-formal automotive, textile, electronics and ICT courses for jobseekers and students and for workers as part of their professional development. However, participation lags behind expectations due to concerns about infrastructure, equipment and staff quality (OECD and Asian Development Bank, 2020).

CLCs are characterized by broad-spectrum learning provision that adapts to local needs. In the United Republic of Tanzania, Folk Development Colleges, founded in the 1970s, provide adult education, community development and vocational training. They offer one- to two-year vocational programmes, recognized and validated by the Tanzanian Vocational Education and Training Authority, as well as short courses on campus and in the community. Each college addresses community needs, making its resources and facilities available to members (Rogers, 2019).

CLCs in Myanmar are involved in multiple learning activities, including second-chance education and thematic training on issues such as health, communication skills in family relationships, and agricultural techniques and practices, which are of particular relevance for farmers. The centres also act as a multilevel network enabling social contacts between villages and external actors, with whom they can share knowledge and skills and mobilize resources (Pham Le, 2018). In other contexts, they may respond to emerging and specific needs, relying on their ability to reach people in the most marginalized areas. In Mexico, 32 Digital Inclusion Centres (Puntos Mexico Conectado - Centros de Inclusión Digital) have been established, one in each state and Mexico City, to provide free basic digital skills programmes and entrepreneurship courses, especially to youth and girls (OECD, 2020).

#### FOCUS 12.3: EDUCATION LEAVE IS A TOOL TO PROMOTE ADULT EDUCATION IN HIGH-INCOME COUNTRIES

SDG 4 aims to promote 'lifelong learning opportunities for all' and global indicator 4.3.1 aims to capture that aspect, which, somewhat surprisingly, is not explicitly included in any individual target. However, apart from the fact that there remain considerable data gaps for this indicator, it is important to directly monitor the extent to which all adults enjoy the opportunity to participate in education and training.<sup>1</sup>

Time and cost constraints have been found to impede workers' access to continuing training. Paid study leave is an instrument enabling workers to dedicate more time to training without losing their jobs and income. Accordingly, the ILO considers study leave a key tool to promote the right to continuing education. ILO Convention 140 on paid study leave, approved in 1974 and ratified by 35 countries, defines it as leave granted to workers for educational purposes, for a specified period, during working hours and with payment of adequate financial entitlements.

No systematic cross-country comparative information on the regulation of education leave exists outside of a dated overview for Europe (Cedefop, 2012). Most countries regulate study leave through legislation, although a few, such as the Netherlands, rely almost entirely on collective bargaining agreements, of which there may be hundreds, with varying rules. Even where study leave is regulated through national legislation and paid for by the state, statistics on take-up at the national level may be unavailable, especially in federal states such as Germany and Mexico.

To fill this data gap, a study for this report analysed the situation in 27 countries where data could be obtained. Countries can be classified into four groups according to the level at which they protect workers' right to training, depending on the existence of specific normative instruments, payment during the leave and the breadth of eligibility criteria determining which workers have the right to such leave. Universally, benefits such as employer financing and guaranteed return to employment are tied to criteria such as minimum job tenure, although the cut-off periods differ widely.

The first group of countries does not guarantee the right to such leave. Agreements regarding it may exist, but there is no regulation of an entitlement. The second group has restricted rights to study leave. Canada and Costa Rica offer leave only for public workers. In Chile, only people between ages 18 and 24 are eligible to request unpaid study leave. In the United Kingdom, only workers in companies with more than 250 employees are eligible and study leave is unpaid. In the third group, with moderate rights to study leave, countries have regulations for paid leave but distinguish between various groups (such as public vs private sector), regions or industries, and typically limit duration to no more than 30 days per year. The fourth group, with highly protected rights to study leave, comprises countries with universal entitlements to paid leave, including of long duration. In Austria, it can last up to one year, Finland up to two and Norway up to three, while Sweden has no defined limits. In Finland, large positive effects of study leave participation on education attainment are observed, especially for the less educated (Kauhanen, 2021).

There is no clear relationship between study leave regulation and overall participation in adult education and training. This is unsurprising, since even the most generous programmes reach only a small fraction of workers. Austria is one of the few countries with publicly available, disaggregated data on study leave take-up. In 2019, 21,147 individuals benefited, almost 60% of them women. However, this amounts to less than 0.5% of a labour force of around 4.5 million. Only 41,440 participated in training leave in France in 2013, out of a labour force of around 30 million, or 0.13%. However, even such low annual participation rates could amount to 5% to 20% participation over the course of a working life.

Indicator 4.3.1 is dominated by participation in non-formal education. Typically, certification requirements exist for course attendance or exams taken to qualify for study leave. A more direct comparison is with the share of employed persons participating in formal education and training. The EU average is only 4.2% (Eurostat, 2021b). It is likely that the contribution of education leave to full-time participation in courses of long duration is higher, especially for freely chosen courses.

<sup>1</sup> This section is based on Batthyány et al. (2021).

In Uruguay, while workers have access to relatively short study leave, especially in the private sector, this entitlement enjoys exceptionally strong legal protection

Indicator 4.3.1 also includes training carried out by employers themselves. On average, 70% of enterprises in the EU provide continuous vocational training to their employees (Eurostat, 2021a). In some countries, including Slovenia, study leave can be used only for courses or studies that are of interest to the employer. Even where choice of course is more flexible in principle, employers' discretionary agreement is typically required, although employees may have a right of appeal to regulatory bodies or unions. France offers exceptionally strong protection: Employers cannot refuse eligible requests for individual training leave up to a year. Even there, however, employers may delay the leave's start by nine months, requiring workers to make their plans almost a year in advance.

Reliance on employer goodwill and support may result in too little take-up of education leave. Collective bargaining agreements enable flexibility that allows employers and workers to jointly agree on how leave should be offered, depending on the organization of the sector, profession or company. In some countries, costs are assumed through dedicated funds. In Austria, they are covered by funds provided by employers, government and workers. Similarly, in France, a tripartite body collects funds paid by private employers for worker training.

In Uruguay, while workers have access to relatively short study leave, especially in the private sector (9 days per year for employees working between 36 and 48 hours per week), this entitlement enjoys exceptionally strong legal protection. Workers become eligible after six months of employment. Employers cannot refuse and must pay the full salary during the leave. The combination of modest entitlements and strong protection represents a feasible strategy for resource-constrained settings. Increasing the duration of study leave as economic development allows is thus a matter of incremental change rather than introducing a new instrument.

99

#### COVID-19

**TVET** was severely affected by COVID-19. A May 2020 global survey found that 90% of respondents had experienced school closures and 98% a disruption of work-based learning. One in two low-income and one in three lower-middle-income countries had to cancel all training. Among 92 countries, in 13 a majority of respondents had used distance learning regularly prior to the pandemic; during the pandemic, the number reached 46, or 50%, ranging from 12% of low-income to 72% of high-income countries. Almost three quarters of respondents reported that certification tests were also delayed (ILO et al., 2020).

Remote teaching presents challenges in TVET. Up to 80% of programmes focus on practical and soft skills, which should be acquired in person (Commonwealth of Learning, 2020). A survey of 27 OECD countries showed that some forms of work-based learning and apprenticeships continued despite the pandemic, and some countries prioritized vocational education when education institutions reopened. In the Netherlands, vocational secondary school students continued learning at school. In Poland, practical classes took place in person with some restrictions, while theoretical classes moved online. About 81% of countries made physical adjustments to schools or classrooms, while in 73% of countries the return to school was in hybrid mode (OECD, 2021a). In Ecuador, students carried out practical activities at home and sent pictures and videos of their finished work to be assessed (Hoftijzer et al., 2021).

Preparing teachers has been a major issue. Palestine focused efforts on TVET teachers' capacity to deliver distance learning, while recognizing the lack of a relevant competency framework (Samara, 2021). A South African capacity development programme, aimed at professionalizing TVET lecturer education, had to adapt its approach to online seminars while addressing unequal access to the internet and devices (Scheepers and Gebhardt, 2021). But in some cases, the pandemic did not interrupt longer-term efforts to upgrade vocational education. Indonesia was in the middle of a reform that included a process for professionalizing vocational teacher education and introducing an in-service modality for practising but uncertified teachers. An online platform has enabled this reform to continue through a learning management system (Setiawan and Hamdani, 2021).

Access to devices and the internet is a hurdle in low- and middle-income countries, including in sub-Saharan Africa (Kamaté and Siahoué, 2020). In Sri Lanka, the share of TVET institutions providing distance learning increased from 36% before the pandemic to 92% providing at least one online course during the pandemic. However, only one in five households owned a desktop or laptop computer and 96% of students able to attend online classes did so mostly with low-tech solutions (Hayashi et al., 2021). Thus it is important for delivery of distance learning to use multiple approaches and not rely solely on high-tech solutions. In India, Swayam Prabha Direct-to-Home television channels offer vocational education classes. In Rwanda, even before the pandemic, non-smart mobile phones combined with interactive voice response were used in training community health workers (Hoftijzer et al., 2020).

Collaboration between government and non-state service providers grew during the pandemic as some countries encouraged remote training provision, using digital tools and platforms, by private companies (OECD, 2020b). Armenia's National Network for Distance Learning, an NGO, trains trainers at technical colleges and schools for the National Centre for Educational Technology Development. Collaboration with unions has supported training of frontline staff. The United Kingdom's NHS Professionals, an association of health and social sector workers, and Skills for Health, the sector skills council for health, established a free online training course on COVID-19 (Skills for Health, 2020).

The pandemic demonstrated that jobs in highly affected sectors crucial to the economy rely on vocational training, which can meet urgent demand as well as build resilience to crises. The Dominican Republic's National Institute for Technical and Vocational Training launched a course on new health-related protocols for the tourism industry (ILO/Cinterfor, 2020).

**Tertiary education** students were more exposed to remote learning than students at other education levels prior to the pandemic, even if many, including TVET students, also did practical courses. In a survey of 53 countries, only 3 reported switching fully to online higher education, 19 had primarily online modalities and 28 used a hybrid approach of remote and face-to-face learning (UNESCO, 2021). But a sub-Saharan African survey found that by mid-2020 over 80% of students Countries varied in their choice of online platforms. In Latin America, 60% of universities adopted Moodle, followed by Google Classroom (30%), own-developed platforms (21%) and Blackboard (7%) (IESALC, 2021). Many universities in the Arab States used Moodle or social media, including Facebook and YouTube, to teach lessons based on electronic publications sent through the university website (Lassoued et al., 2020). Colombia created CO-LAB, a collaborative platform to share good pedagogical practices and digital resources among universities. Egypt launched its first digital platform for university distance learning in partnership with Microsoft. The Russian Federation launched a platform called University 20.35 (OECD, 2021b). In China, universities offered 24,000 online courses on 22 platforms, including 1,291 competitive high-quality courses selected by the Ministry of Education and 401 courses featuring virtual experimental simulations (Sun et al., 2020).

Preparedness for switching to online learning can be evaluated in terms of at least 10 factors: a business continuity plan, emergency management offices, electricity, internet, a learning management system, videoconferencing, digital content resources, teaching and learning units, trained instructors and cubersecurity (Salmi, 2020). Before the pandemic, 80% of Latin American universities had learning management systems and online platforms suitable for distance education, while 8% implemented them after the crisis. However, only 68% of teachers regularly connected to the platforms, compared with 80% of students (IESALC, 2021). In Viet Nam, only 2% of tertiary education students engaged in distance learning in 2016, mostly because regulations prevented tertiary institutions from offering courses exclusively online. During the pandemic, 110 of more than 200 institutions moved online. Older teachers received support in using online platforms and WhatsApp groups for discussion and sharing assignments (Pham and Ho, 2020).

The move to remote learning was most challenging for practical courses requiring site-based resources. In India, 65% of medical students missed being able to take traditional anatomy based on dissection, models, microscopic slides and interaction with mentors; in addition, 83% said they lacked devices and internet bandwidth (Singal et al., 2021). In anglophone sub-Saharan Africa, 73% of students engaged in research reported that laboratory or field research activities had been suspended (Mawazo Institute, 2020). Veterinary courses moved online, showing procedures via video and sometimes with 3D virtual tools (Mahdy, 2020). In pathology, microscope visualizations were used as an alternative to hybrid modes, where students were allowed into laboratories under social distancing rules (Kwon et al., 2020).

The impact of the pandemic on enrolment was ambiguous. Contrary to what is often perceived, enrolment tends to expand when economic opportunities shrink, as experience in the United States from the 2007/08 financial crisis showed (Barr and Turner, 2013). But the challenges of attending university under lockdown restrictions may have had the opposite effect. The latest evidence from the United States suggests that undergraduate enrolment fell by 6.5% between 2019 and 2021 but the effect was highly unequal: Enrolment in four-year private for-profit and two-year public institutions fell by at least 13% while in highly selective universities and in graduate courses enrolment rose despite the potential impact of reduced international student mobility (Figure 12.11). A global review of 57 countries, based on non-comparable data, suggested no clear pattern; the number of countries reporting increases in enrolment was roughly equal to the number reporting decreases for 2019 and 2020. However, Armenia, Hungary and the Bolivarian Republic of Venezuela reported decreases of at least 20% (UNESCO, 2021). The Brazilian university entry examination saw the fewest applicants since 2007, especially among black, brown and indigenous students (Pinheiro, 2021).

The unequal impact of the pandemic extended beyond enrolment and access to living conditions. In EU countries, 41% of students who worked during their studies lost their jobs, 29% temporarily and 12% permanently; 20% of students had financial concerns about cost of living. Yet tuition levels remained unchanged for 75% of students; only 16% benefited from flexible payment plans or cancellation of payments (Farnell et al., 2021). In Latin America, more than 60% of public and 90% of private higher education institutions provided tuition discounts (IESALC, 2021). In adult education, many traditional programmes were suspended and there was a move to digital platforms. Adult learners are likely to drop out in adverse economic conditions (Singh et al., 2021). In Ireland, there was a 25% reduction in certification at National Framework of Qualifications levels 1 to 4 (O'Reilly, 2021). In OECD countries, it was estimated that participation in non-formal learning decreased by 18% and in informal learning by 25%, with losses differing by sector, closure duration and skill level. The reduction in informal and non-formal learning opportunities was over twice as large for medium- and low-skilled workers as for tertiary-educated adults (Paciorek et al., 2021).

Adult educator communities maintained contact with learners during the crisis. Literacy groups in Canada's Quebec province telephoned learners to make sure they understood government measures (Brossard, 2020). The United Kingdom's Learning and Work Institute, an NGO, created a toolkit with partners to evaluate the effectiveness of online learning solutions in building career adaptability skills and learner motivation for people in shrinking sectors and occupations (Sadro et al., 2021). Population group networks have also been important in reaching out to disadvantaged people for education purposes. In Egypt, the Women and Society Association's YouTube channel shares education, social and creative content to help mothers (UIL, 2020b).

Local authorities also responded to COVID-19, drawing on their experience with non-formal and informal learning. In Lima, Peru, the Aprendo en Casa ('I Learn at Home') programme promotes intergenerational learning with content targeted at different family members, including material for formal education of students and educational and recreational television and radio content, in 10 indigenous languages and sign language (UIL, 2020a; UNESCO, 2020).

#### **FIGURE 12.11:**

In the United States, changes in tertiary education enrolment vary by level and institution type Change in tertiary education enrolment, by level, sector and selectivity, United States, 2019–21



Where infrastructure existed for digital learning, positive outcomes were reported from continuing online learning. Across OECD countries, the flexibility and continuity of online learning led to more participation in some adult education courses. Governments supported such learning though free or subsidized courses, often in partnership with other actors targeting learners and teachers. The Ontario School Board in Canada partnered with Apple to provide teachers with digital teaching materials, including free virtual coaching sessions (OECD, 2020a).

However, connectivity, digital skills and availability of adequate devices do not always ensure access to online learning. In some cases, migration to telephone applications filled the gap. Globally, in 2017–18, there were 7.5 times more mobile phone subscriptions than fixed broadband subscriptions, on average (Kovacevic and Jahic, 2020). To address the connectivity challenge, Media Works Cape, a private South African provider of adult education, delivered courses through WhatsApp, which facilitated virtual learning by allowing learners to ask questions and share videos (Carroll, 2020). Before the pandemic, studies showed WhatsApp to be the preferred communication application for learning in sub-Saharan Africa whereas, for example, Facebook consumes more data and thus is more expensive (Madge et al., 2019).

Mahmoud, 20, is currently completing his six-month training contract at Classic Fashion in Irbid, Jordan, which gives refugees an introduction to the labour market.

CREDIT: UNHCR/Mohammad Hawari

## **KEY MESSAGES**

In around half of 91 countries, a majority of adults possess none of the nine core ICT skills monitored under SDG 4. Schooling is a strong predictor of ICT skills: in Cuba, Kiribati and Zimbabwe, those with upper secondary education and above possess at least seven of the nine core skills, on average.

Among 20- to 24-year-olds, 98% of women and 90% of men in Chad but 36% of women and 31% of men in Tunisia reported never having used the internet.

In the 2018 International Computer and Information Literacy Study, the main cross-nationally comparative measure of multidimensional digital literacy, girls performed significantly better than boys in all 12 upper-middleand high-income countries that took part.

Across countries that took part in the 2018 PISA, 35% of 15-year-old students reported never learning about compound interest at school in the past year, with shares ranging from 15% in Finland to 63% in Italy.

Computational thinking – solving problems through logical and algorithmic reasoning – is increasingly included in curricula in Europe and East Asia.

COVID-19 lockdowns and school closures made basic digital skills a prerequisite for learning and skills acquisition. In turn, digital skills have been key in gaining access to up-to-date information, medical appointments and, as in Panama, increasingly digitalized public services.

#### **CHAPTER 13**



# TARGET 4.4 Skills for work

By 2030, substantially increase the number of youth and adults who have relevant skills, including technical and vocational skills, for employment, decent jobs and entrepreneurship

#### **GLOBAL INDICATOR**

**4.4.1** – Percentage of youth/adults with information and communications technology (ICT) skills, by type of skill

#### THEMATIC INDICATORS

**4.4.2** – Percentage of youth/adults who have achieved at least a minimum level of proficiency in digital literacy skills

4.4.3 – Youth/adult educational attainment rates by age group and level of education

Global indicator 4.4.1 measures recent use of information and communication technology (ICT) as an indirect measure of computer-related skills. Household survey respondents report whether they carried out any of nine activities in the previous three months, from sending messages with attachments to connecting and installing new devices to writing a computer programme. The International Telecommunication Union (ITU), as co-custodian agency of the global indicator, introduced some changes to the skill set in 2019, e.g. to adjust to the increasing role of mobile devices relative to computers, but they are not uet reflected in reported data.

The framework for indicator 4.4.1 must be seen as aspirational in relation to current levels of ICT skills. In only 10 of 91 countries with data do a majority of respondents report having the skill in question for at least 5 of the 9 skills (**Figure 13.1**). In around half the countries, no skill is possessed by a majority of adults, even though low- and lower-middle-income countries are under-represented in the underlying ITU data. Policies in developing countries must therefore address scarcity of digital skills as well as deploy technology with lower skill requirements. The successful mobile money service M-Pesa in Kenya, for example, works with SMS text messages and does not require a smartphone (James, 2021).

Indicator 4.4.2 is meant to directly assess the 'Percentage of youth/adults who have achieved at least a minimum level of proficiency in digital literacy skills'. Digital literacy is a complex construct that goes beyond isolated ICT usage skills. While not fully aligned with the emerging consensus on an SDG 4 framework for digital literacy and associated minimum proficiency levels, the 2018 International Computer and Information Literacy Study (ICILS) represents the only currently available attempt to comparatively measure multidimensional digital literacy, although its sample, focusing on grade 8 students, consists almost exclusively of high-income countries. In addition, the 2018 ICILS contains an optional module on computational thinking, a skill that is correlated with, but distinct from, digital literacy (Focus 13.1).

#### 66 -

Policies in developing countries must address scarcity of digital skills as well as deploy technology with lower skill requirements **FIGURE 13.1** 

**Digital skills are unequally distributed across countries** Percentage of adults possessing nine ICT skills, selected countries, 2015–19



Source: UIS database.

,,

The sixth round of the Multiple Indicator Cluster Surveys (MICS) includes a module with questions on the nine ICT skills and whether respondents have ever used a computer or the internet. The data cover adults aged 15 to 49 and allow disaggregation of skill levels by individual characteristics. In low- and middle-income countries, access to ICT devices and the internet represents a ceiling on the prevalence of ICT skills. Even among 20- to 24-year-olds, 98% of women and 90% of men in Chad reported never having used the internet; the respective shares were 61% and 63% in the Lao People's Democratic Republic and 36% and 31% in Tunisia.

Disaggregating prevalence of ICT skills by education attainment reveals stark gaps (**Figure 13.3**). In Cuba, Kiribati and Zimbabwe, those with upper secondary or higher education possess more than seven of the nine skills, on average, comparable to digitally advanced societies. But in Iraq, the Lao People's Democratic Republic and Sierra Leone, even the most educated average fewer than two skills. In most countries, few young people who had not completed at least lower secondary school possessed any ICT skills. Universal secondary education will play an important part in any attempt to reach target 4.4.

**FIGURE 13.3:** 

#### 66

In Iraq, the Lao People's Democratic Republic and Sierra Leone, even the most educated average fewer than two of nine ICT skills



#### 



99
A basic ability to use digital tools effectively is increasingly important in almost all countries, making it one of the few comparable labour market skills that can be monitored on a global scale. However, the scope of target 4.4 – 'relevant skills, including technical and vocational skills, for employment, decent jobs and entrepreneurship' – is clearly much broader. It includes soft skills that are notoriously challenging to measure. Crucially, it also includes financial literacy, a key skill for livelihoods in modern economies and for adult life in general that is often mainly acquired outside school.

The 2018 round of the Programme for International Student Assessment (PISA) survey, conducted by the Organisation for Economic Co-operation and Development (OECD), included an optional financial literacy module, which 20 participating education systems used. Data were collected on behaviours, attitudes and experiences, such as whether students had a bank account or had made payments online, as well as a cognitive assessment. The skills assessment was based on the Core Competencies Framework on Financial Literacy for Youth, developed by the OECD International Network on Financial Education. It concerned the 'basic level of financial literacy - in terms of knowledge, attitudes and skills - that is likely to be needed by young people between the ages of 15 and 18 to fully and safely participate in economic and financial life' (OECD, 2020, p. 43).

Notably, girls were less likely to report classroom activities related to financial topics in all participating countries (**Figure 13.4**), despite the fact that financial education is usually included in mathematics, generally a non-elective subject. It is unclear whether this finding reflects genuine differences in experience – for example, when groups or individuals chose assignment topics – or whether gender bias manifests in different recollection.

Not everyone has the opportunity to learn crucial financial concepts at school. Across participating countries, 35% of students, from 15% in Finland to 63% in Italy, reported never having encountered the notion of compound interest at school in the past year. Students in vocational education, on average, learn less about financial concepts. While 44% of students in general education had studied interest payments at school in the past 12 months and knew what the term meant, 38% of vocational students reported the same. Among 18 financial concepts, vocational students were less familiar with 12, including 'wage' and 'entrepreneur' (OECD 2020). 66

Not everyone has the opportunity to learn crucial financial concepts at schools

"

#### **FIGURE 13.4:**

#### Female students report less exposure to financial topics

Percentage of 15-year-old students who recalled encountering a discussion on 'the rights of consumers when dealing with financial institutions' sometimes or often in a school lesson over the previous 12 months, 2018



GEM StatLink: https://bit.ly/GEM2021\_fig13\_4 Source: GEM Report team analysis of 2018 PISA data. In Mexico, a free business literacy programme totalling 48 hours over 6 weeks had positive effects on rural female microentrepreneurs' profits, even 2.5 years later (Calderon et al., 2018). But evaluations of training to support livelihoods should take a broad view of the effects on participants' capabilities and well-being, rather than focusing narrowly on direct outcomes such as earnings (DeJaeghere et al., 2020). Financial education is particularly promising as part of multipronged interventions. A systematic review in low- and middle-income countries of financial education programmes for young people combined with sexual and reproductive health education showed that financial education could have a positive effect on HIV-related knowledge and attitudes, as well as risk-taking behaviour (Lee et al., 2020). This result stemmed not just from improved social conditions, but also from higher confidence, negotiating ability and self-efficacy.

# FOCUS 13.1: COMPUTATIONAL THINKING IS AN IMPORTANT COMPONENT OF DIGITAL LITERACY

Computational thinking, generally understood as solving problems through logical and algorithmic reasoning, is an important component of digital literacy and is included in the Digital Literacy Global Framework for SDG indicator 4.4.2 (Law et al., 2018). Although often referred to in computer-related settings, it is increasingly recognized as a cognitive process that can be developed and applied in various domains, with or without the use of digital devices.

The 2018 ICILS was the first cross-national assessment of student achievement in computational thinking. It tested students' ability to recognize which real-world problems were appropriate for computational formulation and to develop algorithmic solutions that could be used with a computer (Fraillon et al., 2019). The results highlighted digital divides between and within countries (**Figure 13.5**). Students who were from a privileged socioeconomic background, as represented by the number of books at home, and spoke the language of the test at home scored consistently and significantly higher than their peers.

Results indicated a gender gap in favour of boys, which contrasts with computer and information literacy, the other ICILS domain. Given the construction of the assessments, the contrasting gender patterns may reflect differences in attitudes towards ICT use. While girls are stronger users of ICT for general school-related tasks, boys tend to be better and more confident at performing specialist ICT tasks (e.g. creating programs) (Fraillon et al., 2019). Stronger male performance has also been found in national assessments with similar constructs (Román-González et al., 2017), but researchers who assessed computational skills that were applied differently, as in narrative-based computer games, found that girls outperformed boys (Howland and Good, 2015).



66

Growing recognition of computational thinking as a necessary tool has led several countries, particularly in eastern Asia and in Europe, to include it in national curricula

The growing recognition of computational thinking as a necessary tool for everyone has led several countries, particularly in eastern Asia and in Europe, to include it in national education curricula (Román-González et al., 2017; Seow et al., 2019). In the Republic of Korea, computational thinking practices, such as developing technology-based products, creating algorithms, and writing and evaluating code, are explicitly included in the curriculum (Fraillon et al., 2019). Finland has made algorithmic thinking and programming compulsory from grade 1 as a cross-curricular activity (Seow et al., 2019).

Integrating computational thinking in school must go beyond increasing ICT use. Across ICILS-participating countries, students who more frequently used ICT in school for school-related tasks did not necessarily score higher than their peers (Figure 13.6). Nor is it enough to teach children how to code. Although computer programming tends to be one of the main contexts and applications for computational thinking, research has shown that students with programming experience are not necessarily able to transfer those skills to non-programming environments. Instead, the focus has shifted towards teaching computational thinking as a transferable, higher-level concept applied in multiple domains (Voogt et al., 2015). Examples of non-digital applications of computational thinking in other subjects include identifying population trends in social studies and performing linguistic analysis of sentences (Yadav et al., 2014).

If students are expected to learn such skills in schools, teachers must be trained to teach them. Studies have shown that including computational thinking in teachers' pre-service training helps them gain a deeper understanding of it as a cognitive tool that can be applied more widely and across several disciplines (Mouza et al., 2017; Yadav et al., 2014). Yet when computational thinking is included in teacher training, it tends to be limited to computer science teachers (Yadav et al., 2014). This is reflected in practice. Across ICILS-participating countries, teachers in ICT-related classes reported putting more emphasis on teaching skills related to computational thinking than their peers in other subject areas (Fraillon et al., 2019).

#### **FIGURE 13.6:**

# Using computers is not necessary to develop computational thinking

Share of students ranked in the upper part of the computational thinking scale, by ICT use in school for school-related purposes, selected countries, 2018

99



GEM StatLink: https://bit.ly/GEM2021\_fig13\_6 Source: GEM Report team analysis of 2018 ICILS data.

# COVID-19

Demand for digital skills has soared since the pandemic began, making them a core priority of education, training and skills building. COVID-19 has accelerated the digital transformation in labour markets, with the digital economy expanding and the nature of work rapidly changing (ILO, 2020). In a survey of multinational and large national companies, 84% reported being ready to digitize work processes, potentially moving 44% of their workforce to remote operations (World Economic Forum, 2020).

COVID-19 has also led to increasing digitization of society more broadly, from social connections to product purchase and delivery, and even the fight against the pandemic itself. Digital skills have been key in gaining access to up-to-date information, booking medical appointments and using mobile applications for contact tracing (Binda, 2020). Digitization of public services has also gained speed. In Panama, where the renewal process for citizen IDs required two in-person visits, the government trialled an online platform for a remote renewal process and created an in-kind transfer that delivers digital vouchers linked to ID cards, thus turning them into debit cards for essential purchases during the pandemic (Reyes et al., 2021). Lockdowns and school closures made basic digital skills a prerequisite for learning and skills acquisition.

For some, the unprecedented increase in demand resulted in accelerated digital skills acquisition. The African Development Bank's Coding for Employment platform saw a 40% increase in users within a week after the pandemic hit (Doroba et al., 2020). The focus on improving digital skills can be even stronger among the unemployed. Analysis of the online learning platform Coursera indicates that while the increase in participation of employed learners focused on personal development courses, unemployed learners focused on acquiring new digital skills, such as data analysis, computer science and information technology (World Economic Forum, 2020).

But digital skills-building opportunities were not equally available for those lacking basic literacy or access to the internet and devices. In Nigeria, while private school teachers received devices and training to help them move towards remote teaching, public school teachers received only the schedules of learning programmes available on TV and radio (Azubuike, 2021). In refugee camps, such as that in Kakuma, Kenya, lockdowns halted programmes that delivered digital skills training to young people with no internet connectivity (ILO, 2020). Older adults, who are less likely to be digitally proficient, have also been hit particularly hard by the accelerated digitization of society. A survey of 17 European countries showed that even in high-income settings, less than half those aged 50 or older used the internet before the pandemic (Seifert, 2020). And those without basic skills were left further behind. A study of digital literacy among the elderly in Mexico found that those enrolled in basic levels did not have the necessary skills to continue with the training during the pandemic, leading to a dropout rate of nearly 80%. Those in more advanced levels were able to continue and effectively improve their skills (Martínez-Alcalá et al., 2021).

Governments are being called on to play a larger a role. Only 21% of businesses reporting to the World Economic Forum had access to public funds to support employee reskilling or upskilling (World Economic Forum, 2020). Improving digital skills and closing the digital divide may also require reforms in education and training systems. In 2020, South Africa's Department of Communications and Digital Technologies proposed reviewing the basic education curriculum to include computing, coding and a wide range of digital skills (BusinessTech, 2020). As part of the European Union (EU) Digital Education Action Plan (2021-2027), countries aim to establish common guidelines for teachers on fostering digital literacy, to collect data on student digital skills and to introduce an EU target for student digital competence (European Commission, 2021). Regulations and the coordination of global efforts to curb risks associated with the rush in digital upskilling, such as bullying, technology addiction and misinformation, have also gained new urgency (Jackman et al., 2021).

Non-state actors have stepped in as well. Over 30 million people enrolled in free online courses delivered by LinkedIn Learning, Microsoft Learn and GitHub Learning Lab – some of the most popular of which were related to the digital transformation (Smith, 2021). Microsoft also collaborated with local partners, such as the African banking group Ecobank, to provide small and medium-sized enterprises with the necessary digital skills to transition into the ever-more-digital world (Monteiro, 2021). Mastercard collaborated with USAID to launch Project Kirana in India, aimed at providing digital literacy skills to expand the access of female-owned shops to financial and digital services (Corneille, 2020). In Chile, a public-private partnership, Digital Talent for Chile, was launched in 2019 with the aim of training 16,000 disadvantaged people in skills related to the digital economy; it provided 3,000 scholarships for people who lost their jobs during the pandemic (Fundación Chile, 2020).

Areli, a third-grade teacher in Mexico, has participated in Save the Children-led workshops that have helped teachers be aware of children's emotional needs. Most children in this school have one or two close family members living in the United States.

CREDIT: Caroline Trutmann Marconi/Save the Children

# **KEY MESSAGES**

Cambodia, Congo, the Gambia, Ghana, Haiti, Malawi and Rwanda have, within a generation, come close to eliminating what used to be a large gender gap in upper secondary education completion.

Child deprivation is a strong predicter of education outcomes, over and beyond household wealth. Ghana is slightly wealthier than Bangladesh, but its share of 3- to 4-year-olds with no books or toys is five times higher.

Non-state armed groups adopt one of three approaches towards education provision in areas they control, each with its own equity implications: establishing their own system for educating civilians directly under their control; asserting some control over the system while leaving key elements in place; or leaving control of the education system to existing providers.

In western and central African countries, such as Chad, the Gambia and Togo, no more than 5% of children aged 7 to 14 speak the language of instruction at home.

Phone surveys of 19-year-olds during the COVID-19 pandemic showed that 70% of young women in Ethiopia vs 35% of young men spent more time than before the pandemic doing household chores, while 42% of young women in Peru vs 26% of young men spent more time looking after children.

The percentage of 6- to 14-year-old Syrian refugees in Lebanon attending education fell from 67% in 2020 to 53% in 2021. Among those offered remote learning, 20% could not attend.

# **CHAPTER 14**



target 4.5 Equity

By 2030, eliminate gender disparities in education and ensure equal access to all levels of education and vocational training for the vulnerable, including persons with disabilities, indigenous peoples and children in vulnerable situations

### **GLOBAL INDICATOR**

**4.5.1** – Parity indices (female/male, rural/urban, bottom/top wealth quintile and others such as disability status, indigenous peoples and conflict-affected, as data become available) for all education indicators on this list that can be disaggregated

#### THEMATIC INDICATORS

**4.5.2** – Percentage of students in primary education whose first or home language is the language of instruction

**4.5.3** – Extent to which explicit formula-based policies reallocate education resources to disadvantaged populations

- 4.5.4 Education expenditure per student by level of education and source of funding
- 4.5.5 Percentage of total aid to education allocated to least developed countries

arget 4.5 focuses on inequality in education. The primary measure of inequality in the SDG 4 monitoring framework is the parity index, which serves as global indicator 4.5.1, comparing the observed values of two groups on any indicator. The parity index is just one of several possible inequality measures, which vary in the aspects of inequality they capture best. Tracking progress is complicated by the fact that, even for a single measure such as the parity index, inequality can be analysed in terms of several different indicators (e.g. inequality in attendance, completion, learning), at the various education levels (e.g. inequality in primary, lower secondary, upper secondary education) and for a range of individual characteristics (e.g. inequality by sex, location, wealth). Combining indicators, levels and characteristics leads to a large number of possible permutations. There are also many ways of defining and measuring some of these characteristics, such as wealth, which is a proxy of socioeconomic status (Focus 14.1). Many other characteristics also affect education inequality, including conflict (Focus 14.2) and language (Focus 14.3), for which there are various ways of describing the context learners face.

The parity index highlights the relative situation of disadvantaged and privileged groups. Regarding urban-rural gaps in lower secondary completion, for example, it is clear that the parity index is lower in Ethiopia (0.20), as the gap in completion (12% in rural areas, 59% in urban areas) is larger than in the United Republic of Tanzania (0.35; 17% in rural areas, 48% in urban areas). But a higher value of the parity index can also be observed even if everyone is worse off. Parity is marginally higher in Uruguay (0.95) than in Ecuador (0.92), but completion rates are lower in the former than in the latter, in both urban and rural areas (**Figure 14.1**). This comparison highlights the risk in examining parity indices in isolation from the underlying indicator values.

Gender inequality remains a key concern, even if increasing nuance is required to understand the mosaic of challenges at different levels and in different places. Overall, girls' enrolment in education has improved dramatically over the past 25 years, with 180 million more girls enrolled in primary and secondary education, and impressive gains in closing the gap in completing primary and secondary education.

#### **FIGURE 14.1:**

**The parity index can be higher even if everyone is worse off in absolute terms** Location parity index and lower secondary completion rates by location, selected countries, 2018 or latest available year



Girls' average primary completion rate increased by almost 20 percentage points to 87%. The fastest progress was in Central and Southern Asia, where in 1995 just over half of girls completed primary school, but a generation later almost 90% of their daughters do so.

Where girls remain disadvantaged at the upper secondary level, they experience the largest gender gaps. There are three groups of countries: those with disparity at girls' expense throughout the past 25 years, those with disparity at boys' expense and those that switched from disparity at girls' expense to that of boys. In primary education, the majority of countries that had disparity at girls' expense have moved steadily towards parity. In upper secondary education, the majority of countries had disparity at boys' expense and little progress has been achieved (**Figure 14.2**). Overall, however, progress since 1995 has been in favour of girls. Where they were behind, they have been on average catching up. Where

## 66

Girls' average primary completion rate has increased by almost 20 percentage points over the past 25 years, to reach 87%

the direction of the gap changed, it has typically been in favour of girls. And where they already enjoyed an advantage, it increased on average further.

Gender parity in a region may be seen either in terms of the pooled population or by taking countries as the unit for analysis. The two perspectives may give different results. Viewing the entire population-weighted distribution of parity gives a fuller picture. Around half of all countries in sub-Saharan Africa have no, or only a slight, female disadvantage in gross secondary enrolment. However,

#### **FIGURE 14.2:**

**Girls have improved their relative chance of completion faster than boys, but still face a larger challenge in some countries** Adjusted gender parity index of the completion rate, by education level and type of disparity, 1995–2019



Notes: The colour of the line represents a country group, depending on the type of gender disparity in the past 25 years: countries with disparity at girls' expense, i.e. gender parity index below 1 (green); those with disparity at boys' expense, i.e. gender parity index above 1 (orange); and those that switched from disparity at girls' expense to disparity at boys' expense, i.e. the gender parity index changed from less than 1 to more than 1 (grey). The width of the lines represents the number of countries in the group (the more countries, the thicker the line).

GEM StatLink: https://bit.ly/GEM2021\_fig14\_2

Source: GEM Report team analysis of household survey data.

99

where female enrolment does remain lower, it is much lower, and in the countries with a female disadvantage, such as the Democratic Republic of the Congo, Ethiopia and Nigeria, on average have larger populations (**Figure 14.3**).

Upper secondary education is the level where adolescent girls may be severely disadvantaged (e.g. in Benin, Chad and Niger) but also likely to enjoy an advantage and a rapid shift in conditions to their favour. This is happening in a wide range of countries, including in those that are furthest from SDG 4 relative to their peers in specific regions. Haiti in the Caribbean, Cambodia in South-eastern Asia, the Gambia and Ghana in western Africa, Congo in central Africa, Rwanda in eastern Africa and Malawi in southern Africa have all come close within a generation to eliminating what used to be a large gender gap. Prior to the latest conflict, Yemen was also making rapid progress towards gender parity in upper secondary completion (**Figure 14.4**).

Prior to the Taliban takeover in August 2021, Afghanistan had made large absolute gains for both girls and boys, mirroring rapid progress elsewhere in South

#### FIGURE 14.3:

In half of countries in sub-Saharan Africa, girls are not at a disadvantage in secondary education Gender parity index for the secondary gross enrolment ratio, by cumulative secondary school-age population, sub-Saharan Africa, 2019 or latest available year



GEM StatLink: https://bit.ly/GEM2021\_fig14\_3 Source: UIS database.

#### **FIGURE 14.4:**

**Gender disparity at girls' expense in upper secondary completion is narrowing in all regions** Adjusted gender parity index of upper secondary completion, 1995–2019



GEM StatLink: https://bit.ly/GEM2021\_fig14\_4

Source: GEM Report team analysis of household survey data.

#### FIGURE 14.4 (CONTINUED):

**Gender disparity at girls' expense in upper secondary completion is narrowing in all regions** Adjusted gender parity index of upper secondary completion, 1995–2019



Prior to the Taliban takeover in August 2021, the completion rate for girls in Afghanistan had increased from 6% in 1995 to an estimated 50% in 2018

Asia. The completion rate for girls increased from 6% in 1995 to an estimated 50% in 2018, i.e. a position comparable to India in 1995. Gender parity in primary completion might have been reachable within a generation, but the latest developments suggest that the gains may have been fragile (**Figure 14.5**).

Parity, in the form of equal numbers of two groups, such as boys and girls, in classrooms or passing examinations, is a necessary condition for equitable education but not sufficient. Gender discrimination and other disadvantages run deep and manifest, for instance, in biased curricula and prejudiced everyday classroom interactions. Even with respect to outcomes reflecting individual choice, such as entering science, technology, engineering and mathematics fields in tertiary education, disparities should raise questions over the extent to which choices are freely made or constrained by societal norms and expectations.

#### **FIGURE 14.5:**

66

Impressive progress in gender parity in primary education was achieved in Southern Asia Primary education completion rate, by sex, and gender parity index, 1995–2019 100 Disparity at the expense of boys 2018 🔵 2017 2019 80 Disparity at the expense of girl Bangladesh Female primary completion rate (%) 60 2015 1995 2018 1995 40 Pakistar Adused genter party inter-Lie party • 1995 1995 20 Afghanistan Adjusted gender parity index = 0.5 1995 0 0 20 40 60 80 100 Male primary completion rate (%) GEM StatLink: https://bit.lu/GEM2021 fig14 5

Source: GEM Report team analysis of household survey data.

"

# FOCUS 14.1: HOUSEHOLD WEALTH DOES NOT CAPTURE ALL DIMENSIONS OF POVERTY

Reducing poverty is at the heart of the international community's efforts to leave no one behind in the 2030 Agenda. It can be conceptualized in numerous ways. International statistics on poverty to monitor SDG 1 are based on measures of income or consumption. But the detailed information needed to estimate poverty is often not available in the international multipurpose household surveys that are used to monitor progress in other SDGs, including SDG 4 on education. Instead, the surveys estimate a proxy measure for poverty based on household possessions and house amenities (Hannum et al., 2017). Using this wealth-related measure to disaggregate education indicators has limitations, which affect cross-country comparability.

Being poor in terms of wealth does not equate to being poor in terms of income. Wealth and income are expected to be positively correlated: Higher income allows households to save and accumulate wealth, and higher wealth can increase income through investments or easier access to credit. However, the relationship is often not that strong or clear. In the United States, for example, the correlation between total household income and net financial worth is positive, but relatively low (0.5); if only wage or salary income is considered, the correlation with net financial worth drops further (0.2) (Keister, 2018). Low correlation between wealth and income may mean that households spend their earnings and do not save or that they have low income but high wealth through, for example, inheritance.

The share of children who received the minimum number of meals in the past day can be a proxy for income, as opposed to accumulated wealth. In many countries, the share of children aged 6 to 23 months not receiving the minimum number of meals is about the same across wealth quintiles. In the Democratic Republic of the Congo, for example, the 66% of children not receiving the minimum for their age are almost evenly split among the five wealth quintiles (**Figure 14.6a**). By contrast, there is a considerable wealth gap in the Lao People's Democratic Republic: 40% of children in the poorest households in terms of the wealth index, but only 13% of their peers in the richest households, did not receive the minimum meals (**Figure 14.6b**).

Wealth, which tends to be measured at the household level, does not always capture child-specific deprivation. Although this measure is likely related to household members' living conditions, it does not directly

#### FIGURE 14.6:

#### Income and wealth are not always well correlated



Share of 6- to 23-month-olds who do not receive the minimum meal frequency, by wealth index quintile, 2017–18

Note: Minimum meal frequency, which is used in this figure as a proxy of income, is the percentage of children aged 6 to 23 months who received the recommended minimum number of solid/liquid feeds appropriate for their age the day before the survey.

GEM StatLink: https://bit.ly/GEM2021\_fig14\_6

Source: UNICEF global databases and calculations based on MICS data.

Wealth, which tends to be measured at the household level, does not always capture child-specific deprivation

translate to the individual level. This is particularly true for children, for whom resource allocation may be particularly unequal (Hannum et al., 2017). SDG target 1.2 is the first global poverty target explicitly mentioning children. Growing recognition that children's needs and living standards can differ from those of adults in the same household has led to development of measures of children-specific deprivation (Guio et al., 2018). In the EU Survey of Income and Living Conditions 2009 and 2014, ad hoc modules on deprivation contained questions on children-specific goods, such as shoes, clothes and books, as well as participation in social activities, such as celebrations, school trips and holidays (Eurostat, 2014).

Another example is UNICEF's Multidimensional Overlapping Deprivation Analysis, aimed at identifying and quantifying child poverty through various dimensions, including availability of books and toys. The share of deprived 3- to 4-year-olds varies considerably across low- and lower-middle-income countries and is not determined by the country's wealth. Ghana's GDP per capita, for example, is slightly higher than that of Bangladesh, but its share of 3- to 4-year-olds with no books or toys is nearly five times higher (33% vs 7%) (Figure 14.7a). Within countries, the prevalence of child deprivation is again not fully determined by household wealth. Regarding the population of deprived children, 78% live in the country's poorest households in Palestine, but only 27% in Chad. Indeed, in several countries, 10% of deprived children are in the richest households and over 30% of children in the poorest households are not deprived (Figure 14.7b).

As with household wealth, the level of child deprivation can be an additional strong predictor of education outcomes. In Togo, 3- to 4-year-olds who have no books or toys are significantly less likely to have attended early childhood education than those with at least one such item. About 40% of 3- to 4-year-olds in the country are deprived of both items, and there is no significant difference between their probability of attending early childhood education and that of children living in the 40% poorest households. But controlling for household wealth, being deprived of books and toys in Togo further decreases the probability of having attended early childhood education by four percentage points, on average across wealth levels. The effect is strongest for children in the richest households, where being deprived is associated with a decrease in probability of six percentage points (from 32% to 26%) (**Figure 14.8**). Together, different measures provide a fuller understanding of households' economic situation.

# FOCUS 14.2: MANY CHILDREN ATTEND REBEL-CONTROLLED SCHOOLS – SOMETIMES THEIR ENTIRE SCHOOL LIFE

Target 4.5 calls for monitoring the education of children in vulnerable situations, including those exposed to violent conflict.<sup>1</sup> To date, attention has focused on the number of out-of-school children in conflict-affected countries and on attacks on students, teachers and education facilities. Less attention has been paid to how education provision is itself affected by conflict, especially when it falls under the control of rebels, i.e. non-state armed groups.

Estimating the number of children attending schools controlled by non-state armed groups is a challenge due to data reliability issues. Moreover, rebels' territorial control frequently shifts, and their education provision, even in territory directly under their control, is rarely uniform but more like a patchwork of widely varying initiatives. In addition, for children and educators in settings of violent conflict, not only access to education but also the meaning attached to it is affected. For groups opposing the government, sustaining schooling outside government reach may serve as an act of resistance (Selvik, 2021).

Rebel governance refers to public service provision, institution building, norms and rules, and symbolic appeals developed by non-state armed groups in interactions with civilian populations and international actors (Branch and Mampilly, 2005; Kasfir, 2005; Mampilly, 2011). Non-state armed groups have numerous reasons for choosing to govern and provide public goods, especially education, to civilians; among these are the possibility of material benefits, risk of defection, their own ideological beliefs and, perhaps most importantly, pressure from local communities or international actors (Huang and Sullivan, 2020).

99

<sup>1</sup> This section is based on Mampilly (2021).

#### FIGURE 14.7:



- a. Share of 3- to 4-year-olds who are deprived, 2017–20
- b. Distribution of deprived children by household wealth quintile,

2017-20

Poorest Second Middle Fourth Richest D. R. Congo Sierra Leone Guinea-Bissau Τοαο Gambia Madagascar Ghana C. A. R. Lesotho Chad Zimbabwe Nepal Lao PDR Algeria Irao Tunisia Bangladesh Palestine Mongolia 50 20 10 40 30 Λ ٥ 20 40 60 80 100 % %

Note: The indicator measures the share of children with no toys (homemade or store bought) and with no children-specific or picture books at home.

GEM StatLink: https://bit.ly/GEM2021\_fig14\_7

Source: GEM Report team calculations based on MICS data.

Non-state armed groups may adopt one or more of three broad approaches. First, such a group may establish its own system for educating civilians directly under its control, especially if the group's leaders are suspicious of the existing system and define their stance in opposition to state-controlled schools. Al Shabaab, an Islamist insurgency in Somalia, is an example. It has set up schools on its territories, which emphasize religious education and are viewed primarily as a source of recruits for the insurgency (Hiraal Institute, 2018). Other groups view control over curricula as an opportunity to socialize youth in ways perceived as benefiting their larger struggle (Palmer, 2020).

But direct control of an education system can tie up significant material and human resources while failing to generate community support. Non-state armed groups may seek to assert some control over the system while leaving key elements of it in place, whether curriculum, teaching and administrative staff, or infrastructure. While it means sacrificing some control, this requires far fewer resources. In Colombia, FARC concentrated its

#### **FIGURE 14.8:**



A child deprived of books and toys has a lower chance of attending preschool even after controlling for household wealth Probability of having attended early childhood education for children aged 3 to 4, by household wealth and child deprivation, Togo, 2017

education activities on recruits and cadres rather than the general population (Arjona, 2015). Some rebel groups prefer to leave national curricula unaffected or adopt those of neighbouring countries.

Finally, some non-state armed groups may choose to leave control of, and accountability for, the education system in their territories to existing providers, whether the government, religious organizations, private actors or local community boards, whether tacitly or by striking agreements. However, armed groups may attack such schools and their personnel as a perceived threat to their authority.

Examples illustrate some of the tensions these options entail. The Liberation Tigers of Tamil Eelam (LTTE) controlled large portions of the former North Eastern province during the 26-year-long Sri Lankan civil war, especially from 1991 to 2005, when they established a complex governance system. Under pressure from local

## 66 -

Non-state armed groups may provide education for material benefits, risk of defection, their own ideological beliefs and pressure from local communities or international actors

"

communities whom they relied on for support, the rebels struck a complex agreement with the government to ensure education provision in the areas under their control. An education council coordinated between the government education ministry and local communities, providing access to rebel territory for government-paid teachers. Schools retained a version of the national curriculum supplemented by more ideological material provided by LTTE and students continued taking national government examinations (Mampilly, 2011). Even at the high point of LTTE territorial control in 2002, the government retained a large presence in urban areas with LTTE controlling more remote rural areas. In 2003, it was estimated that there were 648,000 school-age children in the North Eastern province, with a dropout rate of 15%, four times the national average but still low considering the intensity of the violence (NRC, 2005).

For the resource-poor Sudan People's Liberation Movement (SPLM), creating an education system from scratch was beyond its capacity, especially when the collapse of the Derg regime in Ethiopia in 1987 and the closing of refugee camps there triggered a massive influx of civilians into territories the SPLM controlled. Instead, the organization reached out to non-governmental organizations (NGOs) and international agencies. NGOs were required to sign a memorandum of understanding, pledging to follow SPLM rules on where they could open schools and which populations would be eligible to participate. Generally, schools were expected to adhere to the Kenyan or Ugandan curricula, not the Sudanese one. Local communities working with NGOs and overseen by non-military SPLM personnel determined the form of education provision in rebel territories (Mampillu, 2011). Disputed census counts and large refugee movements led to much uncertainty around population levels in South Sudan during the war. Around 2000, the number of primary school-age children was estimated at 1.06 million, with only 318,000 enrolled (Deng, 2003).

An important question is how to promote engagement between state and non-state actors for co-ordination in service delivery. With support from the Myanmar Education Consortium and the World Bank, the Myanmar Education Partnerships Project was initiated in 2018 to overcome a legacy of distrust and promote dialogue between the Ministry of Education and Ethnic Basic Education Providers (EBEPs), non-state actors providing education services to over 420.000 children (Myanmar Ministry of Education, 2020). Through a series of inclusive, participatory and conflict-sensitive workshops, the initiative sought to facilitate, advise and support dialogue for a partnership framework and associated roadmaps towards collaboration between the ministry and two EBEPs: the Karen Education and Culture Department and the Mon National Education Committee. Through the initiative, the role of EBEPs in education provision in Myanmar would be recognized and the needs of children they served reflected in national planning. Dialogue focused on development of structures for accountable coordination and communication, and quality standards to support recognition of teachers, curriculum and student learning in EBEP-administered schools.

66

Nuanced if potentially controversial engagement with non-state armed groups by governments and the international community can improve education opportunities

99

In Afghanistan, after the Taliban's fall from power in 2001, they started opening schools in areas under their control in 2007 due to local community pressure, reflecting the presence of pragmatic voices alongside traditionalist hardliners (Jackson, 2018). They vetted teachers and controlled the curriculum, excluding textbooks on the Afghan culture, constitution and law (Amiri and Jackson, 2021). They established education commissions in Quetta in 2006 and Peshawar in 2007 (Jackson and Amiri, 2019). They also began prohibiting attacks against government-run schools, which had led to the closing of some 4,000 schools during the first 15 years of the Taliban movement. In 2010, they even began negotiations with the education ministry to develop a framework for reopening schools (Giustozzi and Franco, 2011, 2013). In December 2020, a few months before assuming control over the entire country following the US troop withdrawal, the Taliban had reached agreement with the UN to establish community-based education in the four provinces they controlled: Faryab, Helmand, Kandahar and Uruzgan. The programme was intended to increase the number of such schools to 4,000 and reach up to 140,000 children (UNICEF, 2020a).

The example of the Taliban shows that non-state armed groups alone do not determine the nature and scope of education provision. Rather, they face both opportunities and constraints from a wide variety of actors. Education is among the most prized services demanded by civilians; failure to provide it can lead to resentment from local communities.

Even during the darkest days of Taliban rule, local commanders faced pressure from civilian communities, resulting in schools operating in open defiance of the central committee's rules (Feroz and Lakanwal, 2020). Nuanced if potentially controversial engagement with non-state armed groups by governments and the international community can improve education opportunities. While such engagement clearly poses some risk, the benefits for school-age populations may be worth it – especially where whole generations of children would otherwise be left out of education. Indicator 4.5.2, the percentage of students in primary education whose first or home language is the language of instruction, recognizes the challenges of linguistic diversity in education. Many education systems favour the use of national or colonial languages, despite the benefits of local languages as documented by the theory and practice of multilingual learning. Experiencing a language disconnect has detrimental social and cognitive consequences for learners (Adamson, 2020). Home language instruction, especially in the early years of schooling, contributes to cognitive, identity and cultural development and provides a strong foundation for learning of other languages and for use of mathematics (Outhwaite et al., 2020).

Usage of 'first language' and 'home language' is ambiguous and it is not clear whether the terms can be considered interchangeable. Intuitively, home language is the language spoken in the community or household, but multiple languages can be spoken in the same household. The Technical Cooperation Group on SDG 4 Indicators defined 'home language' as the 'most used language at home' and 'language of instruction' as the 'language used most frequently in teaching in the classroom' (UIS, 2018).

However, classroom reality cannot be neatly captured by indicator 4.5.2. For instance, a common, spontaneous, teacher-specific classroom language practice is code-switching or alternating between languages. Some education systems transition from one language to another at a given grade. In such transitional approaches, the use of home language in instruction may be over- or underestimated if the point of measurement is a lower or upper primary school grade, affecting data interpretation. Focusing on the first four years of primary school would be recommended, as that is when the pedagogical advantage of home language instruction is greatest.

To maintain the positive effects of teaching and learning in the home language, the child's home language needs to be sufficiently developed before any transition to second language instruction. In Ghana, during the 2016/17 school year, the 40,000 children who enrolled in the mother tongue-based Complementary Basic Education programme and transitioned to government schools faced different literacy trajectories. Those who transitioned to government schools where the language of instruction was English lost some of the skills acquired through the accelerated mother tongue programme and hence did not perform as well in literacy as their peers who continued their education in their home language (Carter et al., 2020).

Political concerns further complicate practice. In India, the three-language formula, an ambitious language policy adopted in 1968 and reaffirmed in the 2020 National Education Policy, calls on all students to learn three languages in school, at least two of which should be native to India. However, the policy has yielded only limited results. The percentage of Indians who could speak 2 of the 22 scheduled languages increased from 12% in 1971 to 25% in 2011, with large differences among states. In Uttar Pradesh, the percentage increased from 24% to 33%, while in Tamil Nadu it remained constant at 2%. Measuring progress towards indicator 4.5.2 needs to take into account policy factors that affect choices of language as compulsory school subjects (Bhattacharya and Chandrasekhar, 2020).

A guestion in the Foundational Learning Skills module, introduced in the sixth round of the UNICEF Multiple Indicator Cluster Surveys (MICS), provides rich insights. Children are asked whether the language their teachers use in class is the one they speak at home (Hattori et al., 2017). In western and central African countries, such as Chad, the Gambia and Togo, no more than 5% of children aged 7 to 14 speak the language of instruction at home (Figure 14.9). In many countries, wealth is negatively associated with the probability that a child speaks the language of instruction. This may be because poorer children tend to live in linguistically homogeneous rural areas where teachers use the local language, while richer children tend to live in linguistically fragmented urban areas where teachers use a national language (Figure 14.10a). In some countries, the percentage of children who speak the language of instruction at home falls between primary and lower secondary school, usually reflecting national policy. For instance, Kiribati adopted a language policy in 2012 for children to transition from te-Kiribati to English in grade 3 (Kiribati Government, 2016) (Figure 14.10b).

#### FIGURE 14.9:

In many sub-Saharan African countries, only a fraction of students speak the language of instruction at home Percentage of children aged 7 to 14 who at home use the language used by teachers at school, selected low- and middle-income countries, 2017–20



**99** 

# 66

In western and central African countries, such as Chad, the Gambia and Togo, no more than 5% of children aged 7 to 14 speak the language of instruction at home

Language policy information is thus needed to complement and interpret data on language of instruction experiences, such as those MICS provides. Information on language policy can also be used to provide an approximate picture of the linguistic reality students face when data on language of instruction experiences are missing, if combined with other sources, such as Ethnologue: Languages of the World (SIL International, 2021), which not only estimates the population speaking a language but also lists the languages used for education in each country, along with school-age population estimates and enrolment rates. Such a combination of Ethnologue data and language policy information was used to estimate the situation in 21 eastern and southern African countries, where the percentage of children taught in their home language ranged from zero in Angola, Djibouti, Comoros, Mozambique, Rwanda and the United Republic of Tanzania to 96% in Ethiopia. Many countries with a home language policy reach a majority of children, although policies in Uganda and especially Kenya are narrowly focused on a limited number of languages (Wekundah and Trudell, 2021). When compared with MICS data, this approach produced similar results in Zimbabwe but overestimated the situation in Lesotho and underestimated that in Madagascar (**Figure 14.11**).

A similar approach led to a recent estimate that 37% of children in low- and middle-income countries learned in a language other than their home language: 27% spoke a minority written language and 10% a less common language, each with relatively few speakers, which education systems would struggle to address (World Bank, 2021).

#### **FIGURE 14.10:**

A variety of factors are related to the probability that the language of instruction is also the children's home language Percentage of children aged 7 to 14 who at home use the language used by teachers at school, by characteristic, selected low- and middle-income countries, 2017–20

a. By wealth







#### **FIGURE 14.11:**

More than a quarter of the 21 countries in eastern and southern Africa do not offer home language education Percentage of primary school-age children taught in their home languages, estimates by two different methodologies, eastern and southern African countries, 2021



Note: The data for Lesotho, Madgascar and Zimbabwe refer to children aged 7 to 14.

GEM StatLink: https://bit.ly/GEM2021\_fig14\_11

Sources: Wekundah and Trudell (2021) on policy, linguistic, demographic and enrolment data; MICS Survey Findings reports for Lesotho, Madgascar and Zimbabwe.

# COVID-19

The most serious legacy of COVID-19 on education has been its disproportionate impact on disadvantaged learners. First, there was wide inequality between countries in access to learning. In terms of policy, schools in high-income countries were fully closed for 21% of total instruction days, compared with 31% in low- and middle-income countries. Online platforms, which ensure learning continuity better than radio and television, were used in more high-income countries (96%) than middle-(92%) and low-income ones (58%) (Figure 14.12a). In terms of household living conditions, online learning was more possible for students in richer than in poorer countries; at best, 6% of students in Africa would have been able to attend classes online and about 30% on television (Figure 14.12b). The Remote Learning Readiness Index, combining the policy and household dimensions with an emergency preparedness dimension in 67 low- and middle-income countries, estimated that more than 200 million students lived in the 31 countries ranked as least prepared to deliver remote learning (UNICEF 2021b).

Second, there was considerable inequality in access within countries (Human Rights Watch, 2021). Even in those with near universal broadband access, such as the Netherlands, disadvantaged students suffered from worse home conditions (having their own space to study), lower access to devices, less home support (parents directly helping, checking homework, contacting teachers and receiving information from school) and less school support (as they attended schools with fewer resources) (Bol, 2020). In the United States, 37% of parents with an income under US\$30,000 reported that their children had to do their homework on smartphones, compared with 16% of parents with an income over US\$75,000, who likely had access to computers. Moreover, 23% of the poorest, but just 4% of the richest, had to use public Wi-Fi (Schaeffer, 2021).

Such inequality translates into unequal consequences for learning (**Chapter 10**). In the US state of California, analysis of learning assessments in English for students in grades 4 to 8 estimated that the learning lag was 3.8 months for English learners but 2.3 months for

#### **FIGURE 14.12:**

#### Poorer countries have far less access to interactive forms of remote learning

a. Percentage of countries offering remote learning modalities for at least one education level, by income group, 2021



b. Percentage of students potentially reached by different remote learning modalities, by region, 2020



English native speakers, and 3.3 months for poor students but 1.2 months for the non-poor (Pier et al., 2021). Analysis of remote learning's impact on grade 3 to 8 students' examination pass rates in 12 states showed that moving from in-person to fully hybrid or virtual mode exacerbated the negative impact by an average of 10 percentage points in mathematics and four percentage points in English. For a district with no Black or Hispanic students, the switch to fully hybrid or virtual mode lowered pass rates by four percentage points; for a district with a 50% Black and Hispanic student population, the effect amounted to nine percentage points (Halloran et al., 2021).

Recovery may also be inequitable. Only one in four countries is providing incentives, such as cash, food, transport or fee waivers, to help girls or children from disadvantaged families return to school (UNESCO et al., 2021).

Boys and girls may not face the same consequences in terms of access to devices, time use and early pregnancy risks. Some parents in Bangladesh, Jordan and Pakistan expressed reluctance to give girls access to smartphones (UNESCO, 2021). Phone surveys of 19-year-olds during the pandemic showed that 61% of young women in India (vs 35% of young men) and 70% in Ethiopia (vs 35%) spent more time than before the pandemic doing household chores. Likewise, 46% of young women in Viet Nam (vs 34% of young men) and 42% in Peru (vs 26%) spent more time looking after children (Ford, 2021). Few studies have compared early pregnancies before and during the pandemic. Checking antenatal visit records from clinics is thwarted by the estimated 39% decline in such visits due to lockdowns and related restrictions (Townsend et al., 2021). In a Kenyan study, teenage girls' antenatal visits fell by 16% between March 2020 and February 2021 compared with the year before the pandemic (UNESCO. 2021). A related concern is the increase in incidence of gender-related domestic violence reported through hotlines worldwide (Viero et al., 2021).

Remote learning arrangements often left learners with disabilities unsupported. A global survey of parents of children with disabilities found that only 19% of those in need had access to sign language interpretation, yet remote learning programmes were often not available in sign languages. Only 23% had access to transcripts of audio sessions (World Bank and IEI, 2020). A survey in 37 countries found that 38% of parents of children with disabilities were unable to help their children learn, as opposed to 28% of parents of children without disabilities (Save the Children, 2021). Carers' ability to earn and provide for families with children with disabilities was also severely affected (Mbazzi et al., 2020). Finally, the mental toll of school closures, deeply felt by students all over the world, was particularly strong for learners with disabilities. In the United States, the number of university students with disabilities reporting major depressive disorders was roughly twice that of students without disabilities (Sutton, 2021).

Migrant and refugee communities have been severely affected. A survey of Indian domestic migrants found that school closures increased the number of children forced to accompany parents to brick kiln work sites by 67% (Aide et Action, 2021). In some cases, this was because the seasonal hostels set up to support these children had been forced to close (Suffian, 2021). Refugee children were affected in various ways. As learning centres for the Rohingya in Bangladesh remained closed for 18 months, some humanitarian organizations switched to caregiver-led approaches with small groups of up to five learners (Australian Humanitarian Partnership, 2021). Enrolment in Save the Children's programmes in the northern Syrian Arab Republic fell by 30% (Save the Children, 2020). A vulnerability assessment of Syrian refugees in Lebanon, a country that has been affected by multiple economic, social and political crises, found that just the percentage of 6- to 14-year-olds attending education fell from 67% in 2020 to 53% in 2021. and 20% of those offered remote learning were not able to follow it (UNICEF et al., 2021).

Yao language supervisor Martha Makalani from the Literacy for Women in Africa programme monitors a women's literacy class in Malawi, allowing women to learn basic literacy in their mother tongue.

CREDIT: Ari Vitikainen/ Finnish Bible Society.

# **KEY MESSAGES**

In sub-Saharan Africa, female youth literacy increased by less than one percentage point per year between 2015 and 2020. Over a quarter of young women in the region are illiterate.

Global improvement has stagnated: A substantial decline in the absolute number of people with low literacy skills in Eastern and South-eastern Asia to 52 million was offset by an increase in sub-Saharan Africa to over 127 million.

Literacy is overestimated if it is not directly assessed. Almost half of all 20- to 24-year-olds who had completed lower secondary school in 18 low- and lower-middle-income countries could not read a simple sentence.

New adult literacy assessment data from Germany and the United States suggest that literacy skills are not declining, as previously thought. But there are chronic literacy challenges in the United States: More than one in three adults fall below minimum proficiency in 295, or almost 1 in 10, counties.

Analysis of survey and census data on cohorts born between the 1950s and 1980s in 42 sub-Saharan African countries shows basic numeracy skills have stagnated among the poorest adults.

Even before the COVID-19 pandemic, distance education was an unpopular mode of delivery for initial literacy programmes. In Brazil, a 2021 regulation on second-chance programmes for youth and adults clarified that classes corresponding to the primary curriculum had to be delivered in person.

# **CHAPTER 15**



# **TARGET 4.6**

# Literacy and numeracy

By 2030, ensure that all youth and a substantial proportion of adults, both men and women, achieve literacy and numeracy

### **GLOBAL INDICATOR**

**4.6.1** – Percentage of population in a given age group achieving at least a fixed level of proficiency in functional (a) literacy and (b) numeracy skills, by sex

## THEMATIC INDICATORS

- 4.6.2 Youth/adult literacy rate
- 4.6.3 Participation rate of illiterate youth/adults in literacy programmes

A oving beyond the conventional binary opposition between literacy and illiteracy, global indicator 4.6.1 aims to capture a range of proficiency levels in both literacy and numeracy. However, no new data have been reported for global indicator 4.6.1 since 2017. Methodological progress in measuring literacy at multiple proficiency levels (Grotlüschen et al., 2020b) has not yet led to new data collection efforts since the Programme for the International Assessment of Adult Competencies (PIAAC), run by the Organisation for Economic Co-operation and Development (OECD), and the World Bank's Skills towards Employment and Productivity (STEP) surveys.

These two programmes thus remain key sources to understand the state of literacy proficiency. PIAAC's first cycle collected data in 39 mostly high-income countries between 2011 and 2017. The second cycle is scheduled for 2022–23, with results expected in 2024. The STEP surveys adopted the PIAAC scale and administered assessments mostly in urban areas of 17 middle-income countries between 2012 and 2017.

On the PIAAC scale, minimum literacy proficiency means the ability to match text and information and/or paraphrase or make low-level inferences given competing information. This is a higher level of proficiency than that commonly associated with the adult literacy rate, which is thematic indicator 4.6.2, a binary categorization of literacy that is either self-assessed ('Can you read and write?') or directly assessed through a simple sentence reading test. Such data are, however, available for longer periods and most countries.

# 66

The literacy gender gap is 2.1 percentage points among people aged 15 to 24 and 11.5 points among those aged 65 and above

99

Globally, among adults aged 15 and above, 83% of women and 90% of men are literate, a 7 percentage point gap. The gender gap is 2.1 percentage points among people aged 15 to 24 and 11.5 points among those aged 65 and above. Progress over time remains slow. In low-income countries and in sub-Saharan Africa, female youth literacy has increased by less than one percentage point per year. More than one in four young women in sub-Saharan Africa are illiterate (**Table 15.1**).

With improvements in youth literacy outpacing those among older adults, literacy programming will increasingly need to find ways to reach the elderly. Older people make up an increasingly large part of the population in every region except sub-Saharan Africa. In Northern Africa and Western Asia, their share went from 4.7% in 2000 to 5.8% in 2020 and is expected to reach 7.6% by 2030; in Central and Southern Asia, the equivalent figures are 4.3%, 6.1% and 8%. In Nepal, a pilot literacy class for older people, with an 8-month curriculum of 332 hours of in-class instruction, uncovered specific cognitive challenges, such as poor memory, and the need for learning and teaching materials to be tailored to older learners (Ayyappan, 2020).

#### **TABLE 15.1:**

Youth, adult and elderly literacy rates, 2015 and 2019

	Youth (15–24) literacy rate (%)				Adult (15+) literacy rate (%)				Elderly (65+) literacy rate (%)			
	Female		Male		Female		Male		Female		Male	
	2015	2020	2015	2020	2015	2020	2015	2020	2015	2020	2015	2020
World	89.4	90.8	92.5	92.9	81.8	83.3	89.3	90.1	68.5	72.5	81.2	84.0
Cul Calance Africa	70.0	740	70.0	70 5	563	F0 C	71 5	72.0	245	27.0	40.0	50.0
Sud-Sanaran Africa	70.6	74.2	/8.6	/9.5	56.3	59.6	/1.5	72.8	24.5	27.0	48.0	50.6
Northern Africa and Western Asia	87.4	87.5	91.5	90.9	74.1	74.9	86.4	85.8	42.1	45.0	65.5	66.4
Central and Southern Asia	85.2	89.6	90.2	92.4	63.0	67.7	79.5	82.4	28.8	33.0	56.1	61.6
Eastern and South-eastern Asia	98.8	99.0	98.8	99.0	93.4	94.6	97.2	97.6	71.4	79.1	88.6	92.0
Latin America and the Caribbean	98.5	98.8	98.1	98.5	92.7	94.1	93.6	94.9	77.0	81.1	81.4	84.9
Oceania												
Europe and Northern America												
Low income	66.1	70.7	75.4	76.3	50.2	54.1	67.3	68.9	26.7	27.9	47.0	49.2
Lower middle income	85.0	88.2	89.8	91.3	66.3	70.0	80.8	82.9	37.5	40.9	59.9	64.0
Upper middle income	98.5	98.5	98.6	98.6	93.5	94.5	96.7	97.1	74.9	80.9	87.8	91.1
High income												

Source: UIS database.

Slow progress in raising literacy rates means that, in absolute terms, the number of people with no or low literacy skills, especially women, has hardly changed. A substantial decline in Eastern and South-eastern Asia to 52 million illiterate women was offset by an increase in sub-Saharan Africa to over 127 million (**Figure 15.1**).

As literacy measurement shifts from self-reporting to direct assessment, literacy statistics are likely to worsen. For decades, census and household survey data routinely recorded as literate anyone who had completed primary school, without questioning their actual literacy status, much less testing it. However, increasing attention to poor learning outcomes at school has shown that many children and adolescents do not acquire minimum proficiency in reading, even by the end of lower secondary school. The most recent Demographic and Health Surveys (DHS) reflect this more critical understanding that schooling does not necessarily lead to literacy skills acquisition. While still limited to relatively simple literacy tests for operational reasons, the surveys now apply them to all respondents regardless of schooling level.

Results confirmed that assuming all secondary school leavers were literate meant the true literacy level had previously been overestimated. While indicator 4.6.2 defines the youth literacy rate over ages 15 to 24, focusing on those aged 20 to 24 would be more appropriate to fully capture the effect of completed secondary schooling. Almost half of lower secondary completers in the 18 countries with recent DHS data do not reach the basic level of literacy defined as being able to read a simple sentence. Even among upper secondary school completers, 1 in 6 young adults remains illiterate; in Benin, just 64% of upper secondary school graduates could read a simple sentence. Conversely, though, never having been to school need not be associated with illiteracy: 3 in 10 of all young adults who have not been to school - and as many as 1 in 2 in Indonesia and Myanmar - could read (Figure 15.2).

#### **FIGURE 15.1:**

The overall number of illiterate women has hardly changed in 20 years, and in sub-Saharan Africa the number has increased Number of illiterate women aged 15 and above, by region, 1999 and 2019



## 66

Assuming all secondary school leavers are literate means the true literacy level had previously been overestimated

**99** 

FIGURE 15.2:





GEM StatLink: https://bit.ly/GEM2021\_fig15\_2 Source: GEM Report team analysis of DHS data

While the change from assuming lower secondary graduates to be literate to measuring their literacy improves understanding of the true level of literacy, it means comparisons over time can be made only for those with less than secondary schooling. Progress for this group has been slow or non-existent in many countries. For instance, between 2012 and 2018, among people who had never gone to school or had left education before secondary school, the proportion of those who could read a simple sentence increased by five percentage points in Mali but declined by eight in Guinea (Figure 15.3). As long as less than half of all young people in low-income countries complete lower secondary school, achieving substantial increases in youth literacy will depend crucially on the literacy skills these early school leavers acquire.

The fact that no schooling does not equal illiteracy highlights the importance of acquiring literacy outside school. Although indicator 4.6.3 focuses on participation of illiterate youth and adults in literacy programmes, international data for this indicator have not been available. In principle, targeted adult literacy programmes may be distinguished from adult education programmes that aim to confer a primary education equivalent qualification. However, the time it takes to acquire lasting basic literacy is significant, and longitudinal studies regularly confirm the need for follow-up programmes to maintain newly gained literacy skills (Lang, 2021). In the spirit of SDG 4, literacy training should ideally be only a first step on the adult education ladder. In the absence of current data on participation in literacy programmes, adult education programmes at the basic level are worth examining as a proxy.

For a number of countries, enrolment data at the primary education level (or the first level in the International Standard Classification of Education, ISCED 1) can be disaggregated into enrolment in formal initial primary education and other ISCED 1 programmes, and by age. Where reported, enrolment of individuals aged 20 and above in other than formal initial primary ISCED 1 programmes may indicate formal basic adult education. Such enrolment reaches over 56,000 individuals in Colombia and 100,000 in Thailand. In relation to the estimated illiterate population, adult enrolment in other ISCED 1 programmes is 1% or less in the Plurinational State of Bolivia, Honduras, Mozambigue, Qatar and Suriname, 2% in Bahrain and Peru, 3% in Colombia and Thailand, 4% in Saudi Arabia and 8% in the Dominican Republic.

In Costa Rica, the corresponding figure is 28%, suggesting adult basic education programmes reach many more people than those with low literacy skills. The Ministry of Public Education maintains a network

**FIGURE 15.3:** 



In most countries, literacy is at best stagnating among those who do not complete at least lower secondary schooling Youth literacy rate among those with less than secondary education, 2010–14 and 2015–19

of adult education centres (Centros Integrados de Educación de Adultos) and night schools (Colegios Académicos Nocturnos) for young people and adults who did not finish primary or secondary education. Nevertheless, participation of illiterate individuals in literacy programmes is high, reflecting a large-scale literacy initiative, the Plan Nacional de Alfabetización (PLANALFA). Launched in 1998, PLANALFA sought to halve illiteracy by 2015 and eradicate it by 2025. Enrolment in literacy programmes was 7,300 in 2012 (Zúñiga et al., 2015), or 8.4% of the country's 87,000 illiterate people. The UNESCO Institute for Statistics estimated the number of illiterate adults at 84,000 in 2018, so little changed in nearly a decade.

Indirect evidence is available for India, where suspension of formal adult education programmes at the primary level in 2018 resulted in a drop by almost 23 million in total primary enrolment, representing 9% of the estimated 253 million illiterate adults.

Brazil has more complete data, as the annual national household sample survey, Pesquisa Nacional por Amostra de Domicílios (PNAD), captures illiterate adult participation in three types of programmes: regular basic schooling, second-chance basic schooling and adult literacy programmes. The percentage of illiterate adults enrolled as share of all illiterate adults fell between 2012 and 2019 from 2.7% to 1.4% among women and from 2.2% to 0.7% among men. But this is the lower bound of participation: Some who were illiterate at the start of the programme may have gained literacy skills by the time of the survey. Measuring enrolment of currently illiterate adults may thus underestimate these programmes' contribution to reducing illiteracy. Conversely, measuring enrolment of all adults in the programmes will overestimate the participation of those who were illiterate. These two estimates provide lower and upper bounds for the rate at which illiterate individuals participate in adult education programmes (Figure 15.4). The evidence suggests that such participation has most likely declined and at this pace it would be inadequate to help achieve universal literacy by 2030.

66

The fact that no schooling does not equal illiteracy highlights the importance of acquiring literacy outside school

**99** 

#### FIGURE 15.4:

The percentage of illiterate adults participating in literacy programmes has been decreasing in Brazil Illiterate adults/All adults enrolled in adult literacy programmes as share of illiterate adults, by sex, Brazil, 2012–19



Notes: Adults are defined as those aged 25 or above. Literacy programmes for adults include regular basic schooling, second-chance basic schooling and adult education programmes. GEM StatLink: https://bit.ly/GEM2021\_fig15\_4

Source: GEM Report team analysis of PNAD.

In recent years, comparisons of PIAAC results with those of the Adult Literacy and the Life Skills Survey in 2003–07 and the International Adult Literacy Survey in 1994–98 suggested that literacy skills might be declining in high-income countries (Desjardins, 2020). Newer trend evidence for Germany and the United States casts some doubt on this suggestion.

Two rounds of the German Literacy Survey show a positive trend (Grotlüschen et al., 2020a). The proportion of adults aged 18 to 64 with low literacy skills at levels 1 to 3 declined from 14.5% in 2010 to 12.1% in 2018. Meanwhile, the percentage of proficient readers above level 4 increased from 60% to 68%. But additional analysis shows these trends to be driven largely by compositional change in the population, particularly increased education attainment. Analysis of longitudinal data from three additional waves of a national extension of PIAAC showed little average gain in literacy skills at the individual level, although the average hides literacy declines at older ages that is offset by literacy gains at younger ages (Reder et al., 2020).

In the United States, the overall percentage of adults aged 16 to 65 who did not reach minimum proficiency level 2 in literacy skills increased marginally, from 18% in 2012–14 to 19% in 2017 (NCES, 2020a). The decline was greater among younger groups, and literacy actually improved for those with less than secondary school education. However, almost all these changes are within the margin of error. Hence evidence from the two countries does not suggest that literacy skills are declining, whether due to changing work cultures that require less engagement with continuous written text or other reasons.

Within the United States, great variation in community literacy can be observed. The US Skills Map (NCES, 2020b) has used its larger pooled sample to perform small area estimation to generate skills profiles for each of 3,142 counties (Krenzke et al., 2020). Even within a single state, such as Texas, the percentage of adults below minimum literacy proficiency ranges from 9% in Borden County to 70% in Kenedy County. Across the United States, more than one in three adults have below minimum proficiency in 295 counties.

Target 4.6 refers to both literacy and numeracy. Adult numeracy is a multifaceted set of skills that goes well beyond school-type arithmetic problems. The five domains of numeracy – civic, digital, financial and commercial, health and workplace (UIL, 2020) – highlight connections with SDG targets 4.4, 4.7 and beyond. As with literacy, numeracy in these everyday domains is also a

# 66 -

# Even simple numeracy skills data are scarce

social practice. For example, adults with low numeracy skills may apply various strategies, such as counting pills to adhere to treatment protocols (Chamberlin, 2019).

99

However, even simple numeracy skills data are scarce. Aside from the PIAAC surveys, there is no source of internationally comparable adult numeracy skills. Unlike for literacy, where basic data for some countries date back over a hundred years, no directly measured numeracy skills time series exist. Nevertheless, indirect measures can provide a sense of inequality and numeracy trends over time (**Focus 15.1**).

# FOCUS 15.1: BASIC NUMERACY SKILLS HAVE STAGNATED AMONG AFRICA'S POOREST FOR DECADES

Numeracy is important. Calculating numbers and proportions is crucial for many activities in production and trade, and even in household activities and agriculture (Tollnek and Baten, 2017).<sup>1</sup> Health-related activities similarly require numerical judgements about proportions and the exact use of time. Overall, numerical abilities have a positive impact on economic growth, and also on the economic success of migrants in major destination countries (Hanushek and Woessmann, 2012). Despite its importance, numeracy has been evaluated much less than literacy, not least because of data scarcity (Lilenstein, 2018; Pritchett, 2013). This is especially true for sub-Saharan Africa and for disadvantaged populations.

Learning assessments such as the Programme d'analyse des systèmes éducatifs de la CONFEMEN (PASEC) have filled data gaps on numeracy of schoolchildren, although they have been limited by changes in countries participating and measurement methodology over the years (Dickerson et al., 2015). Most recently, the UNICEF Multiple Indicator Cluster Surveys (MICS) have included foundational numeracy assessments for children, and work is progressing on the People's Action for Learning Network's International Common Assessment of Numeracy, a citizen-led assessment. However, these initiatives do not provide information on adult numeracy and trends over past decades. Unlike for literacy, there has been no tradition of including numeracy information in censuses.

There are, however, techniques for extracting indirect estimates of numeracy from past surveys and censuses, even when they did not explicitly include assessments or questions about numeracy. In particular, it is well established that the amount of age heaping in such data provides viable indirect estimates of basic numeracy (A'Hearn et al., 2009; Tollnek and Baten, 2016). Age heaping is a particular kind of age misreporting, where less numerate survey respondents round their age, typically to multiples of five. A proxy measure of basic numeracy can be calculated as the percentage stating their age correctly, after estimating the number of round ages exceeding what would be expected based on demographic patterns.

This measure correlates consistently with other measures of education and learning across cultures, periods and regions, including for developing countries after 1950 (Crayen and Baten, 2010), and even going back to Roman antiquity and the Inca before European contact (Juif and Baten, 2013). This basic indicator reflects the ability to work with simple, low integers, well below even minimum proficiency in the sense of SDG 4. Accordingly, most cross this threshold, even among the poorest. Nevertheless, it is suited for examining historical numeracy trends.

An analysis for this report of MICS and population census data allows the numeracy of cohorts born between the 1960s and the 2010s to be traced for 42 sub-Saharan African countries. The indirect measure of numeracy based on age heaping is highly correlated at the subnational level with results from school-based numeracy assessments (**Figure 15.5**), confirming both numeracy deficits' persistence and the proxy measure's validity. Across all PASEC countries in western and central Africa, the correlation is as high as 0.8, reaching 0.95 in Niger. The high degree of persistence has also been observed in Europe and Latin America (Baten and Hippe, 2018; Baten and Juif, 2014).

Sizeable socioeconomic differences in numeracy are especially seen in Cameroon, the Gambia, Mali, Mauritania, Nigeria, Sudan and Sierra Leone. The difference is as large as 19 percentage points in Nigeria among 50- to 59-year-olds, with 44% numeracy among the poorest compared with 63% among the richest. On average, improvements over time have been marginal and not sustained among the poorest (**Figure 15.6**).

<sup>&</sup>lt;sup>1.</sup> This section is based on Baten (2021).

#### **FIGURE 15.5:**

Indirect estimates of numeracy among adults are highly correlated with children's numeracy skills Age heaping-based numeracy of 40- to 50-year-olds compared with children's mathematics skills, by region, Burkina Faso



These trends apply both for women and men. In general, gender gaps are marginal at this elementary level of numeracy, at less than five percentage points for all but three of the countries analysed. However, urban-rural and wealth gaps are significant. In Sudan, numeracy in rural areas is more than 10 points behind for all age groups. For the age cohort now between 40 and 50, urban-rural differences are particularly large in the Gambia, Mauritania, Nigeria, Sierra Leone and Sudan.

Even bigger, unsurprisingly, is the difference in numeracy between those who benefited from any schooling and the unschooled, reaching 22 percentage points in Senegal and 24 in Cameroon. In Cameroon, Kenya, Malawi, Nigeria and Senegal, improvements in numeracy have been modest at best, both among the unschooled and those with some schooling. For example, numeracy among those who reported no schooling in Senegal was 69% for those born in the 1950s and 70% for those born in the 1970s. Cameroon saw a modest increase in numeracy skills among those who received some schooling, from

#### **FIGURE 15.6:**

Basic adult numeracy has improved only slowly in recent decades, and not at all among the poorest Numeracy in 18 African countries, by birth decade and wealth



Note: The countries are those with MICS samples for which the 1980s could be documented: Benin, Chad, Central African Republic, the Democratic Republic of the Congo, Eswatini, Ghana, Guinea, Lesotho, Madagascar, Malawi, Mali, Mauritania, Mozambique, Sierra Leone, South Sudan, Sudan, Togo and Zimbabwe.

GEM StatLink: https://bit.ly/GEM2021\_fig15\_6 Source: Baten (2021).

84% to 89%. But in most countries, progress was limited or even slightly negative, as among the unschooled in Kenya. This means the overall increase in numeracy was due almost entirely to rising school participation, mirroring findings for literacy (Barakat, 2016).

Notably high levels of numeracy are observed in the Democratic Republic of the Congo, exceeding 95% in the 1960s birth cohorts, even among the poorest. Indeed, elite numeracy was already very high in the 16th century (Baten and Alexopoulou, 2021) in the early kingdoms of Ba-Kongo and Kuba, among others.

66 -

Overall increase in numeracy in Africa was due almost entirely to rising school participation

99

# COVID-19

It is recognized that literacy and numeracy skills are crucial for health literacy and effective vaccination campaigns and must form an integral part of public emergency responses and reconstruction plans (Lopes and McKay, 2020). In addition, numeracy - not just mathematics skills but the broader ability to understand and use quantitative information - was the most consistent predictor of decreased susceptibility to misinformation about COVID-19 (Roozenbeek et al., 2020). A study in India found that women who participated in an adult literacy programme – conducted before COVID-19 spread in India and with no specific material on the pandemic - had considerably higher COVID-19 knowledge than their illiterate counterparts. Over 80% of the newly literate women were aware of the symptoms, compared with 16% among the illiterate control group (Das et al., 2021).

Conceptually, it is clear that numeracy – in the sense of capability for evaluative and analytical numerical reasoning in a range of contexts, with a positive disposition and using a variety of tools (O'Sullivan et al., 2021) – was a prerequisite for making sense of much public debate, policy decisions and advice relating to the pandemic. Indicators and graphs played an outsized role in societal understanding of COVID-19, including fairly sophisticated notions such as conditional rates and rolling averages. At the same time, this created a teaching opportunity, whereby learners could acquire a richer understanding of mathematics and statistics in terms of a real-world phenomenon they were experiencing (Ancker, 2020).

Yet adult literacy and numeracy programmes were hit hard by COVID-19. A rapid assessment by UNESCO in mid-2020 suggested that 90% of adult literacy programmes were partially or even fully suspended (UNESCO, 2020c). Moreover, such programmes were mostly absent from countries' initial education response plans (UNESCO, 2020b). Exceptions include Chad, which incorporated adult and non-formal education in its COVID-19 response plan, and Senegal, which, after developing a distance learning response plan for children and youth, established a working group to focus on basic education for youth and adults (UNESCO, 2020c).

A survey of Latin America and the Caribbean uncovered diverse government responses. Countries with well-established online, radio or television programmes pre-pandemic were better equipped to set up alternative distance education options for learners. However, infrastructure is not enough. In Uruguay, with some of the best internet connectivity and access to devices in the region, the adult education programme does not use uniform materials or curriculum; instead, educators design them locally for each community, making them difficult to adapt to distance learning (Kalman and Carvajal, 2020).

In many countries, partnerships between assorted sectors and actors were fundamental to sustainability of adult learning and education provision. In the United Republic of Tanzania, the Karibu Tanzania Organization, a national non-governmental organization representing more than 50 Folk Development Colleges, cooperated with the Ministry of Education, Science and Technology, industry organizations and the media to produce and disseminate learning materials (UNESCO, 2020a). Globally, however, providers faced serious obstacles. Many adult literacy teachers' salaries went unpaid when classes were suspended. Others lost their jobs. Those who did not were ill-equipped to sustain distance learning (Apena, 2021; UNESCO, 2021).

Even before the pandemic, distance education was an unpopular mode of delivery for initial literacy programmes. The most disadvantaged typically lacked access to the right tools, even in high-income countries (UNESCO, 2020c), or the literacy required to use them. In Latin America, many educators believe learning to read and write relies on in-person interaction, reflecting the Freire critical pedagogy stance that mechanical skills are but a small part of literacy (Kalman and Carvajal, 2020). In Brazil, a regulation published in June 2021 on second-chance programmes for youth and adults clarified that classes corresponding to the primary curriculum had to be delivered in person, with distance learning limited to programmes corresponding to lower or upper secondary education (Brazil Ministry of Education, 2021). Enrolment in primary-level second-chance programmes fell by 10% between 2019 and 2020 (INEP, 2021).

For programmes that could continue to operate, maintaining contact with learners and delivering learning materials was a challenge. Lockdowns shut down not only learning centres, but also internet sources for low-income individuals, such as libraries and restaurants (Smythe et al., 2021). In the Canadian province of Ontario, literacy agencies lent equipment to learners and provided support by telephone (UNESCO, 2020c). Often non-digital solutions were used. In Costa Rica, worksheets and other learning materials could be picked up in person, and advice was provided on how learners could work on them together with their families (Kalman and Carvajal, 2020). Another challenge has been keeping learners engaged amid competing priorities. Increased childcare responsibilities during school closures and the need to work extra hours to compensate for lost wages make it difficult for many to continue their studies. In Sierra Leone, a local adult education programme teaching basic literacy and numeracy to first-time learners addressed this challenge by establishing a community bank within the programme. The bank aimed to ease financial pressure, particularly on women, by providing loans, supporting students' businesses and encouraging saving (Partners in Health, 2021).

Adult literacy and numeracy programmes are likely to become even more important in the post-pandemic period. School closures and ensuing dropout may increase demand for second-chance programmes. In the United States, participation in adult education programmes has historically risen during recessions and is likely to rise after the pandemic (Lotas, 2021). In Latin America, one estimate suggested demand could increase by 17%, projecting that up to 1 million students might drop out of school (Kalman and Carvajal, 2020).


## **KEY MESSAGES**

Only 10 countries reported fully reflecting or including the guiding principles of UNESCO's 1974 Recommendation related to Target 4.7 in all four domains, from policies to assessment, in relation to global citizenship education and education for sustainable development.

Applying UNESCO's Sexuality Education Review and Assessment Tool in 24 countries showed that only 3 provided 'advanced' curriculum content for the 9 to 12 age group.

Around four in five students in OECD countries reported that their school curriculum covered global citizenship and sustainability issues.

Data from the 2019 TIMSS show that learning proficiency in environmental science has stagnated since 2015, with 30% of students reaching proficiency.

A new series of country profiles on climate change communication and education, a partnership between the GEM Report and the Monitoring and Evaluating Climate Communication and Education project, shows that a climate change focus was found in 8 out of 20 national education laws.

COVID-19 has shed light on education systems' failures to pursue the ideals of solidarity and multilateralism, as the world has witnessed responses in the opposite direction, from vaccine nationalism to xenophobic policies and the spread of discriminatory beliefs. COVID-19 also put health literacy at the centre of attention.

### **CHAPTER 16**



### **TARGET 4.7**

# Sustainable development and global citizenship

By 2030, ensure that all learners acquire the knowledge and skills needed to promote sustainable development, including, among others, through education for sustainable development and sustainable lifestyles, human rights, gender equality, promotion of a culture of peace and non-violence, global citizenship and appreciation of cultural diversity and of culture's contribution to sustainable development

### **GLOBAL INDICATOR**

**4.7.1** – Extent to which (i) global citizenship education and (ii) education for sustainable development are mainstreamed in: (a) national education policies, (b) curricula, (c) teacher education and (d) student assessment

### THEMATIC INDICATORS

4.7.2 – Percentage of schools that provide life skills-based HIV and sexuality education

**4.7.3** – Extent to which the framework on the World Programme on Human Rights Education is implemented nationally (as per the UNGA Resolution 59/113)

**4.7.4** – Percentage of students by age group (or education level) showing adequate understanding of issues relating to global citizenship and sustainability

**4.7.5** – Percentage of students in the final grade of lower secondary education showing proficiency in knowledge of environmental science and geoscience

**4.7.6** - Extent to which national education policies and education sector plans recognize a breadth of skills that needs to be enhanced in national education systems

arget 4.7 goes further than the rest of the SDG 4 agenda in addressing what learners need to learn in order to reach the transformational ambitions of SDG 4. Global indicator 4.7.1 examines the 'extent to which (i) global citizenship education and (ii) education for sustainable development are mainstreamed at all levels in: (a) national education policies, (b) curricula, (c) teacher education and (d) student assessment'. The data source for this indicator is country contributions to consultations for UNESCO's 1974 Recommendation concerning Education for International Understanding, Cooperation and Peace and Education relating to Human Rights and Fundamental Freedoms. The 1974 Recommendation sets out guiding principles on teaching peace and non-violence, human rights and fundamental freedoms, cultural diversity and tolerance, and human survival and well-being.

The latest available data are from the sixth consultation in 2016/17. They show that implementation of the 1974 Recommendation remains patchy, with only 10 countries fully reflecting or including the guiding principles in all 4 domains, from policies to assessment.

### 66

Civics textbooks were more likely than history or social studies textbooks to focus on individual human rights rather than collective social justice

Among obstacles, countries mentioned unavailability of resources three or more times as often (29 countries) as lack of demand (10 countries), intrinsic education challenges (7 countries) or differing political priorities (5 countries) to explain their lack of progress (**Figure 16.1**).

99

Countries reported on the subjects in which the guiding principles were mainly taught. Science subjects were the single most common main channel. Notably, civics or ethics, not taught in all countries to begin with, were barely mentioned, even where they exist, as the main channels for teaching the guiding principles (**Figure 16.2**).

### **FIGURE 16.1:**

Countries mainly say lack of resources is why they lag in implementing the 1974 recommendation's guiding principles

Mentions of specific factors as obstacle to or enabling factor for improvement in country responses to the 1974 Recommendation consultation, 2016/17



#### **FIGURE 16.2:**

### Civics and ethics classes are not often used to teach the 1974 recommendation's principles

Countries assigning a ranking greater than 5 out of 10 to a given school subject as one 'in which the guiding principles are mainly taught', 2016/17



Source: GEM Report team analysis of responses to the 2016/17 consultation on the 1974 recommendation.

consultation on the 1974 recommendation

A review of 556 secondary school textbooks from 80 countries between 1950 and 2011 showed that civics textbooks were more likely than history or social studies textbooks to focus on individual human rights rather than collective social justice (Skinner and Bromley, 2019).

The seventh consultation in 2022 will use a refined questionnaire discussed at the 2021 UNESCO General Conference and will require documentary evidence supporting country reports. In principle, climate change falls within indicator 4.7.1's scope, but climate change education deserves more attention (**Focus 16.1**).

Indicator 4.7.2 examines provision of life skills-based HIV and sexuality education. In total, 42 countries report that 100% of schools provide this, while 17, including Algeria, Bangladesh and Costa Rica, report that no school does. Overall, 70% of countries report no data at all. Where empirical school-based information rather than statutory curriculum content is used as a source, the fraction is frequently low, e.g. 2.5% of primary schools in Burkina Faso and 6% in Niger. However, 23% of primary schools in Sierra Leone and 62% in Zambia offer such education (**Figure 16.3**). Many countries report data for only one level of schooling. Where data are reported for schools at all levels, life skills-based HIV and sexuality education tends to be more common in secondary than primary schools. Yet age-appropriate topics for young primary school children include knowledge about the private nature of certain body parts and accepting natural curiosity about them, behaviours that can protect the body, including against HIV. These are among topics suggested by the revised UN International Technical Guidance on Sexuality Education for ages 5 to 8 (UNESCO, 2018). In particular, the guidance recommends covering puberty and menstruation before learners experience them, i.e. for ages 9 to 12. Given the increasingly early onset of the first menstrual cucle, with estimates indicating girls in the 2002 birth cohort in low- and middle-income

66 -

Given the increasingly early onset of the first menstrual cycle, age 12 can be seen as the last opportunity to prepare them

#### **FIGURE 16.3:**



Globally, the likelihood of receiving HIV and sexuality education varies widely Percentage of schools providing life skills–based HIV and sexuality education, 2019 or latest available year "

countries experienced menarche before their 13th birthday, on average (Leone and Brown, 2020), age 12 can be seen as the last opportunity to prepare them.

UNESCO's Sexuality Education Review and Assessment Tool (SERAT), developed in 2012, has been revised to reflect the updated International Technical Guidance. SERAT underlies a recent global progress report on comprehensive sexuality education (UNESCO, 2021b) and provides a more in-depth review of comprehensive sexuality education provision than the single indicator summary of 4.7.2, including by age and specific topics. Among 24 countries, only 3 are assessed as providing 'advanced' curriculum content on sexual and reproductive health for ages 9 to 12, and 5 countries as having 'established' content. Even in countries scoring relatively high, gaps remain. In Malawi, with advanced curricular content both on sexuality and sexual behaviour and on sexual and reproductive health, 9- to 12-year-olds receive little detail on contraceptives or the use of pregnancy tests. Some countries, including Uganda, have a high-quality curriculum but it only begins at age 15, an age by which many of the girls most at risk of early pregnancy have dropped out of school.

While indicators 4.7.1 and 4.7.2 address provision, the monitoring framework for target 4.7 also aims to measure the resulting knowledge outcomes. In particular, it asks what percentage of students show 'an adequate understanding of issues relating to global citizenship and sustainability' (4.7.4) and 'proficiency in knowledge of environmental science and geoscience' (4.7.5). Both indicators are monitored at the lower secondary school level. Following a review, including of alignment with the target 4.7 global content framework, as well as an agreement on minimum proficiency levels, the UNESCO Institute for Statistics (UIS) now reports values for these indicators based on cross-national assessments (Sandoval-Hernandez and Carrasco, 2020). Both indicators have been defined as composites of one cognitive and multiple non-cognitive dimensions.

Data for indicator 4.7.4 draw on the 2016 International Civic and Citizenship Education Study (ICCS) and are reported for 23 upper-middle- and high-income countries. The percentage of students with adequate understanding, based on one cognitive component (consisting of four civic knowledge content domains: society and systems, principles, participation, and identities) and seven non-cognitive components (global-local, multiculturalism, gender equality, peace, freedom, social justice, sustainable development), ranges from around 40% in the Dominican Republic, Latvia and the Netherlands to almost 70% in Croatia, the Republic of Korea and Sweden. However, the seven non-cognitive components are not all positively correlated with the cognitive component. For instance, a higher cognitive score is positively correlated with attitudes of students who are better at identifying what is good for democracy (e.g. electing political leaders, protesting an unfair law) and what is bad for democracy (e.g. media concentration, nepotism in government, influence of government over the judiciary). But it is negatively correlated with expression of highly positive attitudes towards one's country of residence (global-local) (**Figure 16.4**).

The 2018 Programme for International Student Assessment (PISA) of the Organisation for Economic Co-operation and Development (OECD) collected information on cognitive and non-cognitive dimensions of 'global competence' that overlapped partly with the notion of global citizenship addressed by indicator 4.7.4, although UIS did not include it in its reporting because, unlike ICCS, this PISA module is not expected to be regularly repeated. On average, some four in five students in OECD countries reported that their school curriculum covered global issues (OECD, 2020). Gathering information on both learning opportunities and outcomes makes it possible to examine their relationship. Consistently across countries, student awareness, agency and self-efficacy regarding global issues were higher when the number of related learning activities was higher, even after accounting for socioeconomic effects. These positive effects remained small overall, though, and almost half of all learners, on average, failed to demonstrate minimum 'global competence'. On many measures, there was a large gap between students with low and high socioeconomic status, not just in aspects of knowledge or cognitive performance, but also regarding perceptions of agency, awareness, interest and practical action. Such education cannot succeed if it is perceived as only of concern for the better off.

66

Education for global competences cannot succeed if it is perceived as only of concern for the better off

"

#### **FIGURE 16.4:**

Non-cognitive components of student understanding of global citizenship differ in their relationship with knowledge of global citizenship

Correlation between the cognitive and seven non-cognitive components of adequate understanding of issues relating to global citizenship and sustainability, 2016



Notes: Data refer to grade 8 students in 23 countries. The non-cognitive dimensions of understanding of issues related to global citizenship and sustainability refer to attitudes towards country of residence (global-local), equal rights for all ethnic/racial groups (multiculturalism) and gender rights (gender equality), along with personal experiences of bullying and abuse (peace), perceptions of what is good for democracy (freedom), perceptions of the importance of social movement-related citizenship (social justice) and responses to threats to the world's future (sustainable development).

GEM StatLink: https://bit.ly/GEM2021\_fig16\_4 Source: UIS database.

### **FIGURE 16.5:**

### The majority of students are not proficient in scientific knowledge on the environment

Percentage of grade 8 students showing proficiency in knowledge of environmental science and geoscience, cognitive dimension, selected countries, 2015 and 2019



### **TABLE 16.1:**

Education indicators in the periodic reporting on Intangible Cultural Heritage

Thematic area	Core indicator
Transmission and education	4. Extent to which both formal and non-formal education strengthen the transmission of ICH and promote respect for ICH
	5. Extent to which ICH and its safeguarding are integrated into primary and secondary education, included in the content of relevant disciplines, and used to strengthen teaching and learning about and with ICH and respect for one's own and others' ICH
	6. Extent to which post-secondary education supports the practice and transmission of ICH as well as study of its social, cultural and other dimensions
Policies as well as legal and administrative measures	12. Extent to which policies as well as legal and administrative measures in the field of education reflect the diversity of ICH and the importance of its safeguarding and are implemented
Source: UNESCO (2021a).	

Data from the 2019 Trends in International Mathematics and Science Study for indicator 4.7.5 show that learning proficiency in environmental science has remained stagnant, with only about 30% of students reaching proficiency. Country shares ranged from 5% or less in Lebanon, Morocco and South Africa to half or more in Finland, Japan, the Republic of Korea and Singapore (**Figure 16.5**). The UIS uses a composite index that also includes two non-cognitive dimensions: enjoyment (e.g. student likes learning physics) and confidence (e.g. student is confident in their knowledge of earth science). But it is doubtful these should be part of a 'knowledge' indicator.

Much environmental knowledge is indigenous knowledge. Indigenous communities are stewards of 18% of the world's land, not including areas that are claimed but lack legal recognition. Indigenous land includes vast forest areas storing around 300 billion tonnes of carbon, or 33 times the world's energy emissions in 2017 (RRI, 2018). It is recognized that indigenous local knowledge plays an important part in addressing the threat of climate change (IPCC, 2019). Indigenous environmental knowledge forms an inseparable part of indigenous culture, protection of which falls under target 4.7's aim of protecting cultural diversity in its own right. Moreover, including living heritage in education is crucial for achieving target 4.5, and its indicator on the use of home language in education, by creating opportunities to revitalize marginalized languages through arts, drama, poetry and music in schools.

Such traditions, together with the practices, representations, expressions, knowledge, skills – as well as the instruments, objects, artefacts and cultural spaces associated therewith, form the intangible cultural heritage that every community transmits from generation to generation and constantly recreates in response to its environment. The results framework of the 2003 Convention for the Safeguarding of the Intangible Cultural Heritage (ICH) can inform understanding of the extent to which target 4.7's overarching aims are being achieved. Of the 26 core indicators in the framework, 4 relate to education (**Table 16.1**).

As with reporting under the 1974 Recommendation consultation process for SDG indicator 4.7.1, periodic reporting for the ICH Convention is based on country self-assessment. The process is formative, challenging the education sector to engage systematically with the issue of intangible cultural heritage. At its seventh session in 2018, the General Assembly of States Parties to the Convention approved a regional cycle of reporting, beginning with Latin America and the Caribbean in 2021. Out of 27 countries responding in the region, 21, or almost three quarters, reported teaching about protection of natural and cultural spaces and places of memory for expressing intangible cultural heritage as part of the primary or secondary school curriculum.

However, only 12 countries said this was a stand-alone subject; 21 countries used these topics as a means of demonstrating other subjects; only 15 countries integrated intangible cultural heritage and its safeguarding into teacher training. Ecuador adopted a National Strategic Plan for Afro-Ecuadorian Ethnoeducation for 2020-2025 that includes the establishment of Guardians of Knowledge in educational units, as well as teacher training and curriculum interventions.

Almost half the responding countries in the region reported strengthening the practice and transmission of intangible cultural heritage in technical and/or vocational education and training, for example in conservation management.

### FOCUS 16.1: CLIMATE CHANGE EDUCATION AIMS TO EQUIP POPULATIONS TO COPE WITH AND MITIGATE THE EFFECTS OF CLIMATE CHANGE

Climate change education (CCE) can be formal, non-formal or informal, multidisciplinary and at every education level; it aims to help populations understand, address, mitigate and adapt to the impact of climate change. Achieving this requires developing appropriate curricula, teacher training and pedagogies (UNESCO and UNFCCC, 2016). CCE encourages positive mindsets for lasting change and actions to address the causes of climate change and adopt more sustainable lifestules. It also aims to help policymakers become fully aware of the urgent need for mechanisms to tackle climate change at the national and global levels and to increase the resilience of the vulnerable communities that are the most likely to be affected (UNESCO, 2015b). CCE participants are more concerned about environmental issues (UNESCO, 2021c), which often translates into support for pro-environmental policies, environmental activism and active participation (Coan and Holman, 2008; Lubell et al., 2006).

Demand for CCE is clearly expressed in public opinion surveys. Global surveys show that most people are concerned about climate change, are willing to change their lifestyles to mitigate its effects and support further relevant government action. A 2020 survey showed widespread recognition of the urgent need for a collective approach by government and civil society to advance the agenda to address climate change (World Economic Forum, 2020). The results also indicated that more than half of global respondents, particularly in South Asia, had full confidence in climate science. An overwhelming majority of respondents worldwide indicated they saw most government climate regulations as necessary.

A 2021 Pew Research Center survey of more than 16,000 adults in 17 high-income countries found that between 60% and 90% felt somewhat or very concerned about damage they might personally experience as a result of climate change, a marked increase since the original survey conducted in 2015 (Luong et al., 2021). In 16 countries, at least 70% of respondents said they were willing to make some or many changes to their lifestyle to help tackle global climate change, with higher willingness among younger respondents. However, deep ideological divides in climate attitudes were evident in Australia, Canada, the Netherlands and the United States. Such divides are likely to pose significant challenges for implementation of much-needed policy changes, including in education.

Despite concern, many countries are still not fully committed to making climate action a core component of the curriculum. Results of the 2019 curriculum analysis of the Fourth Regional Comparative and Explanatory Study in Latin America and the Caribbean revealed that while more than half the 18 countries analysed mentioned climate change in their curriculum documents, it was only the 11th-most-covered Education for Sustainable Development theme (UNESCO, 2020). Similarly, a UNESCO study revealed that climate change was mentioned in less than half the policy and curricula documents examined (UNESCO, 2021c). However, the 2018 PISA survey reported that of the top seven global issues commonly included in curricula in OECD countries, the one most commonly reported by school principals was global warming and climate change. The study indicated that 88% of students in OECD countries attended a school where the topic was covered in the curriculum (OECD, 2020).

A few countries are adopting promising education changes to ensure CCE of good quality. Countries are making significant efforts to address CCE in a cross-curricular rather than subject-specific way. In France, the Vademecum

### 66

In Italy, climate change education is part of transversal civics education introduced by law to schools from 2020/21

99

on Education for Sustainable Development stated that at the start of the 2020/21 school year, the curricula for cycles 1, 2, 3 and 4 had been strengthened with lessons on climate change, in art and music education, French, moral and civic education, history and geography, the life and earth sciences, technology, modern languages, physics and chemistry.

In Italy, climate change education is part of transversal civics education introduced by law to schools from 2020/21 to support development of knowledge and understanding of social, economic, legal, civic and environmental structures. In the Republic of Korea, the national curriculum framework introduced climate change education at all levels in 2007. In preschool, climate change education is part of scientific exploration. This curriculum encourages 4-year-olds to engage with weather and climate change and 5-year-olds to learn about climate regularities. The framework is anchored in the Environmental Education Promotion Act, which aims to contribute to sustainable development by fostering knowledge and skills to prevent and address environmental problems, including climate change (Republic of Korea Ministry of Education, 2015).

In India, educational resources are being developed to integrate climate change topics into core curricula at the school and undergraduate levels to raise awareness about the causes and effects of climate change. A team has built a repository of teaching resources from around the world on climate change that can be used for discipline-specific topics in mathematics, science, humanities and the social sciences. This new pedagogical approach of integrating CCE into the existing curriculum allows students to develop analytical and communication skills in core subjects (Shashidhara, 2019).

Climate change can evoke emotions such as anxiety, empathy and anger, which influence action. However, few countries put emphasis on non-cognitive dimensions, such as social and emotional components and values. In Rwanda, the 2015 national curriculum framework integrates climate change, environment and sustainability into learning materials that are also meant to develop values for sustainable living, an intention confirmed in the 2019–24 Education Sector Strategic Plan.

Many countries have incorporated CCE in their curricula, both in theory and practice (Chiba et al., 2021), with solution-oriented and participatory approaches to empower learners to be agents of change (UNESCO, 2015b). Successful approaches in Erasmus+ projects on education for sustainable development often include physical activities, such as collecting waste, planting trees and organizing environmental campaigns (European Commission, 2021). Since the mid-1990s, Colombia's Ministry of National Education has managed School Environmental Projects that identify environmental priority situations and focus on actions adapted to school realities, integrating various fields of knowledge, disciplines and skills for interdisciplinary problem solving (Colombia Ministry of National Education, 2005; Mora-Ortiz, 2015).

Strong connections to the environment and climate result in more intentions to act and willingness to change behaviour (Dietz et al., 2020; Zelenika et al., 2018). Hence countries try to prioritize action-oriented teaching and place-based learning, especially in upper secondary education (UNESCO, 2021c). In England (United Kingdom), the youth-led Teach the Future campaign aims for all school buildings to have a net-zero carbon footprint by 2030. To this end, the Department for Education is providing funding to help schools become more sustainable (Burns, 2020).

Overall, in terms of pedagogy, an effective CCE approach would aim to achieve a balance between developing learners who can think critically about CCE and empathetic

66 -

In a sample of 20 countries, a climate change focus was found in only 40% of national education laws

99

students who are committed to taking action to improve their living environment (Chang, 2015). It would promote collaboration with peers and interactions to inspire change. Promising initiatives require a long-term perspective to be able to trigger behaviour change. There is little robust cross-national information on where countries stand on addressing climate change in education systems, as much of the information appears in national documents rather than in Nationally Determined Contributions to the UN Framework Convention on Climate Change Secretariat (SEPN, 2020).

A new series of country profiles on climate change communication and education at the Profiles Enhancing Education Reviews (PEER) website aims to improve the evidence basis. It is the result of a partnership between the Global Education Monitoring Report and the Monitoring and Evaluating Climate Communication and Education (MECCE) project, hosted by the Sustainability and Education Policy Network and financed by the Social Sciences and Humanities Research Council of Canada. It offers a comparative perspective of countries' progress on Article 6 of the United Nations Framework Convention on Climate Change and Article 12 of the Paris Agreement (Action for Climate Empowerment), and SDG target 4.7. A first set of 20 country profiles was published in November 2021, covering all world regions and country income groups. A second set of up to 50 profiles is scheduled to be published in 2022.

These profiles present content on climate change contexts; climate change education (policy, curriculum, teacher education and assessment) in primary, secondary, technical, vocational, higher, teacher and adult education; climate change communication (public awareness, public access to education, public participation); and monitoring. Initial analysis suggests that in all but 1 of the 20 countries analysed, education ministries work on climate change; all but 2 have a national climate change law, strategy or plan that includes education content; and all but 3 have monitoring mechanisms to track progress in CCE. Yet a climate change focus was found in only 40% of national education laws and 45% of education sector plans or strategies. National curriculum frameworks refer to climate change in 75% of countries and just over one third have a law or plan with a clear focus on CCE. Efforts focus on the primary and secondary education levels in 90% of countries; on technical, vocational and higher education in 70%; and on teacher education in 55% (Figure 16.6).

### **FIGURE 16.6:**

There is scope for countries to strengthen how climate change is addressed in education

Percentage of countries with climate change education in laws, plans and policies, 2021



Note: GEM Report and MECCE team analysis of 20 climate change communication and education county profiles: Azerbaijan, Bangladesh, Colombia, Cook Islands, Costa Rica, Dominican Republic, the Gambia, Indonesia, Italy, Morocco, Myanmar, New Zealand, Qatar, Republic of Korea, Rwanda, South Africa, Sweden, Tajikistan, Tuvalu and Zimbabwe.

GEM StatLink: https://bit.ly/GEM2021\_fig16\_6 Source: PEER website.

### COVID-19

Education for sustainable development and global citizenship education is a response to the challenges of a planet that is increasingly interconnected but whose future is at stake. It is meant to enable learners to actively engage in developing more inclusive and secure societies (UNESCO, 2015a). Yet COVID-19 has shed light on systems' failures to pursue the ideals of solidarity and multilateralism, and growing inequality within and between countries raises moral concerns. Even as the nature of the pandemic, with rapid contagion and variant development, requires global solutions, the world has witnessed many responses that go in the opposite direction, from vaccine nationalism to xenophobic policies and the spread of discriminatory beliefs (Farge, 2021; United Nations, 2020).

COVID-19 put health literacy at the centre of attention. The well-known link between education and health awareness was confirmed during the pandemic. In Saudi Arabia, individuals with higher education attainment were more aware and more likely to follow personal protective measures (Bazaid et al., 2020). Health literacy is connected with making decisions based on health care providers' advice (Turhan et al., 2021). It is also linked with lower vaccine hesitancy (Biasio, 2017), a key factor for slowing down COVID-19 contagion rates. Calls to include health literacy as a compulsory subject in basic education curricula have gained ground (Molnar, 2021). The county of San Diego, United States, pledged US\$2 million to improve health literacy in schools (San Diego County Office of Education, 2021). Just as such issues have become more urgent, however, the pandemic has led many governments to increase the curricular focus on 'core' subjects to minimize learning losses in literacy or numeracy. Yet there is danger of this narrowing of focus undermining education's broad humanistic dimension (International Commission on the Futures of Education, 2020). Over 20% of low-income countries and over 40% of lower- and upper-middle-income countries planned to reduce the content of instruction as part of their COVID-19 response (UIS, 2020). In Afghanistan, the COVID-19 response plan envisaged core subjects being continued through distance learning and social science subjects being 'self-learned' (Afghanistan Ministry of Education, 2020). In England (United Kingdom), the Department for Education announced that non-core subjects had to be dropped in favour of English and maths for the 2020/21 academic year (Staton and Hughes, 2020).

Health literacy can also be pursued through informal and non-formal adult education initiatives. In South Africa, participants in a large adult literacy campaign with a health literacy dimension improved their ability to understand health messages (Lopes and McKay, 2020). Initiatives must not only provide accurate health information, but also foster long-term improvement in health literacy through a variety of actors, including health care organizations, community-based partnerships and cross-sector collaborations (Damian and Gallo, 2020). In 2021, the US Department of Health and Human Services distributed US\$250 million in grants to 73 local governments to improve health literacy among underserved populations by working with local community-based organizations (US Department of Health and Human Services, 2021).

Jerrene Gerald, 11, sits with classmates at Princess Margaret Secondary School in St. John's, Antigua. Jerrene and her family were evacuated to Antigua from Barbuda following Hurricane Irma in September 2017.

A A A A A A

CREDIT: UNICEF/Roger LeMoyne

## **KEY MESSAGES**

Some children have better facilities at school than at home. In Liberia, few households have basic hygiene facilities, but 69% of schools do. Conversely, in Honduras, only 12% of schools but 84% of households do.

Internet access at home remains out of reach for most in the least developed countries, but some countries are making efforts to increase access in schools. Bhutan only introduced the internet in 1999 and now has internet access in 98% of its schools.

One in five students in most countries experienced bullying at least monthly, rising to one in three in high-income English-speaking countries.

Attacks on schools continue. Mali saw over 500 attacks on schools in 2020.

The organization of school calendars matters for quality and equity. An overlap between seasonal labour demand and annual school examinations in Bangladesh led to a dropout rate seven percentage points higher for students from agricultural households.

Hygiene measures can ensure it is safe to go to school during COVID-19, but these have cost implications. In early 2021, less than 10% of low-income countries reported having basic measures such as sufficient soap, clean water, masks, and sanitation and hygiene facilities to assure the safety of all learners and staff; the share in high-income countries was 96%.



## TARGET 4.a Education facilities and learning environments

Build and upgrade education facilities that are child, disability and gender sensitive and provide safe, non-violent, inclusive and effective learning environments for all.

### **GLOBAL INDICATOR**

4.a.1 - Proportion of schools offering basic services, by type of service

### **THEMATIC INDICATORS**

- 4.a.2 Percentage of students experiencing bullying in the last 12 months
- 4.a.3 Number of attacks on students, personnel and institutions

## In Côte d'Ivoire, 87% of upper secondary schools but just 17% of households have internet connections

A t the core of SDG target 4.a is the call for 'safe, non-violent, inclusive and effective learning environments for all'. Learning of good quality cannot take place if the environment is unsuitable, much less if it threatens children's well-being. Underlying the current formulation of target 4.a is an assumption that education facilities provide the learning environment, but COVID-19 has been a stark reminder that learners can be located in other environments when learning is remote. Beyond the physical facilities, other aspects of how schools and learning are organized, such as the school calendar, have tangible effects on effectiveness and equity (**Focus 17.1**).

School may be the only place some children have access to water, sanitation and hygiene facilities. Access to water at home is an important prerequisite for freeing children, especially girls, from water-fetching duties that interfere with their ability to attend school. In Nepal, spending one hour a day fetching water lowers girls' primary school completion by 17 percentage points (Dhital et al., 2021). The extent to which schools provide facilities superior or inferior to what students have access to at home varies tremendously by country. In Liberia, few households have hugiene facilities that meet the basic international standard, but 69% of schools do, while in Honduras the shares are 12% of schools and 84% of households. In many of the poorest countries, most children have no access to basic facilities at school or home (Figure 17.1). It is especially concerning that so few schools meet such standards given how basic the standards are, such as running water with soap. In some high-income countries, including the United States, most if not all schools meet such standards, although there are concerns that school infrastructure is crumbling (Pulkkinen, 2021) (Focus 17.2).

Internet access is another crucial factor for education. While most sub-Saharan African countries have some plan or policy on information and communications technology (ICT) (Burns et al., 2019), few have achieved successful implementation. In Côte d'Ivoire, 87% of upper secondary schools but just 17% of households have internet connections. The 2016–25 education sector plan includes provisions for increasing ICT use in secondary schools, such as multimedia rooms and professional development for teachers (Côte d'Ivoire Government, 2017). In Rwanda, 62% of upper secondary schools had internet access in 2019, a low share but twice that of low-income countries as a group in 2016. Rwanda adopted an ambitious ICT in education policy in 2016. At its peak in 2017/18, the programme budget was almost US\$15 million, over 6% of the education budget. The programme included curriculum changes, teacher training and management support, reflecting lessons from the disappointing results of the One Laptop Per Child programme in 2008 (Rwanda Ministry of Education, 2016).

99

Home internet access has been rising rapidly in developing countries in recent years, but remains out of reach for most in the least developed countries. Some of those countries have managed to increase school internet access, however. Bhutan, one of the least developed countries as recently as 2018 (Razzaque, 2020), did not introduce internet or even television until 1999 (BBC, 2019) but now boasts internet access and use for pedagogical purposes in 98% of schools, although only 44% of households (**Figure 17.2**).

Indicator 4.a.1 on basic facilities is expressed as a percentage of schools. It does not indicate how manu students have access to the facilities. Nor does it capture what facilities out-of-school children would have access to if they went to school. One estimate, which applied the percentage of schools in each country to the number of children, suggested that 818 million children lacked access to basic hygiene facilities at school (Joint Monitoring Programme, 2020). In practice, schools differ greatly in size, with some having fewer than a hundred students and others several thousand. Moreover, the indicator measures the existence and guality of facilities, not whether there are enough taps and washbasins. Small schools, primarily located in remote and rural areas, can reasonably be assumed to be less likely to meet the indicator criterion than large schools, which are primarily urban and housed in dedicated buildings. In this case, the number of children attending schools without basic facilities will be significantly lower.



60

40

20

0 0 Rwanda

20

40

Senegal

Samoa

In many low-income countries, both households and schools lack basic amenities Percentage of households and schools with basic facilities, 2019 or latest available year



High-income • Upper-middle-income • Lower-middle-income • Low-income

• Iran, Isl. Rep

100

80

Cabo Verde

60

Households with internet accesss (%)

Note: The internet access level is calculated as a median across school levels.

GEM StatLink: https://bit.ly/GEM2021\_fig17\_1b Sources: ITU and UIS databases.

### **FIGURE 17.2:**

Very few households in the least developed countries provide an environment conducive to online learning Percentage of households with computer and internet access at home, by country development status, 2009–19



"

Estimating the share of schools with basic facilities requires detailed information on the size distribution of schools, for which no comprehensive international data source exists. In Cabo Verde, 22% of primary schools in 2018 lacked basic handwashing facilities. However, the smallest 22% of primary schools accounted for only 2% of primary enrolment. On average, just half of all primary schools in lower-middle-income countries have basic handwashing facilities. Across six lower-middle-income countries, the percentage of enrolment in the smallest 50% of primary schools ranged from 10% in Cabo Verde and 14% in Djibouti to 15% in

66

The percentage of enrolment in the smallest 50% of primary schools is 10% in Cabo Verde

Bhutan and El Salvador, 16% in Kyrgyzstan and 24% in Cambodia (**Figure 17.3**). While the correlation of school size and facilities cannot be estimated accurately, it is clear that the true number of children worldwide lacking access to basic handwashing facilities at school could easily be half of current estimates (around 400 million lower). Nevertheless, even if it leads to overestimating the percentage of children affected, focusing on the number of schools is arguably more equity-oriented, as small schools tend to be remote and disadvantaged.

Evidence continues to grow showing that corporal punishment not only violates children's rights, but also affects education outcomes (Maiti, 2021). Corporal punishment is fully banned in schools in 156 countries (End Corporal Punishment, 2021a). In addition, the Lesotho and Marshall Islands parliaments have made clear that it is unlawful even if not explicitly prohibited.

99

Nine other countries offer some legal protection: In federal systems such as those of Australia, India and the United States, only certain states prohibit corporal punishment in schools, while countries including Samoa allow 'reasonable' punishment but with legal clarification that despite traditional acceptance of some violence in disciplining children, corporal punishment is never considered reasonable (End Corporal Punishment, 2021b).

Students face other risks at school in addition to harm inflicted by teachers. Many also suffer at the hands of their peers. Indicator 4.a.2 monitors the percentage of students who experienced bullying in the past 12 months. While not fully standardized, international learning assessments are currently the preferred source of data on bullying (UIS, 2020), including the latest iterations of the Trends in International Mathematics and Science Study (TIMSS) and the Programme for International Student Assessment (PISA) (**Figure 17.4**).

### 66

In Malaysia and South Africa, more than half of students experienced bullying at least monthly

TIMSS asks about various dimensions of bullying to calculate an overall frequency index. In most countries surveyed, at least one in five students reported experiencing bullying at least monthly. The share rose to one in three in high-income English-speaking countries, where more frequent victimization was also common. In Malaysia and South Africa, more than half of students experienced bullying. Differences between girls and boys were small in most countries, but boys were much more likely to report weekly or monthly bullying in Northern African and Western Asian countries. TIMSS



Smaller schools enrol only a fraction of students Cumulative share of enrolment in primary schools ordered by size, selected countries, 2017–20



conflates students who never and students who almost never experienced bullying, i.e. less often than monthly. Thus some of these students also experienced bullying in the previous year. Using data from PISA, whose index includes the intermediate category 'a few times per year', the UNESCO Institute for Statistics has estimated values aligned with the definition of indicator 4.a.2, which records all those who experience bullying in the past 12 months. By that definition, about half of all students were affected.

These data refer to bullying in the previous 12 months. Earlier estimates were lower; they were based on the Health Behaviour in School-Aged Children (HBSC) survey and the Global School-based Student Health Survey (GSHS), which asked about bullying over shorter reference periods: the 'previous 30 days' (GSHS) or 'last couple of months' (HBSC). The HBSC retained this reference period in the 2017/18 round (Inchley et al., 2018) while the GSHS questionnaire adopted the 12-month reference period, aligned with indicator 4.a.1, starting with the 2018–20 round (CDC and WHO, 2017), promising improved data availability in the coming years.

While bullying is pervasive in all countries, violent conflict in some countries creates additional threats from outside school. Indicator 4.a.3 serves to monitor the number of attacks on students, personnel and institutions. The data for this indicator are compiled by the inter-agency Global Coalition to Protect Education from Attack (GCPEA) and are based on observations and reports by various actors on the ground.

The GCPEA has developed a toolkit to help ensure that the information collected allows for consistent monitoring using agreed definitions and criteria (Kapit et al., 2021).

#### **FIGURE 17.4:**





Percentage of students in lower secondary education experiencing bullying, by frequency and gender, selected countries, 2018–19

Sources: 2019 TIMSS (weekly and monthly prevalence) and 2018 PISA (experience in the past year).

### 66

## Mali saw over 500 attacks on schools in 2020

Mali saw over 500 attacks on schools in 2020; 13 people were harmed. Amid intensifying conflict in Burkina Faso, 233 students and education personnel were harmed in 2020, though the number of incidents was lower. Education in the Democratic Republic of the Congo has been heavily affected by armed conflict since 2015. In Palestine, the Syrian Arab Republic and Yemen, the number of attacks on education in 2020 was lower than in previous years, but remained high in absolute terms. The same was true to a lesser extent in Somalia and Ukraine. But school safety deteriorated significantly in Colombia in 2020 (**Figure 17.5**). In several countries, including Afghanistan, Ethiopia and Myanmar, the overall level of armed conflict intensified in 2021, foreshadowing potential increases in the number of attacks on education to be reported in 2022. In Afghanistan, much will depend on the Taliban regime's approach to girls' education. Factions within its ranks consider its previous obstruction to be harmful to the movement's reputation, but signs in late 2021 were not positive (**Focus 14.2**).

The Safe Schools Declaration, an intergovernmental political commitment to protect students, teachers, schools and universities from attack during times of armed conflict, has now been endorsed by 112 states, with Algeria, Antigua and Barbuda, Ghana, Maldives, Mexico, Saint Vincent and the Grenadines, Timor-Leste and Togo the latest to sign up in 2020 and 2021 (GCPEA, 2021).

### **FIGURE 17.5:**

Conflicts continue to affect attacks on education Number of incidents of attacks on education, 2015–20, selected countries



99

Note: Lightly shaded values represent multiyear aggregates spread equally over the relevant time period.

GEM StatLink: https://bit.ly/GEM2021\_fig17\_5 Source: GCPEA.

### FOCUS 17.1: SCHOOL SCHEDULES ARE PART OF THE LEARNING ENVIRONMENT – AND INFLUENCE OUTCOMES

The organization of school calendars – from distribution of instruction days across weeks and years to duration and organization of the school day itself – can have important consequences for the quality and equity of education systems.

### SCHOOL YEARS ARE ORGANIZED IN A VARIETY OF WAYS

The organization of yearly school calendars is directly linked to total intended instruction time for students, often legislated by national or local authorities. Variation across countries is wide: Among middle- and high-income countries, the average number of instruction days per year in primary education ranges from 162 in France to 219 in Israel (European Commission/EACEA/Eurydice, 2020; OECD, 2019). Part of the cross-country difference is due to the organization of the school week, which varies, for example, from 4 or 4.5 days in France to 6 days in Israel (AMF, 2018; Kadari-Ovadia, 2020). Another factor is the total length of school breaks, from less than 9 weeks in Mexico to nearly 18 in Ireland. There are also differences in the distribution of school breaks through the year. Although students in Luxembourg and Turkey have 15 weeks of vacation a year, it is split into six breaks in Luxembourg but two in Turkey (**Figure 17.6**).

A common pattern is for the longest school break to be at the end of the academic year, often during the summer. The history of this calendar, traditional in Europe and North America and now common across much of the world, is less likely linked to agrarian labour needs, as commonly believed, and more likely associated with a need for standardization as urbanization and income levels increased (Fischel, 2006; Melker and Weber, 2014).

### **FIGURE 17.6:**

**Countries vary widely in school break duration and organization** Number of weeks of school breaks in lower secondary education, selected countries, 2019



Many countries' school calendar's structure is due more to the influence of colonialism than seasons. The Bangladesh school calendar dates to the British colonial period and is not aligned with local agricultural cycles, requiring students to take exams during the peak wet-season harvest period (Ito and Shonchoy, 2020). Schools in Somalia run from September to June, likely an influence of British and Italian rule, unaligned with the country's warmer months and the school year in neighbouring countries, such as Kenya and Uganda (Kenya Ministry of Education, 2018; Uganda Ministry of Education and Sports, 2018; Wickman, 2011). Schools in southern hemisphere territories, such as American Samoa and French Polynesia, follow the northern hemisphere calendar, unlike most of their neighbours (Fischel, 2006).

School calendars are also influenced by cultural and religious practices. It is common for countries with a Christian background to have institutionalized breaks around Christmas and Easter, even if they are increasingly referred to as winter and spring breaks (Burke and Segall, 2011). In Muslim countries, school calendars may change every year to adapt to the moving Ramadan dates (Ito and Shonchoy, 2020). Indigenous schools in Canada, Peru and the United States use ancestral calendars that may take into account moon and seasonal cycles or cultural practices (Huaman, 2020).

School calendars may even differ within a given country. Breaks may vary among subnational units, due either to autonomous decisions or to purposeful staggering by the central authority (European Commission/EACEA/ Eurydice, 2020). Staggered calendars within a single school can allow groups of students go on vacation at different times.

Adaptation of school calendars to local contexts can have important equity implications. Using a natural experiment in calendar shifts, a studu showed that the overlap between seasonal labour demand and annual school examinations in Bangladesh led to a dropout rate seven percentage points higher for students from agricultural households. In India, the difference was estimated to be from five to seven percentage points. During peak harvest seasons, not only were students from agricultural households more likely to miss class, but also fatigue and injuries from field work hindered exam preparation (Ito and Shonchoy, 2020). School calendars' lack of resonance with local cultures has contributed to higher teacher absenteeism and lower attendance rates among children from Scheduled Tribes in India (Brahmanandam and Bosu Babu, 2016; Sujatha, 2002).

Some argue school calendars may also influence learning outcomes, especially for poorer students. The idea of the 'summer slide', a drop in achievement due to the long period away from classes, has led several schools to adopt year-round education that distributes school days more evenly throughout the year. Insufficient evidence makes it difficult to assess the impact of this change on overall performance, and some have questioned the magnitude of summer learning loss compared with loss distributed across the shorter breaks (Finnie et al., 2019; von Hippel and Hamrock, 2019).

### The organization of the school day also matters

Along with the number of school days, their length determines learners' total instruction time. Again, wide variation exists within and between countries – from an average of fewer than 600 hours of compulsory instruction per year in primary education in Latvia and the Russian Federation to over 1,000 hours in Chile, Costa Rica and Denmark (OECD, 2019). Differences across countries have likely been exacerbated by the COVID-19 pandemic. Students in many countries have received fewer instruction hours due to staggering of classes to reduce class size. In England (United Kingdom), by contrast, the government will extend the secondary-level school day to help make up for lost time (Hazell, 2021).

More instruction time is broadly associated with better student performance, but the effect tends to be mediated by factors such as instruction quality, classroom environment, school autonomy and accountability (Lavy, 2015; Woessmann, 2016; Yeşil Dağlı, 2019). It also depends on what 'more' means. Adding a few extra minutes may not make a difference; adding an hour may help; adding three may be counterproductive (Barshay, 2021). If well-used, more instruction time can help foster equity. Across a subset of PISA countries, more instruction time was associated with a greater likelihood of disadvantaged students succeeding academically (Agasisti et al., 2021). More and longer school days are also associated with increased participation of mothers in the labour market (Duchini and Effenterre, 2017; Shure, 2019).

Over time, there has been a global shift towards providing full-day education. Although double-shifting, where schools take in different groups of students for morning and afternoon sessions, remains common in Latin America and Africa, many countries, including Chile and Ghana, have moved to phase it out or abolish it (Ashong-Katai, 2013; Parente, 2020). In Namibia,

### 66

## Delaying the starting time by 50 minutes led to a significant improvement in student performance

the government's plans to phase out double shifting were hampered by COVID-19, which pushed many reopening schools back to it (Nakale, 2020; Rasmeni, 2017).

Increasing the length of the school day increases costs. In addition to requiring more school buildings, more teachers and longer work contracts, there may be significant infrastructure implications for existing schools. In Germany, primary schools traditionally offered classes only in the morning and so did not need to provide meals. As the country undertook nationwide reform to increase school days' duration over the past 15 years, many schools had to build cafeterias. Those that could not do so were often unable to change their schedules (Shure, 2019).

Beyond duration, school starting times also matter. A growing body of literature points to benefits from delaying starting times, particularly at the secondary level. In addition to allowing more sleep time, later starts appear to align better with adolescents' circadian rhythm, with peak alertness in the late morning and evening (Kelley et al., 2015; Kirby et al., 2011). A study using students' random assignments to earlier or later classes in the United States found that delaying the starting time by 50 minutes led to a significant improvement in student performance for all courses, not just the first period (Carrell et al., 2011). Some studies found even short delays helpful. In Hong Kong, China, a 15-minute delay from 7:45am to 8am was associated with greater attentiveness, fewer behavioural difficulties and better peer relationships between secondary school students (Chan et al., 2017). Others support pushing starting time much later. In the United Kingdom, a change from 8:50am to 10am was associated with fewer absences due to illness and improved academic performance (Kelley et al., 2017).

Finally, the organization of the school day is not only about instruction time. Recess has been shown to improve students' level of physical activity, memory and concentration, as well as their socioemotional development and academic performance (Haapala et al., 2017; Zavacky and Michael, 2017). Some countries consider breaks to be part of compulsory instruction time. Denmark and some Spanish regions and autonomous communities regulate recess time by law (OECD, 2019). Still, as daily recess is often seen as a waste of time, many schools do not offer it. In the United States, in the first five years after the 2001 No Child Left Behind Act, which focused on standardized testing, most districts increased time for tested subjects and 20% of districts decreased recess time by an average of 50 minutes per week (McMurrer, 2007). More recently, recess suffered another blow with COVID-19 as reopening schools tried to make up for lost time and found it hard to maintain physical distancing during free time. In Angola, the modified scheduled for the return of primary classes during the pandemic excluded recess (ANGOP, 2021).

99

### FOCUS 17.2: SCHOOL INFRASTRUCTURE MAY BE ABOVE THE MINIMUM STANDARD YET UNACCEPTABLE AND DETERIORATING

Policymakers should not consider investment in school infrastructure and the physical conditions for learning as secondary. This is recognized through SDG target 4.a and its global indicator that monitors school facilities, such as electricity, internet, drinking water and sanitation. However, given the variety of facilities and equipment that can affect learning, indicator 4.a.1 cannot offer a comprehensive basis for monitoring the adequacy of physical conditions within schools. Thus other indicators and potential information sources are needed.

Surveys that sample schools are one potential source of relevant data. TIMSS surveys head teachers on the extent to which inadequate physical conditions harm instruction quality. In nine countries that participated in the 2019 round, over a third of primary school head teachers reported that inadequate buildings or instruction space strongly affected instruction. In the latter case, the share exceeded 50% in Kuwait, Saudi Arabia and Turkey (**Figure 17.7**).

Even among Organisation for Economic Co-operation and Development (OECD) countries, head teachers reported that inadequate buildings or instruction space significantly hindered learning. Over 20% did so in Chile and Slovakia, and over 10% in Finland, France, Hungary, New Zealand and Sweden. Generally, head teachers in high-income countries are more likely to report a negative impact on learning from inadequate buildings and instruction space than from inadequate instructional

### 17

### **FIGURE 17.7:**

#### Instruction quality is hard to maintain in inadequate physical conditions

Percentage of primary school head teachers stating that inadequate physical conditions strongly affect instruction quality, selected countries, 2019



material and supplies. Among the 46 high- and upper-middle-income countries that participated in the Teaching and Learning International Survey (TALIS), only in 6 did lower secondary school head teachers report that a shortage of materials hindered learning more often (OECD, 2018). Inadequate materials and supplies are cited more often in poorer countries.

These figures indicate that many countries fall short of required levels of capital investment and non-recurrent expenditure in schools. Media reports highlight neglected repairs in 'death trap' public schools and instances where teachers pay out of pocket to replace outdated learning materials and equipment (Harris, 2018; Peek, 2018; Sedgwick, 2018). In Brazil, school census data showed that instruction space declined in urban schools between 2013 and 2017 while conditions remained unchanged in rural schools. Infrastructure quality was worse in schools of up to 150 students, which accounted for nearly half of students, and in schools with more disadvantaged students (Gomes and Duarte, 2017; UNESCO, 2019).

In the United States, it has been estimated that public expenditure would need to increase by 50% to meet the US\$146 billion required annually to adequately maintain, upgrade and construct new schools (Filardo, 2016). A government survey in 2019 showed that in 41% of school districts, more than half the schools

66

In Brazil, instruction space declined in urban schools between 2013 and 2017

**99** 

need to update or replace heating, ventilation and air conditioning. One in eight districts required repairs to improve the structural integrity of at least half their schools (US Government Accountability Office, 2020). The federal government contributes little to no funding for primary and secondary schools, which depend on state and district government funding, resulting in regional inequality (Filardo, 2016).

Engineers and other qualified personnel are needed to accurately determine building conditions and needed investment. However, governments generally do not conduct such assessments regularly at the national level, in part due to the complication and costs. In Canada, Ontario's provincial government hires a team of independent engineers to determine each school's repair and renewal costs. Results of these assessments are publicly released in the form of a 'facility condition index' (FCI), comparing the cost of a school's repair and renewal needs with the cost of completely rebuilding the school. Data from the 2015 assessment showed that over a third of the province's schools were found to be in 'poor' condition - an FCI between 0.30 and 0.65 (i.e. the repair cost was estimated at up to 65% of the cost of building a completely new school) (Ontario Ministry of Education, 2019; Sachgau, 2016).

In the absence of expert reviews, assessments from school staff on building conditions provide valuable information. Although such assessments are not as well suited to gauging buildings' structural integrity and resilience to natural disasters, evidence suggests they can be effective in diagnosing the conditions most strongly linked to learning outcomes. A study comparing engineering assessments to those by head teachers found the latter much more closely related to the quality of teaching and learning environments (Roberts, 2009).

The utility of educator facility assessment has relevance for the monitoring of SDG target 4.a. In particular, government-reported administrative data tend to yield more optimistic assessments of school facilities than those by head teachers. Where administrative data point to all primary schools having access to computers, head teacher assessments reveal shortcomings. Over half of head teachers in France, Kuwait, Oman, Portugal and Slovakia reported that inadequate access to computers moderately or strongly affected learning (Figure 17.8), yet the countries' administrative data show all schools having access to computers for pedagogical purposes.



### **FIGURE 17.8:**

### Head teachers underscore that access to equipment is not enough

A central dilemma of the pandemic, as yet unresolved, has been whether it is safe for children to go to school. Without preventive measures, schools are likely to be environments promoting infection, where large proportions of the population spend long hours together in relatively crowded indoor spaces, at an age where those infected are least likely to display recognizable symptoms. Still, almost two years into the pandemic, the net effect of school closures and reopenings on infection dynamics at the societal level remains inconclusive. In OECD countries, infection rates in the population appeared unrelated to the numbers of days schools were closed (OECD, 2021).

What has become clear is that minimizing infection risk in schools is possible. A study closely tracking infections in a large school in Chicago, United States, revealed a lower infection rate for students and staff participating in in-person learning than in the community overall (Fricchione et al., 2021). Adapting the school environment is crucial to providing a safe space for students. The cumulative global experience has resulted in guidance on processes that countries may go through when they reopen schools (UNESCO, 2021). These measures go well beyond the now everyday protective measures of masking, distancing and handwashing. Logical extensions of this hygiene regime include discouraging the sharing of objects and cleaning and disinfecting touched surfaces frequently. Applied conscientiously, these basic measures can significantly reduce infections (Asanati et al., 2021; Park et al., 2021).

Other school measures concern infrastructure use or additions. Even in normal times, lighting, temperature and air quality are sufficiently important to account in some contexts for half of learning outcome improvement (Barrett et al., 2019). With respect to COVID-19, temperature and air quality are even more crucial, as they affect the virus's survival rate and subsequent transmission rates (Huang et al., 2020). Low-tech solutions for improved ventilation include using outdoor spaces and opening windows, where seasonally appropriate, preferably with cross-ventilation. Even natural ventilation can significantly lower the infection risk, especially in combination with masks (Park et al., 2021).

In colder environments, high-efficiency particulate air (HEPA) filters can lower concentrations of viral aerosols. Retrofitting air filters is lengthy and cumbersome and few attempts have been made to deploy HEPA systems in schools at scale, even in high-income countries. However, they could be an important tool where ventilation is impossible or in classrooms with high-risk individuals. If retrofitting HEPA filters is not possible, portable filters can be used while developing guidelines for effective school ventilation and air filtration based on room volume, number and age of occupants and type of activity (Asanati et al., 2021).

Testing regimes vary. Vienna, Austria, took a test-intensive approach in 2021/22, with all students and teachers submitting saliva samples for lab-based polymerase chain reaction (PCR) testing several times a week (Stadt Wien, 2021). Just as important as testing capacity is what happens when infections are detected. To encourage compliance and buy-in from families, rules on quarantine triggered by COVID-19 cases in schools must strike a careful balance between complete closure and full opening, between accurate epidemiological evidence and messaging that is straightforward and simple to understand, and between school autonomy fitting the local context and consistent policy among schools.

Costly measures are difficult to implement in developing countries. In the early phase of the pandemic in 2020, a quarter of all countries – but half of all low-income countries – reported not having enough resources to ensure the safety of all learners and school staff (UNESCO et al., 2020). In early 2021, less than 10% of low-income countries reported having basic measures such as sufficient soap, clean water, masks, and sanitation and hygiene facilities to assure the safety of all learners and staff; the share in high-income countries was 96% (UNESCO et al., 2021). About 80% of low-income countries said external donors provided the funds to implement such measures.

In addition to measures to prevent transmission, extra support is needed to create safe environments, particularly for disadvantaged students, to address their mental health and gender-based violence and other issues that may have arisen or escalated during school closures or face students returning to school (UNESCO et al., 2021). Although systematic data are lacking, qualitative research shows instances of children reporting that school closures removed them from harm, such as bullying by peers (Klein, 2021), while other children faced increased harm being away from school and under restrictions. School reopening must not only minimize the infection risk, but also ensure that students return to a learning environment that is safe for all. Minimum Minimum

Mohammad, 21, is the first in his family to attend university. Through a DAFI (Albert Einstein German Academic Refugee Initiative) scholarship he is studying Arabic at Zarqa University in Jordan.

CREDIT: Mohammad Hawari/UNHCR

## **KEY MESSAGES**

After years of stagnation, overall scholarship aid increased by 30% between 2015 and 2019, predominantly favouring low-income countries. Yet because of an increase in the number of outbound students, the average student from a low-income country benefits from less scholarship aid now than in 2006.

A large share of scholarship aid does not specify recipient countries, making it difficult to monitor whether allocation is equitable.

Student teachers tend to be less mobile than students in other fields of tertiary education because of licensing requirements, lack of internationally recognized diplomas and attachment to local curricula. Among 24 OECD countries, 9% of higher education students were international, ranging from 3% in education to 12% in natural sciences, mathematics and statistics.

Teaching abroad has many benefits but teachers from poorer countries have far fewer opportunities. Only 2.2% of lower secondary teachers in Viet Nam have spent part of their training or careers abroad, compared with over half of teachers in many European countries.

Scholarship aid is still connected to trade and historical colonial ties, but network analysis shows that recipient countries have had access to aid from an increasing number of donors in recent years.

COVID-19 reduced inbound student mobility. With up to a third of students in Australia being international, this put higher education institutions in serious financial jeopardy.

### **CHAPTER 18**



## TARGET 4.b Scholarships

By 2020, substantially expand globally the number of scholarships available to developing countries, in particular least developed countries, small island developing States and African countries, for enrolment in higher education, including vocational training and information and communications technology, technical, engineering and scientific programmes, in developed countries and other developing countries

### **GLOBAL INDICATOR**

**4.b.1** - Volume of official development assistance flows for scholarships, by sector and type of study

C DG target 4.b called for the number of international scholarships for developing countries to 'substantially expand' by 2020. After years of stagnation, overall aid to support student mobility (global indicator 4.b.1) rose by 30% between 2015 and 2019, from US\$3.4 billion to US\$4.4 billion. The increase was largely the result of two factors: increasing bilateral flows with unspecified recipient countries, driven by EU institutions and Japan, which started reporting all of its scholarship aid under this category in 2017; and imputed student costs, largely related to refugee flows after 2015 to Germany, where they were eligible for entry into the country's essentially free tertiary education system (Figure 18.1). A sudden increase in unspecified scholarship aid in 2018 was largely a result of Turkey's inclusion as a donor in the Organisation for Economic Co-operation and Development Creditor Reporting System (CRS) database, although it is not a member of the Development Assistance Committee (DAC) (Box 18.1).

### 66

Total scholarship aid to low-income countries doubled over 2015–19, exceeding the growth in tertiary enrolment

99

Low-income countries benefited more from the aid increase in relative terms. Total scholarship aid to low-income countries doubled over 2015–19, exceeding the growth in tertiary enrolment. But taking both low- and middle-income countries into account, the number of outbound students far outpaced the growth in scholarship aid. On average, per international student, less scholarship aid was available in 2019 than in 2006 (**Figure 18.2**).

To date, the private sector has played a relatively minor role in providing scholarship opportunities for the poorest countries. A mapping of more than

### **FIGURE 18.1:**

Scholarships and imputed student costs increased by US\$1 billion over five years Volume of aid to education disbursed, by type and recipient, 2010–19





**FIGURE 18.2:** 

### Scholarship aid has not kept pace with demand

Relative change in outbound tertiary mobility, tertiary enrolment and scholarship aid received, by country income group, 2006–19



200 scholarship providers for students from sub-Saharan Africa for the 2020 GEM Report found that, among the top 50 providers, corporations, foundations and private donors together accounted for less than one in seven scholarships (**Figure 18.3**). Most came from Mastercard Foundation and the Absa banking group, which together provided 95% of corporate scholarships. China stands out as the largest provider among governments, and is now the second-most-common destination for African student migrants. The global student mobility network is slowly becoming less centralized (Mulvey, 2021) (**Focus 18.1**).

The increasing share of recorded scholarship aid whose recipient countries are unspecified reduces the ability to monitor the amount going to individual countries and country groups. It also makes it harder to analyse the aid distribution. Hardly any scholarship aid has gender as its primary focus, and scholarship aid that does not have a specified recipient country is even more likely to have no gender priority (**Figure 18.4**).

The patchy data available suggest that the aim of a substantial expansion in scholarships has not been met. Student mobility is higher in information and communication technology, technical, engineering and scientific programmes, which are specified in target 4.b, than in education as a field (**Box 18.2**). But there is no evidence to suggest that these fields of study have been prioritized for scholarships.

### **FIGURE 18.3:**

Private actors provide less than one in seven scholarships for sub-Saharan Africa

Share of top 50 scholarship providers to sub-Saharan Africa, by category, 2019



Out of 22 Latin American and Caribbean countries, 8 maintain formal brain circulation networks

Still, target 4.b should not be seen as having lapsed in 2020. The challenge of limited domestic higher education capacity and opportunities in the least developed countries and Small Island Developing States remains. And the passing of the 2020 deadline provides an opportunity to reconsider the target's purpose and spirit. It is the only SDG 4 target with an explicitly international dimension, in terms of both what is to be achieved and how. One concern has been that, narrowly understood, the target benefits a small, select group. However, a broader understanding is that scholarships aim to contribute to development goals, which will benefit many more (**Box 18.3**).

The concept of 'brain drain', where scholarship alumni do not return to their countries of origin, is being replaced by a more sophisticated understanding of 'brain circulation'. Recent estimates suggest that return migration represents a significant part of migration flows to sub-Saharan Africa and Latin America. These migrants, especially younger adults, are more educated than the norm, on average (Chen et al., 2021).

Some countries are recognizing that even highly skilled nationals who will not return in the foreseeable future represent an asset if properly engaged. Out of 22 Latin American and Caribbean countries analysed for the Emigrant Policies Index, 8 maintain formal brain circulation networks (Hoffmann et al., 2017; Pedroza and Palop-García, 2017). An earlier mapping of diaspora policies of 35 countries, representing all world regions, income levels and government types, found that two thirds maintained scientific networks of some kind and half had return obligations for students sent abroad through scholarships (Ragazzi, 2014).

### BOX 18.1:

"

### Turkey has become one of the largest scholarship providers

In the wake of the Syrian conflict, Turkey now hosts the most refugees in the world. As a result, Turkey effectively became one the largest humanitarian donors in absolute terms and as a percentage of GDP, although direct comparisons are complicated by the fact that other donors report only international humanitarian assistance (Development Initiatives, 2020).

In recognition of this development, Turkey gained DAC observer status in 2016, and is included in CRS for 2018 (when almost all its entries were coded as regional or with an unspecified recipient country) and 2019 (with more complete recipient information). Turkey's inclusion in international aid statistics as a donor has resulted in a noticeable break in the trend and exaggerates the recent increase in scholarship aid. While its emergence as a major humanitarian donor was driven by necessity, Turkey has at the same time become one of the biggest donors of scholarship aid by choice.

Turkey offered 17,000 international scholarships in 2019 (Turkish Cooperation and Coordination Agency, 2020), chiefly through the Türkiye Burslari (Turkey Scholarships) programme. These scholarships, offered at undergraduate and postgraduate levels, provide tuition, accommodation, a one-off return flight, health insurance and a monthly stipend (Turkey Ministry of Culture and Tourism, 2021). As a result, excluding imputed student costs, in 2018 Turkey was the single largest scholarship donor in the CRS database (which notably does not include China) at US\$225 million, just ahead of EU institutions. It dropped to fifth place at US\$150 million in 2019, between the United Kingdom at US\$138 million and Saudi Arabia at US\$162 million.

Turkey's scholarships include language courses. The country is also expanding opportunities, not just for prospective students, to study Turkish before arrival through its growing network of Yunus Emre Institutes around the world. It is particularly expanding the network in Africa (Anadolu Agency, 2021) as part of a larger strategy of rapidly increasing the Turkish presence on the continent in trade and diplomatic relations in addition to academic exchange (Mitchell, 2021).

### **FIGURE 18.4:**

### Scholarship aid does not have a strong gender focus Scholarship aid by gender focus, 2010–19



a. Recipient country/region specified

Note: The gender marker in the OECD-DAC CRS database was introduced in 2017. Data prior to 2017 are the result of a one-off pilot exercise to code historic entries.

GEM StatLink: https://bit.ly/GEM2021\_fig18\_4 Source: OECD-DAC CRS database.





Note: The gender marker in the OECD-DAC CRS database was introduced in 2017. Data prior to 2017 are the result of a one-off pilot exercise to code historic entries.

Source: OECD-DAC CRS database.

### FOCUS 18.1: SCHOLARSHIP AID FLOWS ARE BECOMING LESS CONCENTRATED

Scholarship aid flows represent a network of connections between donor and recipient countries.<sup>1</sup> Network analysis, a research field that traces relations between actors, can shed light on how the spatial pattern of scholarship aid flows relates to flows of international students, and to geographical, trade, cultural and historical links (Shields and Menashy, 2019).

The more equal the number of connections among actors, the less centralized the network. By several measures, the scholarship aid network became less centralized in the 2010s and flows have become more dispersed, i.e. more evenly distributed among recipients. In conjunction with the SDG targets, this trend may also be explained by an increasing role of scholarships in cultural diplomacy and the wider exercise of soft power (Campbell and Neff, 2020). Still, recorded scholarship aid flows represent only around 10% of all possible connections between donors and recipients.

Three groups of donors can be identified. Some report scholarship aid to only one or two countries, perhaps reflecting key bilateral relationships. Another group sends aid to some 10 to 20 countries, perhaps due to regional or historical connections. The third group provides scholarship aid to over 20 countries, reflecting a broad or even near universal approach that likely extends beyond key bilateral relationships. Donors are now likely to provide scholarships to more developing countries than in 2015, and, more importantly, recipient countries are less likely to be dependent on one or two key donors.

<sup>1</sup> This section is based on Shields (2021).

Some geographical clustering is still evident. The Czech Republic, Hungary, Poland and Romania provide scholarship aid to a similar set of countries. Among the recipients, Armenia, Belarus, Georgia and Ukraine have many donors in common.

Diversification of recipients can reflect deliberate policy. China's provision of scholarships to Africa started in 1972 with 200 scholarships for students from the United Republic of Tanzania and Zambia (Dong and Chapman, 2008). The government has since considerably increased the number – in 2018, it issued over 63,000 international higher education scholarships (Jing, 2020) – and diversified recipients' nationality. Between 1999 and 2015, nearly all African countries received Chinese government scholarships for students to pursue higher education in China, and 14 countries had over 1,500 recipients each (Ha et al., 2020).

66

The Chinese government issued over 63,000 international higher education scholarships to Africa in 2018

99

### BOX 18.2:

### Teacher international mobility is common in some contexts

Including an international experience in teachers' education and training can help them broaden their world view, develop cultural sensitivity and acquire global competencies (Baecher, 2021; Jaritz, 2011). As SDG target 4.7 calls for promotion of global citizenship and appreciation of cultural diversity in education, fostering these competencies among students must start with developing them in teachers.

Yet education is the least mobile field of study. Among 24 OECD countries, 9% of higher education students were international, ranging from 3% in education to 12% in natural sciences, mathematics and statistics. In Australia, over 25% of all higher education students were international, compared with just 7% of tertiary students in the field of education (**Figure 18.5**). Reasons for low mobility in this field include national licensing requirements, lack of international recognition of diplomas and a belief that teacher education must be anchored in local curricula (Jaritz, 2011; Witt and Liu, 2021). In Europe, students in education tend to be older and hold jobs; they are also more likely to have children and be the first of their family in higher education, all of which help explain their under-representation in enrolment abroad (Vögtle, 2019).

International experiences can be part of teachers' in-service professional development. The value of teacher exchange programmes has long been recognized and is included in the principles of the 1966 ILO/UNESCO Recommendation concerning the Status of Teachers. Exposure to different pedagogical methods and curricula and the opportunity to share expertise can benefit teachers in both source and destination countries (Caravatti et al., 2014).

In over half the countries and economies participating in the Teaching and Learning International Survey, teachers who have been abroad for professional purposes, either as a teacher or during teacher education, engage more often in professional collaboration and report higher levels of self-efficacy (OECD, 2020b). In a survey by Education International of over 1,000 teachers with experiences abroad, nearly all reported a positive effect on their instruction practice, including an enhanced ability to work with students with diverse needs and a broader world view and cultural competency (Caravatti et al., 2014).

The prevalence of teacher experiences abroad, however, is still closely linked with country income level (**Figure 18.6**). Only 2% of lower secondary school teachers in Viet Nam have been abroad for professional purposes, compared with over half in many European countries. Indeed, increasing teachers' mobility in Europe was a key objective of the Bologna process, with incentives provided by programmes such as Erasmus+ offering opportunities to study and teach abroad (European Commission, 2017; Iyevlyev, 2018; OECD, 2020b).

Many smaller exchange programmes have been developed in or with low- and middle-income countries. The Chinese government has partnered with institutions in Australia, Canada, New Zealand, the United Kingdom and the United States to send teachers overseas for professional development and language training (Vasilopoulos and Romero, 2021). The US government-funded Instructional Leadership Institute for Pakistan Educators provides opportunities for Pakistani teachers to go on training programmes in the United States (Woodland, 2021). Partnerships between institutions in various countries also help facilitate exchange programmes. An agreement between universities in Botswana and the United States allows for bidirectional study-abroad experiences for teachers (Rose and Cooper-Duffy, 2021). Institutions in high-income countries may also partner with non-governmental organizations abroad to send student teachers to work alongside local teachers (Tripp et al., 2021).

### BOX 18.2 CONTINUED:

### **FIGURE 18.5:**

### Education is the least mobile field of study

International students as a share of all tertiary education students, by host country and field of study, 2018



GEM StatLink: https://bit.ly/GEM2021\_fig18\_5 Source: OECD (2020a).

### **FIGURE 18.6:**

### Teachers in high-income countries are much more likely to go abroad

Share of lower secondary teachers who have ever been abroad for professional purposes



Chinese scholarship programmes are also open to applicants from any developing country, including specific scholarships for nationals of countries in the Association of Southeast Asian Nations and the European Union, along with Pacific Island states (China Scholarship Council, 2020). Still, bilateral trade and diplomatic relations are strong predictors of China's international scholarship recipients' nationalities (Ha et al., 2020).

Indeed, analysis of how bilateral scholarship aid relates to other networks shows that it tends to be more frequent between countries where international student mobility is generally higher, to countries where donors export goods or services, and to former colonies. The direction of some of these associations is ambiguous. For example, increased scholarship aid may be both an outcome of high levels of student mobility and a contributing factor. The continued association between scholarship aid and past colonial ties should prompt critical discussions on the decolonization of higher education. In some cases, the link is intentional, as with Commonwealth Scholarships offered by the United Kingdom. By contrast, Turkey explicitly views its lack of colonial history as an asset in expanding its ties in Africa (Ünveren, 2021), including by providing scholarships (Box 18.1).

Another interesting question is which recipient countries receive scholarship aid above what would be expected from their position in trade, mobility, cultural and other networks. Some of the least developed countries, including Mali and Mauritania, along with Small Island Developing States such as Comoros and Samoa, attract relatively high scholarship aid.

Overall, application of social network perspectives provides greater insights into scholarship aid trends than would be possible through aggregated trend analysis alone. The social network between donors and recipients is complex and changing rapidly. While the analysis shows that scholarship aid is still connected to trade and historical colonial ties, it also shows that recipient countries have access to aid from an increasing number of donors. This trend represents at least qualitative progress towards target 4.b.

### BOX 18.3:

### Scholarships should benefit more than just the scholars

Target 4.b is a 'means of implementation' rather than an end, and it is important to ask how international scholarships for students from low-income countries contribute to sustainable development generally, and SDG 4 in particular.

Beyond individual benefits, research on scholarships has examined their contribution at higher levels (Mawer, 2018). Assessing their impact often relies on graduates' subjective self-reporting. Robust evaluation requires long-term tracking, which can be difficult and costly. One successful example is the Ford Foundation's 10-year Alumni Tracking Study, administered by the Institute of International Education, with reports and briefs available online. Germany has tracked how scholarship programme alumni experiences compare with the 2030 Agenda (GIZ, 2020). Some alumni associations have built effective networks of advocates for change, affecting policy in the home country (Campbell and Baxter, 2019; Martel, 2017).

Many subject-specific scholarships contribute to technical capacity building in support of SDGs, such as SDG 7 on renewable energy. Some scholarship programmes are not limited to technical subjects; the Rotary Foundation master's in peace studies fellowship is aligned with SDG 16, for instance. Research conducted with scholarship providers for the GEM Report suggests that many applicants are motivated by the SDGs and seek courses aligned with them (Campbell, 2021).

Support given to alumni in terms of professional development, internships and networking opportunities prepares them more ambitiously to contribute to their countries' economic development, hire others and usher innovative ideas to specific fields, contributing to SDG 9. Ghanaian and Nigerian scholarship alumni of the Ford Foundation's International Fellowships Program contribute to teaching new curricula at university level in their home countries (Campbell et al., 2021).

Outside of professional activities, alumni activities and networks contribute to the SDGs through campaigns and volunteer projects, such as mentoring youth in entrepreneurship initiatives. Other scholarship alumni volunteer in activities such as educating citizens on voters' rights, the rule of law and tolerance, thereby contributing to the focus of SDG target 16.6 on developing effective, accountable and transparent institutions at all levels. Indeed, international student mobility creates 'foreigneducated individual actors' (Chankseliani, 2018, p. 281) who can help increase levels of democracy and participation. An alumni association in Ghana 'established working groups around thematic areas, tying many of the focal areas' to the SDGs (Campbell and Lavallee, 2020, p. 416) . Some scholarship providers support their alumni in their activities, e.g. by providing small grants for alumni projects (Campbell and Baxter, 2019).

### COVID-19

Travel restrictions related to COVID-19 reduced international student mobility and financial stability, affecting scholarship aid opportunities. Graduate students were particularly affected (Mercado, 2020). Over 60% of international students at graduate level in the United States depend on international funds (Di Maria, 2020), a situation also typical for other host countries. Both inbound and outbound mobility were affected. In Finland, 48% of international students returned home, and 53% of Finnish students abroad returned to Finland (Finnish National Agency for Education, 2020).

Early evidence suggests the high infection rates in the United States dissuaded students from applying for scholarships at US universities (Reardon, 2020), as did concerns about racial discrimination against Chinese people (Peters et al., 2021). Other popular anglophone international student destinations, such as Australia, New Zealand and the United Kingdom, also experienced decreased inbound mobility. With up to a third of students in Australia being international, this put Australian higher education institutions in serious financial jeopardy (Waters, 2021). The loss in income from fee-paying international students reduces funds available for institutional scholarships for students from poor countries who cannot pay their own tuition.

China was affected by COVID-19 early and imposed strict containment measures, including stopping international travel. This had a significant impact on target 4.b. In recent years, China had become the third-largest receiving country of international students (Waters, 2021). For African students, the decision to study in China strongly reflects scholarship availability and the relative affordability of tuition and living costs (Lei et al., 2021).

Not only were many students no longer able to accept scholarships, but other students and graduates were stranded in host countries when they were expected or expecting to return to their home countries. This happened in Turkey, which has recently adopted ambitious, proactive policies, projects and efforts to attract international students (Peters et al., 2021), but lacked an ecosystem of support networks and services.

Many scholarships do not cover all expenses, and students, including scholarship holders, often rely on part-time work to fully fund their studies and living. Such students were dealt a major financial blow with the closure of bars, restaurants and libraries where many students worked, while others lost their part-time jobs due to the general economic downturn (Bilecen, 2020).

Some universities in the United States, such as the University of Oklahoma, set up pandemic scholarships designed to provide relief to international students financially affected by COVID-19 or associated restrictions (University of Oklahoma, 2020). The Institute of International Education's Emergency Student Fund helped unemployed international students who struggled to pay for food, shelter and medical care during the pandemic (IIE, 2020).

Some scholarship opportunities disappeared because they would have required physical travel to the host country campuses. Sport scholarships, for instance, provide a route into higher education for disadvantaged groups not only domestically, but also internationally. Kenyan students reported being able to study in neither the United States nor in neighbouring Ugandan universities when sports scholarships were cancelled (Odhiambo, 2020).

Distance learning has enabled developing country international students with decent internet and computer access to take up scholarship opportunities for higher education institutions and degree courses abroad despite the pandemic. However, they do not benefit from the cultural enrichment of living abroad, and have lost opportunities to find jobs in host countries through visa programmes enabling students to stay for work upon graduation (Yildırım et al., 2021). Many students from developing countries view the higher fees in host countries as an investment to recoup through a job at host country salary levels. Host countries differ in their plans to revise or keep such visa programmes in place (Bilecen, 2020).

As universities reopen campuses, they need such visa programmes and student social safety nets to regain their attractiveness for international students and affordability for scholarship holders. COVID-19 may end up resulting in a more permanent shift to online learning for some courses and programmes, coupled with reduced fees (Waters, 2021), making programmes more accessible.

Even with such changes, mobility of students from South Asian and sub-Saharan African countries will take the longest to recover (Marginson, 2020), not least because of global vaccine inequality. Young people in these countries are likely to be last in line for vaccination, which may be required for international travel or university admission.
Discriminatory practices, such as the United Kingdom's refusal to accept vaccinations performed in certain countries, even with vaccines donated by COVAX, the global vaccine-sharing initiative that is also financed by the United Kingdom itself, created additional obstacles (Princewill, 2021).

Monitoring of target 4.b is also affected by constrained student mobility. First, comparisons over time must carefully distinguish between direct scholarship aid and imputed student costs if the pandemic's effects are to be properly understood. In 2019, 62% of international aid for higher education students represented imputed costs of students from developing countries attending higher education in donor countries with tuition-free systems, notably France and Germany, rather than scholarships. As imputed student costs arise at the point and time of delivery, reduced international student flows to fee-free destinations automatically reduce this aid component, while scholarships can continue to be awarded during the pandemic, even if actual travel is deferred.

Second, target 4.b calls for increased provision of scholarships enabling 'enrolment ... in developed countries and other developing countries'. The pandemic is likely to accelerate trends towards remote delivery of higher education and phenomena such as local branch campuses of institutions from donor countries. Thus it will be increasingly necessary to clarify the status of scholarship recipients who remain in their home countries but enrol in developed country institutions, either remotely or in person at a branch campus. T. Ezhilarasi teaches a student at a house in the town of Villivakkam, India.

CREDIT: UNICEF/Srishti Bhardwaj

# **KEY MESSAGES**

Sub-Saharan Africa has the lowest percentage of teachers meeting national standards. Pupil/trained teacher ratios are twice as high as the global average. One estimate suggests the region will need to recruit 15 million teachers by 2030 to support universal enrolment.

The qualifications of a majority of the world's teachers remain unknown.

The 2018 TALIS showed that in 48 education systems, 76% of lower secondary school teachers had attended courses or seminars and 72% had read professional literature during the 12 months before the survey.

Teaching out of field is prevalent in much of the world. In at least 40 education systems, over 10% of lower secondary school science and mathematics teachers had received no formal education or training in the subject. In Georgia and Saudi Arabia, less than 60% of science and mathematics teachers had received training in their subjects as part of their formal education.

Teachers tend to be paid less than comparable professionals in high-income countries, but more in some low- and middle-income countries.

In January 2021, nearly one quarter of teachers in the United States indicated the desire to leave their jobs due to COVID-19 conditions, compared with an average national turnover rate of 16% prior to the pandemic.

# **CHAPTER 19**



**TARGET 4.c** Teachers

By 2030, substantially increase the supply of qualified teachers, including through international cooperation for teacher training in developing countries, especially least developed countries and small island developing States

# **GLOBAL INDICATOR**

4.c.1 – Proportion of teachers with the minimum required qualifications, by education level

## **THEMATIC INDICATORS**

4.c.2 – Pupil-trained teacher ratio by education level

**4.c.3** – Percentage of teachers qualified according to national standards by level and type of institution

4.c.4 - Pupil-qualified teacher ratio by education level

**4.c.5** – Average teacher salary relative to other professions requiring a comparable level of qualification

4.c.6 - Teacher attrition rate by education level

**4.c.7** – Percentage of teachers who received in-service training in the last 12 months by type of training

#### **TABLE 19.1:**

Pupil/trained teacher ratio, by region and country income group, 2015 and 2020 or latest available year

	Pre-primary				Secondary	
	2015	2020	2015	2020	2015	2020
World			27.1	27.3	19.8	19.6
	62.5	60.0	58.0	55.8	36.6	33.9
Northern Africa and Western Asia	22.7	24.1	22.0	23.8	17.7	17.1
Central and Southern Asia			41.9	36.7	31.5	24.9
	25.1	24.1	26.0	25.2	19.5	20.7
Europe and Northern America						
	72.6	67.3	56.0	53.0	37.1	34.3
			35.1	33.6	27.3	24.1
Upper-middle-income						
High-income						

99

Source: UIS database.

Evidence continues to accumulate showing that effective teachers have a large impact on student outcomes, although evidence is mixed on the effects of interventions to improve teacher qualifications, skills or working conditions. Awareness of teaching's challenges and appreciation of teaching professionals' contribution have grown as school closures due to COVID-19 forced many parents and other caregivers into roles as substitute teachers.

Global indicator 4.c.1 has been retitled 'Proportion of teachers with the minimum required qualifications, by education level', a refinement approved by the Inter-agency and Expert Group on SDG Indicators in April 2020 and confirmed at the 52nd session of the Statistical Commission in March 2021 (UIS, 2020). The indicator's content has not changed; it aims to capture how many teachers have at least the minimum qualifications required for teaching, according to national standards.

## 66

Pupil/trained teacher ratios are almost twice as high in sub-Saharan Africa as the global average Yet, since data on indicator 4.c.1 remain patchy, there is no estimated global average for all levels. The concept of an average across countries has also been questioned, as definitions vary. This situation has given rise to an initiative for an international standard classification of education for teacher training programmes (**Box 19.1**).

Sub-Saharan Africa is the region with the lowest percentage of teachers meeting national standards: 57% in pre-primary (vs 83% in Latin America and the Caribbean), 67% in primary (vs 85% in Northern Africa and Western Asia) and 61% in secondary education (vs 78% in Central and Southern Asia). Hence pupil/trained teacher ratios (thematic indicator 4.c.2) are almost twice as high in sub-Saharan Africa as the global average, despite some small improvement since 2015 (**Table 19.1**). Many high-income countries do not report such statistics, because tight regulation of the teaching profession fosters the assumption that all teachers meet minimum qualifications as a statutory requirement. But being qualified to teach in general does not mean teachers are necessarily qualified for the subject they teach (**Focus 19.1**).

At the national level, average pupil/teacher ratios obscure major inequality between schools. The 50% of students attending the least well-staffed schools are taught

# BOX 19.1:

## A new initiative aims to improve comparability of teacher qualifications

The problem with the SDG target 4.c indicators is that international definitions of 'trained' and 'qualified' teachers are lacking. Low-income countries, such as Niger, require primary school teachers to hold an upper secondary education diploma in teacher training (Tatto, 2020); high-income countries, such as the Czech Republic and Finland, require lower secondary teachers to have a master's degree (European Commission/ EACEA/Eurydice, 2019).

In 2019, the 40th session of the UNESCO General Conference endorsed development of an International Standard Classification of Teacher Training Programmes (ISCED-T) to support the monitoring of SDG target 4.c. ISCED-T is a framework in which to assemble, compile and analyse cross-nationally comparable statistics on teacher training programmes. It is based on and complements the International Standard Classification of Education, which is the framework for comparing overall education programmes and related qualifications by level and field. A technical advisory panel established by the UNESCO Institute for Statistics (UIS), in partnership with the International Task Force on Teachers for Education 2030, has been charged with its development.

Analysis of a UIS inventory of 814 national teacher training programmes from 196 countries and territories identified five dimensions as the basis for the classification. Single-digit codes were assigned to each dimension: the ISCED level of the training programme (from 2, i.e. lower secondary education, to 8, i.e. a doctoral programme); the teaching level in which graduates are authorized to teach (from 0, pre-primary education, to 9, lower and upper secondary education); education prerequisites for entering the programme (from 1, primary education, to 8, a doctoral programme); duration of the programme (from 1, up to a year, to 7, more than six years); and the teaching practice ratio (from 0, none, to 3, 20% or more) (UIS, 2021). Each programme thus is assigned a five-digit number. Other criteria considered, but not included, were the pathways to the teaching profession (concurrent, consecutive and alternative); type of institution; proportion of academic and pedagogical content; and probation/ induction support.

Following a global consultation with UNESCO Member States and relevant experts, a proposal was presented for adoption to the 41st session of the UNESCO General Conference in November 2021. Once requested revisions by the General Conference have been incorporated, the final ISCED-T will be disseminated and UIS will develop a data collection strategy, guidelines, and training for implementation and roll-out. ISCED-T will mark a significant step in producing internationally comparable data on teacher training programmes, especially pre-service teacher education. However, data improvement will be specific to new teachers (flows); comprehensive data on the qualifications of the entire teacher population (stock) will take longer to materialize.

by 38% of teachers in Cabo Verde, 34% of teachers in Kyrgyzstan and 30% of teachers in Cambodia. As a result, the pooled primary pupil/teacher ratio of 44:1 is considerably lower than the average pupil/teacher ratio from the student perspective, which is 61:1 (**Box 19.2**).

The number of teachers globally has been increasing steadily in primary and lower secondary education, reflecting growing enrolment and efforts to reduce pupil/teacher ratios. However, while data availability on qualification status increased in 2015 in response to the need to report on the SDG global indicator, data coverage has not further improved in recent years (**Figure 19.2**). For a majority of the world's teachers it is not known whether they meet national minimum qualifications, although the number of teachers known to be unqualified is small. Countries with lower administrative and statistical capacity to report teacher qualifications are also more likely to have a relatively high number of unqualified teachers. The number of teachers at the upper secondary education level appears to have declined in the early 2010s (Figure 19.2c). This drop is largely accounted for by reporting from India: After a gap in 2011, it reported almost 800,000 fewer upper secondary school teachers in 2012 than in 2010. This decrease was offset bu increased numbers of post-secondary non-tertiary, lower secondary and, especially, primary school teachers. Rather than actual reallocation, however, there had been a shift in reporting. Before 2010, populous states, such as Andhra Pradesh, Karnataka and Maharashtra, reported grade 8 as part of the secondary level. After the 2009 Right to Education Act, the states were obliged to report class 8 as elementary level, with funding contingent on this reporting. Also, under the Rashtriya Madhyamik Shiksha Abhiyan secondary education programme, new guidelines took effect in 2011 for reporting to the Unified District Information System for Education by schools spanning several education levels, whose reporting method had previously differed.

# Aggregating unequal pupil/teacher ratios

Aggregate data on pupil/teacher ratios can easily be misinterpreted. Imagine 2 classrooms with 1 teacher each, one with 20 pupils and the other with 80 pupils. Overall, with 100 pupils to 2 teachers, the pooled pupil/teacher ratio is 50. This is also the average pupil/teacher ratio that one of the teachers, randomly selected, would experience. From the student perspective, however, the large classroom is much more likely. On average, a randomly selected pupil would experience a pupil/teacher ratio of 68, the result of a calculation accounting for such a pupil having an 80% probability of being in a classroom with 80 pupils and a 20% probability of being in a classroom with 20 pupils.

Both ratios are accurate but correspond to different perspectives. The aggregate pupil/teacher ratio of 50 better captures the teaching conditions experienced by the average teacher and the system's resource intensity. However, the aggregate pupil/teacher ratio of 68 is a better representation of the learning environment experienced by the average pupil (**Figure 19.1**).

#### **FIGURE 19.1:**

A random pupil is more likely than a random teacher to be in an overcrowded classroom



A recent estimate suggests that sub-Saharan Africa will need to recruit 15 million teachers by 2030 to staff schools for universal enrolment at a maximum pupil/ teacher ratio of 40 in primary and 25 in secondary schools (UNESCO, 2021). Of this number, 8.7 million will be needed to fill additional posts to keep up with growing enrolment and 6.3 million to replace teachers leaving the profession. Some of the poorest countries, including Central African Republic, Chad, Malawi, Mozambique and Niger, would need to increase the number of secondary teachers by at least 15% every year.

The percentage of teachers who received in-service training in the past 12 months (thematic indicator 4.c.7) is not routinely reported by most countries. Monitoring currently relies on the Teaching and Learning International Survey (TALIS); in its 2018 round, 48 education systems took part. The study is a rich source of information on teacher professional development, including its modality and content. Traditional in-person seminars and independent study remain the most common forms of teacher professional development: 76% of lower secondary school teachers had attended courses or seminars and 72% had read professional literature during the 12 months before the survey. Participation in online seminars was less common, although in some education systems, notably in the Republic of Korea and in Shanghai, China, participation in such courses was close to universal even before the pandemic. The next round of TALIS in 2024 will offer a chance to observe the extent to which COVID-19 has had a lasting effect on shifting teacher professional development modalities online (Figure 19.3).

Unsurprisingly, teacher in-service training most commonly focuses on subject and curriculum knowledge, as well as pedagogical and assessment practice. In almost all countries, more than half of all lower secondary school teachers participate in professional development activities on such topics in a given year, and almost all teachers did in Viet Nam. By contrast, less than half of all teachers in most countries benefit from recent training on inclusive approaches to multicultural classrooms.

66

Traditional in-person seminars and independent study remain the most common forms of teacher professional development

99

#### **FIGURE 19.2:**

Information on teacher qualifications is incomplete

Number of teachers, by qualification status and education level, 2000–20



#### **FIGURE 19.3:**

**In-person courses and private study remain the most common forms of teacher professional development** Lower secondary school teacher participation rate in professional development during the previous 12 months, by modality, selected education systems, 2018



#### **FIGURE 19.4:**

**Few teachers in France receive training in topics related to inclusive education, while most do in Viet Nam** Lower secondary school teacher participation rate in professional development during the previous 12 months, by content, 2018



"

Teachers in France reported the lowest level of professional development. They were least likely to have participated at all in the past 12 months, and they report the fewest activity types (**Figure 19.4**). Despite high levels of inclusion challenges in France's diverse classrooms, fewer teachers than in any other TALIS country benefit from professional development for teaching in multicultural or multilingual settings or communicating with people from different cultures or countries.

The characteristics that make teacher professional development programmes successful are multifaceted. But a study of 139 in-service teacher training programmes in 14 low- and middle-income countries showed that few applied effective practices associated with higher learning gains (Popova et al., 2018).

# 66

In high-income countries, teachers tend to be paid less than comparable professionals in other sectors UIS has reported new estimates on the teacher salary indicator, which examines how teachers fare relative to other professions requiring a comparable level of qualification (thematic indicator 4.c.5). Generally, average differences among teachers at different education levels within the same country are smaller than differences among countries (**Figure 19.5**).

In high-income countries, teachers tend to be paid less than comparable professionals in other sectors. In Finland, a country known for its strong record in international learning assessments and teachers' high social status, they are paid about 25% less than comparable professionals. By contrast, in the Republic of Korea, teachers are paid about 20% more than other professionals.

By comparison, in low- and middle-income countries it is more likely that teachers earn higher salaries than those of other professionals, although data are available only in a few cases. Such relatively high salaries do not necessarily mean teachers enjoy attractive and competitive compensation packages. Measurement challenges obscure comparisons. For instance, formal labour markets for highly educated workers are weakly developed in other sectors and the income levels of such professionals are not captured to allow a reliable comparison.

#### **FIGURE 19.5:**

Relative to other professionals, teachers tend to earn lower salaries in high-income countries but higher salaries in some low- and middle-income countries

Average teacher salary relative to other professions requiring a comparable level of qualification, latest year available in 2015–19



Comparisons of teachers with other professionals by qualifications should control for other factors. A study of teacher salaries in 15 sub-Saharan African countries found that, after adjusting for education, age, gender and location, the number of countries where teachers earned a salary premium relative to comparable workers dropped from 10 to 5, and teachers had a lower salary in 7. At the same time, teachers were reported as working fewer hours than other workers in all 15 countries. An unconditional comparison of teachers with other workers suggested they were paid more on an hourly basis in all but two countries (Democratic Republic of the Congo and Niger). After adjusting for all factors, they had a premium in only seven countries and a disadvantage in one country (Nigeria) (Evans et al., 2021).

Salary incentives can motivate teachers to teach in schools difficult to staff due to remote location or other factors. But the amount of incentives offered is often not enough. In Peru, an analysis of teacher preferences indicated on applications to a centralized school assignment system found that attempting to fill all vacancies through salary incentives alone would double the salary bill. It would take six times current salary spending to equalize teacher quality across Peru. However, both figures are based on existing teacher stock. An alternative to incentives for reluctant teachers from urban areas to teach in rural schools would be to train more local teachers. Increasing the number of teachers from rural areas by as little as 3% could result in cost estimates that are 30% to 35% lower (Bobba et al., 2021).

The teacher salary indicator is meant to be a proxy for teacher motivation. But a recent analysis of considerable teacher absenteeism in eight eastern and southern African countries suggests many more factors affect motivation (Karamperidou et al., 2020). Even teachers' own reporting indicates the share of those absent from school at least once a week ranges from nearly 10% in Kenya and Rwanda to nearly 30% in South Sudan. Absenteeism

66

In Peru, increasing the number of teachers from rural areas by as little as 3% could result in cost estimates that are 30% to 35% lower

is marginally higher in rural areas than in urban and peri-urban areas and in public than private schools. School factors are less important than personal ones. Teachers say they are absent on health (62%) and family grounds (35%), followed by weather (especially heavy rain and excessive heat), official business and transport issues. The report recommends focusing on teacher monitoring and accountability and on ensuring that any training takes place during holidays and weekends – in contrast to high-income countries, where research has noted the importance of freeing up time in teachers' schedules for training (De Neve and Devos, 2017; Owen, 2014).

# FOCUS 19.1: MANY TEACHERS ARE TRAINED AND QUALIFIED BUT NOT FOR THE SUBJECT THEY TEACH

General measures of teacher qualification and training overlook the phenomenon of out-of-field teaching, in which the subjects are outside teachers' area of expertise or they lack necessary specialist requirements. Often the result of teacher shortages or personnel mismanagement, this understudied issue may have important implications for the quality of teaching and learning.

Teaching out of field is prevalent in much of the world. In at least 40 education systems that participated in the 2018 TALIS, over 10% of lower secondary school science teachers had received no formal education or training in the subject. The same is true for mathematics teachers. In Georgia and Saudi Arabia, less than 60% of science and mathematics teachers have received training in their subjects as part of their formal education.

In at least 16 countries that participated in the 2019 Trends in International Mathematics and Science Study (TIMSS), over 10% of grade 8 students were taught mathematics by teachers without a major in mathematics and/or in mathematics education. Taking the stricter definition of having both, this is true for

#### **FIGURE 19.6:**

In most countries, less than half of grade 8 students are taught mathematics by teachers with a major in mathematics and mathematics education

Grade 8 students taught by teachers with a major in mathematics and/or in mathematics education, selected countries, 2019



GEM StatLink: https://bit.ly/GEM2021\_fig19\_6 Source: IEA (2019).

99

over half the students in most participating countries (**Figure 19.6**). In Australia, there is a 76% chance of a student having at least one out-of-field mathematics teacher between grades 7 and 10, and a 35% chance of having at least two (Prince and O'Connor, 2018).

The prevalence of out-of-field teaching may be even greater in more specialized subjects. In Brazil, over half the sociology, arts, philosophy and physics upper secondary classes are taught by teachers without subject-matter qualification, compared with less than 30% of mathematics, Portuguese, biology and physical education classes (INEP, 2019). In Australia, the subject areas with the greatest share of out-of-field teachers include languages, geography and information and communication technology (Weldon, 2016).

Out-of-field teaching may have detrimental consequences on teaching and learning. Some studies have found that it negatively influences teachers' self-esteem, confidence

# 66

Out-of-field teaching has been found to negatively influence teachers' self-esteem, confidence and relationships within the school community

and relationships within the school community (du Plessis et al., 2014), as well as job satisfaction and likelihood of staying in the profession (Sharplin, 2014). Instructional practices may also differ. Out-of-field teachers tend to rely more on textbooks and pre-prepared material and be less likely to make connections across topics and engage in creative classroom activities (Napier et al., 2020; Van Overschelde and Piatt, 2020). Across all education systems participating in TALIS, teachers whose formal education or training included content of some or all the subjects they teach were more likely to feel well prepared for that element of teaching (**Figure 19.7**).

#### **FIGURE 19.7:**

Teachers who receive content training as part of their formal education are more likely to feel well prepared to teach that content

Share of lower secondary teachers who reported feeling well or very well prepared for the content element of teaching, by inclusion of this element in their formal education or training, 2018



All this can influence student performance. In the 16 TIMSS countries where at least 10% of students had out-of-field teachers, the association between out-of-field teaching and student scores is generally inconclusive, but in all cases where it is statistically significant, it is negative. The fact that the TIMSS assessment takes place in grades 4 and 8 may mask a stronger impact of out-of-field teaching in later grades, when subject-matter specialization increases.

Several studies in the United States point in that direction, though more research in various contexts is needed before conclusive results can be reached (Porsch and Whannell, 2019). Longitudinal data from the state of North Carolina examined secondary students' academic growth and found that those who took classes taught by in-field teachers performed significantly better (Clotfelter et al., 2010). Using New York state census secondary school data, researchers found that chemistry and physics students of out-of-field teachers performed considerably worse than their peers on standardized state examinations, even after controlling for socioeconomic background (Sheppard et al., 2020).

The phenomenon of out-of-field teaching also raises equity concerns, as not everyone is equally likely to be an out-of-field teacher or taught by one. In Brazil, students in rural and poorer regions are more likely to be in a class taught by an out-of-field teacher (INEP, 2019). Similarly, the prevalence of out-of-field teaching is considerably higher in rural and remote locations in Australia, as well as in schools that serve communities of lower socioeconomic status (Weldon, 2016). In Ireland, out-of-field teachers were predominantly deployed to teach less academically able students. In Germany, teaching out of field is more prevalent in schools that do not directly lead to post-secondary education, as opposed to Gymnasien or Gesamtschulen (Price et al., 2019). In the US state of Texas, students who are Black. of lower socioeconomic status, in special education and in rural schools are significantly more likely to be taught

by an out-of-field teacher, while Black teachers and teachers in rural schools are more likely to teach out of field (Van Overschelde and Piatt, 2020).

Some countries offer certificates or professional development programmes to support out-of-field teachers. In Ireland, a national government-sponsored programme, the Professional Diploma in Mathematics for Teaching, was founded in 2012 to help out-of-field mathematics teachers improve their content and pedagogy skills. It has been credited with reducing the incidence of out-of-field teaching from 48% in 2009 to 25% a decade later (Goos et al., 2019). Australia, Germany and the United Kingdom, which have similar programmes for mathematics or science out-of-field teachers, report success in increasing participants' content knowledge (Kenny et al., 2020). Out-of-field teachers may also acquire necessary knowledge through experience. In Australia, mathematics teachers without a related academic degree have been teaching the subject for over seven years, on average (Weldon, 2016).

Not all out-of-field teachers are unhappy or uncomfortable with their positions. Those who are well supported may see it as an important career development step (The Guardian, 2017). Statistics may count teachers as out of field when they teach the subject they are trained in but at a different level of schooling. Moreover, measures of out-of-field teaching may overestimate the number of out-of-field teachers, as the measures tend to focus on matching the subject with teachers' academic degree without considering the continuum in fields of study - e.g. mathematics teachers with a degree in physics are not as out of field as those with a degree in sociology. Positive experiences with out-of-field teaching usually combine a variety of factors, including teachers' interests and a strong support system offered by the school community and school head (Campanini, 2019; du Plessis et al., 2014). These require acknowledging the issue and working towards better understanding of its prevalence and impact and the best strategies to manage it.

99

66 -

In the US state of Texas, students who are Black, of lower socioeconomic status, in special education and in rural schools are significantly more likely to be taught by an out-of-field teacher

# COVID-19

Around the world, teachers have been directly affected by the COVID-19 pandemic. In the United States, well over 1,000 educators had died as of September 2021 (Maxwell, 2021). In India's Uttar Pradesh state and in South Africa, more than 1,600 teachers died (Ndaba, 2021; Rashid, 2021). The pandemic has also posed unprecedented challenges to teachers' professional lives. School closures found many teachers unprepared for the move to remote learning, uncertain about their role and unfamiliar with the technology. School reopenings found teachers without clear support in making up for loss of instructional time, protecting their own health and taking measures to protect students' health.

Countries have differed in their designation of teachers as key or front line workers and in the related consequences. Teachers are particularly exposed to COVID-19 infection, sharing a closed room for hours with large numbers of children who are not likely to be vaccinated. Teachers may face distinct requirements for using masks and being tested or vaccinated in the interest of a safe school environment. Prioritizing teachers in national COVID-19 vaccine roll-out plans is essential to making school reopenings possible and safe.

On the supply side, about two thirds of countries reported that teachers were or would be a priority target for vaccination against COVID-19, either through national immunization measures (59%) or the COVAX initiative (7%), which applies to low- and middle-income countries (UNESCO et al., 2021). However, vaccine hesitancy has affected demand. In Ethiopia, a country with limited vaccine supply that has not prioritized teachers, a study in the city of Gondar found that only 55% of teachers would accept the vaccine (Handebo et al., 2021). In the Canadian province of British Columbia, 90% of over 5,000 teachers reported they were likely or very likely to accept a COVID-19 vaccine (Racey et al., 2021).

In high-income countries, the question of whether to include teachers among occupations for which vaccination is mandatory is politically disputed as a potential human rights violation (Dzehtsiarou, 2021). The few countries and territories to have taken that step include New Zealand and the Australian state of New South Wales (New South Wales Government, 2021; New Zealand Government, 2021). In the United States, 10 states, the District of Columbia and Puerto Rico have introduced vaccination requirements. In Hawaii, the policy helped increase the vaccination rate from 80% to 89%. In addition, the federal government requires teachers at federally run schools and in the federally funded Head Start and Early Head Start programmes to be vaccinated (White House, 2021). Some education authorities, on the other hand, have introduced frequent testing for unvaccinated teachers, as in Hong Kong, China (Cheung, 2021).

One way to limit infection in schools is to reduce the number of students sharing a classroom. About 3 in 10 countries in 2020 and 4 in 10 in 2021 reported recruiting additional teachers to support school reopening efforts (UNESCO et al., 2021). In sub-Saharan Africa, 26% of countries reported hiring more teachers (ADEA et al., 2021). However, the size of such recruitment drives is unclear, as is whether they were intended to support smaller classrooms. They may instead have reflected the resumption of hiring processes that were frozen in 2020, as in Uganda, or replacement of teachers who died or resigned, as in South Africa's Eastern Cape province (Funani, 2021; Xinhua, 2021). Where reductions in class size were achieved, it was not by reducing the pupil/teacher headcount ratio but by double-shifting, as in Rwanda, reducing the number of in-person teaching hours, as in Mozambigue, and making adjustments such as moving learning outdoors or using other school facilities (ADEA et al., 2021).

More generally, however, evidence on teacher attrition has been mixed. In the United Kingdom, the primary and secondary school teacher leaving rates fell by 17% between 2019 and 2020. School closures and lockdowns in March 2020 hindered teacher interviewing, making moving jobs challenging and teachers reluctant to leave due to uncertain prospects in a recession (Worth, 2021). In the US state of North Carolina, average teacher attrition was lower, as is common during recessions, but attrition was higher among teachers close to retirement age, in schools serving disadvantaged students and in schools using hybrid rather than remote modality (Bastian and Crittenden Fuller, 2021). In the state of California, an increase in retirements, resignations and leaves of absence since the onset of the pandemic has caused severe shortages in small, rural districts, particularly in subjects such as mathematics and science (Carver-Thomas et al., 2021).

Absence of attrition in the short term does not mean increased stress and burnout will not take their toll in the medium to long run in low-, middle- and high-income countries (Goldberg, 2021; Mwesigwa, 2021; Reid and Cranston, 2021). A nationally representative study in the United States found in January 2021 that nearly 25% of teachers indicated a desire to leave their jobs at the end of the school year, compared with an average national turnover rate of 16% prior to the pandemic (Steiner and Woo, 2021; Zamarro et al., 2021). In a survey of over 20,000 teachers in 165 countries, 39% stated that their physical, mental and emotional well-being had suffered during the pandemic. On the other hand, 50% of respondents stated that they felt more enthusiastic about their vocation (Pota et al., 2021).

Teacher education has also been affected by school closures and the limitations of remote learning. In some cases, where the infrastructure was available, there were new opportunities. Israel's Ministry of Education set up an initial teacher education programme to attract unemployed graduates to teaching, granting autonomy to training institutes to develop their own programmes. After an initial three-month online training period, trainees started teaching, while their education continued for a year. The ability to teach remotely made the profession more attractive to this target group (Ramot and Donitsa-Schmidt, 2021).

But in most cases, the challenges increased. In Australia and Chile, student teachers expressed concerns about lack of interaction with peers, trainers and students, along with a sense of isolation and the potential effect on their professional development (Blackley et al., 2021; Sepulveda-Escobar and Morrison, 2020). As pre-service programmes moved online, trainees lost their practical classroom experience, as in Ireland (White and McSharry, 2021). In the Russian Federation, only the largest of the country's 300 teacher education institutions had the resources to change their organization, teaching management and online learning platforms (Valeeva and Kalimullin, 2021). In middle-income countries, such as Ghana, online modalities were introduced but could not be rolled out effectively because of trainees' lack of preparedness and access to technology (Salifu and Todd, 2020).

The crisis has raised questions over shifts needed in the content of teacher education. In Germany, young teachers with digital skills adapted more easily to online teaching (König et al., 2020). But beyond technological knowledge, teachers need to respond to new social-emotional and academic needs of students. Mentoring that supports teachers in learning to play these new roles will be needed, along with time for collaboration among educators (Darling-Hammond and Hyler, 2020). In Haiti, 7-year-old Melissa makes a drawing while taking part in the Healing and Education through the Arts (HEART) programme, which provides psychosocial support to children who survived Hurricane Matthew.

CREDIT: Ray-ginald Louissaint Jr/Save the Children



# CHAPTER



# Education in the other SDGs A focus on energy, infrastructure and sustainable consumption

(Global indicators from goals other than SDG 4 that are education-related)

# **GLOBAL INDICATOR**

**1.A.2** – Proportion of total government spending on essential services (education, health and social protection)

**5.6.2** – Number of countries with laws and regulations that guarantee full and equal access to women and men aged 15 years and older to sexual and reproductive health care, information and education

**8.6.1** – Proportion of youth (aged 15–24 years) not in education, employment or training

**4.7.1/12.8.1/13.3.1** – Extent to which (i) global citizenship education and (ii) education for sustainable development are mainstreamed in (a) national education policies, (b) curricula, (c) teacher education, and (d) student assessment

# **KEY MESSAGES**

Education is substantially affected by access to electricity, energy and internet connectivity. A review of 50 impact evaluations, mostly from India and Peru, found that electrification led to a 7% increase in school enrolment, with greater benefits for girls than for boys.

A study of education choices linked to 115,000 roads built in India's flagship road construction programme between 2001 and 2015 found that connecting a village with a new paved road caused a 7% increase in lower secondary school enrolment over the following three years.

A household survey in Nepal found that a literate household head reduced the household's firewood demand by around 8% compared with a non-literate household, possibly because of knowledge of the negative impact of smoke pollution.

Analysis of household choice of solar energy for domestic purposes in Ethiopia, Kenya and Uganda found that education level was among key factors encouraging adoption.

Curricular interventions can improve sustainable consumption and production practices. The Sustainable Energy Network supports teaching on sustainable energy in upper secondary and tertiary education through topics such as renewable and alternative energy systems.

A workforce with new skills and specializations is needed to address sustainability challenges. It has been estimated that building a fully renewable energy sector would require 43 million jobs to be created by 2050.

Sustainable Development Goals 7 (energy), 9 (infrastructure, industrialization and innovation) and 12 (sustainable consumption and production) contribute to efforts towards sustainable economic growth through clean industry and circular economy principles. Progress in some of these areas can also support education. At the same time, improvements in education help countries achieve energy, industrialization and sustainable consumption objectives, particularly development of professional capacity to meet countries' technological advancement needs.

# IMPROVED ENERGY AND INFRASTRUCTURE SUPPORT EDUCATION

Of the 759 million people in the world who lacked access to electricity in 2019, 84% lived in rural areas (IEA et al., 2021). Education is substantially affected by access to electricity, a fact that the COVID-19 pandemic has further emphasized. A review of 50 impact evaluations, mostly from India and Peru, found that electrification led to a 7% increase in school enrolment, with greater benefits for girls than for boys. Similarly affected were years of schooling, attendance, literacy and time allocated to study at home (Jimenez, 2017). A comparison of time usage data among households with and without electricity access in Burkina Faso, Rwanda and Senegal shows that children shifted study time from day to night with electrification, and study time increased cumulatively in Senegal (Peters and Sievert, 2016).

Access to energy at home can play a significant role in allowing children to participate in education activities. In South Asia, women and girls spend one hour a day gathering fuel and around four hours a day cooking (Bloomfield, 2014). Being able to shift from procuring fuel and cooking with inefficient stoves to attending school and studying enabled positive educational outcomes. Bhutan's rural electrification programme helped reduce fuelwood use and led to 0.8 more years of schooling, with stronger effects for girls than boys, potentially due to reduced time burden for girls' chores (Litzow et al., 2019). Energy poverty in the Lao People's Democratic Republic had a negative impact on households' average years of schooling (Oum, 2019). In Nigeria, school enrolment increased, particularly for girls, among households with access to electricity (Nano, 2021).

Access to energy in schools can help improve the learning environment and expand access to learning resources (**Chapter 17**). The Energy Sector Management Assistance Program with support from the Sustainable Energy for All initiative surveyed education facility electrification status in Cambodia, Ethiopia, Kenya, Myanmar, Nepal and Niger. They found, for instance, that 72% of schools in Kenya, but only 22% in Ethiopia, had access to the national public grid. New country surveys and data are to be made systematically available by 2024 (IEA et al., 2020).

Improvements to mobile networks, internet connectivity and road infrastructure can also help improve access to education if support measures are taken. Analysis of staggered entry of 3G into Brazil's 5,570 municipalities showed mobile internet had no effects on test scores for grade 5 and 9 students, suggesting that high-speed internet was not sufficient to improve education outcomes (Bessone et al., 2021). However, a separate analysis of internet access showed possible improvement in test scores if policy initiatives focused on both school and home internet access and incorporated internet-based educational tools (Badasyan and Silva, 2018).

Analysis of 173 classrooms in 15 predominantly rural school districts in the US state of Michigan found that students who lacked internet access at home or depended on mobile phones had worse digital skills, were less likely to complete homework and had worse standardized test scores than peers with home internet. They were less likely to consider post-secondary education and had less interest in careers in science, technology, engineering and mathematics (Hampton et al., 2020).

Roads help alleviate poverty and promote economic and social development, including education outcomes. In Brazil's Tocantins state, improved roads led to higher attendance by girls in two regions and a decline in the perception that road conditions prevent school access (Iimi et al., 2015). In the Colombian department of Antioquia, improved rural roads were associated with improved education performance for rural students (Hincapie-Velez et al., 2017). Residents of peri-urban areas around Accra, Ghana, and Kisumu, Kenya, agreed that primary school accessibility improved substantially after road expansion (Khanani et al., 2021).

66

Access to energy at home can play a significant role in allowing children to participate in education activities

**?**?

A study of education choices linked to 115,000 roads built in India's flagship road construction programme between 2001 and 2015 found that connecting a village with a new paved road caused a 7% increase in lower secondary school enrolment over the following three years and that children stayed in school longer and performed better on standardized examinations (Adukia et al., 2020). However, another study cautioned that improved access to urban markets improved enrolment of younger children but offered older children an incentive to leave school early to join the labour force (Aggarwal, 2018).

# EDUCATION CAN SUPPORT ACHIEVEMENT OF ENERGY AND SUSTAINABILITY OBJECTIVES

Education has the potential to help people and societies make better choices regarding energy and sustainable consumption, although many questions remain about what type of education is most effective in achieving these results and the channels through which knowledge changes not just attitudes but also behaviour.

A systematic review of 160 studies on cooking energy found that education, along with income, influenced the adoption of modern cooking energy sources (ESMAP, 2021). A household survey in Nepal found that a literate household head reduced its firewood demand by around 8% compared with a non-literate household, possibly because of knowledge of the negative impact of smoke pollution (Sharma, 2018). Similarly, education attainment was a driver for changing fuel source in households in Delhi, India, even after controlling for fuel prices and electricity access (Ahmad and Puppim de Oliveira, 2015). A large survey in India's Kerala and Rajasthan states found that education was a predictor of clean cooking fuel adoption, although not as a result of attitudinal changes (Gould and Urpelainen, 2020). Analysis of household choice of solar energy for domestic purposes in Ethiopia, Kenya and Uganda found that education level was among key factors encouraging adoption (Rahut et al., 2018).

Analysis of data from 1990 to 2015 across Organisation for Economic Co-operation and Development countries showed that education increased environmental awareness among citizens and could reduce carbon emissions (Zafar et al., 2020). In Europe, analysis of two waves of Eurobarometer surveys found that education increased individuals' awareness of the need for more environment-friendly behaviour (Meyer, 2015). Analysis of utility-level data from across the United States found education to be a key factor in demand for green electricity (Conte and Jacobsen, 2016). In China, household survey analysis showed that for each additional year of household education, the expressed willingness to pay for environmental protection increased by nearly 30% (Tianyu and Meng, 2020).

The education level and awareness of corporate leaders is a key driver of sustainable production practices (Reisch et al., 2016). An online survey of 766 chief executive officers (CEOs) in nearly 100 countries found that they viewed education as the most critical development issue for the success of their business and for development of skills, knowledge and mindset in the next generation of business leaders to accelerate integration of sustainability into core business (Lacy et al., 2012).

Analysis of Chinese enterprises from 2008 to 2017 found that enterprises with highly educated CEOs were more likely to engage in environmental innovation, particularly in regions with strict environmental pressures (Zhou et al., 2021). In Denmark, analysis of the effect of CEO education on green orientation in private life and corporate decisions found that better-educated CEOs expressed more concern for climate change, were more likely to own energy-efficient cars and helped improve sustainability in corporate actions (Amore et al., 2019). In Viet Nam, an analysis of 810 small and medium-sized enterprises found that CEO education level was positively related to corporate environmental performance (Tran and Pham. 2020). An analysis of pollution-intensive companies in Thailand found education level positively correlated with environmental information disclosure (Li et al., 2019).

**99** 

66

Analysis of data across OECD countries showed that education increased environmental awareness among citizens and could reduce carbon emissions

## AWARENESS CAMPAIGNS CAN PLAY AN IMPORTANT ROLE

Awareness campaigns, a form of adult education, can foster behaviour change that contributes to sustainability. For instance, while governments may supply renewable energy infrastructure, households may not demand such energy because of lack of awareness and the cost of switching. In Bangladesh, the government-owned Infrastructure Development Company Limited launched an intensive customer awareness campaign in 2017, which increased mini-solar grid uptake by 500% (IEA et al., 2020).

In recent years, meat consumption has been acknowledged as a key contributor to greenhouse gas emissions (Rust et al., 2020). A diet with less meat is viewed as the single biggest way individuals can reduce their environmental impact (Poore and Nemecek, 2018). Meat is also related to some types of ill health, so changes in consumption may be difficult to associate with a specific motivation. Overall, education tends to be associated with lower meat consumption. In Belgium's Flanders region, having secondary rather than higher education doubled the probability of being a meat eater rather than a flexitarian (De Backer and Hudders, 2015). In Germany, both individual and average household education levels were associated with reduced meat consumption, an effect stronger for younger households (Einhorn, 2020). In Chile, people who cited environmental reasons for their decision to stop eating meat tended to be more educated (Giacoman et al., 2021).

Responsiveness to public information can also be related to education. A study of consumption patterns in Italy, following a public health warning, showed that educated households reduced their long-term red meat consumption (Carrieri and Principe, 2020). However, a systematic review of 59 interventions found that providing information about the health or environmental consequences of eating meat did not reduce meat consumption, on average; campaigns emphasizing animal welfare were more effective (Bianchi et al., 2018). Slow Food, a grass-roots organization in 160 countries that opposes overproduction and food waste, launched the Meat the Change campaign in January 2020 to change meat-eating habits, promoting sustainable farming and less meat consumption (Slow Food International, 2020).

Waste reduction, especially as regards food, plastic and clothing, is another imperative. Save Food, an initiative of the Food and Agriculture Organization of the United Nations and Messe Düsseldorf, aims to reduce food loss and waste by encouraging dialogue among industry, research, politicians and civil society. One of its projects, focusing on developing educational packages so teachers could teach young people to value food and reduce its loss, involved primary and secondary age groups in Albania, Croatia, Hungary, Turkey and Ukraine. Another project raised awareness and built capacity among small-scale entrepreneurs in Timor-Leste for use of good practice in processing, packing and labelling their products (FAO, 2017).

In the clothing industry, non-governmental organizations generate awareness of the clothing value chain's negative environmental and societal impact. They challenge industry to find systemic solutions to overconsumption, including through production technology and development of materials that pollute less than those now used (Ellen MacArthur Foundation, 2017). The UN Alliance for Sustainable Fashion, launched in 2018 by seven UN agencies in Nairobi, Kenya, maps initiatives and partnerships addressing interlinkage between sustainable fashion and the 17 SDGs. Those related to SDG 4 include initiatives bu the International Labour Organization and partners to advance working conditions, labour rights and competitiveness (Meier, 2021). The 2020 Circular Fashion System Commitment, introduced by the Global Fashion Agenda, aims to foster circular economy principles; 94 companies representing around 12.5% of the global fashion market have signed on (Wu and Li, 2019).

## EDUCATION INSTITUTIONS NEED TO IMPROVE UNDERSTANDING OF ENERGY, TRANSPORT AND OTHER SUSTAINABILITY CHALLENGES

There have been many attempts to frame the curricular challenge of sustainability, as well as suggestions for specific curricular interventions related to sustainable consumption and production practices, including on energy, diet and waste. The Sustainable Energy Network, established by the UN initiative Future Earth, is a framework based on teaching programmes and interdisciplinary textbooks to support teaching on sustainable energy in upper secondary and tertiary education through topics such as renewable and alternative energy systems (Nowotny et al., 2018). National curricula incorporating energy education include the National Energy Education Development Project in the United States, with topics such as energy sources, electricity, transport, efficiency and conservation (NEED, 2021). Education programmes promoting efficient cooking technology also need to include aspects of sustainable energy use. An analysis of 200 households in rural Namibia showed that such interventions can affect energy-related attitudes and behaviours, including openness to solar energy as an alternative to biomass (Lindgren, 2021a, 2021b).

A review of 58 sustainable consumption teaching initiatives by the Sustainable Consumption Research and Action Initiative found that most undergraduate and graduate courses were offered in European countries, often as part of environmental science programmes. Many courses had a holistic, system-wide perspective and goals for transformative social change (Sahakian and Seyfang, 2018).

Universities are critical not only for technology development but also for contributing to its assimilation and absorption. The economic, health and environmental benefits of a rapid transition to clean energy have been promoted through research and dissemination by prominent universities and think tanks (Esposito, 2021). Higher education institutions globally also function as exemplars, leading the implementation of renewable energy and energy efficiency initiatives. An analysis of 157 universities in 13 Latin American countries found that more than 80% incorporated sustainability initiatives, with an emphasis on campus operations (Leal Filho et al., 2021). The Better Buildings Alliance includes 31 colleges and universities that share best practice to advance energy savings (US Department of Energy, 2021).

### EDUCATION INSTITUTIONS NEED TO CONTRIBUTE PROFESSIONAL CAPACITY TO ADDRESS SUSTAINABILITY CHALLENGES

Delivering transformation for equitable access to sustainable energy, infrastructure and production requires a professional workforce with new skills and specializations. For instance, achieving the Paris Agreement target of limiting global warming by 2050 to 1.5°C requires professionals with diverse profiles to implement unprecedented energy transition strategies. The necessary skills, of varying levels of complexity, can partly come from existing industries, but new skill development policies will also be needed. Science, technology, engineering and mathematics professionals will need to collaborate with experts in law, logistics and market regulation. Retraining can help redeploy construction workers and technicians for the manufacture, installation and maintenance of solar water heaters and wind turbines. In the ambitious scenario, the number of jobs in renewable energy

has grown from 7.3 million in 2012 to 16.5 million in 2021 and will need to reach 38 million in 2030 before stabilizing to 43 million in 2050 out of 122 million jobs in the energy sector. The share of renewable energy in total jobs in the energy sector will therefore be 35% in 2050 or double the level in 2021 (IRENA, 2021).

Assessments show substantial skills gaps in developing, designing, financing, building, operating and maintaining renewable energy projects (OECD, 2020). General education curricula will need to instil appreciation of the challenge and interest in renewable energy careers from a young age, and technical, vocational and higher education curricula will need to be adjusted to focus on development of higher skills, notably in engineering (WFEO, 2018).

An assessment of 150 decentralized renewable energy sector companies in India, Kenya and Nigeria showed 210,000 informal jobs in India (vs 95,000 formal jobs), 15,000 in Kenya (vs 10,000 formal) and 9,000 in Nigeria (vs 4,000 formal) (Power for All et al., 2019). Consultations with surveyed companies highlighted the need to significantly improve training in managerial, finance, legal and sales roles to improve the extent to which these jobs are filled locally and absorbed in the formal labour market (Power for All, 2020).

Accelerated design and implementation of capacity development plans are needed. Poor countries lack capacity to absorb skills in the clean energy transition and thus cannot participate equally in low-carbon energy technology value chains. A global review of skills for green jobs showed that the skills gaps will lead to project delays or cancellations, excessive costs and faulty installations (Hafner and Tagliapietra, 2020). A review of 46 countries that submitted voluntary national reviews featuring energy sector analyses to the 2018 High-level Political Forum found that only about a third outlined plans to scale up capacity and education to develop the skills to support energy sector transformation. Togo planned to create solar academies

66

An analysis of 157 universities in 13 Latin American countries found that more than 80% incorporated sustainability initiatives, with an emphasis on campus operations Improved coordination is often needed between the renewable energy industry and education institutions to develop effective curricula

to train 3,000 technicians in 2017–18 (UNDESA, 2018). But a mapping of 32 countries' Nationally Determined Contributions, which are non-binding plans with public policies and measures in response to climate change, showed that almost 75% included training or capacity building measures (ILO, 2020).

66

Several global partnerships finance capacity-building projects to help countries achieve their sustainable energy goals. The Energy Sector Management Assistance Program, a partnership of the World Bank and 19 other entities, supports energy transition activities in low- and middle-income countries, including on clean cooking, renewables, electricity access, accelerated decarbonization and gender gap reduction. In Tuvalu, the programme supported a study on variable renewable energy and integration, and organized workshops to develop the national utility's technical and institutional capacity (ESMAP, 2020). The International Renewable Energy Agency trained national experts in Eswatini to help develop the Energy Masterplan 2034 and a roadmap to a sustainable energy future (KfW Development Bank et al., 2021).

Improved coordination is often needed between the renewable energy industry and education institutions to develop effective curricula and support vocational training courses and apprenticeships (IRENA et al., 2018). As part of Morocco's Nooro I solar power project, the University of Ouarzazate introduced curricula on renewables to develop local research capacity but none of the 100 students who graduated with a degree in renewables found jobs matching their qualifications during the project construction phase (Wuppertal Institute and Germanwatch, 2015).

Even high-income countries face challenges. For instance, Canada is one of the countries that joined the Clean Energy Education and Empowerment initiative, which aims to promote gender equality in the energy sector and increase education and employment opportunities for women. Singapore has instituted a vocational training programme to build the technical competence of the local workforce (UNDESA, 2018). In the United Kingdom, a green apprenticeship fund for small and medium-sized enterprises has been proposed to help fill an expected green skills gap in construction (Watkins and Hochlaf, 2021). The US Department of Energy is promoting workforce development with improved science curricula in education and training systems through the Better Buildings Workforce Accelerator (US Department of Energy, 2021). The National Center for Sustainable Transportation, a consortium of leading US universities, is developing strategies to make careers and training in transport more attractive through first-person testimonies about work in the sector (McRae et al., 2019).

# CONCLUSION

In the race to achieve the SDGs by 2030, there has been laudable progress in improving renewable energy technology, supported by major investment in the transition to solar and wind power. There is also growing awareness of the need to consume and produce sustainably. However, improvement in areas of goals that are not as market oriented – e.g. equitable access to clean cooking technology, expertise on renewables, financial assistance to the least developed countries for capacity building, diverse and equitable workforce development – has been marked by struggle.

Education supports the achievement of sustainability objectives. Education institutions need to improve students' understanding of energy and other sustainability challenges. Public awareness can contribute to broader social change. Professional capacity development needs to take place at an unprecedented pace to support the green transition.

99

Fatoumata Touré, 6, and Fatouma Diakité, 7, attending class in an Islamic school in Man, Côte d'Ivoire. ,

CREDIT: UNICEF/Miléquêm Diarassouba



# CHAPTER



# Monitoring finance



# **KEY MESSAGES**

Education's share of total public expenditure grew from 13.8% in 2000 to 14.1% in 2019. Recent data for 71 countries suggest the share decreased to 13.5% in 2021, suggesting a strong COVID-19 impact.

The cost per student in tertiary education is 93% of GDP per capita in low-income and 41% in lower-middle-income countries, but about 25% in upper-middle- and high-income countries.

About US\$483 billion is lost to cross-border corporate tax abuse by multinational enterprises and offshore tax evasion by wealthy individuals, of which US\$40 billion is lost to poorer countries.

Some countries are quite effective in rolling out programmes that reach the most disadvantaged. Algeria offers an annual education allowance, equivalent to US\$23, to 3 million primary and secondary school students: 38% of the poorest but 10% of the richest receive support.

Aid effectiveness means different things to different people. One definition focuses on country ownership and results, transparency and mutual accountability, and inclusive development partnerships. The share of direct budget support in total aid fell from 6.6% in 2002 to 2.5% in 2019.

Analysis of household budget survey reports from about 100 low- and middle-income countries in the 2010s found that education accounted for 3.2% of total household expenditure, reaching as high as 6% or more in Haiti and Lebanon and in sub-Saharan African countries, including Rwanda, Uganda and Zambia.

The Global Education Monitoring Report team has long argued that to understand global education spending, the three main sources of education financing – domestic public financing, external public financing and private financing – must be examined jointly. In 2021, the team partnered with the World Bank on Education Finance Watch, a new annual report series that reviews the main trends on education spending and complements this chapter (World Bank and UNESCO, 2021).

# **PUBLIC EXPENDITURE**

The Education 2030 Framework for Action called on countries to spend at least 4% of gross domestic product (GDP) and 15% of total public spending on education. While countries are meeting the first benchmark globally – public education expenditure stands at 4.4% of GDP – they are not meeting the second. Education's share of total public spending has remained stagnant over the past 20 years. It grew from 13.8% in 2000 to 14.1% in 2019; in contrast, health's share of spending grew from 9.8% to 10.6%. While the increased priority on health may be the result of rising longevity and costlier medical technology for the elderly, education needs have also been growing due to larger school-age population cohorts and ambitious education targets.

At the country level, of 151 countries with data for 2014–19, 48 countries, or 32%, from all regions missed both benchmarks. The 10 countries with the highest education share in government expenditure are low- or middle-income countries, mainly in sub-Saharan Africa (Burkina Faso, Eswatini, Ethiopia and Sierra Leone) and Central America (Costa Rica, Guatemala, Honduras and Nicaragua). Tunisia and Uzbekistan complete the list (**Figure 21.1a**).

On average, governments in poorer countries with small budgets but large cohorts of children tend to spend less on education as a share of GDP but more as a share of total government spending. Sub-Saharan Africa has the highest share of government spending on education (16.8%) but is slightly below the education spending benchmark as a share of GDP (3.8%). Conversely, Europe and Northern America meets the benchmark as a share of GDP (4.7%) but allocates a low share of total spending to education (11.9%). Latin America and the Caribbean and Central and Southern Asia are the only regions to meet both benchmarks (**Figure 21.1b**).

Despite education being a greater budget priority, poorer countries spend less on education as a percentage of GDP because of lower capacity to raise revenue (**Figure 21.2**). In these countries, reaching the share of GDP benchmark for education spending requires more domestic resources. For example, given current levels of total government spending, Uganda would have to increase the share of education in its budget from 11.5% to 25% to reach the benchmark of 4% of GDP. More effective and fair taxation can contribute to resource mobilization (**Focus 21.1**).

The COVID-19 pandemic and its aftermath threaten to squeeze education budgets through a combination of reduced revenue and increased demands from other sectors. Strong calls have been made during the COVID-19 crisis to protect spending on education, which is as important as other sectors that have benefited from government support packages. Helping keep schools open and offering remedial classes to children whose learning has suffered represents an investment that will avert future social costs from increased early school leaving and lower learning achievement (UNESCO, 2020). Indeed, the equity focus of national education financing policies is slowly gaining more attention (UNESCO, 2021) (**Focus 21.2**).

As countries' financial years differ, budgets were approved at different times relative to the onset of the first COVID-19 wave, hampering estimation of the COVID-19 impact on spending. Analysis of the first post-COVID budgets in 29 countries representing 54% of the school-age population found that education's share in total spending did not change much in relative terms. In absolute terms, budgets increased by 4% post-COVID compared with 1.1% pre-COVID. While 65% of low- and lower-middle-income countries in the sample reduced spending, only 33% of upper-middle- and high-income countries did so. These trends suggest that richer countries were able to increase their total spending (World Bank and UNESCO, 2021). But more recent data collected by the UNESCO Institute for Statistics (UIS) for 71 countries suggest that the median education share in total spending decreased by 0.3 percentage points per year between 2019 and 2021, from 14.1% to 13.5%, suggesting a stronger impact of COVID-19 than first predicted (UIS, 2021) (Figure 21.3).

#### One in three countries spends too little on education

Public education expenditure as share of (i) total public expenditure and (ii) GDP, 2019 or latest available year



## **FIGURE 21.2:**

#### Poor countries spend little on education despite its priority in budgets

Public education expenditure as share of (i) total public expenditure and (ii) GDP, and total revenue as share of GDP, by country income group, 2019 or latest available year



Source: GEM Report team estimates based on UIS and IMF World Economic Outlook data.

Globally, public expenditure per student varies from about US\$2,200 in pre-primary education to about US\$5,000 in tertiary education, in purchasing power terms. Absolute differences between poor and rich countries in spending per student are vast. In primary education, spending per student ranges from US\$168 in low-income to US\$8,363 in high-income countries, a 50-fold difference, and the gaps in pre-primary and lower secondary education are similar. The gaps in spending between high-income and low-income countries are smaller in upper secondary (25 times) and, especially, tertiary education (6 times).

Providing tertiary education in the poorest countries entails high fixed costs and a small number of beneficiaries due to low enrolment ratios. Sub-Saharan African countries thus spend about US\$2,500 per student, similar to Central and Southern Asia and Latin America and the Caribbean. Consequently, public investment at this level is highly regressive, benefiting the more privileged who can afford university studies.

#### **FIGURE 21.3:**

# One impact of COVID-19 may have been lower budget priority on education

Share of education in total public expenditure, selected countries, 2019–21



While some of the differences between countries in spending per student are attenuated if expressed in GDP per capita terms, in tertiary education the difference becomes even more apparent. The cost per student in tertiary education is 93% of GDP per capita in low-income and 41% in lower-middle-income countries, but about 25% in upper-middle- and high-income countries (Figure 21.4). By contrast, high-income countries spend about 17% of GDP per capita per pre-primary education student, while middle-income countries spend about 10% and low-income countries 5%. The gap between low- and high-income countries is 8% vs 18% in primary and 11% vs 22% in lower secondary education. In upper secondary education, countries at all income levels spend almost the same amount per student in GDP per capita terms. Overall, global median public expenditure as a share of GDP is highest in primary education (1.4%) followed by tertiary (0.8%), lower secondary (0.7%), upper secondary (0.6%) and pre-primary education (0.3%).

Even in low-income countries receiving a high share of income from grants (i.e. aid), the main source of government revenue is taxation. In a sample of 70 countries in 2018, the three main total government revenue sources were taxes (77%), non-tax revenue (19%) (which includes loans, royalties, fees and sales) and grants (4%) (**Figure 21.5**). The share of aid can be large in poorer countries: for example, 19% in Rwanda, 25% in Haiti and 30% in Bhutan (UNU-WIDER, 2021).

Countries differ greatly in their amount and sources of tax revenue. In a sample of 100 countries, Chad, Congo, the Democratic Republic of the Congo and Nigeria raised less than 10% of GDP in taxes. Even if they were to spend 20% of the budget on education, it would be far too little to cover their education development needs (Archer, 2016; Lewin, 2020). At the opposite end, Cuba and some European countries, including Austria, Denmark and Italy, raised more than 40%, and France 46%, of GDP in tax (**Figure 21.6a**). Overall, average tax revenue as a share of GDP was 14% in low-income, 18% in lower-middle-, 22% in upper-middle- and 33% in high-income countries.

In the same country sample, individual and corporate income taxes accounted for just one sixth of tax revenue in some countries, including Argentina, Brazil and Costa Rica, but more than 50% in others, including Australia, Lesotho, Namibia, New Zealand, Papua New Guinea and South Africa. Corporate income taxes alone provided about half of tax revenue in Malaysia and Nigeria, but no more than 5% in France, Italy and the United States. More than three quarters of tax revenue in the Lao People's Democratic Republic, Samoa and Togo came from taxes on goods and services (**Figure 21.6b**).

Most large differences between individual countries disappear when comparing groups of countries. For instance, low-, middle- and high-income countries all raised one third of their tax revenue from individual and corporate income taxes, the only difference being that, within this category, the share from individual income taxes was 43% in low- and middle-income countries and 69% in high-income countries. Otherwise, the main difference in tax structures was

#### **FIGURE 21.4:**

**Except in tertiary education, the richer a country, the more it spends per student** Government funding per student as a share of GDP per capita, by education level and country income group, 2019 or latest available year



#### **FIGURE 21.5:**



that the richer the country, the higher the share of revenue from social security contributions and the lower the share from taxes on goods and services (e.g. consumption or trade taxes) (**Figure 21.7**).

The Addis Tax Initiative, a result of the third International Conference on Financing for Development in 2015, is the spearhead of implementation of the 2015 Addis Ababa Action Agenda on domestic resource mobilization (ATI, 2021; United Nations, 2015). With more than 60 countries participating, the initiative aims to strengthen tax systems by broadening the domestic tax base, improving domestic tax compliance and enhancing tax collection capacity through improved tools and procedures to stop cross-border tax evasion and domestic tax avoidance (ATI, 2015). The Addis Ababa conference also led to the Integrated National Financing Frameworks platform for sharing best practices to strengthen alignment of national development plans with financing needs (INFF, 2021).

In addition to initiatives to strengthen domestic tax regimes, recent years have seen concerted efforts to improve international tax policy. As they try to generate revenue, low- and middle-income countries need to overcome structural obstacles. They need to formalize the economy to be able to collect income taxes more effectively and they need to build their institutions to introduce modern value-added tax systems. In the meantime, taxing corporate income is an area in which the Global South can collaborate with the Global North to close loopholes. Multinational companies use various strategies to move profits to lower-tax jurisdictions (Kleinbard, 2011). Developing countries lose much of their tax base through such practices as revenue shifting, debt shifting, transfer pricing and tax deferral. Collectively, these tax avoidance activities, while not illegal, violate the good taxation principle of fairness, as tax should be paid where income is generated and where local people's skills are used.

There are serious methodological and data challenges to investigating tax avoidance losses in low- and middle-income countries (Johannesen and Pirttilä, 2016; UNCTAD, 2015). Yet recent studies have expanded understanding of the issue. Research on German multinational firms' found that internal debt shifting became more widespread the larger the gap between the host-country tax rate and the lowest tax rate the firm's affiliates faced, but that the effects were small (Buettner **FIGURE 21.6:** 

21

# Countries differ in both the size and the composition of their tax revenue

Tax revenue, selected countries, 2018





b. By type of tax and country income group



## FIGURE 21.7:

# Poor countries depend more than rich countries on taxes on goods and services

Composition of tax revenue, by country income group, 2018



and Wamser, 2013). A study of Danish multinationals found that transfer pricing reduced tax revenue by 3.2% (Cristea and Nguyen, 2016).

Research by International Monetary Fund economists on transfers from high-tax to low-tax countries estimated they reduced total corporate income tax revenue by 2.6%, or 0.07% of global GDP. However, the report recognized that the effect might be more substantial in developing countries, where avoidance practices are so complex that available data cannot demonstrate the impact (Beer et al., 2019). A more recent study with data from 210,000 corporations confirmed that the propensity to report zero profit was correlated with incentives to shift profit to countries with lower tax rates; this would partly explain why many developing countries reduce their corporate income tax rates despite the urgent need to improve their tax base (Johannesen et al., 2020).

A review of 79 countries using 2016 data found that tax revenue losses from profit shifting amounted to 0.17% of GDP in lower-middle-income countries, although effects of around 1% of GDP were found in countries including Honduras, India and Zambia – and as high as 3.5% of GDP in Mozambique. El Salvador and Nigeria may be losing around one quarter of their corporate income tax revenue – and the Bolivarian Republic of Venezuela as much as 100% (Janský and Palanský, 2019). The 2021 State of Tax Justice report estimates that US\$483 billion is lost to cross-border corporate tax abuse by multinational enterprises and offshore tax evasion by wealthy individuals. Of that, US\$40 billion is lost to poorer countries (Global Alliance for Tax Justice et al., 2021). For reference, governments in low- and lower-middle-income countries allocate US\$250 billion to education and receive US\$9 billion in aid. These studies confirm that unfair practices are the rule and that restructuring and improving the international tax system for better disclosure, information sharing and transparency are urgently needed.

Prompted by research findings and civil society campaigns as well as the pressure of continuing transformation towards digital globalized economies, 130 countries signed a statement on a two-pillar solution for international tax reform in July 2021 (OECD, 2021b). The first pillar introduces new rights to tax multinational enterprises, whether or not they are physically present in countries, with 25% of profit over a certain margin, estimated at US\$125 billion of profit, reallocated to the countries where the enterprises' customers are located. The second pillar sets a minimum 15% tax on corporate profit to prevent harmful tax competition between countries; it is expected to generate around US\$150 billion in new tax revenue globally (OECD, 2021d). It remains to be seen how these measures will benefit low- and middle-income countries. And education campaigners need to work with countries to ensure that much of the new revenue is used on education.

# FOCUS 21.2: EDUCATION SPENDING SHOULD FOCUS ON EQUITY

SDG thematic indicator 4.5.3 aims to capture countries' efforts to redistribute public resources to support disadvantaged groups. While the indicator was initially conceived to measure formula-based funding mechanisms for disadvantaged groups, its definition has been expanded to encompass all funding mechanisms that redistribute education resources to disadvantaged groups. This requires monitoring of quantitative data (e.g. how many pupils receive how much) and qualitative judgements (e.g. what the mechanism is trying to achieve and how). The indicator recognizes that while inequality in education has been monitored closely over the past 15 years, relatively little attention has been paid to monitoring equity-oriented policies and programmes.

Countries vary in their approaches to mitigating the impact on education of factors such as poverty, gender, ethnicity, disability and remoteness. Drawing on the PEER website, the GEM Report team collected information from 78 low- and middle-income countries covering all SDG regions except Europe and Northern America. The team identified four mechanisms to assess the equity focus of financing policies and programmes and provide a definition for indicator 4.5.3: overall education financing mechanisms, notably from central to local governments; allocations to schools, which take into account their characteristics; education resources to students, such as scholarships; and social policies and programmes targeting students and their families, such as social transfers. For each mechanism, three dimensions - comprehensiveness, coverage and volume - helped characterize the country's efforts to promote equity in education. The analysis concluded that 17 of the 78 countries, or only 1 in 5, maintained a strong equity focus through financing policies; they were mostly upper-middle-income and Latin American countries (UNESCO, 2021).

More than half of low- and middle-income countries' financing mechanisms do not consider sufficiently that some local authorities are more disadvantaged than others. A common mechanism to allocate funds to districts or schools is through capitation grants, i.e. tied to the number of students enrolled. In Myanmar, a formula has been used since 2009 to transfer resources to states and regions based on the number of students and teachers and the budget execution rate in the previous three years (UNICEF, 2018). The government developed and shared operational guidelines with local education officers and head teachers. Grants to schools ranged from US\$400 to US\$15,000 in 2017/18, depending on size. But the formula is not sensitive to the higher needs of remote schools, for instance, to cover high transport costs. Schools have had limited autonomy in use of the grant (World Bank, 2018).

Tajikistan strengthened its education management information system and introduced a per capita financing system in 2010. Schools receive a grant consisting of a fixed component related to type ('minimum standard', which takes into account recurrent expenditure) and a variable component related to number of students, adjusted by type of school. The formula is further adjusted by location (accounting for the district budget) and available school facilities. The grant covers teacher salaries, maintenance and other operational costs. Although schools have autonomy in budget execution, no component promoting equity other than school location has been incorporated in the formula. Making funding proportional to need has resulted in more equitable allocation in terms of pupil/teacher ratios. But the mechanism has not fully addressed inequality related to districts' inability to raise enough resources, even if they benefit from central government transfers (GPE, 2019; IsDB, 2019).

While donors have introduced various types of school grants in low- and middle-income countries, focusing on school improvement or school-based management, such grants have not become part of the public budget and sustained after the programmes end. For example, donor-funded programmes in Cambodia supported additional grants in 2009–12 (Marshall and Bunly, 2017) and 2013–16 (UNESCO Bangkok, 2017) but were not sustained. Nor were supplementary school-based management grants in the Philippines, which were piloted in selected districts, taking school locations into account (Philippines Department of Education, 2015; World Bank, 2020).

Scholarships to students tend to be awarded on the basis of academic performance, which usually exacerbates inequality. Some countries take socioeconomic status into account. In Nepal, where it is estimated that households contribute 49% of total education expenditure, equity is one of five foundations of the School Sector Development Programme. There are scholarship programmes for vulnerable groups, notably girls and Dalits. However, with the number of recipients in 2016 reaching 2.8 million, or 37% of total primary and secondary school enrolment, scholarship amounts are tiny, ranging from US\$3.60 to US\$4.90 per student per year (Nepal Department of Education, 2016; Vertex Consult, 2016).

Other government expenditure besides the education budget can affect equity in education. Conditional and unconditional cash transfer programmes targeting the poor exist in much of the world, often inspired by the experience of Latin American countries. In Indonesia, Program Keluarga Harapan (Family Hope Programme) began providing quarterly cash transfers to very poor households in 2008. Initially equivalent to 15% to 20% of income, their real value fell to 7% within six years. Eligible households' demographic characteristics include having children under age 15 or children aged 16 to 18 who have not completed nine years of education. Conditions for payment include an 85% school attendance rate.
A six-year follow-up evaluation showed enrolment rates among 13- to 15-year-olds rose by up to nine percentage points, equivalent to halving the share of those out of school. Increases of between four and seven percentage points were observed in the secondary school completion rate among 18- to 21-year-olds, with the effect concentrated among young men (Cahyadi et al., 2018). The government scaled up the programme from 3.5 million to 10 million households between 2016 and 2018 (World Bank, 2017), reaching about 14% of the population.

The module on social transfers added to the UNICEF Multiple Indicator Cluster Surveys (MICS) in 2017 enables analysis of education programmes' coverage and their relative effectiveness in reaching disadvantaged households. Observations can be made about three groups of countries in terms of the equity focus of their education financing policies.

First, there is a marked absence of policies and programmes in sub-Saharan African countries, including the Democratic Republic of the Congo and Zimbabwe, likely due to lack of resources. Second, some countries are quite effective in rolling out programmes that reach the most disadvantaged. In Algeria, a large-scale programme offers an annual education allowance, equivalent to US\$23, to 3 million primary and secondary school students from families earning less than US\$60 per month. The allowance is paid at the beginning of the school year to help with school supply purchases. The total outlay of US\$68 million corresponds to 6% of public education expenditure (Algeria Ministry of National Education, 2015). While the programme has relatively low coverage, it is reasonably well targeted (38% of the poorest but 10% of the richest receive support). In Nepal, scholarships have been found to have a progressive impact, although the MICS evidence refers only to the incidence of support, not its volume.

The third group comprises countries with programmes that are large but relatively untargeted. Bangladesh has administered a stipend to primary school students for 20 years, with some limited targeting to poorer subdistricts, and there is a blanket stipend to female secondary school students. In Thailand, a large part of support goes to richer households (59% of the poorest but 44% of the richest receive school tuition support). The same is true for Georgia, whose limited tuition support programme is regressive, favouring the richest (**Figure 21.8**).

#### **FIGURE 21.8:**

**Countries' efforts and success in supporting the education of the poorest households vary considerably** Coverage and targeting effectiveness of programmes offering support for school tuition and other school-related support, by household wealth quintile, selected countries, 2017–19



### **AID EXPENDITURE**

As early as 1970, a UN resolution set 0.7% of gross national income (GNI) as the target for official development assistance (ODA). This target has been a reference point for members of the OECD Development Assistance Committee (DAC) (OECD, 2021a). However, ODA has hovered around 0.3% of GNI since the 1970s. The GEM Report estimates that if the target had been met, US\$3.3 trillion in additional ODA would have been disbursed over 1990–2016.

Of the 30 DAC countries, only Denmark, Luxembourg, Norway, Sweden and the United Kingdom spent more than 0.7% in 2019. In 2020, just as Germany became the next DAC and G7 country to reach the target, the United Kingdom announced it would lower its spending to 0.5% from 2021 (Loft and Brien, 2021). In any case, the United Kingdom is the only country earmarking special drawing rights, an international reserve asset held by the International Monetary Fund, as ODA (Miller and Roger, 2021), so its spending may have been overestimated.

The most recent data on international aid to education are for 2019, the year prior to the onset of COVID-19.<sup>1</sup> Aid disbursement remained constant in 2019, relative to 2018, at US\$15.3 billion. A decrease in aid to basic education by US\$504 million was offset by increases in aid to secondary (by US\$203 million) and post-secondary education (by US\$358 million) (**Figure 21.9a**). As a result of a small decline in total ODA, the share of education continued the upward trend it began in 2015, reaching 8.6% of total aid and 11.6% of aid allocated directly to specific sectors (**Figure 21.9b**).



1. For detailed data on aid to education, see the Aid Tables in the annex.

#### FIGURE 21.9 CONTINUED:

The share of aid allocated to education has risen slightly since 2015

b. Share of education in aid, 2003–19



The amount of aid to basic education per primary school-age child is US\$14 in low-income countries (or 29% of the US\$48 per child spent by governments, although, as argued later, only a fraction of aid goes through governments) and US\$8 in lower-middle-income countries. For a young person of secondary school age, the amount is US\$7 in low-income and US\$3 in lower-middle-income countries.

#### AID EFFECTIVENESS REMAINS AN ELUSIVE CONCEPT

The small amounts of aid disbursed make the question of effectiveness more relevant than ever. As aid effectiveness means different things to different people, there is a risk of contradictory conclusions. Four definitions offer a useful framework on which to build a common understanding among various constituencies (Janus et al., 2020) (**Table 21.1**).

The first definition is popular among development practitioners. It is based on the aid effectiveness principles, agreed in the Paris Declaration on Aid Effectiveness in 2005 and adapted at the Busan High-Level Forum on Aid Effectiveness in 2011, where the Global Partnership for Effective Development Cooperation (GPEDC) was established and indicators were defined (Table 21.2). The assumption behind the principles is that aid's impact on development will increase if donor and country actions reflect development priority ownership, a focus on results, transparency and shared responsibility, and development partnerships. A framework with 12 sets of indicators is monitored every 3 years (OECD and UNDP, 2014, 2016, 2019). While the principles resonate with development partners, this definition of aid effectiveness may be narrow. Adherence to principles may not be necessary for aid to have an impact, nor will adherence be sufficient, given the complexity of the development process. Many of the indicators are subjective and contested, and thus may not reflect actual adherence to the principles.

The second definition is popular with the academic community. Cross-country or time-series research analyses macro effects of aid on, for example, growth or poverty. This type of research has also explored aid's negative side effects, especially how it changes prices or incentives and diverts economic activity to sectors that may undermine countries' long-term growth potential. This approach tends to look beyond aid to the relative impact of all financial flows to a country. However, a focus on growth as an outcome may not be sufficient; information on growth mechanisms is also important. Moreover, it is difficult to attribute any observed impact to aid: Data quality is often inadequate, while the relationship between aid and macro effects is difficult to model, especially as effects are realized with considerable lags and several factors are simultaneously at play.

# TABLE 21.1:Definitions of aid effectiveness

Macro level	1. Principles Paris Declaration GPEDC	2. Macro effects Cross-country data Subnational data
Micro level	4. Organizations Public management Results and adaptation	3. Interventions Outcome monitoring Experimental evaluations
	Activities	Outcomes

Source: Janus et al. (2020).

#### Correspondence between the Paris and Busan monitoring indicators of aid effectiveness

Paris indicators	Busan indicators
<ol> <li>Countries put in place national development strategies with clear strategic priorities.</li> </ol>	
<ol><li>Countries develop reliable national fiduciary systems or reform programmes to achieve them.</li></ol>	Ownership and results
<ol> <li>Coordinated programmes aligned with national development strategies provide support for capacity development.</li> </ol>	<ol> <li>Development cooperation is focused on results that meet developing countries' priorities.</li> </ol>
<ol> <li>Donors align their aid with national priorities and provide the information needed for it to be included in national budgets.</li> </ol>	6. Aid is on budgets which are subject to parliamentary scrutiny.
<ol><li>Country structures are used to implement aid programmes rather than parallel structures created by donors.</li></ol>	
<ol> <li>As their first option, donors use fiduciary systems that already exist in recipient countries.</li> </ol>	9a. Quality. 9b. Use of developing country public financial management and
5b. As their first option, donors use procurement systems that already exist in recipient countries.	procurement systems. 10. Aid is untied.
8. Bilateral aid is not tied to services supplied by the donor.	
<ol> <li>Countries have transparent, measurable assessment frameworks to measure progress and assess results.</li> </ol>	Transparency and mutual accountability         4.       Transparency: Information on development cooperation is publicly available.         5.       Development cooperation is more predictable.         5a.       Annual predictability         = proportion of aid disbursed within fiscal year in which it was scheduled
7. Aid is released according to agreed schedules.	<ul> <li>by cooperation providers.</li> <li>5b. Medium-term predictability <ul> <li>proportion of aid covered by indicative forward spending plans provided at the country level.</li> </ul> </li> <li>7. Mutual accountability strengthened through inclusive reviews.</li> </ul>
12. Regular reviews assess progress in implementing aid commitments.	<ol> <li>Inclusive development partnerships</li> <li>Civil society operates within an environment that maximizes its engagement in and contribution to development.</li> <li>Engagement and contribution of the private sector to development.</li> <li>Gender equality and women's empowerment.</li> </ol>
9. Aid is provided through harmonized programmes coordinated among donors.	
10a. Donors conduct their field missions together with recipient countries.	
10b. Donors conduct their country analytical work together with recipient countries.	
Source: OECD and UNDP (2014).	

The third definition is also popular in more recent academic research. It is at the foundation of the 'what works' agenda, which focuses on the effectiveness of individual interventions and projects. This approach is characterized by rigorous methodology, notably randomized controlled trials, for evaluating project impact, along with aggregation of findings through meta-analyses and systematic reviews.

The main, widely debated weakness of this approach is the generalizability of its findings. It is questioned whether results can be translated at scale outside the highly controlled environment in which projects and evaluations were carried out, without considering political economy factors.

The fourth definition, which may appear the most narrowly conceived, is also the most neglected. It refers to how development organizations are managed and the effect of their procedures on improving aid effectiveness. Do organizations 'practice what they preach'? In other words, do they apply knowledge from results in their operations? Are they are swayed by incremental thinking and political pressures?

#### **FIGURE 21.10:**

## Only a tiny part of bilateral aid goes through recipient governments

Aid to education and to basic education, by channel and donor type, 2017–19



A dimension not captured in this framework, which primarily focuses on the donor-recipient relationship, is the extent to which aid and collective action are effective in delivering global public goods. Donors often fail to coordinate with each other or they free-ride on others' contributions.

The following discussion focuses on the first definition and its four principles and related indicators. These tend to characterize donors' overall policies, not the education-specific ones. Accordingly, the discussion uses additional indicators that help bring the discussion closer to the education context.

#### Ownership: While donors align with national priorities, budget support and use of national budgets has fallen

Ownership issues are related to sustainability. A project-based approach to aid, while appropriate from the donor perspective, may not be closely linked to recipients' priorities. Provisions may be lacking to ensure that governments incorporate and build on project results.

Progress towards the Busan principle of ownership is measured through three sets of indicators: alignment with priorities, whether aid is on budget, and the use of government systems and budgets. Tied aid, a fourth indicator, is not discussed here.

The first indicator is alignment with national development priorities and can refer to stages throughout the process, from objective setting to result evaluation. Such alignment can be observed at the level of country strategy or projects. Overall, the 2018 GPEDC review found that 93% of development partners 'include development priorities that have been jointly identified with the partner country' but only 65% 'use government data and statistics to report on the strategy's results indicators'. At project level, 62% are based on the national development, sector or ministerial plans (OECD and UNDP, 2019). However, such information is not specifically available for the education sector.

The second indicator is the extent to which aid is 'on budget' and therefore subject to parliamentary scrutiny. The term 'on budget' should not be interpreted literally. It means that a development agency has recorded planned funding in the budget approved by parliament. It does not indicate whether the development agency used the recipient government's budget process to disburse funds. According to GPEDC reviews, the proportion of aid recorded on budget and subject to parliamentary scrutiny increased from 57% in 2011 to 66% in 2016 but fell to 61% in 2018 (Li et al., 2018).

The third indicator is the use of government systems and budgets. Analysis of how aid was made available in 2017–19 shows that just 12% of bilateral donor total aid to education and 8% of aid to basic education was channelled via recipient governments; the respective shares increase to 17% and 20% if aid going through the Global Partnership for Education (GPE) is added, as it tends to make disbursements through governments. By contrast, multilateral donors provided 54% of total aid to education and 41% of aid to basic education via recipient governments (**Figure 21.10**).

Bilateral donors used a variety of channels to disburse aid to basic education. In 2019, 66% of Canadian aid, 88% of Norwegian aid and 76% of Swedish aid went through multilateral organizations. France provided 50% of its aid, and the Republic of Korea 54%, through their own government institutions. Switzerland had the largest share of aid to basic education, 52%, channelled through non-governmental organizations. The United States spent the highest share through private sector organizations, 32%; the United Kingdom had the next highest share at 9%.

Another related proxy measure in the OECD-DAC Creditor Reporting System (CRS) database is aid 'type'. Of total aid in 2017–19, core contributions and pooled programmes and funds accounted for 12% and budget support specific to education for 4%. The GPE is a key example of the latter approach (**Box 21.1**). By contrast, project-type interventions accounted for almost 50% and international experts for 9% (**Figure 21.11**).

#### FIGURE 21.11:

Donors favour project-based interventions



GEM StatLink: https://bit.ly/GEM2021\_fig21\_11 Source: GEM Report team analysis based on OECD-DAC CRS database (2021).

#### BOX 21.1:

#### The GPE is an example of efforts to adhere to aid effectiveness principles

The Global Partnership for Education was formed as the main collective effort to address core issues of aid effectiveness in education, such as alignment with national education sector plans and joint sector reviews. Between 2003 and 2020, it approved implementation grants worth a total of US\$6.4 billion, including US\$1 billion approved in 2020 (GPE, 2021). This was the second time since 2013 that approved grants exceeded US\$1 billion. The high number of grants approved in 2013 had enabled the GPE to maintain a high disbursement rate, around US\$500 million per year, between 2015 and 2017. However, the growth of cumulative grant approvals slowed considerably, to single-digit numbers, between 2015 and 2019, and the annual volume of disbursed grants declined by 38% between 2014–17 and 2018–19.

The lag between approvals and disbursements is about three years. In the current circumstances, the high volume of approved grants in 2020 should lead to accelerated disbursements. Total disbursement increased from US\$237 million in 2019 (or 4.6% of total aid to basic education to low- and lower-middle-income or unspecified countries, down from a high of 11.4% in 2014) to US\$411 million in 2020 (**Figure 21.12**). Disbursements were expected to increase faster in 2021.

The transition to the new GPE financing and funding framework may have contributed to the slowdown in disbursements. While there is no direct evidence concerning the GPE, the Multilateral Organization Performance Assessment Network's institutional assessment of another global financing mechanism, the Global Fund to Fight AIDS, Tuberculosis and Malaria, suggested that its transition to a new funding model may have contributed to lower-than-expected disbursements (MOPAN, 2017).

A lead indicator of future GPE activity, donor contributions to the GPE Fund, which follow shortly after pledges made in replenishment rounds, suggests that disbursement levels should pick up even more. At the GPE Financing Conference in February 2018, donors pledged US\$1 billion more for 2018–20 than they had for the preceding three-year period, 2015–17. In the conference's aftermath, total donor contributions to the GPE Fund in 2018 (US\$638 million) were 50% higher than the average donor contribution over 2007–17 (US\$426 million), although only 15% higher than the second-highest year (US\$553 million in 2014). The grants approved in 2019 do not reflect the record donor contribution made to the GPE Fund in 2018. Moreover, slower-than-expected approvals and disbursements in 2018–19 inadvertently helped position the GPE to respond rapidly to the COVID-19 crisis in 2020 and 2021.

In July 2021, at its replenishment for 2021–25, the GPE raised \$4 billion. As it switches to its new strategy and funding model to 2025, it will be essential for it to avoid the obstacles that slowed down disbursements in 2018–19 and to address key questions in relation to effectiveness. On the one hand, the GPE adds costs, since it acts as an intermediary between bilateral donors and multilateral grant agents – just one of which, the World Bank, manages 75% of all grants. On the other hand, the GPE improves decision-making processes, resulting in allocations that are better targeted at the poorest countries than those of most other donors (Akmal et al., 2021).

#### FIGURE 21.12:

# GPE approvals and disbursements slowed down in 2015–19

GPE grant approvals and disbursements per year, 2003–20



Source: GEM Report team analysis based on GPE data.

As noted earlier, one measure of donor preparedness to put aid 'on budget' is the willingness to allocate funds through direct budget support. Between 2002 and 2019, the share of such support fell by two thirds among all donors (from 6.6% to 2.5%) and by three quarters among DAC donors. The European Union disbursed 32% of its aid through direct budget support in 2002, but by 2019 the share was 3% (**Figure 21.13a**). The United Kingdom, which disbursed 19% of its aid in 2003 through direct budget support, stopped using this modality in 2017, as had Sweden a year earlier (**Figure 21.13b**). Yet budget support has been credited with benefits that project support lacks, notably increased accountability and transparency in budgeting and higher leverage effects (Caputo et al., 2011; Dijkstra, 2018; Koch et al., 2017).

Regarding the indicator referring to the percentage of aid going through public financial management and procurement systems, only 53% of development partners use country systems for budget execution, financial reporting, auditing and procurement. Multilateral development banks are the most likely to use these systems (57%) followed by bilateral DAC donors (55%), vertical funds (40%) and UN agencies (12%). Use of the systems is more common in lower-middle-than low-income countries (OECD and UNDP, 2019). Results are not available for aid to education.

# Results focus: Targeting aid where needed remains a challenge

Effective aid should lead to development results. While results can be monitored, it is often difficult, even after painstaking research, to link them to aid. It is challenging to assess the effectiveness of multiple interventions on the basis of available evidence, even if conclusions about the effect of individual interventions are available. Since it is thus more common to define and monitor results at the level of intention, one of the results most closely monitored is effectiveness in targeting.

It is often argued that aid should focus on the countries most in need, not just to achieve equity but also for effectiveness. SDG indicator 4.5.5 focuses on the percentage of aid to education going to low-income countries. Analyses of targeting tend to stop at the national level, yet arguably they may be as relevant at the subnational, population group or even individual level. Little consensus exists on which areas or levels of education aid should focus on. Debates continue over the relative emphasis on ensuring that all children fulfil their right to acquire fundamental literacy and numeracy skills compared with supporting countries in developing all levels of their education systems.

The aid effectiveness agenda is relatively silent on achieving equitable allocation of aid resources. In principle, even taking differences in donor policies into account, disparity in allocations should not be justified. In practice, many considerations lead to a deviation from this principle. In addition, available data sometimes prevent an accurate assessment of equality in aid disbursements.

From a regional perspective, assessment of targeting is hampered by the fact that a considerable and increasing amount of aid is not linked to any particular recipient country. The share of aid to basic education not tied to specific countries increased from 11% in 2009 to 21% in 2019, or from US\$0.6 billion to US\$1.3 billion. This makes it harder to interpret the apparent fall in the share of aid to basic education in sub-Saharan Africa, which went from US\$2 billion to US\$1.4 billion between 2009 and 2014. It has since increased, reaching US\$1.7 billion in 2019.

#### **FIGURE 21.13:**



Direct budget support, a hallmark of aid effectiveness, has fallen out of favour

Two explanations seem plausible. First, betweenUS\$702009 and 2019, aid to unspecified recipients averagedthe ca\$1 billion per year. Of that, average annual disbursementsthe reby the GPE accounted for some US\$366 million,as higof which about 79%, or about US\$290 million, wentSouthto sub-Saharan Africa. Therefore, aid disbursed bythethe GPE (which the OECD-DAC records as havingWithirunspecified recipients) may explain part of the apparentin amounddecline in aid to sub-Saharan Africa. But this doesneed.not suggest that aid to sub-Saharan Africa has beenprimaincreasing. A second explanation is that aid doubled(US\$9from US\$0.8 billion in 2009 to US\$1.6 billion in 2018 inGuineNorthern Africa and Western Asia because conflict anda muchdisplacement crises in Iraq, the Syrian Arab Republicsome

and Yemen led donors to reallocate resources. There are no large regional differences in aid to basic education per capita except for Northern Africa and Western Asia. The Palestinian school system is unique

education per capita except for Northern Africa and Western Asia. The Palestinian school system is uniquely supported by aid through the United Nations Relief and Works Agency for Palestine Refugees in the Near East; the per capita amount is substantial: almost b. Direct budget support as share of total aid, selected bilateral donors 2002–19



US\$70 per child in 2017. Even excluding Palestine from the calculation, Northern Africa and Western Asia is the region with the highest aid at US\$19 per child, twice as high as in sub-Saharan Africa (US\$11) or Central and Southern Asia (US\$7) (**Figure 21.14**).

Within regions or country income groups, the variation in amounts of aid per child is larger, especially relative to need. In sub-Saharan Africa, Liberia received US\$80 per primary school-age child, almost 10 times the average (US\$9) and 5 times the median (US\$18) in the region. Guinea-Bissau and Sao Tome and Principe also received a much higher than average amount. Overall, there is some relationship between need, expressed in terms of average primary completion rate, and amount of aid per capita received (Figure 21.15). Still, considerable differences remain. Burundi and Rwanda have similar primary completion rates but Rwanda received eight times more aid per capita (US\$26 vs US\$3). Conversely, Chad and the United Republic of Tanzania received the same amount of aid to basic education per capita but the latter has a much higher primary completion rate.

#### **FIGURE 21.14:**

# Aid per student is higher in Northern Africa and Western Asia than in other regions

Aid to basic education per school-age child, selected regions, 2009–19



Source: GEM Report team analysis based on OECD-DAC CRS database (2021).

Inevitably, there are limitations to such analysis. Completion rates are a robust and widely available measure of need, but others, such as learning outcomes, should also be considered. Average allocations do not adequately describe distribution of aid within a country and populations most in need. Comparison of aid levels and need levels is also incomplete because it does not include the one fifth of aid to basic education that has unspecified recipients. Still, such comparison is a necessary component of a work plan for a global education financing coordination architecture.

#### Partnerships: It is important to prevent fragmentation

The 2015 Addis Ababa Agenda for Action for SDG financing recognized the importance of multi-stakeholder partnerships in external financing. Education-related examples include the GPE and the Education Cannot Wait fund. These mechanisms are intended to address a common criticism of aid - that it is distributed in fragmented, overlapping or complex channels. In 2019, the average number of donors to low- and middle-income countries was 25, with shares varying from 2 in the Federated States of Micronesia to 37 in Kenua. In 23 of 78 countries, governments had to deal with more than 30 donors. Although the relationship between number of donors and amount of aid is positive, there is variation in donor numbers for a given aid level. For example, the number of donors for Burundi and Niger is 23 each, but Niger receives 5 times as much aid (Figure 21.16). Papua New Guinea and Uzbekistan receive the same amount of aid, around US\$50 million, but Uzbekistan has more than twice as many donor partners.

Excluding the GPE and UNICEF, which are not clearly identifiable in the OECD database, the top five donors in basic education – the European Union, Germany, the United Kingdom, the United States and the World Bank – account for 60% of aid to basic education in low-income countries. However, per-country variation is wide, e.g. they account for 9% in Eritrea and 93% in the Democratic Republic of the Congo (**Figure 21.17**). The GPE allocated some US\$150 million to 13 of 29 low-income countries in 2019, representing 16% of the total amount received by low-income countries that year.

#### **FIGURE 21.15:**

#### Wide disparity among countries exists in aid per student

Aid to basic education per child and primary completion rate, sub-Saharan Africa, 2016



GEM StatLink: https://bit.ly/GEM2021\_fig21\_15

Source: GEM Report team analysis based on OECD-DAC CRS database (2021) and the World Inequality Database on Education.

#### **FIGURE 21.16:**

#### **On average, 25 donors are active in every low-income country** Number of donors and aid to education, low-income countries, 2019



FIGURE 21.17:

The top five donors account for 60% of aid to basic education in low-income countries Composition of aid to basic education in low-income countries, five largest donors, 2017–19



As many donors do not work directly with governments, the presence of large numbers of donors for small-scale aid, with different reporting requirements and planning cycles, can have considerable negative effects. It fragments efforts, raises transaction costs, reduces ownership, stretches capacity and distorts incentives in public administration in recipient countries, especially if donor coordination mechanisms are not well established. While multiple donors can arguably support accountability by forcing each other to perform better, they can also lead to wasted effort. Partnerships, therefore, are one of the four Busan principles, even if the emphasis is on civil society participation, gender equality and private-sector involvement.

It is often asked whether more aid to education should be channelled through multilateral organizations, as their allocation decisions are less affected by historical and geopolitical ties and more likely to align with needs. Bilateral donors channel funding through them because of their capacity to fill gaps in hard-to-reach places. Multilaterals also have relatively strong technical capacity and convening power (Rose et al., 2013). But bilateral DAC donors are the main source of aid to education, accounting for an average of 61% of aid to basic and secondary education between 2003 and 2019. The share declined from about 70% in 2007 to 57% in 2019, but mainly lost ground to non-DAC bilateral donors rather than to multilateral donors (**Figure 21.18**). Again, however, it is not easy to identify trends in the relative shares of bilateral and multilateral donors, as the CRS does not explicitly identify donors such as the GPE and UNICEF.

By comparison, a larger volume and percentage of aid to the health and population sector is disbursed through multilateral donors, such as the Global Fund and Gavi, the Vaccine Alliance. Volume increased from US\$2.2 billion in 2003 to US\$9.9 billion in 2019, while in education it increased from US\$1.3 billion to US\$4 billion (**Figure 21.19**). The respective shares of multilateral donors increased from 19% to 29% in education and from 30% to 44% in the health and population sector. Although incomplete data in education may explain part of the gap, there is a clear difference in the two sectors' aid structures.

#### FIGURE 21.18:

The share of bilateral donors remains high

Distribution of aid to education by type of donor, 2003-19



# Transparency: Aid reporting improvements are still pending

The OECD-DAC CRS database clearly identifies one of the three largest multilateral donors in education, the World Bank, but not the other two, the GPE and UNICEF. A code was created in 2019 to identify the GPE as a donor, but by the time of the January 2021 CRS release, the problem had not been resolved. UNICEF is registered as a donor, but not as a recipient, while many of its programmes are not funded by ODA. These gaps hamper informed analysis of patterns and trends.

The only way to understand the GPE flows is to look at channel records. The GPE averaged a total of US\$459 million per year from bilateral donors to the education sector over the three years from 2017 to 2019. Of this amount, primary education received US\$141 million. If GPE had been recorded as a donor, it would have been the second largest, representing 16% of the total amount received by low-income countries. Further analysis shows that bilateral donors disbursed US\$527 million to the GPE in 2019, but data reported by the GPE show disbursements of US\$226 million. This inconsistency highlights the time lag in disbursement dates from donors to the GPE (captured in the CRS) and from the GPE to grant agents (captured in GPE data).

For 2017–19, the CRS reported disbursements by UNICEF of US\$85 million per year. Yet the 2019 figure in UNICEF's annual report is US\$1.2 billion, or 21% of its total spending. Three factors may account for this discrepancy. First, 28% of UNICEF revenue is from private sources and therefore not recorded as ODA. Second, 41% of its total expenditure, or US\$485 million, is emergency related and may not be recorded as ODA. Finally, like other major multilateral donors, UNICEF does not report earmarked and thematic funds to the CRS (Tortora and Steensen, 2014; UNICEF, 2021).

#### Conclusion

Aid effectiveness is no longer at the top of donor priorities, as it was 15 years ago. Many of the initiatives that mobilized the international community and led to

#### **FIGURE 21.19:**

#### Education lags behind health in aid disbursed through multilateral donors

Volume and share of bilateral and multilateral donors in the education and health and population sectors, 2003–19



the Paris Declaration, such as direct budget support and sector-wide approaches, have fallen out of favour, with increased pressure for short-term 'results' and more attention to risk management. The GPE, established in that period of commitment to aid effectiveness, has been facing more pressure to comply with results and risk management concerns. While the GPEDC, established in 2011, continues its regular review process every three years, monitoring indicators often hide more than they reveal. However, it is worth considering whether sector-specific aspects of aid effectiveness can be addressed within the GPEDC framework.

With respect to the four Busan principles and their application to education, improvement is needed overall. It is hard to assess progress on ownership. Alignment with education sector plans appears to be strong but may be nominal; progress on that front is probably best made at the country level through a well-informed and inclusive joint sector review process. Very little aid is channelled through government systems using government mechanisms. Funders disagree about which parts or levers of the education system to prioritize for support.

While the Busan principle of focus on results is quite broad, it is believed to be the right direction, as aid generally targets the countries – or regions and groups within countries – that are most in need. However, there is still considerable disparity in aid allocation, with some countries favoured and others left out. Good and more timely data are needed to strengthen accountability on how aid responds to need.

The partnership element needs to be strengthened. Multilateral donors channel a much smaller share of aid in education than in health, although support to multilateral donors in education is underestimated due to data issues that undermine transparency. A clear framework is needed, based on benchmarks to identify country needs (**Chapter 9**) and criteria to hold multilateral donors accountable for resource allocation decisions (for which they have their own governing bodies) and their ability to coordinate support for achieving benchmarks. Ensuring provision of the most basic of all global public goods, i.e. regular data of high quality on a core set of SDG 4 indicators, is a worthy objective but it has been a prominent example of aid effectiveness failure. Resolving the challenge of data collection, based on a common plan, would be a quick win and would boost confidence in what a coordination mechanism could achieve.

Ultimately, though, the most important question is whether aid is relevant and focuses on the right priorities through the right approaches. These are difficult, politically charged questions, which require strong leadership to achieve consensus.

## HOUSEHOLD EXPENDITURE

Households spend a significant amount to support their children's education – and more so in poorer than in richer countries. However, it is not easy to piece together evidence to understand these out-of-pocket contributions. The UIS asks countries to submit estimates of household education spending, but for 2012–18 only 68 countries, or 33%, provided at least one observation. Of those, 38 were high-income countries; no data were available for more than 3 in 4 low- and middle-income countries.

The source of information is household income and expenditure (or budget) surveys. The vast majority of countries conduct such surveys. They form the basis for estimating comparable macroeconomic indicators, such as inflation and poverty, and tend to follow common principles. But data sets vary in accessibility and quality. Significant resources are required to standardize them so that comparable data on issues such as education expenditure can be extracted (Chapter 4).

To fill this gap, the GEM Report team relied on the fact that, unlike the data sets, the household budget survey reports are publicly available. In most cases, they include a summary table of how households allocate expenditure to various purposes. Almost all countries separate out the share of education in household consumption expenditure. There are often significant differences between countries in the questions asked, which can affect the estimated shares. For instance, some countries direct questions to the entire household, while others collect information on individual members. The list of items included, how they are grouped and recall periods also differ. However, averaging over many thousands of households cancels out some of these potential biases and errors.

Analysis of reports from about 100 low- and middle-income countries between 2009 and 2020 found that, on average, households allocated 3.2% of their total expenditure to education. The share varied from less than 1% in some south-eastern European countries, including Bosnia and Herzegovina, North Macedonia and Romania, and in several sub-Saharan African countries, including Burundi, Ethiopia and Lesotho, to more than 6% in countries with a high percentage of private schools, such as Haiti and Lebanon, and in other sub-Saharan African countries, including Rwanda, Uganda and Zambia. The country with the largest share was Ghana, where three successive rounds of the Living Standards Survey found that the share of education spending was not only the world's largest but had increased from 8.9% in 2005/06 to 10.6% in 2012/13 and 13.1% in 2016/17 (Ghana Statistical Service, 2008, 2016, 2019).

Household education expenditure as a share of GDP can be estimated by multiplying the share of education in household consumption expenditure with data on household consumption expenditure as a share of GDP. The latter varies by country because of considerable differences in economic, social and political conditions. Globally, household consumption as a share of GDP was 59% in 2020, with shares ranging from less than 40% in oil-producing countries, including Oman, Saudi Arabia and the United Arab Emirates, to close to or even exceeding 100% in poor countries that rely on remittances, such as Haiti, Liberia and Somalia (World Bank, 2021).

This indirect estimate of household education expenditure as a share of GDP can be complemented with direct estimates provided by the UIS and OECD, generating a data set of almost 150 countries. On average, households spend 1.9% of GDP on education, compared with the average 4.5% of GDP by governments; this means households account for 30% of total education spending. Regional shares range from 16% in high-income to 39% in low- and lower-middle-income countries, reaching 37% in Central and Southern Asia and in Latin America and the Caribbean, and 39% in sub-Saharan Africa, but only 12% in Europe and Northern America (**Figure 21.20**).

#### FIGURE 21.20:

#### Households account for 30% of total education spending globally

a. Education expenditure as share of GDP, by source, region and income group, 2010s



GEM StatLink: https://bit.ly/GEM2021\_fig21\_20

Source: GEM Report team analysis based on national household budget survey reports and UIS and OECD data.

There is significant variation in volume of household spending. For instance, in 39% of low-income and 26% of lower-middle-income countries but just 6% of upper-middle-income and 2% of high-income countries, households account for more than 50% of total education spending. There is also significant variation within each income group. For instance, among low-income countries, households account for 5% of total education spending in Ethiopia and 10% in Mozambique but 59% in Uganda and 73% in Liberia. Among lower-middle-income countries, the shares are 5% in Lesotho and Sao Tome and Principe but 71% in Bangladesh and 72% in Nigeria. Among upper-middle-income countries, households account for 7% of total education spending in Romania and 9% in the Russian Federation vs 55% in Jordan and 74% in Lebanon (**Figure 21.21**).



Two small children play with toy xylophones at the Sayariy Warmi early childhood development centre in Sucre, Bolivia.

WARMI

CREDIT: UNICEF/Giacomo Pirozzi



# Statistical tables<sup>1</sup>

Table 1 presents basic information on demographic and education system characteristics as well as on domestic education finance. Tables 2-7 are organized by each of the seven SDG 4 targets (4.1–4.7) and three means of implementation (4.a-4.c). The tables mainly focus on the SDG 4 monitoring framework of 43 internationally comparable indicators: 12 global and 31 thematic indicators. An additional indicator, 'Proportion of children/young people prepared for the future, by sex', is the product of the two global indicators of SDG target 4.1. The UNESCO Institute for Statistics (UIS) reported on all indicators in 2021 except indicator 4.7.3 (Table 1.1).<sup>2,3</sup> The tables also include additional indicators, such as transition from primary to secondary education and student mobility, which are not formally part of the SDG 4 monitoring framework.

## **METHODOLOGICAL NOTES**

Most data in the statistical tables come from the UIS. Where the statistical tables include data from other sources, these are mentioned in footnotes. The most recent UIS data on pupils, students, teachers and education expenditure presented in the tables are from the February 2021 release and refer to the school year or financial year ending in 2019.<sup>4</sup> These statistics refer to formal education, both public and private, by level of education. The statistical tables list 209 countries and territories, all of which are UNESCO Member States or associate members. Most report their data to the UIS using standard questionnaires issued by the UIS itself. For 46 countries, education data are collected by the UIS via the UNESCO-UIS/OECD/Eurostat (UOE) joint data collection questionnaires.<sup>5</sup>

#### **POPULATION DATA**

The population-related indicators used in the statistical tables, including enrolment ratios, number of out-of-school children, adolescents and youth, and number of youth and adults, are based on the 2019 revision of population estimates produced by the UN Population Division (UNPD). Because of possible differences between national population estimates and those of the United Nations, these indicators may differ from those published by individual countries or by other organizations.<sup>6</sup> In the 2019 revision, the UNPD does not provide population data by single years of age for countries with total population of less than 90,000. For these countries, as well as some special cases, population estimates are derived from Eurostat (Demographic Statistics), the Secretariat of the Pacific Community (Statistics and Demography Programme) or national statistical offices.

#### **ISCED CLASSIFICATION**

Education data reported to the UIS are in conformity with the International Standard Classification of Education (ISCED), revised in 2011. Countries may have their own definitions of education levels that do not correspond to ISCED 2011. Differences between nationally and internationally reported education statistics may be due to the use of nationally defined education levels rather than the ISCED level, in addition to the population issue raised above.

<sup>1</sup> The statistical tables are accessible on the GEM Report website at https://en.unesco.org/gem-report/statistical-tables

<sup>2</sup> The Inter-agency and Expert Group on SDG Indicators proposed the 11 SDG 4 global indicators. The UN Statistical Commission adopted them at its 48th session, in March 2017. The United Nations Economic and Social Council adopted them in June 2017. The completion rate (indicator 4.1.2) was added to the list in March 2021 following the 2020 Comprehensive Review.

<sup>3</sup> The Technical Advisory Group on post-2015 education indicators originally proposed the 43 indicators. The Technical Cooperation Group (TCG), whose secretariat is at the UIS, endorsed them, with some changes, to monitor progress towards the SDG 4 targets. Information on indicator methodological developments is accessible at the TCG website, http://tcg.uis.unesco.org/.

<sup>4</sup> This means 2018/19 for countries with a school year that overlaps two calendar years, and 2019 for those with a calendar school year. The most recent reference year for education finance for the UOE countries is the year ending in 2017.

<sup>5</sup> The countries concerned are most European countries, non-European OECD countries, and a changing set of other countries

<sup>6</sup> Where obvious inconsistencies exist between enrolment reported by countries and the United Nations population data, the UIS may decide not to calculate or publish enrolment ratios for some or all levels of education.

#### **ESTIMATES AND MISSING DATA**

Regarding statistics produced by the UIS, both observed and estimated education data are presented throughout the statistical tables. The latter are marked with subscript (i). Wherever possible, the UIS encourages countries to make their own estimates. Where this does not happen, the UIS may make its own estimates if sufficient supplementary information is available. Gaps in the tables may arise where data submitted by a country are found to be inconsistent. The UIS makes every attempt to resolve such problems with the countries concerned, but reserves the final decision on omitting data it regards as problematic. If information for the year ending in 2019 is not available, data for earlier or later years are used. Such cases are indicated by footnotes.

#### AGGREGATES

Figures for regional and other aggregates represent either sums, the percentage of countries meeting some condition, medians or weighted averages, as indicated in the tables, depending on the indicator. Weighted averages take into account the relative size of the relevant population of each country, or more generally of the denominator in the case of indicators that are ratios. The aggregates are derived from both published data and imputed values, for countries for which no recent data or reliable publishable data are available. Aggregates marked with (i) in the tables are based on incomplete country coverage of reliable data (between 33% and 60% of the population [or aggregate denominator value] of a given region or country grouping). GEM Report calculated sums are flagged for incomplete coverage if less than 95% of the population of a given region or country income group is represented among the countries for which data are available.

#### **REGIONAL AND COUNTRY INCOME GROUPS**

In terms of regional groups, the statistical tables use the SDG regional classification of the United Nations Statistical Division (UNSD), with some adjustments. The UNSD classification includes all territories, whether independent national entities or parts of bigger entities. However, the list of countries presented in the statistical tables includes only full UNESCO Member States and associate members, as well as Bermuda and Turks and Caicos Islands, non-member states that were included in the Exploratory Factor Analysis (EFA) statistical tables. The UIS does not collect data for the Faroe Islands, so this territory is not included in the GEM Report despite its status as a UNESCO associate member. In terms of country income groups, the statistical tables use the World Bank groups, which are updated each year on 1 July.

#### SYMBOLS USED IN THE STATISTICAL TABLES

- ± n Reference year differs (e.g. -2: reference year 2017 instead of 2019)
- i Estimate and/or partial coverage
- Magnitude nil or negligible
- ... Data not available or category not applicable

Notes by indicator (**Table I.2**), footnotes to the tables and a glossary provide additional help to interpret the data.

## **TABLE I.1:** SDG 4 monitoring framework indicators

	Indicator
	Target 4.1
410	Proportion of children/young people prepared for the future, bu sex
4.1.1	Proportion of children and young people (a) in Grade 2 or 3; (b) at the end of primary education; and (c) at the end of lower secondary education achieving at least a minimum proficiency level in (i) reading and (ii) mathematics, by sex
4.1.2	Completion rate (primary education, lower secondary education, upper secondary education)
4.1.3	Gross intake ratio to the last grade (primary education, lower secondary education)
4.1.4	Out-of-school rate (primary education, lower secondary education, upper secondary education)
4.1.5	Percentage of children over-age for grade (primary education, lower secondary education)
4.1.6	Administration of a nationally-representative learning assessment (a) in Grade 2 or 3; (b) at the end of primary education; and (c) at the end of lower secondary education
4.1.7	Number of years of (a) free and (b) compulsory primary and secondary education guaranteed in legal frameworks
421	
4.2.1	Proportion of children aged 24-59 months who are developmentally on track in health, learning and psychosocial well-being, by sex
4.2.2	Participation rate in organized learning (one year before the official primary entry age), by sex
4.2.3	Percentage of children druget 5 years experiencing positive and sumulating nome learning environments
4.2.4	Gross early childhood education enrolment ratio in (a) pre-primary education and (b) early childhood educational development
4.2.3	Toront 4.2
121	larget 4.5
4.3.1	Francipation rate of good rate addition and the rot man education and training in the previous 12 months, by sex
4.3.2	Caross enronment ratio for ter using education by sex
4.5.5	
4.4.1	Target 4.4
4.4.1	Proportion of gouth and addits with mormation and communications technology (ic) skins, by type of skin
4.4.2	Vouth/adult educational attainment rates bulane aroun economic activity status levels of education and programme orientation
J	Tarnat 4 5
4.5.1	Parity indices (female/male, rural/urban, bottom/top wealth quintile and others such as disability status, indigenous peoples and conflict-affected, as data become available)
	for all education indicators on this list that can be disaggregated
4.5.2	Percentage of students in a) early grades, b) at the end of primary, and c) at the end of lower secondary education who have their first or home language as language of instruction
4.5.3	Existence of funding mechanisms to reallocate education resources to disadvantaged populations
4.5.4	Education expenditure per student by level of education and source of funding
4.5.5	Percentage of total aid to education allocated to least developed countries
	Target 4.6
4.6.1	Percentage of population in a given age group achieving at least a fixed level of proficiency in functional (a) literacy and (b) numeracy skills, by sex
4.6.2	Youth/adult literacy rate
4.6.3	Participation rate of illiterate youth/adults in literacy programmes
471	
4.7.1	Extent to which (i) global citizenship education and (ii) education for sustainable development are mainstreamed at all levels in: (a) national education policies, (b) curricula, (c) teacher education and (d) student assessment
4.7.2	Percentage of schools that provide life skills-based Hiv and sexuality education
4.7.3	Extent to which the framework on the world Programme on Human Rights douceation is implemented hationally (as per the UNGA Resolution 59/113)
4.7.4	Percentage of scudents by age group for education reven showing adequate interstantioning in issues relating to group a citizensing and sustainability.
4.7.5	Percentage of 15-3 gear out students showing proncency in knowledge or environmental science and geoscience
4.7.0	Extent to which hadonal education policies and education sector plans recognize a breadth of skins that needs to be emilanced in national education systems
4 - 1	Talget 4.a Drepation of schools offering basic services, but upo of service
4.d.1	Proportion of scholars experiences, by type of service
4.a.2	retentage of students experientelling building in the last 12 months in a primary, and by lower secondary education
4.d.J	Target 4.b
4 h 1	Volume of official development assistance flows for scholarships bu sector and tupe of studu
	Target 4.c
4.c.1	Proportion of teachers with the minimum required gualifications, bu education level
4.c.2	Pupil-trained teacher ratio bu education level
4.c.3	Proportion of teachers gualified according to national standards by education level and tupe of institution
4.c.4	Pupil-qualified teacher ratio by education level
4.c.5	Average teacher salary relative to other professions requiring a comparable level of gualification
4.c.6	Teacher attrition rate by education level
4.c.7	Percentage of teachers who received in-service training in the last 12 months bu tupe of training

## **TABLE I.2:** Notes of indicators in the statistical tables

	Indicator Notes
	Table 1
A	Compulsory education by level Number of years during which children are legally obliged to attend school.
В	Free years of education, by level Number of years during which children are legally guaranteed to attend school free of charge.
C	Official primary school starting age Official age at which students are expected to enter primary school. This is expressed in whole years, not accounting for cut-off dates other than the beginning of the school year. The official entrance age to a given programme or level is typically, but not always, the most common entrance age.
D	Duration of each education level Number of grades or years in a given level of education.
E	Official school-age population by level Population of the age group officially corresponding to a given level of education, whether enrolled in school or not.
F	Total absolute enrolment by level Individuals officially registered in a given educational programme, or stage or module thereof, regardless of age.
G	Initial government expenditure on education as percentage of GDP Total general (local, regional and central, current and capital) initial government funding of education. It includes transfers paid (such as scholarships to students), but excludes transfers received, in this case international transfers to government for education (when foreign donors provide education sector budget support or other support integrated in the government budget).
H	Expenditure on education as percentage of total government expenditure Total general (local, regional and central) government expenditure on education (current, capital and transfers), expressed as a percentage of total general government expenditure on all sectors (including health, education, social services, etc.). It includes expenditure funded by transfers from international sources to government.
I	Initial government expenditure per pupil by level, in constant 2016 PPP US\$ and as percentage of GDP per capita Total general (local, regional and central, current and capital) initial government funding of education per student, which includes transfers paid (such as scholarships to students), but excludes transfers received, in this case international transfers to government for education (when foreign donors provide education sector budget support or other support integrated in the government budget).
J	Initial household expenditure on education as percentage of GDP Total payments by households (pupils, students and their families) for educational institutions (including tuition fees, exam and registration fees, contribution to parent-teacher associations or other school funds, and fees for canteen, boarding and transport) and purchases outside of educational institutions (e.g. for uniforms, textbooks, teaching materials and private classes). Initial funding' means that government transfers to households, such as scholarships and other financial aid for education, are subtracted from what households spend.
	Table 2
Α	Out-of-school children, total number and as percentage of corresponding age group Children in the official school age range who are not enrolled in either primary or secondary school.
В	Education completion rate by level Percentage of children three to five years older than the official age of entry into the last grade of an education level who have reached the last grade of that level. For example, the primary completion rate in a country with a six-year cycle where the official age of entry into the last grade is 11 is the percentage of 14- to 16-year-olds who have reached grade 6.
с	<b>Percentage of pupils over-age for grade by level</b> Percentage of pupils in each level of education whose age is two years or more above the intended age for their grade.
D	Gross enrolment ratio in primary education Total enrolment in primary education, regardless of age, expressed as a percentage of the population in the official age group. It can exceed 100% because of early or late entry and/or grade repetition.
E	Primary adjusted net enrolment rate Enrolment of the official age group for primary education in either that level or the levels above, expressed as a percentage of the population in that age group.
F	Gross intake ratio to last grade of primary education Total number of new entrants to the last grade of primary education, regardless of age, expressed as a percentage of the population at the official school entrance age for that grade.
G	Effective transition from primary to lower secondary general education Number of new entrants to the first grade of lower secondary education in the following year expressed as a percentage of the students enrolled in the last grade of primary education in a given year who do not repeat that grade the following year.
Н	Lower secondary total net enrolment rate Number of pupils of the official school age group for lower secondary education who are enrolled in any level of education, expressed as a percentage of the corresponding school age population.
I	Gross intake ratio to last grade of lower secondary education Total number of new entrants to the last grade of lower secondary education, regardless of age, expressed as a percentage of the population at the official school entrance age for that grade.
J	Upper secondary total net enrolment rate Number of pupils of the official school age group for upper secondary education who are enrolled in any level of education, expressed as a percentage of the corresponding school age population.
K	Administration of nationally representative learning assessment in early grades (grade 2 or 3) or final grade of primary or lower secondary The definition includes any nationally representative, national or cross-national formative low-stakes learning assessment.
L	Percentage of students achieving at least a minimum proficiency level in reading and mathematics The minimum proficiency level in reading and mathematics is defined by each assessment. Data need to be interpreted with caution since the different assessments are not comparable. In the absence of assessments conducted in the proposed grade, surveys of student learning achievement in the grade below or above the proposed indicator grade are used as placeholders.

	Indicator
	Notes
	Table 3
A	Percentage of children aged 36 to 59 months who are developmentally on track in health, learning and psychosocial well-being The UNICEF Early Childhood Development Index (ECDI) is collected through the UNICEF Multiple Indicator Cluster Surveys (MICS) and is a measure of fulfilment of developmental potential that assesses children aged 36 to 59 months in four domains: (a) literacy-numeracy, (b) physical development, (c) social-emotional development and (d) learning (ability to follow simple instructions, ability to occupy themselves independently). The percentage of children who are developmentally on track overall is the percentage of children on track in at least three of the four domains.
В	Under-5 moderate or severe stunting rate Proportion of children in a given age group whose height for their age is below minus two standard deviations from median height for age established by the National Center for Health Statistics and the World Health Organization (WHO). (Source: March–August 2019 UNICEF, WHO and World Bank Joint Child Malnutrition Estimates [JME]. Regional aggregates are JME statistical estimates for the reference year, not weighted averages of the observed country values in the country table.)
C	Percentage of children aged 36 to 59 months experiencing positive and stimulating home learning environments Percentage of children 36 to 59 months old with whom an adult has engaged in four or more of the following activities to promote learning and school readiness in the previous three days: (a) reading books to the child, (b) telling stories to the child, (c) singing songs to the child, (d) taking the child outside the home, (e) playing with the child and (f) spending time with the child naming, counting or drawing things. (Source: UNICEF database.)
D	Percentage of children under 5 years living in households with three or more children's books Percentage of children aged 0 to 59 months who have three or more books or picture books. ( <i>Source</i> : UNICEF database.)
E	Gross early childhood education enrolment ratio in pre-primary education Total enrolment in pre-primary education, regardless of age, expressed as a percentage of the population in the official age group. It can exceed 100% because of early or late entry.
F	Adjusted net enrolment rate one year before the official primary school entry age Enrolment of children one year before official primary school entry age in pre-primary or primary education, expressed as a percentage of the population in that age group.
	Table 4
A	Participation rate in adult education and training Participation rate of adults (aged 25 to 64) in formal or non-formal education and training in the last 12 months. Estimates based on other reference periods, in particular 4 weeks, are included when no data are available on the last 12 months.
В	Percentage of youth enrolled in technical and vocational education Youth (aged 15 to 24) enrolled in technical and vocational education at ISCED levels 2–5, as a percentage of the total population of that age group.
C	Share of technical and vocational education in total enrolment by level Total number of students enrolled in vocational programmes at a given level of education, expressed as a percentage of the total number of students enrolled in all programmes (vocational and general) at that level.
D	<b>TVET share of post-secondary non-tertiary (%)</b> Share of technical and vocational education and training (TVET) in post-secondary non-tertiary enrolment (%).
E	Gross graduation ratio from tertiary (%) Number of graduates from first degree programmes (at ISCED 6 and 7) expressed as a percentage of the population of the theoretical graduation age of the most common first degree programme.
F	Gross enrolment ratio in tertiary education Total enrolment in tertiary education, regardless of age, expressed as a percentage of the population in the five-year age group above the official graduation age from upper secondary. It can exceed 100% because of early or late entry and prolonged study.
G	Percentage of adults (15 and over) with specific ICT skills Individuals are considered to have such skills if they have undertaken certain computer-related activities in the last three months: copying or moving a file or folder; using copy and paste tools to duplicate or move information within a document; using basic arithmetic formulas in a spreadsheet; writing a computer program using a specialized programming language.
Н	Percentage of adults (25 and over) who have attained at least a given level of education Number of persons aged 25 and above by the highest level of education attained, expressed as a percentage of the total population in that age group. Primary refers to ISCED 1 or higher, lower secondary to ISCED 2 or higher, upper secondary to ISCED 3 or higher, post-secondary to ISCED 4 or higher.
I	Percentage of population of a given age group achieving at least a fixed level of proficiency in functional literacy/numeracy skills The threshold level corresponds to level 2 on the Programme for the International Assessment of Adult Competencies scale.
J	Youth (15 to 24)/Adult (15 and above) literacy rate
К	Number of youth (aged 15 to 24)/adult (aged 15 and above) illiterates Number of illiterate youth (aged 15 to 24) and adults (aged 15 and above), expressed as a percentage of the total population in that age group. Literacy data are for 2010–16 and include both national observed data from censuses or household surveys and UIS estimates. The latter are based on the most recent national observed data and the Global Age-specific Literacy Projections (GALP) model. As definitions and methodologies used for data collection differ by country, data need to be used with caution.
	Table 5
	Adjusted gender parity index, by indicator The gender parity index (GPI) is the ratio of female to male values of a given indicator. If the female value is less than or equal to the male value, the adjusted gender parity index (GPIA) = GPI. If the female value is greater than the male value, GPIA = 2 - 1/GPI. This ensures the GPIA is symmetrical around 1 and limited to a range between 0 and 2. A GPIA equal to 1 indicates parity between females and males. (Sources: UIS database; GEM Report team calculations based on national and international household surveus.)
Α	Completion rate, by level
В	Percentage of students with minimum level of proficiency at the end of given level
C	Youth and adult literacy rate
D	Percentage of adults (16 and over) achieving at least a fixed level of proficiency in functional literacy and numeracy skills
E	Gross enrolment ratio, by level
F	Location and wealth disparity The location parity index is the ratio of rural to urban values of a given indicator. The wealth parity index is the ratio of the poorest 20% to the richest 20% of values of a given indicator.
F G	Percentage of students with minimum level of proficiency at the end of given level

	Indicator Notes
	Table 6
A	Extent to which (i) global citizenship education and (ii) education for sustainable development are mainstreamed at all levels in (a) national education policies; (b) curricula; (c) teacher education; and (d) student assessment Three levels are distinguished: low (not reflected or little reflected), medium (somewhat reflected) and high (fully reflected). (Source: UNESCO, 2019.)
В	Percentage of schools providing life skills-based HIV/AIDS education Percentage of lower secondary schools providing life skills-based HIV/AIDS education (all institutions).
C	Percentage of students and youth with adequate understanding of HIV/AIDS and sexuality Voith (aged 15 to 24) who know at least two wave to provent infection and roises at least three micropropriates (Source: LINAIDS, 2019.)
	Totul (age) 15 to 247 who know at least two ways to prevent intertuon and reget at least time miscure phone. OwnED, 2015.7
	Cognitive and non-cognitive achievement in global citizenship and sustainability, including issues related to multiculturalism/interculturalism; gender equality; peace, non-violence and human security; freedom; social justice; and sustainable development.
	Percentage of 15-year-old students performing at or above level 2 of proficiency in scientific literacy
	Scientific literacy is defined as (a) scientific knowledge and its use to identify questions, acquire new knowledge, explain scientific phenomena and draw evidence-based conclusions about science-related issues; (b) understanding of the characteristic features of science as a form of human knowledge and enquiry; (c) awareness of how science and technology shape the material, intellectual and cultural environments; and (d) willingness to engage in science-related issues, and with the ideas of science, as a reflective citizen.
D	Percentage of schools with basic drinking water, basic (single-sex) sanitation or toilets, and basic handwashing facilities Basic drinking water means drinking water from an improved source, and water available at the school at the time of the survey. Basic sanitation or toilets means improved sanitation facilities at the school that are single-sex and usable (available, functional and private) at the time of the survey. Basic handwashing facilities means handwashing facilities with water and soap available at the school at the time of the survey.
E	<ul> <li>Percentage of public schools with:</li> <li>Electricity         Regularly and readily available sources of power (e.g. grid/mains connection, wind, water, solar and fuel-powered generator) that enable the adequate and sustainable use of ICT infrastructure by pupils and teachers to support course delivery or independent teaching and learning needs.     </li> </ul>
	<ul> <li>Internet used for pedagogical purposes</li> <li>Internet that is available for enhancing teaching and learning and is accessible by pupils irrespective of the device used. Access can be via a fixed narrowband, fixed broadband or mobile network.</li> </ul>
	Computers Use of computers to support course delivery or independent teaching and learning needs, including to meet information needs for research purposes, develop presentations, perform hands-on exercises and experiments, share information and participate in online discussion forums for educational purposes. The definition includes desktops, laptops and tablets.
F	Percentage of public primary schools with access to adapted infrastructure and materials for students with disabilities Any built environments related to education facilities that are accessible to all users, including those with various types of disability, enabling them to gain access to use and exit from them. Accessibility includes ease of independent approach, entry, evacuation and/or use of a building and its services and facilities (such as water and sanitation) by all of the building's potential users with an assurance of individual health, safety and welfare during the course of those activities.
G	Percentage of students experiencing school-related bullying in lower secondary education Percent of students subjected to bullying in the past 12 months (or alternative period as available in the source data) at the lower secondary level. The definition of bullying includes, when possible, physical, verbal and relational abuse. This scope reflects current research on bullying as well as the definitions for major international student assessments.
Н	Level of attacks on students, teachers or institutions
	Categorical ranking of the extent to which a country is affected by violent attacks, threats or deliberate use of force in a given period (e.g. the last 12 months, a school year or a calendar year) directed against students, teachers and other personnel or against education buildings, materials and facilities, including transport. The indicator focuses on attacks carried out for political, military, ideological, sectarian, ethnic or religious reasons by armed forces or non-state armed groups. Five levels are captured:
	No incidents reported: No reports of attacks on education were identified.
	Sporadic: Fewer than 5 reported attacks or fewer than 5 students and education personnel harmed.
	Affected: 5–99 reported attacks on education or fewer than 5–99 students and education personnel harmed.
	Heavily affected: 100–199 reported attacks or 100–199 students and education personnel harmed.
	Very heavily affected: More than 200 reported attacks or more than 200 students and education personnel harmed.
I	Internationally mobile students, inbound and outbound numbers enrolled and mobility rates Number of students from abroad studying in a given country, expressed as a percentage of total tertiary enrolment in that country. Number of students from a given country studying abroad, expressed as a percentage of total tertiary enrolment in that country.
ſ	Volume of official development assistance for scholarships Total gross disbursement of official development assistance flows (all sectors) for scholarships (all levels). The sum of the values of regions and country income groups does not add up to the global total because some aid is not allocated by country.
	Imputed student costs Costs incurred by donor countries' higher education institutions when they receive students from developing countries.

	Indicator
	Notes
	Table /
Α	Number of classroom teachers Persons employed full-time or part-time in an official capacity to guide and direct the learning experience of pupils and students, irrespective of their qualifications or the delivery mechanism, i.e. face-to-face and/or at a distance. This definition excludes educational personnel who have no active teaching duties (e.g. headmasters, headmistresses or principals who do not teach) or who work occasionally or in a voluntary capacity in educational institutions.
В	Pupil/teacher ratio Average number of pupils per teacher at a given level of education, based on headcounts of pupils and teachers.
C	Percentage of trained classroom teachers Trained teachers are defined as those who have received at least the minimum organized and recognized pre-service and in-service pedagogical training required to teach at a given level of education. Data are not collected for UOE countries.
D	Percentage of qualified classroom teachers Qualified teachers are defined as those who have the minimum academic qualification necessary to teach at a specific level of education according to national standards.
E	Teacher attrition rate Number of teachers at a given level of education leaving the profession in a given school year, expressed as a percentage of teachers at that level and in that school year.
F	Relative teacher salary level Teacher salary relative to other professionals with equivalent academic qualification. Data refer to actual salaries of all teachers relative to earnings for full-time, full-year workers with tertiary education (ISCED 5 to 8). The indicator is defined as a ratio of salary, using annual average salaries (including bonuses and allowances) of teachers in public institutions relative to the wages of workers with similar educational attainment (weighted average) and to the wages of full-time, full-year workers aged 25 to 64 with tertiary education. Values for secondary education are GEM Report team calculations and represent weighted averages of lower and upper secondary values, weighted by the number of teachers at each level.
G	Percentage of teachers who received in-service training in the last 12 months For data representative of teachers at a level of education or grade: proportion of teachers that have received in-service training in the past 12 months (or time period available in the dataset). For data representative of students' teachers: proportion of students' teachers that have received in-service training in the past 12 months (or time period available in the dataset). For cross-national assessments with more than one assessment in the same level of education, the average of all grades is used.

## **TABLE 1:** Education system characteristics and education expenditure

	EDUCATION SYSTEMS																	
		A		В	с		1	)				E				F		1
	Compulsory Free			y I age		Duration	n (years)		School-age population (000,000)					Enrolment (000,000)				
	1 year of pre-primary	9 years of primary- secondary	1 year of pre-primary	12 years of primary- secondary	Official primar school starting	Pre-primary	Primary	Lower secondary	Upper secondary	Pre-primary	Primary	Secondary	Tertiary	Pre-primary	Primary	Secondary	Tertiary	
SDG indicator	4.2.4	4.1.7	4.2.4	4.1.7														1
Reference year									2019									
Region		% of co	untries				Median						Su	ım				
World	23	74	50	53	6	3	6	3	3	352	728	791	586i	217 <sub>i</sub>	739i	601 <sub>i</sub>	228i	
Sub-Sabaran Africa	2	44	18	22	6	3	6	3	3	78	175	145	91,	25;	174;	61.	8;	
Northern Africa and Western Asia	12	97	58	71	6	3	6	3	3	25	56	57	47i	81	56	47i	201	
Northern Africa		83	50	50	6	2	6	3	3	11	29	25	20i	4	29	21	20. 7i	
Western Asia	17	94	61	78	6	3	6	3	3	15	27	31	22i	4i	26i	26i	13i	
Central and Southern Asia	14	64	50	43	6	3	5	4	3	100	188	259	180	61	188	184	47	
Central Asia	20	100	100	40	7	4	4	5	2	6	6	9	6	2	6	8	2	
Southern Asia	11	44	22	44	6	2	5	3	4	93	182	250	174	58	182	176	45	
Eastern and South-eastern Asia	22	78	38	38	6	3	6	3	3	81	178	179	154	67	184	153	73	
Eastern Asia	29	100	57	43	6	3	6	3	3	58	114	112	99	51	116	98	54	
South-eastern Asia	18	64	22	33	6	3	6	3	3	23	64	68	54i	16 <sub>i</sub>	68i	56i	19i	
Oceania	18	65	55i	64i	6	2	6	4	3	1	4	4	3i	1i	5i	4i	2i	
Latin America and the Caribbean	54	83	68	58	6	2	6	3	2	28	59	66	54i	22i	64i	64i	28i	
Caribbean	27	82	47	58	5	2	6	3	2		4i	4i	3i		2i	2i	1i	
Central America	100	86	86	57	6	3	6	3	2		19	19	16		19	17	6	
South America	75	83	92	58	6	3	6	3	3		35i	42 i	35i		38	43	18i	
Europe and Northern America	26	93	63	72	6	3	6	3	3	38	68	82	63i	33i	68i	86i	50i	
Europe	28	93	60	70	6	3	5	4	3	26	40	55	40 <sub>i</sub>	24i	41i	58i	29i	
Northern America	-	100	100	100	6	1	6	3	3	12	27	28	24i	9i	28i	28i	21i	
Low income	6	42	32	20	6	3	6	3	3	54	106	92	58i	11 <sub>i</sub>	108i	36i	6 i	
Middle income	23	70	47	46	6	3	6	3	3	260	544	613	457	175	552	473	168i	
Lower middle	17	60	33	33	6	3	6	3	3	145	313	372	257	87	312	251	62	
Upper middle	28	78	58	56	6	3	6	3	3	115	231	241	199i	87i	240i	222i	106i	
High income	31	94	63	75	6	3	6	3	3	37	78	87	71i	31i	80i	91i	54i	

A Years of compulsory education, by level.

- B Years of free education, by level.
- C Official primary school starting age.
- D Official duration of education levels in years.
- E Official school-age population by level (for tertiary: the five years following upper secondary).
- F Total absolute enrolment by level.
- G Initial government expenditure on education as % of GDP.
- H Initial government expenditure on education as % of total government expenditure.
- I Initial government expenditure per pupil by level, in constant 2017 PPP US\$ and as % of GDP per capita.
- J Initial household expenditure on education as % of GDP.

#### Note: PPP = purchasing power parity.

Source: UIS unless noted otherwise. Data refer to school year ending in 2019 unless noted otherwise.

Aggregates represent countries listed in the table with available data and may include estimates for countries with no recent data.

- (-) Magnitude nil or negligible.
- (...) Data not available or category not applicable.
- ( $\pm$  n) Reference year differs (e.g. -2: reference year 2017 instead of 2019).
- (i) Estimate and/or partial coverage.

	FINANCE														
G	н				I	I .				J					
5 6				Governmen	nt education	n expenditu	re per pupil			_					
of GL	노르		2017 F	PPP US\$			% of GDP	of GDP per capita							
edu (% o	hare mer (%)									que					
nment diture	tion sl jovern diture	rimary	Ð	dary	Ð	rimary	Ð	dary	Ð	ehold e diture GDP)					
Gover	Educa otal <u>o</u> expen	Pre-pi	Prima	econ	fertia	Pre-pi	Prima	econ	fertia	House Stypen					
	1a2			•	4.4	5.4		••							
					2019										
	Median														
		2 4 40			5 202			20							
4.4	14.1	2,140i	3,133i	4,152i	5,382i	12 <sub>i</sub>	16i	<b>20</b> i	25i	1.6					
3.8	16.8	80i	279i	443i	2,722i	3 i	12 <sub>i</sub>	17 <sub>i</sub>	74i	2.4					
3.1i	10.4i				4,567i				21i	1.6					
				3,491i				32i							
3.1i	10.3i	2,190i	6,337i		3,241i	11i	18i		20i						
4.0	15.7	149i	350i	887i	1,584i	2 i	10i	16i	22i	2.4					
5.3	16.4	1,150i			565 i	22i			8i						
3.8	14.3	29	626	778	2,973i	1	11	14	25i						
3.9	12.6i	2,728i	5,604i	10,497 <sub>i</sub>	7,739i	10 <sub>i</sub>	14i	21i	21i	1.9					
3.8	13.5 <sub>i</sub>	4,948i	8,932i	11,329i	8,274	12 <sub>i</sub>	15 <sub>i</sub>		17 <sub>i</sub>						
4.2i	11.4i		4,343i	5,573i	7,204i		13 <sub>i</sub>	19i	25i						
4.5i	13.6i									2.1					
4.7	16.3i	1,336i	2,239i	2,780i	2,287i	11i	14i	17i	25i	2.5					
3.2i	14.1i		2,222i	2,911i		7i	14i	20i	21i						
4.4	22.8	1,049i	2,001i	1,968i	2,214	11i	14i	14i	25						
5.0	16.3	2,399	2,906	2,906	4,291	15	15	17	23						
4.7	11.9	6,492	8,223	9,255	8,839	18	21	23	26	0.5					
4.7	11.9	6,475	8,171	8,985	8,597	18	21	23	26						
1.5i	7.8i	7,179i	10,145	13,386i	14,741	15	18	17	28						
3.6	16.4	59i	201 <sub>i</sub>	266i		2 i	9 <sub>i</sub>	14i		2.1					
4.2i	15.6	1,083i	1,443i	2,481i	2,973i	10i	13i	19i	24i	2.1					
4.5	16.3		563i	994i	2,610i		13i	18i	38i	2.4					
4.1i	14.1i	1,240i	2,231i	2,839i	3,319i	9i	16i	19i	21i	1.7					
4.6	11.8	6,186i	8,380	9,408i	12,697	17	19	21	26	0.7					

	EDUCATION SYSTEMS																
		A	i	в	c		ſ	)			1	E			I	F	
	Comp	ulsory	Fr	ee	age		Duration	n (years)			School-age (000	population ,000)			Enrol (000	ment ,000)	
Country or territory	Years of pre-primary	Years of primary- secondary	Years of pre-primary	Years of primary- secondary	Official primary school starting a	Pre-primary	Primary	Lower secondary	Upper secondary	Pre-primary	Primary	Secondary	Tertiary	Pre-primary	Primary	Secondary	Tertiary
SDG indicator	4.2.4	4.1.7	4.2.4	4.1.7													
Reference year				<u> </u>			1		2019	1	1	1			1		
Sub-Saharan Africa																	
Angola	-	6		6	6	2	6	3	3	2.184	5.862	4,723	2,713	784	5.621	2.034	253
Benin	-	6	-	6	6	2	6	4	3	705	1,912	1,866	1,058	169	2,181	993	132
Botswana	-	-			6	3	7	3	2	164	363	234	203	33	345		51
Burkina Faso	-	10	-	10	6	3	6	4	3	1,946	3,498	3,356	1,868	105	3,234	1,342	133
Burundi	-	-			7	2	6	4	3	743	1,931	1,716	1,032	120	2,213	650	42
Cabo Verde	-	10	-	8	6	3	6	3	3	32	63	60	49	23	64	53	12
Cameroon	-	6	-	6	6	2	6	4	3	1,547	4,251	4,139	2,319	543	4,400	2,207	331
Central African Republic	-	10	•	13	6	3	6	4	3	428	824	861		12	814	138	
Chad	-	10	-	10	6	3	6	4	3	1,618	2,836	2,697	1,286	17	2,469	537	42
Comoros	-	6	•	6	6	3	6	4	3	72	130	129	76	15	124	74	
Congo	-	10	3	13	6	3	6	4	3	479	879	835	433	67	783		55
Côte d'Ivoire	-	10	•	10	6	3	6	4	3	2,301	4,083	4,155	2,477	188	4,004	2,227	247
D. R. Congo	-	6	-	6	6	3	6	2	4	8,779	15,166	12,105	7,037	4/4	16,807	4,619	465
Djibouti	-	10	1	12	5	2	5	4	3	41	94	125	90	4	69	68	
Equal. Guillea	-	0		0	6	2	0 E	4	2	107	104	147		40	33		
Enurea	-	0		0	6	2	7	2	4	207	205	128	121	47	330	108	10
Ethiopia		8		8	7	3	6	1	2	9 373	17 101	15 695	121		16 198	5.029	
Gabon	-	10		10	6	3	5	4	3	180	260	289		2,515	10,150	5,025	
Gambia	-	9	-	9	7	4	6	3	3	298	380	313		126	375		
Ghana	2	9	2	9	6	2	6	3	4	1.608	4,432	4,492	2.879	1.852	4.550	2.851	496
Guinea	-	6	-	6	7	3	6	4	3	1,148	2,091	2,117	1,128	_,	1,777		
Guinea-Bissau	-	9			6	3	6	3	3	175	314	262					
Kenya	-	12	-	12	6	3	6	2	4	4,207	8,318	7,687	5,258	3,200	8,290		563
Lesotho	-	7		7	6	3	7	3	2	144	305	216	212	54	368	136	22
Liberia	-	6	-	6	6	3	6	3	3	422	784	692		510	635	227	
Madagascar	-	5	3	12	6	3	5	4	3	2,291	3,524	4,403	2,687	902	4,649	1,495	144
Malawi	-	8	•	8	6	3	6	4	2	1,699	3,236	2,856		1,361	4,593	990	
Mali	-	9	4	12	7	3	6	3	3	1,953	3,474	2,771	1,607	131	2,477	1,046	83
Mauritania	-	9	3	13	6	3	6	4	3	391	694	677	406	36	677	260	23
Mauritius	-	11	•	13	5	2	6	3	4	26	84	122	96	25	86	122	39
Mozambique	-	-			6	3	7	3	2	2,939	6,097	3,759	2,926		6,941	1,216	214
Namibia	-	/	-	/	/	2	1	3	2	131	416	249	246	43	491		59
Niger	-	-			1	3	6	4	3	2,488	4,1/4	3,699	1,902	1/8	2,667	/8/	80
Nigeria	-	9	-	9	5	1	6	3	3	6,203	33,598	27,795			25,591	10,315	
Sao Tomo and Drincipo	-	6		6	6	3	6	2	3	1,042	36	22	1,137	202	2,512	752	72
Sopogal		11		11	6	3	6	1	3	1 5 1 6	2 7 7 2	2 567	1 486	252	2 172	1 150	105
Seuchallac	-	10		11	6	2	6	3	1	3	2,725 Q	10	6	3	2,172	7	195
Sierra Leone	-	9		9	6	3	6	3	4	663	1.245	1.264	5	127	1.770	492	-
Somalia	-	-			6	3	6	2	4	1,529	2,715	2,298					
South Africa	-	9	-	12	7	4	7	2	3	4,684	7,921	4,996	4,948	824	7,568	4,879	1,178
South Sudan	-	8	-	8	6	3	6	2	4	981	1,790	1,548		111	1,274	164	
Тодо	-	10	-	5	6	3	6	4	3	697	1,294	1,289	723	208	1,634	728	101
Uganda	-	7			6	3	7	4	2	4,516	9,409	6,649		609	8,841		
United Republic of Tanzania	-	7	2	7	7	2	7	4	2	3,566	11,274	7,873	4,981	1,429	10,605	2,338	154
Zambia	-	7	-	7	7	4	7	2	3	2,274	3,565	2,189		160	3,285		
Zimbabwe	-	7			6	2	7	2	4	901	2,949	2,063	1,355				136

	FINANCE													
G	н				1	l i				J				
E C				Governmen	nt education	n expenditu	re per pupil							
f GD	<u>ت</u> م		2017 P	PP US\$			% of GDP	per capita		tion				
edu (% o	nen (%)									duca	a			
nent ture	on sh vern ture	nary		Ð		nary		Ð		old e ture	poo			
endi	cati al go endi	Prin	nary	pudz	tiary	prin	nary	pudã	tiary	endi of GL	Intro			
Gov	Edu exp	Pre.	Prin	Sec	Tert	Pre-	Prin	Sec	Tert	exp (% c	Cou			
	1.a.2		1		4.5	5.4		I						
		1			2019									
										23	460			
2.9-1i	 17.7-1i	261-4	201-4	238-4	1.603-4	9-4	7-4	8-4	53-4	3.7	BEN			
										2.4	BWA			
5.4-1i	22.7-1i	165-3	279-4	321-3	6,083-3	8-3	14-4	16-3	298-3		BFA			
5.1-1	18.8-1									0.7	BDI			
5.2-2	16.4-2	84-2	1,135-2	1,339-2	2,610-2	1-2	17-2	20-2	38-2	0.7	CPV			
3.1-1i	16.9-1i									1.9	CMR			
											CAF			
2.5-2i	16.4-2i		106-1	218-1			7-1	14-1		0.9	TCD			
2.5-4	13.4-4									1.8	COM			
3.5-1i	15.6-1i									1.1	COG			
3.3-1	18.3-1	806-1	508-1	/02-1	5,491-2	16-1	10-1	14-1	111-2	2.3	CIV			
1.5-2i	14.0-2i	4				4				1.5				
3.0-1	14.0-1						20-3			2.5				
											FPI			
											SW/7			
4.7-4	27.1-4	59-4	130-4	278-4		4-4	8-4	17-4		0.3	ETH			
											GAB			
2.4-1i	11.2-1i	4	178-4			4	8-4			2.5	GMB			
4.0-1i	18.6-ıi									9.2	GHA			
2.3-1	14.9-1		157-3				7-3			1.0	GIN			
											GNB			
5.3-1i	19.0-1i	51-4	406-4		2,835-4	1-4	11-4		76-4	3.7	KEN			
7.0-1	14.1-1		618-1	896-1	1,334-1		22-1	32-1	48-1	0.4	LSO			
2.6-1i	8.1-1i	155-3	229-3	287-4		10-3	14-3	18-4		5.7	LBR			
2.8-1i	19.8-1i				1-3				3	3.0	MDG			
4.7-1i	15.8-1i	3	84-3	247-3		3	8-3	24-3			MWI			
3.8-2	16.5-2	41-2	281-2	583-2	3,610-4	2-2	12-2	25-2	165-4	1.9	MLI			
1.9	10.2		330	464	3,619		6	9	68	3.3	MRI			
4.7	16.7	600	3,511	0,/3/	1,977-2	3	10	30	10-2	4.5 0.9	MO7			
1-0.5	10.7-11									1.8	NAM			
3,5-1i	16.8-1i	403-2		139-2	2.254-2	34-2	10-2	12-2	190-2	3,6	NER			
										4.4	NGA			
3.1-1	10.8-1	41-1	83-1	423-1	1,904-1	2-1	4-1	21-1	96-1	5.9	RWA			
5.1-ıi	20.1-ıi									0.3	STP			
4.8-1	21.5-1	460-4	417-4	624-4	4,618-1	15-4	14-4	21-4	136-1	4.0	SEN			
4.4-3	11.7-3	3,277-3	3,842-3	4,140-3	19,135-3	12-3	14-3	15-3	71-3		SYC			
7.7	35.5	-	232	240-2		-	13	14-2		2.6	SLE			
											SOM			
6.5	19.6	818	2,352	2,897	7,027	6	19	23	56	1.6	ZAF			
1.5-3	0.9-1i					0.4-3	5-3	12-3			SSD			
5.4-1i	21.8-1i	76-4	243-3		1,187-2	5-4	16-3		76-2	2.4	TGO			
2.1-1i	11.5-1i									3.2	UGA			
3./-1i	20.5-1									1.4	1ZA			
4.0-1	10.9-1	b / -3	460-2			Z-3	13-2			3.8				
5.9-1i	19.U-1i									J.D	ZWE			

	EDUCATION SYSTEMS																	
		A I		D	C		·	,			School-age	- population	I	Enrolment				
	Comp	ulsory	Fr	ee	ary ng age		Duration	n (years)			(000)	,000)			(000,	,000)		
Country or torritory	fears of ore-primary	fears of orimary- secondary	fears of ore-primary	fears of orimary- secondary	Official prim school starti	<sup>D</sup> re-primary	Primary	-ower secondary	Jpper secondary	<sup>D</sup> re-primary	Drimary	Secondary	<b>Fertiary</b>	<sup>D</sup> re-primary	Primary	Secondary	[ertiary	
SDG indicator	4.2.4	4.1.7	4.2.4	4.1.7									-				-	
Reference year									2019									
Northern Africa and Western Asia																		
Algeria Armonia	•	10	1	12	6	1	5	4	3	956	4,360	4,526	3,009		4,517		1,583	
Azerbaijan	1	9	5	11	6	3	4	5	3	503i	659i	1,021i	692i	207	645	967	218	
Bahrain	-	9	-	12	6	3	6	3	3	68	122	108	85	34	116	102	47	
Cyprus	1	9	1	12	6	3	6	3	3	28i	58i	55i	58i	25	57	55	47	
Georgia		9		12	6	3	6	3	3	173	327	271	237	1,460	317	283	151	
Iraq	•	6	2	12	6	2	6	3	3	2,170	5,876	5,116	3,815					
Israel	3	12	3	12	6	3	6	3	3	508	938	823	609	548	936	837	374	
Jordan Kuwait	•	9	-	12	6	2	5	4	2	460	320	362	213	76	276	302	314	
Lebanon	•	10	3	9	6	3	6	3	3					216	517	406	243	
Libya	-	9	2	12	6	2	6	3	3	265	778	695	543					
Morocco	•	9	•	9	6	2	6	3	3	1,402	3,939	3,618	2,906	762	4,432	2,921	1,120	
Palestine	-	10	1	12	6	2	4	5	3	277	522	886	504	148	497	787	218	
Qatar	•	9	-	9	6	3	6	3	3	82	157	125	184	49	159	117	35	
Saudi Arabia	•	9	-	12	6	3	6	3	3	1,826	3,425	2,843	2,332	390	3,391	3,159	1,653	
Sudan Surian Arab Republic	-	9	2	11	6	2	6	3	3	2,384	6,683	5,970	4,323	1,100	5,118	2,216	653	
Tunisia		9	-	11	6	3	6	3	4	632	1,119	1,114	839	251	1,202	1,047	267	
Turkey		12	3	12	6	3	4	4	4	4,100	5,469	10,854	6,678	1,501	5,105	11,280	7,560	
United Arab Emirates	-	12	2	12	6	2	5	4	3	208	496	558	562	204	554	556	296	
Central and Couthern Asia	•	9	•	9	0	3	0	3	3	2,428	4,495	3,9/1	2,949	30	3,900	1,910		
Afghanistan		9	1	12	7	1	6	3	3	1.090	6.393	5.784	3.826		6.545	3.064	371	
Bangladesh	-	5	-	5	6	3	5	3	4	8,658	14,724	21,456	15,386	3,578	17,338	15,711	3,695	
Bhutan	-	-	-	11	6	2	7	4	2	25	89	82	74	8	94	76i	12	
India Iran Islamic Republic of	-	9	-	9	6	3	5	3	4	69,410 1 414	7 904	6 850	5 760	43,896	8 172	5 684	35,148	
Kazakhstan	-	9	4	11	6	3	4	5	2	1,202	1,508	1,951	1,047	891	1,513	2,024	740	
Kyrgyzstan	1	9	4	11	7	4	4	5	2	640	551	744	517	251	551	698	219	
Maldives	-	7	-	12	6 E	3	7	3	2	22	51	27	46	20	49	21	14	
Pakistan	-	12	-	10	5	2	5	3	4	10,723	25,106	31,988	20,965	8,538	23,558	13,858	1,878	
Sri Lanka	-	11	-	13	5	2	5	4	4	664	1,690	2,737	1,542	464	1,725	2,728	326	
Tajikistan	•	9	4	11	7	4	4	5	2	1,029	880	1,211	849	91	771		265	
lurkmenistan Uzbekistan	-	12	3	12	6 7	3	4	5	2	426	2,507	4,240	463	 874	2,506	4.073	362	
Eastern and South-eastern Asia										2,052	2,507	1,210	2,077	071	2,500	1,075	502	
Brunei Darussalam	-	9			6	3	6	2	5	21	40	45	35	13	39	43	11	
Cambodia	-	-	-	9	6	3	6	3	3	1,066	2,055	1,807	1,512	266	2,163		223	
China DDD Kawa	-	9	-	9	6	3	6	3	3	52,098	102,982	99,105	87,406	46,564	104,325	86,102	46,994	
Hong Kong, China	-	9	-	11	6	3	6	3	3	186	350	316	368	 176	376	341	298	
Indonesia	-	9	-	12	7	2	6	3	3	9,800	27,959	27,959	22,134	5,909i	29,426	24,894	8,037	
Japan	-	9	-	9	6	3	6	3	3					2,824	6,505	6,996	3,862	
Lao PDR Macao, China	-	9	-	9	6	3	5	4	3	46/	34	26	34	19	33	26	34	
Malaysia	-	6	-	11	6	2	6	3	3	1,032	2,966	3,048	2,829	1,004	3,085	2,602	1,218	
Mongolia	-	12	4	12	6	4	5	4	3	307	333	335	237	261	327	295	155	
Myanmar Philippipor	-	5	-	5	5	2	5	4	2	1,777	4,573	5,946	4,954	154	5,300	4,187	932	
Republic of Korea	-	9	3	9	6	3	6	3	3	1,316	2,729	2,844	3,217	1,260	2,719	2,894	3,084	
Singapore	-	6			6	3	6	2	2	113i	233i	158i	223i		234	167	198	
Thailand	-	9	3	12	6	3	6	3	3	2,245	4,780	5,145	4,888	1,787	4,899	6,019	2,411	
Timor-Leste Viet Nam	-	9	-	5	6	3	5	3	3	96 4 651	184	187 9 411		24 4 415	207	162		
	*	-		5	5	~			2	.,551	.,	-,	3,305	., .15	5,507		1,000	

	FINANCE													
G		н				1					J			
5 2	a				Governmer	nt education	expenditu							
catio	je je	. L. I		2017 P	PP US\$			% of GDP	per capita		ation			
edu	(% o	men (%)	_				_				son	e		
ment	iture on st	vern iture	nary	_	arg		nary	_	arg	_	old e iture DP)	f		
verni	oendi Irrati	al go bendi	- prir	mary	puo	tiary	- prir	mary	puo	tiary	useh of GI	ntr		
Ĝ	exp Edi	ex to	Pre	Pri	Sec	Ter	Pre	Pri	Sec	Ter	Ho (% exp	ē		
		1.a.2				4.5	5.4							
						2019								
											1.0	DZA		
2.7-	-2 1	0.4-2	2,467-2	1,259-2		1,225-2	20-2	10-2		10-2	1.9	ARM		
2.5	-1	7.4-1	1,543-1			3,241-1	11-1			22-1	1.0	AZE		
2.3	-2	572	4 3 2 2 2	5,520-4	8,05/-4	 8 403 p		29.2	37 2		2.0	CVD		
5.0	2 1		718	744	1.343		6	7	12		4.0	EGY		
3.5	-1 ]	3.0-1				1,517-1				10-1	1.8	GEO		
											0.3	IRQ		
6.1	-2 1	5.7-2	5,607-2	8,945-2	7,712-2	7,885-2	14-2	23-2	19-2	20-2	1.2	ISR		
3.1	L	10.0	/6	1,330	1,575	952	1	13	16	9	3.9	JOR		
											6.1	LBN		
												LBY		
											2.2	MAR		
			1,913	7,155	7,708	13,556-3	7	25	27	44-3	1.6	OMN		
5.3	-2 1	.6.1-2									3.6	PSE		
2.1	,	0.0									1.6	SAU		
											1.4	SDN		
												SYR		
6.6	-4 2	2.7-4			5,640-4	5,892-4			52-4	55-4	1.6	TUN		
	1										1.4			
5.1	L	10.2									3.5	YEM		
4.1-	-2j 1	5.7-2j	2	231-2	255-2		2	10-2	11-2			AFG		
1.3	3	9.3			381	885			9	20	3.7	BGD		
6.9	-1i 2	1.9-1i	4		2,921-4		4		31-4		1.2	BTN		
		 วา ว								 25	2.4	IND		
4.0	-1	21.5 14.1	1 765-2	1,579-2	2,490-2	2,022-1 2,283	1-3 7_2	0.3	21-2	20-1	0.6	KA7		
6.0-	-2 1	.6.3-2	1,150-2			253-2	22-2			5-2	1.8	KGZ		
3.7	-3 1	0.2-3	2,279	3,036	4,164		12	16	22		1.8	MDV		
5.1	-1 ]	.4.1-1	59-4	350-4	296-4i	684-4	2-4	12-4	10-4i	24-4	3.2	NPL		
2.9	-2 1	4.5-2		338-4	668-4	2,973-2		8-4	16-4	67-2	3.3	PAK		
5.2	-1 1 -4 1	6.4-4	795-4	901-1	00/-1	565-4	28-4	/-1	/-1	20-1	2.1	TJK		
		23.0										TKM		
5.3-	- <sub>2i</sub> 2	2.5-2i										UZB		
4.4	-3 1	.1.4-3	644-3	5,604-3	14,931-3	20,162-3	1-3	9-3	24-3	32-3		BRN		
2.2	-1	9.4-1									1.4	KHM		
												CHN DP/		
3.8	3	 20.3	 6,035	 9,162	 13,851	 20,281	 10	 15	23	33		HKG		
3.6	-4 2	0.5-4		1,380-4	1,093-4	2,158-4		13-4	11-4	21-4	2.5	IDN		
3.2	-2	8.4-2	3,862-2	8,702-2	9,664-2	8,274-2	10-2				1.1	JPN		
											3.1	LAO		
2.7-	-1 ] )	.4.4-1 17 4				27,151-1 7,204	 4			20-1	0.6	MAC		
4.1	-2 1	2.6-2	1,594-2	1,517-2	د <i>ا</i> درد 	368-2	14-2	13-2		3-2	3.5	MNG		
1.9	)	10.6		361-1	476-1	771-1		8-1	10-1	16-1	2.8	MMR		
											1.9	PHL		
4.6	-2		6,255-2	11,229-2	11,329-2	5,773-2	16-2	28-2	28-2	14-2	1.3	KOR		
				10,528-1	20,284-1	23,903-1		1/-1	ZI-1	25-1	0.8	SGP THA		
6.8-	-1j	 7.9-1i									2.8	TLS		
4.2	-1 ]	6.1-1									2.1	VNM		

			EDUCATION SYSTEMS									_				_		
	AB			В	C		I	D		E F						F		I
	Comp	ulsory	Fr	ee	age		Duration	n (years)			School-age (000	population ,000)			Enrol (000	ment ,000)		
	ears of re-primary	ears of rimary- econdary	ears of re-primary	ears of rimary- econdary	official primary chool starting	he-primary	rimary	ower econdary	lpper econdary	he-primary	rimary	econdary	ertiary	he-primary	rimary	econdary	ertiary	
Country or territory	> <u>~</u>	> <u>~</u> 0	> <u>~</u>	> <u>~</u> 0	0 s	<u> </u>	•	~ L	⊃ s	<u> </u>	<u> </u>	s	-		<u> </u>	S		I
Reference uear	4.2.4	4.1.7	4.2.4	4.1.7					2019									
Oceania																		
Australia	-	10	1	13	5	1	7	4	2	330	2,291	1,872	1,556	534	2,248	2,380	1,677	
Cook Islands	-	12	2	13	5	2	6	4	3	1	2	2	1	0.4	2	2		1
Fiji	-	-			6	3	6	4	3	54	104	110	74	20	122			
Kiribati	-	9	•	9	6	3	6	3	4	9	17	16			17			1
Marshall Islands Micronesia E S	-	8	-	8	6	2	6	4	2	3	9	9	6	2	14	6	2	
Nauru	2	12	2	12	6	3	6	4	2	1	2	1		0.3	2	1		l
New Zealand	-	10	2	13	5	2	6	4	3	122	378	439	323	115	388	490	268	ΙĒ
Niue	-	11	1	12	5	1	6	4	3	-	0.2	0.2	0.1	-	0.2	0.2		
Palau Paran New Grines	-	12	•	12	6	3	6	2	4	1	1	1						1
Papua New Guinea	-	- 8			5	4	6	2	4	856	1,434	30		358	34	26		l
Solomon Is	-	-			6	3	6	3	4	59	103	100		55	107			ł
Tokelau	-	11			5	2	6	4	3	-	0.1	0.2		0.1	0.2	0.2		IĒ
Tonga	2	13	-	8	6	2	6	5	2	5	15	16		2	17	16		
Tuvalu	-	9			6	3	6	4	3	1	2	2		1	2	1		l
vanuatu	-	-			6	2	0	4	5	16	46	46		14	46	21		
Latin America and the Caribbean		12		12	-	2	7	2	2	0.4	1	1	1	0.4	2	1		
Anguilla Antiqua and Barbuda	-	12	-	12	5	2	7	3	2	3	10	7	8	0.4	10	8		l
Argentina	2	12	3	12	6	3	6	3	3	2,250	4,412	4,278	3,483	1,731	4,776	4,582	3,190	1Ē
Aruba	2	11	2	11	6	2	6	2	3	2	7	7	8				1	1
Bahamas	-	12	2	12	5	2	6	3	3	10	34	39	33	4	30	27		I.
Barbados	-	11	2	11	5	2	6	3	2	6	19	18	19	5	20	19		ł
Bolivia P S	- 7	12	2	12	6	2	6	2	4	474	40	47	1 037	368	1 387	1 242	10	i
Brazil	2	12	2	12	6	2	5	4	3	5,464i	14,013	21,402i	16,415i	5,158	15,952	22,864	8,742	i ľ
British Virgin Islands	-	12	-	12	5	2	7	3	2	1	2	2	2	1	3	2	0.3	1
Cayman Islands	1	11	2	12	5	2	6	3	3	1	5	5	4	1	4	3		1
Chile	-	12	2	12	6	3	6	2	4	739	1,520	1,481	1,381	620	1,540	1,522	1,255	ł
Costa Rica	2	11	2	11	6	2	6	4	2	143	428	356	4,359	1,815	4,264	4,908	2,390	ł
Cuba	-	9	3	12	6	3	6	3	3	377	736	754	715	371	751	772	296	l
Curaçao	2	12			6	2	6	2	4	4	12	13						l
Dominica	-	12	-	12	5	2	7	3	2	2	6	5	7	1	6	5		l
Dominican Republic	3	12	3	12	6	3	6	2	4	578i	1,157i	1,149i	929i	332	1,300	941	557	
El Salvador	3	9	3	12	7	3	6	3	3	342	689	1,000 698	1,558 649	230	663	522	191	ł
Grenada	-	12	2	12	5	2	7	3	2	4	13	8	9	4	13	9	9	iľ
Guatemala	3	9	3	12	7	3	6	3	2	1,209	2,334	1,947	1,683	590	2,366	1,195	367	l
Guyana	-	6	-	6	6	3	6	3	2	45	86	73	82					I.
Haiti	-	6	-	6	6	3	6	3	4	754	1,472	1,628	1,059					ł
lamaica	-	6	-	6	6	3	6	3	2	142	1,197	233	263	109	233	203	75	i
Mexico	2	12	2	12	6	3	6	3	3	6,719	13,408	13,462	10,986	4,900	14,061	14,161	4,562	1
Montserrat	-	12	-	12	5	2	7	3	2	0.1	0.4	0.3	0.2	0.1	0.4	0.3		
Nicaragua	1	6	•	9	6	3	6	3	2	398	786	623	607					1
Panama	2	9	2	12	6	2	6	3	3	156	456	435	337	95	419	323	161	
Peru	3	12	3	12	6	3	6	3	2	1.589	3,175	2,602	2,557	1,680	3,715	2,826	1.896	ł
Saint Kitts and Nevis	-	12	-	12	5	2	7	3	2	1	5	4	4	1	5	4	4	ľ
Saint Lucia	-	10	•	10	5	2	7	3	2	4	15	12	15	3	16	11	2	l
Saint Vincent/Grenadines	-	12	2	12	5	2	7	3	2	3	12	9	9	4	13	10	2	
Sint Maarten	2	11	2	11	6	3	6	2	3	2	3	2	3				0.2	$\left\{ \right\}$
Trinidad and Tobago	-	7	-4	D-4	5	2	7	4	3	37	136	93	49 86	32	bδ	58		l
Turks and Caicos Islands	2	11			6	2	6	3	2	1	3	3	2	1	4	2	0.3	ľ
Uruguay	2	12	2	12	6	3	6	3	3	143	282	286	257	136	297	356	162	l
Venezuela, B. R.	3	11	3	11	6	3	6	3	2	1,598	3,268	2,615	2,661	1,190	3,285	2,391		

FINANCE												
	G	н				1					J	
	no (do				Governme	nt education	expenditu	_				
	lucati of GI	) nt e		2017 P	PPP US\$			% of GDP	per capita		catio	
	ent ed re (%	shan rnme re (%	Ð		_		Ð		_		d edu	ode
	nditu	ation gove nditu	orima	ary	ndarı	ary	orima	ary	ndarı	ary	aeholo nditu f GDP	itry o
	Gove	Educ total expe	Pre-p	Prim	Seco	Terti	Pre-p	Prim	Seco	Terti	Hous expe (% of	Cour
		1.a.2			<u> </u>	4.5	5.4					
						2019						
	5.1-2	13.6-2	5,402-2	9,521-2	7,734-2	8,981-2	11-2	19- <sub>2</sub>	15-2	18-2	2.0	AUS
	4.4-3										7 8	COK
1											2.0	KIR
		14.7										MHL
	12.4-4	22.3-4										FSM
1	6.3-2	 16.7-2	 8,175-2	 8,130-2	 8,757-2	 10,472-2	 20-2	 20-2	 21-2	 25-2	1.5	NZL
						2				2		NIU
												PLW
1	4.2-3	9.2-1i 13.0-3	 106-3	 549-3	 797-3		 2-3	 9-3	 13-3		1.4	WSM
												SLB
											2.2	TKL
											2.5	TUV
	4.5-2	12.2-2	3-4	409-4	636-4		0.1-4	13-4	20-4		2.2	VUT
												AIA
	4.9-1	 12.5-1	 3.296-1	 3.131-1	 4.066-1	 3.653-1	 14-1	 13-1	 17-1	 16-1	2.1	AIG
	5.5-3	21.4-3								88-3		ABW
												BHS
	3.2	22 1-1		3,532	3,157		 18-1	23	20	 28-1	1.5	BRB BLZ
											1.9	BOL
	6.3-2	16.5-2		3,025-2	3,243-2	5,480-2		20-2	22-2	37-2	2.5	BRA
	2.5-1						0.1-4	8-1	16-1	/1-1		CVM
	5.4-2	21.3-2	5,411-2	4,450-2	4,521-2	4,928-2	22-2	18-2	19-2	20-2	2.6	CHL
	4.5	14.1	1,117	2,906	2,906	2,360	7	19	19	15	2.1	COL
	7.0	24.6	2,140	4,270	4,845	7,178	11	21	24	36	2.5	CRI
												CUW
	5.6	8.3	1,168	2,431	3,041	-	10	21	26	•		DMA
			1,446	3,135	2,472		8	17	13	53.4	3.1	DOM
	3.6-1	14.3-1	859-1	1,347-1	1,254-1	978-1	10-1	15-1	14-1	11-1	4.3	SLV
	3.2-2	14.0-2	976-2	1,359-2	1,810-2	864-2	6-2	8-2	11-2	5-2		GRD
	3.2	23.8	1,049	1,140	491	1,475-4	12	13	6	18-4		GTM
	2.8-1	14.6-1									6.9	HTI
1	6.1-1	23.2-1				2,214-4				41-4	5.5	HND
	5.2	17.3	524	2,206	2,780	3,397-4	5	23	29	36-4	2.6	JAM
	4.5-2	1/.0-2 		2,000-2	2,002-2		32	14	30	۲ J-2 	1.2	MSR
1	4.4-2	22.4-2									3.6	NIC
											1.7	PAN
	3.8	17.8	1,543	1,440-3	1,945	 1,381-2	12	12-3	12-3	 11-2	2.6	PER
1	2.6-4	8.6-4	3,340-4	1,386-4	4,626-3	1,551-4	13-4	5-4	17-3	6-4		KNA
	3.3	14.1	1	1,995	3,185	1	1 2	13	21	1		LCA
	5./-1	19.U-1 	391-4	2,239-1	2,5/6-1		5-4	18-1 	2U-1			SXM
												SUR
											0.7	TTO
	2.9-1 5.0-1	11.9-1	3,255-1	2,965-1	3,549-1	 5,542-2	1/-1	0-1 13-1	18-1 16-1	88-4 25-2	2.2	URY
							18-4	18-4	15-4			VEN

	EDUCATION SYSTEMS																	
		A		в	с	D E									F			
	oulsory	Fi	ree	je		Duratio	n (years)			School-age (000	population (000)	1		Enrol (000	lment ,000)			
Country or territory	Years of pre-primary	Years of primary- secondary	Years of pre-primary	Years of primary- secondary	Official primary school starting ag	Pre-primary	Primary	Lower secondary	Upper secondary	Pre-primary	Primary	Secondary	Tertiary	Pre-primary	Primary	Secondary	Tertiary	
SDG indicator	4.2.4	4.1.7	4.2.4	4.1.7														
Reference year								1	2019	1	1	1	1	1	1			
Furner and Northern America																		
Albania	-	11	2	12	6	2	c c	4	2	104	162	257	722	70	167	256	120	
Andorra	-	11	-	12	6	3	6	4	2	104	102	237	235	79	107	230	139	
Austria	1	12	1	10	6	3	4	4	4	258	336	690	496	2	339	687	430	
Relarus	-	9	-	11	6	3	4	5	7	364	456	657	430	3/9	478	649	389	
Relaium	-	12	3	17	6	3	6	2	4	393	806	775	653	450	871	1 179	516	
Bermuda	-	13	1	13	5	1	6	3	4	1	4	5	4	0.4	4	4	1	
Bosnia and Herzegovina	-	9	-	9	6	3	5	4	4	90			221	23	157	240	89	
Bulgaria	2	9	4	12	7	4	4	4	4	258	291	537	330	221	263	490	236	
Canada	-	10	1	12	6	1	6	3	3	393	2,384	2,335	2,314		2,407	2,654	1,623	
Croatia	-	8	-	8	7	4	4	4	4	157	169	323	244	115	162	332	165	
Czechia	-	9	-	13	6	3	5	4	4	325	572	826	516	366	584	787	329	
Denmark	-	10	-	10	6	3	7	3	3	173	452	405	383	178	468	531	311	
Estonia	-	9	4	12	7	4	6	3	3	58	92	77	65	55	88	84	46	
Finland	1	9	1	12	7	4	6	3	3	237	373	356	326	208	369	546	295	
France	-	10	3	12	6	3	5	4	3	2,356i	4,166i	5,840i	3,872i	2,543	4,302	6,110	2,619	
Germany	-	13	-	13	6	3	4	6	3	2,306	2,982	7,082	4,447	2,359	2,987	6,949	3,128	
Greece	1	9	2	12	6	2	6	3	3	177	619	641	537	152	643	668	767	
Hungary	3	10	3	12	7	4	4	4	4	353	374	778	563	311	374	809	283	
Iceland	-	10			6	3	7	3	4	13	33	30	24	13	32	35	18	
Ireland	-	10			5	2	8	3	2	128i	563i	323i	299i	123	564	492	231	
Italy	-	12	-	8	6	3	5	3	5	1,527	2,768	4,594	2,949	1,491	2,871	4,630	1,896	
Latvia	2	9	6	12	7	4	6	3	3	85i	121i	109i	88i	77	122	117	82	
Liechtenstein	1	8			7	2	5	4	3	1	2	3	2i	1	2	3	1	
Lithuania	-	10	-	12	7	4	4	6	2	118i	114i	209i	160 i	103	117	233	118	
Luxembourg	2	10	3	13	6	3	6	3	4	20	38	47	38	18	38	49	7	
Malta	-	11	2	13	5	2	6	3	4	9	25	28	25	10	27	31	16	
Monaco	-	11	3	12	6	3	5	4	3					1	2	3	1	
Montenegro	-	9	-	9	6	3	5	4	4	22	38	63	42	16	39	5/	23	
Netherlands	1	12	2	12	6	3	6	3	3	528	1,093	1,195	1,021	4/5	1,1/5	1,632	890	
Norun	-	13	-	13	b	3	5	4	4	104	114	189	249	29	110	150	50	
Norway	-	10	-	10	5	5	1	5	5	1 402	440	383	348	1 2 0 1	447	449	289	
Poidilu	1	9	4	12	/ F	4	6	5	5	1,493	2,383	620	2,1/0	1,301	622	2,392	254	
Portugal Pepublic of Moldova	-	11	2	12	7	2	0	5	נ ר	152:	156	260	208:	122	1/0	707	81	
	-	10	4	12	6	3	5	4	4	554	1 030	1 650	1.056	521	9/8	1 458	539	
Russian Federation	-	11	1	11	7	/	1	ч с	7	7 637	7 059	10 / 90	6 877	6 3 8 7	6 978	10 242	5 775	
San Marino	-	10	-	13	6	3	5	3	5	1	7	10,490	7	1	7	7	1	
Serbia	-	8	-	17	7	4	4	4	4	263	265	560	368	168	264	529	250	
Slovakia	-	10	1	13	6	3	4	5	4	169	228	488	318	166	229	442	144	
Slovenia	-	9	-	13	6	3	6	3	4	65	130	132	99	61	129	147	77	
Spain	-	10	3	10	6	3	6	3	3	1,259	2,909	2,782	2,252	1,296	3,043	3,371	2.052	
Sweden	1	9	1	12	7	4	6	3	3	473	710	651	595	463	893	935	431	
Switzerland	2	9	2	9	7	2	6	3	4	174	497	591	500	174	515	609	307	
Ukraine	-	11	-	11	6	3	4	5	2					1,094	1,725	2,445	1,602	
United Kingdom	-	11	2	13	5	2	6	3	4	1,633	4,921	5,249	4,019	1,765	4,893	6,174	2,467	
United States	-	12	1	12	6	3	6	3	3	12,016i	24,648i	25,053i	21,452i	8,669	24,958	24,871	18,942	

	FINANCE													
	G	Н			C	ا 					J			
H	noi (qC				Governmen	it education	expenditu	re per pupil			E			
	ducat 6 of G	re of ent 6)		2017 P	PPP US\$			% of GDP	per capita		Icatio			
	ent e. ure (%	n shai ernm ure (%	ar,		r,		ard		₽		ld edu ure	code		
	ernm endit	l gov	prim	hary	ondar	iary	prim	lary	ondar	iary	seho enditu f GDI	ntry		
	Gove	Educ tota expe	Pre-	Prin	Seco	Terti	Pre-	Prin	Seco	Terti	Hou expe	Coul		
		1.a.2		I		4.5	.4		<u> </u>					
						2019								
	3.6-2	12.4-2		4,534-2	1,058-2	1,829-2		34-2	8-2	14-2	2.5	ALB		
1	3.2	10.9					14	12	14	16		AND		
	5.4-2	11.0-2	10,028-2	12,872-2	15,101-2	19,478-2	18-2	23-2	27-2	35-2	0.3	AUT		
	4.8-2	12.3-2	6,016-2		6,674-2	3,354-2	32-2		36-2	18-2	0.9	BLR		
IJ	6.4-2	12.4-2	8,866-2	10,978-2		16,480-2	17-2	21-2		32-2	0.3	BEL		
	1.5-2	7.8-2					17-4	8-4	12-4	24-2		BMU		
IJ						3,284-3				24-3	0.5	BIH		
	4.1-2	12./-2	6,459-2	4,529-2	4,/11-2	4,894-2	30-2	21-2	22-2	22-2	0.9	BGR		
				8,380-4		15,509-2		18-4		33-2 21	1.4	LIDV		
	3.9-2	0.0-2	5,152-2	11,220-2 E 960 -		0,009-2	19-2	41-2		21-2	0.4	CZE		
1	7.8 2	15.3 2	15 033 2	13 786 2	12 852 2	25 477 2	27 2	24 2	23-2	45 2	0.4	DNK		
i	5.0 2	17.8 2	15,055-2	6 6 5 3 3	6 308 3	11 838 5	21-2	19 2	19 2	36 2	0.4	FST		
1	6 4-2	11 9-2		9 870-2	11 074-2	15 029-2	21-2	20-2	23-2	31-2	0.1	FIN		
1	5.5-2	9.7-2	8.505-2	8,171-2	11.894-2	14.196-2	19-2	18-2	26-2	31-2	0.7	FRA		
1	4.9-2	11.0-2	9,439-2	9,469-2	12,593-2	17,933-2	18-2	18-2	23-2	33-2	0.6	DEU		
11			5,157-2	5,745-2	6,402-2	2,576-2	17-2	19-2	22-2	9-2	1.3	GRC		
1	4.7-2	9.9-2	6,688-2	5,154-2	6,332-2	8,269-2	22-2	17-2	21-2	28-2	0.6	HUN		
	7.7-2	17.8-2	13,544-2	13,560-2	11,748-2	15,063-2	24-2	24-2	21-2	26-2	0.2	ISL		
	3.5-2	13.4-2	3,196-4	8,223-2	8,716-2	16,415-2	4-4	10-2	11-2	21-2	0.5	IRL		
	4.0-2	8.3-2	7,981-2	8,728-2	9,381-4	10,384-2	19-2	21-2	23-4	24-2	0.6	ITA		
	4.4-2	12.0-2	6,117-2	6,270-2	6,867-2	4,314-2	21-2	22-2	24-2	15-2	0.5	LVA		
IJ												LIE		
	3.8-2	11.8-2	5,779-2	6,194-2	5,788-2	5,382-2	17-2	18-2	17-2	16-2	0.4	LTU		
IJ	3.6-2	8.5-2	18,904-2	19,023-2	22,417-2	45,567-2	16-2	16-2	19-2	39-2	0.1	LUX		
	4.8-2	13.4-2	/,439-2	/,183-2	12,270-2	17,038-2	18-2	2/-2	29-2	41-2	0.7	MLI		
	1.5-2	5.0					Z-3	5-3	5-3		1.6	MCO		
l	5.2 2	12.4 2	6 492 2	9.410 2	12 378 5		12 2	17 2		32.5	1.0	NID		
	3.2-2	12.4-2	0,732-2	5,410-2	12,370-2	11,031-2	12-2	11-2	22-2	52-2	0.3	MKD		
1	7.9-2	15.9-2	14.091-2	14.851-2	17.548-2	26.184-2	21-2	22-2	26-2	39-2	0.1	NOR		
1	4.6-2	11.1-2	5,721-2	7,055-2	6,321->	8,062-2	19-2	23-2	21-2	27-2	0.6	POL		
1	5.0-2	11.1-2	5,489-2	8,094-2	9,902-2	7,759-2	16-2	24-2	30-2	23-2	0.8	PRT		
1	6.1	19.5	3,852	3,423	3,223	4,177	29	26	24	31	0.7	MDA		
	3.1-2	10.1-2	3,284-2	2,223-2	4,555-2	7,301-2	12-2	8-2	16-2	26-2	0.2	ROU		
	4.7-2	13.5-2				5,510-2				20-2	0.5	RUS		
	3.6-1	14.5-1	15,595-1	14,810-1	14,180-1	6,566-1	26-1	24-1	23-1	11-1		SMR		
l	3.6-1	8.8-1	457-1	6,854-4	5,720-1	4,738-1	3-1	44-4	33-1	27-1	1.0	SRB		
	3.9-2	9.5-2	5,373-2	6,536-2	6,485-2	8,285-2	17-2	21-2	21-2	26-2	0.5	SVK		
	4.8-2	10.9-2	6,654-2	8,347-2	8,516-2	9,010-2	18-2	22-2	23-2	24-2	0.4	SVN		
	4.2-2	10.2-2	6,264-2	6,772-2	7,588-2	8,668-2	16-2	17-2	19-2	22-2	0.8	ESP		
	/.6-2	15.7-2	14,030-2	11,564-2	12,499-2	22,510-2	26-2	22-2	23-2	42-2	0.2	SWE		
	5.1-2	15.5-2	13,093-2	10,954-2	10,808-2	25,713-2	20-2	25-2	25-2	38-2 24	0.3	LIKP		
	5.4-2	13.9 -	4,412-2	3,004-2	9,070-2	4,100-2	30-2	23 2	21 -2	30 -	2.0	GPD		
	J.4-2	13.0-2	7 179-2	11 910.	13 386.5	13 974-2	12-2	20-2	21-2	59-2	1.1			
Л			1,113-2	11,710-2	13,300-2	13,517-2	++1	20-2	23-2		1.7	005		

## TABLE 2: SDG 4, Target 4.1 - Primary and secondary education

By 2030, ensure that all girls and boys complete free, equitable and quality primary and secondary education leading to relevant and effective learning outcomes

A         B         C         D         E         F         G         H         I         J           out-of-school-nite (000,000)         0ut-of-school-nite (000,000)         0ut-of-school-nite (									PAR	ICIPATIO	N / COMP	LETION								
Sub-         Sub- <th< th=""><th></th><th></th><th></th><th></th><th>۵</th><th></th><th></th><th></th><th>B</th><th></th><th> </th><th>c</th><th>D</th><th>F</th><th>F</th><th>6</th><th>н</th><th></th><th></th><th></th></th<>					۵				B			c	D	F	F	6	н			
Out-of-school-liden         Out-of-school-schol-school-school-school-school-school-school-schol-school-school-					-				b			C C	U		r				,	
Image         Image <th< th=""><th></th><th colspan="2">Out-of-school children (000,000)</th><th colspan="2">Out-of-school children (000,000)</th><th>of-school (%)</th><th>  rate</th><th>Co</th><th>mpletion (%)</th><th>rate</th><th>Over- grad</th><th>age for le (%)</th><th colspan="2">(%) (%) (%) (%) (%) (%) (%) (%) (%) (%)</th><th>secondary (%)</th><th>er secondary</th><th>secondary (%)</th><th></th></th<>		Out-of-school children (000,000)		Out-of-school children (000,000)		of-school (%)	rate	Co	mpletion (%)	rate	Over- grad	age for le (%)	(%) (%) (%) (%) (%) (%) (%) (%) (%) (%)		secondary (%)	er secondary	secondary (%)			
SDG indicator         U         L1.1		Primary	Lower secondary	Upper secondary	Primary	Lower secondary	Upper secondary	Primary	Lower secondary	Upper secondary	Primary	Lower secondary	GER primary	NERA prima	GIR last prin	Transition fr lower secone	NERT lower	GIR last lowe grade (%)	NERT upper	
Reference year         2019         2019         Velghted verage           Reference year         Sum         Sum         Sum         Velghted verage         Sum         Velghted verage	SDG indicator					4.1.4			4.1.2		4.	1.5			4.1.3			4.1.3		
Region         Sum         Sum         Velighted average         Velighted average         Velighted average           World         58,         61,         137,         8,         15,         35,         84         72         51         9,         10,         93,         90,         86         85,         76,         65,         56           Sub-Saharan Africa         31,         29,         38,         18,         36,         57,         64         41         28,         23,         31-11         85,         69,         64,         64,         44,         43,         77,         88,         78,         87,         75,         70,         88,         74,         70,         78,         89,         76,         70,         33,         90,         86,         74,         70,         -7         79,         89,         76,         74,         70,         -7         79,         89,         76,         74,         74,         70,         -7         79,         89,         76,         74,         74,         70,         -7         79,         89,         76,         74,         74,         70,         54,         89,         84,         60,         7	Reference year						2019									2019				
Vord         561         61         137         81         15         35         84         72         51         91         101         931         90         86         851         76         65         56           Sub-Saharan Africa         311         291         38         181         361         571         64         411         28         231         31-11         851         691         64         641         441         431         77           Morthern Africa         3         1         41         9         111         301         87         68         391         81         111         92         831         78         871         751         701         83           Western Asia         31         2,1         41         101         141         301         831         651         433         71         85         92         93         79         891         76         70         73         79         52         6         6          91         91         851         80         551         79         54         69         89         0.2         0.4         97         104         98	Region		Sum					Weighte	d average						Wei	ghted ave	rage			
Sub-Saharan Africa       31       29       38       18       36       571       64       41       28       23       31-11         Northern Africa       3       1.       41       9       11.       301       87       68       391       81       11.       92       881       78       871       75.       70.       8         Northern Africa       3       1.       44       9       11.       301       87       68       391       81       11.       92       881       78       871       75.       70.       8         Central adSouthern Asia       31       17.       65.       71       15.       45.       87       79       52       6       6        91       91       85.       89       55.       79       54.       89       92       83.       78.       87.       79.       54.       89       92       83.       74.       74.       74	World	58i	<b>61</b> i	137 <sub>i</sub>	<b>8</b> i	15 <sub>i</sub>	35i	84	72	51	<b>9</b> i	<b>10</b> i	93i	<b>90</b> i	86	<b>85</b> i	<b>76</b> i	<b>65</b> i	56	
International Western Asia         51         10         11 <th< td=""><td>Sub-Saharan Africa</td><td>31i</td><td>29i</td><td>38i</td><td>18i</td><td>36i</td><td>57i</td><td>64</td><td>41</td><td>28</td><td>23i</td><td>31-1</td><td>85i</td><td>69i</td><td>64</td><td>64i</td><td>44i</td><td>43i</td><td>77</td><td></td></th<>	Sub-Saharan Africa	31i	29i	38i	18i	36i	57i	64	41	28	23i	31-1	85i	69i	64	64i	44i	43i	77	
Northern Africa         3         1i         4i         9         11i         30i         89         70         35i         9i         15         92         93         79         89i         76         70i         33           Western Asia         3i         2i         4i         10i         14i         30i         83i         65i         43i         7i         8i         92         83i         78i         86i         74i         70i            Central Asia         -         0.1         0.4         1         2         16         100         98         89         0.2         0.4         97         104         98         98         98         98         974         88i         91         85i         79         54i         89           Southern Asia         6i         9i         17i         3i         9i         20i         95         83         58         4i         9i          9i         88         91i         89i         80i         50           Eastern Asia         3i         5i         7i         15i         95         86         57         3i         9i         96         104i<	Northern Africa and Western Asia	5i	4i	8i	10	13	30i	87	68	39i	8i	111	92	88i	78	87i	75i	70i	8	
Western Asia312141101141301831651431718192831781861741701-Central Asia1311716517115145187795266971049898998460Southern Asia131171651711514618678526791918518055179Eastern Asia6191713191201958358419191858517954189Eastern Asia4141913171151958659919185818517155Oceania0.110.110.413151261601131161111979319518017494Latin America and the Caribbean0.110.110.2180654271121969785858384998469836970149898879318007494Latin America0.110.110.2180654271121969785878369	Northern Africa	3	11	4i	9	11i	30i	89	70	35i	9i	15	92	93	79	89i	76	70i	33	
Central and Southern Asia       131       171       651       71       151       451       87       79       52       6       6        91       91       851       80       55.       79         Central Asia       ·       0.1       0.4       1       2       16       100       98       89       0.2       0.4       97       104       98       98       99       84       60         Southern Asia       131       171       655       71       151       466       86       78       52       6       7        91       91       851       79       541       89         Eastern Asia       4i       4i       4i       4i       12i       29i       95       83       58       4i       91       88       91i       80i       56       43i         Southreastern Asia       3i       5i       9i       4i       12i       29i       94       80       57       3i       9i       96       104i       85       88i       85i       71i       55         Oceania       0.1i       0.1i       0.4i       3i       7i       21i       94	Western Asia	3 i	2i	4i	10 <sub>i</sub>	14i	30i	83i	65i	43i	7i	8i	92	83i	78i	86i	74i	70i	-	
Central Asia       -       0.1       0.4       1       2       16       100       98       89       0.2       0.4       97       104       98       98       99       84       60         Souther Asia       131       171       651       71       151       461       86       78       52       6       7        91       91       851       79       541       89         Eastern and South-eastern Asia       61       91       171       31       91       201       95       83       58       41       91        91       91       851       79       541       89         Eastern Asia       41       41       91       31       71       151       95       86       59          961       90       931       921       851       433         South-eastern Asia       0.1       0.14       31       51       94       80       57       31       91       96       1041       85       881       851       711       55       6       601        131       161       111       97       931       80	Central and Southern Asia	13 <sub>i</sub>	17i	65i	7i	15i	45i	87	79	52	6	6		91	91	85i	80	55i	79	
Souther Asia       131       171       651       71       151       461       86       78       52       6       7        91       91       851       79       541       89         Eastern and South-eastern Asia       6i       9i       17i       3i       9i       20i       95       83       58       4i       9i        91       81       81       80i       50       50         Eastern Asia       4i       4i       9i       3i       7i       15i       95       86       59         96i       90       93i       92i       85i       433       50         Oceania       0.1i       0.1i       0.4i       3i       5i       26i       60i        13i       16i       11i       97       93i       87i       80i       79i       69       60       10i       10i       97i       93i       87i       80i       79i       69i       60       60       10i       10i       10i       80i       79i       60       70i       10i       10i       80i       79i       60       70i       10i       10i       10i       10i	Central Asia	-	0.1	0.4	1	2	16	100	98	89	0.2	0.4	97	104	98	98	99	84	60	
Eastern and South-eastern Asia         6i         9i         17i         3i         9i         20i         95         83         58         4i         9i          9ji         88         91i         89i         80i         50           Eastern Asia         4i         4i         9i         3i         7i         15i         95         86         59           96i         90         93i         92i         85i         43           South-eastern Asia         3i         5i         9i         4i         12i         29i         94         80         57         3i         9i         96         104i         85         88i         85i         71i         55           Oceania         0.1i         0.1i         0.4i         3i         7i         21i         94         81         63         8i         14i         98         98i         87         93i         80i         7ji         65           Caribbean         0.1i         0.1i         0.2i           80i         56         4         7i         12i           80i         99i         98i         87	Southern Asia	13 <sub>i</sub>	17i	65i	7 i	15i	46i	86	78	52	6	7		91	91	85i	79	54i	89	
Eastern Asia       4i       4i       9i       3i       7i       15i       95       86       59         9i       9i       9i       4i       12i       29i       94       80       57       3i       9i       96       104i       85       88i       85i       71i       55         Oceania       0.1i       0.1i       0.4i       3i       5i       26i       60i        13i       16i       11i       9f       93i       92i       85i       43         Cocania       0.1i       0.1i       0.4i       3i       5i       26i       60i        13i       16i       11i       9f       93i        95i       80i       74i       94         Latin America and the Caribbean       2i       2i       6i       3i       94       80       56       42       7i       12i         82i                 96       97       85       87       83       69       100i	Eastern and South-eastern Asia	6i	9i	17i	3і	9i	20i	95	83	58	4i	9i		99i	88	91i	89i	80i	50	
South-eastern Asia         3i         5i         9i         4i         12i         29i         94         80         57         3i         9i         96         104i         85         88i         85i         71i         55           Oceania         0.1i         0.1i         0.4i         3i         5i         26i         60i          13i         16i         11i         97         93i          95i         80i         74i         94           Latin America and the Caribbean         2i         6i         3i         7i         21i         94         81         63         8i         14i         98         98i         87         93i         80i         74i         94           Caribbean         0.1i         0.1i         0.2i           80         65         42         7i         12i              80i         65         42         7i         12i	Eastern Asia	4i	4i	9i	3і	7i	15i	95	86	59				96i	90	93i	92i	85i	43	
Oceania         0.1i         0.4i         3i         5i         26i         60i          13i         16i         11i         97         93i          95i         80i         74i         94i           Latin America and the Caribbean         2i         2i         6i         3i         7i         21i         94         81         63         8i         14i         98         98i         87         93i         80i         7ji         69           Caribbean         0.1i         0.1i         0.2i           80i         65         42         7i         12i          82i            52i           Central America         1         1         3         4         13i         31         94         80i         56i         44         7i         96i         97i         85i         87         83i         69         100i           South America         1i         1i         2i         6i         100i         99i         94i         2i         3i         99i         98i         99i         98i         95i         93i         30i           Europe <td>South-eastern Asia</td> <td>3 і</td> <td>5i</td> <td>9i</td> <td>4i</td> <td>12i</td> <td>29i</td> <td>94</td> <td>80</td> <td>57</td> <td>3і</td> <td>9i</td> <td>96</td> <td>104i</td> <td>85</td> <td>88i</td> <td>85i</td> <td>71i</td> <td>55</td> <td></td>	South-eastern Asia	3 і	5i	9i	4i	12i	29i	94	80	57	3і	9i	96	104i	85	88i	85i	71i	55	
Latin America and the Caribbean         2i         2i         6i         3i         7i         21i         94         81         63         8i         14i         98         98i         87         93i         80i         79i         69           Caribbean         0.1i         0.1i         0.2i           80         65         42         7i         12i          82i           82i           82i           82i            52i           Central America         1         1         3         4         13         31         94i         80i         56i         44         7i         96i         97i         85i         87         83i         69         100i           South America         1i         1i         2i         6i         100i         99i         94i         2i         3i         99i         99i         99i         98i         95i         94i         30i           Europe         1i         1i         2i         7i           1i         2i         99i         94i	Oceania	0.1i	0.1i	0.4i	3і	5i	2 G i	60i		13i	16i	11i	97	93i		95i	80i	74i	94	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	Latin America and the Caribbean	2 i	2i	6i	3і	7i	21i	94	81	63	8i	14i	98	98i	87	93i	80i	79i	69	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	Caribbean	0.1 <sub>i</sub>	0.1 <sub>i</sub>	0.2i				80	65	42	7i	12 <sub>i</sub>			82				52	
	Central America	1	1	3	4	13	31	94	80	56	4	7	96	97	85	87	83	69	100	
	South America	1,	1 <sub>i</sub>	31	2i	4i	16i	96	83	69	7	14	98i	99i	87	96i	88i	85i	83	
Europe         1i         1i         2i         2i         2i         7i           1i         2i         99         98i          98i         95i         93i         30           Northern America         0.1i         0.1i         1i         1i         1i         100         99         94         3i         4i         100i         99         94i         3i         4ii         100i         99         93i         30          100i         99         99i         93i         31         30          100i         99         99i         93i         30           100i         99         99i         93i         30           100i         99         99i         93i         30           100i         99         99i         93i         30            100i         99         99i         93i         30             100i         90         99i         95i         33	Europe and Northern America	1,	1 <sub>i</sub>	2 <sub>1</sub>	1 <sub>i</sub>	2 i	6 i	100 <sub>i</sub>	99i	94i	2 i	3 i	99i	99i	99i	98i	95i	94i	30	
Northern America         0.1i         0.1i         1i         1i         1i         1i         1i         1ii         99         94         3i         4i          100i         99         99i         95i         96i         33           Low income         20i         19i         25i         19i         37i         60i         57         34         20         26i         27-ii         84i         65i         60         63i         40-ii         40i         81i         81i         81i         81i         81i         81i         81i         81i         81i         95i         92i         88i         86i         79i         65i         63i         40-ii         81i         91i         81i         81i         91i	Europe	1	11	2i	2i	2i	7 i				1i	2i	99	98i		98i	95i	93i	30	
Low income         20i         19i         25i         19i         37i         60i         57         34         20         26i         27-ii         84i         65i         60         63i         40-ii         40i         81           Middle income         37i         42i         109i         7i         14i         35i         89         78         54         7i         10i         95i         92i         88         86i         79i         65i         63           Lower middle         29i         31i         88i         9i         18i         45i         85i         75         51         9i         10i         93i         90         88         82i         74         55i         70           Upper middle         8i         10i         22i         4i         8i         19i         95         84i         60i         4i         9i         97i         97i         89i         92i         87i         81i         57i	Northern America	0.1i	0.1i	1i	1i	1i	4i	100	99	94	Зі	4i		100i	99	99i	95i	96i	33	
Middle income         37 <sub>1</sub> 42 <sub>1</sub> 109 <sub>1</sub> 7 <sub>1</sub> 14 <sub>1</sub> 35 <sub>1</sub> 89         78         54         7 <sub>1</sub> 10 <sub>1</sub> 95 <sub>1</sub> 92 <sub>1</sub> 88         86 <sub>1</sub> 79 <sub>1</sub> 65 <sub>1</sub> 63           Lower middle         29 <sub>1</sub> 31 <sub>1</sub> 88 <sub>i</sub> 9 <sub>1</sub> 18 <sub>i</sub> 45 <sub>1</sub> 85         75         51         9 <sub>i</sub> 10         93 <sub>i</sub> 90         88         82 <sub>i</sub> 74         55 <sub>i</sub> 70           Upper middle         8 <sub>i</sub> 10 <sub>i</sub> 29 <sub>i</sub> 81         19 <sub>i</sub> 95         84         60         4 <sub>i</sub> 9 <sub>i</sub> 97 <sub>i</sub> 89         92 <sub>i</sub> 87 <sub>i</sub> 81 <sub>i</sub> 57	Low income	20i	19i	25i	19i	37i	60i	57	34	20	26i	27-1	84i	65i	60	63i	40-1	40i	81	
Lower middle       29i       31i       88i       9i       18i       45i       85       75       51       9i       10       93i       90       88       82i       74       55i       70         Upper middle       8i       10i       22i       4i       8i       19i       95       84       60       4i       9i       97i       89       92i       87i       81i       57i	Middle income	37	42	109	71	14	35	89	78	54	71	10	95	92	88	86	79	65	63	
Upper middle         8i         10i         22i         4i         8i         19i         95         84         60         4i         9i         97i         97i         89         92i         87i         81i         57	Lower middle	29	31	88	9i	18	45	85	75	51	9i	10	93	90	88	821	74	55	70	
	Upper middle	81	10	22	4	81	19	95	84	60	4	91	97	97i	89	92	87i	81	57	-
High income 1i 1i 3i 2i 2i 7i 99i 2i 3i 99i 98i 98i 95i 93i 32	High income	1	1	31	21	21	71	99i			21	31	99i	98i		98i	95i	93i	32	

A Out-of-school children, total number (million) and out-of-school rate as percentage of the corresponding age group.

B Completion rate by level [Source: UIS and GEM Report analysis of household surveys].

- C Percentage of pupils who are at least two years over-age for their current grade, by level.
- D Gross enrolment ratio (GER) in primary education.
- E Primary adjusted net enrolment rate (NERA) (%).
- F Gross intake ratio (GIR) to last grade of primary education (%).
- G Effective transition rate from primary to lower secondary general education (%).
- H Lower secondary total net enrolment rate (NERT) (%).
- I Gross intake ratio (GIR) to last grade of lower secondary education (%).
- J Upper secondary total net enrolment rate (NERT) (%).
- K Administration of nationally representative learning assessment in early grades (grade 2 or 3), or final grade of primary or lower secondary.
- L Percentage of students achieving at least a minimum proficiency level in reading and mathematics.

Source: UIS unless noted otherwise. Data refer to school year ending in 2019 unless noted otherwise.

Aggregates represent countries listed in the table with available data and may include estimates for countries with no recent data.

(-) Magnitude nil or negligible.

(...) Data not available or category not applicable.

(± n) Reference year differs (e.g. -2: reference year 2017 instead of 2019).

(i) Estimate and/or partial coverage.

LEARNING																					
		H	(			L															
	ionallu ro	Administ	ration of		ont		Achie	ving minin	num profic	iency:											
1141	lonally re	lesentati	ve rearring	y assessiii																	
Early	grades	End of p	orimary	secor	ndary	Early g	grades	secor	secondary												
Reading	Mathematics	Reading	Mathematics	Reading	Mathematics	Reading	Mathematics	Reading	Mathematics	Reading	Mathematics										
		4.1	L.6				4.1	.1													
					19																
		% of co	untries				Weighted	l average													
56	55	79	82	79	83																
77	75	88	85	90	92																
 8	-	50	75	58	83																
33	-	17	67	50	67																
-	-	61	78	61	89																
79	71	79	79	64	71																
60	40	40	40	40	40																
89	89	100	100	78	89																
50	39	83	83	83	83																
43	43	57	57	86	86																
55	36	100	100	82	82																
94	94	100	100	94	94																
69	71	74	69	60	62																
52	57	57	52	39	43																
100	100	100	100	100	100																
83	83	92	83	75	75																
30	33	83	87	93	93																
 30	35	84	88	95	95																
33	-	67	67	67	67																
81	74	71	74	77	81																
63	59	80	84	79	84																
70	62	85	87	79	83																
57	57	76	81	79	84																
32	35	81	81	79	84																
Nor         B         C         D         E         F         G         H         I         J           Or         See         See <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th>PART</th> <th>ICIPATION</th> <th>і / СОМРІ</th> <th>LETION</th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th>									PART	ICIPATION	і / СОМРІ	LETION									
--	--	----------	------------------------	--------------------	--------------------	---------------	--------------------	--------------------	-------------------	--------------------	--------------------	-------------------	--------------------	-------------	--------------	---------------------	--------------------------------	--------------	----------------------------	--------------	--
Image: Problem interment of the state of the s				J	4				В			c	D	E	F	G	н	1	J		
Desimal         <																					
Det         Det         Loo         Loo <thloo< th=""> <thloo< th=""> <thloo< th=""></thloo<></thloo<></thloo<>																ary to	ŋ (%)	ary	<b>"</b>		
Image         Image <th< th=""><th></th><th>Out-o</th><th>f-school c (000,000</th><th>hildren )</th><th>Out</th><th>of-school (%)</th><th>rate</th><th>Co</th><th>mpletion r (%)</th><th>ate</th><th>Over- grad</th><th>age for le (%)</th><th>(%)</th><th>(%) fi</th><th>ary (%)</th><th>om prima ary (%)</th><th>econdar</th><th>r second</th><th>econdar</th><th></th></th<>		Out-o	f-school c (000,000	hildren )	Out	of-school (%)	rate	Co	mpletion r (%)	ate	Over- grad	age for le (%)	(%)	(%) fi	ary (%)	om prima ary (%)	econdar	r second	econdar		
Soli adecinine         IMA 1.0         A 1.1.2         A 1.2.         A 1.2.         Colspan="2">Colspan="2"           Solitaria         Solitari         Solitaria <th colspa="&lt;/th"><th></th><th>Primary</th><th>Lower secondary</th><th>Upper secondary</th><th>Primary</th><th>Lower secondary</th><th>Upper secondary</th><th>Primary</th><th>Lower secondary</th><th>Upper secondary</th><th>Primary</th><th>Lower secondary</th><th>GER primary</th><th>NERA primari</th><th>GIR last prim</th><th>Transition fro lower second</th><th>NERT lower s</th><th>GIR last lowe grade (%)</th><th>NERT upper s</th><th></th></th>	<th></th> <th>Primary</th> <th>Lower secondary</th> <th>Upper secondary</th> <th>Primary</th> <th>Lower secondary</th> <th>Upper secondary</th> <th>Primary</th> <th>Lower secondary</th> <th>Upper secondary</th> <th>Primary</th> <th>Lower secondary</th> <th>GER primary</th> <th>NERA primari</th> <th>GIR last prim</th> <th>Transition fro lower second</th> <th>NERT lower s</th> <th>GIR last lowe grade (%)</th> <th>NERT upper s</th> <th></th>		Primary	Lower secondary	Upper secondary	Primary	Lower secondary	Upper secondary	Primary	Lower secondary	Upper secondary	Primary	Lower secondary	GER primary	NERA primari	GIR last prim	Transition fro lower second	NERT lower s	GIR last lowe grade (%)	NERT upper s	
Reference yant         U        <	SDG indicato	r				4.1.4			4.1.2		4.	1.5			4.1.3			4.1.3			
Solution Mile         Solution	Reference yea	r			1		2019	1			1			1	1	2019					
popsi         popsi <t< th=""><th>Sub-Saharan Africa</th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th></t<>	Sub-Saharan Africa																				
Bern         Dis         Dis <thdis< th=""> <thdis< td="" th<=""><td>Angola</td><td></td><td></td><td></td><td></td><td></td><td></td><td>57i</td><td>36i</td><td>17<sub>i</sub></td><td></td><td></td><td></td><td></td><td>63i</td><td></td><td></td><td></td><td>No</td><td></td></thdis<></thdis<>	Angola							57i	36i	17 <sub>i</sub>					63i				No		
Betwine iso         image	Benin	121		369-4i	6		56-4i	48-1	19-1	12i	12	30-4	94	64	39-1		46-3	44-4i	Yes		
Barbar         Temp         Barbar         Temp         Barbar         Temp         Set	Botswana							98i	90i	61i					93 i				Yes		
Banda         11         25         40         8         34         65         50         27         9         29         52         53         55         65         30         35         Ver           Converton         47         889         178         8         13         77         -         -         -         0         0         0         80         47         48         47         48         47         48         47         48         47         48         47         48         47         48         47         48         47         48         47         48         47         48         47         48         47         48         47         48         47         48         47         48         47         48         47         48         47         48	Burkina Faso	748	901	856	22	46	66	43i	17i	6i	23	59	78	65	40 i	54	41	34	Yes		
Cashe Neether         A+         A+         B+	Burundi	141	257	440	8	34	65	50i	27i	9i	29	52	92	59	53i	66	30	35	Yes		
Camboo         Camboo<	Cabo Verde	4-1	4-1	8-1	0-1	27.	Z/-1				10-1	29-1	94-1	8/-1 CE		67 -1	68-1	/3-1	Yes	_	
Conversion regions	Central African Pepublic	547	000-3	//0-4	0	5/-3	54-4	70i 27	49	101	15	20-4	92	41 2	04i 45	00-3	4/-3	40-4	No	-	
Concos         23.a         14.a         25.a         19.a         19.a         17.a         47.a         23.a         27.a         -a         88.a         77.a         60.a         81.a         -a         50.a         No           Congo         131.a         -         -         16.a         -         70.4         21.0         77.0         22.1         67.0         51.3         73.4         -         -         -         70.4         45.4         47.4         47.4         -         -         -         70.4         45.4         47.4         47.4         70.4         50.5         27.1         41.4         27.4         45.4         47.4         47.4         70.4         50.5         27.1         41.4         27.4         47.4         47.4         70.4         50.5         27.1         41.4         27.4         45.4         49.4         50.4 <td>Chad</td> <td>724</td> <td>978</td> <td>811</td> <td>26</td> <td>62</td> <td>78</td> <td>27</td> <td>13</td> <td>5</td> <td>30</td> <td>41</td> <td>74</td> <td>41-1</td> <td>49</td> <td>38</td> <td>15-1</td> <td>22</td> <td>Yes</td> <td></td>	Chad	724	978	811	26	62	78	27	13	5	30	41	74	41-1	49	38	15-1	22	Yes		
Coroge131-LLLTTSS	Comoros	23-1	14-1	25-1	18-1	19-1	50-1	77i	47i	23i	27-2		82-1	77-2	60i	81-1		50-1	No		
Cale Prior     199     109     97     5     44     58     60     31     13     14     57     15     16     13     13     13     14     14     14     14     10<	Congo	131-1			16-1			76i	51-4	23-4	14-1	31-1	84-1						Yes		
DR. Orgond         i.e.	Côte d'Ivoire	199	1,069	967	5	44	58	60i	31i	13i	11	31	95	79	51i	56	53	42	Yes		
Diploci         31.         35.         36.         35.         36.         37	D. R. Congo							67-1	54-1	31-1				70-4	82-1				Yes		
Equit         Bit         I         I         T<         T< </td <td>Djibouti</td> <td>31+1</td> <td>35-4</td> <td>36-4</td> <td>33+1</td> <td>48-4</td> <td>66-4</td> <td>86i</td> <td>62i</td> <td>32i</td> <td>7+1</td> <td>22+1</td> <td>67+1</td> <td>63+1</td> <td>72 i</td> <td>52-4</td> <td>50+1</td> <td>34-4</td> <td>No</td> <td></td>	Djibouti	31+1	35-4	36-4	33+1	48-4	66-4	86i	62i	32i	7+1	22+1	67+1	63+1	72 i	52-4	50+1	34-4	No		
Entra         242-         95-         138-         47-         36-         a         a         a         a         a         a         b         a         b         a         b         a         b         a         b         a         b         a         b         a         b         a         b         a         b         a         b         a     <	Equat. Guinea	84-4			55-4			73i	34i	9i	39-4	49-4	45-4	41-4	47 i		24-4		No		
Examin         34-9         2.4         8-4         8-5         27.1         41         27.4         84-7         74.4         94-7         74.4         94-7         74.4         74.7         74	Eritrea	242-1	95-1	138-1	47-1	36-1	49-1				34-1	48-1	53-1	60-1		64-1	51-1	51-1	Yes		
Chunga         Zabol /         Zabol //         Zabol /         Zabol //         Zabol /         Zabol //         Zabol // <thzabol <="" th=""> <thzabol <="" th=""> <th< td=""><td>Eswatini</td><td>34-1i</td><td>2-4i</td><td>8-4i</td><td>16-1i</td><td>3-4i</td><td>16-4i</td><td>6/i</td><td>48i</td><td>31i</td><td>41-1</td><td>6/-1</td><td>84-1i</td><td>94-1</td><td>/1i</td><td>9/-4i</td><td>6/-2</td><td>84-4i</td><td>NO</td><td></td></th<></thzabol></thzabol>	Eswatini	34-1i	2-4i	8-4i	16-1i	3-4i	16-4i	6/i	48i	31i	41-1	6/-1	84-1i	94-1	/1i	9/-4i	6/-2	84-4i	NO		
Code         Code <th< td=""><td>Cabon</td><td>2,307-4i</td><td>4,030-4i</td><td>3,330-4j</td><td>14-4i</td><td>4 / - 4i</td><td>/ 4-4j</td><td>221</td><td>271</td><td>14i</td><td>ZZ-4</td><td>20-4</td><td>00-4i</td><td>54-4</td><td>491</td><td>0 0 - 4i</td><td>29-4</td><td>20-4j</td><td>Voc</td><td></td></th<>	Cabon	2,307-4i	4,030-4i	3,330-4j	14-4i	4 / - 4i	/ 4-4j	221	271	14i	ZZ-4	20-4	00-4i	54-4	491	0 0 - 4i	29-4	20-4j	Voc		
Cannom         Dat         Dat <thdat< <="" td=""><td>Gambia</td><td>54</td><td></td><td></td><td></td><td></td><td></td><td>68:</td><td>52;</td><td>30;</td><td></td><td></td><td></td><td></td><td>76;</td><td></td><td>56</td><td></td><td>Voc</td><td></td></thdat<>	Gambia	54						68:	52;	30;					76;		56		Voc		
and Gaine42s -icei	Ghana	35	208	517	1	11	28	74i	521	36-1	31-1	41	99	94-1	70	89	78		Yes		
Guinea-Bissau	Guinea	425-3			22-3			56i	37i	20i	16-3	30-3	78-3	60-3	661				Yes		
kenga <t< td=""><td>Guinea-Bissau</td><td></td><td></td><td></td><td></td><td></td><td></td><td>24i</td><td>13i</td><td>9i</td><td></td><td></td><td></td><td></td><td>54i</td><td></td><td></td><td></td><td>No</td><td></td></t<>	Guinea-Bissau							24i	13i	9i					54i				No		
Lescho         7.a         2.a         2.a         1.7.a         3.a.         7.5.         3.6.         2.a.         3.0.a         5.0.a.         98.a.         8.6.a.         97.a.         8.1.a.         97.a.         8.1.a.         97.a.         87.a.         87.a	Kenya							75i	68i	42i				100-3	90 i		79-s		Yes		
Liberia159-212.81.81.171-279-261-261-26744-2YesMaday817641.322306151.2011.4455988070.70.81.8180.70.83.80.70.83.80.70.83.80.70.83.80.70.83.70.	Lesotho	7-2i	23-3	29-3	2-2i	17-3	34-3	75i	36 <sub>i</sub>	22i	30-2	50-2	98-2i	86-3	47 i	83-3	47-2	66-3	Yes		
Madagacar       Bit       Tef4       L132       Z       30       64       51,       29,       11,       44       55       98       63       58,       70       35       36       Yes         Malawi        342       771       89,       41.4       53       75,       51,       32,       15,       36       98       80       80,<	Liberia	159-2			21-2			28i	18 <sub>i</sub>	11 <sub>i</sub>	71-2	79-2	79- <sub>2</sub>	61-2	67i		44-2		Yes		
Malawi        M342       S71       <	Madagascar	81	764	1,132	2	30	64	51i	29i	11 <sub>i</sub>	44	55	98	63	58i	70	35	36	Yes		
Main       1,43-1       1/19-1       893-1       1/1-1       1/2-1       5/3-1       5/3-1       5/3-1       5/3-1       5/3-1       5/3-1       1/1-1       1/1-1       1/2-1       5/3-1 <t< td=""><td>Malawi</td><td></td><td>342</td><td>571</td><td></td><td>19</td><td>69</td><td>47i</td><td>22i</td><td>15<sub>i</sub></td><td>36</td><td></td><td>98</td><td>80</td><td>47i</td><td>81</td><td></td><td>31</td><td>No</td><td>_</td></t<>	Malawi		342	571		19	69	47i	22i	15 <sub>i</sub>	36		98	80	47i	81		31	No	_	
Madritalia         110         165         110         165         12         23	Mali	1,343-1	719-1	893-1	41-1	53-1	75-1	57i	32i	17i	11-1	17-2	59-1	50-2	56i	47-1	30-2	25-1	Yes		
Machados         1         1         1         1         1         1         0         1         1         0         1         1         1         0         1         1         0         1         1         0         1         0         1         0         1         1         0         1         1         1         0         1         1         1         1         0         1<	Mauritania	156	2:	163	23	28	21:	53i 100:	53i 01:	24i 57:	39	42	00	/3	991	72	46	39	NO		
Number Mambia6-10.1 <th< td=""><td>Mozambique</td><td>143</td><td>860-4</td><td>827-4</td><td>2</td><td>43-4</td><td>69-4</td><td>44;</td><td>15</td><td>6</td><td>38</td><td>45</td><td>99</td><td>55</td><td>34</td><td>57-4</td><td>74</td><td>31-4</td><td>Ves</td><td>-</td></th<>	Mozambique	143	860-4	827-4	2	43-4	69-4	44;	15	6	38	45	99	55	34	57-4	74	31-4	Ves	-	
Ninger1000 <td>Namibia</td> <td>6-1i</td> <td>000-4</td> <td>027-4</td> <td>2-1i</td> <td></td> <td>05-4</td> <td>821</td> <td>521</td> <td>37i</td> <td>26-1</td> <td>48-2</td> <td>98-1i</td> <td>94-1</td> <td>64i</td> <td>J7-4</td> <td>77-2</td> <td>J1-4</td> <td>No</td> <td>-</td>	Namibia	6-1i	000-4	027-4	2-1i		05-4	821	521	37i	26-1	48-2	98-1i	94-1	64i	J7-4	77-2	J1-4	No	-	
Nigeria $76_1$ $66_1$ $56_1$ $1$ $86_1$ $1$ <td>Niger</td> <td>1,647</td> <td>1,287-2</td> <td>1,087-2</td> <td>41</td> <td>65-2</td> <td>86-2</td> <td>41</td> <td>10</td> <td>21</td> <td>5</td> <td>24</td> <td>59</td> <td>62</td> <td>24i</td> <td>35-2</td> <td>18</td> <td>14-2</td> <td>Yes</td> <td>-</td>	Niger	1,647	1,287-2	1,087-2	41	65-2	86-2	41	10	21	5	24	59	62	24i	35-2	18	14-2	Yes	-	
Rwanda12132393645054-42615133379497964250YesSao Tome and Principe2-21-42-46-210-417-48760 $8_1$ 15-243-294-284-26990-474-283-44YesSenegal6811261261473011674.181-26162379911082YesSenegal0.2-121180.40.40.598999911082YesSiera Leone6-3268-1432-11-349-165-169472016-235-299836851-151-235-1YesSouth Africa0.40.588-1999911082YesSouth Sudan164420-1988747-281-121-1988747-281-221-188-1991.0.88-180-180-179-1YesSouth Sudan164410113-147-257-27347-214-114-121-19147-214-121-181-114-121-181-114-121-181-114-114-114	Nigeria							76i	66i	56i					86i				Yes		
SaoTome and Principe $2 - 2$ $1 - 4$ $2 - 4i$ $6 - 2$ $10 - 4$ $17 - 4$ $87$ $60$ $8_1$ $15 - 2$ $43 - 2$ $94 - 2i$ $84 - 2$ $69$ $90 - 4$ $74 - 2$ $83 - 4i$ $Yes$ Senegal $681i$ $$ $61i$ $$ $20i$ $$ $20i$ $$ $20i$ $$ $47i$ $30i$ $11i$ $6i$ $$ $61i$ $61i$ $62i$ $$ $37i$ $$ $Yes$ Seycheles $0.2$ $ 10i$ $2i$ $1i$ $12i$ $1ii$ $65i$ $60i$ $1ii$ $6ii$ $6iii$ $6iii$ $6iii$ $6iiii$ $6iiii$ $6iiii$ $6iiiii$ $6iiiii$ $6iiiiii$ $6iiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiii$	Rwanda	121	32	393	6	4	50	54-4	26i	15 <sub>i</sub>	33	37	94	97		96	42	50	Yes		
Senegal       681i        Constrained       1       Constrained	Sao Tome and Principe	2-2	1-4	2-4i	6-2	10-4	17-4i	87	60	8i	15-2	43-2	94-2i	84-2	69	90-4	74-2	83-4i	Yes		
Seychelles $0.2$ $ 1$ $2$ $1$ $18$ $$ $$ $0.4$ $0.5$ $98$ $99$ $$ $99$ $110$ $82$ $Yes$ Sieral Leone $6-3$ $268-1$ $432-1$ $1-3$ $49-1$ $65-1$ $69$ $47$ $20$ $16-2$ $35-2$ $99$ $83$ $68$ $51-1$ $51-2$ $35-1$ $Yes$ Somlai $$ $$ $$ $$ $$ $331$ $251$ $81$ $$ $$ $$ $751$ $$ $$ $$ $Yes$ South Africa $845-1$ $282-1$ $597-1$ $11-1$ $14-1$ $21-1$ $981$ $871$ $81-1$ $24-1$ $89-1$ $90-1$ $89$ $86-1$ $80-1$ $7-1$ $Yes$ South Africa $1684$ $2094$ $62-4i$ $56-4i$ $64-4i$ $10_1$ $13_1$ $61-1$ $7+1$ $91-4$ $88-1$ $90-1$ $88-1$ $90-1$ $88-1$ $80-1$ $7-2$ $7+1$ $7+1$ $10ga$ $16+1$ $152-2$ $265-2$ $1+1$ $21-2$ $57-2$ $7+1$ $47-2$ $41-1$ $16+1$ $27+1$ $91-4$ $88-4i$ $$ $7-9-2$ $52+1$ $43-2$ $7+2$ $10ga$ $1-1$ $1-14$ $1-12$ $1-12$ $7-12$ $7-12$ $47-2$ $48-2$ $87-2$ $1-18$ $87-2$ $1-18$ $7-12$ $7-12$ $7-12$ $8-12$ $1-18$ $1-18$ $7-12$ $1-18$ $7-12$ $1-18$ $7-12$ </td <td>Senegal</td> <td>681i</td> <td></td> <td></td> <td>26i</td> <td></td> <td></td> <td>47</td> <td>30</td> <td>11</td> <td>6</td> <td></td> <td>74i</td> <td>61</td> <td>62</td> <td></td> <td>37</td> <td></td> <td>Yes</td> <td></td>	Senegal	681i			26i			47	30	11	6		74i	61	62		37		Yes		
Sierral Leone       6-3       268-1       432-1       1-3       49-1       65-1       69       47       20       16-2       35-2       99       83       68       51-1       51-2       35-1       Yes         Somalia             331       251       81          75          Yes         South Africa       845-1       282-1       597-1       11-1       14-1       21-1       981       87.1       61.1       24-1       89.1       99-1       88.1       90-1       89.1       80-1       80-1       79.1       Wes       Wes         South Sudan       10.88       282-4       625-4i       62-4i       61-4i       10.1       13.1       61.1       24-1       99-1       88.1        80-1       80-1       80-1       79-1       <	Seychelles	0.2	•	1	2	1	18				0.4	0.5	98	99		99	110	82	Yes		
Sommala            3.1       2.51       8.1          751          Ves         SouthAfrica       845-1       282-1       597-1       11-1       14-1       21-1       981       871       471       8-1       24-1       89-1       90-1       891       86-1       80-1       79-1       Ves       Ves         SouthAfrica       1.641       202-4       62-41       66-41       101       131       61       77-4       91-4       38-41        90-1       881        79-2       52-11       44-3       Ves         Togo       16+1       15+2       262-41       62-41       56-4       64-41       101       131       47-2       141       16+1       27+1       99-1       88+1        79-2       52+1       43-4       Ves       Ves <td>Sierra Leone</td> <td>6-3</td> <td>268-1</td> <td>432-1</td> <td>1-3</td> <td>49-1</td> <td>65-1</td> <td>69</td> <td>47</td> <td>20</td> <td>16-2</td> <td>35-2</td> <td>99</td> <td>83</td> <td>68</td> <td>51-1</td> <td>51-2</td> <td>35-1</td> <td>Yes</td> <td></td>	Sierra Leone	6-3	268-1	432-1	1-3	49-1	65-1	69	47	20	16-2	35-2	99	83	68	51-1	51-2	35-1	Yes		
South Sudan         643-1         262-1         397-1         11-1         14-1         21-1         961         87/1         87/1         89-1         90-1         891         80-1         80-1         79-1	Somalia Courth Africa							33i	25i	8i					/5i				Yes		
Journal Laboration       Journal L	South Sudan	045-1	202-1	675	11-1 62 r	14-1 56 r	21-1 64 ···	98i 10-	0/i 12.	4/i	0-1 77 .	24-1 01 .	38	90-1	09i 20.	00-1	0U-1	79-1	res Voc		
Loss       Loss <thloss< th="">       Loss       Loss</thloss<>	Togo	1,000-4	157-2	265-2	1-1	21-2	57-2	73:	47-2	14:	16+1	27-1	99.1	88+1	231	79_2	57+1	43-2	Yes		
Junted Republic of Tanzania       1435       3,363-ai       1,783-ai       Table       72-ai       86-ai       78i       31i       12i       9-ai       18-ai	Uganda	1011	132-2	205-2	±71	L 1-2	57-2	44-3	39	20	34-2	48-2	551	53-2		, 5-2	26-2	13-2	Yes		
Zambia       496-2         15-2        74i       48i       30i       27-2        85-2        65i         Yes         Zimbiawe          89       54       14         61i         Yes	United Republic of Tanzania	1,435	3,363-a	1,783-3	13	72-3i	86-3i	78i	31	121	9-3	18-3	87	68	40i	28-3i	30-1	 14-3i	Yes		
Zimbabwe	Zambia	496-2			15-2			74i	48i	30i	27-2		85-2		65i				Yes		
	Zimbabwe							89	54	14					61				Yes		

						LEAR	NING						
l				ĸ						L			
l			Administ	ration of				Achio		-	rioncu		
l	nat	ionally re	presentati	ve learning	g assessm	ent		Acilie	ving mini (9	%)	Liency		
	Farlu	urades	End of	orimaru	End of	f lower	Farlu	nrades	End of	orimaru	End of	lower	
l	Lang	Jiddes		primary 	Jeco		Lang		LIGUI		3000		
	ading	athematics	ading	athematics	ading	athematics	ading	athematics	ading	athematics	ading	athematics	untry code
	Re	Ŵ	Re	Ŵ	Re	Ŵ	Re	Ŵ	Re	Ŵ	Re	Ŵ	S
			4.1	1.6					4.	1.1			
l						20	19						
1													
1	No	No	No	No	No	No							AGO
]	Yes	Yes	Yes	Yes	Yes	Yes	38	62	46	19			BEN
	Yes	Yes	Yes	Yes	Yes	Yes							BWA
	Yes	Yes	Yes	Yes	Yes	Yes	34	61	33	25			BFA
Į.	Yes	Yes	Yes	Yes	Yes	Yes	79	99	4	18			BDI
l	Yes	Yes	Yes	Yes	Yes	Yes							CPV
Į.	Yes	Yes	Yes	Yes	Yes	Yes	39	58	30	11			CMR
ł	No	No	Yes	Yes	Yes	Yes							CAF
ł	Yes	Yes	Yes	Yes	Yes	Yes	34	64	8	2			TCD
ł	NO	NO	Yes	Yes	Yes	Yes							COM
ł	Yes	Yes	Yes	Yes	Yes	Yes	63	86	34	8			COG
ł	Yes	Yes	Yes	Yes	Yes	Yes	33	58	22	3			CIV
ł	No	No	No	No	No	No	42	11	9	3			
t	No	No	Voc	Voc	Voc	Voc							GNO
ł	Voc	Voc	Voc	Voc	Voc	Voc							FPI
í	No	No	Ves	Ves	Vec	Ves							SW/7
1	Ves	Ves	Ves	Ves	Ves	Ves			55-2	73_2	 29_2	18-2	FTH
í	Yes	Yes	Yes	Yes	Yes	Yes	66	88	76	23	25 5	10 5	GAB
1	Yes	Yes	Yes	Yes	Yes	Yes	5-1	4-1					GMB
Í	Yes	Yes	Yes	Yes	Yes	Yes	6-2	8-2					GHA
1	Yes	Yes	Yes	Yes	Yes	Yes	23	60	22	7			GIN
ĺ	No	No	No	No	Yes	Yes							GNB
1	Yes	Yes	Yes	Yes	Yes	Yes	53-1	42-1	44-1	29-1			KEN
ĺ	Yes	Yes	Yes	Yes	Yes	Yes	13-1	1-1					LSO
1	Yes	Yes	No	No	Yes	Yes							LBR
1	Yes	Yes	Yes	Yes	Yes	Yes	13-1	4-1	6	6			MDG
1	No	No	Yes	Yes	Yes	Yes							MWI
	Yes	Yes	Yes	Yes	Yes	Yes							MLI
	No	No	Yes	No	No	No							MRT
	No	No	Yes	Yes	Yes	Yes							MUS
	Yes	Yes	Yes	Yes	Yes	Yes							MOZ
Į.	No	No	Yes	Yes	Yes	Yes							NAM
l	Yes	Yes	Yes	Yes	Yes	Yes	44	67	14	8			NER
Į.	Yes	Yes	Yes	Yes	Yes	Yes	17-2	11-2					NGA
l	Yes	Yes	Yes	Yes	Yes	Yes							RWA
Į.	Yes	Yes	Yes	Yes	Yes	Yes							STP
l	Yes	Yes	Yes	Yes	Yes	Yes	48	79	41	27	9-4	8-4	SEN
ļ	Yes	Yes	Yes	Yes	Yes	Yes							SYC
ł	Yes	Yes	Yes	Yes	Yes	Yes	6-2	6-2					SLE
ł	Yes	No	No	No	Yes	Yes							SOM
	Yes	Yes	Yes	Yes	No	Yes	22-3	16					ZAF
ł	Yes	Yes	No	No	Yes	Yes							SSD
ł	Yes	Yes	Yes	Yes	Yes	Yes	24	4/	19	10			IGU
ł	Tes Ver	Yes	res	Tes Ver	res	Yes	55-4	21-4	52-4	23-4			UGA
ł	Tes Ver	res Ver	res	Tes Ver	res Ver	res Vee	20-3	30-3				 2	TLA
ł	Tes Ver	Yes	res	Tes Ver	res	Yes		 F			D-4	Z-4	
	162	162	162	162	162	162	20	2					LWE

								PART	ICIPATIO	I / COMPL	ETION								
			1	A				В			с	D	E	F	G	н	1	J	
	Outro	freeboole	hildron	0*	of-school		60	malation		0.000					ary to	ry (%)	dary	ĥ	
	Out-o	(000,000	)   	Out	(%)	rate		(%)		grad	e (%)	ıry (%)	lary (%)	imary (%)	from prim ndary (%)	er secondai	wer secon	er seconda	
	Primary	Lower secondary	Upper secondary	Primary	Lower secondary	Upper secondary	Primary	Lower secondary	Upper secondary	Primary	Lower secondary	GER prima	NERA prim	GIR last pr	Transition lower seco	NERT lowe	GIR last lo grade (%)	NERT uppe	
SDG indicator					4.1.4	<u>I</u>		4.1.2		4.	1.5			4.1.3			4.1.3		
Reference year						2019									2019				
Northern Africa and Western As	ia																		
Algeria	16			0.4			97i	65i	41i	5	22	100	101	67i		83		No	
Armenia	15	19i	11,	9	10i	11 <sub>i</sub>	100i	98i	65-3	1	1	91	93	98i	90i	92	89i	No	
Azerbaijan Bahrain	3	1i 2	1i 6	8i 7	0.2i 4	13	99i	96i	90i	2	3-2	92i	100i 100	97i	100i 96	85-1i 93	100i 87	NO	
Cyprus	0.3-1i	0.1-1i	2-1i	1-1i	0.3-1i	7-1i				0.3-1	2-1	99-1i	102-1i		100-1i	96-1i	93-1i	No	
Egypt	91	128	1,209	1	2	23	94i	84i		2-1	3	99	105	89i	98	88	77	Yes	
Georgia	2	•	7	1	-	6	99i	97i	77-1	1	1	99	93	98i	100	114	94	No	
Iraq			7,1				/6i	4/i	31i					62i				No	-
Jordan	262	257	187	19	30	46	 99i	 94i	57-1	1	2	81	82	 95i	 70	64	54	No	-
Kuwait	7-3	11-4	24-4	3-3	6-4	18-4				3	4	83	85		94-4	92	82-4	No	
Lebanon										8	12							No	
Libya																		No	
Morocco	16	6	496	0.4	2	28	80i	50i	28i	14	31	97	97	62i	91	64 106	90	No	
Palestine	18	14	78	3	3	25	 99i	 90i	 66i	0.3	1	97	96	 91i	97	93	75	No	-
Qatar	3	3		2	5					1	4	98	95		95	95		No	
Saudi Arabia	59-3	15	48	2-3	1	4				6	8	98-3	96		99	105	96	No	_
Sudan Sudan Arab Depublic	2,131-1	687-1	1,462-1	33-1	34-1	52-1	76i	60i	33i	26-1	33-1	67-1	64-1	78i	66-1		48-1	No	
Synan Arab Republic							95	76;	50;	6-1	16-1	99_1	95-2	54i 80i		77-1		Yes	
Turkey	267-1	334-1	921-1	5-1	6-1	17-1				2-1	2-1	95-1	89-1		94-1	87-1	83-1	No	
United Arab Emirates	1	0.3	4	0.2	0.1	2				2-2	5-2	100	112		100	102	98	No	
Yemen	650-3	543-3	1,019-3	16-3	28-3	56-3	76i	61i	43i	9-3	11-3	84-3	72-3	81i	72-3	53-3	44-3	No	
Central and Southern Asia																			
Afghanistan			1,481-1i			56-1i	64i	48i	28i		11-1		86-1	76i		55-1	44-1i	Yes	
Bangladesh			4,81/-1i	 4.1	 12 v	38-1i 78 1i	83	60	29		28.1	96.1		/8		85.1	62-1i 72-1i	Yes	
India	J+1	7-1j	0-1			20-1	93i	86	59i	5	7		92	92i		83	/ 2 - 1	Yes	-
Iran, Islamic Republic of	17-2	157-2	854-2	0.2-2	5-2	26-2	96i	88i	69i	3-2	3-2	100-2	99- <sub>2</sub>	92i	95-2	90-2	74-2	No	
Kazakhstan	145+1	0.3+1	5-1	10+1	7+1	1-1	100 <sub>i</sub>	100i	95i	2+1	2+1	90+1	102+1	100 <sub>i</sub>	100+1	104+1	99-1	No	
Kyrgyzstan	3	11	55	0.5	2	28	100i	99i	84i	0.4	1	100	109	99i	98	98	72	Yes	
Maldives	103	1		2	3		100i 80	96i 72	25i 28	36	11-2	98	92	96i	91	111	 81	Yes	
Pakistan		4/	490	4			58i	52i	20		-45		73	90i		49		Yes	-
Sri Lanka	9-1	2-1	211-1	1-1	0.1-1	16-1				1-1	1-1	99-1	102-2		100-1	96-2	84-1	Yes	
Tajikistan	4-2			1-2			99i	92i	70 <sub>i</sub>	2	2	99-2	95-2	94 <sub>i</sub>		96-2		Yes	
Turkmenistan							99	99	94i					100				Yes	
UZDERISLATI	15	52	220	1	1	14	100i	991	911	-	0.1	99	100	991	99	95	00	INO	
Eastern and South-eastern Asia	0.2	0.4	6	1	2	19				1	2	00	100		07	107	87	Voc	
Cambodia	190	120-4i		9	13-4i		 74i	 46i	 22i	18	18	91	91	 63i	87-4i	58		Yes	
China							95i	86i	58i					90i				Yes	
DPR Korea																		Yes	
Hong Kong, China	8-1i	1i	1i	2-1i	1i	1i				2i	6i	97i	107		99i	106	99i	No	
lapan	1,222-1	2,299-1	۵,⊥۵/-۱	0-1	10-1	∠5-1	97i	8/i	22!	U.3-1	9-1	94-1	LUZ-1	9Ui	64-1	90-2	//-1	NO	
Lao PDR	65	163	186	8	28	44	85i	63i	 40i	9	22	92	92	 74i	72	65	56	Yes	
Macao, China	0.4	0.3	2	1	2	13				2	11	99	104		98	98	87	No	
Malaysia	10-2i	206-2	605-2	0.4-2i	13-2	37-2	100i	97i	62i	2	-	100-2i	99- <sub>2</sub>	97i	87-2	85	63-2	No	
Mongolia	2	15	27	1	8	21	99i	94i	80i	1	1	99	107	95i	92	95	79	Yes	
Myanmar Philippines	92-1	848-1 900	884-1 474 -	Z-1 2	21-1	43-1 21 -	82i	52i 74:	22i 70:		9-1	98-1	95-1 106	63i 82:	/9-1 80	65-1	5/-1 79.	N0 Voc	
Republic of Korea	25-1	38-1	61-1	1-1	3-1	4-1				0.2-1	0.3-1	99-1	95-1		97-1	 100-1	96-1	No	
Singapore	0.5-1i	0.4-1i	1j	0.2-1i	1-1j	0.1-1i				0.3-1	1-1	100-1i	100-1i		99-1i	100-1i	100-1i	No	
Thailand			603-4i			21-4i	99	87	66	2	2		94	89		81-1	79-4i	Yes	
Timor-Leste	9	10	23	5	10	24	80-3	66-3	52-3	22	32	95	105	82-3	90	91	76	Yes	
viet Nam							981	8/i	201	1-3		99	11U-1	891		98-1		NO	

					LEAR	NING						
		H	(						L			
		Administ	ration of				Achie	ving minir	num profi	ciency		
nat	ionally re	presentati	ve learnin	g assessm	ent			(9	%)	1		
Earlu	arades	End of a	orimaru	End of seco	flower ndaru	Earlu	orades	End of	orimaru	End of seco	f lower ndaru	
	,								- v			
6	natic	5	natic	5	natic	5	natic	6	natic	5	natic	) code
adin	athei	adin	athe	adin	ather	adin	athe	adin	ather	adin	ather	untri
å	ž	Å	ž	ž	ž	ž	ž	- <sup>3</sup>	ž	ž	ž	ů.
		4.1	1.6					4.:	1.1			
					20	)19						
NI-	N.	N.	Mar	Ma a	No					21	10	D74
No	No	No	Yes	Yes	Yes				 64	21-4	19-4 50-4	ARM
No	No	Yes	Yes	Yes	Yes			81-3	72			AZE
No	No	Yes	Yes	No	Yes			69-3	54		55	BHR
No	No	Yes	Yes	Yes	Yes				77	56-1	63	CYP
Yes	No	No	Yes	No	Yes				27		21-4	EGY
No	No	No	No	No	No			00-3	00	30-1	39-1	IRQ
No	No	Yes	Yes	Yes	Yes			91-3		69-1	66-1	ISR
No	No	No	No	Yes	Yes					59-1	41-1	JOR
No	No	Yes	Yes	No	Yes				21		21	KWT
N0	No	No	Yes	Yes	Yes				27	32-1	35-4	LBN
No	No	Yes	Yes	Yes	Yes			36-3	 18	27-1		MAR
No	No	Yes	Yes	No	Yes			59-3	33		27	OMN
No	No	No	No	No	Yes							PSE
No	No	Yes	Yes	Yes	Yes			66-3	40	49-1	37	QAT
No	No	Yes	Yes	Yes	Yes			63-3	23	48-1	15	SAU
No	No	No	No	No	Yes							SYR
Yes	No	No	Yes	Yes	Yes	47-1				28-4	25-4	TUN
No	No	Yes	Yes	Yes	Yes				70	74-1	56	TUR
No	No	Yes	Yes	Yes	Yes			68-3	53	57-1	50	ARE
No	No	No	Yes	No	No							YEM
Ves	Ves	Ves	Ves	No	No	22-2	24-2					AFG
Yes	Yes	Yes	Yes	Yes	Yes	47-2	34-2	44-2	32-2	54-4	57-4	BGD
Yes	Yes	Yes	Yes	Yes	Yes					56-4		BTN
Yes	Yes	Yes	Yes	Yes	Yes	47-2	53-2	46-2	44-2	38-2	40-2	IND
No	No	Yes	Yes	No	Yes			66-3	39		37	IRN
INU Yes	Yes	Yes	Yes	Yes	Yes	39-1	30-1	40-2	40-2	48-2	35-2	KGZ
/es	Yes	Yes	Yes	Yes	Yes							MDV
Yes	Yes	Yes	Yes	Yes	Yes			80-1	68-1	100-2	98-2	NPL
Yes	Yes	Yes	Yes	Yes	Yes		14-3	52-3	8			PAK
Yes Vec	Yes	Yes	Yes	Yes	Yes			55-4	/3-4	21-3	51-3	LKA TIV
Yes	Yes	No	No	No	No	71	53					TKM
No	No	No	No	No	No							UZB
Yes	Yes	Yes	Yes	Yes	Yes					48-1	52-1	BRN
Yes	Yes	Yes	Yes	Yes	Yes			11	19	8-4	10-4	KHM
Yes	Yes	No	No	Yes	Yes	82-3	85-4			80-3	79-4	CHN
No	No	Yes	Yes	Yes	Yes	54-2	00-2	 99-3	 96	87-1	 91-1	HKG
No	No	Yes	Yes	Yes	Yes				18-4	30-1	28-1	IDN
No	Yes	Yes	Yes	Yes	Yes							JPN
Yes	Yes	Yes	Yes	No	No			2	8			LAO
N0	No	Yes	No	Yes	Yes			98-3 59		89-1 5.4 -	95-1 50 -	MAC
Yes	No	Yes	Yes	Yes	Yes	44-1		00	04	54-1	59-1	MNG
No	No	Yes	Yes	No	No			11	12			MMR
Yes	No	Yes	Yes	Yes	Yes			10	17	19-1		PHL
No	No	No	Yes	Yes	Yes				95	85-1	85-1	KOR
N0 Voc	No	Yes	Yes	Yes	Yes			97-3	96	89-1	92	SGP THA
Yes	Yes	Yes	Yes	Yes	Yes					4U-1	++ / -1	TLS
No	No	Yes	Yes	Yes	Yes			82	92	86-4	81-4	VNM

								PART	ICIPATION	N / COMPL	ETION								1
			4	۹.				В			c	D	E	F	G	н	I	J	1
	Out-of	f-school c	hildren	Out	-of-school	rate	Co	mpletion r	ate	Over-a	age for			()	imary to 6)	lary (%)	ndary	dary	
		(000,000	)		(%)			(%)		grad	e (%)	ry (%)	ary (%)	imary (%	from pri ndary (9	r second	wer seco	er second	
	Primary	Lower secondary	Upper secondary	Primary	Lower secondary	Upper secondary	Primary	Lower secondary	Upper secondary	Primary	Lower secondary	GER prima	NERA prim	GIR last pr	Transition lower seco	NERT lowe	GIR last lov grade (%)	NERT uppe	
SDG indicator		I			4.1.4			4.1.2	<u> </u>	4.:	1.5			4.1.3			4.1.3		ĺ
Reference year						2019									2019				
Oceania																			
Australia	10-1	29-1	47-1	0.4-1	2-1	8-1				0.2-1	2-1	100-1			98-1		92-1	Yes	
Cook Islands	3	3	0.2	1-3	1-3	29				0.1	1	99	120		99-3	97	71	Yes	
Fiji	1-3			1-3			98i	93i	85i	2	2	99	109	95i		103-3		Yes	
Kiribati Marchall Islands	1-2	 2		4-2			94	78	17	2-2	10-2	96-2	101-3	83		95-3		Yes	h
Micronesia E S	1	Z	1	10	21	45				9	19	90	88		09	 79	57	Yes	
Nauru	3	0.1	-	3-3	9	6				2	1	99	117		91	87		Yes	1
New Zealand	1-1	3-1	4-1	0.3-1	1-1	2-1				0.2-1	0.3-1	100-1			99-1		98-1	No	1
Niue										-	-	94	112-3			104-4		Yes	
Palau																		Yes	Į.
Papua New Guinea	86-3	53-3	320-3	/-3	14-3	46-3	56i	50-1	8i	47-3 0	50-3 10	93-3	1/-3		86-3	62-3 103	54-3 90 -	Yes	
Solomon Is	7		Z-3	7		TO-3				75	75	93	86			70	JU-3	Yes	
Tokelau			-			41				1	-	97	119				59	Yes	11
Tonga	0.2-4	1-4	2-4	1-4	5-4	38-4	98	92	36	0.2-4	2-4	99-4		94	95-4		62-4	Yes	
Tuvalu	0.2	0.3	0.3	15	29	50				-	-	85	79-1		71	61-3	50	Yes	
Vanuatu	3-4	1-4	7-4	8-4	3-4	44-4	82i	45i	12 <sub>i</sub>			92-4		55i	97-4		56-4	Yes	
Latin America and the Caribbean	ı																		
Anguilla	-		-	1		4				1	1	99	96				96	Yes	
Antigua and Barbuda	0.1-1	0.1-1	0.4-1	1-1	1-1	13-1				2-1	13-1	99-1	96-1		99-1	99-1	8/-1	Yes	
Aruba	10-1	2-1	257-1	0.4-1	0.1-1	11-1	501	/ 21	051	J-1	12-1	100-1	55-1	101	100-1	51-1	09-1	No	
Bahamas										6-1								No	11
Barbados	0.2	1	0.4	1	5	5	99i	99i	95i	0.2	3	99	90	100 <sub>i</sub>	95		95	No	
Belize	0.2	3	6	0.5	10	36	96-3	61-3	49-3	8	15	100	103	63-3	90	70	64	No	
Bolivia, P. S.	75	62-1	200	5	13-1	22	98i	92i	67i	3	8	95	92	93i	93	86	78	Yes	
Brazii British Virgin Islands	7 2 - 11	01-4	0.2-3	2-1	Z-1i 5-4	20-3	90-1	00-1	0/-1	5-1	19-1	99-1	80-1	90-1	90-1i 88-3	98-3	80-3	No	
Cayman Islands	1-1	0.5-1	1-1	10-1	20-1	25-1				0.2-1	0.4-1	90-1	90-1		80-1	80-1	75-1	Yes	1
Chile	19-1	26-1	46-1	1-1	5-1	5-1	96-2	96-2	86-2	4-1	8-1	99-1	96-1	100-2	95-1	93-1	95-1	Yes	
Colombia	37	176	359-1	1	6	21-1	93-1	77-1	72-1	13	21	99	107	82-1	94	78	85	Yes	
Costa Rica	1	4	9	0.1	2	6	99-1	73-1	58-1	5	19	100	103	74-1	98	74	94	Yes	
Cuba	6	38	70	1	10	18	100	95	65	0.4	1	99	87	95	90	89	82	Yes	h
Dominica	0.2	-		 4		 18				 5_2						91-4	 87	INU Ves	
Dominican Republic	46i	33i	127i	4i		22i	91i	85i	55i	12	20	96i	93i	93i	94i	84i	78i	Yes	l
Ecuador	25-1	41-1	178-1	1-1	4-1	19-1	99-1	91-1	72-1	3-1	7-1	99-1	104-1	93-1	96-1	97-1	81-1	Yes	
El Salvador	96-1	60-1	126-1	14-1	17-1	34-1	90-1	75-1	59-1	13-1	20-1	86-1	87-1	84-1	83-1	77-1	66-1	Yes	
Grenada	0.1-1		0.1-2	1-1		3-2				2-1	9-1	99-1	123-1			107-1	97-2	No	
Guatemala	249	386-1	680-1	11	33-1	59-1	82i	53i	3/i 64:	15	25	89	/9	65i	6/-1	56	41-1	Yes	
Haiti							53-2	30i	13					911				No	
Honduras	151	234	233	13	38	56	85i	55i	461	9	16	87	79	65i	62	44	44	Yes	11
Jamaica		25	23		18	24	100 <sub>i</sub>	97i	92i	1	3			97i	82	84	76	Yes	
Mexico	94-1	527-1	1,754-1	1-1	8-1	26-1	98-1	89-1	59-1	1-1	2-1	99-1	102-1	90-1	92-1	91-1	74-1	Yes	
Montserrat	3	-	-	3-3	7	16				-	0.5	92	97		93	110	84	Yes	
Nicaragua	50 .						/8i	52i	39i 63:					68i				Yes	h
Paraquay		20-2	130-3	13-2	12-2	32-3	94-1	79-1	64-1	14-3	14-3	07-2	50-2	 84-1			68-3	Yes	ŀ
Peru	48-1	28-1	185-1	1-1	2-1	18-1	96i	87i	82i	4	7	98	99	90i	98-1	102	92i	Yes	
Saint Kitts and Nevis	0.1-3		0.1-3	1-3		4-3				1-3	2-3	99-3	98-3			111-3	96-3	Yes	[[
Saint Lucia	0.3	1	1	2	10	21	100i	98i	90i	1	2	98	100	98i	90	92	79	Yes	
Saint Vincent/Grenadines	0.1-2	0.1-1	1-1	0.5-2	2-1	15-1				1-1	14-1	97-ıi	106-1		98-1	92-1	85-1	Yes	
Sint Maarten							 Q.C.	 5 C.	 70.						 QE .			No	ł
Trinidad and Tobago	0	0-4	11-4	12	10-4	20-4	96:	95;	201	10	22	00	00	99;	03-4	4/	02-4	No	ŀ
Turks and Caicos Islands		0.4-1	0.4-1		20-1	32-1				4-1	3-1	99-1	 129-1		80-1	106-1	68-1	No	1Ē
Uruguay	0.4-1	0.4-2	18-2	0.1-1	0.3-2	12-2	97 <sub>i</sub>	66i	42i	3-1	14-1	100-1	103-1	68i	100-2		88-2	Yes	
Venezuela, B. R.	325-2	232-2	249-2	10-2	14-2	23-2				8-2	12-2	90-2	93-2		86-2	75-2	77-2	No	1

	LEARNING K L													
				(						L				
l			Administ	ration of				Achie	ving minin	num profi	ciency			
l	nat	ionally re	presentati	ve learnin	g assessm	ent			J (9	%)				
					End of	lower					End of	flower		
l	Early	grades	End of	primary	seco	ndary	Early	grades	End of	primary	seco	ndary		
l		tics		tics		tics		tics		tics		tics	ode	
l	ing	iema	ing	iema	ing	lema	ing	lema	ing	lema	ing	lema	tryc	
l	Read	Math	Read	Math	Read	Math	Read	Math	Read	Math	Read	Math	Coun	
ł			4	1.6					4	1 1				
1							10							
						20	19							
	Voc	Voc	Voc	Voc	Voc	Voc	94.5	70		68	80.1	78 1	ALIS	
l	Yes	Yes	Yes	Yes	Yes	Yes	54-3						COK	
	Yes	Yes	Yes	Yes	Yes	Yes							FJI	
	Yes	Yes	Yes	Yes	Yes	Yes	29-1	12-1					KIR	
ł	Yes	Yes	Yes	Yes	Yes	Yes							MHL	
í	Yes	Yes	Yes	Yes	Yes	Yes							NRU	
l	No	No	Yes	Yes	Yes	Yes			 90-3	56	81-1	78-1	NZL	
	Yes	Yes	Yes	Yes	No	No							NIU	
l	Yes	Yes	Yes	Yes	Yes	Yes							PLW	
ł	Yes	Yes	Yes	Yes	Yes	Yes							PNG	
í	Yes	Yes	Yes	Yes	Yes	Yes	71-4	76-4	58-4	90-4			SLB	
1	Yes	Yes	Yes	Yes	Yes	Yes							TKL	
	Yes	Yes	Yes	Yes	Yes	Yes							TON	
ł	Yes	Yes	Yes	Yes	Yes	Yes							TUV	
	Yes	Yes	Yes	Yes	Yes	Yes							VUI	
	Vac	Vac	Vac	Vac	Vac	Voc							A1A	
í	Yes	Yes	Yes	Yes	Yes	Yes							ATG	
1	Yes	Yes	Yes	Yes	Yes	Yes					48-1	31-1	ARG	
	No	No	No	No	No	No							ABW	
ł	No	No	No	No	No	No							BHS	
ł	NO	NO	Yes	Yes No	NO	NO							BRB BLZ	
ĺ	Yes	Yes	Yes	Yes	No	No	48-4	38-4	15-4	8-4			BOL	
]	Yes	Yes	Yes	Yes	Yes	Yes					50-1	32-1	BRA	
Į.	No	No	No	No	No	No							VGB	
ł	Yes	Yes	Yes	Yes	Yes	Yes							CYM	
1	Yes	Yes	Yes	Yes	Yes	Yes					50-1	35-1	COL	
1	Yes	Yes	Yes	Yes	Yes	Yes					58-1	40-1	CRI	
	Yes	Yes	Yes	Yes	Yes	Yes	97	97	100	100	100	100	CUB	
ł	No	No	No	No	No	No							CUW	
ł	Yes	Yes	Yes	Yes	N0 Ves	N0 Ves					 21-1	9_1	DMA	
1	Yes	Yes	Yes	Yes	Yes	Yes					49-4	29-4	ECU	
	Yes	Yes	Yes	Yes	Yes	Yes							SLV	
	No	No	No	No	No	No							GRD	
	Yes	Yes	Yes	Yes	Yes	Yes					30-4	11-4	GIM	
	No	No	No	No	No	No							HTI	
]	Yes	Yes	Yes	Yes	Yes	Yes					30-4	15-4	HND	
	Yes	Yes	Yes	Yes	Yes	Yes	85-2	67-2					JAM	
ł	Yes	Yes	Yes	Yes	Yes	Yes					55-1	44-1	MEX	
	Yes	Yes	Yes	Yes	Yes	Yes							NIC	
ĺ	Yes	Yes	Yes	Yes	Yes	Yes					36-1	 19-1	PAN	
1	Yes	Yes	Yes	Yes	Yes	Yes					32-4	8-4	PRY	
l	Yes	Yes	Yes	Yes	Yes	Yes							PER	
	Yes	Yes	Yes	Yes	Yes	Yes							KNA	
1	Yes	Yes	Yes	Yes	No	No							VCT	
	No	No	No	No	No	No							SXM	
	Yes	Yes	Yes	No	No	Yes	30-1	12-1					SUR	
l	No	Yes	Yes	Yes	Yes	Yes			80-3		58-4	48-4	TTO	
	Yes	Yes	Yes	Yes	Yes	Yes					58-1	49-1	URY	
1	No	No	No	No	No	No							VEN	
а.														

								PART	ICIPATION	V / COMPL	ETION								
				A				В			c	D	E	F	G	н	1	J	
	Out-o	f-school c (000,000	hildren )	Out	-of-school (%)	rate	Con	mpletion r (%)	ate	Over-a grad	age for e (%)	(%)	(%) fi	ary (%)	om primary to ary (%)	econdary (%)	r secondary	secondary	
	Primary	Lower secondary	Upper secondary	Primary	Lower secondary	Upper secondary	Primary	Lower secondary	Upper secondary	Primary	Lower secondary	GER primary	NERA primar	GIR last prim	Transition fro lower second	NERT lower s	GIR last lowe grade (%)	NERT upper s	
SDG indicator					4.1.4			4.1.2		4.	1.5			4.1.3			4.1.3		
Reference year						2019									2019				
Europe and Northern America																			
Albania	3	5	23	2	4	18	95- <sub>2</sub>	94-2	80-2	2	3	98	103	99-2	96	93	82	Yes	
Andorra										2	5							No	
Austria	0.1-1	1-1	34-1	7-1	0.3-1	10-1				5-1	8-1	100-1	100-1		100-1	97-1	90-1	No	
Belarus	6-1	5-1	2-1	1-1	1-1	1-1	100i	99i	89i	1-1	1-1	99-1	105-1	99i	99-1	98-1	99-1	Yes	
Belgium	4-1	2-1	8-1	1-1	1-1	1-1				1-1	5-1	99-1			99-1	93-1	99-1	Yes	
Bermuda										4	4		92-4			87-4		No	
Bulgaria			30			12 -	100i	99i		1	5.			99i			/9	NO	
Canada	40-1	34-1 1 4	73 1	0.2 %	014	12-1 6 1				1-1	J-1	0/-1 100 v	0/-1		0/-1		94 1	No	
Croatia	3-1	1-4	24-1	2-1	1-1	14-1				0.2-1	0.4-1	98-1	 95-1		99_1	 99_1	86-1	No	
Czechia	2-1	5-1	11-1	0.4-1	1-1	3-1				4-1	5-1	100-1	99-1		99-1	94-1	97-1	No	
Denmark	3-1	1-1	22-1	1-1	1-1	10-1				0.3-1	1-1	99-1	101-1		99-1	99-1	90-1	Yes	
Estonia	2-1	0.5-1	0.4-1	2-1	1-1	1-1				1-1	4-1	98-1	97-1		99-1	102-1	99-1	Yes	
Finland	5-1	0.1-1	7-1	1-1	0.1-1	4-1						99-1	100-1		100-1	99-1	96-1	No	
France	2-1i	27-1i	115-ıi	1i	1-1i	5-1i					1-1	100-1i			99-1i	100-1i	95-1i	Yes	
Germany	20-1	215-1	350-1	1-1	5-1	14-1						99-1	101-1		95-1		86-1	Yes	
Greece	9-1	13-1	16-1	1-1	4-1	5-1				1-1	4-1	99-1	98-1		96-1	94-1	95-1	No	
Hungary	17-1	12-1	47-1	4-1	3-1	12-1				1-1	3-1	96-1	99-1		97-1	93-1	88-1	No	
Iceland	7-1	7-1	2-1	0.1-1	0.2-1	13-1				7-1	7-1	100-1	99-1		100-1	98-1	87-1	Yes	
Ireland	<sup>1i</sup>	1-1i	2-1i	1i	1-1i	1-1i				1	0.2-1	100-1i	97-1i		99-1i	99-1i	99-1i	Yes	
Italy	89-1	35-1	155-1	3-1	2-1	5-1				0.4-1	2-1	97-1	98-1		98-1	99-1	95-1	No	
LdLVId	1-1i	L-1i	2-1i	1-1i	1-1i	4-1i				1-1 0.1.c	5-1 1 x	99-1i	98-1i		99-1i	99-1i	90-1i 84 si	N0 Voc	
Liethuania	11	0.1-1	0.2-1	0.1-1	4-1	10-1i 3. si				0.1-2	1-2	100-1	10/-1		100 4	102-1i QQ 4i	97 1	Yee	
Luxemboura	0.4-1	1-1	5-1	1-1	4-1	19-1				2-1	8-1	99-1	80-1		96-1	115-1	81-1	No	
Malta	0.1-2	0.2	1	0.2-2	2	8				0.5	0.5	100	108		98	104	92	No	
Monaco										-+1	0.2+1							No	
Montenegro	-	2	3	-	8	11	97-1	95-1	86-1	1	1	100	97	98-1	92	90	89	No	
Netherlands	4-1	15-1	2-1	0.4-1	3-1	0.3-1						100-1			97-1		100-1	No	
North Macedonia	1-1			1-1			98	94	83	1-1	1-1	99-1	93-1	96		86-1		No	
Norway	7-1	1-1	15-1	1	1-1	8-1				1	1	100-1	99-1		99-1	98-1	92-1	Yes	
Poland	51-1	21-1	40-1	2-1	2-1	4-1						98-1	94-1		98-1	100-1	96-1	No	
Portugal	2-1	0.4-1	3-1	0.4-1	0.1-1	1-1				5-1	11-1	100-1	100-1		100-1	95-1	99-1	No	
Republic of Moldova	16i	29i	261	10 <sub>i</sub>	16;	35i	99i	97i	71 <sub>i</sub>	0.3	0.5	90i	88i	97i	84i	84i	65i	No	
komania	136-1	/4-1i	162-1	13-1	9-1i	20-1				Z-1	4-1	8/-1	85-1		91-1i	88-1	8U-1	NO	
KUSSIAN FEDERATION	11-1 0.1	10-1i	83-1i 1	U.Z-1	U.Z-1i 1	5-1j 54						100-1	10Z-1		100-1i	103-1	9/-1i	NO No	
Sorbia	U.1 E.	£.	25.	5 n.	1 2.	54 15.			 Q 7	1	1	90.	00.		99	102	40	NO	
Slovakia	10 1	12,	22.1	4 .	5 1	11 1	100	30	60	1	T	96 1	92 1	27	95 1	81 1	89.1	No	
Slovenia	0.3-1	1.5-1	1-1	0.7_1	2-1	7-1				1-1	1-1	100-1	97-1		98-1	95_1	98-1	No	
Spain	92-1	5-1	22-1	3-1	0.4-1	2-1				0.2-1	7-1	97-1	99-1		100-1	98-1	98-1	No	
Sweden	1-1	0.5-4	4-3	0,1-1	0.2-4	1-3				0.1-1	0.3-1	100-1	104-1		100-4	109-1	99-3	No	
Switzerland	0.4-1	2-1	63-1	0.1-1	1-1	18-1				0.1-1	1-1	100-1	96-1		99-1	95-1	82-1	No	
	the second se																		
Ukraine							100 <sub>i</sub>	99i	95i	1	1			100 <sub>i</sub>				No	
Ukraine United Kingdom	 32-1	 3-1	 128-1	 1-1	 0.1-1	 4-1	100 <sub>i</sub>	99i 	95i	-1	1	 99-1	 101-1	100 <sub>i</sub>	 100-1	 101-1	 96-1	No Yes	

					LEAR	NING						
			<i>.</i>									
							A		-			
nat	ionallu re	Administ presentati	ration of ve learnin	a assessm	ent		Achie	ving minir (9	num profi %)	ciency		
	<b>.</b>			Endo	lower					End of	lower	
Early	grades	End of	primary	secor	ndary	Early	grades	End of	primary	seco	ndary	
					5						5	
	atic		atic		atic		atic		atic		atic	code
ling	hem	ling	men	ling	hem	ling	hem	ling	mer	ling	hem	trg
Read	Matl	Read	Matl	Read	Matl	Read	Matl	Read	Matl	Read	Matl	Cour
												-
		4	1.0					4.	1.1			
					20	19						
Yes	Yes	Yes	Yes	Yes	Yes				62	48-1	58-1	ALB
No	No	No	No	No	No							AND
No	No	Yes	Yes	Yes	Yes			98-3	84	76-1	79-1	AUT
Yes	Yes	Yes	No	Yes	Yes	82	73			77-1	71-1	BLR
Yes	Yes	Yes	Yes	Yes	Yes			97-3	80	79-1	80-1	BEL
No	No	No	No	No	No							BMU
No	No	No	Yes	Yes	Yes				40	46-1		BIH
No	No	Yes	Yes	Yes	Yes			95-3	71	53-1	56-1	BGR
No	No	Yes	Yes	Yes	Yes			96-3	69	86-1	84-1	CAN
No	No	Yes	Yes	Yes	Yes				/0	/8-1	69-1	HRV
No	No	Yes	Yes	Yes	Yes			97-3	78	79-1	80-1	CZE
Yes	Yes	Yes	Yes	Yes	Yes	97-3	/5			84-1	85-1	DNK
Yes	Yes	Yes	Yes	Yes	Yes					89-1	90-1	EST
NO	NO	Yes	Yes	Yes	Yes			98-3	/8	86-1	85-1	FIN
Yes	Yes	Yes	Yes	Yes	Yes			94-3	5/	79-1	79-1	FRA
Yes	Yes	Yes	Yes	Yes	Yes			95-3	/5	/9-1	/9-1	DEU
NO	NO	Yes	Yes	Yes	Yes					75 -1	04-1 69	GRC
Voc	Voc	Vec	Vec	Vec	Voc			97-3	/4	7.5-1	70 .	
Voc	Voc	Vec	Vec	Vec	Voc					00.	73-1 94 -	IDL
No	No	Voc	Voc	Voc	Voc	30-3	04			77 .	67	ITA
No	No	Voc	Voc	Voc	Voc			90-3	85	78 1	83.1	
Voc	Voc	Voc	Voc	Voc	Vac			55-3	05	70-1	03-1	LIF
Ves	Ves	Ves	Ves	Ves	Ves			97_2		76-1	74-1	ITU
No	Ves	Ves	Ves	Ves	Ves			51 5	01	71-1	73-1	
No	No	Yes	Yes	Yes	Yes			73-3	69	64-1	62-4	MIT
No	No	No	No	Yes	Yes							MCO
No	No	No	Yes	Yes	Yes				43	56-1	54-1	MNE
No	No	Yes	Yes	Yes	Yes			99-3	84	76-1	84-1	NLD
No	Yes	Yes	Yes	Yes	Yes				52	45-1	39-1	MKD
Yes	Yes	Yes	Yes	Yes	Yes	99-3	82		65	81-1	81-1	NOR
No	No	Yes	Yes	Yes	Yes			98-3	73	85-1	85-1	POL
No	No	Yes	Yes	Yes	Yes			97-3	74	80-1	77-1	PRT
No	No	Yes	Yes	Yes	Yes					57-1	50-1	MDA
No	No	Yes	Yes	Yes	Yes					59-1	53-1	ROU
No	No	Yes	Yes	Yes	Yes			99-3	91	78-1	78-1	RUS
No	No	No	No	No	No							SMR
No	No	No	Yes	Yes	Yes				68	62-1	60-1	SRB
No	No	Yes	Yes	Yes	Yes			93-3	71	69-1	75-1	SVK
No	No	Yes	Yes	Yes	Yes			96-3	75-4	82-1	84-1	SVN
No	No	Yes	Yes	Yes	Yes			97-3	65	84-4	75-1	ESP
No	No	Yes	Yes	Yes	Yes			98-3	74	82-1	81-1	SWE
No	No	No	No	Yes	Yes					76-1	83-1	CHE
No	No	Yes	Yes	Yes	Yes					74-1		UKR
Yes	Yes	Yes	Yes	Yes	Yes			97-3	83	83-1	81-1	GBR
Yes	No	Yes	Yes	Yes	Yes			96-3	77	81-1	73-1	USA

#### TABLE 3: SDG 4, Target 4.2 - Early childhood

By 2030, ensure that all girls and boys have access to quality early childhood development, care and pre-primary education so that they are ready for primary education

	A	В	C	D	E	F
	Children under 5 developmentally on track (%)	Under-5 stunting (%)	Stimulating home environment (%)	Children under 5 with 3+ children's books (%)	GER pre-primary (%)	NERA one year before primary entry (%)
SDG indicator	4.2.1		4.2.3		4.2.4	4.2.2
Reference year			20	19		
Region			Weighte	d average		
World		22+1			<b>62</b> i	<b>73</b> i
Sub-Saharan Africa		32+1			31i	43i
Northern Africa and Western Asia		18+1			3 3 i	52i
Northern Africa		21+1			41	56i
Western Asia		14+1			27i	48i
Central and Southern Asia		30+1			61	
Central Asia		10+1			38	58
Southern Asia		31+1			62	
Eastern and South-eastern Asia		13+1			83	88i
Eastern Asia		5+1			88	
South-eastern Asia		27+1			70i	93i
Oceania		41+1			76i	81i
Latin America and the Caribbean		11+1			78i	96i
Caribbean		12+1				
Central America		17+1			67	95
South America		9+1			85i	96i
Europe and Northern America		4+1			86i	93i
Europe		4+1			93i	93i
Northern America		3+1			72i	91i
Low income		35+1			20i	39i
Middle income		22+1			67	77i
Lower middle		29+1			60	71i
Upper middle		11+1			76i	85i
High income		3+1			83i	90i

A Percentage of children aged 36 to 59 months who are developmentally on track in health, learning and psychosocial well-being [UNICEF Early Childhood Development Index (ECDI)].

B Under-5 moderate or severe stunting rate (%) [Source: UNICEF, WHO, World Bank Joint Child Malnutrition Estimates (JME)].
 (Regional aggregates are weighted averages of statistical JME estimates for the reference year, not of the observed country values in the country table; Eastern Asia excludes Japan, Oceania excludes Australia and New Zealand, Northern America is based only on United States.)

C Percentage of children aged 36 to 59 months experiencing positive and stimulating home learning environments [Source: UNICEF database].

D Percentage of children under age 5 living in households with three or more children's books [Source: UNICEF database].

E Gross enrolment ratio (GER) in pre-primary education.

F Adjusted net enrolment rate (NERA) one year before the official primary school entry age.

Source: UIS unless noted otherwise. Data refer to school year ending in 2019 unless noted otherwise.

Aggregates represent countries listed in the table with available data and may include estimates for countries with no recent data.

(-) Magnitude nil or negligible.

(...) Data not available or category not applicable.

(± n) Reference year differs (e.g. -2: reference year 2017 instead of 2019).

(i) Estimate and/or partial coverage.

	A	В	с	D	E	F	
	_	()				e	
	er 5 ally or	ting (9	iome (%)	er 5 ren's	ary (%	ar befo J (%)	
	n unde ment	stun	ting h ment	child (%)	-prim	ne yez	
	ildrer velop ick (%	der-5	mula	ildrer th 3+ oks (9	R pre	RA or mary	е Р
Country or territory	ta é E	5	Sti	호호호	B	Pri	il coc
SDG indicator	4.2.1		4.2.3		4.2.4	4.2.2	ounti
Reference year			20	)19			0
Sub-Saharan Africa		20			10	65	460
Angola		38+1 21.			40-3	05-3i	AGO
Botswana	24-1	2 1+1 2 3+1	39-1	Z-1	24	00-1 21-4i	BLN BW/A
Burkina Faso		26+1			6	19	BFA
Burundi	40-2	58+1	58-2	0.1-2	17	49	BDI
Cabo Verde		10+1			73-1	81-1	CPV
Cameroon		27+1			36	44	CMR
Central African Republic		40+1	39	0.4	3-2		CAF
Chad	32-4	35+1	56	-	1	14	TCD
Comoros		23+1			22-1	30-1	COM
Congo	61-4	18+1	59-4	3-4	14-1	29-1	COG
Côte d'Ivoire		18+1	29-3	1-3	8	22	CIV
D. R. Congo		41+1	44-1	7-1	6-1		COD
Djibouti		34+1			10+1	13+1	ILU
Equat. Guinea		20+1			43-4	27.	GNQ
Enuea		22			2 3 - 1	Z / -1	CN/7
Ethiopia		35+1			29-4	37-4	FTH
Gabon		14+1			254	57 4	GAB
Gambia	67-1	16+1	16-1	1-1	43	61	GMB
Ghana		14+1	36-1	7-1	117	87	GHA
Guinea	49-3	29+1	31-3	0.4-3		42-3	GIN
Guinea-Bissau		28+1	44	-			GNB
Kenya		19+1			76-3		KEN
Lesotho	73-1	32+1	27-1	3-1	39-3	42-3	LSO
Liberia		28+1			125-2	79-2	LBR
Madagascar	67-1	40+1	25-1	1-1	40	59	MDG
Malawi		37+1			84-4		MWI
Mali	62-4	26+1	55-4	0.3-4	7-1	45-1	MLI
Mauritania	60-4	24+1	44-4	1-4	10-4		MRI
Mauritius		9+1 20.			90	91	MOZ
Namihia		18.1			34 1	69.1	NAM
Niner		47.1			7	74	NER
Nigeria	61-3	35+1	63-2	6-2		21	NGA
Rwanda	71-4	33+1	44-4	1-4	28	53	RWA
Sao Tome and Principe		12+1	43	6	50-3	52-4	STP
Senegal	67-2	17+1	20	1	17	16	SEN
Seychelles		7+1			95	92	SYC
Sierra Leone	51-2	27+1	28-2	2-2	19	42	SLE
Somalia		27+1					SOM
South Africa		23+1			18-1	72-1	ZAF
South Sudan		31+1			11-4	21-4i	SSD
Тодо	52-2	24+1	28-2	2	30+1	99+1	TGO
Uganda	65-3	28+1	53-3	Z-3	14-2		UGA
Vinced Republic of Tanzania		5Z+1 22			42	5/	
Zamula		3Z+1 72			0-3		ZIVIB 7\\/F
	/1	∠ J+1	53	3			ZWE
Northern Africa and Western Asi	а	0	63	0			D74
Argeria		9+1	63	8			DZA
Armenia		9+1			39 A1.	49	
Rabrain		5.1			52	741	BHD
Cuprus		511			85-1i	98-1i	CYP

	Α	В	С	D	E	F	
Country or territory	Children under 5 developmentally on track (%)	Under-5 stunting (%)	Stimulating home environment (%)	Children under 5 with 3 + children's books (%)	GER pre-primary (%)	NERA one year before primary entry (%)	J code
SDG indicator	4.2.1		4.2.3		4.2.4	4.2.2	l II
Reference uear			20	)19			5
Northern Africa and Western Asi	a continued	1					-
Fount		22.1			29	37	FGV
Georgia		6+1	77-1	56-1	25	57	GEO
Iran	79-1	12+1	46-1	3-1			IRQ
Israel					111-1	100-1	ISR
Jordan	71-1	7+1	92-1	16-1	29	45	JOR
Kuwait		6+1			60	81-1	KWT
Lebanon		10+1					LBN
Libya		44+1					LBY
Morocco		13+1	36-1		55	65	MAR
Oman		12+1			56	86	OMN
Palestine		8+1	76+1	12+1	54	65	PSE
Qatar		5+1			61	91	QAT
Saudi Arabia		4+1			21	50	SAU
Sudan		34+1			47-1	40-1	SDN
Syrian Arab Republic		30+1					SYR
Tunisia	82-1	9+1	74-1	24-1	45-3		TUN
Turkey	74-1		65-1	29-1	37-1	76-1	TUR
United Arab Emirates					99	100	ARE
Yemen		37+1			2-3	4-3	YEM
Central and Southern Asia							
Afghanistan		35+1					AFG
Bangladesh	74	30+1	64	6	41-1		BGD
Bhutan		22+1			33+1	41+1	BTN
India		31+1			63		IND
Iran, Islamic Republic of		6+1		36-2	54-3	51-3	IRN
Kazakhstan	86-4	7+1	86-4	51-4	74+1	78+1	KAZ
Kyrgyzstan	72-1	11+1	88-1	21-1	40	90	KGZ
Maldives	93-2	14+1	96-2	59-2	86	93	MDV
Nepal		30+1	/6	3	87	87	NPL
Pakistan		3/+1			81	93i	PAK
Sri Lanka Talihistan		16+1			69-1		LKA
Tajikistan		15+1			10-2	12-2	TI/AA
Iurkmenistan	91-3	0+1	91	52			
OZDERISLAII		10+1			11	40	UZD
Eastern and South-eastern Asia				1			
Brunei Darussalam		13+1			63	83	BRN
Cambodia		30+1			25	54	KHM
		5+1			89		CHN
UPR Korea	88-2	18+1	95-2	50-2			PRK
Hong Kong, China Indonesia					62	97	IDN
	00-1	52+1			02-1	90-1	IDN
	80 5	30					LAO
Macao China	0 5-2	30+1	74-2	7-2	91	89	MAC
Malausia		21+1	25-3	56-3	98	99-4	MYS
Mongolia	76-1	7+1	58-1	2.9-1	86	96	MNG
Muanmar		25+1	52-3	4-3	9-1	12-1	MMR
Philippines		29+1			105	86	PHL
Republic of Korea		2+1			94-1	99-1	KOR
Singapore		3+1					SGP
Thailand	91-3	12+1	93	34	79	99	THA
Timor-Leste	53-3	49+1	81-3	4-3	25	50	TLS
Viet Nam		22+1			96	100-1	VNM

	A	В	С	D	E	F	
Country or territory	Children under 5 developmentally on track (%)	Under-5 stunting (%)	Stimulating home environment (%)	Children under 5 with 3+ children's books (%)	GER pre-primary (%)	NERA one year before primary entry (%)	code
SDG indicator	4.2.1		4.2.3		4.2.4	4.2.2	ntry
Reference year			20	19		1	Cou
Oceania							
Australia		2+1			166-1	86-1	AUS
Cook Islands					84	98	СОК
Fiji		8+1			38	99	FJI
Kiribati		15+1	78	4			KIR
Marshall Islands	79-2	32+1	72-2	18-2	44	69	MHL
Micronesia, F. S.					29	68	FSM
Nduru New Zealand		10+1			94	95	NZI
Niue					129	81	NIU
Palau							PLW
Papua New Guinea		48+1			43-3	71-3	PNG
Samoa		7+1	87+1	9+1	40	35	WSM
Solomon Is		29+1			93	66	SLB
Tokelau					151	90	TKL
Tonga		3+1	90	24	46-4		TON
Vapuatu		20.1			83	93	VUT
		2 9+1			50-4	02-4	VUI
Latin America and the Caribbean					0.0	0.2	A1A
Anguina Antigua and Barbuda					90 70 -	95	
Ancigua ano barbooa		8+1			77-1	98-1	ARG
Aruba							ABW
Bahamas							BHS
Barbados		7+1			86	92	BRB
Belize	82-4	13+1	88-4	44-4	48	83	BLZ
Bolivia, P. S.		13+1			77	92	BOL
Brazil		6+1			97-1i	100-1i	BRA
British Virgin Islands					132-2	95-1	VGB
Chile		 7.1			87 1	90-1	CHI
Colombia		12+1			82	98	COL
Costa Rica		9+1	77-1	39-1	96	96	CRI
Cuba		7+1	89	42	97	98	CUB
Curaçao							CUW
Dominica					86	86	DMA
Dominican Republic		6+1			57i	93i	DOM
Ecuador		23+1			66-1	95-1	ECU
El Salvador Grenada		11+1			b/-1 100 ·	82-1	GPD
Guatemala		43+1			49	84	GTM
Guyana		9+1	91+1				GUY
Haiti	65-2	20+1	54-2	8-2			HTI
Honduras		20+1			40	77	HND
Jamaica		8+1			76	93	JAM
Mexico	80	12+1	71	29	73-1	99-1	MEX
Montserrat					76	90	MSR
Nicaragua		14+1					NIC
Panama		15+1 5			6Z-2	/ b-2 60 a	PAN
Peru	02-3	11+1	0-1-3	2 J-3	106	100	PER
Saint Kitts and Nevis					90-3	89-3	KNA
Saint Lucia		3+1			72	98	LCA
Saint Vincent/Grenadines					111-1	95-2	VCT
Sint Maarten							SXM
Suriname	77-1	8+1	66-1	26-1	94	89	SUR
Irinidad and Tobago		9+1			85		TTO
TURKS and Calcos Islands		 6 .			109-1	90-1	
Venezuela R P		0+1			90-1 70 -	100-1 86 -	VEN
Chickberg, D. N.		T T ± 1			,0-2	50-2	111

							1
	A	В	c	D	E	F	
Country or territory	Children under 5 developmentally on track (%)	Under-5 stunting (%)	Stimulating home environment (%)	Children under 5 with 3+ children's books (%)	GER pre-primary (%)	NERA one year before primary entry (%)	try code
SDG indicator	4.2.1		4.2.3		4.2.4	4.2.2	no
Reference year			20	019			0
Europe and Northern America							
Albania		10+1	78-1		76	97-1	ALB
Andorra							AND
Austria					103-1	100-1	AUT
Belarus	87	4+1	97	91	99-1	98-1	BLR
Belgium		2+1			114-1	98-1	BEL
Bermuda					69-4		BMU
Bosnia and Herzegovina		9+1			25	28	BIH
Bulgaria		6+1			79-1	81-1	BGR
Canada							CAN
Croatia					69-1	95-1	HRV
Czechia		2+1			110-1	90-1	CZE
Denmark					100-1	98-1	DNK
Estonia		1+1			93-1	88-1	EST
Finland					85-1	97-1	FIN
France					106-1i	100-1i	FRA
Germany		2+1			109-1	99-1	DEU
Greece		2+1			78-1	95-1	GRC
Hungary					86-1	87-1	HUN
Iceland					96-1	97-1	ISL
Ireland					93-1i	99-1i	IRL
Italy					93-1	93-1	ITA
Latvia					95-1i	98-1i	LVA
Liechtenstein					99-1i	99-1i	LIE
Lithuania					88-1i	97-1i	LTU
Luxembourg					92-1	99-1	LUX
Malta					111	97	MLT
Monaco							MCO
Montenegro	90-1	8+1	90-1	58-1	74	77	MNE
Netherlands		2+1			89-1	99-1	NLD
North Macedonia		4+1	88	55	42-1	49-1	MKD
Norway					96-1	97-1	NOR
Poland		2+1			88-1	95-1	POL
Portugal		3+1			91-1	94-1	PRT
Republic of Moldova		5+1			88i	95i	MDA
Romania		10+1			88-1	82-1	ROU
Russian Federation					86-1	90-1	RUS
San Marino					94	93	SMR
Serbia		5+1	95	78	64i	91i	SRB
Slovakia					97-1	84-1	SVK
Slovenia					92-1	94-1	SVN
Spain					96-1	95-1	ESP
Sweden					97-1	100-1	SWE
Switzerland					103-1	100-1	CHE
Ukraine		16+1					UKR
United Kingdom					107-1	100-1	GBR
United States		3+1			72-1i	90-1i	USA

# **TABLE 4:** SDG 4, Target 4.3 – Technical, vocational, tertiary and adult education

By 2030, ensure equal access for all women and men to affordable and quality technical, vocational and tertiary education, including university

#### SDG 4, Target 4.4 - Skills for work

By 2030, substantially increase the number of youth and adults who have relevant skills, including technical and vocational skills, for employment, decent jobs and entrepreneurship

	Α	В	c	D	E	F		G			1	н		
	idult (%)	led	condary		) ratio		ç	% of adults 15+ with ICT skills	•		% of ad having atta	ults 25+ ined at least		
	Participation in a education & trair	% of youth enroll in TVET	TVET share of se enrolment (%)	TVET share of post-secondary non-tertiary (%)	Gross graduatior from tertiary (%)	GER tertiary (%)	Copy & paste within document	Use formula in spreadsheet	Write computer program	Primary	Lower secondary	Upper secondary	Post-secondary	
SDG indicator	4.3.1	4.3.3				4.3.2		4.4.1			4.4	4.3		
Reference year			20	19						2019				
Region			Weighte	d average					W	eighted avera	ge			
World		<b>5</b> i	<b>10</b> i			<b>39</b> i	<b>33</b> i	<b>16</b> i	<b>4</b> i	<b>84</b> i	<b>70</b> i	<b>57</b> i	<b>26</b> i	
Sub-Saharan Africa		1i	6-1i			9 <sub>i</sub>								
Northern Africa and Western Asia		9i	13i			47i	43	19	7	87i	69i	57i	23i	
Northern Africa		7i	14i			35i	38	14	8		69i	67i	13i	
Western Asia		10i	12i			58i	51i	24	6	88	70	52	28	
Central and Southern Asia		2 i	3			26		7						
Central Asia		16i	20			27	19	21		100	99	95	64	
Southern Asia		1i	2			26		7						
Eastern and South-eastern Asia		7 i	16			47								
Eastern Asia		7	17			54								
South-eastern Asia		7 i	13i			34i	42	19	3	76	53	36i	14i	
Oceania		9i	22i			70i				100	93	76	48	
Latin America and the Caribbean		7 i	13i			53i	23i	19	5	82	61	47	19	
Caribbean			18				22	16	6					
Central America			27			38		26	7	81	60	35	16	
South America			8			61i	24	17	4	83	61	50	19	
Europe and Northern America		11i	16i			78i	46i	34i	5 i	98	91	78	37	
Europe		18 <sub>i</sub>	23i			73i	46	34	5	98	88	71	30	
Northern America		-1	0.4i			87i				99	96	89	47	
Low income		1i	6-1i			10-1i								
Middle income		5i	10			37i		12i	3 <sub>i</sub>					
Lower middle		2 <sub>1</sub>	4			24	28i	10						
Upper middle		8i	16i			53i		18 <sub>i</sub>	3 i					
High income		10 <sub>i</sub>	15 <sub>i</sub>			76i	48i	34i	6	97	89	76	38	

A Participation rate of adults (25 to 64) in formal or non-formal education and training in the last 12 months (%).

Estimates based on other reference periods, in particular 4 weeks, are included in the country when no data are available on the last 12 months, but not in regional aggregates.

B Percentage of youth (15 to 24) enrolled in technical and vocational education and training (TVET) programmes (ISCED levels 2 to 5) (%).

C Share of technical and vocational education and training (TVET) in total secondary enrolment (%).

D Share of technical and vocational education and training (TVET) in post-secondary non-tertiary enrolment (%).

E Gross graduation ratio from first degree programmes in tertiary education (ISCED levels 6 and 7).

F Gross enrolment ratio (GER) in tertiary education.

G Percentage of adults (15 and over) with specific information and communication technology (ICT) skills.

H Percentage of adults (25 and over) who have attained at least a given level of education.

I Percentage of population achieving at least a fixed level of proficiency in functional literacy and numeracy skills.

J Literacy rate, among youth (15 to 24) and adults (15 and above).

K Number of youth and adult illiterates, and percentage female.

Source: UIS unless noted otherwise. Data refer to school year ending in 2019 unless noted otherwise.

Aggregates represent countries listed in the table with available data and may include estimates for countries with no recent data.

(-) Magnitude nil or negligible.

(...) Data not available or category not applicable.

(± n) Reference year differs (e.g. -2: reference year 2017 instead of 2019).

(i) Estimate and/or partial coverage.

#### SDG 4, Target 4.6 - Literacy and numeracy

By 2030, ensure that all youth and a substantial proportion of adults, both men and women, achieve literacy and numeracy

	I		J		ŀ	(	
% ach profici	ieving ency in	Literacy	rate (%)		Illite	rates	
				% fe	male	Nun (000)	nber ,000)
Literacy	Numeracy	Youth	Adults	Youth	Adults	Youth	Adults
4.	6.1	4.	6.2				
			20	19			
		Weighte	d average			Su	ım
		92	86	55	63	99	773
		76	66	56	60	50	211
		89	80	57	63	9	71
		89	73	51	62	4	44
55i	50i	89	86	62	63	5	27
		91	75	56	64	34	365
		100	100	45	64	-	0.1
		90	74	56	64	34	365
		99	96	48	69	3	75
		100	97	46	73	1	45
		98	94	48	64	3	30
						Ti	- 11
		99	94	43	55	2	28
							3і
 49	40	99	94			0.4i	7i
 		99	95			1	18
 		100i	98i	46i		0.4i	2i
81	71						
		73i	61i	55i	60i	36i	152i
		93	86	55	64	62	587
		89	76	56	63	56	488
 		99	96	49	66	6	100
						0.1	5
							21

	A	В	с	D	E	F		G			I	н		
	dult ing (%)	p	ondary		ratio			% of adults 15- with ICT skills	•		% of ad having atta	ults 25+ ined at least		
Country or territory	Participation in ac education & traini	% of youth enrolle in TVET	TVET share of sec enrolment (%)	TVET share of post-secondary non-tertiary (%)	Gross graduation from tertiary (%)	GER tertiary (%)	Copy & paste within document	Use formula in spreadsheet	Write computer program	Primary	Lower secondary	U pper secondary	Post-secondary	
SDG indicator	4.3.1	4.3.3				4.3.2		4.4.1	1		4.	4.3		
Reference year			20	)19						2019				
Sub-Sabaran Africa														
Angola			14-3			9-3								
Benin		1-3	3-3			13-1								-
Botswana						25								
Burkina Faso		1	2		1-1	7								1
Burundi		3	9			4-1				18-2	9-2	6-2	2-2	
Cabo Verde		1-1	2-1	100-1	14-1	24-1	38-4	23-4	5-4	52-4	29-4	20-4	12-4	1
Cameroon		7-3	22-3	27-3		14-1								
Central African Republic			4-2											1
Chad		2	1			3-4								
Comoros		1	1											
Congo						13-2								
Côte d'Ivoire		2	5			10	12	3	1					-
D. R. Congo			19-4		5-3	7-3				64-3	51-3	27-3	9-3	
Djibouti			8+1				16-2	12-2	5-2					
Equat. Guinea														
Eritrea		0.5-1	1-1	75-2		3-3								
Eswatini	2-3	4	4-3		4									
Ethiopia		2-4i	7-4											
Gabon														
Gambia										39-4	31-4	23-4	8-4	
Ghana	2-2	1	3		10	17								
Guinea					6-2					18-1	13-1	7-1	6-1	
Guinea-Bissau														
Kenya				49-3		11-2								
Lesotho		1-4i	2-2		4-1	10-1								
Liberia														
Madagascar		1	3	100	4-1	5-1								
Malawi	1-2	-												
Mali	1-3	4-1	12-1	100-1		6-4				16-1	10-1	7-1	6-1	
Mauritania		0.2	1		4i	6								
Mauritius	2-2	2-1	11	36-1	29-2	41-2								
Mozambique		1-4	9-2		4-1	7-1				46-2	15-2	9-2	2-2	
Namibia	7-1			100-1	13-1	24-1								
Niger		1-2	7-2	100-2	4	4	2-1	1-1	0.3-1					
Nigeria														
Rwanda	3-2	4	13	100	6	6				36-1	13-1	10-1	4-1	_
Sao Tome and Principe		5-4i	6-2			13-4								
Senegal	6-4		5			13				22-2	18-2	11-2	10-2	_
Seychelles		20	11	100	8-3	19								
Sierra Leone				100-4										
Somalia														
South Africa	3-1	5-1	7-1	100-1	11-1	24-1				86-2	72-2	61-2	14-2	
South Sudan														
Тодо		3-2	6-2	100+1		14	3-2	1-2	0.5-2					
Uganda														
United Republic of Tanzania		0.1-2	0.4	9		3								
Zambia	3-2						52-1	23-1	7-1					
Zimbabwe						10-4				82-2	65-2	12-2	9-2	

		I	J	J			к			
	% ach profici	ieving ency in	Literacy	rate (%)		Illite	rates			
l							Nur	nber		
l					% fe	male	(000	,000)		
	Literacu	Numeracu	Youth	Adults	Youth	۵dults	Youth	Adults	code	
l		5 1		5 7	Touch	Addits	routin	Addito	itry	
l	4.0	0.1	4.0	20	10				Cour	
				20	15				-	
									460	
1			61-1	47-1i	61-1	61-1	890-1	 3.810-1i	BEN	
ĺ								5,010	BWA	
1			58-1i	41-ıi	53-1j	58-1i	1,652-1i	6,391-ıi	BFA	
1			88-2	68-2	62-2	63-2	248-2	1,851-2	BDI	
1			98-4	87-4	34-4	68-4	2-4	48-4	CPV	
			85-1i	77-1i	59-1i	62-1i	745-1i	3,317-1i	CMR	
			38-1i	37-1i	58-1i	60-1i	623-1i	1,627-1i	CAF	
			31-3	22-3	57-3	56-3	2,021-3	5,903-3	TCD	
			78-1i	59-1i	49-1i	57-1i	35-1i	207-1i	COM	
Į.			82-1i	80-1i	59-1i	65-1i	176-1i	602-1i	COG	
ł			58-1i	47-1i	57-1i	56-1i	2,154-1i	7,691-1i	CIV	
Į.			85-3	77-3	69-3	75-3	2,181-3	9,561-3	COD	
ł									DJI	
ļ									GNQ	
ł			93-1i	77-1i	54-1i	67-1i	42-1i	470-1i	ERI	
ł			95-1i	88-1i	36-1i	52-1i	11-1i	81-11	SWZ	
ł			/ 3-2i	5 Z-2i	51-2i	58-2i	6,2/3-2i	30,147-2i	EIH	
ł			90-1i	85-1i	4Z-1i	5 5 - 1i	3 / -1i	205-1i	GAB	
ł			07-4	70	55-4	60	140-4	2 8 0 4	GMB	
ł			54.	19-1	55.	65 .	433-1	3,034-1i 4174 -	GIN	
í			J4-1	40-1	0.0-1	03-1	1,131-1	4,1/4-1	GNR	
1			88-1i	87-1i	 49-1i	 60-1i	 1 291-1i	 5 714-1i	KEN	
í			00 1	02 1	13 1	00 1	1,231 1	5,7 1 1 1	150	
1			55-21	48-21	60-21	64-21	409-21	1 423-21	LBR	
Í			81-1	75-1	51-11	55-1i	1.017-1	3.926-1	MDG	
1			73-4i	62-4i	50-4i	61-4i	934-4i	3,471-4i	MWI	
1			50-1	35-1	57-1	59-1	1,831-1	6,422-1	MLI	
1			64-2i	53-2i	59-2i	61-2i	297-2i	1,190-2i	MRT	
1			99-1i	91-ıi	33-1i	62-1i	2-1i	90-1i	MUS	
]			71-2	61-2	61-2	67-2	1,659-2	6,178-2	MOZ	
			95-1i	92-1i	40-1i	53-1i	24-1i	131-ıi	NAM	
			43-1i	35-1i	56-1i	57-1i	2,440-1i	7,291-ıi	NER	
Į.			75-1i	62-1i	63-1i	62-1i	9,365-1i	41,764-ıi	NGA	
			86-1	73-1	43-1	59-1	327-1	1,968-1	RWA	
Į.			98-1i	93-1i	48-1i	73-1i	1-1i	9-1i	STP	
ł			69-2	52-2	60-2	66-2	923-2	4,236-2	SEN	
Į.			99-1i	96-1i	21-ıi	43-1i	0.1-ıi	3-1i	SYC	
l			67-1i	43-1i	56-1i	58-1i	518-1i	2,561-ıi	SLE	
ł									SOM	
	ð / -2	98-2	95-2	ŏ/-2	55-2	54-2	404-2	5,229-2	ZAF	
1			48-1i	35-1i	5U-1i	55-1i	1,15/-1i	4,181-1i	22D	
ł	NumerayNumerayNumerayNumerayNumerayNumerayNumerayNumerayNumerayNumerayII									
ł			86 .	7.2 .	40-1i	67 .	34U-1i 1 /17 .	6.240 .	T74	
			97_1	87-1	53_1	65-1	285-1	1 266-**	7MB	
1			52-11	0,-11	55-11	0.5-11	205-11	1,200-11	ZWF	
1										

	Α	В	с	D	E	F		G			I	н	
	S     P     S     % of adults 15+     % of adults 25+       H = 2     P     P     P     P       H = 2     P     P     P     P												
Countru or territoru	Participation in a education & train	% of youth enroll in TVET	TVET share of sec enrolment (%)	TVET share of post-secondary non-tertiary (%)	Gross graduation from tertiary (%)	GER tertiary (%)	Copy & paste within document	Use formula in spreadsheet	Write computer program	Primary	Lower secondary	U pper secondary	Post-secondary
SDG indicator	431	433				432		441			4	4 3	
Reference year	4.5.1	4.5.5	20	)19		4.5.2				2019			
Nothern Africa on dWastern Asia													
Northern Africa and Western Asia					29 1	53	18 1	9,	7,				
Armenia		 8i			44	51	10-1			 99-2	 97-2	 90-2	47-2
Azerbaijan		15 <sub>i</sub>	15	100	21-1j	32i	64-1	21-1	1-1	98-2	96-2	89-2	30-2
Bahrain		4	7	100	29	56	58	36	18	87-1	80-1	65-1	32-1
Cyprus	48-3	7-1i	9-1		25-1i	81-1i	45-4	28	4	96-1	82-1	73-1	39-1
Egypt	1-2	12	22		18-3	39-1	58	16	9		73-2	67-2	13-2
Georgia	Z-2	5	5	100	35	64	33	11		99-2	98-2	92-2	59-2
Israel	53-4		20-1			61-1	Z D-1	/-1	D-1	96-4		81-4	47-4
Jordan		1	3			33							
Kuwait		4	2-4			55	60-2	38	13	62-1	56-1	31-1	19-1
Lebanon			16										
Libya													
Morocco		6	9	100	18	39	49	22	9				
Uman Palestine		2	0.4		22	40	90	28	6	84-4 95 -	64 -	50-4	21-4
Qatar	2-1	0.7	1	100	7	19	44	25	5	88-2	68-2	41-2	20-1
Saudi Arabia		4	1	100	41	71	68	47	14	81-2	69-2	54-2	31-2
Sudan			2-1			17-4	4-3	2-3	2-3				
Syrian Arab Republic						43							
Tunisia			9-3	100-з	25-3	32	23	18	16	74-3	45-3		15-3
Turkey	21-3	25-1	23-1		34-1	113-1		19-2	3	90-2	61-2	39-2	19-2
United Arab Emirates		1	2	100	15-2	53	59-1	34-1	17	91-1	83-1	69-1	55-1
remen		3	0.5-3										
Central and Southern Asia													
Afghanistan		1-1 2	I-1	55-1	10-1	10-1							
Bhutan		5	4	100		16	1	0.4	0.2	22.5	45	17 .	10
India		- 11	1	100+1		29		8-1		52-2	20-2	17-2	11-2
Iran, Islamic Republic of		6-2	13-2		37-1	63-1	21-2	7-2	1-2		70-3	48-3	23-3
Kazakhstan	17-2	19+1	10+1	100+1	69+1	71+1	14	40	6	100-1	99-1	97-1	79-1
Kyrgyzstan		6	8	100	29	42							
Maldives	9-3		6			31-2							
Nepal		2	1		9-1	13	 E	 ว					
Sri Lanka	1-3	4-1	4-1	100-1		21	3	2	1	49-2	83-1	63-1	9-2
Tajikistan						31-2					95-2	81-2	23-2
Turkmenistan		2i		100		14							
Uzbekistan		24-1	34			13	22-1	10-1		100-1	100-1	96-1	64-1
Eastern and South-eastern Asia													
Brunei Darussalam		7	11		26	31	60	42	28				
Cambodia						15	29	9	1	23-4	12-4	9-4	6-4
China		7 i	18	71	34	54							
DPR Korea				100-1	21-1	27-1							
Hong Kong, China		3i	2	69		81	54	36	1	96-2	/9-2	63-2	29-2
lanan		13-1	11-1		21-1	20-1	7-1	6-1	0.5-1	/ 0-1	J L-1	33-1	10-1
Lao PDR	1-2	4	1	100	9	 14							
Macao, China		1	3		71	100	46	38	4	90-3	73-3	52-3	26-3
Malaysia		5	10			43	59	27	8		74-3	58-3	21-3
Mongolia	1-2	6	10	100	51-1	66-1	15	12	3				
Myanmar	0.4-2	0.3-1	0.2-1	100-2		19-1							
Philippines		4	10	100-2		35-2	6	2	1	84-2	59-2		20-2
Kepublic of Korea		24-1	9-1		52-2	96-1	82	46	6	96-4	86-4	7.6-4	40-4
Thailand	0.5-3	24-1i 6-4		72-1	25-4i	09-1i 49-3	21	40	1	67-1	46-1	33-1	J 0-1
Timor-Leste		5	10										
Viet Nam	0.2-4			100-3	22	29							

	I		J			к		
%	achieving	Literacu	rate (%)		Illite	erates		
pit pit		Literacy			inte			
				% fe	male	(000	,000)	
		v						qe
Literac	y Numeracy	Youth	Adults	Youth	Adults	Youth	Adults	try co
	4.6.1	4.	6.2	10				Count
			20	13				•
		97-1i	81-1i	52-1i	66-1i	156-ıi	5,484-1i	DZA
		100-2	100-2	40-2	67-2	1-2	6-2	ARM
		100-2	100-2	67-2	68-2	1-2	16-2	AZE
		100-1	97-1	96-1	67-1	1-1	32-1	BHR
		 88-2	 71-2	 54-2	 59-2		 18.519-2	EGY
		100-2	99-2	67-2	59-2	2-2	21-2	GEO
		94-2	86-2	60-2	69-2	487-2	3,321-2	IRQ
72-4	68-4							ISR
		99-1i	98-1i	38-1i	61-1i	13-1i	116-1i	JOR
		99-1 100-1i	96-1	25-1	48-1 68-1i	4-1 3-1i	250-1	LBN
								LBY
		98-1i	74-1i	56-1i	69-1i	134-ıi	6,885-1i	MAR
		99-1	96-1	29-1	51-1	9-1	161-1	OMN
		99	97	52	77	7	80	PSE
		95-2i	93-2i	17-2i	17-2i	21-2i	154-2i	QAT
		73-1	61-1	49-1i	57-1i	2.296-1	9,774-1i	SDN
								SYR
								TUN
53-4	49-4	100-2	96-2	81-2	85-2	33-2	2,380-2	TUR
		99-4i	93-4i	62-4i	19-4i	6-4i	539-4i	ARE
								T L M
		65	43	61	60	2 701	12.054	AEG
		95	43-1	37	55	1 581	30.005	BGD
		93-2	67-2	49-2	60-2	10-2	183-2	BTN
		92-1i	74-1i	56-1i	64-1i	20,538-1i	252,864-1i	IND
		98-3	86-3	54-3	66-3	229-3	8,700-3	IRN
74-2	73-2	100-1i	100-1i	70-1i	63-1i	2-1i	29-1i	KAZ
		100-1i	100-1i 98 p	3/-1i 24.2	62-1i 29 a	3-1i	18-1i 9 .	KGZ
		92-1i	68-1i	62-1i	23-3 71-1i	481-1	6.275-1i	NPL
		75-2	59-2	62-2	64-2	10,534-2	54,876-2	PAK
		99-1	92-1	39-1	59-1	38-1	1,331-1	LKA
								TJK
							 ר	TKM
		100-1	100-1	50-1	100-1	-1	Z-1	UZB
_		100	07	25	64	0.2	0	DDN
		100-1i	97-1	35-1i 47	67 i	U.Z-1i	9-1i	BRN
		92-4 100-1i	97-1	47-4	75-1i	375-1i	37.038-1i	CHN
								PRK
								HKG
		100-1	96-1	50-1	69-1	133-1	8,527-1	IDN
			 QE .					JPN
		92-4 100-2	97-2	37-2	75-2	0.7-2	19_2	MAC
		97-1	95-1	46-1	61-1	177-1	1,234-1	MYS
		99-1i	98-1i	33-1i	44-1i	6-1i	35-1i	MNG
		85-3i	76-зі	51-зі	61-3i	1,468-3i	9,360-зі	MMR
		99-4	98-4	39-4	49-4	182-4	1,257-4	PHL
74 -			97 1					KUR
		98-1	94-1	37-1	63-1	176-1	3,589-1	THA
		84-1i	68-1i	46-1i	56-1i	45-1i	252-1i	TLS
		98-1i	95-1i	50-1i	65-1i	224-1i	3,670-1i	VNM

	A	В	C	D	E	F		G			I	H		
	dult ing (%)	pa	ondary		ratio			% of adults 15+ with ICT skills			% of ad having atta	ults 25+ ined at least		
Country or territory	Participation in ac education & train	% of youth enroll in TVET	TVET share of sec enrolment (%)	TVET share of post-secondary non-tertiary (%)	Gross graduation from tertiary (%)	GER tertiary (%)	Copy & paste within document	Use formula in spreadsheet	Write computer program	Primary	Lower secondary	U pper secondary	Post-secondary	
SDG indicator	4.3.1	4.3.3				4.3.2		4.4.1			4.4	4.3		1
Reference year			20	19						2019				1
Oceania														
Australia		13-1	29-1	100-1	65-1	108-1				100-1	93-1	78-1	48-1	
Cook Islands	1-3	-												
Fiji	1-3			100-3							87-2	45-2		
Kiribati Marchall Islands			 ว											
Micronesia, F. S.			 	45										
Nauru		1		100										
New Zealand	67-4	10-1	14-1	84-1	46-1	83-1						70-3	46-3	_
Niue		4	5											
Papua New Guinea		2-3	 9-3											
Samoa		3			11	14								1
Solomon Is														
Tokelau T		-												
Tuvalu		Z-4 3	3-4 9	δ1-4										-
Vanuatu		1-4	2-4	74-4										
Latin America and the Caribbean														
Anguilla		-	-	-										
Antigua and Barbuda		2-1	4-1											
Argentina	6-2	1			8-1	92-1				93-1		57-1	20-1	
Aruba Babamac						16-3								
Barbados		-		53										
Belize		3	9		7-4	25								
Bolivia, P. S.	5-2	27	63							72-4	59-4	43-4	24-4	
Brazil		4-1i	4-1	100-1		53-1i	20-1	12-1	3-1	80-1	60-1	47-1	17-1	_
British virgin Islands Cauman Islands		1-3	4-1			10-1				99-4	95-4	90-4		
Chile	47-4	14-1	12-1		16-1	91-1		43-2	12-2	88-2	80-2	59-2	22-2	
Colombia			8		27	55	33	23	5	79-1	54-1	50-1	21-1	
Costa Rica		9	25			58		25-1	4-1	82-1	55-1	40-1	22-1	
Curacao		15	21	100	1/-1	41-1	22	22	0 4-2					
Dominica		-	-	3										1
Dominican Republic	6-2	5i	10		31-2i	60-2i	22-4	11-4	7-4	71-3	67-3	49-3	23-3	
Ecuador	3-1	11-1	14-1		36-1	48-1	27	20	5	83-2	53-2	44-2	14-2	
El Salvador Grenada		7-1 2-1	18-1	 100-1	67-3	29-1				59-2	43-2	30-2	8-2	
Guatemala	3-2	9-1	29		5-4	22-4								
Guyana	2-2			90										
Haiti														
lamaica	5-2	-	54	 80	11	25		 6-2		60-1	31-1	Z 3 - 1	10-1	
Mexico	30-2	13-1	28-1		27-2	42-1		26	7	83-1	63-1	36-1	16-1	1
Montserrat														
Nicaragua														
Panama	4-2	/-2 5-3	1/-2		25-3	48-3				76-1	 50-1	39-1	 15-1	
Peru	34-2	1	2			71-2	31	20	3	82-1	64-1	58-1	22-1	
Saint Kitts and Nevis		3		100-3		87-4								
Saint Lucia		1	2	34		15								
Saint Vincent/Grenadines		2	2	31-2		24-4				91-2		42-2	4-2	
Suriname		 18-4	44-1		2-4									1
Trinidad and Tobago														
Turks and Caicos Islands		7-1			9-4	11-4								
Uruguay Venezuela P. P.	6-1	11-2	24-1			63-2				91-1	57-1	30-1	13-1	
venezoela, D. N.		2-4	5-2							د-ر ر	1-1-3	0 2 - 3	3-ل ر	

	% ach	l Nieving	Literacu	J urate (%)		Illite	(		
	pronei		Literacy		% fe	male	Nui (000	nber 1,000)	
	Literacy	Numeracy	Youth	Adults	Youth	Adults	Youth	Adults	y code
	4.	6.1	4.	6.2					ntr
				20	19				Õ
									AUS
1									СОК
			100-2	99-2	36-2	49-2	0.4-2	6-2	FJI
1									KIR
									MHL
									FSM
_									NRU
									NZL
						50.4	- 4		
			55-4	57-4	2 3-4	50-4	-4	0.5-4	PNG
1			 99-1i	 99-1i	31-1i	40-1i	0.3-1j	 1-ıi	WSM
									SLB
									TKL
			99-1i	99-1i	40-1i	46-1i	0.1-1i	0.4-1i	TON
_									TUV
			96-1i	88-1i	45-1i	53-1i	2-1i	22-1i	VUT
									AIA
_				99-4		29-4		1-4	ATG
			100-1	99-1	24-1	49-1	35-1	333-1	ARG
			100-1	90-1i	0U-1j	0 0 - 1j	U.1-1i	Z -1i	RHS
									BRB
1									BLZ
			99-4	92-4	49-4	77-4	13-4	548-4	BOL
			99-1	93-1	35-1	50-1	270-1	11,168-1	BRA
									VGB
									CYM
	46-4	38-4	99-2	96-2	49-2	52-2	28-2	531-2	CHL
			99-1	90-1	40-1	49-1	100-1	1,0/J-1 8/1 1:	COL
1					41-1	11-64			CUB
									CUW
									DMA
			99-3	94-3	48-3	50-3	22-3	462-3	DOM
	28-2	23-2	99-2	93-2	39-2	56-2	23-2	851-2	ECU
			98-1	89-1	44-1	63-1	26-1	515-1	SLV
									GRD
1									GUY
1			83-3i	62-3i	51-3i	56-3i	366-31	2.741-3i	HTI
1			97-1	87-1	26-1	50-1	71-1	838-1	HND
									JAM
	49-2	40-2	99-1	95-1	44-1	61-1	151-1	4,273-1	MEX
_									MSR
			92-4i	83-4i	37-4i	51-4i	102-4i	744-4i	NIC
			99-1	95-1	61-1	50-1	0-1 22.	139-1	PAN
	 29_2		99_1	94-1	54-1	75-1	2 3 -1 5 7_1	1 334-1	PER
									KNA
									LCA
1									VCT
									SXM
			99-1i	94-1i	57-1i	65-1i	1-1i	24-1i	SUR
									TTO
									TCA
	99-3		99-1	99-1	37-1	40-1	63.3	35-1 615 x	VEN
			J J - 5	J1-3	5-0-5	7 2-5	0 3 - 3	0 ± J = 3	V LIN

	Α	В	С	D	E	F		G			ŀ	1		
	dult ing (%)	pa	ondary		ratio			% of adults 15+ with ICT skills			% of adu having attai	ılts 25+ ned at least		
Country or territory	Participation in ac education & train	% of youth enrolls in TVET	TVET share of sec enrolment (%)	TVET share of post-secondary non-tertiary (%)	Gross graduation from tertiary (%)	GER tertiary (%)	Copy & paste within document	Use formula in spreadsheet	Write computer program	Primary	Lower secondary	U pper secondary	Post-secondary	
SDG indicator	4.3.1	4.3.3				4.3.2		4.4.1			4.4	1.3		
Reference year			20	)19	1					2019				
Europe and Northern America														
Albania	9.2	5	8		/3	60	13	7	2					
Andorra	5-3	5	10	100		00	15	1	6.2	07.5			22.5	1
Austria	60.5	78.1	34.1	100	35.1	87.1	63.4	46.4	Q	57-3	72-3	47-3 80 a	31 5	
Relarus	00-3	10 1	13.	100-1	33-1	87.1	41	20	2			00-2	J1-2	
Belgium	45-2	25-1	43-1	93_1	50-1	79_1	65-2	45	4	96-2	87-2	69-2	36-2	
Bermuda	15 5	3	15 1	551	1	19-1	052	15		502	07 2	87-3	55-3	
Bosnia and Herzegovina	9-3	22	38		31	40	22	8	2	88-1	81-1	64-1	12-1	
Bulgaria	25-3	17-1	33-1	100-1	48-1	72-1			1		95-2	76-2	25-2	
Canada			5-1		40-1	70-1						84-3	60-3	
Croatia	32-3	22-1	38-1		44-1	68-1	54	43	9					
Czechia	46-3	26-1	35-1	36-1	43-1	64-1	56-4	45	6	100-2	100-2	91-2	21-2	
Denmark	50-3	12-1	21-1		56-1	81-1	68-3	54	14		94-1	79-1	37-1	
Estonia	44-3	12-1	23-1	100-1	45-1	70-1	55-3	44-3	7-2			88-1	40-1	
Finland	54-3	20-1	48-1	100-1	58-1	90-1	69-3	48	9			76-2	36-2	
France	51-3	19-1i	18-1	55-1	47-1i	68-1i			6	98-2	84-2	70-2	30-2	
Germany	52-3	21-1	19-1	93-1	41-1	70-1	57	35	5	100-1	96-1	83-1	36-1	
Greece	17-3	13-1	15-1	100-1	45-1	143-1	52-2	38	4	91-3	65-3	55-3	27-3	
Hungary	56-3	18-1	20-1	100-1	33-1	50-1	53-3	37-3	4	100-3	97-3	76-3	29-3	
Iceland		9-1	18-1	99-1	51-1	73-1	82-2	71-2	13-2					
Ireland		8-1i	27-1	100-1		77-1i	53-1	36-1	6-1		86-2	71-2	43-2	
Italy	42-3	21-1	33-1	100-3	40-1	64-1	42-3	31-3	4-3	95-4	78-4	49-4	15-4	
Latvia	48-3	17-ıi	20-1	100-1	49-1i	93-1i	53-4	32	3		100-4	90-1	44-1	
Liechtenstein		23-1i	34-1		14-1i	38-1i								
Lithuania	28-3	9-1i	9-1	100-1	62-1i	74-1i	45-3	42	5	99-2	96-2	87-2	55-2	
Luxembourg	48-3	22-1	33-1	100-1	8-1	19-1	82-4	69-2	11-2			69-4		
Malta	36-3	10	16		45-2	65	50-4	41	8	99-1	82-1	45-1	30-1	
Monaco			11+1	100+1										
Montenegro		23	34		36	54		28	4					
Netherlands	64-3	23-1	37-1		52-1	87-1	72	54	9	99-1	90-1	71-1	33-1	
North Macedonia	13-3		28-1	100-1	2/-1	43-1	32-3	21-3	3-3					
Norway	60-3	1/-1	28-1	100-1	55-1	83-1	/8	60	11	100-2	99-2	/8-2	40-2	
Poland	26-3	20-1	28-1	100-1	45-1	69-1	34-4	28	4	99-3	85-3	85-3	28-3	
Portugal	46-3	10-1	24-1	100-1	51-2	00-1	4 / -3	37	8	92-1	54-1	3/-1	19-1	
Republic of Moldova		101	13	100	34-1i	391				99-1	9/-1	/ 5-1		-
Romania Dussian Enderstian	/-3		28-1	100-1	58-3	51-1 or	22-2	15	1	99-2	91-2	b / -2	18-2	
Russiali Federacion		10-1i 2	14-1	100-1	22.	0 D - 1 E 1	21	24	1			 E 4 .		
Sarbia		25.	25		3 3 - 1	58	34.4			97-1	0.0-1	72 .	22 -1	
Slovakia	46 3	231	30 .	100	35.1	45 1	J <del>4</del> - 4	24-2	4	100 -2	90-2	87 2	23-2	
Slovenia	46 2	35.1	45 1	100-1	48 1	77 1		42 2	-	100-2	98 2	83.2	23-2	
Snovellia	40-3	15 1	19,	100 1	43 1	91 1	52.4	42-3	7	97 1	78 1	50 1	31 1	
Sweden	64 2	13.1	21.	76 1	40 1	72 1	64	46	, 11	100 2	91 2	76 2	30 2	
Switzerland	69 2	23.1	37 1	77,	54 1	61 1	04	57	10	100-2	97.	86 1	J J-2	
Ilkraine	0 9 - 3	23-1	7	100	J4-1	01-1			10		37-1	00-1		
United Kingdom	52-3	18-1	32-1	100		61-1	65-2	46	9	100-2	100-2	77-2	44-2	
United States	59-2			100-1		88-1i				99-1	96-1	90-1	45-1	
		1	1	1						1	1		1 C C C C C C C C C C C C C C C C C C C	

	% ach	l ieving	Litoracu	J		Illita	K		
l	protici	ency in	Literacy	rate (%)		mite	rates		
					% fe	male	Nur (000	nber ,000)	
	Literacy	Numeracy	Youth	Adults	Youth	Adults	Youth	Adults	) code
l	4.0	6.1	4.6	5.2					ntr
l		·		20	19				Cou
			0.0	0.0	26	60	2		
ł			99-1i	98-1i	26-1i	60-1i	3-1i	44-1i	ALB
ł									AND
ł									AUT
ł			100-1i	100-1i	4 Z - 1i	61-1i	1-1i	19-1i	BER
ł									BEL
ł									BINIU
ł									BCD
í									CAN
ł									HPV
í			100-3	100-3	36-3	40-3	7-3	15-3	C7E
1			100 3	100 3	50 5	10 3	2 3	15 5	DNK
í									FST
1									FIN
Í									FRA
1									DEU
1	73-4	71-4	99-1i	98-1i	54-1i	65-1i	9-1i	187-ıi	GRC
1	81-2	82-2							HUN
1									ISL
1									IRL
			100-1i	99-1i	37-1i	63-1i	4-1i	444-1i	ITA
]			100-1i	100-1i	37-1i	47-1i	0.3-1i	2-1i	LVA
									LIE
	84-4	82-4							LTU
									LUX
			99-1i	95-1i	30-1i	37-1i	0.3-1i	21-1i	MLT
Į.									MCO
l			99-1i	99-1i	55-1i	77-1i	1-ıi	6-1i	MNE
Į.									NLD
l									MKD
Į.									NOR
ł									POL
ł			100-1i	96-1i	44-1i	68-1i	4-1i	343-1i	PRT
ł									MDA
ł			99-1i	99-1i	4 / -1i	63-11	12-1i	190-1i	ROU
ł			100-1	100-1i	41-1j	54-1i	4 Z - 1i	323-1i	RUS
ł			100-1	100-1	32-1	59-1	1	1	SMR
ł			100-3	22-3	40-3	19-3	3-3	00-3	SKR
ł									SVK
	/ )-4	/ 4-4							SVN
ł			100-1	90-1	44-1	0/-1	13-1	023-1	ESP
									SWE
ł									
ł									GRP
1		71 .							LISA
1	01-2	1 1-2							AC U

#### TABLE 5: SDG 4, Target 4.5 - Equity

By 2030, eliminate gender disparities in education and ensure equal access at all levels of education and vocational training for the vulnerable, including persons with disabilities, indigenous peoples and children in vulnerable situations

	GENDER														
		A				В			с		D			E	
				G	PIA in minim	um proficien	cy								
		GPIA in				End o	flower	GP	IA in	GPIA i	n adult				
		completion		End of	primary	seco	ndary	litera	cy rate	profic	ciency	G	PIA in gross	enrolment ra	tio
	Primary	Lower secondary	Upper secondary	Reading	Mathematics	Reading	Mathematics	Youth	Adults	Literacy	Numeracy	Pre-primary	Primary	Secondary	Tertiary
SDG indicator		-				4.5.1									
Reference year								2019							
Region								Median							
World	<b>1.00</b> i	<b>1.01</b> i	1.04i			1.15i	<b>1.00</b> i	0.98	0.92			<b>0.99</b> i	0.98i	0.99i	1.13i
Sub-Saharan Africa	1.03i	0.89i	0.79i					0.93	0.81			0.99i	0.96	0.88-1i	0.76
Northern Africa and Western Asia	1.00i	1.02	1.15	 1.17i	1.02i	1.32	1.06	0.96	0.87			0.99i	0.96	0.96i	1.04
Northern Africa	1.02i	1.15	1.30i			1.31	1.10	0.99	0.84			0.99	0.98	1.00i	1.15
Western Asia	1.00 <sub>i</sub>	1.00i	1.12 <sub>i</sub>	1.17 <sub>i</sub>	1.02i	1.33 <sub>i</sub>	1.04	0.94	0.90			0.99j	0.94i	0.93i	0.98j
Central and Southern Asia	1.00i	0.99i	0.91 <sub>i</sub>					0.96	0.82			1.01	0.99	1.01	1.03
Central Asia	1.00	0.99	0.86					1.00	1.00			0.98	0.99	0.99	1.02
Southern Asia	0.94i	0.95i	0.92i	1.04i	1.03i			0.96	0.81			1.01	0.99	1.01	1.04
Eastern and South-eastern Asia	1.02i	1.07i	1.11i	1.21i	1.03i	1.23i	1.04i	1.00	0.97			0.99	1.00	1.02i	1.15i
Eastern Asia			1.12i			1.08i	1.01i	1.00	0.97			1.00	1.01	1.02	1.14
South-eastern Asia	1.03i	1.08i	1.10i	1.23i	1.06	1.27	1.07i	1.00	0.97			0.96i	0.98i	1.03i	1.19i
Oceania												0.98i	0.97i	0.93i	1.27i
Latin America and the Caribbean	1.02i	1.06i	1.07i			1.13i	0.81i	1.00	0.99			1.01i	0.98i	1.04i	1.23i
Caribbean												1.03	0.99i	1.03	1.44i
Central America	1.01	1.04	1.08			1.11	0.82	1.00	0.99			1.02	0.99	1.06	1.17
South America	1.02	1.05	1.06			1.12i	0.75i	1.00	1.00			1.01	0.98	1.03	1.14i
Europe and Northern America	1.00	1.00	1.04	1.01i	0.99i	1.13	1.00					1.00i	1.00i	0.99i	1.22i
Europe	1.00	1.00	1.05	1.01i	0.99i	1.13	1.00	1.00i				0.99i	1.00i	0.99i	1.19i
Northern America	1.00i	1.01i	1.01	1.01	0.97	1.09	0.99			1.02i	0.97i	1.01i	1.00i	0.99i	1.27i
Low income	0.99	0.89	0.76					0.92;	0.78;			1.00:	0.97:	0.83-*	0.64.1
Middle income	1.01;	1.02	1.07			1 22;	1.00;	0.921	0.92			0.99	0.921	1 01	1 13
Lower middle	1.01	1.021	0.98			1.221	1.001	0.96	0.92			1.00	0.99	1.01	1.15
	1.05	1.021	1.08			1.22	1.00;	1.00	0.04			0.99	0.99	1.00	1.00
High income	1.01	1.04	1.001	1.02		1.221	1.001	1.00	1.00;			1.00:	1.00:	0.99;	1.10

A Adjusted gender parity index (GPIA) in school completion rate by level.

B Adjusted gender parity index (GPIA) in percentage of students with minimum level of proficiency at the end of given level.

C Adjusted gender parity index (GPIA) in youth and adult literacy rate.

D Adjusted gender parity index (GPIA) in percentage of adults aged 16 and over achieving at least a fixed level of proficiency in functional literacy and numeracy skills.

E Adjusted gender parity index (GPIA) in gross enrolment ratio by level.

F Adjusted parity index for location (rural-urban) and wealth (poorest to richest quintile) in school completion by level.

G Adjusted parity index for wealth (poorest to richest quintile) in achievement of minimum proficiency.

Source: UIS and GEM Report analysis of household surveys. Data refer to school year ending in 2019 unless noted otherwise. Aggregates represent countries listed in the table with available data and may include estimates for countries with no recent data. (-) Magnitude nil or negligible.

(...) Data not available or category not applicable.

(± n) Reference year differs (e.g. -2: reference year 2017 instead of 2019).

(i) Estimate and/or partial coverage.

F           Disparity in primary completion         Disparity in lower secondary completion         Disparity in upper secondary completion           Adjusted parity index         % of poorest completing         % of poorest Adjusted parity index         % of poorest completing         % of poorest completing         % of poorest completing         % of poorest completing	G Wealth disparity in minimum pro End of primary Secon Si terme ippe at the secon se	oficiency f lower ndary system tem tem tem tem tem tem tem tem tem
Disparity in primary completion     Disparity in lower secondary completion     Disparity in upper secondary completion       Adjusted parity index     % of poorest completing     % of poorest Adjusted parity index     % of poorest completing     % of poorest completing	Wealth disparity in minimum pro End of primary Secon Site ipp attack W Bu Site ipp attack W Bu Site ipp attack W Bu Site ipp attack W Bu Site ipp attack W Bu Site ipp attack Site ipp attack Site Site Site Site Site Site Site Site	oficiency f lower ndary system tem tem tem tem tem
Adjusted parity index         % of poorest completing         Adjusted parity index         % of poorest completing         % of poorest Adjusted parity index         % of poorest completing	End of primary Secon End of primary Secon Secon End of primary Secon End of primary Secon Secon End of primary Secon Secon End of primary Secon	f lower ndary system tice transfer tran
Adjusted parity index completing Adjusted parity index completing Adjusted parity index completing	End of primary secon built secon secon built secon built secon bui	lawer hatthematics
	Reading Mathematics Reading	Aathematics
eatth eatth eatth eatth eatth eatth eatth	<u>∞</u> ≥ ∞	
		~
4.5.1		
2019		
Median		
0.97i 0.92i 73i 80i 0.90i 0.74i 39i 0.70i 0.51i	0.61i	0.59i
0.65 0.35i 31i 32i 0.40 0.15i 11i 7i 0.27i 0.07i 3i 1i		
1.00i 0.96i 0.97i 0.82i 0.84i 0.69i	0.80i 0.65i 0.47i	0.52
0.98i 0.92i 0.83i 0.57i 0.69i 0.37i	0.34i	0.40
1.00 <sub>i</sub> 0.99 <sub>i</sub> 0.98 <sub>i</sub> 0.93 <sub>i</sub> 0.86 <sub>i</sub> 0.71 <sub>i</sub>	0.80i 0.66i 0.48i	0.53
0.99 <sub>i</sub> 0.92 89 89 0.95 <sub>i</sub> 0.74 69 74 0.65 <sub>i</sub> 0.29 <sub>i</sub> 21 <sub>i</sub> 22 <sub>i</sub>		
1.00         1.00         99         100         1.00         0.98         97         97         0.92         0.81         81         85		
0.97 <sub>i</sub> 0.79 72 79 0.92 <sub>i</sub> 0.57 47 53 0.49 <sub>i</sub> 0.24 <sub>i</sub> 16 <sub>i</sub> 8 <sub>i</sub>		
0.96i 0.86i 71i 89i 0.89i 0.61i 40i 68i 0.68i 0.49i 31i 32i	0.45i	0.51i
0.80i 0.75i	0.89i	0.89i
0.94i 0.75i 69i 77i 0.76i 0.46i 36i 51i 0.58i 0.29i 24i 26i	0.41	0.47i
0.97i 0.95i 89i 93i 0.88i 0.75i 65i 68i 0.70i 0.47i 38i 37i	0.43i	0.26i
<u> </u>		
0.93         0.88         86         88         0.80         0.65         56         57         0.63         0.36         25         30	0.35	0.20
0.99 0.96 93 94 0.90 0.79 68 80 0.74 0.62 49 52	0.44i	0.27i
1.00 <sub>i</sub> 1.00 <sub>i</sub> 1.00 <sub>i</sub> 0.97 0.97 <sub>i</sub> 0.88	0.93 <sub>i</sub> 0.78 <sub>i</sub> 0.70	0.68
1.00i 1.00i 1.00i 0.97 0.98i 0.85	0.93i 0.79i 0.70	0.68
1.00i 0.99i 0.95i 0.93	0.92 0.70 0.81	0.71
0.63 0.35 29i 32i 0.38 0.13i 10i 9i 0.27i 0.07i 2i 1i		
0.97i 0.92i 84i 92i 0.89i 0.68i 55i 62i 0.68i 0.39i 21i 31i	0.44i	0.40i
0.89i 0.73i 62i 65i 0.61i 0.33i 23i 24i 0.45i 0.20i 13i 9i		
0.99i 0.96i 94i 95i 0.93i 0.82i 75i 80i 0.80i 0.56i 41i 48i	0.46i	0.45i
1.00i 1.00i I.00i 0.98i I.0.95i 0.90i I.	0.92i 0.76i 0.72	0.68

								GENDER							
		Α				В			с		D				
				6	PIA in minim	um proficien	~								
		CDIA in		J		Code Code	() ()	CD	IA :	CDIA	a a du la				
		completion		End of	primaru	End of Seco	r Iower ndaru	litera	iA in cu rate	Drofie	n adult ciencu	G	PIA in aross e	nrolment ra	tio
					<u>n</u>		່ <u>ທ</u>			,					
		Ð	Ð		atic		latic				5	Jary		Ę	
	lary	er	er	ding	them	ding	them	£	ম	racy	Jera	prin	lary	buda	iary
Countru or territoru	Prin	Low	Upp	Rea	Mat	Rea	Mat	Yout	Adu	Lite	N	Pre-	Prin	Secc	Tert
SDG indicator		I				451									
Deference uear						4.5.1		2019				I			
								1015							
Sub-Saharan Africa	0.00	0.76	0.65									0.00	0.07	0.64	0.02
Angola	0.89-4	0.76-4	0.65-4									0.89-3	0.87-4	0.64-3	0.83-3
Benin	0.8/-1	0.54-1	0.45-1					0./4-1i	0.58-1j			1.02	0.93	0.76-3	0.47-1
Botswana Dudina Fasa												1.03-4	0.98-4		1.32
Burkina Faso								0.88-1i	0.05-1			1.00	0.99	1.03	0.56
	1.10-2	0.80-2	0.84-2					0.94-2	0.80-2			1.04	1.01	1.13	0.61-1
Cabo Verde								1.01-4	0.89-4			1.01-1	0.93-1	1.09-1	1.33-1
Cameroon	0.98-1	0.83-1	0.81-1					0.94-1i	U.8/-1i			1.02	0.90	0.86-3	0.89-1
Central African Republic								0.60-1i	0.52-1i			1.04-2	0.78-3	0.6/-2	
Chad	0.76	0.47	0.30					0.55-3	0.45-3			0.96	0.78	0.53	0.29-4
Comoros								1.00-1i	0.82-1i			1.03-1	1.00-1	1.06-1	
Congo	1.04-4	0.79-4	0.69-4					0.92-1i	0.87-1i			1.08-1	0.97-1		0.67-2
Cote d'Ivoire	0.88-3	0.62-3	0.82-3					0.83-1i	0./5-1i			1.03	0.94	0.79	0.75
D. R. Congo	1.00-1	0.95-1	0.77-1					0.88-3	0.75-3			1.05-1	0.94-1	0.64-4	0.56-3
Djibouti												0.85+1	0.96+1	1.03+1	
Equat. Guinea												1.02-4	0.99-4		
Eritrea								0.99-1i	0.82-1i			0.99-1	0.86-1	0.91-1	0.71-3
Eswatini								1.02-1i	1.00-1i				0.92-1	0.99-3	
Ethiopia	1.03-3	1.12-3	1.18-3					0.98-2i	0.75-2i			0.95-4	0.91-4	0.96-4	
Gabon								1.04-ıi	0.97-1i						
Gambia	1.12-1	1.09-1	0.88-1					0.91-4	0.67-4			1.07	1.10		
Ghana	1.06-1	1.11-1	0.97-1					0.99-1i	0.89-1i			1.02	1.01	1.00	0.85
Guinea	0.75-1	0.61-1	0.51-1					0.62-1	0.51-1				0.82-3		
Guinea-Bissau	0.95	0.94	0.71												
Kenya				1.12-1	0.84-1			1.01-ıi	0.92-1i			0.97-3	1.00-3		0.74-2
Lesotho	1.25-1	1.39-1	1.28-1									1.04-3	0.95-2	1.26-2	1.35-1
Liberia								0.70-2i	0.54-2i			1.01-2	0.99-2	0.77-4	
Madagascar	1.14-1	1.01-1	0.97-1					0.99-1i	0.94-1i			1.10	1.02	1.04	0.95-1
Malawi								1.01-4i	0.79-4i			1.01-4	1.03	0.83	
Mali	0.81-1	0.71-1	0.60-1					0.75-1	0.56-1			1.03-1	0.90-1	0.82-1	0.42-4
Mauritania	0.86-4	0.73-4	0.57-4					0.80-2j	0.68-2i			1.21-4	1.06	1.05	0.61
Mauritius								1.01-1i	0.96-1i			1.00	1.03	1.02	1.29-2
Mozambique	0.96-4	0.89-4	0.82-4					0.85-2	0.69-2				0.93	0.89-2	0.81-1
Namibia	1.05-4	1.14-4	1.13-4					1.02-1i	1.00-ıi			1.03-1	0.97-1		1.49-1
Niger								0.70-1i	0.61-1i			1.06	0.88	0.75-2	0.64
Nigeria	1.00-1	0.89-1	0.76-1					0.84-1i	0.74-ıi				0.94-3	0.90-3	
Rwanda	1.22-4	1.16-4	0.84-4					1.05-1	0.89-1			1.04	0.98	1.12	0.84
Sao Tome and Principe	1.09	0.98	1.05					1.00-1i	0.93-1i			1.09-3	0.97-2	1.13-2	1.04-4
Senegal						1.11-4	0.86-4	0.84-2	0.61-2			1.11	1.12	1.12	0.75
Seychelles								1.01-ıi	1.01-ıi			1.02	1.05	1.06	1.55
Sierra Leone	1.03-2	0.89-2	0.60-2					0.89-1i	0.67-ıi			1.10	1.03	0.97-2	
Somalia	0.81-3	0.79-3	0.54-3												
South Africa	1.03-3	1.06-3	1.13-3					1.03-2	0.99-2	0.99-2	0.99-2	1.01-1	0.96-1	1.07-1	1.32-1
South Sudan	0.82-2	1.14-2	1.60-2					0.98-1i	0.72-1i			0.95-4	0.71-4	0.54-4	
Тодо	0.92-2	0.71-2	0.38-2					0.87-4	0.66-4			1.03+1	0.97+1	0.73-2	0.53
Uganda	1.07-3	0.87-3	0.79-3	1.01-4	0.85-4			1.01-ıi	0.86-1i			1.04-2	1.03-2		
United Republic of Tanzania	1.09-2	1.11-2	0.67-2					0.97-4	0.88-4			1.00	1.02	1.07	0.66
Zambia	1.03-1	0.89-1	0.82-1			1.46-4	1.26-4	0.99-1i	0.92-1i			1.07-3	1.02-2		
Zimbabwe	1.06	1.02	0.71												0.84-4

								N/WEALTH	LOCATIO							
	fisionau	G 		Weelsh				Discout		: 	 	Dissori	41au			Di
	ficiency	minimum pro	disparity in	wealth	npietion	condary cor	ty in upper se	Dispari	npletion	condary con	ty in lower se	Dispari	tion	nary comple	sparity in prin	Dis
	ndary	Seco	primary	End of	oleting	comp	arity index	Adjusted p	leting	comp	arity index	Adjusted p	oleting	com	oarity index	Adjusted (
ę	.9		Ŋ													
20	emati	Ð	emati	Ð			_	5			-	5			_	5
ount	lathe	eadin	lathe	eadin			ealth	ocatio			ealth	ocatio			ealth	ocatio
Ŭ	2	ě	2	ž	ш	۶	3	<u> </u>	ш.	٤	3	Ľ	ш	۶	3	Ĕ
								.5.1	4							
								019	2							
										-						
AGO					1-4	2-4	0.04-4	0.1/-4	3-4	5-4	0.06-4	0.20-4	1/-4	21-4	0.21-4	0.37-4
BEIN BW/A					0.2-1	I-1	0.02-1	0.25-1	3-1	D-1	0.08-1	0.45-1	10-1	Z4-1	0.20-1	0.70-1
BFA																
BDI							0.05-2	0.23-2		12-2	0.20-2	0.49-2	32-2	24-2	0.45-2	0.69-2
CPV																
CMR					0.3-1	3-1	0.02-1	0.21-1	5-1	13-1	0.11-1	0.41-1	30-1	36-1	0.35-1	0.66-1
CAF																
TCD							0.04	0.15			0.08	0.25			0.19	0.42
COM																
COG					0.2-4	1-4	0.01-4	0.09-4	0-4	/-4	0.08-4	0.30-4	40-4	42-4	0.43-4	0.61-4
COD					3	10 1	0.03-3	0.13-3	31,	31 1	0.08-3	0.27-3	39,	44 1	0.52-3	0.30-3
DII						10-1	0.25-1	0.50-1	51-1	51-1	0.40-1	0.00-1	55-1		0.05-1	0.75-1
GNQ																
ERI																
SWZ																
ETH					2-3	1-3	0.06-3	0.13-3	5-3	3-3	0.08-3	0.19-3	28-3	28-3	0.33-3	0.50-з
GAB																
GMB							0.17-1	0.27-1	22-1	18-1	0.29-1	0.40-1	50-1	42-1	0.55-1	0.63-1
GHA					10-1	10-1	0.14-1	0.46-1	25-1	21-1	0.29-1	0.60-1	54-1	51-1	0.61-1	0.82-1
GIN					3							0.16-1		23-1	0.20-1	0.40-1
							0.17	0.32			0.10	0.32			0.32	0.40
L SO								0.41-1	19-1		0.16-1	0.46-1	79-1	40-1	0.60-1	0.80-1
LBR																
MDG								0.34-1		4-1	0.04-1	0.38-1	21-1	14-1	0.20-1	0.68-1
MWI					2-3	4-3	0.07-3	0.28-3	5-3	7-3	0.11-3	0.31-3	29-3	22-3	0.35-3	0.58-3
MLI					0.2-4			0.22-1			0.14-1	0.30-1	21-1	29-1	0.32-1	0.50-1
MRT					5-4	9-4	0.13-4	0.35-4	24-4	19-4	0.26-4	0.52-4	21-4	34-4	0.32-4	0.58-4
MUS																
MOZ							4	0.09-4			0.03-4	0.23-4			0.08-4	0.40-4
NAM							0.41-4	0.64-4			0.72-4	0.82-4			0.86-4	0.92-4
NGA					6.1		0.13.1	0.48 1	14 1		0.19.1	0.58 1		26.1	0.78.1	0.65 1
RWA					4-4	2-4	0.08-4	0.30-4	17-4	11-4	0.15-1	0.49-4	38-4	26-4	0.48-4	0.05-1
STP							0.33	0.89			0.47	0.94			0.81	0.99
SEN	0.36-4	0.28-4			2	3	0.07	0.23	3	11	0.12	0.29	27	26	0.35	0.54
SYC																
SLE					1-2	2-2	0.03-2	0.16-2	7-2	10-2	0.11-2	0.30-2	34-2	31-2	0.37-2	0.54-2
SOM								0.10-3				0.59-3				0.67-3
ZAF					24-3	18-3	0.26-3	0.61-3	74-3	70-3	0.74-3	0.91-3	94-3	88-3	0.92-3	0.96-3
SSD																
IGU								0.29-2		3U-2	0.12 -	0.00-2	55-2 17 -	b/-2	0.76 -	0.50 -2
T7A					J-3	0.4.4	0.07-3	0.54-3	4-3	Q 4	0.12-3	0.38-3	67 4	54 4	0.20-3	0.59-3
ZMB	0.04-4	0.04-4			-4	01-4	0.01-4	0.27-1	12-1	20-1	0.17-1	0.46-1	40-1	38-1	0.42-1	0.69-1
ZWE								0.21	18	20	0.22	0.51	81	75	0.79	0.88

								GENDER							
		Α			I	3			c		D		1	E	
				G	PIA in minim	um proficien	cy								
		GPIA in				End of	lower	GPI	Ain	GPIA i	n adult				
		completion	1	End of	primary	seco	ndary	literae	cy rate	profic	ciency	G	PIA in gross e	nrolment ra	io
		_	_		tic		tic				_	Ð		_	
	Ð	dary	dary	Бĩ	ema	Ę	ema	_	5	ອົ	eracy	rima	Ę	idary	Ð
Country on torritory	rima	owei	ppei	eadi	Math	eadi	Math	outh	dult	itera	m	re-p	rima	econ	ertia
Country or territory	<u>م</u>	-D or	Ν	~	~	~	~	~	A		z	<u>م</u>	<u>م</u>	Ň	
SDG indicator						4.5.1									ļ
Reference year								2019							
Northern Africa and Western Asia															
Algeria	1.02	1.20	1.38			1.46-4	1.16-4	1.00-1i	0.86-1i				0.96		1.40
Armenia	1.00-1	0.98-1	1.08-1	1.06 .	1.04-4		1.07-4	1.00-2	1.00-2			1.04	1.01	1.05	1.25
Bahrain				1.00-3	 1 13-4			0.99-1	0.96-1			1.11	0.99	1.001	1.14
Cuprus	1.00-2	1.00-2	1.09-2		0.99-4	1.32-1						0.97-1j	0.99-1j	0.98-1j	1.08-1
Egypt	0.99-2	1.02-2	1.09-2				1.13-4	0.97-2	0.86-2			1.00	1.01	0.99	1.04-1
Georgia	1.00-1	1.00-1	1.08-1	1.07-3	1.02-4	1.37-1	1.04-1	1.00-2	1.00-2				1.01	1.01	1.12
Iraq	0.94-1	1.01-1	1.15-1					0.97-2	0.88-2						
Israel	1.00-1	1.00-1		1.05-3		1.22-1	1.09-1			1.01-4	0.92-4	0.99-1	1.01-1	1.02-1	1.30-1
Jordan	1.01-1	1.02-1	1.24-1			1.35-1	1.01-1	1.00-1i	0.99-1i			0.99	0.98	1.02	1.15
Lebanon					1.03-4		0.04-4	1.01-1	0.98-1			1.06	1.13	1.00-4	1.51
Libua						1.22-1	0.55-4	1.00-1	0.00-1						
Morocco					1.01-4	1.31-1	1.07-4	0.99-1i	0.78-1i			0.89	0.97	0.93	1.03
Oman				1.27-3	1.18-4		1.27-4	1.01-1	0.96-1			1.02	1.08	0.91	1.46
Palestine	1.00	1.15	1.22					1.00	0.97			0.99	1.00	1.09	1.39
Qatar				1.18-3	0.99-4	1.41-1	1.03-4	1.02-2i	1.02-2i			0.99	1.03		1.87
Saudi Arabia				1.34-3	1.38-4	1.44-1	0.91-4	1.00-2	0.95-2			1.05	1.01	0.90	1.07
Sudan								1.01-1i	0.86-1i			1.00-1	0.93-1	1.02-1	1.02-4
Syrian Arab Republic	1.02 .														1.11
Turkeu	1.05-1	1.1.1.11	1.30-1		0.98-4	1.20-4	1.05-4	1 00-2	0.95-2	0.84-4		0.96-1	0.99-1	0.98-1	0.90-1
United Arab Emirates				1.17-3	1.00-4	1.33-1	1.07-4	0.99-4i	1.03-4j			0.99	1.00	0.99	0.95
Yemen												0.90-3	0.87-3	0.73-3	
Central and Southern Asia															
Afghanistan	0.60-4	0.52-4	0.45-4					0.76-1i	0.54-ıi				0.67-1	0.57-1	0.35-1
Bangladesh	1.14	1.16	0.87			0.98-4	0.84-4	1.02	0.93			1.04-1	1.07-1	1.14	0.72
Bhutan								1.00-2	0.76-2			1.00+1	1.02+1	1.11-1i	1.06+1
India				1.04-2	1.00-2			0.97-1i	0.80-1i			1.04	1.02	1.02	1.10
Iran, Islamic Republic of				1.25-3	1.04-4		1.01-4	1.00-3	0.89-3			1.03-3	1.05-2	0.96-2	0.88-1
Kazakhstan	1.00-4	1.00-4	1.02-4	1.01-3	1.02-4	1.31-1	1.00-1	1.00-1i	1.00-ii	1.03-2	1.01-2	0.98+1	1.00+1	1.00+1	1.1/+1
Nyryyzstan Maldives	1.00-1	1.00-1	1.79 -1					1.00-1	1.00-1			1.01	1.03	0.93	1.21
Nepal	1.02-2	1.05-2	1.2.5-2					0.97-1i	0.76-1i			0.91	1.03	1.07	1.05
Pakistan	0.87-1	0.82-1	0.97-1	1.04-3	1.03-3			0.83-2	0.65-2			0.86	0.86	0.87	0.87-1
Sri Lanka								1.01-1	0.98-1			1.06-1	0.99-1	1.04-1	1.38
Tajikistan	0.99-2	0.97-2	0.76-2									0.87-2	0.99-2		0.76-2
Turkmenistan	1.00	0.98	0.74										0.98	0.98	0.82
Uzbekistan								1.00-1	1.00-1			0.94	0.99	0.99	0.83
Eastern and South-eastern Asia															
Brunei Darussalam						1.23-1	1.07-1	1.00-1i	0.98-1i			1.01	1.01	1.03	1.36
Cambodia				1.41	1.16	1.31-4	0.83-4	1.01-4	0.87-4			1.04	0.97		0.94
DPR Korea	1.05-3	1.00-3	1.13-3					1.00-1	0.97-1			1.00	1.01	1.01-4	0.51-1
Hong Kong, China				1.01-3	1.00-4	1.10-1	1.03-1					1.04	1.05	0.98	1.10
Indonesia	1.02-2	1.06-2	0.99-2		1.03-4	1.31-1	1.13-1	1.00-1	0.97-1			0.90-1i	0.97-1	1.02-1	1.13-1
Japan															
Lao PDR	0.98-2	0.96-2	0.94-2	1.33	1.08			0.96-4	0.88-4			1.03	0.97	0.94	1.10
Macao, China				1.00-3		1.06-1	1.00-1	1.00-3	0.97-3			0.97	0.98	1.00	1.27
Malaysia				1.24	1.10	1.23-1	1.07-1	1.00-1	0.97-1			1.03	1.01-2	1.07	1.23
Muapmar	1.01-1	1.03-1	1.12-1					1.UL-1i	1.00-1i			0.99	0.98	1.01	1.29-1
Philippines	1.05-3	1.15-1	1.11-1	1.21	1.02	1.34-1		1.00-4	1.00-4			0.99	0.90-1	1.09	1.23-1
Republic of Korea			1.03-3		1.00	1.08-1	1.01-1					1.00-1	1.00-1	0.99-1	0.80-1
Singapore				1.02-3	1.01-4	1.07-1	1.03-4	1.00-1	0.97-1	0.96-4	0.93-4		1.00-1i	0.99-1i	1.13-ıi
Thailand	1.00	1.10	1.14			1.38-1	1.16-1	1.01-1	0.97-1			1.00	1.00	0.97	1.29-3
Timor-Leste	1.10-3	1.10-3	1.10-3					1.03-ıi	0.89-1i			1.04	0.98	1.09	
Viet Nam				1.05	1.00	1.11-4	1.04-4	1.00-ıi	0.97-1i			1.02	1.02		1.20-3

							LOCATIO	N/WEALTH								
					I	F								G		
Dis	parity in prin	nary comple	tion	Dispari	ty in lower se	condary con	npletion	Dispari	ity in upper se	condary cor	npletion	Wealth	disparity in	minimum pro	oficiency	
Adjusted p	oarity index	% of p comp	pleting	Adjusted p	oarity index	% of p comp	oorest oleting	Adjusted (	parity index	% of p comp	oorest leting	End of	primary	End o seco	f lower ndary	a
cation	ealth			cation	ealth			cation	ealth			ading	athematics	ading	athematics	ountry cod
2	Ä	٤	ш	2	Ä	×	ш.	2	Ä	٤	ш	Re	Z	Re	×	Ů
							4.	.5.1 n19								
							21	515							1	
0.98	0.92			0.83	0.57			0.69	0.37					0.61-4	0.49-4	DZA
1.00-1	0.99-1	99-3	99-3	0.98-1	0.98-1	85-3	99-3	0.84-1	0.91-1	45-3	55-3		0.94-4		0.99-4	ARM
												0.90-з				AZE
												0.72-3	0.66-4		0.69-4	BHR
1.00-2	0.99-2			1.01-2	0.99-2			0.92-2	0.73-2				0./9-4	0.58-1		EGV
1.00-1	1.00-1	100-1	100-1	0.96-1	0.93-1	95-1	91-1	0.84-1	0.69-1	 54-1	52-1	0.97-3	0.65-4	0.39-1	0.40-1	GEO
0.87-1	0.58-1	62-1	45-1	0.76-1	0.32-1	26-1	19-1	0.80-1	0.24-1	15-1	11-1					IRQ
1.00-1	1.00-1			1.00-1	0.99-1							0.77-3		0.57-1	0.53-1	ISR
1.01-1	0.92-1	88-1	93-1	1.02-1	0.66-1	64-1	66-1	0.88-1	0.31-1	20-1	31-1			0.60-1	0.52-1	JOR
													0.50-4	0.25-1	0.44-4	LBN
																LBY
													0.32-4	0.33-1	0.27-4	MAR
												0.79-з	0.95-4		0.70-4	OMN
1.00	0.99			0.97	0.87			0.96	0.74							PSE
												0.76-3	0.38-4	0.40-1	0.70-4	SAU
																SDN
																SYR
0.93-1	0.89-1	85-1	92-1	0.72-1	0.55-1	49-1	56-1	0.52-1	0.30-1	17-1	32-1			0.34-4	0.31-4	TUN
													0.63-4	0.71-1	0.52-4	
																YEM
0.68-4	0.60-4	56-4	32-4	0.56-4	0.41-4	36-4	13-4	0.46-4	0.29-4	21-4	5-4					AFG
0.99	0.77	62	79	0.95	0.52	38	49	0.78	0.24	16	8					BGD
																BTN
0.97-3	0.82-3	81-3	80-3	0.92-3	0.62-3	62-3	56-3	0.65-3	0.18-3	18-3	9-3					IND
1.00-4	1.00-4	 100-4		1.00-4	0.99-4	 100-4	 99-4	0.96-4	0.90-4	 88-4	 89-4	1.00-3	0.82-4	0.56-1	0.75-1	KAZ
1.00-1	1.01-1	100-1	100-1	0.99-1	0.97-1	96-1	97-1	0.92-1	0.81-1	74-1	81-1					KGZ
0.98-2	0.98-2	96-2	97-2	0.93-2	0.86-2	75-2	89-2	0.49-2	0.34-2	16-2	22-2					MDV
	0.86-3	83-3	81-3		0.62-3	56-3	59-3									NPL
0.68-1	0.31-1	39-1	19-1	0.59-1	0.15-1	22-1	4-1	0.44-1	0.03-1	3-1	1-1	0.72-3	0.72-3			PAK
1.00-2	0.99-2	 99-2	96-2	0.98-2	0.96-2	95-2	95-2	0.93-2	0.82-2	77-2	55-2					TJK
1.00	1.00	98	99	1.04	1.08	98	97	0.30	0.17	86-3	95-3					TKM
																UZB
														0.40-1	0.47-1	BRN
														0.22-4	0.19-4	KHM
0.55-3	0.90-3			0.00-3	0.09-3			0.00-3	0.75-3							PRK
												1.00-3	0.96-4	0.89-1	0.89-1	HKG
0.97-2	0.91-2	87-2	94-2	0.89-2	0.68-2	64-2	69-2	0.68-2	0.35-2	31-2	32-2		0.44-4	0.39-1	0.37-1	IDN
																JPN
0.85-2	0.64-2	6/-2	58-2	0.56-2	0.19-2	21-2	12-2	0.36-2	0.07-2	5-2	4-2					LAO
												0.55-3		0.45-1	0.30-1	MYS
0.99-1	0.97-1	95-1	98-1	0.89-1	0.84-1	79-1	90-1	0.68-1	0.53-1	44-1	61-1					MNG
0.91-3	0.70-3	64-3	65-3	0.47-3	0.18-3	18-3	9-3	0.31-3	0.04-3	1-3	2-3					MMR
0.98-1	0.81-1	71-1	89-1	0.92-1	0.54-1	40-1	68-1	0.89-1	0.51-1	42-1	56-1			0.11-1		PHL
								0.99-3	0.93-3				0.93-4	0.82-1	0.28	KOR
0.99	1.00	 96	 99	0.95	0.73	63	 74	0.82	0.49	33	49	0.52-3	0.00-4	0.03-1	0.20-4	THA
0.82-3	0.62-3	57-3	63-3	0.63-3	0.37-3	33-3	35-3	0.48-3	0.23-3	18-3	20-3					TLS
														0.84-4	0.78-4	VNM

								GENDER								
		Α				В		.	с		D		1	E		
				G	PIA in minim	um proficien	icy									
		GPIA in				End o	flower	GP	IA in	GPIA i	n adult					
		completion		End of	primary	seco	ndary	litera	cy rate	profi	ciency	G	PIA in gross e	enrolment ra	tio	
		_	_		tic		tic				_	Ð		_		
	Ē	r idary	r idary	ē	ema	бu	ema	_	s s	5	eracy	rima	Ē	ıdarı	Ð	
Country or torritory	rima	owe	lppe	teadi	Math	teadi	Math	outh	dult	itera	l m	re-p	rima	ecor	ertia	
Country of territory SDG indicator	<u> </u>	~	2%	<u> </u>		451		_			-	-	<u> </u>		-	
Deference year						4.J.1		2019								
Reference gear					1			2013			1					
Oceania					0.07	1 11	0.00					0.00	1.00	0.04	1.20	
Cook Islands					0.97-4	1.11-1	0.99-1					1 10	0.96	1.04	1.20-1	ł
Fiji								1.00-2	1.00-2			0.97	0.96			Ì
Kiribati	1.04	1.22	1.38										1.07-2			ľ
Marshall Islands												1.01	0.96	1.07	1.11	
Micronesia, F. S.												0.99	0.96			ļ
Nauru												1.03	0.99	1.03		ł
New Zealand				1.05-3	0.99-4	1.11-1	0.99-1					0.98-1	1.01-1	1.06-1	1.32-1	ł
Palau								1.01-4	1.00-4			0.04	1.15	1.00		ľ
Papua New Guinea	1.02-1	1.00-1	0.89-1									0.99-3	0.91-3	0.73-3		Ì
Samoa								1.01-ı	1.00-1i			1.06	1.01	1.09-3	1.46	ĺ
Solomon Is												1.02	0.99			ļ
Tokelau												0.88	1.03	0.99		
longa	1.00	1.10	1.14					1.00-1i	1.00-1i			1.06-4	0.99-4	1.03-4		ł
Vanuatu								 1 01-1i				0.89	0.92	1.25		ł
Latin America on data Carit have								1.01-1	0.50-1			0.57-4	0.57-4	1.05-4		
Latin America and the Caribbean												1.08	0.99	0.97		
Antiquia and Barbuda									1.01-4			1.10-1	0.99-1	0.96-1		ł
Argentina						1.11-1	0.78-1	1.01-1	1.00-1			1.02-1	1.00-1	1.04-1	1.40-1	ľ
Aruba								1.00-1i	1.00-1i						1.48-3	
Bahamas																
Barbados												0.99	0.96	1.04		ļ
Belize	1.01-3	1.15-3	1.06-3									1.03	0.97	1.02	1.40	
Bolivia, P. S. Brozil	1.02	1.02	1.02	1.13-4	0.84-4			1.00-4	0.92-4			1.02	0.99	0.98		ł
British Virgin Islands	1.05-1	1.00-1	1.15-1			1.20-1	0.00-1	1.00-1	1.00-1			1.11-2	1.00-1	1.10-3	1.44-1	h
Cayman Islands												0.98-1	0.97-1	1.01-1		ľ
Chile	1.02-2	1.02-2	1.06-2			1.13-1	0.74-4	1.00-2	1.00-2	0.90-4	0.70-4	0.98-1	0.97-1	1.00-1	1.14-1	
Colombia	1.02-1	1.06-1	1.07-1			1.07-1	0.75-1	1.00-1	1.00-1			1.02	0.97	1.05	1.14	ļ
Costa Rica	1.01-1	1.05-1	1.09-1			1.11-1	0.80-1	1.00-1i	1.00-1i			1.00	1.01	1.07	1.18	ļ
Cuba	1.00	1.02	1.09									0.99	0.97	1.01	1.37-1	
Dominica												1.06				ł
Dominican Republic	1.06-1	1.11-1	1.24-1			1.37-1	0.94-1	1.00-3	1.00-3			1.00 1.02i	0.95 <sub>1</sub>	1.08	1.44-2j	Ì
Ecuador	1.00-1	1.01-1	1.05-1			1.09-4	0.71-4	1.00-2	0.98-2	0.96-2	0.77-2	1.05-1	1.02-1	1.03-1	1.13-1	ľ
El Salvador	1.04-1	1.00-1	1.04-1					1.00-1	0.96-1			1.02-1	0.97-1	0.99-1	1.12-1	
Grenada												1.03-1	0.98-1	1.03-1	1.20-1	ļ
Guatemala	0.95-4	0.87-4	0.91-4			1.15-4	0.84-4					1.02	0.98	0.96	1.15-4	ļ
uuyana Haiti	1.02	1.04	1.04													
Honduras	1.07-1	1.09-1	1.24-1			 1.11-4	0.66-4	1.03-1	1.00-1			1.04	1.00		1.28	ŀ
Jamaica												1.04		1.02	1.43-4	
Mexico	1.01-1	1.03-1	1.07-1			1.11-1	0.88-1	1.00-1	0.98-1	0.99-2	0.80-2	1.03-1	1.01-1	1.08-1	1.04-1	ĺ
Montserrat												1.15	1.12	1.08		
Nicaragua								1.04-4i	1.00-4i							ļ
Panama	1.01-1	1.07-1	1.11-1			1.16-1	0.82-1	1.00-1	0.99-1			1.02-2	0.98-2	1.05-2	1.36-3	ł
Paraguay	1.02	1.08	1.09			1.12-4	0.56-4	1.01-1	0.99-1		0.74 >	1.01-3				
Saint Kitts and Nevis	1.00-1	1.01-1	1.01-1					1.00-1	0.94-1	0.09-2	0.74-2	0.80-3	0.90	1.03-3	1.50-4	ŀ
Saint Lucia												1.02	1.02	0.97	1.53	
Saint Vincent/Grenadines												1.02-1	0.99-1	1.03-1	1.40-4	ľ
Sint Maarten															1.70-4	
Suriname	1.11-1	1.28-1	1.29-1					1.00-1i	0.97-1i			1.05	1.00	1.24-4		ļ
Trinidad and Tobago				1.11-3		1.28-4	1.16-4					1.02				ļ
IURKS and Calcos Islands												1.15-1	1.00-1	1.03-1		
Venezuela B R	1.02-1	1.13-1	1.2/-1			1.1/-1	0.93-1	1.01-1	1.01-1			1.00-1	0.99-1	1.10-1		ł
								1.91.9	2.00 5			2.01 2	0.00 2	2.37 2		L.

								LOCATIO	N/WEALTH				I				
							F .								G	<i>.</i>	
	Dis	parity in prin	nary comple % of r	tion	Dispari	ty in lower se	condary con % of n	npletion	Dispari	ty in upper se	condary cor % of n	npletion	Wealth	disparity in I	minimum pro	oficiency f lower	
Ad	ljusted p	arity index	com	oleting	Adjusted p	oarity index	comp	leting	Adjusted p	oarity index	comp	leting	End of	primary	seco	ndary	
	cation	ealth			cation	ealth			cation	ealth			ading	athematics	ading	athematics	untry code
	2	Å	٤	ш	2	Ň	¥	<u>ц</u>	의 [1	Ň	۶	ш	Re	×	Re	۶	Ů
								4.	)19								
	-																
														0.52-4	0.76-1	0.71-1	AUS
																	EUK FJI
	0.97	0.92	85	93	0.85	0.69	52	75	0.28								KIR
_																	MHL
																	NRU
_													0.83-3	0.41-4	0.75-1	0.70-1	NZL
																	PLW
0	.74-1	0.45-1	41-1	36-1	0.74-1	0.33-1	31-1	20-1	0.41-1								PNG
																	SLB
																	TKL
	1.00	1.00	97	97	0.95	0.74	88	86	0.99	0.71	5	23					TON
																	VUT
																	AIA
															0.36-1	0.20-1	ARG
																	ABW
_																	BRB
0	.99-3	0.92-3	88-3	94-3	0.68-3	0.38-3	23-3	44-3	0.69-3	0.25-3	16-3	21-3					BLZ
0	1.03 .96-1	0.96	95-1 89-1	9/-1 94-1	0.87-1	0.90	8/-1 68-1	88-1 80-1	0.68-1	0.67	59-1 38-1	55-1 47-1			0.45-1	 0.26-1	BOL
																	VGB
0			93_2	96-2		0.94-2	 91-2	95_2								0.28-4	CYM
0	.95-1	0.98-1	84-1	91-1	0.79-1	0.76-1	55-1	67-1	0.64-1	0.62-1	49-1	52-1			0.44-1	0.34-1	COL
1	00-1	0.96-1	95-1	95-1	0.93-1	0.63-1	48-1	55-1	0.88-1	0.38-1	22-1	29-1			0.50-1	0.37-1	CRI
		1.00			0.96				0.79								COR
																	DMA
0	.98-1 .99-1	0.94-1	88-1 98-1	96-1 98-1	0.93-1	0.90-1	78-1 83-1	86-1 85-1	0.72-1	0.56-1	40-1 55-1	47-1			0.22-1	0.12-1	DOM ECU
0	.92-1	0.86-1	83-1	85-1	0.76-1	0.67-1	64-1	58-1	0.58-1	0.35-1	28-1	32-1					SLV
0				54 4											0.25.4		GRD
	0.99	0.97			0.97	0.91			0.97	0.93							GUY
0	.61-2	0.26-2	17-2	24-2	0.46-2	0.12-2	7-2	9-2	0.30-2	0.02-2	1-2	1-2					HTI
0	.91-1	U. / b - 1 	/ 5-1	/0-1	0.45-1	0.32-1	Z5-1 		0.39-1	0.20-1		10-1 			0.35-4	0.20-4	JAM
0	.99-1	0.97-1	94-1	97-1	0.90-1	0.82-1	78-1	78-1	0.67-1	0.47-1	38-1	37-1			0.47-1	0.44-1	MEX
_																	MSR
0	.94-1	0.91-1	89-1	91-1	0.83-1	0.72-1	66-1	75-1	0.70-1	0.45-1	39-1	41-1			0.27-1	0.15-1	PAN
0	0.97	0.92	86-1	93-1	0.82	0.74	67-1 82 -	69-1 80 -	0.65	0.48	41-1	36-1			0.34-4	0.15-4	PRY
0					0.09-1	0.00-1	02-1	00-1	0.02-1								KNA
																	LCA
																	VCT SXM
0	.88-1	0.69-1	60-1	77-1	0.67-1	0.30-1	16-1	32-1	0.49-1								SUR
													0.82-3		0.60-4	0.51-4	TTO
1		0.98-1	96-1	97-1	0.95-1	0.53-1	46-1	58-1	0.75-1	0.18-1	 10-1	 17-1			0.46-1	0.39-1	URY
																	VEN

								GENDER							
		Α				в			с		D		1	E	
				G	PIA in minim	um proficien	cu								
		GPIA in				Endo	flower	GPI	IA in	GPIA i	n adult				
		completion		End of	primary	seco	ndary	literad	cy rate	profi	ciency	GI	PIA in gross e	enrolment rat	io
					<u>.</u>		<u>.</u> 2					5			
	يط ل	lary	lary	5	mat	5	smat			5	acy	mar	يد ت	dary	7
	imar	wer	ond	adin	athe	adin	athe	ŧ	hults	terac	mer	e-pri	imar	conc	rtiar
Country or territory	2	s E	L s	ž	Σ	ž	Σ	۶	Ac	5	ž	2	<u>r</u>	Š	Te
SDG indicator						4.5.1									
Reference year								2019							
Europe and Northern America															
Albania	1.02-2	1.01-2	1.04-2			1.35-1	1.06-1	1.01-ıi	0.99-1i			1.00	1.03	1.01	1.36
Andorra															
Austria	1.01-2	1.02-2	0.98-2	1.01-3		1.13-1	0.99-1					0.99-1	0.99-1	0.96-1	1.16-1
Belarus	1.00					1.13-1	0.99-1	1.00-1i	1.00-1i			0.96-1	1.00-1	0.99-1	1.16-1
Belgium	1.04-4	1.08-4	1.05-4			1.08-1	0.97-1					1.00-1	1.00-1	1.11-1	1.24-1
Bermuda												0.85-4	0.98-4	1.11-4	1.33-1
Bosnia and Herzegovina						1.30-1						0.97			1.33
Bulgaria				1.02-3	1.03-4	1.27-1	1.03-1					0.99-1	0.99-1	0.97-1	1.19-1
Canada			1.01-2	1.01-3	0.95-4	1.09-1	1.00-1						1.00-1	1.01-1	1.25-1
Croatia	1.00-2	0.99-2	1.00-2		0.92-4	1.10-1	0.98-1					0.99-1	1.00-1	1.05-1	1.28-1
Czechia	1.00-2	1.01-2	1.08-2	1.02-3	0.97-4	1.13-1	1.01-1	1.00-3	1.00-3			0.97-1	1.01-1	1.00-1	1.28-1
Estopia	1.00-2	1.00-2	1.19-2			1.11-1	1.01-1					0.98-1	1.00-1	1.00-1	1.27-1
Estonia	1.00-2	1.03-2	1.08 -2			1.07-1	1.00-1					0.99-1	1.00-1	1.03-1	1.34-1
Franco	1.00-2	1.00-2	1.05 -2	1.01-3	0.07.	1.11.	1.04-1					1.00	0.99-1	1.00	1.10-1
Germanu	1.01-2	1.02-2	1.03-2	1.02-3	0.97-4	1.11-1	1.00 1					0.99.1	1 01 1	0.94.1	1.20-1
Greece	1 00-2	1 01-2	1 01-2	1.01-3	0.50-4	1 22-1	1.00-1	1 00-1i	0.99_1i	1.05-4	0.94-4	1.01-1	1.01-1	0.95-1	1.00-1
Hungaru	0.99-2	0.98-2	1.01 2	1 01-3	0.99-4	1 12-1	0.95-4	1.00 1	0.55 1	1.03 4	1 01-2	0.97-1	0.99-1	1 00-1	1 18-1
Iceland	1.00-4	1.00-4	1.32-4			1.19-1	1.07-1					1.01-1	1.00-1	0.99-1	1.46-1
Ireland	1.00-4	1.01-4	1.04-4			1.07-1	1.00-1					0.99-1i	0.99-1i	1.12-1j	1.11-ıj
Italy	1.00-2	1.00-2	1.02-2	1.02-3	0.88-4	1.11-1	0.96-4	1.00-1j	1.00-1j			0.98-1	0.97-1	0.99-1	1.26-1
Latvia	1.00-4	1.01-4	1.11-4	1.00-3		1.16-1	1.00-1	1.00-1i	1.00-1j			0.99-1i	1.00-1i	0.99-1i	1.33-ıi
Liechtenstein												0.90-1i	1.01-1i	0.83-1i	0.56-1i
Lithuania	1.00-2	0.99-2	1.12-2	1.01-3	1.03-4	1.18-1	1.05-1			1.02-4	0.99-4	0.99-1i	1.00-1i	0.96-1i	1.27-1i
Luxembourg		0.97-2	1.00-2			1.13-1	0.97-1					0.98-1	0.98-1	1.02-1	1.13-1
Malta				1.11-3		1.26-1	1.03-4	1.01-1i	1.03-ıi			0.99	1.01	1.00	1.29
Monaco															
Montenegro	1.04-1	1.04-1	1.08-1			1.24-1	0.94-1	1.00-1i	0.99-1i			0.95	1.00	1.02	1.25
Netherlands	1.00-2	1.05-2	1.03-2	1.01-3	0.97-4	1.13-1	1.02-1					1.02-1	1.00-1	1.01-1	1.12-1
North Macedonia						1.41-1	1.09-1					1.02-1	1.00-1	0.98-1	1.24-1
Norway					0.99-4	1.16-1	1.05-1					1.00-1	1.00-1	0.96-1	1.32-1
Poland	1.00-2	1.00-2	1.02-2	1.01-3	1.02-4	1.11-1	1.02-1					1.00-1	0.99-1	0.9/-1	1.34-1
Portugal	1.01-2	1.00-2	1.10-2	1.01-3	0.9/-4	1.10-1	1.00-1	1.00-1i	0.98-1i			0.99-1	0.97-1	1.00-1	1.12-1
Republic of Moldova						1.20-1	1.02-1	1.00 .				1.00	0.971	1.00	1.20i
Romania Dussian Federation	1.00-2	1.00-2	0.98-2			1.22-1	0.98-1	1.00-1	0.99-1i			1.00-1	0.99-1	1.00-1	1.21-1
San Marino	1.00-1	1.00-1	1.01-1	1.01-3	1.00-4	1.12-1	1.00-1	1.00 -1	1.00-1			1.00	1 10	0.97-1	0.77
Sarbia								1.00-1	0.00 -1			1.00	1.10	1.01.	1.28
Slovakia	1 00 4	1 01 4	0.98 4	1 01 1	1.01-4	1 1 2 4	1.01-1	1.00-3	0.33-3			0.98 1	1.00	1.01	1 34 1
Slovenia	1.00-4	1.01-4	1.03-2	1.01-3	1.00-4	1 16-1	1.01-1			1.07-4	0.98-4	0.98-1	1.00-1	1.07-1	1 31-1
Spain	0.94-2	1.03-2	1.05-2	1.02-3	0.93-4	1.08-4	1.00-1	1.00-1	0.99-1	1.02 4	0.50 4	1.00-1	1.01-1	1.02-1	1.17-1
Sweden	1.00-2	1.00-2	1.06-2	1.01-3	1.00-4	1.11-1	1.02-1					0.99-1	1.02-1	1.06-1	1.37-1
Switzerland	1.00-4	1.00-4	0.96-4			1.12-1	0.99-1					0.98-1	0.99-1	0.95-1	1.03-1
Ukraine						1.16-1									
United Kingdom	1.00-4	1.00-4	1.10-4			1.07-1	0.97-1					1.00-1	1.00-1	1.03-1	1.27-1
United States	1.00	1.01	1.02	1.01-3	0.98-4	1.09-1	0.98-1			1.02-2	0.97-2	1.01-ıi	1.00-1i	0.99-1i	1.27-ıi

								LOCATIO	N/WEALTH				I				
ŀ					1	I	•		1					(	G		
ŀ	Dis	parity in prim	nary complet	tion	Dispari	ty in lower se	condary com	pletion	Dispari	ty in upper se	condary cor	npletion	Wealth	disparity in r	ninimum pro	ficiency	
l	Adjucted p	aritu indox	% of p	oorest	Adjusted a	aritu indax	% of po	oorest	Adjusted r	aritu indax	% of p	oorest	End of	nimaru	End of	f lower ndoru	
l	Aujusteu p	arity muex	comp		Aujusteu p	arity muex	compi	leting	Aujusteu p		comp		Elidoi	primary vo	Seco		e
					_				_					latic		atic	cod
l	Ition	£			Ition	£			Ition	ŧ			ding	them	ding	them	ntry
l	Loca	Wea	×	L.	Loca	Wea	×	ш	Loca	Wea	×	ш	Rea	Mat	Rea	Mat	Cou
I.								4.	5.1		1			1	1	1	
l								20	019								
	0.07 .	0.04 .	01 .	96 .	0.00 .	0.88 5	80 .	86.5	0.85 .	0.62 .	60 .	60 .			0.51 .	0.75 .	AL R
1	0.57-2	0.54-2	51-2	50-2	0.33-2	0.00-2	03-2	00-2	0.03-2	0.02-2	00-2	00-2			0.31-1	0.75-1	
Í	1 00-2				0.99-2	0.85-2			1 11-2	0.75-2			0.92-3		0.70-1	0.70-1	AUT
1	1.00	1.00											0.52 5		0.61-1	0.54-1	BLR
ĺ					0.94-4	0.93-4			0.94-4	0.84-4					0.68-1	0.67-1	BEL
1																	BMU
1															0.50-1		BIH
]													0.91-3	0.70-4	0.40-1	0.45-1	BGR
									0.95-2	0.94-2			0.89-3	0.70-4	0.85-1	0.81-1	CAN
	1.01-2	1.00-2			1.01-2	1.00-2			0.98-2	0.90-2				0.83-4	0.80-1	0.68-1	HRV
Į.	1.01-2	0.98-2			1.01-2	0.96-2			1.00-2	0.81-2			0.86-3	0.69-4	0.68-1	0.66-1	CZE
1	1.00-2	0.99-2			1.01-2	0.98-2			0.86-2	1.11-2					0.78-1	0.80-1	DNK
Į.	1.00-2	1.00-2			0.98-2	1.02-2			0.97-2	1.07-2					0.90-1	0.88-1	EST
ł	1.00-2	1.00-2			1.00-2	1.00-2			0.84-2	1.03-2			0.98-3	1.03-4	0.85-1	0.80-1	FIN
Į.	1.00-2	0.98-2			1.01-2	0.96-2			1.02-2	0.80-2			0.92-3	0.52-4	0.70-1	0.64-1	FRA
ł													0.82-3	0.68-4	0.70-1	0.68-1	DEU
ł	1.00-2	1.00-2			1.01-2	0.97-2			0.94-2	0.96-2					0.63-1	0.57-1	GRC
ł	1.01-2	1.00-2			1.02-2	0.99-2			0.92-2	0.95-2			0.94-3	0.60-4	0.58-1	0.53-4	HUN
ł	1.00-4	1.00-4			1.00-4	1.00-4			0.65-4	0.04					0.73-1	0.78 -	ISL
í	1.00-4	1.00-4			1.01-4	1.00 -4			1.05-4	0.94-4				0.76 .	0.72 .	0.78-1	IRL
ł	1.00-2	1.00-2			1.00-2	0.00 /			0.01.	0.70-2			0.96-3	0.70-4	0.72-1	0.04-4	
í	1.00-4	1.00-4			1.01-4	0.33-4			0.34-4	0.33-4			0.50-3		0.70-1	0.70-1	LIF
1	0.98-2	0.99-2			0.96-2	0.98-2			0.83-2				0.91-3	0.79-4	0.68-1	0.65-1	ITU
Í	0.501	0.55 1			1.04-2	0.90-2			1.10-2	0.75-2			0.515	0.75 4	0.58-1	0.59-1	LUX
1													0.93-3		0.64-1		MLT
1																	МСО
1	1.02-1	0.89-1	84-1	94-1	1.03-1	0.78-1	70-1	86-1	1.00-1	0.54-1	52-1	58-1			0.63-1	0.60-1	MNE
1						0.93-2				1.02-2			0.92-3	0.94-4	0.73-1	0.78-1	NLD
	1.02	0.97	97	98	1.03	0.84	79	88	1.02	0.56	63	49			0.45-1	0.39-1	MKD
														0.79-4	0.81-1	0.78-1	NOR
	1.00-2	1.00-2			0.98-2	0.97-2			0.97-2	1.02-2			0.96-3	0.81-4	0.81-1	0.78-1	POL
Į.	1.01-2	1.00-2			1.00-2	0.86-2			0.93-2	0.79-2			0.97-3	0.85-4	0.71-1	0.65-1	PRT
															0.44-1	0.38-1	MDA
Į.	0.99-2	0.98-2			0.98-2	0.96-2			0.83-2	0.63-2					0.47-1	0.40-1	ROU
ł	1.00-1				1.00-1				0.98-1	1.00-1			0.99-3	0.96-4	0.79-1	0.76-1	RUS
ł																	SMR
		0.97	100	93		0.93	95	92		0.64	63	59		0.89-4	0.62-1	0.60-1	SRB
1	1.00-4	0.98-4			1.00-4	0.9/-4			1.02-4	0.00			0.59-3		0.56-1	0.5/-1	SVK
ł		1.00-2			0.07	1.00-2				0.90-2			0.99-3	0.99-4	0.77	0.69	SVIN
1	1.00 -				1.00 -	1.00 -			1.04 -	0.01-2			0.90-3	0.57-4	0.77 -	0.00-1	CIVE
ł	1.00-2	1.00-2			1.00-2	1.00-2			1.04-2	0.90-2			0.32-3	0.03-4	0.68 -	0.75-1	CHE
1	1.00-4	1.00-4			1.01-4	1.00-4			1.11-4	0.33-4					0.00-1	0.70-1	LIKR
ł	1.00-4	1.00-4			1.00-4	1.00-4			1.02-4	0.90-4					0.81-1	0.76-1	GBR
1		1.00			2.00 4	0,99			2.52 4	0.92			0.96-3	0.71-4	0.76-1	0.62-1	USA

#### TABLE 6: SDG 4, Target 4.7 - Education for sustainable development and global citizenship

By 2030, ensure that all learners acquire the knowledge and skills needed to promote sustainable development, including, among others, through education for sustainable development and sustainable lifestyles, human rights, gender equality, promotion of a culture of peace and non-violence, global citizenship and appreciation of cultural diversity and of culture's contribution to sustainable development

			A		В		c	
	Exten and e	t to which globa ducation for su are main	al citizenship edu stainable develo Istreamed	ucation pment	viding HIV/AIDS		% of students an youth with understanding o	nd of
	Education policies/ frameworks	Curriculum	In-service teacher training	Student assessment	% of schools pro life skills-based education	HI V/AIDS and sexuality	Global citizenship	Scientific literacy
SDG indicator		4.	7.1		4.7.2	4.1	7.4	4.7.5
Reference year		20	017			20	19	
Region		% of c	ountries			Me	dian	
World	<b>100</b> i	84i	<b>86</b> i	<b>86</b> i				
Sub-Saharan Africa						36 <sub>1</sub>		
Northern Africa and Western Asia	100 <sub>i</sub>	89i	100 <sub>i</sub>	100 <sub>i</sub>	100 <sub>i</sub>		33 <sub>i</sub>	
Northern Africa	100 <sub>i</sub>	100 <sub>i</sub>	100 <sub>i</sub>	100i	-1			
Western Asia	100i	86i	100i	100i	100i		32i	
Central and Southern Asia	100i	50i	83i	100i	100i			
Central Asia	100i	-i	100i	100i	61i			
Southern Asia	100i	75i	75i	100i	100i	23i		
Eastern and South-eastern Asia	100 <sub>i</sub>	71 <sub>i</sub>	83 <sub>i</sub>	86i	98 <sub>i</sub>		15 <sub>i</sub>	
Eastern Asia	100 <sub>i</sub>	100 <sub>i</sub>	100 <sub>i</sub>	67i	96i		14 <sub>i</sub>	
South-eastern Asia	100 <sub>i</sub>	50i		100 <sub>i</sub>	100 <sub>i</sub>	29i		
Oceania					60 <sub>i</sub>			
Latin America and the Caribbean		71 <sub>i</sub>						
Caribbean					100i			
Central America	100i	75i	100i	50i				
South America	100i	57i	67i	71i				
Europe and Northern America	100 <sub>i</sub>	100 <sub>i</sub>	88i	93i				56i
Europe	100	100	88i	92i				56i
Northern America		100i	100i	100i	100i		29	
						21		
Low income						31i		
Middle income	100i	/6i	89i	84i				
Lower middle								
Upper middle	100	/8i	90i	82i				
High income	100i	97i	87i	84i				

A Extent to which (i) global citizenship education and (ii) education for sustainable development (including climate change education) are mainstreamed at all levels in (a) national education policies; (b) curricula; (c) teacher education; and (d) student assessment [Source: UNESCO, 2019].

(Low = reflected not at all or little/not included in student assessment. Medium = reflected somewhat. High = fully reflected/included in student assessment.) B Percentage of lower secondary schools providing life skills-based HIV/AIDS education.

C Percentage of students and youth with adequate understanding of issues relating to global citizenship and sustainable development [Sources: UIS, UNAIDS].

D Percentage of primary schools with water, sanitation and hygiene (WASH): basic drinking water, basic (single-sex) sanitation or toilets, and basic handwashing facilities.

E Percentage of public schools with electricity, and computers or internet used for pedagogical purposes.

F Percentage of public primary schools with access to adapted infrastructure and materials for students with disabilities.

G Level of bullying.

H Level of attacks on students, teachers or institutions [Source: Global Coalition to Protect Education from Attack].

I Internationally mobile students, inbound and outbound numbers enrolled (thousand) and inbound and outbound mobility rates (as a percentage of total tertiary enrolment in the country).

J Volume of official development assistance flows (all sectors) for scholarships (all levels) and imputed student costs, total gross disbursements (million constant 2018 US\$).

K Region totals include flows unallocated to specific countries. World total includes flows unallocated to specific countries or regions.

Note: ICT = information and communication technology.

Source: UIS unless noted otherwise. Data refer to school year ending in 2019 unless noted otherwise.

Aggregates represent countries listed in the table with available data and may include estimates for countries with no recent data.

(-) Magnitude nil or negligible.

(...) Data not available or category not applicable.

(± n) Reference year differs (e.g. -2: reference year 2017 instead of 2019).

(i) Estimate and/or partial coverage.

#### SDG 4, Means of implementation 4.a - Education facilities and learning environments

By 2030, build and upgrade education facilities that are child, disability and gender sensitive and provide safe, non-violent, inclusive and effective learning environments

#### SDG 4, Means of implementation 4.b - Scholarships

By 2020, substantially expand globally the number of scholarships available to developing countries

		D			E		F	G	н			I			J
	% of scho	ols with WAS	H facilities	% of s peda	chools with agogical purp	ICT for poses	:h adapted nd materials for isabilities	_	on education	Mob rate	Internation tertiary pility (%)	nally mobile students Nun (00	nber DO)	Official de assistan (000	velopment ce, in US\$ ,000)
	Basic drinking water	Basic sanitation or toilets	Basic handwashing	Electricity	Internet	Computers	% of schools wit infrastructure a students with d	Level of bullying	Level of attacks	punoqui	Outbound	Inbound	Outbound	Scholarships	Imputed student costs
				4.a.1				4.a.2	4.a.3					4.	b.1
					2019							20	19		
				Median					-	Me	dian		Su	im	
	<b>78</b> i	78i	66-1i	<b>73</b> i	<b>40</b> i	<b>48</b> i				2-1	<b>2</b> -1	5,571-1	5,571-1	<b>1,035</b> i	<b>2,549</b> i
	44-2i	64i	37-2i	31 <sub>i</sub>	4i	13 <sub>i</sub>				2-1j	5-1i	138-ıi	390-1	154	340
	87-2i	90-2i	93-2i	88-2i	80i	85i				3-1j	3-1i	514-1	658-1	151 <sub>i</sub>	750i
	85-2i	89-2i	94-2i	86-2i	77i	86i				1-1i	2-1i	86-1i	168-1	71	391
	89i	90i	92i	91i	83i	84i	100i			3-1	4-1	428-1	490-1	80i	359i
	84	79	52-1i	56-1	12	22				0.2-1	2-1	104-1	880-1	94	404
	95-1i	92i	93i	100	70i	96i	17i			2-1	13-1	34-1	196-1	20	28
	84	78	51-ıi	55-1	11	20				0.2-1	2-1	70-1	684-1	74	376
	79i	74i	82i	90i	77i	69i	59i			1-1	2-1	694-1	1,504-1	164i	517i
	97	97	96	97	95	95				1-1	2-1	501-1	1,180-1	35i	383i
	67i	59i	73i	85i	66i	53i	23i			1-1i	2-1i	192-ıi	324-1	129i	134i
	87-2i	86-2i	95-2i	92-2i	62-2i	75-2i	18 <sub>i</sub>			26-1	2-1	505-1	31-1	36i	2 i
	100 <sub>i</sub>	83-2i	100 <sub>i</sub>	89-1i	43-1i	61-1i	32i			1-1	1-1	190-1	356-1	67i	184i
	100i	100i	100i	100 i	99i	100i	62i				20i		35i	11i	10i
	88i	75i		98i	23	38i	30 i			1i	2	10i	58i	14	46
	67i			96	43i	75i				0.5i	2	148i	260i	42i	128i
	98i	99i	96i	100 <sub>i</sub>	100 <sub>i</sub>	100 <sub>i</sub>				7-1	2-1	3,426-1	1,135-1		
	97-2i	99i	97-2i	100-2i	100 <sub>i</sub>	100 <sub>i</sub>				8-1	3-1	2,214-1	1,001-1		
			100i	100i	100i	100i	100i			6-1	1-1	1,212-1	134-1		
	49- <sub>2i</sub>	62-2j	43-2j	32-ıj		2 <sub>1</sub>				1	5-1j	47i	273-1	99	293
	78i	77i	63-ıi	73 <sub>i</sub>	33i	42 <sub>1</sub>				1-1	2-1	1,380-1	3,351-1	601	2,164
٦	78i	76i	53-1i	55-1i	16;	27i				0.5-1	2-1	275-1	1,360-1	350	1,038
٦	79i	79i	81i	94i	62-1i	67i				1-1	2-1	1,105-1	1,991-1	252i	1,126
	96-2i	97-2i	95-2i	98-2i	100i	100i				8-1	2-1	4,141-1	1,331-1		

	Extent to wh for sust	ich global citizer tainable develop	A nship education a ment are mainst	and education reamed	viding HIV/AIDS 📼	% oʻ wit	C f students and y th understanding	outh g of
Country or territory	Education policies/ frameworks	Curriculum	In-service teacher training	S tudent assessment	% of schools prov life skills-based F education	HIV/AIDS and sexuality	Global citizenship	Scientific literacy
SDG indicator		4.	7.1		4.7.2	4.3	7.4	4.7.5
Reference year				20	17			
Sub-Saharan Africa								
Angola						32-4		
Benin								
Botswana						47-3	19-4	
Burkina Faso					21			
Burundi	High	Low	Low	High	100-2			
Cabo Verde					100-2			
Central African Republic	High	rign	Medium	High		39-1		
Chad	Medium	Medium	Medium	High				
Comoros	Wedlull	Medium	Wedlull	ngn				
Congo						33-4		
Côte d'Ivoire	Medium	High	Medium	High		27-4		
D. R. Congo	Medium	Medium	Low	High	4			
Djibouti								
Equat. Guinea								
Eritrea								
Eswatini					100-1			
Ethiopia	Medium	High	Medium	High		31-3		
Gabon								
Gambia								
Ghana								
Guinea						22-1		
Guinea-Bissau								
Kenya								
Lesotho								
Liberia								
Madagascar						47 - 4		
Mali	High	High	Medium	High		16 1		
Mauritania	nign	riigii	Medium	ngn		58-2		
Mauritius	High	Low	Medium	Low		5-0-3		
Mozambigue	nign	LOW	medium	LOW		31-4		
Namibia	Medium	High	Medium	Low				
Niger					100	22-3		
Nigeria						41-1		
Rwanda					100	64-4		
Sao Tome and Principe					100-2			
Senegal	High	High	Medium	High		28-3		
Seychelles					87			
Sierra Leone					50			
Somalia								
South Africa						46-3	19	
South Sudan								
Togo								
Uganda						46-3		
United Republic of Tanzania								
	High	High	High	High		42-1		
ZIMDADWe						40-4		

	D		E			oted terials	G	G H I E Internationally mobile E tertiary students					J Official development		
% of schoo	ols with WASI	H facilities	% of schools with ICT for pedagogical purposes			vith adap and mat rith disal	Ð	(s on edu	Mobility rate (%)		Number (000)		assistance, in US\$ (000,000)		
Basic drinking water	Basic sanitation or toilets	Basic handwashing	Electricity	Internet	Computers	% of schools w infrastructure for students w	Level of bullyi	Level of attach	Inbound	Outbound	Inbound	Outbound	Scholarships	Imputed student costs	Country code
4.a.1 4.a.2 4.								4.a.3					4.	b.1	
				2019					2019						
20-з			22-3	3-3	7-3					6-3		13-ıi	3	2	AGO
40			28				49-3	Sporadic+1	5-1	6-1i	6-1	8-1i	2	11	BEN
							95-4	Affected+1	2	5-2	1	3-1i	1	0.1	BWA
59	60	29	21	0.2	6	38		Very heavy+1	2	5-1i	3	6-1i	2	7	BFA
39	35	20	9	- 10	-	-		Affected+1	5-1	8-1i	2-1 0.2	4-1i	1	2	BDI
34 -	9Z-1 20 -	/ 0 - 1	21	10-1	42-1				1-1	3 Z - 1i	0.2-1	4-1i 26 si	11	64	CMP
J4-2	55-2		4_2						J-1	0-11	5-1	20-11	2	2	CAF
15-1		20	3					Affected+1		14-4		6-1	1	4	TCD
			41-2	8-2	31-2							6-1i	5	5	COM
			24-1		12-1					18-2		10-ıj	6	12	COG
42		31	46					Affected+1	2-2	6-2	7	14-ıi	5	23	CIV
4		4	9-4	4	4	4		Very heavy+1	0.4-3	2-3	2-3	12-1i	5	5	COD
91+1	96+1	91+1	95-2									2-1i	1	3	DJI
												1-1i	0.5	0.2	GNQ
	26-1	3-1	29-1					Sporadic+1		20-3		2-1i	2	1	ERI
79-2	100-3		100-1	16-3	15-3	12-3					4	2-1i	0.5	0.1	SWZ
								Heavy+1				8-1i	13	6	ETH
												7-1i	3	14	GAB
84-1	84-1		34		21							2-1i	1	0.4	GMB
33-1	91-2	35-1	25-1	8-1	3-1			Sporadic+1	1	4-1i	7	16-1i	10	13	GHA
25-3		85-3	14-3	3	3			Affected+1			0.4-2	9-1i	4	15	GIN
												3-1i	1	8	GNB
			83-3					Affected+1	1-2	3-2	/	16-1i	8	/	KEN
								 Canandia	0.4-1	14-1i	0.1-1	3-1i 1	0.4	-	LSO
59-2	20-3	02-2	0					Sporadic+1				L-1j	2	0.1	LDK
			0	0.1	0			Affoctod.	1-1	J-1j	Z-1	J-1i	2	0.3	MUM
07-2	17_2	20-2	16-2		5			Veru heavu	1-4	10-4	1-4	10-1	3	9	MU
51	28		44		14-2				1-2	24_1i	03	5-1	1	4	MRT
100	100	91	100	40	100	29		Ancecedit	5-2	22-2	2-2	8-1i	2	6	MUS
	48-3i	15-si					45-4	Affected+1	0.4-1	1-1i	1-1	3-1i	3	4	MOZ
			73-1						6-2	9-2	3-1	5-1i	1	1	NAM
14	20	17	7	1	2	3		Affected+1	5	6-1i	4	5-1i	1	3	NER
								Very heavy+1				76-1i	9	26	NGA
52	71	67	61	35	83	23		Sporadic+1	4	6-1i	3	5-1i	4	3	RWA
88-2	72-2	88-2	87-2		59-2					28-4i		1-1i	1	1	STP
78		40	44	13	27			Sporadic+1	8	8-1j	15	14-ıi	6	39	SEN
100	100	100	100	100	100	7	47-4		-	48-1i	-	1-1i			SYC
32	64	55	13	1	2	10		Affected+1				1-1i	0.4	0.4	SLE
								Affected+1				7-1i	1	1	SOM
							92-4	Affected+1	4-1	1-1i	42-1	9-1i	8	3	ZAF
								Affected+1				1-1i	0.4	0.1	SSD
40+1	64	19+1 47	23+1	U.5+1	Z+1	Z+1		Sporadic+1		/ -1i		/ -1i	2	10	IGO
		41-2						ATTECTED+1				0-1j	4	2	UGA
		2U-3i	3/		100		21	Sporadic .		4-3		/ -1j	5	1	TZA ZMD
02-2		00-3	50-2	0-2	00-2	4-3		Affocted -				2-1i	2	2	21VID
								Allected+1	U.D-4	13-4	1-4	20-1	3	3	ZWE
			A		В		с								
----------------------------------	------------------------------------	--	--------------------------------------	--------------------------	---	--------------------------	-------------------------------------	----------------------							
					S										
	Extent to wh for sus	ich global citizer tainable develop	nship education a ment are mainst	and education treamed	eroviding ed HIV/AID:	% o wi	f students and y th understandin	outh g of							
	:ducation olicies/ rameworks	urriculum	n-service eacher raining	tudent ssessment	6 of schools p ife skills-base ducation	IIV/AIDS and exuality	ilobal itizenship	cientific iteracy							
Country or territory	ш с с	0		a v	ہ <u>ت</u> %	± •		 475							
SUG INDICATOR		4.	/.1		4./.2	4.	/.4	4.7.5							
Northern Africa and Western Asia				20	1/										
Algeria					-1										
Armenia	Medium	High	Medium	High	100	18-3									
Azerbaijan															
Bahrain					100		37								
Equpt	Medium	High	Medium	 High	3	5-4									
Georgia	High	High	Medium	High											
Iraq	Medium	Low	Medium	High											
Israel							30								
Kuwait	High	High	High	 High	100		30								
Lebanon															
Libya															
Morocco	Medium	High	Medium	High											
Palestine	піді1 	nign 	weardm	nigi1	3										
Qatar	High	Medium	Medium	High	100		30								
Saudi Arabia					100		32								
Sudan															
Syrian Arab Republic															
Turkey	 High	High	Medium	 High			43								
United Arab Emirates							34								
Yemen															
Central and Southern Asia															
Afghanistan	 Madium					2-4									
Bangladesn Bhutan	Medium	LOW	LOW	High	100-1										
India						26-3									
Iran, Islamic Republic of	Medium	Medium	Medium	High			38								
Kazakhstan															
Kyrgyzstan	 Hiab	 Hiab	 Modium	 Hiab	100-2										
Nepal	nigii	nigii	Medium	nigii	100-2										
Pakistan	Medium	High	High	High											
Sri Lanka					100-1										
Tajikistan	High	Low	Medium	High											
lurkmenistan	 Hiab		 Medium	 Hiab	71										
Eastern and South-oastern Asia	riigit	LUW	WEUIUIII	riigit	21										
Brunei Darussalam															
Cambodia	High	High	Medium	High											
China					81										
DPR Korea															
Hong Kong, China					96i		16	56-3							
Japan	Hiah	 Hiah	 Hiah	 Hiah			 14								
Lao PDR															
Macao, China					100										
Malaysia					100	41-4	16								
Mongolia	Medium	High	Medium	Low											
Philippines	nigi1	LUW	LUW	niyi1	100	1 / -3 									
Republic of Korea	Medium	Medium	Medium	High			14	69-3							
Singapore	Medium	High	Medium	High	89-1		33								
Thailand	High	Low		High	100	46-3	13-4								
Timor-Leste						11-3									

	D		E			F	G	H		Internation	l nally mobile			1	
% of scho	ols with WAS	H facilities	% of s peda	% of schools with ICT for pedagogical purposes fin fit fit		th adapted ind materials th disabilitie	5	on educatio	Mol	tertiary bility e (%)	students Nur (0	nber 00)	Official de assistan (000	evelopment ice, in US\$ ),000)	
Basic drinking water	Basic sanitation or toilets	Basic handwashing	Electricity	Internet	Computers	% of schools wi infrastructure a for students wi	Level of bullyin	Level of attacks	punoqu	Outbound	Inbound	Outbound	Scholarships	Imputed student costs	Country code
		<u> </u>	4.a.1	1	<u> </u>		4.a.2	4.a.3					4.	.b.1	
				2019				1			20	19			
								Affected+1	1-1	2-1i	8	30-1i	19	103	DZA
100	100	100	100	54	95		40-3	Affected+1	2	27-1i	5	D-1i 44-1i	8	10	ARM
100	100	100	100	100	100	100	29-3		14	13-1i	7	6-1i			BHR
									24-1	56-1i	11-1	26-1i			CYP
	100-3	100-3	100-3	71	95		75-4	Sporadic+1	2-3	1-2	51-3	39-1i	13	52	EGY
100	100	100	100	100	100		42-1	 Affected	8	8-1i	12	11-1i 32_1i	5	18	GEO
100-3	100-3	100-3	100-3	85-3i	 85-зі		24			4-1j		16-1			ISR
36	36	36-1	36	13	13		62-1		14-1	8-1i	41	26-1i	14	16	JOR
100	100	100	100	100	100	100	79-4			21-ıi		25-1i			KWT
60-3i	92- <sub>3i</sub>	100	100	91	67		18-2	Sporadic+1	10	8-1i	23	18-1i	4	28	LBN
	91	 81		 83		 20		Affected+1		 5-1i		10-1i 52-1i	22	141	LBY
100	100	100	100	100	100		86-4		3	14-1i	3	16-1i			OMN
99	100	100	100	91	94	59		Heavy+1	-	12-1i	-	27-1i	5	21	PSE
100	100	100	100	100	100	100	64-1		35	27-1i	12	9-1i			QAT
100	100	100	100	100	100	100	54-1		4	5-1i	73	77-1i			SAU
90-3	/ 5-3		54-3					Heavu+1		Z-4 7-3		13-1i 64-1i	15	142	SYR
98-1	100-2	100-1	100-1	49-1	96-1			Sporadic+1	2-1	9-1i	6-1	24-1i	13	84	TUN
							45-1	Heavy+1	2-1	1-1i	125-1	48-1i	17	87	TUR
100	100	100	100	100	100	100	59-1		49-3	6-2	225	12-1i			ARE
							42	Heavy+1				26-1i	2	15	YEM
															150
60-1 70 -	26-1	5-1					44	Heavy+1		8-1i		30-1i 50. r	6	8	AFG RCD
80+1	57-3	25-3	95+1	61+1	21+1		30-3	Allected+I		38-1		5-1i	2	0.2	BTN
94	89	86	65	6	17	69		Very heavy+1	0.1	1-1i	47	375-1i	19	180	IND
							77-4	Sporadic+1	1-1	2-1i	21-1	56-1i	12	93	IRN
			100+1			7-1	49-1	Sporadic+1	3	13-1i	41+1	84-1i	8	13	KAZ
		100-2	100-2	41-2	89-2			Sporadic+1	9	5-1i	20	11-1i 2 r	5	4	KGZ
39-3i	100-2	100-2	100			100-2	51-4	Sporadic+1		20-2		82-1i	6	14	NPL
52-3i								Very heavy+1		3-1i		59-1i	15	50	PAK
87-1	90-1	87-1	100-1	16-1	51-1		39-3	Sporadic+1	0.5	8-1i	2	24-1i	7	3	LKA
									1-2	7-2	2-2	20-1i	2	2	TJK
100	100	100	100	28	99			Sporadic+1	0.3		0.2	45-1i 27 si	1	1	I KM
30-1	05	05	100	07	57-1	21			0.2-1	12-11	1-1	57-11	4	/	020
	100	100	100				81-1		3	25-1i	0.4	3-1i			BRN
	48-3i	49-3i								3-1j		6-1i	13	3	KHM
100	99	98	99	98	98			Affected+1	0.4	2-1	201	993-1	23	376	CHN
								Sporadic+1		0.3-1i		1-1i	0.1	0.4	PRK
100	100	100	100	99i	99i	95i	54-1	 Cooradic -	14	12-1i	43	36-1i			HKG
20-1	5U-1	09-1	93-1		40-1		35-1	Sporadic+1	5-1	1-1i 1-1i	183-1	32-1i	30	45	JPN
			50				13-4		0.5	6-1i	0.5	7-1i	11	0.4	LAO
100	100	100	100	100	100	78	58-1		52	8-1i	18	3-1i			MAC
92-1	100-1	92-1	93	92	100-2	40-1	66-1		7	5-1i	82	62-1i	5	14	MYS
				71-3					1-1	8-1i	2-1	12-1i	13	6	MNG
/ 5-1 59	58	50-1 83	27-2	29	78	L-1 6	5U-3 88-1	Affected+1	1	L-1i 0.5-2	U.5-1	10-1i 19-1i	15	4	PHI
100-3	100-3	100-3	100-3	100-3	100-3		20-1	meeteuri	3-1	3-1i	85-1	102-1			KOR
100-1	100-1	100-1	100-1	100-1	100-1	93-1	64-1				52-1	24-1i			SGP
100		100	100	100	100		50-1	Affected+1	1-3	1-3	32-3	33-1i	6	10	THA
68		68	84				31-4					3-1i	5	2	TLS
							0Z-1		0.4	4-3	1	109-1i	26	56	VNM

			A		В		с	
					8			
	Extent to whi for sus	ich global citizer tainable develop	ship education a ment are mainst	and education creamed	iding IN/AI	% of wit	f students and y th understandin	outh a of
	s			L 1	ls prov ased F	2		
	ation ies/ ework	calum	rvice ler ing	ent smen	school kills-b ation	AIDS a ality	al enship	cy
Country or territory	Educ polici fram	Curri	In-se teach traini	Stude asses	% of : life sl educi	HIV// sexua	Globa citize	Scien litera
SDG indicator		4.	7.1	1	4.7.2	4.1	7.4	4.7.5
Reference year				20	)17			
Oceania								
Australia		High		High			23	
Cook Islands Fiii	High	Medium		High	100			
Kiribati								
Marshall Islands								
Micronesia, F. S.								
Nauru New Zealand	 Hiab	 Hiab	 Hiab	 Hiab	20		 18	
Niue					100			
Palau								
Papua New Guinea						25-2		
Samoa Solomon Is					100			
Tokelau					-			
Tonga								
Tuvalu	Medium	High	Medium	Low	9			
Vanuatu								
Latin America and the Caribbean					100			
Anguilla Antigua and Barbuda					100-1	86-2		
Argentina		High	Medium	 High			 14-4	
Aruba								
Bahamas								
Barbados								
Bolivia, P. S.	 Hiah	 Hiah	 Hiah	 Hiah		43-3		
Brazil								
British Virgin Islands								
Cayman Islands					100-1			
Colombia	High	Medium	Low	LOW		30-4	13	58-3
Costa Rica					47			
Cuba					100			
Curaçao								
Dominica					100			
Ecuador	 Hiah	 Low		 Hiah				J D - 3
El Salvador	Medium	Low	Medium	Low				
Grenada					92-1			
Guatemala	High	Medium	High	Low		22-4		
Guyana Haiti	Medium	 Medium	 Medium	Hiab		37-2		
Honduras	Medium	High	Medium	High				
Jamaica								
Mexico	High	High	Medium	High				46-3
Montserrat					100			
Panama								
Paraguay								
Peru	High	High	Medium	High		75-3		45-3
Saint Kitts and Nevis	Medium	High	Medium					
Saint Lucia					88			
Sint Maarten					50-1			
Suriname								
Trinidad and Tobago Turks and Caicos Islands	Medium	High 	Low	Low				
Uruguay	Medium	Low	Medium	Low	100-1			
Venezuela, B. R.								

	D			E		F	G	н			I			J	
				ted				.5		Internation	nally mobile				
			0/ - F -	ع من				lucat		tertiary	students		Official de	evelopment	
% of schoo	ols with WASI	H facilities	peda	agogical purp	DOSES	h ada nd mi h disi	_	on ed	rate	(%)	(0	nber 00)	(000	),000)	
	5	Ę				s wit Irear s wit	lying	acks					s		de
Ð	tion o	ashir	city	t.	Iters	chool ructu dent	ef bul	of att	Ę	Pun	Ę	Pun	rship	고보	ry co
asic rinkir ater	asic anitat oilets	asic andw	lectri	Iterne	duo	of sc Ifrast or stu	evel o	evelo	pour	utbo	pour	utbo	chola	nput( tuden osts	ount
≥	e s t	۵ź	ш	5	Ŭ	2 I. %				0	5	0	Ň		U
			4.a.1				4.a.2	4.a.3	_				4.	.b.1	
				2019							20	119		1	
100 2	100 2	100 -	100 2	100 -	100 2		63 1		27.1	1	445 1	13 1			ALIS
100-3	22	100-3	100-3	100-3	100-3	100	31-4			T-1		0.2-1j	0.1	-	COK
			98-3				30-3					1-1i	4	0.1	FJI
												1-1i	3	-	KIR
68	65	63	74	30	93	21-3			6		0.1	0.3-1i	-	-	MHL
100	100											0.2-1	1		NRU
							67-1		20-1	2-1i	53-1	5-1i			NZL
100	100	100	100	100	100	100						1i	0.4		NIU
												0.1-ıi	0.1	-	PLW
												1-1i	10	0.1	PNG
100	100	100	100	33	15	15			4-1	44-1i	0.1	1-1i 2 .:	/		WSM SLB
100			100	-	100	-	40					0.1-1i	-	-	TKL
							38-2					1-1i	3	-	TON
50	70	100	100	-	70	-						0.4-1i	1	-	TUV
												2-1i	2	2	VUT
100	100	100	100	100	100	100	26-3					0.1-1i			AIA
100-1	100-1	100-1	100-1	90-1	90-1	5-1						1-1i	0.2	-	ATG
			97-1	43-1	05-1		6Z-1		3-2	20.3-2	0.3-2	9-1i 0.4-1i	4	9	ARG ARW
										20-3		4-1j			BHS
100	100	100	100									1-1i			BRB
										9-1i		1-ıi	0.3	0.1	BLZ
												20-1i	1	3	BOL
		95-2	96-2	62-2	54-2	28-2	56-1		0.2-1	1-1i	21-1	67-1i	18	48	BRA
100-1	100-1	91-1 100-1	100-1	100-1	100-1	20-1			1/-3	103-11	U.1-3	0.4-1			CVM
							54-1		0.5-1	1-1i	6-1	17-1i			CHL
			100	37	94		59-1	Heavy+1	0.2-1	2-1i	5	47-1i	8	37	COL
91	73	85	99	84	95	65	52-1			2-1i		3-1i	1	3	CRI
100-1	100-1	100	100	16	100					1-1i		2-1i	2	2	CUB
												0.Z-1i			DMA
100	90-3i	100		23-3	100	100	66-1		2-2	1-2	10-2	4-1i	1	1	DOM
40-1	83-3i	83-1	79-1	39-1	75-1				1-4	3-4	6-1	23-1i	3	9	ECU
82-2			98-1	23-1	61-1	30-1			1-1	2-1i	1-1	5-1i	1	2	SLV
100-1		100-1	100-1	72-1	72-1	22-1			85-1	5-1i	8-1	0.5-1i	-	-	GRD
	76-зі			9-3	12-3		23-4			1-4		4-1i	1	2	GTM
								Sporadic+1				2-1i 10-1i	5	6	HTI
88			91	16-3	16-3	5-3			1-1	2-1j	2	5-1j	1	1	HND
90	95	100	100	79	85	12-2	26-2			6-4		5-1i	1	0.5	JAM
	75-зі			39-3			51-1		0.2-1	1-1i	7-1	34-1i	9	36	MEX
100	100	100	100	100	100	-						~-1i	0.1		MSR
												3-1i	1	1	NIC
67-3	02-31	62-3	94-3	 5-3	 5-3		17-2			Z-3		4-1i 14-1i	1	1	PAN
55-1			89	50	76	33	52-1			2-2		34-1i	4	13	PER
79-3		79-3	100-3							13-4i		1-1i			KNA
99	99	99	99	99	99	98			18	46-1i	0.4	1-1i	1	0.1	LCA
100-1	100-1	100-1	100-1	100-1	100-1	100-1				33-4i		1-1i	0.2	-	VCT
									36-4	49-4i	0.1-4	0.1-1i			SXM
												1-1i 3-1i	1	0.1	TTO
100-1	100-1	100-1	100-1	93-1	97-1					 54-4i		0.2-1i			TCA
100-1	83-зі		100-1	100-1	100-1	100-1	55-1			3-2		5-1i			URY
97-3	90-3		99-3					Sporadic+1				21-1i	2	8	VEN

	Extent to wh for sus	ich global citizer tainable develop	A Iship education a ment are mainst	nd education reamed	d HIV/AIDS 🚥	% o wi	C f students and y th understandin	jouth g of
Country or territory	Education policies/ frameworks	Curriculum	In-service teacher training	Student assessment	% of schools pr life skills-base education	HIV/AIDS and sexuality	Global citizenship	Scientific literacy
SDG indicator		4.	7.1		4.7.2	4.1	7.4	4.7.5
Reference year				20	17			
Europe and Northern America								
Albania	High	High	Low	High	95			
Andorra	Medium	High	Medium	High	100			
Austria	Medium	High	Medium					
Belarus								
Belgium	Medium	High	Low	High				49-3
Bermuda					100-3			
Bosnia and Herzegovina	Medium	Medium	Medium					
bulgaria	нідп	High	Medium	High		13-3		55-3
		nign	Medium	nign			28	
Czechia	High	Hiab	Medium	 L ow				07-3
Denmark	Medium	High	Medium	High				57_2
Estonia	Medium	Medium	Medium	Hinh				54-3
Finland	Medium	High	Mediam	High	 100i			65-3
France	High	Medium	Hiah	High				
Germanu	High	High	High	High				57-3
Greece	Medium	High	Medium	High				
Hungary	Medium	High	Medium	High				
Iceland								
Ireland	High	High	Medium	High			24	
Italy							24	62-3
Latvia	High	High	High	High				41-3
Liechtenstein								
Lithuania	High	High	Low	High				54-3
Luxembourg								
Malta	High	High	Medium	High				50-3
Monaco	High	Medium	Medium	High	100+1			
Montenegro								
Netherlands	High	High						42-3
North Macedonia	High	High		High				
Norway							24	64-3
Poland	High	High	High	High				
Portugal	High	нідп		нідп				
	High	 Hiab	Medium	 Hich	100			
Runalla Pussion Endoration	High	High	High	High				52 5
	nigit	підп	nigii	nigli				52-3
Serbia	High	 Hiab	 Medium	High	100			
Slovakia	High	Medium	Medium	Low				
Slovenia	nigh	MECIUITI	Medium	LUW				60-2
Spain	High	High	Medium	 High				00-3
Sweden	High	High	High	High				68-3
Switzerland	riigii	nign	ngn	nign				00-3
Ukraine								
United Kingdom							21	
United States							30	

	D			E			G	Н			I			J	
						s s		5		Internation	nally mobile				
						ted erial		Icati		tertiary	students		Official d	evelopment	
% of ccho	ale with WAC	LI facilities	% of s	chools with	ICT for	adap I mat disal		nedt	Mot	bility	Nun	nber	assistar	nce, in US\$	
/6 01 SCI100			peua	igogicai purț	Joses	with with	ing	ks of	Idu	( /0)			(00	,000)	a
	n or	hing	5		s	ools v cture	hlluc	attac		Ψ		Ţ	hips		cod
er king	its c	dvas	tricit	rnet	pute	sch	lof	alof	pun	noq	pun	noq	olars	uted s	ntry
Basi drin wat	Basi sani toile	Basi	Elec	Inte	Com	% of infra for s	Leve	Leve	qui	Out	qui	Out	Sche	stud cost	Cou
			4 2 1				4.a.2	4.a.3						h1	
			4.4.1	2010							20	10		.0.1	
				2013							20	19			
E 2	74	62	0.0	E1	40	7	40 -		2	14	2	10	7	20	AL D
100	100	100	96	100	49	100	49-1		41	24-1i	0.3	10-1i 2 1i	1	20	
100	100	100	100	100	100		53-1		17-1	5-1i	75-1	22-1			AUT
100-1	100-1	100-1	100-1	87-1	100-1		42-1		4-1	6-1i	17-1	23-1	3	35	BLR
100-1		100-1	100-1	100-1	100-1		54-1		10-1	3-1i	54-1	16-1i			BEL
		100-3	100-3	100-3	100-3	100-3			10-1	225-1i	0.1-1	2-1i			BMU
							45-1		7	16-1i	6	15-ıi	3	31	BIH
							56-1		6-1	11-ıi	15-1	25-1i			BGR
							57-1		14-1	3-1i	225-1	48-1i			CAN
							42-1		3-1	6-1i	5-1	10-1i			HRV
							58-1		14-1	4-1i	45-1	12-1i			CZE
100-з	100-3	100-з	100-3	100-3	100-3		61-1		11-1	2-1i	33-1	6-1i			DNK
100-з	100-3	100-з	100-3	100-3	100-3		55-1		10-1	8-1i	4-1	4-1i			EST
100i	100i	100i	100i	100i	100i	100i	51-1		8-1	4-1i	24-1	11-ıi			FIN
100-1	100-1	100-1	100-1	98-2i	99-2i		49-1	Affected+1	9-1	4-1i	230-1	99-1i			FRA
100-1	100-1	100-1	100-1				57-1		10-1	4-1i	312-1	123-ıi			DEU
							52-1	Affected+1	3-1	5-1i	26-1	39-1i			GRC
100-з	100-3	100-з	100-з	100-з	100-3		47-1		11-1	5-1i	32-1	13-1i			HUN
							37-1		8-1	19-1i	1-1	3-1i			ISL
							59-1		10-1	6-1i	22-1	15-ıi			IRL
100-з	100-3	100-3	100-3	70-зі			47-1		6-1	4-1i	107-1	76-1i			ITA
100-3	100-3	100-з	100-3	100-3	100-3	18-3	71-1		9-1	6-1i	8-1	5-1i			LVA
									88-1	128-1i	1-1	1-ıi			LIE
							48-1		5-1	9-1i	6-1	10-1i			LTU
							52-1		48-1	164-ıi	3-1	12-1i			LUX
							63-1		10-1	7-1i	2	1-1i			MLT
100+1	100+1	100+1	100+1	100+1	100+1	100+1			25+1	39-1i	0.2+1	0.4-1i			MCO
							45-1			22-1i		5-1i	1	3	MNE
100	100	100	100	100	100		46-1		12-1	2-1i	104-1	18-1i			NLD
							23		5-1	9-1i	3-1	6-1i	2	11	MKD
100+1	100+1	100+1	100+1	100+1	100+1		49-1		4-1	6-1i	12-1	17-1i			NOR
100-3	100-3	100-3	100-3	100-3	100-3		5/-1		4-1i	2-1i	54-1i	26-1i			POL
100-1	100-1	100-1	100-1	100-1	100-1		39-1		8-1	4-1i	28-1	15-1i			PRI
100	100	100	100	99	100		60-1		6	22-1i	5	19-1i 20	2	6	MDA
							6/-1	 Ca Ka	5-1	/-1i	29-1	38-1i			RUU
							03-1	Sporadic+1	5-1	1-1i	262-1	58-1i			RUS
100	100	100	100	100	100	100			00 E	129-1i	12	1-1i			SIVIR
							40-1		о О	0-1j	12	10-1j	9	25	SKD
100-3	100-3	100-3	100-3	100-3	100-3	14-3	J0-1 19 -		0-1	22-11	12-1	2 -1 -1 -1 - 2			SVK CV/N
100-3	100-3	100-3	100-3	100-3	100-3		40-1		4-1	4-11	71.	J-11			FCD
100+1	100+1	100+1	100+1	100+1	100+1		44-1 50 -			2-11	71-1	42-1			C///E
100 2	100 2	100 -	100 2	100 -	100 -		56 1		18 ,	4-1i 5 si	54 1	16 1			CHE
100-3	100-5	81	100-3	79	94		52-1	 Affected	3	4-1	55	72-1	10	125	LIKB
		51	100	, ,		T	62-1	Sporadic+1	18-1	2-1i	452-1	39-1	10	125	GBR
							57-1	Sporadicer	5-1	0.4-1	987-1	84-1			USA
							57 1		51	0.1 1	507 1	011			034

#### TABLE 7: SDG 4, Means of implementation 4.c - Teachers

By 2030, substantially increase the supply of qualified teachers, including through international cooperation for teacher training in developing countries, especially least developed countries and small island developing States

		PI	RE-PRIMA	RY					PRIMARY	,					S	ECONDAR	Y		
	A	в	c	D	E	A	В	c	D	E	F	G	A	в	с	D	Е	F	G
	v	_	s s	s		y,	-	s	s				<b>5</b>		s	s			
	Classroom teacher (000)	Pupil/teacher ratio	% of trained classroom teacher.	% of qualified classroom teacher.	Teacher attrition rate (%)	Classroom teacher (000)	Pupil/teacher ratio	% of trained classroom teacher.	% of qualified classroom teacher.	Teacher attrition rate (%)	Relative teacher salary level	% receiving in-service training	Classroom teacher (000)	Pupil/teacher ratio	% of trained classroom teacher.	% of qualified classroom teacher.	Teacher attrition rate (%)	Relative teacher salary level	% receiving in-service training
SDG indicator			4.c.1	4.c.3	4.c.6			4.c.1	4.c.3	4.c.6	4.c.5	4.c.7			4.c.1	4.c.3	4.c.6	4.c.5	4.c.7
Reference year			2019						2019							2019			
Region	Sum		Me	dian		Sum			Me	dian			Sum			Mee	lian		
World	<b>11,579</b> i	<b>17</b> i	<b>89</b> i	<b>87</b> i		<b>32,646</b> i	20	<b>81</b> i	<b>92</b> i	<b>4</b> -1i			<b>36,526</b> i	<b>13</b> i	<b>78</b> i	<b>92</b> i			
Sub-Saharan Africa	801-1i	26	45-2i	69-1i		4,577i	37	65i	78i	6i			2,785-1i	22i	51-3i	77-2i			
Northern Africa and Western Asia	415i	16	84-2i	88i		2,827i	16	85i	93i	5i		89i	3,158i	11 <sub>i</sub>	84i	97i			90i
Northern Africa	178i	25i	85i	92i		1,302	23	87i	98	5i			1,261i	17i	88i	98i	2i		83i
Western Asia	237i	16	83-2i	85i		1,525i	14	86-зі	90-2i	5i		90i	1,897i	11	100i	96-2i			91
Central and Southern Asia	2,380	13i	92i	92		6,434	26	75	92	1-1i			8,708	15	79	89			
Central Asia	217i	11i	82i	86i		285	22	98	94	5-2i			829	10	97	100			
Southern Asia	2,164	14i	88i	93		6,149	28	74	92	1-1i			7,879	20	77	88			
Eastern and South-eastern Asia	4,007	17	97i	86	8i	10,703	18	98	95	5	1 <sub>i</sub>	94i	10,367	13	95i	95	4 <sub>i</sub>		98i
Eastern Asia	3,013	15	97i	91	8i	7,094	16	96i	96	5		93i	7,413	12	93i	94	4i		98i
South-eastern Asia	994i	17	93i	74i	6i	3,609	22	97i	92	5 i	1		2,955i	17	95i	96i			97i
Oceania	62-2i	14i	100 <sub>i</sub>	100 <sub>i</sub>		202-2i	22	92i	96i				158i	15 <sub>i</sub>	69i	92i			
Latin America and the Caribbean	1,124i	19i	76-4i	98i		3,083i	17	83i	99i				3,912i	13	84-1i	98i			
Caribbean		14i	74i	98i		169i	14	82	100				157i	10	72	98			
Central America		19	93 i	100i		818	23	95i	98i	4i	1i		1,060	14	96i	99i			
South America		19i				1,472i	19i	94i	97i		11		2,022i	17i	90i	93i			89i
Europe and Northern America	2,776i	12i				4,808i	13i					79i	7,311i	9i					95i
Europe	2,171 <sub>i</sub>	12i				2,882i	13 <sub>i</sub>					78i	5,452i	9i					95i
Northern America	614-ıi		100 <sub>i</sub>	100 <sub>i</sub>		1,896-ıi		100 <sub>i</sub>	100 <sub>i</sub>		1 <sub>i</sub>	91	1,842-ıi		100 <sub>i</sub>	99i			89
Low income	402-1i	27	53-2i	66-1i	5i	2,829i	39	75i	85i	6i			1,695-ıi	27i	58-4i	86-зі			
Middle income	8,978	17i	93i	86i		24,135	22	84i	91	4i			27,621	15 <sub>i</sub>	83i	91i			
Lower middle	3,392	20 <sub>i</sub>	82-4i	89i		10,668	27	75	88i	3-1i			11,936	17 <sub>i</sub>	76	88			
Upper middle	5,586i	15 <sub>i</sub>	90i	84i		13,467	17	96i	93	5i			15,685i	13	93i	93i			92i
High income	2,191 <sub>i</sub>	13 <sub>i</sub>				5,682i	12 <sub>i</sub>	99i	100 <sub>i</sub>			88i	7,178 <sub>i</sub>	10 <sub>i</sub>		100i			95i

A Number of classroom teachers.

B Pupil/teacher ratio, headcount basis.

C Percentage of teachers with the minimum required qualifications (received at least the minimum organized and recognized pre-service and in-service pedagogical training) to teach at a given level of education.

D Percentage of teachers qualified according to national standards.

E Teacher attrition rate (%).

F Ratio of actual teacher salaries to comparable workers [Sources: OECD; for secondary: GEM Report weighted average of OECD lower secondary and upper secondary data].

G Percentage of teachers (primary/lower secondary) who received in-service training in the last 12 months.

Source: UIS unless noted otherwise. Data refer to school year ending in 2019 unless noted otherwise.

Aggregates represent countries listed in the table with available data and may include estimates for countries with no recent data.

(-) Magnitude nil or negligible.

(...) Data not available or category not applicable.

(± n) Reference year differs (e.g. -2: reference year 2017 instead of 2019).

(i) Estimate and/or partial coverage.

n         n			PRE-PRIMARY							PRIMARY	1					S	ECONDA	RY			
number         number<		A	В	c	D	E	Α	В	c	D	E	F	G	A	В	c	D	E	F	G	
Obside in the netwoork of a set o	Country or territory	Classroom teachers (000)	Classroom teachers       (000)       (000)       Pupil/teacherratio       Pupil/teacherratio       (135500m teachers       (135500m teachers       (135500m teachers       (135500m teachers       (135500m teachers       (135500m teachers       (235500m teachers       (235500m teachers					Pupil/teacher ratio	% of trained classroom teachers	% of qualified classroom teachers	Teacher attrition rate (%)	Relative teacher salary level	% receiving in-service training	Classroom teachers (000)	Pupil/teacher ratio	% of trained classroom teachers	% of qualified classroom teachers	Teacher attrition rate (%)	Relative teacher salary level	% receiving in-service training	Country code
Interner part         Interne	SDG indicator			4.c.1	4.c.3	4.c.6			4.c.1	4.c.3	4.c.6	4.c.5	4.c.7			4.c.1	4.c.3	4.c.6	4.c.5	4.c.7	
SpecimentaUUU	Reference year		1	2019				1		2019		1			1		2019				
AppleDesc	Sub-Saharan Africa																				
bend	Angola	12-3	63-3		72-3		96-3			63-3	15-3			76-3	27-3	51-4	52-3				AGO
Bowsension         is         is<         is<        is<	Benin	5	31	25-1	100	25	53	41	71	100	6			90-3	11-3	18-3	69-3				BEN
Backsarbon         S         L         M         Total         M        M        <	Botswana						15-4	24-4												64-4i	BWA
Bundi         J         M <td>Burkina Faso</td> <td>5</td> <td>21</td> <td>43</td> <td>71-2</td> <td>8</td> <td>82</td> <td>39</td> <td>89</td> <td>96</td> <td>8</td> <td>1<sub>i</sub></td> <td></td> <td>60</td> <td>22</td> <td>61</td> <td>99</td> <td>3</td> <td></td> <td></td> <td>BFA</td>	Burkina Faso	5	21	43	71-2	8	82	39	89	96	8	1 <sub>i</sub>		60	22	61	99	3			BFA
Cholweden         1a         1a         1b         3b         1b         1b        <	Burundi	3	44	100-1	90	5-1	52	43	100	100	11-3			26	25	100-1	99	3			BDI
Carneron         18         100         100         100         100         100         100         100         100         100         100         10         100 <td>Cabo Verde</td> <td>1-1</td> <td>16-1</td> <td>30-1</td> <td>30-1</td> <td></td> <td>3-1</td> <td>21-1</td> <td>99-1</td> <td>94-1</td> <td>4-1</td> <td></td> <td></td> <td>3-1</td> <td>15-1</td> <td>96-1</td> <td>93-1</td> <td>13-1</td> <td></td> <td></td> <td>CPV</td>	Cabo Verde	1-1	16-1	30-1	30-1		3-1	21-1	99-1	94-1	4-1			3-1	15-1	96-1	93-1	13-1			CPV
Certrig Antron meganolis         1.0         3.0         1.0 <td>Cameroon</td> <td>28</td> <td>20</td> <td>67-2</td> <td>61-2</td> <td></td> <td>97</td> <td>46</td> <td>81-2</td> <td>73-2</td> <td>9-2</td> <td>1-1i</td> <td></td> <td>115-3</td> <td>19-3</td> <td>53-4</td> <td>54-3i</td> <td></td> <td></td> <td></td> <td>CMR</td>	Cameroon	28	20	67-2	61-2		97	46	81-2	73-2	9-2	1-1i		115-3	19-3	53-4	54-3i				CMR
Chai         Chai         Cas         Cas <thcas< t<="" td=""><td>Central African Republic</td><td>0.3-3</td><td></td><td></td><td>100-3</td><td></td><td>10-3</td><td>83-3</td><td></td><td>100-3</td><td></td><td></td><td></td><td>4-2</td><td>32-2</td><td>45-3</td><td></td><td></td><td></td><td></td><td></td></thcas<>	Central African Republic	0.3-3			100-3		10-3	83-3		100-3				4-2	32-2	45-3					
Condos         1         2a-         2a- <td>Comoroc</td> <td>1.</td> <td>20</td> <td>24-3</td> <td>83</td> <td></td> <td>45</td> <td>20.</td> <td></td> <td>80</td> <td></td> <td></td> <td></td> <td>23</td> <td>23</td> <td>44-3</td> <td>51</td> <td></td> <td></td> <td></td> <td>COM</td>	Comoroc	1.	20	24-3	83		45	20.		80				23	23	44-3	51				COM
mage         n	Condo	1-1	20-1	50-2	44-2		78-1	28-1						26-1	0-1						COG
D.R. Congo         18+         26+         13+         100-         1         101-         1         101-         1         101-<	Côte d'Ivoire	10	20	100	100		96	42	100	100	7			77	29	100	100				CIV
Dipotent	D. R. Congo	18-1	26-1	13-1	100-1		544-1	31-1	92-1	100-1				378-1		58-1	100-1				COD
Equ 	Djibouti	0.2-1			100-1		2+1	30+1	100-1	100-1	3-2			2		100-4	100	6-1			DJI
Introde     1     <	Equat. Guinea	2-4	17-4	89-4			4-4	23-4	37-4	61-4											GNQ
Examini            9-1         26-1         8-2         9-1           7-3         16-3         7-	Eritrea	2-1	29-1	42-1		4-2	9-1	39-1	84-1	84-1				7-1	35-1		84-2				ERI
Ethopa         23         3         3         4         5         5         5         5         5         5         6         5         5         5         5         5         6         5 </td <td>Eswatini</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>9-1</td> <td>26-1</td> <td>88-z</td> <td>92-1</td> <td></td> <td>1-2i</td> <td></td> <td>7-3</td> <td>16-3</td> <td>73-4</td> <td>73-3</td> <td></td> <td></td> <td></td> <td>SWZ</td>	Eswatini						9-1	26-1	88-z	92-1		1-2i		7-3	16-3	73-4	73-3				SWZ
Gabon         in         in<         in	Ethiopia	23-2			100-2																ETH
GambáGaSSS <td>Gabon</td> <td></td> <td>GAB</td>	Gabon																				GAB
Ghana         62         30         59         55-          169         27         62         60-	Gambia	3	36	69-2	69-2	19-1	10	37	88	88				8		96	96				GMB
GuineaGuin	Ghana	62	30	59	55-1		169	27	62	60-1				188	15	77-1	77-1				GHA
Guine-Bissaul         in	Guinea						38-3	47-3	75-3	92-3	22-3										GIN
Nernga         111-1         199-         1         <	Guinea-Bissau																				GNB
Lesonio         3-3         16-3         10-4         10-4         11-4         3-2         8-3         n <td>Kenya</td> <td>111-3</td> <td>29-3</td> <td></td> <td></td> <td></td> <td>267-4i</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>199-4i</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>KEN</td>	Kenya	111-3	29-3				267-4i							199-4i							KEN
Loci         Loci <thloci< th="">         Loci         Loci         <thl< td=""><td>Liberia</td><td></td><td>37 5</td><td>55 2</td><td>55 2</td><td>5.5</td><td>78 2</td><td>22</td><td>70 3</td><td>70 2</td><td></td><td></td><td></td><td>18 2</td><td>23-2</td><td>67.4</td><td>91-3 64 4</td><td></td><td></td><td></td><td>LSU</td></thl<></thloci<>	Liberia		37 5	55 2	55 2	5.5	78 2	22	70 3	70 2				18 2	23-2	67.4	91-3 64 4				LSU
material         image	Madagascar	41	27	44	99	J-2	127	37	15	100	0-2			87	18	20-1	85-1				MDG
Mair         7-1         20-1          100-1          65-1         38-1            58-2            Mau           Mauritania         2-4         19-4            17         41         97          16           9         <	Malawi	32-4	42-4		100-4		83	55		100-1				15	68		58				MWI
Maritania         2-4         19-4           17         41         97          16          9         93          3          Mar           Mauritus         2         12         100         100         1         66         15         100         100         4         1          10         12         48         100         14          Mus           Mauritus         2         12         10         12.1         57         98         100          1         10         12.         48.         100         14.          Mus           Mamibia         2         17         17         18         17         18         17         18         17         18         17         18         17         18         17         18         17         18         19         18         18         18         10         18         18         18         18         18         18         18         18         18         18         18 <t< td=""><td>Mali</td><td>7-1</td><td>20-1</td><td></td><td>100-1</td><td></td><td>65-1</td><td>38-1</td><td></td><td></td><td></td><td></td><td></td><td>58-2</td><td></td><td></td><td></td><td></td><td></td><td></td><td>MLI</td></t<>	Mali	7-1	20-1		100-1		65-1	38-1						58-2							MLI
Madritius $2$ $12$ $100$ $100$ $1$ $6$ $15$ $100$ $10$ $4$ $1$ $$ $10$ $12$ $48$ $100$ $14$ $$ $$ $MOS$ Mozambique $$ $$ $$ $$ $$ $$ $121$ $57$ $98$ $100$ $$ $$ $33.2$ $37.2$ $85.4$ $100.3$ $$ $$ $MOS$ Namibia $2$ $23-1$ $$ $76.1$ $$ $20.1$ $25.1$ $$ $90.1$ $$ $$ $11.2$ $$ $I$ </td <td>Mauritania</td> <td>2-4</td> <td>19-4</td> <td></td> <td></td> <td></td> <td>17</td> <td>41</td> <td>97</td> <td></td> <td>16</td> <td></td> <td></td> <td>9</td> <td>29</td> <td>93</td> <td></td> <td>3</td> <td></td> <td></td> <td>MRT</td>	Mauritania	2-4	19-4				17	41	97		16			9	29	93		3			MRT
Mozambique            121         57         98         100           37-2         85-41         100-3           Ma           Namibia         2-1         23-1          76-1          20-1         25-1          90-1           1         1         1         Na          Na          Na          Na          Na          Na          90-1           Na          Na          Na         Na         Na         Na          Na          Na          Na          Na          Na          Na          Na          Na          Na          Na          Na          Na          Na          Na          Na          Na          Na          Na         Na         Na         Na         Na         Na         Na         Na	Mauritius	2	12	100	100	1	6	15	100	100	4	1i		10	12	48	100	14			MUS
Namibia $2_{-1}$ $2_{3-1}$ $$ $76_{-1}$ $$ $20_{-1}$ $25_{-1}$ $$ $90_{-1}$ $$ $$ $$ $11_{-2}$ $$ <t< td=""><td>Mozambique</td><td></td><td></td><td></td><td></td><td></td><td>121</td><td>57</td><td>98</td><td>100</td><td></td><td></td><td></td><td>33-2</td><td>37-2</td><td>85-4i</td><td>100-3</td><td></td><td></td><td></td><td>MOZ</td></t<>	Mozambique						121	57	98	100				33-2	37-2	85-4i	100-3				MOZ
Niger         6         32         36-1         94         2         67         40         62-1         99         2           91         1         1         1.00-1         1.2-2          N         N           Nigería	Namibia	2-1	23-1		76-1		20-1	25-1		90-1				11-2							NAM
Nigeria	Niger	6	32	36-1	94	2	67	40	62-1	99	2			29-1			100-1	12-2			NER
Rwanda       7       41       50       89       10       44       57       95       99       3         30-2        58-2       80-2       5-2         RWA         Sao Tome and Principe       1-4        28-4         1-2       31-2       27-7          1-3        36-4       26-4          Strp         Sengal       0.2       18       86       88       6       1       15       84       89       9          10       89       95       7         Strp         Sengelels       0.2       18       86       88       6       1       15       84       89       9         11       10       89       95       7         Strp       Strp <td>Nigeria</td> <td></td> <td>NGA</td>	Nigeria																				NGA
Sao tome and Principe       1-4        28-4         1-2       31-2       21-2          1-3        36-4       26-4          Stippe         Sengal       12       22       38       100        65       34       75       100      4         56-4        72-4       76-4          Strippe         Segchelles       0.2       18       86       88       6       1       15       84       89       9         1       10       89       95       7         Strippe         Sorralia	Rwanda	7	41	50	89	10	44	57	95	99	3			30-2		58-2	80-2	5-2			RWA
Selegal122238100 $\dots$ 653475100 $4$ $\dots$ $\dots$ $-5-4$ $\dots$ $1$ $12-4$ $7-4$ $n$ $1-6$ $1-6-4$ $\dots$ $\square$ <th< td=""><td>Sao Tome and Principe</td><td>1-4</td><td></td><td>28-4</td><td></td><td></td><td>1-2</td><td>31-2</td><td>2/-2</td><td></td><td></td><td></td><td></td><td>1-3</td><td></td><td>36-4</td><td>26-4</td><td></td><td></td><td></td><td>SIP</td></th<>	Sao Tome and Principe	1-4		28-4			1-2	31-2	2/-2					1-3		36-4	26-4				SIP
Segurations       0.2       18       60       60       60       61       15       64       69       9        1.1       10       89       95       7          Streta         Sierra Leone       6       23       52       52       52       52       34       87       64       17         22-3        7.4       37-3          Ster         Somalia <t< td=""><td>Senegal</td><td>12</td><td>10</td><td>38</td><td>100</td><td></td><td>65</td><td>34</td><td>/5</td><td>100</td><td>4</td><td></td><td></td><td>56-4</td><td></td><td>/ Z-4i</td><td>/6-4</td><td></td><td></td><td></td><td>SEN</td></t<>	Senegal	12	10	38	100		65	34	/5	100	4			56-4		/ Z-4i	/6-4				SEN
Stein actional       0       23       32       32       32       32       32       33       04       04       17         22-3        70-4       57-3   <	Seychelles	0.2	18	80	88	27	1	27	64	89	9			1	10	89	95	1			SYC
South Africa	Somalia	U	20	JZ	JZ	52	40	11	04	04	1/			22-3		10-4	J/-3				SOM
South Sudan       3-4       35-4       m	South Africa						249-4					 7_1i		197_2		100-2	80-4			91-1	74F
Togo     81     2741     641     3941     741     3941     7641     4341     741     5	South Sudan	3-4	35-4		87-4		27-4			84-4i		2-11		6-4i		100-3	64-4i			51-1	SSD
Uganda     28-2     22-2     60-2     40-2      207-2     43-2     80-2	Тодо	8+1	27+1	64+1	39+1		42+1	39+1	76+1	43+1	-+1			30+1							TGO
United Republic of Tanzania       12       116       50-3       78        196       54       99-3       98       0.3-11         106       22        91          TA         Zambia               106       22        91         TA         Zimbabwe                    91          TA         Zimbabwe                       TA         TA         Zimbabwe </td <td>Uganda</td> <td>28-2</td> <td>22-2</td> <td>60-2</td> <td>40-2</td> <td></td> <td>207-2</td> <td>43-2</td> <td>80-2</td> <td></td> <td>UGA</td>	Uganda	28-2	22-2	60-2	40-2		207-2	43-2	80-2												UGA
Zambia           78-2       42-2       99-2       94-2	United Republic of Tanzania	12	116	50-3	78		196	54	99-3	98	0.3-1j			106	22		91				TZA
Zimbabwe	Zambia						78-2	42-2	99-2	94-2											ZMB
	Zimbabwe																				ZWE

	PRE-PRIMARY								PRIMARY	1					S	ECONDAR	Y			
	A	В	c	D	E	A	В	c	D	E	F	G	A	В	c	D	E	F	G	
Country or territory	Classroom teachers (000)	Pupil/teacher ratio	% of trained classroom teachers	% of qualified classroom teachers	Teacher attrition rate (%)	Classroom teachers (000)	Pupil/teacher ratio	% of trained classroom teachers	% of qualified classroom teachers	Teacher attrition rate (%)	Relative teacher salary level	% receiving in-service training	Classroom teachers (000)	Pupil/teacher ratio	% of trained classroom teachers	% of qualified classroom teachers	Teacher attrition rate (%)	Relative teacher salary level	% receiving in-service training	Country code
SDG indicator			4.c.1	4.c.3	4.c.6			4.c.1	4.c.3	4.c.6	4.c.5	4.c.7			4.c.1	4.c.3	4.c.6	4.c.5	4.c.7	
Reference year			2019						2019							2019				
Northern Africa and Western Asia																				
Algeria						223	20	100-4	100	13-1										DZA
Armenia	8	6	82-2	100		7	21	74	100	31		55-3i	25	10	75	100			63-зі	ARM
Azerbaijan	11	18	94	97		41	16	100	100				123	8		100				AZE
Bahrain	2	14	100	100	5	9	12	100	100	9		91-4i	10	10	100	100	7		90-4i	BHR
Cyprus	2-1	15-1				5-1	12-1					92-4i	7-1	8-1					92-1	CYP
Egypt	60	25	83	100		531	25	85	100	3		 07	594	16	83	100	3		83-4i	EGY
ling						30	9			4		0 Z - 4I	23	1					94-1	IPO
Israel						77-1	12-1				1								96-1	ISR
Jordan	8	16	100	100	-	70	16	100	100	-			58	14	100	100	21		58-4i	JOR
Kuwait	9	8	75-4	74-4		33	8	79-4	77-4			85-4i	46						93-4i	KWT
Lebanon	15	15		92		41	13		93				50-з						79-4i	LBN
Libya																				LBY
Morocco						172	26	100	100	1		15-4i	155	19	100	100	0.1		84-1i	MAR
Uman	4	20	100	26	 6 .	28	10	100	100		 ٦.	89-4i	41	11	100	100			8/-4i	UMN
Qatar	4	10	100	100	10-1	13	17	100	100	6	21	 91_4i	10	10	100	100	14		 91_4i	OAT
Saudi Arabia	23	17	100	100	10 2	231	15	100	100			85-4i	233	14	100	100			86-1	SAU
Sudan	38-1	29-1																		SDN
Syrian Arab Republic																				SYR
Tunisia	16-3	15-3	100-3	100-4		71-1	17-1	100-1	100-1				87-1		100-1	100-1				TUN
Turkey	84-1	18-1				297-1	17-1					95-1	688-1	16-1					93-1	TUR
United Arab Emirates	8	27	100	100		30	19	100	100			97-1	53	10	100	100			98-1	ARE
Yemen	1-3	26-3		54-3		145-3	27-3		59-3											YEM
Central and Southern Asia						10.1														186
Afghanistan						134-1	49-1		79-1				92-1	33-1		79-1				AFG
Bangladesh		 0 .				5//-1i		50-2i	100-1i	5-3			407	39	61 100	100-1	1-2			BGD
India	1 417	31	100+1	100+1		4 339	22+1	73	97	1-2			6 103	21	76	87	 3-2			IND
Iran, Islamic Republic of						286-2	29-2	100-2	100-2			97-4i	299-2	19-2	98-2	100-2			97-4i	IRN
Kazakhstan						90+1	17+1	100+1	100+1	7-2		94-4i	244+1	8+1	100+1	100+1			98-1	KAZ
Kyrgyzstan						21	26	95- <sub>2</sub>					59	12	75-2					KGZ
Maldives	1	14	89-1	19	8-2	5	10	89	44	0.4-2	1 <sub>i</sub>		4	5	94	75				MDV
Nepal	51	19	83	88	2	201	20	97	97	2			123	28	83	89				NPL
Pakistan						514	46	//					655-1i							PAK
JII Lalika Tajikistan	30-1	13-1	0/-1 100 p	57.2		70-1	22-1	00-1 100 p	97 5	1-2	1-1j		100-1	10-1	/9-1	/0-1				TIK
Turkmenistan						22	26	99	100				76		99	100				TKM
Uzbekistan	81	11	96	100	1-3	119	21	100	100	2-3			375	11	99	100	3-3			UZB
Eastern and South-eastern Asia																				
Brunei Darussalam	1	15	64	100	8	4	10	87	100	4			5	8	90	92	5			BRN
Cambodia	8	33	98	98		52	42	100	100											KHM
China	2,786	17		91		6,364	16		96	5			6,475	13		94	3			CHN
DPR Korea						74-1	20-1		100-1				124-1			100-1				PRK
Hong Kong, China	14	12	97	100	8	29	13	96	100	2		88-4i	31	11	96	100	4		99-1i	HKG
Indonesia	466-1i	13-1i 20.		60-1i		1,//2i			8/i			51-4i	1,63/-1	15-1		96-1				IDN
Lao PDR	102-1	20-1	90	 42-1i	 1-2	36	1J-1 21	 97	 90-1i	 2-2		33-1	39	11-1	98;	 81-2i			03-1	LAO
Macao, China	1	14	99	100	5	2	14	99	100	0.5-1			3	10	93	100	5		 99-1i	MAC
Malaysia	62	16	97-1	100	14-1i	235		97	99	3	0.4-1i		229	11	93-1	95	1		99-1i	MYS
Mongolia	8	33	96	92	17	11	30	89	94	1	0.3-1i		22	13	87	94	5			MNG
Myanmar	10-1	15-1	81-1	100-1		218-1	24-1	95-1	91-1	12-1	1-ıi		154-1	27-1	89-1	97-1				MMR
Philippines	73	33	100	99	4	514	26	100	100	4	1-1i		451	25	100	100	2			PHL
Republic of Korea	99-1	13-1				166-1	16-1				1	99-1	230-1	13-1					98-1	KOR
Singapore						10-1 379	14-1	98-1 100	100-1		1-1i	90-4i	15-1 220	11-1 26	98-1	100-1 100			98-1	SGP THA
Timor-Leste		36		35		8	27	100	77				6	20	100	84			JT-41	TLS
Viet Nam	262	17	100			391	22	100		3		96-1							97-1	VNM

		PRE-PRIMARY							PRIMARY	'					S	ECONDAR	RY			
	A	В	c	D	E	A	В	c	D	E	F	G	A	В	c	D	E	F	G	
Country or territory	Classroom teachers (000)	Pupil/teacher ratio	% of trained classroom teachers	% of qualified classroom teachers	Teacher attrition rate (%)	Classroom teachers (000)	Pupil/teacher ratio	% of trained classroom teachers	% of qualified classroom teachers	Teacher attrition rate (%)	Relative teacher salary level	% receiving in-service training	Classroom teachers (000)	Pupil/teacher ratio	% of trained classroom teachers	% of qualified classroom teachers	Teacher attrition rate (%)	Relative teacher salary level	% receiving in-service training	Country code
SDG indicator			4.c.1	4.c.3	4.c.6			4.c.1	4.c.3	4.c.6	4.c.5	4.c.7			4.c.1	4.c.3	4.c.6	4.c.5	4.c.7	
Reference year			2019						2019							2019				
Oceania																				
Australia											1	88-4i							99-1	AUS
Cook Islands	· ·	19	100	100		0.1	17	100	100				0.1	15	100	100				COK
Fiji						6	20	92	100											FJI
Kiribati Mawaka II Jalaw da						1-2	25-2	73-3	100-2											KIR
Micronesia E S									 Q1											FSM
Nauru	-	23	100-3	92		0.1	27	100-3	96				-	60		100				NRU
New Zealand	15-1	8-1				26-1	15-1				1	87-4i	36-1	14-1					98-1	NZL
Niue	3		100-3	100-3		3		92-3	100-3				4		100-4	100-4				NIU
Palau																				PLW
Papua New Guinea	9-3	42-3				36-3	36-3						15-3	34-3						PNG
Samoa	0.4	10	100-1	100-3																WSM
Solomon Is	2	29		26-2	11	4	25	57	82	1			Z-4		/6-4	84-4 70				SLB
Торда	0.2-4	11-4	00	100		1-4	22-4	97-4	92-4				1-4	15-4	59-4	80-4				TON
Tuvalu	0.1	12	100	100		0.1	16	78	100				0.1	8	61	100				TUV
Vanuatu	1-4	16-4	46-4	52-4		2-4	27-4		72-4				1-4	21-4		79-4				VUT
Latin America and the Caribbean																				
Anguilla	•	32				0.2	10						0.1	9						AIA
Antigua and Barbuda	0.4-4		65-4	100-4		1-1	12-1	53-1	100-1				1-1	9-1	48-1	98-1				ATG
Argentina																				ARG
Aruba																				ABW
Bahamas	0.2-1	21-1	82-2	82-2		2-1	19-1	90-1	90-1				2-1	12-1	83-1	83-1				BHS
Belize	0.5	15	52	48		3	14	73-2	27-2		1;		2	10	52	33				BKB BLZ
Bolivia, P. S.	11-1		83-1		5-1	77-1		90-1		5-1	1-1		67-1		89-1		3-1			BOL
Brazil	318-1	16-1				799-1	20-1						1,380-1	17-1					87-1	BRA
British Virgin Islands	0.1-3					0.2-1	10-1	95-1	100-1				0.2-1	7-1	99-1	100-1				VGB
Cayman Islands						0.3-1	16-1	100-1	100-1				0.3-1	11-1	100-1	100-1				CYM
Chile	25-1	25-1		99-2		88-1	17-1		99-3		1	64-4i	84-1	18-1		100-2			87-1	CHL
Colombia Costa Dice	49	37				185	23	97	97	 ר			188	26	99	99			91-1	COL
Cuba	12	12	90	97	1	45 81	9	100	98	1	Li		30	10	100	99 74	3			CLIR
Curação																				CUW
Dominica	0.1	12	43	39-3		1	12	63	100-3				0.5	10	45	52-3				DMA
Dominican Republic	18	19	90-1	90-1		66	20	95-1	95-1	1	3 i		53	18	83-4	83-4			96-1i	DOM
Ecuador	33-1	19-1		89-1	9-1	80-1	24-1		89-1	7-1	1-1i		92-1	21-1		93-1	7-1			ECU
El Salvador	8-1	28-1	95-1	100-1	4-1	25-1	27-1	95-1	100-1	9-2	1-1i		19-1	28-1	92-1	100-1	4-1			SLV
Grenada	0.3-1	12-1	38-1	30-3	5-1	1-1	20	63-1	100-1	/-1			1-1 0.9	13-1	40-1	100-1	/-1			GRD
Guyana																				GUY
Haiti																				HTI
Honduras	12	19				44	25			4-3	2i		47	15						HND
Jamaica	10	11	100	100	3-1	11	21	100	100	13-1			13	16	100	100	14-1			JAM
Mexico	238-1	21-1	85-1			572-1	25-1	95-1			1		834-1	17-1	96-1				89-1	MEX
Montserrat	-	6	69	100	1	-	15	76	100	10-1			-	9	46	100	-1			MSR
Nicaragua	 6 a	15 5	100 2	100 2					90.5				24.2	14 2					96 1	DAN
Paraguau	0-2	13-2	100-2	100-3		19-2	22-2	55-2	50-2				24-2	14-2		04-2			50-11	PRY
Peru	89	19				213	17						209	14	91-1	82-1			96-1i	PER
Saint Kitts and Nevis	0.2-4			100-4		0.4-3	14-3	72-3	99-3	14-4			1-3	8-3	62-3	100-3	5-4			KNA
Saint Lucia	1-3					1	15	88	100				1	11	72	98				LCA
Saint Vincent/Grenadines	0.4-1	8-1				1-1	14-1	61-1	27-1				1-1	14-1	58-4	54-1				VCT
Sint Maarten																				SXM
Trinidad and Tobago	1	12	76	98-1 76		5	13	99-1	98-1				5-4		/1-4	6U-4				SUR
Turks and Caicos Islands	1	43-1	/0	70		0.2-1	 18-1	43-1	55-1				0.2-1	10-1	98-4	 90-1				TCA
Uruguay						28-1	11-1	100-1	100-1		1-1i									URY
Venezuela, B. R.																				VEN

		PRE-PRIMARY							PRIMARY						S	ECONDAR	Y			
	A	В	c	D	E	A	В	c	D	E	F	G	A	В	c	D	Е	F	G	
	assroom te achers 00)	ıpil/teacher ratio	of trained assroom teachers	of qualified assroom teachers	acher attrition te (%)	assroom te achers 00)	ıpil/teacher ratio	of trained assroom teachers	of qualified assroom teachers	acher attrition te (%)	elative teacher lary level	receiving service training	assroom te achers 00)	ıpil/teacher ratio	of trained assroom teachers	of qualified assroom teachers	acher attrition te (%)	elative teacher Iary level	receiving service training	untry code
Country or territory	ΰS	2	<del>ت</del> %	ت % ت	Te Te	58	2	<u>ن</u> %	ڭ %	Te	sa Re	≓. %	59	2	<del>ت</del> %	°,	Te	sa Re	i: %	Ů
SDG indicator			4.c.1	4.c.3	4.c.6			4.c.1	4.c.3	4.c.6	4.c.5	4.c.7			4.c.1	4.c.3	4.c.6	4.c.5	4.c.7	
Reference year			2019						2019							2019				
Europe and Northern America																				
Albania	5	16	86-1	68	0.4	10	17	90-1	88	3	1i		24	11		97-2	4		98-1i	ALB
Andorra	0.2	13	100	100	2	0.4	11	100	100	7			1	8	100	100	10			AND
Austria	24					32							74						99-1	AUT
Belarus	44-1	8-1	93-1	46-1	2-1	22-1	19-1	100-1	100-1	6-1			76-1	9-1	97-1	100-1				BLR
Belgium	36-1	12-1				73-1	11-1					74-4i	132-1	9-1					94-1	BEL
Bermuda	0.1-3		100-3	100-3		0.4-3		100-3	100-з				1-3		100-3	99-3				BMU
Bosnia and Herzegovina	2	14				9	17						27	9						BIH
Bulgaria	18-1	12-1				19-1	14-1					59-4i	39-1	13-1					96-1	BGR
Canada												88-4i							79-4i	CAN
Croatia	9-1	12-1				12-1	13-1					88-4i	52-1	6-1					98-1	HRV
Czechia											1	/8-4i							9/-1	CZE
Denmark	32-1	5-1				4/-1	10-1					91-1	54-1	10-1					94-1	DNK
Estonia						8-1	11-1						9-1	10-1					98-1	EST
Finiand	20					27					1	4Z-4i	40					0.76	93-1	FIN
France	211					247					1	90-1	400	12				0.75	03-1	FRA
Germany	311-1	0-1				245-1	12-1 0.				1	/ 3 - 41	70 -	12-1					97-1	CDC
Hupgaru	26 3	10-1				27 .	9-1				1	54	21 s	9-1					05.	
Iceland	20-3	5,				37-3	10 1				1	J4-4	01-3						96 1	
Ireland	J-1	J-1				J-1	10-1					79 4							83 4	IDL
Italu	131-1	11_1				253-1						62-4	461-1	10-1					93_1	ITA
Latvia	9		100-1			11		100-1				02 4	13	10 1	100-1				99_1	
Liechtenstein	0.1-1	8-1	100-1			0.3-1	8-1	100-1					0.3-1	9_1	100-1				55-1	LIF
Lithuania	11-1	9-1				9_1	14-1					 84-4i	30-1	8-1					99_1	ITU
Luxembourg	2-1	10-1				5-1	8-1						5-1	9-1						LUX
Malta	1-2					2-2							4-2						91-1	MLT
Monaco	0.1+1	18+1	86+1	100+1	7-2	0.2+1	11+1	75+1	99+1	4+1			0.5+1	7+1	81+1	98+1				MCO
Montenegro																				MNE
Netherlands	31					101					1	65-4j	115					0.95	98-1	NLD
North Macedonia						7-1	15-1						19-1	8-1						MKD
Norway	16					51					1	52-4i	51					0.77	94-1	NOR
Poland	107					253					1	99-4i	258					0.61		POL
Portugal	15-1	16-1				51-1	12-1					71-4i	83-1	9-1					88-1	PRT
Republic of Moldova	11	12	100	90		8	18	100	99				22	10	100	98				MDA
Romania	35-1	15-1				49-1	19-1						124-1	12-1					89-1	ROU
Russian Federation						322-1	22-1	99-1i				97-4i							98-1	RUS
San Marino	0.1	6	98			0.2	7	90					0.3	6		100				SMR
Serbia	14	12		100		19	14		100			84-4i	67	8		100				SRB
Slovakia	14-1	12-1				15-1	16-1					57-4i	40-1	11-1					92-1	SVK
Slovenia	7-1	9-1				9-3					1	80-4i	15-3						98-1	SVN
Spain	99		100-1	100-1		236		100-1	100-1		1	95-1	304		100-1	100-1		1.03	92-1	ESP
Sweden	83-1	6-1				70					1	96-1	75					0.79	95-1	SWE
Switzerland	15-1	12-1				52-1	10-1						63-1	10-1						CHE
Ukraine						108	16	89					312	8	93-1					UKR
United Kingdom	28-1	63-1				279-1	18-1						369-1	17-1					100-1i	GBR
United States	613-2					1,769-2					1	93-4i	1,695-2						98-1	USA

Rima, 13, goes to school in Bethlehem, Palestine. Her education is frequently disrupted but she has never lost hope of building a better future.

CREDIT: Jonathan Hyams/Save the Children

469

# Aid tables

#### INTRODUCTION

Data in the following four tables on official development assistance (ODA) are derived from the International Development Statistics (IDS) database of the Organisation for Economic Co-operation and Development (OECD). The IDS database records information provided annually by all members of the OECD Development Assistance Committee (DAC), as well as a growing number of non-DAC donors. Figures for ODA come from the DAC database, while figures for aid to education come from the Creditor Reporting System (CRS), a database of individual projects. Figures in the DAC and CRS databases are expressed in constant 2019 US dollars. The DAC and CRS databases are available at www.oecd.org/dac/stats/idsonline.htm.

In 2019, the methodology of defining ODA changed:

- The cash-flow approach, used for Tables 2 to 4, includes both grants and loans that (a) are undertaken by the official sector, (b) have promotion of economic development and welfare as their main objective and, for loans, (c) are at concessional financial terms (having a grant element of at least 25%).
- The new grant-equivalent approach, which is used for Table 1, counts only grants and the grant element of concessional loans as ODA.

The DAC glossary of terms and concepts is available at www.oecd.org/dac/financing-sustainable-development/ development-finance-data/dac-glossary.htm.

#### **AID RECIPIENTS AND DONORS**

The DAC list of ODA recipients consists of all low- and middle-income countries, based on the World Bank income classification. For further information, see www.oecd.org/development/financing-sustaina ble-development/development-finance-standards/ historyofdaclistsofaidrecipientcountries.htm.

Bilateral donors are countries that provide development assistance directly to recipient countries. Most are DAC

members. Bilateral donors also contribute substantially to the financing of multilateral donors through contributions recorded as multilateral ODA.

Multilateral donors are international institutions with government membership that conduct many or all of their activities supporting development and aid recipient countries. They include multilateral development banks (e.g. World Bank, regional development banks), UN agencies and regional agencies.

- 'Bilateral flows' refers to bilateral donors contracting with multilateral donors to deliver a programme.
- 'Multilateral flows' refers to bilateral donor contributions pooled with other contributions and disbursed at the discretion of the multilateral donor to fund its own programmes and running costs.

For a list of bilateral and multilateral donors, see the 'Donors' worksheet at www.oecd.org/ dac/financing- sustainable-development/ development-finance-standards/DAC-CRS-CODES.xls.

## TABLE 1: DEVELOPMENT ANDHUMANITARIAN ASSISTANCE

ODA comprises bilateral and multilateral development assistance, both sector allocable and non-allocable (e.g. general budget support, humanitarian aid, debt relief). ODA disbursements are reported as follows:

- Total ODA
  - As volume, in million US dollars
  - As share of gross national income (GNI)
- Contributions to multilateral donors (a subset of total ODA)
  - As volume, in million US dollars
  - As share of total ODA disbursements.

Reported humanitarian assistance is a subset of total ODA from the OECD CRS database. It has been estimated using the cash-flow approach.

#### TABLES 2 AND 3: DEVELOPMENT ASSISTANCE TO EDUCATION BY DONOR AND BY RECIPIENT

Direct aid to education is aid reported in the CRS database as direct allocations to the education sector. Four education levels are distinguished:

- Basic covers primary education, basic life skills for youth and adults, and early childhood education.
- Secondary covers general secondary education and vocational training.
- Post-secondary covers tertiary education as well as advanced technical and managerial training.
- Level unspecified refers to any activity that cannot be attributed solely to the development of a particular level of education, such as education research and teacher training. General education programme support is often reported in this subcategory.

Total aid to education adds to direct aid a component of general budget support (i.e. aid provided to governments without being earmarked for specific projects or sectors). It is reported as follows:

- Total aid to education is direct aid to education plus 20% of general budget support.
- Total aid to basic education is direct aid to basic education plus 50% of 'level unspecified' and 10% of general budget support.
- Total aid to secondary education is direct aid to secondary education plus 25% of 'level unspecified' and 5% of general budget support.
- Total aid to post-secondary education is direct aid to post-secondary education plus 25% of 'level unspecified' and 5% of general budget support.

The share of education in total ODA is calculated using total ODA as reported in Table 1.

# TABLE 4: DEVELOPMENT ASSISTANCETO EDUCATION BY DONOR - TOP 3RECIPIENTS

This table reports the amount and share of bilateral and multilateral donor assistance to education and to basic education allocated to the top three recipients of assistance from each donor.

						OFFIC	CIAL DEVE	LOPMEN Disburs	T ASSISTA sements	NCE (ODA	1)****						т	али них		١N
				То	tal						Of which,	contributi	ions to m	ultilateral	s			ASSIS	TANCE	
Donor	Cor	istant 201	9 US\$ mill	ions	As a sha	re of gros (9	s nationa %)	l income	Con	stant 201	9 US\$ mill	ions		As a share disburse	of total ne ments (%)	t	Cor	stant 201	9 US\$ mill	ions
	2017	2018	2019	2020	2017	2018	2019	2020	2017	2018	2019	2020	2017	2018	2019	2020	2017	2018	2019	2020
Australia	2903	3021	2888	2582	0.23	0.23	0.21	0.19	596	574	660	612	21	19	23	24	197	176	222	91
Austria	1287	1129	1230	1237	0.30	0.26	0.28	0.31		660	783	736		59	64	57	60	26	38	55
Belgium	2275	2229	2175	2235	0.45	0.44	0.42	0.48	924	946	1041	1245	41	43	49	57	171	183	154	215
Canada	4411	4657	4725	5091	0.26	0.27	0.26	0.30	1196	1142	1495	1218	27	25	33	25	653	658	561	599
Czechia	330	301	309	293	0.15	0.13	0.13	0.13	243	202	213	223	74	67	69	76	8	17	16	16
Denmark	2468	2470	2554	2567	0.74	0.71	0.71	0.73	729	744	784	917	30	30	31	36	349	343	412	363
Finland	1116	950	1131	1223	0.42	0.36	0.42	0.47	500	491	529	604	45	52	48	49	69	49	55	86
France**	10857	11652	12211	13545	0.43	0.45	0.43	0.60		5240	4790	4804		43	38	34	82	104	156	121
Germany	25181	24210	24198	27511	0.67	0.63	0.61	0.74		6021	5617	6237		24	23	22	2716	2565	2166	2877
Greece	312	276	368	235	0.16	0.13	0.18	0.13	228	239	225	230	73	87	61	98	13	6	4	
Hungary	154	278	312	424	0.11	0.21	0.21	0.27	113	152	153	218	74	55	49	52	0	7	10	16
Iceland	64	68	61	66	0.28	0.28	0.25	0.29	13	12	10	12	21	18	16	18	4	6	5	2
Ireland	857	910	973	933	0.32	0.31	0.32	0.31	353	394	405	445	41	43	42	48	118	122	120	116
Italy	5926	4957	4373	4062	0.30	0.24	0.21	0.23	2911	2826	2975	2980	49	58	70	71	269	224	116	138
Japan	15744	14429	15588	15777	0.23	0.20	0.22	0.26	3263	3471	3794	2987	30	39	36	25	778	600	495	391
Kuwait*		823	394			0.17	0.25				1			0	0		10			
Luxembourg	446	464	472	428	1.00	0.98	1.03	1.02	126	127	110	139	28	27	23	32	57	60	60	55
Netherlands	5238	5526	5292	5143	0.60	0.61	0.59	0.59	1492	1827	1867	1708	29	33	35	33	300	282	208	336
New Zealand**	431	542	555	526	0.23	0.28	0.28	0.27	76	91	99	97	18	17	18	18	35	38	33	31
Norway	4117	3919	4298	4660	0.99	0.94	1.03	1.11	996	947	983	1154	24	24	23	25	537	473	526	525
Poland	722	743	777	785	0.13	0.13	0.14	0.14	470	506	553	584	67	69	73	76	47	35	19	19
Portugal	409	397	410	367	0.18	0.17	0.16	0.16	274	247	264	238	70	66	69	68	13	7	10	6
Republic of Korea	2079	2206	2463	2251	0.14	0.14	0.15	0.14	566	584	606	486	27	26	24	21	94	122	124	139
Romania*		244	254	298		0.11	0.10	0.13		186	190	228		76	75	77	6	8	9	5
Russian Federation*		1007	1227			0.06	0.07			374	535			37	44					
Saudi Arabia*	1978	4361	1944	1454		0.61	0.26	0.20	328	28	33	44	17	1	3	3	64	779	717	546
Slovakia	124	134	116	135	0.13	0.13	0.11	0.14	87	103	94	99	70	77	81	73	1	0	2	1
Slovenia	79	81	88	86	0.16	0.16	0.17	0.17	53	53	57	58	67	65	65	67	2	2	2	2
Spain	2607	2780	2944	2891	0.19	0.18	0.19	0.22		1810	1906	1919		73	70	72	62	60	69	101
Sweden	5287	5663	5205	6095	1.02	1.07	0.96	1.13	1649	2041	1736	2660	31	36	33	44	450	466	490	519
Switzerland	3133	3049	3099	3371	0.47	0.44	0.42	0.50	805	753	739	828	26	25	24	23	336	319	329	556
Turkey*		8363	8667	8773		1.10	1.15	1.12		174	198	96		2	2	1	6207	7139	7574	8024
United Arab Emirates*		3790	2240	1651		0.99	0.61	0.45		74	112	23		2	4	1	394	1176	537	393
United Kingdom**	17727	18961	19377	17434	0.70	0.70	0.70	0.72	6008	6891	6303	5998	37	37	33	37	1876	1698	1965	1532
United States	36740	34762	33492	35071	0.18	0.16	0.15	0.16	4928	3922	4167	6083	14	11	13	18	7271	7214	8167	8735
EU Institutions	15283	15802	14937	18730					360	346	353	243	2	2	2	1	2114	1935	2132	2491
TOTAL ***	170715	186390	183077	189542	0.34	0.33	0.31	0.36	89053	90720	90246	96422	24	25	25	25	23263	24980	25381	26631

#### **TABLE 1:** Development and humanitarian assistance

Source: OECD-DAC (2021).

\* Not part of the Development Assistance Committee (DAC) but included in its Creditor Reporting System (CRS) database.

\*\* Includes funds disbursed to overseas territories.

\*\*\* Includes ODA from other bilaterals and multilaterals not listed above.

\*\*\*\* ODA disbursements and contributions to multilaterals are calculated using a new grant-equivalent methodology except for humanitarian assistance.

Shares of gross national income and shares of total net disbursements are calculated using the previous cash-flow methodology.

(...) Data not available.

## **TABLE 2:** Development assistance to education by donor

	TOTAL ODA							DIRECT ODA								SHARE						
	Educ	ation	Ba educ	sic ation	Secor educa	ndary ation	Po seco educ	st- ndary ation	Educ	ation	Ba educ	sic ation	Seco educ	ndary ation	Po secor educ	st- 1dary ation	Educ in se allocat	ation ector ble ODA	Ba educa in tota to edu	sic ation II ODA cation	Secor educa in tota to edu	ndary ation al ODA cation
Donor			Cons	tant 2019	9 US\$ mil	lions					Cons	tant 201	9 US\$ mi	llions					9	6		
	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019
Australia	208	180	121	95	45	34	43	51	207	178	85	57	27	15	25	32	10	10	58	53	22	19
Austria	154	160	3	5	18	19	133	135	154	160	2	3	17	18	132	134	49	49	2	3	12	12
Belgium	109	115	20	23	36	35	53	57	109	115	15	15	33	31	50	52	15	16	19	20	33	31
Canada	230	279	114	150	66	77	50	52	228	278	61	99	40	52	24	26	11	15	49	54	29	28
Czechia	8	8	1	1	1	1	6	7	8	8	0	1	1	0	6	6	21	20	13	13	12	8
Denmark	123	87	76	43	20	18	26	26	121	87	38	9	1	1	7	9	12	9	62	49	17	21
Finland	45	52	28	33	7	10	10	9	45	52	19	21	3	4	6	3	15	14	62	63	16	20
France**	1301	1416	155	215	180	257	966	945	1221	1318	104	121	154	210	940	897	20	20	12	15	14	18
Germany	2473	2845	365	470	492	554	1617	1821	2422	2775	167	252	393	445	1518	1712	18	20	15	17	20	19
Greece	2	2	1	1	0	0	1	0	2	2	1	1	0	0	1	0	55	45	65	81	0	0
Hungary	61	101	3	4	1	3	57	95	61	101	1	0	0	1	56	94	55	70	6	4	2	3
Iceland	0	5	0	3	0	1	0	1	0	5	0	1	0	0	0	0	1	15	60	63	20	20
Ireland	42	46	26	30	8	6	7	10	42	46	20	24	5	3	4	7	16	16	63	65	20	14
Italy	118	137	45	39	22	25	50	73	118	137	21	14	10	13	38	61	16	18	38	28	19	18
Japan	656	782	191	243	116	128	350	411	588	722	72	109	56	61	290	344	6	6	29	31	18	16
Kuwait*	199	28	66	14	59	/	/3	/	199	28	0	0	26	0	40	0	10	4	33	50	30	25
Luxembourg	51	50	11.4	15	3/	31	3	4	51	50	/	10	35	29	1	71	- 22	20	21	30	/2	63
Netherlands	1/8	99	114	19	1/	8	4/	72	1/8	99	111	1/	15	/	46	/	/	4	64	19	9	8
New Zealand**	74	72	10	15	4	5	54	52	244	242	12	11	2	3	52	51	23	21	22	21 70	5	10
Poland	00	12/	205	2/1	3/	1	40 87	121	00	12/	232	1	0	20	30 87	121	10	65	2	70	1	0
Poldilu	50	62	12	14	10	10	22	10	50	62	2	0	2	2	26	22	42	55	24	2	17	15
Penublic of Korea	214	250	67	69	50	76	97	104	214	250	55	54	44	68	90	97	14	15	24	78	73	30
Romania*	46	52	2	0	4	4	40	48	46	57	0	0	3	4	39	48	96	97	4	0	9	7
Russian Federation*	10	16	-	8	•	4	10	4	10	6	0	3	5	2	35	1	50	11		50	5	27
Saudi Arabia*	907	329	393	86	205	43	309	200	410	255	2	5	9	3	114	160	60	33	43	26	23	13
Slovakia	4	3	1	1	1	1	2	2	4	3	1	0	1	1	2	2	32	27	24	15	20	22
Slovenia	11	14	0	0	0	1	11	13	11	14	0	0	0	1	11	13	57	60	0	0	3	8
Spain	59	58	23	21	17	18	20	19	59	58	9	9	10	12	13	13	12	11	38	36	28	31
Sweden	155	140	94	85	19	16	41	38	155	140	75	68	9	8	32	30	6	6	61	61	12	12
Switzerland	133	138	49	55	58	57	26	27	131	137	31	34	50	47	17	16	10	10	37	40	44	41
Turkey*	371	433	16	61	9	33	345	339	366	429	4	2	3	4	339	309	38	53	4	14	2	8
United Arab Emirates*	479	230	233	105	115	53	131	72	81	78	5	4	2	2	17	21	53	26	49	46	24	23
United Kingdom**	911	1017	472	560	204	229	235	228	911	1017	318	423	128	160	158	159	10	11	52	55	22	22
United States	1665	1423	1393	1119	57	108	215	196	1637	1406	1345	1052	33	75	191	162	9	9	84	79	3	8
TOTAL bilaterals***	11491	11195	4382	3903	1917	1921	5192	5370	10347	10650	2818	2665	1135	1302	4410	4750	14	14	38	35	17	17
African Development Fund	116	97	9	19	44	38	63	40	116	75	0	0	39	28	58	30	6	5	8	20	38	39
Asian Development Bank	276	429	117	173	139	217	20	38	276	429	83	121	122	191	4	12	12	13	42	40	50	51
EU Institutions	1267	1107	610	441	294	323	362	342	1156	995	271	130	125	167	193	186	8	8	48	40	23	29
International Monetary Fund (Concessional Trust Funds)	238	295	119	147	60	74	60	74	0	0	0	0	0	0	0	0						
UN Relief and Works Agency for Palestine Refugees	450	433	450	433	0	0	0	0	450	433	450	433	0	0	0	0	80	80	100	100	0	0
UNICEF	85	81	53	52	17	15	15	14	85	81	23	24	1	2	0	0	10	18	63	64	20	19
World Bank (International Development Association)	1281	1552	620	656	395	450	267	447	1281	1552	421	354	295	299	167	296	9	9	48	42	31	29
TOTAL multilaterals***	3769	4122	2018	1992	952	1150	799	979	3419	3694	1283	1099	585	704	432	532	8	9	54	48	25	28
TOTAL	15260	15316	6400	5896	2869	3072	5991	6349	13766	14344	4102	3764	1720	2006	4842	5283	12	12	42	38	19	20

Source: OECD-DAC, CRS database (2021).

\* Not part of the Development Assistance Committee (DAC) but included in its Creditor Reporting System (CRS) database.

\*\* Includes funds disbursed to overseas territories.

\*\*\* Includes ODA from other bilaterals and multilaterals not listed above.

(-) Nil.

(...) Data not available.

## **TABLE 3:** Development assistance to education by recipient

	TOTAL ODA						DIRECT ODA								SHARE							
Region	Basic Education education				Post- Secondary secondary education education		Basic Education education		Secondary education		Post- secondary education		Educ in se allocat	ation ector ble ODA	Ba educa total educ	asic ation in ODA to cation	Secondary education in total ODA to education					
Countru			Cons	stant 201	9 US\$ mi	llions					Cons	tant 201	9 US\$ mi	llions	1					%		
	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019
	2604	2000	177.4	1620	0.00		1000		222.4	2200	-				764	0.21						
Sub-Saharan Africa	3604	3651	1724	1639	860	904	1020	1108	3224	3306	1214	1085	605	627	764	831	9	9	48	45	24	25
Unallocated within the region	/3	114	34	53	14	24	26	3/	/3	114	22	33	8	15	20	29	3	4	46	4/	19	21
Angola	2/	2/	16	18	3	2	8	/	2/	2/	15	16	2	2	10	6	12	14	61	65	10	9
Benin	/1	59	26	22	22	14	24	22	58	5/	16	16	1/	11	19	19	14	10	36	38	31	24
Botswana	4	8	2	3	1	4	1	2	4	8	1	2	0	3	1	1	4	10	46	35	19	4/
Burkina Faso	105	100	49	48	30	25	21	2/	8/	84	32	28	21	15	18	18	- 11	10	46	48	28	25
Burunai	1/	24	5		6	4	12	9	1/	24	2	/	5	2	5	10	5	5	2/	46	3/	1/
Cabo verde	28	26	1	5	9	8	12	12	2/	125	10	11	5	5	9	10	32	1/	26	20	32	33
Cameroon	152	162	40	43	19	22	94	97	99	125	10	10	4	/	79	81	20	15	26	26	12	14
Central African Republic	25	34	11	19	5	6	9	9	6	26	1	10	1	1	4	5	8	9	43	56	22	16
Chad	76	80	3/	39	19	20	20	21	2/	55	8	16	4	8	5	10	22	25	48	49	25	25
Comoros	1/	16	3	3	2	ſ	12	13	1/	16	2	2	2	1	12	13	20	22	18	15	12	5
Congo	31	38	10	10	3	6	18	23	31	28	9	3	2	2	18	20	23	30	32	25	8	15
Cote d'Ivoire	106	94	40	24	25	20	41	50	84	82	24	9	1/	13	33	42	13	8	38	26	24	22
D. R. Congo	153	144	12	66	53	48	21	30	153	144	55	45	45	3/	19	20	8	8	4/	46	35	33
Djibouti	22	29	10	13	4	6	8	10	16	18	6	/	2	3	6	/	15	14	45	45	1/	21
Equat. Guinea	2	16	1	8	0	4	1	4	2	2	1	1	0	0		-	19	169	52	50	8	23
Eritrea	5	26	0	9	2	8	2	9	5	10	0	0	2	4	2	5	6	35	11	33	38	32
Eswatini	8	9	5	5	1	2	2	2	8	9	2	2	0	0	1	1	6	10	60	55	19	20
Ethiopia	308	285	217	170	53	67	38	48	308	264	191	146	40	55	25	35	8	8	70	60	17	24
Gabon	38	40	10	10	7	7	21	22	21	23	1	0	2	2	16	17	74	54	26	26	18	19
Gambia	28	20	11	9	4	4	13	8	21	15	4	3	0	1	9	5	15	12	41	42	13	18
Ghana	129	108	53	36	40	35	36	37	91	97	28	20	28	27	24	29	13	11	41	33	31	33
Guinea	93	56	39	27	27	5	26	24	78	51	30	23	22	3	22	22	19	11	42	48	29	9
Guinea-Bissau	17	17	5	2	1	2	10	13	16	17	3	1	1	1	9	12	12	15	32	14	9	11
Kenya	141	111	62	39	39	32	39	40	141	111	47	32	32	29	32	36	5	3	44	35	28	29
Lesotho	7	6	4	4	2	1	1	1	7	6	3	3	1	1	0	1	4	4	61	63	22	18
Liberia	44	46	31	35	10	8	2	3	41	44	30	33	9	7	1	2	8	9	72	76	23	17
Madagascar	63	59	28	2/	15	14	20	18	51	50	18	19	10	10	15	13	10	10	45	46	24	24
Malawi	131	108	/4	60	31	26	25	22	125	108	62	46	26	19	19	15	11	10	5/	55	24	24
Mali	157	195	94	109	30	40	33	46	123	127	68	61	1/	16	20	22	14	15	60	56	19	20
Mauritania	48	35	/	10	8	1/	33	9	39	35	1	/	5	15	30	/	12	8	15	28	1/	4/
Mauritius	11	14	1	2	2	3	8	9	11	14	1	1	2	2	8	8	9	20	9	1/	18	21
Mozambique	199	195	122	114	48	52	28	29	198	180	94	80	34	35	13	11		13	62	59	24	27
Namibia	28	23	19	12	4	1	5	4	28	23	1/	9	3	6	4	3	15	13	68	50	16	31
Niger	103	136	48	6/	41	46	14	23	92	128	30	35	32	30	5	/	11	12	4/	49	39	34
Nigeria	100	239	96	108	72	58	59	/3	10.4	239	64	20	56	45	43	59	9	9	42	45	32	24
Rwanda	109	132	50	48	2/	53	31	31	104	125	42	39	23	48	2/	2/	10	12	46	36	25	40
Sao Tome and Principe	6	1	2	3	1	2	2	3	6	1	1	1	0	1	2	2	12	14	39	43	19	21
Senegal	152	169	50	53	31	38	/1	68	150	165	35	50	24	32	63	62	15	12	33	3/	21	23
Sierra Leone	46	60	24	31	15	20	10	9	36	55	13	19	10	14	2	3	10	- 11	52	52	33	33
Somalia	45	39	23	16	12	9	10	14	39	32	13	/	/	4	5	9	6	5	52	41	26	23
South Africa	61	/4	24	32	10	13	27	28	61	/4	15	18	5	6	23	21	5	/	39	44	16	18
South Sudan	58	56	50	41	3	5	5	10	58	56	45	35	1	2	2	7	11	8	86	74	5	9
Uganda	44	34 133	47	ð 45	30	8 27	20	18 61	126	31	41	5 27	27	0 18	46	52	1/	8	28	34	21	23
U. R. Tanzania	120	184	113	88	42	58	32	37	120	184	79	58	25	43	15	22	7	9	60	48	22	32
Zambia	39	44	14	20	14	14	10	9	39	44	7	13	11	11	6	5	4	5	37	46	37	33
Zimbabwe	40	19	23	5	11	6	6	7	40	19	21	3	10	5	5	6	6	3	58	28	27	33
Northern Africa and Western Asia	3810	3237	1721	1363	666	486	1423	1389	2874	2855	866	911	239	260	996	1163	21	18	45	42	17	15
Unallocated within the region	42	44	33	19	4	13	5	13	42	44	32	7	3	7	5	7	5	5	79 C	42	9	29
Ацена	٥Cı	د دا	5	5	5	4	132	120	٥Cı	133			5	5	ا د ا	125	04	01	2	2	Ζ.	5

	TOTAL ODA					DIRECT ODA							SHARE									
Region	Educ	ation	Ba educ	isic ation	Seco educ	ndary ation	Po secoi educ	st- ndary ation	Educ	ation	Ba educ	isic ation	Seco educ	ndary ation	Po secor educ	st- ndary ation	Educ in se allocat	ation ector ole ODA	Ba educa total C educa	sic tion in )DA to ation	Secon educat total O educa	idary tion in IDA to ation
Country			Cons	tant 201	9 US\$ mi	llions					Cons	tant 201	9 US\$ mi	llions					9	6		
	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019
Armenia	74	103	5	45	1	20	18	38	74	30	4	5	0	0	18	18	10	47	20	/13	4	20
Azerbaijan	34	43	4	3	10	13	20	27	34	43	2	1	9	11	19	26	15	18	11	8	30	20
Egypt	401	309	157	89	92	59	152	160	349	258	24	24	25	27	85	128	14	11	39	29	23	19
Georgia	77	71	20	18	17	13	40	39	77	64	4	4	9	6	32	32	11	12	26	26	22	18
Iraq	117	80	55	39	25	11	37	30	55	63	16	24	6	3	18	23	14	7	47	49	21	13
Jordan	380	402	288	282	25	40	67	80	371	349	257	230	10	13	51	54	18	20	76	70	7	10
Lebanon	2/5	283	168	196	42	30	65	58	2/5	283	132	185	24	24	4/	52	31	3/	61	69	15	1
Morocco	316	314	46	60	68	69	201	185	304	303	18	44	54	61	187	177	21	20	15	19	77	77
Palestine	486	494	372	353	51	54	64	87	449	431	308	283	19	19	32	52	34	37	76	71	10	11
Sudan	44	110	17	43	10	26	17	41	36	36	9	1	6	5	13	20	13	23	39	39	22	24
Syrian Arab Republic	196	283	60	68	11	11	125	204	196	283	44	51	2	2	117	195	19	32	31	24	5	4
Tunisia	149	202	8	37	19	43	122	123	149	193	4	7	17	28	120	108	11	14	6	18	13	21
Turkey	328	240	96	61	90	57	142	123	328	240	3	32	44	42	96	108	13	14	29	25	27	24
Yemen	/94	115	389	4/	197	23	208	45	39	90	8	13	6	6	1/	27	/4	10	49	41	25	20
Central and Southern Asia	2366	2534	974	941	464	542	927	1051	2327	2460	714	590	334	366	797	875	12	12	41	37	20	21
Afabanistan	20	252	150	3	4	87	21	4	26	212	112	112	4	5	70	56	ð	4	53	24 51	16	45
Bangladesh	612	720	351	401	177	206	85	113	612	720	272	274	137	142	45	49	14	16	57	56	29	29
Bhutan	7	13	2	5	2	4	2	4	7	13	1	1	1	2	1	2	6	7	38	37	27	34
India	408	446	103	45	52	36	254	365	396	413	59	13	30	21	232	350	8	8	25	10	13	8
Iran, Islamic Republic of	93	112	1	1	1	1	90	110	93	112	0	0	1	1	90	109	69	69	1	1	1	1
Kazakhstan	73	51	3	2	2	2	68	47	73	51	0	0	1	1	67	46	64	54	4	4	3	3
Kyrgyzstan Maldivos	152	72	27	25	19	8	106	39	152	72	14	16	13	4	100	35	35	20	18	34	13	12
Nenal	200	5 191	117	85	45	58	38	49	200	190	93	48	34	39	2	30	14	15	58	44	73	30
Pakistan	374	392	184	158	75	72	116	161	373	392	140	104	53	45	94	134	17	13	49	40	20	18
Sri Lanka	76	67	22	18	25	28	29	21	76	67	17	13	22	26	26	19	11	9	29	26	33	42
Tajikistan	25	47	9	14	9	17	8	15	25	47	5	6	6	13	5	11	6	13	36	31	34	37
Turkmenistan	6	10	1	0	3	5	3	5	6	10	0	0	3	5	3	4	27	37	10	5	45	50
Uzbekistan	23	46	3	3	6	16	15	26	23	46	0	1	5	15	14	25	2	4	12	/	25	36
Eastern and South-eastern Asia	1618	1740	343	326	376	459	899	955	1618	1740	164	134	286	363	810	859	13	15	21	19	23	26
Unallocated within the region	6	9	3	4	2	3	2	2	6	9	2	3	2	2	2	1	3	4	43	45	30	32
Cambodia	653	633	57	17	184	4/	34 418	38	653	633	33	4	160	176	394	427	45	48	8	38 2	24	34 29
DPR Korea	1	1	0	0	0	0	1	0	1	1	0	0	0	0	1	0	4	2	0	1	0	1
Indonesia	182	153	54	32	25	21	103	100	182	153	14	14	5	12	83	92	6	10	30	21	14	14
Lao PDR	102	109	58	41	30	47	13	21	102	109	49	26	25	40	9	14	16	17	57	37	29	43
Malaysia	35	35	3	3	1	1	31	30	35	35	1	1	0	1	30	29	48	34	8	9	3	4
Mongolia	50	78	11	20	11	18	28	41	50	78	8	10	9	13	27	36	13	21	23	25	21	22
Philippines	122	59	25	50 22	23	04 11	87	26	122	50	14	19	1/	48	83	29	5	9	30 74	32	5	40
Thailand	36	43	8	8	3	5	24	29	36	43	5	5	2	3	22	27	9	11	23	19	9	12
Timor-Leste	34	44	17	25	7	7	11	12	34	44	6	13	1	1	5	6	18	21	49	56	19	17
Viet Nam	198	281	19	51	54	53	125	177	198	281	7	8	48	31	119	155	7	12	10	18	27	19
Oceania	231	253	93	95	56	73	82	85	197	232	50	49	35	50	61	62	11	12	40	38	24	29
Unallocated within the region	29	27	3	5	11	36	17	25	35	83	2	10	10	30	15	19	9	9	11	19	34	7
Cook Islands	2	2	1	1	0	0	1	1	1	1	0	0	0	0	1	0	3	4	34	38	18	23
HJI Kiribati	19	1/	/	6	2	2	9	9	19	16	3	3	0	0	/	/ 2	1/	15	39	38	13	10
Marshall Islands	14	10	0	7	2	3	2	3	9	5	7	0	0	0	0	0	51	38	74	53	12	23
Micronesia, F. S.	11	12	6	6	2	2	3	3	3	2	1	1	0	0	0	0	25	28	55	55	21	22
Nauru	2	3	0	0	2	2	0	1	2	3	0	0	2	2	0	1	6	6	4	14	79	62
Niue	4	4	1	2	1	1	2	2	1	2	0	0	0	0	1	1	15	14	38	39	19	20
Palau	14	1	7	0	3	0	4	0	1	1	0	0	0	0	0	0	74	3	50	54	24	9
Papua New Guinea	46	47	15	22	17	12	14	13	46	47	8	14	13	8	11	9	6	7	33	46	36	26
Solomon Is	16 74	13 19	10	4	6	5	8 7	8 6	74	13	4	2	4	4	/	/	13	Q II	43	29 39	8 27	9 79
Tokelau	6	3	4	2	1	1	1	1	1	2	1	1	0	0	0	0	24	11	60	55	20	23

	TOTAL ODA						DIRECT ODA								SHARE							
Region	ation	Ba	isic ation	Seco educ	ndary ation	Po seco educ	ost- ndary ation	Educ	ation	Ba educ	sic ation	Seco educ	ndary ation	Po seco educ	ost- ndary ation	Educ in se allocat	ation ector ble ODA	Ba educa total ( educ	isic tion in ODA to ration	Secon educa total C educa	ndary tion in DDA to ration	
Country			Cons	stant 201	9 US\$ mi	llions			Constant 2019		9 US\$ mi	illions						%				
	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019	2018	2019
Торда	8	7	2	2	2	2	4	3	8	7	2	2	2	2	3	3	10	7	30	79	25	78
Tuvalu	4	4	2	1	0	1	1	2	3	3	2	1	0	0	1	2	16	14	56	37	6	13
Vanuatu	20	21	7	8	4	5	8	8	20	21	6	4	4	2	7	6	18	19	38	39	22	22
Latin America and the Caribbean	831	873	342	305	143	188	345	380	820	813	252	191	98	132	300	323	9	11	41	35	17	22
Unallocated within the region	22	33	11	7	1	3	10	24	22	33	10	6	1	2	10	23	2	4	50	21	5	9
Antigua and Barbuda	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	4	2	1	14	62	7
Argentina	31	6/	/	25	5	0	18	28	31	30	2	2	2	2	0	0	6	10	42	38	17	20
Bolivia P S	30	2	7	7	7	6	16	10	30	2	3	3	5	4	14	9	4	3	45	30	74	24
Brazil	99	103	18	18	9	9	72	76	99	103	4	3	2	1	65	69	15	19	18	18	9	8
Colombia	75	76	14	14	9	9	52	53	75	76	9	9	6	7	50	50	4	9	19	18	12	12
Costa Rica	14	13	6	6	3	2	6	5	14	13	4	4	2	1	5	5	12	19	40	45	18	13
Cuba	9	11	1	1	0	2	8	8	9	11	1	0	0	1	7	7	5	8	14	14	4	14
Dominica	4	7	2	3	1	2	1	2	3	1	0	0	0	0	0	1	34	32	44	43	22	24
Dominican Republic	21	24	12	12	5	8	4	4	21	24	11	12	4	8	3	3	16	15	5/	52	24	34
Ecuador	33 49	53 77	13	15	21	8 46	9	30	33 49	44	17	15	3 20	4	14 9	20	8 19	73	38	28	43	64
Grenada	0	4	0	1	0	2	0	1	0	1	0	0	0	1	0	0	1	35	20	37	13	43
Guatemala	79	55	60	35	11	12	7	7	79	55	56	30	9	10	5	4	20	15	77	65	14	23
Guyana	7	8	2	3	4	4	1	1	7	8	1	2	4	3	1	1	6	7	26	35	54	49
Haiti	93	59	58	34	16	10	18	15	84	58	45	29	10	7	12	12	12	10	63	58	18	17
Honduras	43	53	34	34	5	15	4	4	43	53	32	32	4	14	3	3	6	10	80	65	11	28
Jamaica	9	12	6	7	1	2	2	3	9	8	5	5	1	1	2	1	8	12	61	60	16	18
Mexico	66	65	10	9	/	6	49	49	66	65	4	4	3	3	46	46	10	9	15	15	10	10
Nicaragua	3	3	76	18	16	10	6	6	3	3	21	13	12	0	3	1	13	2	47	48	24	24
Panama	4/	4	20	7	0	0	2	2	47	4	21	7	0	0	2	4	8	6	51	47	5	5
Paraguau	26	28	13	14	6	7	7	7	26	28	3	4	1	2	2	2	12	15	49	50	23	23
Peru	43	41	13	12	8	8	23	22	43	41	7	6	5	5	20	19	8	8	31	28	17	18
Saint Lucia	1	2	0	0	0	0	1	1	1	2	0	0	0	0	1	1	11	12	29	22	12	19
Saint Vincent/Grenad.	1	1	1	0	0	0	0	0	1	1	1	0	0	0	0	0	7	2	62	51	10	12
Suriname	2	2	0	0	1	1	1	1	2	2	0	0	1	1	1	1	12	11	2	2	43	51
Venezuela, B. R.	16	17	4	5	2	2	10	10	16	17	2	3	1	1	9	10	42	32	26	28	12	12
Europe and Northern America	606	749	126	115	72	85	408	549	526	683	52	33	35	45	371	508	12	15	21	15	12	11
Unallocated within the region	97	193	36	30	7	16	54	147	97	193	29	8	4	5	51	137	9	18	37	15	8	8
Albania	45	54	4	4	10	8	31	41	45	54	4	4	9	8	30	41	10	18	10	8	21	15
Belarus	41	46	2	2	4	2	35	42	41	46	0	0	3	1	34	41	37	23	4	4	10	5
Bosnia and Herzegovina	49	73	6	17	5	11	38	46	49	50	2	2	3	3	36	38	11	14	12	23	10	15
Montenegro	5	6	0	0	1	0	4	5	5	6	0	0	0	0	4	5	3	4	9	6	10	/
North Macedonia	21	24	5	5	3	3	13	10		24	4	3	2	14	12	15	10	12	26	22	15	12
Serbia	5/	00	50	7	21	15	42	49 50	55	50	3	3	5	14	3/	4/	18	20	36	31	13	10
	152	184	15	19	4	12	122	152	152	18.4	10	10	2	7	130	148	15	14	10	11	3	6
Unspecified by region	2194	2279	1077	1112	231	335	886	833	2179	2255	790	770	88	164	743	662	9	9	49	49	11	15
Low income	3387	2945	1736	1387	762	665	889	893	2375	2583	1006	896	397	420	525	648	13	12	46	44	23	22
Lower middle income	5975	6192	2355	2214	1233	1380	2387	2599	5703	5921	1577	1379	843	962	1998	2181	15	14	38	36	21	24
Upper middle income	3362	3380	1095	1037	591	584	1677	1758	3185	3067	627	647	357	389	1443	1563	18	23	32	30	16	19
High income	32	23	10	5	7	5	14	12	18	23	3	3	4	4	10	11	20	7	23	28	38	21
Unallocated by income	2504	2776	1203	1253	276	437	1024	1087	2485	2749	888	839	119	230	867	880	8	11	43	37	19	25
TOTAL	15260	15316	6400	5896	2869	3072	5991	6349	13766	14344	4102	3764	1720	2006	4842	5283	12	12	42	38	19	20

Source: OECD-DAC, CRS database (2021).

The country groupings by level of income are as defined by the World Bank but include only countries shown in the table.

They are based on the list of countries by income group as revised in July 2020.

All data represent gross disbursements.

Sector allocable ODA does not include budget support.

#### **TABLE 4:** Development assistance to education by donor - top 3 recipients

			EDUCATION		BASIC EDUCATION					
Donor		Recipient	Constant 2019 US\$ millions	Recipient %	Recipient	Constant 2019 US\$ millions	Recipient %			
Bilateral	Australia	Papua New Guinea	24.3	13	Unspecified by region	13.0	17			
		Unspecified by region	19.2	10	Lebanon	10.3	14			
		Myanmar	14.8	8	Indonesia	7.9	10			
	Austria	Turkey	20.9	13	Mexico	1.1	56			
		Bosnia and Herzegovina	20.3	13	Republic of Moldova	0.2	11			
		Serbia	11.6	7	Albania	0.1	5			
	Belgium	Unspecified by region	23.6	21	Unspecified by region	7.4	47			
		D. R. Congo	14.8	13	Viet Nam	1.4	9			
		Uganda	11.8	11	South Africa	1.2	8			
	Canada	Unspecified by region	25.8	11	Unspecified by region	10.7	14			
		Jordan	22.4	9	Senegal	7.5	10			
		Senegal	20.1	8	Afghanistan	6.8	9			
	Czechia	Ethiopia	0.9	10	Ukraine	0.1	38			
		Ukraine	0.8	9	Syrian Arab Republic	0.1	37			
		Cambodia	0.5	6	Ethiopia	0.0	11			
	Denmark	Unspecified by region	70.3	73	Afghanistan	12.9	48			
		Afghanistan	13.0	13	Myanmar	6.7	25			
		Myanmar	6.7	7	Unspecified by region	3.3	12			
	Finland	Mozambique	10.0	21	Mozambique	8.5	42			
		Ethiopia	5.7	12	Palestine	2.5	12			
	-	Unspecified by region	5.1	11	Ethiopia	1.5	7			
	France	Morocco	173.3	14	Lebanon	10.0	10			
		Algeria	120.7	10	Unspecified by region	9.6	9			
	<b>C</b>	China	93.2	/	Niger	8.4	8			
	Germany	China	428.9	18	Jordan	37.2	20			
			164.0	1	Lebanon	19./	10			
	6	Unspecified by region	149.1	6	Mozambique	15.5	8			
	Greece	Albania	0.3	18	Albania	0.3	22			
		Ukraine	0.3	16	Ukraine	0.3	20			
	11	Egypt	0.3	15	Egypt	0.2	15			
	Huliyaly	Julian Arab Depublic	0.0	10		0.5	11			
		China	4.2	6		0.1	10			
	Iceland		3.7	56	Malawi	0.0	85			
	ICEIdilu	Malawi	0.6	31	Afahanistan	0.0	9			
		Kenua	0.0	7	South Africa	0.0	4			
	Ireland	Unspecified by region	12.3	29	Unspecified by region	8.6	45			
	incluito	Mozambique	77	18	Mozambique	23	12			
		Uganda	47	11	Uganda	19	10			
	Italu	Unspecified by region	21.9	18	Ethiopia	2.4	12			
		Mozambique	7.0	6	Jordan	2.3	11			
		India	7.0	6	Lebanon	1.7	8			
	Japan	Unspecified by region	156.4	25	Guinea	5.6	7			
		Indonesia	38.3	6	Papua New Guinea	5.0	6			
		Egypt	37.5	6	Myanmar	4.3	5			
	Kuwait*	China	31.8	36						
		Lebanon	19.8	22						
		Mauritania	8.6	10						
	Luxembourg	Niger	8.8	18	Niger	3.6	40			
		Senegal	6.5	14	Syrian Arab Republic	1.0	11			
		Cabo Verde	5.4	11	Northern Africa/W. Asia, unallocated	0.9	10			
	Netherlands	Unspecified by region	81.7	66	Unspecified by region	22.8	49			
		Lebanon	12.4	10	Lebanon	9.0	19			
		Ethiopia	5.6	5	Syrian Arab Republic	2.7	6			
	New Zealand**	Solomon Is	6.9	10	Solomon Is	3.4	31			
		Oceania, unallocated	6.8	10	Timor-Leste	3.0	27			
		Samoa	6.4	9	Oceania, unallocated	2.0	18			
	Norway	Unspecified by region	176.5	50	Unspecified by region	147.2	62			
		Malawi	18.9	5	Malawi	12.5	5			
		Ethiopia	16.1	5	South Sudan	10.8	5			
	Poland	Ukraine	57.7	56	Ukraine	1.0	66			
		Belarus	22.2	22	Georgia	0.1	8			
		India	3.4	3	United Republic of Tanzania	0.1	6			

			EDUCATION		BASIC EDUCATION					
Donor		Recipient	Constant 2019 US\$ millions	Recipient %	Recipient	Constant 2019 US\$ millions	Recipient %			
	Portugal	Timor-Leste	12 5	22	Guinea-Bissau	0.0	73			
	rorcogui	Mozambique	11.6	22	Mozambique	0.0	25			
		Cabo Verde	10.0	18	Cabo Verde	0.0	1			
	Republic of Korea	Unspecified by region	34.6	15	Unspecified by region	5.9	11			
	Republic of Rorea	Viet Nam	13.7	6	Philippines	2.6	5			
		Muanmar	10.8	5	lordan	2.5	5			
	Romania*	Republic of Moldova	34.5	79	Unspecified by region	0.0	100			
	Homania	Serbia	21	5		0.0	100			
		Ukraine	13	3						
	Saudi Arabia	Faunt	88.9	37	Yemen	20	97			
	Subarniublu	Vemen	36.5	13	Malausia	0.2	8			
		Philippines	24.5	9	Lebanon	0.0	0			
	Slovakia	Serhia	10	25	Lebanon	0.0	33			
	Siovakia	Kenua	0.9	23	Kenua	01	18			
		likraine	0.5	Q	Iran	0.1	15			
	Clavania	Bosnia and Horzogovina	4.0	24	Cambia	0.0	100			
	Siuveilla	North Macadonia	4.0	27	Gambia	0.0	100			
		Sorbia	3.0	32						
	Capin	Serbia	2.1	12	11-1+1	1.4	16			
	эраш	Morocco	0.5	12		1.4	0			
		Dispective D C	4.5	о Г		0.7	0			
	<b>C</b> . 1	Bolivia, P. S.	3.0	5	D. R. Congo	0.6	1			
	Sweden	Unspecified by region	46.4	35	Unspecified by region	39.6	64			
		United Republic of Tanzania	22.9	1/	Argnanistan	14.3	23			
	C h l l	Afghanistan	22.1	1/	Syrian Arab Republic	2.4	4			
	Switzerland	Unspecified by region	26.8	20	Unspecified by region	4.8	15			
		Sub-Saharan Africa, unallocated	7.3	6	Niger	4.0	12			
		Niger	6.5	5	Mali	3.8	11			
	Turkey	Unspecified by region	84.9	32	Bosnia and Herzegovina	0.4	18			
		Europe/N. America, unallocated	41.1	16	Republic of Moldova	0.4	18			
		Kyrgyzstan	36.8	14	Albania	0.2	11			
	United Arab Emirates*	Jordan	21.2	25	United Republic of Tanzania	0.6	10			
		Unspecified by region	11.6	14	Comoros	0.5	10			
		Palestine	7.9	9	Unspecified by region	0.4	7			
	United Kingdom**	Unspecified by region	343.7	36	Unspecified by region	162.4	43			
		Pakistan	155.9	16	Pakistan	73.5	20			
		Bangladesh	46.0	5	Lebanon	19.9	5			
	United States	Unspecified by region	208.6	13	Unspecified by region	190.1	15			
		Afghanistan	99.4	6	Jordan	74.8	6			
		Jordan	89.2	6	Pakistan	52.2	4			
Multilateral	African Development Fund	Uganda	21.7	21	Benin					
		Unspecified by region	15.5	15	Namibia					
		Ghana	14.8	14	Equat. Guinea					
	Asian Development Bank	Bangladesh	150.3	46	Bangladesh	60.3	83			
		Nepal	47.8	15	Nepal	11.9	16			
		Viet Nam	43.8	13	Marshall Islands	0.3	0			
	EU Institutions	Unspecified by region	354.5	32	Unspecified by region	90.1	39			
		Turkey	86.7	8	Bangladesh	21.5	9			
		Europe/N. America, unallocated	47.2	4	Nepal	14.1	6			
	UN Relief and Works Agency	Palestine	289 5	67	Palestine	289 5	67			
	for Palestine Refugees	lordan	102.9	77	lordan	107.9	77			
		Lehanon	48 3	10	Lebanon	48 3	10			
			6.0	0	India	76	0			
	UNICLI	D. N. Congo	61	7	Afabanistan	2.0	3			
		Ethiopia	0.1	7	Nigor	1.2	4			
	World Dapk	Danaladash	4.5	5	Randadash	1.2	4			
	(International Development		324.ŏ	12	Ethiopia	122.7	UC			
	Association)	Ethiopia	1/3.9	CI 0		03.0 70 E	17			
			103.4	0	DUUId	10.0	1/			

Source: OECD-DAC, CRS database (2021).

## Glossary

**Attainment rate.** Number of persons in a particular age group by the highest level of education attained, expressed as a percentage of the total population in that age group (see **Completion rate**).

**Age-specific enrolment ratio.** Enrolment of a given age or age group, regardless of the level of education in which pupils or students are enrolled, expressed as a percentage of the population of the same age or age group. An example is global indicator 4.2.2, the participation rate in organized learning (one year before the official primary entry age).

**Completion rate.** Percentage of children aged three to five years older than the official age of entry into the last grade of an education level who have reached the last grade of that level. For example, the primary completion rate in a country with a 6-year cycle where the official age of entry into the last grade is 11 years is the percentage of 14- to 16-year-olds who have reached grade 6.

**Conflict-affected country.** For a given year, any country with 1,000 or more battle-related deaths (including fatalities among civilians and military actors) over the preceding 10-year period and/or more than 200 battle-related deaths in any 1 year over the preceding 3-year period, according to the Uppsala Conflict Data Program Battle-Related Deaths Dataset.

**Constant price.** Price of a particular item adjusted to remove the overall effect of general price changes (inflation) since a given baseline year.

**Early childhood care and education.** Services and programmes that support children's survival, growth, development and learning – including health, nutrition and hygiene, and cognitive, social, emotional and physical development – from birth to entry into primary school.

**Early Childhood Development Index.** Index of fulfilment of developmental potential that assesses children aged 36 to 59 months in four domains: literacy/ numeracy, and physical, social-emotional, and cognitive development. The information is collected through the UNICEF Multiple Indicator Cluster Surveys. A child is 'on track' overall if it is 'on track' in at least three of the four domains. The index is currently being revised. Education levels according to the International Standard Classification of Education (ISCED), which is the classification system designed to serve as an instrument for assembling, compiling and presenting comparable indicators and statistics of education both within countries and internationally. The system, introduced in 1976, was revised in 1997 and 2011.

- Pre-primary education (ISCED level 0). Programmes at the initial stage of organized instruction, primarily designed to introduce very young children, aged at least 3 years, to a school-type environment and provide a bridge between home and school. Upon completion of these programmes, children continue their education at ISCED 1 (primary education).
- Primary education (ISCED level 1). Programmes generally designed to give pupils a sound basic education in reading, writing and mathematics, and an elementary understanding of subjects such as history, geography, sciences, art and music.
- Secondary education (ISCED levels 2 and 3). Lower secondary education (ISCED 2) is generally designed to continue the basic programmes of the primary level but the teaching is typically more subject-focused, requiring more specialized teachers for each subject area. The end of this level often coincides with the end of compulsory education. Teaching in upper secondary education (ISCED 3) is often organized even more along subject lines and teachers typically need a higher or more subject-specific qualification.
- Post-secondary non-tertiary education (ISCED level 4). It provides learning experiences building on secondary education, preparing for labour market entry as well as tertiary education.
- Tertiary education (ISCED levels 5–8). It builds on secondary education, providing learning activities in specialized fields of education. It aims at learning at a high level of complexity and specialization. It comprises:
  - Level 5: Short-cycle tertiary education, often designed to provide participants with professional knowledge, skills and competences. It is practically based and occupationally specific, and prepares students to enter the labour market.

- Level 6: Bachelor's, often designed to provide participants with intermediate academic and/or professional knowledge, skills and competences, leading to a first degree or equivalent qualification.
- Level 7: Master's or equivalent level, often designed to provide participants with advanced academic and/or professional knowledge, skills and competences, leading to a second degree or equivalent qualification.
- Level 8: Doctoral or equivalent level, designed primarily to lead to an advanced research qualification.

**Education for Sustainable Development.** A type of education that aims to enable learners to constructively and creatively address present and future global challenges and create more sustainable and resilient societies.

**Global Citizenship Education.** A type of education that aims to empower learners to assume active roles to face and resolve global challenges and to become proactive contributors to a more peaceful, tolerant, inclusive and secure world.

**Gross domestic product (GDP).** The value of all final goods and services produced in a country in one year.

**Gross enrolment ratio.** Enrolment in a specific level of education, regardless of age, expressed as a percentage of the population in the official age group corresponding to this level of education. It can exceed 100% because of early or late entry and/or grade repetition.

**Gross intake rate.** Total number of new entrants to a given grade of primary education, regardless of age, expressed as a percentage of the population at the official school entrance age for that grade.

**Gross national income.** The value of all final goods and services produced in a country in one year (GDP) plus income that residents have received from abroad, minus income claimed by non- residents.

Information and communication technology skills. Individuals are considered to have such skills if they have undertaken certain computer-related activities in the last three months: copying or moving a file or folder; using copy and paste tools to duplicate or move information within a document; sending emails with attached files; using basic arithmetic formulas in a spreadsheet; connecting and installing new devices; finding, downloading, installing and configuring software; creating electronic presentations with presentation software; transferring files between a computer and other devices; and writing a computer program using a specialized programming language.

Literacy. According to UNESCO's 1958 definition, the term refers to the ability of an individual to read and write with understanding a simple short statement related to his/her everyday life. The concept of literacy has since evolved to embrace several skill domains, each conceived on a scale of different mastery levels and serving different purposes.

**Literacy rate.** Number of literate people in a particular age group, expressed as a percentage of the total population in that age group.

- Adult. Aged 15 and above.
- Youth. Aged 15 to 24.

Minimum proficiency level. Benchmark of basic knowledge in mathematics and reading, measured through learning assessments. Until such time as common standards are validated by the international community or countries, the definitions of minimum proficiency published by agencies specialized in cross-national learning assessments are being used.

**Net attendance rate.** Number of students in the official age group for a given level of education who attend school at that level, expressed as a percentage of the population in that age group.

**Net enrolment rate.** Enrolment of the official age group for a given level of education, expressed as a percentage of the population in that age group. There are two additional variations of this indicator:

- Adjusted net enrolment rate. Enrolment of the official age group for a given level of education either at that level or the levels above, expressed as a percentage of the population in that age group.
- Total net enrolment rate. Enrolment of the official age group in any level of education, expressed as a percentage of the population in that age group.

**New entrants.** Students entering a given level of education for the first time; the difference between enrolment and repeaters in the first grade of the level.

Never been to school rate. Percentage of children aged three to five years older than the official entrance age into primary education who have never been to school. For example, in a country where the official entrance age is 6 years, the indicator is calculated over the age group 9 to 11 years.

**Out-of-school number.** Those not enrolled, defined over the following populations:

- Children of official primary school age.
- Adolescents of official lower secondary school age.
- Youth of official upper secondary school age.

**Out-of-school rate.** Those of the official age group for a given level of education not enrolled, expressed as a percentage of the population in that age group.

**Over-age for grade rate.** The percentage of students in each level of education (primary, lower secondary and upper secondary) who are two years or more above the intended age for their grade.

**Parity index.** A measure of inequality defined as the ratio of the values of an education indicator of two population groups. Typically, the numerator is the value of the disadvantaged group and the denominator is the value of the advantaged group. An index value between 0.97 and 1.03 indicates parity. A value below 0.97 indicates disparity in favour of the advantaged group. A value above 1.03 indicates disparity in favour of the disadvantaged group. An adjusted parity index is symmetrical around 1 and limited to a range between 0 and 2. Groups can be defined by:

- Gender. Ratio of female to male values of a given indicator.
- Location. Ratio of rural to urban values of a given indicator.
- Wealth/income. Ratio of the poorest 20% to the richest 20% of a given indicator.

**Private institutions.** Institutions that are not operated by public authorities but are controlled and managed, whether for profit or not, by private bodies such as non-government organizations, religious bodies, special interest groups, foundations or business enterprises.

**Public expenditure on education.** Total current and capital expenditure on education by local, regional and national governments for public and private institutions.

**Pupil/teacher ratio.** Average number of pupils per teacher at a specific level of education.

**Purchasing power parity.** An exchange rate adjustment that accounts for price differences between countries, allowing international comparisons of real output and income.

**Qualified teacher.** Teacher who has the minimum academic qualification necessary to teach at a specific level of education in a given country.

**Teacher attrition rate**. Number of teachers at a given level of education leaving the profession in a given school year, expressed as a percentage of teachers at that level and in that school year.

#### Technical and vocational education and training.

Programmes designed mainly to prepare students for direct entry into a particular occupation or trade (or class of occupations or trades).

**Trained teacher.** Teacher who has fulfilled at least the minimum organized teacher-training requirements (preservice or in-service) to teach a specific level of education according to national policy or law.

**Transition rate.** Number of new entrants to the first grade of an education level in a given year, expressed as a percentage of the number of students who were enrolled in the final grade of the previous education level in the previous year and who do not repeat that grade the following year.

## Abbreviations

ADB	Asian Development Bank
AIDS	Acquired immunodeficiency syndrome
ALE	Adult learning and education
ALS	Alternative Learning System (Philippines)
ANEP	Administración Nacional de Educación Pública (Uruguay)
ASEAN	Association of Southeast Asian Nations
ASER	Annual Status of Education Report (India)
BNCC	Base Nacional Comum Curricular (Brazil)
CARICOM	Caribbean Community
CCE	Climate change education
CEO	Chief executive officer
CERES	Centro de Estudios de la Realidad Económica y Social (Uruguay)
CLC	Community learning centre
COVID-19	Corona Virus Disease of 2019
CPF	Comparative Panel File
CRS	Creditor Reporting System (OECD)
CSO	Civil society organization
DAC	Development Assistance Committee (OECD)
DAFI	Deutsche Akademische Flüchtlings Initiative Albert Einstein (scholarships)
DET	Duolingo English Test
DFID	Department for International Development (UK, to 2 September 2020)
DHS	Demographic and Health Survey
EBEP	Ethnic Basic Education Provider (Myanmar)
ECCE	Early childhood care and education
ECDI	Early Childhood Development Index
EFA	Education for All
EOF	Education Outcomes Fund
ESC	Education Contracting Services (Philippines)
EU	European Union
Eurostat	Statistical office of the European Union
FAO	Food and Agriculture Organization of the United Nations
FCI	Facility condition index (Canada)
GBC-Education	Global Business Coalition for Education
GCE	Global Campaign for Education
GCPEA	Global Coalition to Protect Education from Attack
GDP	Gross domestic product
GEM Report	Global Education Monitoring Report
GEO	Global Education Observatory

GER	Gross enrolment ratio
GNI	Gross national income
GPE	Global Partnership for Education
GPEDC	Global Partnership for Effective Development Cooperation
GPI	Gender parity index
GSHS	Global School-based Student Health Survey
HBSC	Health Behaviour in School-Aged Children (survey)
HEPA	High-efficiency particulate air (filters)
HIV	Human immunodeficiency virus
IAEG	Inter-agency and Expert Group (on SDG Indicators)
IBE	International Bureau of Education (UNESCO)
ICCS	International Civic and Citizenship Education Study
ICETEX	Instituto Colombiano de Crédito Educativo y Estudios Técnicos en el Exterior
ICF	International Classification of Functioning, Disability and Health (WHO)
ICH	Intangible Cultural Heritage (UNESCO)
ICILS	International Computer and Information Literacy Study
ICT	Information and communication technology
IEA	International Association for the Evaluation of Educational Achievement
IFC	International Finance Corporation (World Bank)
ILO	International Labour Office/Organization
IMF	International Monetary Fund
ISCED	International Standard Classification of Education
ISSP	International Social Survey Programme
ITU	International Telecommunication Union
LLECE	Laboratorio Latinoamericano de Evaluación de la Calidad de la Educación
LTTE	Liberation Tigers of Tamil Eelam (Sri Lanka)
MDG	Millennium Development Goal
MECCE	Monitoring and Evaluating Climate Communication and Education
MICS	Multiple Indicator Cluster Surveys
MILO	Monitoring Impacts on Learning Outcomes (Africa)
NCERT	National Council of Educational Research and Training (India)
NER	Net enrolment rate
NGO	Non-governmental organization
ODA	Official development assistance
OECD	Organisation for Economic Co-operation and Development
OHCHR	Office of the United Nations High Commissioner for Human Rights
PAL	People's Action for Learning
PASEC	Programme d'analyse des systèmes éducatifs de la CONFEMEN
PCR	Polymerase chain reaction (testing)
PEER	Profiles Enhancing Reviews in Education
PFI	Private finance initiative

ΡΙΑΑΟ	Programme for the International Assessment of Adult Competencies (OECD)
PIRLS	Progress in International Reading Literacy Study
PISA	Programme for International Student Assessment (OECD)
PLANALFA	Plan Nacional de Alfabetización (Costa Rica)
PNAD	Pesquisa Nacional por Amostra de Domicílios (Brazil)
PPP	Public-private partnership (in text) or purchasing power parity (in statistical tables)
PRONATEC	Programa Nacional de Acesso ao Ensino Técnico e Emprego (Brazil)
PSL	Partnership Schools for Liberia
RTE Act	Right to Education Act (India; Right of Children to Free and Compulsory Education)
SCOPE	Scoping Progress in Education
SDG	Sustainable Development Goal
SEA-PLM	Southeast Asia Primary Learning Metrics
SEED	Sustainable Education & Enterprise Development (Nigeria)
SERAT	Sexuality Education Review and Assessment Tool (UNESCO)
SHS VP	Senior High School Voucher Program (Philippines)
SPLM	Sudan People's Liberation Movement
STEP	Skills Toward Employment and Productivity (World Bank)
TALIS	Teaching and Learning International Survey (OECD)
TCG	Technical Cooperation Group
TIMSS	Trends in International Mathematics and Science Study
TOEFL	Test of English as a Foreign Language
TVET	Technical and vocational education and training
UIL	UNESCO Institute for Lifelong Learning
UIS	UNESCO Institute for Statistics
UK	United Kingdom
UN	United Nations
UNAIDS	Joint United Nations Programme on HIV/AIDS
UNCRC	United Nations Convention on the Rights of the Child
UNDP	United Nations Development Programme
UNESCO	United Nations Educational, Scientific and Cultural Organization
UNHCR	United Nations High Commissioner for Refugees
UNICEF	United Nations Children's Fund
UNPD	United Nations Population Division
UNSD	United Nations Statistical Division
UOE	UIS/OECD/Eurostat
US	United States
USAID	United States Agency for International Development
VIEW	Visualizing Indicators of Education for the World
WASH	Water, sanitation and hygiene
WHO	World Health Organization (United Nations)
WIDE	World Inequality Database on Education

#### 485 ANNEX ·

.

## References

#### **THEMATIC SECTION**

#### **CHAPTER 1**

- Adamson, F., Aubry, S., de Koning, M. and Dorsi, D. (eds). 2021. *Realizing the Abidjan Principles on the Right to Education: Human Rights, Public Education, and the Role of Private Actors in Education.* Cheltenham, UK, Edward Elgar. (NORRAG Series on International Education and Development.)
- Banerji, R. and Chavan, M. 2016. Improving literacy and math instruction at scale in India's primary schools: the case of Pratham's Read India program. *Journal of Educational Change*, Vol. 17, No. 4, pp. 453–75.
- Baum, D., Lewis, L., Lusk-Stover, O. and Patrinos, H. 2014. What Matters Most for Engaging the Private Sector in Education: A Framework Paper. Washington, DC, World Bank. (SABER Working Paper 8.)
- Busemeyer, M. R. 2012. Inequality and the political economy of education: an analysis of individual preferences in OECD countries. *Journal of European Social Policy*, Vol. 22, No. 3, pp. 219–40.
- Busemeyer, M. R., Garritzmann, J. L. and Neimanns, E. 2020. A Loud but Noisy Signal? Public Opinion and Education Reform in Western Europe. Cambridge, UK, Cambridge University Press.
- CHIP Training and Consulting. 2020. Report of Profiles of Slums/Underserved Areas of 8 Largest Cities of Pakistan. Islamabad, UNICEF and Gavi.
- Chu, J. and Channa, A. 2019. Evaluation of the 'PEAS-DES Inspect & Improve' Project: Baseline Report. Slough, UK, National Foundation for Educational Research.
- Edlund, J. and Lindh, A. 2021. Popular Support for Public Education in Global Perspective Paris, UNESCO. (Background Paper for Global Education Monitoring Report 2021/2.)
- Emmerson, B. 2020. Expert Report on the Right to Education and the Abidjan Principles. London, Global Schools Forum.
- Escola Nova 21. 2020. Escola Nova 21. Barcelona, Spain, Escola Nova 21. www.escolanova21.cat/escola-nova-21-en. (Accessed 17 August 2021.)
- Fuller, B. F. and Elmore, R. 1996. Who Chooses? Who Loses? Culture, Institutions and the Unequal Effects of School Choice. New York, Teachers College Press.
- Gonzalez, I. 2014. Private schools gain popularity in Cuba. Al Jazeera, 4 January. www.aljazeera.com/features/2014/1/4/ private-schools-gain-popularity-in-cuba. (Accessed 17 August 2021.)
- Green, A. 2013. Education and State Formation: Europe, East Asia and the USA. London, Palgrave Macmillan.
- Hui, M. D. 2019. Tenfold increase in private education cost over last decade in North Korea. *Daily NK.* www.dailynk.com/ english/tenfold-increase-in-private-education-cost-over-last-decade-in-north-korea. (Accessed 17 August 2021.)
- IBE. 2006. Thinkers on Education. Geneva, Switzerland, UNESCO International Bureau of Education. www.ibe.unesco. org/en/document/thinkers-education. (Accessed 17 August 2021.)
- Le Grand, J. 2003. Motivation, Agency, and Public Policy: Of Knights and Knaves, Pawns and Queens. Oxford, UK, Oxford University Press.
- McCall, L. 2016. Political and Policy Responses to Problems of Inequality and Opportunity: Past, Present, and Future. Kirsch, I. and Braun, H. (eds), *The Dynamics of Opportunity in America: Evidence and Perspectives*. Cham, Switzerland, Springer.
- OECD. 2019. PISA 2018 Results (Volume II): Where All Students Can Succeed. Paris, Organisation for Economic Co-operation and Development.
- OHCHR. 1999. General Comment No. 13: The Right to Education. New York, United Nations Committee on Economic, Social and Cultural Rights.
- Singh, K. 2015. Report of the Special Rapporteur on the Right to Education: Protecting the Right to Education against Commercialization. New York, Human Rights Council, United Nations General Assembly. (A/HRC/29/30.)
- Skelton, A., Nolan, A., Mowbray, J., Kothari, J., Sepúlveda, M., Smirnova, M., Zinigrad, R., Fredman, S. and Epal Ratjen, S.
   2019. The Abidjan Principles: Guiding Principles on the Human Rights Obligations of States to Provide Public Education and to Regulate Private Involvement in Education. Abidjan, Côte d'Ivoire.
- Srivastava, P. 2020. Framing Non-state Engagement in Education. Paris, UNESCO. (Think piece for Global Education Monitoring Report 2021/2.)
- Stanfield, J. 2021. Parental Choice and the Right to Education: Revisiting Article 26 of the Universal Declaration of Human Rights. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)

Street Child. 2021. Construction, Curriculum and Instruction: Critical Non-State Services Supporting Children in Conflict and Crisis Contexts. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)

Swift, A. 2003. How Not to Be a Hypocrite: School Choice for the Morally Perplexed Parent. London, Routledge. UNESCO. 2015. Incheon Declaration and Framework for Action for the Implementation of Sustainable Development Goal 4.

Paris, UNESCO.

United Nations. 1948. Universal Declaration of Human Rights. New York, United Nations.

\_\_\_\_. 1966. International Covenant on Economic, Social and Cultural Rights. New York, United Nations.

USAID, Results for Development and Education in Crisis and Conflict Network. 2018. Affordable Non-State Schools in El Salvador. Washington, DC, US Agency for International Development.

Winthrop, R. 2018. Leapfrogging Inequality: Remaking Education to Help Young People Thrive. Washington, DC, Brookings Institution.

#### **CHAPTER 2**

Abayasekara, A. 2018. Shadow Education in Sri Lanka. Colombo, Institute of Policy Studies.

Abdulkadiroğlu, A., Pathak, P. A., Schellenberg, J. and Walters, C. R. 2020. Do parents value school effectiveness? American Economic Review, Vol. 110, No. 5, pp. 1502–39.

Acholla, D. A. 2021. Low-fee Private Schooling: A Framework to Define the Scope of these Non-state Actors in Education. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)

Afghanistan Ministry of Education. 2016. National Education Strategic Plan 2017–2021. Kabul, Ministry of Education.

Alcott, B. and Rose, P. 2015. Schools and learning in rural India and Pakistan: Who goes where, and how much are they learning? Prospects, Vol. 45, No. 3, pp. 345–63.

Anderson, M. L., Gallagher, J. and Ritchie, E. R. 2018. School meal quality and academic performance. Journal of Public Economics, Vol. 168, pp. 81–93.

Andrews, J. 2016. School Performance and Leadership: School Performance in Multi-academy Trusts and Local Authorities – 2015. London, Education Policy Institute.

Ansari, A. 2021. Collaboration or competition? Evaluating the impact of public private partnerships (PPPs) on public school enrolment. International Journal of Educational Research, Vol. 107.

Asadullah, M. N. 2018. Madrasah for girls and private school for boys? The determinants of school type choice in rural and urban Indonesia. International Journal of Educational Development, Vol. 62, pp. 96–111.

ASER. 2019. Annual Status of Education Report (Rural) 2018. New Delhi, ASER Centre.

Austin, M. J. 2020. Charter school competition. Berends, M., Primus, A. and Springer, M. G. (eds), Handbook of Research on School Choice. New York, Routledge, pp. 146–159.

Azam, M., Kingdon, G. and Wu, K. B. 2016. Impact of private secondary schooling on cognitive skills: evidence from India. *Education Economics*, Vol. 24, No. 5, pp. 465–80.

Bahous, R., Bacha, N. N. and Nabhani, M. 2011. Multilingual educational trends and practices in Lebanon: a case study. International Review of Education, Vol. 57, No. 5, pp. 737–49.

Bakshi, P. 2020. Unpacking Inclusion in Education: Lessons from Afghanistan for Achieving SDG 4. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2020.)

Bamattre, R. 2019. Bordering state and society: community schools in Zambia. Clothey, R. and Heidemann, K. (eds), Another Way: Decentralization, Democratization and the Global Politics of Community-Based Schooling. Leiden, Netherlands, Brill, pp. 97–114. (Pittsburgh Studies in Comparative and International Education.)

Bangladesh Directorate of Primary Education. 2019. Bangladesh Primary Education Annual Sector Performance Report 2019. Dhaka, Directorate of Primary Education.

Barakat, S., Hardman, F., Rohwerder, B. and Rzeszut, K. 2014. The Evidence for the Sustainable Scale-up of Low-cost Private Schools in South West Asia. London, Evidence for Policy and Practice Information and Co-ordinating Centre, Social Science Research Unit, Institute of Education, University of London.

Bates, A., Choi, T.-H. and Kim, Y. 2019. Outsourcing education services in South Korea, England and Hong Kong: a discursive institutionalist analysis. Compare: A Journal of Comparative and International Education, Vol. 51, No. 2, pp. 259–77.

Baum, D. R. and Riley, I. 2019. The relative effectiveness of private and public schools: evidence from Kenya. School Effectiveness and School Improvement, Vol. 30, No. 2, pp. 104–30.

Berry, J. and Mukherjee, P. 2019. Pricing Private Education in Urban India: Demand, Use and Impact. New Delhi, J-PAL South Asia.

Bhorkar, S. and Bray, M. 2018. The expansion and roles of private tutoring in India: from supplementation to supplantation. International Journal of Educational Development, Vol. 62, pp. 148–56.

- Böhlmark, A. and Lindahl, M. 2015. Independent schools and long-run educational outcomes: evidence from Sweden's large-scale voucher reform. *Economica*, Vol. 82, No. 327, pp. 508–51.
- Borghans, L., Golsteyn, B. H. and Zölitz, U. 2015. Parental preferences for primary school characteristics. The BE Journal of Economic Analysis & Policy, Vol. 15, No. 1, pp. 85–117.
- Borkowski, A., Sunny, B. and Zapata, J. 2021. Non-state Education in South Asia: Understanding the Effect of Non-State Actors Engagement on the Quality, Equity, and Safety of Education Service Delivery. Florence, Italy, UNICEF Office of Research – Innocenti.
- Bose, S., Ghosh, P. and Sardana, A. 2020. Exit at the Bottom of the Pyramid: Empirical Explorations in the Context of Elementary Schooling in Delhi. New Delhi, National Institute of Public Finance and Policy. (NIPFP Working Paper 306.)
- Botta Somparé, E. and Wotem Somparé, A. 2018. La condition enseignante en Guinée: des stratégies de survie dans le champ scolaire et universitaire guinéen [Teacher conditions in Guinea: survival strategies in Guinean schools and universities]. Cahiers de la recherche sur l'éducation et les savoirs, No. 17, pp. 21–45.
- Brandt, K. 2018. Private Beats Public: A Flexible Value-added Model with Tanzanian School Switchers. Helsinki, United Nations University World Institute for Development Economics Research (WIDER Working Paper 2018/081.)
- Bray, M. 2017. Schooling and its supplements: changing global patterns and implications for comparative education. Comparative Education Review, Vol. 61, No. 3, pp. 469–91.
- Bray, M., Kobakhidze, M. N. and Kwo, O. 2020. Shadow Education in Myanmar: Private Supplementary Tutoring and Its Policy Implications. Hong Kong, China, University of Hong Kong. (Comparative Education Research Centre Monograph in Comparative and International Education and Development 13.)
- Broussard, M. 2014. Why poor schools can't win at standardized testing. *The Atlantic.* www.theatlantic.com/education/ archive/2014/07/why-poor-schools-cant-win-at-standardized-testing/374287. (Accessed 8 June 2021.)
- BSIC. 2018. A New Class Representative: Kroton to Acquire Somos for \$1.8bn. Milan, Italy, Bocconi Students Investment Club. https://bsic.it/new-class-representative-kroton-acquire-somos-1-8bn. (Accessed 8 June 2021.)
- Buchbinder, N. 2020. Edtech in Latin America: Have we Been Successful in Expanding ICT Availability and Use Through Schools? Paris, UNESCO. (Global Education Monitoring Report Fellowship Paper.)
- Bunnell, T., Donnelly, M. and Lauder, H. 2020. The clustering in 'global universities' of graduates from 'Elite Traditional International Schools': a surprising phenomenon? *Globalisation, Societies and Education*, Vol. 19, No. 5, pp. 558–67.
- Burkholder, K. 2020. Education for street-connected children in Kenya: marginalization, challenges, and recommendations. Watzlawik, M. and Burkholder, A. (eds), *Educating Adolescents Around the Globe: Becoming Who You Are in a World Full of Expectations.* Cham, Switzerland, Springer, pp. 97–115.
- Cambridge Education. 2021. Can Private Schools Improve Equitable Access to Schooling for Vulnerable Families? Perceptions and Attitudes from Parents in Kinshasa and Lagos. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Campaign for Popular Education. 2005. Education Watch Report 2003/4: Quality with Equity The Primary Education Agenda. Dhaka, Campaign for Popular Education.
- Canada Government. 2018. Building a National Textbook and Book Industry. Ottawa, Government of Canada. www.international.gc.ca/world-monde/stories-histoires/2019/mali-book-livre.aspx?lang=eng (Accessed 17 August 2021.)
- Central Bank of Sri Lanka. 2020. Annual Report (Volume 1): Economic and Social Infrastructure. Colombo, Central Bank of Sri Lanka.
- Chakrani, B. 2017. Between profit and identity: analyzing the effect of language of instruction in predicting overt language attitudes in Morocco. Applied Linguistics, Vol. 38, No. 2, pp. 215–33.
- Cherng, S. and Barch, F. 2021. Teachers' Working Conditions and Job Satisfaction Differences in State and Nonstate Schools: An Analysis of TALIS 2018 Data. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Chesters, J. 2018. The marketisation of education in Australia: does investment in private schooling improve postschool outcomes? Australian Journal of Social Issues, Vol. 53, No. 2, pp. 139–57.
- Chilangwa, B. 2019. Four Key Approaches to Scaling Community Schools in Zambia. Washington, DC, Brookings Institution. www.brookings.edu/blog/education-plus-development/2019/08/08/four-key-approaches-to-scalingcommunity-schools-in-zambia. (Accessed 12 August 2021.)

Chingos, M. M., Kuehn, D., Monarrez, T., Wolf, P. J., Witte, J. F. and Kisida, B. 2019. The Effects of Means-Tested Private School Choice Programs on College Enrollment and Graduation. Washington, DC, Urban Institute.

Choi, Y. 2018. The heterogeneous effects of shadow education on SAT scores. *Development and Society*, Vol. 47, No. 3, pp. 451–76.

Chudgar, A., Chandra, M. and Razzaque, A. 2014. Alternative forms of teacher hiring in developing countries and its implications: a review of literature. *Teaching and Teacher Education*, Vol. 37, pp. 150–61.

Clark Tuttle, C., Gill, B., Gleason, P., Knechtel, V., Nichols-Barrer, I. and Resch, A. 2013. KIPP Middle Schools: Impacts on Achievement and Other Outcomes. Washington, DC, Mathematica Policy Research.

Cohn, R. M. 2020. Effects of Public-school Choice on Private Schools: Evidence from Open Enrollment Reform. Waterloo, Ontario, Canadian Labour Economics Forum. (Conference Volume CLEF Working Paper 23.)

Cole, R. 2017. Estimating the impact of private tutoring on academic performance: primary students in Sri Lanka. Education Economics, Vol. 25, No. 2, pp. 142–57.

Competition Policy International. 2018. Brazil: CADE clears purchase of Somos by Kroton. Boston, Mass., Competition Policy International. www.competitionpolicyinternational.com/brazil-cade-approves-without-restriction-the-purchase-of-somos-by-kroton. (Accessed 8 June 2021.)

Cordes, S. A. 2018. In pursuit of the common good: the spillover effects of charter schools on public school students in New York City. Education Finance and Policy, Vol. 13, No. 4, pp. 484–512.

Cortina, R. 2017. Indigenous Education Policy, Equity, and Intercultural Understanding in Latin America. New York, Palgrave Macmillan.

Costa, K. 2021. Projeto que permite educação domiciliar é aprovado na CCJ da Câmara [Draft law that allows homeschooling is approved by the Chamber's Constitution and Justice Commission]. Brasília, Radioagência Nacional. https:// agenciabrasil.ebc.com.br/radioagencia-nacional/educacao/audio/2021-06/projeto-que-permite-educacaodomiciliar-e-aprovado-na-ccj-da-camara. (Accessed 17 August 2021.)

Côte d'Ivoire Ministry of National Education. 2020. Liste des Manuels Agrees et Recommandes: Annee Scolaire 2020–2021 [List of Approved and Recommended Textbooks: School Year 2020–2021]. Abidjan, Department of Pedagogy and Continuing Education, Ministry of National Education, Technical Education and Professional Training.

Crawfurd, L. 2017. School management and public-private partnerships in Uganda. *Journal of African Economies*, Vol. 26, No. 5, pp. 539–60.

\_\_\_\_. 2018. Contracting Out Schools at Scale: Evidence from Pakistan. Oxford, UK, Research on Improving Systems of Education. (RISE Working Paper 18/022.)

Crawfurd, L., Patel, D. and Sandefur, J. 2021. Heterogeneous effects of low-cost private schools: experimental evidence from Delhi. Washington, DC, Centre for Global Development. (Unpublished.)

d'Aiglepierre, R. and Bauer, A. 2018. The choice of Arab-Islamic education in sub-Saharan Africa: findings from a comparative study. International Journal of Educational Development, Vol. 62, pp. 47–61.

Daun, H. 2018. Islamic education around the world: commonalities and varieties. Daun, H. and Arjmand, R. (eds), Handbook of Islamic Education. Cham, Switzerland, Springer, pp. 529–53.

Davis, M. and Heller, B. 2019. No excuses charter schools and college enrollment: new evidence from a high school network in Chicago. Education Finance and Policy, Vol. 14, No. 3, pp. 414–40.

Day Ashley, L., Mcloughlin, C., Aslam, M., Engel, J., Wales, J., Rawal, S., Batley, R., Kingdon, G., Nicolai, S. and Rose, P.
 2014. The Role and Impact of Private Schools in Developing Countries: A Rigorous Review of the Evidence. London, Department for International Development. (Education Rigorous Literature Review.)

de Andrade Silverio, G. and Araújo de Sousa, A. 2014. Organic foods from family farms in the National School Food Program: perspectives of social actors from Santa Catarina, Brazil. *Revista de Nutrição*, Vol. 27, No. 3, pp. 289–300.

de Lesseux, W. 2018. Sénégal: les éditeurs locaux à la peine sur le marché des manuels scolaires [Senegal: local publishers struggling in the textbook market]. Paris, Radio France Internationale. www.rfi.fr/fr/emission/20181003-senegalediteurs-locaux-peine-marche-manuels-scolaires. (Accessed 17 August 2021.)

De Talancé, M. 2020. Private and public education: do parents care about school quality? Annals of Economics and Statistics, No. 137, pp. 117–44.

Delprato, M. and Chudgar, A. 2018. Factors associated with private-public school performance: analysis of TALIS-PISA link data. International Journal of Educational Development, Vol. 61, pp. 155–72.

Dixon, P., Egalite, A. J., Humble, S. and Wolf, P. J. 2019. Experimental results from a four-year targeted education voucher program in the slums of Delhi, India. *World Development*, Vol. 124, art. 104644.

Dixon, P. and Humble, S. 2017. How school choice is framed by parental preferences and family characteristics: a study of Western Area, Sierra Leone. *Journal of School Choice*, Vol. 11, No. 1, pp. 95–110.

- Doherty, C. and Dooley, K. 2018. Responsibilising parents: the nudge towards shadow tutoring. British Journal of Sociology of Education, Vol. 39, No. 4, pp. 551–66.
- Dunning, C. 2013. Is local spending better? The controversy over USAID procurement reform. Washington, DC, Center for American Progress. www.americanprogress.org/article/is-local-spending-better. (Accessed 17 August 2021.)
- Edmonds, R. and Macloeo, S. 2021. COVID-19 and Street-Connected Children: Impacts, Responses and Opportunities. London, Consortium for Street Children.
- Edwards Jr, D. B. and Termes, A. 2019. 'Concession Schools' in Bogotá: the limits of economic efficiency of charter programs. Revista Colombiana de Educación, No. 76, pp. 91–116.
- Egalite, A. J. and Wolf, P. J. 2016. A review of the empirical research on private school choice. Peabody Journal of Education, Vol. 91, No. 4, pp. 441-54.
- Eggleston, C. and Field, J. 2021. Census Bureau's Household Pulse Survey Shows Significant Increase in Homeschooling Rates in Fall 2020. Washington, DC, US Census Bureau. www.census.gov/library/stories/2021/03/homeschooling-onthe-rise-during-covid-19-pandemic.html. (Accessed 17 August 2021.)
- Eigbiremolen, G. O. 2020. Estimating private school premium for primary school children in Ethiopia: evidence from individual-level panel data. *Progress in Development Studies*, Vol. 20, No. 1, pp. 26–44.
- Eigbiremolen, G. O., Ogbuabor, J. E. and Nwambe, C. S. 2020. Estimating private school effects for school children in Peru: evidence from individual-level panel data. *Journal of International Development*, Vol. 32, No. 2, pp. 131–49.
- Elacqua, G., Schneider, M. and Buckley, J. 2006. School choice in Chile: Is it class or the classroom? Journal of Policy Analysis and Management, Vol. 25, No. 3, pp. 577–601.
- Ellison, S. and Aloe, A. M. 2019. Strategic thinkers and positioned choices: parental decision making in urban school choice. *Educational Policy*, Vol. 33, No. 7, pp. 1135–70.
- Epple, D., Romano, R. E. and Urquiola, M. 2017. School vouchers: a survey of the economics literature. *Journal of Economic Literature*, Vol. 55, No. 2, pp. 441–92.
- Espindola, J. 2019. Low-fee private schools in developing nations: some cautionary remarks. Global Justice: Theory Practice Rhetoric, Vol. 12, No. 1, pp. 55–77.
- Evans, D. K., Yuan, F. and Filmer, D. 2020. Are Teachers in Africa Poorly Paid? Evidence from 15 Countries. Washington, DC, World Bank. (Policy Research Working Paper 9358.)
- Feldhoff, T. 2017. Child poverty in a rich country: measuring and influencing policies in contemporary Japan. Education About Asia, Vol. 22, No. 3, pp. 20–26.
- Ferrare, J. J. 2020. Charter school outcomes. Berends, M., Primus, A. and Springer, M. G. (eds), Handbook of Research on School Choice. New York, Routledge, pp. 160–73.
- FICCI and Nielsen India. 2016. K–12 Book Publishing Market Report 2016. New Delhi, Federation of Indian Chambers of Commerce and Industry/Nielsen India Private Limited.
- Figlio, D. and Karbownik, K. 2016. Evaluation of Ohio's EdChoice Scholarship Program: Selection, Competition, and Performance Effects. Columbus, Ohio, Thomas B. Fordham Institute.
- Fitz, J. A. and Nikolaidis, A. 2020. A democratic critique of scripted curriculum. *Journal of Curriculum Studies*, Vol. 52, No. 2, pp. 195–213.
- Fleming, D. J., Lavertu, S. and Crawford, W. 2018. High school options and post-secondary student success: the Catholic school advantage. *Journal of Catholic Education*, Vol. 21, No. 2, pp. 1–25.
- Frontlist. 2019. Private Publishers Are Concerned about the Educational Textbook Market. New Delhi, Frontlist. www.frontlist. in/private-publishers-are-concerned-about-the-educational-textbook-market. (Accessed 4 June 2021.)
- Gaddis, J. E. 2020. The big business of school meals. Kappan, Vol. 102, No. 2.
- Gary, N. 2018. Par-delà les frontières, "la francophonie est une aubaine pour l'édition" [Beyond borders, «the Francophone community is a boon for publishing»]. Paris, ActuaLitté. https://actualitte.com/article/20633/interviews/par-dela-les-frontieres-la-francophonie-est-une-aubaine-pour-l-edition. (Accessed 17 August 2021.)
- GCE. 2016. Private Profit, Public Loss: Why the Push for Low-fee Private Schools Is Throwing Quality Education Off Track. Johannesburg, Global Campaign for Education.
- Gemignani, R., Shojo, M. and Wodon, Q. 2014. What drives the choice of faith-inspired schools by households? Qualitative evidence from two African countries. The Review of Faith and International Affairs, Vol. 12, No. 2, pp. 66–76.
- Gerrard, J. and Barron, R. 2020. Cleaning public education: the privatisation of school maintenance work. Journal of Educational Administration and History, Vol. 52, No. 1, pp. 9–21.

- Green, F., Anders, J., Henderson, M. and Henseke, G. 2020. Private benefits? External benefits? Outcomes of private schooling in 21st century Britain. *Journal of Social Policy*, Vol. 49, No. 4, pp. 724–43.
- Guill, K., Lüdtke, O. and Köller, O. 2020. Assessing the instructional quality of private tutoring and its effects on student outcomes: analyses from the German National Educational Panel Study. *British Journal of Educational Psychology*, Vol. 90, No. 2, pp. 282–300.
- Hachette Livre. 2019. Publishers 2019. Vanves, France, Lagardère.
- Hajar, A. 2018. Exploring year 6 pupils' perceptions of private tutoring: evidence from three mainstream schools in England. Oxford Review of Education, Vol. 44, No. 4, pp. 514–31.
- Hamid, A. F. A. 2017. Islamic education in Malaysia. Daun, H. and Arjmand, R. (eds), Handbook of Islamic Education. Cham, Switzerland, Springer, pp. 745–61.
- Hammad, W. and Shah, S. 2018. Dissonance between the 'international' and the conservative 'national': challenges facing school leaders in international schools in Saudi Arabia. *Educational Administration Quarterly*, Vol. 54, No. 5, pp. 747–80.
- Hares, S. and Crawfurd, L. 2021. The Role and Impact of Private Schools, School Chains and PPPs in Low and Middle Income Countries. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Härmä, J. 2021. Low-fee Private Schooling and Poverty in Developing Countries. London, Bloomsbury.
- Harris, A. B. and Brunjes, B. M. 2021. Developing capacity and reducing risk? An analysis of federal international contracting. International Public Management Journal.
- Hernández, M. 2019. White middle-class families and sociocultural diversity in schools: a literature review. Urban Review, Vol. 51, No. 2, pp. 270–300.
- Hofflinger, A., Gelber, D. and Cañas, S. T. 2020. School choice and parents' preferences for school attributes in Chile. Economics of Education Review, Vol. 74, art. 101946.
- Hofflinger, A. and von Hippel, P. T. 2020. Does achievement rise fastest with school choice, school resources, or family resources? Chile from 2002 to 2013. Sociology of Education, Vol. 93, No. 2, pp. 132–52.
- Hogan, A., Thompson, G., Sellar, S. and Lingard, B. 2018. Teachers' and school leaders' perceptions of commercialisation in Australian public schools. *Australian Educational Researcher*, Vol. 45, No. 2, pp. 141–60.
- Hohman, J. and Slasinski, C. 2019. Michigan School Privatization Survey 2018. Midland, Mich., Mackinac Center for Public Policy.
- Holmström, C. 2020. Friskolor i aktiebolagsform [Independent schools in the form of limited companies]. Stockholm, Economifakta. www.ekonomifakta.se/Fakta/Valfarden-i-privat-regi/Skolan-i-privat-regi/friskolor-iaktiebolagsform. (Accessed 21 July 2021.)
- Hsieh, C.-T. and Urquiola, M. 2006. The effects of generalized school choice on achievement and stratification: evidence from Chile's voucher program. *Journal of Public Economics*, Vol. 90, No. 8–9, pp. 1477–503.
- Hsu, K. J. and Schuller, M. 2020. Humanitarian aid and local power structures: lessons from Haiti's 'shadow disaster'. *Disasters*, Vol. 44, No. 4, pp. 641–65.
- Humanity and Inclusion. 2021. Sahel: working toward inclusive education for girls with disabilities. Silver Spring, Md., Humanity and Inclusion. www.hi-us.org/news\_africa\_sahel\_region\_inclusive\_education\_girls\_disabilities. (Accessed 12 August 2021.)
- IEA. 2021. Public and Private Secondary School: International Variation in Social Composition, Resources and Student Literacy from Two International Comparative Studies. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- ISC. 2021. ISC Census and Annual Report 2021. London, Independent Schools Council.
- ISC Research. 2021. Data on international schools. Faringdon, UK, ISC Research. https://iscresearch.com/data. (Accessed 17 August 2021.)
- ISSEK. 2018. 10 facts about Russian teachers. Moscow, Institute for Statistical Studies and Economics of Knowledge, Higher School of Economics University. https://issek.hse.ru/en/news/225245595.html. (Accessed 17 August 2021.)
- Jabbar, H. 2018. Recruiting 'talent': school choice and teacher hiring in New Orleans. Educational Administration Quarterly, Vol. 54, No. 1, pp. 115–51.
- Jabbar, H. and Lenhoff, S. W. 2020. Parent decision-making and school choice. Berends, M., Primus, A. and Springer, M. G. (eds), Handbook of Research on School Choice. New York, Routledge, pp. 351–64.
- Jabbar, H. and Li, D. 2016. Multiple choice: how public school leaders in New Orleans's saturated market view private school competitors. *Education Policy Analysis Archives*, Vol. 24, art. 94.

Jackson, C. K. 2012. School competition and teacher labor markets: evidence from charter school entry in North Carolina. *Journal of Public Economics*, Vol. 96, No. 5–6, pp. 431–48.

Jackson, G. M. 2017. Common themes in Australian and New Zealand home education research. Gaither, M. (ed.), The Wiley Handbook of Home Education, Chichester, UK, John Wiley & Sons, pp. 326–61.

Jayachandran, S. 2014. Incentives to teach badly: after-school tutoring in developing countries. Journal of Development Economics, Vol. 108, pp. 190–205.

Johnson, D. and Hsieh, P. J. 2019. A longitudinal study of learning, progression and personal growth in Sierra Leone. Oxford, UK, University of Oxford.

Joshi, P. 2014a. Continuing to exercise choice after selecting a school: insights into parent decision-making in local education markets in Nepal. International Journal of Educational Development, Vol. 37, pp. 57–67.

\_\_\_\_. 2014b. Parent decision-making when selecting schools: the case of Nepal. Prospects, Vol. 44, No. 3, pp. 411–28.

\_\_\_\_. 2016. Experiencing and responding to private competition: the importance of subjectivity and intermediate outcomes. *Comparative Education Review*, Vol. 60, No. 3, pp. 571–600.

\_\_\_\_. 2020. Do private schools improve public school quality or increase stratification? International Journal of Educational Development, Vol. 77, art. 102219.

Kamal, K. 2018. Education System Profiles: Education in the United Arab Emirates. New York, World Education Services.

- Kaplan, V. 2004. History teaching in post-Soviet Russia: coping with antithetical traditions. Eklof, B., Holmes, L. E. and Kaplan, V. (eds), Educational Reform in Post-Soviet Russia. London, Routledge, pp. 263–87.
- Kim, S. and Hong, S. 2018. The effects of school contexts and student characteristics on cognitive and affective achievement in South Korea. *Asia Pacific Education Review*, Vol. 19, No. 4, pp. 557–72.
- Kim, Y. C. and Jung, J.-H. 2019. Shadow Education as Worldwide Curriculum Studies. Cham, Switzerland, Springer.

Kippels, S. and Ridge, N. 2019. International and Other Migrant Schools in Gulf Cooperation Council Countries. Paris, UNESCO. (Background paper for Global Education Monitoring Report on Migration, Displacement and Education in the Arab States 2019.)

Klemm, K. and Hollenbach-Biele, N. 2016. Nachhilfeunterricht in Deutschland: Ausmaß – Wirkung – Kosten [Tutoring in Germany: Extent – Impact – Costs]. Gütersloh, Germany, Bertelsmann Stiftung.

Kornhall, P. and Bender, G. 2019. School Segregation in Sweden: Evidence from the Local Level. Vilnius, PPMI Group. (NESET Ad Hoc Report 1/2019.)

Krafft, C., Elbadawy, A. and Sieverding, M. 2019. Constrained school choice in Egypt. International Journal of Educational Development, Vol. 71, art. 102104.

- Kumar, D. and Choudhury, P. K. 2020. Determinants of private school choice in India: all about the family backgrounds? *Journal of School Choice*, Vol. 15, No. 4, pp. 576–602.
- Kumar, S. M. 2018. Comparing private and government schools in India: the devil is in the maths. Applied Economics *Letters*, Vol. 25, No. 6, pp. 409–14.
- Kunzman, R. and Gaither, M. 2020. Homeschooling: an updated comprehensive survey of the research. Other Education, Vol. 9, No. 1, pp. 253–336.
- L.E.K. Consulting. 2020. Private Schools for Public Goods: Exploring the Potential of Privately-run Schools to Benefit Societies. Boston, Mass., L.E.K. Consulting.
- Lange, M.-F., Lauwerier, T. and Locatelli, R. 2021. The Impact of Privatization on Teachers in Francophone sub-Saharan African Countries. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)

Languille, S. 2016. The scramble for textbooks in Tanzania. African Affairs, Vol. 115, No. 458, pp. 73–96.

- LaRocque, N. 2008. Public–Private Partnerships in Basic Education: An International Review. Reading, UK, CfBT Education Trust.
- Lauwerier, T. and Akkari, A. 2019. Les enseignants d'Afrique de l'Ouest francophone face à des approches curriculaires pensées pour des contextes exogènes: le cas de l'approche par compétences au Burkina Faso et au Sénégal [Teachers in francophone West Africa faced with curricular approaches meant for exogenous contexts: the case of the competency-based approach in Burkina Faso and Senegal]. *Formation et profession*, Vol. 27, No. 1, pp. 5–19.
- Lavado, P., Cueto, S., Yamada, G. and Wensjoe, M. 2019. The effect of a charter school in Peru (Fe y Alegría) on school achievement: exploiting a school lottery selection as a natural experiment. *Studies in Economics and Econometrics*, Vol. 43, No. 3, pp. 95–110.
- Lewin, K. 2007. The Limits to Growth of Non-government Private Schooling in sub-Saharan Africa. Brighton, UK, Consortium for Research on Educational Access, Transitions and Equity. (CREATE Research Monograph 5.)
- Lipcan, A., Crawfurd, L. and Law, B. 2018. Learning in Lagos: Comparing Student Achievement in Bridge, Public and Private Schools. Abuja, EDOREN.
- Macdonald, D., Johnson, R. and Lingard, B. 2020. Globalisation, neoliberalisation, and network governance: an international study of outsourcing in health and physical education. *Discourse: Studies in the Cultural Politics of Education*, Vol. 41, No. 2, pp. 169–86.
- Maietta, O. W. and Gorgitano, M. T. 2016. School meals and pupil satisfaction: evidence from Italian primary schools. Food Policy, Vol. 62, pp. 41–55.
- Marcotte, D. E. and Dalane, K. 2019. Socioeconomic segregation and school choice in American public schools. Educational Researcher, Vol. 48, No. 8, pp. 493–503.
- Marshall, J. H. and Fukao, T. 2019. Shadow education and inequality in lower secondary schooling in Cambodia: understanding the dynamics of private tutoring participation and provision. *Comparative Education Review*, Vol. 63, No. 1, pp. 98–120.
- Martin, J.-Y. 2003. Les écoles spontanées en Afrique subsaharienne : champ éducatif et contre-champ scolaire [Spontaneous schools in sub-Saharan Africa: education and counter-education]. *Cahiers d'études africaines*, Vol. 43, No. 169–70, pp. 19–39.
- Martin, J. and Dunlop, L. 2019. For-profit schools in England: the state of a nation. *Journal of Education Policy*, Vol. 34, No. 5, pp. 726–41.
- Martindale, N. 2019. Does outsourcing school systems degrade education workforces? Evidence from 18,000 English state schools. British Journal of Sociology of Education, Vol. 40, No. 8, pp. 1015–36.
- Menashy, F. and Zakharia, Z. 2017. Investing in the Crisis: Private Participation in the Education of Syrian Refugees. Brussels, Education International. (Education International Research Paper.)
- \_\_\_\_. 2021. Private Engagement in Education in Emergencies: Rights and Regulations. New York, Inter-agency Network for Education in Emergencies.
- Mishra, N. 2019. Schools continue to use pvt publishers' books. *The Tribune*, 2 January. www.tribuneindia.com/news/ archive/chandigarh/schools-continue-to-use-pvt-publishers-books-707889. (Accessed 4 June 2021.)
- Morrison, J. R., Ross, S. M. and Cheung, A. C. 2019. From the market to the classroom: how ed-tech products are procured by school districts interacting with vendors. *Educational Technology Research and Development*, Vol. 67, No. 2, pp. 389–421.
- Moskwa, D. 2019. How to profit from education in Russia. New Eastern Europe, Vol. 35, No. 1, pp. 77-82.
- Mottaleb, K. A. and Pallegedara, A. 2019. Spending privately for education in Nepal: Who spends more on it and why? International Journal of Educational Development, Vol. 69, pp. 39–47.
- Mufti, I. 2017. NCERT books full of errors, say Chandigarh schools. *Hindustan Times*, 20 February. www.hindustantimes. com/punjab/ncert-books-full-of-errors-say-chandigarh-schools/story-PMcWXKqoiRUj4lCdooifOH.html. (Accessed 17 August 2021.)
- Mullis, I. V. S., Martin, M. O., Goh, S. and Prendergast, C. 2017. PIRLS 2016 Encyclopedia: Education Policy and Curriculum in Reading. Boston, Mass., Boston College, TIMSS & PIRLS International Study Center.
- Narwana, K. 2019. Hierarchies of access in schooling: an exploration of parental school choice in Haryana. *Millennial Asia*, Vol. 10, No. 2, pp. 183–203.
- Nazaryan, L. 2021. Armenia. Paris, UNESCO. (Information gathering template prepared for the Global Education Monitoring Report on Inclusion and Education in Central and Eastern Europe, the Caucasus and Central Asia 2021.)
- Niakaté, H. 2018. Les livres scolaires en Côte d'Ivoire, un business qui vire au casse-tête [Textbooks in Côte d'Ivoire, a business that turns into a puzzle]. *Le Monde*, 13 February. www.lemonde.fr/afrique/article/2018/02/13/leslivres-scolaires-en-cote-d-ivoire-un-business-qui-vire-au-casse-tete\_5256326\_3212.html. (Accessed 17 August 2021.)
- Nigeria UBEC. 2019. 2018 Natonal Personnel Audit (NPA) Report on Public and Private Basic Education Schools in Nigeria. Abuja, Universal Basic Education Commission.
- Nkeck Bidias, S. R. 2019. Le Manuel Scolaire au Cameroun: Enjeux, Diagnostic et Esquisses de Solutions pour une Politique Sectorielle [Textbooks in Cameroon: Issues, Diagnosis and Outlines of Solutions for a Sectoral Policy]. Yaoundé, Syndicat National Autonome de l'Enseignement Secondaire/Friedrich Ebert Stiftung.
- Nsabimana, J. d. D. 2018. Education board moves to create publishing house for textbooks. The New Times, 28 December. www.newtimes.co.rw/section/read/226483. (Accessed 17 August 2021.)
- OECD. 2017. Ukraine. Paris, Organisation for Economic Co-operation and Development. (OECD Reviews of Integrity in Education.)

- \_\_\_\_. 2019a. Balancing School Choice and Equity: An International Perspective Based on PISA. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2019b. PISA 2018 Results: What School Life Means for Students' Lives Volume III. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2020a. Education at a Glance 2020: OECD Indicators. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2020b. PISA 2018 Results: Effective Policies, Successful Schools Volume V. Paris, Organisation for Economic Co-operation and Development.
- Ömeroğulları, M., Guill, K. and Köller, O. 2020. Effectiveness of private tutoring during secondary schooling in Germany: Do the duration of private tutoring and tutor qualification affect school achievement? *Learning and Instruction*, Vol. 66, art. 101306.
- Pakistan Ministry of Federal Education and Professional Training. 2021. Pakistan Education Statistics 2017-18. Islamabad, Ministry of Federal Education and Professional Training.
- Pallegedara, A. and Mottaleb, K. A. 2018. Patterns and determinants of private tutoring: the case of Bangladesh households. International Journal of Educational Development, Vol. 59, pp. 43–50.
- Pane, J. 2018. Strategies for Implementing Personalized Learning While Evidence and Resources Are Underdeveloped. Arlington, Va., RAND Corporation.
- Parkin, M. and Irby, P. 2019. Education Contracting Analysis: Selling to K–12 and Higher Education Markets. Herndon, Va., Deltek.
- Patel, D. and Sandefur, J. 2020. A *Rosetta Stone for Human Capital*. Washington, DC, Center for Global Development. (CGD Working Paper 550.)
- Pearson. 2019. Focused on Delivery: Pearson Annual Report and Accounts 2019. London/New York, Pearson.
- Permoser, J. M. and Stoeckl, K. 2020. Reframing human rights: the global network of moral conservative homeschooling activists. *Global Networks*, Vol. 21, No. 4, pp. 681–702.
- Petts, R. J., Carlson, D. L. and Pepin, J. R. 2020. A gendered pandemic: childcare, homeschooling, and parents' employment during COVID-19. *Gender, Work and Organization*, Vol. 28, No. S2, pp. 515–34.
- Pianta, R. C. and Ansari, A. 2018. Does attendance in private schools predict student outcomes at age 15? Evidence from a longitudinal study. *Educational Researcher*, Vol. 47, No. 7, pp. 419–34.
- Population Council. 2020. Abriendo Oportunidades Strengthens Government Education Programming for Indigenous Adolescent Girls in Guatemala. New York, Population Council. (Research Utilization and Impact.)
- Prosveshcheniye. 2017. Prosveshcheniye: Transforming a Publishing House into an Educational Holding Solutions for National and Regional Educational Challenges. Saint Petersburg, Russian Federation, Prosveshcheniye. https://prosv.ru/\_data/pages/29/prosvescheniye\_presentation\_eng.pdf. (Accessed 8 June 2021.)

Rahman, M. A. 2016. Low-Cost Private Schools: A Case Study in Jakarta. Jakarta, Center for Indonesian Policy Studies.

- Ramachandran, V. and Walz, J. 2015. Haiti: where has all the money gone? *Journal of Haitian Studies*, Vol. 21, No. 1, pp. 26–65.
- Rao, S. S. 2017. Production of an 'educational'city: shadow education economy and re-structuring of Kota in India. Pink, W. T. and Noblit, G. W. (eds), Second International Handbook of Urban Education. Cham, Switzerland, Springer, pp. 417–43.
- Ravitch, D. 2020. Chicago Public Schools decides to kick out privatized custodial services. *Diane Ravitch's Blog: A Site to Discuss Better Education for All.* New York, Diane Ravitch. https://dianeravitch.net/2020/05/23/chicago-public-schools-decides-to-kick-out-privatized-custodial-services. (Accessed 10 February 2021.)
- Ray, B. D. 2017. A systematic review of the empirical research on selected aspects of homeschooling as a school choice. *Journal of School Choice*, Vol. 11, No. 4, pp. 604–21.
- Read, T. 2015. Where Have All the Textbooks Gone? Toward Sustainable Provision of Teaching and Learning Materials in Sub-Saharan Africa. Washington, DC, World Bank. (Directions in Development: Human Development.)
- Rho, E. 2013. Contracting revisited: determinants and consequences of contracting out for public education services. Public Administration Review, Vol. 73, No. 2, pp. 327–37.
- Riep, C. 2015. Corporatised Education in the Philippines: Pearson, Ayala Corporation, and the emergence of affordable private education centers (APEC). Brussels, Education International.
- Riep, C. and Machacek, M. 2016. Schooling the Poor Profitably: The Innovations and Deprivations of Bridge International Academies in Uganda. Brussels, Education International.
- Riep, C. B. 2017. Making markets for low-cost schooling: the devices and investments behind Bridge International Academies. *Globalisation, Societies and Education, Vol.* 15, No. 3, pp. 352–66.

Rogez, O. 2020. La lente émergence du livre scolaire africain [The slow emergence of the African textbook]. Paris, Radio France Internationale. www.rfi.fr/fr/podcasts/20200904-la-lente-émergence-livre-scolaire-africain. (Accessed 17 August 2021.)

- Rohde, L. A., Campani, F., Oliveira, J. R. G., Rohde, C. W., Rocha, T. and Ramal, A. 2019. Parental reasons for school choice in elementary school: a systematic review. *Journal of School Choice*, Vol. 13, No. 3, pp. 287–304.
- Rolleston, C. and Moore, R. 2018. Young Lives School Survey, 2016–17: Value-added Analysis in India. Oxford, UK, University of Oxford.
- Romero, M., Sandefur, J. and Sandholtz, W. A. 2020. Outsourcing education: experimental evidence from Liberia. American Economic Review, Vol. 110, No. 2, pp. 364–400.
- Rose, P. M. 2007. Supporting Non-state Providers in Basic Education Service Delivery. Brighton, UK, Consortium for Research on Educational Access, Transitions and Equity. (CREATE Research Monograph 4.)
- S. Chand and Company Ltd. 2021. S. Chand Business Segments. New Delhi, S. Chand and Company Ltd. https://schandgroup.com/segments. (Accessed 8 June 2021.)
- Sabarwal, S., Sununtnasuk, C. and Ramachandran, D. 2020. Low-Cost Private Schools in Tanzania: A Descriptive Analysis. Washington, DC, World Bank. (Policy Research Working Paper 9360.)
- Sakellariou, C. 2017. Private or public school advantage? Evidence from 40 countries using PISA 2012-Mathematics. Applied Economics, Vol. 49, No. 29, pp. 2875–92.
- Santos, H. and Elacqua, G. 2016. Socioeconomic school segregation in Chile: parental choice and a theoretical counterfactual analysis. CEPAL Review, No. 119, pp. 123–37.
- Scheunpflug, A. and Wenz, M. 2021. Faith Based Non-State Actors in Selected African Countries: A Comparative Analysis of Catholic and Protestant Schooling in the Democratic Republic of Congo, Rwanda, Tanzania, Madagascar and Cameroon. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Sellar, S. and Hogan, A. 2019. Pearson 2025: Transforming Teaching and Privatising Education Data. Brussels, Education International.
- Sieverding, M., Krafft, C. and Elbadawy, A. 2019. An exploration of the drivers of private tutoring in Egypt. Comparative Education Review, Vol. 63, No. 4, pp. 562–90.
- Singh, R. 2021. Teachers' Working Conditions in State and Non-state Schools. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Sivasubramaniam, M. 2021. Philanthropic and Faith-based Non-state actors in the Low-fee Private School Sector in the Global South: Blurring the Lines of Philanthropy and Enterprise. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Smart, A. and Jagannathan, S. 2018. Textbook Policies in Asia: Development, Publishing, Printing, Distribution, and Future Implications. Manila, Asian Development Bank.
- Smith, E. and Nelson, J. 2015. Using the Opinions and Lifestyle Survey to examine the prevalence and characteristics of families who home educate in the UK. *Educational Studies*, Vol. 41, No. 3, pp. 312–25.
- Sperka, L. 2020. (Re)defining outsourcing in education. Discourse: Studies in the Cultural Politics of Education, Vol. 41, No. 2, pp. 268–80.
- Srivastava, P. 2007. Neither Voice nor Loyalty: School Choice and the Low-Fee Private Sector in India. New York, National Center for the Study of Privatization in Education. (Occasional Paper 134.)
- \_\_\_\_. 2020. Framing non-state engagement in education. Paris, UNESCO. (Think piece for Global Education Monitoring Report 2021/2.)
- Šťastný, V., Chvál, M. and Walterová, E. 2021. An ordinary moonlighting activity? Determinants of the provision of private tutoring by Czech schoolteachers. *International Journal of Educational Development*, Vol. 81, No. 2, art. 102351.
- Stern, J. M. B. and Heyneman, S. P. 2013. Low-fee private schooling: the case of Kenya. Srivastava, P. (ed.), Low Fee Private Schooling: Aggravating Equity or Mediating Disadvantage. Oxford, UK, Symposium Books, pp. 105–28. (Oxford Series in Comparative Education.)
- Street Child. 2021. Construction, Curriculum and Instruction: Critical Non-state Services Supporting Children in Conflict and Crisis Contexts. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Sullivan, A., Parsons, S., Green, F., Wiggins, R. D., Ploubidis, G. and Huynh, T. 2018. Educational attainment in the short and long term: Was there an advantage to attending faith, private, and selective schools for pupils in the 1980s? Oxford Review of Education, Vol. 44, No. 6, pp. 806–22.
- Sun, L., Shafiq, M. N., McClure, M. and Guo, S. 2020. Are there educational and psychological benefits from private supplementary tutoring in Mainland China? Evidence from the China Education Panel Survey, 2013–15. International Journal of Educational Development, Vol. 72, art. 102144.

- Sutton Trust. 2019. Private Tuition Polling 2019. London, Sutton Trust. www.suttontrust.com/our-research/private-tuition-polling-2019. (Accessed 17 August 2021.)
- Termes, A., Edwards Jr, D. B. and Verger, A. 2020. The development and dynamics of public–private partnerships in the Philippines' education: a counterintuitive case of school choice, competition, and privatization. *Educational Policy*, Vol. 34, No. 1, pp. 91–117.
- Thierry, R. 2020. Quand l'édition africaine s'émancipe [When African publishing is emanicpated]. Paris, Institut national de l'audiovisuel. https://larevuedesmedias.ina.fr/quand-ledition-africaine-semancipe. (Accessed 17 August 2021.)
- Timberlake, M. T., Thomas, A. B. and Barrett, B. 2017. The allure of simplicity: scripted curricula and equity. *Teaching and Teacher Education*, Vol. 67, pp. 46–52.
- Ting, S.-H. and Lee, D. P.-Y. 2019. Determinants of primary school choice in Malaysia: school proximity and ethnicityrelated reasons. *Journal of School Choice*, Vol. 13, No. 2, pp. 228–54.
- Tooley, J. 2013. School Choice in Lagos State. Newcastle, UK, Newcastle University.
- Tooley, J. and Longfield, D. 2016. Affordability of private schools: exploration of a conundrum and towards a definition of 'low-cost'. Oxford Review of Education, Vol. 42, No. 4, pp. 444–59.
- Trinh, K. 2019. New Textbooks for Grade 1 Students Approved. Hanoi, Viet Nam News. https://vietnamnews.vn/ society/548893/new-textbooks-for-grade-1-students-approved.html. (Accessed 17 August 2021.)
- US Department of Education. 2019. Homeschooling in the United States: Results from the 2012 and 2016 Parent and Family Involvement Survey (PFI-NHES: 2012 and 2016). Washington, DC, National Center for Education Statistics.
- Ugbodaga, K. 2018. 11,895 private schools in Lagos are illegal LASG. P.M. News, 25 May. www.pmnewsnigeria. com/2018/05/25/11895-private-schools-in-lagos-are-illegal-lasg. (Accessed 17 August 2021.)
- UNESCO. 2013. Module 5: Development of Textbooks and Other Teaching and Learning Materials. IBE (ed.), Training Tools for Curriculum Development: A Resource Pack. Geneva, Switzerland, International Bureau of Education.
- \_\_\_\_. 2019. Global Education Monitoring Report 2019: Migration, Displacement and Education Building Bridges, Not Walls. Paris, UNESCO.
- UNICEF and WHO. 2018. Drinking Water, Sanitation and Hygiene in Schools: Global Baseline Report 2018. New York/Geneva, Switzerland, UNICEF/World Health Organization.
- Urquiola, M. 2016. Competition among schools: traditional public and private schools. Hanushek, E. A., Machin, S. and Woessmann, L. (eds), Handbook of the Economics of Education, Vol. 5. Amsterdam, Elsevier, pp. 209–38.
- USAID. 2019. Non-governmental organizations (NGOs). Washington, DC, US Agency for International Development. www.usaid.gov/partnership-opportunities/ngo. (Accessed 12 August 2021.)
- van der Berg, S., van Wyk, C., Burger, R., Kotzé, J., Piek, M. and Rich, K. 2017. *The Performance of Low Fee Independent Schools in South Africa: What Can Available Data Tell?* Stellenbosch, South Africa, University of Stellenbosch. (Economic Working Paper 01/17.)
- VCIOM. 2019. ЕГЭ как способ поступить в вуз: удобства и сложности [Unified State Exam as a way to enter university: advantages and difficulties]. Moscow, All-Russian Public Opinion Research Center. https://wciom.ru/analyticalreviews/analiticheskii-obzor/ege-kak-sposob-postupit-v-vuz-udobstva-i-slozhnosti. (Accessed 17 August 2021.)
- Verger, A., Fontdevila, C. and Zancajo, A. 2018. Constructing low-fee private schools as an educational model for the Global South: from local origins to transnational dynamics. Verger, A., Novelli, M. and Altinyelken, H. K. (eds), Global Education Policy and International Development: New Agendas, Issues and Policies, 2nd ed. London, Bloomsbury, pp. 255–76.
- Vinayashree, J. 2017. Delayed supply, poor print quality of NCERT books anger parents. *The Times of India*, 26 July. https://timesofindia.indiatimes.com/city/chennai/delayed-supply-poor-print-quality-of-ncert-books-angerparents/articleshow/59762779.cms. (Accessed 17 August 2021.)
- von Stumm, S. and Plomin, R. 2021. Does private education make nicer people? The influence of school type on social–emotional development. *British Journal of Psychology*, Vol. 112, No. 2, pp. 373–88.
- Vrangbæk, K., Petersen, O. H. and Hjelmar, U. 2015. Is contracting out good or bad for employees? A review of international experience. *Review of Public Personnel Administration*, Vol. 35, No. 1, pp. 3–23.
- Walford, G. 2011. Low-fee private schools in England and in less economically developed countries: what can be learnt from a comparison? *Compare*, Vol. 41, No. 3, pp. 401–13.
- Wamalwa, F. M. and Burns, J. 2018. Private schools and student learning achievements in Kenya. Economics of Education Review, Vol. 66, pp. 114–24.
- Wells, A. S., Keener, A., Cabral, L. and Cordova-Cobo, D. 2019. The more things change, the more they stay the same: the resegregation of public schools via charter school reform. *Peabody Journal of Education*, Vol. 94, No. 5, pp. 471–92.

Williamson, B. and Hogan, A. 2020. Commercialisation and Privatisation in/of Education in the Context of COVID-19. Brussels, Education International.

Wischenbart, R. 2019. Global 50: the world ranking of the publishing industry 2019. Publishers Weekly, 5 September.

- Wodon, Q. 2019. More schools, larger schools, or both? Patterns of enrollment growth in K12 Catholic schools globally. Journal of Catholic Education, Vol. 22, No. 1, p. 27.
- \_\_\_\_. 2020a. Are new secondary schools built where they are needed most in Uganda? Comparing Catholic with public and other private schools. The Review of Faith and International Affairs, Vol. 18, No. 2, pp. 44–60.
- \_\_\_\_. 2020b. Global Catholic Education Report 2020: Achievements and Challenges at a Time of Crisis. Rome, International Office of Catholic Education.
- \_\_\_\_. 2020c. How well do Catholic and other faith-based schools serve the poor? A study with special reference to Africa – Part II: Learning. International Studies in Catholic Education, Vol. 12, No. 1, pp. 3–20.
- \_\_\_\_. 2021. Global Catholic Education Report 2021: Education Poverty, Learning Pluralism, and the Right to Education. Washington, DC, Global Catholic Education, International Office of Catholic Education, International Federation of Catholic Universities, World Organization of Former Students of Catholic Education and World Union of Catholic Teachers.
- Woldetsadik, G. and Raysarkar, C. 2017. Textbook Provision for All in Ethiopia: Lessons Learned from the General Education Quality Improvement Project (GEQIP) 1. Washington, DC, World Bank.
- World Bank. 2008. Textbooks and School Library Provision in Secondary Education in Sub-Saharan Africa. Washington, DC, World Bank. (Africa Human Development Series Working Paper 126.)
- Zancajo, A. 2019. Education markets and schools' mechanisms of exclusion: the case of Chile. Education Policy Analysis Archives, Vol. 27, art. 130.
- Zhang, W. 2020. Regulating Private Supplementary Tutoring: Actors, Dynamics, and Lessons from International Experiences. Paris, UNESCO. (Global Education Monitoring Report Fellowship Paper.)
- Zhang, W. and Bray, M. 2018. Equalising schooling, unequalising private supplementary tutoring: access and tracking through shadow education in China. Oxford Review of Education, Vol. 44, No. 2, pp. 221–38.
- Zheng, X., Wang, C., Shen, Z. and Fang, X. 2020. Associations of private tutoring with Chinese students' academic achievement, emotional well-being, and parent-child relationship. *Children and Youth Services Review*, Vol. 112, art. 104934.
- Zook, C. 2017. Infographic: Textbook Costs Skyrocketed 812% in 35 Years. Lancaster, Pa., Applied Education Systems. www.aeseducation.com/blog/infographic-the-skyrocketing-cost-of-textbooks-for-schools-students. (Accessed 9 June 2021.)
- Zuilkowski, S., Piper, B. and Ong'ele, S. A. 2020. Are low-cost private schools worth the investment? Evidence on literacy and mathematics gains in Nairobi primary schools. *Teachers' College Record*, Vol. 122, No. 1.

## **CHAPTER 3**

- Abdous, K. 2020. Privatisation of Education in Morocco: A Multi-speed Education System and a Polarised Society Summary. Brussels, Education International. (Education International Research.)
- Abdulkadiroğlu, A., Angrist, J. D., Hull, P. D. and Pathak, P. A. 2016. Charters without lotteries: testing takeovers in New Orleans and Boston. American Economic Review, Vol. 106, No. 7, pp. 1878–920.
- Afridi, M. 2018. Equity and Quality in an Education Public–Private Partnership: A Study of the World Bank-supported PPP in Punjab, Pakistan. Nairobi, Oxfam. (Oxfam Research Reports.)
- Agence Bujumbura News. 2017. Fermeture des écoles avec moins de 20% de réussites aux évaluations nationales [Schools with less than 20% pass rate in national examinations closed]. Agence Bujumbura News, 7 April. https://bujumburanewsblog.wordpress.com/2017/04/07/fermeture-des-ecoles-avec-moins-de-20-dereussites-aux-evaluations-nationales. (Accessed 17 August 2021.)
- Ahimbisibwe, P. 2018. Govt to stop funding to 800 private USE schools. *Daily Monitor*, 17 January. www.monitor.co.ug/ uganda/news/national/govt-to-stop-funding-to-800-private-use-schools-1736078. (Accessed 17 August 2021.)
- Al Qasimi Foundation for Policy Research. 2021. Non-State Education in Arab States. Paris, Al Qasimi Foundation for Policy Research. (Background paper for Global Education Monitoring Report 2021/2.)
- Altai Consulting. 2018. Study on Understanding the role of Non-State Education Providers in Somalia: Final Report. Paris, Altai Consulting.
- Ambast, S. and Gaur, A. 2017. Legal developments. Sarin, A., Dongre, A. and Wad, S. (eds), *State of the Nation: RTE* Section 12 (1) (C) 2017. Ahmedabad/New Delhi, Indian Institute of Management Ahmedabad/Centre for Policy Research/Central Square Foundation, pp. 64–68.

Ambast, S., Gaur, A. and Sangai, A. 2017. Regulation of Private Schools in India. New Delhi, Vidhi Centre for Legal Policy.

- Andrabi, T., Das, J., Khwaja, A. I., Vishwanath, T. and Zajonc, T. 2007. Learning and Educational Achievements in Punjab Schools (LEAPS): Insights to Inform the Education Policy Debate. Washington, DC, World Bank.
- Ansari, A. H. 2020. Cream skimming? Evaluating the access to Punjab's public–private partnership programs in education. International Journal of Educational Development, Vol. 72, art. 102126.
- Arab News. 2013. Islamic studies to be taught in all expat schools, orders ministry. Arab News, 8 May. www.arabnews.com/news/450843. (Accessed 17 August 2021.)
- Australian Tutoring Association. 2021. *Member code of conduct*. Dundas, NSW, Australian Tutoring Association. https://ata.edu.au/about-us/member-code-of-conduct.
- Baker, J. 2019. Seven years after Gonski, why is school funding still inequitable? Sydney Morning Herald, 17 May. www.smh.com.au/education/seven-years-after-gonski-why-is-school-funding-still-inequitable-20190516p51034.html. (Accessed 17 August 2021.)
- Balarin, M. 2015. Low-fee Private Schools in Peru. Geneva, Switzerland, NORRAG. www.norrag.org/low-fee-private-schools-in-peru. (Accessed 17 August 2021.)
- Balsera, M. R. 2017. Tax, Privatisation and the Right to Education: Influencing Education Financing and Tax Policy to Transform Children's Lives. Johannesburg, South Africa, ActionAid.
- Bangladesh Ministry of Education. 2010. National Education Policy 2010. Dhaka, Ministry of Education.
- Barman, P. D. 2020. Stop private tuition, or else leave jobs, Tripura govt warns school teachers. Hindustan Times, 7 June.
- Barrera-Osorio, F. and Raju, D. 2015. Evaluating the impact of public student subsidies on low-cost private schools in Pakistan. The Journal of Development Studies, Vol. 51, No. 7, pp. 808–25.
- Barrett, N. and Harris, D. 2015. Significant Changes in the New Orleans Teacher Workforce. New Orleans, La., Education Research Alliance for New Orleans. (Policy Brief.)
- Baum, D. R., Cooper, R. and Lusk-Stover, O. 2018. Regulating market entry of low-cost private schools in Sub-Saharan Africa: Towards a theory of private education regulation. *International Journal of Educational Development*, Vol. 60, pp. 100–12.
- Bertoni, M., Gibbons, S. and Silva, O. 2020. School choice during a period of radical school reform: evidence from academy conversion in England. *Economic Policy*, Vol. 35, No. 104, pp. 739–95.
- Béteille, T., Tognatta, N., Riboud, M., Nomura, S. and Ghorpade, Y. 2020. *Ready to Learn: Before School, In School, and Beyond School in South Asia.* Washington, DC, World Bank. (South Asia Development Forum.)
- Binci, M., Rasulova, S., Iversen, V., Outhred, R., Lipcan, A., Bahri, S., Uppal, V., Nnodu, I. and MacAuslan, I. 2016. Developing Effective Private Education Nigeria (DEPEN): Baseline Report. Abuja, Education Data, Research and Evaluation in Nigeria (EDOREN).
- Bishop Accountability. 2021. Bishop Accountability. Waltham, Mass. www.bishop-accountability.org. (Accessed 17 August 2021.)
- Black, S. E. and Machin, S. 2011. Housing valuations of school performance. Handbook of the Economics of Education, Vol. 3. Amsterdam, Elsevier, pp. 485–519.
- Bloomberg News. 2021. China bans for-profit school tutoring in sweeping overhaul. Bloomberg News, 24 July.
- Braithwaite, J., Coglianese, C. and Levi-Faur, D. 2007. Can regulation and governance make a difference? Regulation & Governance, Vol. 1, No. 1, pp. 1–7.
- Brandt, C. and De Herdt, T. 2020. Reshaping the reach of the state: the politics of a teacher payment reform in the DR Congo. Journal of Modern African Studies, Vol. 58, No.1, pp. 23–43.
- Bray, M. and Kwo, O. 2014. Regulating Private Tutoring for Public Good: Policy Options for Supplementary Education in Asia. Hong Kong/Bangkok, Comparative Education Research Centre, University of Hong Kong/UNESCO Regional Bureau for Education in Asia and the Pacific. (CERC Monograph Series in Comparative and International Education and Development No. 10.)
- British Education Suppliers Association. 2021. *Key UK education statistics*. London, British Education Suppliers Association. www.besa.org.uk/key-uk-education-statistics. (Accessed 17 August 2021.)
- Buchanan, N. K. and Fox, R. A. 2008. Every school a school of choice: school choice in Ireland as viewed through American eyes. *Irish Educational Studies*, Vol. 27, No. 3, pp. 267–79.
- Buerger, C. and Harris, D. N. 2020. The impact of government contracting out on spending: the case of public education in New Orleans. The American Review of Public Administration, Vol. 51, No. 2, pp. 139–54.
- Burgess, S., Greaves, E. and Vignoles, A. 2020. School Places: A Fair Choice? School Choice, Inequality and Options for Reform of School Admissions in England. London, Sutton Trust.
- Cambridge Education. 2021. Religious Networks in the DRC: Help or Hindrance to Education? . Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)

Cameron, E. 2019. The sweeping incrementalism of Partnership Schools for Liberia. *Journal of Education Policy*, Vol. 35, No. 6, pp. 856–70.

CapPlus. 2017. Banking on Education: Low-Cost Private Schools' Demand for Finance. Skokie, Ill., CapitalPlusExchange.

Carrasco, A., Gutierrez, G. and Flores, C. 2017. Failed regulations and school composition: selective admission practices in Chilean primary schools. *Journal of Education Policy*, Vol. 32, No. 5, pp. 642–72.

Center for Popular Democracy and Coalition for Community Schools. 2015. System Failure: Louisiana's Broken Charter School Law – Underinvestment in Oversight Leaves Louisiana's Charter Schools Vulnerable to Financial Fraud and Academic Failures. New Orleans, La., Center for Popular Democracy and Coalition for Community Schools.

- Central Square Foundation. 2020. State of the Sector Report on Private Schools in India. New Delhi, Central Square Foundation/Omidyar Network India.
- Centre for Civil Society. 2019. Anatomy of K-12 Governance in India: Evidence from States on Regulation of Private Schools. New Delhi, Centre for Civil Society.

Che, C. 2021. After online tutoring, why is China cracking down on private schools? *SupChina*, 9 September. https://supchina.com/2021/09/09/after-online-tutoring-why-is-china-cracking-down-on-private-schools. (Accessed 1 November 2021.)

Cheruiyot, K. 2019. Government orders closure of unregistered schools countrywide. The Star, 29 September.

- Civil Network OPORA. 2016. Опитування: Як ставляться до зовнішнього незалежного оцінювання його учасники? [Survey: What Do the EIT Participants Think about the EIT?]. Kyiv, Civil Network OPORA.
- Clarke, M. 2021. China's Crackdown on Private Tutoring: How Will This Affect the TEFL Industry? Vital Consular Blog. Batley, UK, Vital Consular. https://blog.vitalconsular.com/china-private-tutoring-ban-trialled-affect-the-teflindustry. (Accessed 17 August 2021.)
- CMEPT. 2014. Rapport alternatif au groupe de pré-session du Comité des Droits Economiques Sociaux et Culturels de l'Organisation des Nations Unies à l'occasion de sa considération d'une liste de questions au Maroc lors de la 55e session du comité [Alternative Report to the Pre-sessional Group of the United Nations Committee on Economic, Social and Cultural Rights on the Occasion of its Consideration of a List of Questions in Morocco during the 55th Session of the Committee]. Rabat, Coalition Marocaine pour l'Education Pour Tous.
- Coco, J., Stovall, A. and Marcell Williams, L. 2020. *Returning to School in the Wake of Disaster: Post-Katrina Lessons for the COVID Era*. New York, Center for Learner Equity. www.centerforlearnerequity.org/news/returning-to-schoolin-the-wake-of-disaster-post-katrina-lessons-for-the-covid-era. (Accessed 17 August 2021.)
- Cohodes, S. R. and Parham, K. S. 2021. Charter Schools' Effectiveness, Mechanisms, and Competitive Influence. Cambridge, Mass., National Bureau of Economic Research. (NBER Working Paper 28477.)
- CONFEMEN. 2010. Rapport PASEC Union des Comores 2008/2009: Diagnostic et Préconisations pour une Scolarisation Universelle de Qualité [2008/2009 PASEC Report on the Union of the Comoros: Diagnosis and Recommendations for Quality Universal Schooling]. Moroni, Ministère de l'Éducation Nationale et de la Recherche.
- Council of Europe. 2018. 12 Principles of Good Governance. Strasbourg, Council of Europe. www.coe.int/en/web/goodgovernance/12-principles. (Accessed 17 August 2021.)
- Crawfurd, L. and Hares, S. 2021. The Role and Impact of Private Schools, School Chains and PPPs in Low and Middle Income Countries. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Crawfurd, L. and Pugatch, T. 2020. Teacher Labor Markets in Developing Countries. Bonn, Germany, Institute of Labor Economics. (IZA Discussion Paper 12985.)
- Cunningham, J. 2014. Accountability in Private School Choice Programs. Washington, DC, National Conference of State Legislatures.
- D'Agostino, T. J., Dowd, R. and Mugo, J. 2019. Faith-Based Education in Changing Social, Economic, and Political Contexts: Perspectives from Catholic Educators in Kenya. *The Review of Faith & International Affairs*, Vol. 17, No. 4, pp. 76–88.
- Day Ashley, L., Skinner, R., Meyer, A. and Perry, T. 2020. Private Education and Disadvantaged Children in India: A Literature Review of Three Models of Private School Provision. Birmingham, UK, Save the Children.
- Deccan Chronicle. 2014. Karnataka Minister wants to shut 1,400 schools. *Deccan Chronicle*, 30 October. www. deccanchronicle.com/141030/nation-current-affairs/article/karnataka-minister-wants-shut-1400-schools. (Accessed 17 August 2021.)
- Department for Education. 2021. School Admissions Code: Mandatory Requirements and Statutory Guidance for Admission Authorities, Governing Bodies, Local Authorities, Schools Adjudicators and Admission Appeals Panels. London, Department for Education.

Deshotels, M. 2015. New Orleans RSD compared to traditional schools. *Lousiana Educator*, 21 May. https://louisianaeducator.blogspot.com/2015/05/new-orleans-rsd-compared-to-traditional.html. (Accessed 17 August 2021.)

Dolton, P., Marcenaro, O., De Vries, R. and She, P.-W. 2018. Global Teacher Status Index 2018. London, Varkey Foundation.

- Dongre, A., Sarin, A. and Singhal, K. 2019. Understanding School Choices under RTE's 25% Mandate. London/Oxford, Ideas for India. www.ideasforindia.in/topics/human-development/understanding-school-choices-under-rte-s-25-mandate1.html. (Accessed 17 August 2021.)
- Doyle, D. M., Muldoon, M. and Murphy, C. 2020. Education in Ireland: accessible without discrimination for all? International Journal of Human Rights, Vol. 24, No. 10, pp. 1701–20.
- Education International. 2017. Privatisation and Its Impact on Teachers' Conditions of Employment. Brussels, Education International.
- Edwards Jr, D. B. and Termes, A. 2018. Los colegios en concesión de Bogotá: los límites de la eficiencia económica de los programas chárter [Concession schools in Bogotá: the limits of charter programmes' economic efficiency]. *Revista Colombiana de Educación*, Vol. 1, No. 76, pp. 91–116.
- Elacqua, G. 2012. The impact of school choice and public policy on segregation: evidence from Chile. International Journal of Educational Development, Vol. 32, No. 3, pp. 444–53.
- Elacqua, G., Iribarren, M. L. and Santos, H. 2018. Private Schooling in Latin America: Trends and Public Policies. Washington, DC, Education Division, Social Sector, Inter-American Development Bank. (IDB Working Paper IDB-TN-01555.)
- Elacqua, G., Jaimovich, A. and Román, A. 2019. The Effects of Accountability on the Allocation of School Resources: Regression Discontinuity Evidence from Chile. Washington, DC, Inter-American Development Bank. (IDB Working Paper IDB-WP-1074.)
- Eyles, A. and Machin, S. 2019. The introduction of academy schools to England's education. Journal of the European Economic Association, Vol. 17, No. 4, pp. 1107–46.
- Eyles, A., Machin, S. and McNally, S. 2017. Unexpected school reform: academisation of primary schools in England. Journal of Public Economics, Vol. 155, pp. 108–21.
- Eyles, A., Machin, S. and Silva, O. 2018. Academies 2: the new batch the changing nature of academy schools in England. *Fiscal Studies*, Vol. 39, No. 1, pp. 121–58.
- Falabella, A. 2020. The ethics of competition: accountability policy enactment in Chilean schools' everyday life. Journal of Education Policy, Vol. 35, No. 1, pp. 23–45.
- France 24. 2021. China's tutoring groups fined millions as scrutiny grows. France 24, 1 June.
- Gaddis, J. E. 2020. The big business of school meals. *Kappan*, Vol. 102, No. 2.
- Gannon, K. 2017. Islamic schools in Pakistan plagued by sex abuse of children. *Associated Press*, 22 November. https://apnews.com/article/pakistan-middle-east-international-news-asia-pacific-islam-ddd9660f63ae44339 66684823f79d3e9. (Accessed 17 August 2021.)
- Goss, P., Sonnemann, J., Griffiths, K. and Chivers, C. 2016. Circuit Breaker: A New Compact on School Funding. Melbourne, Grattan Institute.
- Gumus-Dawes, B., Luce, T. and Orfield, M. 2013. The state of public schools in post-Katrina New Orleans: the challenge of creating equal opportunity. Orfield, G. and Frankenberg, E. (eds), *Educational Delusions? Why Choice Can Deepen Inequality and How to Make Schools Fair.* Berkeley, Calif., University of California Press, pp. 159–85.
- Han, C., Raymond, M. E., Woodworth, J. L., Negassi, Y., Richardson, W. P. and Snow, W. 2020. Lights Off: Practice and Impact of Closing Low-Performing Schools 2017, Volume I. Stanford, Calif., Center for Research on Education Outcomes, Stanford University.
- Hanif, S., Ali, M. H. and Shaheen, F. 2021. Religious extremism, religiosity and sympathy toward the Taliban among students across madrassas and worldly education schools in Pakistan. *Terrorism and Political Violence*, Vol. 33, No. 3, pp. 489–504.
- Härmä, J. 2019. Ensuring quality education? Low-fee private schools and government regulation in three sub-Saharan African capitals. International Journal of Educational Development, Vol. 66, pp. 139–46.
- Härmä, J. and Adefisayo, F. 2013. Scaling up: challenges facing low-fee private schools in the slums of Lagos, Nigeria. Srivastava, P. (ed.), Low-fee Private Schooling: Aggravating Equity or Mitigating Disadvantage? Oxford, Symposium Books, pp. 129–52. (Oxford Studies in Comparative Education.)
- Hassan, S. 2021. Religious Maltreatment, Educational Neglect and Politics. New York, Freedom of Mind Resource Center. https://freedomofmind.com/religious-maltreatment-educational-neglect-politics. (Accessed 17 August 2021.)
- Hernández, M. 2019. Is There No Excuse? The Effects of the New Orleans School Reforms on School Discipline. New Orleans, La., Education Research Alliance for New Orleans.

Honduras Ministry of Education. 2020. Programa Hondureño de Educación Comunitaria–PROHECO [Honduras Community Education Program–PROHECO]. Comayagüela, Ministry of Education. www.transparencia. se.gob.hn/estructura/atribuciones-por-unidad-administrativa/nivel-central/h-programa-hondureño-deeducación-comunitaria-proheco. (Accessed 17 August 2021.)

Hossain, N., Subrahmanian, R. and Kabeer, N. 2002. The Politics of Educational Expansion in Bangladesh. Brighton, UK, Institute of Development Studies. (IDS Working Paper 167.)

Human Rights Watch. 2019. Senegal: Failure to End Abuses in Quranic Schools. New York, Human Rights Watch.

IICSA. 2019. Inquiry Publishes Report into Ealing Abbey and St Benedict's School. London, Independent Inquiry into Child Sexual Abuse. (Accessed 17 August 2021.)

India Comptroller and Auditor General. 2017. Report of the Comptroller and Auditor General of India on Implementation of Right of Children to Free and Compulsory Education Act, 2009. New Delhi, Comptroller and Auditor General. (Report No. 23 of 2017.)

Indus Action. 2018. The Bright Spots: Status of Social Inclusion through RTE Section 12(1)(c) 2018. New Delhi, Indus Action. \_\_\_\_\_. 2019. The Bright Spots: Status of Social Inclusion through RTE Section 12(1)(c) 2019. New Delhi, Indus Action.

International Task Force on Teachers for Education 2030. 2020. A Review of the Use of Contract Teachers in sub-Saharan Africa. Paris, UNESCO.

IPSOS Mori and the Key. 2018. State of Education Survey Report. London, IPSOS Mori/The Key.

Ireland Department of Education and Skills. 2020. Establishing a New School. Dublin, Department of Education and Skills. www.education.ie/en/Schools-Colleges/Information/Establishing-a-New-School. (Accessed 17 August 2021.)

Ireland Government. 2018. Education (Admission to Schools) Act 2018. Dublin, Office of the Attorney General. www.irishstatutebook.ie/eli/2018/act/14/section/11/enacted/en/html. (Accessed 17 August 2021.)

ISER. 2016. A Threat or Opportunity? Public–Private Partnerships in Education in Uganda. Kampala, Initiative for Social and Economic Rights.

Iyer, G. and Counihan, C. 2018. When a right goes wrong: the unintended consequences of India's Right to Education Act. *Economic Affairs*, Vol. 38, No. 3, pp. 367–79.

Jackson, E. 2016. Non-State Actors in Basic Education: Indonesia Case Study. Canberra, Department of Foreign Affairs and Trade. (Education Analytics Service.)

Jha, J. 2019. Right to free and compulsory education in Nepal: a study with special reference to India's Right of Children to Free and Compulsory Education Act, 2009. Dehradun Law Review, Vol. 11, No. 1, pp. 41–57.

Jones, P., Isaacson, J. and Morgan-McDermott, G. 2021. A New Regulatory Landscape for International Schools in China. London, Farrer. www.farrer.co.uk/news-and-insights/a-new-regulatory-landscape-for-international-schoolsin-china. (Accessed 1 November 2021.)

Joshi, R. 2020. Can social integration in schools be mandated: evidence from the Right to Education Act in India. International Journal of Educational Development, Vol. 77, art. 102228.

Kahlenberg, R. D. 2020. Why Taxpayers Should Not Be Forced to Support Private Religious Education. New York, The Century Foundation. https://tcf.org/content/commentary/taxpayers-not-forced-support-private-religiouseducation. (Accessed 17 August 2021.)

Kenny, D. 2020. The virtues of unprincipled constitutional compromises: Church and State in the Irish Constitution. European Constitutional Law Review, Vol. 16, No. 3, pp. 417–39.

Kingdon, G. and Muzammil, M. 2018. Per Pupil Expenditure in the Government Schools of Uttar Pradesh, and the Rate of Reimbursement to Private Schools under the Right To Education Act: An Update. Delhi, Centre for Civil Society.

Ladd, H. F. and Fiske, E. B. 2016. England Confronts the Limits of School Autonomy. New York, National Center for the Study of Privatization in Education, Teachers' College, Columbia University. (NCSPE Working Paper 232.)

Lange, M.-F., Lauwerier, T. and Locatelli, R. 2021. The Impact of Privatization on Teachers in Francophone Sub-Saharan African Countries. (Background paper for Global Education Monitoring Report 2021/2.)

LaRocque, N. and Sipahimalani-Rao, V. 2019. School Education in Pakistan: A Sector Assessment. Manila, Asian Development Bank.

Levin, H. M. and Belfield, C. 2015. Guiding the development and use of cost-effectiveness analysis in education. J ournal of Research on Educational Effectiveness, Vol. 8, No. 3, pp. 400–18.

Lowrie, B. 2017. Islamic Faith School's Gender Segregation Ruled Unlawful. London, EachOther. https://eachother.org.uk/ islamic-faith-schools-gender-segregation-ruled-unlawful. (Accessed 17 August 2021.)

Maphanga, C. 2020. 2 illegal schools shut down in Joburg, staff questioned by police. News24, 8 October.

Marchetta, F. and Dilly, T. 2019. Supporting Education in Africa: Opportunities and Challenges for an Impact Investor. Monaco, Fondation pour les études et recherches sur le développement international/Investisseurs et Partenaires. (Feasibility study.)

Maussen, M. and Bader, V. 2015. Non-governmental religious schools in Europe: institutional opportunities, associational freedoms, and contemporary challenges. *Comparative Education*, Vol. 51, No. 1, pp. 1–21.

Mehendale, A., Mukhopadhyay, R. and Namala, A. 2015. Right to education and inclusion in private unaided schools: an exploratory study in Bengaluru and Delhi. *Economic and Political Weekly*, Vol. 50, No. 7, pp. 43–51.

Mizala, A. and Schneider, B. 2019. Promoting quality education in Chile: the politics of reforming teacher careers. Journal of Education Policy, Vol. 35, No. 4, pp. 529–55.

Mkutu, K. and Opondo, V. 2021. The complexity of radicalization and recruitment in Kwale, Kenya. Terrorism and Political Violence, Vol. 33, No. 1, pp. 26–48.

Morris, R. 2014. The admissions criteria of secondary free schools. Oxford Review of Education, Vol. 40, No. 3, pp. 389–409. Muralidharan, K. and Sundararaman, V. 2015. The aggregate effect of school choice: evidence from a two-stage experiment in India. Quarterly Journal of Economics, Vol. 130, No. 3, pp. 1011–66.

NAO. 2021. School Funding in England. London, National Audit Office. (Report by the Comptroller and Auditor General.)

National Governance Association. 2020. Governing in a Multi Academy Trust. Birmingham, UK, National Governance Association.

- National Secular Society. 2018. Unsafe Sex Education: The Risk of Letting Religious Schools Teach Within the Tenets of their Faith. London, National Secular Society.
- \_\_\_\_. 2019. Government Warns Independent Faith Schools over Sex Segregation. London, National Secular Society. www.secularism.org.uk/news/2019/06/government-warns-independent-faith-schools-over-gendersegregation. (Accessed 17 August 2021.)

Netherlands Government. 2021. *Public-authority and Private Schools*. The Hague, Government of the Netherlands. www.government.nl/topics/freedom-of-education/public-authority-and-private-schools. (Accessed 17 August 2021.)

New York Times. 2021. My son's yeshiva is breaking the law. *The New York Times*, 7 April. www.nytimes.com/video/ opinion/10000007670190/yeshiva-schools-education.html. (Accessed 2 November 2021.)

O'Shea, C., Palcic, D. and Reeves, E. 2019. Comparing PPP with traditional procurement: the case of schools procurement in Ireland. Annals of Public and Cooperative Economics, Vol. 90, No. 2, pp. 245–67.

O'Donoghue, J., Crawfurd, L., Makaaru, J., Otieno, P. and Perakis, R. 2018. A Review of Uganda's Universal Secondary Education Public Private Partnership Programme. London, Education Partnerships Group.

- OECD. 2017. OECD Reviews of Integrity in Education: Ukraine 2017. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2019. PISA 2018 Results (Volume I): What Students Know and Can Do. Paris, Organisation for Economic Co-operation and Development.

Olneck-Brown, B. 2020. Interactive Guide to School Choice Laws. Washington, DC, National Conference of State Legislatures. www.ncsl.org/research/education/interactive-guide-to-school-choice.aspx. (Accessed 17 August 2021.)

Omoeva, C. and Gale, C. 2016. Universal, but not free: household schooling costs and equity effects of Uganda's Universal Secondary Education policy. International Journal of Educational Development, Vol. 50, No. C, pp. 41–50.

Ouédraogo, H. 2018. Enseignement privé au Burkina: plusieurs écoles et établissements privés jugés «non conformes» par le MENA [Private education in Burkina: several private schools and establishments judged 'non-compliant' by the MENA]. Les Echos du Faso, 7 September.

Philippines Commission on Audit. 2019. Government Assistance to Students and Teachers in Private Education (GASTPE). Quezon City, Philippines, Commission on Audit. (Performance Audit Report 2018-02.)

Phillips, K. 2018. A culture of clerical immunity in Myanmar is putting children at risk of abuse. *Time*, 6 December.

- Pratap, K. V. and Chakrabarti, R. 2017. *Public-Private Partnerships in Infrastructure*. Cham, Switzerland, Springer. Premium Times. 2021. Osun shuts 600 illegal schools. *Premium Times*, 9 June.
- ProPublica. 2021. Credibly accused priests. ProPublica. www.propublica.org/datastore/dataset/credibly-accusedpriests. (Accessed 17 August 2021.)
- Punjab Education Foundation. 2021. New School Program. Lahore, Punjab Education Foundation. www.pef.edu.pk/nsp/ index.aspx. (Accessed 17 August 2021.)

Puozaa, C. Y. 2016. Outsourcing and the pupil transportation industry in Minnesota: an economic evaluation. PhD dissertation, Department of Economics, University of Mississipi. (Unpublished.)

Qadri, H. M.-u.-D. 2018. Foreign, political and financial influences on religious extremism: a study of madrassas in Punjab, Pakistan. Counter Terrorist Trends and Analyses, Vol. 10, No. 4, pp. 5–11.

RFI. 2017. French Catholic school closed in abuse inquiry. Paris, Radio France Internationale. www.rfi.fr/en/ france/20170603-french-catholic-school-closed-abuse-inquiry. (Accessed 17 August 2021.)

. 2019. Sénégal: la question des écoles coraniques reste très sensible [Senegal: the issue of Koranic schools remains very sensitive]. Paris, Radio France Internationale. www.rfi.fr/fr/afrique/20191212-senegal-daaras-ecolescoraniques-eleves-enchaines-coran-religion-islam. (Accessed 17 August 2021.)

Roberts, N. and Danechi, S. 2021. School Places in England: Admissions and Appeals. London, House of Commons. (Briefing Paper 07147.)

Romero, M., Sandefur, J. and Sandholtz, W. 2020. Outsourcing education: experimental evidence from Liberia. American Economic Review, Vol. 110, No. 2, pp. 364–400.

Rossignoli, S. 2021. The Dual Face of Non-State Schools in Peru, Tanzania, and the Philippines: A Case-Study Analysis. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)

Saguin, K. I. 2019. Designing effective governance of education. Policy Design and Practice, Vol. 2, No. 2, pp. 182–97.
 Sanders, R. 2018. A failure on all fronts: what we really need to know about school unification. The New Orleans Tribune.
 Santos, H. and Elacqua, G. 2016. Socioeconomic school segregation in Chile: parental choice and a theoretical counterfactual analysis. CEPAL Review, No. 119, pp. 123–38.

Sarasvati, N. 2020. How reservation in private schools isn't working for poor children. India Spend, 29 October.

Scheunpflug, A. and Wenz, M. 2021. Faith Based Non-state Actors in Selected African Countries: A Comparative Analysis of Catholic and Protestant Schooling in the Democratic Republic of Congo, Rwanda, Tanzania, Madagascar and Cameroon. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)

Senghor, C. 2020. Au Sénégal, des maîtres coraniques s'insurgent contre la violence dans les « daaras » [In Senegal, Koranic teachers rise up against violence in 'daaras']. La Croix Africa, 10 February.

Sharma, K. 2021. Only 16 states/UTs teach underprivileged kids for free in pvt schools, child rights body finds. *ThePrint*. https://theprint.podbean.com/e/theprintpod-only-16-statesuts-teach-underprivileged-kids-for-free-in-pvt-schools-child-rights-body-finds. (Accessed 12 November 2021.) (Podcast.)

Sherratt, F., Sherratt, S. and Ivory, C. 2020. Challenging complacency in construction management research: the case of PPPs. Construction Management and Economics, Vol. 38, No. 2, pp. 1–15.

Shewbridge, C., Fuster, M. and Rouw, R. 2019. Constructive Accountability, Transparency and Trust between Government and Highly Autonomous Schools in Flanders: Background and Lessons from the Flemish Strategic Education Governance Learning Seminar. Paris, Organisation for Co-operation and Development. (OECD Education Working Paper.)

Shrinivasa, M. 2020. Karnataka: private unaided schools must pay teachers, staff. The Times of India, 13 September. Sindh Government. 2002. Sindh Private Educational Institutions (Regulation and Control) Ordinance, 2001. Karachi,

Government of Sindh.

Singh, B. B. 2019. Private schools in Jajarkot defy scholarship provision. Kathmandu Post, 6 November.

Somparé, E. B. and Somparé, A. W. 2018. La condition enseignante en Guinée : des stratégies de survie dans le champ scolaire et universitaire guinéen [The condition of teachers in Guinea: survival strategies in Guinean schools and universities]. Cahiers de la Recherche sur L'Education et les Savoirs, Vol. 17.

Sood, A. and Prakash, K. 2020. 38 private schools in Punjab get notice for seeking fee. The Tribune, 8 April.

Street Child. 2021. Construction, Curriculum and Instruction: Critical Non-State Services Supporting Children in Conflict and Crisis Contexts. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)

Terrier, C., Pathak, P. A. and Ren, K. 2021. From immediate acceptance to deferred acceptance: effects on school admissions and achievement in England. London, Centre for Economic Performance. (Discussion Paper 1815)

Tezel Mccarthy, A. 2017. Non-state actors and education as a humanitarian response: role of faith-based organizations in education for Syrian refugees in Turkey. *Journal of International Humanitarian Action*, Vol. 2, No. 1.

The Learning Landscape. 2015. New Orleans, Louisiana: Alternative Governance Arrangements. Sudbury, Mass., Bellwether Education Partners. https://thelearninglandscape.org/new-orleans-louisiana-alternative-governancearrangements. (Accessed 17 August 2021.)

Think Mauritius. 2019. Rethinking the Education System in Mauritius. Ebene, Mauritius, Think Mauritius.

Thompson, O. 2011. The estimated cost impact of privatizing student transportation in Minnesota school districts. *Public Choice,* Vol. 146, No. 3–4, pp. 319–39.

Titeca, K. and de Herdt, T. 2011. Real governance beyond the 'failed state': negotiating education in the Democratic Republic of the Congo. African Affairs, Vol. 110, No. 439, pp. 213–31.

- Titheradge, N. 2018. Abuse concerns over unregistered schools. BBC News. www.bbc.com/news/uk-43126598. (Accessed 17 August 2021.)
- Treviño, E., Mintrop, R., Villalobos, C. and Órdenes, M. 2018. What Might Happen if School Vouchers and Privatization of Schools Were to Become Universal in the U.S.: Learning from a National Test Case – Chile. Boston, Mass., NEPC. (Policy brief.)
- Trucano, M. 2014. Paying teacher salaries with mobile phones. World Bank Blogs, 21 February.

Tutors' Association. 2021. About the Tutors' Association. Marlow, UK, The Tutors' Association. https://thetutorsassociation.org.uk/about.

- UNESCO. 2008. EFA Global Monitoring Report 2009: Overcoming Inequality Why Governance Matters. Paris, UNESCO.
- \_\_\_\_. 2019. Global Education Monitoring Report 2019: Migration, Displacement and Education Building Bridges, not Walls. Paris, UNESCO.
- UNICEF. 2016. Violence against Children in Education Settings in South Asia. Kathmandu, UNICEF Regional Office for South Asia.
- \_\_\_\_. 2020. Integrating Faith for Social and Behaviour Change into Pagoda Structures for a Systems Approach to Capacity Development. Phnom Penh, UNICEF Cambodia. (Change for Children: Global Initiative on Social and Behaviour Change.)
- قرار مجلس الوزراء رقم 92 صادر بتاريخ 8002/7/6م. بشأن اللائحة التنظيمية للتعليم الخاص .908 OD2/7/6م. بشأن اللائحة التنظيمية للتعليم الخاص . [Cabinet Resolution No. 29 Issued on 6/7/2008 AD Concerning the Regulation of Private Education]. Abu Dhabi, Government of the United Arab Emirates.
- USAID, Results for Development and Education in Crisis and Conflict Network. 2018. Affordable Non-State Schools in El Salvador. Washington, DC, USAID.
- Verger, A., Moschetti, M. C. and Fontdevila, C. 2021. How and why policy design matters: understanding the diverging effects of public–private partnerships in education. Adamson, F., Aubry, S., Koning, M. d. and Dorsi, D. (eds), Realizing the Abidjan Principles on the Right to Education: Human Rights, Public Education, and the Role of Private Actors in Education. Cheltenham, UK, Edward Elgar.
- Wad, S., Dongre, A. and Sarin, A. 2017. Lottery logic in the four states. Sarin, A., Dongre, A. and Wad, S. (eds), State of the Nation: RTE Section 12 (1) (C) 2017. Ahmedabad/New Delhi, Indian Institute of Management Ahmedabad/Centre for Policy Research/Central Square Foundation, pp. 57–63.
- West, A. and Wolfe, D. 2018. Academies, the School System in England and a Vision for the Future. London, London School of Economics and Political Science. (Clare Market Paper 23.)
- Wodon, Q. 2020a. Global Catholic Education Report 2020. Rome, International Office of Catholic Education.
- \_\_\_\_. 2020b. Global Catholic Education Report 2020: Achievements and Challenges at a Time of Crisis. Rome, International Office of Catholic Education.
- \_\_\_\_. 2021a. Faith-based Schools, Education Pluralism, and the Right to Education: Three Papers from the Global Catholic Education Project. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- \_\_\_\_. 2021b. Potential Impact of the COVID-19 Crisis on Catholic Schools and Universities: A Preliminary Assessment. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Wokadala, J. and Barungi, M. 2015. Benefit incidence analysis of government spending on public–private partnership schooling under Universal Secondary Education policy in Uganda. *Africa Education Review*, Vol. 12, No. 3, pp. 381–97.

World Bank. 2016. SABER EPS Mauritania Country Report. Washington, DC, World Bank.

- \_\_\_\_. 2020. Philippines Basic Education Public Expenditure Review. Washington, DC, World Bank.
- Zafar, S. A. 2020. Learning in safety. *The News*, 15 July. www.thenews.com.pk/print/686878-learning-in-safety. (Accessed 17 August 2021.)
- Zakharia, Z. 2016. Bilingual Education in the Middle East and North Africa. Garcia, O., Lin, A. and May, S. (eds), *Bilingual and Multilingual Education*. Cham, Switzerland, Springer, pp. 281–93.
- Zancajo, A., Fontdevila, C., Verger, A. and Bonal, X. 2021. Regulating Public–Private Partnerships, Governing Non-state Schools: An Equity Perspective. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Zhang, W. 2020. Non-State Actors in Education: The Nature, Dynamics and Policy Implications of Private Supplementary Tutoring. Paris, UNESCO. (Global Education Monitoring Report Fellowship Paper.)
- Zhu, J. and Yang, Y. 2021. China planning new crackdown on private tutoring sector. Euronews, 12 May.
- Zimbabwe Ministry of Primary and Secondary Education. 2021. Primary and Secondary Education Statistics Report 2020. Harare, Ministry of Primary and Secondary Education.

#### **CHAPTER 4**

- Acerenza, S. and Gandelman, N. 2019. Household education spending in Latin America and the Caribbean: Evidence from income and expenditure surveys. *Education Finance and Policy*, Vol. 14, No. 1, pp. 61–87.
- Acholla, D. A. 2021. Low-fee Private Schooling: A Framework to Define the Scope of these Non-state Actors in Education. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Ahmed, Z. S. and Shahzad, R. 2021. The role of peace education in countering violent extremism in Pakistan: an assessment of non-governmental efforts. *Conflict, Security and Development, Vol. 21, No. 3, pp. 199–222.*
- Alvarez, H., Elacqua, G., Mendez, C., Munevar, I. and Vasquez, D. 2021. Colegios Privados en Tiempos de COVID-19 [Private Schools in the Time of COVID-19]. Washington, DC, Inter-American Development Bank. (Hablemos de Politica Educativa: America Latina y el Caribe 8.)
- Asadullah, M. N. 2018. Madrasah for girls and private school for boys? The determinants of school type choice in rural and urban Indonesia. International Journal of Educational Development, Vol. 62, pp. 96–111.
- Aslam, M., Rawal, S. and Saeed, S. 2017. Public–Private Partnerships in Education in Developing Countries: A Rigorous Review of the Evidence. London, Ark Education Partnerships Group.
- Attridge, S. and Engen, L. 2019. Blended Finance in the Poorest Countries: The Need for a Better Approach. London, Overseas Development Institute.
- Azam, M. 2016. Private tutoring: evidence from India. Review of Development Economics, Vol. 20, No. 4, pp. 739-61.
- Azzam, Z. 2017. Dubai's private school fees framework: a critical discussion. Journal of Research in International Education, Vol. 16, No. 2, pp. 115–30.
- Bangladesh Ministry of Education. 2018. Pocket Book on Bangladesh Education Statistics 2017. Dhaka, Bangladesh Bureau of Educational Information and Statistics, Ministry of Education.
- Berlanga, C., Morduchowicz, A., Scasso, M. and Vera, A. 2020. Reabrir las Escuelas en América Latina y el Caribe: Claves, Desafíos y Dilemas para Planificar el Retorno Seguro a las Clases Presenciales [Reopening Schools in Latin America and the Caribbean: Keys, Challenges and Dilemmas for Planning a Safe Return to In-Person Classes]. Washington, DC, Inter-American Development Bank. (Technical Note 2075.)
- Braniff, L. 2016. Digital Finance and Innovations in Financing for Education. Washington, DC, Consultative Group to Assist the Poor. (CGAP Working Paper.)
- Bray, M., Kobakhidze, M. N. and Kwo, O. 2020. Shadow Education in Myanmar: Private Supplementary Tutoring and Its Policy Implications. Hong Kong, Comparative Education Research Centre, University of Hong Kong. (CERC Monograph in Comparative and International Education and Development 13.)
- Brown, C., Sargrad, S. and Benner, M. 2017. Hidden Money: The Outsized Role of Parent Contributions in School Finance. Washington, DC, Center for American Progress.
- Campbell, S., DiGiuseppe, M. and Murdie, A. 2019. International development NGOs and bureaucratic capacity: facilitator or destroyer? *Political Research Quarterly*, Vol. 72, No. 1, pp. 3–18.
- Carvalho, S. and Hares, S. 2020. The Economic Shock of COVID-19 May Hit Private School and Contract Teachers Hardest. Washington, DC, Center for Global Development. www.cgdev.org/blog/how-are-international-donorsresponding-education-needs-during-covid-pandemic. (Accessed 24 November 2020.)
- CDC Group. 2019. Framework: Maximising the Impact of Education Investments. London, CDC Group.
- Centre for Civil Society. 2021. Ease of Operations for Budget Private Schools in India. Paris, UNESCO. (Background paper for Global Education Monitoring Report on Non-state Actors in South Asia 2022.)
- Chadha, N. and Nandwani, B. 2020. Examining the role of corporate social responsibility funding in Indian education. Avelar, M. and Patil, L. (eds), New Philanthropy and the Disruption of Global Education. Geneva, Switzerland, NORRAG, pp. 102–05.
- Choi, E. J. and Hwang, J. 2020. Transition of son preference: evidence from South Korea. Demography, Vol. 57, pp. 627–52.
- Col, B. and Patel, S. 2019. Going to haven? Corporate social responsibility and tax avoidance. *Journal of Business Ethics,* Vol. 154, No. 4, pp. 1033–50.
- Coughlan, S. 2017. Schools 'depending on parents' direct debits'. BBC News, 15 April. www.bbc.com/news/ education-39601310. (Accessed 12 February 2021.)
- Damgaard, J., Elkjaer, T. and Johannesen, N. 2019. What Is Real and What Is Not in the Global FDI Network? Washington, DC, International Monetary Fund. (IMF Working Paper 19/274.)
- Datta, S. and Kingdon, G. G. 2019. Gender Bias in Intra-Household Allocation of Education in India: Has It Fallen over Time? Bonn, Institute of Labor Economics. (IZA Discussion Paper 12671.)
- Demirgüç-Kunt, A., Klapper, L., Singer, D. and Van Oudheusden, P. 2014. Global Findex Database 2014: Measuring Financial Inclusion around the World. Washington, DC, World Bank.

Dey, C. and Gibbon, J. 2018. New development: private finance over public good? Questioning the value of impact bonds. Public Money and Management, Vol. 38, No. 5, pp. 375–78.

DFID. 2018. DFID Education Policy: Get Children Learning. London, Department for International Development.

- Dore, B. 2013. Private, aided schools to get development funds after 8 years. *Hindustan Times*, 26 January. www. hindustantimes.com/mumbai/private-aided-schools-to-get-development-funds-after-8-yrs/storyym2P8MWXSPDrxAIjCF8ToJ.html. (Accessed 12 February 2021.)
- Dupuy, K., Palik, J. and Østby, G. 2020. Walking the Talk? Financing for Education in Emergencies: 2015–2018. Oslo, Policy Research Institute Oslo/Save the Children.

EACHRights. 2018. Kenya Compliance Advisor Ombudsman Complaint. Nairobi, The East African Centre for Human Rights. Ecorys. 2019a. Independent Evaluation of the UK Department for International Development's Development Impact Bonds

(DIBs) Pilot Programme: Full Report. Birmingham, UK, Ecorys.

- Education Partnerships Group. 2019. Analysis of the Ivorian Government's Tuition Fee Subsidy Scheme. London, Education Partnerships Group.
- Edwards, S. 2019. World's Largest Education DIB Reports First-year Results. Washington, DC, Devex.
- Elffers, L. and Jansen, D. 2019. De opkomst van schaduwonderwijs in Nederland: wat weten we en welke vragen liggen nog open? [The Rise of Shadow Education in the Netherlands: What Do We Know and Which Questions Are Still Open?]. Amsterdam, University of Amsterdam.

Enterprise. 2019. Where things stand with the PPP schools program. Enterprise, 2 December. https://enterprise.press/ stories/2019/12/02/where-things-stand-with-the-ppp-schools-program-7509. (Accessed August 6 2021.)

- European Parliament. 2018. Resolution of 13 November 2018 on EU Development Assistance in the Field of Education. Strasbourg, France, European Parliament. (2018/2081(INI).)
- Feigenberg, B., Yan, R. and Rivkin, S. 2019. Illusory gains from Chile's targeted school voucher experiment. *Economic Journal*, Vol. 129, No. 623, pp. 2805–32.
- Giacomin, V., Jones, G. and Salvaj, E. 2019. Why Does Business Invest in Education in Emerging Markets? Why Does it Matter? Cambridge, Mass., Harvard Business School. (HBS Working Paper 20-039.)
- Giordono, L. and Pugatch, T. 2015. Informal Fee Elimination and Student Performance: Evidence from The Gambia. Bonn, Germany, Institute for the Study of Labor. (IZA Discussion Paper 9560.)
- Giridharadas, A. 2019. Winners Take All: The Elite Charade of Changing the World. New York, Vintage.
- Global Chinese Philanthropy Initiative. 2017. Chinese and Chinese American Philanthropy. Los Angeles, Calif., Global Chinese Philanthropy Initiative.
- GPE. 2019. Private Sector Engagement Strategy: June 2019. Washington, DC, Global Partnership for Education. (BOD/2019/06 DOC 08.)
- Gustafsson-Wright, E., Boggild-Jones, I., Segell, D. and Durland, J. 2017. Impact Bonds in Developing Countries: Early Learnings from the Field. Washington, DC, Brookings Institution.
- Gustafsson-Wright, E., Gardiner, S. and Putcha, V. 2015. The Potential and Limitations of Impact Bonds: Lessons from the First Five Years of Experience Worldwide. Washington, DC, Brookings Institution.
- Haiti Ministry of National Education and Professional Training. 2018. Plan Décennal d'Éducation et de Formation 2017–2027 [10-Year Education and Training Plan 2017–2027]. Port-au-Prince, Ministry of National Education and Professional Training.
- Hares, S. and Crawfurd, L. 2021. The Role and Impact of Private Schools, School Chains and PPPs in Low and Middle Income Countries. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Härmä, J. 2020. Low-Fee Private Schooling and Poverty in Developing Countries. London, Bloomsbury.
- Härmä, J. and Siddhu, G. 2017. Parental Fee Default: Extent, Determinants and Implications. Lagos, Nigeria, DEEPEN.
- Hedges, S., Winton, S., Rowe, E. and Lubienski, C. 2020. Private actors and public goods: a comparative case study of funding and public governance in K–12 education in 3 global cities. *Journal of Educational Administration and History*, Vol. 52, No. 1, pp. 103–19.
- IDB. 2011. IDB Donates \$50 Million for Education in Haiti. Washington, DC, Inter-American Development Bank. www.iadb.org/en/news/idb-donates-50-million-education-haiti. (Accessed 17 August 2021.)
- IFC. 2020. IFC Freeze on Investment in K-12 Private, Fee-charging Education. Washington, DC, International Finance Corporation. www.ifc.org/wps/wcm/connect/Industry\_EXT\_Content/IFC\_External\_Corporate\_Site/ Education. (Accessed 17 August 2021.)

IMF. 2021. Policy Responses to COVID-19: Policy Tracker. Washington, DC, International Monetary Fund. www.imf.org/en/Topics/imf-and-covid19/Policy-Responses-to-COVID-19. (Accessed 20 May 2021.)

- India Ministry of Education. 2021. UDISE+ Publications and Statistics. New Delhi, Department of School Education and Literacy, Ministry of Education. https://udiseplus.gov.in/#/Publication (Accessed 10 January 2021.)
- Institute for Health Metrics and Evaluation. 2021. Generating Estimates of Household Spending on Education. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Irwin, T. C., Mazraani, S. and Saxena, S. 2018. How to Control the Fiscal Costs of Public–Private Partnerships. Washington, DC, International Monetary Fund. (How To Note 18/04.)
- Iversen, E. and Begue, A. 2017. The Effects of Privatisation on Girls' Access to Free, Quality Public Education in Malawi, Mozambique, Liberia, Tanzania and Nepal. Johannesburg, South Africa, ActionAid.
- JICA. 2015. JICA Position Paper in Education Cooperation. Tokyo, Japan International Cooperation Agency. \_\_\_\_\_. 2016. JICA's Position Paper on SDGs: Goal 4. Tokyo, Japan International Cooperation Agency.
- \_\_\_\_. 2016. JICA'S Position Paper on SDGS: Goal 4. Tokyo, Japan International Cooperation Agency.
- Johnson, P. D. and Saich, T. 2018. Values and Vision: Perspectives on Philanthropy in 21st Century China. Cambridge, Mass., Ash Centre, Harvard Kennedy School.
- Joshi, R. 2018. Public Financing for Low-cost Private Schools. Jakarta, Center for Indonesian Policy Studies.
- Kasozi-Mulindwa, S. and Okware, B. 2019. Corporation tax on private primary and secondary schools: implications on access and quality education. *American Journal of Economics*, Vol. 9, No. 3, pp. 133–39.
- Kenny, C., Kalow, J., Leo, B. and Ramachandran, V. 2018. Comparing Five Bilateral Development Finance Institutions and the IFC. Washington, DC, Center for Global Development.
- Khan, A., Wales, J., Nicolai, S. and Caron, C. 2020. Strengthening Coordinated Education Planning and Response in Crises: Democratic Republic of the Congo Case Study. London, Overseas Development Institute.
- Khan, Z. 2020. How tax evasion by the ultra-rich is everyone's problem. *Data Wrapper*, 9 January. https://blog. datawrapper.de/weekly-rich-people-tax-charities-donations. (Accessed 27 May 2021.)
- Kippels, S. and Ridge, N. 2019. International and Other Migrant Schools in Gulf Cooperation Council Countries. Paris, UNESCO. (Background paper for Global Education Monitoring Report on Inclusion and Education in Arab States 2019.)
- Koutou, N. G. C. and Goi Bi, Z. T. 2019. Etude sur la privatisation de l'école en Côte d'Ivoire [Study on school pivatization in Côte d'Ivoire]. Brussels, Education International.
- Le Nestour, A., Carvalho, S. and Minardi, A. L. 2020. Unequal Burdens: The Impact of Shocks on Household Education Spending. Washington, DC, Center for Global Development. www.cgdev.org/blog/unequal-burdens-impactshocks-household-education-spending. (Accessed 12 February 2021.)
- Leigland, J. 2018. Public–private partnerships in developing countries: the emerging evidence-based critique. World Bank Research Observer, Vol. 33, No. 1, pp. 103–34.
- Liao, X. and Huang, X. 2018. Who is more likely to participate in private tutoring and does it work? Evidence from PISA (2015). East China Normal University Review of Education, Vol. 1, No. 3, pp. 69–95.
- Lindsjö, K. 2018. The financial burden of a fee free primary education on rural livelihoods: a case study from rural Iringa Region, Tanzania. Development Studies Research, Vol. 5, No. 1, pp. 26–36.
- Mattern, M. and Garcia, A. 2021. In Uganda, Solar Home Systems Help Students Stay in School. Washington, DC, Consultative Group to Assist the Poor. www.cgap.org/blog/uganda-solar-home-systems-help-students-stayschool. (Accessed 25 May 2021.)
- Mbole, F. and Kimathi, D. 2019. What Solutions Would Make a Difference in Education Finance? Nairobi, FSD Kenya. https:// fsdkenya.org/blog/what-solutions-would-make-difference-in-education-finance. (Accessed 17 August 2021.)
- McCluskey, R. 2015. Is Responsible Tax Behaviour the Next Frontier of CSR? Brighton, UK, International Centre for Tax and Development, Institute of Development Studies. www.ictd.ac/blog/is-responsible-tax-behaviour-the-next-frontier-of-csr. (Accessed 17 August 2021.)
- Mehendale, A. and Singh, A. 2020. India Education Outcomes Fund: A Case Study. Geneva, Switzerland, NORRAG. (Working Paper 12.)
- Meiring, G. and Krugel, L. 2018. The Responsible Taxpayer through a New Lens of Transparency: Working towards a Common Purpose. London, PricewaterhouseCoopers.
- Menashy, F. and Zakharia, Z. 2020. Private engagement in refugee education and the promise of digital humanitarianism. Oxford Review of Education, Vol. 46, No. 3, pp. 313–30.
- \_\_\_\_. 2021. Private Engagement in Education in Emergencies: Rights and Regulations. New York, Inter-agency Network for Education in Emergencies.
- Mizala, A. and Schneider, B. 2020. Promoting quality education in Chile: the politics of reforming teacher careers. Journal of Education Policy, Vol. 35, No. 4, pp. 529–55.

Moschetti, M. C. and Verger, A. 2020. Opting for private education: public subsidy programs and school choice in disadvantaged contexts. *Educational Policy*, Vol. 34, No. 1, pp. 65–90.

- Murnane, R. J., Waldman, M. R., Willett, J. B., Bos, M. S. and Vegas, E. 2017. The Consequences of Educational Voucher Reform in Chile. Cambridge, Mass., National Bureau of Economic Research. (NBER Working Paper 23550.)
- Narodowski, M. and Moschetti, M. 2015. The growth of private education in Argentina: evidence and explanations. Compare, Vol. 45, No. 1, pp. 47–69.

Neilson, C. 2013. Targeted vouchers, competition among schools, and the academic achievement of poor students. New Haven, Conn., Yale University. (Unpublished.)

- Niazi, M. and Doorly, A. 2020. Estimating the Impact of COVID-19 on the Non-State Education Sector in Low- and Middle-Income Countries: A Rapid Review. London, Global Schools Forum.
- Nicolai, S., Diwakar, V., Khan, A., Mansour-Ille, D. and Anderson, A. 2020. Strengthening Coordinated Education Planning and Response in Crisis Contexts: Synthesis Report. London, Overseas Development Institute.
- Nusche, D., Miron, G., Santiago, P. and Teese, R. 2015. *Flemish Community of Belgium*. Paris, Organisation for Economic Co-operation and Development. (OECD Reviews of School Resources.)
- OECD. 2016. Netherlands 2016: Foundations for the Future. Paris, Organisation for Economic Co-operation and Development. (OECD Reviews of National Policies for Education.)
- \_\_\_\_. 2017. The Funding of School Education: Connecting Resources and Learning. Paris, Organisation for Economic Co-operation and Development. (OECD Reviews of School Resources.)
- \_\_\_\_. 2018a. Education at a Glance 2018. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2018b. Private Philanthropy for Development. Paris, OECD Development Centre.
- \_\_\_\_. 2018c. Responsive School Systems: Connecting Facilities, Sectors and Programmes for Student Success. Paris, Organisation for Economic Co-operation and Development. (OECD Reviews of School Resources.)
- \_\_\_\_. 2019a. India's Private Giving: Unpacking Domestic Philanthropy and Corporate Social Responsibility. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2019b. Philanthropy and Education: Quality Education For All Lessons and Future Priorities. Paris, OECD Development Centre.
- \_\_\_\_. 2019c. PISA 2018 Results (Volume I): What Students Know and Can Do. Paris, Organisation for Economic Co-operation and Development. (Table B3.6.2: Financial Incentives and Disincentives for School Choice.)
- \_\_\_\_. 2020a. How Much Is Spent per Student on Educational Institutions? Paris, Organisation for Economic Co-operation and Development. (Education at a Glance 2020.)
- \_\_\_\_. 2020b. PISA 2018 Results (Volume V): Effective Policies, Successful Schools. Paris, Organisation for Economic Co-operation and Development.
- <u>2021. Private Philanthropy for Development. Paris, Organisation for Economic Co-operation and Development.</u>
- Olsen, A. S. and Prado, J. 2020. COVID-19 y la transición de la educación privada a la pública en Ecuador [COVID-19 and the transition from private to public education in Ecuador]. Washington, DC, Inter-American Development Bank. https://blogs.iadb.org/educacion/es/covid-19-y-la-transicion-de-la-educacion-privada-a-la-publica-en-ecuador. (Accessed 17 August 2021.)
- Opportunity EduFinance. 2020. Impacts of COVID-19 on the Affordable Non-State School Sector: Understanding the Challenges Schools, Teachers and Parents Face as a Result of the COVID-19 Pandemic. London, Opportunity EduFinance.
- Pallegedara, A. and Kumara, A. S. 2020. Spending privately for education despite having a free public education policy: evidence from Sri Lankan household surveys. *International Journal of Social Economics*, Vol. 47, No. 5.
- Pallegedara, A. and Mottaleb, K. A. 2018. Patterns and determinants of private tutoring: the case of Bangladesh households. International Journal of Educational Development, Vol. 59, pp. 43-50.
- Patil, L. and Brakman Reiser, D. 2021. Emerging Roles and Risks of Philanthropy in Global Education. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Patrinos, H. A., Barrera-Osorio, F. and Guáqueta, J. 2009. The Role and Impact of Public–Private Partnerships in Education. Washington, DC, World Bank.
- Philippines Department of Education. 2021. PPP for School Infrstructure Project (PSIP) Phase 1: DepEd's First PPP Project. Quezon City, Department of Education/Public-Private Partnership Center of the Philippines. (PPP Learning Series.)
- Quality Education India Development Impact Bond. 2020. Top Marks for India's Largest Development Impact Bond. London, British Asian Trust. https://qualityeducationindiadib.com/year-2-results. (Accessed 17 August 2021.)

- Rizvi, A. 2019. Six out of 10 parents believe school fees in Dubai are too high. *The National*, 4 September. www.thenational.ae/uae/six-out-of-10-parents-believe-school-fees-in-dubai-are-too-high-1.906019. (Accessed 5 July 2021.)
- Rodríguez Takeuchi, L. 2020. Intrahousehold Inequalities in Education Spending. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2020.)
- Romero, M. J. and Vervynckt, M. 2017. Unpacking the dangerous illusion of PPPs. Kishimoto, S. and Petitjean, O. (eds), Reclaiming Public Services: How Cities and Citizens Are Turning Back Privatisation. Paris/Amsterdam, Transnational Institute/Multinationals Observatory/Austrian Federal Chamber of Labour/European Federation of Public Service Unions/Ingeniería Sin Fronteras Cataluña/Public Services International/Public Services International Research Unit/We Own It/Norwegian Union for Municipal and General Employees/Municipal Services Project/Canadian Union of Public Employees, pp. 105–17.
- Ron Balsera, M. 2020. The World Bank's Milestone Decision to not Fund For-profit Providers. London, University College London. https://blogs.ucl.ac.uk/ceid/2020/04/15/balsera. (Accessed 6 August 2021.)
- Rosell, J. and Saz-Carranza, A. 2020. Determinants of public–private partnership policies. Public Management Review, Vol. 22, No. 8, pp. 1171–90.
- Rowe, E. and Perry, L. B. 2020. Private financing in urban public schools: inequalities in a stratified education marketplace. Australian Educational Researcher, Vol. 47, pp. 19–37.
- Sakaue, K. 2018. Informal fee charge and school choice under a free primary education policy: panel data evidence from rural Uganda. International Journal of Educational Development, Vol. 62, pp. 112–27.
- Sakellariou, C. and Patrinos, H. A. 2009. The equity impact of public finance of private education provision in Côte d'Ivoire. International Journal of Educational Development, Vol. 29, No. 4, pp. 350–56.
- Samhita Social Ventures. 2018. Decoding CSR Trends in India: Looking Back to Look Forward. Mumbai, India, Samhita Social Ventures.
- Santiago, P., Fiszbein, A., García Jaramillo, S. and Radinger, T. 2017. *Chile*. Paris, Organisation for Economic Co-operation and Development. (OECD Reviews of School Resources.)
- Sarin, A., Khun, S., Daulet Singh, B., Khanghta, P., Dongre, A., Joshi, E., Sengupta, A. and Rahman, F. 2015. State of the Nation: RTE Section 12(1)(c). Ahmedabad/New Delhi, Indian Institute of Management Ahmedabad/Centre for Policy Research/Central Square Foundation/Vidhi Centre for Legal Policy.
- Shaxson, N. 2019. Tackling tax havens. Finance and Development, Vol. 56, No. 3, pp. 6–10.
- Sieverding, M., Krafft, C. and Elbadawy, A. 2019. An exploration of the drivers of private tutoring in Egypt. *Comparative Education Review*, Vol. 63, No. 4, pp. 562–90.
- Sivasubramaniam, M. 2021. Philanthropic and Faith-based Non-state Actors in the Low-fee Private School Sector in the Global South: Blurring the Lines of Philanthropy and Enterprise. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Smith, W. C. and Baker, T. 2017. From Free to Fee: Are For-profit, Fee-charging Private Schools the Solution for the World's Poor? Washington, DC, Results Educational Fund.
- Srivastava, P. and Read, R. 2019. Philanthropic and impact investors: private sector engagement, hybridity and the problem of definition. Ridge, N. Y. and Terway, A. (eds), *Philanthropy in Education: Diverse Perspectives and Global Trends*. Cheltenham, UK, Edward Elgar, pp. 15–36. (NORRAG Series on International Education and Development.)
- Stuardo, G. M. and Cayuela, J. W. 2019. The difficult process in Chile: redefining the rules of the game for subsidized private education. Ornelas, C. (ed.), Politics of Education in Latin America: Reforms, Resistance and Persistence. Leiden, Netherlands, Brill, pp. 72–100. (Comparative and International Education: Diversity of Voices 49.)
- Tax Justice Network. 2020. The States of Tax Justice 2020: Tax Justice in the Time of COVID-19. Chesham, UK, Tax Justice Network.
- Thomas, A. 2020. SC declines parents' plea for moratorium on school fees. *Hindustan Times*, 10 July. www.hindustantimes.com/india-news/sc-declines-parents-plea-for-moratorium-on-school-fees/story-POhfEcNaFcWlRzAGH5NHII.html. (Accessed 17 August 2021.)
- Thompson, G., Hogan, A. and Rahimi, M. 2019. Private funding in Australian public schools: a problem of equity. Australian Educational Researcher, Vol. 46, No. 5, pp. 893–910.
- UNESCO. 2012. Education for All Global Monitoring Report 2012: Youth and Skills Putting Education to Work. Paris, UNESCO. \_\_\_\_\_. 2018. Walk Before You Run: The Challenges of Results-based Payments in Aid to Education. Paris, UNESCO.
  - (Global Education Monitoring Report Policy Paper 33.)
- \_\_\_\_. 2020. Global Education Monitoring Report 2020: Inclusion and Education All Means All. Paris, UNESCO.

Unterhalter, E. and Robinson, L. 2020. Precise evasions: PPPs, aid and education. Gideon, J. and Unterhalter, E. (eds), Critical Reflections on Public Private Partnerships. Abingdon, UK, Routledge, pp. 98–114.

- USAID. 2018a. Public Financing of Education in Haiti 2010–2018. Washington, DC, US Agency for International Development. \_\_\_\_\_. 2018b. USAID Education Policy. Washington, DC, US Agency for International Development.
- van Tonder, N. 2019. PPPs crucial to the future of education. *BusinessBrief*, 23 May. www.bbrief.co.za/2019/05/23/ppps-crucial-to-the-future-of-education. (Accessed August 6 2021.)
- Varkey Foundation, UNESCO and Dubai Cares. 2015. Business Backs Education: Creating a Baseline for Corporate CSR Spend on Global Education Initiatives. Dubai, Varkey Foundation/UNESCO/Dubai Cares.
- Wei Yi. 2018. 2017 China Educational Finance Household Survey: Current Status of China's Household Educational Expenditure. Beijing, China Institute for Educational Finance Research, Peking University. (Research Briefing.)
- Winchip, E. 2020. The interplay of government, private funding and civil actors in the United Arab Emirates: response to 'private actors and public goods: a comparative case study of funding and public governance in K-12 education in 3 global cities'. Journal of Educational Administration and History, Vol. 52, No. 1, pp. 120–23.
- World Bank. 2018. Haiti School Finance. Washington, DC, World Bank. (SABER Country Report.)
- \_\_\_\_. 2020. Bangladesh Transforming Secondary Education for Results Operation. Washington, DC, World Bank. (Implementation Status and Results Report 6.)
- Zuilkowski, S., Piper, B. and Ong'ele, S. 2020. Are low-cost private schools worth the investment? Evidence on literacy and mathematics gains in Nairobi primary schools. *Teachers' College Record*, Vol. 122, No. 1.

### **CHAPTER 5**

- ACIJ. 2019. Los Efectos de la Creciente Participación de Actores Privados en la Educación Inclusiva: El Caso de las Personas con Discapacidad [The Effects of the Growth of Private Actors on Inclusive and Equitable Education for All: The Case of Persons with Disabilities]. Buenos Aires, Asociación Civil por la Igualdad y la Justicia.
- ActionAid. 2018. Scaling-up Domestic Resources for Financing SDG 4: A Taxing Business? Johannesburg, South Africa, ActionAid.
- ActionAid Uganda, Africa Network Campaign for Education for All et al. 2018. NGOs says Bridge International Academies Must Respect the Right to education & Comply with Govt.'s Order to Close Its Schools. London, Business and Human Rights Resource Centre. www.business-humanrights.org/en/latest-news/ngos-says-bridgeinternational-academies-must-respect-the-right-to-education-comply-with-govts-order-to-close-itsschools. (Accessed 17 August 2021.)
- Alcott, B., Rose, P., Sabates, R., Alonso, M. L. and Cherfils, M. 2018. Experience and Lessons of Learning Intervention Programmes across the PAL Network Members. Cambridge, UK, REAL Centre, University of Cambridge. (Research and Policy Paper 18/4.)
- Alfavita. 2021. Δημοσκόπηση Marc: Υπέρ της αξιολόγησης των εκπαιδευτικών 7 στους 10 [Marc opinion poll: 7 in 10 in favour of teacher appraisal]. Athens, Alfavita. www.alfavita.gr/ekpaideysi/353953\_dimoskopisi-marc-yper-tis-axiologisis-ton-ekpaideytikon-7-stoys-10. (Accessed 17 August 2021.)
- Alsemgeest, L. 2015. Arguments for and against financial literacy education: where to go from here? International Journal of Consumer Studies, Vol. 39, No. 2, pp. 155–61.
- Alvarado, L. K. A., Soler, S. C. G. and González, J. C. 2021. The effect of teacher strikes on academic achievement: evidence from Colombia. International Journal of Educational Development, Vol. 82, art. 102369.
- Ark. 2017. Partnership Schools for Liberia. London, Ark. https://arkonline.org/Partnership\_Schools\_For\_Liberia. (Accessed 17 August 2021.)
- \_\_\_\_. 2021. About Us: Transforming Lives through Education. London, Ark. https://arkonline.org/about-us/what-we-do. (Accessed 17 August 2021.)
- Aslam, M., Rawal, S. and Saeed, S. 2017. Public–Private Partnerships in Education in Developing Countries: A Rigorous Review of the Evidence. London, Ark Education Partnerships Group.
- Au, W. and Lubienski, C. 2016. The role of the Gates Foundation and the philanthropic sector in shaping the emerging education market: lessons from the US on privatization of schools and education governance. Verger, A., Lubienski, C. and Steiner-Khamsi, G. (eds), World Yearbook of Education 2016: The Global Education Industry. New York, Routledge, pp. 27–43.
- Avelar, M. and Ball, S. J. 2019. Mapping new philanthropy and the heterarchical state: the mobilization for the national learning standards in Brazil. International Journal of Educational Development, Vol. 64, pp. 65–73.

- Baker, M. 2013. Industrial actions in schools: strikes and student achievement. Canadian Journal of Economics/Revue canadienne d'économique, Vol. 46, No. 3, pp. 1014–36.
- Bakir, A., Blodgett, J. G. and Salazar, R. J. 2017. Corporate sponsorships in schools: altruism and ethical judgments. Journal of Promotion Management, Vol. 23, No. 1, pp. 80–99.
- Ball, S. J. 2017. Exploring educational privatization, FreshEd. https://freshedpodcast.com/stephenball/ (Accessed 17 August 2021.)
- Baum, D., Lewis, L., Lusk-Stover, O. and Patrinos, H. 2014. What Matters Most for Engaging the Private Sector in Education: A Framework Paper. Washington, DC, World Bank. (SABER Working Paper Series.)
- Bogliaccini, J. A. and Rodríguez, F. 2015. Education system institutions and educational inequalities in Uruguay. CEPAL Review, Vol. 116, pp. 85–100.
- Bordoli, E., Martinis, P., Moschetti, M., Conde, S. and Alfonzo, M. 2017a. Educational Privatisation: A Latent Phenomenon in Uruguay. Brussels, Education International. (Unite for Quality Education.)
- . 2017b. Privatización educativa en Uruguay: políticas, actores y posiciones [Education privatizatión in Uruguay: policies, actors and positions]. Brussels, Education International. (Investigaciones Internacional de la Educación.)
- Brazil Ministry of Education and Culture. 2016. MEC justifica ação contra lei que impede opinião em Alagoas [Ministry of Education justifies action against law impeding opinion in Alagoas]. Brasília, Ministry of Education and Culture. http://portal.mec.gov.br/component/tags/tag/escola-livre. (Accessed 30 September 2021.)
- Brehm, W. and Silova, I. 2019. Five generations of NGOs in education: from humanitarianism to global capitalism. Davies, T. (ed.), Routledge Handbook of NGOs and International Relations. New York/Abingdon, UK, Routledge, pp. 283–96.
- Brion, C. 2020. Low-fee private schools: case studies from Ghana. International Journal of Education Policy and Leadership, Vol. 16, No. 3.
- Busemeyer, M. R. and Thelen, K. 2020. Institutional sources of business power. World Politics, Vol. 72, No. 3, pp. 448–80.
- Cabalin, C. 2015. Mediatizing higher education policies: discourses about quality education in the media. *Critical Studies in Education*, Vol. 56, No. 2, pp. 224–40.
- Cancion, N. 2018. Bandeira de Bolsonaro, veto a abordagem de gênero sofre derrotas em série na Justiça [Supported by Bolsonaro, vetoes to gender discussions suffer a series of defeats in the judicial system]. Folha de S. Paulo, 12 November. www1.folha.uol.com.br/educacao/2018/11/bandeira-de-bolsonaro-veto-a-abordagem-degenero-sofre-derrotas-em-serie-na-justica.shtml. (Accessed 30 September 2021.)
- Cappelli, P. 2020. STF julga inconstitucional lei de Alagoas inspirada no movimento Escola Sem Partido [Supreme Court judges unconstitutional the law inspired by the movement School Without Parties in Alagoas]. *O Globo*, 22 August. https://oglobo.globo.com/brasil/stf-julga-inconstitucional-lei-de-alagoas-inspirada-no-movimentoescola-sem-partido-24601475. (Accessed 30 September 2021.)
- Carrns, A. 2021. Pandemic helps stir interest in teaching financial literacy. *The New York Times*, 2 April. www.nytimes.com/2021/04/02/your-money/financial-literacy-courses.html. (Accessed 17 August 2021.)
- Central Square Foundation. 2020. State of the Sector Report on Private Schools in India. New Delhi, Central Square Foundation/Omidyar Network India.
- Ciriaco, M. 2019. Cuando los libros escolares mienten [When textbooks lie]. *La República*, 16 July. https://larepublica.pe/sociedad/2019/07/16/cuando-los-libros-escolares-mienten. (Accessed 17 August 2021.)
- Coleman, V. 2021. Digital Divide in UK Education during COVID-19 Pandemic: Literature Review. Cambridge, UK, Cambridge Assessment.
- Congressional Black Caucus. 2017. The Congressional Black Caucus Engagement Packet on the Nomination of Betsy DeVos for Secretary of Education. Washington, DC, Congressional Black Caucus.
- Connell, R. 2013. The neoliberal cascade and education: an essay on the market agenda and its consequences. *Critical Studies in Education*, Vol. 54, No. 2, pp. 99–112.
- Coons, J. A. 2018. Funding its own organizational learning: the impact of National Education Association grants in supporting and sustaining affiliate change efforts. PhD dissertation, University of Washington, Seattle, Wash. (Unpublished.)
- Costin, C. and Pontual, T. 2020. Curriculum reform in Brazil to develop skills for the twenty-first century. Reimers, F. M. (ed.), Audacious Education Purposes: How Governments Transform the Goals of Education Systems. Cham, Switzerland, Springer, pp. 47–64.
- Cowen, J. M. and Strunk, K. O. 2015. The impact of teachers' unions on educational outcomes: what we know and what we need to learn. *Economics of Education Review*, Vol. 48, pp. 208–23.
- Crawford-Garrett, K. and Thomas, M. A. 2018. Teacher education and the global impact of Teach for All. Noblit, G. W. (ed.), Oxford Research Encyclopedia of Education. Oxford, UK, Oxford University Press.

- Croso, C. and Magalhães, G. M. 2016. Privatização da educação na América Latina e no Caribe: tendências e riscos para os sistemas públicos de ensino [Privatization of education in Latin America and the Caribbean: trends and risks for public education systems]. *Educação and Sociedade*, Vol. 37, No. 134, pp. 17–33.
- Curtiss Wyss, M. and Perlman Robinson, J. 2021. Improving Children's Reading and Math at Large Scale in Côte d'Ivoire: The Story of Scaling PEC. Washington, DC, Brookings Institution.
- DeBray-Pelot, E. and McGuinn, P. 2009. The new politics of education: analyzing the federal education policy landscape in the post-NCLB era. *Educational Policy*, Vol. 23, No. 1, pp. 15–42.
- DeCoster, B. 2019. #SABERexposed 'Engaging the private sector'. Brussels, Education International. www.ei-ie.org/en/ woe\_homepage/woe\_detail/16527/saberexposed-"engaging-the-private-sector"-by-brendan-decoster. (Accessed 17 August 2021.)
- Dickinson, G. M. and Dolmage, W. R. 1996. Education, religion, and the courts in Ontario. Canadian Journal of Education/ Revue canadienne de l'éducation, Vol. 21, No. 4, pp. 363–83.
- Donnelly, M. 2021. No More Blue Notes for Panama Homeschoolers. Purcellville, Va., Home School Legal Defense Association. https://hslda.org/post/no-more-blue-notes-for-panama-homeschoolers. (Accessed 22 September 2021.)
- Educapital. 2020. Covid19: The New Mover and Shaker of the Edtech Market. Paris, EduCapital.
- Education International. 2021a. La Política Educativa en Costa Rica: Gobernar Mediante Alianzas Público-Privadas [Education Polícy in Costa Rica: Governing through Public–Private Alliances]. San José, Education International Latin America.
- \_\_\_\_. 2021b. La Política Educativa en República Dominicana: Participación Privada a la Vista y Paciencia del Estado [Education Policy in the Dominican Republic: Private Participation Supervised and Tolerated by the State]. San José, Education International Latin America.
- \_\_\_\_. 2021c. La Política Educativa en Uruguay: Experimentos y Alianzas Empresariales Para Lucrar [Education Polícy in Uruguay: Experiments and Business Alliances for Profit]. San José, Education International Latin America.
- Edwards Jr, D. B., Rappeport, A., Sperduti, V. and Caravaca, A. 2021. The Influence of the World Bank on Policy Formation, Policy Implementation, and Private Education: A Systematic Review of the Literature. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- EPG. 2021. Our Story. London, Education Partnerships Group. https://epg.org.uk/our-story. (Accessed 17 August 2021.)
- Equal Education. 2018. Equal Education's Submission on the Western Cape Provincial School Education Amendment Bill, 2018. Khayelitsha, South Africa, Equal Education.
- Escola Sem Partido. 2021. Anteprojetos [Draft projects]. Brasília, Escola Sem Partido. http://escolasempartido.org/ anteprojeto. (Accessed 30 September 2021.)
- European Commission, EACEA and Eurydice. 2018. Teaching Careers in Europe: Access, Progression and Support. Brussels, European Commission/EACEA/Eurydice.
- Fagundez, I. 2018. Mesmo sem lei, Escola sem Partido se espalha pelo país e já afeta rotina nas salas de aula [Even without laws, Nonpartisan School spreads across the country and affects classroom routines]. BBC News Brasil, 5 November. www.bbc.com/portuguese/brasil-46006167. (Accessed 30 September 2021.)
- Fang, J., Yang, G. and Wan, X. 2020. 'Pro-tobacco propaganda': a case study of tobacco industry-sponsored elementary schools in China. *Tobacco Control*, Vol. 29, No. 4, pp. 447–51.
- FAO. 2013. FAO Strategy for Partnerships with the Private Sector. Rome, Food and Agriculture Organization of the United Nations.
- Fernandes, L. I. and Ferreira, C. A. 2021. O Movimento Escola Sem Partido: ascensão e discurso [The Nonpartisan School movement: rise and discourse]. *Humanidades em Diálogo*, Vol. 10, pp. 194–209.
- García Mathewson, T. and Butrymowicz, S. 2020. Ed tech companies promise results, but their claims are often based on shoddy research. *The Hechinger Report*, 20 May. https://hechingerreport.org/ed-tech-companies-promiseresults-but-their-claims-are-often-based-on-shoddy-research. (Accessed 17 August 2021.)
- GHEX. 2020. Global Home Education Exchange. Global Home Education Exchange. https://ghex.world. (Accessed 21 September 2021.)
- Global Business Coalition for Education. 2019. Engaging the Investment Community in Education: The Case for Education Bonds to Raise New Resources for Sustainable Development Goal 4. New York/Washington, Global Business Coalition for Education/The Education Commission.
- Global Schools Forum. 2021. About Us. London, Global Schools Forum. www.globalschoolsforum.org/page/AboutUs. (Accessed 17 August 2021.)
- GPE. 2019. GPE Private Sector Engagement Strategy Paper (2019-2022): Engaging the Private Sector to Support the Delivery of GPE 2020. Washington, DC, Global Partnership for Education.

- Grau, E. S. C. and Olmedo, A. 2012. Neoliberalismo creación de 'sentido común': crisis educativa y medios de comunicación en Chile [Neoliberal creation of 'common sense': educational crisis and the media in Chile]. Revista de Curriculum y Formación del Profesorado, Vol. 16, No. 3, pp. 145–68.
- GSG. 2021. Tying Funding to Results. London, Global Steering Group for Impact Investment.
- GSG and The Education Commission. 2019. The Education Outcomes Fund for Africa and the Middle East: A Game-changing Way to Drive Results in Education. London/Washington, Global Steering Group for Impact Investment/The Education Commission.
- Han, E. S. 2020. The myth of unions' overprotection of bad teachers: evidence from the district-teacher matched data on teacher turnover. *Industrial Relations*, Vol. 59, No. 2, pp. 316–52.
- Hill, P. T. and Jochim, A. E. 2009. Political perspectives on school choice. Berends, M., Primus, A. and Springer, M. G. (eds), Handbook of Research on School Choice, 1st ed. New York and London, Routledge, pp. 3–18.
- Hochschulwatch. 2021. Hochschulwatch. Berlin, Transparency International. http://netaction.github.io/hochschulwatch. (Accessed 17 August 2021.)
- Hogan, A., Thompson, G., Sellar, S. and Lingard, B. 2018. Teachers' and school leaders' perceptions of commercialisation in Australian public schools. *Australian Educational Researcher*, Vol. 45, No. 2, pp. 141–60.
- HolonIQ. 2019. Global Education in 10 Charts. New York, HolonIQ.
- HSLDA. 2021. Making Homeschooling Possible Together. Purcellville, Va., Home School Legal Defense Association.
- Huffington Post. 2020. Scuole paritarie, una questione di tutti [Schools with parity, an issue for all].
  - Huffington Post Italia, 18 May. www.huffingtonpost.it/entry/scuole-paritarie-una-questione-di-tutti\_ it\_5ec23988c5b691d3e0907698. (Accessed 17 August 2021.)
- Human Rights and Youth Rights Commission. 2020. Commentaires sur le Projet de loi nº 12, Loi visant à préciser la portée du droit à la gratuité scolaire et à permettre l'encadrement de certaines contributions financières pouvant être exigées [Comments on Bill 12, An Act to Clarify the Scope of the Right to Free Education and to Allow the Regulation of Certain Financial Contributions that May Be Required]. Montréal, Qué., Commission des droits de la personne et des droits de la jeunesse.
- Istat. 2019. Education and Training: Schools. Rome, National Statistics Institute. http://dati.istat.it/Index.aspx?lang=en& SubSessionId=cd4ad893-b46a-4e8c-82ff-3fd8b602ed17. (Accessed 20 September 2021.)
- India Ministry of Human Resource Development. 2020. National Education Policy 2020. New Delhi, Ministry of Human Resource Development.
- Italy Parliament. 2021. Legge 23 luglio 2021, N. 106. Conversione in legge, con modificazione, del decreto-legge 25 maggio 2021, n. 73 recante misure urgenti connesse all'emergenza da COVID-19, per le imprese, il lavoro, i giovani, la salute e i servizi territoriali [Law 23 July 2021, No. 106. Conversion into Law, with amendments, of Decree 25 May 2021, No. 73 concerning urgent measures linked to the COVID-19 emergency, for enterprises, labour market, youth, health and territorial services]. Rome, Italian Parliament.
- Jaume, D. and Willén, A. 2019. The long-run effects of teacher strikes: evidence from Argentina. Journal of Labor Economics, Vol. 37, No. 4, pp. 1097–139.
- \_\_\_\_. 2021. The effect of teacher strikes on parents. Journal of Development Economics, Vol. 152.
- Junemann, C. and Ball, S. J. 2013. ARK and the revolution of state education in England. *Education Inquiry*, Vol. 4, No. 3, art. 22611.
- Junemann, C. and Olmedo, A. 2020. New policy intercessors: philanthropy and public private partnerships. Gideon, J. and Unterhalter, E. (eds), *Critical Reflections on Public Private Partnerships*. Abingdon, UK, Routledge, pp. 115–133.
- Junqueira, D. 2016. Aprovada em Alagoas, 'escola sem partido' terá de superar batalha jurídica para ser implementada [Approved in Alagoas, 'nonpartisan school' will have to win legal battles to be implemented]. R7, 15 August. https://noticias.r7.com/educacao/aprovada-em-alagoas-escola-sem-partido-tera-de-superar-batalhajuridica-para-ser-implementada-15082016#/foto/1. (Accessed 30 September 2021.)
- Kasman, M., Heuberger, B. and Hammond, R. 2018. A Review of Large-Scale Youth Financial Literacy Education Policies and Programs. Washington, DC, Brookings Institution.
- Keidanren. 2020. Society 5.0 に向けて求められる初等中等教育改革 第二次提言—ダイバーシティ&インクルージョン を重視した 初等中等教育の実現— [Second Proposal for Reform of Elementary and Secondary Education Required for Society 5.0: Realization of Elementary and Secondary Education with an Emphasis on Diversity and Inclusion]. Tokyo, Keidanren.
- \_\_\_. 2021. About Keidanren. Tokyo, Keidanren. www.keidanren.or.jp/en/profile/pro001.html. (Accessed 30 September 2021.)
- Kirst, M. W. 2007. Politics of charter schools: competing national advocacy coalitions meet local politics. *Peabody Journal of Education*, Vol. 82, No. 2–3, pp. 184–203.

- KPMG. 2020. Q1'20 Venture Pulse Report: Global trends A Global Overview of Key Findings Uncovered from the Q4'19 Venture Pulse Report. London, KPMG Private Enterprise. https://home.kpmg/xx/en/home/campaigns/2020/04/q1venture-pulse-report-global.html. (Accessed 17 August 2021.)
- Lakasas, A. 2021. Εκπαιδευτικοί: Γυρίζουν την πλάτη στους συνδικαλιστές [Teachers: They turn their backs on the trade unionists]. I Kathimerini, 27 October. www.kathimerini.gr/society/561558448/ekpaideytikoi-gyrizoyn-tin-platistoys-syndikalistes-metexetastea-i-fason-axiologisi. (Accessed 1 November 2021.)
- Lara, K. E. 2021. ¿En qué consiste la educación en casa y qué impacto tendrá su regulación en Panamá? [What is homeschooling and what impact will its regulation have in Panama?]. *Panamá América*, 11 August. www.panamaamerica.com.pa/sociedad/en-que-consiste-la-educacion-en-casa-y-que-impacto-tendra-suregulacion-en-panama-1192268. (Accessed 17 August 2021.)
- Legistorm. 2020. Congressional Caucus on Education Innovation and Opportunity. Washington, DC, Legistorm. www.legistorm.com/organization/summary/123129/Congressional\_Caucus\_on\_Education\_Innovation\_ and\_Opportunity.html. (Accessed 17 August 2021.)
- LendED. 2021. Access to Quality Home Learning Resources. London, British Educational Suppliers Association. www.lended.org.uk. (Accessed 17 August 2021.)
- Leponiemi, L., Ruoho, N. and Tuominen, S. 2020. Development of an Innovation-friendly Education System. Helsinki, HundrED. (HundrED Research Report 14.)
- Linkiesta. 2021. Maggioranza e opposizione: sugli aiuti alle scuole paritarie i parlamentari sono tutti d'accordo [Majority and opposition: all parliamentarians agree to support schools with parity]. *Linkiesta*, 16 June,. www.linkiesta.it/2020/06/scuole-paritarie-private-decreto-rilancio-discussione-camera. (Accessed 17 August 2021.)
- Lubienski, C. 2019. Advocacy networks and market models for education. Parreira do Amaral, M., Steiner-Khamsi, G. and Thompson, C. (eds), Researching the Global Education Industry: Commodification, the Market and Business Involvement. Cham, Switzerland, Palgrave Macmillan, pp. 69–86.
- Lubienski, C., Scott, J. T., Rogers, J. and Welner, K. G. 2012. *Missing the Target? The Parent Trigger as a Strategy for Parental Engagement and School Reform*. Boulder, Colo., National Education Policy Center. (NEPC Policy Memo.)
- MacLellan, D. 2012. Faith-based schooling and the politics of education: a case study of Ontario, Canada. Politics and Religion Journal, Vol. 6, No. 1, pp. 37–60.
- Marianno, B. D. 2018. Down but not out: the National Education Association in federal politics. *Educational Policy*, Vol. 32, No. 2, pp. 234–54.
- Mármol Queraltó, J. 2021. A Critical Analysis of Investors' Logic in Business Discourse. Lancaster, UK, Lancaster University. (Universities and Unicorns Project.)
- Marrero, A. 2014. How public is Uruguay's public education? Global Dialogue, Vol. 4, No. 1.
- Martinis, P. 2020. The Onslaught of Privatisation on Uruguayan Education. Brussels, Education International. www.ei-ie. org/en/woe\_homepage/woe\_detail/16819/the-onslaught-of-privatisation-on-uruguayan-education-bypablo-martinis. (Accessed 17 August 2021.)
- Mayo-Wilson, E., Grant, S. and Supplee, L. H. 2021. Clearinghouse standards of evidence on the transparency, openness, and reproducibility of intervention evaluations. *Prevention Science*.
- McShane, M. 2020. A window into homeschooling advocacy: the HSLDA. *Forbes*, 1 May. www.forbes.com/sites/ mikemcshane/2020/05/01/a-window-into-homeschooling-advocacy-the-hslda. (Accessed 17 August 2021.)
- MEXT. 2020. The Image of the Transformation of Learning Brought by '1 Device for 1 Student with a High-speed Network'. Tokyo, Ministry of Education, Culture, Sport, Science and Technology.
- Michigan Campaign Finance Network. 2006. DeVos Dollars and 'Education Choice'. Lansing, Mich., Michigan Campaign Finance Network. https://mcfn.org/node/14/devos-dollars-and-education-choice. (Accessed 17 August 2021.)
- Moschetti, M., Martínez Pons, M., Bordoli, E. and Martinis, P. 2019. The increasing role of non-state actors in education policy-making: evidence from Uruguay. *Journal of Education Policy*, Vol. 35, No. 3, pp. 367-93.
- Moschetti, M. C. and Verger, A. 2020. Opting for private education: public subsidy programs and school choice in disadvantaged contexts. *Educational Policy*, Vol. 34, No. 1, pp. 65–90.
- Mouvement l'école ensemble. 2019. L'injuste système d'éducation québécois : L'équité du système d'éducation québécois comparée à celle des autres systèmes d'éducation provinciaux en vertu de données inédites de l'enquête PISA [Quebec's Unfair Education System: The Equity of Quebec's Education System Compared to Other Provincial Education Systems Using Unpublished PISA Data]. Gatineau, Qué., Mouvement l'école ensemble.
- \_\_\_\_. 2021. Mythes [Myths]. Gatineau, Qué., Mouvement l'école ensemble,. www.ecoleensemble.com/mythes. (Accessed 28 September 2021.)

Muñoz, F. 2020. Género y educación en Perú [Gender and education in Peru]. Paris, UNESCO. (Background paper for Global Education Monitoring Report on Inclusion and Education in Latin America and the Caribbean 2020.)

- Narodowski, M., Gottau, V. and Moschetti, M. 2016. Quasi-state monopoly of the education system and socioeconomic segregation in Argentina. *Policy Futures in Education*, Vol. 14, No. 6, pp. 687–700.
- Nathan, A. 2019. Why are Karnataka's schoolchildren unhappy with the mid-day meal? *The Hindu*, 31 May. www. thehindu.com/news/national/karnataka/why-are-karnatakas-schoolchildren-unhappy-with-the-mid-daymeal/article27378176.ece. (Accessed 17 August 2021.)
- National Assembly of Québec. 2020. Bill 12 (2019, Chapter 9): An Act to Clarify the Scope of the Right to Free Education and to Allow the Regulation of Certain Financial Contributions that May be Required. Montreal, Que., Québec Official Publisher.
- National Foundation for Educational Research. 2020. Final Evaluation of the Civil Society Education Fund, 2016–2019 (CSEF III). Slough, UK, National Foundation for Educational Research.
- OECD. 2017. Education Policy in Greece: A Preliminary Assessment. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2018. Private Philantropy for Development. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2021. Education Policy Outlook: Brazil With a Focus on National and Subnational Policies. Paris, Organisation for Economic Co-Operation and Development.
- Olmedo, A. 2013. From England with love... ARK, heterarchies and global 'philanthropic governance'. Journal of Education Policy, Vol. 29, No. 5, pp. 575–97.
- PAL Network. 2020. 2020 Annual Report. Nairobi, People's Action for Learning Network.
- Pane, J. 2018. Strategies for Implementing Personalized Learning While Evidence and Resources Are Underdeveloped. Arlington, Va., RAND Corporation.
- Papakonstantinou, P. and Kolympari, T. 2019. A bone of contention: teacher evaluation system in Greece. International Journal of Management in Education, Vol. 13, No. 1, pp. 40–58.
- Parliamentary Monitoring Group. 2018. Western Cape Provincial School Education Amendment Bill: Public Hearing. Cape Town, South Africa, Parliamentary Monitoring Group.
- Patil, L. and Brakman Reiser, D. 2021. Emerging Roles and Risks of Philanthropy in Global Education. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Pearce, C. 2017. Public Good over Private Profit: A Toolkit for Civil Society to Resist the Privatisation of Education. Johannesburg, South Africa, Global Campaign for Education.
- Peepul India. 2020. Transforming Lives through Education: Annual Report 2019–2020. New Delhi, Peepul India.
- Polanin, J., Caverly, S., Pollard, E., Bzura, R., Davis, E., Goldston, C., Gnedko-Berry, N., Lindsay, J., Nolan, E. and Song,
   M. 2021. A Reintroduction to the What Works Clearinghouse. Washington, DC, National Center for Education Evaluation, Institute of Education Sciences.
- Québec Government. 2020. Effectif scolaire de la formation générale des jeunes, selon diverses variables, années scolaires 2005–2006 à 2020–2021 [Enrolment in general youth education, by various variables, school years 2005–2006 to 2020–2021]. Montréal, Qué., Government of Québec.
- Québec Ministry of Education. 2021a. Écoles Privées [Private Schools]. Montréal, Qué., Government of Quebec. http://www.education.gouv.qc.ca/parents-et-tuteurs/ecoles-privees. (Accessed 28 September 2021.)
- \_\_\_\_. 2021b. Établissements dienseignement privés agréés aux fins de subventions: règles budgétaires pour liannée scolaire 2021–2022. Éducation préscolaire et enseignement primaire et secondaire [Private Educational Institutions Approved for Subsidies: Budgetary Rules for the 2021–2022 School Year. Pre-School, Primary and Secondary Education]. Montréal, Qué., Government of Québec.
- Ready, D. 2018. The Future of Brazil's 'Common Core'. Philadelphia, Pa., Consortium for Policy Research in Education Knowledge Hub. https://cprehub.org/blogs/future-brazil%E2%80%99s-%E2%80%9Ccommoncore%E2%80%9D. (Accessed 17 August 2021.)
- Reckhow, S. and Tompkins-Stange, M. 2018. Financing the education policy discourse: philanthropic funders as entrepreneurs in policy networks. *Interest Groups and Advocacy*, Vol. 7, No. 3, pp. 258–88.
- Reeve, E., Thow, A. M., Bell, C., Engelhardt, K., Gamolo-Naliponguit, E. C., Go, J. J. and Sacks, G. 2018. Implementation lessons for school food policies and marketing restrictions in the Philippines: a qualitative policy analysis. *Globalization and Health*, Vol. 14, No. 8, pp. 1–14.

- Reeves, T. C. and Lin, L. 2020. The research we have is not the research we need. Educational Technology Research and Development, Vol. 68, No. 4, pp. 1991–2001.
- Robertson, S. L. and Verger, A. 2012. Governing education through public private partnerships. Robertson, S. L., Mundy, K., Verger, A. and Menashy, F. (eds), Public Private Partnerships in Education: New Actors and Modes of Governance in a Globalizing World. Cheltenham, UK, Edward Elgar, pp. 21–42.
- Savedoff, W. D. 2019. What Is 'Country Ownership'? A Formal Exploration of the Aid Relationship. Washington, DC, Center for Global Development. (Working Paper 519.)
- Sawchuk, S. 2016. Outside the classroom, TFA leaves increasingly complex political trail. *Education Week*, 15 January. www.edweek.org/teaching-learning/outside-the-classroom-tfa-leaves-increasingly-complex-politicaltrail/2016/01. (Accessed 17 August 2021.)
- Sayed, Y. and Soudien, C. 2021. Managing a progressive educational agenda in post-apartheid South Africa: the case of education public-private partnerships. Zajda, J. (ed.), Third International Handbook of Globalisation, Education and Policy Research. Cham, Switzerland, Springer, pp. 117–38.
- Schoenfeld, A. H. 2006. What doesn't work: the challenge and failure of the What Works Clearinghouse to conduct meaningful reviews of studies of mathematics curricula. *Educational Researcher*, Vol. 35, No. 2, pp. 13–21.
- Schultz, A. 2018. How educators, philanthropists, and investors are revolutionizing education. *Barron's*, 22 September. www.barrons.com/articles/how-educators-philanthropists-and-investors-are-revolutionizingeducation-1537639200. (Accessed 1 August 2021.)
- Sivasubramaniam, M. 2021. Philanthropic and Faith-based Non-state Actors in the Low-fee Private School Sector in the Global South: Blurring the Lines of Philanthropy and Enterprise. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Skelton, A., Nolan, A., Mowbray, J., Kothari, J., Sepúlveda, M., Smirnova, M., Zinigrad, R., Fredman, S. and Epal Ratjen, S.
   2019. The Abidjan Principles: Guiding Principles on the Human Rights Obligations of States to Provide Public Education and to Regulate Private Involvement in Education. Abidjan, Côte d'Ivoire.
- Slavin, R. E. 2008. What works? Issues in synthesizing educational program evaluations. *Educational Researcher*, Vol. 37, No. 1, pp. 5–14.
- Srinivasaraju, S. 2020. The allegations against Akshaya Patra, and why a probe is needed. *The Wire*, 20 November. https://thewire.in/government/akshaya-patra-trustees-resignation-explainer. (Accessed 17 August 2021.)
- Srivastava, P. 2020. Framing Non-state Engagement in Education. Paris, UNESCO. (Think piece for Global Education Monitoring Report 2021/2.)
- Strauss, V. 2013. Global education market reaches \$4.4 trillion and is growing. *Washington Post*, 9 February. www.washingtonpost.com/news/answer-sheet/wp/2013/02/09/global-education-market-reaches-4-4trillion-and-is-growing. (Accessed 17 August 2021.)
- Takayama, K. 2013. Untangling the global-distant-local knot: the politics of national academic achievement testing in Japan. *Journal of Education Policy*, Vol. 28, No. 5, pp. 657–75.
- Tarlau, R. and Moeller, K. 2019. 'Philanthropizing' consent: how a private foundation pushed through national learning standards in Brazil. *Journal of Education Policy*, Vol. 35, No. 3, pp. 337–66.
- Teach for All. 2021. Teach for All Overview. New York, Teach for All.
- Technavio Research. 2020. Global K-12 testing and assessment market 2020-2024: increasing use of analytics to boost growth. *Business Wire*, 5 March. www.businesswire.com/news/home/20200305005463/en/Global-K-12-Testing-Assessment-Market-2020-2024-Increasing. (Accessed 17 August 2021.)
- Thapliyal, N. 2018. Resisting educational privatisation on screen: a critical analysis of two activist documentaries from India and the USA. Postcolonial Directions in Education, Vol. 7, No. 2, pp. 148–73.
- UNESCO. 2018. Walk before You Run: The Challenges of Results-based Payments in Aid to Education. Paris, UNESCO. (Global Education Monitoring Report Policy Paper 33.)
- UNICEF. 2020. Funding Compendium 2020. New York, UNICEF.
- United Nations. 2015. Guidelines on a Principle-Based Approach to the Cooperation Between the United Nations and the Business Sector. New York, United Nations.
- Uruguay Government. 2020. Ley 19889 [Law 19889]. Montevideo, Government of Uruguay. www.impo.com.uy/bases/ leyes/19889-2020. (Accessed 17 August 2021.)
- Uruguay Ministry of Education and Culture. 2008. Ley 18437 [Law 18437]. Montevideo, Ministry of Education and Culture.
- Vergari, S. 2007. The politics of charter schools. Educational Policy, Vol. 21, No. 1, pp. 15–39.

- Verger, A., Fontdevila, C., Rogan, R. and Gurney, T. 2019. Manufacturing an illusory consensus? A bibliometric analysis of the international debate on education privatisation. *International Journal of Educational Development*, Vol. 64, pp. 81–95.
- Verger, A., Fontdevila, C. and Zancajo, A. 2016. The Privatization of Education: A Political Economy of Global Education Reform. New York, Teachers' College Press. (International Perspectives on Education Reforms.)
- Vigneault, S. 2020. The UN to hold Quebec accountable for its segregated school system. *Our Schools/Our Selves*, Vol. Summer–Fall, pp. 30–31.
- Vivas, J. 2020. Snack company continues support programs to feed public schools. *Manila Bulletin*, 20 July. https://mb.com.ph/2020/07/20/snack-company-continues-support-programs-to-feed-public-schools. (Accessed 17 August 2021.)
- Wang, Y. 2020. Understanding Congressional coalitions: a discourse network analysis of Congressional hearings for the Every Student Succeeds Act. Education Policy Analysis Archives, Vol. 28, No. 1.
- Western Cape Education Department. 2020. Annual Report 2019/2020: Department of Education. Cape Town, South Africa, Western Cape Education Department.
- WHO. 2016. Framework of Engagement with Non-State Actors. Geneva, Switzerland, World Health Organization.
- \_\_\_\_. 2018a. Guide for Staff on Engagement with Non-State Actors. Geneva, Switzerland, World Health Organization.
- \_\_\_\_. 2018b. Handbook for Non-State Actors on Engagement with WHO. Geneva, Switzerland, World Health Organization.
- Wills, G. 2014. The Effects of Teacher Strike Activity on Student Learning in South African Primary Schools. Cape Town, Economic Research Southern Africa. (ERSA Working Paper 402.)
- Wilton Park. 2017. No Child Left Behind: What Is the Complementary Role of Non-state Actors in the Delivery of Quality Basic Education in Low Resource Environments? London, Foreign, Commonwealth and Development Office. (WP1577.)

# **CHAPTER 6**

- Abumiya, M. I. 2015. New Trends in Preschool Education and Childcare in Japan: Transition to a 'Comprehensive Support System for Children and Child-rearing'. Tokyo, National Institute for Educational Policy Research.
- Admas, F. 2019. Quality of early childhood education in private and government preschools of Addis Ababa, Ethiopia. International Journal of Early Childhood, Vol. 51, No. 2, pp. 163–76.
- Alfers, L. 2016. 'Our Children Do Not Get the Attention They Deserve': A Synthesis of Research Findings on Women Informal Workers and Child Care from Six Membership-based Organizations. Washington, DC, Women in Informal Employment: Globalizing and Organizing. (WIEGO Child Care Initiative.)
- Aljabreen, H. 2020. Montessori, Waldorf, and Reggio Emilia: a comparative analysis of alternative models of early childhood education. International Journal of Early Childhood, Vol. 52, pp. 337–53.
- Anderson, K., Raikes, A., Kosaraju, S. and Solano, A. 2017. National Early Childhood Care and Education Quality Monitoring Systems. Washington, DC, Center for Universal Education, Brookings Institution.
- Aran, M. A., Munoz Boudet, A. M. and Aktakke, N. 2016. Can regulations make it more difficult to serve the poor? The case of childcare services in Istanbul, Turkey. Journal of Human Development and Capabilities, Vol. 17, No. 4, pp. 558–82.
- Araujo, M. C., López-Boo, F. and Puyana, J. M. 2013. Overview of Early Childhood Development Services in Latin America and the Caribbean. Washington, DC, Inter-American Development Bank.
- Arup and Bernard van Leer Foundation. 2020. Proximity of Care: A New Approach to Designing for Early Childhood in Vulnerable Urban Contexts. London/The Hague, Arup/Bernard van Leer Foundation.
- Bahamas Ministry of Finance. 2019. Government embarks on preschool to university free education strategy. Nassau, Ministry of Finance. www.bahamasbudget.gov.bs/2020/news-centre/latest-news/2019/06/19/governmentembarks-preschool-university-free-education-strategy. (Accessed 17 July 2021.)
- Bahrain Government. 1998. Private Educational and Training Institutions Law. Manama, Ministry of Education.
- Bastos, P. and Straume, O. R. 2013. Preschool Education in Brazil: Does Public Supply Crowd out Private Enrollment? Washington, DC, Inter-American Development Bank. (Working Paper 463.)
- Baum, D. 2021. Non-State Actors in Early Childhood Education: Implications for Education Equity and Quality. Paris, UNESCO. (Global Education Monitoring Report Fellowship Paper.)
- Baum, D. R., Cooper, R. and Lusk-Stover, O. 2018. Regulating market entry of low-cost private schools in sub-Saharan Africa: towards a theory of private education regulation. *International Journal of Educational Development*, Vol. 60, pp. 100–12.
- Bidwell, K. and Watine, L. 2014. Exploring Early Education Programs in Peri-urban Settings in Africa: Nairobi, Kenya Study Summary. New Haven, Conn., Innovations for Poverty Action.

- Blome, A. 2018. Von Kinder, Küche, Kirche zu Kinder, Karriere, KiTa? Geschlechterrollen, Familienpolitik und Religion im Wandel der Zeit [From children, kitchen, church to children, career, daycare? Gender roles, family politics and religion through the ages]. Sozialer Fortschritt, Vol. 67, No. 3, pp. 453–75.
- Bold, C. 2006. A Matter of Belonging: How Faith-based Organisations Can Strengthen Families and Communities to Support Orphans and Vulnerable Children. London, Christian Aid.
- Bouchard, I., Cheung, L. and Pacheco, G. 2021. Evaluating the impact of 20 hours free early childhood education on mothers' labour force participation and earnings. *New Zealand Economic Papers*, Vol. 55, No. 2, pp. 188–202.
- Brett, R. 2018. Roma and Traveller Children with a Parent in Prison: A Follow-Up Report with Case Studies and Recommendations. Montrouge, France, Children of Prisoners Europe.
- BRIDGE, Ilifa Labantwana, National ECD Alliance, Nelson Mandela Foundation, Smartstart and South African Congress for Early Childhood Development. 2020. The Plight of the ECD Workforce: An Urgent Call for Relief in the Wake of COVID-19. Johannesburg, South Africa, BRIDGE.
- Britto, P. R., Engle, P. L. and Super, C. 2013. Handbook of Early Childhood Development Research and Its Impact on Global Policy. New York, Oxford University Press.
- Britto, P. R., Lye, S. J., Proulx, K., Yousafzai, A. K., Matthews, S. G., Vaivada, T., Perez-Escamilla, R., Rao, N., Ip, P., Fernald,
  L. C. H., MacMillan, H., Hanson, M., Wachs, T. D., Yao, H., Yoshikawa, H., Cerezo, A., Leckman, J. F. and Bhutta, Z.
  A. 2017. Nurturing care: promoting early childhood development. *The Lancet*, Vol. 389, No. 10064, pp. 91–102.
- Burroughs, B. 2017. YouTube kids: the app economy and mobile parenting. Social Media + Society, Vol. 3, No. 2.
- Caddy, E. 2017. There Are at Least 2700 Informal Child Daycares in Nairobi: Tiny Totos is Working to Help Them Upgrade Their Services. London, TheirWorld. https://theirworld.org/voices/tiny-totos-kenya-helping-informal-daycares-innairobi-upgrade-services. (Accessed 17 August 2021.)
- Cárdenas, M. and Cadena, A. M. 2020. How to Prioritize Early Childhood? A Note on the Recent Experience in Colombia. Washington, DC, Center for Global Development. (Policy Paper 174.)
- Center on the Developing Child. 2021. What Is Early Childhood Development? A Guide to the Science. Cambridge, Mass., Harvard University. https://developingchild.harvard.edu/guide/what-is-early-childhood-development-aguide-to-the-science. (Accessed 17 August 2021.)
- Cheruiyot, J. K. 2019. An assessment of the challenges of children in prison with their mothers: a case of Langata Women Maximum Prison. Interdisciplinary Journal on the African Child, Vol. 1, No. 1.
- Chile Government. 2021. ¿Qué es Chile Crece Contigo? [What is Chilee Crece Contigo?]. Santiago, Ministry of Social Development. www.crececontigo.gob.cl/acerca-de-chcc/que-es. (Accessed 17 July 2021.)
- Chile Undersecretary of Early Childhood Education. 2019. Informe de Caracterizacion de la Educacion Parvularia: Oficial 2019 [Early Childhood Education Characterization Report: Official 2019]. Santiago, Undersecretary of Early Childhood Education.
- CLADE and OMEP. 2020. El Derecho a la Educación y al Cuidado en la Primera Infancia: Perspectivas desde América Latina y el Caribe [The Right to Education and Care in Early Childhood: Perspectives from Latin America and the Caribbean]. São Paulo, Brazil, Latin American Campaign for the Right to Education/World Organization for Early Childhood Education.
- Clark, H., Coll-Seck, A. M., Banerjee, A., Peterson, S. and others. 2020. A future for the world's children? A WHO– UNICEF–Lancet Commission. *The Lancet*, Vol. 395, No. 10224, pp. 605–58.
- Datla, A. 2018. Cuna Más: Peru's Home Visiting Program Evolves into a Comprehensive Early Childhood Development Strategy. Cambridge, Mass., Harvard Kennedy School.
- \_\_\_\_. 2021. Crianca Feliz: Brazil's Ambitious Early Childhood Program. Cambridge, Mass., Harvard Kennedy School.
- Deloitte. 2018. A New Era of Education: China Education Development Report 2018. Beijing, Deloitte China.
- Denmark Ministry of Children and Education. 2020. The Strengthened Pedagogical Curriculum: Framework and Content. Copenhagen, Ministry of Children and Education.
- Department of Education. 2019. Survey of Childcare and Early Years Providers: Main Summary England. London, Department of Education.
- Devercelli, A. E. and Beaton-Day, F. 2020. Better Jobs and Brighter Futures: Investing in Childcare to Build Human Capital. Washington, DC, World Bank.
- Diker, G. 2001. Organización y Perspectivas de la Educación Inicial en Iberoamérica: Principales Tendencias [Organization and Perspectives of Initial Education in Ibero-America: Main Trends]. Madrid, Organization of Ibero-American States.
- Dwyer, J. G. 2014. Jailing black babies. Utah Law Review, No. 3, pp. 465–541.
- Dýrfjörð, K. and Magnúsdóttir, B. R. 2016. Privatization of early childhood education in Iceland. Research in Comparative and International Education, Vol. 11, No. 1, pp. 80–97.

- ECDA. 2020. About Us. Singapore, Early Childhood Development Agency. www.ecda.gov.sg/pages/aboutus.aspx. (Accessed 17 August 2021.)
- Education Counts. 2021. ECE Finances. Wellington, Ministry of Education. www.educationcounts.govt.nz/statistics/ finances. (Accessed 1 November 2021.)
- Education International. 2020. Canada: The Federation of Childcare Workers of Quebec Calls an Indefinite General Strike. Brussels, Education International. www.ei-ie.org/en/item/23505:canada-the-federation-of-childcareworkers-of-quebec-calls-an-indefinite-general-strike. (Accessed 17 August 2021.)
- Edwards, D. B., Okitsu, T. and Mwanza, P. 2019. Low-fee private schools, the state, and globalization: a market analysis within the political sociology of education and development. *Education Policy Analysis Archives*, Vol. 27, No. 133.
- Eswatini Government. 2018. National Education and Training Sector Policy. Mbabane, Ministry of Education and Training. Ethiopia Federal Ministry of Education. 2020. Education Statistics Annual Abstract September 2019–March 2020. Addis Ababa, Federal Ministry of Education.
- European Commission/EACEA/Eurydice. 2019. Key Data on Early Childhood Education and Care in Europe: 2019 Edition. Brussels, European Commission. (Eurydice Report.)
- \_\_\_\_. 2021a. Belgium: French Community Educational Guidelines for Children ≥ 2.5 Years. Brussels, European Commission. https://eacea.ec.europa.eu/national-policies/eurydice/belgique-communaute-fran%C3%A7aise/educationalguidelines-children-%E2%89%A5-25-years\_pt-pt. (Accessed 17 August 2021.)
- \_\_\_\_. 2021b. France: Early Childhood Education and Care. Brussels, European Commission. https://eacea.ec.europa.eu/ national-policies/eurydice/france/early-childhood-education-and-care\_en. (Accessed 17 August 2021.)
- \_\_\_\_. 2021c. France: Home-based Provision. Brussels, European Commission. https://eacea.ec.europa.eu/national-policies/ eurydice/france/home-based-provision\_en. (Accessed 17 August 2021.)
- European Public Service Union. 2017. Unions Discuss Quality Employment in Childcare Services. Brussels, European Public Service Union. www.epsu.org/article/unions-discuss-quality-employment-childcare-services. (Accessed 17 July 2021.)
- Eurostat. 2021. Living Conditions in Europe: Childcare Arrangements. Luxembourg, Eurostat. https://ec.europa.eu/eurostat/ statistics-explained/index.php?title=Living\_conditions\_in\_Europe\_-\_childcare\_arrangements#Childcare\_ and\_education\_arrangements. (Accessed 17 August 2021.)
- Evans, D. K. and Kosec, K. 2012. Early Child Education: Making Programs Work for Brazil's Most Important Generation. Washington, DC, World Bank.
- Expertisecentrum Kinderopvang. 2021. Expertisecentrum kinderopvang: Van, voor en mét de sector [Childcare Expertise Centre: From, for and with the sector]. Utrecht, Netherlands, Expertisecentrum Kinderopvang. https://expertisecentrumkinderopvang.nl. (Accessed 17 August 2021.)
- Fédération Française des Entreprises de Crèches. 2021. Qui sommes-nous [Who are we]. Boulogne-Billancourt, France, French Federation of Nursery Enterprises. (Accessed 17 August 2021.)
- Finland Government. 2018. Varhaiskasvatuslaki [Early Childhood Education and Care Act]. Helsinki, Ministry of Justice. www.finlex.fi/fi/laki/ajantasa/2018/20180540. (Accessed 17 August 2021.)
- FLACSO. 2020. Informe Final: Evaluación REDCUDI [Final Report: Evaluation of the National Child Care and Development Network]. San José, Social Development and Family Allowances Fund/National Care Network/Latin American Faculty of Social Sciences.
- Gallagher, A. 2017. Growing pains? Change in the New Zealand childcare market 2006–2016. New Zealand Geographer, No. 73, pp. 15–24.
- Ghana Ministry of Finance. 2019. Medium Term Expenditure Framework (MTEF) for 2019-2022: Ministry of Education – Programme Based Budget Estimates for 2019. Accra, Ministry of Finance.
- Goshin, L. S., Byrne, M. W. and Blanchard-Lewis, B. 2014. Preschool outcomes of children who lived as infants in a prison nursery. *Prison Journal*, Vol. 94, No. 2, pp. 139–58.
- Gould, E. and Blair, H. 2020. Who's Paying Now? The Explicit and Implicit Costs of the Current Earlycare and Education System. Washington, DC, Economic Policy Institute.
- Gromada, A., Richardson, D. and Rees, G. 2020. Childcare in a Global Crisis: The Impact of COVID-19 on Work and Family Life. Florence, Italy, UNICEF Office of Research – Innocenti. (Research Brief 18.)
- Gupta, S. 2020. Do Preschools Add 'Value'? Evidence on Achievement Gaps from Rural India. Brighton, UK, University of Sussex Business School. (Working Paper 20-2020.)

- Halter, E. 2018. Parental prisoners: the incarcerated mother's constitutional right to parent. Journal of Criminal Law and Criminology, Vol. 108, No. 3, pp. 539–68.
- Haskins, A. R. 2014. Unintended consequences: effects of paternal incarceration on child school readiness and later special education placement. *Sociological Science*, Vol. 1, pp. 141–58.
- Hayden, J. and Wai, S. 2013. Community-based approaches to early childhood development: a matter of degree. Brito,
   P. R., Engle, P. L. and Super, C. M. (eds), Handbook of Early Childhood Development Research and Its Impact on
   Global Policy. New York, Oxford University Press.
- Herrera, C., Vives, A., Carvallo, C. and Molina, H. 2016. Protección integral de la infancia a través del subsistema 'Chile Crece Contigo': análisis de una política para romper el ciclo intergeneracional de la pobreza y la inequidad [Comprehensive protection of children through the 'Chile Grows With You' subsystem: analysis of a policy to break the intergenerational cycle of poverty and inequity]. González Contró, M., Mercer, R. and Minujin., A. (eds), Lo Esencial no Puede Ser Invisible a los Ojos: Pobreza e Infancia en América Latina [The Essential Cannot Be Invisible to the Eyes: Poverty and Childhood in Latin America]. Mexico City, Universidad Nacional Autónoma de México.
- Home Grown. 2020. Home-based Child Care Fact Sheet. https://homegrownchildcare.org/wp-content/uploads/2020/12/ HomeGrown\_Child-Care-Fact-Sheet\_final.pdf. (Accessed 17 August 2021.)
- Hughes, R., Kitsao-Wekulo, P., Muendo, R., Bhopal, S., Kimani-Murage, E., Hill, Z. and Kirkwood, B. 2021. Who actually cares for children in slums? Why we need to think, and do, more about paid childcare in urbanizing sub-Saharan Africa. *Philosophical Transactions of the Royal Society B*, Vol. 376, No. 1827.
- IFC. 2019. Tackling Childcare: A Guide for Employer-Supported Childcare. Washington, DC, International Finance Corporation.
- \_\_\_\_. 2020. Tackling Childcare: Employer-Supported Childcare in Cambodia. Washington, DC, International Finance Corporation.
- IFC, Australian Aid and UNICEF. 2018. Tackling Childcare: The Business Case for Employer-supported Childcare in Sri Lanka. Washington, DC, International Finance Corporation.
- IFC and Bright Horizons. 2019. The Benefits and Challenges of a Workplace Creche: Employer-supported Childcare in India. Washington, DC, International Finance Corporation and Bright Horizons Family Solutions.
- ILO. 2018. Women and Men in the Informal Economy: A Statistical Picture. Geneva, Switzerland, International Labour Organization.
- ILO and IFC. 2018. Towards Gender Equality: Lessons from Factory Compliance Assessments. Geneva, Switzerland/ Washington, DC, International Labour Organization/International Finance Corporation. (Better Factories Cambodia.)
- ILO and WIEGO. 2019. Extending Childcare Services to Workers in the Informal Economy: Policy Lessons from Country Experiences. Geneva, Switzerland/Washington, DC, International Labour Organization/Women in Informal Employment: Globalizing and Organizing.
- India Ministry of Human Resource Development. 2020. National Education Policy 2020. New Delhi, Ministry of Human Resource Development.
- INEP. 2020. Matrículas na educação infantil crescem 12,6% nos últimos cinco anos [Enrolment in kindergarten grow 12.6% in the last five years]. Brasilia, National Institute for Educational Studies and Research Anísio Teixeira. http://portal. inep.gov.br/artigo/-/asset\_publisher/B4AQV9zFY7Bv/content/id/6836830. (Accessed 17 August 2021.)
- Ionescu, M., Trikic, Z. and Pinto, L. M. 2019. Towards Integrated Early Childhood Systems: Building the Foundations Toolkit. Leiden, Netherlands, International Step by Step Association.
- Jamil, B. R. and Saeed, S. 2018. In Punjab, Variations in Entry into Pre-primary Education Can Lead to Lower Learning Outcomes. Washington, DC, Global Partnership for Education. www.globalpartnership.org/blog/punjabvariations-entry-pre-primary-education-can-lead-lower-learning-outcomes. (Accessed 17 July 2021.)
- Japan Government. 2014. Cabinet Office: Outline of Duties. Tokyo, Cabinet Office.
- Joo, Y. S., Magnuson, K., Duncan, G. J., Schindler, H. S., Yoshikawa, H. and Ziol-Guest, K. M. 2020. What works in early childhood education programs?: A meta-analysis of preschool enhancement programs. *Early Education and Development*, Vol. 31, No. 1, pp. 1–26.
- Jordan Ministry of Education and UNICEF. 2020. Out-of-School Children Initiative: Country Report on Out-of-School Children. Amman, Ministry of Education/UNICEF.
- Josephson, K., Guerrero, G. and Coddington, C. 2017. Supporting the Early Childhood Workforce at Scale: The Cuna Más Home Visiting Program in Peru. Washington, DC, Results for Development.
- Kaisahang Buhay Foundation. 2021. KBF Services. Quezon City, Philippines, Kaisahang Buhay Foundation. www.kbf.ph/ services. (Accessed 17 August 2021.)

Kamerman, S. B. 2006. A Global History of Early Childhood Education and Care. Paris, UNESCO. (Background paper for EFA Global Monitoring Report 2007.)

Kaneko, M., Lombardi, J. and Weisz, A. 2020. Support Programs for Home-based Child Care: A Global Study. London/San Francisco, Calif., Spring Impact/Echidna Giving.

Karch, A. 2013. Early Start: Preschool Politics in the United States. Ann Arbor, Mich., University of Michigan Press.

- Kim, J., Hailu, B., Rose, P., Rossiter, J., Teferra, T. and Woldehanna, T. 2022. Persistent inequalities in early years' access and learning: evidence from a large-scale expansion of pre-primary education in Ethiopia. *Early Childhood Research Quarterly*, Vol. 58, pp. 103–14.
- King, K. M., Crouch, L., Wils, A. and Baum, D. R. 2020. How well are we measuring access to early childhood education?
   Wiseman, A. W. (ed.), Annual Review of Comparative and International Education 2019, Vol. 39. Bingley, UK, Emerald Publishing, pp. 171–89. (International Perspectives on Education and Society.)
- Knijn, T. and Lewis, J. 2017. ECEC: Childcare markets in the Netherlands and England. Unger, B., van der Linde, D. and Getzner, M. (eds), *Public or Private Goods?* Cheltenham, UK, Edward Elgar.
- Kumpulainen, K. 2018. Respecting Children and Families: A Case Study of the Finnish Early Childhood Educationand Care System. New York, Teachers College. (The Early Advantage: International Case Studies of Early Childhood Education and Care System.)

LaingBuisson. 2020. Childcare UK Market Report. London, LaingBuisson/Cainreagle/Office for National Statistics.

- Lancet, The. 2016. Advancing early childhood development: from science to scale. *The Lancet*, Vol. 389, No. 10064. (The 2016 *Lancet* Early Childhood Development Series.)
- Li, H., Yang, W. and Chen, J. J. 2016. From 'Cinderella' to 'Beloved Princess': the evolution of early childhood education policy in China. International Journal of Child Care and Education Policy, Vol. 10, No. 1.
- Lieber, C. 2020. No, my toddler doesn't need to learn to code. *The New York Times*, 21 July. www.nytimes. com/2020/07/21/parenting/stem-toys-kids.html. (Accessed 17 August 2021.)
- Liu, C. and Hwang, G.-J. 2021. Roles and research trends of touchscreen mobile devices in early childhood education: review of journal publications from 2010 to 2019 based on the technology-enhanced learning model. Interactive Learning Environments.
- Loizillon, A. and Leclercq, F. 2016. New Horizons: A Review of Early Childhood Care and Education in Asia and the Pacific. Bangkok, UNESCO Regional Bureau for Education in Asia and the Pacific.
- Malik, R., Hamm, K., Schochet, L., Novoa, C., Workman, S. and Jessen-Howard, S. 2018. America's Child Care Deserts in 2018. Washington, DC, Center for American Progress.
- Martin, E. 2017. Hidden consequences: the impact of incarceration on dependent children. National Institute of Justice Journal, No. 278.
- Marzonetto, G. and Rodríguez, C. 2017. La coordinación institucional de políticas de cuidado infantil en la Argentina: desafío necesario para el abordaje de las desigualdades [Institutional coordination of childcare policies in Argentina: a necessary challenge to address inequalities]. *Cuadernos de Economía Crítica*, Vol. 4, No. 7, pp. 43–69.
- Miller, L., Cameron, C., Dalli, C. and Barbour, N. 2017. The SAGE Handbook of Early Childhood Policy. London, SAGE.
- Morgan, P. L., Wang, Y. and Woods, A. D. 2021. Risk and protective factors for frequent electronic device use of online technologies. *Child Development*, Vol. 92, No. 2, pp. 704–14.
- Murray, J., Farrington, D. P. and Sekol, I. 2012. Children's antisocial behavior, mental health, drug use, and educational performance after parental incarceration: a systematic review and meta-analysis. *Psychological Bulletin*, Vol. 138, No. 2, pp. 175–210.
- NAEYC. 2021. Strategic Direction. Washington, DC, National Association for the Education of Young Children.
- National Day Nurseries Association. 2019. About the National Day Nurseries Association (NDNA). Huddersfield, UK, National Day Nurseries Association. www.ndna.org.uk/NDNA/All\_About\_Us/About\_Us.aspx. (Accessed 17 August 2021.)
- Neuman, M. J. and Devercelli, A. E. 2013. What Matters Most for Early Childhood Development: A Framework Paper. Washington, DC, World Bank. (SABER Working Paper 5.)
- Neuwelt-Kearns, C. and Ritchie, J. 2020. Investing in Children? Privatisation and Early Childhood Education in Aotearoa New Zealand. Auckland, New Zealand, Child Poverty Action Group.
- New Zealand Ministry of Education. 2020. ECE Funding Handbook. Wellington, Ministry of Education. www.education. govt.nz/early-childhood/funding-and-data/funding-handbooks/ece-funding-handbook/the-ece-fundingsubsidy. (Accessed 17 July 2021.)
- Norway Ministry of Education and Research. 2014. OECD: Thematic Review of Early Childhood Education and Care Policy in Norway – Background Report. Oslo, Ministry of Education and Research.

- Novoa, C. 2020. How Child Care Disruptions Hurt Parents of Color Most. Washington, DC, Center for American Progress. www.americanprogress.org/issues/early-childhood/news/2020/06/29/486977/child-care-disruptions-hurtparents-color. (Accessed 17 August 2021.)
- OECD. 2016. Education in Colombia. Paris, Organisation for Economic Co-operation and Development. (Reviews of National Policies for Education.)
- \_\_\_\_. 2018. Engaging Young Children: Lessons from Research about Quality in Early Childhood Education and Care. Paris, Organisation for Economic Co-operation and Development. (Starting Strong.)
- \_\_\_\_\_. 2019a. Education at a Glance 2019. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2019b. PF3.3: Informal Childcare Arrangements. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2020. Is Childcare Affordable? Paris, Organisation for Economic Co-operation and Development. (Policy Brief on Employment, Labout and Social Affairs.)
- Padinha, T. A. and Goia, M. R. 2021. Private bilingual schools and teacher qualifications. Cadernos de Pesquisa, Vol. 51.
- Paschall, K. 2019. Nearly 30 Percent of Infants and Toddlers Attend Home-based Child Care as Their Primary Arrangement. Bethesda, Md., Child Trends. www.childtrends.org/blog/nearly-30-percent-of-infants-and-toddlers-attendhome-based-child-care-as-their-primary-arrangement. (Accessed 17 August 2021.)
- Penn, H. 2019. Putting Childcare at the Heart of the Social Market Economy. Brussels, Wilfried Martens Centre for European Studies. (Policy Brief.)
- Pesando, L. M., Wolf, S., Behrman, J. R. and Tsinigo, E. 2020. Are private kindergartens really better? Examining preschool choices, parental resources, and children's school readiness in Ghana. *Comparative Education Review*, Vol. 64, No. 1, pp. 107–36.
- Philippines Congress. 2012. Republic Act No. 10157: An Act Institutionalizing the Kindergarten Education into the Basic Edication System and Appropriating Fund Therefor. Manila, Congress of the Philippines.
- \_\_\_\_. 2013. Republic Act No. 10410: An Act Recognizing the Age from Zero (0) to Eight (8) Years as the First Crucial Stage of Educational Development and Strengthening the Early Childhood Care and Development System, Appropriating Funds Therefor and for Other Purposes. Manila, Congress of the Philippines.
- Philippines ECCD Council. 2015. Guidelines on the Registration and Granting of Permit and Recognition to Public and Private Child Development Centers/Learning Centers Offering Early Childhood Programs for 0 to 4 Years Old Filipino Children. Pasig City, Philippines, Early Childhood Care and Development Council.
- \_\_\_\_. 2019. Revised Philippine ECCD Checklist: Technical and Administration Manual. Pasig City, Philippines, Early Childhood Care and Development Council.
- Poehlmann-Tynan, J. and Turney, K. 2021. A developmental perspective on children with incarcerated parents. Child Development Perspectives, Vol. 15, No. 1, pp. 3–11.
- Ponguta, L. A., Aggio, C., Moore, K., Hartwig, E., Jang, B., Markovic, J., Hasanova, L., Bosworth, J. and Grover, D. 2019. Exploratory analysis of decentralized governance and its implications for the equity of early childhood education services in four countries of Europe and Central Asia. *Early Years*, Vol. 39, No. 3, pp. 326–42.
- Rao, N., Cohrssen, C., Chan, S. and Richards, B. 2020. An Evaluation of the Early Childhood Care and Development Programme Bhutan. Thimphu, UNICEF/Bhutan Ministry of Education.
- Rao, N. and Lau, G. L. C. 2018. Responsive policymaking and implementation: enhancing ECEC system quality and equity in Hong Kong. Kagan, S. L. (ed.), *Early Childhood Systems that Lead by Example: A Comparative Focus on International Early Childhood Education*. New York, Teachers College Press/National Center on Education and the Economy.
- Rauf, D. S. 2020. Early education startup closes on \$50 million fundraising round. EDWeek Market Brief, 12 October. https://marketbrief.edweek.org/marketplace-k-12/early-education-startup-closes-50-million-fundraisinground. (Accessed 17 August 2021.)
- Rede Nacional Primeira Infância. 2018. *Quem somos [Who are we]*. Brasília, Rede Nacional Primeira Infância. http:// primeirainfancia.org.br/quem-somos. (Accessed 17 August 2021.)
- Reid Chassiakos, Y., Radesky, J., Christakis, D., Moreno, M. A. and Cross, C. 2016. Children and adolescents and digital media. *Pediatrics*, Vol. 138, No. 5.
- Rideout, V. and Robb, M. B. 2020. The Common Sense Census: Media Use by Kids Age Zero to Eight. San Francisco, Calif., Common Sense Media.
- Ruutiainen, V., Alasuutari, M. and Karila, K. 2021. Selectivity of clientele in Finnish private early childhood education and care. Nordic Journal of Studies in Educational Policy, Vol. 7, No. 2, pp. 91–105.

Samoa Ministry of Education, Sports and Culture. 2019. Education Statistical Digest. Apia, Ministry of Education, Sports and Culture.

- \_\_\_\_. 2020. Education Sector COVID-19 Response Plan: Implementing the COVID Basic Education Response Programme. Apia, Ministry of Education, Sports and Culture.
- Saudi Arabia Government. 2021. *Qurrah*. Riyadh, Human Resources Development Fund. https://qurrah.sa. (Accessed 17 August 2021.)
- Schulte, B. and Durana, A. 2016. The New America Care Report. Washington, DC, New America.
- Schweinhart, L. J. 2016. Use of early childhood longitudinal studies by policy makers. International Journal of Child Care and Education Policy, Vol. 10.
- SEAMEO INNOTECH. 2020. Quality Assurance in Early Childhood Care and Development (ECCD) in Southeast Asia. Quezon City, Philippines, Regional Center for Educational Innovation and Technology, Southeast Asian Ministers of Education Organization.
- Seychelles Government. 2011. The Seychelles Framework for Early Childhood Care and Education: Winning for Children A Shared Commitment. Victoria, Department of Education.
- Shdaimah, C., Palley, E. and Miller, A. 2018. Voices of child care providers: an exploratory study on the impact of policy changes. International Journal of Child Care and Education Policy, Vol. 12.
- Simon, A., Owen, C., Hollingworth, K. and Rutter, J. 2015. Provision and Use of Preschool Childcare in Britain: Summary of Research Findings. London, UCL Institute of Education.
- Sitati, E. M., Ndirangu, M., Kennedy, B. and Rapongo, G. S. 2016. Implementation of early childhood development education service standard guidelines on physical facilities in public and private early childhood education centres: Kakamega County, Kenya. *Early Child Development and Care*, Vol. 186, No. 11, pp. 1765–78.
- Slot, P. 2018. Structural Characteristics and Process Quality in Early Childhood Education and Care: A Literature Review. Paris, Organisation for Economic Co-operation and Development. (Education Working Paper 176.)
- Smith, P. S. and Gampell, L. 2011. Children of Imprisoned Parents. Copenhagen, Danish Institute for Human Rights. South Africa Department of Social Development. 2020. Social Development Sets up Workstreams to Conduct Risk
- Assessment and State of Readiness for the Early Childhood Development (ECD) Centres. Pretoria, Department of Social Development. www.dsd.gov.za/index.php/latest-news/21-latest-news/183-social-development-sets-upworkstreams-to-conduct-risk-assessment-and-state-of-readiness-for-the-early-childhood-development-ecdcentres. (Accessed 17 August 2021.)
- Stack, N. 2013. Making a case for early intervention: the role of developmental neuroscience. Wasserman, L. H. and Zambo, D. (eds), Early Childhood and Neuroscience: Links to Development and Learning, Educating the Young Child, Vol. 7. Dordrecht, Netherlands, Springer, pp. 157–70.
- Statistics South Africa. 2019. General Household Survey 2018. Pretoria, Statistics South Africa. (Statistical Release P0318.)
- Stella, C., Sequeira, V. C. and Rosa, T. L. 2016. Mothers and babies in a Brazilian prison: a study exploring the impact of prison conditions. Forensic Research & Criminology International Journal, Vol. 2, No. 3, pp; 104–10.
- Step by Step. 2019. Strengthening the Role of Civil Society in Assuring Equity and Excellence in Early Childhood. Skopje, Foundation for Educational and Cultural Initiatives Step by Step. www.stepbystep.org.mk/en/proektzajaknuvanje-na-ulogata. (Accessed 17 July 2021.)
- Strehmel, P. 2019. Measures of personnel development in different types of German early childhood education (ECE) enterprises. Strehmel, P., Heikka, J., Hujala, E., Rodd, J. and Waniganayake, M. (eds), *Leadership in Early Education in Times of Change: Research from Five Continents*. Opladen, Germany, Verlag Barbara Budrich. (International Leadership Research Forum Early Education Research monograph 3.)
- Sultana, R. 2009. Jordan's Early Childhood Development Initiative: Making Jordan Fit for Children. Amman, UNICEF Middle East and North Africa Regional Office. (Learning Series 2.)
- Suzuki, W. 2019. Japan's free childcare program no panacea for daycare waitlists. *Nippon.com*, 11 June. www.nippon. com/en/in-depth/d00489/japan's-free-childcare-program-no-panacea-for-daycare-waitlists.html. (Accessed 17 August 2021.)
- Tan, C. T. 2017. Enhancing the quality of kindergarten education in Singapore: policies and strategies in the 21st century. International Journal of Child Care and Education Policy, Vol. 11, No. 7, pp. 1–22.
- The Knesset. 2019. Education Committee Calls for Speedy Transfer of Authority to Supervise Child Daycare Centers from the Labor Ministry to the Education Ministry. Jerusalem, The Knesset. https://m.knesset.gov.il/en/News/ PressReleases/Pages/press16719h.aspx. (Accessed 17 August 2021.)
- Trinidad and Tobago Ministry of Education. 2021. Early Childhood Care and Education (ECCE) Division. Port of Spain, Ministry of Education. www.moe.gov.tt/e-c-c-e. (Accessed 17 August 2021.)

Tshomo, P. 2017. Early Childhood Care and Development in Bhutan: A Case for Investment. Thimphu, Ministry of Education/UNICEF.

Uganda Ministry of Education and Sports. 2017. Education Abstract 2017. Kampala, Ministry of Education and Sports.

- UN Women. 2019. Investing in Early Childhood Education and Care in Kyrgyz Republe: An Assessment of Care Deficits, Costs and Impact on Employment, Gender Equality and Fiscal Returns. New York, UN Women.
- UNESCO. 2016a. Asia-Pacific Regional Report: Financing for Early Childhood Care and Education (ECCE). Bangkok, UNESCO Regional Bureau for Education in Asia and the Pacific.
- \_\_\_\_. 2016b. Global Education Monitoring Report 2016: Education for People and Planet Creating Sustainable Futures for All. Paris, UNESCO.
- \_\_\_\_. 2019. Regional Guidelines on Innovative Financing Mechanisms and Partnerships for Early Childhood Care and Education (ECCE). Bangkok UNESCO Regional Bureau for Education in Asia and the Pacific/SEAMEO Regional Center for Early Childhood Care Education and Parenting.
- \_\_\_\_. 2020. The Impact of COVID-19 on ECCE Sector: Lesson Learned and Promising Practices from the Asia-Pacific. Bangkok, UNESCO Regional Bureau for Education in Asia and the Pacific.
- \_\_\_. 2021. Right to Pre-primary Education: A Global Study. Paris, UNESCO.
- UNICEF. 2017. 2017 Status Report on Early Childhood Care and Education in Pacific Island Countries. Suva, UNICEF/Pacific Regional Council for Early Childhood Care and Education.
- \_\_\_\_. 2019. A World Ready to Learn: Prioritizing Quality Early Childhood Education. New York, UNICEF.
- UNICEF, ILO and WIEGO. 2021. Family-friendly Policies for Workers in the Informal Economy: Protecting and Ensuring Social Protection and Care Systems for All Children and Families in the Context of COVID-19 and Beyond. New York/Geneva, Switzerland, UNICEF/International Labour Organization/Women in Informal Employment: Globalizing and Organizing.
- United Arab Emirates Ministry of Education. 2018. Institutions Compliance Inspection Manual for Early Childhood Education. Abu Dhabi, Ministry of Education.
- Vaala, S. E. and LaPierre, M. A. 2014. Marketing genius: the impact of educational claims and cues on parents' reactions to infant/toddler DVDs. *Journal of Consumer Affairs*, Vol. 48, No. 2, pp. 323–50.
- Vaknin, D. 2020. Early Childhood Education and Care in Israel Compared to the OECD: Enrollment Rates, Employment Rates of Mothers, Quality Indices, and Future Achievement. Jerusalem, Taub Center for Social Policy Studies in Israel. (Early Childhood Research Paper 3.)
- Vargas-Barón, E. 2015. Policies on Early Childhood Care and Education: Their Evolution and Some Impacts. Paris, UNESCO. (Background paper for EFA Global Monitoring Report 2015.)
- Wallonia-Brussels Federation. 2020. Référentiel des Compétences Initiales; Tronc Commun [Reference of Initial Competences: Common Core]. Brussels, Fédération Wallonie-Bruxelles.
- Warnasuriya, R., Sosale, S. and Dey, S. 2020. Integrating Early Childhood Care and Education in Sri Lanka: From Global Evidence to National Action. Washington, DC, World Bank.
- Warner, J. 2015. Infants in orange: an international model-based approach to prison nurseries. Hastings Women's Law Journal, Vol. 26, No. 1, pp. 65–91.
- Wills, G. and Kika-Mistry, J. 2021. Early Childhood Development in South Africa during the COVID-19 Pandemic: Evidence from NIDS-CRAM Waves 2-5. Cape Town, University of Cape Town. (National Income Dynamics Study Coronavirus Rapid Mobile Survey Wave 5: 14.)
- Wolf, S., Raza, M., Kim, S., Aber, J. L., Behrman, J. and Seidman, E. 2018. Measuring and predicting process quality in Ghanaian pre-primary classrooms using the Teacher Instructional Practices and Processes System (TIPPS). Early Childhood Research Quarterly, Vol. 45, pp. 18–30.
- Wolfe, J., Kandra, J., Engdahl, L. and Shierholz, H. 2020. Domestic Workers Chartbook: A Comprehensive Look at the Demographics, Wages, Benefits, and Poverty Rates of the Professionals Who Care for Our Family Members and Clean Our Homes. Washington, DC, Economic Policy Institute.
- Wong, J. M. S. and Rao, N. 2015. The evolution of early childhood education policy in Hong Kong. International Journal of Child Care and Education Policy, Vol. 9, No. 1.
- World Bank. 2012. Vanuatu Early Childhood Deveopment. Washington, DC, World Bank. (SABER Country Report.)

- \_\_\_\_. 2019b. Why Childcare? Washington, DC, Women, Business and the Law, World Bank and Gender Secretariat, International Finance Corporation. (Women, Business and the Law Brief.)
- \_\_\_\_. 2020. The Promise of Education in Indonesia. Jakarta, World Bank.
- \_\_\_\_. 2021. Eswatini Education Sector Analysis. Washington, DC, World Bank.
- Yoshikawa, H., Wuermli, A. J., Raikes, A., Kim, S. and Kabay, S. B. 2018. Toward high-quality early childhood development programs and policies at national scale: directions for research in global contexts. *Social Policy Report*, Vol. 31, No. 1.
- Zahar, Z. and Khondker, K. 2017. From small to scale: the expansion of pre-primary in Bangladesh. Early Childhood Matters, pp. 61–65.
- Zeanah, C. H., Humphreys, K. L., Fox, N. A. and Nelson, C. A. 2017. Alternatives for abandoned children: insights from the Bucharest Early Intervention Project. *Current Opinion in Psychology*, Vol. 15, pp. 182–88.
- Zonji, S. 2018. A network for early childhood development. Annals of the New York Academy of Sciences, Vol. 1419, No. 1, pp. 20–22.

### **CHAPTER 7**

Ahmad, A. B. and Shah, M. 2016. The rise of private higher education in Kurdistan. Shah, M. and Nair, C. S. (eds), A Global Perspective on Private Higher Education. Cambridge, Mass., Chandos, pp. 219–28.

- Ahmed, J. U. 2015. Massification to marketization of higher education: private university education in Bangladesh. Higher Education for the Future, Vol. 3, No. 1, pp. 76–92.
- Almeida de Carvalho, C. H. 2013. A mercantilização da educação superior brasileira e as estratégias de mercado das instituições lucrativas [The commodification of Brazilian higher education and the market strategies of profitable institutions]. *Revista Brasileira de Educação*, Vol. 18, pp. 761–76.

Altbach, P. 2015. AHELO: the myth of measurement and comparability. International Higher Education, No. 82, pp. 2–3.

- Altbach, P., de Wit, H. and Woldegiyorgis, A. A. 2021. Public vs. Private Participation in Higher Education: Realities and Debates. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Amde, W. K., Sanders, D., Chilundo, B., Rugigana, E., Haile Mariam, D. and Lehmann, U. 2018. Exploring multiple job holding practices of academics in public health training institutions from three sub-Saharan Africa countries: drivers, impact, and regulation. *Global Health Action*, Vol. 11, No. 1.
- Amini-Philip, C. 2019. Moonlighting activities and lecturers' wellbeing in Nigerian universities. Advances in Social Sciences Research Journal, Vol. 6, No. 7, pp. 339–46.
- Arke, R. 2020. For-profit Education. Washington, DC, Center for Responsive Politics. www.opensecrets.org/industries/ background.php?cycle=2020&ind=H5300. (Accessed 17 July 2021.)
- Asian Development Bank. 2012. Private Higher Education Across Asia: Expanding Access, Searching for Quality. Manila, Asian Development Bank. (Higher Education in Dynamic Asia.)
- Assomull, A., Abdo, M. and Pelley, R. 2015. Driving Grades, Driving Growth: How Private Capital in Education Is Increasing Access, Inspiring Innovation and Improving Outcomes. Doha, World Innovation Summit for Education.
- Australia DFAT. 2016. TPP Outcomes: Education Services. Canberra, Department of Foreign Affairs and Trade.
- Bahceli, Y. 2020. Facing Pandemic Squeeze, Universities Hit Bond Markets for Cheap Cash. Reuters, 11 August.

www.reuters.com/article/us-health-coronavirus-university-bonds-idUSKCN25721A. (Accessed 17 July 2021.)

- Bangladesh University Grants Commission. 2018. Strategic Plan for Higher Education in Bangladesh: 2018–2030. Dhaka, University Grants Commission.
- Benavides, M. and Hagg Watanabe, F. 2020. Complexité et inégalités de l'offre universitaire privée au Pérou : Regard sur la diversité sociodémographique des étudiants et des conditions d'insertion professionnelle des diplômés d'universités privées [Complexity and Inequalities of the Private University Offer in Peru: A Look at the Socio-demographic Diversity of Students and the Professional Integration Conditions of Private University Graduates]. Paris, Agence française de développement. (Papier de recherche 163.)
- Bennett, B. 2002. The new style boards of governors: are they working? *Higher Education Quarterly*, Vol. 56, No. 3, pp. 287–302.
- Bernasconi, A. 2013. El motivo del lucro en la educación superior [The profit motive in higher education]. International Higher Education, No. 71, pp. 9–11.
- BRAC. 2021. Bangladesh Country Report. Paris, UNESCO. (Background paper for Global Education Monitoring Report on Non-state Actors in South Asia 2022.)

Brazil Ministry of Education and Culture. 2021. Perguntas frequentes [Frequently asked questions]. Brasília, Ministry of Education and Culture. http://portal.mec.gov.br/cotas/perguntas-frequentes.html. (Accessed 17 July 2021.)

Buckner, E. 2011. The role of higher education in the Arab state and society: historical legacies and recent reform patterns. Journal of Comparative & International Higher Education, Vol. 3, pp. 21–26.

\_\_\_\_. 2018. The growth of private higher education in North Africa: a comparative analysis of Morocco and Tunisia. Studies in Higher Education, Vol. 43, No. 7, pp. 1295–306.

\_\_\_\_. 2021. The Non-State Tertiary Sector and Inequalities in Tertiary Attendance and Completion. Paris, UNESCO. (Global Education Monitoring Report Fellowship Paper.)

Campbell, A. 2021. Examining Non-state Actors' Contributions to International Higher Education Scholarships. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)

Carpenter, J. 2017. Christian universities grow in Africa. International Higher Education, No. 88, pp. 25–26.

Casanova, A. M., Kandri, S.-E., Khan, M. A. and Valenzuela, C. 2015. UNIMINUTO: Bringing Entrepreneurship, Innovation and Jobs to Marginalized Communities in Colombia through Tertiary Education. Washington, DC, International Finance Corporation.

Caulier, S. 2020. Recherche publique et entreprises privées : les liens se resserrent [Public research and private companies: the links are growing closer]. *Le Monde*, 26 February. www.lemonde.fr/emploi/article/2020/02/26/ recherche-publique-et-entreprises-privees-les-liens-se-resserrent\_6030860\_1698637.html. (Accessed 12 August 2021.)

CAUT. 2018. CAUT Report on Board of Governors Structures at Thirty-One Canadian Universities. Ottawa, Canadian Association of University Teachers.

CHANCEN. 2020. CHANCEN schreiben [CHANCEN letter]. Berlin, CHANCEN.

\_\_\_\_. 2021. Income Share Agreements. Berlin, CHANCEN. https://chancen-eg.de/en/income-share-agreements. (Accessed 17 July 2021.)

Chapman, B. 2016. Income contingent loans in higher education financing. IZA World of Labor, No. 227.

Chattopadhyay, S. 2019. State–market dynamics in higher education financing. Varghese, N. V. and Panigrahi, J. (eds), India Higher Education Report 2018: Financing of Higher Education. New Delhi, SAGE/National Institute of Educational Planning and Administration, pp. 25–46.

Chau, Q., Dang, B. L. and Nguyen, X. A. 2020. Patterns of ownership and management in Vietnam's private higher education: an exploratory study. *Higher Education Policy*.

Chiaverini, T. 2018. Planos para a educação devem enfraquecer professores e beneficiar Guedes [Education plans will weaken teachers and benefit Guedes]. *Exame*, 16 December. https://exame.com/brasil/plano-para-aeducacao-deve-enfraquecer-professor-e-beneficiar-guedes. (Accessed 17 July 2021.)

Clark, P. 2016. Reforming Higher Education in Albania. Cheltenham, UK, Higher Education Statistics Agency. www.hesa.ac.uk/blog/11-10-2016/reforming-higher-education-albania. (Accessed 17 July 2021.)

Club of Mozambique. 2018. Mozambique: Government Pledges Crackdown on 'Turbo Teachers'. Maputo, Club of Mozambique. https://clubofmozambique.com/news/mozambique-government-pledges-crackdown-onturbo-teachers-watch. (Accessed 16 August 2021.)

Corbucci, P. R., Kubota, L. C. and Barbosa Meira, A. P. 2016. Reconfiguração Estrutural da Educação Superior Privada no Brasil: Nova Fase da Mercantilização do Ensino [Structural Reconfiguration of Private Higher Education in Brazil: A New Phase in the Commodification of Education]. Brasília, Instituto de Pesquisa Econômica Aplicada. (Texto para Discussão 2256.)

Cunha, J. 2018. Conglomerados do ensino superior avançam sobre a educação básica [Higher education conglomerates advance on basic education]. *Folha de S.Paulo*, 17 June. www1.folha.uol.com.br/mercado/2018/06/ conglomerados-do-ensino-superior-avancam-sobre-a-educacao-basica.shtml. (Accessed 12 August 2021.)

Davies, H. 2018. Comprehensive Economic and Trade Agreement with Canada: CETA – How Far Has It Come and Where Is It Headed? Geneva, Switzerland/Brussels, European University Association.

de Mello, J. M. P. and Duarte, I. F. 2020. The effect of the availabilty of student credit on tuitions: testing the Bennett hypothesis using evidence from a large-scale student loan program in Brazil. *Economía*, Vol. 20, No. 2, pp. 179–222.

de Wet, P. 2019. You can now go to jail for faking a degree on your CV – or claiming a qualification you don't have on LinkedIn or Twitter. Business Insider South Africa, 21 August. www.businessinsider.co.za/national-qualificationsframework-amendment-act-offences-for-fraud-2019-8. (Accessed 17 July 2021.)

- Debter, L. 2017. The world's biggest for-profit college company, Laureate Education, raises \$490 million in public debut. *Forbes*, 1 February. www.forbes.com/sites/laurengensler/2017/02/01/laureate-education-initial-public-offering. (Accessed 17 July 2021.)
- Dia, H. and Goudiaby, J. A. 2020. Le paradoxe de l'enseignement supérieur privé au Sénégal : réduire les inégalités tout en les maintenant [The Paradox of Private Higher Education in Senegal: Reducing Inequalities while Maintaining Them]. Paris, Agence française de développement. (Papier de recherche 154.)
- Dilas, D. B., Mackie, C., Huang, Y. and Trines, S. 2019. *Education in Indonesia*. New York, World Education News and Reviews. https://wenr.wes.org/2019/03/education-in-indonesia-2. (Accessed 16 August 2021.)
- Doan, D., Kang, J. and Zhu, Y. 2020. Financing Higher Education in Vietnam: Student Loan Reform. London, Centre for Global Higher Education. (Working Paper 51.)
- Dobbins, M. and Knill, C. 2009. Higher education policies in Central and Eastern Europe: convergence toward a common model? *Governance*, Vol. 22, No. 3, pp. 397–430.
- Douglas-Gabriel, D. 2018. Former executives of defunct for-profit college firm ITT settle fraud charges with SEC. *Washington Post*, 9 July. www.washingtonpost.com/news/grade-point/wp/2018/07/09/itts-top-executivessettle-fraud-charges-with-sec. (Accessed 17 August 2021.)
- Durham, E. R. and Sampaio, H. 2000. O setor privado de ensino superior na América Latina [The higher education private sector in Latin America]. *Cadernos de Pesquisa*, No. 110, pp. 7–37.
- Eaton, C., Howell, S. and Yannelis, C. 2018. When investor incentives and consumer interests diverge: private equity in higher education. *Review of Financial Studies*, Vol. 33, No. 9, pp. 4024–60.
- Educación 2020. 2018. Estos son 7 aspectos que debes saber sobre el lucro en la educación superior [These are 7 things you should know about profit in higher education]. Santiago, Educación 2020.
- El-Galil, T. 2021. Libya Closes 20 Private Universities and Colleges. London, Al-Fanar Media. www.al-fanarmedia. org/2021/08/libya-closes-20-private-universities-and-colleges. (Accessed 30 August 2021.)
- Elder, S. and Kring, S. 2016. Young and Female: A Double Strike? Gender Analysis of School-to-work Transition Surveys in 32 Developing Countries. Geneva, Switzerland, International Labour Organization. (Work4Youth Publication 32.)
- Erebara, G. 2014. Albania closes suspected 'diploma factories'. *Balkan Insight*, 7 August. https://balkaninsight. com/2014/08/07/albania-axes-suspected-diploma-mills. (Accessed 17 July 2021.)
- ESSA. 2019. Transforming Education Together: Impact Report 2016–2019. London, Education Sub Saharan Africa.
- Ferreyra, M. M., Avitabile, C., Álvarez, J. B., Haimovich Paz, F. and Urzúa, S. 2017. At a Crossroads: Higher Education in Latin America and the Caribbean. Washington, DC, World Bank. (Directions in Development: Human Development.)
- FIUC. 2021. Strategic Plan 2019–2021: A Global Voice for Common Future. Paris, Fédération Internationale des Universités Catholiques.
- Fried, J., Glass, A. and Baumgartl, B. 2007. Shades of privateness: non-public higher education in Europe. Wells, P. J., Sadlak, J. and Vläsceanu, L. (eds), *The Rising Role and Relevance of Private Higher Education in Europe*. Bucharest, UNESCO European Centre for Higher Education, pp. 585–664.
- Gérard, E. 2020. L'expansion de l'enseignement supérieur privé et le creusement des inégalités sociales: Analyses à partir de l'Argentine, de l'Inde, du Mexique, du Pérou, de la République Démocratique du Congo, du Sénégal, du Vietnam [The Expansion of Private Higher Education and Growing Social Inequalities: Analyses from Argentina, India, Mexico, Peru, the Democratic Republic of the Congo, Senegal, Vietnam]. Paris, Agence française de développement. (Papier de recherche 156.)
- Gérard, E., Grediaga, R. and Lopez, M. 2020. Universités privées au Mexique : entre reproduction, production et réduction des inégalités [Private universities in Mexico: between reproduction, production and reduction of inequalities]. Paris, Agence française de développement. (Papier de recherche 155.)
- Gilbert, T. and Hrdlicka, C. 2017. A hedge fund that has a university. *Wall Street Journal*, 13 November. www.wsj.com/articles/a-hedge-fund-that-has-a-university-1510615228. (Accessed 16 August 2021.)
- Goldin, C. and Cellini, S. R. 2014. Does federal student aid raise tuition? New evidence on for-profit colleges. American Economic Journal: Economic Policy, Vol. 6, No. 4, pp. 174–206.
- Green, E. L. 2018. DeVos ends Obama-era safeguards aimed at abuses by for-profit colleges. *The New York Times*, 10 August. www.nytimes.com/2018/08/10/us/politics/betsy-devos-for-profit-colleges.html. (Accessed 16 August 2021.)
- Guasco Peixoto, M. 2018. O parentesco desastroso para a educação [The disastrous kinship for education]. *Carta Capital*, 12 November. www.cartacapital.com.br/opiniao/o-parentesco-desastroso-para-a-educacao. (Accessed 12 August 2021.)

- Halperin, D. 2016. Friends in High Places: Who Endorses America's Troubled For-Profit Colleges? Washington, DC, Republic Report. www.republicreport.org/2016/friends-in-high-places-who-endorses-americas-troubled-for-profitcolleges. (Accessed 16 August 2021.)
- Henaff, N., Thai, H. T. T. and Bich, L. 2020. Enseignement supérieur au Vietnam : privatisaiton, démocratisation et inégalités [Higher education in Vietnam: privatization, democratization and inequalities]. Paris, Agence française de développement. (Papier de recherche 158.)
- Henry, O., Panirahi, J. and Sabharwal, N. S. 2020. Enseignement supérieur et inégalités sociales en Inde [Higher education and social inequalities in India]. Paris, Agence française de développement. (Papier de recherche 194.)
- Howard, C. 2020. Higher-education Bonds in a COVID-19 World. Westlake, Tex., Charles Schwab. www.schwab.com/ resource-center/insights/content/higher-education-bonds-covid-19-world. (Accessed 9 February 2021.)
- Hurtado, M. E. 2020. Laureate International pulls out of its five universities. *University World News*, 25 September. www.universityworldnews.com/post.php?story=20200925114344469. (Accessed 16 August 2021.)
- ICETEX. 2020. Fondos en administración: comunidades indígenas [Administered funds: indigenous communities]. Bogotá, ICETEX. https://portal.icetex.gov.co/Portal/Home/HomeEstudiante/fondos-en-administracion-Listado/ fondo-comunidades-indigenas. (Accessed 17 July 2021.)
- IFC. 2018. Private Education in Emerging Markets: An Investor's Perspective. Washington, DC, International Finance Corporation. (IFC Education Newsletter.)
- India University Grants Commission. 2021. UGC schemes for differently-abled student. New Delhi, University Grants Commission. www.ugc.ac.in/subpage/scheme\_differently\_abled.aspx. (Accessed 9 February 2021.)
- INEP. 2020. Microdados do Censo da Educação Superior [Microdata of the Higher Education Census]. Brasília, Instituto Nacional de Estudos e Pesquisas Educacionais Anísio Teixeira. www.gov.br/inep/pt-br/acesso-a-informacao/ dados-abertos/microdados/censo-da-educacao-superior. (Accessed 1 November 2020.)
- Ireland Higher Education Authority. 2021. Fund for Students with Disabilities. Dublin, Higher Education Authority. https://hea.ie/funding-governance-performance/funding/student-finance/fund-for-students-withdisabilities. (Accessed 10 February 2021.)
- Irene, B. and Hussain, T. 2020. The accessibility and affordability of education in Sub-Saharan Africa: the debate on low-cost and private higher education institutions. Adeyemo, K. S. (ed.), *The Education Systems of Africa*. Cham, Switzerland, Springer, pp. 859–85.
- Jablecka, J. 2007. Poland. Wells, P. J., Sadlak, J. and Vläsceanu, L. (eds), The Rising Role and Relevance of Private Higher Education in Europe. Bucharest, UNESCO European Centre for Higher Education, pp. 257–303.
- Jakiel, L. B. 2016. The gainful employment rule and for-profit higher education in the United States. Shah, M. and Nair, C. S. (eds), A Global Perspective on Private Higher Education. Cambridge, Mass., Chandos, pp. 299–312.
- Jamjoom, Y. 2016. Private higher education and graduate employability in Saudi Arabia. Shah, M. and Nair, C. S. (eds), A Global Perspective on Private Higher Education. Cambridge, Mass., Chandos, pp. 189–204.
- Jie, G. 2018. Xiamen University Malaysia: A Chinese branch campus. International Higher Education, No. 95, pp. 9–11.
- Johnstone, D. B. 2004. The economics and politics of cost sharing in higher education: comparative perspectives. Economics of Education Review, Vol. 23, No. 4, pp. 403–10.
- Júnior, A. 2015. Professores universitários moçambicanos contra regime de exclusividade [Mozambican university professors against exclusivity]. Washington, DC, VOA Português. www.voaportugues.com/a/professores-universitariosmocambicanos-contra-regime-de-exclusividade/3055143.html. (Accessed 12 August 2021.)
- Kajsiu, B. 2015. Higher education in Albania: the never ending challenge. International Higher Education, No. 82, pp. 15–16.
- Kalla, M. 2021. A Strategic Approach to Employability: Universities that Make it Happen. Washington, DC, International Finance Corporation. https://vimeo.com/showcase/7832892/video/525100694. (Accessed 17 March 2021.)
- Kaplan, A. E. 2021. Voluntary Support of Education. Washington, DC, Council for Advancement and Support of Education.
- Katsomitros, A. 2018. The emerging university bonds market. *World Finance*, 20 April. www.worldfinance.com/markets/ the-emerging-university-bonds-market. (Accessed 16 August 2021.)
- Kean, T. and Soe, H. K. 2018. New rules for private schools. *Frontier Myanmar*, 16 July. www.frontiermyanmar.net/en/ new-rules-for-private-schools. (Accessed 31 August 2021.)
- Kelchen, R. 2020. How Much Do Private Colleges Rely on Auxiliary Revenue Sources? Knoxville, Tenn., Kelchen on Education. https://robertkelchen.com/2020/06/01/how-much-do-private-colleges-rely-on-auxiliary-revenue-sources. (Accessed 16 August 2021.)
- Khan, A. A. 2017. University leaders demand action on fake universities. University World News, 27 January. www.universityworldnews.com/post.php?story=20170126173524178. (Accessed 16 August 2021.)
Knight, P. 2006. So, are universities public or private? *The Guardian*, 20 June. www.theguardian.com/education/2006/ jun/20/highereducation.comment. (Accessed 12 August 2021.)

Knobel, M. and Verhine, R. 2017. Brazil's for-profit higher education dilemma. International Higher Education, No. 89, pp. 23–24.

Korka, M. and Nicolescu, L. 2007. Romania. Wells, P. J., Sadlak, J. and Vlăsceanu, L. (eds), The Rising Role and Relevance of Private Higher Education in Europe. Bucharest, UNESCO European Centre for Higher Education, pp. 345–98.

Kreighbaum, A. 2019. DeVos issues final repeal of gainful employment. *Inside Higher Ed*, 2 July. www.insidehighered. com/quicktakes/2019/07/02/devos-issues-final-repeal-gainful-employment. (Accessed 12 August 2021.)

Kuzmin, A. 2018. 'One Language Fits All' for Higher Education in Latvia: A Reason for UN Concern. Prague, International Centre for Ethnic and Linguistic Diversity Studies. www.icelds.org/2018/11/26/one-language-fits-all-forhigher-education-in-latvia-a-reason-for-un-concern. (Accessed 16 August 2021.)

Kwiek, M. 2016. From growth to decline? Demand-absorbing private higher education when demand is over. Shah, M. and Nair, C. S. (eds), A Global Perspective on Private Higher Education. Cambridge, Mass., Chandos, pp. 53–79.

Kwiek, M. 2018. Private higher education in developed countries. Teixeira, P. N. and Shin, J. C. (eds), Encyclopedia of International Higher Education Systems and Institutions. Dordrecht, Netherlands, Springer.

Laureate. 2021. First Quarter 2021 Earnings Presentation. Baltimore, Md., Laureate.

Leray de Lima, K. A. 2018. A Atuação de Grandes Grupos Educacionais no Ensino Superior Privado: Realidades e Repercussões [The Role of Large Educational Groups in Private Higher Education: Realities and Repercussions]. M. Ed. thesis, Pontifícia Universidade Católica de São Paulo, Brazil. (Unpublished.)

Levy, D. 1986. Higher Education and the State in Latin America: Private Challenges to Public Dominance. Chicago, Ill., University of Chicago Press.

\_\_\_\_. 2006. The unanticipated explosion: private higher education's global surge. Comparative Education Review, Vol. 50, No. 2, pp. 217–40.

\_\_\_\_. 2007. A recent echo: African private higher education in an international perspective. Journal of Higher Education in Africa/Revue de l'enseignement supérieur en Afrique, Vol. 5, No. 2–3, pp. 197–220.

- \_\_\_\_. 2013. The decline of private higher education. *Higher Education Policy*, Vol. 26, No. 1, pp. 25–42.
- \_\_\_\_. 2015a. For-profit versus nonprofit private higher education. International Higher Education, No. 54, pp. 12–13.

\_\_\_\_. 2015b. Squeezing the nonprofit sector. International Higher Education, No. 71, pp. 10–12.

- Levy, D., Bernasconi, A., Buckner, E., Casta, A., Chau, Q., Gupta, A., Kinser, K., Mizikaci, F., Navarro, E., Salto, D., Silas, J. C., Tamrat, W., Teixeira, P., Wang, Y., Yonezawa, A. and Zilka, G. 2020. How COVID-19 Puts Private Higher Education at Especially High Risk – and Not: Early Observations plus Propositions for Ongoing Global Exploration. Albany, N.Y., Program for Research on Private Higher Education. (Working Paper 22.)
- Levy, D. and Tamrat, W. 2021. Viewing the Stunning Growth of Nonstate Higher Education: Ethiopia and East Africa. (Background paper for Global Education Monitoring Report 2021/2.)
- Liu, X. and Zhang, L. 2013. Flexibility at the core: what determines employment of part-time faculty in academia. Relations Industrielles/Industrial Relations, Vol. 68, No. 2, pp. 312–39.
- Lundh, A., Lexchin, J., Mintzes, B., Schroll, J. B. and Bero, L. 2018. Industry sponsorship and research outcome: systematic review with meta-analysis. *Intensive Care Medicine*, Vol. 44, No. 10, pp. 1603–12.
- Makoni, M. 2015. Academics barred from moonlighting to boost earnings. University World News, 4 December. www.universityworldnews.com/post.php?story=20151203112529221 (Accessed 9 February 2021.)
- Marinoni, G., van't Land, H. and Jensen, T. 2020. The Impact of COVID-19 on Higher Education Around the World: IAU Global Survey Report. Paris, International Association of Universities.
- Martello, A. 2017. Cade reprova compra da Estácio pela Kroton Educacional [The Administrative Council for Economic Defense rejects the purchase of Estácio by Kroton Educacional]. G1 Globo, 28 June. https://g1.globo.com/ economia/negocios/noticia/cade-reprova-compra-da-estacio-pela-kroton-educacional.ghtml. (Accessed 12 August 2021.)
- Mathot, T. 2019. And It's Signed! New Partnership with the Ghanaian Government. Vancouver, BC, Brighter Investment. www.brighterinvestment.com/blog/new-partnership-with-the-ghanaian-government. (Accessed 16 August 2021.)
- MEXT. 2014. Selection for the FY 2014: Top Global University Project. Tokyo, Ministry of Education, Culture, Sports, Science and Technology.
- Mohanty, B. K. 2020. On government table: education for profit and repatriation. *The Telegraph*, 23 February. www.telegraphindia.com/india/on-government-table-education-for-profit-and-repatriation/cid/1748008. (Accessed 16 August 2021.)

Mok, K. H. 2021. Managing neo-liberalism with Chinese characteristics: the rise of education markets and higher education governance in China. International Journal of Educational Development, Vol. 84, art. 102401.

- Moura, M. 2019. Universidades públicas realizam mais de 95% da ciência no Brasil. São Paulo, Brazil, Universidade Federal de São Paulo. www.unifesp.br/noticias-anteriores/item/3799-universidades-publicas-realizam-mais-de-95-da-ciencia-no-brasil. (Accessed August 31 2021.)
- Mulokozi, C. 2015. Teachers' moonlighting and its impact on their job performance in Dar Es Salaam region secondary schools. M. Ed. thesis, Open University of Tanzania. (Unpublished.)
- Mungiu-Pippidi, A. and Dusu, A. E. 2011. Civil society and control of corruption: assessing governance of Romanian public universities. International Journal of Educational Development, Vol. 31, No. 5, pp. 532–46.
- Muzammil, M. 2019. Growth and expansion of private higher education. Varghese, N. V. and Panigrahi, J. (eds), *India Higher Education Report 2018: Financing of Higher Education*. New Delhi, SAGE/National Institute of Educational Planning and Administration, pp. 339–62.
- NACUBO. 2019. Private Colleges Now Use Nearly Half of Tuition Revenue for Financial Aid. Washington, DC, National Association of College and University Business Officers. www.nacubo.org/Press-Releases/2019/Private-Colleges-Now-Use-Nearly-Half-of-Tuition-Revenue-For-Financial-Aid. (Accessed 16 August 2021.)
- NCES. 2019. Undergraduate Enrollment. Washington, DC, National Center for Education Statistics. https://nces.ed.gov/programs/coe/indicator\_cha.asp#info. (Accessed 16 August 2021.)
- Nganga, G. 2019. Universities face ongoing struggle as fees hike rejected. University World News, 25 May. www.universityworldnews.com/post.php?story=20190522082856329. (Accessed 12 August 2021.)
- Nicolescu, L. 2007. Institutional efforts for legislative recognition and market acceptance: Romanian private higher education. Slantcheva, S. and Levy, D. (eds), *Private Higher Education in Post-Communist Europe: In Search of Legitimacy*. New York, Palgrave Macmillan, pp. 201–22.
- O Sul. 2019. O ministério da educação estuda como flexibilizar o ensino superior privado [The Ministry of Education is seeking how to make private higher education more flexible]. O Sul, 5 October. www.osul.com.br/oministerio-da-educacao-estuda-como-flexibilizar-o-ensino-superior-privado. (Accessed 17 July 2021.)
- OECD. 2019. Philanthropy and Education: Quality Education For All Lessons and Future Priorities. Paris, Organisation for Economic Co-operation and Development. (Network of Foundations Working for Development.)
- Oliveira, C. 2015. Convênios com fundações privadas desvirtuam função social das universidades [Agreements with private foundations distort the social role of universities]. São Paulo, Brazil, Agência Universitária de Notícias, São Paulo University. www.usp.br/aun/antigo/exibir.php?id=6846. (Accessed 12 August 2021.)
- Onsongo, J. 2007. The growth of private universities in Kenya: implications for gender equity in higher education. Journal of Higher Education in Africa/Revue de l'enseignement supérieur en Afrique, Vol. 5, No. 2–3, pp. 111–33.
- Owens, T. L. 2017. Higher education in the sustainable development goals framework. European Journal of Education, Vol. 52, No. 4, pp. 414–20.
- Pachuashvili, M. 2009. The Politics of Higher Education: Governmental Policy Choices and Private Higher Education in Post-Communist Countries. Ph.D. thesis, Central European University, Budapest. (Unpublished.)
- \_\_\_\_. 2011. Governmental policies and their impact on private higher education development in post-communist countries: Hungary, Latvia, Lithuania and Georgia, 1990–2005. Journal of Comparative Policy Analysis: Research and Practice, Vol. 13, No. 4, pp. 397–410.
- Pennsylvania Attorney General. 2019. Attorney General Shapiro Annunces For-profit College Company Will Provide \$493 Million in Debt Relief for over 179,000 Students Nationwide. Harrisburg, Pa., Office of Attorney General. (Accessed 30 August 2021.)
- Pereira, C. A., Araujo, J. F. F. E. and de Lourdes Machado-Taylor, M. 2018. The Brazilian higher education evaluation model: 'SINAES' sui generis? International Journal of Educational Development, Vol. 61, pp. 5–15.
- Pompeu, C., Augusto, L. and Frazao, H. 2016. Lista de maiores financiadores expõe laços econômicos e políticos [List of top funders exposes economic and political ties]. *Estadão*, 26 September. https://politica.estadao. com.br/noticias/geral,lista-de-maiores-financiadores-expoe-lacos-economicos-e-politicos,10000078203. (Accessed 12 August 2021.)
- Provini, O. 2019. Negotiating the marketization of higher education in East Africa: a comparative analysis of Tanzania and Kenya. *Higher Education*, Vol. 77, No. 2, pp. 323–42.
- Quang, C.-D. 2017. A gloomy outlook for private higher education. University World News, 19 May. www.universityworldnews.com/post.php?story=20170515200538925. (Accessed 16 August 2021.)

- Ravi, S., Gupta, N. and Nagaraj, P. 2019. *Reviving Higher Education in India*. New Delhi, Brookings India. (Research paper 112019-01.)
- Raxhimi, A. 2019. Mass student protests bring down Albania's science minister. *Nature*, 16 January. www.nature.com/articles/d41586-019-00144-7. (Accessed 12 August 2021.)
- Romania Insider. 2017. Romanian universities investigated for fraudulently granting diplomas to foreigners. *Romania Insider*, 26 July. www.romania-insider.com/romania-fraudulent-university-graduations. (Accessed 16 August 2021.)
- \_\_\_\_. 2019. Corruption scandal hits biggest private university in Romania, pro-rector arrested for bribery. Romania Insider, 19 December. www.romania-insider.com/spiru-haret-university-romania-corruption. (Accessed 16 August 2021.)
- Sakyi, E. and Agomor, K. 2020. Moonlighting in Ghana's higher education institutions: exploring lecturers' experiences at the Ghana Institute of Management and Public Administration (GIMPA). Journal of Applied Research in Higher Education, Vol. 13, No. 1, pp. 180–94.
- Salmi, J. 2017. The Tertiary Education Imperative: Knowledge, Skills and Values for Development. Rotterdam, Netherlands, Sense. (Global Perspectives on Higher Education 38.)
- \_\_\_\_. 2020. Higher Education and Inclusion. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2020.)
- Salmi, J., Hâj, C. M. and Alexe, D. 2015. Equity from an institutional perspective in the Romanian higher education system. Curaj, A., Deca, L., Egron-Polak, E. and Salmi, J. (eds), *Higher Education Reforms in Romania: Between the Bologna Process and National Challenges.* Cham, Switzerland, Springer, pp. 63–86.
- Salmi, J. and Sursock, A. 2018. Access and Completion for Underserved Students: International Perspectives. Boston, Mass., Boston College Center for International Higher Education (CIHE Perspectives 9.)
- Salmon, J. 2020. Financing higher education through equity, not debt: the case for income share agreements. *Journal of School Choice*, Vol. 14, No. 3, pp. 371–94.
- Salto, D. J. 2018. To profit or not to profit: the private higher education sector in Brazil. *Higher Education*, Vol. 75, No. 5, pp. 809–25.
- Samuels, R. 2013. Why Public Higher Education Should Be Free: How to Decrease Cost and Increase Quality at American Universities. New Brunswick, NJ, Rutgers University Press.
- SEDLAC. 2018. Socio-Economic Database for Latin America and the Caribbean. La Plata, Argentina/Washington, DC, Center of Distributive, Labor and Social Studies/World Bank https://datacatalog.worldbank.org/dataset/ socio-economic-database-latin-america-and-caribbean. (Accessed 17 July 2021.)
- Seeber, M. 2016. Non-university higher education. Teixeira, P. N. and Shin, J. C. (eds), Encyclopedia of International Higher Education Systems and Institutions. Dordrecht, Netherlands, Springer.
- Sirois, G., Morales-Perlaza, A. and Dembélé, M. 2021. La privatisation de la formation initiale des enseignants : cartographie et études de cas du phénomène en Afrique subsaharienne, Amérique latine et Asie du Sud [The Privatization of Initial Teacher Training: Mapping and Case Studies of the Phenomenon in sub-Saharan Africa, Latin America and South Asia]. (Background paper for Global Education Monitoring Report 2021/2.)
- Slantcheva, S. and Levy, D. 2007. Introduction: private higher education in post-Communist Europe in search of legitimacy. Slantcheva, S. and Levy, D. (eds), *Private Higher Education in Post-Communist Europe: In Search of Legitimacy*. New York, Palgrave Macmillan.
- Tamrat, W. 2017. Private higher education in Africa: old realities and emerging trends. International Journal of African Higher Education, Vol. 4, No. 2, pp. 17–40.
- \_\_\_\_. 2018. Family-owned private higher education institutions in Africa. International Higher Education, No. 95, pp. 23–24.
- \_\_\_\_. 2021. COVID-19 has dealt a blow to Ethiopia's private higher education institutions. *The Conversation*, 11 February. https://theconversation.com/covid-19-has-dealt-a-blow-to-ethiopias-private-higher-educationinstitutions-153398. (Accessed 16 August 2021.)
- Tamrat, W. and Levy, D. 2017. Unusual in growth and composition: Ethiopian private higher education. International Higher Education, No. 90, pp. 19–21.
- Tamrat, W. and Teferra, D. 2019. Private higher education in Ethiopia: risks, stakes and stocks. Studies in Higher Education, Vol. 45, pp. 677–91.
- Tanhueco-Tumapon, T. 2020. Quality assurance in higher education. *The Manila Times*, 5 March. www.manilatimes. net/2020/03/05/campus-press/quality-assurance-in-higher-education-2/700186. (Accessed 16 August 2021.)
- Tarlea, S. 2017. Higher education governance in Central and Eastern Europe: a perspective on Hungary and Poland. European Educational Research Journal, Vol. 16, No. 5, pp. 670–83.
- Task Force on Higher Education and Society. 2000. Higher Education in Developing Countries: Peril and Promise. Washington, DC, World Bank.

- Taylor, A. 2016. Universities are becoming billion-dollar hedge funds with schools attached. *The Nation*, 8 March. www.thenation.com/article/archive/universities-are-becoming-billion-dollar-hedge-funds-with-schoolsattached. (Accessed 16 August 2021.)
- Teixeira, P. N., Biscaia, R., Rocha, V. and Cardoso, M. F. 2016. What role for private higher education in Europe? Reflecting about current patterns and future prospects. Shah, M. and Nair, C. S. (eds), A Global Perspective on Private Higher Education, Cambridge, Mass., Chandos, pp. 13–28.
- The Princeton Review. 2021. ABA-Accredited Law School. Framingham, Mass., The Princeton Review. www.princetonreview.com/law-school-advice/law-school-accreditation. (Accessed 16 August 2021.)
- Tho, V. 2021. Former university principal in Hanoi prosecuted for issuing over 400 fake degrees. *Tuoi Tre News*, 31 July. https://tuoitrenews.vn/news/society/20210731/former-university-principal-in-hanoi-prosecuted-for-issuing-over-400-fake-degrees/62351.html. (Accessed 31 August 2021.)
- TimesLIVE. 2018. Fake colleges on the rise and destroying SA education. *TimesLIVE*, 9 January. www.timeslive.co.za/ sebenza-live/features/2018-01-09-fake-colleges-on-the-rise-amp-destroying-sa-education. (Accessed 17 July 2021.)
- Todos Pela Educação. 2019. Formação inicial de professores [Initial teacher training]. São Paulo, Brazil, Todos Pela Educação.
- Tomusk, V. 2003. The war of institutions, episode I: the rise, and the rise of private higher education in Eastern Europe. Higher Education Policy, Vol. 16, No. 2, pp. 213–38.
- Unangst, L. 2017. Germany's innovative strategies to enroll refugees. *Inside Higher Ed*, 18 January. www.insidehighered. com/blogs/world-view/germanys-innovative-strategies-enroll-refugees. (Accessed 16 August 2021.)
- UNESCO. 2017. Six Ways to Ensure Higher Education Leaves No One Behind. Paris, UNESCO. (Global Education Monitoring Report Policy Paper 30.)
- UNIMINUTO. 2020. Informe de gestión UNIMINUTO 2020-1 [UNIMINUTO Management Report 2020/1]. Bogotá, Corporación Universitaria Minuto de Dios.
- US Department of Education. 2018. Law Enforcement Actions against Predatory Colleges. Washington, DC, Department of Education. https://sites.ed.gov/naciqi/files/2018/05/NACIQI-Enclosure-1-law-enforcement.pdf. (Accessed 30 August 2021.)
- US Department of Justice. 2015. For-profit College Kaplan to Refund Federal Financial Aid under Settlement with United States. Washington, DC, Department of Justice. www.justice.gov/usao-wdtx/pr/profit-college-kaplan-refundfederal-financial-aid-under-settlement-united-states. (Accessed 30 August 2021.)
- US Federal Trade Commission. 2016. DeVry University Agrees to \$100 Million Settlement with FTC. Washington, DC, Federal Trade Commission. www.ftc.gov/news-events/press-releases/2016/12/devry-university-agrees-100-millionsettlement-ftc. (Accessed 30 August 2021.)
- \_\_\_\_. 2019. FTC Obtains Record \$191 Million Settlement from University of Phoenix to Resolve FTC Charges It Used Deceptive Advertising to Attract Prospective Students. Washington, DC, Federal Trade Commission. www.ftc.gov/newsevents/press-releases/2019/12/ftc-obtains-record-191-million-settlement-university-phoenix. (Accessed 30 August 2021.)
- UWN. 2009. Romania: Diplomas of private university illegal. *University World News*, 19 July. www.universityworldnews. com/post.php?story=20090717083926472. (Accessed 17 July 2021.)
- Van Damme, D. 2015. Global higher education in need of more and better learning metrics: why OECD's AHELO project might help to fill the gap. European Journal of Higher Education, Vol. 5, No. 4, pp. 425–36.
- Varghese, N. V. 2016. Governance Reforms in Higher Education: A Study of Selected Countries in Africa. Paris, UNESCO International Institute for Educational Planning. (IIEP Research Paper.)
- Varghese, N. V. and Panigrahi, J. (eds). 2019. India Higher Education Report 2018: Financing of Higher Education. New Delhi, SAGE/National Institute of Educational Planning and Administration.
- Vîiu, G.-A. and Miroiu, A. 2015. The quest for quality in higher education: is there any place left for equity and access? Curaj, A., Deca, L., Egron-Polak, E. and Salmi, J. (eds), *Higher Education Reforms in Romania*: Between the Bologna Process and National Challenges. New York, Springer, pp. 173–89.
- Wangenge-Ouma, G. 2018. Financing Higher Education in Africa: An Overview. Teixeira, P. N. and Shin, J. C. (eds), Encyclopedia of International Higher Education Systems and Institutions. Dordrecht, Netherlands, Springer.
- Web of Science Group. 2019. Research in Brazil: Funding Excellence. Philadelphia, Pa., Clarivate Analytics.
- Welch, A. 2021. Private Higher Education in East and Southeast Asia: Growth, Challenges, Implications. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Wesangula, D. 2015. Kenya's shuttling lecturers: university shortages are taking a toll. *The Guardian*, 29 July. www.theguardian.com/global-development-professionals-network/2015/jul/29/kenyas-shuttling-lecturersuniversity-shortages-are-taking-toll. (Accessed 16 August 2021.)

World Bank. 2017. Higher Education in Côte d'Ivoire. Washington, DC, World Bank. (Policy Brief.)

- \_\_\_\_. 2019a. Bangladesh Tertiary Education Sector Review: Skills and Innovation for Growth. Washington, DC, World Bank.
- \_\_\_\_. 2019b. World Bank scales-up its support for regional higher education centers for excellence in Africa. Washington, DC, World Bank. www.worldbank.org/en/news/press-release/2019/04/03/world-bank-scales-up-its-support-forregional-higher-education-centers-for-excellence-in-africa. (Accessed 16 August 2021.)
- Xiaoying, M. and Abbott, M. 2016. The issue of contractible quality, quality assurance, and information asymmetries in higher education. Shah, M. and Nair, C. S. (eds), A *Global Perspective on Private Higher Education*. Cambridge, Mass., Chandos, pp. 1–11.
- Zha, Q. 2006. The resurgence and growth of private higher education in China. *Higher Education Perspectives*, pp. 54–68.
- Ziderman, A. 2017. Experience with student loans, higher education. Shin, J. C. and Teixeira, P. (eds), Encyclopedia of International Higher Education Systems and Institutions. Dordrecht, Netherlands, Springer.
- Zuniga, T., Chibai, J. and Kalla, M. 2021. A Strategic Approach to Employability: Universities that Make It Happen. Washington, DC, International Finance Corporation. https://vimeo.com/showcase/7832892. (Accessed 17 March 2021.)

### **CHAPTER 8**

- Aarkrog, V. and Wahlgren, B. 2015. Assessment of prior learning in adult vocational education and training. International Journal for Research in Vocational Education and Training, Vol. 2, No. 1, pp. 41–58.
- Adams, S. 2019. Game of tongues: how Duolingo built a \$700 million business with its addictive language-learning app. Forbes, 16 July. www.forbes.com/sites/susanadams/2019/07/16/game-of-tongues-how-duolingo-built-a-700-millionbusiness-with-its-addictive-language-learning-app (Accessed 17 August 2021.)
- AFDB. 2020. African Economic Outlook 2020: Developing Africa's Workforce for the Future. Abidjan, African Development Bank.

Afeti, G. 2018. Revitalising technical and vocational education and training in Africa: issues outstanding. Journal of Vocational, Adult and Continuing Education and Training, Vol. 1, No. 1, pp. xi–xviii.

- Aitchison, J. 2017. CONFINTEA VI Mid-Term Review 2017: The Status of Adult Learning and Education in Sub-Saharan Africa. Hamburg, Germany, UNESCO Institute for Lifelong Learning.
- Andersson, P. and Muhrman, K. 2019. Marketization of Vocational Adult Education in Sweden. Paper for Crossing Boundaries in VET, Valencia, Spain, 2–3 May.
- Arias, O., Evans, D. K. and Santos, I. 2019. The Skills Balancing Act in Sub-Saharan Africa: Investing in Skills for Productivity, Inclusivity, and Adaptability. Washington, DC, World Bank. (Africa Development Forum.)
- Arribas, M. G. and Papadakis, N. 2019. Governance Arrangements for Vocational Education and Training in ETF Partner Countries: Analytical Overview 2012–17. Turin, Italy, European Training Foundation.
- Australia Department of Employment, Skills, Small and Family Business. 2017. Review of the VET Student Loans Course List and Loan Caps Methodology. Canberra, Department of Employment, Skills, Small and Family Business.
- Australian National Audit Office. 2016. Administration of the VET FEE-HELP Scheme. Canberra, Australian National Audit Office. www.anao.gov.au/work/performance-audit/administration-vet-fee-help-scheme. (Accessed 17 August 2021.)
- Bangladesh Technical Education Board. 2012. National Skills Quality Assurance System: Manual 3 Registration of Training Organizations and Accreditation of Learning and Assessment Programs. Dhaka, Bangladesh Technical Education Board/European Union/International Labour Organization.
- Bateman, A. and Coles, M. 2013. Qualifications Frameworks and Quality Assurance of Education and Training. Washington, DC, World Bank.
- Bateman, A. and Liang, X. 2016. National Qualification Framework and Competency Standards: Skills Promotion and Job Creation in East Asia and Pacific. Washington, DC, World Bank. (Korea-World Bank Partnership Facility.)
- Beare, K. 2019. How Many People Learn English? New York, ThoughtCo. www.thoughtco.com/how-many-people-learnenglish-globally-1210367. (Accessed 22 July 2021.)
- Béteille, T., Tognatta, N., Riboud, M., Nomura, S. and Ghorpade, Y. 2020. *Ready to Learn: Before School, in School, and Beyond School in South Asia.* Washington, DC, World Bank. (South Asia Development Forum.)
- Bhutani Vij, A. 2020. The neoliberal agenda of workplace learning in NGOs: contestation, narratives and resistance. Studies in Continuing Education, Vol. 43, No. 2, pp. 208–22.
- British Council. 2019. Annual Report and Accounts 2018-19. London, British Council.
- Brunello, G. and Wruuck, P. 2020. Employer Provided Training in Europe: Determinants and Obstacles. Bonn, Germany, Institute of Labor Economics. (IZA Discussion Paper 12981.)

Busemeyer, M. R. and Trampusch, C. 2012. The Political Economy of Collective Skill Formation. Oxford, UK, Oxford University Press.

- Cedefop. 2020. European Skills Index: Country Pillars Czechia. Thessaloniki, Greece, European Centre for the Development of Vocational Training. www.cedefop.europa.eu/en/publications-and-resources/datavisualisations/european-skills-index/country/czechia. (Accessed 22 July 2021.)
- Cedefop, European Commission and ICF. 2019. European Inventory on Validation of Non-Formal and Informal Learning 2018: Final Synthesis Report. Luxembourg, European Centre for the Development of Vocational Training/European Commission/ICF.
- Cell-Ed. 2020. Essentials Skills on the Go. Palo Alto, Calif., Cell-Ed. www.cell-ed.com. (Accessed 22 July 2021.)
- Chankseliani, M. and Anuar, A. M. 2019. Cross-country comparison of engagement in apprenticeships: a conceptual analysis of incentives for individuals and firms. International Journal for Research in Vocational Education and Training, Vol. 6, No. 3, pp. 261–83.
- Chankseliani, M., Keep, E. and Wilde, S. 2017. People and Policy: A Comparative Study of Apprenticeship across Eight National Contexts. Doha, World Innovation Summit for Education. (RR.9.2017.)
- Chinen, M., de Hoop, T., Alcázar, L., Balarin, M. and Sennet, J. 2017. Vocational and business training to improve women's labour market outcomes in low- and middle-income countries: a systematic review. *Campbell Systematic Reviews*, Vol. 16.
- Chłoń-Domińczak, A., Holzer-Żelażewska, D. and Maliszewska, A. 2019. Vocational Education and Training in Europe: Poland. Thessaloniki, Greece, European Centre for the Development of Vocational Training. (ReferNet VET in Europe.)

Cronquist, K. and Fiszbein, A. 2017. English Language Learning in Latin America. Washigton, DC, Inter-American Dialogue.

- CVTS. 2005, 2015. Enterprises providing training by type of training and size class: % of all enterprises. Luxembourg, Eurostat. https://appsso.eurostat.ec.europa.eu/nui/show.do?dataset=trng\_cvt\_01s&lang=en. (Accessed 22 July 2021.)
- . 2015. Cost of CVT courses by type and size class: % of total labour cost of all enterprises. Luxembourg, Eurostat. https://appsso.eurostat.ec.europa.eu/nui/show.do?dataset=trng\_cvt\_01s&lang=en. (Accessed 22 July 2021.)
- Dunbar, M. 2013. Engaging the Private Sector in Skills Development. Oxford, UK, Health and Education Advice and Resource Team.
- EAEA. 2014. RENEWAL: State of the Art Report on the Implementation of the European Agenda for Adult Learning. Brussels, European Association for the Education of Adults.
- EIBIS. 2018. Average Share of Investment in Different Asset Types: 2018. Luxembourg, European Investment Bank. https://data.eib.org/eibis/graph;jsessionid=8C088B76DA8A8204ACED9D4F01D18267. (Accessed 17 August 2021.)
- El Salvador Ministry of Education. 2015. Ejes Estratégicos del Plan Nacional de Educación en Función de la Nación: Gestión 2014–2019 [Strategic Axes of the National Education Plan for the Nation: 2014–2019]. San Salvador, Ministry of Education.
- Elder, S. and Kring, S. 2016. Young and Female: A Double Strike? Gender Analysis of School-to-Work Transition Surveys in 32 Developing Countries. Geneva, Switzerland, International Labour Organization. (Work4Youth 32.)

ETF. 2013. Good Multilevel Governance for Vocational Education and Training. Turin, Italy, European Training Foundation.

- \_\_\_\_. 2020a. Public–Private Partnerships for Skills Development: A Governance Perspective Volume I. Turin, Italy, European Training Foundation.
- \_\_\_\_. 2020b. Public–Private Partnerships for Skills Development: A Governance Perspective Volume II, Case-studies. Turin, Italy, European Training Foundation.
- Eurostat. 2021. Distribution of Non-formal Education and Training Activities by Provider. Luxembourg, Eurostat. https://ec.europa.eu/eurostat/databrowser/view/trng\_aes\_170/default/table?lang=en. (Accessed 17 August 2021.)
- Fejes, A. and Holmqvist, D. 2019. Procurement as a market in adult education. Dahlstedt, M. and Fejes, A. (eds), Neoliberalism and Market Forces in Education: Lessons from Sweden. London, Routledge, pp. 156–69. (Routledge Research in Education Policy and Politics.)
- Fejes, A., Runesdotter, C. and Wärvik, G.-B. 2016. Marketisation of adult education: principals as business leaders, standardised teachers and responsibilised students. *International Journal of Lifelong Education*, Vol. 35, No. 6, pp. 664–81.
- Fernandez-Stark, K. and Bamber, P. 2018. Skills development for economic and social upgrading: the case of Asian developing countries in the global value chains. Sakamoto, A. and Sung, J. (eds), Skills and the Future of Work: Strategies for Inclusive Growth in Asia and the Pacific. Bangkok, ILO Regional Office for Asia and the Pacific, pp. 62–99.

- Fitzpayne, A. and Pollack, E. 2018. Lifelong Learning and Training Accounts: Helping Workers Adapt and Succeed in a Changing Economy. Washington, DC, Aspen Institute. (Issue Brief.)
- Flake, R., Kumar, P., Brings, C., Simelane, T. and Ta, M. 2017. The Role of the Private Sector in Vocational and Educational Training: Developments and Success Factors in Selected Countries. Bonn, Germany, Deutsche Gesellschaft für Internationale Zusammenarbeit. (Economic Policy Forum Working Paper.)
- Fondation des Alliances Françaises. 2020. Rapport d'Activité 2019 [2019 Activity Report]. Paris, Fondation des Alliances Françaises.
- Foubert, T. and Folisi, F. 2019. Policy Dialogue in Vocational Education: What Role for Civil Society Organisations? A Pilot Survey in the Southern and Eastern Mediterranean. Turin, Italy, European Training Foundation.
- Germany Federal Office for Migration and Refugees. 2015. Konzept für einen bundesweiten Jugendintegrationskurs [Concept for a Nationwide Youth Integration Course]. Nürnberg, Germany, Bundesamt für Migration und Flüchtlinge.
- Glick, P. J., Huang, C. and Mejia, N. 2015. The Private Sector and Youth Skills and Employment Programs in Low and Middle-Income Countries. Washington, DC, World Bank.
- Hanemann, U. 2015. The Evolution and Impact of Literacy Campaigns and Programmes 2000–2014. Hamburg, Germany, UNESCO Institute for Lifelong Learning.
- \_\_\_\_. 2021. Non-State Actors in Non-Formal Youth and Adult Education. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Haolader, F. A., Foysol, K. M. and Clement, C. K. 2017. Technical and vocational education and training (TVET) in Bangladesh: systems, curricula, and transition pathways. Pilz, M. (ed.), Vocational Education and Training in Times of Economic Crisis. Cham, Switzerland, Springer, pp. 201–27.
- Hawley-Woodall, J. 2019. European Inventory on Validation of Non-Formal and Informal Learning 2018: Thematic Report Bridging the Gap: Validation Creating Routes and Links between Sectors. Luxembourg, European Centre for the Development of Vocational Training/European Training Foundation/European Commission/ICF.
- Hawley-Woodall, J., Duell, N., Scott, D., Finlay-Walker, L., Arora, L. and Carta, E. 2015. Skills Governance in the EU Member States: Synthesis Report for the EEPO. Brussels, European Commission.
- ILO. 2019a. Intermediary Organizations in Apprenticeship Systems. Geneva, Switzerland, International Labour Organization.
- \_\_\_\_. 2019b. Quality Apprenticeships: Addressing Skills Mismatch and Youth Unemployment. Geneva, Switzerland, International Labour Organization.
- \_\_\_\_. 2019c. Sector Skills Councils. Jakarta, International Labour Organization.
- \_\_\_\_. 2020a. Employer Organizations in the Governance of TVET and Skills Systems. Geneva, Switzerland, International Labour Organization.
- \_\_\_\_. 2020b. Global Employment Trends for Youth 2020: Technology and the Future of Jobs. Geneva, Switzerland, International Labour Organization.
- \_\_\_\_. 2020c. What Is Skills Mismatch and Why Should We Care? Geneva, Switzerland, International Labour Organization. www.ilo.org/skills/Whatsnew/WCMS\_740388/lang--en/index.htm. (Accessed 16 April 2021.)
- Imdorf, C. 2017. Understanding discrimination in hiring apprentices: how training companies use ethnicity to avoid organisational trouble. *Journal of Vocational Education and Training*, Vol. 69, No. 3, pp. 405–23.
- Jeon, S. 2019. Unlocking the Potential of Migrants: Cross-country Analysis. Paris, Organisation for Economic Co-operation and Development. (OECD Reviews of Vocational Education and Training.)
- Kaňáková, M., Czesaná, V. and Šímová, Z. 2019. Vocational Education and Training in Europe: Czech Republic. Thessaloniki, Greece, European Centre for the Development of Vocational Training. (Cedefop ReferNet VET in Europe.)
- Katz, B. and Elliott, D. 2020. CareerWise: Case Study of a Youth Apprenticeship Intermediary. Washington, DC, Urban Institute.
- Konings, J. and Vanormelingen, S. 2015. The impact of training on productivity and wages: firm-level evidence. Review of Economics and Statistics, Vol. 97, No. 2, pp. 485–97.
- Krishnan, A. and Gelb, S. 2018. Skills Development Funds: Lessons from Asia. London, Overseas Development Institute. (Supporting Economic Transformation Briefing Paper.)
- Kuczera, M. and Field, S. 2018. Apprenticeship in England, United Kingdom. Paris, Organisation for Economic Co-operation and Development. (OECD Reviews of Vocational Education and Training.)
- Loewen, S., Crowther, D., Isbell, D. R., Minhye Kim, K., Maloney, J., Miller, Z. F. and Rawal, H. 2019. Mobile-assisted language learning: a Duolingo case study. *ReCALL*, Vol. 31, No. 3, pp. 293–311.

- Lukyanova, L. and Veramejchyk, G. 2017. Transformation of adult education institutions in Eastern Europe: experiences of Belarus, Ukraine and Moldova. Avramovska, M., Hirsch, E. and Schmidt-Behlau, B. (eds), Adult Education Centres as a Key to Development: Challenges and Success Factors. Bonn, Germany, DVV International, pp. 66–86. (International Perspectives in Adult Education.)
- Mehrotra, S. 2017. Technical and vocational education in Asia: what can South Asia learn from East/South East Asia? Indian Journal of Labour Economics, Vol. 59, pp. 529–52.
- \_\_\_\_. 2018. From the informal to the formal economy: skills initiatives in India. Sakamoto, A. and Sung, J. (eds), Skills and the Future of Work: Strategies for Inclusive Growth in Asia and the Pacific. Bangkok, ILO Regional Office for Asia and the Pacific.
- Musset, P., Field, S., Mann, A. and Bergseng, B. 2019. Vocational Education and Training in Estonia. Paris, Organisation for Economic Co-operation and Development. (OECD Reviews of Vocational Education and Training.)
- Nambiar, D. 2021. The Functioning and Regulation of Public–Private Partnerships in Skills Development: The Case of India's National Skills Development Corporation. Paris, UNESCO. (Background paper for Global Education Monitoring Report on Non-state Actors in South Asia Education 2022.)
- NCVER. 2019. Employers' Use and Views of the VET System. Adelaide, Australia, National Centre for Vocational Education Research. www.ncver.edu.au/research-and-statistics/collections/employers-use-and-views-of-the-vetsystem. (Accessed 17 August 2021.)
- Nedelkoska, L. and Quintini, G. 2018. Automation, Skills Use and Training. Paris, Organisation for Economic Co-operation and Development. (OECD Social, Employment and Migration Working Paper 202.)
- Needham, S. 2019. TVET policy in South Africa: caught between neo-liberalism and privatisation? Journal of Vocational, Adult and Continuing Education and Training, Vol. 2, No. 2, pp. 62–101.
- Nesterova, Y. and Capsada-Munsech, Q. 2021. National and Subnational Approaches to Regulating Private Technical and Vocational Education and Training: Comparative Insights from Asia and Africa. Paris, UNESCO. (Background Paper for Global Education Monitoring Report 2021/2.)
- Nguyen, T. Q., Nguyen, A. T., Tran, A. L., Le, H. T., Le, H. H. T. and Vu, L. P. 2020. Do workers benefit from on-the-job training? New evidence from matched employer-employee data. *Finance Research Letters*, Vol. 40.
- NILE and UIL. 2016. Synthesis Report on the State of Community: Learning Centres in Six Asian Countries. Hamburg, Germany, National Institute for Lifelong Education/UNESCO Institute of Lifelong Learning.
- OECD. 2017. Better Use of Skills in the Workplace: Why It Matters for Productivity and Local Jobs. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2018. Share of students enrolled by institution. Paris, Organisation for Economic Co-operation and Development. https://stats.oecd.org/Index.aspx?DataSetCode=EAG\_ENRL\_SHARE\_INST. (Accessed 17 August 2021.)
- \_\_\_\_. 2019a. Individual Learning Accounts: Panacea or Pandora's Box? Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2019b. OECD Skills Strategy Flanders: Assessment and Recommendations. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2019c. OECD Skills Strategy Poland: Assessment and Recommendations. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2020a. Education at a Glance 2020: OECD Indicators. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2020b. Effective Adult Learning Policies: Challenges and Solutions for Latin American Countries. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2021. Adult education and learning: barriers to participation. Paris, Organisation for Economic Co-operation and Development. https://stats.oecd.org/index.aspx?queryid=79325. (Accessed 17 August 2021.)
- Palmer, R. 2017. Financing TVET in the East Asia and Pacific Region: Current Status, Challenges and Opportunities. Washington, DC, World Bank.
- \_\_\_\_. 2020. A Review of Skills Levy Systems in Countries of the Southern African Development Community. Geneva, Switzerland, International Labour Organization.
- Pilz, M., Uma, G. and Venkatram, R. 2015. Skills development in the informal sector in India: the case of street food vendors. International Review of Education, Vol. 61, pp. 191–209.
- Polish Agency for Enterprise Development. 2021. Sektorowe Rady ds. Kompetencji: Buduj Kompetencje w Swojej Branży! [Sectoral Skills Councils: Build competences in your Industry!]. Warsaw, Polish Agency for Enterprise Development. www.parp.gov.pl/component/site/site/sektorowe-rady-ds-kompetencji - rady. (Accessed 22 July 2021.)

Robinson-Pant, A., Binesse, H., Maleki, M. N., Millora, C. and Wang, Q. 2021. Adult Literacy and Learning for Social Change: Innovation, Influence and the Role of Non-State Actors. Case Studies from Afghanistan, China, the Philippines and Senegal. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)

Robinson, C. 2019. Literacy in my language? Principles, practices, prospects. Hogan-Brun, G. and O'Rourke, B. (eds), Palgrave Handbook of Minority Languages and Communities. London, Palgrave Macmillan, pp. 405–29.

- Robinson, C. and Vũ, T. A. T. 2019. Literacy, languages and development in Africa: a policy perspective. *International Review of Education*, Vol. 65, No. 3, pp. 443–70.
- Rogers, A. 2019. Second-generation non-formal education and the sustainable development goals: operationalising the SDGs through community learning centres. *International Journal of Lifelong Education*, Vol. 38, No. 5, pp. 515–26.
- Rossel, C. and CEAAL. 2017. CONFINTEA VI Mid-Term Review 2017: The Status of Adult Learning and Education in Latin America and the Caribbean. Hamburg, Germany, UNESCO Institute for Lifelong Learning.
- Sanchez Puerta, M. L., Valerio, A., Hoftijzer, M., Rizvi, A. and Avato, J. 2016. Employer Survey Snapshot 2016: Highlights from Six Low and Middle-Income Countries. Washington, DC, World Bank. (STEP Skills Measurement.)
- Santori, D., Ball, S. J. and Junemann, C. 2016. Financial markets and investment in education. Verger, A., Lubienski, C. and Steiner-Khamsi, G. (eds), The Global Education Industry. London, Routledge, pp. 193–210.
- Singapore Ministry of Manpower. 2019. Labour Force in Singapore 2019. Singapore, Manpower Research and Statistics Department, Ministry of Manpower.
- SkillsFuture Singapore. 2020a. Skills Development Levy (SDL) System. Singapore, SkillsFuture Singapore. https://sdl.ssg.gov.sg. (Accessed 17 August 2021.)

\_\_\_\_. 2020b. SkillsFuture Credit. Singapore, SkillsFuture Singapore. www.skillsfuture.sg/credit. (Accessed 22 July 2021.) SkillsFuture Singapore and Workforce Singapore. 2020. 500,000 Individuals and 14,000 Enterprises Benefitted from

- SkillsFuture Programmes in 2019. Singapore, SkillsFuture Singapore/Workforce Singapore. www.ssg-wsg.gov. sg/news-and-announcements/500-000-individuals-and-14-000-enterprises-benefitted-from-skill.html (Accessed 17 August 2021.)
- Streck, D. R. and Zanini Moretti, C. 2018. Latin America: adult and popular education in dialogue. Milana, M., Webb, S., Holford, J., Waller, R. and Jarvis, P. (eds), *The Palgrave International Handbook on Adult and Lifelong Education and Learning*. London, Springer, pp. 443–60.
- Stromquist, N. P. and Lozano, G. 2018. Popular universities: their hidden functions and contributions. Milana, M., Webb, S., Holford, J., Waller, R. and Jarvis, P. (eds), The Palgrave International Handbook on Adult and Lifelong Education and Learning. London, Springer, pp. 779–96.
- Tenazas, N. M., Yamauchi, F., Tiongco, M. M. and Igarashi, T. 2016. Alternative Learning System Study: Alternative and Inclusive Learning in the Philippines. Washington, DC, World Bank.
- The Asia Foundation. 2020. Go Digital ASEAN. New York, The Asia Foundation. https://asiafoundation.org/emergingissues/go-digital-asean. (Accessed 17 August 2021.)
- Triyono, M. B. and Moses, K. M. 2019. Technical and vocational education and training and training in Indonesia. Bai, B. and Paryono (eds), Vocational Education and Training in ASEAN Member States: Current Status and Future Development. Singapore, Springer, pp. 45–79. (Perspectives on Rethinking and Reforming Education.)
- Uganda Business and Technical Examinations Board. 2019. Conduct of Examinations and Assessment Regulations. Kampala, Uganda Business and Technical Examinations Board.
- Uganda Ministry of Education and Sports. 2011. Skilling Uganda: BTVET Strategic Plan 2011–2020. Kampala, Ministry of Education and Sports.
- \_\_\_\_. 2019. Technical Vocational Education and Training Policy. Kampala, Ministry of Education and Sports.
- UIL. 2018. Cell-Ed: Innovative Education through Cell Phones United States of America. Hamburg, Germany, UNESCO Institute for Lifelong Learning. https://uil.unesco.org/case-study/effective-practices-database-litbase-0/ cell-ed-innovative-education-through-cell-phones. (Accessed 17 August 2021.)
- \_\_\_\_. 2019a. Fourth Global Report on Adult Learning and Education: Leave No One Behind Participation, Equity and Inclusion. Hamburg, Germany, UNESCO Institute for Lifelong Learning.
- \_\_\_\_. 2019b. Tata Consultancy Services' Adult Literacy Programme: Computer-based Functional Literacy, India. Hamburg, Germany, UNESCO Institute for Lifelong Learning. https://uil.unesco.org/case-study/effective-practicesdatabase-litbase-0/tata-consultancy-services-adult-literacy-programme. (Accessed 17 August 2021.)
- \_\_\_\_. 2020. Trends in Adult Learning and Education in the Arab States. Hamburg, Germany, UNESCO Institute for Lifelong Learning.

- UIL, Cedefop and ETF. 2015. Global Inventory of Regional and National Qualifications Frameworks: Volume II. Hamburg, Germany, European Centre for Development of Vocational Training/European Training Foundation/UNESCO Institute for Lifelong Learning.
- \_\_\_\_. 2019. Global Inventory of Regional and National Qualifications Frameworks 2019: Volume I. Hamburg, Germany, European Centre for Development of Vocational Training/European Training Foundation/UNESCO Institute for Lifelong Learning.
- UNESCO-UNEVOC, BIBB and SFIVET. 2019. Bridging Innovation and Learning in TVET Thematic Workshop: New Qualifications and Competencies in TVET. Bonn, UNESCO-UNEVOC International Centre for Technical and Vocational Education and Training/German Federal Institute for Vocational Education and Training/Swiss Federal Institute for Vocational Education and Training.
- UNESCO. 2016a. Recommendation Concerning Technical and Vocational Education and Training. Paris, UNESCO. \_\_\_\_\_. 2016b. Recommendation on Adult Learning and Education 2015. Paris, UNESCO.
- \_\_\_\_. 2017a. Global Education Monitoring Report 2017/8: Accountability in Education Meeting Our Commitments. Paris, UNESCO.
- \_\_\_\_. 2017b. Reading the Past, Writing the Future: Fifty Years of Promoting Literacy. Paris, UNESCO.
- \_\_\_\_. 2018. Funding Skills Development: The Private Sector Contribution. Paris, UNESCO.
- \_\_\_\_. 2019. Beyond Commitments: How Countries Implement SDG 4. Paris, UNESCO.
- UNESCO and ILO. 2018. Taking a Whole of Government Approach to Skills Development. Paris/Geneva, Switzerland, UNESCO/International Labour Organization.

UNESCO and Mongolia Ministry of Labor and Social Protection. 2019. TVET Policy Review: Mongolia. Ulaanbaatar, UNESCO. Viet Nam News. 2018. VinFast opens technical training centre. Viet Nam News, 8 February. https://vietnamnews.vn/ economy/422706/vinfast-opens-technical-training-centre.html. (Accessed 17 August 2021.)

\_\_\_\_\_. 2020. VinFast works with local colleges to train auto engineers. *Viet Nam News*, 29 May. http://bizhub.vn/corporatenews/vinfast-works-with-local-colleges-to-train-auto-engineers\_315931.html. (Accessed 17 August 2021.)

VSO Tanzania. 2014. Pathway to Vocational Employment in the Emerging Tanzanian Gas Sector. Dar Es Salaam, VSO Tanzania. Walker, J. and Sarkodie, G. 2019. Adult education as corporate social responsibility in Canadian mining companies:

- performing the good citizen for greater self-regulation. Compare, Vol. 49, No. 2, pp. 298–317.
- White, I., De Silva, N. and Rittie, T. 2018. Unaccredited Training: Why Employers Use It and Does It Meet Their Needs? Adelaide, Australia, National Centre for Vocational Education Research.
- World Bank. 2018a. Improving Youth Employability through Informal Apprenticeship. Washington, DC, World Bank.
- \_\_\_\_. 2018b. A Second Chance to Develop the Human Capital of Out-of-School Youth and Adults: The Philippines Alternative Learning System. Washington, DC, World Bank. (Philippines Education Note 1.)
- \_\_\_\_. 2019a. Firms offering formal training. Washigton, DC, World Bank. https://databank.worldbank.org/reports. aspx?source=2&series=IC.FRM.TRNG.ZS. (Accessed 17 August 2021.)
- \_\_\_. 2019b. World Development Report 2019: The Changing Nature of Work. Washington, DC, World Bank.
- Zarestky, J. and Ray, S. M. 2019. Adult education programmes of NGOs operating in non-Western contexts: a review of empirical literature. International Journal of Lifelong Education, Vol. 38, No. 6, pp. 657–72.
- Ziderman, A. 2016. Funding Mechanisms for Financing Vocational Training: An Analytical Framework. Bonn, Germany, Institute of Labor Economics. (IZA Policy Paper 110.)

# **MONITORING SECTION**

- Abreh, M. K., Agbevanu, W. K., Alhassan, A. J., Ansah, F., Bosu, R. S., Crawfurd, L., Mills, C. A., Minardi, A. L. and Nyame, G. 2021. What Happened to Dropout Rates after COVID-19 School Closures in Ghana? Washington, DC, Center for Global Development. www.cgdev.org/blog/what-happened-dropout-rates-after-covid-19-school-closuresghana. (Accessed 17 August 2021.)
- Agness, D., Dupas, P., Fafchamps, M., Getahun, T. and Lestant, E. 2021. Households' Attitudes towards School Reopening during COVID-19 in Ethiopia. London, International Growth Centre. (COVID-19-20109-ETH-1.)

- Akmal, M., Fry, L., Ghatak, N., Hares, S., Jha, J., Minardi, A. L. and Nyamweya, N. 2020. Who Is Going Back to School? A Four-country Rapid Survey in Ethiopia, India, Nigeria, and Pakistan. Washington, DC, Center for Global Development. www.cgdev.org/blog/who-going-back-school-four-country-rapid-survey-ethiopia-indianigeria-and-pakistan. (Accessed 17 August 2021.)
- Asanov, I., Flores, F., McKenzie, D., Mensmann, M. and Schulte, M. 2021. Remote-learning, time-use, and mental health of Ecuadorian high-school students during the COVID-19 quarantine. *World Development*, Vol. 138, art. 105225.
- Aurino, E., Wolf, S., Behrman, J. R., Aber, J. L., Tsinigo, E., Suntheimer, N. and Jordan, J. 2021. The Experience of Educators during the COVID-19 Pandemic: Surveys from during and after School Closures. Accra, Innovation for Poverty Action Ghana. (IPA Ghana Preliminary Results Brief.)
- Discenza, A. R. and Walsh, K. 2021. Global Review of Impacts of the COVID-19 Pandemic on Labour Force Surveys and Dissemination of Labour Market Statistics. Geneva, Switzerland, International Labour Organization.
- Gourlay, S., Kilic, T., Martuscelli, A., Wollburg, P. and Zezza, A. 2021. High-frequency phone surveys on COVID-19: good practices, open questions. Food Policy, Vol. 105, art. 102153.
- Grewenig, E., Lergetporer, P., Werner, K., Woessmann, L. and Zierow, L. 2021. COVID-19 and educational inequality: how school closures affect low-and high-achieving students. *European Economic Review*, Vol. 140, art. 103920.
- ICF. 2020. COVID-19 Update: Some DHS Surveys Return to the Field; Others Postponed until 2021. Rockville, Md., DHS Program Office, ICF. https://dhsprogram.com/Who-We-Are/News-Room/COVID-19-Update-Some-DHSsurveys-return-to-the-field-others-postponed-until-2021.cfm. (Accessed 17 August 2021.)
- Ikeda, M. and Yamaguchi, S. 2021. Online learning during school closure due to COVID-19. Japanese Economic Review, Vol. 72, pp. 471–507.
- Mangiavacchi, L., Piccoli, L. and Pieroni, L. 2021. Fathers matter: intrahousehold responsibilities and children's wellbeing during the COVID-19 lockdown in Italy. *Economics and Human Biology*, Vol. 42, art. 101016.
- Mbaye, S., Le Nestour, A., Moscoviz, L. and Chery, J. 2021. What Happened to Senegalese Students after the COVID-19 School *Closure*? Washington, DC, Center for Global Development. www.cgdev.org/blog/what-happened-senegalesestudents-after-covid-19-school-closure. (Accessed 17 August 2021.)
- Mongolia National Statistics Office and UNICEF. 2021. Mongolia 2020–2021 Wave 3: February 15–March 1, 2021. Ulaanbaatar, National Statistics Office of Mongolia/UNICEF. (MICS Plus.)
- Sharp, C., Nelson, J., Lucas, M., Julius, J., McCrone, T. and Sims, D. 2020. Schools' Responses to COVID-19: The Challenges Facing Schools and Pupils in September 2020. Slough, UK, National Foundation for Educational Research.
- UIS. 2020a. 7th Annual Meeting of the Technical Cooperation Group on the Indicators for SDG4-Education 2030 (TCG). Montreal, Que., Technical Cooperation Group, UNESCO Institute for Statistics. http://tcg.uis.unesco.org/ seventh-meeting-of-the-tcg. (Accessed 15 October 2021.)
- \_\_\_\_. 2020b. 8th Meeting of the TCG. Montreal, Que.,Technical Cooperation Group, UNESCO Institute for Statistics. http://tcg.uis.unesco.org/tcg8. (Accessed 11 November 2021.)
- \_\_\_\_. 2020c. Survey of COVID-19 Impact on National Education Planning Units. Montreal, Que., UNESCO Institute for Statistics. (Information Paper 66.)
- \_\_\_\_. 2021a. Continental Overview: Bridging CESA and SDG 4 in Africa. Montreal, Que., UNESCO Institute for Statistics.
- \_\_\_\_. 2021b. COVID-19 Impact. Montreal, Que., Global Education Observatory, UNESCO Institute for Statistics. https://geo.uis.unesco.org/covid-19/covid-19-impacts. (Accessed 11 November 2021.)
- \_\_\_\_. 2021c. SDG Benchmarks. Montreal, Que., Global Education Observatory, UNESCO Institute for Statistics. https://geo.uis.unesco.org/sdg-benchmarks. (Accessed 11 November 2021.)
- \_\_\_\_. 2021d. Setting Benchmarks to Achieve SDG 4 Targets. Montreal, Que.,Technical Cooperation Group, UNESCO Institute for Statistics. http://tcg.uis.unesco.org/benchmarks. (Accessed 15 October 2021.)
- \_\_\_\_. 2021e. UIS Data Release Features New SDG 4 Indicators and Disaggregated Dimensions. Montreal, Que., UNESCO Institute for Statistics. http://uis.unesco.org/en/news/uis-data-release-features-new-sdg-4-indicators-anddisaggregated-dimensions. (Accessed 11 November 2021.)
- UNESCO. 2015. Incheon Declaration and Framework for Action for the Implementation of Sustainable Development Goal 4. Paris, UNESCO.
- \_\_\_\_. 2017. Strengthening Peer Learning of Education Policies for SDG 4: the Role of Regional Organizations. Paris, UNESCO.
- \_\_\_\_. 2020. Extraordinary Session of the Global Education Meeting: Education post-COVID-19 2020 Global Education Meeting Declaration. Paris, UNESCO.
- UNESCO and UNICEF. 2021. Situation Analysis on the Effects of and Responses to COVID-19 on the Education Sector in Asia. Bangkok, UNESCO/UNICEF. https://bangkok.unesco.org/content/situation-analysis-effects-and-responsescovid-19-education-sector-asia-0. (Accessed 15 October 2021.)

UNESCO, UNICEF and World Bank. 2020. What Have We Learnt? Overview of Findings from a Survey of Ministries of Education on National Responses to COVID-19. Paris/New York/Washington, DC, UNESCO/UNICEF/World Bank.

- UNESCO, UNICEF, World Bank and OECD. 2021. What's Next? Lessons on Education Recovery: Findings from a Survey of Ministries of Education amid the COVID-19 Pandemic. Paris/Montreal, Que.,/New York/Florence/Washington, DC, UNESCO, UNESCO Institute for Statistics/UNICEF/UNICEF Office of Research – Innocenti/World Bank/OECD.
- UNFCCC. 2021a. Nationally Determined Contributions (NDCs). Bonn, Germany, United Nations Framework Convention on Climate Change. https://unfccc.int/process-and-meetings/the-paris-agreement/nationally-determinedcontributions-ndcs/nationally-determined-contributions-ndcs. (Accessed 15 October 2021.)
- \_\_\_\_. 2021b. Nationally Determined Contributions under the Paris Agreement: Synthesis Report by the Secretariat. Bonn, Germany, United Nations Framework Convention on Climate Change.
- UNICEF. 2021. MICS Plus. New York, UNICEF. https://mics.unicef.org/mics-plus/mics-plus-results. (Accessed 17 August 2021.)
- UNICEF and ITU. 2020. How Many Children and Young People Have Internet Access at Home? Estimating Digital Connectivity during the COVID-19 Pandemic. New York/Geneva, Switzerland, UNICEF/International Telecommunication Union.
- United Nations. 2014. The Road to Dignity by 2030: Ending Poverty, Transforming All Lives and Protecting the Planet – Synthesis Report of the Secretary-General on the post-2015 Agenda. New York, United Nations.
- UNSD. 2021. SDG Indicators: Metadata Repository Target 4.1. New York, United Nations Statistical Division. https://unstats.un.org/sdgs/metadata?Text=&Goal=&Target=4.1 (Accessed 17 August 2021.)
- World Bank. 2020. Monitoring COVID-19 Impacts on Households in Lao PDR: Results Snapshot from a Rapid Monitoring Phone Survey of Households. Washington, DC, World Bank. (Report 1.)
- \_\_\_\_. 2021. LSMS-supported High-frequency Phone Surveys on COVID-19. Washington, DC, World Bank. www.worldbank.org/en/ programs/lsms/brief/lsms-launches-high-frequency-phone-surveys-on-covid-19. (Accessed 17 August 2021.)

- Abadía Alvarado, L. K., Soler, S. C. G. and González, J. C. 2021. Gone with the pandemic: effects of COVID-19 on academic performance in Colombia. Universitas Económica, Vol. 21, No. 4.
- ACER and UIS. 2021. Study Design: COVID-19 Monitoring Impacts on Learning Outcomes. Melbourne, Australia/Montreal, Que., Australian Council of Educational Research/UNESCO Institute for Statistics.
- Akos, P., Rose, R. A. and Orthner, D. 2015. Sociodemographic moderators of middle school transition effects on academic achievement. *Journal of Early Adolescence*, Vol. 35, No. 2, pp. 170–98.
- Alkema, L., Chou, D., Hogan, D., Zhang, S., Moller, A.-B., Gemmill, A., Fat, D. M., Boerma, T., Temmerman, M. and Mathers, C. 2016. Global, regional, and national levels and trends in maternal mortality between 1990 and 2015, with scenario-based projections to 2030: a systematic analysis by the UN Maternal Mortality Estimation Inter-Agency Group. *The Lancet*, Vol. 387, No. 10017, pp. 462–74.
- Angrist, N., de Barros, A., Bhula, R., Chakera, S., Cummiskey, C., DeStefano, J., Floretta, J., Kaffenberger, M., Piper, B. and Stern, J. 2021. Building back better to avert a learning catastrophe: estimating learning loss from COVID-19 school shutdowns in Africa and facilitating short-term and long-term learning recovery. International Journal of Educational Development, Vol. 84, art. 102397.
- Ardington, C., Wills, G. and Kotze, J. 2021. COVID-19 learning losses: early grade reading in South Africa. International Journal of Educational Development, Vol. 86, art. 102480.
- Asakawa, S. and Ohtake, F. 2021. Impact of Temporary School Closure due to COVID-19 on the Academic Achievement of Elementary School Students. Osaka, Japan, Graduate School of Economics, Osaka University. (Discussion Paper in Economics and Business 21-14.)
- ASER. 2021. Karnataka Rural: Analysis Based on Data from Households 24 out of 30 Districts. New Delhi, Annual Status of Education Report.
- Asian Development Bank. 2021. Learning and Earning Losses from COVID-19 School Closures in Developing Asia: Special Topic of the Asian Development Outlook 2021. Manila, Asian Development Bank.
- Bau, N., Das, J. and Chang, A. Y. 2021. New evidence on learning trajectories in a low-income setting. International Journal of Educational Development, Vol. 84, art. 102430.
- Bonke, J. and Esping-Andersen, G. 2011. Family investments in children: productivities, preferences, and parental child care. European Sociological Review, Vol. 27, No. 1, pp. 43–55.
- British Council. 2021. Investigación y análisis acerca del cierre de las escuelas en América: 2020 [Research and analysis on school closures in America: 2020]. Buenos Aires, British Council Argentina/Varkey Foundation.

Brown, P. H. 2006. Parental education and investment in children's human capital in rural China. Economic Development and Cultural Change, Vol. 54, No. 4, pp. 759–89.

Buchanan, T., McFarlane, A. and Das, A. 2018. Educational attainment and the gender gap in childcare in Canada: a decomposition analysis. *Margin: The Journal of Applied Economic Research*, Vol. 12, No. 4, pp. 458–76.

Buchmann, M. and Steinhoff, A. 2017. Social inequality, life course transitions, and adolescent development: introduction to the special issue. *Journal of Youth and Adolescence*, Vol. 46, pp. 2083–90.

Carlana, M. and Ferrara, E. L. 2021. Apart but Connected: Online Tutoring and Student Outcomes during the COVID-19 Pandemic. Bonn, Germany, Institute of Labor Economics. (IZA Discussion Paper 14094.)

- CGD. 2021. The Pathway to Progress on SDG 4: A Symposium A Collection of Essays. Washington, DC, Center for Global Development.
- Cha, S.-E. and Song, Y.-J. 2017. Time or money: the relationship between educational attainment, income contribution, and time with children among Korean fathers. *Social Indicators Research*, Vol. 134, No. 1, pp. 195–218.
- Chen, J. 2020. Self-abandonment or seeking an alternative way out: understanding Chinese rural migrant children's resistance to schooling. British Journal of Sociology of Education, Vol. 41, No. 2, pp. 1–16.
- Crawfurd, L., Hares, S. and Minardi, A. L. 2021. New Data on Learning Loss in Pakistan. Washington, DC, Center For Global Development. www.cgdev.org/blog/new-data-learning-loss-pakistan. (Accessed 17 August 2021.)
- Dahir, A. L. 2020. Kenya's unusual solution to the school problem: cancel the year and start over. *The New York Times*, 5 August. www.nytimes.com/2020/08/05/world/africa/Kenya-cancels-school-year-coronavirus.html. (Accessed 17 August 2021.)
- Daily Star. 2021. Four urgent steps to put students on track for successful learning. 25 September. www.thedailystar. net/views/opinion/news/four-urgent-steps-put-students-track-successful-learning-2183326. (Accessed 1 October 2021.)
- Dang, H.-A., Oseni, G., Zezza, A. and Abanokova, K. 2021. Impact of COVID-19 on Learning: Evidence from Six Sub-Saharan African Countries. Washington, DC, World Bank.
- Dharamshi, A., Barakat, B., Alkema, L. and Antoninis, M. 2021. Adjusted Bayesian completion rates (ABC) estimation. SocArXiv, 21 January.
- Dotti Sani, G. M. and Treas, J. 2016. Educational gradients in parents' child-care time across countries, 1965–2012. Journal of Marriage and Family, Vol. 78, No. 4, pp. 1083–96.
- Dube, R. 2020. Bolivia decision to cancel school because of COVID-19 upsets parents. *The Wall Street Journal*, 4 August. www.wsj.com/articles/bolivia-decision-to-cancel-school-because-of-covid-19-upsets-parents-11596577822. (Accessed 17 August 2021.)
- Eccles, J. S. and Wigfield, A. 2002. Motivational beliefs, values, and goals. Annual Review of Psychology, Vol. 53, No. 1, pp. 109–32.
- Eulich, W. 2020. Bolivia cancels school year. Parents ask: What now? *The Christian Science Monitor*, 17 September. www.csmonitor.com/World/Americas/2020/0917/Bolivia-cancels-school-year.-Parents-ask-What-now. (Accessed 17 Augut 2021.)
- France Ministry of National Education. 2021a. 800 000 élèves évalués en début de sixième en 2020 : des performances en hausse, mais toujours contrastées selon les caractéristiques des élèves et des établissements [800,000 students assessed at the start of grade 6 in 2020: performance on the rise, but still affected by student and school characteristics]. Paris, Ministry of National Education, Youth and Sport. www.education.gouv.fr/800-000eleves-evalues-en-debut-de-sixieme-en-2020-des-performances-en-hausse-mais-toujours-309160. (Accessed 17 August 2021.)
- \_\_\_\_\_. 2021b. Évaluations repères 2020 de début de CP et de CE1 : baisse des performances par rapport à 2019, notamment en français en CE1, et hausse des écarts selon les secteurs de scolarisation [2020 benchmark assessments at the start of grades 1 and 2: lower performance compared to 2019, especially in French in grade 2, and increase in the gaps by sector]. Paris, Ministry of National Education, Youth and Sport. (Accessed 17 August 2021.)
- Friedlander, E. W. 2020. The home literacy environment in rural Rwanda and its relationship to early grade reading. Scientific Studies of Reading, Vol. 24, No. 2, pp. 123–40.
- Gölz, N. and Wohlkinger, F. 2019. Determinants of students' idealistic and realistic educational aspirations in elementary school. Zeitschrift für Erziehungswissenschaft, Vol. 22, No. 6, pp. 1397–431.
- Gonida, E. N. and Cortina, K. S. 2014. Parental involvement in homework: relations with parent and student achievement-related motivational beliefs and achievement. *British Journal of Educational Psychology*, Vol. 84, No. 3, pp. 376–96.

Gore, J., Fray, L., Miller, A., Harris, J. and Taggart, W. 2021. The impact of COVID-19 on student learning in New South Wales primary schools: an empirical study. *Australian Educational Researcher*, Vol. 48, No. 4, pp. 605–37.

Guryan, J., Hurst, E. and Kearney, M. 2008. Parental education and parental time with children. *Journal of Economic* Perspectives, Vol. 22, No. 3, pp. 23–46.

Gustafsson, M. and Nuga, C. 2020. How Is the COVID-19 Pandemic Affecting Educational Quality in South Africa? Evidence to Date and Future Risks. Stellenbosch, Cape Town and Johannesburg, South Africa, National Income Dynamics Study – Coronavirus Rapid Mobile Survey. (Insight Brief.)

- Hammerstein, S., König, C., Dreisörner, T. and Frey, A. 2021. Effects of COVID-19-related school closures on student achievement: a systematic review. *Frontiers in Psychology*, Vol. 12, art. 746289.
- Hayashi, R., Sudarshana, H. D., Jayasundara, A., Garcia, M., Balasuriya, A. and Hirokawa, T. 2021. COVID-19 Impact on Technical and Vocational Education and Training in Sri Lanka. Manila, Asian Development Bank. (ADB Brief 168.)
- Helbling, L. A., Tomasik, M. J. and Moser, U. 2019. Long-term trajectories of academic performance in the context of social disparities: longitudinal findings from Switzerland. *Journal of Educational Psychology*, Vol. 111, No. 7, p. 1284–99.
- House of Commons. 2021. COVID-19: Support for Children's Education. London, House of Commons. https://publications. parliament.uk/pa/cm5802/cmselect/cmpubacc/240/24002.htm. (Accessed 17 August 2021.)
- Human Rights Watch. 2021. Africa: Rights Progress for Pregnant Students. New York, Human Rights Watch. www.hrw.org/news/2021/09/29/africa-rights-progress-pregnant-students. (Accessed 11 October 2021.)
- IADB. 2020. Hablemos de Política Educativa América Latina y el Caribe:Educación más allá del COVID-19 [Let's Talk about Educational Policy in Latin America and the Caribbean: Education beyond COVID-19]. Washington, DC, Inter-American Development Bank.
- IEA and UNESCO. 2020. Measuring Global Education Goals: How TIMSS Helps. Amsterdam/Paris, International Association for the Evaluation of Educational Achievement/UNESCO.
- INVALSI. 2021. Rilevazioni Nazionali Degli Apprendimenti 2020–21 [National Learning Surveys 2020–21]. Rome, Istituto nazionale per la valutazione del sistema educativo di istruzione e di formazione.
- Kaffenberger, M. 2019. A Typology of Learning Profiles: Tools for Analysing the Dynamics of Learning. Oxford, UK, Research on Improving Systems of Education. (Insights.)
- Kalita, B. 2020. Odisha to reduce school syllabus in view of COVID-19. NDTV, 7 August. www.ndtv.com/education/ odisha-reduce-school-syllabus-in-view-of-covid-19. (Accessed 17 August 2021.)
- Kao, G. and Tienda, M. 1998. Educational aspirations of minority youth. American Journal of Education, Vol. 106, No. 3, pp. 349–84.
- Kim, J., Rose, P., Tiruneh, D. T., Sabates, R. and Woldehanna, T. 2021. Learning Inequalities Widen following COVID-19 School Closures in Ethiopia. Oxford, UK, Research on Improving Systems of Education. https://riseprogramme.org/ blog/learning-inequalities-widen-COVID-19-Ethiopia. (Accessed 17 August 2021.)
- Lam, S. f., Jimerson, S., Shin, H., Cefai, C., Veiga, F. H., Hatzichristou, C., Polychroni, F., Kikas, E., Wong, B. P. and Stanculescu, E. 2016. Cultural universality and specificity of student engagement in school: the results of an international study from 12 countries. British Journal of Educational Psychology, Vol. 86, No. 1, pp. 137–53.
- Leh Wi Lan. 2021. Sierra Leone Back to School Study 2020: Status of Pupil Learning Outcomes in Junior and Senior Secondary Schools of Sierra Leone Post-school Closure. Cambridge/Oxford, UK, Mott Macdonald/Oxford Policy Management.
- Lichand, G., Dória, C. A., Neto, O. L. and Cossi, J. 2021. The Impacts of Remote Learning in Secondary Education: Evidence from Brazil during the Pandemic. Washington, DC, Inter-American Development Bank.
- Malawi Government. 2013. Malawi Education Act. Lilongwe, Government of Malawi.
- Meeter, M. 2021. Primary school mathematics during Covid-19: no evidence of learning gaps in adaptive practicing results. *Trends in Neuroscience and Education*, Vol. 25, art. 100163.
- Mukhopadhyay, A. and Sahoo, S. 2016. Does access to secondary education affect primary schooling? Evidence from India. *Economics of Education Review*, Vol. 54, pp. 124–42.
- Mullis, I. V. S., Martin, M. O., Foy, P., Kelly, D. L. and Fishbein, B. 2020. TIMSS 2019 International Results in Mathematics and Science. Chestnut Hill, Mass./Amsterdam, TIMSS & PIRLS International Study Center, Lynch School of Education and Human Development, Boston College/International Association for the Evaluation of Educational Achievement.

- Nigeria National Bureau of Statistics. 2019. *Living Standards Survey* 2018–2019. Abuja, National Bureau of Statistics. https://microdata.worldbank.org/index.php/catalog/3827. (Accessed 17 August 2021.)
- Nugroho, D., Pasquini, C., Reuge, N. and Amaro, D. 2020. COVID-19: How Are Countries Preparing to Mitigate the Learning Loss as Schools Reopen? Trends and Emerging Good Practices to Support the Most Vulnerable Children. Florence, Italy, UNICEF Office of Research – Innocenti. (Innocenti Research Brief 2020-20.)
- Oduor, A. and Gikandi, B. 2020. Learners will not repeat classes, State declares. *The Saturday Standard*, 6 November. www.standardmedia.co.ke/education/article/2001392858/learners-will-not-repeat-classes-state-declares. (Accessed 17 August 2021.)
- Ortiz-Ospina, E., Giattino, C. and Roser, M. 2020. *Time Use*. Oxford, UK, Our World in Data. https://ourworldindata.org/ time-use. (Accessed 18 June 2021.)
- PASEC. 2015. PASEC2014 : Performances des Systèmes Éducatifs en Afrique Subsaharienne Francophone Compétences et Facteurs de Réussite au Primaire [PASEC2014: Education System Performance in Francophone Africa – Competencies and Learning Factors in Primary Education]. Dakar, Conférence des Ministres de l'Éducation des États et Gouvernements de la Francophonie.
- \_\_\_\_. 2020. PASEC2019 : Qualité des Systèmes Éducatifs en Afrique Subsaharienne Francophone Performances et Environnement de l'Enseignement-Apprentissage au Primaire [PASEC2019: Education System Quality in Francophone Africa – Performance and Teaching-Learning Environment in Primary Education]. Dakar, Conférence des Ministres de l'Éducation des États et Gouvernements de la Francophonie.
- Patall, E. A., Cooper, H. and Robinson, J. C. 2008. Parent involvement in homework: a research synthesis. Review of Educational Research, Vol. 78, No. 4, pp. 1039–101.
- Philippines Department of Education. 2021. DepEd Issues Guidelines on Conduct of Remedial and Advancement Classes during Summer 2021. Manila, Department of Education. www.deped.gov.ph/2021/07/16/deped-issues-guidelines-onconduct-of-remedial-and-advancement-classes-during-summer-2021. (Accessed 17 August 2021.)
- Radhakrishnan, K., Angrist, N., Bergman, P., Cullen, C., Matsheng, M., Ramakrishnan, A., Sabarwal, S. and Sharma, U. 2021. Learning in the Time of COVID-19: Insights from Nepal. Washington, DC, World Bank.
- Rahman, T. and Ahmed, R. 2021. Combatting the Impact of COVID-19 School Closures in Bangladesh. Washington, DC, World Bank. https://blogs.worldbank.org/endpovertyinsouthasia/combatting-impact-covid-19-school-closuresbangladesh. (Accessed 17 August 2021.)
- Robinson, K. and Harris, A. L. 2014. The Broken Compass: Parental Involvement with Children's Education. Cambridge, Mass., Harvard University Press.
- Sengeh, D. 2021. The Power of Radio in Sierra Leone: Ensuring No Child Is Left Behind. Paris, UNESCO. https:// gemreportunesco.wpcomstaging.com/2021/07/16/the-power-of-radio-in-sierra-leone-ensuring-no-child-isleft-behind. (Accessed 17 August 2021.)
- Silverstein, J. 2021. Measuring, Visualising, and Simulating Solutions to the Learning Crisis: New Evidence from Learning Profiles in 18 Countries. Oxford, UK, Research on Improving Systems of Education. (Insights.)
- Somsanith, P. and Noonan, R. 2020. Preschool and primary education. Noonan, R. (ed.), Education in the Lao People's Democratic Republic: On Track for the Twenty-First Century. Singapore, Springer, pp. 63–86.
- Spitzer, M. and Musslick, S. 2021. Academic performance of K-12 students in an online-learning environment for mathematics increased during the shutdown of schools in wake of the COVID-19 pandemic. *PLoS ONE*, Vol. 16, No. 8, art. 0255629.
- Statistics South Africa. 2018. General Household Survey 2018. Pretoria, Statistics South Africa. (Statistical Release P0318.)
- Tomasik, M. J., Helbling, L. A. and Moser, U. 2021. Educational gains of in-person vs. distance learning in primary and secondary schools: a natural experiment during the COVID-19 pandemic school closures in Switzerland. International Journal of Psychology, Vol. 56, No. 4, pp. 566–76.
- Uganda Bureau of Statistics. 2019. Uganda National Panel Survey 2018/19: Household Questionnaire. Kampala, Uganda Bureau of Statistics.
- UIS. 2018. Final Report of the Results of the Consensus Building Meeting on Proficiency Levels. Montreal, Que., UNESCO Institute for Statistics.
- \_\_\_\_. 2020. Rosetta Stone project. Montreal, Que., UNESCO Institute for Statistics. http://tcg.uis.unesco.org/rosettastone. (Accessed 17 August 2021.)
- UIS, UNICEF and World Bank. 2021. Learning Data Compact: UNESCO, UNICEF, and the World Bank Unite to End the Learning Data Crisis. Montreal, Que., UNESCO Institute for Statistics.
- UIS and USAID. 2020a. Global Proficiency Framework for Mathematics Grades 1 to 9. Montreal, Que.,/Washington, DC, UNESCO Institute for Statistics/US Agency for International Development.

- \_\_\_\_. 2020b. Global Proficiency Framework for Reading Grades 1 to 9. Montreal, Que.,/Washington, DC, UNESCO Institute for Statistics/US Agency for International Development.
- \_\_\_\_. 2020c. Policy Linking for Measuring Global Learning Outcomes: Toolkit Linking Assesments to the Global Proficiency Framework. Montreal, Que.,/Washington, DC, UNESCO Institute for Statistics/US Agency for International Development.
- UNESCO. 2019. HerAtlas: Monitoring the Right to Education for Girls and Women. Paris, UNESCO. https://en.unesco.org/ education/girls-women-rights. (Accessed 17 August 2021.)
- \_\_\_\_. 2020. Global Education Monitoring Report 2020: Inclusion and Education All Means All. Paris, UNESCO.
- UNESCO and UNICEF. 2021a. China Case Study: Situation Analysis on the Effects of and Responses to COVID-19 on the Education Sector in Asia. Bangkok/Kathmandu, UNESCO Asia and Pacific Regional Bureau for Education/ UNICEF East Asia and Pacific Regional Office and Regional Office for South Asia.
- \_\_\_\_. 2021b. India Case Study: Situation Analysis on the Effects of and Responses to COVID-19 on the Education Sector in Asia. Bangkok/Kathmandu, UNESCO Asia and Pacific Regional Bureau for Education/UNICEF East Asia and Pacific Regional Office and Regional Office for South Asia.
- \_\_\_\_. 2021c. Indonesia Case Study: Situation Analysis on the Effects of and Responses to COVID-19 on the Education Sector in Asia. Bangkok/Kathmandu, UNESCO Asia and Pacific Regional Bureau for Education/UNICEF East Asia and Pacific Regional Office and Regional Office for South Asia.
- \_\_\_\_. 2021d. Situation Analysis on the Effects of and Responses to COVID-19 on the Education Sector in Southeast Asia: Sub-regional Report. Bangkok/Kathmandu, UNESCO Asia and Pacific Regional Bureau for Education/UNICEF East Asia and Pacific Regional Office and Regional Office for South Asia.
- UNESCO, UNICEF and World Bank. 2021a. COVID-19 Learning Losses: Rebuilding Quality Learning for All in the Middle East and North Africa. Paris/Amman/Washington, DC, UNESCO/UNICEF Regional Office in the Middle East and North Africa/World Bank.
- \_\_\_\_. 2021b. The State of the Global Education Crisis: A Path to Recovery. Paris/New York/Washington, DC, UNESCO/ UNICEF/World Bank.
- UNESCO, UNICEF, World Bank and OECD. 2021c. What's Next? Lessons on Education Recovery: Findings from a Survey of Ministries of Education amid the COVID-19 Pandemic. Paris/Montreal, Que.,/New York/Florence/Washington, DC, UNESCO, UNESCO Institute for Statistics/UNICEF/UNICEF Office of Research – Innocenti/World Bank/ OECD.
- USAID. 2021. Policy Linking for Measuring Global Learning Outcomes. Washington, DC, US Agency for International Development. www.edu-links.org/resources/policy-linking-measuring-global-learning-outcomes. (Accessed 11 October 2021.)
- van de Werfhorst, H. G. 2021. Inequality in learning is a major concern after school closures. Proceedings of the National Academy of Sciences, Vol. 118, No. 20, art. 2105243118.
- Varkey Foundation. 2018. Global Parents' Survey. London, Varkey Foundation.

Whizz Education. 2021. Measuring the Impact of COVID-19 on Learning in Rural Kenya. London, Whizz Education.

World Bank. 2021a. Accelerating Learning Recovery. Washington, DC, World Bank. (Brief.)

\_\_\_\_. 2021b. Adapting the Curriculum. Washington, DC, World Bank. (World Bank Brief/Note.)

- You, D., Hug, L., Ejdemyr, S., Idele, P., Hogan, D., Mathers, C., Gerland, P., New, J. R. and Alkema, L. 2015. Global, regional, and national levels and trends in under-5 mortality between 1990 and 2015, with scenario-based projections to 2030: a systematic analysis by the UN Inter-agency Group for Child Mortality Estimation. *The Lancet*, Vol. 386, No. 10010, pp. 2275–86.
- Zierer, K. 2021. Effects of pandemic-related school closures on pupils' performance and learning in selected countries: a rapid review. *Education Sciences*, Vol. 11, No. 6, art. 252.

- Atiles, J. T., Almodóvar, M., Vargas, A. C., Dias, M. J. A. and León, I. M. Z. 2021. International responses to COVID-19: challenges faced by early childhood professionals. European Early Childhood Education Research Journal, Vol. 29, No. 1, pp. 66–78.
- Attanasio, O., Paes de Barros, R., Carneiro, P., Evans, D., Lima, L., Olinto, P. and Schady, N. 2017. Impact of Free Availability of Public Childcare on Labour Supply and Child Development in Brazil. New Delhi, International Initiative for Impact Evaluation. (Impact Evaluation Report 58.)

Brossard, M., Cardoso, M., Kamei, A., Mishra, S., Mizunoya, S. and Reuge, N. 2020. Parental Engagement in Children's *Learning.* Florence, Italy, UNICEF Office of Research – Innocenti. (Innoccenti Research Brief 2020-09.)

- Cabrera-Hernandez, F. and Padilla-Romo, M. 2020. Hidden violence: how COVID-19 school closures reduced the reporting of child maltreatment. *Latin American Economic Review*,Vol. 29.
- Dong, C., Cao, S. and Lia, H. 2020. Young children's online learning during COVID-19 pandemic: Chinese parents' beliefs and attitudes. *Children and Youth Services Review*, Vol. 118, art. 105440.
- European Commission. 2011. Communication from the Commission: Early Childhood Education and Care: Providing All Our Children with the Best Start for the World of Tomorrow. Brussels, European Commission.
- \_\_\_\_. 2014. Proposal for Key Principles of a Quality Framework for Early Childhood Education and Care: Report of the Working Group on Early Childhood Education and Care Under the Auspices of the European Commission. Brussels, European Commission.
- \_\_\_\_. 2020. Inclusion for All: Commission Presents Action Plan on Integration and Inclusion 2021–2027. Brussels, European Commission. https://ec.europa.eu/commission/presscorner/detail/en/ip\_20\_2178. (Accessed 17 August 2021.)
- \_\_\_\_. 2021a. Commission Staff Working Document Accompanying the Document Proposal for a Council Recommendation Establishing a European Child Guarantee. Brussels, European Commission.
- \_\_\_\_. 2021b. Toolkit for Inclusive Early Childhood Education and Care: Providing High Quality Education and Care to All Young Children. Brussels, European Commission.
- European Council. 2018. Council Recommendation of 22 May 2018 on Promoting Common Values, Inclusive Education, and the European Dimension of Teaching. Brussels, Council of the European Union.
- \_\_\_\_. 2019. Council Recommendation of 22 May 2019 on High-Quality Early Childhood Education and Care Systems. Brussels, Council of the European Union.
- Eurostat. 2021. Pupils from age 4 to the starting age of compulsory education at primary level, by sex as % of the population of the corresponding age group. Luxembourg, Eurostat. https://ec.europa.eu/eurostat/databrowser/view/educ\_uoe\_enra10/default/table?lang=en. (Accessed 17 August 2021.)
- Fletcher, K. L. and Reese, E. 2005. Picture book reading with young children: a conceptual framework. *Developmental Review*, Vol. 25, No. 1, pp. 39–64.
- Galevski, M., Adona, V. J. A., Barbosa, B. B., Ben Yahmed, Z., Currimjee, A., Ibrahim, R., Song, C., Tazi, S. and Yacoub, R. 2021. COVID-19 and the Early Years: A Cross-Country Overview of Impact and Response in Early Childhood Development. Washington, DC, World Bank.
- Ghana Ministry of Women and Children's Affairs. 2004. Early Childhood Care and Development Policy. Accra, Ministry of Women and Children's Affairs.
- Hanno, E. C., Hayhurst, E. W., Fritz, L., Gardner, M., Turco, R. G., Jones, S. M., Lesaux, N. K., Hofer, K., Checkoway, A. and Goodson, B. 2021. Persevering through the Pandemic: Key Learnings about Children from Parents and Early Educators. Cambridge, Mass., Saul Zaentz Early Education Initiative, Harvard Graduate School of Education.
- Howard-Jones, A. R., Bowen, A. C., Danchin, M., Koirala, A., Sharma, K., Yeoh, D. K., Burgner, D. P., Crawford, N. W., Goeman, E., Gray, P. E., Hsu, P., Kuek, S., McMullan, B. J., Tosif, S., Wurzel, D. and Britton, P. N. 2021. COVID-19 in children: epidemiology, prevention and indirect impacts. *Journal of Paediatrics and Child Health*.
- IBGE. 2018. Educação 2017: PNAD Contínua [Education 2017: Continuous PNAD]. Rio de Janeiro, Brazil, Instituto Brasileiro de Geografia e Estatística.
- IEA and UNESCO. 2020. *Measuring Global Education Goals: How TIMSS Helps*. Amsterdam/Paris, International Association for the Evaluation of Educational Achievement/UNESCO.
- Laire, C. 2016. Early Childhood Development in Cuba: Sharing the Experience of a Scaled-up Integrated System that Promotes the Best Start in Life for Every Child. Havana, UNICEF.
- Luxembourg National Youth Service. 2020. Quality beyond Regulations in Early Childhood Education and Care (ECEC): Country Background Report of Luxembourg. Luxembourg, Service National de la Jeunesse.
- McCoy, D. C., Cuartas, J., Behrman, J., Cappa, C., Heymann, J., Bóo, F. L., Lu, C., Raikes, A., Richter, L., Stein, A. and Fink, G. 2021. Global estimates of the implications of COVID-19-related preprimary school closures for children's instructional access, development, learning, and economic wellbeing. *Child Development*, Vol. 92, No. 5, pp. 883–99.
- McKenna, M., Soto-Boykin, X., Cheng, K., Haynes, E., Osorio, A. and Altshuler, J. 2021. Initial development of a national survey on remote learning in early childhood during COVID-19: establishing content validity and reporting successes and barriers. *Early Childhood Education Journal*.
- Mozambique Ministry of Education and Human Development. 2020. Plano Estratégico da Educação 2020–2029 [Education Strategic Plan 2020–2029]. Maputo, Ministry of Education and Human Development.

- Nugroho, D., Jeon, Y., Kamei, A. and Boo, F. L. 2021. It's Not Too Late to Act on Early Learning: Understanding and Recovering from the Impact of Pre-primary Education Closures During COVID-19. Florence, UNICEF Office of Research – Innocenti. (Innocenti Research Brief 2021-03.)
- OECD. 2017. Starting Strong 2017: Key OECD Indicators on Early Childhood Education and Care. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2018. How Does Access to Early Childhood Services Affect the Participation of Women in the Labour Market? Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2019. Education at a Glance 2019: OECD Indicators. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2020a. Education at a Glance 2020: OECD Indicators. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2020b. Is Childcare Affordable? Paris, Organisation for Economic Co-operation and Development. (Policy Brief on Employment, Labour and Social Affairs.)
- Oktavianingsih, E. and Arifiyanti, N. 2021. School readiness for early childhood in face-to-face learning in pandemic Covid-19. Indonesian Journal of Educational Assessment, Vol. 4, No. 1, pp. 30–37.
- Pascal, C. and Bertram, T. 2021. What do young children have to say? Recognising their voices, wisdom, agency and need for companionship during the COVID pandemic. *European Early Childhood Education Research Journal*, Vol. 29, No. 1, pp. 1–5.
- Rao, N., Umayahara, M., Yang, Y. and N., R. 2021. Ensuring access, equity and quality in early childhood education in Bangladesh, China, India and Myanmar: challenges for nations in a populous economic corridor. International Journal of Educational Development, Vol. 82, art. 102380.
- Szente, J. 2020. Live virtual sessions with toddlers and preschoolers amid COVID-19: implications for early childhood teacher education. Journal of Technology and Teacher Education, Vol. 28, No. 2, pp. 373–80.
- UNESCO. 2021a. The Impact of COVID-19 on Early Childhood Education in the Asia-Pacific and Sub-Saharan Africa: Insights from the Results of Rapid Regional Personnel Survey. Bangkok/Dakar, UNESCO.
  - \_. 2021b. Right to Pre-primary Education: A Global Study. Paris, UNESCO.
- UNESCO, UNICEF and World Bank. 2020. What Have We Learnt? Overview of Findings from a Survey of Ministries of Education on National Responses to COVID-19. Paris/New York/Washington, DC, UNESCO/UNICEF/World Bank.
- UNESCO, UNICEF, World Bank and OECD. 2021. What's Next? Lessons on Education Recovery: Findings from a Survey of Ministries of Education amid the COVID-19 Pandemic. Paris/Montreal, Que.,/New York/Florence/Washington, DC, UNESCO, UNESCO Institute for Statistics/UNICEF/UNICEF Office of Research – Innocenti/World Bank/OECD.

United Nations. 2006. Convention on the Rights of the Child: General Comment No. 7. Geneva, Switzerland, United Nations. Varmuza, P., Perlman, M. and White, L. A. 2019. Understanding early childhood education and care utilization in Canada:

- implications for demand and oversight. International Journal of Child Care and Education Policy, Vol. 13, art. 7. Walker, I. 2019. Inside Brazil's Hidden Daycare Economy. London, BBC. www.bbc.com/worklife/article/20190306-inside
  - brazils-secret-daycare-economy. (Accessed 17 August 2021.)
- Winter, P. 2010. Engaging Families in the Early Childhood Development Story: Neuroscience and Early Childhood Development – Summary of Selected Literature and Key Messages for Parenting. Canberra, Ministerial Council for Education, Early Childhood Development and Youth Affairs.
- World Bank and UNICEF. 2020. Costing Study on Early Childhood Education and Development (ECED) in Nepal: A Case for Investment in ECED. Washington, DC/Kathmandu, World Bank/UNICEF.
- Yoshikawa, H., Wuermli, A. J., Britto, P. R., Dreyer, B., Leckman, J. F., Lye, S. J., Ponguta, L. A., Richter, L. M. and Stein, A.
  2020. Effects of the global coronavirus disease-2019 pandemic on early childhood development: short- and long-term risks and mitigating program and policy actions. *The Journal of Pediatrics*, Vol. 223, pp. 188–93.

- Aivazova, N. 2013. Role of Apprenticeships in Combating Youth Unemployment in Europe and the United States. Washington, DC, Peterson Institute for International Economics. (Policy Brief 13-20.)
- Ali Asadullah, M. 2019. Quadratic indirect effect of national TVET expenditure on economic growth through social inclusion indicators. SAGE Open, Vol. 9, No. 1, pp. 1–13.
- Alvarado Calderón, G. and Mora Hernández, R. 2020. Educación técnica secundaria pública en Costa Rica: 1950–2014 [Public technical secondary educacion in Costa Rica: 1950–2014]. Revista Actualidades Investigativas en Educación, Vol. 20, No. 2.
- Ambrose, B. W., Cordell, L. and Ma, S. 2015. The Impact of Student Loan Debt on Small Business Formation. Philadelphia, Pa., Federal Reserve Bank of Philadelphia. (Research Department Working Paper 15-26 ).

- Arribas, M. G. and Papadakis, N. 2019. Governance Arrangements for Vocational Education and Training in ETF Partner Countries: Analytical Overview 2012–17. Turin, Italy, European Training Foundation.
- Barr, A. and Turner, S. E. 2013. Expanding enrollments and contracting state budgets: the effect of the Great Recession on higher education. Annals of the American Academy of Political and Social Science, Vol. 650, No. 1, pp. 168–93.
- Batthyány, K., Bruno, F. and Perrotta, V. 2021. International Review of Study/Training Leave Policies. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Beirute Brealey, T. 2018. Principales Cambios en la Oferta de Educación Técnica Presentados en el Periodo 2006–2018 y su Pertinencia para Jóvenes en Zonas de Alta Vulnerabilidad [Main Changes in Technical Education Provision in the Period 2006–2018 and Their Relevance for Youth in Areas of High Vulnerability]. San José, Consejo Nacional de Rectores. (Informe Estado de la Educación.)
- Black, S. E., Denning, J. T., Dettling, L. J., Goodman, S. and Turner, L. J. 2020. Taking it to the Limit: Effects of Increased Student Loan Availability on Attainment, Earnings, and Financial Well-Being. Cambridge, Mass., National Bureau of Economic Research. (Working Paper 27658.)
- Boatman, A., Evans, B. J. and Soliz, A. 2017. Understanding loan aversion in education: evidence from high school seniors, community college students, and adults. *AERA Open*, Vol. 3, No. 1.
- Bonomelli, F. 2021. Engagement in Apprenticeships: A Cross-national Analysis of School-to-Work Transition Surveys. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Brazil Ministry of Economy. 2017. Diagnóstico FIES [Diagnostic of the Higher Education Student Financing Fund]. Brasília, Ministry of Economy.
- Britton, J., van der Erve, L. and Higgins, T. 2019. Income contingent student loan design: lessons from around the world. *Economics of Education Review*, Vol. 71, pp. 65–82.
- Brossard, V. 2020. Les populations en situation de précarité au temps de la COVID [People living precariously in COVID-19 times]. Montreal, Que., Institut de coopération pour l'éducation des adultes. https://icea.qc.ca/fr/actualites/lespopulations-en-situation-de-pr%C3%A9carit%C3%A9-au-temps-de-la-covid. (Accessed 17 August 2021.)
- Burundi Ministry of Education and AFD. 2018. Programme d'appui a la mise oeuvre du plan transitoire de l'education du Burundi « Twige Neza » [Support program for the Implementation of Burundi's 'Twige Neza' Transitional Education plan]. Gitega/ Paris, Ministry of Education, Technical and Professional Training/Agence Française de Développement.
- Callender, C. 2008. The impact of term-time employment on higher education students' academic attainment and achievement. *Journal of Education Policy*, Vol. 23, No. 4, pp. 359–77.
- Camacho-Calvo, S., García-Fallas, J., Kemper, J. M., Maldonado-Mariscal, K. and Vargas-Porras, A. 2019. Country Case Study on Technical Vocational Education and Training (TVET) in Costa Rica. Zurich, Switzerland, KOF Swiss Economic Institute. (LELAM-TVET4Income Working Paper 8.)
- Cardozo, S. 2008. Políticas Educativas, Logros y Desafíos del Sector en Uruguay 1990–2008 [Education Sector Policies, Achievements and Challenges in Uruguay 1990–2008]. Montevideo, Comité de Coordinación Estratégica de Infancia y Adolescencia. (Cuadernos del ENIA.)
- Carroll, J. 2020. It's time to reimagine adult literacy in a post-COVID-19 world. *Mail & Guardian*, 8 September. https://mg.co.za/education/2020-09-08-its-time-to-reimagine-adult-literacy-in-a-post-covid-19-world. (Accessed 17 August 2021.)
- Cedefop. 2012. Training Leave Policies and Practice in Europe. Luxembourg, European Centre for the Development of Vocational Training. (Research Paper 28.)
- Cerdan-Infantes, P. 2018. Additional Financing for Access and Quality in Higher Education Project: Combined Project Information Documents/Integrated Safeguards Datasheet. Washington, DC, World Bank.
- Chaker, A. 2017. Community learning centres in Morocco: an unprecedented experience. Avramovska, M., Hirsch, E. and Schmidt-Behlau, B. (eds), Adult Education Centres as a Key to Development: Challenges and Success Factors. Bonn, Germany, DVV International, pp. 19–28. (International Perspectives in Adult Education.)
- Chapman, B. 2016. Income contingent loans in higher education financing. IZA World of Labor, No. 227.
- Chapman, B. and Dearden, L. 2018. The international transformation in student loan systems. University World News, 11 April. www.universityworldnews.com/post.php?story=20180411082843685. (Accessed 17 August 2021.)
- Chapman, B. and Lounkaew, K. 2015. An analysis of Stafford loan repayment burdens. *Economics of Education Review*, Vol. 45, pp. 89–102.
- Commonwealth of Learning. 2020. Strategies for Blended TVET in Response to COVID-19. Burnaby, BC, Commonwealth of Learning.
- CPF. 2021. Comparative Panel File: Harmonizing Comparative Lifecourse Data. The Hague, Netherlands/Cologne, Germany, Netherlands Interdisciplinary Demographic Institute/University of Cologne. https://www.cpfdata.com/ (Accessed 17 August 2021.)

- D'Amato, P. 2021. Interactive: Explore who gains most from canceling student debt. *The Hechinger Report*, 23 February. https://hechingerreport.org/interactive-explore-who-gains-most-from-cancelling-student-debt. (Accessed 30 August 2021.)
- Darolia, R. 2014. Working (and studying) day and night: heterogeneous effects of working on the academic performance of full-time and part-time students. *Economics of Education Review*, Vol. 38, pp. 38–50.
- Dearden, L. and Nascimento, P. M. 2019. Modelling alternative student loan schemes for Brazil. *Economics of Education Review*, Vol. 71, pp. 83–94.
- Dolado, J. 2015. No Country for Young People? Youth Labour Market Problems in Europe. London, Centre for Economic Policy Research Press. (VoxEU.org eBook.)
- Düchs, G. and Ingold, G.-L. 2016. Gut geparkt ist noch nicht studiert: Statistiken zum Physikstudium an den Universitäten in Deutschland 2016 [Well parked, you haven't studied yet: statistics on physics studies at universities in Germany in 2016 ]. *Physik Journal*, Vol. 15, No. 8/9, pp. 28–33.
- Elliott, W. and Nam, I. 2013. Is student debt jeopardizing the short-term financial health of U.S. households? Federal Reserve Bank of St. Louis Review, Vol. 95, No. 5, pp. 405–24.
- ETF. 2020. Armenia: Education, Training and Employment Developments 2020. Turin, Italy, European Training Foundation. (Country Fiche.)
- Eurofound. 2020. Addressing Household Over-indebtedness. Luxembourg, European Foundation for the Improvement of Living and Working Conditions.
- Eurostat. 2021a. Enterprises providing training by type of training and size class. Luxembourg, Eurostat. https:// ec.europa.eu/eurostat/databrowser/view/TRNG\_CVT\_01S/bookmark/table?lang=en&bookmarkId=61c7c617-8f8e-4961-8c51-e234183efdc5. (Accessed 17 August 2021.)
- \_\_\_\_. 2021b. Participation rate in education and training by labour status. Luxembourg, Eurostat. https://ec.europa.eu/ eurostat/databrowser/view/TRNG\_AES\_103/bookmark/table?lang=en&bookmarkId=ea8df588-79ce-4ad0-9fbe-716548d6945f. (Accessed 17 August 2021.)
- Farnell, T., Skledar Matijević, A. and Šćukanec Schmidt, N. 2021. The Impact of COVID-19 on Higher Education: A Review of Emerging Evidence An Analytical Report. Brussels, European Commisison. (NESET Report.)
- Federal Reserve. 2020. Student Loans Owned and Securitized. St. Louis, Mo., Federal Reserve Economic Data. https://fred.stlouisfed.org/series/SLOAS. (Accessed 16 September 2021.)
- Field, S. and Guez, A. 2018. Pathways of Progression: Linking Technical and Vocational Education and Training with Post-secondary Education. Paris, UNESCO.
- Filizola, P. 2019. Fies 2019: esvaziado após reformulação, financiamento atrai menos alunos e reduz opções para mais pobres [Fies 2019: emptied after reformulation, funding attracts fewer students and reduces options for the poorest]. BBC News Brasil, 6 February. www.bbc.com/portuguese/brasil-46968457. (Accessed 3 September 2021.)
- Fletcher, C., Webster, J. and Di, W. 2020. PLUS Borrowing in Texas: Repayment Expectations, Experience, and Hindsight by Minority-Serving Institution Status. San Antonio, Tex., Trellis Company.
- FNDE. 2020. Pedido de Informação [Information request]. Brasília, Fundo Nacional de Desenvolvimento da Educação. (Unpublished.)
- Furuta, K. 2021. Parental perceptions of university cost, fear of debt, and choice of high school in Japan. British Journal of Sociology of Education, Vol. 42, No. 5–6, pp. 667–85.
- Gartenschlaeger, U. 2017. Introduction: Community learning centres as spaces for lifelong learning in the 21st Century. Avramovska, M., Hirsch, E. and Schmidt-Behlau, B. (eds), Adult Education Centres as a Key to Development: Challenges and Success Factors. Bonn, Germany, DVV International, pp. 8–18. (International Perspectives in Adult Education.)
- Govinda, R. 2017. CONFINTEA VI Mid-Term Review 2017: The Status of Adult Learning and Education in Asia and the Pacific. Hamburg, Germany, UNESCO Institute for Lifelong Learning.
- Guzmán, J. 2011. Educación Técnica y Formación Profesional en Costa Rica: Avances y Desafíos [Technical Education and Professional Training in Costa Rica; Progress and Challenges]. San José, Consejo Nacional de Rectores. (Informe Estado de la Educación.)
- Hayashi, R., Sudarshana, H. D., Jayasundara, A., Garcia, M., Balasuriya, A. and Hirokawa, T. 2021. COVID-19 Impact on Technical and Vocational Education and Training in Sri Lanka. Manila, Asian Development Bank. (ADB Brief 168.)
- Henehan, K. 2020. Can Training Help Workers Change Their Stripes? Retraining and Career Change in the UK. London, Resolution Foundation.
- Hoftijzer, M., Levin, V., Santos, I. and Weber, M. 2020. TVET Systems' Response to COVID-19: Challenges and Opportunities Washington, DC, World Bank.

Hoftijzer, M., Levin, V. and Weber, M. 2021. COVID-19 Highlights the Urgency of TVET Reforms. Washington, DC, World Bank. https://blogs.worldbank.org/education/covid-19-highlights-urgency-tvet-reforms. (Accessed 17 August 2021.)

ICETEX. 2020. Solicitud de Información [Information request]. Bogota, ICETEX. (Unpublished.)

IESALC. 2021. Closing Now to Reopen Better Tomorrow: Pedagogical Continuity in Latin American Universities during the Pandemic. Caracas, UNESCO International Institute for Higher Education in Latin America and the Caribbean.

- ILO. 1974. Paid Educational Leave Convention. Geneva, Switzerland, International Labour Organization. (Convention 140.) \_\_\_\_\_. 2019. State of Skills: Uruguay. Geneva, Switzerland, International Labour Organization.
- \_\_\_\_. 2020. Global Employment Trends for Youth 2020: Technology and the Future of Jobs. Geneva, Switzerland, International Labour Organization.
- ILO, UNESCO and World Bank. 2020. ILO-UNESCO-WBG Joint Survey on Technical and Vocational Education and Training (TVET) and Skills Development during the Time of COVID-19. Geneva, Switzerland/Paris/Washington, DC, International Labour Organization/UNESCO/World Bank.
- ILO/Cinterfor. 2020. Cinterfor 2020 Report: A Year of Challenges and Lessons Learned. Montevideo, ILO Inter-American Centre for Knowledge Development in Vocational Training.
- INEEd. 2014. Informe sobre el Estado de la Educación en Uruguay [Report on the State of Education in Uruguay]. Montevideo, Instituto Nacional de Evaluación Educativa.
- INEP. 2018. Microdados do Censo da Educação Superior [Microdata of the Higher Education Census]. Brasília, Instituto Nacional de Estudos e Pesquisas Educacionais Anísio Teixeira. www.gov.br/inep/pt-br/acesso-a-informacao/ dados-abertos/microdados/censo-da-educacao-superior. (Accessed 9 November 2020.)
- Ingresa. 2019. *Cuenta Pública Año 2019 [Public Account 2019]*. Santiago, Higher Education Credit System. \_\_\_\_\_. 2020. Solicitud de Información [Information request]. Santiago, Higher Education Credit System. (Unpublished.)
- IPEA. 2019. Ipea promove debate sobre financiamento do ensino superior [IPEA promotes debate on higher education financing]. Brasília, Instituto de Pesquisa Econômica Aplicada. www.ipea.gov.br/portal/index.php?option=com\_ content&view=article&id=34875. (Accessed 16 September 2020.)
- Jørgensen, C. H. 2015. Recent Innovations in VET in Denmark: Responses to Key Challenges for VET. Roskilde, Denmark, Nord-VET.
- Kalenkoski, C. M. and Pabilonia, S. W. 2010. Parental transfers, student achievement, and the labor supply of college students. *Journal of Population Economics*, Vol. 23, No. 2, pp. 469–96.
- Kamaté, A. and Siahoué, A. 2020. Africa: How Can Technical and Vocational Education Contribute to the Response to Future Pandemics? Abidjan, Association for the Development of Education in Africa. www.adeanet.org/en/blogs/ africa-how-can-technical-vocational-education-contribute-response-future-pandemics. (Accessed 17 August 2021.)
- Kauhanen, A. 2021. The effects of an education-leave program on educational attainment and labor-market outcomes. Education Economics, Vol. 29, No. 6, pp. 651–69.
- Kolodner, M. 2021. Public colleges shock students by sending them to costly debt collection agencies. *The Hechinger Report*, 1 June. https://hechingerreport.org/public-colleges-shock-students-by-sending-them-to-costly-debtcollection-agencies. (Accessed 30 August 2021.)
- Kovacevic, M. and Jahic, A. 2020. COVID-19 and Human Development: Exploring Global Preparedness and Vulnerability. New York, United Nations Development Programme.
- Kwon, R., Zhang, M. L. and VandenBussche, C. J. 2020. Considerations for remote learning in pathology during COVID-19 social distancing. *Cancer Cytopathology*, Vol. 128, No. 9, pp. 642–47.
- Lassoued, Z., Alhendawi, M. and Bashitialshaaer, R. 2020. An exploratory study of the obstacles for achieving quality in distance learning during the COVID-19 pandemic. *Education Sciences*, Vol. 10, No. 9, art. 232.
- Latham, A. 2020. 2020 Student Loan Industry Study. Santa Ana, Calif., Supermoney. www.supermoney.com/studies/ student-loan-industry-study. (Accessed September 4 2021.)
- Lowrey, A. 2020. Go ahead, forgive student debt. *The Atlantic*, 21 November. www.theatlantic.com/ideas/ archive/2020/11/why-biden-should-forgive-student-loan-debt/617171. (Accessed 30 August 2021.)
- Lucca, D. O., Nadauld, T. and Shen, K. 2017. Credit Supply and the Rise in College Tuition: Evidence from the Expansion in Federal Student Aid Programs. New York, Federal Reserve Bank of New York. (Staff Report 733.)

- Lukyanova, L. and Veramejchyk, G. 2017. Transformation of adult education institutions in Eastern Europe: Experiences of Belarus, Ukraine and Moldova. Avramovska, M., Hirsch, E. and Schmidt-Behlau, B. (eds), Adult Education Centres as a Key to Development: Challenges and Success Factors. Bonn, Germany, DVV International, pp. 66–83 (International Perspectives in Adult Education.)
- Madge, C., Breines, M. R., Dalu, M. T. B., Gunter, A., Mittelmeier, J., Prinsloo, P. and Raghuram, P. 2019. WhatsApp use among African international distance education (IDE) students: transferring, translating and transforming educational experiences. *Learning, Media and Technology*, Vol. 44, No. 3, pp. 267–82.
- Mahdy, M. A. 2020. The impact of COVID-19 pandemic on the academic performance of veterinary medical students. Frontiers in Veterinary Science, Vol. 7.
- Mawazo Institute. 2020. Survey Report: The Impact of COVID-19 on Africa's Higher Education System. Nairobi, Mawazo Institute.
- Mezza, A., Ringo, D. and Sommer, K. 2019. Can Student Loan Debt Explain Low Homeownership Rates for Young Adults? Washington, DC, Board of Governors of the Federal Reserve System. (Consumer and Community Context.)
- Musset, P., Field, S., Mann, A. and Bergseng, B. 2019. Vocational Education and Training in Estonia. Paris, Organisation for Economic Co-operation and Development. (Reviews of Vocational Education and Training.)
- National Student Clearinghouse Research Center. 2021. National Student Clearinghouse Research Center's regular updates on higher education enrollment: October 26, 2021. Herndon, Va., National Student Clearinghouse Research Center. (Accessed 8 November 2021.)
- NCES. 2020. Digest of Education Statistics: Table 331.20. Washington, DC, National Center for Education Statistics. https://nces.ed.gov/programs/digest/d20/tables/dt20\_331.20.asp. (Accessed 30 August 2021.)
- Nepal Ministry of Education and UNESCO. 2017. Reading the Past, Writing the Future: A Report on National Literacy Campaign and Literate Nepal Mission. Kathmandu, Ministry of Education, Non-Formal Education Centre/UNESCO.
- NILE and UIL. 2016. Synthesis Report on the State of Community Learning Centres in Six Asian Countries: Bangladesh, Indonesia, Mongolia, Republic of Korea, Thailand and Viet Nam. Hamburg, Germany, National Institute for Lifelong Education/UNESCO Institute for Lifelong Learning.
- O'Reilly, N. 2021. What Is Happening in Adult Learning and Education (ALE)? Dublin, AONTAS.
- OECD. 2015. Brazil Policy Brief: Education and Skills. Developing Skills and Education for Growth. Paris, Organisation for Economic Co-operation and Development. (OECD Better Policies Series.)
- \_\_\_\_. 2017. Education in Costa Rica. Paris, Organisation for Economic Co-operation and Development. (Reviews of National Policies for Education.)
- \_\_\_\_. 2019. Education at a Glance 2019: OECD Indicators. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2020a. Effective Adult Learning Policies: Challenges and Solutions for Latin American Countries. Paris, Organisation for Economic Co-operation and Development. (Skills Studies.)
- \_\_\_\_. 2020b. The Potential of Online Learning for Adults: Early Lessons from the COVID-19 Crisis. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2020c. VET in a Time of Crisis: Building Foundations for Resilient Vocational Education and Training Systems. Paris, Organisation for Economic Co-operation and Development. www.oecd.org/coronavirus/policy-responses/ vet-in-a-time-of-crisis-building-foundations-for-resilient-vocational-education-and-training-systemsefff194c. (Accessed 17 August 2021.)
- \_\_\_\_. 2021a. Implications of the COVID-19 Pandemic for Vocational Education and Training. Paris, Organisation for Economic Co-operation and Development.
- \_\_\_\_. 2021b. The State of Higher Education One Year into the COVID-19 Pandemic. Paris, Organisation for Economic Co-operation and Development.
- OECD and Asian Development Bank. 2014. Education in Indonesia: Rising to the Challenge. Paris, Organisation for Economic Co-operation and Development.
- Oviatt, D. P., Baumann, M. R., Bennett, J. M. and Garza, R. T. 2017. Undesirable effects of working while in college: work-school conflict, substance use, and health. *Journal of Psychology*, Vol. 151, No. 5, pp. 433–52.
- Paciorek, A., Manca, F. and Borgonovi, F. 2021. Adult Learning and COVID-19: How Much Informal and Non-formal Learning Are Workers Missing? Paris, Organisation for Economic Co-operation and Development.
- Parliamentary Monitoring Group. 2020. NSFAS, CHE & SAQA 2020/21 Annual Performance Plans. Cape Town, South Africa, Parliamentary Monitoring Group. (Annual Performance Plan of Government Departments and Entities 20/2021.)

Pham, H.-H. and Ho, T.-T.-H. 2020. Toward a 'new normal'with e-learning in Vietnamese higher education during the post COVID-19 pandemic. *Higher Education Research and Development*, Vol. 39, No. 7, pp. 1327–31.

- Pham Le, A. T. 2018. The contributions of community learning centres (CLCs) to personal and community development in Myanmar. International Review of Education, Vol. 64, pp. 607–31.
- Pinheiro, L. 2021. Enem 2021 tem 4 milhões de inscritos, menor número desde 2007 [ENEM 2021 has 4 million applicants, the lowest number since 2007]. G1 Globo, 15 July. https://g1.globo.com/educacao/enem/2021/noticia/2021/07/15/enem-2021-tem-4-milhoes-de-inscritos.ghtml. (Accessed 17 August 2021.)
- Portela, M. J. O. and Gebremedhin, A. 2020. Prospective Evaluation of GPE's Country-level Support to Education: Country Level Evaluation – Ethiopia. Montreal, Que., Washington, DC/Hove, UK, Universalia/Results for Development/Itad.
- Pulkkinen, L. 2021. 'Millions upon millions' in employer-funded education benefits go unused. The Hechinger Report, 21 June. https://hechingerreport.org/millions-upon-millions-in-employer-funded-education-benefits-go-unused. (Accessed 17 August 2021.)
- Rambla, X., Castioni, R. and Sepúlveda, L. 2020. The making of TVET systems in middle-income countries: insights on Brazil and Chile. Journal of Education and Work, Vol. 33, No. 1, pp. 67–80.
- Rogers, A. 2019. Second-generation non-formal education and the sustainable development goals: operationalising the SDGs through community learning centres. International Journal of Lifelong Education, Vol. 38, No. 5, pp. 515–26.
- Rutledge, M. S., Sanzenbacherand, G. T. and Vitagliano, F. M. 2018. Do Young Adults with Student Debt Save Less for Retirement? Boston, Mass., Center for Retirement Research at Boston College. (Issue Brief 18-13.)
- Sabzalieva, E. 2020. Responding to Major Institutional Change: The Fall of the Soviet Union and Higher Education in Central Asia. PhD thesis, Toronto, Ont., Ontario Institute for Studies in Education. (Unpublished.)
- Sadro, F., Marsh, L., Klenk, H., Egglestone, C. and Friel, S. 2021. Understanding the Difference You Make: An Evaluation Toolkit for Adult Learning and Career Change Programmes. London, Learning and Work Institute. https:// learningandwork.org.uk/resources/research-and-reports/understanding-the-difference-you-make-anevaluation-toolkit-for-adult-learning-and-career-change-programmes. (Accessed 17 August 2021.)
- Salmi, J. 2020. COVID's Lessons for Global Higher Education: Coping with the Present while Building a More Equitable Future. Indianapolis, Ind., Lumina Foundation.
- Samara, M. 2021. Towards e-learning in TVET: setting and developing an e-competence framework (E-CF) for TVET teachers in Palestine. TVET@Asia, No. 16.
- Scheepers, J. and Gebhardt, J. 2021. Teaching practices and early leaving from vocational education and training: an empirical approach and intervention project. Nägele, C., Kersh, N. and Stalder, B. E. (eds.), Trends in Vocational Education and Training Research, Vol. IV: Proceedings of the European Conference on Educational Research. Bremen, Germany European Research Network of Vocational Education and Training.
- Setiawan, A. and Hamdani, A. 2021. Current development of vocational professional teacher education in Indonesia. TVET@Asia, No. 16.
- Singal, A., Bansal, A., Chaudhary, P., Singh, H. and Patra, A. 2021. Anatomy education of medical and dental students during COVID-19 pandemic: a reality check. *Surgical and Radiologic Anatomy*, Vol. 43, No. 4, pp. 515–21.
- Singh, J., Matthees, B. and Odetunde, A. 2021. Leaning online education during COVID-19 pandemic–attitudes and perceptions of non-traditional adult learners. *Quality Assurance in Education*, Vol. 29, No. 4, pp. 408-21.
- Skills for Health. 2020. Skills for Health and NHS professionals supporting the urgent training needs of front-line staff. *Further Education News*, 9 April. www.fenews.co.uk/press-releases/45307-skills-for-health-and-nhs-professionals-supporting-the-urgent-training-needs-of-front-line-staff. (Accessed 17 August 2021.)
- Sun, L., Tang, Y. and Zuo, W. 2020. Coronavirus pushes education online. Nature Materials, Vol. 19, No. 6, pp. 687–87.
- Triyono, M. B. and Moses, K. M. 2019. Technical and vocational education and training and training in Indonesia. Bai, B. and Paryono (eds), Vocational Education and Training in ASEAN Member States: Current Status and Future Development. Singapore, Springer, pp. 45–79. (Perspectives on Rethinking and Reforming Education.)
- Turek, K., Kalmijn, M. and Leopold, T. 2021. The Comparative Panel File: harmonized household panel surveys from seven countries. *European Sociological Review*, Vol. 37, No. 3, pp. 505–23.
- UIL. 2010. Belem Framework for Action: Harnessing the Power and Potential of Adult Learning and Education for a Viable Future. Hamburg, Germany, UNESCO Institute for Lifelong Learning.
- \_\_\_\_. 2020a. UNESCO Learning Cities' Responses to COVID-19: Outcomes of Webinar on 9 April Latin America and the Caribbean. Hamburg, UNESCO Institute for Lifelong Learning. https://uil.unesco.org/unesco-learning-citiesresponses-covid-19-outcomes-webinar-9-april-latin-america-and-caribbean. (Accessed 17 August 2021.)
- \_\_\_\_. 2020b. Adult Learning and Education and COVID-19. Hamburg, Germany, UNESCO Institute for Lifelong Learning. (UNESCO COVID-19 Education Response: Education Sector Issue Note 2.6.)

- UNESCO-UNEVOC and National Council for the Federal Network of Vocational Institutions. 2018. TVET Country Profile: Brazil. Bonn, Germany, UNESCO-UNEVOC International Centre for Technical and Vocational Education and Training.
- UNESCO. 2015. Incheon Declaration and Framework for Action for the Implementation of Sustainable Development Goal 4. Paris, UNESCO.
- 2016. Community-Based Lifelong Learning and Adult Education: Role of Community Learning Centres as Facilitators of Lifelong Learning. Paris/Bangkok, UNESCO.
- \_\_\_\_. 2020b. How Cities Are Utilizing the Power of Non-formal and Informal Learning to Respond to the COVID-19 Crisis. Paris, UNESCO. (UNESCO COVID-19 Education Response: Issue Note 6.2.)
- \_\_\_\_. 2021. COVID-19: Reopening and Reimagining Universities: Survey on Higher Education through the UNESCO National Commissions. Paris, UNESCO.
- Woldegiorgis, E. T. 2008. Cost Sharing in Ethiopia: An Analysis of Challenges and Prospects. Thesis for European master in higher education, Universidade de Aveiro. (Unpublished.)
- Woodhall, M. 2007. Funding Higher Education: The Contribution of Economic Thinking to Debate and Policy Development. Washington, DC, World Bank. (Education Working Paper 8.)
- World Bank. 2014. Republic of Burundi: Skills Development for Growth. Washington, DC, World Bank.
- Wright, E. and Mulvey, B. 2021. Internships and the graduate labour market: how upper-middle-class students 'get ahead'. British Journal of Sociology of Education, Vol. 42, No. 3, pp. 339–56.
- Yizengaw, T. 2007. Implementation of cost sharing in the Ethiopian higher education landscape: critical assessment and the way forward. *Higher Education Quarterly*, Vol. 61, No. 2, pp. 171–96.
- Ziderman, A. 2017. Experience with student loans, higher education. Shin, J. C. and Teixeira, P. (eds), Encyclopedia of International Higher Education Systems and Institutions. Dordrecht, Netherlands, Springer.

- Azubuike, O. B. 2021. Education, digital skills acquisition and learning during COVID-19 in Nigeria. Johannesburg, South Africa, Africa Portal. www.africaportal.org/features/education-digital-skills-acquisition-and-learning-during-covid-19-nigeria. (Accessed 22 October 2021.)
- Binda, T. L. 2020. New Voices in Africa: COVID-19 as a Catalyst for Digital Transformation. Paris, Institut Montaigne. www.institutmontaigne.org/en/blog/new-voices-africa-covid-19-catalyst-digital-transformation. (Accessed 22 October 2021.)
- BusinessTech. 2020. These are the skills government wants South African schools to cover. BusinessTech, 23 September. https://businesstech.co.za/news/technology/435649/these-are-the-skills-government-wantssouth-african-schools-to-cover. (Accessed 25 October 2021.)
- Calderon, G., Cunha, J. M. and Giorgi, G. D. 2018. Business literacy and development: evidence from a randomized controlled trial in rural Mexico. *Economic Development and Cultural Change*, Vol. 68, No. 2, pp. 507–40.
- Corneille, E. 2020. Mastercard and USAID Partner under the W-GDP to Launch Project Kirana. London, IBS Intelligence. https://ibsintelligence.com/ibsi-news/mastercard-and-usaid-partner-under-the-w-gdp-to-launch-projectkirana. (Accessed 25 October 2021.)
- DeJaeghere, J., Morris, E. and Bamattre, R. 2020. Moving beyond employment and earnings: reframing how youth livelihoods and wellbeing are evaluated in East Africa. *Journal of Youth Studies*, Vol. 23, No. 5, pp. 667–85.
- Doroba, H., Mbanugo, T. and Edosio, U. 2020. The Relevance of Digital Skills in the COVID-19 Era. Abidjan, African Development Bank. www.afdb.org/fr/news-and-events/relevance-digital-skills-covid-19-era-36244. (Accessed 22 October 2021.)
- European Commission. 2021. Digital Education Action Plan (2021-2027): Resetting Education and Training for the Digital Age. Brussels, European Commission. https://ec.europa.eu/education/education-in-the-eu/digital-educationaction-plan\_en. (Accessed 25 October 2021.)
- Fraillon, J., Ainley, J., Schulz, W., Friedman, T. and Duckworth, D. 2019. Preparing for Life in a Digital World: IEA International Computer and Information Literacy Study 2018 International Report. Amsterdam, International Association for the Evaluation of Educational Achievement.
- Fundación Chile. 2020. 1,200 New Openings for Cost-free Courses Are Made Available to Satisfy the High Demand for Workers in the Digital Area. Santiago, Fundación Chile. https://fch.cl/en/news/1200-new-openings-for-cost-freecourses-are-made-available-to-satisfy-the-high-demand-for-workers-in-the-digital-area. (Accessed 22 October 2021.)

Howland, K. and Good, J. 2015. Learning to communicate computationally with Flip: a bi-modal programming language for game creation. Computers and Education, Vol. 80, pp. 224–40.

- ILO. 2020. Digital Skills and the Future of Work: Challenges and Opportunities in a Post COVID-19 Environment. Geneva, Switzerland, International Labour Organization.
- Jackman, J. A., Gentile, D. A., Cho, N.-J. and Park, Y. 2021. Addressing the digital skills gap for future education. *Nature Human Behaviour*, Vol. 5, No. 5, pp. 542–45.
- James, J. 2021. Confronting the scarcity of digital skills among the poor in developing countries. Development Policy Review, Vol. 39, No. 2, pp. 324–39.
- Law, N., Woo, D., de la Torre, J. and Wong, G. 2018. A Global Framework of Reference on Digital Literacy Skills for Indicator 4.4.2. Montreal, Que., UNESCO Institute for Statistics. (Information Paper 51.)
- Lee, N., Beeler Stücklin, S., Lopez Rodriguez, P., El Alaoui Faris, M. and Mukaka, I. 2020. Financial education for HIV-vulnerable youth, orphans, and vulnerable children: a systematic review of outcome evidence. *Campbell Systematic Reviews*, Vol. 16, No. 1, art. 1071.
- Martínez-Alcalá, C. I., Rosales-Lagarde, A., Pérez-Pérez, Y. M., Lopez-Noguerola, J. S., Bautista-Díaz, M. L. and Agis-Juarez, R. A. 2021. The effects of COVID-19 on the digital literacy of the elderly: norms for digital inclusion. *Frontiers in Education*, Vol. 6, art. 716025.
- Monteiro, L. 2021. Ecobank Group Partners with Microsoft to Digitally Upskill Africa's SMEs. London, IBS Intelligence. https://ibsintelligence.com/ibsi-news/ecobank-group-partners-with-microsoft-to-digitally-upskill-africassmes. (Accessed 25 October 2021.)
- Mouza, C., Yang, H., Pan, Y.-C., Ozden, S. Y. and Pollock, L. 2017. Resetting educational technology coursework for pre-service teachers: a computational thinking approach to the development of technological pedagogical content knowledge (TPACK). Australasian Journal of Educational Technology, Vol. 33, No. 3, pp. 61–76.
- OECD. 2020. PISA 2018 Results (Volume IV): Are Students Smart About Money? Paris, Organisation for Economic Co-operation and Development.
- Reyes, A., Roseth, B. and Vera-Crossio, D. 2021. Technology, Identification, and Access to Social Programs: Experimental Evidence from Panama. Washington, DC, Inter-American Development Bank. (IDB Working Paper 1261.)
- Román-González, M., Pérez-González, J.-C. and Jiménez-Fernández, C. 2017. Which cognitive abilities underlie computational thinking? Criterion validity of the Computational Thinking Test. Computers in Human Behavior, Vol. 72, pp. 678–91.
- Seifert, A. 2020. The digital exclusion of older adults during the COVID-19 pandemic. Journal of Gerontological Social Work, Vol. 63, No. 6–7, pp. 674–76.
- Seow, P., Looi, C.-K., How, M.-L., Wadhwa, B. and Wu, L.-K. 2019. Educational policy and implementation of computational thinking and programming: case study of Singapore. Kong, S.-C. and Abelson, H. (eds), Computational Thinking Education. Singapore, Springer, pp. 345–61.
- Smith, B. 2021. Building a more inclusive skills-based economy: the next steps for our global skills initiative. *Official Microsoft Blog*, 20 March. https://blogs.microsoft.com/blog/2021/03/30/building-a-more-inclusive-skillsbased-economy-the-next-steps-for-our-global-skills-initiative. (Accessed 22 October 2021.)
- Voogt, J., Fisser, P., Good, J., Mishra, P. and Yadav, A. 2015. Computational thinking in compulsory education: towards an agenda for research and practice. *Education and Information Technologies*, Vol. 20, No. 4, pp. 715–28.
- World Economic Forum. 2020. The Future of Jobs Report 2020. Geneva, Switzerland, World Economic Forum.
- Yadav, A., Mayfield, C., Zhou, N., Hambrusch, S. and Korb, J. 2014. Computational thinking in elementary and secondary teacher education. ACM Transactions on Computing Education, Vol. 14, No. 1, art. 5.

- Adamson, L. 2020. Language of instruction: a question of disconnected capabilities. *Comparative Education*, Vol. 57, No. 2, pp. 187–205.
- Aide et Action Southeast Asia. 2021. India: COVID-19 Subjecting Children of Seasonal Migrants to Child Labour. Phnom Penh, Aide et Action. https://seac.aide-et-action.org/india-covid-19-subjecting-children-of-seasonal-migrants-tochild-labour. (Accessed 17 August 2021.)
- Amiri, R. and Jackson, A. 2021. Taliban Attitudes and Policies towards Education. London, Centre for the Study of Armed Groups, Overseas Development Institute. (Working Paper 601.)

Arjona, A. 2015. Civilian resistance to rebel governance. Arjona, A., Kafsir, N. and Mampilly, Z. (eds), *Rebel Governance in Civil War*. Cambridge, UK, Cambridge University Press, pp. 180–202.

- Australian Humanitarian Partnership. 2021. Delivering COVID-safe Education to Rohingya Camps in Bangladesh. Cox's Bazar, Bangladesh, Australian Humanitarian Partnership.
- Bannink-Mbazzi, F., Nalugya, R., Kawesa, E., Nimusiima, C., King, R., Hove, G. v. and Seeley, J. 2021. The impact of COVID-19 measures on children with disabilities and their families in Uganda. *Disability and Society*.
- Bhattacharya, L. and Chandrasekhar, S. 2020. India's Search for Link Language and Progress towards Bilingualism. Mumbai, India, Indira Gandhi Institute of Development Research. (Working Paper 2020-015.)
- Bol, T. 2020. Inequality in homeschooling during the Corona crisis in the Netherlands: first results from the LISS Panel. SocArXiv.
- Branch, A. and Mampilly, Z. C. 2005. Winning the war, but losing the peace? The dilemma of SPLM/A civil administration and the tasks ahead. *Journal of Modern African Studies*, Vol. 43, No. 1, pp. 1–20.
- Carter, E., Sabates, R., Rose, P. and Akyeampong, K. 2020. Sustaining literacy from mother tongue instruction in complementary education into official language of instruction in government schools in Ghana. *International Journal of Educational Development*, Vol. 76, art. 102195.
- Deng, L. B. 2003. Education in Southern Sudan: War, Status and Challenges of Achieving Education for All Goals. Paris, UNESCO. (Background paper for Education for All Global Monitoring Report 2003/4.)
- Eurostat. 2014. Child specific material deprivation rate by age (children aged 1 to 15). Luxembourg, Eurostat. https://ec.europa.eu/eurostat/databrowser/view/ilc\_chmd01/default/table?lang=en. (Accessed 17 August 2021.)
- Feroz, E. and Lakanwal, A. R. 2020. In rural Afghanistan, some Taliban gingerly welcome girls schools. *Foreign Policy*, 4 May. https://foreignpolicy.com/2020/05/04/afghanistan-taliban-girls-schools. (Accessed 17 August 2021.)
- Ford, K. 2021. Supporting Vulnerable Adolescent Girls to Continue Their Education Should Be Prioritised in Recovery Plans for Covid-19. Paris, UNESCO. https://gemreportunesco.wpcomstaging.com/2021/06/18/supporting-vulnerableadolescent-girls-to-continue-their-education-should-be-prioritised-in-recovery-plans-for-covid-19. (Accessed 17 August 2021.)
- Giustozzi, A. and Franco, C. 2011. The Battle for the Schools: The Taleban and State Education. Berlin, Afghanistan Analysts Network. (Thematic Report 08/2011.)
- \_\_\_\_. 2013. The Ongoing Battle for the Schools: Uprisings, Negotiations and Taleban Tactics. Berlin, Afghanistan Analysts Network. (Briefing Paper 02/2013.)
- Guio, A.-C., Gordon, D., Marlier, E., Najera, H. and Pomati, M. 2018. Towards an EU measure of child deprivation. Child Indicators Research, Vol. 11, No. 3, pp. 835–60.
- Halloran, C., Jack, R., Okun, J. C. and Oster, E. 2021. Pandemic Schooling Mode and Student Test Scores: Evidence from US States. Cambridge, Mass., National Bureau of Economic Research. (Working Paper 29497.)
- Hannum, E., Liu, R. and Alvarado-Urbina, A. 2017. Evolving approaches to the study of childhood poverty and education. Comparative Education, Vol. 53, No. 1, pp. 81–114.
- Hattori, H., Cardoso, M. and Ledoux, B. 2017. Collecting Data on Foundational Learning Skills and Parental Involvement in Education. New York, UNICEF. (MICS Methodological Paper 5.)
- Hiraal Institute. 2018. The Fighters Factory: Inside Al-Shabab's Education System. Mogadishu, Hiraal Institute.
- Huang, R. and Sullivan, P. L. 2020. Arms for education? External support and rebel social services. *Journal of Peace Research*, Vol. 58, No. 4, pp. 794–808.
- Human Rights Watch. 2021. 'Years Don't Wait for Them': Increased Inequalities in Children's Right to Education Due to the Covid-19 Pandemic. New York, Human Rights Watch. www.hrw.org/report/2021/05/17/years-dont-wait-them/ increased-inequalities-childrens-right-education-due-covid. (Accessed 17 August 2021.)
- Inter-Agency Coordination Lebanon, UNICEF, UNHCR and WFP. 2021. Vulnerability Assessment of Syrian Refugees in Lebanon: VASyR 2021 Preliminary Findings. Beirut, Inter-Agency Coordination Lebanon/UNICEF/United Nations High Commissioner for Refugees/World Food Programme.
- Jackson, A. 2018. Life under the Taliban Shadow Government. London, Overseas Development Institute.
- Jackson, A. and Amiri, R. 2019. Insurgent Bureaucracy: How the Taliban Makes Policy. Washington, DC, US Institute of Peace. (Peaceworks 153.)
- Kasfir, N. 2005. Guerrillas and civilian participation: the National Resistance Army in Uganda, 1981–86. Journal of Modern African Studies, Vol. 43, No. 2, pp. 271–96.
- Keister, L. 2018. Income and Wealth Are not Highly Correlated: Here Is Why and What It Means. Washington, DC, American Sociological Association. www.wipsociology.org/2018/10/29/income-and-wealth-are-not-highly-correlatedhere-is-why-and-what-it-means. (Accessed 3 March 2021.)

Kiribati Government. 2016. Kiribati Development Plan 2016–19. Tarawa, Government of Kiribati.

Mampilly, Z. C. 2011. Rebel Rulers: Insurgent Governance and Civilian Life during War. Ithaca, NY, Cornell University Press.

- \_\_\_\_. 2021. Counting School Children Under Rebel Control: Four Case Studies. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Myanmar Ministry of Education. 2020. Myanmar COVID-19 National Response and Recovery Plan for the Education Sector. Naypyidaw, Myanmar, Ministry of Education.
- NRC. 2005. Profile of Internal Displacement: Sri Lanka. Geneva, Switzerland, Global IDP Project, Norwegian Refugee Council.
- Okasheh, H. 2020. Reversing Gains: Brief on the Impact of COVID-19 on Education in Syria. London, Save the Children.
- Orsander, M. and Mendoza, P. 2020. The Hidden Imapct of COVID-19 on Children and Families with Disabilities. London, Save the Children.
- Outhwaite, L. A., Gulliford, A. and Pitchford, N. J. 2020. Language counts when learning mathematics with interactive apps. *British Journal of Educational Technology*, Vol. 51, No. 6, pp. 2326–39.
- Palmer, A. 2020. Rebel schools: the strategic timing of education provision in Afghanistan. New York, Wilf Family Department of Politics, New York University. (Unpublished.)
- Pier, L., Christian, M., Tymeson, H. and Meyer, R. H. 2021. COVID-19 Impacts on Student Learning: Evidence from Interim Assessments in California. Stanford, Calif., Policy Analysis for California Education.
- Schaeffer, K. 2021. What We Know about Online Learning and the Homework Gap amid the Pandemic. Washington, DC, Pew Research Center. www.pewresearch.org/fact-tank/2021/10/01/what-we-know-about-online-learning-andthe-homework-gap-amid-the-pandemic. (Accessed 1 November 2021.)
- Selvik, K. 2021. Education activism in the Syrian civil war: resisting by persisting. Comparative Education Review, Vol. 65, No. 3.
- SIL International. 2021. Ethnologue: Languages of the World. Dallas, Tex., SIL International. www.ethnologue.com. (Accessed 26 April 2021.)
- Suffian, M. 2021. Closure of seasonal hostels forces migrant children to work as daily wage laborers, education takes backstage. *India Today*, 26 July. www.indiatoday.in/education-today/news/story/closure-of-seasonal-hostels-migrant-children-work-as-daily-wage-laborers-1832875-2021-07-26. (Accessed 17 August 2021.)
- Sutton, H. 2020. COVID-19 disproportionately impacts students with disabilities across all sectors. Disability Compliance for Higher Education, Vol. 26, No. 6, p. 9.
- Townsend, R., Chmielewska, B., Barratt, I., Kalafat, E., van der Meulen, J., Gurol-Urganci, I., O'Brien, P., Morris, E., Draycott, T., Thangaratinam, S., Le Doare, K., Ladhani, S., von Dadelszen, P., Magee, L. A. and Khalil, A. 2021. Global changes in maternity care provision during the COVID-19 pandemic: a systematic review and metaanalysis. *EClinicalMedicine*, Vol. 37, art. 100947.
- UIS. 2018. SDG Indicator 4.5.2: Recommended Approaches for Measurement. Montreal, Que., Technical Cooperation Group for SDG 4 Indicators, UNESCO Institute for Statistics. (TCG5/REF/5.)
- UNESCO. 2021. When Schools Shut: Gendered Impacts of COVID-19 School Closures. Paris, UNESCO.
- UNESCO, UNICEF, World Bank and OECD. 2021. What's Next? Lessons on Education Recovery: Findings from a Survey of Ministries of Education amid the COVID-19 Pandemic. Paris/Montreal, Que.,/New York/Florence/Washington, DC, UNESCO, UNESCO Institute for Statistics/UNICEF/UNICEF Office of Research – Innocenti/World Bank/OECD.
- UNICEF. 2020a. Community-based Education Will Reach up to 140,000 Hardest to Reach Children and Those Living in Conflict Zones. Kabul, UNICEF. www.unicef.org/afghanistan/press-releases/community-based-education-will-reach-140000-hardest-reach-children-and-those-living. (Accessed 17 August 2021.)
- \_\_\_\_. 2020b. COVID-19: Are Children Able to Continue Learning During School Closures? A Global Analysis of the Potential Reach of Remote Learning Policies. New York, UNICEF.
- \_\_\_\_. 2021. Ensuring Equal Access to Education in Future Crises: Findings of the New Remote Learning Readiness Index. New York, UNICEF.
- Viero, A., Barbara, G., Montisci, M., Kustermann, K. and Cattaneo, C. 2021. Violence against women in the Covid-19 pandemic: a review of the literature and a call for shared strategies to tackle health and social emergencies. *Forensic Science International*, Vol. 319, art. 110650.
- Wekundah, M. and Trudell, B. 2021. Measuring Thematic Indicator 4.5.2: Challenges and Alternatives. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- World Bank. 2021. Loud and Clear: Effective Language of Instruction Policies for Learning. Washington, DC, World Bank. World Bank and IEI. 2021. Pivoting to Inclusion: Leveraging Lessons from the COVID-19 Crisis for Learners with Disabilities.
  - Washington, DC, World Bank/Inclusive Education Initiative.

- A'Hearn, B., Baten, J. and Crayen, D. 2009. Quantifying quantitative literacy: age heaping and the history of human capital. *Journal of Economic History*, Vol. 69, No. 3, pp. 783–808.
- Ancker, J. 2020. The COVID-19 pandemic and the power of numbers. Numeracy, Vol. 13, No. 2, art. 2.
- Apena, T. T. 2021. Sustainability of adult education programmes in Nigeria: a post pandemic threat to national development. KIU Journal of Education, Vol. 16, No. 1, pp. 133–40.
- Ayyappan, A. 2020. Ageing Nepal: Basic Literacy Classes for Older Persons. Hamburg, UNESCO Institute for Lifelong Learning. https://uil.unesco.org/case-study/effective-practices-database-litbase-0/ageing-nepal-basicliteracy-classes-older-persons. (Accessed 17 August 2021.)
- Barakat, B. 2016. Improving adult literacy without improving the literacy of adults? A cross-national cohort analysis. World Development, Vol. 87, pp. 242–57.
- Baten, J. 2021. Indirect Survey Estimates of Basic Numeracy among Disadvantaged Groups in sub-Saharan Africa. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Baten, J. and Alexopoulou, K. 2021. Elite violence and elite numeracy in Africa from 1400 CE to 1900 CE. European Review of Economic History, art. heab013.
- Baten, J. and Hippe, R. 2018. Geography, land inequality and regional numeracy in Europe in historical perspective. *Journal of Economic Growth*, Vol. 23, No. 1, pp. 79–109.
- Baten, J. and Juif, D. 2014. A story of large landowners and math skills: inequality and human capital formation in long-run development, 1820–2000. Journal of Comparative Economics, Vol. 42, No. 2, pp. 375–401.
- Brazil Ministry of Education. 2021. Resolução No 1, de 28 de Maio de 2021 [Resolution Number 1, of May 28, 2021]. Brasília, Diário Oficial da União.
- Chamberlin, S. 2019. Education as an Individual and Social Support Strategy for Managing HIV Care Adherence in Malawi. Paper for 2019 Comparative and International Education Society Annual Conference, San Francisco, Calif., 14–18 April.
- Crayen, D. and Baten, J. 2010. Global trends in numeracy 1820–1949 and its implications for long-term growth. Explorations in Economic History, Vol. 47, No. 1, pp. 82–99.
- Das, A., Wadhwa, S., Bansal, T. and Retnakaran, D. 2021. Literacy and response to COVID-19. Economic and Political Weekly, Vol. 56, No. 34.
- Desjardins, R. 2020. Micro and macro drivers affecting adult literacy proficiency profiles across countries. International Review of Education, Vol. 66, No. 2, pp. 289–340.
- Dickerson, A., McIntosh, S. and Valente, C. 2015. Do the maths: an analysis of the gender gap in mathematics in Africa. Economics of Education Review, Vol. 46, pp. 1–22.
- Grotlüschen, A., Buddeberg, K., Dutz, G., Heilmann, L. and Stammer, C. 2020a. Low literacy in Germany: results from the second German literacy survey. European Journal for Research on the Education and Learning of Adults, Vol. 11, No. 1, pp. 127–43.
- Grotlüschen, A., Desjardins, R. and Liu, H. 2020b. Literacy and numeracy: global and comparative perspectives. International Review of Education, Vol. 66, No. 2, pp. 127–37.
- Hanushek, E. A. and Woessmann, L. 2012. Do better schools lead to more growth? Cognitive skills, economic outcomes, and causation. *Journal of Economic Growth*, Vol. 17, No. 4, pp. 267–321.
- INEP. 2021. Censo da Educação Básica 2020: Resumo Técnico [Basic Education Census 2020: Technical Summary]. Brasília, Instituto Nacional de Estudos e Pesquisas Educacionais Anísio Teixeira.
- Juif, D.-T. and Baten, J. 2013. On the human capital of Inca Indios before and after the Spanish Conquest: was there a 'pre-colonial legacy'? Explorations in Economic History, Vol. 50, No. 2, pp. 227–41.
- Kalman, J. and Carvajal, E. 2020. Adult Literacy Programs in the Context of COVID-19: A Regional Snapshot from Latin America and the Caribbean. Santiago, UNESCO.
- Krenzke, T., Mohadjer, L., Li, J., Erciulescu, A., Fay, R., Ren, W., van de Kerckhove, W., Li, L., Rao, J. N. K., Xie, H. and Provasnik, S. 2020. Program for the International Assessment of Adult Competencies (PIAAC): State and County Estimation Methodology Report. Washington, DC, Institute of Education Sciences, National Center for Education Statistics, US Department of Education.
- Lang, S. 2021. Key factors influencing the maintenance of adult learners' literacy skills levels: a follow-up study of three participant cohorts of the literacy programme in Cambodia. *International Review of Education*, Vol. 67, No. 5, pp. 611–36.
- Lilenstein, A. 2018. Integrating Indicators of Education Quantity and Quality in Six Francophone African Countries. Stellenbosch, South Africa, Department of Economics, University of Stellenbosch. (Working Paper 09/2018.)

- Lopes, H. and McKay, V. 2020. Adult learning and education as a tool to contain pandemics: the COVID-19 experience. International Review of Education, Vol. 66, No. 4, pp. 575–602.
- Lotas, S. 2021. The COVID-19 pandemic from an adult literacy practitioner-scholar perspective: where we were, where we are, and where we should be going. Adult Literacy Education, Vol. 3, No. 2, pp. 50–54.
- NCES. 2020a. Highlights of the 2017 U.S. PIAAC Results Web Report. Washington, DC, Institute of Education Sciences, National Center for Education Statistics, US Department of Education. https://nces.ed.gov/surveys/piaac/ current\_results.asp. (Accessed 17 August 2021.)
- \_\_\_\_. 2020b. U.S. Skills Map. Washington, DC, Institute of Education Sciences, National Center for Education Statistics, US Department of Education. https://nces.ed.gov/surveys/piaac/skillsmap. (Accessed 17 August 2021.)
- O'Sullivan, K., O'Meara, N., Goos, M. and Conway, P. 2021. How COVID-19 has reinforced the importance of a numerate society. Irish Educational Studies, Vol. 40, No. 2, pp. 341–47.
- Partners in Health. 2021. Never Too Late to Learn: Providing Adult Education in Sierra Leone. Boston, Mass., Partners in Health. www.pih.org/article/never-too-late-learn-providing-adult-education-sierra-leone. (Accessed 20 October 2021.)
- Pritchett, L. 2013. The Rebirth of Education: Schooling Ain't Learning. Washington, DC, Center for Global Development.
- Reder, S., Gauly, B. and Lechner, C. 2020. Practice makes perfect: practice engagement theory and the development of adult literacy and numeracy proficiency. *International Review of Education*, Vol. 66, No. 2, pp. 267–88.
- Roozenbeek, J., Schneider, C., Dryhurst, S., Kerr, J., Freeman, A., Recchia, G., van der Bles, A. M. and van der Linden, S. 2020. Susceptibility to misinformation about COVID-19 around the world. *Royal Society Open Science*, Vol. 7, No. 10.
- Smythe, S., Wilbur, A. and Hunter, E. 2021. Inventive pedagogies and social solidarity: the work of community-based adult educators during COVID-19 in British Columbia, Canada. *International Review of Education*, Vol. 67, No. 1, pp. 9–29.
- Tollnek, F. and Baten, J. 2016. Age-heaping-based human capital estimates. Diebolt, C. and Haupert, M. (eds), Handbook of Cliometrics. Berlin, Springer, pp. 131–54.
- Tollnek, F. and Baten, J. 2017. Farmers at the heart of the 'human capital revolution'? Decomposing the numeracy increase in early modern Europe. *Economic History Review*, Vol. 70, No. 3, pp. 779–809.
- UIL. 2020. Adult Numeracy: Assessment and Development. Hamburg, UNESCO Institute for Lifelong Learning. (Policy Brief 13.)
- UNESCO. 2020a. Adult Learning and Education and COVID-19. Hamburg, Germany, UNESCO Institute for Lifelong Learning. (UNESCO COVID-19 Education Response: Education Sector Issue Note 2.6.)
- \_\_\_\_. 2020b. Concept Note: Literacy Teaching and Learning in the COVID-19 Crisis and Beyond The Role of Educators and Changing Pedagogies. Paris, UNESCO.
- \_\_\_\_. 2020c. Youth and Adult Literacy in the Time of COVID-19: Impacts and Revelations. Paris, UNESCO. (Background paper for International Literacy Day 2020.)
- Zúñiga, M. G., Torres, S. G. and Solano, J. C. 2015. Directrices Didácticas para la Alfabetización en Costa Rica [Literacy Teaching Guidelines in Costa Rica]. San José, Faculty of Education, University of Costa Rica/Department of Youth and Adults, Ministry of Education.

- Afghanistan Ministry of Education. 2020. Alternate Education Service Delivery COVID-19: Emergency Response Plan. Kabul, Ministry of Education.
- Bazaid, A. S., Aldarhami, A., Binsaleh, N. K., Sherwani, S. and Althomali, O. W. 2020. Knowledge and practice of personal protective measures during the COVID-19 pandemic: a cross-sectional study in Saudi Arabia. PloS ONE, Vol. 15, No. 12.
- Biasio, L. R. 2017. Vaccine hesitancy and health literacy. Human Vaccines and Immunotherapeutics, Vol. 13, No. 3, pp. 701–02.
- Burns, J. 2020. Climate change: schools failing us, say pupils. BBC News, 24 February. www.bbc.com/news/ education-51492942. (Accessed 21 June 2021.)
- Chang, C. H. 2015. Teaching climate change: a fad or a necessity? International Research in Geographical and Environmental Education, Vol. 24, No. 3, pp. 181–83.

Chiba, M., Sustarsic, M., Perriton, S. and Edwards Jr, D. B. 2021. Investigating effective teaching and learning for sustainable development and global citizenship: implications from a systematic review of the literature. International Journal of Educational Development, Vol. 81, art. 102337.

Coan, T. G. and Holman, M. R. 2008. Voting green. Social Science Quarterly, Vol. 89, No. 5, pp. 1121–35.

- Colombia Ministry of National Education. 2005. Educar para el Desarrollo Sostenible [Educating for Sustainable Development]. Bogotá, Colombia, Ministry of National Education. www.mineducacion.gov.co/1621/ article-90893.html. (Accessed 17 August 2021.)
- Damian, A. J. and Gallo, J. J. 2020. Promoting Health Literacy during the COVID-19 Pandemic: A Call to Action for Healthcare Professionals. Cambridge, Mass., Shorenstein Center on Media, Politics and Public Policy, Harvard Kennedy School. https://misinforeview.hks.harvard.edu/article/promoting-health-literacy-during-the-covid-19pandemic-a-call-to-action-for-healthcare-professionals. (Accessed 11 October 2021.)
- Dietz, T., Shwom, R. L. and Whitley, C. T. 2020. Climate change and society. Annual Review of Sociology, Vol. 46, pp. 135–58.
- European Commission. 2021. Education for Environmental Sustainability. Brussels, European Commission. https:// ec.europa.eu/education/education-in-the-eu/european-education-area/education-for-environmentalsustainability\_en. (Accessed 17 August 2021.)
- Farge, E. 2021. Vaccine nationalism puts world on brink of 'catastrophic moral failure': WHO chief. Reuters, 18 January. www.reuters.com/article/us-health-coronavirus-who-idUSKBN29N0TB. (Accessed 11 October 2021.)
- International Commission on the Futures of Education. 2020. Education in a Post-COVID World: Nine Ideas for Public Action. Paris, UNESCO.
- IPCC. 2019. Climate Change and Land: An IPCC Special Report on Climate Change, Desertification, Land Degradation, Sustainable Land Management, Food Security, and Greenhouse Gas Fluxes in Terrestrial Ecosystems. Geneva, Switzerland, Intergovernmental Panel on Climate Change.
- Leone, T. and Brown, L. J. 2020. Timing and determinants of age at menarche in low-income and middle-income countries. *BMJ Global Health*, Vol. 5, No. 12, art. 003689.
- Lopes, H. and McKay, V. 2020. Adult learning and education as a tool to contain pandemics: the COVID-19 experience. International Review of Education, Vol. 66, No. 4, pp. 575–602.
- Lubell, M., Vedlitz, A., Zahran, S. and Alston, L. T. 2006. Collective action, environmental activism, and air quality policy. Political Research Quarterly, Vol. 59, No. 1, pp. 149–60.
- Luong, K. T., Maibach, E. and Kotcher, J. 2021. Pew's new global survey of climate change attitudes finds promising trends but deep divides. *The Conversation*, 14 September. https://theconversation.com/pews-new-globalsurvey-of-climate-change-attitudes-finds-promising-trends-but-deep-divides-167847. (Accessed 14 September 2021.)
- Molnar, A. 2021. Health literacy during pandemic and education. Disaster Medicine and Public Health Preparedness.
- Mora-Ortiz, J. R. 2015. Los Proyectos Ambientales Escolares: herramientas de gestión ambiental [School
  - Environmental Projects: environmental management tools]. Revista Bitácora Urbano Territorial, Vol. 25, No. 2, pp. 67–74.
- OECD. 2020. PISA 2018 Results (Volume VI): Are Students Ready to Thrive in an Interconnected World? Paris, Organisation for Economic Co-operation and Development.
- Republic of Korea Ministry of Education. 2015. National Curriculum Framework of Early Childhood Education. Sejong City, Republic of Korea, Ministry of Education.
- RRI. 2018. A Global Baseline of Carbon Storage in Collective Lands: Indigenous and Local Community Contributions to Climate Change Mitigation. Washington, DC, Rights and Resources Initiative.
- San Diego County Office of Education. 2021. \$2 million in funding to expand health literacy in schools. San Diego, Calif., San Diego County Office of Education. www.sdcoe.net/news/Pages/2-million-funding-to-expand-healthliteracy-in-schools.aspx. (Accessed 28 October 2021.)
- Sandoval-Hernandez, A. and Carrasco, D. 2020. A Measurement Strategy for SDG Thematic Indicators 4.7.4 and 4.7.5 Using International Large Scale Assessments in Education. Montreal, Que., UNESCO Institute for Statistics.
- SEPN. 2020. The Monitoring and Evaluation of Climate Change Education (MECCE) Project. Saskatoon, Sask., Sustainability and Education Policy Framework. https://sepn.ca/mecce. (Accessed 21 June 2021.)
- Shashidhara, L. S. 2019. Integrating climate change education across the curriculum. *Resonance*, Vol. 24, No. 9, pp. 1025–28.
- Skinner, N. and Bromley, P. 2019. Individual and collective social justice education: comparing emphases on human rights and social movements in textbooks worldwide. *Comparative Education Review*, Vol. 63, No. 4, pp. 502–28.

Staton, B. and Hughes, L. 2020. Schools in England could drop some non-core subjects next year. Financial Times, 29 June. www.ft.com/content/f3e98ae9-badb-4ad5-ae3b-0dab037bfd28. (Accessed 11 October 2021.)

- Turhan, Z., Dilcen, H. Y. and Dolu, İ. 2021. The mediating role of health literacy on the relationship between health care system distrust and vaccine hesitancy during COVID-19 pandemic. *Current Psychology*.
- US Department of Health and Human Services. 2021. HHS Announces Awardees of \$250 Million to Fight COVID-19 and Improve Health Literacy among Racial and Ethnic Minority and Vulnerable Communities. Washington, DC, Office of Minority Health, Department of Health and Human Services. www.hhs.gov/about/news/2021/06/28/hhsannounces-awardees-of-250-million-to-fight-covid-19-and-improve-health-literacy.html. (Accessed 11 October 2021.)
- UIS. 2020. UNESCO-UNICEF-World Bank Survey on National Education. Responses to COVID-19 School Closures: 1st Iteration. Montreal, Que., UNESCO Institute for Statistics. http://covid19.uis.unesco.org/joint-covid-r1. (Accessed 11 October 2021.)
- UNESCO. 2015a. Global Citizenship Education: Topics and Learning Objectives. Paris, UNESCO.
- \_\_\_\_. 2015b. Not Just Hot Air: Putting Climate Change Education into Practice. Paris, UNESCO.
- \_\_\_\_. 2018. International Technical Guidance on Sexuality Education: An Evidence-informed Approach. Paris/Geneva, Switzerland/New York, UNESCO/UNAIDS/UNFPA/UNICEF/UN Women/WHO.
- \_\_\_\_. 2020. Global Citizenship Education and Education for Sustainable Development in Latin America and the Caribbean: 2030 Education Agenda Components in the Curricula of the Countries Participating in Regional Comparative and Explanatory Study (ERCE 2019). Paris/Santiago, UNESCO/OREALC/UNESCO Santiago.
- \_\_\_\_. 2021a. 2003 Convention Periodic Reporting: Online Form Adapted to the Overall Results Framework. Paris, UNESCO. https://ich.unesco.org/en/perioding-reporting-ich-10-2020-01081?edit\_form=764#1. (Accessed 17 August 2021.)
- \_\_\_\_. 2021b. The Journey towards Comprehensive Sexuality Education: Global Status Report Highlights. Paris, UNESCO.
- \_\_\_\_. 2021c. Learn for Our Planet: A Global Review of How Environmental Issues Are Integrated in Education. Paris, UNESCO. UNESCO and UNFCCC. 2016. Action for Climate Empowerment: Guidelines for Accelerating Solutions through Education,
  - Training and Public Awareness. Paris/Bonn, Germany, UNESCO/UN Framework Convention on Climate Change Secretariat.
- United Nations. 2020. COVID-19: UN Counters Pandemic-related Hate and Xenophobia. New York, United Nations. www.un.org/fr/node/71022. (Accessed 28 October 2021.)
- World Economic Forum. 2020. Toward a More Sustainable World: A Global Study of Public Opinion. Cologny, Switzerland, SAP + Qualtrics. www.weforum.org/about/a-global-study-of-public-opinion-davos-2020. (Accessed 22 June 2021.)
- Zelenika, I., Moreau, T., Lane, O. and Zhao, J. 2018. Sustainability education in a botanical garden promotes environmental knowledge, attitudes and willingness to act. *Environmental Education Research*, Vol. 24, No. 11, pp. 1581–96.

- Agasisti, T., Avvisati, F., Borgonovi, F. and Longobardi, S. 2021. What school factors are associated with the success of socio-economically disadvantaged students? An empirical investigation using PISA data. Social Indicators Research, Vol. 157, No.2, pp. 749–81.
- AMF. 2018. Enquête 2018: Les temps périscolaires après les réformes successives [2018 Survey: extracurricular schedule after successive reforms]. Paris, Association des Maires de France et des Présidents d'Intercommunalité.
- ANGOP. 2021. Aulas no ensino primário podem retomar a 10 de fevereiro [Primary school classes can resume on February 10]. Agência Angola Press, 11 January. www.angop.ao/noticias/educacao/aulas-no-ensino-primario-retorna-emfevereiro. (Accessed 23 April 2021.)
- Asanati, K., Voden, L. and Majeed, A. 2021. Healthier schools during the COVID-19 pandemic: ventilation, testing and vaccination. *Journal of the Royal Society of Medicine*, Vol. 114, No. 4, pp. 160–63.
- Ashong-Katai, L. N. D. 2013. Abolition of the double shift system of schooling in Ghana: policy and its implementation in public basic schools. MPhil thesis in Comparative and International Education, University of Oslo. (Unpublished.)
- Barrett, P., Treves, A., Shmis, T. and Ambasz, D. 2019. The Impact of School Infrastructure on Learning: A Synthesis of the Evidence. Washington, DC, World Bank.

Barshay, J. 2021. Could more time in school help students after the pandemic? The Hechinger Report, 24 May. https:// hechingerreport.org/proof-points-could-more-time-in-school-help-students-after-the-pandemic. (Accessed 27 March 2021.)

BBC. 2019. Bhutan profile: Media. London, BBC. www.bbc.com/news/world-south-asia-12484025. (Accessed 23 April 2021.)

- Brahmanandam, T. T. and Bosu Babu, T. 2016. Educational status among the scheduled tribes: issues and challenges. *NEHU Journal*, Vol. XIV, No. 2, pp. 69–85.
- Burke, K. J. and Segall, A. 2011. Christianity and its legacy in education. *Journal of Curriculum Studies*, Vol. 43, No. 5, pp. 631–58.
- Burns, M., Santally, M. I., Halkhoree, R., Sungkur, K. R., Juggurnath, B. and Rajabalee, Y. B. 2019. Information and Communication Technologies in Secondary Education in Sub-Saharan Africa: Policies, Practices, Trends, and Recommendations. Toronto, Ont., Mastercard Foundation. (Background paper for Secondary Education in Africa: Preparing Youth for the Future of Work.)
- Carrell, S. E., Maghakian, T. and West, J. E. 2011. A's from Zzzz's? The causal effect of school start time on the academic achievement of adolescents. American Economic Journal: Economic Policy, Vol. 3, No. 3, pp. 62–81.
- CDC and WHO. 2017. GSHS Questionnaire. Atlanta, Ga./Geneva, Switzerland, Centres for Disease Control and Prevention/ World Health Organization. www.who.int/teams/noncommunicable-diseases/surveillance/systems-tools/ global-school-based-student-health-survey/questionnaire. (Accessed 17 August 2021.)
- Chan, N. Y., Zhang, J., Yu, M. W. M., Lam, S. P., Li, S. X., Kong, A. P. S., Li, A. M. and Wing, Y. K. 2017. Impact of a modest delay in school start time in Hong Kong school adolescents. *Sleep Medicine*, Vol. 30, pp. 164–70.
- Côte d'Ivoire Government. 2017. Plan Sectoriel Education/Formation 2016–2025 [Education/Training Sector Plan 2016–2025]. Abidjan, Ministry of Education, Technical Education and Professional Training/Ministry of Higher Education/ Scientific Research Task Force.
- Dhital, R. P., Ito, T., Kaneko, S., Komatsu, S. and Yoshida, Y. 2021. Household access to water and education for girls: the case of villages in hilly and mountainous areas of Nepal. *Oxford Development Studies*.
- Duchini, E. and Effenterre, C. V. 2017. La réforme des rhythmes scolaires: un révelateur des inégalités présentes sur le marché du travail? [The Reform of School Calendars: A Revelation of the Inequalities Present on the Labor Market?]. Paris, Institut des Politiques Publiques. (Les notes de l'IPP 26.)
- End Corporal Punishment. 2021a. Legality of Corporal Punishment Worldwide. New York, Global Partnership to End Violence Against Children. https://endcorporalpunishment.org/global-progress/global-table-of-legality. (Accessed 17 August 2021.)
- European Commission/EACEA/Eurydice. 2019. The Organisation of School Time in Europe: Primary and General Secondary Education – 2019/20. Luxembourg, European Commission/EACEA/Eurydice. (Eurydice Facts and Figures.)
- \_\_\_\_. 2020. The Organisation of School Time in Europe: Primary and General Secondary Education 2020/21. Luxembourg, European Commission/EACEA/Eurydice. (Eurydice Facts and Figures.)
- Filardo, M. 2016. State of Our Schools: America's K-12 Facilities. Washington, DC, 21st Century School Fund.
- Finnie, R. K. C., Peng, Y., Hahn, R. A., Johnson, R. L., Fielding, J. E., Truman, B. I., Muntaner, C., Fullilove, M. T., Zhang, X. and the Community Preventive Services Task Force. 2019. Examining the effectiveness of year-round school calendars on improving educational attainment outcomes within the context of advancement of health equity: a community guide systematic review. *Journal of Public Health Management and Practice*, Vol. 25, No. 6, pp. 590–94.
- Fischel, W. A. 2006. 'Will I see you in September?' An economic explanation for the standard school calendar. *Journal of Urban Economics*, Vol. 59, No. 2, pp. 236–51.
- Fricchione, M. J., Seo, J. Y. and Arwady, M. A. 2021. Data-driven reopening of urban public education through Chicago's tracking of COVID-19 school transmission. *Journal of Public Health Management and Practice*, Vol. 27, No. 3, pp. 229–32.
- GCPEA. 2021. Safe Schools Declaration Endorsements. New York, Global Coalition to Protect Education from Attack. https://ssd.protectingeducation.org/endorsement. (Accessed 13 October 2021.)
- Gomes, C. A. T. and Duarte, M. R. T. 2017. School infrastructure and socioeconomic status in Brazil. Sociology and Anthropology, Vol. 5, No. 7, pp. 522–32.
- Haapala, E. A., Väistö, J., Lintu, N., Westgate, K., Ekelund, U., Poikkeus, A.-M., Brage, S. and Lakka, T. A. 2017. Physical activity and sedentary time in relation to academic achievement in children. *Journal of Science and Medicine in* Sport, Vol. 20, No. 6, pp. 583–89.

- Harris, J. 2018. Our schools are broke so why aren't we talking about it? *The Guardian*, 1 June. www.theguardian.com/ commentisfree/2018/jun/01/schools-broke-class-sizes-teachers-brexit. (Accessed 17 August 2021.)
- Hazell, W. 2021. Education catch-up: longer school day, more tutoring and better teaching 'three pillars' of recovery plan. *iNews*, 10 May. https://inews.co.uk/news/education/education-catch-up-longer-school-day-tutoring-better-teaching-recovery-plan-994579. (Accessed 5 March 2021.)
- Huaman, E. S. 2020. Small indigenous schools: indigenous resurgence and education in the Americas. Anthropology and Education Quarterly, Vol. 51, No. 3, pp. 262–81.
- Huang, Z., Huang, J., Gu, Q., Du, P., Liang, H. and Dong, Q. 2020. Optimal temperature zone for the dispersal of COVID-19. Science of the Total Environment, Vol. 736, art. 139487.
- Inchley, J., Currie, D., Cosma, A. and Samdal, O. 2018. *Health Behaviour in School-aged Children (HBSC) Study Protocol: Background, Methodology and Mandatory Items for the 2017/18 Survey.* St Andrews, UK, Child and Adolescent Health Research Unit.
- Ito, S. and Shonchoy, A. S. 2020. Seasonality, Academic Calendar and School Drop-outs in Developing Countries. Miami, Fla., Florida International University. (Department of Economics Working Paper 2013.)
- Joint Monitoring Programme. 2020. Progress on Drinking Water, Sanitation and Hygiene in Schools: Special Focus on COVID-19. New York, WHO/UNICEF Joint Monitoring Programme for Water Supply, Sanitation and Hygiene.
- Kadari-Ovadia, S. 2020. Israeli Treasury favors moving to a five-day school week. Haaretz, 6 February. www.haaretz. com/israel-news/.premium-israeli-treasury-favors-moving-to-a-five-day-school-week-1.8500729. (Accessed 27 April 2021.)
- Kapit, A., Marston, J. and Tsolakis, M. 2021. Toolkit for Collecting and Analyzing Data on Attacks on Education. New York, Global Coalition to Protect Education from Attack.
- Kelley, P., Lockley, S. W., Foster, R. G. and Kelley, J. 2015. Synchronizing education to adolescent biology: 'Let teens sleep, start school later'. *Learning, Media and Technology*, Vol. 40, No. 2, pp. 210–26.
- Kelley, P., Lockley, S. W., Kelley, J. and Evans, M. D. R. 2017. Is 8:30 a.m. still too early to start school? A 10:00 a.m. school start time improves health and performance of students aged 13–16. *Frontiers in Human Neuroscience*, Vol. 11, art. 588.
- Kenya Ministry of Education. 2018. 2019 Term Dates for Schools and Colleges. Nairobi, Ministry of Education.
- Kirby, M., Maggi, S. and D'Angiulli, A. 2011. School start times and the sleep-wake cycle of adolescents: a review and critical evaluation of available evidence. *Educational Researcher*, Vol. 40, No. 2, pp. 56–61.
- Klein, R. 2021. While learning online, many students received a surprising pandemic respite from cyberbullying. The Hechinger Report, 30 September. https://hechingerreport.org/while-learning-online-many-studentsreceived-a-surprising-pandemic-respite-from-cyberbullying. (Accessed 13 October 2021.)
- Lavy, V. 2015. Do differences in schools' instruction time explain international achievement gaps? Evidence from developed and developing countries. *The Economic Journal*, Vol. 125, No. 588, pp. F397–F424.
- Maiti, A. 2021. Effect of corporal punishment on young children's educational outcomes. Education Economics, Vol. 29, No. 4, pp. 411–23.
- McMurrer, J. 2007. Choices, Changes, and Challenges: Curriculum and Instruction in the NCLB Era. Washington, DC, Center on Education Policy.
- Melker, S. and Weber, S. 2014. Agrarian roots? Think again. Debunking the myth of summer vacation's origins. Public Broadcasting Service, 7 September. www.pbs.org/newshour/education/debunking-myth-summer-vacation. (Accessed 28 April 2021.)
- Nakale, A. 2020. Double session school system returns. New Era, 7 June https://neweralive.na/posts/double-session-school-system-returns. (Accessed 28 April 2021.)
- OECD. 2018. TALIS 2018 data. Paris, Organisation for Economic Co-operation and Development. www.oecd.org/ education/talis/talis-2018-data.htm. (Accessed 19 July 2021.)
- \_\_\_\_. 2021. The State of School Education: One Year into the COVID Pandemic. Paris, Organisation for Economic Co-operation and Development.
- Ontario Ministry of Education. 2019. School Facility Condition Index. Toronto, Ont., Ministry of Education. www.edu.gov.on.ca/eng/parents/fci.html. (Accessed 17 August 2021.)
- Parente, C. d. M. D. 2020. Multiple-shift schooling: international context and the Brazilian case. Revista Tempos e Espaços em Educação, Vol. 13, No. 32, pp. 1–20.

- Park, S., Choi, Y., Song, D. and Kim, E. K. 2021. Natural ventilation strategy and related issues to prevent coronavirus disease 2019 (COVID-19) airborne transmission in a school building. *Science of the Total Environment*, Vol. 789, art. 147764.
- Peek, L. 2018. America's deathtrap schools. *The New York Times*, 7 April. www.nytimes.com/2018/04/07/opinion/ sunday/americas-deathtrap-schools.html. (Accessed 17 August 2021.)
- Pulkkinen, L. 2021. America's schools are crumbling. Fixing them could save lives (and the planet). *The Hechinger Report*, 28 June. https://hechingerreport.org/americas-schools-are-crumbling-fixing-them-could-save-lives-and-the-planet. (Accessed 17 August 2021.)
- Rasmeni, M. 2017. No more platoon schools: Namport commits more funding for classrooms. Namibia Economist, 10 August. https://economist.com.na/27191/education/no-more-platoon-schools-namport-commits-morefunding-for-classrooms. (Accessed 28 April 2021.)
- Razzaque, M. A. 2020. Graduation of Bhutan from the Group of Least Developed Countries: Potential Implications and Policy Imperatives. Bangkok, UN Economic and Social Commission for Asia and the Pacific. (Macroeconomic Policy and Financing for Development Division Working Paper 20/04.)
- Roberts, L. W. 2009. Measuring school facility conditions: an illustration of the importance of purpose. Journal of Educational Administration, Vol. 47, No. 3, pp. 368–80.
- Rwanda Ministry of Education. 2016. ICT in Education Policy. Kigali, Ministry of Education.
- Sachgau, O. 2016. \$15 billion needed to fix Ontario schools. Toronto Star, 26 August. www.thestar.com/news/ gta/2016/08/26/15-billion-needed-to-fix-ontario-schools.html. (Accessed 17 August 2021.)
- Sedgwick, J. 2018. 25-year-old textbooks and holes in the ceiling: inside America's public schools. *The New York Times*, 16 April. www.nytimes.com/2018/04/16/reader-center/us-public-schools-conditions.html. (Accessed 17 August 2021.)
- Shure, N. 2019. School hours and maternal labor supply. Kyklos, Vol. 72, No. 1, pp. 118–51.
- Stadt Wien. 2021. Wien etabliert 'Alles gurgelt!' an Schulen [Vienna implements 'Everyone gargles!' at school]. Vienna, Municipal Government. www.wien.gv.at/presse/2021/08/27/wien-etabliert-alles-gurgelt-an-schulen. (Accessed 12 October 2021.)
- Sujatha, K. 2002. Education among scheduled tribes. Govinda, R. (ed.), India Education Report: A Profile of Basic Education. New Delhi, Oxford University Press.
- Uganda Ministry of Education and Sports. 2018. Schools' and Other Institutions' Calendar 2019. Kampala, Ministry of Education and Sports.
- UIS. 2020. SDG Indicator 4.A.2 (School Bullying): Methodological Note. Montreal, Que., UNESCO Institute for Statistics. (Technical Cooperation Group WG/GAML/10.)
- UNESCO. 2019. Qualidade da Infraestrutura das Escolas Públicas do Ensino Fundamental no Brasil: Indicadores com Dados Públicos e Tendências de 2013, 2015 e 2017 [Infrastructure Quality of Public Elementary Schools in Brazil: Indicators with Public Data and Trends for 2013, 2015 and 2017]. Brasília, UNESCO.
- \_\_\_\_. 2021. COVID-19 Response: Health, Safety and Resurgence Protocols Ensuring Safe School Reopening, Operation and Resurgence Planning. Paris, UNESCO.
- UNESCO, UNICEF and World Bank. 2020. What Have We Learnt? Overview of Findings from a Survey of Ministries of Education on National Responses to COVID-19. Paris/New York/Washington, DC, UNESCO/UNICEF/World Bank.
- UNESCO, UNICEF, World Bank and OECD. 2021. What's Next? Lessons on Education Recovery: Findings from a Survey of Ministries of Education amid the COVID-19 Pandemic. Paris/Montreal, Que.,/New York/Florence/Washington, DC, UNESCO, UNESCO Institute for Statistics/UNICEF/UNICEF Office of Research – Innocenti/World Bank/OECD.
- US Government Accountability Office. 2020. K–12 Education: School Districts Frequently Identified Multiple Building Systems Needing Updates or Replacement. Washington, DC, Government Accountability Office. (GAO-20-494.)
- von Hippel, P. T. 2020. What Does Research Say about Staggered School Calendars? Los Angeles, Calif., Center on Education Policy, Equity and Governance. (The Answer Lab.)
- von Hippel, P. T. and Hamrock, C. 2019. Do test score gaps grow before, during, or between the school years? Measurement artifacts and what we can know in spite of them. *Sociological Science*, Vol. 6, pp. 43–80.
- Wickman, F. 2011. School's out forever. *Slate*, 2 September. https://slate.com/news-and-politics/2011/09/if-you-go-to-school-on-the-equator-when-does-your-summer-vacation-end.html. (Accessed 28 April 2021.)
- Woessmann, L. 2016. The importance of school systems: evidence from international differences in student achievement. *Journal of Economic Perspectives*, Vol. 30, No. 3, pp. 3–32.

- Yeşil Dağlı, Ü. 2019. Effect of increased instructional time on student achievement. Educational Review, Vol. 71, No. 4, pp. 501–17.
- Zavacky, F. and Michael, S. L. 2017. Keeping recess in schools. Journal of Physical Education, Recreation and Dance, Vol. 88, No. 5, pp. 46–53.

- Anadolu Agency. 2021. Rwandan students to learn Turkish with Yunus Emre Institute. Daily Sabah, 20 April. www.dailysabah.com/turkey/education/rwandan-students-to-learn-turkish-with-yunus-emre-institute. (Accessed 17 August 2021.)
- Baecher, L. (ed.). 2021. Study Abroad for Pre- and In-Service Teachers: Transformative Learning on a Global Scale. New York, Routledge.
- Bilecen, B. 2020. Commentary: COVID-19 pandemic and higher education international mobility and students' social protection. International Migration, Vol. 58, No. 4, pp. 263–66.
- Campbell, A. C. and Baxter, A. R. 2019. Exploring the attributes and practices of alumni associations that advance social change. International Journal of Educational Development, Vol. 66, pp. 164–72.
- Campbell, A. C., Kelly-Weber, E. and Lavallee, C. 2021. University teaching and citizenship education as sustainable development in Ghana and Nigeria: insight from international scholarship program alumni. *Higher Education*, Vol. 81, No. 1, pp. 129–44.
- Campbell, A. C. and Lavallee, C. A. 2020. A community of practice for social justice: examining the case of an international scholarship alumni association in Ghana. *Journal of Studies in International Education*, Vol. 24, No. 4, pp. 409–23.
- Campbell, A. C. and Neff, E. 2020. A systematic review of international higher education scholarships for students from the Global South. Review of Educational Research, Vol. 90, No. 6, pp. 824–61.
- Caravatti, M.-L., Lederer, S. M., Lupico, A. and Van Meter, N. 2014. *Getting Teacher Migration and Mobility Right*. Brussels, Education International.
- Chankseliani, M. 2018. The politics of student mobility: links between outbound student flows and the democratic development of post-Soviet Eurasia. International Journal of Educational Development, Vol. 62, pp. 281–88.
- Chen, C., Bernard, A., Rylee, R. and Abel, G. 2021. Brain circulation: the educational profile of return migrants. Population Research and Policy Review.
- China Scholarship Council. 2020. Introduction to Chinese Government Scholarships. Beijing, Study in China. www.campuschina.org/content/details3\_74776.html. (Accessed 5 March 2021.)
- Development Initiatives. 2020. Global Humanitarian Assistance Report 2020. Bristol, UK, Development Initiatives.
- Di Maria, D. L. 2020. US colleges report a 43% decline in new international student enrollment, and not just because of the pandemic. *The Conversation*, 19 November. https://theconversation.com/us-colleges-report-a-43decline-in-new-international-student-enrollment-and-not-just-because-of-the-pandemic-149885. (Accessed 17 August 2021.)
- Dong, L. and Chapman, D. W. 2008. The Chinese Government Scholarship Program: an effective form of foreign assistance? International Review of Education, Vol. 54, No. 2, pp. 155–73.
- European Commission. 2017. Erasmus+: School education staff mobility The benefits. Brussels, European Commission. https://ec.europa.eu/programmes/erasmus-plus/news/school-education-staff-mobility-benefits\_en. (Accessed 7 July 2021.)
- Finnish National Agency for Education. 2020. Impact of COVID-19 on Higher Education Student Mobility in Finland. Helsinki, Finnish National Agency for Education. www.oph.fi/en/news/2020/impact-covid-19-higher-educationstudent-mobility-finland. (Accessed 17 August 2021.)
- GIZ. 2020. Alumniportal Deutschland. Bonn, Deutsche Gesellschaft für Internationale Zusammenarbeit. www.giz.de/en/worldwide/30728.html. (Accessed 17 August 2021.)
- Ha, W., Lu, K. and Wo, B. 2020. Do Chinese government foreign student scholarships target natural resources in Africa? Higher Education Policy, Vol. 33, No. 3, pp. 479–509.
- Hoffmann, B., Pedroza, L. and Palop-García, P. 2017. *Emigrant Policies Index (EMIX) Dataset*. Hamburg, German Institute of Global and Area Studies. https://data.gesis.org/sharing/#!Detail/10.7802/1499. (Accessed 17 August 2021.)
- IIE. 2020. Assisting International Students in the COVID-19 Crisis. New York, Institute of International Education. www.iie.org/en/Donate/Where-to-give/IIE-Emergency-Student-Fund/IIE-COVID-19-Response. (Accessed 17 August 2021.)

- Iyevlyev, O. 2018. Professional pedagogical mobility of educators in the European context. Comparative Professional Pedagogy, Vol. 8, No. 3, pp. 57–63.
- Jaritz, G. 2011. Developing a culture of (inter)national mobility in initial teacher training: expectations, limitations and ways forward. Goetz, T., Jaritz, G. and Oser, F. (eds), *Pains and Gains of International Mobility in Teacher Education*. Rotterdam, SensePublishers, pp. 7–24.
- Jing, C. 2020. BRICS students education in China from 2010 to 2018: development, problems and recommendations. Conjuntura Austral, Vol. 11, No. 53, pp. 67–91.
- Lei, J., Xiao, L. and Li, B. 2021. African students' motivations for studying in China's higher education. Asia Pacific Education Review, Vol. 22, No. 2, pp. 319–32.
- Marginson, S. 2020. Global HE as we know it has forever changed. *Times Higher Education*, 26 March. www.timeshighereducation.com/blog/global-he-we-know-it-has-forever-changed. (Accessed 17 August 2021.)
- Martel, M. 2017. Tracing the spark that lights a flame: a review of methodologies to measure the outcomes of international scholarships. Dassin, J. R., Marsh, R. R. and Mawer, M. (eds), *International Scholarships in Higher Education: Pathways to Social Change.* Cham, Switzerland, Springer, pp. 281–304.
- Mawer, M. 2018. Magnitudes of impact: a three-level review of evidence from scholarship evaluation. Dassin, J. R., Marsh, R. R. and Mawer, M. (eds), International Scholarships in Higher Education: Pathways to Social Change. Cham, Switzerland, Springer, pp. 257–80.
- Mercado, S. 2020. International student mobility and the impact of the pandemic. AACSB Insights for Everyone, 11 June. https://www.aacsb.edu/insights/articles/2020/06/covid-19-and-the-future-of-international-studentmobility. (Accessed 17 August 2021.)
- Mitchell, C. 2021. Erdogan's ambition drives Turkey's Africa surge. African Business, 17 March. https://african.business/2021/03/trade-investment/erdogans-ambition-drives-turkeys-africa-surge. (Accessed 17 August 2021.)
- Mulvey, B. 2021. 'Decentring' international student mobility: the case of African student migrants in China. Population, Space and Place, Vol. 27, No. 3, art. 2393.
- Odhiambo, W. 2020. Sport scholarship students suffer under Covid-19 lockdown. University World News, 11 June. www.universityworldnews.com/post.php?story=20200609072143783. (Accessed 17 August 2021.)
- OECD. 2018. TALIS 2018 data. Paris, Organisation for Economic Co-operation and Development. www.oecd.org/education/talis/talis-2018-data.htm. (Accessed 19 July 2021.)
- \_\_\_\_. 2020a. Share of international, foreign and all students enrolled by field of education. Paris, Organisation for Economic Co-operation and Development. https://stats.oecd.org/index.aspx/Images/Index.aspx?DataSetCode=EAG\_ENRL\_MOBILES\_FIELDS. (Accessed 17 August 2021.)
- \_\_\_\_. 2020b. TALIS 2018 Results (Volume II): Teachers and School Leaders as Valued Professionals. Paris, Organisation for Economic Co-operation and Development.
- Pedroza, L. and Palop-García, P. 2017. Diaspora policies in comparison: an application of the Emigrant Policies Index (EMIX) for the Latin American and Caribbean region. *Political Geography*, Vol. 60, pp. 165–78.
- Peters, M. A., Hollings, S., Zhanga, M., Quainoo, E. A., Wang, H., Huang, Y., Zhou, S., Laimeche, A., Omary Chunga, J., Ren, Z., Ogunniran, M. O., Khomera, S. W., Zheng, W., Xu, R., Mou, C. and Green, B. 2021. The changing map of international student mobility. *Access: Contemporary Issues in Education*, Vol. 41, No. 1, pp. 7–28.
- Princewill, N. 2021. Anger over UK quarantine rules for fully vaccinated travelers from India and many African countries. CNN, 2 October. https://edition.cnn.com/travel/article/covishield-vaccine-countries-travel-cmd-intl/index.html. (Accessed 13 October 2021.)
- Ragazzi, F. 2014. A comparative analysis of diaspora policies. Political Geography, Vol. 41, pp. 74–89.
- Rose, A. J. and Cooper-Duffy, K. 2021. Implementation of an interprofessional study abroad experience in Botswana for special educators and speech-language pathologists. Baecher, L. (ed.), Study Abroad for Pre- and In-Service Teachers: Transformative Learning on a Global Scale. New York, Routledge, pp. 146–65.
- Shields, R. 2021. Social Network Analysis Perspectives on Bilateral Scholarship Aid Flows. Paris, UNESCO. (Background paper for Global Education Monitoring Report 2021/2.)
- Shields, R. and Menashy, F. 2019. The network of bilateral aid to education 2005–2015. International Journal of Educational Development, Vol. 64, pp. 74–80.
- Tripp, L. O., Love, A., Barry, N., Thomas, C. M. and Russell, J. 2021. A qualitative analysis of teacher candidates' study abroad experiences in Malawi. Baecher, L. (ed.), *Study Abroad for Pre- and In-Service Teachers : Transformative Learning on a Global Scale.* New York, Routledge, pp. 129–45.
- Turkey Ministry of Culture and Tourism. 2021. Türkiye Burslari [Turkish Bursaries]. Ankara, Ministry of Culture and Tourism. https://turkiyeburslari.gov.tr. (Accessed 17 August 2021.)
- Turkish Cooperation and Coordination Agency. 2020. *Turkish Development Assistance Report 2019*. Ankara, Turkish Cooperation and Coordination Agency.
- University of Oklahoma. 2020. International Student Pandemic Scholarship. Norman, Okla., University of Oklahoma. www.ou.edu/cis/scholarships/international-students/international-student-pandemic-scholarship. (Accessed 17 August 2021.)
- Ünveren, B. 2021. Turkey seeks to strengthen Africa relations with 'benevolence'. *Deutsche Welle*, 4 February. www.dw.com/en/turkey-seeks-to-strengthen-africa-relations-with-benevolence/a-56452857. (Accessed 17 August 2021.)
- Vasilopoulos, G. and Romero, G. 2021. Post-method design for an English language teacher-training study abroad program: from western China to Canada. Baecher, L. (ed.), Study Abroad for Pre- and In-Service Teachers : Transformative Learning on a Global Scale. New York, Routledge, pp. 81–96.
- Vögtle, E. M. 2019. What Deters Students of Education and Teacher Training from Enrolment Abroad? Warsaw, Eurostudent. (Intelligence Brief 2.)
- Waters, J. 2021. COVID-19 and International Student Mobility: Some Reflections. London, London School of Economics. https://blogs.lse.ac.uk/seac/2021/01/05/covid-19-and-international-student-mobility-some-reflections. (Accessed 17 August 2021.)
- Witt, A. and Liu, W. 2021. Transforming teacher preparation for global readiness: scaling up education abroad. Baecher, L. (ed.), Study Abroad for Pre- and In-Service Teachers : Transformative Learning on a Global Scale. New York, Routledge, pp. 31–43.
- Woodland, R. H. 2021. Design principles for international teacher study abroad programs: the case of the Instructional Leadership Institute for Pakistani Educators (ILIPE). Baecher, L. (ed.), Study Abroad for Pre- and In-Service Teachers : Transformative Learning on a Global Scale. New York, Routledge, pp. 71–80.
- Yıldırım, S., Bostancı, S. H., Yıldırım, D. Ç. and Erdoğan, F. 2021. Rethinking mobility of international university students during COVID-19 pandemic. *Higher Education Evaluation and Development*, Vol. 15, No. 2, pp. 98–113.

### **CHAPTER 19**

- ADEA, AU/CIEFFA and APHRC. 2021. School Reopening in Africa during the COVID-19 Pandemic. Abidjan, Association for the Development of Education in Africa/African Union International Centre for Girls' and Women's Education in Africa/African Population and Health Research Center.
- Bastian, K. and Crittenden Fuller, S. 2021. Four Key Lessons on Teacher and Principal Attrition during the Pandemic. Raleigh, NC, EducationNC. www.ednc.org/perspective-four-key-lessons-on-teacher-and-principal-attrition-duringthe-pandemic. (Accessed 17 August 2021.)
- Blackley, S., Wilson, S., Sheffield, R., Murcia, K., Brown, P., Tang, K.-S., Cooper, M. and Williams, J. 2021. How have COVID-19-related changes to tuition modes impacted face-to-face initial teacher education students? *Issues in Educational Research*, Vol. 31, No. 2, pp. 421–39.
- Bobba, M., Ederer, T., Leon-Ciliotta, G., Neilson, C. A. and Nieddu, M. 2021. Teacher Compensation and Structural Inequality: Evidence from Centralized Teacher School Choice in Peru. Bonn, Germany, Institute of Labor Economics. (IZA Discussion Paper 14581.)
- Campanini, M. 2019. The Out-of-field Teaching Phenomenon. Sydney, Independent Education Union of Australia. http://publications.ieu.asn.au/2019-september-ie/articles4/teaching-phenomenon. (Accessed 17 August 2021.)
- Carver-Thomas, D., Leung, M. and Burns, D. 2021. California Teachers and COVID-19: How the Pandemic is Impacting the Teacher Workforce. Palo Alto, Calif., Learning Policy Institute.
- Cheung, E. 2021. Coronavirus: Hong Kong looks at making unvaccinated teachers undergo regular COVID-19 testing. South China Morning Post, 3 July. www.scmp.com/news/hong-kong/health-environment/article/3139711/ coronavirus-hong-kong-looks-making-unvaccinated. (Accessed 15 October 2021.)
- Clotfelter, C. T., Ladd, H. F. and Vigdor, J. L. 2010. Teacher credentials and student achievement in high school: a crosssubject analysis with student fixed effects. *Journal of Human Resources*, Vol. 45, No. 3, pp. 655–81.
- Darling-Hammond, L. and Hyler, M. E. 2020. Preparing educators for the time of COVID... and beyond. European Journal of Teacher Education, Vol. 43, No. 4, pp. 457–65.

De Neve, D. and Devos, G. 2017. How do professional learning communities aid and hamper professional learning of beginning teachers related to differentiated instruction? *Teachers and Teaching*, Vol. 23, No. 3, pp. 262–83.

- du Plessis, A. E., Gillies, R. M. and Carroll, A. 2014. Out-of-field teaching and professional development: a transnational investigation across Australia and South Africa. International Journal of Educational Research, Vol. 66, pp. 90–102.
- Dzehtsiarou, K. 2021. Compulsory vaccination: what does human rights law say? *The Conversation*, 14 September. https://theconversation.com/compulsory-vaccination-what-does-human-rights-law-say-167735. (Accessed 15 October 2021.)
- European Commission/EACEA/Eurydice. 2019. Teachers' and School Heads' Salaries and Allowances in Europe: 2017/18. Brussels, European Commission/EACEA/Eurydice. (Eurydice Facts and Figures.)
- Evans, D. K., Yuan, F. and Filmer, D. 2021. *Teacher Pay in Africa: Evidence from 15 Countries*. Washington, DC, Center for Global Development. (CGD Working Paper 538.)
- Funani, Y. 2021. Eastern Cape recruiting teachers ahead of reopening of schools. SABC News, 3 February. www.sabcnews.com/sabcnews/eastern-cape-recruiting-teachers-ahead-of-reopening-of-schools. (Accessed 17 August 2021.)
- Goldberg, E. 2021. As pandemic upends teaching, fewer students want to pursue it. *The New York Times*, 27 March. www.nytimes.com/2021/03/27/us/covid-school-teaching.html. (Accessed 14 October 2021.)
- Goos, M., Hannigan, A., O'Donoghue, J. and Ní Ríordáin, M. 2019. Building Teacher Knowledge for Quality Mathematics Education in Ireland. Limerick, Ireland, University of Limerick.
- Handebo, S., Wolde, M., Shitu, K. and Kassie, A. 2021. Determinant of intention to receive COVID-19 vaccine among school teachers in Gondar city, northwest Ethiopia. *PLoS ONE*, Vol. 16, No. 6.
- IEA. 2019. Trends in International Mathematics and Science Study. Amsterdam, International Association for the Evaluation of Educational Achievement.
- INEP. 2019. Microdados do Censo da Educação Superior [Microdata of the Higher Education Census]. Brasilia, Instituto Nacional de Estudos e Pesquisas Educacionais Anísio Teixeira. www.gov.br/inep/pt-br/acesso-a-informacao/ dados-abertos/microdados/censo-da-educacao-superior. (Accessed 17 August 2021.)
- Karamperidou, D., Brossard, M., Peirolo, S. and Richardson, D. 2020. Time to Teach: Teacher Attendance and Time on Task in Eastern and Southern Africa. Florence, Italy, UNICEF Office of Research – Innocenti.
- Kenny, J., Hobbs, L. and Whannell, R. 2020. Designing professional development for teachers teaching out-of-field. Professional Development in Education, Vol. 46, No. 3, pp. 500–15.
- König, J., Jäger-Biela, D. J. and Glutsch, N. 2020. Adapting to online teaching during COVID-19 school closure: teacher education and teacher competence effects among early career teachers in Germany. *European Journal of Teacher Education*, Vol. 43, No. 4, pp. 608–22.
- Maxwell, L. A. 2021. We feel your grief: remembering the 1,000 plus educators who've died of COVID-19. Education Week, 3 September. www.edweek.org/teaching-learning/we-feel-your-grief-remembering-the-1-000-pluseducators-whove-died-of-covid-19/2021/09. (Accessed 14 October 2021.)
- Mwesigwa, A. 2021. 'I'll never go back': Uganda's schools at risk as teachers find new work during COVID. The Guardian, 30 September. www.theguardian.com/global-development/2021/sep/30/ill-never-go-back-ugandas-schools-at-risk-as-teachers-find-new-work-during-covid. (Accessed 14 October 2021.)
- Napier, J. B., Luft, J. A. and Singh, H. 2020. In the classrooms of newly hired secondary science teachers: the consequences of teaching in-field or out-of-field. *Journal of Science Teacher Education*, Vol. 31, No. 7, pp. 802–20.
- Ndaba, B. 2021. Angie Motshekga: 1650 teachers have died of COVID-19 between March 2020 and February 2021. IOL, 24 July. www.iol.co.za/news/politics/angie-motshekga-1-650-teachers-have-died-of-covid-19-between-march-2020-and-february-2021-ff1beb76-4827-44be-bb08-90c9f03fb6ea. (Accessed 14 October 2021.)
- New South Wales Government. 2021. Vaccination Requirements for School Sites. Sydney, Australia, Government of New South Wales. https://education.nsw.gov.au/covid-19/advice-for-families/vaccination-requirements-for-school-sites. (Accessed 15 October 2021.)
- New Zealand Government. 2021. Education Sector Vaccinations. Wellington, Government of New Zealand. https://covid19. govt.nz/covid-19-vaccines/vaccinations-and-work/mandatory-vaccinations-for-workers/education-sectorvaccinations. (Accessed 15 October 2021.)
- OECD. 2018. TALIS 2018 Data. Paris, Organisation for Economic Co-operation and Development. www.oecd.org/ education/talis/talis-2018-data.htm. (Accessed 19 July 2021.)
- Owen, S. 2014. Teacher professional learning communities: going beyond contrived collegiality toward challenging debate and collegial learning and professional growth. Australian Journal of Adult Learning, Vol. 54, No. 2, pp. 54–77.

- Popova, A., Evans, D. K., Breeding, M. E. and Arancibia, V. 2018. Teacher Professional Development around the World: The Gap between Evidence and Practice. Washington, DC, World Bank. (Policy Research Working Paper 8572.)
- Porsch, R. and Whannell, R. 2019. Out-of-field teaching affecting students and learning: what is known and unknown. Hobbs, L. and Törner, G. (eds), Examining the Phenomenon of 'Teaching Out-of-field': International Perspectives on Teaching as a Non-specialist. Singapore, Springer, pp. 175–91.
- Pota, V., Hennessy, S., Koomar, S., Kreimeia, A., Zubairi, A., Aerts, C. and Gault, C. 2021. Turning to Technology: A Global Survey of Teachers' Responses to the Covid-19 Pandemic. London, T4 Education/EdTech Hub.
- Price, A., Vale, C., Porsch, R., Rahayu, E., Faulkner, F., Ríordáin, M. N., Crisan, C. and Luft, J. A. 2019. Teaching out-of-field internationally. Hobbs, L. and Törner, G. (eds), *Examining the Phenomenon of 'Teaching Out-of-field'*: International Perspectives on Teaching as a Non-specialist. Singapore, Springer, pp. 53–83.
- Prince, G. and O'Connor, M. 2018. Crunching the Numbers of Out-of-field Teaching. Melbourne, Australia, Australian Mathematical Sciences Institute. (AMSI Occasional Paper 1.)
- Racey, C. S., Donken, R., Porter, I., Albert, A., Bettinger, J. A., Mark, J., Bonifacio, L., Dawar, M., Gagel, M. and Kling, R.
  2021. Intentions of public school teachers in British Columbia, Canada to receive a COVID-19 vaccine.
  Vaccine: X, Vol. 8.
- Ramot, R. and Donitsa-Schmidt, S. 2021. COVID-19: education policy, autonomy and alternative teacher education in Israel. *Perspectives in Education*, Vol. 39, No. 1.
- Rashid, O. 2021. 1,621 teachers died of COVID-19 during U.P. Panchayat polls, says Union. *The Hindu*, 17 May. www.thehindu.com/news/national/other-states/many-teachers-died-of-covid-19-during-up-panchayat-pollssays-union/article34575965.ece. (Accessed 14 October 2021.)
- Reid, N. and Cranston, J. 2021. Provinces should act fast to avert a teacher shortage now and after COVID-19. *The Conversation*, 15 March. https://theconversation.com/provinces-should-act-fast-to-avert-a-teachershortage-now-and-after-covid-19-154930. (Accessed 14 October 2021.)
- Salifu, M. and Todd, R. 2020. Ghana's Teacher Education System and Responding to COVID-19. Accra, Transforming Teacher Education and Learning. http://resourceshub.ncte.edu.gh/news-view/ghanas-teacher-educationsystem-and-responding-to-covid-19. (Accessed 17 August 2021.)
- Sepulveda-Escobar, P. and Morrison, A. 2020. Online teaching placement during the COVID-19 pandemic in Chile: challenges and opportunities. *European Journal of Teacher Education*, Vol. 43, No. 4, pp. 587–607.
- Sharplin, E. D. 2014. Reconceptualising out-of-field teaching: experiences of rural teachers in Western Australia. Educational Research, Vol. 56, No. 1, pp. 97–110.
- Sheppard, K., Padwa, L., Kelly, A. M. and Krakehl, R. 2020. Out-of-field teaching in chemistry and physics: an empirical census study. *Journal of Science Teacher Education*, Vol. 31, No. 7, pp. 746–67.
- Steiner, E. D. and Woo, A. 2021. Job-related Stress Threatens the Teacher Supply: Key Findings from the 2021 State of the US Teacher Survey. Santa Monica, Calif., RAND Corporation.
- Tatto, M. T. 2020. Evaluating Global Progress on Improving Teacher Quality: ISCED-T and Other Possible Metrics. Paris, International Task Force on Teachers for Education 2030. https://teachertaskforce.org/blog/evaluatingglobal-progress-improving-teacher-quality-isced-t-and-other-possible-metrics. (Accessed 17 August 2021.)
- The Guardian. 2017. Secret Teacher: too many of us teach subjects we're not qualified for. 18 May. www.theguardian. com/teacher-network/2017/sep/16/the-secret-teacher-too-many-us-teaching-subjects-not-qualified. (Accessed 17 August 2021.)
- UIS. 2020. Current Status and Next Steps for SDG Indicators 4.c.1 to 4.c.4. Montreal, Que., UNESCO Institute for Statistics. (Working Group on Teachers, Technical Cooperation Group WG/T/4.)
- \_\_\_\_. 2021. International Standard Classification of Teacher Training Programmes Montreal, Que., UNESCO Institute for Statistics.
- UNESCO. 2021. Closing the Gap: Ensuring There Are Enough Qualified and Supported Teachers. Paris, International Task Force on Teachers for Education 2030.
- UNESCO, UNICEF, World Bank and OECD. 2021. What's Next? Lessons on Education Recovery: Findings from a Survey of Ministries of Education amid the COVID-19 Pandemic. Paris/Montreal, Que.,/New York/Florence/Washington, DC, UNESCO, UNESCO Institute for Statistics/UNICEF/UNICEF Office of Research – Innocenti/World Bank/ OECD.
- Valeeva, R. and Kalimullin, A. 2021. Adapting or changing: the COVID-19 pandemic and teacher education in Russia. Education Sciences, Vol. 11, No. 8.
- Van Overschelde, J. and Piatt, A. 2020. U.S. Every Student Succeeds Act: negative impacts on teaching out-of-field. Research in Educational Policy and Management, Vol. 2, No. 1, pp. 1–22.

- Weldon, P. 2016. Out-of-field Teaching in Australian Secondary Schools. Camberwell, Australia, Australian Council for Educational Research. (Policy Insights 6.)
- White House. 2021. White House Report: Vaccination Requirements Are Helping Vaccinate More People, Protect Americans from COVID-19, and Strengthen the Economy. Washington, DC, The White House.
- White, I. and McSharry, M. 2021. Preservice teachers' experiences of pandemic related school closures: anti-structure, liminality and communitas. *Irish Educational Studies*, Vol. 40, No. 2, pp. 319–27.
- Worth, J. 2021. What Impact Did COVID-19 Have on Teacher Retention? Slough, UK, National Foundation for Educational Research. www.nfer.ac.uk/news-events/nfer-blogs/what-impact-did-covid-19-have-on-teacher-retention. (Accessed 14 October 2021.)
- Xinhua. 2021. Uganda to recruit over 3,000 teachers ahead of reopening of schools. 17 February. www.xinhuanet.com/ english/2021-02/17/c\_139748424.htm. (Accessed 17 August 2021.)
- Zamarro, G., Camp, A., Fuchsman, D. and McGee, J. 2021. Understanding How COVID-19 Has Changed Teachers' Chances of Remaining in the Classroom. Fayetteville, Ark., College of Education and Health Professions, University of Arkansas. (EDRE Research Brief 2021-01.)

## **CHAPTER 20**

- Adukia, A., Asher, S. and Novosad, P. 2020. Educational investment responses to economic opportunity: evidence from Indian road construction. *American Economic Journal: Applied Economics*, Vol. 12, No. 1, pp. 348–76.
- Aggarwal, S. 2018. Do rural roads create pathways out of poverty? Evidence from India. Journal of Development Economics, Vol. 133, pp. 375–95.
- Ahmad, S. and Puppim de Oliveira, J. A. 2015. Fuel switching in slum and non-slum households in urban India. *Journal of Cleaner Production*, Vol. 94, pp. 130–36.
- Amore, M. D., Bennedsen, M., Larsen, B. and Rosenbaum, P. 2019. CEO education and corporate environmental footprint. *Journal of Environmental Economics and Management*, Vol. 94, pp. 254–73.
- Badasyan, N. and Silva, S. 2018. The impact of internet access at home and/or school on students' academic performance in urban areas in Brazil. International Journal of Education Economics and Development, Vol. 9, No. 2, pp. 149–71.
- Bessone, P., Dahis, R. and Ho, L. 2021. The Impact of 3G Mobile Internet on Educational Outcomes in Brazil. Cambridge, Mass., Massachussets Institute of Technology.
- Bianchi, F., Dorsel, C., Garnett, E., Aveyard, P. and Jebb, S. A. 2018. Interventions targeting conscious determinants of human behaviour to reduce the demand for meat: a systematic review with qualitative comparative analysis. *International Journal of Behavioral Nutrition and Physical Activity*, Vol. 15, No. 1, pp. 1–25.
- Bloomfield, E. 2014. Gender and Livelihoods Impacts of Clean Cookstoves in South Asia. Washington, DC, Global Alliance for Clean Cookstoves,.
- Carrieri, V. and Principe, F. 2020. WHO and for How Long? An Empirical Analysis of the Consumers' Response to Red Meat Warning. Bonn, Germany, Institute of Labor Economics. (IZA Discussion Paper 13882.)
- Conte, M. N. and Jacobsen, G. D. 2016. Explaining demand for green electricity using data from all US utilities. *Energy Economics*, Vol. 60, pp. 122–30.
- De Backer, C. J. S. and Hudders, L. 2015. Meat morals: relationship between meat consumption consumer attitudes towards human and animal welfare and moral behavior. *Meat Science*, Vol. 99, pp. 68–74.
- Einhorn, L. 2020. Food, Classed? Social Inequality and Diet: Understanding Stratified Meat Consumption Patterns in Germany. Cologne, Germany, International Max Planck Research School on the Social and Political Constitution of the Economy. (Studies on the Social and Political Constitution of the Economy.)
- Ellen MacArthur Foundation. 2017. A New Textiles Economy: Redesigning Fashion's Future. Cowes, UK, Ellen MacArthur Foundation. (Circular Fibres Initiative.)
- ESMAP. 2020. Supporting Tuvalu's Move towards 100% Renewable Energy. Washington, DC, World Bank. (Energy Sector Management Assistance Program Impact 21.)
- \_\_\_\_. 2021. What Drives the Transition to Modern Energy Cooking Services? A Systematic Review of the Evidence. Washington, DC, World Bank. (Energy Sector Management Assistance Program Technical Report 015/21.)
- Esposito, D. 2021. Studies Converge on Benefits of a Rapid Clean Energy Transition. San Francisco, Calif., Energy Innovation Policy and Technology.
- FAO. 2017. Global Initiative on Food Loss and Waste. Rome, Food and Agriculture Organization of the United Nations.

Giacoman, C., Arancibia, P. A. and Alfaro, J. 2021. Choosing to stop consuming meat for environmental reasons: exploring the influence of gender and social status variables in Chile. *British Food Journal*, Vol. 123, No. 9.

- Gould, C. F. and Urpelainen, J. 2020. The role of education and attitudes in cooking fuel choice: evidence from two states in India. Energy for Sustainable Development, Vol. 54, pp. 36–50.
- Hafner, M. and Tagliapietra, S. (eds). 2020. The Geopolitics of the Global Energy Transition. Cham, Switzerland, Springer. (Lecture Notes in Energy 73.)
- Hampton, K., Fernandez, L., Robertson, C. and Bauer, J. M. 2020. *Repercussions of Poor Broadband Connectivity for Students in Rural and Small Town Michigan.* Paper for 48th Research Conference on Communication, Information and Internet Policy, 25–26 September.
- Hincapie-Velez, G. D., Montoya-Gomez, I. and Bustamante, J. J. 2017. The rural roads impact on education performance in Antioquia (Colombia): an ordered probit model. *Ecos de Economía*, Vol. 21, No. 44, pp. 72–85.
- IEA, IRENA, UNSD, World Bank and WHO. 2020. *Tracking SDG 7: The Energy Progress Report 2020*. Paris/Abu Dhabi/New York/Washington, DC/Geneva, Switzerland, International Energy Agency/International Renewable Energy Agency/UN Statistical Division/World Bank/World Health Organization.
- \_\_\_\_. 2021. Tracking SDG 7: The Energy Progress Report 2021. Paris/Abu Dhabi/New York/Washington, DC,/Geneva, Switzerland, International Energy Agency/International Renewable Energy Agency/UN Statistical Division/ World Bank/World Health Organization.
- Iimi, A., Lancelot, E., Manelici, I. and Ogita, S. 2015. Social and Economic Impacts of Rural Road Improvements in the State of Tocantins, Brazil. Washington, DC, World Bank. (Policy Research Working Paper 7249.)
- ILO. 2020. *Skills for a Greener Future: A Global View.* Geneva, Switzerland, International Labour Organization. IRENA. 2021. *World Energy Transitions Outlook: 1.5°C Pathway.* Abu Dhabi, International Renewable Energy Agency. IRENA, European Commission and ILO. 2018. *Accelerating SDG 7 Achievement: Interlinkages between Energy and Jobs.*
- Abu Dhabi/Brussels/Geneva, Switzerland, International Renewable Energy Agency/European Commission/ International Labour Organization. (Policy Brief 13.)
- Jimenez, R. 2017. Development Effects of Rural Electrification. Washington, DC, Inter-American Development Bank. (Policy Brief 261.)
- KfW Development Bank, GIZ and IRENA. 2021. The Renewable Energy Transition in Africa: Powering Access, Resilience and Prosperity. Frankfurt/Eschborn, Germany/Abu Dhabi, KfW Development Bank/GIZ/International Renewable Energy Agency.
- Khanani, R. S., Adugbila, E. J., Martinez, J. A. and Pfeffer, K. 2021. The impact of road infrastructure development projects on local communities in peri-urban areas: the case of Kisumu, Kenya and Accra, Ghana. International Journal of Community Well-Being, Vol. 4, No. 1, pp. 33–53.
- Lacy, P., Haines, A. and Hayward, R. 2012. Developing strategies and leaders to succeed in a new era of sustainability: findings and insights from the United Nations Global Compact-Accenture CEO Study. *Journal of Management Development*, Vol. 31, No. 4, pp. 346–57.
- Leal Filho, W., Amaro, N., Avila, L. V., Brandli, L., Damke, L. I., Vasconcelos, C. P., Hernandez-Diaz, P. M., Frankenberger, F., Fritzen, B. and Velazquez, L. 2021. Mapping sustainability initiatives in higher education institutions in Latin America. Journal of Cleaner Production, Vol. 315.
- Li, D., Lin, A. and Zhang, L. 2019. Relationship between chief executive officer characteristics and corporate environmental information disclosure in Thailand. Frontiers of Engineering Management, Vol. 6, No. 4, pp. 564–74.
- Lindgren, S. 2021a. Cookstove implementation and education for sustainable development: a review of the field and proposed research agenda. *Renewable and Sustainable Energy Reviews*, Vol. 146.
- \_\_\_\_. 2021b. Sustainability education for Namibian youth to advance efficient cookstoves and energy development. Consilience: The Journal of Sustainable Development, Vol. 24.
- Litzow, E. L., Pattanayak, S. K. and Thinley, T. 2019. Returns to rural electrification: evidence from Bhutan. World Development, Vol. 121, pp. 75–96.
- McRae, G., McFadden, M. and Ullman, H. 2019. 'I See Myself in that Career': Exploring Methods to Attract the Next Generation Transportation Workforce. Davis, Calif., National Center for Sustainable Transportation.
- Meier, L. 2021. Synthesis Report on United Nations System-wide Initiatives Related to Fashion, UN Alliance for Sustainable Fashion.
- Meyer, A. 2015. Does education increase pro-environmental behavior? Evidence from Europe. *Ecological Economics*, Vol. 116, pp. 108–21.
- Nano, E. 2021. Electrifying Nigeria: the Household-Level Impact of Access to Electricity on Consumption, Education and Employment. Geneva, Switzerland, Graduate Institute of International and Development Studies.

- NEED. 2021. NEED Curriculum. Manassas, Va., National Energy Education Development Project. www.need.org/aboutneed/our-curriculum. (Accessed 17 August 2021.)
- Nowotny, J., Dodson, J., Fiechter, S., Gür, T. M., Kennedy, B., Macyk, W., Bak, T., Sigmund, W., Yamawaki, M. and Rahman, K. A. 2018. Towards global sustainability: education on environmentally clean energy technologies. *Renewable* and Sustainable Energy Reviews, Vol. 81, pp. 2541–51.
- OECD. 2020. Making the Green Recovery Work for Jobs, Income and Growth. Paris, Organisation for Economic Co-operation and Development.
- Oum, S. 2019. Energy poverty in the Lao PDR and its impacts on education and health. Energy Policy, Vol. 132, pp. 247–53.
- Peters, J. and Sievert, M. 2016. Impacts of rural electrification revisited: the African context. Journal of Development Effectiveness, Vol. 8, No. 3, pp. 327–45.
- Poore, J. and Nemecek, T. 2018. Reducing food's environmental impacts through producers and consumers. *Science*, Vol. 360, No. 6392, pp. 987–92.
- Power for All. 2020. Policy Brief: Action Agenda to Build the Workforce and Skills Needed for Universal Energy Access. San Francisco, Calif., Power for All.
- Power for All, Schneider Electric Foundation and Rockefeller Foundation. 2019. Powering Jobs Census 2019: The Energy Access Workforce. San Francisco, Calif., Power for All.
- Rahut, D. B., Mottaleb, K. A., Ali, A. and Aryal, J. 2018. The use and determinants of solar energy by sub-Saharan African households. International Journal of Sustainable Energy, Vol. 37, No. 8, pp. 718–35.
- Reisch, L. A., Cohen, M. J., Thøgersen, J. B., Tukker, A. and Krause, K. 2016. Sustainable Consumption: Research Challenges. Stockholm, Swedish Foundation for Strategic Environmental Research.
- Rust, N. A., Ridding, L., Ward, C., Clark, B., Kehoe, L., Dora, M., Whittingham, M. J., McGowan, P., Chaudhary, A. and Reynolds, C. J. 2020. How to transition to reduced-meat diets that benefit people and the planet. *Science of the Total Environment*, Vol. 718.
- Sahakian, M. and Seyfang, G. 2018. A sustainable consumption teaching review: from building competencies to transformative learning. *Journal of Cleaner Production*, Vol. 198, pp. 231–41.
- Sharma, B. P. 2018. Household fuel transition and determinants of firewood demand in Nepal. Economic Journal of Development Issues, Vol. 25, No. 1-2, pp. 83–95.
- Slow Food International. 2020. Meat the Change: Eat Less Meat, but of Better Quality. Bra, Italy, Slow Food International. https://meatthechange.slowfood.com/en. (Accessed 14 August 2021.)
- Tianyu, J. and Meng, L. 2020. Does education increase pro-environmental willingness to pay? Evidence from Chinese household survey. *Journal of Cleaner Production*, Vol. 275, art. 122713.
- Tran, N. and Pham, B. 2020. The influence of CEO characteristics on corporate environmental performance of SMEs: evidence from Vietnamese SMEs. *Management Science Letters*, Vol. 10, No. 8, pp. 1671–82.
- US Department of Energy. 2021. Partnering for the Future: Leadership, Innovation and Proven Solutions Progress Report 2021. Washington, DC, US Department of Energy. (Better Buildings Initiative.)
- UNDESA. 2018. Analysis of the Voluntary National Reviews Relating to Sustainable Development Goal 7: Ensuring Access to Affordable, Reliable, Sustainable and Modern Energy for All. New York, UN Department of Economic and Social Affairs.
- Watkins, O. and Hochlaf, D. 2021. Skills for a Green Recovery: A Call to Action for the UK Construction Sector. London, Institute for Public Policy Research.
- WFEO. 2018. WFEO Engineering 2030: A Plan to Advance the Achievement of the UN Sustainable Development Goals through Engineering. Paris, World Federation of Engineering Organizations/UNESCO. (Progress Report 1.)
- Wu, J. X. and Li, L. 2019. Sustainability initiatives in the fashion industry. Beltramo, R., Romani, A. and Cantore, P. (eds), Fashion Industry: An Itinerary between Feelings and Technology. London, IntechOpen, pp. 3–20.
- Wuppertal Institute and Germanwatch. 2015. Social CSP: Energy and Development Exploring the Local Livelihood Dimension of the Noor<sub>o</sub> I CSP Project in Southern Morocco. Wuppertal/Bonn, Germany, Wuppertal Institute for Climate, Environment and Energy/Germanwatch. (Final report to the Federal Ministry for Economic Cooperation and Development.)
- Zafar, M. W., Shahbaz, M., Sinha, A., Sengupta, T. and Qin, Q. 2020. How renewable energy consumption contribute to environmental quality? The role of education in OECD countries. *Journal of Cleaner Production*, Vol. 268, art. 122149.
- Zhou, M., Chen, F. and Chen, Z. 2021. Can CEO education promote environmental innovation: Evidence from Chinese enterprises. Journal of Cleaner Production, Vol. 297, art. 126725.

### **CHAPTER 21**

- Akmal, M., Ali, A., Hares, S. and Sandefur, J. 2021. *Is the Global Partnership for Education Redundant?* Washington. DC, Center for Global Development. www.cgdev.org/publication/global-partnership-education-redundant. (Accessed 17 August 2021.)
- Algeria Ministry of National Education. 2015. Le financement de l'education [Education financing]. Algiers, Ministry of National Education. www.education.gov.dz/fr/systeme-educatif-algerien/le-financement-de-leducation. (Accessed 17 August 2021.)
- Archer, D. 2016. Domestic Tax and Education. New York, International Commission on Financing Global Education Opportunity. (Background paper for The Learning Generation: Investing in Education for a Changing World.)
- ATI. 2015. Financing for Development Conference: The Addis Tax Initiative Declaration. Bonn, Germany, Addis Tax Initiative.
- \_\_\_\_. 2021. Addis Tax Initiative: Tax Systems that Work for People and Advance the SDGs. Bonn, Germany, Addis Tax Initiative. www.addistaxinitiative.net. (Accessed 17 August 2021.)
- Beer, S., de Mooij, R. and Liu, L. 2019. International corporate tax avoidance: a review of the channels, magnitudes, and blind spots. *Journal of Economic Surveys*, Vol. 34, No. 1.
- Buettner, T. and Wamser, G. 2013. Internal debt and multinational profit shifting: empirical evidence from firm-level panel data. *National Tax Journal*, Vol. 66, No. 1, pp. 63–96.
- Cahyadi, N., Hanna, R., Olken, B. A., Prima, R. A., Satriawan, E. and Syamsulhakim, E. 2018. Cumulative Impacts of Conditional Cash Transfer Programs: Experimental Evidence from Indonesia. Cambridge, Mass., National Bureau of Economic Research. (NBER Working Paper 24670.)
- Caputo, E., de Kemp, A. and Lawson, A. 2011. Assessing the Impacts of Budget Support: Case Studies in Mali, Tunisia and Zambia. Paris, Organisation for Economic Co-operation and Development. (Evaluation Insights 2.)
- Cristea, A. D. and Nguyen, D. X. 2016. Transfer pricing by multinational firms: new evidence from foreign firm ownerships. American Economic Journal: Economic Policy, Vol. 8, No. 3.
- Dijkstra, G. 2018. Budget Support, Poverty and Corruption: A Review of the Evidence. Stockholm, Expertgruppen för biståndsanalys. (Rapport 2018:04.)
- Ghana Statistical Service. 2008. Ghana Living Standards Survey Report of the Fifth Round (GLSS 5). Accra, Ghana Statistical Service.
- \_\_\_\_. 2016. Ghana Living Standards Survey Round 6 (GLSS6): Main Report. Accra, Ghana Statistical Service.
- \_\_\_\_. 2019. Ghana Living Standards Survey (GLSS) 7: Main Report. Accra, Ghana Statistical Service.
- Global Alliance for Tax Justice, Public Services International and Tax Justice Network. 2021. The State of Tax Justice 2021. London and Ferney-Voltaire, France, Global Alliance for Tax Justice/Public Services International/Tax Justice Network.
- GPE. 2019. Tajikistan Education Sector Analysis. Washington, D.C., Global Partnership for Education.
- \_\_\_\_. 2021. Results Report 2021: Final Results Report on GPE's 2016-2020 Strategy. Washington, DC, Global Partnership for Education.
- INFF. 2021. About INFFs. New York, Integrated National Financing Frameworks. https://inff.org/about#origins-of-theinff. (Accessed 17 August 2021.)
- IsDB. 2019. Project Appraisal Document: The IsDB/GPE Project for Support to Implementation of the National Education Development Strategy of the Republic of Tajikistan. Jeddah, Islamic Development Bank.
- Janský, P. and Palanský, M. 2019. Estimating the scale of proft shifting and tax revenue losses related to foreign direct investment. International Tax and Public Finance, Vol. 26, pp. 1048–103.
- Janus, H., Marschall, P. and Öhler, H. 2020. Bridging the Gaps: An Integrated Approach to Assessing Aid Effectiveness. Bonn, Germany, German Development Institute. (DIE Briefing Paper 12/2020.)
- Johannesen, N. and Pirttilä, J. 2016. Capital Flight and Development: An Overview of Concepts, Methods, and Data Sources. Helsinki, United Nations University, World Institute for Development Economics Research.
- Johannesen, N., Tørsløv, T. and Wier, L. 2020. Are less developed countries more exposed to multinational tax
- avoidance? Method and evidence from micro-data. World Bank Economic Review, Vol. 34, No. 3, pp. 790–809. Kleinbard, E. D. 2011. Stateless income. Florida Tax Review, Vol. 11, No. 9, pp. 699–774.
- Koch, S., Leiderer, S., Faust, J. and Molenaers, N. 2017. The rise and demise of European budget support: political economy of collective European Union donor action. *Development Policy Review*, Vol. 35, No. 4, pp. 455–73.
- Lewin, K. M. 2020. Taxing matters: fiscal reform, public goods and aid. NORRAG Special Issue, Vol. 5, pp. 14–17.
- Li, X., Gu, J., Leistner, S. and Cabral, L. 2018. Perspectives on the Global Partnership for Effective Development Cooperation. IDS Bulletin, Vol. 49, No. 3.
- Loft, P. and Brien, P. 2021. *Reducing the UK's Aid Spend in 2021*. London, House of Commons. (Commons Library Research Briefing.)

Marshall, J. H. and Bunly, S. 2017. School grants and school performance in rural Cambodia. *Journal of Development Effectiveness*, Vol. 9, No. 3, pp. 1–24.

Miller, M. and Roger, L. 2021. The UK, Special Drawing Rights and Fiscal Gimmickry. London, Overseas Development Institute. https://odi.org/en/insights/the-uk-special-drawing-rights-and-fiscal-gimmickry. (Accessed 2 September 2021.)

MOPAN. 2017. Global Fund to Fight AIDS, Tuberculosis and Malaria (The Global Fund): Institutional Assessment Report. Paris, Multilateral Organisation Performance Assessment Network. (MOPAN 2015–2016 Assessments.)

Nepal Department of Education. 2016. *Status Report:* 2015/016. Bhaktapur, Nepal, Ministry of Education. OECD. 2021a. The 0.7% ODA/GNI Target: A History. Paris, Organisation for Economic Co-operation and Development. www.oecd.org/development/stats/the07odagnitarget-ahistory.htm. (Accessed 20 September 2021.)

\_\_\_\_. 2021b. 130 Countries and Jurisdictions Join Bold New Framework for International Tax Reform. Paris, Organisation for Economic Co-operation and Development. www.oecd.org/newsroom/130-countries-and-jurisdictions-joinbold-new-framework-for-international-tax-reform.htm. (Accessed 17 August 2021.)

\_\_\_\_. 2021c. Global Revenue Statistics Database. Paris, Organisation for Economic Co-operation and Development. https://stats.oecd.org/Index.aspx?DataSetCode=RS\_GBL. (Accessed 17 August 2021.)

\_\_\_\_. 2021d. Two-Pillar Solution to Address the Tax Challenges Arising from the Digitalisation of the Economy. Paris, Organisation for Economic Co-operation and Development. (OECD/G20 Base Erosion and Profit Shifting Project.)

OECD and UNDP. 2014. Making Development Co-Operation More Effective: 2014 Progress Report. Paris/New York, Organisation for Economic Co-operation and Development/United Nations Development Programme.

\_\_\_\_. 2016. Making Development Co-operation More Effective: 2016 Progress Report. Paris/New York, Organisation for Economic Co-operation and Development/United Nations Development Programme.

\_\_\_\_. 2019. Making Development Co-operation More Effective: 2019 Progress Report. Paris/New York, Organisation for Economic Co-operation and Development/United Nations Development Programme.

Philippines Department of Education. 2015. School-Based Management Grant. Pasig, Philippines, Department of Education. www.deped.gov.ph/2015/10/28/school-based-management-grant. (Accessed 17 August 2021.)

Rose, P., Steer, L., Smith, K. and Zubairi, A. 2013. Financing for Global Education: Opportunities for Multilateral Action. Paris/Washington, DC, UNESCO/Centre for Universal Education, Brookings Institution.

Tortora, P. and Steensen, S. 2014. Bilateral Providers' and Multilatearal Organisations' Practices on Earmarked Funding. Paris, Organisation for Economic Co-operation and Development.

UIS. 2021. Monitoring GEM Commitments Using the Joint Survey of National Education Responses to COVID-19. Montreal, Que., UNESCO Institute for Statistics.

UNCTAD. 2015. World Investment Report 2015: Reforming International Investment Governance. Geneva, Switzerland, UN Conference on Trade and Development.

UNESCO. 2020. Act Now: Reduce the Impact of COVID-19 on the Cost of Achieving SDG 4. Paris, UNESCO. (Global Education Monitoring Report Policy Paper 42.)

\_\_\_\_. 2021. How Committed? Unlocking Financing for Equity in Education. Paris, UNESCO. (Global Education Monitoring Report Policy Paper 44.)

UNESCO Bangkok. 2017. Ensuring Adequate, Efficient and Equitable Finance in Schools in the Asia-Pacific Region. Bangkok, UNESCO.

UNICEF. 2018. Myanmar 2018 Education Budget Brief. New York, UNICEF.

\_\_\_\_. 2021. Global Annual Results Report 2020: Goal Area 2. New York, UNICEF.

United Nations. 2015. Addis Ababa Action Agenda of the Third International Conference on Financing for Development: Annex. New York, United Nations. https://sustainabledevelopment.un.org/frameworks/addisababaactionagenda. (Accessed 17 August 2021.)

UNU-WIDER. 2021. UNU-WIDER Government Revenue Dataset. Helsinki, United Nations University, World Institute for Development Economics Research. www.wider.unu.edu/database/data-and-resources-grd. (Accessed 1 October 2021.)

Vertex Consult. 2016. A Study on the Effectiveness of the Scholarship Provided at School Level and Identification of Measures for its Improvement. Sanothimi, Nepal, Department of Education.

World Bank. 2017. Program Appraisal Document on a Proposed Loan in the Amount of US\$200 million to the Republic of Indonesia for a Social Assistance Reform Program. Washington, DC, World Bank.

\_\_\_\_. 2018. Myanmar: Additional Financing for the Decentralizing Funding to Schools Project. Washington, DC, World Bank.

\_\_\_\_. 2020. Philippines Basic Education Public Expenditure Review. Washington, DC, World Bank.

\_\_\_\_. 2021. Households and NPISHs final consumption expenditure (% of GDP). Washington, DC, World Bank. https:// data.worldbank.org/indicator/NE.CON.PRVT.ZS. (Accessed 1 October 2021.)

World Bank and UNESCO. 2021. Education Finance Watch 2021. Washington, DC/Paris, World Bank/UNESCO.

## **ANNEX** • REFERENCES

# Non-state actors in education

## WHO CHOOSES? WHO LOSES?

Non-state actors' role extends beyond provision of schooling to interventions at various education levels and influence spheres. Alongside its review of progress towards SDG 4, including emerging evidence on the COVID-19 pandemic's impact, the 2021/2 *Global Education Monitoring Report* urges governments to see all institutions, students and teachers as part of a single system. Standards, information, incentives and accountability should help governments protect, respect and fulfil the right to education of all, without turning their eyes away from privilege or exploitation. Publicly funded education does not have to be publicly provided but disparity in education processes, student outcomes and teacher working conditions must be addressed. Efficiency and innovation, rather than being commercial secrets, should be diffused and practised by all. To that end, transparency and integrity in the public education policy process need to be maintained to block vested interests.

The report's rallying call – Who chooses? Who loses? – invites policymakers to question relationships with non-state actors in terms of fundamental choices: between equity and freedom of choice; between encouraging initiative and setting standards; between groups of varying means and needs; between immediate commitments under SDG 4 and those to be progressively realized (e.g. post-secondary education); and between education and other social sectors.

Supporting the fifth *Global Education Monitoring Report* are two online tools: PEER, a policy dialogue resource describing non-state activity and regulations in the world's education systems; and VIEW, a new website consolidating sources and providing new completion rate estimates over time.



